### Using Machine Learning for Real-time Activity Recognition and Estimation of Energy Expenditure

by Emmanuel Munguia Tapia

Master of Science Massachusetts Institute of Technology, 2003 Cambridge, Massachusetts

Bachelor of Science Instituto Politécnico Nacional Escuela Superior de Ingeníeria Mecánica y Eléctrica Mexico, DF. 2001

Submitted to the Program in Media Arts and Sciences, School of Architecture and Planning, in partial fulfillment of the requirements for the degree of Doctor of Philosophy at the

#### MASSACHUSETTS INSTITUTE OF TECHNOLOGY

June 2008

© Massachusetts Institute of Technology 2008. All rights reserved.

Author

Program in Media Arts and Sciences March 30, 2008

Certified by

Kent Larson Principal Research Scientist MIT Department of Architecture

Accepted by

Deb Roy Chair, Departmental Committee on Graduate Studies Program in Media Arts and Sciences

### Using Machine Learning for Real-time Activity Recognition and Estimation of Energy Expenditure

by Emmanuel Munguia Tapia

Submitted to the Program in Media Arts and Sciences, School of Architecture and Planning on March 30, 2008, in partial fulfillment of the requirements for the degree of Doctor of Philosophy at the MASSACHUSETTS INSTITUTE OF TECHNOLOGY

#### Abstract

Obesity is now considered a global epidemic and is predicted to become the number one preventive health threat in the industrialized world. Presently, over 60% of the U.S. adult population is overweight and 30% is obese. This is of concern because obesity is linked to leading causes of death, such as heart and pulmonary diseases, stroke, and type 2 diabetes. The dramatic rise in obesity rates is attributed to an environment that provides easy access to high caloric food and drink and promotes low levels of physical activity. Unfortunately, many people have a poor understanding of their own daily energy (im)balance: the number of calories they consume from food compared with what they expend through physical activity. Accelerometers offer promise as an objective measure of physical activity. In prior work they have been used to estimate energy expenditure and activity type. This work further demonstrates how wireless accelerometers can be used for real-time automatic recognition of physical activity type, intensity, and duration and estimation of energy expenditure. The parameters of the algorithms such as type of classifier/regressor, feature set, window length, signal preprocessing, sensor set utilized and their placement on the human body are selected by performing a set of incremental experiments designed to identify sets of parameters that may balance system usability with robust, real-time performance in low processing power devices such as mobile phones. The algorithms implemented are evaluated using a dataset of examples of 52 activities collected from 20 participants at a gymnasium and a residential home. The algorithms presented here may ultimately allow for the development of mobile phonebased just-in-time interventions to increase self-awareness of physical activity patterns and increases in physical activity levels in real-time during free-living that scale to large populations.

KEYWORDS: Activity recognition, context awareness, energy expenditure, physical activity, wearable sensors, obesity, mobile phone, pattern recognition, machine learning, ubiquitous, pervasive, just-in-time.

Thesis Supervisor: Kent Larson, Principal Research Scientist in Architecture

## Using Machine Learning for Real-time Activity Recognition and Estimation of Energy Expenditure

by Emmanuel Munguia Tapia Doctor of Philosophy June 2008

Advisor	
	Kent Larson
	Principal Research Scientist
	MIT Department of Architecture
Reader	
	Dr. Stephen S. Intille
	Research Scientist
	MIT Department of Architecture
Reader	
	Dr. Joseph A. Paradiso
	Associate Professor of Media Arts and Sciences
	MIT Media Laboratory
Reader	
	Prof. Alex (Sandy) Pentland
	Toshiba Professor in Media Arts and Sciences
	MIT Media Laboratory
Reader	
	Dr. William L. Haskell
	Professor of Medicine at Stanford University
	Stanford Prevention Research Center

## Acknowledgments

#### Stephen

I am grateful for the time and advice that you provided over the past six years. This thesis owes much of its contents to your ideas and guidance.

#### Kent

Thank you for offering me the opportunity to study at MIT and for guiding me through my learning process at the institution.

In the following paragraphs, I write the acknowledgements to the most important people in my life. Those who have loved, supported, and encouraged me unconditionally over the course of my life. I also dedicate the contents of this thesis to my sister Anaid Munguia Tapia. Since my heart does not know how to speak or write in English, the following paragraphs are written in Spanish, the native language of my heart.

#### Mi hermana Anaid

Hermana mia, estoy eternamente agradecido por tu amor y cariño incondicional. Tú has sido una verdadera inspiración para mi trabajo y me has enseñado que hay cosas más importantes que ser un buen estudiante o trabajador. Me has enseñado que la vida es más que triunfar económicamente o lograr un mejor estatus social. Gracias por tu paciencia durante todos esos años tan difíciles en los cuales no sabíamos porque las cosas estaban pasando y que era lo que finalmente Dios nos tenía preparado. Tambien quiero pedirte disculpas por permanecer tantos años alejado de ti cuando debí de estar a tu lado, cuidándote y haciendo tu condición mas pasadera. Espero y puedas comprender que lo hice por un mejor futuro económico para todos. Aunque la verdad, creo que tienes razón. Ningún título de ninguna institucion en el planeta, incluyendo MIT, me puede ahora regresar el tiempo que perdí a tu lado. Te amo hermana mia con todo mi corazón y espero tenerte cerca de mi en un futuro muy cercano.

#### Mi mamá Josefina y papá Leonardo

Gracias por enseñarme a soñar, a soñar en grande y sobre todo a protejer mis sueños de personas que piensa que son tan grandes que son imposibles de realizar. Me siento afortunado de haber sido su hijo, gracias por cuidarme y educarme con tanto amor. Madre, gracias por cuidar de mis hermanos cuando mas lo necesitaban y permitirme ir a estudiar a la capital en busca de mis sueños, cuando mi apoyo tanto económico como emocional hacían tanta falta en el hogar durante esos duros momentos de enfermedad y desolación. Gracias a su gran fortaleza y apoyo, el día de hoy puedo decir que me graduo como un doctor en filosofía del Instituto Tecnológico de Massachusetts.

#### Mi esposa Margarita

Amor mio, gracias por tu maravilloso amor, por ser mi compañera, mi fortaleza e inspiración. Gracias tambien por brindarme todo tu apoyo durante todos esos años que permanecimos separados de tal manera que el día de hoy pudiera cumplir mi sueño de terminar un doctorado en MIT. Espero poder cumplir con la promesa que hice ante Dios el día de nuestra boda: brindarte una vida llena de paz, dicha y felicidad. ¡Te amo!

# Contents

1	INTRODUCTION	. 33
	1.1 THE OPPORTUNITY	. 34
	1.2 CHALLENGES OF ESTIMATING ENERGY BALANCE	. 35
	1.3 System Goals	. 37
	1.4 EXPERIMENTAL GOALS	. 37
	1.5 ORGANIZATION OF THESIS	. 39
2	THE SYSTEM GOAL	. 40
	2.1 MOBILE PHONE ENERGY BALANCE INTERVENTIONS	. 40
	2.1.1 Mobile Phone Energy Balance Interventions Wish List	. 40
	2.1.2 Required Components	. 41
	2.1.2.1 Automatic Recognition of Physical Activities	. 41
	2.1.2.2 Estimation of Energy Expenditure	. 42
	2.1.3 Scenarios Exploiting Real-Time Feedback	. 43
	2.1.3.1 Why These Interventions Have not Been Possible Before?	. 45
	2.1.3.2 Ensuring Ease-Of-Use	. 45
	2.1.3.3 Gathering Person-Specific Training Data from End Users	. 46
	2.2 OTHER APPLICATIONS	. 46
3	RESEARCH CHALLENGES AND PRIOR WORK	. 47
	3.1 Assumptions	. 47
	3.2 CHALLENGES FOR EACH SYSTEM COMPONENT	. 47
	3.2.1 Off-the-Shelf Devices to Recognize Physical Activity	. 47
	3.2.2 Physical Activity Recognition Algorithms	. 49
	3.2.3 Existing Devices and Techniques to Measure Energy Expenditure	. 53
	3.2.4 Energy Expenditure Estimation Algorithms	. 59
	3.2.5 Exploiting Real-Time Physical Activity Feedback to Motivate Behavior	
	Change	61
	3.2.6 Ensuring Ease-Of-Use	. 62
	3.3 SUMMARY OF TECHNOLOGICAL LIMITATIONS TO ENABLE REAL-TIME PA TYPE	3
	AND INTENSITY DETECTION	. 63
4	APPROACH AND PROCEDURE	. 66
	4.1 OVERVIEW OF RESEARCH APPROACH	66
	4.2 OVERVIEW OF SYSTEM DESIGN	66
	4.2.1 Wearable Sensors	. 67
	4.2.1.1 MIT Environmental Sensors (MITes)	. 67
	4.2.1.2 The MT1 Actigraph	. 67
	4.2.1.3 The Polar Chest Strap Heart Rate Monitor	. 68
	4.2.1.4 The Cosmed K4b2 Indirect Calorimeter	. 68
	4.3 ACTIVITY RECOGNITION ALGORITHMS	. 69
	4.4 ENERGY EXPENDITURE ESTIMATION ALGORITHMS	. 71
	4.5 INTERACTIVE TRAINING OF ALGORITHMS IN REAL-TIME	. 72

	4.6 D	ATA COLLECTION	73
	4.6.1	Boston Medical Center Protocol	73
	4.6.2	Stanford and Boston University Protocol	73
	4.6.3	MIT Protocol	76
	4.6.4	MIT Energy Expenditure Dataset	78
5	EVAL	UATION	79
	5.1 Ri	EPORTING AND ANALYZING ACTIVITY RECOGNITION RESULTS	79
	5.1.1	Standard Quantitative Measures	79
	5.1.2	Neglected Performance Measures	80
	5.1.	2.1 Complexity of the Activities to Recognize	80
	5.1.	2.2 Training Data Requirements of an Algorithm	81
	5.1.	2.3 Sensor Data Requirements of the Algorithm	82
	5.1.	2.4 Usability Factors Imposed by an Algorithm	82
	5.1.3	Comparison Challenges	83
	5.1.4	Comparison Strategy in this Work	84
	5.2 Ri	EPORTING AND ANALYZING ENERGY ESTIMATION RESULTS	85
	5.2.1	Standard Quantitative Measures	85
	5.2.2	Neglected Measures	86
	5.2.3	Comparison Challenges	88
	5.2.4	Comparison Strategy in this Work	88
	5.3 TF	RAINING PROCEDURES AND ASSUMPTIONS	89
	5.3.1	Subject Dependent Evaluation	90
	5.3.2	Subject Independent Evaluation	90
	5.3.3	The Addition of the Garbage or Unknown Class	90
	5.3.4	Transitions Between Activities	90
	5.3.5	Assumption Made During the Estimation of Energy Expenditure	91
	5.4 A	CTIVITY RECOGNITION ALGORITHM EXPERIMENTS	91
	5.4.1	Cubic Spline Interpolation	92
	5.4.2	Band-pass and Low-pass Filtering of Accelerometer Signals	92
	5.4.3	Non-overlapping Sliding Windows Segmentation	93
	5.4.4	Can Fast Run-Time Classifiers Produce Acceptable Performance?	94
	5.4.5	Can Features Computed Over Each Sensor Reduce Computational	107
	516	What is the Optimal Sliding Window Langth to Use?	10/
	5.4.0 5.4.7	Which Eastures Provide the Most Discriminatory Information and Hoy	115
	5.4.7	I owest Computational Cost?	122
	548	Does the Incorporation of Heart Rate Data Improve Recognition of	122
	5.7.0	Activities?	134
	5.4.	8.1 How Well can Activities be Recognized from Heart Rate Data?	135
	5.4.	8.2 How Well can Activities be Recognized by Combining Acceleration	on
	2.11	and Heart Rate Data?	142
	5.4.9	How Well Can All Activities Be Recognized Using the Selected Classi	fier.
		Window Length, Feature Set, and Signal Preprocessing Techniques?	144
	5.4.	9.1 How Much Training Data is Required During Subject Dependent	
		Training?	147

	5.4.9.	2 What is a Good Compromise on the Number and Placement of Sensors
		to be Work for Recognition of Activities?
	5.4.9.	3 How Well Can All Activities be Recognized Without Differentiating
		Among Intensity Levels?
	5.4.9.	4 How Well Can Postures, Ambulation, and Two MET Intensity
		Categories be Recognized?
	5.4.9.	5 How Well can Postures and Ambulation be Recognized?
	5.4.9.	6 How Well can Postures be Recognized?
	5.4.9.	7 Summary of Results
5.5	Ноч	W DOES THE ACTIVITY RECOGNITION ALGORITHM AND ITS INTERACTIVE
	TRA	INING INTERFACE PERFORM IN REAL-TIME?
5.5	5.1	Real-Time and 10 Fold Cross-Validation Performance
5.5	5.2	Training Usability
5.6	Ene	BRGY EXPENDITURE ALGORITHM EXPERIMENTS
5.6	5.1	Overview
5.6	5.2	MIT Energy Expenditure Dataset Validation
5.6	5.1	How Well Has Energy Expenditure been Estimated From Accelerometer
		and Heart Rate Data in the Past?
5.6	5.2	How Well Can Energy Expenditure be Estimated Using Leading
		Algorithms from Prior Work?
	5.6.2.	1 How Well Can Energy Expenditure be Estimated Using a State-of the-
		Art Actigraph-Based Algorithm?
	5.6.2.	2 How Well Can Energy Expenditure be Estimated Using Simple Linear
		Regression Algorithms?
	5.6.2.	3 How Well Can Energy Expenditure be Estimated Using the
		Compendium of Physical Activities?
	5.6.2.	4 Estimating Energy Expenditure Using One Linear Regression Model
		per Activity. 205
	5.6.2.	5 How Well Can Energy Expenditure be Estimated Using One Non-
		Linear Regression Model per Activity?
	5.6.2.	6 Summary of Baseline Results
5.6	5.3	Can Fast Run-Time Regression Algorithms Produce Acceptable
		Performance?
5.6	5.4	Does Band-Pass Filtering of the Acceleration Signals Improve
		Performance?
5.6	5.5	Can Feature Computation Per Axis Improve Performance Over Feature
		Computation per Sensor?
5.6	5.6	What is the Optimal Sliding Window Length to Use?
5.6	5.7	Which Features Improve Energy Expenditure Estimation and Maintain
		Computational Cost Low?
5.6	5.8	How Well can Energy Expenditure Be Estimated by Computing the Highest
		Performing Features Over Subsets of Accelerometers?
5.6	5.9	Does Heart Rate Data Improve Energy Expenditure Estimation?
5.6	5.10	How Well can Energy Expenditure be Estimated by Combining
		Acceleration and Heart Rate Data?

	5.6.11	How Well Can Energy Expenditure be Estimated Using the Selected	
		Window Length, Feature Set, and Signal Preprocessing Techniques?	. 245
	5.6.1	1.1 How Well Can Energy Expenditure be Estimated Using Linear	
		Regression?	. 246
	5.6.1	1.2 How Well Can Energy Expenditure be Estimated Using One Linea	ar
		Regression Model Per Activity?	248
	561	1.3 How well Can Energy Expenditure be Estimated if the Number of	. 210
	5.0.1	Activities to Recognize is Reduced?	253
	561	1.4 How Well Can Energy Expenditure be Estimated by Predicting M	
	5.0.1	Volues per Activity?	256
	561	Values per Activity :	250
	5612	How Wall Can Time Spont in Dhysical Activity Level be Decognized?	250
	3.0.12	How went can time spent in Physical Activity Level be Recognized?.	. 200
6	CONCL	USIONS	. 263
	6.1 SUM	IMARY OF RESULTS	. 263
	6.1.1	Activity Recognition	. 263
	6.1.2	Energy Expenditure Estimation	. 266
	6.2 Pri	MARY CONTRIBUTIONS	. 270
	6.3 Rev	/isiting the Design Goals	. 272
	6.4 Uni	RESOLVED ISSUES FOR LONG-TERM DEPLOYMENT	. 274
	6.4.1	Hardware Limitations for Mobile Interventions	. 274
	642	Usability Issues and Impact on Algorithms	274
	65 Fur	TIRE RESEARCH DIRECTIONS	275
	651	Activity Recognition	275
	652	Energy Expenditure Estimation	. 275
	0.3.2		270
A	PPENDIX	A1: PERFORMANCE MEASURES FOR ACTIVITY	
R	ECOGNII	FION	. 277
A	PPENDIX	A2: CLUSTERING OF ACTIVITIES FOR PRESENTATION OF	
R	ESULTS		. 279
	DDENDIV	A 2. THE 41 FEATURES EVELOPED IN THIS WORK	101
A	PPENDIA	A3: THE 41 FEATURES EAPLORED IN THIS WORK	202
A	PPENDIX	A4: AMOUNT OF DATA IN THE MIT DATASET	. 287
A	PPENDIX	A5: SLIDING WINDOW LENGTH FOR ACTIVITY	
R	ECOGNII	FION	. 289
A	PPENDIX	A6: FEATURE SELECTION FOR ACTIVITY RECOGNITION	. 298
A	PPENDIX	A7: ACTIVITY RECOGNITION USING HEART RATE DATA	. 325
٨	PPFNDIX	A8. ACTIVITY RECOGNITION - FINAL PERFORMANCE	330
		A0. DEAL TIME INTED ACTIVE TO A DUDIC STUDY	202
A	FENDIX	A7: NEAL-IIIVIE INTEKAUTIVE IKAINING STUDY	302
A	PPENDIX	A10: BOSTON MEDICAL CENTER DATA COLLECTION	
P	ROTOCO	L	. 367
Δ	PPFNDIV	ALL STANFORD AND ROSTON UNIVERSITY DATA	

APPENDIX A12: MIT DATA COLLECTION PROTOCOL	374
APPENDIX A13: ACTIVITY RECOGNITION USING DIFFERENT CLASSIFIERS	. 381
APPENDIX A14: DESCRIPTION OF THE 52 ACTIVITIES CONTAINED IN THE MIT DATASET	. 385
APPENDIX A15: FEATURE COMPUTATION PER SENSOR VS. FEATURE COMPUTATION PER AXIS	. 387
APPENDIX A16: SUMMARY OF RECENT WORK IN ACTIVITY RECOGNITION FROM ACCELEROMETER DATA	. 392
APPENDIX B1: ESTIMATING ENERGY EXPENDITURE USING THE 2- REGRESSION CROUTER ALCORITHM	396
APPENDIX B2: ESTIMATING ENERGY EXPENDITURE USING SIMPLE REGRESSION ALGORITHMS	404
APPENDIX B3: ACTIVITIES UTILIZED TO PREDICT ENERGY EXPENDITURE USING FOLIATION (E)	408
APPENDIX B4: COMPARISON OF ACTIVITIES CONTAINED IN THE MIT ENERGY EXPENDITURE DATASET AND THE ONES FOUND IN THE COMPENDIUM OF PHYSICAL ACTIVITIES	410
APPENDIX B5: ESTIMATING ENERGY EXPENDITURE USING THE COMPENDIUM OF PHYSICAL ACTIVITIES	. 413
APPENDIX B6: ESTIMATING ENERGY EXPENDITURE USING LINEAR REGRESSION AND THE ACABSAREA FEATURE COMPUTED OVER ONE MINUTE WINDOWS.	E . <b>423</b>
APPENDIX B7: ESTIMATING ENERGY EXPENDITURE USING ONE LINE REGRESSION MODEL PER KNOWN ACTIVITY	AR 428
APPENDIX B8: ESTIMATING ENERGY EXPENDITURE USING ONE NON- LINEAR REGRESSION MODEL PER KNOWN ACTIVITY	. 432
APPENDIX B9: ESTIMATING ENERGY EXPENDITURE USING LINEAR A NON-LINEAR REGRESSION ALGORITHMS	.ND .436
APPENDIX B10: ESTIMATING ENERGY EXPENDITURE WHEN BAND-PA FILTERING THE ACCELEROMETER SIGNAL AND WHEN NOT	.SS . 440
APPENDIX B11: FEATURE COMPUTATION PER SENSOR VS. FEATURE COMPUTATION PER AXIS FOR ENERGY EXPENDITURE ESTIMATION .	. 442
APPENDIX B12: WINDOW LENGTH SELECTION FOR ENERGY EXPENDITURE ESTIMATION	. 444
APPENDIX B13: FEATURE SELECTION FOR ENERGY EXPENDITURE ESTIMATION	448
APPENDIX B14: ENERGY EXPENDITURE ESTIMATION USING HEART RATE DATA	. 452

APPENDIX B15: FEATURE SELECTION OVER SUBSETS OF	
ACCELEROMETERS FOR ENERGY EXPENDITURE ESTIMATION	457
APPENDIX B16: MIT ENERGY EXPENDITURE DATASET STATISTICS	461
APPENDIX B17: DATA COLLECTION SESSIONS INCLUDED IN THE MIT	Г
ENERGY EXPENDITURE DATASET	468
APPENDIX B18: PERFORMANCE MEASURES FOR ENERGY EXPENDIT	URE
ESTIMATION	470
APPENDIX B19: ACTIVITY-DEPENDENT REGRESSION MODELS	472
BIBLIOGRAPHY	481

## **List of Figures**

Figure 2-1: Non-exercise activity (NEAT) partitioned according to activity performed over the course of a day
<ul> <li>Figure 2-2: Daily exercise activity partitioned per activity performed. Graphs like this one could be presented at the end of the day to increase self-awareness related to physical activity of individuals.</li> <li>Figure 4-1: Wearable sensors used during the data collections. (a) Wireless accelerometers, (b) heart rate transceiver, (c) MT1 Actigraphs, (d) Polar WearLink chest strap heart rate monitor, (e) HJ-112 pedometer, and (f) the bodybugg armband.</li> </ul>
Figure 4-2: Example of the data collection setup for the Boston Medical Center Protocol. The indirect calorimeter used was the Parvo Medics TrueOne 2400. Participant's face has been blurred to preserve anonymity
Figure 4-3: Five triaxial wireless accelerometers, a polar chest strap heart rate monitor, a MITes heart rate transceiver with 9V battery attached, and a USB wireless receiver, and (b) Placement of the sensors on the body. Green dots indicate accelerometer locations and blue the heart rate monitor location
Figure 4-4: Example of how participants wore the sensors used during the data collections at the (a) gymnasium (shown here) and (b) home. Participant's face has been blurred to preserve anonymity
Figure 5-1: Energy expenditure data collected using the Cosmed K4b2 indirect calorimeter from subject MIT-001. (a) data collected at the gymnasium from 32 activities (b) same gym data with non-steady-state data eliminated and treadmill activities highlighted in red, (c) same data when non-steady-state data has been eliminated and the remaining data has been concatenated (note that in this plot, the time axis has a different scale than for plot a and b) and (d) energy expenditure data collected in a real home while performing 19 activities
Figure 5-2: True positive rate per activity for all the classifiers using the <i>MaxAcceleration</i> feature set and subject dependent evaluation. The grayscale image is scaled so that the maximum true positive rate of 99.7% is represented by the color white and the minimum of 45.7% by the color black. In other words, poor areas of performance are shown in black
Figure 5-3: False positive rate per activity for all the classifiers using the <i>MaxAcceleration</i> feature set and subject dependent evaluation. The grayscale image is scaled so that the minimum false positive rate of 0% is represented by the color white and the maximum of 7.63% by the color black. In other words, poor areas of performance are shown in black
Figure 5-4: True positive rate per activity for all the classifiers using the <i>MaxAcceleration</i> feature set and subject independent evaluation. The grayscale image is scaled so that the maximum true positive rate of 96.7% is represented by the color white and the minimum of 4.5% by the color black. In other words, poor areas of performance are shown in black.
Figure 5.5: Folse positive rate per activity for all the classifiers using the

Figure 5-5: False positive rate per activity for all the classifiers using the *MaxAcceleration* feature set and subject independent evaluation. The grayscale image is scaled so that the minimum false positive rate of 0.01% is represented by

- Figure 5-6: Comparison of true positive rate per activity for the NB and C4.5 Classifiers using the *MaxAccelerationSet1* without the *Unknown* Class evaluated using subject dependent training. The grayscale image is scaled so that the maximum true positive rate of 99.9% is represented by white and the minimum of 67.7% by black. ...... 102
- Figure 5-8: Comparison of true positive rate per activity for the NB and C4.5 Classifiers using the *MaxAccelerationSet1* without the *Unknown* Class evaluated using subject dependent training. The grayscale image is scaled so that the maximum true positive rate of 98.8% is represented by white and the minimum of 11.3% by black. ...... 103
- Figure 5-10: Confusion matrices for the NB and C4.5 classifier without the *unknown* class using subject dependent evaluation and the *MaxAcceleration* feature set. .... 105
- Figure 5-11: Confusion matrices for the NB and C4.5 classifier without the *unknown* class using subject independent evaluation and the *MaxAcceleration* feature set. 106
- Figure 5-12: True positive rate per activity using the *MaxAcceleration* feature set computed per sensor and per axis and evaluated using subject dependent training with the NB and C4.5 classifiers. The grayscale image is scaled so that the maximum true positive rate of 100% is represented by white and the minimum of 60.3% by black. In other words, poor areas of performance are shown in black... 109
- Figure 5-14: True positive rate per activity using the *MaxAcceleration* feature set computed per sensor and per axis and evaluated using subject independent training with the NB and C4.5 classifiers. The grayscale image is scaled so that the maximum true positive rate of 98.8% is represented by white and the minimum of 4.2% by black. In other words, poor areas of performance are shown in black..... 111

- Figure 5-24: Graphic representation of the segmentation of heart rate and accelerometer data using the sliding window approach. Note that the windows for accelerometer and heart rate data are of different lengths and that the heart rate sliding window extends from the end of the end of the accelerometer window backwards in time. 135
- Figure 5-26: Confusion matrices for the C4.5 classifier when recognizing the 51 activities in the MIT dataset without including the *unknown* class during (a) subject dependent and (b) independent training. The feature set used is the *invariant reduced* computed per axis over windows of 5.6s in length. The maximum number of confusions per cell is 95 and 262 for subject dependent and independent training respectively.... 146
- Figure 5-27: Graphical representation of the experiment performed to measure the performance of utilizing different amounts of training data during subject dependent training. The right side of the table represents the duration of an individual activity example performed by one of the participants. The start of the activity is represented by the (start) tag and the end by the (end) tag. The 25% of the data located at the end

- Figure 5-37: Confusion matrices for the C4.5 classifier when recognizing postures and ambulation without intensity levels and without considering the *unknown* class during subject dependent and independent evaluation. The feature set used is the *invariant reduced* feature set computed per axis over windows of 5.6s in length over all accelerometers. The maximum number of errors in a given cell of a confusion matrix is 26 and 168 for subject dependent and independent training respectively.

- Figure 5-44: Pseudo code for the 2-regression Crouter et al. [34] Actigraph-based energy expenditure estimation algorithm. CV stands for coefficient of variation and is computed by dividing the standard deviation by the mean over a window of 10s. 194

- Figure 5-53: Root mean squared error per activity as a gray scale image when multivariable linear regression is used to estimate energy expenditure over several subset of accelerometer-based features computed per sensor over windows of 5.6s in length. The image is scaled to show the lowest RMSE error of 0.2MET in white and the largest of 2.2MET in black. Subset labeled as S1 corresponds to the *ACFFTPeaks* and *ACAbsArea* features, S2 corresponds to the *ACFFTPeaks*,

# List of Tables

Table 3-1: Most popular ambulatory monitoring technologies available off-the-shelf to estimate energy expenditure and measure physical activity. NA stands for non
applicable
Table 3-2: Comparison of advantages and disadvantages of existing methods used to
estimate energy expenditure
Table 4-1: Physical characteristics of the 20 study participants. Values are means (SD)
with range shown in perenthesis, n is the number of subjects
Table 4.2. Diversional advantation of the 16 mention and a drive the MIT encourse
Table 4-2: Physical characteristics of the 16 participants included in the MIT energy
expenditure datasets. Values are means $\pm$ SD with range shown in parenthesis and n is the number of subjects 78
Table 5-1: Comparison of two state-of-the-art activity recognition algorithms along all
the performance dimensions discussed in Sections 5.1.1 and 5.1.2. The comparison
is difficult, and it is not clean what clean the would not form better under similar
is difficult, and it is not clear what algorithm would perform better under similar
conditions
Table 5-2: Parameters used in the design of the Band-Pass Chebyshev Type I infinite
impulse response (IIR) Filter designed to filter the accelerometer signals between
0.1 and 20Hz and the equiripple low-pass finite impulse response filter (FIR)
designed to filter the accelerometer signals below 1Hz. All frequency values are in
Hz
Table 5-3: Brief description of the classifiers explored in this section, their parameters
used during the experiments presented in this section and references to papers
describing them in full detail
Table 5.4: The total training time and classification times (in seconds) required to train
the elegifiers using the date from subjects MIT 002 to MIT 020 (10 subjects) and
the classifiers using the data from subjects MIT-002 to MIT-020 (19 subjects) and
classify the activity examples of subject MIT-001. The features used are the
ACAbsArea (21) and MaxAcceleration (247) computed over sliding windows of
5.6s
Table 5-5: True positive rate and false positive rate (shown in parenthesis) clustered per
activity category while classifying the 52 activities contained in the MIT dataset in a
subject dependent manner. Classifiers were trained by computing the
<i>MaxAcceleration</i> (247) feature set per accelerometer axis over sliding windows of
5.6s 97
Table 5-6: True positive rate and false positive rate (shown in parenthesis) clustered per
activity entergory while classifying the 52 activities contained in the MIT detection of
activity category while classifying the 52 activities contained in the wint dataset in a
subject independent manner. Classifiers were trained by computing the
<i>MaxAcceleration</i> (247) feature set per axis over sliding windows of 5.6s
Table 5-7: True positive rate and false positive rate (shown in parenthesis) clustered per
activity category while classifying the 51 activities contained in the MIT dataset
using the NB and C4.5 classifier without the unknown class. Classifiers were trained
by computing the <i>MaxAcceleration</i> (247) feature set per axis over sliding windows
of 5.6s
Table 5-8: Two techniques used to summarize the acceleration (motion) experienced by a
sensor: The sum over all acceleration axes $(r, y)$ and $(r)$ and the signal vector
senser. The sum over an accordance axes (x, y, and z) and the signal vector
magnitude 107

- Table 5-11: Window lengths explored while selecting the optimal window length to use for the activity recognition algorithm. The length is shown in number of acceleration samples and corresponding time in seconds, assuming a sampling rate of 45Hz... 116

 Table 5-16: Performance of the features that capture the spectral content of motion computed per axis using the C4.5 classifier and evaluated in a subject dependent and independent manner.

 126

 Table 5-17: Final ordering of the features according to their importance (decreasing order of importance from left to right) according to all the experiments performed in this section.

 126

- Table 5-25: Performance of recognizing the 52 activities contained in the MIT dataset (including the *unknown* class) from heart rate features only over windows of 5.6s using the C4.5 decision tree classifier during subject independent evaluation...... 137
- Table 5-27: Performance of combining features that attempt to describe the fitness level of an individual with heart rate features (*ScaledHR*) during subject independent evaluation using the C4.5 decision tree classifier. The target activities were the 51 activities contained in the MIT Dataset, without including the unknown class. .... 139

- Table 5-41: Subject independent performance of the C4.5 classifier when recognizing postures and ambulation using the *invariant reduced* feature set computed per axis over windows of 5.6s in length using different subsets of accelerometers. The *unknown* class was not included in this experiment. The *unknown* class was not

- Table 5-47: True positive and false positive rate (shown in parenthesis) during subject dependent training for the C4.5 classifier when recognizing postures without the *unknown* class over different sensor combinations. The feature set used is the *invariant reduced* feature set computed per axis over windows of 5.6s in length.. 175
- Table 5-48: True positive and false positive rate (shown in parenthesis) during subject independent training for the C4.5 classifier when recognizing postures without the *unknown* class over different sensor combinations. The feature set used is the *invariant reduced* feature set computed per axis over windows of 5.6s in length. 175
- Table 5-50: The three sensor combinations and the three single sensors with higher performance while recognizing different sets of activities from the MIT dataset using the final implementation of the activity recognition algorithm. The

performance is higher for sensor combinations shown at the top and lower for the Table 5-51: Performance obtained by recognizing participant's activities using a C4.5 classifier with the invariant reduced feature set computed per axis over window lengths of 5.6s in length. Performance was measured using 10-fold cross-validation. Table 5-52: Activities confused the most found when participants tested the performance Table 5-53: Average energy expenditure readings in METs for 14 activities (mean and standard deviation) collected using the Cosmed K4b2 indirect calorimeter from 16 subjects at MIT and the Parvo Medics indirect calorimeter from 2 subjects at the Table 5-54: Summary of some relevant results in energy expenditure estimation from accelerometer and heart rate data after year 2001. The numbers in the column Table 5-55: Error statistics for estimating energy expenditure for 52 activities using the 2regression Crouter et al. Actigraph-based algorithm with respect to the energy expenditure measured using the Cosmed K4b2 indirect calorimeter for the MIT Table 5-56: Root Mean Squared Error and Mean Absolute Error (shown in parenthesis) per activity category when estimating energy expenditure using the 2-regression Crouter et al. Actigraph-based algorithm. Errors are computed with respect to the energy expenditure measured using the Cosmed K4b2 indirect calorimeter for the Table 5-57: Performance over all activities when estimating energy expenditure for 11 activities using the 2-regression Crouter et al. algorithm over the Boston University Table 5-58: Performance per activity when estimating energy expenditure using the 2regression Crouter et al. algorithm over the Boston University dataset (n=2). RMSE stands for root mean squared error, MAE for mean absolute error, and MAED for Table 5-59: Form of the simple regression equations explored to estimate energy expenditure from the MIT energy expenditure dataset. Appendix A3 explains how Table 5-60: Regression equations and their coefficients learned from the energy Table 5-61: Correlation coefficient (r), root mean squared error, and mean absolute error (shown in parenthesis) obtained when estimating energy expenditure using the Table 5-62: Performance of estimating energy expenditure using the Compendium of Physical Activities over the *comparable* and *closest* activity sets. Energy Table 5-63: Root mean squared error and mean absolute error (shown in parenthesis) per activity category when estimating energy expenditure using the Compendium of 

- Table 5-68: Comparison of the performance obtained using the energy expenditure estimation methods explored in this section. All methods estimate energy expenditure over sliding windows of one minute in length. The methods that utilize linear or non-linear regression are trained using the ACAbsArea feature computed per sensor.

   209

Table 5-82: Root mean squared error and mean absolute error (shown in parenthesis) obtained when estimating energy expenditure in a subject independent manner using linear regression and the *ACAbsArea* feature computed per sensor using windows of 5.6s in length over different subsets of accelerometers. Energy expenditure is estimated for the 51 activities contained in the MIT energy expenditure dataset. 234

Table 5-83: Root mean squared error and mean absolute error (shown in parenthesis) obtained when estimating energy expenditure in a subject independent manner using linear regression and the *ACFFTPeaks* feature computed per sensor using windows

of 5.6s in length over different subsets of accelerometers. Energy expenditure is estimated for the 51 activities contained in the MIT energy expenditure dataset. . 234

- Table 5-84: Root mean squared error and mean absolute error (shown in parenthesis)obtained when estimating energy expenditure in a subject independent manner usinglinear regression and the ACFFTPeaks + ACModVigEnergy + ACMCR featurescomputed per sensor using windows of 5.6s in length over different subsets ofaccelerometers. Energy expenditure is estimated for the 51 activities contained in theMIT energy expenditure dataset.235
- Table 5-86: Root mean squared error and mean absolute error (shown in parenthesis)

   obtained when estimating energy expenditure in a subject independent manner using

   linear regression and the *Invariant reduced* feature set computed per sensor using

   windows of 5.6s in length over different subsets of accelerometers. Energy

   expenditure is estimated for the 51 activities contained in the MIT energy

   expenditure dataset.

   236

- Table 5-91: Root mean squared error and mean absolute error (shown in parenthesis) obtained while estimating energy expenditure using linear regression when the most discriminating accelerometer (*invariant reduced* feature set) and heart rate feature (*ScaledHR*) are used and computed per sensor over all the seven accelerometers. Energy expenditure was predicted over the 51 activities contained in the MIT energy expenditure dataset. The definition of the *Invariant* and *fast to compute* feature sets

- Table 5-98: Root mean squared error and mean absolute error (shown in parenthesis) obtained when energy expenditure is estimated by first recognizing 11 activities in a subject independent manner and activity dependent linear regression models trained in a subject independent manner are applied. When activities are recognized using

- Table 5-99: Root mean squared error and mean absolute error (shown in parenthesis) obtained when energy expenditure is estimated by first recognizing 11 activities in a subject dependent manner and activity dependent linear regression models trained in a subject independent manner are applied. When activities are recognized using the *fast to compute* feature set, energy expenditure is predicted using the *ACAbsArea* feature. When the *invariant reduced* feature set is used to recognize activities, energy expenditure is predicted using the *ACFFTPeaks* feature. 254

- Table 5-104: Accuracy in recognizing time spent sedentary, light, moderate, vigorous, and very vigorous physical activity using the different energy expenditure estimation algorithms explored in this section when accelerometers at the hip, dominant wrist, and dominant foot are utilized. 261

### **1** Introduction

Due to its dramatic increase over the past decades, obesity is now considered a global epidemic that may dramatically impact health in the industrialized world [1]. The prevalence of obesity from 1960 to 1994 in the U.S alone increased approximately 50% from 13.4% to 22.3%. At present, 65% of adult Americans aged 20 years and older are considered overweight (a body mass index [BMI]  $\geq 25 \text{ kg/m}^2$ ) and 30% are considered obese (BMI  $\geq 30 \text{ kg/m}^2$ ). Further, 16% of children and teenagers (age 6-19) are presently considered overweight and the percentages are rising [2].

If this trend in obesity continues and no action is taken, the majority of the U.S. adult population could be overweight within a few generations [3]. Extrapolating from existing data from the World Health Organization [1], approximately half (45-50%) of the U.S. adult population could be obese by year 2025. This is an alarming statistic because obesity is linked to leading causes of death in the U.S., such as heart and pulmonary diseases, stroke, and type 2 diabetes [2]. Obesity is also a risk factor for chronic conditions such as high blood cholesterol, hypertension, and osteoarthritis [4].

Obesity is a complex condition caused by the interaction of many factors such as such as genetic makeup, neuroendocrine disorders, emotions, and even secondary effects from medical treatments. The rise in obesity, however, is generally believed to result from a caloric imbalance. Most Americans have (1) high caloric intake due to easy access to foods and beverages with a high caloric content and (2) extremely low levels of physical activity relative to that caloric intake [5-7]. Increased calorie intake may have resulted from pervasive advertising and ubiquity of the fast food industry [8], easy access to energy-dense packaged foods such as snacks and soft beverages [9], and the expanding portion sizes of meals and food packages [10]. Advances in transportation and household appliances have also contributed by minimizing everyday physical activity and encouraging sedentary behaviors. Many leisure activities such as sports and outdoor activities have been increasingly replaced by sedentary behaviors such as television viewing, arcade videogame playing (particularly in children), and internet surfing [11, 12]. In fact, a study by Harris Interactive released in 2003 [13] found that Americans 13 years and older spend on average eight hours a day sitting and four hours a day watching TV, playing video games, or surfing the web.

Energy intake and energy expenditure are determined by the individual's daily behavior and, consequently, addressing behavior change has been an important focus of work in treating obesity. For example, clinical behavioral interventions attempt to modify behavior related to physical activity and diet by educating individuals about the benefits of daily exercise, healthier food choices, and portion sizes. These interventions have shown some level of success in reducing body mass index (BMI) and increasing physical activity levels in the short-term (e.g. [14]). Nevertheless, this success may primarily result from the network of health care professionals and researchers constantly monitoring each individual's behavior, making the interventions too costly to scale to the entire population. When people are left to manage their own weight, without the constant support of professionals, they struggle to maintain newly acquired healthy behaviors [15, 16]. In fact, ninety five percent of people who have lost weight during these interventions regain approximately one third of the lost weight during the following year and are usually back to their initial weight within three to five years [17]. Ultimately, if *cost-effective* long-term behavior change is required, individuals must initiate and sustain healthy habits related to physical activity and diet without continuous monitoring by health professionals.

Daily information about physical activity level, energy expenditure, and energy intake is central to weight control since energy balance is defined by the interaction between these variables [6, 12, 18]. Presently, many people have little or no idea how much daily meaningful physical activity they get and how that compares to the minimum standards suggested by medical professionals and their estimated caloric intake. Consequently, there is a need for tools that allow people to measure their physical activity intensity and intake over the course of a day to allow them to make informed decisions and perform energetic trade-offs (e.g. increase physical activity while overeating or reduce food consumption while sedentary). Moreover, tools that automatically inform individuals about how small changes in behavior (e.g. in non-exercise activity) could positively impact their daily energy expenditure might allow people to plan personalized modifications to daily routine that are more likely to be conducive to incremental adoption and subsequently sustained over long periods of time.

Unfortunately, existing technologies in the area of physical activity are mostly designed for those individuals who have already achieved a high level of physical fitness, such as athlete runners. For example, the Nike+iPod sport kit [19] allows individuals to track the speed, distance, pace and calories burned while running by slipping a motion sensor in the Nike+iPod ready shoe and snapping a wireless receiver unit into the iPod nano MP3 player. The Adidas+Polar training system [20], another recently introduced technology, integrates a heart rate monitor on a shirt, a stride sensor on a shoe, and a wristwatch computer to monitor work out zones based on heart rate, speed and distance data. These new portable technologies demonstrate the use of real-time biofeedback as a way to motivate behavior modification, since heart rate data and energy expenditure is used to maintain the work out at particular zones. Still, the challenge remains to come up with technologies that can be used by those who have difficulty maintaining a healthy weight and minimum levels of physical activity every day.

In summary, there is a need for weight maintenance technologies that (1) use automation so that they can scale well in cost to the majority of the population, (2) present information about physical activity levels, energy expenditure, and energy intake in real-time over the course of a day to help people to make more informed decisions related to physical activity and diet, (3) create opportunities for interventions that permit incremental changes that do not necessitate sudden and drastic modifications to daily routine, (4) teach individuals skills that will help them in maintaining their weight in the long term, and that (5) are inexpensive, easy to use and unobtrusive so that people are willing to use them longitudinally during free-living.

#### **1.1 The Opportunity**

As the popularity of portable handheld computers such as mobile phones increases and their cost decreases, opportunities for novel healthcare applications arise. Mobile phones are often carried with people nearly everywhere they go, and people generally keep them functioning and charged [21]. Consequently, they can be used to gather and deliver

tailored health-related information continuously over long periods of time during freeliving conditions.

One important area where mobile phones and wearable accelerometers can be applied in combination is in creating valid and reliable measures of physical activity and energy expenditure. Automatic detection of physical activity and/or energy expenditure would enable new types of health assessment and intervention tools that help people maintain their energy balance and stay physically fit and healthy. For example, mobile phones could be used to run algorithms that automatically recognize physical activities and estimate energy expenditure from body-worn accelerometers and display this information as behavioral feedback in real-time. Indeed, with the advent of accelerometer-enabled mobile phones (e.g. [22-25]), some applications such as mobile phone based pedometers and activity level monitors have started to emerge [25, 26].

Another powerful extension of mobile technology is to use it to deliver "just-in-time" interventions at the point of decision, for example, to encourage a positive behavior change [27, 28]. In this scenario, accelerometer-based mobile phones or mobile phones combined with wearable accelerometers distributed at strategic body segments can be used to detect activities of interest (e.g. *walking slowly*) and encourage increases in intensity levels (e.g. *brisk walking* over *walking slowly*). Obviously, for these types of applications to be possible, activity recognition algorithms running on mobile phones have to be capable of recognizing the intensity of physical activity.

A new area of research where mobile phones and wearable accelerometers might also be applied is non-exercise activity thermogenesis (NEAT) [29]. Recent results suggest that small changes to daily routine such as *walking upstairs* vs. *riding the elevator*, *sitting fidgeting feet* vs. *sitting* and *brisk walking* vs. *walking* can accumulate over the course of a day to meaningful amounts.

Mobile phones could also be used to acquire information related to food intake, body weight and body composition automatically and use it to provide useful estimates of energy balance over time. In fact, Some mobile phone based commercial applications such as MyFoodPhone [30] and Sensei [31] have already started to appear. These applications allow individuals to collect dietary information on phones and, if desired, receive useful feedback from dietitians. Unfortunately, better applications are still required that do not either rely on experts to manually analyze data or end-users to painstakingly enter information about what they eat using a phone.

#### **1.2 Challenges of Estimating Energy Balance**

Unfortunately, the accurate measurement of physical activity, energy expenditure, and energy intake is challenging and, at present, there is no technology that allows people to measure these variables comfortably, accurately, and continuously over the course of a day and obtain real-time feedback. Existing technologies capable of measuring these variables accurately that are used in research (e.g. diet and physical activity diaries, indirect calorimetry, doubly labeled water, and chemical analysis of duplicate food samples) impose a considerable degree of burden to the end user due to (1) the need of maintaining detailed daily records related to physical activity and diet or (2) the use of intrusive and expensive medical equipment available only at specialized laboratories. When methods more suitable for free-living conditions are used to estimate these variables (e.g. recall surveys, end of study interviews, and non-detailed diaries), they provide too coarse of an estimate to allow for useful consumer-based applications.

Motion sensors (accelerometers) are commonly used in exercise physiology research to measure physical activity quantitatively and estimate energy expenditure during freeliving. A common method used during large scale medical research studies is to place a single accelerometer (e.g. Actigraph [32]) at the hip to obtain an estimate of physical activity level (e.g. *light, moderate, vigorous*) and energy expenditure due to overall body motion (e.g. ambulation). This method may perform poorly on some activities that involve primarily upper or lower body motion, which can be difficult to detect from a single accelerometer at the hip [33, 34]. Another disadvantage of this and another existing technologies (e.g. [33, 35, 36]) is that they do not provide any information about the type of activity being performed.

Recognizing activity type (e.g. *washing dishes* vs. *wiping a surface*) across individuals is challenging because individuals perform activities differently. Although there has been extensive research in the area, most algorithms implemented have been evaluated off-line and consequently, it is not clear if they are capable of real-time performance (e.g. [37-40]). Those few algorithms that recognize activities in real-time either recognize a limited set of activities involving mainly postures and ambulation (e.g. [41-44]) or are in a prototype research stage (e.g. [45, 46]). Furthermore, to the best of the author's knowledge, there are presently no technologies or algorithms available to automatically recognize the intensity of physical activity (e.g. *walking at 3mph vs. walking at 4mph*). As a result, the only available methods to capture to capture information about the intensity of physical activity are direct observation and physical activity diaries.

Estimation of energy expenditure presents some additional challenges over physical activity recognition. For example, energy expenditure needs to be estimated in a subject independent manner due to the unavailability of the necessary equipment to allow people to collect energy expenditure data about their activities during free-living. Furthermore, one of the main challenges in estimating energy expenditure is inter-individual variations since two individuals performing the same activity would present different energy expenditure measurements depending on their fitness level, age, and gender. Another challenge is that resistance or work load effort involved in activities such as *walking uphill* or *carrying a heavy box* are difficult to detect from accelerometers. Heart rate data, on the contrary, is able to detect these changes in physical effort, but it suffers from inter-individual variations due to different fitness levels of individuals and intra-individual variations due to emotional states, nicotine, and temperature among others [47-49].

There are still some questions in physical activity recognition and energy expenditure estimation research that prior work has partially addressed or not addressed at all. Some of these questions include: What features computed over the accelerometer data allow better recognition of activities and estimation of energy expenditure? What sliding window lengths (or epochs) provide the highest performance? Do multiple accelerometers at different body segments improve performance? How does the performance of relatively simple classifiers amenable for real-time performance compare to more complex state-of-the-art classification algorithms? Do non-linear regression techniques improve energy expenditure estimates significantly? Do activity dependent regression models improve energy expenditure estimation? Does the combination of accelerometer and heart rate data improve performance? What is the minimum number of
accelerometers required to recognize activities and estimate energy expenditure? Where these accelerometers should be placed to maximize usage comfort and performance? The work presented in this thesis explores some possible answers to all of the above questions by performing a set of experiments to find a reasonable compromise that balances the various criteria required to create activity recognition and energy expenditure algorithms amenable for real-time performance in low processing power devices.

### **1.3 System Goals**

The main goal of the work presented in this thesis is to demonstrate the viability of technology that can reliably detect information about activity type and intensity, and estimate energy expenditure from accelerometers in real-time. Such technology might eventually enable the development of a consumer-based energy expenditure meter that is easy to use, unobtrusive, inexpensive, always on and near the user, accurate, and that can be used longitudinally during free living conditions. Knowledge of physical activity and energy expenditure information in real-time over the course of a day could increase individuals' self-awareness and allow them to perform energetic trade-offs that might help in maintaining a healthy weight (e.g. exercise more when overeating or eating less when sedentary).

## **1.4 Experimental Goals**

In this work, algorithms for automatic detection of physical activity type and intensity and energy expenditure estimation using multiple wireless accelerometers are evaluated on a dataset collected from 20 participants. The dataset consists on data collected at a gymnasium under relatively controlled laboratory conditions and at a residential home under less controlled free-living conditions. During the data collections, participants wore seven wireless accelerometers, a heart rate monitor, and a portable indirect calorimeter to collect data about motion patterns associated with activities, heart rate, and energy expenditure. This dataset is particularly challenging (for the algorithms implemented) since it consists on data collected for 52 different activities, 26 of which have different intensity levels and 18 of them which include examples of the unconstrained motion found in household activities. The parameters of the activity recognition and energy expenditure algorithms such as type of classifier/regressor, feature set, window length, sensor set utilized and their placement on the human body are selected by performing a set of incremental experiments. These experiments have the goal of identifying a set of parameters that could enable robust real-time performance. Once the parameters are selected, the activity recognition algorithm is evaluated on various sets of activities. First, the algorithm is evaluated over all the 52 activities contained in the dataset. Then, it is evaluated again over all activities but without differentiating among activities containing different intensity levels. Later, the algorithm is evaluated over activities involving postures and ambulation, and then over activities involving postures, ambulation and two MET intensity levels, and finally; the algorithm is evaluated by only recognizing postures. The results presented in each of the experiments are clustered according to five activity categories to better understand the performance of the algorithm: Postures,

ambulation, exercise activities, resistance exercise activities, and household activities. To the best of the author knowledge, the dataset collected for this work is larger and more complex that other datasets used in activity recognition studies published to date.

This thesis explores the following:

- The development of algorithms that recognize physical activities and estimate energy expenditure from accelerometer data in real-time amenable for implementation in low-processing power devices such as mobile phones.
- The development of algorithms that recognize not only the physical activity type (e.g. *walking* vs. *cycling*) but also the intensity (e.g. *walking at 3mph* vs. *walking at 4mph*) of some activities.
- The development of a system that simultaneously recognizes activities and estimates energy expenditure from a set of three accelerometers worn at the hip, dominant wrist, and dominant foot.
- The exploration of the impact in performance of different signal processing techniques (e.g. band-pass filtering, data smoothing) and feature computation methods (e.g. feature computation per axis and per sensor) during activity recognition and estimation of energy expenditure.
- The examination of the impact in performance per activity of varying the sliding window length used during physical activity recognition and estimation of energy expenditure.
- The exploration of the subsets of features with highest performance (out of a set of 41 features) during the recognition of physical activities and estimation of energy expenditure.
- The study of the impact in performance achieved by incorporating heart rate data during physical activity recognition (e.g. to better recognize the intensity of activities) and estimation of energy expenditure.
- The exploration of the minimum set of sensors to utilize and their locations in the human body (out of a total set of seven) that maximize performance and usage comfort during recognition of physical activities and estimation of energy expenditure.
- The exploration of how well can different sets of activities be recognized in increasing order of complexity such as postures (4 activities), postures+ambulation (8), postures+ambulation+MET intensity (11), all activities with no intensities (31), and all 52 activities (including exercise and household activities and the garbage or unknown activity).
- The evaluation of activity recognition and energy expenditure estimation algorithms in a subject dependent and independent manner.
- The evaluation of the training data requirements for subject dependent recognition of activities.
- The comparison in performance between complex state-of-the-art classifiers/regressors and simpler classifiers/regressors amenable for real-time performance during the recognition of activities and estimation of energy expenditure.

- The real-time evaluation of the final version of the activity recognition algorithm implemented during a short study where individuals interactively provide the training data required to recognize 10 activities of their choice.
- The investigation of the performance achieved if activity dependent regression models are used to estimate energy expenditure.
- The exploration of the difference in performance between estimating energy expenditure using regression techniques and scoring of activities using the Compendium of Physical Activities.
- The development of a system to recognize activities and estimate energy expenditure that might enable mobile phone based interventions for obesity in the near future that are (1) scalable to large populations due to the use of readily available mobile phones and low-cost sensors, and (2) that can be used longitudinally during free-living conditions due to their ease-of-use and low burden.

# **1.5 Organization of Thesis**

This thesis is organized as follows. Chapter 2 presents a usage scenario and extended examples of how the work developed in this thesis might be used in mobile phone energy measurement interventions. The chapter also describes the system components required to achieve such interventions and their desired characteristics. In Chapter 3, background in prior research and methods in the areas of physical activity recognition and energy expenditure estimation is introduced. Chapter 3 also highlights some of the research challenges and open questions in these areas and explains the existing technological limitations that have prevented the development of mobile phone physical activity detection and energy expenditure estimation interventions. Chapter 4 presents an overview of the system designed and implemented in this work, the research approach followed to collect the data to develop and evaluate the algorithms, and the incremental set of experiments designed to select the parameters of the algorithms presented. Chapter 5 starts by discussing how results are reported and analyzed for the activity recognition and energy expenditure estimation algorithms and continues by presenting the results obtained in each experiment designed to select the parameters of these algorithms. In addition, Chapter 5 presents the evaluation of the final version of each algorithm in full detail. Finally, Chapter 6 presents the main contributions of this work, a discussion of unresolved issues concerning long term deployment of the systems implemented, and some recommendations for future work.

# 2 The System Goal

As the popularity of portable handheld computers such as mobile phones increases and their cost decreases, novel healthcare opportunities arise. Mobile phones are often carried with people nearly everywhere they go [21] and people generally keep them functioning and charged. Consequently, they can be used to deliver and gather tailored health-related information in free-living situations. The work presented in this thesis attempts to take advantage of this opportunity by developing activity recognition and energy expenditure estimation algorithms that are amenable for real-time implementation in mobile phones. These algorithms, in combination with a small set of wireless accelerometers could allow the development of a consumer-based energy expenditure meter that is easy-to-use, unobtrusive, fun and that could be utilized longitudinally during free-living conditions. This section explains the different technology components required to achieve such scenario, and explores different alternatives to utilize the real-time information related to physical activity and energy expenditure during free-living.

# 2.1 Mobile Phone Energy Balance Interventions

Most interventions for obesity have relied on healthcare professionals (e.g. clinical behavioral interventions), required drastic modifications to daily routine (e.g. low calorie diets and exercise routines), or imposed a high level of burden on users (e.g. dietary records). Moreover, most technology currently available to asses physical activity is either only utilized by people who have already achieved a good level of physical fitness (e.g. iPod+Nike sport kit [19]) during short periods of time or by medical professionals during research studies (e.g. Actigraph [32]).

As a result, one of the main goals of this work is to provide a technical foundation on which others can develop mobile phone energy expenditure interventions that can be used longitudinally during free-living, that are inexpensive, and scalable to large populations because they do not rely on humans and because they utilize technologies readily available (mobile phones and wireless accelerometers).

# 2.1.1 Mobile Phone Energy Balance Interventions Wish List

The technical contributions of this work support the development of a real-time physical activity detection and energy expenditure device. This might enable interventions to be considered that take advantage of real-time physical activity type and intensity detection to achieve the following properties:

Presents real-time information at convenient times or at the point of decision. The
intervention might provide real-time information about physical activity type and
intensity over the course of a day. Knowledge of physical activity type might make it
possible to determine good times to present feedback (e.g., during bouts of physical
activity).

- Provides objective feedback. People sometimes have difficulty judging if they have met guidelines for physical activity, such as 30 minutes of "brisk walking." An objective measure of energy expenditure or time spent in certain types of activities might help people make more informed decisions about how and when they get physical activity.
- Provides tailored feedback. By tracking physical activity type and intensity over long periods of time, interventions may then be able to present tailored feedback to motivate behavior change (e.g., today's activity is 10% less vigorous than yesterday's).
- Permits incremental behavior change. Real-time feedback about physical activity
  may enable new interventions focused on rewarding positive behaviors rather than
  only suggesting new behaviors that require drastic modifications to daily routine.
  Interventions may help people integrate more intensity into the activities they
  already do, rather than simply suggesting that people make radical changes to
  schedules that they feel they cannot achieve (e.g., going to the gym for 30 minutes a
  day).

### **2.1.2 Required Components**

This section briefly discusses the system components required to implement mobile phone based interventions that utilize physical activity type, intensity, and energy expenditure information. The necessary system components are (1) an activity recognition system and (2) an energy expenditure estimation system. Here, it is assumed that the system is implemented on a mobile phone platform. In this work we demonstrate the technology is feasible for real-time performance on a PC but anticipate that real-time performance should be possible on emerging mobile phone technology as well. Implementation on a phone would permit real-time interventions based on automatically detected physical activity type and intensity.

### 2.1.2.1 Automatic Recognition of Physical Activities

The physical activity recognition system recognizes a variety of everyday postures, physical activities, household activities and common exercise routines from a small set of wearable accelerometers using pattern classification techniques. This systems needs to be capable of (1) real-time performance with short classification delays in mobile phones, (2) recognition of activity type (e.g. *walking* vs. *cycling*), intensity (e.g. *walking at 5mph* vs. *walking at 6mph*), duration (e.g. seconds, minutes, hours), and frequency (e.g. daily, weekly, etc), (3) recognition of activities in a subject independent manner or in a subject dependent manner but with small training data requirements (e.g. few minutes per activity). Finally, the algorithm should only utilize a small number of accelerometers that can be worn comfortably at strategic but convenient body locations. This maximizes ease of use during free-living.

This thesis will present an activity recognition algorithm that recognizes various sets of physical activities (up to 52) from just three triaxial accelerometers worn at the hip, dominant wrist, and dominant foot (on top of shoe laces). These sensors could be

embedded in convenient easy-to-wear devices such as wristwatches, shoe pods or simply put inside the pocket in the case of the accelerometer at the hip. The algorithm was trained over data collected from 20 participants performing 52 activities at two locations: a gymnasium and an instrumented residential home. The algorithm is capable of real-time performance with short classification delays (5.6s) on existing laptop computers. The algorithm is also amenable for implementation in mobile phones due to its low computational requirements. Section 5.4 presents in-depth details on this physical activity recognition algorithm implemented.

### 2.1.2.2 Estimation of Energy Expenditure

The energy expenditure estimation system combines data collected from multiple wearable sensors (e.g. accelerometers or heart rate monitors) and user specific information, such as age, gender, weight, and height to generate estimates of energy expenditure (e.g. in METs or kcal/min) over the course of a day. The energy expenditure estimation system needs to be capable of (1) real-time performance with short estimation delays in mobile phones, (2) estimation of energy expenditure accurately for all types of activities, including activities with upper body, lower body, and overall body motion, and (3) estimation of energy expenditure in a subject independent manner, since the equipment required to collect energy expenditure data is expensive and unavailable to most people. Similarly the algorithm should provide the estimates from a small set of sensors strategically placed on the human body to maximize comfort and performance.

The energy expenditure estimation algorithm implemented in this work estimates energy expenditure by applying different models depending on the types of activities being performed by individuals. The activity dependent models were trained on data collected from 16 individuals (6hrs per individual) collected at a gym and at a residential home. The algorithm performs in real-time with short classification delays (5.6s) and is able to estimate energy expenditure for activities that have proven difficult in the past. These activities include upper body activities (e.g. *bicep curls* and *bench weight lifting*) and non-ambulatory lower body activities (e.g. *cycling*). This energy expenditure estimation algorithm also relies on the data collected by three wireless accelerometers placed at the hip, dominant wrist, and dominant foot. This was found to be the good sensor combination to capture upper body, lower body and overall body motion. As stated previously, these three sensors could be ultimately embedded in easy-to-wear devices such as wristwatches, shoe pods, and belt-clips.

Presently, existing techniques that produce the most accurate measurements of energy expenditure are either only suitable for laboratory due to the specialized equipment and expertise required (e.g. indirect calorimetry), used in small population studies due to their high cost (e.g. doubly labeled water), or do not provide continuous measurement with low user burden (e.g. activity diaries). The algorithm presented in this work allows the automatic estimation of energy expenditure and recognition of activity context during free-living from a small set of wearable accelerometers. This system improves on currently utilized accelerometer-based techniques by improving energy expenditure estimation for activities involving upper body and lower body activity and by also recognizing the activity type and intensity being performed.

## 2.1.3 Scenarios Exploiting Real-Time Feedback

This section explores how the technical results for physical activity type and intensity detection and energy expenditure estimation presented later in this thesis might be used in novel, real-time interventions.

## Real-Time Personalized Behavioral Feedback Mobile-phone Screensaver

Personalized behavioral feedback related to physical activity and energy expenditure could be presented as background images or screen savers on mobile phones over the course of a day. In this way, users could have access to this information in real-time by simply staring at the phone for a couple of seconds, such as when answering a phone call or checking on the time. The main advantage of this approach is the low level of burden imposed on users since no action other than staring at the phone's screen is necessary to initiate the feedback. Another advantage is that users can choose to ignore this information if they are busy, so this intervention is unlikely to be perceived as intrusive or burdensome. The information presented might be helpful to facilitate self-regulation by allowing individuals to perform energetic trade-offs during the course of the day, and to encourage a positive behavior change.

## End-of-day Behavioral Feedback Graphs

Another method that can be utilized to provide daily behavioral feedback related to physical activity and energy expenditure is to display end-of-day behavioral feedback graphs. These graphs could show the energy expended through non-exercise and exercise activity partitioned per activity performed. These graphs would have the objective of increasing individuals' understanding on how energy is spent while performing different activities over the course of a day. Some examples of end-of-day behavioral graphs are shown in Figure 2-1 and Figure 2-2.

### Encouraging Increases in Non-exercise Physical Activity

A relatively new and promising area of research is non-exercise activity thermogenesis (NEAT). This are of research suggests that small behavior modifications to daily routine (e.g. *sitting fidgeting legs* vs. *sitting, standing* vs. *sitting,* and *brisk walking* vs. *walking slowly*). can sum up over the course of a day and boost overall energy expenditure and thus have a protective effect against weight gain [29, 50, 51]. This is because most of the energy expended everyday comes from non-exercise activity. Thus, if a mobile phone could recognize non-exercise activities and their associated energy expenditure, it could suggest small changes in daily routine that might impact daily energy expenditure positively. The main advantage of this type of intervention is that small behavioral modifications are likely to be not perceived as drastic changes but have the potential to help in controlling weight problems.

# Non-Exercise Physical Activity



Figure 2-1: Non-exercise activity (NEAT) partitioned according to activity performed over the course of a day.



Exercise Physical Activity



### Just-in-time Interventions to Increase Physical Activity Levels

Real-time information about activity type, intensity and duration might allow the development of just-in-time interventions to foster increases in physical activity levels. For example, a mobile phone capable of recognizing different intensities of *walking* (e.g. *walking* at 2, 3, and 4mph) could encourage someone already *walking* to either increase the *walking* speed or the duration of *walking* to boost overall energy expenditure. The phone could also provide positive behavioral feedback by present information about the extra number of calories burned by increasing activity intensity or duration.

### 2.1.3.1 Why These Interventions Have not Been Possible Before?

The types of interventions previously mentioned have not been possible yet due to hardware and algorithmic limitations. For example, some of the hardware limitations include unavailability of real-time data due to the lack of power efficient wireless communication protocols to receive/send data from multiple sensors simultaneously and continuously and inconvenient sensors form factor for longitudinal use. Some of the algorithmic or software limitations that have prevented real-time interventions like the ones described are: (1) Coarse estimates since most existing technologies to measure physical activity provide too coarse of an estimate to allow useful applications (e.g. pedometers and single accelerometers at the hip), (2) Poor energy expenditure estimation performance over upper and lower body activity such as when using a single accelerometer at the hip, (3) Limited or no contextual information since existing technologies available to measure physical activity provide limited or not contextual information about the type, intensity, and duration of the physical activities being performed. (4) Lack of efficient algorithms that run in real-time on low-processing power devices. Most algorithms available for recognizing activities and estimating energy expenditure have been implemented to run offline and have not been tested for real-time performance. Section 3.3 presents a detailed discussion on why these types of interventions have not been previously possible after a discussion of the advantages and disadvantages of existing technology to measure physical activity and estimate energy expenditure.

### 2.1.3.2 Ensuring Ease-Of-Use

Mobile phone based applications that recognize activities, estimate energy expenditure, and trigger interventions would have to be easy to use so that they are used longitudinally during free-living. Ensuring ease-of-use involves addressing factors such as usage comfort of the sensors, the number of sensors utilized and their location in the human body, training time requirements of the algorithms if subject dependent training is used, capability of real-time performance, real-time recognition delay, and invariance to small changes in sensor location and orientation during the installation process. Low training data requirements during subject dependent training is important because it determines the time end-users would have to spend providing examples of the activities to recognize. The longer this time, the more likely end-users will find this training procedure difficult to complete or annoying. Similarly, real-time performance during subject dependent training is particularly important since it would allow individuals to test the performance of the algorithms quickly so the procedure is not perceived as burdensome. Finally, invariance to small changes in sensor orientation and location during installation on the body is important since these will occur during free-living conditions and the algorithms have to be able to perform well despite these variations. Section 3.2.6 describes prior work addressing some factors involved in creating systems easy to use and describes how the work presented this thesis addresses them.

### 2.1.3.3 Gathering Person-Specific Training Data from End Users

Previous work in activity recognition (e.g. [38]) suggests that subject independent recognition of activities is significantly more difficult than subject dependent recognition of activities. As a result, activity recognition algorithms are likely to perform better if users provide examples of the activities to recognize. Thus, activity recognition algorithms using subject dependent training will require efficient and intuitive user interfaces that allow users to interactively train and test the algorithms. Furthermore, intuitive user interfaces would also be required to allow individuals to 'fix' the recognition algorithms when they do not function properly. This could be achieved by either requesting more training data for activities with low performance or by allowing end-user to modify the inner workings of the algorithms. Section 5.4.9.1 explains how training data requirements for subject dependent recognition of activities are evaluated in this work. Section 5.5 presents the evaluation of the user interface implemented in this work to train and test the activity recognition algorithms implemented in real-time.

# **2.2 Other applications**

The algorithms implemented in this work to automatic recognize activities from sensor data can also be exploited by other context-aware applications in other domains. For example, they can be utilized in context sensitive reminders [52-56], context aware experience sampling for health research [57, 58], physical therapy or recovery [59, 60], sports training [60-63], personal health tracking [64-66], interactive games [67-69], autism research [70, 71], availability and interruptability prediction [72, 73], and automatic annotation of important events [73-76]. Thus, all these application areas would also benefit from activity recognition algorithms that can run in real-time in existing or future mobile phones.

# **3** Research Challenges and Prior Work

This section discusses prior work in the areas of activity recognition and energy expenditure estimation from wearable sensors such as accelerometers and heart rate monitors, highlighting the advantages and disadvantages of each approach and challenges that remain. Finally, the section states the assumptions in which the work presented in this thesis relies, and emphasizes the user interface challenges associated with creating mobile phone physical activity and energy expenditure interventions.

# **3.1 Assumptions**

Mobile phones are carried with people nearly anywhere they go and their popularity is increasing among all socio-economic groups [21]. As a result, this work attempts to take advantage of this trend and develop some of the required technology to achieve mobile phone interventions that use physical activity type and intensity detection in the near future. One of the assumptions of this work is that the majority of mobile phones by year 2015 will have a CPU capable of running at a speed of least at 1GHz. This processing power could allow running activity recognition and energy expenditure algorithms like the ones presented in this work in a manner that is not too perceptible to end users (e.g. no apparent delays observed). It is also assumed that the accelerometers required by the system presented in this work to sense human motion could be embedded on devices already worn at convenient locations such as wrist watches, mobile phones, belt clips, and shoe pods. This would make the accelerometers easy and comfortable to wear continuously. Another assumption is that by year 2015, power efficient and standard wireless communication protocols (e.g. similar to WiBree [77] or ZigBee [78]) will exist that will allow the wireless wearable accelerometers to easily communicate with mobile phones.

# **3.2 Challenges for Each System Component**

This subsection describes the research challenges and open questions in the areas of activity recognition and energy expenditure estimation. The subsection also discusses some of the most recent work in these areas from a hardware and algorithmic perspective.

### 3.2.1 Off-the-Shelf Devices to Recognize Physical Activity

One popular approach to the assessment of physical activity in free-living populations is ambulatory monitoring. Ambulatory monitoring can be classified as (1) the use of various types of motion sensors (e.g. pedometers and accelerometers [79, 80]), (2) the use physiological sensors such as heart rate (HR) [81, 82] and body temperature, and (3) the use of combinations of these (e.g. [48, 83, 84]).

Pedometers [85, 86] work by measuring the vertical motion generated at the hip by locomotion using a spring-suspended lever arm switch. They primarily assess locomotion or ambulatory activities by counting steps. However, pedometers only provide information about the number of steps performed and no information about upper body or non-ambulatory lower body activity.

Accelerometers assess physical activity by measuring the intensity of body segment, limb, or trunk acceleration that is proportional to muscular force. The intensity of motion measured is usually presented in the form of counts over a specific time period (or epoch) usually of 60 seconds that represent an estimate of the overall physical activity level. Unfortunately most commercially available activity monitors (accelerometers) either measure only rough physical activity levels (e.g. *light, moderate* and *vigorous*) or a small set activities (e.g. postures and ambulation). For example, the Actigraph activity monitor [32] is a uniaxial accelerometer normally mounted at the hip that provides activity counts at pre-defined time intervals (e.g. 60s) and that can be used to detect time spent in different activity levels (e.g. light, moderate, vigorous) using a variety of offline algorithms [87, 88]. Since this activity monitor is normally placed at the hip, it is best suited to measure physical activity involving overall body motion, such as ambulatory activities. A recently introduced algorithm by Crouter et al. [34] uses data from this device to recognize sedentary, ambulatory (e.g. walking and running) and lifestyle activities from raw activity counts and the coefficient of variation computed over 10s windows. Nevertheless, the algorithm runs offline and has difficulties detecting nonambulatory lower body activity such as *cycling* mostly due to the difficulty of measuring non-ambulatory lower body activity using a single accelerometer at the hip as discussed by the authors.

Hearth rate (HR) monitors (e.g. [89]) are physiological sensors that can indirectly measure physical activity because heart rate (HR) is proportional to the intensity of movement and oxygen supplied to skeletal muscles [90]. Most heart rate monitors come in chest strap form factor and work by measuring the time intervals between peaks of the ECG signal (heart beats). Hear rate information can be utilized to measure time spent in different intensity levels (e.g. *light, moderate, vigorous*) using mostly subject specific regression equations (e.g. [91, 92]). Although heart rate monitors are adequate for short duration exercise sessions, their use and acceptance for prolonged periods of time is presently questionable because they are uncomfortable and their electrodes can cause skin irritation and dermatitis. Recently, Adidas, in collaboration with Polar, introduced a new chest-based heart rate monitor in a shirt form factor that may be more comfortable to wear than other existing chest strap monitors [93]. However, the technology is relatively expensive and is mainly targeted for people already physically fit such as runners and athletes.

One of the few commercially available device capable of recognizing activities such *ambulation, stationary biking, resting,* and *weight lifting* from several physiological sensors is the bodybugg armband [42, 94]. This device is worn at the dominant upper arm (biceps area) and uses a biaxial accelerometer and four additional sensors (skin temperature, galvanic skin response, heat flux, and ambient temperature) to recognize activities and estimate energy expenditure. To the best of the author's knowledge, only one published paper exists that evaluates the performance of the bodybugg at recognizing activities ([94]). In this technical report, the authors report recognition accuracies of

99.8% for ambulatory activities, 99.8% for stationary biking, 99.3% for resting, and 97.6% for bench weight lifting over 350 hours of data collected from several studies (number of participants not specified). The authors also point that the latest version of the proprietary algorithm (4.2) achieves accuracies of 90% for lying down and sleeping, 99.8% for road biking, and a total accuracy of 96.9%. The algorithms to detect activity are proprietary, and the web-based software available to end-users of the bodybugg system [95] only report contextual information for the following activities: *Lying down*, *sleeping*, and four MET intensity levels (*sedentary*, *moderate*, *vigorous*, and *very vigorous*). Finally, recent studies evaluating the performance of the device in estimating energy expenditure have found that the device significantly underestimate energy expenditure for lower body activities [96]. This is not surprising because the bodybugg is worn at the upper arm and consequently may not fully measure non-ambulatory lower body activity. This same limitation can be expected to impact any algorithm that attempts to use bodybugg data to detect specific lower body activities.

The Intelligent Device for Energy Expenditure (IDEEA) monitor [41] is another offthe-shelf device that, according to its creator, can detect postures (e.g. *lie, recline, sit, stand*, and *lean*), ambulatory activities (e.g. *walk, run, ascending/descending stairs*) and transitions between some activities [97]. Unfortunately, to the best of the author's knowledge, there is no independent validation of the performance of this device in recognizing activities. Moreover, the recognition algorithms are proprietary and cannot be re-used or improved upon. The device uses five biaxial accelerometers placed at the limbs and hip to recognize the activities of interest. In its current implementation, the device restricts the wearer's movements because wires are run from a data collection unit placed at the hip to each of the five accelerometers.

In summary, there are few off-the-shelf devices that allow the detection of physical activities, and those that are available detect only time spent at different physical activity levels or a limited set of postures and ambulatory activities. Only the IDEAA performs specific activity detection in real-time, but a cumbersome wired system is required.

### 3.2.2 Physical Activity Recognition Algorithms

Over the past couple of years, a wide variety of algorithms using supervised classifiers have been applied to the problem of recognizing physical activities from accelerometer data. The classifiers utilized in the algorithms have included generative classifiers such as dynamic and static Bayesian networks (e.g. [38, 39, 98, 99]), a variety of discriminative classifiers such nearest neighbors, decision trees, and support vector machines (e.g. [38, 40]), ensembles of classifiers (e.g. [37, 40]), and combinations of all of the above (e.g. [37]).

Among the most popular classifiers applied to the problem are dynamic Bayesian networks (DBNs). A DBN is formed when a static Bayesian network is 'unrolled' over several times slices (time intervals) and graphical dependencies are specified among them. These dependencies among consecutive time slices allow DBNs to capture first order temporal information of the problem at hand. DBNs also allow common sense knowledge of the problem to be encoded in the internal structure of the network by manually specifying the nodes and links of the network. For example, the work by Raj et al. [98, 99], handcrafted the internal structure of a DBNs to simultaneously recognize

human activities from accelerometer data and location from GPS traces. Nevertheless, the results presented [98] indicate that the performance of the handcrafted DBN was no different from the performance of a hidden Markov Model (simpler DBN) learned automatically from the data and ignoring GPS traces for activity recognition. In addition, this approach is not scalable since it depends on experts coding common sense knowledge in the network structure that is likely to change depending on the activities to recognize. Due to the high computational complexity necessary to perform exact inference in DBNs, approximate inference algorithms such as particle filters [100] are often utilized [98, 99]. The number of particles used by this algorithm to perform inference can be adjusted depending on the processing power available on a particular device (e.g. in handheld devices); Nevertheless, its performance degrades as the number of particles is reduced.

Simpler DBNs such as hidden Markov models (HMMs) that have shown excellent performance in speech recognition applications have also been widely applied to classify activities from accelerometer data (e.g. [39, 45, 101-103]). The most common approach used is to train one HMM per activity to recognize using the Baum-Welch algorithm [104]. Once the models are trained, the classification is performed by choosing the model (HMM) that results in the highest log-likelihood over the observation sequence (sequence of feature vectors) as computed using the forward-backwards algorithm [104]. Even though this approach successfully incorporates intra-activity sequential information, its main disadvantage is its high computational requirements because one HMM per activity to recognize is required. Moreover, the forward-backwards algorithm has to be run as many times as there are activities (i.e., HMM models) to recognize so it is computationally expensive. Another disadvantage of this approach is that the number of internal hidden states needs to be specified a priori either using expert knowledge of the activities structure or learning it from training data using a cross-validation procedure.

Another approach to recognizing activities using HMMs is to use a single HMM where each internal state represents one of the activities to recognize [37]. In this approach, sequences of activities performed can be inferred online using the Viterbi algorithm [104] or particle filtering [100]. The main advantage of this approach is the incorporation of information about the transitions between activities (inter-activity sequential information) via the transition matrix of the system and the reduction of computational complexity (with respect to the use of one HMM per activity). One obvious disadvantage of this approach is that since only one hidden state is used to represent each activity, some internal temporal structure for each activity is lost (intra-activity sequential information).

Recent work has also attempted to recognize activities from accelerometer data applying a hybrid approach that combines discriminative and generative classifiers [37]. Discriminative classifiers differentiate among activities of interest by building a decision boundary or mathematical function that separates the features representing each class as much as possible with respect a given criterion. Generative classifiers on the other hand, first attempt to create a model that describes how the data for each activity is being generated before building a mathematical function to discriminate among the activities in the feature space. The combination of discriminative and generative classifiers is useful because discriminative classifiers are known to outperform generative classifiers in classification tasks [105], and generative approaches such as HMMs incorporate first order temporal information on how features or even activities transition over time. Consequently, the approach improves discrimination among the activities of interest and performs temporal smoothing over the activity classifications. The results presented in the work by Lester et al. [37] show that overall accuracy was improved 4% with respect to the performance of the discriminative classifier alone (Adaboost combined with decision stumps). Unfortunately, this slightly higher performance comes at the expense of higher computational requirements, because the computational complexity of discriminative and generative classifiers is combined in this approach. Thus, for practical applications, it is questionable if the small improvement achieved in performance justifies the extra computation incurred by combining these two algorithms.

Other less computationally expensive Bayesian networks such as the naïve Bayesian classifier [106] have also been applied to detect activities from wearable accelerometers [38, 40, 45, 107, 108]. In the naïve Bayesian classifier, the class (or activity) node is the parent to all attribute nodes (or features) and thus, its main assumption is that all the attribute variables are conditionally independent given the class. It calculates the most probable class given the data (attributes) using Bayes rule. Despite its simplicity, this classifier has obtained excellent results with respect to more complex classifiers in realistic datasets [109-111], One disadvantage of simple Bayesian networks is that they do not capture temporal information automatically unless it is encoded in the features extracted. The main advantage is that naïve Bayesian classifiers are fast to train and also perform fast classifications in comparison with DBNs.

Decision tree (DT) classifiers such as the C4.5 algorithm [112] are among the most used to recognize activities from wearable accelerometers [38, 40, 45, 107, 113, 114]. This is because of the following reasons: (1) They learn classification rules that are believed to be easier to interpret than the ones learned by other methods such as neural networks (although for real-world problems, these rules can be quite complex and not trivial to interpret) (2) they incorporate information gain feature selection during the learning process that identifies the most discriminatory features to use, and (3) they perform fast classifications making them suitable for real-time implementation. A disadvantage of decision trees is that they tend to overfit the data if they are trained on small datasets and may not combine probabilistic evidence as well as other methods. Furthermore, decision trees are static classifiers that do not incorporate any temporal transition information of the modeled activities unless it is encoded in the features used.

When the activity recognition results presented in prior work are analyzed, it is found that their evaluation suffers from some limitations. For example, some pieces of work only use total classification accuracy as the performance measure and completely ignore the number of false positives generated by the algorithm as well as other important performance measures such as those discussed in Section 5.1. For instance, the work in [115] is among the few ones to quantify the number of false positives as number of insertions and substitutions over the recognized sequences of activities. Other prior work reports results on small datasets collected at laboratory settings from as few as five participants (e.g. [37, 40, 101, 103, 116]) or researchers themselves (e.g. [101, 103]), possibly introducing an unintentional bias towards simplification of how activities are performed. Most prior work only performs offline evaluation of the algorithms; exceptions are [41, 43, 45, 46, 59, 76, 117, 118]. There are also some important questions that have only been partially addressed in prior work. For example, what is a good compromise between classifier and feature set complexity to allow an acceptable level of

real-time performance over a given set of activities? How the sliding window length impacts performance for different types of activities? What is the minimum set of sensors to use and where should they be worn to recognize a given set of activities? How sensor placement impacts performance for different activities? The answer to these questions strongly depends on the activities to recognize, so it is necessary to explore these questions over a large set of activities. Finally, no prior work to the best of the author's knowledge addressed the problem of recognizing the intensity of physical activity (e.g. *running* at 5mph vs. 6mph). The following paragraphs describe some few pieces of work that have explored some of the important questions raised in this paragraph.

The work by Bao and Intille [38] has been one of the few pieces of work to evaluate the activity recognition algorithms proposed on data collected for a relatively large set of 20 activities performed by 20 participants in non-laboratory settings during seminaturalistic conditions. For each participant, an average of 90 minutes of data was collected. The authors report accuracies ranging from 85-95% for *ambulatory* activities, *postures*, and more complex activities such as *brushing teeth* and *folding laundry* using the C4.5 decision tree classifier. The same work also determined that user-independent recognition of twenty activities is possible from five biaxial accelerometers with an overall accuracy of 73%. The authors also perform experiments to determine the minimum number of accelerometers required to recognize the twenty activities of interest and find that it is possible to achieve an accuracy of ~81% just using two biaxial accelerometers located at the thigh and dominant wrist.

Identifying a minimal or reasonable number of sensors to recognize activities and their placement on the human body has been explored in at least three pieces of prior work [39, 45, 116]. The work by Kern et al. [116] analyzed 18.7 minutes of data collected for eight activities (sit, stand, walk, upstairs, downstairs, handshake, writing and typing) by one of the authors and found that sensors located at the dominant part of the body (e.g. right side for right handed people) are better at discriminating most of the eight activities. Similarly, the combination of a sensor at the hip and another at the ankle achieves the best performance in recognizing lower body activities. The work also found that during subject dependent training, single sensors at the lower body (e.g. hip, knee, and ankle) achieved similar performance so that they could potentially be used interchangeably to recognize lower body activities. When the performance of single sensors was analyzed in recognizing upper body motion, it was found that the single sensor with best performance was the one placed at the shoulder, and the best sensor combination was shoulder and wrist. The work by Blum [45] performed a similar analysis and found that just two accelerometers at the hip and dominant wrist are sufficient to recognize eight postures with an accuracy of 85% from a dataset consisting of 24 hours of data collected by the author. The work by Olguin Olguin [39] also performed an analysis of the number of sensors to use and their location using HMMs to recognize the following eight activities: Sit-down, Run, Squat, Walk, Stand, Crawl, Lay down, and Hand movements. Three accelerometers located at the hip, chest, and dominant wrist was used to recognize the activities. As expected the sensor combination with highest overall accuracy of 92% was the wrist+hip+chest sensor combination on data collected from three participants performing each activity three times. The results also indicate that the single accelerometer with best overall accuracy (62%) is the one at the chest. Adding another accelerometer either at the hip or dominant wrist improved overall accuracy to ~81%.

The work by Ravi et al. [40] explored the performance of different classifiers in recognizing activities such as decision tables, decision trees, support vector machines, nearest-neighbor, and naïve Bayes classifiers individually and in different meta-classifier configurations such as boosting, bagging, stacking, and plurality voting. The results presented were evaluated over data collected from two participants performing eight activities (ambulatory activities, sit-ups, vacuuming and brushing teeth) several times while wearing a single triaxial accelerometer at the pelvic region. In general, it was found that meta-classifiers outperformed base-level or single classifiers as expected. It was also found that boosted support vector machines (SVMs) achieve the highest overall accuracy of 73% followed by the naïve Bayesian classifier with an overall accuracy of 64% during subject independent evaluation. For subject dependent evaluation, the highest overall accuracy was obtained for plurality voting (99.6%) followed by Boosted SVMs (99.4%). The naïve Bayesian classifier also achieved a good overall accuracy of 98.9% during subject dependent training. By analyzing the difference between the performance of the naïve Bayes classifier and the Boosted SVMs of 9% for subject independent training and 0.5% for subject dependent training, the authors concluded that it was clear that metaclassification offers a significant improvement during subject independent evaluation.

The work by Huynh and Schiele [119] analyzed a the importance of five features and different window lengths (0.25-4s) to recognize six activities: walking, standing, jogging, skipping, hopping, and riding a bus. The data used to perform the experiments consisted on 200 minutes of data collected by two participants. The main finding of the paper, as expected intuitively, is that there is no single window length or feature that maximizes performance for all activities. Instead, the choice of window length and feature set to use depends on the activities to recognize. It was also found that one of the best features to use is the FFT coefficients, but that the coefficients to use depend on the activities being recognized. This is expected, since different activities are performed at different speeds and thus, are represented by different frequencies (FFT coefficients). The findings for the window length experiment performed in this work suggest that the best overall performance for all activities was obtained with window lengths between one and two seconds. This might be explained by the fact that the fundamental period of the activities explored in this work can be captured with these window lengths. It was also found that standing requires relatively short window lengths (e.g. 0.25 - 0.5s). Also, longer window lengths are required for activities with longer fundamental durations such as *skipping* and hopping.

The work presented in this thesis improves on prior work by exploring how various algorithmic and feature computation tradeoffs impact activity type recognition in a relatively large and complex dataset incorporating examples of 52 activities collected from 20 participants at two locations (gymnasium and home).

#### 3.2.3 Existing Devices and Techniques to Measure Energy Expenditure

Real-time and reliable measurement of energy expenditure (EE) would create new health intervention opportunities, as described in Section 2.1.3. Presently, the methods that generate the most accurate measurement of energy expenditure are: (1) direct calorimetry, (2) indirect calorimetry [120], (3) doubly labeled water (DLW) [121], and (4) physical activity scoring using direct observation and the Compendium of Physical Activities [122]. Unfortunately, none of these methods can be applied to large-scale

population studies because of their high cost, intrusiveness, complexity, and specialized resources and personnel required. For example, direct calorimetry estimates energy expenditure by directly measuring the heat released by living organisms at rest or during physical activity. Heat released from the body can be measured by confining subjects to a metabolic chamber equipped with specialized sensors. It can also be measured using infrared thermo sensitive cameras, but this technique is primarily employed in measuring EE in preterm infants [123]. Although direct calorimetry produces high quality estimates of EE, it is expensive, intrusive and difficult to perform since specialized equipment and personnel are required. Indirect calorimetry, also known as respirometry, estimates energy expenditure by measuring the volume of oxygen consumption (VO<sub>2</sub>) and carbon dioxide production (VCO<sub>2</sub>) in living organisms. The relationship between energy expenditure and oxygen uptake is linear because every cell in the human body requires oxygen to generate the energy required for cellular work (ATP). Indirect calorimetry is measured using either (1) closed respirometry or (2) open respirometry. In closed respirometry,  $O_2$  uptake and  $CO_2$  production are measured by placing the subject in a small sealed room (e.g. [124]). The main disadvantage of this method is its high cost and the confinement of the subject to a laboratory room that restricts and modifies his/her behavior. In open respirometry,  $O_2$  uptake and  $CO_2$  production are analyzed (after being collected using face masks or plastic canopies) using metabolic carts [125, 126] or portable indirect calorimeters (e.g. [125, 127]). The main disadvantage of open respirometry is that the equipment is expensive, uncomfortable, and obtrusive even when portable indirect calorimeters (e.g. [127, 128]) are used. Despite its disadvantages, indirect calorimetry is considered a reference method to measure EE and is one of the most widely used by the medical community.

The use of the doubly labeled water technique (DLW) is considered the 'gold-standard' in measuring total energy expenditure (TEE) in free living conditions because it does not interfere with individual's daily activities. This method consists of the oral administration of two stable isotopes (oxygen 18 (<sup>18</sup>O) and deuterium (<sup>2</sup>H)) to participants and the monitoring of their elimination rates from the body via urine samples. In this way, total carbon dioxide production can be determined and used to compute an estimate of total energy expenditure. Although the main advantage of this method is its non-intrusiveness, its main disadvantage is that it only provides an overall estimate of the total energy expenditure at the end of the measurement period (e.g. usually 7-18 days [47]) and no information about physical activity type, intensity, or duration. One important disadvantage of all the previously discussed techniques to estimate energy expenditure is that they provide no contextual information about the physical activities being performed. Knowledge of the physical activities performed by individuals and their associated intensities is important because it can be used to (1) improve the energy expenditure estimates, (2) to motivate behavior change, and (3) to better understand relationships between physical activity and behavior.

A method to estimate energy expenditure that provides rich contextual information about the activities performed is physical activity scoring using the Compendium of Physical Activities [122]. This technique involves recording the duration, intensity and frequency of physical activities and scoring these parameters by consulting the Compendium of Physical Activities [122]. The Compendium of Physical Activities contains the energy cost incurred in performing various activities and intensities in

METs. METs are defined as the number of calories expended by an individual while performing an activity in multiples of his/her resting metabolic rate (RMR). Thus, METs can be converted to calories by measuring or estimating and individual's RMR. Some limitations of this method in estimating energy expenditure are that (1) the energy expenditure values listed on the Compendium of Physical Activities only represent the average energy cost for the physical activities and the subject population explored in the Compendium of Physical Activities study, and (2) that other conditions on how the activity is performed such as terrain inclination and intensity levels not included in the Compendium of Physical Activities cannot be evaluated [47]. Furthermore, the accuracy of the energy estimate produced by this method strongly depends on the quality of the RMR estimate and the physical activity information collected. Despite these limitations, this method has been one of the most widely used in published medical research. The intensity, duration and frequency of physical activities can be best compiled through direct observation. Direct observation [129], considered the "gold standard" for assessment in medical and psychological research studying behavior in natural settings, does not suffer from selective recall if performed by trained observers. Direct observation techniques include a person following the subject and the use of continuous video, audio or sensor data recordings to acquire information related to the behaviors of interest in natural environments. Even though direct field observation can provide valuable qualitative and quantitative measures, it is costly, time-consuming, and disruptive. This technique also raises privacy concerns since researchers need to invade private settings such as the home.

Other less accurate methods that researchers have used to measure physical activity and energy expenditure in large subject population studies are: (1) self-reported information such as recall surveys, end of study interviews, and diaries, and (2) ambulatory monitoring using motion and/or physiological sensors. Recall surveys such as questionnaires (e.g. [130-133]) are widely used for assessment of behavior in naturalistic settings. However, this method is known to suffer from recall and selective reporting biases - users can often not remember what they did and/or do not report what they actually did. Furthermore, most questionnaires used in epidemiological studies don't provide detailed information on the type, intensity or duration of activities. End of study interviews consist of interviewing the participants at the conclusion of a study. Interviews have shown to be particularly effective for critiquing ideas or gathering information about the participants' tasks and activities if performed properly. Often however, participants know more than they say in a single or even several interviews [134], and will tend to have difficulty understanding and recalling how context impacts their behavior (i.e. exhibiting selective recall and selective reporting biases [135]). In fact, several studies have shown that is harder for participants to recall *moderate* or *light* activities than vigorous bouts of activities [136, 137]. While using diaries, participants write down what they do during the day either as they do it or at regular, finely-spaced intervals [138]. This method completely relies on the ability of participants to recall (from short and long term memory) the information relevant to the behaviors of interest. Diaries provide better data than questionnaires, recall surveys or end of study interviews but they are burdensome for the user and the quality of the data depends entirely on the participant's compliance. Recent improvements over paper diaries include web-based diaries, PDAbased diaries [139], and electronic experience sampling [140, 141]. These improved diaries are less susceptible to subject recall errors than other self-report feedback elicitation methods [135, 142]. Nonetheless, these also suffer from the disadvantages of self-report methods such as subject inconvenience, reporting bias, poor memory, and poor compliance [143]. The physical activity information collected using self-report can be converted to energy expenditure estimates by scoring the information using the Compendium of Physical Activities. Nevertheless, these EE estimates are usually coarse and less accurate than EE estimates computed from detailed physical activity information collected via direct observation.

One of the most popular methods to estimate energy expenditure in free-living populations is ambulatory monitoring using pedometers, accelerometers, and physiological sensors such as heart rate, body temperature, body flux, and galvanic skin response. Ambulatory monitoring offers some advantages over self-reported data such as reduced report bias and subject burden (when compared with diaries).

The number of steps measured using pedometers can be combined with demographic information (weight, height, age and gender) to produce rough estimates of energy expenditure due to ambulatory activity. Ambulatory activities such as walking account for a major proportion of daily energy expenditure [144], so pedometers are considered a good starting point for computing a daily energy expenditure estimate. One disadvantage of pedometers is that they are incapable of measuring energy expenditure due to stress effort in activities such as lifting heavy objects from the floor.

Similarly, the activity counts or acceleration values collected using accelerometers can be combined with demographic information and regression techniques [34, 47, 145-152] or physical models of the human body [153, 154] to produce energy expenditure estimates. The complexity of accelerometry to measure energy expenditure ranges from the use of single uniaxial accelerometers [35, 36, 155, 156], biaxial and triaxial accelerometers ([151, 157-161]), to the use of multiple biaxial or triaxial accelerometers [41, 49, 146, 148, 162, 163]. The previously discussed Actigraph activity monitor [32] is one of the most popular uniaxial accelerometers used to estimate energy expenditure in the medical community [34, 164, 165]. The only accelerometer-based device available off-the-shelf that measures energy expenditure using multiple accelerometers (five biaxial) is also the previously discussed IDEEA monitor [41]. As explained before, the system requires wires that restrict the wearer's physical movements and its current cost (~\$4000) is prohibitive for large scale deployments. One disadvantage of accelerometers is that they cannot distinguish between the energy costs of performing activities involving different resistance levels or work loads such as carrying light vs. a heavy loads and walking downhill vs. uphill. Currently most accelerometers available off-the-shelf allow the collection of data in non-volatile memories from 1 to 42 days depending on the sampling rates used and are relatively accessible ranging from \$70 to \$400 dollars. However, no off-the-shelf accelerometers make the data available wirelessly for real-time processing. This is unfortunate because real-time data could allow the development of powerful interventions to increase energy expenditure in free-living. Table 3-1 presents a summary of the most popular ambulation monitoring devices used to estimate energy expenditure.

Heart rate information (e.g. beats-per-minute) can be converted to energy expenditure by generating subject dependent or independent regression equations between heart rate and volume of oxygen uptake  $(VO_2)$  as collected from laboratory experiments

Device	Sensors Used	Body locations	Storage Capacity (days)	Cost (\$)
Pedometer Omron HJ-112 [85]	Spring-suspended lever arm switch	Waist	7	25
Pedometer Yamax Digiwalk CW-200 [86]	Spring-suspended lever arm switch	Waist, neck, ankle	0	20
Actigraph GT1M (previously CSA) [166]	Uniaxial accelerometer	Wrist, ankle or waist	42*	400-1500
Tri-trac (CT1 and RT3) [167]	Triaxial accelerometer	Waist	21	100-500
Actitrac, Digitrac and Biotrainer [157]	Actitrac and Biotrainer (biaxial), Digitrac (triaxial)	Wrist, ankle or waist	5-62	70
Actiwatch [168] (also ActiCal)	Uniaxial accelerometer and Light	Wrist, ankle or waist	11-44	1075
X-Max CalTrac [35]	Biaxial accelerometer	Waist	0	70-90
Tracmor [169]	Triaxial accelerometer	Wrist, ankle or waist	NA	NA
Actillume Actigraph [155]	Uniaxial accelerometer And light	Wrist, ankle or waist	NA	NA
Bodybugg armband [158]	Biaxial accelerometer + other sensors	Dominant upper arm	14	400-600
IDEEA [41]	Five biaxial accelerometers	Body segment, limb or hip.	2.5	4000

Table	3-1:	Most	popular	ambulatory	monitoring	technologies	available	off-the-shelf	to	estimate
energy expenditure and measure physical activity. NA stands for non applicable.										

[32, 47, 170, 171]. The main challenges in using heart rate information to estimate energy expenditure are intra-individual and inter-individual variations in heart rate. For example, intra-individual variations in heart rate include changes in heart rate due to emotional states, nicotine, digestion, altitude and temperature [172]. Inter-individual variations in heart rate are mainly due to differences among individuals in fitness level, age, sex, and gender [173]. Heart rate also exhibits a delayed response to physical activity and remains altered once a physically demanding activity has finished. Finally heart monitors do not provide information about the type or frequency of physical activities (e.g. running at 6mph for 5min).

There have been several attempts to estimate energy expenditure by combining several sensor modalities [83, 96, 174, 175]. For instance, one of the few device of this kind available off-the-shelf to estimate energy expenditure during free-living is the previously discussed bodybugg armband [158]. The bodybugg combines information from its five sensors with proprietary algorithms to estimate energy expenditure. Even though energy expenditure comparisons between the bodybugg and indirect calorimetry have shown low error rates (10-15%), two recent validation studies suggest that the device's energy expenditure estimate may not be consistent across different types of activities [176, 177]. Furthermore, the energy expenditure algorithm used by the bodybugg Armband has been found to significantly underestimate lower body activity (e.g. cycle ergometry) and overestimate upper body activity (e.g. arm ergometry) [96]. This is because its location at the upper arm prevents it from detecting lower body motion generated by non-ambulatory activities. In late 2007, a new digital watch was introduced that allows the bodybugg system to display real-time information about energy interventions, the bodybugg system presently does not make the real-time data available to other applications nor does it use the data to enable physical activity interventions.

Method	Advantages	Disadvantages
Direct calorimetry	-Very accurate	-Expensive
		-Intrusive
		-Restricts daily activity
		-Requires specialized equipment and expertise.
Indirect Calorimetry	-Very accurate	-Expensive
		-Cumbersome to wear (portable versions)
		-Restrictive (room calorimetry)
		Requires specialized equipment and expertise.
Doubly labeled water	-Very accurate	-Expensive (\$1500 per person)
	-Unobtrusive since does not interfere with	-Only measures average TEE over periods of 1-3
	subject's daily activities.	weeks.
		-Requires specialized equipment and expertise.
Physical Activity	-Provides qualitative and quantitative	-Expensive (direct observation)
Scoring Using the	information.	-Disruptive (direct observation)
Compendium of	-Provides contextually rich information	-Time consuming (direct observation)
Physical Activities	(activity data)	- Suffer from selective recall and reporting bias
	-Relatively low cost if self-reported	(self report)
	information is used	- Burdensome for subjects (if diaries or
	-Relatively easy to administer if self-report	experience sampling is used)
	is used	-Possible compliance problems (diaries and
		experience sampling)
Pedometry	-Small and easy to wear technology	-Crude EE estimates based on ambulatory
	-Low cost and off the shelf availability.	activity.
		-Does not measure upper body activity or
		muscular contraction.
Accelerometry	-Small and easy to wear technology	-Only measures motion of body segment they are
	-Moderate cost and off the shelf	attached to.
	availability.	-Do not measure EE related to muscle
	-Can provide contextual information	contraction, work load, or resistance effort
		-Conversion of acceleration counts to EE is
		challenging.
Heart rate monitors	-Moderate cost	-HR varies within individuals due to factors other
	- Physiological parameter that provides	than physical activity such as emotional states,
	intensity, frequency and duration	nicotine, temperature, etc.
	information about physical activity.	-HR varies across individuals due to differences
	-Low burden for short data collections.	in fitness level, age, gender, etc.
		-Compliance issues in longitudinal studies.
		-Not clear how to convert HR to EE in a subject
		independent manner

 Table 3-2: Comparison of advantages and disadvantages of existing methods used to estimate energy expenditure.

In summary, the energy expenditure estimation methods using various forms of ambulatory monitoring previously reviewed all have their strengths and weaknesses, and while they are adequate for some purposes, none are satisfactory for accurately determining energy expenditure over weeks or months of a free-living population. As a result, this work explores how energy expenditure estimation can be improved to be more accurate and amenable for longitudinal use in free-living. This is achieved by investigating the following questions: (1) Which features computed over the accelerometer data provide better estimates? (2) What sliding window lengths (epochs) are better for estimating energy expenditure? (3) Do multiple accelerometers at different Body segments improve the estimates? (4) Do non-linear regression techniques improve the estimates significantly? (5) Do activity dependent regression models improve energy expenditure estimation? Finally this work also explores if the combination of acceleration and heart rate data improves the energy expenditure estimates.

#### **3.2.4 Energy Expenditure Estimation Algorithms**

There exists a large body of algorithmic approaches to estimate energy expenditure from accelerometer data. These approaches can be broadly classified in (1) those that use physical models of the human body, and (2) those that use regression algorithms. The algorithms that use physical models of the human body to estimate energy expenditure (e.g. [153, 154]) usually attempt to first estimate velocity or position information from accelerometer data by integrating the accelerometer signals. Unfortunately, this is a difficult task since errors in the acceleration signal accumulate rapidly over time due to the integration (summation) of values containing small amounts of error. Once velocity and/or position are known, these algorithms use kinetic motion and/or segmental body mass to estimate energy expenditure. Some algorithms such as [154] attempt to circumvent the problem of integrating the accelerometer signal to estimate velocity or position by learning the coefficients of the kinetic equations describing the human body motion directly from training data (acceleration and energy expenditure data). Although the use of physical models of the human body in the estimation of energy expenditure makes perfect sense, no prior work to the best of the author's knowledge has shown that this approach has considerable advantages over the use of regression algorithms.

Regression algorithms, on the contrary, estimate energy expenditure by directly mapping accelerometer data to energy expenditure using linear and/or non-linear models. In their simplest form, these algorithms attempt to estimate energy expenditure from a single accelerometer (usually an Actigraph [32]) placed at the hip using simple linear regression (e.g. [145-148]). Some prior work also uses multiple linear regression models to improve the estimates [149]. Estimating energy expenditure from an accelerometer at the hip using linear regression may not fully capture the complex relationship between acceleration and energy expenditure [34, 178]. As a result, some work has explored the idea of using non-linear regression models to estimate energy expenditure from a single accelerometer at the hip [34, 150, 151, 179]. Most prior work estimates energy expenditure over sliding windows (epochs) of one minute in length and uses only the number of acceleration counts (per minute) to estimate energy expenditure. However, some recent work suggests that the utilization of shorter window lengths might improve the estimates of energy expenditure [34, 180]. Other recent work [34, 152] also suggests that computing more complex features over the accelerometer signal such as the coefficient of variation, the inter-quartile interval, the power spectral density over particular frequencies, kurtosis, and skew can improve the estimates of energy expenditure by capturing motion information that would otherwise be lost if simple accelerometer counts are used to estimate energy expenditure.

Presently, the two state-of-the-art algorithms to estimate energy expenditure from a single accelerometer mounted at the hip are the work by Crouter et al. and Rothney [34, 152, 181]. The main idea behind the algorithm presented by Crouter et al. [34] is to classify activities into three categories before estimating energy expenditure: (1) *sedentary* activities, (2) ambulatory activities such as *walking* and *running*, and (3) *lifestyle* activities. Once the activities are recognized, different regression models are applied for each activity type to estimate energy expenditure. The algorithm recognizes sedentary activities by simply setting a threshold over the acceleration counts. Once they are recognized, they are assigned an energy expenditure equivalent of 1 MET. Ambulatory activities such as *walking* and *running* are differentiated from *lifestyle* 

activities by setting a threshold over the coefficient of variation (CV) as computed over 10s sliding windows. If *walking* and/or *running* are detected, a linear regression model is applied to estimate their energy expenditure; otherwise, an exponential regression model is applied to estimate the energy expenditure associated with *lifestyle* activities. The work by Rothney [152, 181] estimates energy expenditure by first computing features over the accelerometer signal such as the coefficient of variation, the inter-quartile interval, the power spectral density over particular frequencies, kurtosis, and skew and then uses these features to train an artificial neural network (non-linear regression model) to estimate energy expenditure. The artificial neural network is trained on nearly 24 hours of data collected from 102 participants which makes this work one of the most extensive with respect to amount of data used. This work also incorporates demographic data as features into the model (neural network) to compensate for inter-individual variations in energy expenditure. It is important to consider inter-individual variations because two individuals performing the same activity might generate similar acceleration signals but different energy expenditure signals due to differences in age, gender, height, weight and ethnicity. One clear disadvantage of these two methods to estimate energy expenditure is that they have difficulties capturing upper body and non-ambulatory lower body motion because of the use of a single accelerometer at the hip. For example, the work by Crouter et al. [34] completely excluded the *cycling* activity from analysis (even when data was collected for it) because the accelerometer at the hip did not produce any readings for this activity. Furthermore, the largest errors obtained in this work occurred for activities involving upper body motion such as *basketball*, *racquetball*, *vacuuming*, and *mowing* the lawn. The work Rothney [152, 181] did not present performance results per activity, so the performance over the *biking* activity cannot be determined. Other prior work also indicates that accelerometers mounted at the hip significantly underestimate lower body activities such as cycling and sliding [33].

There has also been some prior work attempting to combine the advantages of accelerometers and heart rate monitors in estimating energy expenditure [48, 83, 84, 174, 175, 182-186]. For example, the work by Strath et al. [83, 174] combined accelerometer and heart rate data by employing two different HR-EE regression equations depending on upper body or lower body activity. Upper body activity was detected using one accelerometer at the arm and lower body activity by an accelerometer at the leg. More recent work by Brage et al. [46, 175] takes advantage of the synchronization between heart rate and accelerometer data by applying a branched regression model technique to combine heart rate and accelerometer data. The branched modeling technique weights two different regression equations differently depending on the intensity of motion experienced by an accelerometer at the hip and the intensity of physical activity detected by the heart rate monitor. One equation models accelerometer data only and another heart rate data only. The work by Strath et al. and Brage et al. showed significant improvements over the isolated use of accelerometer or heart rate data. Section 5.6.1 will later summarize some of the most recent results obtained when estimating energy expenditure from accelerometers and heart rate monitors.

#### 3.2.5 Exploiting Real-Time Physical Activity Feedback to Motivate Behavior Change

Existing technologies that provide automatic feedback related to physical fitness unobtrusively during free-living are mostly designed for those individuals who have already achieved a good level of physical fitness such as athlete runners. For example, the Nike+iPod sport kit [19] allow individuals to track the speed, distance, pace and calories burned while running by slipping a motion sensor in the Nike+iPod ready shoe and snapping a wireless receiver unit into the iPod nano MP3 player. Another recently introduced technology is the Adidas+Polar training system [20] that integrates a heart rate monitor on a shirt, a stride sensor on a shoe, and a wristwatch computer to monitor work out zones based on heart rate, speed and distance data. The utilization of these new portable technologies is a good example of real-time biofeedback as a way to motivate behavior modification since heart rate data and energy expenditure is used to maintain the work out at particular zones. Still, the challenge remains to come up with technologies that can be used by those who have difficulty maintaining healthy levels of physical activity every day.

Algorithms that automatically recognize activities and estimate energy expenditure using portable devices such as mobile phones offer the potential to provide longitudinal feedback related to physical activity and diet during free-living conditions. In its simplest form, feedback can be presented in the form of graphs showing how a person is getting meaningful physical activity. Mobile technologies that recognize activities or estimate energy expenditure could deliver "just-in-time" interventions at the point of decision [27, 28, 187]. For example, mobile devices could motivate increases in physical activity by detecting physical activities of interest (e.g. walking) and utilizing positive feedback combined with persuasive techniques [188] to encourage increases in activity levels (e.g. walking at 4mph vs. walking at 3mph.). Recent applications have started to explore this idea. The work by Bickmore et al. [189] developed an animated PDA-based advisor to motivate increases in physical activity over the course of a day. The authors are planning to extend this work by adding an accelerometer to provide positive feedback in real-time after a bout of *walking* or to encourage compliance when users have previously committed to exercise at a given date/time. One recently developed application exploring real-time feedback on mobile phones is UbiFit Garden [190]. In this work, a hip-worn accelerometer-based sensing device [44] is combined with a mobile phone to provide real-time feedback related physical activity performed over the course of a day. The feedback is graphically represented as a garden with flowers representing levels of physical activity and butterflies representing physical activity goals.

Another area in which real-time activity recognition and energy expenditure algorithms can be applied to provide valuable feedback to users in non-exercise activity thermogenesis (NEAT) [51, 191]. This relatively new area of research has attracted much interest because it suggests that small increments in non-exercise physical activity such as *standing*, *brisk walking*, *using stairs* and *fidgeting* can accumulate over the course of a day to boost overall energy expenditure. In fact, a recent study [192] demonstrated that the caloric difference between obese sedentary individuals and obese sedentary individuals increasing NEAT activity levels can be as high as 300kcal per day. Other studies also suggest that replacing TV watching by moderate non-exercise physical activity recognition algorithms

combined with energy expenditure algorithms might be used to provide detailed feedback about the number of calories burned per activity performed so that individuals can better understand how they expend their daily energy, and perhaps, plan increases in nonexercise activity that fit their daily routine.

Finally, another interesting application of real-time feedback is to combine it with fun games to make it more amenable to children and adults [6]. This approach attempts to replace the sedentary behaviors usually involved in playing video games with workout routines or small increases in overall body motion. For instance, Dance Dance revolution is a video game that takes advantage of these concepts to motivate physical activity while playing a videogame. As arrows move across the screen representing dance movements, the player steps on the corresponding arrows printed on a 3-foot-square metal mat that replaces the conventional game pad. The game has been so successful that has sold about 2.5 million copies, and by 2009 more than 750 public schools will begin using the video game to motivate physical activity among their students [195]. The new Nintendo Wii console [67] can be used in a similar way requiring users to perform arm-gestures to control the gaming interaction. Real-time sensing may create new opportunities for games that encourage NEAT during everyday activities such as television watching [196].

#### 3.2.6 Ensuring Ease-Of-Use

Ensuring ease-of-use involves addressing factors such as comfort of the sensors, the number of sensors used and their location in the human body, training time requirements of the algorithm, capability of real-time recognition of activities, and real-time recognition delay. Wearability of sensors has received little attention in prior work because sensors have been mainly used during pilot research studies of relatively short duration. For example, most existing wireless accelerometers capable of broadcasting data in real-time (e.g. [44, 197, 198]) have cumbersome form factors and casings ill-suited for longitudinal use. Some existing off-the-shelf devices (e.g. [32, 36, 157]) have more convenient form factors, but they are incapable of making the data available in real-time. The system presented in this work uses wireless accelerometers (MITes [199]) that have smaller form factors than most existing accelerometers and suitable casing and battery life to allow continuous data collection over the course of a day during research studies.

Previous work making recommendations for the minimum number of sensors to use and their locations on the body has been performed the analysis over a limited set of activities and little data collected from one to three subjects [39, 45, 116]. The exception is the work in [38], which analyzes results over different combinations of sensors for 20 activities on data collected from 20 participants. This work continues to explore that question using a significantly larger dataset of 52 activities on data collected from 20 participants at two locations: a gymnasium and a home.

To the best of the author's knowledge, no prior work has explored the training data requirements of activity recognition algorithms because it has been assumed that subject independent recognition of activities will be possible one day. On the contrary, this work explores if it is possible to create subject-dependent activity recognition algorithms that require small amounts of training data and if the data required has the potential to be provided interactively. As a result, this work performs experiments to determine the minimum amount of training data required to recognize an activities reliably during subject dependent training.

Few accelerometer-based activity recognition algorithms presented in prior work have been tested in real-time. For example, the e-watch [43, 117] is one of the few research technologies with a convenient form factor capable of recognizing activities in real time. However, given its location and form factor (wristwatch), it can mainly recognize upper body activities and ambulatory activities involving overall body motion. Other research technologies capable of real-time performance use a single accelerometer at the hip to perform the inference (e.g. [59, 118]). As a result, they are constrained to recognize activities involving overall posture, and motion such as walking and running. The few real-time activity recognition systems that use several accelerometers to perform the recognition (e.g. [41, 45, 76]) recognize activities that have been fixed a priory, and thus the systems do not allow users to specify and train new activities to recognize. In fact, to the best of the author's knowledge, no prior work has presented a real-time system to recognize activities that allows users to specify and train the activities that they want to be recognized. The closest system is presented by Feldman et al. [46], but the activities targeted by the system are mainly upper body activities (hand gestures). Therefore, this work develops an activity recognition and energy expenditure estimation system that uses one or several accelerometers and evaluates the performance of the activity type recognition algorithm in real-time during a short study where participants train and test the recognition algorithms to recognize different activities of their interest themselves.

Finally, real-time classification delay is important to consider because the longer the user has to wait for a recognition result, the less likely it is that users will have the patience to train and test the performance of the algorithms in real-time. Similarly, longer recognition delays degrade the performance of other applications (e.g. interventions) that may use the recognition results and require high-precision prompting. This work includes an analysis of how the sliding window length impacts the recognition performance; a window length is selected that minimizes classification delay while maximizing performance.

## 3.3 Summary of Technological Limitations to Enable Real-Time PA Type and Intensity Detection

In summary, the combination of easy-to-use and easy-to-wear sensors and activity recognition and energy expenditure algorithms that can run in existing mobile phones would allow the development of interventions that (1) have the potential to scale to large populations due to the utilization of mobile phones and low-cost sensors that can be mass produced easily. (2) that are easy to use and impose minimum burden on the user, (3) that provide end-user value on a daily basis by presenting real-time information at the point of decision (related to physical activity and energy expenditure) that is usually unknown to individuals, so that people continue to use them for months or years with little effort, (4) that could potentially incorporate persuasive techniques to engage the user in his/her interactions with the system, and (5) that are fun, easy-to-use and engaging.

The types of interventions previously mentioned have not been developed yet due to hardware and algorithmic limitations. For example, some of the hardware limitations that have prevented these interventions are: (1) Unavailability of real-time data. Most existing accelerometers available to measure physical activity store the data in onboard memories and do not make the data available in real-time. The only few accelerometers that make the data available in real-time are relatively expensive technologies in a prototype stage such as [197, 198]. Making the data available in real-time is important because other applications can use it to trigger just-in-time interventions. (2) Unavailability of technologies to receive data from multiple sensors simultaneously. Most existing technologies that allow the collection of data from multiple accelerometers (e.g. [32]) rely in offline analysis of the data. This is because data is stored locally in the devices and thus, data has to be first downloaded from the multiple devices and synchronized offline. There are also few technologies that allow the simultaneous reception of data from several sensor types (e.g. accelerometers and heart rate monitors). This prevents real-time applications that use multiple data types in real-time. Naturally, if data cannot be received from multiple accelerometers and/or sensor types in real-time, applications that use this data cannot be implemented. (3) Inconvenient form factors. Existing accelerometers have inconvenient form factors to be used inconspicuously for prolonged periods of time.

Some of the algorithmic or software limitations that have prevented real-time interventions are: (1) Coarse estimates. Most existing technologies to measure physical activity provide too coarse of an estimate to allow useful applications. For example, pedometers [85] only provide rough estimations of ambulatory energy expenditure and single accelerometers at the hip have difficulties capturing upper body and nonambulatory lower body motion. As a result, they usually underestimate energy expenditure during lifestyle activities [200]. (2) Limited or no contextual information. Technologies available to measure physical activity provide limited or not contextual information about the type, intensity, and duration of the physical activities being performed (e.g. Actigraph [32], and the bodybugg [42]). Contextual information about physical activities is important for educating the user about physical activity patterns, for motivating a positive behavior change, for estimation of energy expenditure, and for the development of powerful physical activity interventions that are triggered based on the activities being performed. (3) Lack of efficient algorithms that run in real-time on lowprocessing power devices. Most algorithms available for recognizing activities and estimating energy expenditure have been implemented to run offline and have not been tested for real-time performance. As a result, data has to be downloaded from the data collection devices before feedback related to physical activity and energy expenditure can be obtained. This is inconvenient for both end-users and researchers during large-scale longitudinal studies.

There are also some open questions that have prevented the development of real-time interventions related to physical activity. For example, the number of sensors and their placement in the human body required to recognize activities and to estimate energy expenditure is still unknown. Obviously, the answer to this question depends on the target activities, but there is no prior work that analyzes the answer to this question over a large set of activities. It is also not clear what computationally efficient algorithms achieve the best performance at recognizing activities and estimating and estimating energy expenditure and how their performance compare to more complex state-of-the-art

algorithms. It is also unclear what set of features need to be computed to recognize activities and estimate energy expenditure efficiently and with good performance from accelerometer data. Finally, the impact of utilizing different sliding window lengths in recognizing activities and estimating energy expenditure has not being explored in full detail. The work presented in this thesis explores the answer to these questions by evaluating the recognition and energy expenditure estimation for a set of 52 activities and subsets of them.

# **4** Approach and Procedure

This section presents an overview of the design of the activity recognition and energy expenditure estimation systems presented in this work. The section also describes the research approach followed to collect the necessary data to develop and evaluate these algorithms

# 4.1 Overview of Research Approach

The procedure followed in this work to develop the activity recognition and energy expenditure algorithms consisted of three steps. (1) First, activity and energy expenditure data were collected from 20 participants at a gymnasium and residential home to develop train and test the algorithms. At the gymnasium, data about exercise related physical activity were collected under relatively controlled conditions because several stationary exercise machines are used to collect the data (e.g. cycling, rowing, and treadmill machines). During the home data collection, participants performed a variety of everyday household activities under more naturalistic conditions. (2) Once the data were collected, a set of systematic experiments was performed to determine a reasonable set of activity recognition and energy expenditure algorithm parameters that enable real-time performance. Some of these parameters include the classifier (or regression algorithm), the signal processing techniques, the sliding window length, and the feature set to use. Also, experiments were performed to determine the minimum set of sensors to use, sensor placement on the body, and the impact on performance when heart rate data were incorporated. The experiments were organized so that each incrementally answers a relevant question about the algorithm parameters, starting from the most restrictive parameters (e.g. classifier or regression algorithm, feature set) and moving to the least restrictive parameters (sensor modality, window length). (3) Finally, once all the parameters were selected by running offline experiments, the final activity recognition algorithm was implemented on a laptop computer and its real-time performance was evaluated in a small feasibility demonstration.

# 4.2 Overview of System Design

The system presented in this work consists of two mayor components: (1) A real-time physical activity recognition system, and (2) a real-time energy expenditure estimation system. Both systems recognize activities and estimate energy expenditure from data provided by several wireless accelerometers placed at different body locations. In some cases a heart rate monitor is also used. The physical activity recognition system recognizes a variety of everyday postures, exercise routines, and household activities. The energy expenditure estimation system converts overall body motion (acceleration) and heart rate into energy expenditure estimates (in METs) over the course of a day. The following sections describe the wearable sensors used and the data collection protocols

followed to collect the necessary data to develop and evaluate the activity recognition and energy expenditure estimation algorithms.

### 4.2.1 Wearable Sensors

The wearable sensors used to collect the necessary data to develop and evaluate the activity recognition and energy expenditure algorithms were: (1) a wireless sensing platform called MIT environmental sensors (MITes) [199], and (2) some off-the-shelf sensors such as the MT1 Actigraph [32], the HJ-112 pedometer [201], the bodybugg armband [202], and the Cosmed K4b2 indirect calorimeter [203]. The following sections describe these sensors in more detail.

### 4.2.1.1 MIT Environmental Sensors (MITes)

The MIT environmental sensors (MITes) [199] were used to collect the accelerometer and heart rate data. The wireless accelerometers measure 3-axis acceleration in a range of  $\pm 2G$  with a 9-bit resolution. They are small (3.2x2.5x0.6cm), light (8.1g including battery), and easy to wear without constraining the wearer's movements. The battery life on a CR2032 battery is up to 31 hours. The MITes toolkit also includes a heart rate transceiver powered by a 9V battery that converts the beats per minute (BPM) data from a Polar wearable chest strap transmitter (WearLink) to the same format as the accelerometer data and forwards it to the same receiver used by the accelerometers in real-time. The wireless receiver included in this kit is small (see Figure 4-3), and can be easily attached to any USB port of a PC/Computer or mobile phone with USB host capabilities. The wireless link uses 16-Bit CRC error checking and consequently, the probability of data corruption due to noise is low. When CRC detects packet corruption, the packet is discarded. The raw data broadcasted by the MITes sensor nodes (accelerometers and heart rate transceiver) are time stamped after reception by the data collection software running on the laptop computer where the wireless receiver is attached. The sampling rate of the sensors depends on the number of sensors used per receiver (samplingRate=180Hz/numberOfSensors). As a result, during some data collection protocols followed in this work, two receivers are used (four sensors per receiver) to achieve a relatively high sampling rate of 45Hz when collecting data from seven accelerometers and a heart rate monitor. Figure 4-1a shows an image of the MITes wireless accelerometers and Figure 4-1b an image of the heart rate transceiver with corresponding 9V battery attached.

### 4.2.1.2 The MT1 Actigraph

The MT1 Actigraph accelerometer [32] is a small ( $3.8 \times 3.7 \times 1.8$ cm.) and lightweight (27g) uniaxial accelerometer that can measure accelerations in the range of 0.05-2 G sampled at 30Hz and band-limited between 0.25-2.5 Hz with a 12-bit resolution. This accelerometer and its previous version (the CSA activity monitor) are the commonly used for physical activity and energy expenditure studies. The MT1 Actigraph is shown in



Figure 4-1: Wearable sensors used during the data collections. (a) Wireless accelerometers, (b) heart rate transceiver, (c) MT1 Actigraphs, (d) Polar WearLink chest strap heart rate monitor, (e) HJ-112 pedometer, and (f) the bodybugg armband.

Figure 4-1c. During the data collection protocols, participants wore either one Actigraph at the hip or two Actigraphs placed at the dominant wrist and dominant side of the hip. The Actigraphs were set to record data using one second epochs (1s windows), and they were synchronized immediately before the data collection with the computer collecting MITes data.

### 4.2.1.3 The Polar Chest Strap Heart Rate Monitor

The Polar WearLink chest strap heart rate monitor [204] was used during the data collections to measure participants' heart rate in beats-per-minute (BMP). This heart rate monitor has been found to be a valid instrument for measuring heart rate in several studies [205-207]. The Polar WearLink chest strap is shown in Figure 4-1d. Before utilizing the heart rate data collected using the WearLink chest strap, a 15s running average filter was applied to the data to minimize noisy readings.

### 4.2.1.4 The Cosmed K4b2 Indirect Calorimeter

The Cosmed K4b2 indirect calorimeter [125] is an apparatus that measures oxygen uptake (VO<sub>2</sub>) and carbon dioxide (VCO<sub>2</sub>) production by means of a face mask. VO<sub>2</sub> values (in ml/min) can be converted to resting metabolic equivalents (METs) by dividing by the subject body weight and 3.5 (One MET is equivalent to 3.5ml of VO<sub>2</sub> per kilogram per minute). The device weighs less than 1kg and is composed of a face mask, a data collection unit, and a rechargeable battery. Figure 4-4 shows an example of how

participants wore the Cosmed K4b2 system. The data collection unit and battery pack were affixed to the participant's chest using a special harness consisting of adjustable belts. Before the data collection, the equipment was calibrated according to the manufacturer's specifications and synchronized with the time of the computer saving MITes data. Data were recorded in the unit's memory and later downloaded to a PC computer. The Cosmed device provides breath-by-breath data that are usually averaged over 30-60s intervals during energy expenditure research studies (e.g. [200]). In this work, a 15s running average filter was applied to minimize noisy readings. To account for the weight of the Cosmed K4b2 device and the other sensors used, 0.99Kg was added to the participant's weight in all MET calculations. The Cosmed K4b2 device has been found to be a valid and accurate device to measure energy expenditure in prior studies [200, 208, 209]. However, the quality of the measurements obtained strongly depends on the quality of the attachment of the face mask to participants.

Prior work utilizing the Cosmed K4b2 indirect calorimeter to measure energy expenditure has found standard deviations per activity ranging from 0.13 to 1.63MET. For example, the work by Bassett et al. [200] measured standard deviations per activity ranging from 0.31 to 1.58 MET for a set of 20 activities including household and exercise activities with moderate intensities. The work by Strath et al. [174] found standard deviations between 0.4 to 1.1MET over 14 lifestyle activities, and the work by Crouter et al. [34] standard deviations between 0.13 and 1.63MET for 18 exercise and lifestyle activates. The standard deviations obtained for the data collections performed in this work are discussed in Section 5.6.2.

### 4.3 Activity Recognition Algorithms

The process of recognizing activities from sensor data can be described in five steps: (1) signal processing, (2) signal or data segmentation, (3) feature computation, (4) classification, and (5) temporal smoothing. The first step is signal processing and, in the case of wireless sensor data, it usually includes band-pass filtering of the signals to eliminate undesirable noise and signal interpolation to fill out sensor values lost during wireless transmission. Depending on the sensors used, signal processing could also include signal calibration to compensate for slight sensor-to-sensor variations in output values due to hardware differences.

Signal segmentation refers to the process of grouping sensor values over time. This is necessary because features are computed not over single sensor values but over sequences of sensor values accumulated over time. Some common approaches to signal segmentation are the use of sliding windows, overlapping sliding windows, and signal spotting. The sliding windows technique consists on accumulating the sensor data over windows of specific length, where there is no time overlap between consecutive windows. Overlapping sliding windows, on the contrary, allow for some overlap between consecutive windows (e.g. 50%). Overlapping sliding windows are sometimes used because they are believed to minimize the edge conditions that arise when partitioning the data into independent sequential windows. Sliding windows and overlapping sliding windows have shown to be useful for analyzing periodic signals generated by repetitive motion such as *running* [37, 38, 40, 45, 103, 210, 211]. The length of the windows used

in these two segmentation techniques can vary depending on the activity being recognized from less than one second to a minute or longer. The length of the window introduces a real-time recognition delay equivalent to the duration of the window. The longer the window duration is, the longer the real-time recognition delay will be. Signal spotting [115, 212, 213] is widely applied in gesture recognition where the goal is to identify the start and end times of rarely occurring movements of interest from a stream of non-relevant signals or movements. In a simple manner, signal spotting could be implemented by, for example, setting a threshold over the variance of the signals for a given window of time. This would allow the identification of 'chunks' of signal data where the amount of motion observed is significant.

Features must be computed from the segmented sensor data to capture important temporal and spatial relationships. How these features are computed impacts the performance of the pattern classification algorithms. In fact, the features extracted are as important as, if not more important than the classifiers used to obtain good performance. Examples of time and frequency domain features often extracted include variance, mean, correlations, energy, entropy, and FFT coefficients [37, 38, 40, 103, 210, 211], among others.

Once features are computed, pattern classification algorithms are employed to discriminate among the activities of interest. There exist a wide variety of generative and discriminative classification algorithms, with different computational requirements and discriminatory power. Discriminative classifiers differentiate among activities of interest by building a decision boundary or mathematical function that separates the features representing each class as much as possible with respect a given criterion. Generative classifiers on the other hand, first attempt to create a model that describes how the data for each activity is being generated before building a mathematical function to discriminate among the activities in the feature space. Some examples of generative classifiers include Bayesian networks (e.g. the naïve Bayes classifier [106]) and dynamic Bayesian networks (e.g. hidden Markov models [214]), and some examples of discriminative classifiers include boosting classifiers (e.g. Adaboost [215]), decision tree algorithms (e.g. C4.5 classifier [112]) and support vector machines [216].

The final and sometimes optional step in activity recognition consists on performing temporal smoothing. Since classifications are performed over small windows of data and usually those windows are each considered independently of the others, the classification results usually contain spurious classifications. For example, if *running* is performed over ten consecutive windows, a classifier might confuse *running* with *walking* for a small subset of these windows. The confusion indeed makes sense since both activities involve similar motion patterns; nevertheless, the fact that most of those 10 windows are classified as *running* suggests that the activity being performed is *running*. One reason for the generation of spurious classifications in the previous scenario is that temporal information about the activity being performed is not incorporated. In other words, the classifier does not know that instantaneously switching back and forth between running and *walking* is difficult and, for other activities, not even possible. One simple strategy to partially mitigate this problem has been the use of a majority filter. This filter consists on simply reporting the activity with most classifications (majority class) over the past nnumber of windows as the latest classification result. Another possible strategy is to incorporate information about the transition probability between activities. For example,

it is more likely to transition from *sitting* to *standing* than from *sitting* to *walking*. In fact some algorithms such as hidden Markov models attempt to implicitly include this information.

The design goals for the activity recognition algorithm developed in this work in decreasing order of importance are (1) real-time performance, (2) recognition of physical activity type and intensity, (3) subject independent performance (if possible) or subject dependent performance with small training data requirements, (4) minimization of the number of sensors required to recognize activities while maintaining good performance, (5) placement of sensors in the most comfortable body locations, (6) as much invariance to small changes in the sensor placement on the human body as possible, and (7) small real-time classification delay. Section 5.4 presents a set of incremental experiments specially designed to explore the feasibility of the aforementioned design goals.

### 4.4 Energy Expenditure Estimation Algorithms

The standard procedure used to estimate energy expenditure from accelerometer data in prior work consists of the following steps: (1) collect accelerometer and energy expenditure data for a specific set of activities of interest by having n number of subjects wear an accelerometer (e.g. an Actigraph [32]) at the hip and an indirect calorimeter (e.g. the portable Cosmed K4b2 indirect calorimeter [125]. Another option is to place participants inside room indirect calorimeters (e.g. Vanderbilt indirect calorimetry room [124]) (2) Eliminate the portions of data where energy expenditure does not reach steady state by removing a portion of the data (e.g. 30-40%) at the beginning of each activity. (3) Partition the data collected into training data (usually 60-70% of total data) and test data (usually 30-40% of data). (4) Compute the absolute value of the acceleration signal and sum it sample by sample over one-minute windows (if not already computed by the activity monitor). This feature is usually referred as activity "counts" and represents the overall motion experienced by an accelerometer over a one minute interval. If the accelerometer used has multiple acceleration axes (e.g. if a biaxial or triaxial accelerometer is used), the summation is usually performed over all axis to produce a single value representing the overall motion experienced by the sensor in all directions. (5) Compute the mean value over the same one minute windows for the ground truth energy expenditure data collected using the indirect calorimeter (given in METs or Kcal/min). (3) Employ a regression algorithm to map activity counts to energy expenditure using the training data. (4) Test the performance of the regression algorithm by predicting energy expenditure over the test data using the model learned from the training data.

In summary, the standard procedure followed to estimate energy expenditure from accelerometer data is a simplified version of the procedure presented in the previous section to recognize activities. The main differences are the following: (1) regression algorithms such as multivariable linear regression are used instead of classification algorithms to predict energy expenditure, (2) the segmentation technique usually employed is non-overlapping sliding windows of 60s in length, and (3) in most prior work the only feature computed from the accelerometer data is number of activity counts. When multiple accelerometers are used, multivariable regression algorithms such as

multivariable linear regression are employed to produce the mapping between acceleration and energy expenditure.

The design goals for the energy expenditure estimation algorithm presented in this work are the same as the goals presented in Section X for the activity type recognition algorithm. This work extends the standard procedure used estimate energy expenditure by exploring (1) if non-linear regression algorithms and activity dependent regression models improve energy expenditure estimates, (2) if shorter window lengths with shorter real-time estimation delays produce acceptable estimates, (3) if computing more complex features over the accelerometer data improve the estimates, and (4) if a minimum set of multiple sensors at strategic and comfortable body locations improve the estimates. One of the main goals is also to identify regression algorithms, window lengths, features, and accelerometers (number and placement) that produce an energy expenditure estimation algorithm amenable for real-time performance.

### **4.5 Interactive Training of Algorithms in Real-Time**

Previous work on activity recognition (e.g. [38]) suggests that subject independent recognition of activities is more difficult than subject dependent recognition of activities. This means that a recognition algorithm performs better if all users provide examples of all the activities to recognize. Thus, it might be expected that some degree of subject dependent training will be necessary to recognize activities during free-living conditions reliably.

This work evaluates the activity recognition algorithms implemented using both -subject dependent and independent training. The training data requirements during subject dependent training are also evaluated. This analysis is important because it gives an idea of the minimum amount of training data that users might have to provide to the recognition algorithms to achieve a reasonable performance. Finally, the activity recognition algorithm developed in this work is implemented and evaluated in real-time. This is achieved by having several participants interactively train and test the performance of the algorithm over activities specified by participants themselves. The main goal of this real-time evaluation is to determine how difficult it would be for endusers to interactively train and test the activity recognition algorithms in practice. The energy expenditure estimation algorithm, on the other hand, is evaluated primarily using subject independent training because of the unavailability of the specialized equipment (portable indirect calorimeter) required to collect the energy expenditure data during freeliving. Nevertheless, once the final algorithm is presented, its performance is also evaluated during subject dependent training.

The main goal of the user interface of the application developed to test the performance of the activity recognition algorithms in real-time is to allow users to (1) type in the activities they want the activity recognition algorithm to recognize, (2) provide the required training data automatically by simply performing the activities of interest sequentially for a given period of time (e.g. 2 minutes), and (3) test the performance of the recognition algorithm immediately after the training process has finished. More details on this interactive application can be found in Section 5.5.
# 4.6 Data Collection

Three different datasets were used during the development of the activity recognition and energy expenditure estimation algorithms in this work. The first one, referred as the Boston Medical Center Protocol, was a small dataset consisting of 14 activities collected from two participants by researchers at the Boston Medical Center. This dataset was used to validate the energy expenditure data collected for developing the energy expenditure estimation algorithm in this work. The second dataset, referred as the Stanford and Boston University protocol, consists of data collected during two independent data collection sessions at those institutions where acceleration and heart rate data was collected from participants to develop the activity recognition algorithms. Finally, the third dataset referred as the MIT dataset, is the most extensive data set since accelerometer, heart rate, and energy expenditure data was collected from 20 participants at two locations (a gymnasium and residential home) for 52 different activities. The following sections discuss each of the data collection sessions in more detail. Researchers interested in using these datasets should contact the author or visit http://architecture.mit.edu/house n/data

# 4.6.1 Boston Medical Center Protocol

This is a dataset consisting of 14 activities collected from two participants by researchers at the Boston Medical Center. The activities for which data was collected are shown in Appendix A10. This data was used to validate the energy expenditure data collected in the MIT protocol for the development of the energy expenditure estimation algorithm. This comparison was possible because the 14 activities contained in this protocol are a subset of the 52 activities contained in the MIT protocol. Section 5.6.2 presents the results of the validation study. The main difference between the Boston Medical Center protocol and the MIT protocol is that different indirect calorimeters were used during the data collections. The Boston Medical Center protocol used the Parvo Medics TrueOne 2400 metabolic measurement system [126], and the MIT protocol used the Cosmed K4b2 indirect calorimeter [125]. The number and placement of all other wearable sensors remained the same for both protocols. For more details on the placement of the other sensors refer to Section 4.6.3. Figure 4-2 shows an example of the data collection setup during the *cycling* activity for the Boston Medical Center protocol.

# 4.6.2 Stanford and Boston University Protocol

For this protocol, a total of 21 participants with ages ranging between 18 and 65 years old and with varying levels of physical fitness were recruited at two separate medical laboratories: (1) Stanford Medical School and (2) The Boston Medical Center. The participants were screened for any medical condition that would contraindicate moderate to vigorous intensity exercise using the Physical Activity Readiness Questionnaire (PAR-Q) [217] and the Stanford Brief Physical Activity Survey (SBAS) [133] before inclusion in the study.



Figure 4-2: Example of the data collection setup for the Boston Medical Center Protocol. The indirect calorimeter used was the Parvo Medics TrueOne 2400. Participant's face has been blurred to preserve anonymity.

During the data collections, researchers placed five accelerometers on each subject, with one at each of the following locations: top of the dominant wrist just behind the wrist joint, side of the dominant ankle just above the ankle joint, outside part of the dominant upper arm just below shoulder joint, on the upper part of the dominant thigh, and on the dominant hip. The wireless accelerometers were attached to subjects using cotton elastic sweat bands or non-restrictive adhesive bandages. All the accelerometers were  $\pm 10G$  except the accelerometer on the hip, which was  $\pm 2G$ . Prior work suggests that body extremities can reach  $\pm 12G$  in rare circumstances [218], but at the hip  $\pm 2G$  should be sufficient to avoid saturation. The heart rate monitor was worn on the chest. The location of each sensor is indicated in Figure 4-3. After the sensors were placed, each participant was asked to sit still and, after a stabilization period, resting HR was measured by measuring pulse for one minute. The participant's age-predicted maximum HR (MHR=220-age) was calculated. During the data collection, if the participant reached 85% of this value, participants were instructed to stop the activity being performed and take a two minute rest break. That activity was not attempted again.

A combined 21 participants each performed 30 gymnasium activities, with 12 and 9 datasets being collected from each site, respectively. The list of activities broken down by type and intensity differences is shown in Appendix A11. Each activity was performed for two minutes. The research assistant would wait until the participant had started the activity and appeared to be engaged confidently in the activity before indicating the activity was started using annotation software. In most of the gym activities, the gym equipment could be set at a particular speed and/or resistance level. For cycling or rowing, the "light," "moderate," and "hard" resistance settings were made by setting the speed (e.g., rpm) and then as the participant peddled or rowed, adjusting the resistance



Figure 4-3: Five triaxial wireless accelerometers, a polar chest strap heart rate monitor, a MITes heart rate transceiver with 9V battery attached, and a USB wireless receiver, and (b) Placement of the sensors on the body. Green dots indicate accelerometer locations and blue the heart rate monitor location.

Variable	Men (n = 8)	Women $(n = 12)$	All Participants $(n = 20)$
Age (yr)	$27.8 \pm 7.2$ [22.0 - 42.0]	$29.2 \pm 6.4 \ [18.0 - 40.0]$	$28.6 \pm 6.6 \ [18.0 - 42.0]$
Body Mass (Kg)	85.1 ± 11.0 [73.2 - 103.8]	75.5 ± 17.3 [60.7 - 119.1]	79.3 ± 15.6 [60.7 - 119.1]
Height (m)	$1.9 \pm 0.0 \ [1.79 - 1.9]$	$1.7 \pm 0.0 \ [1.54 - 1.82]$	$1.7 \pm 0.1 \ [1.54 - 1.9]$
Fat percent (%)	$12.2 \pm 4.6$ [5.5 - 20.3]	30.7 ± 7.9 [19.8 - 52.3]	23.3 ± 11.4 [5.5 - 52.3]
Water percent (%)	$58.9 \pm 4.7$ [50.0 - 65.5]	48.2 ± 4.9 [35.4 - 54.4]	52.7 ± 7.2 [35.4 - 65.5]
Bone mass (Kg)	$3.6 \pm 0.3$ [3.2 - 4.0]	$2.7 \pm 0.5$ [2.2 - 3.9]	$3.0 \pm 0.6$ [2.2 - 4.0]

Table 4-1: Physical characteristics of the 20 study participants. Values are means (SD) with range shown in parenthesis. n is the number of subjects.

until the participant reported that they were in the desired intensity level.

The activities for this protocol were selected because (1) they are listed in the Compendium of Physical Activities [219], (2) because they include examples of upper, lower, and overall body motion, (3) and because these activities are of practical interest to the medical community because they are prescribed as part of weight management programs. From Appendix A11, we can observe that the activities with different intensity levels are *walking*, *cycling*, and *rowing*. For *walking*, intensity was varied by changing the treadmill speed (e.g. 2, 3, and 4 mph) and inclination (e.g. 4, 8, and 12 degrees). These walking speeds and inclinations were used for all participants. For cycling, the cycle speed (e.g. 60, 80, and 100rpm) and the cycle resistance level were set to settings that participants subjectively considered equivalent to light, moderate, and hard. Finally, for rowing, the rowing speed was maintained constant at 30 strokes per minute (spm) while the resistance was varied until reaching levels that participants considered light, moderate, and hard. The average duration of each activity was 1.9min except for *jumping jacks, sit-ups, push-ups, and bicep curls* that have an average duration of 0.87min. This is because these activities are physically demanding, and most participants were not able to perform them for at least 2min. The Stanford University dataset differs slightly from the Boston Medical Center dataset because due to lab constraints, data for the *move weight*, and *calisthenics* activities were not collected, and the *rowing* activity was replaced by the arm ergometry activity. Nevertheless, the Stanford University dataset still contains the

same number of activities with different intensity levels as the Boston Medical Center dataset (*walking*, *cycling*, and *arm ergometry*). The activity recognition results utilizing the datasets collected at Stanford and Boston University are presented in the work by Munguia Tapia et al. (2007) [220].

#### 4.6.3 MIT Protocol

For this data collection protocol, total of 20 participants with an age range between 18 and 42 years old and with varying levels of physical fitness were recruited at MIT. The participants were screened for any medical condition that would contraindicate moderate to vigorous intensity exercise using the Physical Activity Readiness Questionnaire (PAR-Q) [217] and the Stanford Brief Physical Activity Survey (SBAS) [133] before inclusion in the study. Each participant's age, gender, ethnicity, height, weight, and body composition parameters were recorded at the beginning of the study. Each participant's height was measured using a measuring tape and body weight and composition parameters using a Tanita Body Composition Scale Model BC-536 [221]. Table 4-1 shows the physical characteristics of the 20 participants included in the study.

The participants were asked to perform a script of physical activities at two locations while they wore a variety of sensors that measured body motion (acceleration), heart rate, and energy expenditure. One data collection session took place at MIT Zesiger Sports and Fitness Center and the other at a residential home called the PlaceLab [222]. Participants were expected to complete both data collection sessions, which lasted approximately three hours each.

During the data collections, participants wore seven wireless accelerometers [199], a chest strap heart rate monitor [199, 223], two Actigraph activity monitors [166], a HJ-122 pedometer [85], a bodybugg armband [158], and a K4b2 portable indirect calorimeter [125]. The wireless accelerometers and heart rate monitor collected data about participant's motion, and heart rate, and the portable indirect calorimeter 'ground truth' energy expenditure data. The wireless accelerometers were placed at the following locations: At the feet (on top of the shoe laces), wrists, hip (non-dominant side), dominant upper arm, and dominant thigh. The Actigraph monitors were placed at the dominant wrist, and at the dominant side of the hip. The bodybugg armband was placed at the dominant upper arm, near the bicep muscle, the pedometer at the hip, and the heart rate monitor on the chest. Accelerometers were held in place using cotton elastic sweat bands or non-restrictive adhesive bandages. The K4b2 Cosmed indirect calorimeter [125] was calibrated according to the manufacturer's specifications and affixed to the participant's chest using a special harness consisting of adjustable belts. The location of each sensor is shown in Figure 4-4.

The Actigraph is commonly used to estimate energy expenditure by the medical community in free-living and was used for comparison purposes. The bodybugg data was not utilized in this work due to the lack of access to expensive proprietary software (\$5,000). The wireless accelerometers and heart rate transceiver used in this study are part of an existing wireless sensing platform called MIT environmental sensors or (MITes)[199]. All other sensors are available off-the-shelf [89, 125, 158, 166]. A member of our research group followed the participants to label (using the data collection laptop computer) the start and end times activities as they were performed.



Figure 4-4: Example of how participants wore the sensors used during the data collections at the (a) gymnasium (shown here) and (b) home. Participant's face has been blurred to preserve anonymity.

During both data collection sessions, participants were asked to lie down for a period of five minutes once sensors were properly installed on the subject so that resting metabolic rate and resting heart rate could be measured. The participant's age-predicted maximum HR (MHR=220-age) was also calculated. During the experiment, if a participant reached 85% of this value, he or she was instructed to stop the activity being performed and take a 2 minute rest break. That activity was not attempted again.

After lying down for five minutes, subjects were asked to perform a script of activities depending on the location of the data collection. Appendix A12 presents the script of activities for the gymnasium and the home, respectively. Appendix A12 also presents a detailed list of the activities and a brief explanation of the data collection conditions for each activity. Each activity was performed for three to four minutes, except for physically demanding activities that were executed for as long as the subject's physical fitness allowed. A research assistant would wait until the participant had started the activity and appeared to be engaged confidently in the activity before indicating that the activity was started using the annotation software.

One important difference between this protocol and the Stanford and Boston University protocol is that this study does not define the intensity levels light, moderate, and hard depending on the subject's exertion perception. In other words, the intensity levels (speed of execution of activity, resistance level, or weight loads used) were set in advance and used for all subjects. The intent was to minimize inter-individual variations due to different fitness levels of individuals. Appendix A4 describes the speeds of execution, resistance level, and weight loads used for each activity involving different intensity levels.

Variable	Men (n = 7)	Women $(n = 9)$	All Participants (n = 16)
Age (yr)	25.71 ± 4.61 [22.0 - 35.0]	29.56 ± 7.16 [18.0 - 40.0]	$27.88 \pm 6.30$ [18.0 - 40.0]
Body Mass (Kg)	84.30 ± 11.66 [73.2 - 103.8]	$72.28 \pm 12.05$ [60.7 - 101.8]	77.54 ± 13.03 [60.7 - 103.8]
Height (m)	$1.85 \pm 0.04 \ [1.79 - 1.9]$	$1.67 \pm 0.09 \ [1.54 - 1.82]$	$1.75 \pm 0.12$ [1.54 - 1.9]
Fat percent (%)	12.11 ± 4.99 [5.5 - 20.3]	28.14 ± 4.67 [19.8 - 34.2]	21.13 ± 9.44 [5.5 - 34.2]
Water percent (%)	59.11 ± 5.05 [50.0 - 65.5]	49.66 ± 2.88 [46.5 - 54.4]	54.07 ± 6.24 [46.5 - 65.5]
Bone mass (Kg)	$3.56 \pm 0.29 \ [3.2 - 4.0]$	$2.70 \pm 0.56$ [2.2 - 3.9]	$3.10 \pm 0.62$ [2.2 - 4.0]

 Table 4-2: Physical characteristics of the 16 participants included in the MIT energy expenditure datasets. Values are means ± SD with range shown in parenthesis and n is the number of subjects.

Finally, the set of activities contained in this protocol were included because they are common everyday activities or exercises for which data could be collected in the gym or home setting available, and because most of them are listed in the Compendium of Physical Activities [219]. Appendix A4 presents the amount of training data collected for each activity.

#### 4.6.4 MIT Energy Expenditure Dataset

After visually inspecting all of the datasets collected during the MIT protocol, 13 data collection sessions with suspiciously low energy expenditure readings were identified. These low energy expenditure readings might indicate an improper attachment of the Cosmed K4b2 indirect calorimeter face mask during the data collection. Improper attachment of the face mask is known to produce low energy expenditure readings because air (VO<sub>2</sub> and VCO<sub>2</sub>) escapes from the mask and, consequently, is not measured by the indirect calorimeter data collection unit. Some subjects did express at times that the face mask felt too tight, and as a result, it had to be loosened, increasing the likelihood of air leaks. In other cases, the face mask did not properly match the subject's face anatomy, even when the closest face mask size was selected from those available with the Cosmed K4b2 equipment.

In total, 13 sessions out of a total of 40 were removed from analysis. These sessions removed included data collections at the gymnasium, the residential home, or both. Appendix B17 presents a list of the data collection sessions included in the MIT energy expenditure dataset. A data collection session was eliminated if any of the activities performed (during the data collection), presented an energy expenditure value below ~40% of the values observed for other participants or with respect to the Compendium of Physical Activities. All sessions were removed before performing any of the experiments presented in this work. The new dataset created after eliminating the sessions with low readings is referred as the MIT energy expenditure dataset, and contains data for 16 different subjects. Table 4-2 presents the characteristics of the 16 subjects included in the MIT energy expenditure dataset. Appendix B16 presents the average energy expenditure (in METs and Kcal/min) and average heart rate per activity when non-steady state data is eliminated and when it is not.

# **5** Evaluation

This section presents a summary of the standard quantitative measures in which activity recognition and energy estimation algorithms are commonly evaluated. Useful evaluation measures that are important to consider but are usually neglected when evaluating the algorithms are discussed. The measures used to evaluate the algorithms developed in this work are introduced, and the section discusses why is it difficult to compare results obtained with different algorithms developed in previous work and what will be considered to be a "good" result in this thesis.

# 5.1 Reporting and Analyzing Activity Recognition Results

This section explains what performance measures will be used to evaluate the activity recognition algorithms developed in this work and why.

# 5.1.1 Standard Quantitative Measures

Activity recognition algorithms are often evaluated based on how well they recognize all the target activities of interest with respect to overall standard performance measures. Such measures might include accuracy, true positive rate, false positive rate, true negative rate, false negative rate, precision, recall, F-Measure, area under ROC curve, and some others recently introduced [224, 225]. Appendix A1 presents a brief description of some of these standard evaluation measures. However, even when improving the recognition over all of the activities of interest is important, overall improvements as represented by specific performance measures can be deceiving. For example, overall performance can increase even when the performance on some individual activities might, in fact, decrease. Furthermore, when improvements indeed occur over all or some activities without affecting performance on others, these improvements can be so small (e.g. 95.5% over 94.2% accuracy) that the increase in the complexity of the algorithm might not justify the improvement obtained. Consequently, it can be argued that *epsilon* improvements in overall performance do not necessarily represent significant improvements over results obtained with previously developed algorithms unless the new algorithms provide some other benefit, such as faster run-time or training performance or better interactive training potential.

The target domain dictates the relative importance of recognizing particular activities. For example, an algorithm might improve the recognition accuracy of *ascending stairs* which might allow better differentiation between *walking* and *ascending stairs*. This improved differentiation might be considered a strong result by exercise physiologists interested in estimating energy expenditure during free living given the significantly different energy expenditure costs associated with these activities.

Therefore, the algorithms developed in this work will be evaluated by (1) paying special attention to the individual performance on activities and classes of activities that are considered important for the medical community (e.g. cycling, ascending/descending

stairs, postures, moderate intensity, among others) and (2) utilizing other important evaluation measures commonly neglected while evaluating results in previous work.

# 5.1.2 Neglected Performance Measures

The end goal of this work is to move towards a consumer-based system for helping people measure and motivate behavior changes. Therefore, factors beyond recognition accuracy must be considered. Additional dimensions along which algorithms can be evaluated are briefly explained below.

# 5.1.2.1 Complexity of the Activities to Recognize

Algorithms can be evaluated based on the complexity of the activities they recognize. The complexity of activities can vary depending on the number of activities, the types of activities, and the complexity of the training data collected for those activities (e.g. lab vs. free-living). It is extremely important to consider the complexity of the activities being studied when evaluating algorithms, because by carefully picking the number of activities, the types of activities, and the data collection method, excellent results can be obtained by many classification algorithms.

- *Number of activities to recognize*: Recognizing five activities is easier than recognizing twenty or fifty activities. As the number of activities increases, the classifier has to learn how to discriminate among a larger set of activities, which is usually harder. Discrimination is also harder if activities are "similar" to one another. In this work the recognition rate as compared to chance is listed when reporting all results.
- Complexity of the types of activities to recognize: Activities that are static in nature such as postures (e.g. lying down and standing still), are easier to recognize than activities that are periodic in nature such as running and cycling. Furthermore, activities that involve different intensity levels (e.g. running at 4mph vs. running at 5mph) are also harder to recognize because of their motion similarity and therefore similarity in the feature space. Likewise, activities involving highly unconstrained motions impacted by objects in the environment such as cleaning and making the bed are more difficult to recognize than periodic activities when only accelerometer data from limb motion is available.
- Complexity of the training data collected for the activities: Activities for which training and test data is collected in laboratory settings are usually easier to recognize than activities for which training and test data is collected during free-living conditions. Subjects will usually behave differently and in less constrained ways outside of formal laboratory settings. This work utilizes data collected for 52 activities at a gym during relatively controlled conditions (since exercise equipment such as a treadmill, cycling machine, etc was utilized) and from a residential home during less controlled conditions. It can be argued that this is a relatively complex dataset containing a large set of activities. However, direct observation was utilized during both data collections to acquire the labels of the activities being performed

by participants. Thus, the results presented in this thesis would require validation over datasets collected when direct observation is not utilized.

In this work, the performance of the activity recognition algorithms will be evaluated on different subsets of activities, ranging from relatively easy (postures in a gym) to difficult (cleaning activities collect in a home setting). First, a worse-case scenario of activity recognition complexity will be used when selecting the parameters of the recognition algorithm (e.g. classifier, window length, features, etc). This worse-case scenario consists of recognizing 52 different activities, 26 of which have different intensity levels and 18 of them include examples of the unconstrained motion found in household activities. Once the parameters are selected, the recognition performance of the algorithm will be evaluated on various groups of activities. First, the algorithm will be evaluated by only recognizing postures. Then, it will be evaluated on postures and ambulation, then on exercise activities with different intensity levels (periodic motion), and finally for household activities (unconstrained motion). The algorithm will also be evaluated when recognizing all the activities but without differentiating among activities containing different intensity levels. The datasets on which the algorithms are evaluated were collected in both (1) relatively controlled laboratory conditions and (2) less controlled free-living conditions. To the best of the author knowledge, the dataset collected for this work is larger and more complex that other datasets used in activity recognition studies published to date.

# 5.1.2.2 Training Data Requirements of an Algorithm

Algorithms can also be evaluated based on the type of training data available (e.g. subject dependent vs. independent) and the amount of training data that they require.

- Subject independent recognition of activities: Ideally, an activity recognition algorithm would be trained on a given subject population and then recognize activities accurately on unseen populations without requiring further person-specific training data. Unfortunately, previous work (e.g. [38]) strongly suggests that subject independent recognition of activities is difficult to achieve for diverse sets of activities due to the high variability in the way that individuals perform activities. Body type can also impact sensor readings (e.g., hip sensors at different angles of tilt due to body shape).
- Amount of training data required for subject dependent recognition of activities: In general, previous work on activity recognition suggests that algorithms will perform better with more person-specific training data. For many activities, providing this data can be time consuming and burdensome, so ideally training data requirements would be kept to a minimum. For example, an algorithm that takes only 1min to train might allow end-users to iteratively test and perhaps improve the recognition of poorly recognized activities by providing more training examples in-situ. Algorithms with long training times (e.g. several hours or days) might exhaust end-user's patience while subject dependent training is required.

This work will evaluate the performance of the algorithms developed using subject independent and subject dependent training. The differences in performance using both training methods will be reported overall and per activity. Moreover, the training data requirements for subject dependent training will also be evaluated by training the algorithms using different amounts of training data to determine the minimum amount required to obtain good recognition performance.

# 5.1.2.3 Sensor Data Requirements of the Algorithm

The number of sensors required to recognize activities, the types of sensors used, and their location on the human body can also dramatically impact algorithm performance:

- Number of sensors required to recognize activities: Algorithms that recognize activities from a small set of sensors are easier to use in real-world applications and have lower computational requirements (since fewer sensor signals need to be analyzed) than algorithms that use large sets of sensors. Fewer sensors usually mean the technology can be deployed more affordably.
- Intrusiveness of the sensors required to recognize activities: A recognition algorithm that requires sensors such as ECG or EMG monitors that must be inconveniently stuck on the body may be perceived as more difficult to use, more uncomfortable, and more of a burden than an algorithm that uses sensors such as accelerometers that can be easily and comfortably slipped on the body.
- Location of the sensors required to recognize activities: Sensors worn at socially accepted and comfortable locations on the body are more likely to be used for longer periods of time. Sensors that might be integrated into existing clothing or devices already worn (watches, shoes) and carried (phones) are preferable.

This thesis will evaluate tradeoffs between different types (e.g. accelerometers vs. heart rate monitors), numbers, and locations of sensors to determine sets of sensors to use that provide good recognition with minimal usability burden.

# 5.1.2.4 Usability Factors Imposed by an Algorithm

Some other usability factors in addition to usage comfort of the sensors and training time requirements of the algorithm are:

- *Real-time recognition of activities*: An algorithm capable of recognizing activities in real-time, particularly from hand-held devices such as mobile phones, would permit many new health interventions using just-in-time feedback to be created [28].
- *Real-time recognition delay (lag)*: Some real-time recognition algorithms introduce lag between activity occurrence and detection. Shorter recognition delays may allow for better point of decision interventions.

This work will analyze the usability factor imposed by the activity recognition algorithm carefully, since one of the main goals of this work is to develop an algorithm amenable for longitudinal use during free-living. First, it will be determined if reasonable recognition of activities can be obtained using classifiers suitable for real-time performance in low-processing power devices such as mobile phones. Secondly, it will be decided if subject independent recognition of activities is possible with reasonable performance. If not, the training data and training time requirements of the algorithm will be evaluated in a subject dependent manner. Experiments will also be conducted to find the minimum number of least intrusive sensors to wear and their more comfortable location on the human body.

# **5.1.3** Comparison Challenges

In general, it is difficult to compare the performance of newly developed algorithms with respect to the performance obtained by algorithms in previous work. This is because different pieces of work explore different types and numbers of activities, datasets are collected from different subject populations under different conditions (e.g. lab vs. non-lab), and the duration of the data collections is different (activity examples have different lengths).

For example, Table 5-1 presents the comparison of two state-of-the-art activity recognition algorithms: The work by Bao and Intille [38] and the work by Lester et al. [37]. The work by Bao and Intille is one of the first pieces of work to explore a relatively large set of activities (20) collected from a large number of subjects (20) during seminaturalistic and controlled conditions. The algorithm presented recognizes activities from up to five accelerometers and is evaluated using subject dependent and subject independent training. Moreover, experiments are performed to determine the minimum set of sensors with best performance and their locations on the human body. One of the main finding is that an accuracy of 81% can be obtained for a complex set of activities (postures, ambulation, and household activities) using only two sensors located at the thigh and the wrist. The work also indicates that some activities are well recognized using subject independent training but others do appear to require subject dependent training. The work by Lester et al. explores the recognition of ten activities from eight sensor types located at the shoulder. The algorithm is evaluated on the data collected from two subjects during everyday naturalistic conditions. Their main finding is that using different sensor types at a single location can offset the need to place sensors at different locations, making the system easier to use and wear. The algorithm developed is a complex combination of discriminative classifiers (e.g. a modified version of AdaBoost) and generative classifiers (hidden Markov Models). The work obtains a final recognition accuracy of 95% while recognizing ten activities involving mainly postures and ambulation.

From Table 5-1, it can be seen that comparing these two activity recognition algorithms is a complex task. Some of the performance dimensions were not evaluated, the number of activities explored and their complexity is different, and the datasets were collected under different conditions. As a result, it is challenging to determine what a meaningful

Performance Dimension	Bao and Intille '04 ([38])	Lester et al. '05 ([37])		
Overall performance	Yes (84% accuracy)	Yes (95% accuracy)		
Performance per activity	Yes (True positive rate)	Yes (precision and recall)		
Complexity of activities to recognize				
Number of activities	20	10		
Complexity of activity type	Activities included Postures (static),	Activities included Postures (static) and		
	ambulation (periodic), and household activities (unconstrained motion).	ambulation (periodic)		
Complexity of training/testing data	20 subjects, data collected under semi-	2 subjects, 12 hours of total data		
	naturalistic and controlled conditions, data	collected under naturalistic conditions.		
	collection boards did not restrict subject's	On average, one hour of data per		
	movement. 33 hours of total data with an	activity.		
	average of 2.6min of data per activity.			
Training data requirements				
Subject independent recognition	Yes	Yes		
Subject dependent recognition	Yes	No		
Amount of training data required	No	No		
Sensor data requirements				
Number of sensors	Five biaxial accelerometers located at hip,	Eight sensors located on the shoulder: a		
	wrist, arm, ankle and thigh	triaxial accelerometer, microphone,		
		IR/Visible light, high frequency light,		
		barometric pressure, humidity,		
		temperature and compass.		
Intrusiveness of sensors	Yes (only unobtrusive sensors used)	Yes (only unobtrusive sensors used)		
Location of sensors	Yes, accelerometers at thigh and wrist	No		
Ucability.	achieved best performance			
Comfort	Analysis performed to determine best	Ves only sensors at a single location		
Connort	sensor location but data collection boards	(shoulder) are used No analysis on		
	are bulky and heavy.	multiple locations is performed though.		
Training data requirements	No	No		
Training time requirements	No	No		
Real-time performance	No (offline analysis only)	No (offline analysis only)		
Real-time classification delay	No (offline analysis only)	No (offline analysis only)		
Interpretability of recognizer	No	No		
Ability to fix errors	No	No		
Cost	No	No		

Table 5-1: Comparison of two state-of-the-art activity recognition algorithms along all the performance dimensions discussed in Sections 5.1.1 and 5.1.2. The comparison is difficult, and it is not clear what algorithm would perform better under similar conditions.

improvement is over these works when only considering one of the performance dimensions (e.g. overall accuracy).

# **5.1.4** Comparison Strategy in this Work

In this work, the activity recognition algorithms will be evaluated across all the performance dimensions discussed in Sections 5.1.1 and 5.1.2. The experiments performed will mostly be used to compare incrementally to previous results obtained in this work, particularly when selecting the parameters (e.g. classifier, window length, features) of the recognition algorithm. The goal is to converge on a functional system that is balanced in all dimensions as much as possible, where testing is done on a relatively large and complex set of activities relative to past work. Where possible, comparisons will be made against results presented in previous work by other authors, but they will be presented cautiously given how challenging this task is as explained in the previous section.

This work will utilize the following performance measures that can be computed from the confusion matrix to evaluate recognition algorithms: overall accuracy, true positive rate per activity, false positive rate per activity, F-Measure per activity, and the confusion matrix itself. These standard performance measures are briefly explained in Appendix A1. Furthermore, the results presented in each experiment will be clustered based on activity categories that are useful for interpreting the results. These categories are: (1) postures, (2) ambulation, (3) exercise activity, (4) resistance exercise activity, and (5) household activity. Appendix A2 explains what activities are considered under each clustering category. As stated in the previous sections, subject dependent and independent evaluation will be performed when necessary.

Based on the results obtained in previous work, this work will consider a good result to recognize the set of 52 activities contained in the MIT dataset with overall accuracies equal or greater than 80% in a subject dependent, or even better, independent manner. Another strong result would be to recognize the activities from only one or two sensors, with an overall decrease in accuracy of less than 5% and no significant decrease in performance per activity. Under the same conditions (number of activities, types of activities, number of subjects, number of sensors, evaluation criteria, etc.), this work will also consider an overall performance improvement of 15% or greater as significant improvement over previous work provided that the complexity of the algorithms is comparable or even lower.

# **5.2 Reporting and Analyzing Energy Estimation Results**

The same evaluation dimensions discussed in the previous section can also be used to evaluate the performance of energy expenditure estimation algorithms. However, there are four main differences with respect to evaluating activity recognition algorithms: (1) The performance measures used are different since energy estimation involves numeric prediction, (2) there is less agreement in the literature on which performance measures to use while reporting results, (3) results presented are more difficult to compare against one another since energy expenditure can be predicted as net energy or gross energy and reported in different units (e.g. Kcal/Min, METs), and (4) most previous work has been done on data collected at a laboratory from few activities (usually *walking* and *running* on a treadmill) under steady-state assumption of energy expenditure.

# **5.2.1 Standard Quantitative Measures**

Energy expenditure estimation algorithms can be evaluated by comparing two numeric values: the ground truth energy expenditure values acquired usually from an indirect calorimeter and the predicted energy expenditure values estimated by the algorithm from the sensor data. In this scenario, errors can occur due to erroneous readings from the indirect calorimeter itself and the estimation errors performed by the algorithm. The errors produced by the estimation algorithm can be evaluated by computing the difference (error) between the calorimeter values and the predicted values according to a performance measure. However, there is little agreement as to which error measures to use to report results in the literature. For example, some of the most often used error

measures are: root mean squared error (RMSE), standard error of the estimate (SEE), Pearl's correlation coefficient (r), squared Pearl's correlation coefficient ( $r^2$ ), mean absolute error (MAE), maximum absolute error deviation (MAED), relative absolute error (RAE), root relative squared error (RRSE), percentage difference, difference in total energy expenditure, and mean error scores, among others. Often, scatter plots and Bland-Altman plots are also used to present the results. This makes comparison of existing and previous work difficult, as explained in the next section.

#### **5.2.2 Neglected Measures**

The same performance dimensions used to evaluate activity recognition algorithms can be used to evaluate energy expenditure estimation algorithms. However, one important difference with respect to activity recognition is that most previous work in estimating energy expenditure from wearable sensors has been performed over data collected at the laboratory from a limited set of activities. Algorithms in prior work have most often been trained and tested using data collected from treadmill activities such as *walking* and *running* at different speeds. The datasets may not reflect the variations found in other exercise or lifestyle activities. Consequently, it is no surprise that regression models trained only on these activities have been found to slightly overestimate the energy cost of *walking* and light activities and greatly underestimate the energy expenditure associated with moderate-intensity lifestyle activities [200]. There are few exceptions to this trend of training and testing algorithms on treadmill activities [34, 152]. In work by [34], models were trained over data collected for 18 activities from 20 subjects and in a work by [152], data was collected for 12 activities over nearly 24hours for 102 subjects.

Another potential limitation of previous work in energy expenditure estimation is that the regression models are usually trained assuming steady-state energy expenditure conditions. That is, researchers eliminate the periods of time in which energy expenditure is not in steady state before training and testing the algorithms. Steady-state is usually defined as a coefficient of variation (CV) of less or equal to 5 or 10% computed over windows of 1-5mins [47]. Another common practice is to ignore 30 to 40% of the data located at the beginning and/or end of each activity to eliminate non-steady state conditions. This may be problematic because energy expenditure might not reach steadystate conditions during free-living. Figure 5-1a shows real energy expenditure data collected from subject MIT-001 performing 32 activities at a gymnasium. It can be seen that energy expenditure reaches steady state at the end of most periodic activities such as walking, running, cycling, and rowing, but it never reaches steady state for some physically demanding activities such ascending stairs and calisthenics activities. Figure 5-1b shows the same energy expenditure data with non-steady state periods eliminated (30% of the beginning of each activity eliminated) and Figure 5-1c the same data concatenated. What is more, Figure 5-1c also shows the data corresponding to *walking* and *running* at different speeds highlighted in red, which would normally be the only data used to train and test energy expenditure estimation algorithms. Finally, Figure 5-1d shows the energy expenditure data collected over more naturalistic conditions while the same subject performs 19 activities in a real home for three hours. It can be clearly seen that energy expenditure almost never reaches the steady state condition.



Figure 5-1: Energy expenditure data collected using the Cosmed K4b2 indirect calorimeter from subject MIT-001. (a) data collected at the gymnasium from 32 activities (b) same gym data with non-steady-state data eliminated and treadmill activities highlighted in red, (c) same data when non-steady-state data has been eliminated and the remaining data has been concatenated (note that in this plot, the time axis has a different scale than for plot a and b) and (d) energy expenditure data collected in a real home while performing 19 activities.

#### **5.2.3** Comparison Challenges

Comparing the performance of energy expenditure estimation algorithms is even more difficult than comparing activity recognition results for three reasons: (1) There is less agreement in the literature on which performance measures to use while reporting results as explained in Section 5.2.1, (2) energy expenditure can be predicted as net energy or gross energy expenditure, and (3) the results can be reported in different units (e.g. Kcal/min, METs).

Energy expenditure can be estimated either as net energy expenditure or gross energy. gross energy expenditure is the total energy expended by the body at any given time and is composed of resting energy expenditure and kinetic energy expenditure (due to body motion). Net energy expenditure corresponds only to the kinetic energy or energy spent during actual motion. Thus, it can be estimated from gross energy by subtracting the resting metabolic rate (resting energy). Consequently, energy expenditure estimation algorithms that estimate net energy cannot be directly compared to algorithms that estimate gross energy expenditure easily unless resting metabolic rate is known. Another factor that complicates comparison across algorithms is that energy expenditure can be predicted and reported in different units. For example, most algorithms that include subject characteristics (e.g. gender, weight, and age) in their regression equations to compensate for variations across subjects report energy expenditure in Kcal/Min. Other algorithms try to reduce the impact of differences across subjects by predicting energy expenditure in units that are normalized with respect to one or some subject characteristics (e.g. weight) such as METs (35 ml of oxygen per kg per min). Conversion between values therefore requires knowledge of subject characteristics that are usually reported using summarization statistics over groups of subjects (e.g. means and standard deviations) in the literature.

Table 5-54 shows a summary of the most recent state-of-the-art work in energy expenditure estimation algorithms. From the results column in this table, it can be seen how difficult it is to compare results across different pieces of work. Some results are presented using different performance measures such as SEE, MAED, MAE, RMSE, r,  $r^2$ , and reported in different units such as METs, Kcal/min, and Kcal. Furthermore, some of these pieces of work do not provide the necessary data to convert among different units. Consequently, this work will primarily compare the results obtained by experimenting with different regression algorithms, feature sets, and sliding window lengths incrementally over the dataset collected for this work during the development of the final algorithm. When possible, results will be compared against results obtained in previous work, but this will be avoided as much as possible given the difficulty of doing so as explained in this section.

# **5.2.4 Comparison Strategy in this Work**

In this work, the energy expenditure estimation algorithms will be evaluated across all the performance dimensions discussed in Sections 5.2.1 and 5.2.2. In addition, the following error measures will be used while evaluating the energy expenditure estimation algorithms: the root mean squared error or standard error of the estimate (RMSE or SEE), Pearl's correlation coefficient (r), mean absolute error (MAE), and maximum absolute

error deviation (MAED). Appendix B1 presents a brief description of each of these measures and the formulas for computing them. Special attention will be put in improving the energy expenditure estimation for some activities whose energy expenditure estimation has proven difficult in previous work and that are important for the medical community. Some examples of these activities include lower body activities such as *cycling* and *sitting fidgeting feet and legs*, and upper body activities such as *bicep curls*, *bench weight lifting, doing dishes, washing windows,* and *scrubbing surfaces*. The experiments are performed incrementally, where parameters are selected (e.g. regression algorithm, feature set, sliding window length) based on improvement over prior results on the same dataset. The goal of this systematic procedure is to converge on a system that is balanced in all dimensions.

This work will compare the performance of accelerometers and a Polar heart rate monitor to determine if one modality or the combined modalities provide the best performance. Recent work [152] suggests that energy expenditure estimation can be improved by (1) computing features over the raw accelerometer data and (2) utilizing non-linear regression models such as neural networks to perform the estimation. However, an open question is which of these two steps contributes the most to the improvement of energy expenditure estimation. Thus, this work will compute a large set of features over the accelerometer and heart rate data and evaluate the performance over subsets of those features using linear regression and non-linear regression algorithms. The work will also evaluate the performance of different subsets of accelerometers worn at different body locations to find a reasonable compromise set of sensors that enable good EE estimation without undue subject burden. Most previous work on energy expenditure estimation provides estimates over sliding windows of one minute. Intuitively, one might expect that better results and smaller real-time estimation delays can be obtained by utilizing smaller window lengths [34, 180]. As a result, this work will evaluate the impact of estimating energy expenditure over shorter windows of time to determine if they provide better performance and shorter estimation delays. State-of-theart work in estimating energy expenditure [34, 96] also suggests that activity dependent regression models might be required to improve energy expenditure estimation. Therefore, this work will evaluate the performance of estimating energy expenditure by creating regression models that depend on the activity being performed. Section 5.6.1 will later explore some of the latest results obtained in recent work when accelerometer and heart rate data is utilized to estimate energy expenditure. The section will also discuss what might be considered a good result with respect to prior work.

# **5.3 Training Procedures and Assumptions**

This section presents the training and testing procedures used to evaluate the performance of the activity recognition and energy expenditure algorithms in this work. The section also explains some of the assumptions made during the evaluation of the algorithms.

# **5.3.1 Subject Dependent Evaluation**

One way to test algorithms on relatively small datasets of examples is using 10-fold stratified cross-validation [226]. For subject dependent evaluation, cross validation was performed over each subject's data and the results obtained were averaged over all the subjects. This training/testing technique gives an idea of how well the algorithm recognizes activities or estimate energy expenditure when subjects provide 2-4min of example data for each activity to train the algorithm. The use of 10-fold stratified cross-validation was chosen because previous work suggests that 10 is a reasonable number of folds to use to evaluate the performance classifiers and regression algorithms [226]. This technique is known as individual calibration (IC) in the medical literature.

# 5.3.2 Subject Independent Evaluation

Ideally, an algorithm that recognizes activities or estimates energy expenditure would not require training data from a particular individual; instead it would use a corpus of training data acquired in advance from other individuals. To evaluate if subject-independent training is possible, the algorithms were trained with the data of all the subjects but one and tested the performance on the left-out subject. This procedure was repeated for all the subjects and the results were averaged. This technique is also known as group calibration (GC) in the medical community.

# 5.3.3 The Addition of the Garbage or Unknown Class

It is a common practice in the activity recognition field to include an additional class called the *garbage, zero* or *unknown* class to the activities or classes of interest that contains all the periods of time with no associated labels during the data collection. The addition of this class is believed to provide a more realistic evaluation of the activity recognition algorithms because the classifier has to discriminate between the actual activities of interest and periods of time where the subject performs activities that are not relevant. In some cases, however, when the classes or activities of interest are mutually exclusive, the addition of the garbage class damages the classification performance because it can include examples of the activities of interest that were not labeled during the data collection. In this work, in an attempt to evaluate the algorithms in the most realistic conditions, the garbage class was added to many experiments even though the set of activities explored in this work is mutually exclusive.

# **5.3.4 Transitions Between Activities**

Transitions between activities are particularly problematic when (1) utilizing heart rate data to recognize activities and (2) when estimating energy expenditure. The reason is that heart rate and energy expenditure are not stable during the beginning and end of activities. This is particularly true for physically demanding activities such as *ascending stairs* and performing *sit-ups*, where heart rate and energy expenditure keep increasing over time and steady-state conditions are almost never reached. Moreover, heart rate and

energy expenditure can remain altered for relatively long periods of time once physically demanding activities have been finished. As a result, there is high variability in the features computed over heart rate data that might be difficult to model by the recognition algorithms. Similarly, *standing still* might exhibit different levels of energy expenditure depending if someone was previously *sitting* on a chair or *running* on a treadmill at 5mph. In this work, the accelerometer and heart rate data associated with non-steady-state energy expenditure were not eliminated during the training and testing of energy expenditure estimation algorithms in an effort to make the evaluation as realistic as possible.

# **5.3.5** Assumption Made During the Estimation of Energy Expenditure

This work makes the assumption that energy expenditure may not reach steady state for a particular activity. In other words, segments of data were ground truth energy expenditure did not reach steady state are not eliminated from the training data. This assumption is opposite and more realistic than the steady-state assumption normally made by the medical community while estimating energy expenditure. Using this assumption, the energy expenditure algorithms presented in this work are tested over worst case scenario conditions. Furthermore, if non-steady state data were to be eliminated, performance of the algorithms presented would most likely increase due to a reduction in the complexity of the data. The algorithms developed to estimate energy expenditure predict gross energy expenditure. This means that the algorithms developed produce energy expenditure estimates for sedentary activities involving no motion. That is, the energy expenditure algorithms estimate resting metabolic rate (energy expenditure values close to  $\sim$ 1MET) when no motion is observed for a given activity. Finally, energy expenditure is predicted in METs and not in Kcal/min or other unit of energy. This is because MET units reduce inter-individual variations by normalizing energy expenditure with respect to the weight of the participants.

# **5.4 Activity Recognition algorithm Experiments**

In general, it is not possible to optimize all aspects (parameters) of an algorithm at once. Consequently, this section presents a set of systematic experiments to determine the classifier, signal processing techniques, sliding window length, and feature set to utilize in order to select a reasonable and practical set of parameters to achieve real-time performance. The section also presents experiments to determine the minimum set of sensors to use, the impact on recognition rates of adding heart rate monitoring, and where accelerometers they be worn on the human body to maximize usage comfort and performance. The experiments are organized so that each answers a relevant question about the algorithm parameters incrementally, starting from the most restrictive parameters (e.g. classifier, feature set) to the least restrictive parameters (sensor modality and location). In each case, decisions are made that will support both good recognition rates *and* real-time performance.

The experiments in general will be performed under a worse-case activity recognition scenario where the algorithm has to discriminate among the 51 activities contained in the MIT dataset. Furthermore, a garbage or unknown class is added containing all the unlabeled segments of data in the MIT dataset. The addition of this class is a common procedure in the activity recognition community and is mainly used to allow the classifier to distinguish between activities of interest and random activities that might be contained in a dataset or performed by the subject in real-life. However, the unknown class might contain examples of the activities of interest that were just not labeled during the data collection that might damage the classification performance. In the experiments presented in this section, the addition of the unknown class has the sole purpose of evaluating the recognition algorithms under worse-case scenario conditions. In summary, the experiments performed can be assumed to be worse-case activity recognition conditions due to the following facts: (1) The large number of activities (52) to recognize, (2) the inclusion of 26 activities with different intensity levels, (3) the inclusion of 18 household activities containing examples of unconstrained motion, (4) the inclusion of a garbage class (unknown activity) containing all the time periods with no associated activity labels, and (5) the attempt to recognize all activities in a subject independent manner (which is more challenging that recognizing activities in a subject dependent manner).

The results presented in this section will be also clustered under activity categories that are helpful while analyzing the results. These activity categories are: Postures, ambulation, exercise, resistance exercise, and household activities. All the activities explored in this work are assumed to be mutually exclusive.

Finally, all the experiments presented in this section utilize the following signal preprocessing and segmentation strategies:

### 5.4.1 Cubic Spline Interpolation

Since wireless accelerometer are used to collect the human motion information, acceleration samples (sensor values) can be lost during wireless transmission (from the sensing nodes on the body to the wireless receiver) due to environmental noise or body blocking of the signals (i.e. blocking of signal due to the high water content of the human body). Consequently, interpolation of the signal is required to fill out the missing sensor values (samples). This work utilizes cubic spline interpolation to fill out samples lost during wireless transmission when the number of total samples lost is less than 20% of the number of samples contained in a given window. If the number of samples lost is greater than 20%, the total window of data is discarded since there is not enough information to interpolate the signal reliably. The threshold of 20% was chosen after visualizing the results of interpolating the signal when different percentages of samples are lost. The signal quality after interpolation when 20% of the samples are lost is of reasonable quality. Larger thresholds tend to distort the signal shape and quality.

#### 5.4.2 Band-pass and Low-pass Filtering of Accelerometer Signals

Once the accelerometer signals are interpolated, their information is separated into motion and posture information by applying a band-pass filter between the frequencies of

Band-Pass Chebyshev Type I IIR Filter	Value	Low-Pass Equiripple FIR Filter	Value		
Parameters		Parameters			
Sampling Frequency (Fs)	45	Sampling Frequency (Fs)	45		
First Stopband Frequency	0.1	Filter Order 10			
First Passband Frequency	0.2	Passband Frequency (Fpass) 1			
Second Passband Frequency	18	Stopband Frequency (Fstop) 5			
Second Stopband Frequency	20	Passband Weight (Wpass) 1			
First Stopband Attenuation (dB)	60	Stopband Weight (Wstop) 1			
Second Stopband Attenuation (dB)	80	Density factor (dens) 20			
Passband Ripple (dB)	1				

Table 5-2: Parameters used in the design of the Band-Pass Chebyshev Type I infinite impulse response (IIR) Filter designed to filter the accelerometer signals between 0.1 and 20Hz and the equiripple low-pass finite impulse response filter (FIR) designed to filter the accelerometer signals below 1Hz. All frequency values are in Hz.

0.1 to 20Hz and a low-pass filter with a cutoff frequency of 1Hz respectively. Band-pass filtering the signal between 0.1 and 20Hz has to goals: (1) to eliminate the static acceleration or DC component of the signal that captures posture information about the orientation of the sensor with respect to ground (<0.1Hz) and (2) to eliminate the signal components generated by non-human motion and high frequency noise (>20Hz). Lowpass filtering at a cutoff frequency of 1Hz has the opposite purpose: To eliminate most of the signal generated by dynamic human motion and to preserve the information generated by static human motion or posture information. Again, this information is represented by the DC component of the accelerometer signal. The band-pass filter applied is a Chebyshev Type I infinite impulse response filter (IIR) designed using the CHEBY1 MATLAB [227] function and also re-implemented in Java. The low-pass filter applied is an equiripple finite impulse response filter (FIR) designed using the FIRPM function in MATLAB and re-implemented in Java. Table 5-2 shows the design parameters of the filters. The reason why motion and posture information are separated by applying these filters is that different types of features will be computed over these two signals to better capture the motion and posture information. Appendix A3 presents the list of features explored in this work and a brief explanation on how they are computed. The prefix "AC" in the features indicates they were computed over the band-pass filtered accelerometer signal and the prefix "DC" indicates that they were computed over the low-pass filtered accelerometer signal.

# 5.4.3 Non-overlapping Sliding Windows Segmentation

After the accelerometer signals have been interpolated and filtered, they are segmented into non-overlapping sliding windows. This segmentation consists in simply partitioning the acceleration signal into consecutive windows of fixed length. For most experiments, the window length used is 5.6s since it is the optimal window length found as explained in Section 5.4.6. The use of non-overlapping sliding windows was preferred over 50% overlapping windows, a common applied segmentation technique while recognizing activities from accelerometer data, because of its lower computational requirements since half of the classifications are required for real-time performance.

Classifier	Classifier Description	Parameters	Ref
Nearest Neighbor (NN)	A memory-based classifier with a non-linear decision boundary that classifies examples based on their similarity with the training data. New data points are classified by simply returning the class of the closest training point.	Nearest neighbors classifier with no distance weighing, the Euclidean distance as similarity function, and brute-force linear search of neighbors.	[228]
Naïve Bayesian (NB)	A generative probabilistic classifier with a linear decision boundary. Experimental testing has demonstrated that naive Bayes networks are surprisingly good classifiers on some problem domains, despite their strict independence assumptions between attributes and the class.	Naïve Bayes classifier using a one dimensional continuous Gaussian distribution ( $\mu$ , $\sigma$ ) per feature per class. Prior distributions are learned from the data.	[229]
LogitBoost	A Boosting algorithm based on the log likelihood-loss and Newton optimization. LogitBoost is a statistical version of AdaBoost, of one of the most successful boosting algorithms, that has been found to perform better in more realistic and noisier datasets. LogitBoost produces more accurate classifiers from a sequence of base weak classifiers learned iteratively from reweighed versions of the original data. During each iteration, the weight of instances difficult to classify is increased. Consequently, weak classifiers learned later in the process focus their classification effort on examples that have been found difficult to classify in previous iterations.	LogitBoost with decision stumps (decision trees with two leaf nodes) as base classifiers, with 10 boosting iterations, shrinkage set to 1.0, a weight threshold of 100, likelihood threshold of -1.79E308 and no resampling.	[215]
C4.5	A state-of-the-art decision tree classifier with non-linear decision boundaries that are parallel to the features axis. The algorithm grows the decision tree one node at a time by selecting the best attribute to split on based on the information gain criterion. Once the tree is built, instances are classified by transversing the tree from the root to the corresponding leaf node that contains the example classification.	C4.5 decision tree classifier using pruning with a confidence (certainty) factor of 25% (0.25), subtree rising when pruning, minimum of two instances per leaf, and no Laplace smoothing on leaves.	[112]

Table 5-3: Brief description of the classifiers explored in this section, their parameters used during the experiments presented in this section and references to papers describing them in full detail.

# 5.4.4 Can Fast Run-Time Classifiers Produce Acceptable Performance?

This section explores if classifiers amenable for real-time performance due to their fast training and classification times can achieve similar results than other popular or state-of-the-art classifiers with longer training and classification times. The Weka toolkit [226] was used to compare the performance of the following classifiers: The nearest neighbor classifier (NN) [228], the Naïve Bayesian (NB) classifier [229], the LogitBoost classifier [215], and the C4.5 decision tree classifier [112]. The NN classifier was chosen because it is one of the oldest and most widely used algorithms in existence. The NB classifier because it is a probabilistic classifier that has been found to perform extremely well in a variety of realistic datasets despite its simplicity, the LogitBoost (leading boosting algorithm) in a variety of classifiers due to its fast classification, high performance, and the interpretability of the classifiers due to its fast classification, high performance, and the interpretability of these classifiers and the values of their parameters used during the experiments.

The performance and computational requirements of the classifiers shown in Table 5-4 is tested under two extreme conditions: (1) a best-case scenario where a small set of features is needed to recognize the activities of interest, and (2) a worse-case scenario where a large set of features is required to recognize the activities. In the best case scenario, the classifiers were trained and tested using the *ACAbsArea* feature set computed over each of the three axis (x, y, and z) of the seven accelerometers giving a

	Feature Set	NN	NB	LogitBoost	C4.5
Total training time	ACAbsArea	0.03s	0.8s	457s	54s
(Average time per instance)		(7.8ms)	(0.02ms)	(11.1ms)	(1.3ms)
Total classification time	ACAbsArea	24s	1.3s	0.09s	0.14s
(Average time per instance)		(11.3ms)	(0.6ms)	(0.04ms)	(0.06ms)
Total training time	MaxAcceleration	0.8s	58s	33000s	1662s
(Average time per instance)		(0.01ms)	(1.4ms)	(816.7ms)	(40.6ms)
Total classification time	MaxAcceleration	2823s	53s	0.16s	0.17s
(Average time per instance)		(1319.0ms)	(25.1ms)	(0.07ms)	(0.08ms)

Table 5-4: The total training time and classification times (in seconds) required to train the classifiers using the data from subjects MIT-002 to MIT-020 (19 subjects) and classify the activity examples of subject MIT-001. The features used are the *ACAbsArea* (21) and *MaxAcceleration* (247) computed over sliding windows of 5.6s.

total of 21 features (Features used, such as *ACAbsAreas* are fully described in Appendix A3). In the worse-case scenario, the classifiers where evaluated by computing the following set of features per acceleration axis: *ACAbsArea, DCArea, ACTotalAbsArea, DCMean, ACAbsMean, DCTotalMean, ACTotalSVM, ACRange, ACSegmentalforce, ACTotalSegmentalForce, ACVar, ACAbsCV, ACEnergy, ACEntropy, ACFFTPeaks, ACDomFreqRatio, ACBandEnergy, ACModVigEnergy, ACLowEnergy, ACCorr, ACKur, ACSkew, ACMCR, ACQ1, ACQ2, ACQ3, ACIQR, and ACPitch.* This feature set is referred as the *MaxAcceleration* set and consists of a total of 247 features. Appendix A3 presents the description of each of these features and how to compute them.

Features are computed over sliding windows of 5.6s in length after interpolating and band-pass and low pass filtering the raw accelerometer signals as explained in Section 5.4.2. Section 5.4.6 will later explain why this window length is a good one to use. The classifiers are attempting to discriminate among the 52 activities contained in the MIT dataset (including the *unknown* class) in a subject dependent and independent manner. This allows to testing the performance and computational requirements of the classifiers in a worse-case activity recognition scenario, as explained in the introduction.

Since the main goal of this section is to identify the classifiers more amenable for realtime performance, the section starts by discussing the training and testing time requirements of the classifiers. Table 5-4 presents the training and classification times (in milliseconds) required by the classifiers when recognizing the 52 activities contained in the MIT dataset using *ACAbsArea* and the *MaxAccelerationSet* feature sets in a subject dependent and independent manner using a 1GHz Intel core microprocessor. The total training time shown is the time required to train the classifiers using the data from subjects MIT-002 to MIT-020 (19 subjects) and the total classification time is the time required to classify the activity examples of subject MIT-001. The average training and classification time per activity example is shown in parenthesis.

From Table 5-4, it can be seen that the NN classifier achieves the lowest training times since training only requires the computation of a matrix containing the distances among all training examples. The classification times for the NN classifier however, are the longest ones from all the classifiers explored because every single example to classify has to be compared against all the examples available in the training data. When computing the *MaxAcceleration*, the NN classifier requires a classification time of 1.3s that might be unacceptable in some activity recognition applications. What is more, in order to work, the NN classifier requires storing in memory all the training examples available (46,216 in the MIT dataset). This might be a limitation in handheld devices with limited RAM

memory such as existing mobile phones. Clearly in its current implementation, the NN algorithm is not too amenable for real-time performance in small mobile devices. The NB classifier has the second fastest training times from all the classifiers since training involves only computing the sample means and standard deviations over each feature per class. Its classification times are midrange, since they are 52 times faster than the NN classifier but 335 times slower than the classification times of the LogitBoost and C4.5 classifiers when using the MaxAcceleration feature set. The classification model generated by the NB classifier requires to store in memory the parameters of a univariate Gaussian distribution for each of the features and for each of the classes. When using the MaxAcceleration feature set, the number of parameters is 25,688. Furthermore, classification of new examples using the NB classifier requires the evaluation of 12,844 Gaussian distributions, a task that might prove difficult to achieve in low-processing power devices. The LogitBoost and C4.5 decision tree classifiers exhibit the fastest classification times from all the classifiers explored. This is because classification involves only the evaluation of simple if-then rules generated by the decision stumps in the case of the LogitBoost classifier and the decision tree in the case of the C4.5 classifier. One important disadvantage of the LogitBoost over the C4.5 classifier is its extremely long training times. The LogitBoost classifier took 9.3hours to train while the C4.5 classifier only 27.7min when using the *MaxAcceleration* feature set. This is because training the LogitBoost classifier requires the training of weak classifiers (decision stumps) in repeated iterations (10 in this experiment) to create the final ensemble of classifiers. Long training times might be particularly unacceptable if subject dependent training is required to recognize activities, since end-users might not be able to test the performance of the classifier immediately after the system has been trained. It is important to remember that the training and testing times presented in this section for the *MaxAcceleration* feature set is a worse-case scenario where the largest and most complex feature set is utilized (all accelerometer-based features shown in Appendix A3). Training and testing times will obviously reduce as the number of features is reduced. Section 5.5 will later show that real time performance can be achieved by utilizing a subset of the most discriminant features contained in the MaxAcceleration set.

In summary, the two classifiers more amenable for real-time performance are (1) the C4.5 decision tree classifier and the (2) NB classifier. The C4.5 classifier is a good choice when classifying activities from low-processing power hand-held devices due to its fast classification times, mid-size classification models, and medium range training times. The NB classifier is a good option when low training times are particularly important and processing requirements are available to evaluate the Gaussian functions required by the algorithm in real-time. Now, it is necessary to compare the performance of the C4.5 and NB classifiers against those obtained by the NN and LogitBoost classifiers.

Table 5-5 and Table 5-6 present the performance of evaluating the classifiers using the *MaxAcceleration* feature set using subject dependent and independent training respectively over the MIT dataset. First, it can be seen that the performance of subject dependent training is considerably higher (~81-88%) than the performance obtained using subject independent training (~33-59%). The probability of random guessing in this scenario is 1.9% for 52 activities. The best overall performance in both cases is achieved using the LogitBoost classifier. This is because weak classifiers learned in the final iterations of boosting concentrate their classification efforts on instances that are difficult

Activity Category	NN	NB	LogitBoost	C4.5
All (accuracy)	$82.5 \pm 2.1$	$83.1 \pm 2.1$	$88.2 \pm 1.4$	$81.7 \pm 1.5$
Postures	91.9±6.3	89.2±6.8	95.1±4.7	92.2±5.8
	(0.0±0.0)	(0.1±0.22)	(0.0±0.0)	(0.1±0.0)
Ambulation	88.4±8.4	93.8±6.3	88.9±7.3	84.4±9.1
	(0.2±0.1)	(0.1±0.1)	(0.1±0.0)	(0.2±0.1)
Exercise	83.9±11.5	90.9±9.2	90.6±8.5	88.6±9.9
	(0.2±0.1)	(0.1±0.1)	(0.0±0.0)	(0.15±0.1)
Resistance	78.5±13.8 88.7±10.0		87.0±9.6	84.8±10.7
Exercise	(0.3±0.2)	(0.2±0.2)	(0.1±0.0)	(0.2±0.1)
Household	84.0±9.7	84.1±7.7	82.2±9.6	75.1±9.9
	(0.4±0.3)	(0.5±0.4)	(0.1±0.1)	(0.4±0.2)
Unknown	$70.55 \pm 7.03$	$63.53 \pm 8.43$	86.89 ± 3.94	$74.81 \pm 5.54$
	$(3.7 \pm 1.1)$	$(3.5 \pm 0.7)$	$(7.71 \pm 1.38)$	$(7.63 \pm 1.35)$

Table 5-5: True positive rate and false positive rate (shown in parenthesis) clustered per activity category while classifying the 52 activities contained in the MIT dataset in a subject dependent manner. Classifiers were trained by computing the *MaxAcceleration* (247) feature set per accelerometer axis over sliding windows of 5.6s.

Activity Category	NN	NB	LogitBoost	C4.5
All (accuracy)	$48.9 \pm 5.1$	$33.9 \pm 4.5$	$59.59 \pm 4.43$	$49.4\pm4.9$
Postures	50.7±16.2	36.3±14.2	71.9±28.2	66.8±31.3
	(0.3±0.2)	(1.6±0.7)	(0.2±0.3)	(0.2±0.3)
Ambulation	44.7±26.7	54.6±31.6	54.3±30.1	41.4±26.0
	(0.7±0.6)	$(1.2\pm1.1)$	(0.7±0.7)	$(0.8\pm0.8)$
Exercise	44.5±24.8	48.2±29.8	51.8±31.5	39.8±33.6
	(0.6±0.4)	(0.8±0.7)	(0.6±0.6)	(0.5±0.6)
Resistance	34.1±21.5	34.5±28.6	40.1±31.5	31.5±29.6
Exercise	(0.7±0.5)	(1.1±0.9)	(0.8±0.8)	(0.8±0.7)
Household	45.3±19.6	32.1±18.8	49.7±25.2	37.9±24.4
	(1.3±0.7)	(1.6±0.8)	(0.6±0.6)	(0.9±0.7)
Unknown	$48.2\pm5.6$	$11.7 \pm 5.3$	$73.5\pm5.2$	$65.3\pm6.6$
	(13.0 + 4.0)	(2.4 + 2.8)	(17.8 + 7.2)	(24.0 + 7.2)

Table 5-6: True positive rate and false positive rate (shown in parenthesis) clustered per activity category while classifying the 52 activities contained in the MIT dataset in a subject independent manner. Classifiers were trained by computing the *MaxAcceleration* (247) feature set per axis over sliding windows of 5.6s.

to classify in previous iterations, giving the classifier and advantage when handling activity examples difficult to classify. The overall performance of NN, NB and C4.5 is quite comparable during subject dependent training (see Section 5.3.1 and 5.3.2 for a description of subject dependent vs. independent training procedures). One important difference of the NB classifier is that it tends to classify ambulation, exercise, resistance exercise and household activities slightly better than the other classifiers. However, its performance while classifying the unknown activity is the worst of all the classifiers during both, subject dependent and independent training. One explanation might be that the NB classifier is recognizing more unlabeled activity examples contained in the unknown class as real activities of interest that the other classifiers. This can be seen from the large number of false positives generated by the other classifiers for the unknown class, since they classify the other activities of interest as the unknown class. This can be a side effect of the large number of examples contained in the unknown class with respect to the number of examples contained in other classes (Appendix A4 shows the number of training examples available per class). The NB classifier might be able to recognize the activities of interest better with respect to the *unknown* class given the high probability seen by the evidence (features) for the other activities despite the high prior probability assigned to the *unknown* class. The other classifiers, on the contrary, decide to classify some of the examples of the activities of interest as belonging to the *unknown* class. The NB classifier also has more problems when recognizing postures as compared to the other classifiers. This is because it confuses most postures with the lying down posture given that this posture has a higher number of examples with respect to the other postures.

During subject independent training, the NB classifier also has a lower recognition rate for postures and for the unknown class with respect to the other classifiers for the same reasons previously explained. It also tends to recognize ambulation, exercise and resistance exercise slightly better than the C4.5 decision tree classifier. The overall performance of the C4.5 classifier is the second highest from all the classifiers and is comparable to the performance obtained with the NN classifier. Interestingly, the C4.5 classifier presents the largest false positive rate for the unknown class, reflecting the fact that it might be influenced by the large number of training examples available for this class. Finally, the lowest recognition performance from all the activity categories during subject independent training is obtained for resistance exercise activities. These activities are difficult to discriminate from accelerometer data since most of them do not involve changes in the motion patterns of the activity, but changes in the effort required to perform the activities. Interestingly, this low performance is not so obvious during subject dependent training for this category. It is possible that different subjects perform slight changes in the motion patterns of an activity when different resistance levels are used that the classifiers are able to successfully learn to discriminate among the activities.

In order to better understand the performance differences among classifiers, figures were generated to graphically highlight the per class performance across classifiers. Figures Figure 5-2 through Figure 5-5 show the performance per class for the classifiers while using subject dependent and independent training respectively. The figures show the performance per class as a grayscale image normalized with respect to the lowest and highest performance across all classifiers. For example, in Figure 5-2, the maximum true positive rate of 99.7% is represented by the color white and the minimum of 45.7% by the color black. With this coding scheme, areas of poor performance can be easily identified as dark (black) areas in the image.

Figure 5-2 shows the true positive rate per activity for all the classifiers as evaluated using subject dependent training. From the image it is easy to see that the classifier with best performance per class is the LogitBoost classifier (since it has lighter colors overall), followed by the NB classifier. As discussed previously, the NB classifier has difficulties recognizing the *unknown* class and outperforms the LogitBoost and C4.5 classifiers while recognizing household activities. One possible explanation for this behavior is that the Gaussian distributions used by the NB classifier to represent features are able to better learn distinctive motion patterns per subject such as the circular motion of hands while *scrubbing a surface*, or while *washing windows* that are not learned by the other classifiers due to the high motion variability in these activities.

Figure 5-3 shows the false positive rate per activity as a grayscale image for all the classifiers. Similarly, poor areas of performance are identified by dark areas. It can be seen that largest number of false positives (7.63%) is generated by the LogitBoost and C4.5 classifiers for the *unknown* class. Again, this is because some examples of real



Figure 5-2: True positive rate per activity for all the classifiers using the *MaxAcceleration* feature set and subject dependent evaluation. The grayscale image is scaled so that the maximum true positive rate of 99.7% is represented by the color white and the minimum of 45.7% by the color black. In other words, poor areas of performance are shown in black.



Figure 5-3: False positive rate per activity for all the classifiers using the *MaxAcceleration* feature set and subject dependent evaluation. The grayscale image is scaled so that the minimum false positive rate of 0% is represented by the color white and the maximum of 7.63% by the color black. In other words, poor areas of performance are shown in black.



Figure 5-4: True positive rate per activity for all the classifiers using the *MaxAcceleration* feature set and subject independent evaluation. The grayscale image is scaled so that the maximum true positive rate of 96.7% is represented by the color white and the minimum of 4.5% by the color black. In other words, poor areas of performance are shown in black.



Figure 5-5: False positive rate per activity for all the classifiers using the *MaxAcceleration* feature set and subject independent evaluation. The grayscale image is scaled so that the minimum false positive rate of 0.01% is represented by the color white and the maximum of 24% by the color black. In other words, poor areas of performance are shown in black.

activities are classified as the *unknown* class. The *unknown* class contains examples of the other activities of interest that were just not labeled during the data collections. The C4.5 classifier also generates a larger number of false positives for household activities when compared to the NB classifier. However, the number of false positives for the NB

classifier is concentrated in the *unknown* class. In general, the NN classifier presents a high number of false positives for resistance exercise activities such as *bench weight lifting* and *bicep curls*.

Figures Figure 5-4 and Figure 5-5 show the true positive rate and false positive rate per activity for all the classifiers using the MaxAcceleration feature set and subject independent evaluation. These figures highlight three main differences between the NB and the other classifiers: (1) the NB classifier tends to recognize some periodic activities such as ascending stairs, descending stairs, walking at 2mph, and cycling at 60 and 100rpm better than the other classifiers. This is also true for household activities including periodic motion such as scrubbing a surface, stacking groceries, sweeping, vacuuming, and writing. This is perhaps because the Gaussian distributions capture the periodicity of the motion across different subjects better than the non-linear decision boundaries used by the other classifiers. (2) In activities involving similar motion patterns but different resistance levels (e.g. light, moderate, hard), the classifier tends to classify better the activity with the medium resistance level (moderate). This is because activities with moderate intensity represent the average amount of motion for all the activity intensities. The other classifiers, on the contrary, learn to discriminate equally well among all the different intensity levels of the same activity. (3) Finally, the false positive rate in the NB classifier tends to be more uniformly distributed across all the activities. In contrast, the other classifiers tend to concentrate the number of false positives in a single class: the unknown class.

Since one of the main differences between the NB and the C4.5 classifiers seems to be the recognition performance over the *unknown* class, a new set of experiments was performed to compare these classifiers when the *unknown* class is left out. Table 5-7 shows the performance of these classifiers during subject dependent and independent training. The feature set used to train the classifiers was the *MaxAcceleration* set computed over windows of 5.6s in length. It can be seen that the overall performance of the NB classifier is slightly better during both subject dependent and independent training. During subject dependent training, the C4.5 classifier recognizes postures and exercise activities slightly better than the NB classifier. The NB classifier, on the other hand, classifies ambulation and household activities slightly better than the C4.5 classifier during subject independent training, the same conditions are true except that now, the NB classifier also outperforms the C4.5 decision tree classifier during exercise and resistance exercise activities.

To better compare the performance of the NB and the C4.5 decision tree classifier, grayscale images comparing their performance per class were again generated. These images are shown in Figures Figure 5-6 through Figure 5-9. Figure 5-10 and Figure 5-11 show the confusion matrices for subject dependent and independent training using the *MaxAcceleration* feature set.

In general, as shown by figures Figure 5-6 and Figure 5-7, both classifiers have difficulties recognizing household activities during subject dependent training. For example, the classifiers confuse activities involving standing and walking such as *making the bed* and *taking out trash*. They also confuse activities involving standing and upper body motion such as *washing windows* and *wiping/dusting*. The classifiers also confuse activities involving walking and upper body motion such as *sweeping* and *mopping*. In summary, the NB classifier has more problems in recognizing *sweeping, wiping/dusting*,

	Subject D	ependent	Subject Independent		
Activity	NB	C4.5	NB	C4.5	
Category					
All (accuracy)	$90.6\pm1.8$	$89.4\pm2.2$	$57.8 \pm 4.4$	$53.3\pm5.3$	
Postures	90.9±6.1	95.9±4.4	72.4±17.1	79.5±26.1	
	(0.1±0.1)	(0.1±0.1)	(1.0±0.5)	(0.3±0.4)	
Ambulation	94.7±5.8	90.9±7.1	64.7±27.8	49.0±29.1	
	(0.1±0.1)	(0.2±0.1)	$(0.9\pm1.0)$	(1.2±1.1)	
Exercise	91.9±8.9	93.2±7.6	53.9±27.0	49.6±32.4	
	(0.1±0.1)	(0.1±0.1)	(0.8±0.7)	(0.9±0.9)	
Resistance	89.7±9.7	89.8±8.8	43.3±26.7	36.0±31.9	
Exercise	$(0.2\pm0.2)$	$(0.2\pm0.2)$	(1.0±0.8)	(1.3±1.1)	
Household	86.2±7.5	82.9±8.4	49.7±23.2	50.4±24.5	
	$(0.4\pm0.3)$	$(0.4\pm0.2)$	$(0.9\pm0.6)$	$(1.2\pm0.9)$	

Table 5-7: True positive rate and false positive rate (shown in parenthesis) clustered per activity category while classifying the 51 activities contained in the MIT dataset using the NB and C4.5 classifier without the *unknown* class. Classifiers were trained by computing the *MaxAcceleration* (247) feature set per axis over sliding windows of 5.6s.



Figure 5-6: Comparison of true positive rate per activity for the NB and C4.5 Classifiers using the *MaxAccelerationSet1* without the *Unknown* Class evaluated using subject dependent training. The grayscale image is scaled so that the maximum true positive rate of 99.9% is represented by white and the minimum of 67.7% by black.



Figure 5-7: Comparison of false positive rate per activity for the NB and C4.5 Classifiers using the *MaxAccelerationSet1* without the *Unknown* Class evaluated using subject dependent training. The grayscale image is scaled so that the minimum false positive rate of 0.0% is represented by white and the maximum of 1.1% by black.



Figure 5-8: Comparison of true positive rate per activity for the NB and C4.5 Classifiers using the *MaxAccelerationSet1* without the *Unknown* Class evaluated using subject dependent training. The grayscale image is scaled so that the maximum true positive rate of 98.8% is represented by white and the minimum of 11.3% by black.



Figure 5-9: Comparison of false positive rate per activity for the NB and C4.5 Classifiers using the *MaxAccelerationSet1* without the *Unknown* Class evaluated using subject dependent training. The grayscale image is scaled so that the minimum false positive rate of 0.0% is represented by white and the maximum of 4.1% by black.

*taking out trash*, and *bench weight lifting* (hard and moderate). The C4.5 Classifier presents more difficulties recognizing *making the bed*, *wiping/dusting*, *taking out trash*, *moping* and *sweeping*. Figure 5-7 also shows that the number of false positives is more evenly distributed among all the classes that in the NB classifier. In the NB classifier, the number of false positives in concentrated in the *washing windows* activity.

During subject independent training, the performance of both classifiers as shown by Figure 5-8 is not easy to compare. Both classifiers are good at recognizing some sets of activities while bad at recognizing others. The NB classifier has more difficulties recognizing *bench weight lifting* light, *bicep curls* (light and hard), *rowing* moderate, and *watching TV* than the C4.5 classifier. At the same time, the C4.5 classifier has more difficulties recognizing *gardening*, and *walking* activities in general than the NB classifier. Figure 5-8 and Figure 5-9 show that the NB classifier is particularly bad at recognizing postures. For example, it confuses activities involving different postures such as *watching TV*, *playing video games*, *kneeling*, and *sitting* with *lying down*.

In summary, the overall performance of the NB classifier is slightly better during subject dependent (1.3%) and independent training (4.5%) than the one of the C4.5 classifier when the *unknown* class is not included. However, both classifiers have strengths and weaknesses. For example, the NB classifier seems is good at recognizing periodic activities but is bad at recognizing postures. The C4.5 classifier is good at recognizing postures, and its performance over activities with different intensity levels is more uniformly distributed than when using the NB classifier. When the *unknown* class is eliminated, the C4.5 classifier has a false positive rate per activity more evenly distributed across all activities than the NB classifier. When taking into account the classification time, the C4.5 classifier has an advantage since it can classify new examples 312 times faster than the NB classifier. The NB classifier, in contrast, can be



Figure 5-10: Confusion matrices for the NB and C4.5 classifier without the *unknown* class using subject dependent evaluation and the *MaxAcceleration* feature set.



Figure 5-11: Confusion matrices for the NB and C4.5 classifier without the *unknown* class using subject independent evaluation and the *MaxAcceleration* feature set.

trained 28 times faster than the C4.5 classifier. Taking into account all the advantages and disadvantages of both classifiers, this work will utilize the C4.5 decision tree classifier as the final classification algorithm for the remainder of this thesis.

Lastly, the answer to the question posed by this section is yes. Fast run time classifiers such as the C 4.5 decision tree classifier and the NB classifier can achieve performances similar to the ones obtained by popular classifiers such as the NN classifier and state-of-the-art classifiers such as the LogitBoost Classifier. Upcoming sections will perform experiments to determine the set of parameters (e.g. feature set and window length) required to make the training process of the C4.5 classifier more amenable for real time performance without sacrificing recognition performance too much.

#### 5.4.5 Can Features Computed Over Each Sensor Reduce Computational Requirements Without Sacrificing Performance?

In the previous section, as in most previous work, features were computed over each of the axis (x, y, and z) of the accelerometer signal. Nevertheless, computing features over each axis adds substantial computational load, particularly when complex features are computed over a large number of sensors. For example, if the FFT coefficients are being computed as features when using three triaxial accelerometers, the FFT algorithm would have to be run nine times (3 accelerometers times 3 axes) every time a new example has to be classified. As a result, it is worth exploring if it is possible to compute features per sensor and achieve similar performance as when computing features per axis. If this is in fact possible, at least a threefold improvement in computational performance would be obtained in the feature computation step. Further computational improvements would also be obtained during training and classification since fewer features are required.

Features can be computed per sensor by first summarizing the overall acceleration (motion) experienced by each sensor in all axes. This can be done by computing the sum of the acceleration over all axes (x, y, and z) sample by sample. Another technique is to compute the signal vector magnitude (SVM) over all axes. However, the SVM is computationally expensive since it requires the computation of square and root square operations. The formulas for the two techniques are shown in Table 5-8.

Sum	Sum = x + y + z
Signal Vector Magnitude	$SVM = \sqrt{x^2 + y^2 + z^2}$

Table 5-8: Two techniques used to summarize the acceleration (motion) experienced by a sensor: The sum over all acceleration axes (x, y, and z) and the signal vector magnitude.

In this section, the performance of computing features per sensor is evaluated by summarizing the motion experienced by each sensor using the sum operation. The experiments compare the performance of the NB and C4.5 classifiers when features are computed per axis and per sensor using the *MaxAcceleration* feature set over sliding windows of 5.6s. The classifiers are trying to recognize the 51 activities in the MIT dataset without including the *unknown* class. The *unknown* class was not added during these experiments to maximize the interpretability of the results per class. The experiments are evaluated using subject dependent and independent training.

		Subject Dependent Evaluation				
Activity Category	NB Per Sensor	NB Per Axis	Change in Performance	C4.5 Per Sensor	C4.5 Per Axis	Change in Performance
All (accuracy)	$88.5\pm3.6$	$90.6\pm1.8$	-2.16	$87.3\pm3.8$	$89.4 \pm 2.2$	-2.06
Postures	94.7±4.4	90.9±6.1	3.8	96.5±3.9	95.9±4.4	0.6
	(0.0±0.1)	(0.1±0.1)	(-0.1)	(0.1±0.1)	(0.1±0.1)	(0)
Ambulation	91.8±6.3	94.7±5.8	-2.9	87.4±8.2	90.9±7.1	-3.5
	$(0.2\pm 0.2)$	(0.1±0.1)	(0.1)	(0.4±0.2)	(0.2±0.1)	(0.2)
Exercise	90.0±9.2	91.9±8.9	-1.9	91.9±7.6	93.2±7.6	-1.3
	$(0.2\pm 0.2)$	(0.1±0.1)	(0.1)	(0.2±0.2)	(0.1±0.1)	(0.1)
Resistance	85.9±10.9	89.7±9.7	-3.8	86.4±9.3	89.8±8.8	-3.4
Exercise	(0.4±0.3)	(0.2±0.2)	(0.2)	(0.3±0.2)	(0.2±0.2)	(0.1)
Household	81.6±9.8	86.2±7.5	-4.6	80.0±9.8	82.9±8.4	-2.9
	(0.6+0.4)	(0.4+0.3)	(0.2)	(0.6+0.4)	(0.4+0.2)	(0.2)

Table 5-9: True positive rate, false positive rate (shown in parenthesis) and change in performance while computing features per sensor and per axis clustered per activity category while classifying the 51 activities contained in the MIT dataset using the NB and C4.5 classifier without the *unknown* class. Classifiers were trained by computing the *MaxAccelerationSet1* (247 over sliding windows of 5.6s using subject dependent training.

	Subject Independent Evaluation					
Activity	NB	NB	Change in	C4.5	C4.5	Change in
Category	Per Sensor	Per Axis	Performance	Per Sensor	Per Axis	Performance
All (accuracy)	$47.6\pm6.0$	$57.9 \pm 4.4$	-10.23	$45.5\pm7.7$	$53.3 \pm 5.3$	-7.88
Postures	42.7±22.0	72.4±17.1	-29.7	58.7±38.4	79.5±26.1	-20.8
	(1.4±0.8)	(1.0±0.5)	(0.4)	(0.7±1.0)	(0.3±0.4)	(0.4)
Ambulation	47.8±34.8	64.7±27.8	-16.9	37.8±26.6	49.0±29.1	-11.2
	(1.3±1.4)	$(0.9 \pm 1.0)$	(0.4)	(1.4±1.6)	(1.2±1.1)	(0.2)
Exercise	42.2±30.8	53.9±27.0	-11.7	39.1±28.5	49.6±32.4	-10.5
	$(1.1\pm1.1)$	(0.8±0.7)	(0.3)	(1.2±1.5)	(0.9±0.9)	(0.3)
Resistance	30.9±27.5	43.3±26.7	-12.4	25.9±25.3	36.0±31.9	-10.1
Exercise	(1.3±1.2)	(1.0±0.8)	(0.3)	(1.6±1.8)	(1.3±1.1)	(0.3)
Household	32.5±24.1	49.7±23.2	-17.2	32.3±26.8	50.4±24.5	-18.1
	$(1.2 \pm 1.0)$	(0.9±0.6)	(0.3)	(1.6±1.5)	(1.2±0.9)	(0.4)

Table 5-10: True positive rate, false positive rate (shown in parenthesis) and change in performance while computing features per sensor and per axis clustered per activity category while classifying the 51 activities contained in the MIT dataset using the NB and C4.5 classifier without the *unknown* class. Classifiers were trained by computing the *MaxAccelerationSet1* (247 over sliding windows of 5.6s using subject independent training.

Table 5-9 presents a comparison of the performance of the NB and C4.5 classifiers while computing features per sensor and per axis using subject dependent training. It can be seen that the overall decrease in performance for both classifiers is very close and corresponds to a decrease of approximately 2%. The decrease in performance for the NB classifier is larger for household activities (-4.6%) and resistance exercise activities (-3.8%). One explanation for the performance decrease in resistance exercise activities is that subjects perform characteristic motions in several axes while struggling with different load or resistance levels. An interesting result is that the performance for postures increases when computing features per sensor for both classifiers during subject dependent training. It seems that computing features per axis while recognizing postures in a subject dependent manner introduces more variability in the features leading to a decrease in performance. Moreover, the smallest decrease in performance for both classifiers occurs for exercise activities. Exercise activities include activities with very characteristic periodic motions differentiated by the use of particular limbs such as *bicep*


Figure 5-12: True positive rate per activity using the *MaxAcceleration* feature set computed per sensor and per axis and evaluated using subject dependent training with the NB and C4.5 classifiers. The grayscale image is scaled so that the maximum true positive rate of 100% is represented by white and the minimum of 60.3% by black. In other words, poor areas of performance are shown in black.

*curls, cycling, and calisthenics* activities (*crunches, sit-ups*) and ambulation (*walking* and *running*). Consequently, they can be well differentiated without analyzing the motion per axis.

Table 5-10 presents the performance while computing features per sensor and per axis for both classifiers using subject independent training. The overall decrease in performance of approximately 9% for both classifiers is larger than the decrease observed during subject dependent training. The decrease for the NB classifier is 10% and for the C4.5 classifier is 7.8% (slightly lower). In contrast to subject dependent training, it can be seen that a decrease in performance occurs for all the activity categories. The most dramatic decrease occurs for postures (29% for NB and 20% for C4.5) and household activities (17% for NB and 18% for C4.5). It can be concluded that analyzing motion per axis is more important during subject independent training since this extra information compensates for the high variability found in the performance of activities across subjects. The smallest decrease in performance occurs again for exercise activities because they are mainly discriminated by analyzing the motion of different body limbs moving at different speeds.

In an effort to better identify the differences per activity while computing features per sensor and per axis, images were generated to highlight the differences in performance



Figure 5-13: False positive rate per activity using the *MaxAcceleration* feature set computed per sensor and per axis and evaluated using subject dependent training with the NB and C4.5 classifiers. The grayscale image is scaled so that the minimum false positive rate of 0.0% is represented by white and the maximum of 1.9% by black. In other words, poor areas of performance are shown in black.

per activity. The images are grayscale images normalized so that the worse performance per activity is shown in black and the best performance per activity is shown in white.

Table 5-10 shows the true positive rate for both classifiers when features are computed per sensor and per axis using subject dependent training. One of the first differences one might notice is that both classifiers are having difficulties recognizing *wiping/dusting*. When inspecting the confusion matrices for the C4.5 classifier shown in Figure 5-16 and Figure 5-17, it can be seen that this activity is being confused with *ironing*, *doing dishes*, and *washing windows*. These are all activities involving the standing posture and upper body motion. The reason why the classifiers are even able to differentiate (to some extent) between these activities when motion is analyzed per sensor is the computation of features that capture posture information such as the DCAreas and DCMeans features (these features are explained in Appendix A3). For example, the classifiers are able to differentiate between wiping/dusting and washing windows due to the different posture of the dominant arm during the activities. Analyzing motion by axis helps in discriminating between these activities better as can be observed from Figure 5-12 and Figure 5-13. For the same reasons, the C4.5 classifier also confuses washing windows with wiping/dusting and *scrubbing a surface*. The activities least affected by not analyzing motion per axis are activities involving periodic motion and movement of distinctive limbs such as *cycling*, calisthenics, walking and running.



Figure 5-14: True positive rate per activity using the *MaxAcceleration* feature set computed per sensor and per axis and evaluated using subject independent training with the NB and C4.5 classifiers. The grayscale image is scaled so that the maximum true positive rate of 98.8% is represented by white and the minimum of 4.2% by black. In other words, poor areas of performance are shown in black.

Figure 5-13 presents the false positive rate per activity for both classifiers. According to the figure, the number of false positives for the NB classifier is concentrated in the *washing windows* activity since it is confused with other activities involving standing and upper body motion as previously explained. In general, it can be seen that analyzing motion per axis decreases the number of false positives per class.

Figure 5-14 and Figure 5-15 show the true positive and false positive rate per class when evaluating the classifiers in a subject independent manner. In general, the performance per class follows the same behavior as the one observed during subject dependent training. However, the performance of some activities involving different resistance levels such as *cycling*, *walking* (at different inclinations), and *bicep curl* is dramatically affected by not computing features per axis in both classifiers. As explained before, subjects might perform motions in different axis while struggling with different loads that might help in the discrimination between these activities. In addition, the performance on some activities with different postures such as standing, playing video games and weeding is also highly affected for the NB classifier. In the previous section, it was explained that the NB classifier has problems recognizing postures, and it seems that



Figure 5-15: False positive rate per activity using the *MaxAcceleration* feature set computed per sensor and per axis and evaluated using subject dependent training with the NB and C4.5 classifiers. The grayscale image is scaled so that the minimum false positive rate of 0.0% is represented by white and the maximum of 5.9% by black. In other words, poor areas of performance are shown in black.

computing features per axis helps this classifier to better differentiate between activities involving different postures. When analyzing the confusion matrix for the C4.5 classifier during subject independent training (shown in Figure 5-17a), it can also be observed that this classifier is confusing activities involving different postures such as *watching TV*, *playing video games*, and *typing* when features are computed per sensor. Furthermore, it can also be seen that the C4.5 classifier also confuses *walking* at different speeds with *running*.

Finally, Figure 5-14 shows the true positive rate for both classifiers when subject independent training is used. Again, the true positive rate per class tends to be higher when features are computed per axis. It can also be seen that the C4.5 classifier has a false positive rate per class more evenly distributed across all activities and that the false positive rate for the NB classifier is higher for the activity *lying down*.

In summary, performance is affected less during subject dependent training than during subject independent training when features are computed per sensor. The decrease in performance with respect to feature computation per axis was ~2% for subject dependent training and ~9% for subject independent training. Using both training methods, the activities most affected where the ones involving similar postures (e.g. *watching TV* and *playing video games*), similar motion of limbs (e.g. *wiping/dusting* vs. *ironing*), and



Figure 5-16: Confusion matrices comparing the performance of the C4.5 classifier when the *MaxAcceleration* feature set is computed per sensor and per axis using subject dependent training.



Figure 5-17: Confusion matrices comparing the performance of the C4.5 classifier when the *MaxAcceleration* feature set is computed per sensor and per axis using subject independent training.

activities involving different resistance levels (e.g. bicep curls moderate vs. hard). The activities least affected by feature computation per sensor were activities involving characteristic motion of a particular limb (e.g. cycling vs. running) or activities involving motion at different speeds (e.g. running at 5mph vs. 6mph). Consequently, feature computation per sensor can be used with only a 2% loss in recognition performance with subject dependent training, particularly when the activities of interest can be differentiated by the use of a particular limb, or motion at a particular speed without a considerable decrease in performance. From this point on, results will be presented utilizing feature computation per axis to better differentiate between the activities of interest and show the best possible performance during subject independent evaluation. While this is more computationally expensive, it might improve activity recognition performance in latter sections that analyze other parameters of the activity recognition algorithm. Finally, at three-fold increase in computational performance can be achieved by switching from feature computation per axis to feature computation per sensor at expense of some decrease in the performance for some activities if the amount of processing power is limited in some applications (e.g. in real-time applications implemented to run on mobile phones).

One possible criticism of the experiments performed in this section is that the analysis was performed using a large set of features (*MaxAcceleration* feature set). This might be a problem because the training and testing data might not be enough to reliably evaluate the difference in performance between feature computation per sensor and per axis. This is less of a problem for the C4.5 classifier since it performs feature selection during the training process and only utilizes the set of most discriminant features in the models generated. However, future work is required to analyze the difference between feature computation per sensor and per axis over subsets of accelerometer features to see if the results presented in this section hold when the number of features is decreased (at least for the naïve Bayes classifier).

## 5.4.6 What is the Optimal Sliding Window Length to Use?

Acceleration signals need to be broken down into windows of specific lengths so that features summarizing the mathematical properties of the signals can be computed. Selecting an appropriate sliding window length is a trade-off between the quality of some of the features computed and the real-time classification delay. The quality or resolution of some features such as the FFT transformation and the Pearson's correlation coefficients strongly depend on the window length used. Usually, the longer the window length, the better these features can be estimated. On the other hand, the longer the window length, the longer the end-user of an activity recognition system has to wait for a recognition result. Another trade-off involved in selecting the window length is the ability to capture the motion patterns associated with the activities of interest. Long windows might introduce additional variability in the motion patterns observed for some highly unconstrained activities (e.g. household activities) while very short windows might not be able to capture the fundamental variability that characterizes a given activity (e.g. periodicity of motion during *walking* or *running*).

In this section, the most appropriate sliding window length is determined by measuring the performance of the C4.5 decision tree classifier over the two features whose quality is

Number of Samples	Time (s)
64	1.4
128	2.8
256	5.6
512	11.3
1024	22.7
2048	45.5
4098	91

Table 5-11: Window lengths explored while selecting the optimal window length to use for the activity recognition algorithm. The length is shown in number of acceleration samples and corresponding time in seconds, assuming a sampling rate of 45Hz.

most likely to vary with the window length: The FFT Peaks (*ACFFTPeaks*) and the Pearson's correlation coefficients (*ACCorr*). As a baseline, the performance is also computed utilizing the *ACAbsAreas* feature. The features will be first computed over window lengths ranging from 1.4 to 91 seconds (64 -2048 accelerometer samples) using feature computation per sensor, and later on a reduced set of window lengths using feature computation per axis. This procedure is followed to minimize the time and number of experiments to run. Window lengths shorter that 1.4s were not considered because they are intuitively too short to capture the repetitive motion patterns found in some periodic activities such as *walking* slowly at 2mph. Similarly, window lengths longer than 91s were not considered due to the extremely long real-time classification delay they introduce into the system.

The length of the sliding windows is constrained to be a power of two by the algorithms required to compute the Fourier and Wavelet transformations efficiently. Table 5-11 shows the windows lengths explored in this section in number of acceleration samples and corresponding time in seconds assuming a sampling rate of 45Hz. Figure 5-18 and Figure 5-19 present the true positive rate per activity category while evaluating the performance of the C4.5 decision tree classifier over different window lengths using the *ACAbsAreas* and *FFTCorr* feature set (*ACFFTPeaks* and *ACCorr* features) computed per sensor during subject dependent and independent training. Appendix A5 presents the same results in a tabular form.

The figures illustrate that during subject dependent training, the true positive rate for all activities reaches a maximum at a window length of 22.7s for the ACAbsArea feature and 5.6s for the *FFTCorr* feature set. After these maxima, the true positive rate for all the activities starts declining sharply. This is because increasing the window length reduces the number of activity examples available for training. For example, increasing the window length from 5.6s to 45s reduces the number of training examples per activity by a factor of 8. Since the MIT dataset contains between 1 and 4.5min of data for each activity, a window length of 45s reduces the number of training examples to be only between 1 and 6 per activity. In general, it can be observed that the performance for most activities increases as the window length increases. This makes sense since the quality or resolution of features such as the FFT and the correlation coefficients usually increases with increasing window lengths. The increase in performance when longer window lengths are used is more dramatic for household and ambulation activities when using the ACAbsArea feature and for household and resistance exercise activities while using the FFTCorr feature set. In fact, the performance over household activities keeps increasing even when the performance over all the activities has started decreasing for the *FFTCorr* 



Figure 5-18: True positive rate per activity category when computing the *ACAbsAreas* feature per sensor over sliding windows of varying lengths using the C4.5 classifier during (a) subject dependent and (b) independent evaluation.



Figure 5-19: True positive rate per activity category when computing the *FFTCorr* feature set (*ACFFTPeaks* and *ACCorr* features) per sensor over sliding windows of varying lengths using the C4.5 classifier during (a) subject dependent and (b) independent evaluation.



Figure 5-20: True positive rate per activity using the *FFTCorr* feature set computed per sensor and evaluated using subject independent training with the C4.5 classifier. The grayscale image is scaled so that the maximum true positive rate of 100% is represented by white and the minimum of 0% by black. In other words, poor areas of performance are shown in black.

feature set. This indicates, as one might expect, that longer window lengths (e.g. 22-45s) are necessary to capture the high motion variability found in household activities.

The only activity category whose performance does not increase with longer window lengths is postures. The performance over postures reaches a maximum at a window length of 2.8s when the *ACAbsArea* feature is used and at 5.6s when the *FFTCorr* feature set is used. This is because longer window lengths introduce more motion variability that can be thought as noise for the static nature of the posture information represented by the DC level or static component of acceleration signals.

During subject independent training, the performance over all activities increases with increasing window lengths until a window length of 44.5s is reached in both feature sets (*ACAbsArea* and *FFTCorr*). After a window length of 44.5s, the performance decreases for ambulatory activities when the *ACAbsArea* feature is used and for resistance exercise activities when the *FFTCorr* feature set is used. After inspecting the performance per activity, one of the major problems of using long window lengths was found: Poor performance over short duration activities. For example, physically demanding activities such as *bench weight lifting* and *bicep curls* (included in the resistance exercise category) were only performed for less than a minute. Consequently, during training, the C4.5 classifier has few or no training examples for some of these activities.



Figure 5-21: True positive rate per activity category when computing the *FFTCorr* feature set (*ACFFTPeaks* and *ACCorr* features) per axis over sliding windows of varying lengths using the C4.5 classifier during (a) subject dependent and (b) independent evaluation.

To better understand this problem, a grayscale image was created to highlight the difference in performance per class using the *FFTCorr* feature set and subject independent training.

The image (shown in Figure 5-20) presents the best true positive rate of 100% as white and the worse of 0% as black. This image also shows that the performance over short duration activities such as ascending and descending stairs, and physically demanding activities performed for short periods of time such as *crunches* and *sit-ups* gets progressively worse as the window length is increased. This limitation of long window lengths is of practical concern, since activities shorter than the window length will be most likely merged with activities of longer duration performed before or after the short duration activity or confused with other activities by the recognition algorithm. For example, ascending stairs for 20 seconds after having walked for three minutes will be most likely recognized as walking if sliding window lengths longer than 30 or 40s are used. Figure 5-20 also shows that the performance for some periodic activities such as cycling, walking and running gets progressively better with longer window lengths. As explained before, this is because their periodicity is better captured by the FFT transformation as its resolution increases with increasing window lengths. Finally, the reason why the performance for most activities (including postures) increases with increasing window lengths during subject independent evaluation is that differences in the way subjects performed the activities are smoothed out when longer windows of time are used, leading to an improved performance. The performance also increases because the number of training examples is decreased (to some extent although there are still enough training examples during subject independent evaluation) as longer window lengths are used, lowering the chances of misclassification. Unlike the subject dependent scenario, here, the performance for most activities keeps increasing, even after window lengths of 91s. This is because during subject independent training the data of all the subjects minus one is used for training and enough training data exists to train the algorithm using long window lengths. Figure 5-21 shows the true positive rate for different window lengths when the FFTCorr feature set is computed per axis and the performance is evaluated in a subject dependent and independent manner using the C4.5 classifier. From the figure, it is clear that feature computation per axis exhibits the same performance behavior observed for feature computation per sensor in Figure 5-18 and Figure 5-19. The figure also shows that during subject dependent training a window length of 5.6s represents the optimal choice for recognizing postures and near the optimal choice for overall performance. During subject independent training, the performance over all activities and activity categories also increases with increasing window lengths as observed before.

In conclusion, one of the main findings of this section is that the optimal window length to utilize depends on the activity being recognized. This finding is in agreement with prior work by [119] that also found that the best window length to utilize depends on the activity being recognized. However, utilizing one window length per activity is computationally expensive since all the features used need to be re-computed over every different window length utilized. Thus, this work will perform activity recognition using only a single window for all activities. After analyzing the performance of different window lengths during subject dependent and independent evaluation, it has been determined that the best single window length to use in this work is 5.6s. This is because

the performance of this window length using the *ACAbsArea* and *FFTCorr* feature sets is either near optimal (*ACAbsArea*) or optimal (*FFTCorr*) during subject dependent training. Furthermore, this short window length allows optimal performance while recognizing postures during subject dependent evaluation. This short window of 5.6s length also allows the accurate recognition of short duration activities such as *bench weight lifting*, *bicep curls*, *ascending and descending stairs*, and intense calisthenics activities such as *crunches* and *sit-ups*. More importantly, the classification delay introduced by this window length is short enough to allow fast interventions at the point of decision that could be triggered almost as soon as the activity of interest is recognized. One obvious disadvantage of such short duration window is an expected lower performance on some household activities with high motion variability such as *making the bed*, *taking out trash*, *stacking groceries*, *gardening* and *weeding*. Intuitively, the high variability in motion presented by these activities would be better captured using long windows since these activities can be performed in different ways from day to day by the same individual and naturally by different individuals under different household layouts.

## 5.4.7 Which Features Provide the Most Discriminatory Information and Have the Lowest Computational Cost?

Features are computed over acceleration signals to capture specific motion characteristics (e.g. magnitude, periodicity, spectral content) that might allow better discrimination between the activities of interest. Nevertheless, it is important to carefully consider the type and the number of features used since some of them might not provide enough discriminant information and others might be too computationally expensive to achieve real-time performance.

Therefore, this section presents experiments to determine the subset of features (from the ones shown in Appendix A3) that discriminate better between the activities of interest and that have the lowest computational cost. This is performed by first ordering the features according to their importance (information they provide) using the information gain criterion. This technique is known as information gain feature selection and will be performed when computing features per sensor and per axis over windows of 5.6s in length (a good compromise window length as reported in section 5.4.3). These experiments are performed in a best case scenario where all the seven accelerometers are used for recognizing activities. Once subsets of potentially high discriminating features are found, their performance will be evaluated using the C4.5 decision tree classifier in a subject dependent and independent manner. As in previous experiments, the classifier attempts to discriminate between the 52 activities contained in the MIT dataset, including the unknown class. Finally, once the best feature subsets are identified their performance will be evaluated utilizing only some subsets of the sensors (accelerometers). This will help to identify the feature set that provides the most information when only few sensors are used to recognize activities.

A complete list of all the features explored in this section is shown in Appendix A3. These features consist on a superset of features used in previous work that have shown good performance as well as some new features not explored before. Table 5-12 presents a list of the features and a brief explanation of the information they attempt to capture. Features are computed after preprocessing the acceleration signals to better differentiate

Information captured by the features	Features
Measures of body posture	Mean (DCMean), mean over all axes or sensors (DCTotalMean), area under signal
	(DCArea), and mean distances between axes (DCPostureDist).
Measures of motion shape	Means over absolute value of signal (ACAbsMean), area under absolute value of
	signal (ACAbsArea), area under absolute value of signal summarizing overall
	motion (ACTotalAbsArea), mean of signal vector magnitudes over all samples
	(ACTotalSVM), entropy of the signal (ACEntropy), skewness of the signal
	(ACSkew), kurtosis of the signal (ACKur), and quartiles of the signal (ACQ1,
	ACQ2, ACQ3).
Measures of motion variation	Variance of the signal (ACVar), Coefficient of variation of the absolute value of
	the signal (ACAbsCV), inter-quartile range of signal (ACIQR), and range of the
	signal (ACRange).
Measures of motion spectral content	Fast Fourier transform coefficients ( <i>ACFFTCoeff</i> ), Fast Fourier Transform Peaks
	(ACFFTPeaks), and fast Wavelet transform coefficients (FWTCoeff).
Measures of motion energy	Total energy of signal (ACEnergy), energy between frequencies 0.3-3.5Hz
	(ACBandEnergy), energy between frequencies 0-0.7Hz (ACLowEnergy), energy
	between frequencies 0.71-10Hz (ACModVigEnergy), heart rate above resting heart
	rate ( <i>HRAboveRest</i> ), and heart rate scaled between resting heart rate and heart rate
	while running at 5mph on a treadmill ( <i>ScaledHR</i> ).
Measure of trend in physical activity	The slope of the regression line computed over the heart rate data ( <i>HR1rend</i> )
Measures of motion periodicity	Pitch of the signal ( <i>ACPitch</i> ), ratio of energy in dominant frequency and energy in
	the other bands of the spectrum (ACDomFreqRatio), mean crossing rate of the
	signal (ACMCR).
Measures of motion similarity across body	Correlation coefficients among all accelerometer signals (ACCorr).
segments	
Measures of force employed per body	Segmental force computed by multiplying the sum over the absolute value of the
segment	accelerometer signal by the mass of the corresponding body segment (ACSF), sum
Manager of autient days doubt	of all the segmental force for all body segments (ACTotalSF).
Measures of subject characteristics	Gender, age, neight, and weight.
Measure of subject fitness	Body mass index (BMI), fat percentage (FatPercent), heart rate reserve (HRR).

 Table 5-12: Features explored in this work and the information they capture about motion or physical activity. A complete description of the features can be found in Appendix A3.

Information Ga	in Per Sensor	Information Gain Per Axis		
Feature (1-14)	Feature(15-28)	Feature (1-14)	Feature (15-28)	
ACTotalSF (1)	ACModVigEnergy (7)	ACTotalAbsArea (1)	DCMean (21)	
ACTotalAbsArea (1)	DCMean (7)	ACTotalSVM (1)	ACLowEnergy (21)	
ACTotalSVM (1)	DCPostureDist (21)	ACAbsArea (21)	DCPostureDist (21)	
ACSF (5)	ACMCR (7)	ACAbsMean (21)	ACModVigEnergy (21)	
ACAbsArea (7)	ACLowEnergy (7)	ACIQR (21)	ACFFTPeaks (210)	
ACAbsMean (7)	ACFFTPeaks (70)	ACQ3 (21)	ACDomFreqRatio (21)	
ACIQR (7)	ACDomFreqRatio (7)	ACQ1 (21)	ACBandEnergy (21)	
ACQ3 (7)	ACBandEnergy (7)	ACRange (21)	ACQ2 (21)	
ACQ1 (7)	ACQ2 (7)	ACPitch (21)	ACEnergy (21)	
DCTotalMean (1)	ACEnergy (7)	DCTotalMean (1)	ACAbsCV (21)	
ACPitch (7)	ACAbsCV (7)	ACMCR (21)	ACKur (21)	
ACRange (7)	ACKur (7)	ACEntropy (21)	ACCorr (210)	
ACEntropy (7)	ACCorr (21)	ACVar (21)	ACSkew (21)	
ACVar(7)	ACSkew (7)	DCArea (21)		
DCArea (7)				

Table 5-13: Acceleration features in decreasing order of importance according to the information gain criterion for (a) feature computation per sensor and (b) feature computation per axis. Features were computed over sliding windows of 5.6s over the MIT dataset. Feature 1 is the most important.

between motion information and posture information. The features intended to capture motion information are computed over the accelerometer signals after applying a bandpass filter between the frequencies of 0.1 and 20Hz. It has been shown in prior work that human acceleration has amplitudes below 12G and frequencies below 20Hz for most activities [218]. This preprocessing has two goals: (1) eliminate the DC or static component of the acceleration signal due to the orientation of the sensors with respect to ground (posture information) and (2) filter high frequency noise and motion not

Information captured by the features Features						
Measures of body posture	DCTotalMean, DCArea, DCMean, and DCPostureDist.					
Measures of motion shape	ACTotalAbsArea, ACTotalSVM, ACAbsArea, ACAbsMean, ACQ3, ACQ					
	ACEntropy, ACQ2, ACKur, and ACSkew.					
Measures of motion variation	ACIQR, ACRange, ACVar, and ACAbsCV,					
Measures of motion spectral content	ACFFTPeaks					
Measures of motion energy	ACLowEnergy or ACModVigEnergy, ACBandEnergy, and ACEnergy,					
Measures of motion periodicity	ACPitch, ACMCR, and ACDomFreqRatio.					
Measures of motion similarity across body	ACCorr					
segments						
Measures of force employed per body	ACTotalSF and ACSF.					
segment						

Table 5-14: Features ordered in decreasing order of importance (from left to right) according to the information gain criterion clustered according to the information they capture from the accelerometer signals.

generated by the human body. The features intended to capture posture information are computed over the accelerometer signals after low-pass filtering them at a cut-off frequency of 1Hz. This has the purpose of eliminating most of the signal information due to body motion and preserving the information due to static acceleration or posture. Features that capture motion information start with the prefix "AC" and those that capture posture information start with the prefix "DC". Table 5-13 shows the acceleration features in decreasing order of importance according to the information gain criterion when features are computed over all the accelerometers per sensor and per axis respectively. The first thing to notice is that the ordering of the features in both tables is almost identical, suggesting that the importance of the features does not change significantly when features are computed per sensor and per axis. Overall, features that capture information about the motion of activities (starting with prefix "AC") are ranked with higher importance than features that capture information about the activity posture (starting with prefix "DC"). This is because most of the activities contained in the dataset are activities that can be differentiated just by analyzing their distinctive motion signatures. Furthermore, the top ranking features are the ones that capture information about the overall motion of the human body while performing an activity such as the ACTotalSF, ACTotalAbsArea, and ACTotalSVM. This is because these features are computed over signals that summarize the overall motion (acceleration) experienced by the human body while performing the activities of interest. The reason why the ACTotalSF feature is outperforming the ACTotalAbsArea feature is that the ACTotalSF feature multiplies the acceleration signal per body segment by the mass of the body segment, thus, including additional information as to which body segment is generating the motion. Interestingly, the tables also show that some of the most discriminant features such as the ACSF, ACAbsArea, and the ACAbsMean are some of the features with the lowest computational requirements. The column labeled as "Information Gain Per Axis" in Table 5-13 does not show the ACTotalSF and ACSF features because its computation does not depend on computing features per axis or per sensor by definition and where already shown in column "Information Gain Per Sensor" in Table 5-13.

After analyzing Table 5-13, the acceleration features where rearranged so that their importance is shown according to the information they capture. The rearranged features are shown in Table 5-14. In this table, the features are ordered according to their importance from the most important at the left to the least important at the right of the table. When generating Table 5-14, the ordering of the features importance matched

Features subsets	Evaluation	All Activities	Postures	Ambulation	Exercise	Resistance Exercise	Household
ACEnergy	Subject Dependent	41.9 ± 10.2	20.3±14.0 (1.3±0.5)	62.7±15.5 (0.8±0.4)	44.0±15.3 (1.0±0.4)	35.0±13.6 (1.2±0.4)	14.8±11.6 (1.3±0.6)
ACLowEnergy	Subject Dependent	38.8 ± 8.8	18.6±14.8 (1.5±0.7)	44.5±15.7 (1.1±0.4)	41.0±15.0 (0.9±0.4)	27.7±12.4 (1.1±0.5)	9.1±7.0 (1.2±0.6)
ACModVigEnergy	Subject Dependent	39.6 ± 9.4	19.2±14.2 (1.4±0.5)	47.0±13.7 (1.1±0.4)	44.9±15.1 (0.9±0.4)	31.7±12.9 (1.2±0.5)	9.7±7.2 (1.4±0.7)
ACBandEnergy	Subject Dependent	51.1 ± 7.9	35.7±17.1 (1.0±0.5)	56.0±15.9 (1.0±0.4)	57.7±17.9 (0.7±0.3)	42.5±17.4 (1.0±0.4)	27.1±13.8 (1.2±0.5)
ACEnergy	Subject Independent	27.9 ± 4.9	7.75±8.02 (1.25±0.51)	19.9±18.7 (1.2±0.9)	14.5±13.9 (0.8±0.5)	9.1±10.6 (0.9±0.5)	3.6±4.2 (0.8±0.4)
ACLowEnergy	Subject Independent	31.0 ± 5.6	12.80±8.43 (1.63±0.70)	15.7±12.5 (1.3±0.7)	14.4±11.4 (0.7±0.5)	9.1±8.7 (0.7±0.4)	1.6±2.2 (0.4±0.2)
ACModVigEnergy	Subject Independent	30.8 ± 4.8	9.7±8.16 (1.25±0.52)	20.9±17.0 (1.2±0.7)	20.8±15.1 (0.7±0.5)	12.1±10.6 (0.9±0.5)	3.1±3.5 (0.8±0.3)
ACBandEnergy	Subject Independent	34.1 ± 4.5	$18.4\pm5.6$ (1.00+0.4)	$20.6\pm17.1$ (1.2+0.72)	$16.9\pm16.5$ (0.7+0.5)	9.0±10.2 (0.8+0.5)	$7.8\pm7.6$ (0.8+0.4)

Table 5-15: Performance of the features that capture motion energy using the C4.5 classifier when features are computed per sensor over the MIT dataset evaluated in a subject dependent and independent manner.

perfectly for the two columns labeled as "Information Gain Per Sensor" and "Information Gain Per Axis" in Table 5-13 except for two of the features that capture the energy of motion: ACLowEnergy and ACModVigEnergy. The ACModVigEnergy shows higher ranking in column "Information Gain Per Sensor" but lower ranking in column "Information Gain Per Axis" with respect to the ACLowEnergy feature. Consequently, their importance cannot be disambiguated from these columns in Table 5-13. In order to disambiguate their importance, an experiment was run by measuring the performance of all the features that capture energy motion over the MIT dataset using the C4.5 classifier when features are computed per sensor using subject dependent and independent evaluation. The results are shown in Table 5-15. From the table, it can be seen that the most discriminant feature is the ACBandEnergy, followed by the ACEnergy or ACLowEnergy, and the ACModVigEnergy. Even when the ordering of importance for some of these features is not completely disambiguated by this experiment, one thing is clear: the best feature to use to capture the energy of motion is the ACBandEnergy feature. This is because this feature is computed over the frequencies in which the majority of human activity is believed to lie [218]. From Table 5-15, it can also be seen that the ACLowEnergy feature is good at discriminating sedentary or low energy activities such as postures while the ACModVigEnergy features is good at discriminating activities with higher motion energy such as ambulation and exercise activity, just as one might intuitively expect. One reason why the ACLowEnergy feature outperforms the ACEnergy and ACModVigEnergy features during subject independent evaluation is that this feature includes frequency zero or the DC component of the acceleration signal, thus, giving it an advantage over the other features while recognizing postures. However, this also makes the ACLowEnergy feature more dependent on the magnitude of the accelerometer signal. This dependency might be undesirable since the magnitude of the acceleration signal changes with slight variations in the way accelerometers (sensors) are worn.

Two features that were not included during the information gain feature selection experiment in Table 5-13 were the Fast Fourier transform coefficients (*ACFFTCoeff*) and

Features subsets	Evaluation	All Activities	Postures	Ambulation	Exercise	Resistance Exercise	Household
(Feature size)							
	Subject	$55.1 \pm 8.0$	45.9±15.8	60.2±14.2	61.4±16.5	53.2±16.0	33.3±12.2
	Dependent		(0.9±0.4)	(0.7±0.3)	(0.6±0.2)	(0.8±0.3)	(1.4±0.6)
ACFFTPeaks	Subject	$65.5\pm6.4$	54.1±14.7	73.6±11.2	73.1±12.5	64.1±14.4	49.6±15.2
(70)	Dependent		(0.7±0.3)	(0.6±0.2)	(0.4±0.2)	(0.6±0.3)	(1.0±0.4)
ACFWTCoeff	Subject	$28.7\pm6.4$	24.0±9.3	17.6±9.7	23.7±12.1	17.3±8.9	11.4±6.5
(1785)	Dependent		(1.4±0.4)	(1.5±0.5)	(1.4±0.4)	(1.6±0.5)	(1.9±0.7)
ACFFTCoeff	Subject	$36.8\pm3.8$	23.2±14.8	24.3±18.6	25.5±21.0	17.9±16.1	$14.2 \pm 11.9$
(889)	Independent		(0.9±0.4)	(1.0±0.7)	(0.7±0.5)	(0.9±0.6)	(1.2±0.5)
ACFFTPeaks	Subject	$41.6 \pm 4.3$	27.2±19.6	28.2±22.7	28.2±24.3	20.1±18.4	19.8±16.6
(70)	Independent		(0.8±0.5)	(0.9±0.7)	(0.6±0.6)	(0.8±0.6)	(1.0±0.6)
ACFWTCoeff	Subject	$26.6 \pm 3.5$	17.1±10.0	$14.9 \pm 11.0$	16.7±14.4	11.6±10.4	6.9±5.9
(1785)	Independent		(1.0±0.4)	(1.3±0.5)	(0.9±0.4)	(1.0±0.5)	(1.4±0.4)

Table 5-16: Performance of the features that capture the spectral content of motion computed per axis using the C4.5 classifier and evaluated in a subject dependent and independent manner.

Information captured by the features	Features
Measures of body posture	DCArea, DCMean, DCPostureDist, and DCTotalMean,,
Measures of motion shape	ACAbsArea, ACAbsMean, ACQ3, ACQ1, ACQ2, ACTotalAbsArea, ACTotalSF,
	ACTotalSVM A, ACEntropy, ACKur, and ACSkew.
Measures of motion variation	ACIQR, ACVar, ACRange, and ACAbsCV,
Measures of motion spectral content	ACFFTPeaks, ACFFTCoeff, and FWTCoeff.
Measures of motion energy	ACBandEnergy ACEnergy, ACLowEnergy, or ACModVigEnergy.
Measures of motion periodicity	ACPitch, ACMCR, and ACDomFreqRatio.
Measures of motion similarity across body	ACCorr
segments	
Measures of force employed per body	ACTotalSF and ACSF.
segment	

Table 5-17: Final ordering of the features according to their importance (decreasing order of importance from left to right) according to all the experiments performed in this section.

the Fast Wavelet transform coefficients (ACFWTCoeff). These features were not included in the experiment to prevent their large vector sizes (889 and 1785 respectively) to interfere with the ordering of the results. Therefore, a new experiment was run to determine which of the features that capture the motion spectral content is more discriminating. This was achieved by measuring the performance of these features computed per sensor over the MIT dataset in a subject dependent and independent manner. The results in Table 5-16 shows that the most discriminating feature is the ACFFTPeaks during both subject dependent and independent training. It can also be observed that the order of importance of the ACFFTCoeff and the ACFWTCoeff features corresponds to their inverse ordering with respect to their sizes. This indicates that their performance is low because of their large size and not because of the information they provide. In fact, the ACFFTPeaks feature summarizes just some of the information contained in the ACFFTCoeff feature. It is thus, clear that the large amount of information contained in the ACFFTCoeff and the ACFWTCoeff features needs to be summarized to prevent poor performance when the number of features is large with respect to the amount of training data available. One way to achieve this is to utilize feature extraction techniques such as linear discriminant analysis (LDA) or principal component analysis (PCA) to reduce the size of the features. However, this will not be explored in this work due to the high computational requirements of these techniques.

Finally, in order to corroborate the ordering of importance of the remaining features in Table 5-14, their individual performance was tested over the MIT dataset using feature

computation per sensor and subject independent evaluation. Appendix A6 presents the results obtained. The analysis shows that the order of importance for the features that capture the periodicity of motion (*ACPitch*, *ACMCR*, and *ACDomFreqRatio*) is the same as the one shown in Table 5-14. The ordering of the features that capture posture information also remains the same except for the *DCMean* and *DCArea* features. The experiment shows that the *DCMean* feature slightly outperforms the *DCArea* feature in capturing posture information (2% improvement). This is because the *DCArea* feature is more affected by motion than the *DCMean* feature. While the *DCArea* feature takes into account all the motion over a given window, the *DCMean* feature only computes the average value of the motion over the window, making it more robust to motion variability and noise. However, the *DCMean* feature is more computationally expensive than the *DCArea* feature since it requires the computation of averages. As a result, the order of importance of these features was not modified.

Finally, the appendix shows that the ordering for some of the features that capture information about the motion shape needs to be modified. Table 5-14 shows that the performance of features computed over signals summarizing the overall motion experienced by the whole body such the *ACTotalAbsArea* and the *ACTotalSVM* features have more importance than the same features computed over signals summarizing motion per sensor or axis such as the *ACAbsArea and ACAbsMean*. Appendix A6 on the contrary, shows that the *ACAbsArea and ACAbsMean* features are more informative. Intuitively, this makes sense since these features incorporate the additional information of which body segment is generating the motion. Consequently, the ordering of importance of all the accelerometer features explored in this work according to all the experiments performed in this section.

Once the order of importance of all the features with respect to the information they capture was determined, experiments were performed over different subsets of the most discriminant features of each category (information they capture) to determine the best performing ones. Five feature subsets were selected for presenting their results because one or more of the following reasons: Their performance was among the highest, they were fast to compute, they were more invariant to the magnitude of the acceleration signals, or because they could serve as a comparison baseline. These feature subsets were labeled as follows: (1) All features: This feature set includes all the features computed over the accelerometer data and serves as a baseline of the performance that can be achieved if all the features are used without regard to computational speed. (2) Fast to compute: This feature set includes the best performing features found over most categories that are fast to compute. (3) Fast to compute reduced: This set includes the two features that showed the best performance from all the features and were the easiest to compute. This set also serves as a baseline of the performance that can be obtained with the simplest possible features. (4) Invariant total: This set includes all the features that are more invariant to the magnitude of the acceleration signal. Invariance to the magnitude of the acceleration signal is important because it can change due to hardware differences among sensors or with differences in the positioning of the sensors on the body. (5) Invariant reduced: This feature set is a subset of the Invariant total feature set containing only the most discriminant features. The performance of these five feature

Features subsets	All Activities	Postures	Ambulation	Exercise	Resistance Exercise	Household
All Features:	$80.6\pm3.6$	89.3±7.3	81.6±9.0	87.2±10.1	81.9±11.6	72.2±10.4
MaxAcceleration		(0.1±0.1)	(0.3±0.1)	(0.2±0.1)	(0.3±0.1)	(0.5±0.3)
Fast to compute:	$81.5\pm3.5$	91.8±6.9	81.1±9.8	87.5±9.4	82.1±11.2	74.1±10.5
ACAbsArea, DCArea,		(0.1±0.1)	(0.4±0.2)	$(0.2\pm0.1)$	(0.3±0.2)	(0.5±0.3)
ACVar, ACRange,						
ACMCR						
Fast to compute reduced:	$82.2 \pm 3.3$	92.0±5.9	81.0±8.9	88.0±9.4	82.7±10.2	75.0±9.9
ACAbsArea, DCArea,		(0.1±0.1)	(0.4±0.2)	(0.2±0.1)	(0.3±0.2)	(0.5±0.3)
Invariant: total	$80.6 \pm 3.8$	89.4±8.2	81.5±9.4	87.3±10.0	82.0±11.3	71.8±11.3
DCPostureDist, ACVar,		$(0.2\pm0.1)$	$(0.4\pm0.2)$	$(0.2\pm0.1)$	$(0.3\pm0.2)$	$(0.6\pm 0.3)$
ACBandEnergy,						
ACLowEnergy,						
ACModVigEnergy,						
ACEntropy, ACFFTPeaks,						
ACPitch, ACMCR and						
ACCorr						
Invariant reduced	$80.7 \pm 3.6$	90.4±7.8	80.7±9.4	87.9±10.0	82.2±11.3	73.1±11.0
DCPostureDist, ACVar,		$(0.2\pm0.1)$	$(0.4\pm0.2)$	$(0.2\pm0.1)$	$(0.3\pm0.2)$	$(0.5\pm0.3)$
ACBandEnergy,						
ACFFTPeaks,			1			

Table 5-18: Performance of the four best feature subsets found in this work while computing featur	es
per sensor and evaluating the results in a subject dependent manner.	

Features subsets	All	Postures	Ambulation	Exercise	Resistance	Household
	Activities				Exercise	
All Features:	$44.8\pm6.1$	42.9±35.2	29.0±25.7	27.4±30.1	20.0±25.0	23.8±22.7
MaxAcceleration		(0.4±0.5)	(0.9±1.0)	(0.6±0.7)	(0.8±0.8)	(0.9±0.7)
Fast to compute:	$41.6\pm6.5$	41.6±33.3	24.4±24.4	23.7±28.1	16.9±22.2	21.8±23.3
ACAbsArea, DCArea,		$(0.4\pm0.6)$	$(1.0\pm1.0)$	(0.7±0.8)	(0.9±0.9)	(0.9±0.8)
ACVar, ACRange,						
ACMCR						
Fast to compute reduced:	$40.1\pm5.9$	39.78±31.23	$22.48 \pm 24.87$	$24.5\pm29.0$	16.3±23.7	$18.7 \pm 20.9$
ACAbsArea, DCArea,		(0.49±0.60)	$(1.02 \pm 1.14)$	$(0.7\pm0.8)$	(0.9±1.0)	(0.9±0.8)
Invariant total	$43.9\pm6.9$	40.4±33.6	$28.5 \pm 25.7$	27.7±29.9	$20.2\pm24.4$	23.3±22.5
DCPostureDist, ACVar,		$(0.4\pm0.5)$	(1.0±0.9)	(0.7±0.7)	(0.9±0.8)	(1.0±0.8)
ACBandEnergy,						
ACLowEnergy,						
ACModVigEnergy,						
ACEntropy,						
ACFFTPeaks, ACPitch,						
ACMCR and ACCorr						
Invariant reduced	$43.4\pm5.8$	38.7±35.6	28.3±26.0	28.6±30.8	20.6±25.5	23.5±23.6
DCPostureDist, ACVar,		$(0.5\pm0.8)$	$(1.0\pm1.0)$	$(0.7\pm0.8)$	$(0.9\pm0.9)$	$(1.0\pm0.8)$
ACBandEnergy,						
ACFFTPeaks,						

 Table 5-19: Performance of the four best feature subsets found in this work while computing features per sensor and evaluating the results in a subject independent manner.

Features subsets	All	Postures	Ambulation	Exercise	Resistance	Household
	Activities				Exercise	
All Features:	$81.76 \pm 1.57$	92.2±5.9	84.4±9.2	88.7±9.9	84.8±10.8	75.2±10.0
MaxAcceleration		$(0.1\pm0.1)$	(0.3±0.1)	$(0.2\pm0.1)$	$(0.2\pm0.1)$	(0.4±0.2)
Fast to compute:	$82.61 \pm 1.70$	93.4±5.2	85.2±9.4	$88.4 \pm 9.8$	$84.2 \pm 11.1$	77.4±10.0
ACAbsArea, DCArea,		(0.1±0.1)	(0.3±0.1)	(0.2±0.1)	$(0.2\pm0.1)$	(0.4±0.2)
ACVar, ACRange,						
ACMCR						
Fast to compute reduced:	$83.27 \pm 1.64$	94.0±5.1	86.1±8.0	89.4±8.6	85.6±9.6	77.7±9.3
ACAbsArea, DCArea,		(0.1±0.1)	$(0.2\pm0.1)$	(0.1±0.1)	$(0.2\pm0.1)$	(0.4±0.2)
Invariant: total	$80.45 \pm 2.02$	90.3±8.3	82.9±9.8	88.2±9.5	84.1±10.4	72.8±10.3
DCPostureDist, ACVar,		(0.1±0.1)	(0.3±0.1)	(0.2±0.1)	$(0.2\pm0.1)$	(0.5±0.2)
ACBandEnergy,						
ACLowEnergy,						
ACModVigEnergy,						
ACEntropy, ACFFTPeaks,						
ACPitch, ACMCR and						
ACCorr						
Invariant reduced	$81.0\pm1.9$	93.1±6.0	82.4±10.4	88.8±9.7	84.0±10.8	74.3±10.5
DCPostureDist, ACVar,		$(0.1\pm0.1)$	(0.3±0.1)	$(0.2\pm0.1)$	(0.3±0.1)	(0.4±0.2)
ACBandEnergy,						
ACFFTPeaks,						

 Table 5-20: Performance of the four best feature subsets found in this work while computing features per axis and evaluating the results in a subject dependent manner.

Features subsets	All Activities	Postures	Ambulation	Exercise	Resistance Exercise	Household
All Features:	$49.4\pm4.9$	66.9±31.4	41.4±26.0	39.8±33.7	31.5±29.7	37.9±24.4
MaxAcceleration		(0.3±0.4)	(0.8±0.8)	(0.6±0.6)	(0.8±0.8)	(0.9±0.7)
Fast to compute:	$48.2\pm3.8$	64.6±29.4	33.8±26.1	40.9±34.3	28.1±29.3	39.3±23.4
ACAbsArea, DCArea,		(0.3±0.4)	$(0.9\pm0.8)$	(0.7±0.7)	$(0.9\pm0.9)$	(0.9±0.6)
ACVar, ACRange,						
ACMCR						
Fast to compute reduced:	$47.0 \pm 4.7$	$60.0 \pm 32.5$	35.2±27.3	$41.9 \pm 35.1$	29.9±30.3	36.1±24.0
ACAbsArea, DCArea,		(0.3±0.4)	(0.9±0.8)	$(0.6\pm0.6)$	$(0.8\pm0.8)$	(1.0±0.7)
Invariant: total	$47.3 \pm 5.9$	59.2±34.4	38.4±27.0	37.0±30.3	28.3±26.8	35.9±23.3
DCPostureDist, ACVar,		$(0.4\pm0.5)$	$(0.8\pm0.8)$	$(0.6\pm0.6)$	$(0.8\pm0.7)$	$(0.9\pm0.6)$
ACBandEnergy,						
ACLowEnergy,						
ACModVigEnergy,						
ACEntropy,						
ACFFTPeaks, ACPitch,						
ACMCR and ACCorr						
Invariant reduced	$47.0 \pm 4.1$	54.5±34.0	38.4±26.1	38.4±31.9	29.4±28.2	37.2±23.4
DCPostureDist, ACVar,		$(0.4\pm0.5)$	$(0.9\pm0.8)$	$(0.6\pm0.7)$	$(0.8\pm0.8)$	$(0.9\pm0.6)$
ACBandEnergy,						
ACFFTPeaks,						

 Table 5-21: Performance of the four best feature subsets found in this work while computing features per axis and evaluating the results in a subject independent manner.

subsets using feature computation per sensor, per axis, subject dependent and independent evaluation is presented in Table 5-18 through Table 5-21.

Overall, Table 5-18 and Table 5-20 illustrate that the best performing feature sets for subject dependent training in decreasing order of performance are: *Fast to compute reduced, fast to compute, invariant reduced, invariant total,* and *all features.* It is

surprising that features so easy to compute such as the ACAbsArea and the DCArea contained in the Fast to compute reduced set can achieve the best performance during subject dependent training. One explanation might be that subjects perform activities in such a distinctive manner that these simple features are enough to discriminate among all the activities of interest well enough. The performance per activity during subject dependent training (as shown in Appendix A6) is also consistently higher for the fast to compute reduced feature set than for the other feature sets except for few activities such as callisthenic crunches, Cycling hard at 80rmp, running at 4mph, walking at 3mph (3 grade incline), and vacuuming. Nevertheless, the difference in performance on these activities with respect to the all features and invariant total feature sets ranges only from 2.0 to 5.3%.

The best performing feature sets during subject independent evaluation from Table 5-19 and Table 5-21 in decreasing order of performance are: *All features, invariant total, invariant reduced, fast to compute,* and *fast to compute reduced.* It is clear, thus, that for subject independent training more features are required to capture the high variability found in the way activities are performed from individual to individual. In this scenario, the overall decrease in performance obtained by using the *fast to compute* feature set with respect to the performance of the *all features* and *invariant reduced* feature sets is ~2% during both feature computation per sensor and per axis.

When the differences in performance are analyzed per activity during subject independent training, it is found that all the feature subsets explored have difficulties recognizing activities involving the same speed and type of motion but different resistance levels. These activities include *bench weight lifting* and *bicep curls* with different weight loads, *walking* at the same speed (3mph) but different inclination grades, and *rowing* at the same speed but with different resistance levels. This is expected, since the accelerometer signals look alike given that the activities are performed at the same speed and with a very similar motion patterns. Section 5.4.8 will later explore if this activities can be better discriminated by incorporating heart rate information. All the feature sets have also difficulties recognizing activities that were performed with high degree of variability from subject to subject such as *gardening*, *weeding*, and *taking out trash*. These activities are also difficult to discriminate because they include short periods of time where other activities being recognized are executed such as *walking*, *standing*, and *kneeling*.

When the performance per activity of the subsets is compared against the performance of the *all features* subset, it is found that they have different strengths and weaknesses. Figure 5-22 and Figure 5-23 show the differences in true positive and false positive rate per activity coded as a grayscale image. The grayscale images show the performance per activity scaled so that the best performance is shown in white and the worst performance is shown in black. In other words, poor areas of performance are shown as dark regions in the image. Again, it can be seen that the best performing feature subset is the *all features* set. When comparing the differences of the *fast to compute* and *invariant reduced* feature



Figure 5-22: True Positive Rate per Activity when the different subsets of features are computed per sensor using the C4.5 decision tree classifier and subject independent evaluation. The grayscale image is scaled so that the maximum true positive rate of 79.1% is represented by white and the minimum of 1.3% by black. In other words, poor areas of performance are shown in black.

sets against the *all features* set it is found that the *invariant reduced* feature set has more problems recognizing activities involving different postures than the *fast to compute* feature set. These activities include *sitting*, *standing*, *kneeling*, *bench weight lifting*, *bicep* curls, and rowing at 30spm. On the contrary, the fast to compute feature set has more difficulties than the *invariant reduced* feature set in recognizing some activities involving periodic motion such as carrying groceries, ascending and descending stairs, cycling at 80rmp, calisthenics sit-ups, rowing moderate at 30spm, and typing. These differences in performance can be explained by the fact that the *fast to compute* feature set includes the ACAbsArea and DCArea features that capture posture information better than the invariant reduced feature set. Similarly the invariant reduced feature set includes the ACFFTPeaks feature that provides powerful information about the periodicity of activities, thus giving it an advantage over the *fast to compute* feature set. One might expect that a new feature set created by merging the *fast to compute* and *invariant* reduced feature sets would address the short comings of both feature sets. However, this new feature set would be more computationally expensive and not invariant to the magnitude of the acceleration signal.

To better understand the impact of utilizing fewer sensors to recognize activities on the feature sets, an experiment was performed using the *fast to compute* and *invariant reduced* feature sets to recognize activities using the C4.5 classifier over different subsets



Figure 5-23: True Positive Rate per Activity when the different subsets of features are computed per axis using the C4.5 decision tree classifier and subject independent evaluation. The grayscale image is scaled so that the maximum true positive rate of 93.8% is represented by white and the minimum of 4.8% by black. In other words, poor areas of performance are shown in black.

of sensors. Table 5-22 and Table 5-23 present the performance of these two feature sets when recognizing activities using (1) three sensors worn at the dominant wrist, hip, and dominant foot, (2) a single sensor worn at the hip, (3) a single sensor worn at the dominant wrist, and (4) a single sensor worn at the dominant foot evaluated in a subject dependent and independent manner. From the tables, we can see that during subject dependent training, the *fast to compute* feature set achieves the best performance when all the sensors are used and when three sensors located at the wrist, hip and foot are used. Nevertheless, the *invariant reduced* feature set has a slightly higher performance than the *fast to compute* feature sets during subject dependent training is very similar and both show approximately the same decrease in performance when the number of sensors is reduced. Therefore, it can be concluded from the results presented in this work that both feature sets achieve similar performance when recognizing activities in a subject dependent manner.

The same behavior is found during subject independent training. *The fast to compute* feature set performs well when at least one sensor per limb is used and *the invariant reduced* feature set outperforms considerably *the fast to compute* feature set when only one sensor is used. The difference in performance is higher (3 and 9%) when only one

Features subsets	Sensor subset	All Activities	Postures	Ambulation	Exercise	Resistance Exercise	Household
	4.11	00.5 0.0	00.0.0.0		00.0.0.5	04.1 10.4	72.0.10.2
Invariant	All	$80.5 \pm 2.0$	90.3±8.3	82.9±9.8	88.2±9.5	84.1±10.4	/2.8±10.3
Reduced			$(0.1\pm0.1)$	$(0.3\pm0.1)$	$(0.2\pm0.1)$	$(0.2\pm0.1)$	$(0.5\pm0.2)$
Fast to	All	$82.6 \pm 1.7$	93.4±5.2	85.2±9.4	$88.4 \pm 9.8$	84.2±11.1	77.4±10.0
Compute			$(0.1\pm0.1)$	(0.3±0.1)	$(0.2\pm 0.1)$	$(0.2\pm0.1)$	(0.4±0.2)
Invariant	Hip	$79.2\pm2.6$	91.6±7.7	81.6±9.1	85.9±11.7	80.8±12.4	74.0±10.1
Reduced	DWrist		$(0.1\pm0.1)$	(0.3±0.2)	$(0.2\pm0.1)$	$(0.3\pm0.2)$	$(0.4\pm0.2)$
	DFoot		. ,	. ,	, í	· · · · ·	
Fast to	Hip	$80.4 \pm 2.1$	90.6±7.4	83.9±9.3	86.8±9.7	81.9±11.0	74.0±10.1
Compute	DWrist		$(0.1\pm0.1)$	(0.3±0.1)	$(0.2\pm0.1)$	$(0.3\pm0.2)$	$(0.4\pm0.2)$
-	DFoot		. ,	. ,	, í	· · · · ·	
Invariant	Hip	$73.5 \pm 2.9$	86.5±9.8	80.4±11.0	81.2±13.6	74.3±15.8	61.7±13.0
Reduced	_		(0.2±0.1)	(0.3±0.2)	(0.3±0.2)	(0.4±0.2)	(0.6±0.2)
Fast to	Hip	$72.4 \pm 2.7$	87.8±9.3	80.3±11.8	80.3±13.9	73.3±14.8	58.8±13.3
Compute			(0.1±0.1)	(0.4±0.2)	(0.3±0.2)	(0.4±0.2)	(0.7±0.3)
Invariant	DWrist	$65.2 \pm 3.8$	85.0±11.9	68.1±13.1	69.3±15.6	61.5±15.8	54.2±13.4
Reduced			(0.2±0.1)	(0.5±0.2)	(0.4±0.2)	(0.6±0.2)	(0.7±0.3)
Fast to	DWrist	$67.0 \pm 3.7$	85.8±11.5	69.7±14.1	69.4±17.1	61.6±17.3	57.3±12.6
Compute			(0.2±0.1)	(0.5±0.2)	(0.4±0.2)	(0.5±0.3)	(0.7±0.3)
Invariant	DFoot	$68.8\pm3.8$	88.2±12.1	68.1±11.5	78.1±14.0	67.1±16.8	56.9±15.4
Reduced			(0.2±0.1)	(0.5±0.2)	(0.3±0.2)	(0.5±0.2)	(0.7±0.3)
Fast to	DFoot	$66.4 \pm 3.5$	91.0±8.4	65.3±12.0	77.4±14.5	64.5±15.8	51.9±14.6
Compute			(0.1+0.1)	(0.6+0.2)	(0.3+0.2)	(0.5+0.2)	(0.8+0.3)

Table 5-22: Performance of the two feature subsets with highest performance and lowest computational requirements when features are computed per axis and subject dependent evaluation is used. DWrist stands for dominant wrist and DFoot for dominant foot. The activities to recognize are the 52 activities contained in the MIT dataset including the *unknown* class.

Features	Sensor	All	Postures	Ambulation	Exercise	Resistance	Household
subsets	subset	Acuvities				Exercise	
Invariant	All	$47.3 \pm 5.8$	59.2±34.4	38.4±27.0	37.0±30.3	28.3±26.8	35.9±23.3
Reduced			(0.4±0.5)	$(0.8\pm0.8)$	$(0.6\pm0.6)$	(0.8±0.7)	$(0.9\pm0.6)$
Fast to	All	$48.2\pm3.8$	64.6±29.4	33.8±26.1	40.9±34.3	28.1±29.3	39.3±23.4
Compute			(0.3±0.4)	(0.9±0.8)	(0.7±0.7)	$(0.9\pm0.9)$	$(0.9\pm0.6)$
Invariant	Hip	$43.8\pm5.4$	50.1±33.2	33.6±25.7	33.6±29.7	27.2±27.0	34.7±23.0
Reduced	DWrist DFoot		(0.5±0.7)	(0.9±0.8)	(0.7±0.7)	(0.8±0.7)	(0.9±0.6)
Fast to	Hip	$44.3 \pm 4.9$	52.0±31.1	33.3±27.2	35.9±32.3	28.5±28.2	32.5±23.4
Compute	DWrist		$(0.3\pm0.4)$	$(0.9\pm0.9)$	$(0.6\pm0.6)$	$(0.8\pm0.8)$	(1.0±0.7)
	DFoot		. ,		. ,		. ,
Invariant	Hip	$36.1 \pm 4.1$	35.0±22.9	28.0±22.9	26.2±24.8	20.5±20.6	22.5±17.8
Reduced			(0.8±0.7)	(1.0±0.8)	(0.7±0.6)	(0.8±0.7)	(1.2±0.6)
Fast to	Hip	$26.5\pm5.5$	19.1±18.5	$20.8\pm22.8$	$17.4 \pm 19.7$	10.1±14.7	11.7±14.2
Compute			$(0.8\pm0.8)$	(1.0±0.9)	(0.7±0.7)	$(0.8\pm0.6)$	$(1.4\pm0.8)$
Invariant	DWrist	$36.9 \pm 5.2$	30.8±31.1	$28.0\pm22.6$	26.9±21.6	25.4±19.3	29.6±17.7
Reduced			$(0.8\pm0.8)$	(1.0±0.9)	(0.7±0.5)	(0.8±0.5)	(1.1±0.7)
Fast to	DWrist	$35.1 \pm 4.8$	31.2±26.6	$23.8\pm20.8$	23.7±21.9	$21.8 \pm 18.1$	$28.6 \pm 18.4$
Compute			(0.8±0.7)	$(1.0\pm0.8)$	$(0.7\pm0.6)$	$(0.9\pm0.6)$	(1.1±0.7)
Invariant	DFoot	$33.9\pm3.6$	41.1±27.4	26.3±18.6	28.1±22.1	$20.5 \pm 18.7$	17.2±13.4
Reduced			(0.7±0.7)	(1.1±0.8)	(0.7±0.5)	(0.8±0.6)	(1.2±0.7)
Fast to	DFoot	$27.1 \pm 4.1$	29.7±20.1	24.9±20.2	23.7±18.9	14.1±16.5	$7.2\pm8.9$
Compute			$(0.6\pm0.8)$	(1.1±0.9)	$(0.7\pm0.8)$	(1.0±0.9)	$(1.4\pm1.1)$

Table 5-23: Performance of the two feature subsets with highest performance and lowest computational requirements when features are computed per axis and subject independent evaluation is used. DWrist stands for dominant wrist and DFoot for dominant foot. The activities to recognize are the 52 activities contained in the MIT dataset including the *unknown* class.

sensor is worn at the hip and at the dominant foot. The results indicate that the *invariant reduced* feature set has important advantages over the *fast to compute* feature set, particularly during subject independent training.

It can be concluded from the results presented in this section, that the best feature set to use is the *invariant reduced* feature set. Even when this feature set is more computationally expensive than the *fast to compute* feature set, it has important advantages such as invariance to the magnitude of the accelerometer signal and higher performance during subject independent training. If invariance to the magnitude of the accelerometer signal is not important and if activities are recognized in a subject dependent manner, the *fast to compute* feature set can be used to recognize activities with extremely low computational requirements. For the remaining of this thesis, the invariant reduced feature set will be used to recognize activities in future experiments.

#### 5.4.8 Does the Incorporation of Heart Rate Data Improve Recognition of Activities?

Accelerometers are good at recognizing activities with different postures or distinctive motions of the body segments at particular speeds. However, they may not be well suited for recognizing activities that involve similar motion signatures but different resistance work or effort. Examples of some of these activities include *cycling at 80rmp* at different resistance levels, performing *bicep curls* with different weights at the same speed of motion, and *walking on a treadmill at 3mph* with different inclination grades. On the other hand, heart rate has a linear relationship with energy expenditure and can detect changes in effort or resistance load [47, 48, 81]. This section explores whether combining accelerometer and heart rate data improves the recognition of activities involving the same motion characteristics (e.g. speed) but different resistance work.

First, a baseline is obtained by recognizing activities utilizing only features computed from heart rate data. This also serves the purpose of identifying the most discriminant features based on heart rate. Later, the most discriminating heart rate features are incorporated to the best set of accelerometer-based features found in the previous sections. As in previous experiments, the activities to recognize are the 52 activities contained in the MIT dataset, including the *unknown* class.

In general, heart rate data from devices that are relatively convenient to wear for moderate periods of time such as the Polar chest strap tends to be noisy, particularly for activities with high degree of motion such as *running* on a treadmill at 5 or 6mph. This is because the quality of heart rate data depends on the good attachment of the heart rate monitor to the chest of the subject. The more the heart rate monitor moves, the noisier the heart rate data. Thus, in order to reduce the noise, a 15s running average filter is applied over the heart rate data before the segmentation and feature computation steps. This filter is applied over windows of 15s because it was found via informal testing to reduce noise considerably while minimizing the delay introduced in some activities with rapid changes in heart rate such as ascending and descending stairs. Heart rate data is then segmented by accumulating the data over sliding windows of specific lengths. When the accelerometer data is included, the heart rate window length extends from the end of the acceleration window backwards in time (see Figure 5-24). Accelerometer features are computed over windows of 5.6s (optimal window length found in Section 5.4.6), so the two types of features are being computing using different window lengths.



Figure 5-24: Graphic representation of the segmentation of heart rate and accelerometer data using the sliding window approach. Note that the windows for accelerometer and heart rate data are of different lengths and that the heart rate sliding window extends from the end of the end of the accelerometer window backwards in time.

Heart rate windows and their associated acceleration windows are discarded when no sensor values are available for heart rate over a given window. The results presented in this section are evaluated using subject dependent and independent training.

## 5.4.8.1 How Well can Activities be Recognized from Heart Rate Data?

This subsection has two goals: (1) to identify the heart rate features with highest discriminant power and (2) to establish a baseline as to how well can the activities of interest be recognized from heart rate data alone. The heart rate features explored in this section are: *HRMean*, *HRAboveRest*, *ScaledHR*, *HRVar*, and *HRTrend*. Appendix A3 provides an explanation of how these features are computed. These features attempt to capture either the level of physical effort associated with an activity (*HRMean*, *HRAboveRest*, *and ScaledHR*), its variability (*HRVar*), or tendency (*HRTrend*).

First, information gain feature selection was performed over the heart rate features. This sorts the features in order of importance according to the information they provide. The ordering of the features from most important to least important obtained from the

procedure is as follows ScaledHR, HRAboveRest, HRMean, HRTrend, and HRVar. The three most important features ScaledHR, HRAboveRest, and HRMean capture the same information. However, the better performance of *ScaledHR* is expected since this feature normalizes the heart rate data of an individual to lie between resting heart rate (a value of zero for this feature) and the average heart rate value obtained while the subject runs on a treadmill at 5mph (a value of 1 for this feature). Using this normalization, two individuals with different fitness level performing the same activity could have different heart rate values but relative to their heart rate values at rest and while running at 5mph, they could be performing in the same intensity zone. Thus, this normalization helps to minimize the inter-individual variations in heart rate values due to differences in the fitness level. Ideally, this feature would normalize heart rate between resting heart (RHR) rate and maximal heart rate (MHR). Unfortunately, for this to be possible, a maximal physical exertion test would be required. These tests are inconvenient in practice because they require an exhaustive exercise session where the physical condition of the subject is push to the limit to measure heart rate during maximal exertion. Some individuals not physically fit or with other medical conditions might not even be able to perform such a test. Furthermore, these tests would be impractical to impose for a consumer-based mobile phone application – the ultimate goal. The least important features are HRTrend and HRVar (an experiment will later characterize how least important they are). These features capture the variability or tendency of the heart rate data. The HRTrend feature, in particular, tells if heart rate data is increasing (positive value for the feature), decreasing (negative value) or in steady state (near a value of zero) over time. This might be important information to capture because heart rate for some physically demanding activities such as *ascending stairs*, *crunches* and *sit-ups* constantly increases continuously as the activity is performed, possibly for the duration of the activity if a person is not very fit and the maximal exertion is not reached Consequently, when features such as ScaledHR are computed over these activities, they show high variability in their values that leads to a low recognition performance by the C4.5 classifier. One possible explanation for the low ranking of these features (HRVar and HRTrend) is that they were computed over a short duration window (5.6s). Later in this section, experiments will be performed to measure the impact of utilizing longer window lengths over these features.

To corroborate the ordering of importance found using information gain feature selection, the performance over individual features was tested using the C4.5 classifier during subject dependent and independent training. The results are illustrated in Table 5-24 and Table 5-25. The ordering during subject dependent training is the same as the one found during information gain feature selection. For subject independent training, the ordering with respect to overall performance is reversed for the *ScaledHR* and *HRAboveRest* features. However, when analyzing the result per activity category it can be seen that the *ScaledHR* feature is outperforming the *HRAboveRest* feature in most activity categories. The main difference in overall performance between these features is due to the performance over the *unknown* class. The *HRAboveRest* feature achieves a true positive rate of 88.0% for this class while the *ScaledHR* feature achieves only a true positive rate of 80.3%. When the true positive rate is analyzed per activity during subject dependent evaluation, it is found that the activities with highest performance are *lying down* (87-89%), *running at 5mph* (7-31%), and *running at 6mph* (12-29%).

Heart Rate Features (Number of features)	All	Postures	Ambulation	Exercise	Resistance Exercise	Household
ScaledHR (1)	37.7 ± 7.0	27.4±17.4 (1.0±0.7)	26.6±20.5 (1.1±0.6)	30.3±22.7 (1.0±0.6)	24.5±18.8 (1.0±0.6)	16.3±14.0 (1.2±0.6)
HRAboveRest (1)	37.4 ± 7.4	26.8±18.1 (1.0±0.8)	27.2±20.8 (1.1±0.6)	29.9±22.5 (1.0±0.6)	24.5±18.8 (1.1±0.6)	16.2±14.4 (1.2±0.7)
HRMean (1)	37.4 ± 6.6	26.5±18.5 (1.0±0.6)	27.4±20.9 (1.1±0.6)	30.6±22.5 (1.0±0.6)	24.9±18.7 (1.10±0.6)	16.8±13.8 (1.2±0.7)
HRVar (1)	29.7 ± 8.5	6.8±9.1 (0.9±1.0)	1.4±2.8 (0.5±0.5)	6.5±9.3 (0.6±0.5)	5.9±8.7 (0.5±0.4)	2.0±3.6 (0.5±0.5)
HRTrend (1)	29.3 ± 8.3	4.9±5.9 (0.9±0.9)	1.9±3.3 (0.5±0.4)	6.3±8.6 (0.7±0.6)	6.2±8.3 (0.7±0.6)	$1.7\pm2.9$ (0.5±0.4)

Table 5-24: Performance of recognizing the 52 activities contained in the MIT dataset (including the *unknown* class) from heart rate features only over windows of 5.6s using the C4.5 decision tree classifier during subject dependent evaluation.

Heart Rate Features (Vector size)	All	Postures	Ambulation	Exercise	Resistance Exercise	Household
ScaledHR (1)	$30.0 \pm 6.3$	15.2±4.9	5.5±4.2	4.4±4.3	1.0±1.9	1.3±1.6
		$(0.9\pm0.8)$	$(0.4\pm0.4)$	$(0.5\pm0.5)$	$(0.4\pm0.3)$	$(0.4\pm0.2)$
HRAboveRest (1)	$31.4\pm7.0$	15.2±4.2	1.8±2.9	$1.85 \pm 3.1$	0.7±1.4	0.9±1.2
		(0.8±0.8)	(0.2±0.2)	(0.3±0.2)	(0.2±0.1)	$(0.2\pm0.1)$
HRMean (1)	$27.8\pm7.3$	4.2±3.2	1.7±3.0	1.6±2.7	0.6±1.1	0.6±1.4
		$(1.1\pm1.4)$	(0.2±0.2)	(0.3±0.3)	(0.2±0.2)	$(0.2\pm0.2)$
HRVar(1)	$29.9\pm7.6$	0.0±0.0	0.0±0.0	0.8±1.9	0.8±2.0	0.0±0.0
		(0.0±0.0)	(0.0±0.0)	$(0.0\pm0.0)$	$(0.0\pm0.0)$	(0.0±0.0)
HRTrend (1)	$29.8\pm7.9$	0.0±0.0	0.0±0.0	0.0±0.0	0.0±0.0	0.0±0.0
		(0.0±0.0)	(0.0±0.0)	$(0.0\pm0.0)$	$(0.0\pm0.0)$	(0.0±0.0)

Table 5-25: Performance of recognizing the 52 activities contained in the MIT dataset (including the *unknown* class) from heart rate features only over windows of 5.6s using the C4.5 decision tree classifier during subject independent evaluation.

These activities are better discriminated because heart rate reaches steady state for a good proportion of the activity length and because they represent the extreme values that heart rate data can reach. The overall performance of the *HRVar* and *HRTrend* features during subject independent evaluation is deceiving. From Table 5-25, it can be seen that their performance for all activity categories is zero, while their overall performance is around 29%. This is because the *unknown* class, which is the class with largest number of examples (see Appendix A4), has a true positive rate of near 100% when both features are used. To prevent this from happening in the following experiments, the *unknown* class is no longer included for the remaining of this section.

Given that the main difference found between the *HRAboveRest* and *ScaledHR* feature is the performance over the *unknown* class, new experiments were run to better understand the performance per activity when this class is not included. Table 5-26 presents the new results obtained. The table shows that the performance using both features is very similar during subject dependent training. When the true positive rate is analyzed per class for both features, it is found that the best performance is achieved for *lying down* (87%), *cycling hard at 80rmp* (65%), *cycling light at 60rmp* (66.5%), *running at 5 and 6mph* (56-64%), and *walking* at different speeds and inclinations (47-55%). Again, this is because these activities include periods where the heart rate data reaches steady state and that the classifier utilizes to discriminate among the activities.

Features	All	Postures	Ambulation	Exercise	Resistance Exercise	Household
ScaledHR	$38.4\pm7.8$	37.7±22.3	39.0±22.2	38.3±23.9	34.2±20.6	24.0±18.0
Subject Dependent		$(1.5\pm1.0)$	(1.7±0.8)	(1.4±0.8)	(1.6±0.9)	(2.0±1.1)
HRAboveRest	$38.0 \pm 7.7$	37.6±21.7	39.5±22.1	37.7±23.6	34.2±20.2	23.6±17.3
Subject Dependent		$(1.5 \pm 1.0)$	(1.7±0.8)	(1.4±0.8)	(1.6±0.9)	$(2.1\pm1.1)$
ScaledHR	$13.8\pm3.2$	16.2±5.9	14.2±9.3	9.0±7.5	4.1±5.2	4.3±4.9
Subject Independent		(1.7±1.2)	(2.3±1.3)	(1.6±1.0)	(1.7±0.8)	(1.7±0.9)
HRAboveRest	$11.9 \pm 3.32$	17.5±7.0	5.7±7.4	4.8±7.2	3.8±5.4	4.3±5.0
Subject Independent		(1.7±1.3)	$(2.2\pm1.0)$	(1.7±0.8)	(1.8±0.8)	(1.7±0.8)

Table 5-26: Performance of recognizing the 51 activities contained in the MIT dataset (without including the *unknown* class) in a subject independent manner using the C4.5 decision tree classifier trained over the *ScaledHR* and *HRAboveRest* features over a heart rate sliding window of 5.6s

The activities with poorest performance are those whose heart rate value keeps increasing as the activity is performed such as *ascending stairs* (2.9%) and *bench weight lifting hard* (8.7%). When the confusion matrix is analyzed for subject dependent training (shown in Appendix A7), it can be seen that there exists a high degree of confusion among household activities.

This is because heart rate has similar values for most of these activities and there is no single heart rate value characterizing each of the activities. Interestingly, the household activities better recognized were sedentary activities where the heart rate value reaches steady state such as *ironing* (36%), *playing video games* (34%), *typing* (30%), and *writing* (48%). The confusion matrix also shows that sedentary and ambulatory activities are being confused with household activities. This is because most household activities include short periods of time where postures and ambulatory activities are executed.

The results using subject independent evaluation show in Table 5-26, suggest that the *ScaledHR* feature has a slightly higher performance (~2%) than the *HRAboveRest* feature. This better performance is more apparent for ambulation an exercise activities where the true positive rate doubles for the *ScaledHR* feature. This can be explained by the fact that the *ScaledHR* feature helps in normalizing the heart rate signal so that differences in fitness level of individuals are minimized. The notable low performance during subject independent training with respect to subject dependent training is due mainly to the different fitness level of the individuals. Two individual performing the same activity, but with different fitness levels, would have heart rate readings with different values. Although the *ScaledHR* feature attempts to mitigate this problem, it is clear that new ways to compensate for this variation among individuals are required.

One way to mitigate the differences in the fitness level of individuals would be to include features that attempt to capture this information such as the body weight of an individual (*weight*) and the percentage of fat of an individual (*FatPercent*). Therefore, an experiment was run to determine if the addition of these features improves subject independent recognition of activities when heart rate data alone is used. Table 5-27 presents the results. The table also includes another feature that attempts to better describe the fitness level of an individual. This feature is labeled as *FitnessIndex* and is computed by dividing the number of steps by the average heart rate value (in beats-perminute) of a subject while *running* on a treadmill at 5mph. This feature is an approximation to the fitness index suggested in [230]. The results show that comparing

Features	All	Postures	Ambulation	Exercise	Resistance Exercise	Household
ScaledHR	$13.8 \pm 3.2$	16.2±5.9	14.2±9.3	9.0±7.5	4.1±5.2	4.3±4.9
		(1.7±1.2)	(2.3±1.3)	$(1.6\pm1.0)$	(1.7±0.8)	(1.7±0.9)
ScaledHR +	$12.9\pm5.3$	19.0±13.5	$14.4{\pm}18.1$	9.4±12.8	5.7±10.0	5.2±9.3
Weight		$(2.6\pm2.7)$	(3.0±2.9)	(2.1±1.9)	$(2.2\pm1.9)$	(2.6±2.3)
ScaledHR +	$12.0 \pm 3.9$	19.3±15.3	11.2±13.3	7.5±10.3	4.1±8.5	4.1±7.3
FatPercent		(2.7±3.0)	(2.7±2.6)	(2.1±2.2)	$(2.2\pm2.1)$	$(2.5\pm2.1)$
ScaledHR +	$13.8\pm3.6$	20.5±15.5	14.1±17.9	10.4±16.4	6.7±13.9	6.5±10.0
FitnessIndex		$(2.6\pm2.5)$	(2.8±2.4)	(2.0±1.9)	(2.2±1.9)	(2.5±2.1)

Table 5-27: Performance of combining features that attempt to describe the fitness level of an individual with heart rate features (ScaledHR) during subject independent evaluation using the C4.5 decision tree classifier. The target activities were the 51 activities contained in the MIT Dataset, without including the unknown class.

the difference in overall performance is again deceiving. For example, the overall performance for *ScaledHR+Weight* is lower than for the *ScaledHR* feature alone,

but the performance per activity category shows that the *ScaledHR+Weight* features have a slightly higher performance ranging from 1 to 4%. Cautious analysis of the performance per activity when the *Weight*, *FatPercent*, and *FitnessIndex* features are added to the *ScaledHR* feature (Tables can be found in Appendix A7) show that the addition of these features changes the distribution of the performance per activity little. As a result, there is no clear advantage of incorporating these features. One explanation why these features do not improve recognition of activities is that the number of subjects is relatively low (20) to capture the fitness differences across individuals.

Since the performance of heart rate features such as ScaledHR, HRVar, and HRTrend can vary depending on the window length used, an additional experiment was run to measure this effect. During this experiment, the heart rate window length was varied from 5.6s to 45.5s while the performance over these features was measured using the C4.5 decision tree classifier. Figure 5-25 presents the results obtained from the experiment for the ScaledHR feature. The results for the HRVar and HRTrend can be found in Appendix A7. During subject dependent training, Figure 5-25 shows that the window length with higher performance over all and per activity category is 5.6s. After this window length, the performance for all the activities and each activity category drops significantly. The main reason why performance decreases as the window length is increased is a reduction in the number of training examples available. Longer window lengths also introduce more variability in the heart rate data readings for some activities such as postures and thus, further decreases performance as seen in Figure 5-25a. During subject independent training, on the contrary, the performance per activity seems to improve as observed when increasing the window length for accelerometer-based features. This is because longer window lengths have a smoothing effect over the heart rate data that reduces inter-subject variability. The effect of increasing the window length for the HRVar and HRTrend features (as shown in Appendix A7) during subject dependent training has a similar effect as the one observed for the *ScaledHR* feature. The only difference is that for these features, a little improvement in performance (~1-2%) is observed for the ambulation, exercise, resistance exercise, and household activities at a window length of 11.3s. Nevertheless, the decrease in performance at this window length



Figure 5-25: Performance of the *ScaledHR* feature computed over varying window lengths using the C4.5 classifier and evaluated using (a) subject dependent training and (b) subject independent training.

(11.3s) for postures is about 4.3%. For subject independent training, it is found that the performance using the *HRVar* and *HRTrend* features decreases as the window length increases. This is because the variability and trend of the heart rate data is less distinctive for each activity as the window length is increased since increasing window lengths have a smoothing effect on the data. In summary it can be concluded that the optimal window length to use for heart rate data is 5.6s. This window length has the highest performance during subject dependent training for the *ScaledHR* feature and is the optimal window length to use for the *HRVar* and *HRTrend* features. The performance during subject independent training improves for the *ScaledHR* feature as the window length increases, but the overall improvement obtained is only about 2% when increasing the window length from 5.6s to 45.5s.

Given the low performance obtained while recognizing activities from heart rate data alone, one might predict that combining heart rate with accelerometer data might not substantially improve recognition of activities unless the C4.5 classifier could learn to utilize it only to discriminate among activities involving different effort or resistance level. Although this option will be explored in the next section, another alternative is to force the C4.5 classifier to use heart rate data only to discriminate among these activities. This can be achieved by first classifying the activity type using accelerometer data, and then utilizing a classifier trained only on heart rate data to discriminate among the intensity levels of an activity. Consequently, to test this idea, we trained five independent classifiers utilizing the ScaledHR feature to discriminate among the different intensity levels of an activity in a subject independent manner. Table 5-28 presents the overall accuracy of each classifier and Appendix A7 presents the accuracies per activity. It can be seen that even just recognizing among the intensity level of few activities using heart rate data is difficult. The best accuracy achieved is 51% for the classifier that discriminates between the intensity levels of *bench weight lifting*, and the worse accuracy of 29% is for the rowing at 30spm classifier. The performance of the rowing at 30spm classifier is perhaps the worst because the resistance level of the rowing machine used during the data collection changes only slightly compared to the other machines used. Thus, heart rate readings do not change dramatically for the different intensity settings. In fact, the resistance of a rowing machine is mostly set by the pulling effort and expertise of each individual at the machine. It follows that even when this hierarchical approach is followed to incorporate heart rate data, the discrimination among intensity levels of an activity would still be poor for subject independent recognition. Again, the main problem is the diverse heart rate readings obtained for each activity due to differences in subjects' fitness. Another option would be to utilize subject dependent training to discriminate among the intensity levels of an activity, but this requires the subject to provide examples of the activities of interest. The main challenge then would be to come up with strategies to make the training process easy and amenable to end-users.

This section has shown that recognizing activities from heart rate data alone is difficult, particularly in a subject independent manner. Even when the best performing feature is used (*ScaledHR*), the best total accuracy obtained is  $38.4 \pm 7.8$  for subject dependent training and  $13.8 \pm 3.2$  for subject independent training. The main problem found so far is that the values of heart rate data are different for two individuals performing the same activity but with different fitness level and without requiring burdensome training

Intensity Activity Classifier	Intensities	Total Accuracy
Bicep Curls	Weight Load: Light (0.9Kg), moderate (2.2Kg), and hard	$42.1 \pm 10.8$
	(3.6Kg)	
Bench Weight Lifting	Weight Load: Light (0.9Kg), moderate (3.1Kg), and hard	$51.3 \pm 20.0$
	(7.7Kg)	
Cycling at 80rmp	Resistance level: Light (2), moderate(7), and hard (13)	$41.0\pm15.5$
Rowing at 30spm	Resistance level: Light (2), moderate(5), and hard (8)	$29.7 \pm 11.2$
Walking at 3mph	Treadmill inclination: 0, 3, 6, 9	$43.2 \pm 11.6$

Table 5-28: Performance of five independent C4.5 classifiers trained utilizing the *ScaledHR* feature to discriminate among the different intensity levels of an activity. The *ScaledHR* feature was computed over windows of 5.6s in length.

procedures measuring maximal expenditure we have no mechanism to normalize them. After plotting misclassification histograms, it was also found that the errors were concentrated at the beginning and end of activities. This is because heart rate lags physical activity and remains altered once the activity has finished (errors at the end of activity or beginning of the next one). Furthermore, for vigorous activities of short duration such as *ascending stairs*, *crunches*, and *sit-ups*, heart rate constantly increases resulting in classification errors all across the activity. Finally, another problem with heart rate data while recognizing sedentary activities is that heart rate can also be altered by emotional states, stress, biological processes, and even smoking. This problem was not apparent in our dataset but it might be of potential concern in practice.

The poor performance obtained by recognizing activities from heart rate data alone in this section might be a consequence of the static nature of the classification approach followed, where no temporal information about the variability of heart rate over time is incorporated. More computationally expensive classification techniques such as hidden Markov models (HMMs) or dynamic Bayesian networks (DBNs) might improve the classification results obtained in this section by incorporating temporal information about how hear rate changes over time as activities change. This approach was not explored in this work due to its high computational requirements since one of the main goals of this work is to come up with classification strategies amenable for real-time performance in existing handheld devices.

# 5.4.8.2 How Well can Activities be Recognized by Combining Acceleration and Heart Rate Data?

In this section, the best performing heart rate feature found (*ScaledHR*) in the previous section is added to the best set of acceleration-based features (*invariant reduced* feature set) found in previous sections. Table 5-29 presents the results of combining both feature sets using the C4.5 classifier during subject dependent and independent evaluation. Both feature sets are computed over windows of 5.6s. The detailed results per activity can be found in Appendix A7.

Table 5-29 shows that the improvement in overall performance when the *ScaledHR* feature is incorporated is approximately 1.6% during both subject dependent and independent evaluation. When the performance per class is inspected for subject dependent training (shown in Appendix A7), it is found that the C4.5 classifier indeed learns to improve the performance of activities involving different levels of effort or

Features subsets	Evaluation	All	Postures	Ambulation	Exercise	Resistance Exercise	Household
Invariant Reduced	Subject	87.93 ±	96.9±4.1	88.8±8.4	92.6±7.7	88.0±9.5	80.6±9.0
	Dependent	2.08	$(0.1\pm0.1)$	$(0.2\pm0.2)$	$(0.1\pm0.1)$	$(0.2\pm0.2)$	$(0.4\pm0.2)$
Invariant Reduced	Subject	$89.49 \pm$	96.9±4.1	91.7±7.3	93.3±7.6	90.4±8.7	82.0±8.5
+ ScaledHR	Dependent	1.81	(0.1±0.1)	$(0.2\pm0.1)$	$(0.1\pm0.1)$	(0.2±0.2)	(0.4±0.2)
Invariant Reduced	Subject	$50.63 \pm$	77.0±24.0	46.6±27.1	46.6±31.6	34.1±29.9	43.8±25.6
	Independent	5.18	(0.5±0.7)	$(1.2\pm1.0)$	$(0.9\pm0.9)$	$(1.2\pm1.0)$	(1.3±0.9)
Invariant Reduced	Subject	$52.28 \pm$	76.2±24.6	51.8±28.8	46.4±31.5	36.6±31.4	43.7±25.8
+ ScaledHR	Independent	5.84	(0.5±0.7)	$(1.0\pm1.0)$	(0.8±0.9)	$(1.1\pm1.0)$	(1.3±0.8)

Table 5-29: Performance of recognizing activities when the most discriminating accelerometer (invariant reduced feature set) and heart rate feature (*ScaledHR*) are computed per axis and fed to the C4.5 classifier to recognize the 51 activities contained in the MIT dataset without including the *unknown* class.

resistance load without changing the performance for the other activities. The performance improves for cycling at 80rmp hard and moderate (improvement between +1.6 and +4.5%), Rowing at 30spm light, moderate and hard (+1.9-2.9%), walking at 3mph on a treadmill at inclinations of 0, 3, 6, and 9 (+3.7-12.5%), wiping/dusting (+4.6%), and taking out trash (+3.2%). stacking groceries (+3%), sweeping (+2.3%), Vacuuming (+1.5%), The performance also increases for washing windows, carrying groceries, and scrubbing a surface, but the increase is less than +1%. During subject independent evaluation, the performance also increases between +2.4% and +28.7% for some activities such as cycling at 80rmp hard, rowing at 30spm moderate, running at 5mph, walking on a treadmill at different inclinations, and taking out trash. However, even when the improvement seems considerable for some activities (e.g. 28.7% for walking at 3mph with an inclination of 9), other activities such as bicep curls hard and moderate, rowing at 30spm hard, and running at 6mph suffer a decrease in performance as high as 8%. Therefore, the improvement achieved by incorporating heart rate data during subject independent training is less obvious than the one obtained for subject dependent training. The confusion matrices for the combination of the invariant reduced and *ScaledHR* features are shown in Appendix A7. From these matrices, it can be seen that most of the confusion during subject dependent training happens among household activities. In subject independent training, confusions are concentrated among household activities, and between ambulation and postures and household activities.

In conclusion, the overall improvements achieved while incorporating heart rate data during both, subject dependent and independent training is only 1.6%. The improvement is more obvious for subject dependent activities where the performance is increased between 1.6 and 12.5% for activities involving different levels of effort or resistance load. Nevertheless, the improvement obtained might not justify (1) the burden associated with wearing a heart rate monitor or (2) the need for personalized training. As a result, from this point on, this thesis will only utilize accelerometer-based features (*invariant reduced* feature set) to recognize the activities of interest.

Evaluation Method	All	Postures	Ambulation	Exercise	Resistance	Household
					Exercise	
Subject dependent	$87.9 \pm 2.0$	96.9±4.1	88.8±8.4	92.6±7.7	88.0±9.5	80.6±9.0
without Unknown class		$(0.1\pm0.1)$	(0.2±0.2)	(0.1±0.1)	(0.2±0.2)	$(0.4\pm0.2)$
Random guess: 1.96%						
Subject dependent	$81.0 \pm 2.0$	93.1±6.0	82.4±10.4	88.8±9.7	84.0±10.8	74.3±10.5
with unknown class		(0.1±0.1)	(0.3±0.1)	(0.2±0.1)	(0.3±0.1)	$(0.4\pm0.2)$
Random guess: 1.92%						
Subject independent	$50.6 \pm 5.2$	77.0±24.0	46.6±27.1	46.6±31.6	34.1±29.9	43.8±25.6
without unknown class		(0.5±0.7)	(1.2±1.0)	(0.9±0.9)	$(1.2\pm1.0)$	(1.3±0.9)
Random guess: 1.96%						
Subject independent	$47.0 \pm 4.2$	54.5±34.0	38.4±26.1	38.4±31.9	29.4±28.2	37.2±23.4
with unknown class		(0.4±0.5)	(0.9±0.8)	(0.6±0.7)	(0.8±0.8)	(0.9±0.6)
Random guess: 1.92%						

Table 5-30: True positive and false positive rate (shown in parenthesis) of the C4.5 classifier when recognizing the 51 activities contained in the MIT dataset either including or not including the *unknown* class. The feature set used is the *invariant reduced* computed per axis over windows of 5.6s in length over all the seven accelerometers. The accuracy of random guessing is 1.92% (for 52 activities) when the unknown class is included and 1.96% (for 51 activities) when it is not.

#### 5.4.9 How Well Can All Activities Be Recognized Using the Selected Classifier, Window Length, Feature Set, and Signal Preprocessing Techniques?

This section presents the performance of the final implementation of the activity recognition algorithm using the set of parameters incrementally selected in the previous sections. These parameters consist of the C4.5 decision tree classifier, the *invariant reduced* feature set (*ACVar*, *ACFFTPeaks*, *ACBandEnergy*, and *DCPostureDist*), feature computation per axis, and sliding windows of 5.6s in length.

First, the performance of the algorithm is evaluated over the 51 activities contained in the MIT dataset using subject dependent and independent training. The results are presented for both cases -- when the *unknown* class is included and when it is not. The inclusion of the *unknown* class is used to test the performance of the recognition algorithm under somewhat more realistic conditions where there are examples of untrained or ill-defined activities. Once these results are presented, the training data requirements for subject dependent training are evaluated by training the algorithm with decreasing amounts of data. Finally, an experiment is performed to determine a good number of accelerometers to use and their locations in the human body, considering both recognition performance and everyday convenience.

Table 5-30 presents the true positive and false positive rate obtained during subject dependent and independent training while the *unknown* class is included and when it is not. The table illustrates that the best performance overall (87.9%) and per activity category is obtained for subject dependent training when the *unknown* class is not used. When the *unknown* class is added, overall performance drops 7% for subject dependent training and 3.6% for subject independent training. This is expected, since the *unknown* class contains examples of the activities of interest that were just not labeled during the data collection. The *unknown* class may also contain examples of activities very similar to the target classes. The difference in performance of 34-37% between subject dependent and independent training is considerable. This reflects the higher difficulty of
recognizing activities across individuals since each individual perform activities differently.

During subject dependent training, the best performance is obtained for postures and the worse performance is obtained for household activities. Postures are recognized well because the accelerometer signals they generate are static in nature (DC component of the signal), so there is not much variation to model. Moreover, the window length of 5.6s used is optimal for postures, as explained in Section 5.4.6. Obviously, variations in the posture signals increase when multiple individuals are considered thus decreasing performance between 20-39% during subject independent training. Household activities are the most difficult to recognize during subject dependent training because they involve a high degree of variation in the signal even when a single individual performs the activities. Some of these activities such as wiping/dusting, making the bed, taking out trash, and stacking groceries even involve multiple postures and ambulation modes that change based on how the person interacts with objects in the environment. Consequently they are confused with postures and ambulatory activities, as seen from the confusion matrices shown in Figure 5-26. Another reason why household activities are not well recognized is because they may require windows longer than 5.6s due to their high motion variability, and they may involve execution of sequences of events (e.g. standing, walking). Finally, from Table 5-30, it can be seen that the true positive rate for ambulation, exercise, and resistance exercise activities is very close, ranging from 82 to 92% during subject dependent training. During subject independent training, the performance for ambulation and exercise activities is also close (46% when the unknown class is used and 38% when it is not); however, the performance is worse for resistance exercise activities. This is due to a combination of two factors: (1) High variability in the way activities are performed across individuals and (2) the difficulty in recognizing activities with similar motion signatures (e.g. speed of execution) but different levels of resistance or work load from accelerometers.

When the performance per activity is evaluated from Appendix A8, it is found that the activities with lowest performance are resistance exercise activities such as *bench weight* lifting and bicep curls with different weights, rowing at 30spm and cycling at 80rmp at different resistance levels, walking at 3mph at different inclination grades, and some household activities such as weeding, gardening, making the bed, and taking out trash. As explained before, this is because accelerometers are not good at detecting changes in effort or resistance load, and because household activities are highly variable, even within an individual. Surprisingly, during subject dependent training, some resistance exercise activities such as *bench weight lifting* and *bicep curls* with different weights, and *cycling* at 80rpm at different resistance levels are recognized with high true positive rates ranging from 88.4 to 99.5%. Even activities such as rowing at 30spm at different resistance levels and walking at 3mph at different inclination grades are relatively well recognized with true positive rates ranging from 76 to 88.9%. This seems to indicate that changes in weight load, resistance level, or grade of inclination induce a change in the motion signature of the activity that is distinctive of each individual. For example, someone struggling with a heavier weight during *bicep curls* might unconsciously reduce the performance speed of the activity. This reduction in speed is detected by the accelerometers and captured by the features so that the classifier learns to differentiate among the different intensity levels of an activity. One important caveat here is that this



Figure 5-26: Confusion matrices for the C4.5 classifier when recognizing the 51 activities in the MIT dataset without including the *unknown* class during (a) subject dependent and (b) independent training. The feature set used is the *invariant reduced* computed per axis over windows of 5.6s in length. The maximum number of confusions per cell is 95 and 262 for subject dependent and independent training respectively.

slight reduction in motion patterns detected by the accelerometers could be different if sensors are worn with slight different variations in orientation and position from day to day. Therefore, future work should analyze the impact of slight variations in sensor placement and orientation on data collected over multiple days for the same person during subject dependent training.

Figure 5-26 presents the confusion matrices for subject dependent and independent training as a grayscale image scaled to highlight large number of classification confusions in black and low number of classification confusions in white. It can be seen that most confusions happen among household activities for both, subject dependent and independent training. In particular, activities involving postures and ambulation such as *taking out trash, making the bed, sweeping, vacuuming, mopping, wiping/dusting* are either confused among each other or with ambulatory activities. Ambulatory activities such as *carrying groceries, walking around block, walking at 2mph*, and *walking at 3mph* are also being confused among each other. Finally, activities involving similar posture and upper body motion such as *wiping/dusting, washing windows, ironing*, and *doing dishes* are also being confused, particularly during subject independent training.

### 5.4.9.1 How Much Training Data is Required During Subject Dependent Training?

The performance of subject dependent training is consistently and considerably better than subject independent training. These results strongly suggest that for activity recognition to be practical with a large number of diverse target activities such as those used here, some level of personalized training will be desirable. Thus, one of the main questions to answer is how much training data is required to achieve a reasonable performance while minimizing training time. This section explores this question by measuring the performance the final activity recognition algorithm implemented when varying amounts of training data are used.

For this experiment, the amount of data collected from each activity example provided by participants was partitioned in 75% training data and 25% testing data. The 25% of the data located at end of each activity example is utilized to test the performance of the recognition algorithm while 75% of the data at the beginning of each activity example is used to train the recognition algorithm. During the analysis, the performance of the recognition algorithm is measured when the 75% of the data located at the beginning of each activity is decreased from its original length of 100% to a final length of 10% in decrements of 20% and used to train the recognition algorithm. During this procedure, true positive rate is evaluated on the 25% of the data located at the end of each activity example. Figure 5-27 illustrates the procedure followed by representing how the data from an activity example is utilized to train and test the recognition algorithm during this analysis. This procedure was followed to keep the amount of testing data constant while the amounts of training data were varied.

Ideally, the amount of training data for each activity would be the same to perform a fair comparison among all activities. This is not the case during the analysis performed in this section since most activity examples have different lengths (as shown in Table 5-31).

	(St	art)		Activ	ity Ex	ample	(end)
			7	5%			25%
Experiment 1: 100% training data						100%	
Experiment 2: 80% training data					80%		
Experiment 3: 60% training data				60%			
Experiment 4: 40% training data			40%				
Experiment 5: 20% training data		20%					
Experiment 6: 10% training data	10%						

Figure 5-27: Graphical representation of the experiment performed to measure the performance of utilizing different amounts of training data during subject dependent training. The right side of the table represents the duration of an individual activity example performed by one of the participants. The start of the activity is represented by the (start) tag and the end by the (end) tag. The 25% of the data located at the end of the activity example (shown in blue) is used for testing the algorithm while the varying amounts of the 75% of the data at the start of the activity are used for training (shown in red). Unused segments of data per experiment are shown in white.

Percentage of	Percentage of	Length of activity training examples in minutes							
training data used	total activity length	Lying down	Postures	Physically demanding activities	Rest of Activities				
-	100%	5	2	1.5 - 2.0	3 - 3.5				
100%	75%	3.7	1.5	1.1 – 1.5	2.2 - 2.6				
80%	60%	3.0	1.2	0.9 - 1.2	1.8 - 2.1				
60%	45%	2.2	0.9	0.6 - 0.9	1.3 - 1.5				
40%	30%	1.5	0.6	0.4 - 0.6	0.9 - 1.0				
20%	15%	0.7	0.3	0.2 - 0.3	0.4 - 0.5				
10%	7.5%	0.3	0.15	0.1 - 0.15	0.2 - 0.3				

Table 5-31: Percentage of training data used in comparison with the average total amount of data available per activity and corresponding duration (in minutes) of training data with respect to some activity categories.

activity	CV	100%	80%	60%	40%	20%	10%
Category							
All	$87.9\pm2.0$	$80.8\pm3.4$	$76.2\pm4.7$	$70.5\pm4.7$	$63.7\pm5.5$	$48.2\pm4.8$	$34.4\pm4.7$
Postures	96.9±4.1	96.6±11.9	95.3±12.9	87.2±23.9	76.8±34.6	42.1±34.6	24.3±1.5
	(0.1±0.1)	(0.2±0.3)	(0.3±0.4)	(0.4±0.6)	$(0.7 \pm 1.1)$	(1.5±1.2)	(2.7±1.1)
Ambulation	88.8±8.4	92.1±12.8	90.7±11.9	83.1±16.6	69.0±25.4	37.0±30.4	0.0±0.0
	(0.2±0.2)	(0.3±0.4)	(0.2±0.3)	$(0.4\pm0.5)$	(0.7±0.7)	(0.9±0.9)	$(0.0\pm0.0)$
Exercise	92.6±7.7	84.2±20.4	80.8±24.4	75.9±25.3	67.8±29.8	56.2±31.3	47.7±33.5
	(0.1±0.1)	(0.3±0.4)	(0.4±0.5)	(0.6±0.6)	(0.7±0.7)	$(0.9\pm0.8)$	(1.6±1.4)
Resistance	88.0±9.5	78.4±23.2	73.7±27.1	68.3±27.5	58.9±31.0	44.7±30.8	37.3±29.6
Exercise	(0.2±0.2)	(0.4±0.5)	(0.6±0.6)	(0.8±0.7)	(0.9±0.8)	$(1.2\pm1.0)$	$(1.8\pm1.4)$
Household	80.6±9.0	73.1±21.1	69.2±23.9	65.0±25.6	59.0±29.6	43.9±29.3	28.4±26.8
	(0.4±0.2)	(0.6±0.6	$(0.8\pm0.8)$	$(0.9\pm0.9)$	(1.0±0.9)	$(1.5 \pm 1.4)$	$(1.9\pm1.5)$

Table 5-32: True positive rate and false positive rate (shown in parenthesis) during subject dependent recognition of activities when varying amounts of training data are used. The target activities are the 51 activities contained in the MIT dataset without including the *unknown* class. CV stands for 10-fold stratified cross-validation. All seven accelerometers are utilized in this experiment.

One possible way to force the activity examples to have the same length would be to cut their length so it is the same as the one for the activity example with shortest duration. However, this procedure was not followed during this analysis because some physically demanding activities such as *bench weight lifting hard*, *cycling at 80rmp hard*, *running at 6mph*, *sit-ups*, and *crunches* have very short durations ranging from 20s to less than a minute. Reducing the duration of all activity examples to this short duration would make the analysis performed in this section difficult do to the unavailability of enough data. Another possibility could be to eliminate the short duration activities from the analysis



Figure 5-28: True positive rate during subject dependent recognition of activities when varying amounts of training data are used. The target activities are the 51 activities contained in the MIT dataset without including the *unknown* class. In this experiment, all seven accelerometers are utilized.

but this would prevent the presentation of results for these activities. Table 5-31 presents the average amount of training data available per activity in minute so that the results presented in this section can be better interpreted.

Appendix A4 presents the average amount of training data available per activity in the MIT energy expenditure dataset. Table 5-31 shows a summary of this information as well as the duration of each activity example as a function of the percentage of training data used. It can be seen that the total amount of data available for each activity is on average 5min for *lying down* since heart rate and resting metabolic rate were measured during this activity, 2min for the remaining the postures, 1.5-2.0min for some physically demanding activities such as *bench weight lifting hard*, *cycling at 80rmp hard*, *running at 6mph*, *situps*, and *crunches* and 3.0-3.5min for all other activities. Data for some physically demanding activities such as *sit-ups*, *crunches*, and *bench weight lifting hard* were collected over several repetitions of the activity, thus, leading to multiple short duration examples (<1min in length). There are also three examples for the *ascending* and *descending stairs* activities per data collection (participant) because the length of the stairs from the ground floor to the 4th<sup>rd</sup> floor allowed only the collection of 1min of data continuously for these activities.

Table 5-32 and Figure 5-28 present the results of varying the amount of training data while recognizing the 51 activities contained in the MIT dataset when the *unknown* class is not included. Table 5-32 also shows the performance of 10-fold stratified cross-validation (CV) as a comparison baseline. The performance using CV is higher than when using 100% of the training data (75% of activity duration) because there is more training and testing data available during CV as observed in practice. From Table 5-32,

we can observe that, as expected, overall performance and performance per activity category decreases as the amount of training data is decreased. The overall accuracy of 80% while using 100% of the training data suggests that reasonable recognition results can be achieved when just two and three minutes of data for most activities is provided. The activity categories with lower true positive rates are ambulation, resistance exercise, and exercise activity. The poorer performance over these activities is due to the difficulty involved in discriminating among the intensity levels of the activities from accelerometer data, as explained previously, and the smaller amounts of data available for physically demanding activities. The decrease in overall accuracy of only 4.6% when the percentage of training data is reduced from 100% to 80% suggests that the amount of training data can be as little as 2min per activity without significant decrease in performance. The true positive rate overall and per activity category starts decreasing sharply when only 40% of the training data is used. In this setting, there is about 1min of training data for most activities and approximately 0.5min of data for postures and physically demanding activities. Thus, with a window length of 5.6s, there are only between 5 and 10 training examples per activity. It is thus, expected to observe a sharp decrease in performance after this point since having less than 5 examples per activity to train a classifier seems unreasonable.

In conclusion, the experiment suggests that it is possible to provide only 2min of training data per activity and achieve an overall accuracy of 76% when discriminating between 51 activities, and true positive rates per activity category ranging from 69 to 95%. Nevertheless, it might be possible to obtain higher performance even with less training data if discrimination among the intensity levels of an activity is not required or if the number of activities to recognize is reduced. Section 5.5 will later evaluate the real-time performance of subject dependent recognition of activities during a short study where participants provide 2min of training data per activity to recognize.

# **5.4.9.2** What is a Good Compromise on the Number and Placement of Sensors to be Work for Recognition of Activities?

This section evaluates the performance of the activity recognition algorithm when different combinations of sensors (accelerometers) are used during subject dependent and independent training. The main objective is to determine a good compromise on the number and placement of sensors to use, balancing recognition performance with usage comfort and sensor cost.

Table 5-33 and Table 5-34 present the results obtained while recognizing the 51 activities in the MIT dataset (without the *unknown* class) using different combinations of accelerometers with the C4.5 classifier, the *invariant reduced* feature set computed per axis, and windows of 5.6s in length. The *unknown* class is not used during this analysis to prevent its large number of examples (with respect to the other activities) from altering the results obtained. Figure 5-29 and Figure 5-30 present the true positive rate as a grayscale image scaled so that the best true positive rate is shown in white and the worse is shown in black. In other words, the image highlights the difference in performance per activity category by showing good areas of performance in white and poor areas of performance in black. The prefix "D" in the sensor location label (e.g. DWrist) used in

Sensor Combination	All	Postures	Ambulation	Exercise	Resistance	Household
					Exercise	
All sensors	$87.9\pm2.0$	96.9±4.1	88.8±8.4	92.6±7.7	88.0±9.5	80.6±9.0
		(0.1±0.1)	(0.2±0.2)	(0.1±0.1)	(0.2±0.2)	(0.4±0.2)
Hip + DWrist + DFoot	$86.3\pm1.6$	96.6±4.2	$87.8 \pm 8.5$	$90.5 \pm 8.4$	85.4±9.9	79.0±9.7
		$(0.1\pm0.1)$	(0.3±0.2)	$(0.2\pm0.1)$	(0.3±0.2)	(0.5±0.3)
Hip + DWrist	$83.6\pm2.3$	95.7±4.3	86.5±9.2	86.3±12.9	81.6±13.2	75.4±10.4
		(0.1±0.1)	(0.3±0.2)	$(0.2\pm0.2)$	(0.3±0.2)	(0.6±0.3)
Hip + DFoot	$84.8\pm2.3$	96.6±3.8	87.1±8.4	90.4±8.3	84.4±10.9	76.0±11.1
		(0.1±0.1)	(0.3±0.2)	$(0.2\pm0.1)$	(0.3±0.2)	(0.6±0.3)
DWrist + DThigh	$81.6\pm2.6$	95.5±5.8	81.5±10.6	87.5±9.9	80.6±11.5	72.6±10.6
		(0.1±0.1)	(0.4±0.2)	$(0.2\pm0.2)$	(0.4±0.2)	(0.6±0.3)
DWrist + DFoot	$81.1 \pm 2.1$	$95.8 \pm 4.1$	79.9±9.6	86.9±10.5	78.2±13.1	73.6±10.7
		(0.1±0.1)	(0.4±0.2)	$(0.2\pm0.2)$	(0.4±0.3)	(0.6±0.3)
Hip	$80.5 \pm 2.7$	94.8±5.5	86.5±8.8	85.7±12.9	79.2±14.3	68.7±11.8
		(0.1±0.1)	(0.3±0.2)	(0.2±0.2)	(0.4±0.3)	(0.7±0.3)
DWrist	$70.7\pm3.7$	91.0±8.6	73.5±11.8	73.1±14.1	65.0±15.7	61.5±12.7
		(0.1±0.1)	(0.6±0.3)	(0.5±0.3)	(0.7±0.3)	(0.9±0.3)
DFoot	$74.8\pm3.7$	93.3±7.7	73.6±11.0	84.1±12.3	72.4±15.7	63.4±14.8
		(0.1±0.1)	(0.6±0.3)	(0.3±0.2)	(0.5±0.3)	(0.8±0.4)
DUpperArm	$74.4\pm3.9$	88.0±9.8	81.7±10.4	79.1±12.0	69.8±14.9	61.2±12.6
		(0.2±0.2)	(0.4±0.2)	(0.4±0.2)	(0.6±0.3)	(0.9±0.3)
DThigh	$74.6 \pm 2.9$	93.1±7.1	76.1±12.3	84.8±11.9	73.9±15.1	59.4±13.3
		$(0.1\pm0.1)$	(0.5±0.3)	$(0.3\pm0.2)$	(0.5±0.3)	$(0.9\pm0.4)$

Table 5-33: Performance of the C4.5 classifier using the invariant reduced feature set computed per
axis over windows of 5.6s in length using different subsets of accelerometers while recognizing the 51
activities contained in the MIT dataset in a subject dependent manner (without including the
unknown class). The guessing accuracy is 1.96% for all the activity categories shown in the table.

Sensor Combination	All	Postures	Ambulation	Exercise	Resistance	Household
					Exercise	
All sensors	$50.6\pm5.2$	77.0±24.0	$46.6 \pm 27.1$	46.6±31.6	34.1±29.9	43.8±25.6
		(0.5±0.7)	$(1.2\pm1.0)$	(0.9±0.9)	$(1.2\pm1.0)$	(1.3±0.9)
Hip + DWrist + DFoot	$46.6\pm7.5$	66.9±32.3	39.6±27.3	39.2±31.1	30.9±28.6	42.9±24.0
		(0.6±1.0)	(1.3±1.1)	(1.0±0.9)	$(1.2\pm1.1)$	(1.3±1.0)
Hip + DWrist	$42.7\pm5.9$	46.1±34.3	37.9±27.3	35.4±27.1	29.1±23.7	42.9±23.5
		(1.0±1.1)	(1.4±1.3)	$(1.1\pm1.0)$	$(1.2\pm1.0)$	(1.3±0.9)
Hip + DFoot	$41.0\pm6.5$	57.5±33.6	35.6±28.3	39.1±31.7	28.5±29.2	32.4±22.5
		$(0.8\pm1.1)$	(1.3±1.2)	$(1.0\pm1.0)$	$(1.2\pm1.1)$	$(1.6\pm1.1)$
DWrist + DThigh	$46.5\pm4.9$	55.4±34.2	44.1±24.1	45.6±29.5	33.0±27.3	39.7±21.7
		(0.8±0.9)	(1.2±0.9)	$(0.8\pm0.8)$	(1.1±0.9)	(1.4±0.9)
DWrist + DFoot	$44.0\pm7.3$	62.8±29.5	37.9±25.0	36.5±28.1	30.4±24.9	40.1±23.4
		(0.6±0.9)	(1.3±1.1)	(1.0±0.9)	(1.1±0.9)	(1.5±1.0)
Hip	$36.2\pm6.2$	40.1±27.7	37.5±26.3	35.3±29.1	26.5±24.9	28.2±20.2
		(1.0±1.0)	$(1.4 \pm 1.1)$	$(1.0\pm1.0)$	$(1.2\pm1.0)$	(1.7±0.9)
DWrist	$36.6\pm7.6$	35.3±33.1	32.6±26.2	31.7±23.6	28.3±21.8	36.2±20.1
		(1.0±1.0)	(1.3±1.1)	(1.1±0.8)	(1.2±0.8)	(1.5±1.0)
DFoot	$33.0 \pm 5.4$	46.9±32.1	32.6±20.2	33.7±22.9	24.7±21.9	22.6±16.8
		(1.1±1.4)	$(1.4 \pm 1.1)$	(1.1±0.9)	(1.2±0.9)	(1.7±1.0)
DUpperArm	$37.2 \pm 3.5$	39.2±24.3	42.3±23.0	43.4±22.9	28.2±20.9	24.2±16.5
		(1.0±0.9)	(1.3±1.0)	(0.9±0.9)	$(1.2\pm1.0)$	(1.8±0.9)
DThigh	$29.0\pm4.6$	19.4±20.4	38.4±23.8	40.4±27.6	23.7±23.8	17.3±14.7
-		$(1.4 \pm 1.3)$	$(1.3\pm0.9)$	$(1.0\pm1.1)$	$(1.3\pm1.2)$	$(2.0\pm1.3)$

Table 5-34: Performance of the C4.5 classifier using the *invariant reduced* feature set computed per axis over windows of 5.6s in length using different subsets of accelerometers while recognizing the 51 activities contained in the MIT dataset in a subject independent manner (without including the *unknown* class). The guessing accuracy is 19.6% for all the activity categories shown in the table.



Figure 5-29: True Positive Rate per sensor combination using the C4.5 classifier when features are computed per axis over windows of 5.6s during subject dependent evaluation. The grayscale image is scaled so that the maximum true positive rate of 96.9% is represented by white and the minimum of 59.4% by black. In other words, poor areas of performance are shown in black.



Figure 5-30: True Positive Rate per sensor combination using the C4.5 classifier when features are computed per axis over windows of 5.6s during subject independent evaluation. The grayscale image is scaled so that the maximum true positive rate of 77.0% is represented by white and the minimum of 17.3% by black. In other words, poor areas of performance are shown in black.

the tables and figures stands for "dominant" and implies that the sensor was worn at the dominant limb of the subject (e.g. right wrist if subject was right handed). Refer to Figure 4-4 for an image of the placement of sensor during the data collections.

The results show that the three sensor subsets with best performance (after the one using all the sensors) during subject dependent evaluation are the Hip+DWrist+DFoot, Hip+DWrist, and Hip+DFoot. For subject independent evaluation, the best three subsets are Hip+DWrist+DFoot, DWrist+DThigh, and DWrist+DFoot. Therefore. the Hip+DWrist+DFoot combination has the higher performance overall and per activity for both, subject dependent and independent training. The performance of this sensor combination drops only 1.6% during subject dependent training and 4% during subject independent training with respect to the performance obtained using all seven sensors. The fact that the Hip+DWrist+DFoot sensor combination achieves the best performance intuitively makes sense, since this sensor combination captures upper body motion (DWrist), lower body motion (DFoot), and overall body motion at the Hip. The other sensor combinations explored do not capture motion information at least in one the aforementioned locations (upper, lower or overall). When the Hip+DWrist+DFoot sensor combination is used, the activity categories that suffer the greatest decrease in performance (with respect to the all sensors combination) are ambulation (1-7%), exercise (2.1-7%) and resistance exercise (2.6-7.4%). The reason is that activities involving resistance or load effort such as bench weight lifting, bicep curls, cycling at 80rpm (different resistance levels), walking at 3mph (different inclinations), and rowing at 30spm (different resistance levels) are better recognized when more sensors are used. As explained in previous sections, changes in resistance or load induce changes in the motion signature of an activity (e.g. execution speed) that allow discrimination between the intensity levels. When the number of sensors is decreased, these slight differences in motion are more difficult to detect, leading to poorer performance on these activities. The performance per activity for the Hip+DWrist and Hip+DFoot sensor combinations was analyzed to determine why these combinations are the two second best performing ones during subject dependent training even when they do not measure both lower body or upper body motion (other than through the hip . It was found that for the Hip+DWrist sensor combination, the performance over activities involving lower body motion such as cycling, sitting fidgeting feet and legs, ascending and descending stairs indeed decreases, particularly during subject independent training. However, during subject dependent training, the decrease in performance for these activities ranges only from 1% to 8%. It appears that during subject dependent training, the small changes induced in the acceleration signal at the hip while performing lower body activities are enough to allow good performance over these activities. For the Hip+DFoot sensor combination, it was found that the performance over activities involving upper body motion such as *bench* weight lifting, doing dishes, playing video games, scrubbing a surface, typing, writing, washing windows, wiping and dusting was also lower. Nevertheless, the performance on some activities also including upper body motion such as *ironing* and *sitting fidgeting* hands and arms did not change. Again, it appears that changes in the acceleration signal at the hip induced while performing upper body activities are enough to allow good performance over some of these activities.

During subject independent training, the two second best performing sensor combinations were DWrist+DThigh and DWrist+DFoot. Both sensor combinations place

one sensor at the upper body and one sensor at the lower body. In particular, the performance of the DWrist+DThigh sensor combination is as good as the performance obtained for the Hip+DWrist+DFoot sensor combination (0.04% performance difference). The accelerometer located at the thigh (DThigh) provides more reliable information than the accelerometer located at the foot (DFoot) because it experiences less variation in the signal. The foot has a higher degree of motion (or freedom) than the thigh and consequently, produces more variable signals during ambulation, exercise, and resistance exercise.

Unfortunately, the thigh is a problematic location to wear a sensor on because the sensor can be easily knocked off during ambulation and because the attachment of a sensor at this body location is difficult. For example, if bandages are used, they tend to loosen up over time and slip from this body location. If other elastic materials are used, they need to be placed so tight that they might become uncomfortable to wear.

Finally, when the performance of single sensors is analyzed during subject dependent training, it is found that the sensors with best performance in decreasing order are Hip, DFoot, and DThigh. The sensor at the hip has a consistently higher performance per activity category than the DFoot and DThigh sensors. The performance per activity category for the DFoot and DThigh sensors is very similar. From Table 5-33, it can be observed that the activity category that suffers the largest decrease in performance when single sensors are used to recognize activities is household activities. This is an expected result, since household activities involve a high degree of variation in upper body and lower body motion during their execution. During subject independent training, the single sensors with higher performance in decreasing order of importance are DUpperArm, DWrist, and DHip. The difference in overall accuracy when using these sensors is less than 1% though. The single sensor at the hip has more problems recognizing resistance exercise activities than the DUpperArm sensor but it is slightly better at recognizing household activities. The DWrist sensor on the other hand, is better at recognizing household activities than the DUpperArm and Hip sensors. This is because most household activities involve some degree of upper body motion. Finally, a note of caution is required while interpreting the relatively good performance obtained during subject dependent evaluation of the single sensors placed at the Hip, DFoot, and DThigh. During the real-time study performed in Section 5.5, it was found that the performance over these sensors is not as good as the one implied by the results obtained during the experiments performed in this section. The results of the real-time evaluation indicate, as expected, that the sensor placed at the hip has difficulties recognizing activities involving upper body and non-ambulatory lower body activities. Similarly, the sensors at the dominant foot and thigh present difficulties recognizing upper body activities. Consequently, the good performance obtained for these sensors in this section might be a result of overfitting, even when stratified crossvalidation was utilized to evaluate their performance during subject dependent training.

In summary, it was found that the sensor combination Hip+ DWrist+DFoot obtains a performance close to the one obtained using all the seven sensors during subject dependent and independent evaluation. This sensor combination was also found to consistently outperform the other sensor combinations explored in this section. This is an intuitive result since this sensor combination captures upper, lower, and overall body motion. The two second best performing sensor combinations during subject dependent

training were the ones placing a sensor at the hip and another one either at the upper body (Hip + DWrist) or lower body (Hip + DFoot). During subject independent training, the two second best performing sensor combinations were the ones placing one sensor at the upper body (DWrist) and another sensor at the lower body (e.g. DThigh or DFoot). Based on these experiments, for this set of activities, wearing three sensors located at the hip, dominant wrist, and dominant foot seems to provide a reasonable compromise between performance and ease-of-use. During free-living, the sensor at the wrist could be worn as a bracelet, the sensor at the hip as a clip on addition to the belt or pants, and the sensor at the foot as a shoe pod with clip on capabilities so it can be conveniently attached to the shoe laces. Furthermore, in the future, these sensors could be miniaturized and embedded in wristwatches, shoes, and perhaps, even in the elastic material surrounding the hip utilized in underwear (a convenient location for the hip sensor provided it is small enough to be comfortable to wear).

### 5.4.9.3 How Well Can All Activities be Recognized Without Differentiating Among Intensity Levels?

In the previous section, it was found that some of the activities presenting the lowest performance were activities involving different intensity levels, particularly when the intensity changes due to the use of different resistance levels or work loads. As a result, this section explores how well activities can be recognized without differentiating among the intensity levels of an activity. This is performed by merging all activities with different intensity levels into a single activity. For example, all the intensity levels for *cycling* at different speeds and resistance levels are merged into a single class called *cycling*. The same procedure is repeated for *walking*, *running*, *rowing*, *bench weight lifting*, *bicep curls*, and *sitting*, leaving a total of 31 activities to be recognized. This gives a guessing accuracy of 3.2% when the *unknown* class is not incorporated and 3.1% when it is. The column labeled as "All Activities with No Intensity" in Appendix A2 shows a detailed list of the activities whose intensity levels were merged. The activity recognition algorithm used uses the following parameters: the C4.5 classifier, the *invariant reduced* feature set, feature computation per axis, and a sliding window length of 5.6s.

Table 5-35 presents the performance obtained during subject dependent and independent training while the *unknown* class is included and when it is not. The highest performance overall and per class is again obtained during subject dependent training when the *unknown* class is not used. When the *unknown* class is incorporated, overall performance drops 8% for subject dependent training, and 12% for subject independent training. The decrease in performance during subject dependent training is similar to the one obtained while recognizing the intensity level of activities (7%). For subject independent training, however; the decrease in performance is three times greater than the one obtained when intensity levels are being recognized. Note that during these comparisons, decreases in performance are computed when the number of activities being recognized is the same (51 for the previous section and 31 for this section), so the difference in random guessing is not relevant. This is because the *unknown* class is being confused more with *walking* and *sitting* now that their intensity levels have been merged. During the *unknown* activity (segments of data not labeled during the data collection),

Evaluation Method	All	Postures	Ambulation	Exercise	Resistance	Household
					Exercise	
Subject dependent	$91.4\pm1.6$	97.3±2.9	94.4±4.8	97.9±2.9	92.4±4.8	79.5±9.1
without Unknown class		(0.1±0.1)	(0.2±0.2)	(0.1±0.1)	$(0.2\pm0.2)$	(0.5±0.2)
Random guessing: 3.2%						
Subject dependent	$83.3\pm1.9$	93.9±5.0	87.5±7.2	94.2±4.7	87.3±6.6	73.1±10.4
with unknown class		$(0.2\pm 0.1)$	(0.6±0.2)	(0.2±0.1)	(0.4±0.2)	(0.4±0.2)
Random guessing: 3.1%						
Subject independent	$72.0\pm5.7$	83.9±13.5	78.4±20.3	82.0±25.0	75.6±21.3	43.5±25.9
without unknown class		(0.6±0.8)	(0.8±0.7)	(0.4±0.7)	(0.8±0.9)	(1.2±0.9)
Random guessing: 3.2%						
Subject independent	$59.9\pm6.6$	60.4±33.5	63.1±24.9	67.6±30.6	65.2±25.1	34.8±24.0
with unknown class		(0.5±0.6)	(1.0±0.6)	(0.3±0.3)	(0.8±0.5)	(0.8±0.5)
Random guessing: 3.1%						

Table 5-35: True positive rate and false positive rate (shown in parenthesis) of the C4.5 classifier when recognizing 31 activities without intensity levels during subject dependent and independent evaluation. The feature set used is the *invariant reduced* feature set computed per axis over windows of 5.6s in length over all the seven accelerometers. The probability of random guessing for all activity categories presented in this table is 3.1% when the *unknown* class is used and 3.2% when it is not.

subjects were *standing* or *sitting* while resting, or *walking* while transitioning from one activity to another (e.g. transitioning from one exercise machine to another during the gym data collection or transitioning among rooms during the household data collection).

The improvement in performance of +21% obtained for subject independent training when the intensity levels of activities are not being recognized is substantial although this comparison is unfair since the probability of random guessing has increased from 1.96% to 3.2% now that intensity levels are not recognized. The improvement during subject dependent training; however, is only +3.5%. As explained in the previous section, this is because during subject dependent training, the recognition algorithm is able to differentiate well between the intensity levels of an activity because changes in resistance level or work load induce changes in the motion signature of activities that are detected by the accelerometers.

Table 5-29 shows the confusion matrices for subject dependent and independent training when the *unknown* class is not used. The grayscale images were normalized to highlight the activities being confused the most (shown in black), and the ones confused the least (shown in white). For subject dependent training, the activities confused the most are *sweeping* with *mopping*, since both include similar upper body motion and walking, and *making the bed* with *taking out trash*, since both include upper body motion and sequences of standing and walking. The confusion matrix for subject dependent training also shows that most of the confusions happen among household activities. As explained before, they involve high degree of variability in their motion and performance as well as sequences of standing and walking behaviors. Finally this confusion matrix also shows that *walking* is being slightly confused with *ascending* and *descending stairs*. This is expected due to the similarity of motion involved in these activities.

The confusion matrix for subject independent training shows that the activities being confused the most are activities with similar postures such as *sitting*, *watching TV*, and *playing video games*, and activities involving similar posture and upper body motion such as *writing*, *typing*, and *playing video games*. Moreover, *walking* is also being confused with *running* and with *descending stairs*. This is expected since they all involve similar motion patterns that can be confused, particularly during subject independent evaluation.



Figure 5-31: Confusion matrices for the C4.5 classifier when recognizing 31 activities without intensity levels and without the *unknown* class during (a) subject dependent and (b) independent evaluation. The feature set used is the invariant *reduced* feature set computed per axis over windows of 5.6s in length. The maximum number of errors in a given cell of a confusion matrix is 73 and 169 for subject dependent and independent training respectively.

Sensor Combination	All	Postures	Ambulation	Exercise	Resistance	Household
					Exercise	
All sensors	$91.4\pm1.6$	97.3±2.9	94.4±4.8	97.9±2.9	92.4±4.8	79.5±9.1
		(0.1±0.1)	(0.2±0.2)	(0.1±0.1)	(0.2±0.2)	(0.5±0.2)
Hip + DWrist + DFoot	$90.2 \pm 1.6$	97.6±3.1	92.4±6.3	97.3±2.9	91.3±5.6	77.1±10.4
		(0.1±0.1)	(0.3±0.2)	(0.1±0.1)	(0.3±0.2)	(0.5±0.3)
Hip + DWrist	$88.4 \pm 2.2$	97.0±4.2	90.7±6.1	96.9±3.3	89.2±5.2	73.1±10.7
_		(0.1±0.1)	(0.3±0.2)	(0.1±0.1)	(0.3±0.2)	(0.6±0.3)
Hip + DFoot	$88.9 \pm 1.9$	97.2±3.2	92.2±5.9	97.4±3.2	90.3±6.4	73.8±11.6
		(0.1±0.1)	(0.3±0.3)	(0.1±0.1)	(0.3±0.2)	(0.6±0.3)
DWrist + DThigh	$87.8 \pm 2.2$	96.8±3.5	87.9±5.4	97.7±3.9	88.3±6.8	71.3±10.9
_		(0.1±0.1)	(0.4±0.2)	(0.1±0.1)	(0.3±0.2)	(0.7±0.3)
DWrist + DFoot	$87.4 \pm 2.1$	96.8±3.3	85.4±6.4	96.5±3.8	88.4±6.9	71.2±11.7
		(0.1±0.1)	(0.4±0.2)	(0.1±0.1)	(0.3±0.2)	(0.7±0.3)
Hip	$85.9\pm2.2$	95.1±5.3	91.0±6.0	96.9±3.4	86.2±6.1	66.6±11.9
		(0.1±0.1)	(0.4±0.2)	(0.1±0.1)	(0.4±0.3)	(0.7±0.3)
DWrist	$80.4 \pm 3.4$	90.9±7.6	79.9±8.8	93.4±5.5	82.1±7.3	58.2±12.8
		$(0.2\pm 0.2)$	$(0.8\pm0.4)$	(0.3±0.2)	(0.7±0.3)	(0.9±0.4)
DFoot	$83.4 \pm 3.1$	94.5±6.8	84.1±8.0	95.5±4.8	85.3±8.8	62.3±15.1
		(0.1±0.1)	(0.5±0.3)	(0.2±0.2)	(0.4±0.3)	(0.9±0.4)
DUpperArm	$82.6\pm3.9$	88.9±8.4	88.6±6.1	96.1±4.2	84.1±7.2	59.3±14.5
		(0.3±0.2)	(0.4±0.2)	(0.2±0.2)	(0.5±0.3)	(0.9±0.4)
DThigh	$82.9 \pm 2.8$	94.8±6.2	86.6±5.9	97.7±3.3	83.9±7.5	58.8±13.1
		$(0.2\pm0.2)$	$(0.4\pm0.2)$	$(0.1\pm0.1)$	$(0.4\pm0.2)$	$(0.9\pm0.4)$

Table 5-36: Performance of the C4.5 classifier while recognizing 31 activities with no intensity levels and without the *unknown* class using different sensor combinations during subject dependent training. The feature set used is the *invariant reduced* feature set computed per axis over windows of 5.6s in length.

Sensor Combination	All	Postures	Ambulation	Exercise	Resistance Exercise	Household
All sensors	72.0 ± 5.7	83.9±13.5	78.4±20.3	82.0±25.0	75.6±21.3	43.5±25.9
		(0.6±0.8)	(0.8±0.7)	(0.4±0.7)	(0.8±0.9)	(1.2±0.9)
Hip + DWrist + DFoot	$67.6 \pm 10.6$	72.5±28.1	67.8±24.2	71.5±30.9	69.6±26.1	43.2±25.0
_		(1.0±1.5)	(1.3±1.3)	(0.5±0.7)	(0.9±0.8)	(1.3±1.0)
Hip + DWrist	$64.5\pm8.7$	46.9±26.9	65.1±26.3	70.0±30.0	67.0±25.8	43.1±22.8
		(1.3±1.2)	(1.3±1.1)	(0.8±1.3)	(1.2±1.1)	(1.3±0.9)
Hip + DFoot	$61.0\pm8.4$	60.2±30.9	64.2±23.1	70.1±32.1	63.2±26.2	31.4±22.8
		(1.3±1.5)	(1.3±1.1)	(0.5±0.7)	$(1.1 \pm 1.0)$	(1.6±1.3)
DWrist + DThigh	$67.5\pm5.7$	63.4±29.8	72.2±19.7	77.3±27.7	70.4±21.0	39.5±22.2
		$(1.2\pm1.2)$	(1.0±1.0)	(0.4±0.7)	(0.9±0.7)	(1.4±0.8)
DWrist + DFoot	$65.5\pm9.6$	$75.2\pm25.0$	65.4±22.0	65.5±26.3	68.8±23.9	39.9±23.9
		$(1.0\pm1.2)$	(1.3±1.1)	(0.6±1.0)	(1.0±0.9)	(1.4±0.9)
Hip	$55.8\pm9.3$	41.9±23.4	64.9±25.2	62.1±30.9	58.0±27.7	27.2±18.9
		(1.6±1.2)	(1.4±1.0)	$(0.9\pm1.4)$	(1.5±1.3)	(1.7±1.0)
DWrist	$57.4 \pm 10.4$	44.1±32.6	57.1±24.6	65.3±29.0	$62.8 \pm 24.9$	34.9±18.9
		(1.5±1.3)	(1.7±1.1)	(1.0±1.1)	(1.4±1.0)	(1.5±1.0)
DFoot	$52.9\pm6.4$	$54.7 \pm 32.0$	57.5±21.0	55.3±24.2	57.4±24.3	21.5±17.5
		(1.9±2.0)	(1.6±1.3)	(1.1±1.2)	(1.4±1.1)	(1.7±1.0)
DUpperArm	$57.5 \pm 4.3$	$40.4 \pm 20.8$	79.9±20.2	72.3±28.5	63.9±20.9	22.9±16.6
		(1.4±1.2)	(0.8±0.8)	(0.8±1.1)	(1.4±1.1)	(1.7±1.0)
DThigh	$49.0\pm5.9$	28.0±23.6	68.1±20.1	65.5±27.7	52.8±19.4	16.2±13.8
		$(1.9\pm1.4)$	$(1.1\pm1.0)$	$(1.0\pm1.3)$	$(1.6\pm1.2)$	$(2.0\pm1.4)$

Table 5-37: Performance of the C4.5 classifier while recognizing 31 activities with no intensity levels and without the *unknown* class using different sensor combinations during subject independent training. The feature set used is the *invariant reduced* feature set computed per axis over windows of 5.6s in length.



Figure 5-32: True Positive Rate per sensor combination using the C4.5 classifier when features are computed per axis over windows of 5.6s during subject dependent evaluation. The grayscale image is scaled so that the maximum true positive rate of 97.9% is represented by white and the minimum of 58.2% by black. In other words, poor areas of performance are shown in black.





As in subject dependent training, most confusions also happen among household activities.

Figure 5-38 and Figure 5-39 present the true positive and false positive rate obtained when different combinations of sensors are used to recognize activities. As in the previous section, Figure 5-38 and Figure 5-39 present the true positive rate as a grayscale image, scaled to highlight the differences in performance. The best performance is shown in white while the worse performance is shown in black. From these tables and figures, it can be observed that the best performance overall and per class is obtained using the Hip+DWrist+DFoot sensor combination. The decrease in performance for this sensor combination with respect to the performance obtained using all the sensors is 1.2% for subject dependent training and 4.4% for subject independent training. Again, the performance decreases more for subject independent training because more sensors are required to capture inter-individual variations in the way activities are performed. The second best performing sensor combinations for subject dependent training are Hip+DWrist, and Hip+DFoot. These are the same sensor combinations with higher performance found in the previous section when the intensity level of activities was being discriminated. For subject independent training, the sensor combinations with higher performance are also the same: DWrist+DThigh, and DWrist+DFoot. The performance of the combination DWrist+DThigh is also very close to the performance obtained with the Hip+DWrist+DFoot sensor combination (0.1% difference). So far, it can be concluded that the best sensor combination for recognizing activities during subject dependent and independent training is Hip+DWrist+DFoot. Nonetheless, during subject independent training, the DWrist+DThigh sensor combination can be also used with a very little decrease in overall performance. The main difference between these sensor combinations is that the DWrist+DThigh sensor combination achieves a higher performance ranging from +0.8 to +4.4% for the ambulation, exercise, and resistance exercise activity categories. This can be explained by the fact that the DThigh sensor experiences less variability than the DFoot sensor while activities are performed. The Hip+DWrist+DFoot sensor combination on the other hand, is significantly better at recognizing postures (+9% better), and slightly better at recognizing household activities (+3.7% better). It seems that the sensor at the hip is responsible for the better performance over these activities.

When the performance of single sensors is analyzed from Figure 5-32 and Figure 5-33, it can be seen that again, the activities that the activities with lower performance are household activities. The single sensors with higher overall performance in decreasing order are Hip, DFoot, and DThigh during subject dependent training and DUpperArm, DWrist, and Hip during subject independent training. This is also the same ordering of sensors found in the previous section, when the intensity level of activities is being recognized. The relatively good single sensor performance of the sensor at the hip during subject dependent training is due to the fact that upper body and lower body activity induce small changes in the acceleration signal at this sensor that allow the recognition of some of these activities. The performance of the DUpperArm sensor during subject independent training can be explained by the fact that this sensor is able to sense upper body motion with less variability than the sensor at the wrist, and that it can also detect changes in posture almost as well as the sensor at the hip. The sensor at the hip has the advantage of recognizing household activities slightly better (4.3%) than the DUpperArm

sensor because the overall body motion experienced in some activities *stacking groceries*, *gardening* and *weeding* is better detected at this body location. The DWrist sensor is also better than the DUpperArm sensor in recognizing household activities and postures during subject independent training.

In conclusion, when the intensity level of activities is not of interest, the activities in the MIT dataset can be recognized with an overall accuracy of 91.4% during subject dependent training and with an accuracy of 72% during subject independent training. The improvement in performance is higher (+21%) during subject independent training. The best sensor combination to use to recognize these activities is again the Hip+DWrist+DFoot combination. If a single sensor is to be used to recognize activities, its location depends on the activities of interest. For example, if ambulatory, exercise, or resistance exercise activities are to be recognized the best single sensor to utilize is the DUpperArm. If postures and household activities need to be recognized, the DWrist sensor seems to be the best one to utilize. Finally, the sensor that provides average performance over all activity categories is the sensor at the hip. These results need to be interpreted with caution, since the location of the sensors to use strongly depends on the activities being recognized. The relatively higher performance obtained for the DUpperArm and DWrist sensors with respect to the performance of the Hip sensor might be due to the fact that most activities contained in the MIT dataset involve upper body motion.

# 5.4.9.4 How Well Can Postures, Ambulation, and Two MET Intensity Categories be Recognized?

This section explores how well postures, ambulation, and two MET intensity categories can be recognized utilizing all and several sensor combinations during subject dependent and independent training. The activity recognition algorithm is the same as the one used in the previous sections and has the same parameters: the C4.5 classifier, the invariant reduced feature set, feature computation per axis, and a sliding window length of 5.6s. The activities explored in this section are 11 in total and consist of the following activities: Lying down, standing, sitting, kneeling, walking at 2mph and 3mph, running at 4, 5, and 6mph, and the *moderate* and *vigorous* MET intensity categories. Appendix A2 shows a detailed list of the activities that were merged into the *moderate* and *vigorous* intensity categories according to their associated number of METs from the Compendium of Physical Activities [122]. The random guessing probability for these activities is 9%. Exploring this set of activities and these two MET intensity categories (moderate and vigorous) makes sense from a medical point of view for the following reasons: (1) Most daily energy expenditure is spent in sedentary or light activities such as postures and ambulation. Consequently, if these activities are well recognized in a subject independent manner, better energy expenditure algorithms can be created that estimate energy depending on the activity being performed. Furthermore, if walking and running speeds are also recognized, one might expect further improvements in the estimation of energy expenditure. (2) When medical interventions are designed to foster an increase in physical activity levels, it is important to know if the target population is exercising at *moderate* or *vigorous* intensity levels. If they are, there may be no need for an

<b>Evaluation Method</b>	All	Postures	Ambulation	Moderate	Vigorous	Unknown
Subject dependent	$96.5 \pm 1.1$	96.5±3.8	96.0±4.5	96.4±1.7	93.3±3.5	-
without unknown class		(0.1±0.1)	(0.1±0.1)	(2.4±1.2)	(1.3±0.6)	
Random guessing: 9%						
Subject dependent	$89.5 \pm 1.5$	94.1±5.9	88.2±8.2	84.0±3.0	89.6±4.1	$89.7 \pm 2.1$
with unknown class		(0.1±0.1)	(0.3±0.1)	(3.8±0.8)	(1.3±0.4)	$(8.7 \pm 1.7)$
Random guessing: 8.3%						
Subject independent	$81.3 \pm 4.7$	91.3±9.3	64.4±33.6	86.8±5.4	66.0±15.4	-
without unknown class		(0.5±0.8)	(1.0±1.1)	(13.0±5.4)	(5.3±2.9)	
Random guessing: 9%						
Subject independent	$70.8 \pm 5.2$	72.3±25.0	49.5±32.8	58.3±8.4	59.3±13.9	$82.4 \pm 3.3$
with unknown class		(0.4±0.5)	(0.7±0.7)	(8.0±1.9)	(2.6±0.8)	$(29.9 \pm 7.7)$
Random guessing: 8.3%						

Table 5-38: Performance of the C4.5 classifier when recognizing postures, ambulation, and two MET intensity categories when the *unknown* class is included and when it is not. The feature set used is the *invariant reduced* feature set computed per axis over windows of 5.6s in length over all the seven accelerometers during. The probability of random guessing is 9% when the *unknown* class is not utilized and 8.3% when it is utilized.

	Su	Subject Dependent		Su	bject Independ	ent
Class	TP Rate	FP Rate	F-Measure	TP Rate	FP Rate	F-Measure
Lying down	$99.9\pm0.3$	$0.0\pm0.0$	$99.7\pm0.3$	$99.3 \pm 1.3$	$0.1 \pm 0.4$	$99.1 \pm 1.9$
Standing	$93.3\pm7.0$	$0.1 \pm 0.1$	$94.5\pm6.0$	$94.2 \pm 12.1$	$0.2 \pm 0.7$	$92.8 \pm 13.6$
Sitting	96.1 ± 3.9	$0.2\pm0.2$	$96.2\pm3.2$	$73.7 \pm 19.3$	$1.6 \pm 1.8$	$73.3 \pm 16.8$
kneeling	$96.6 \pm 3.9$	$0.0\pm0.1$	$97.3\pm3.2$	$97.9\pm4.4$	$0.1 \pm 0.2$	$97.4 \pm 5.1$
Walking - Treadmill 2mph -	$97.8\pm2.4$	$0.2\pm0.1$	$96.4\pm2.8$	$75.8\pm34.2$	$0.3 \pm 0.4$	$77.7\pm29.5$
Treadmill 0						
Walking - Treadmill 3mph -	$99.0 \pm 1.2$	$0.2 \pm 0.1$	$98.9\pm0.8$	$89.0\pm23.3$	$0.7 \pm 1.0$	$89.6\pm20.5$
Treadmill 0						
Running - Treadmill 4mph -	$97.7 \pm 2.3$	$0.1\pm0.1$	$97.0\pm2.6$	$53.0\pm36.1$	$1.9 \pm 2.3$	$47.9\pm30.6$
Treadmill 0						
Running - Treadmill 5mph -	94.7 ± 3.3	$0.1 \pm 0.1$	$94.9\pm2.8$	$52.8 \pm 34.1$	$1.4 \pm 1.1$	$47.5 \pm 26.2$
Treadmill 0						
Running - Treadmill 6mph -	$91.0\pm13.2$	$0.1 \pm 0.1$	$91.9\pm10.2$	$51.3 \pm 40.0$	$0.8 \pm 0.9$	$45.0 \pm 33.4$
Treadmill 0						
Moderate	$96.4 \pm 1.7$	$2.4 \pm 1.2$	$96.4 \pm 1.4$	$86.8 \pm 5.4$	$13.0 \pm 5.4$	$83.8 \pm 3.6$
Vigorous	$93.3 \pm 3.5$	$1.3 \pm 0.6$	$93.8 \pm 2.9$	$66.0 \pm 15.4$	$5.3 \pm 2.9$	$69.2 \pm 10.7$

Table 5-39: Performance per activity while recognizing postures, ambulation, and two MET intensity categories using the final implementation of the activity recognition algorithm. The algorithm consists of the C4.5 classifier using the *invariant reduced* feature set computed per axis over windows of 5.6s in length. The probability of random guessing is 9% for all activities shown in this table.

intervention. However, if they are not, it might be important to know what activities are being performed (e.g. postures, ambulation type and intensity) to plan the intervention according. An intervention might encourage the transition from sedentary activities to ambulatory or exercise activities, or an intervention might promote a person to increase in intensity of ambulatory activities during everyday living (e.g. encourage *walking* at 3mph over *walking* at 2mph).

Table 5-38 presents the results obtained during subject dependent and independent training while the *unknown* class is used and when it is not. Again, presenting results incorporating the *unknown* class has the sole purpose of testing the performance of the recognition algorithm in a worse-case difficulty scenario. The best performance of 96.5% is achieved, as expected, during subject dependent training when the *unknown* class is not used. It can be observed that the overall accuracy in recognizing the 11 activities explored is excellent during subject dependent training (96.5%) and reasonably good during subject independent training (81.3%). The performance when the *unknown* class is



Figure 5-34: Confusion matrices for the C4.5 classifier when recognizing postures, ambulation, and two MET intensity categories when the unknown class is not included during (a) subject dependent and (b) independent evaluation. The feature set used is the *invariant reduced* feature set computed per axis over windows of 5.6s in length. The maximum number of errors in a given cell of a confusion matrix is 247 and 1330 for subject dependent and independent training respectively.

Sensor Combination	All	Postures	Ambulation	Moderate	Vigorous
All sensors	96.5 ± 1.1	96.5±3.8	96.0±4.5	96.4±1.7	93.3±3.5
		(0.1±0.1)	(0.1±0.1)	(2.4±1.2)	(1.3±0.6)
Hip + DWrist + DFoot	$95.3 \pm 1.1$	95.5±3.8	96.4±5.1	94.9±1.5	91.4±2.6
_		(0.1±0.2)	(0.1±0.1)	(3.3±0.8)	(1.9±0.7)
Hip + DWrist	$94.9 \pm 1.4$	94.7±5.1	94.0±7.2	94.8±1.4	90.7±4.1
		(0.1±0.2)	(0.2±0.1)	(3.5±1.4)	(2.1±0.8)
Hip + DFoot	$95.8 \pm 1.0$	95.5±4.0	96.4±4.1	95.4±1.6	92.7±3.0
		(0.1±0.2)	(0.1±0.1)	(2.7±0.7)	(1.7±0.7)
DWrist + DThigh	$94.3 \pm 1.4$	95.6±4.1	92.7±5.9	94.4±1.7	89.6±4.7
		(0.1±0.2)	(0.2±0.2)	(3.7±1.2)	(2.4±0.7)
DWrist + DFoot	$93.8 \pm 1.6$	95.5±3.7	94.3±6.7	93.3±1.8	88.2±4.4
		(0.1±0.1)	(0.2±0.1)	(4.3±1.5)	(2.7±0.8)
Hip	$94.8 \pm 1.5$	92.7±5.5	95.1±5.2	94.4±1.6	91.4±4.0
		(0.2±0.2)	(0.2±0.1)	(3.2±1.4)	(2.2±0.8)
DWrist	$89.6 \pm 3.0$	91.3±7.6	87.9±10.8	89.4±3.3	82.4±6.7
		(0.3±0.2)	(0.4±0.3)	(7.1±2.5)	(4.2±1.4)
DFoot	$92.9 \pm 2.2$	94.1±7.4	93.5±4.6	92.5±2.3	86.6±5.1
		(0.2±0.2)	(0.2±0.2)	(4.6±1.7)	(3.1±1.0)
DUpperArm	$93.8 \pm 1.8$	91.0±6.9	95.6±3.9	93.3±1.8	89.7±4.2
		(0.3±0.2)	(0.2±0.1)	(4.0±1.5)	(2.5±0.9)
DThigh	$94.2 \pm 1.3$	95.2±4.8	93.0±5.7	93.7±1.7	90.7±3.8
_		(0.2±0.2)	(0.2±0.2)	(3.7±1.3)	(2.2±0.6)

Table 5-40: Subject dependent performance of the C4.5 classifier when recognizing postures and ambulation using the *invariant reduced* feature set computed per axis over windows of 5.6s in length using different subsets of accelerometers. The *unknown* class was not included in this experiment. The probability of random guessing is 9% for all activity categories shown in this table.

Sensor Combination	All	Postures	Ambulation	Moderate	Vigorous
All sensors	$81.3 \pm 4.7$	91.3±9.3	64.4±33.6	86.8±5.4	66.0±15.4
		(0.5±0.8)	(1.0±1.1)	(13.0±5.4)	(5.3±2.9)
Hip + DWrist + DFoot	$76.2 \pm 8.4$	80.7±25.2	62.1±33.7	85.6±6.1	58.4±16.2
_		(0.7±0.9)	(1.3±1.2)	(15.7±6.1)	(7.5±5.6)
Hip + DWrist	$72.5 \pm 7.6$	52.3±26.3	59.6±29.3	84.0±7.6	60.0±14.6
-		(1.4±1.8)	(1.5±1.5)	(14.4±4.8)	(8.8±7.4)
Hip + DFoot	$72.9\pm8.9$	75.2±25.5	57.3±33.2	81.9±8.5	58.0±17.5
-		(1.3±1.4)	(1.5±1.6)	(15.9±7.2)	(7.0±3.3)
DWrist + DThigh	$78.6 \pm 5.3$	69.7±28.3	72.9±25.2	83.2±7.5	67.2±11.4
		(1.1±1.2)	(0.9±1.0)	(12.1±3.1)	(6.9±4.3)
DWrist + DFoot	$74.5 \pm 6.9$	83.3±23.1	57.3±34.5	82.2±6.9	57.5±11.7
		(0.7±1.0)	(1.6±1.8)	(15.7±4.4)	(7.4±2.6)
Hip	$72.0 \pm 7.4$	51.0±24.7	60.7±30.2	82.8±7.9	57.6±12.6
_		(1.4±1.4)	(1.5±1.4)	(15.7±5.0)	(8.0±6.6)
DWrist	$68.5\pm6.6$	52.6±28.4	48.5±29.3	81.3±4.6	55.2±10.6
		(1.7±1.7)	(1.5±1.5)	(18.5±5.6)	(8.6±2.2)
DFoot	$67.1 \pm 9.1$	67.7±30.2	50.1±32.2	79.8±8.1	45.8±15.8
		(1.5±1.4)	(1.8±1.9)	(19.6±6.4)	(8.5±3.9)
DUpperArm	$75.7 \pm 4.0$	55.0±22.8	71.7±26.7	83.3±6.6	63.8±11.3
		(1.4±1.4)	(1.0±1.2)	(16.3±5.0)	(5.8±2.0)
DThigh	$67.9 \pm 7.4$	40.8±30.4	69.6±28.2	73.4±9.3	69.1±13.6
-		$(2.5\pm2.2)$	$(0.8\pm0.9)$	(20.6±8.4)	(7.4±5.7)

Table 5-41: Subject independent performance of the C4.5 classifier when recognizing postures and ambulation using the *invariant reduced* feature set computed per axis over windows of 5.6s in length using different subsets of accelerometers. The *unknown* class was not included in this experiment. The *unknown* class was not included in this experiment. The *unknown* class was not included in this experiment. The probability of random guessing is 9% for all activity categories shown in this table.



Figure 5-35: True positive rate per sensor combination while recognizing postures, ambulation, and two MET intensity categories using the C4.5 classifier when features are computed per axis over windows of 5.6s during subject dependent evaluation. The grayscale image is scaled so that the maximum true positive rate of 96.5% is represented by white and the minimum of 82.4% by black. In other words, poor areas of performance are shown in black.



Figure 5-36: True Positive Rate per sensor combination while recognizing postures, ambulation, and two MET intensity levels using the C4.5 classifier when features are computed per axis over windows of 5.6s during subject independent evaluation. The grayscale image is scaled so that the maximum true positive rate of 91.3% is represented by white and the minimum of 40.8% by black. In other words, poor areas of performance are shown in black.

included drops between 7-10% because it contains unlabeled examples of *sitting*, *standing*, and *walking* at different speeds.

Table 5-39 presents the performance per activity and Figure 5-23 the confusion matrices for subject dependent and independent training. The confusion matrices are grayscale images scaled to highlight large number of confusions in black and low number of confusions in white. During subject dependent training, the true positive rate over postures, ambulation, and the moderate MET category is very similar (~96%). The activity category with lower performance is vigorous, with a performance of 93.6%. This is because the vigorous intensity category is being confused with the moderate intensity category during both subject dependent and independent evaluation as shown by the confusion matrices (Figure 5-34). This confusion happens because these intensity categories include activities with similar motion patterns such as cycling, crunches and sit-ups that were just partitioned due to their associated MET intensities into the moderate and vigorous MET intensity categories. Figure 5-23 also shows that there is a considerable degree of confusion between lying down and the vigorous MET intensity level. This is because the examples for the activities crunches and sit-ups were merged into the vigorous MET intensity level. These two activities include examples of lying down, particularly when participants rested for a couple of seconds between repetitions. When the performance per activity is inspected for subject dependent training, it can be seen that the lowest performance is obtained for running at 5mph (94.7%) and 6mph (91%), standing (93%), and the vigorous category (93%). The performance is relatively low for running at 6mph (91%) because most subjects only performed this activity for less than 2min due to its physical difficulty. Finally the confusion matrix (Figure 5-34) for subject dependent training also illustrates that sitting is being confused with the moderate intensity category. This is because some activities included in the moderate intensity category such as bench weight lifting and bicep curls are performed in a in a sitting position.

When the performance per activity is analyzed for subject independent training, it is found that the worse performance is obtained while discriminating among the intensity levels or speeds of running (TP rate between 51-53%) and for the vigorous intensity category (66%). Inspection of the confusion matrix (Figure 5-23) also reveals confusions among the different speeds of running, confusion between sitting and the moderate intensity category, and confusion between the moderate and vigorous categories for the same reasons explained before.

Finally, the performance of recognizing postures, ambulation, and the two MET intensity categories was evaluated when utilizing different sensor combinations. Table 5-40 and Table 5-41 present the results obtained. Figure 5-35 and Figure 5-36 also present the true positive rate as a grayscale image normalized to show the best performance in white and the worse performance in black. These tables and figures show that the three best performing sensor combinations for subject dependent training in decreasing order are Hip+DFoot, Hip+DWrist+DFoot, and DHip. This ordering is different from the one obtained in previous sections, and suggest that when the number of activities to recognize is decreased the number of sensors can also be decreased without significant decreases in performance. When comparing the results per activity category for the Hip+DFoot and for the Hip+DWrist+DFoot sensor combinations, it is found that their performance is very similar. It seems that the DWrist sensor becomes less useful

when gymnasium and household activities with high energy upper body motion are merged into the moderate MET intensity category. The sensor at the hip detects this upper body motion indirectly well enough to allow discrimination, at least during subject dependent training. The fact that the Hip sensor is the third best performing sensor also suggests that this sensor captures lower body motion indirectly well enough to discriminate between the different intensity levels of walking and running. It is thus clear that the best single sensor to utilize while recognizing the set of activities explored in this section is the sensor at the hip. When the performance per activity for this sensor is compared against the performance of the Hip+DFoot sensor combination, it is found that it has a slightly lower performance for the postures (-2.8%) and ambulation (-1.3%) categories. In general, the change in performance obtained while evaluating different sensor combinations to recognizing activities is higher for subject independent training than for subject dependent training.

The three best performing sensor combinations found for subject independent training in decreasing order are DWrist+DThigh, Hip+DWrist+DFoot, and DUpperArm. The order is also different from the one obtained in previous sections. Now, the combination of a sensor at the dominant wrist and at the dominant thigh (DWrist+DThigh) slightly outperforms the combination Hip+DWrist+DFoot. This increased performance is distributed across the ambulation (+10%) and the *vigorous* MET intensity category (+8.8%). The Hip+DWrist+DFoot sensor combination; however, is better at recognizing postures (+10%). The best single sensor to utilize during subject independent training is the DUpperArm, followed by the Hip sensor, and the DWrist sensor. The DUpperArm sensor is better in recognizing ambulation and the vigorous intensity category that the sensor at the Hip. This is because it better recognizes the upper body activity found in these activities. The sensor at the hip (Hip) is better at recognizing ambulation than the sensor wrist (DWrist). This is because the sensor at the wrist experiences more variability in the accelerometer signals due to the higher degree of freedom of the wrist.

In summary, the best compromise sensor combination to use for detecting postures, ambulation, and the two MET intensity categories is still the Hip+DWrist+DFoot sensor combination. However, because the number of activities explored has been reduced or merged into the *moderate* and *vigorous* intensity categories, the number of sensors can be reduced. For example the Hip+DFoot or DHip sensor combinations can be used during subject dependent training and the DWrist+DThigh or DUpperArm combinations during subject independent training with little decrease in performance.

#### 5.4.9.5 How Well can Postures and Ambulation be Recognized?

This section explores how well postures and ambulation with no intensity levels can be recognized from several combinations of accelerometers using the final implementation of the activity recognition algorithm. Thus, this experiment is performed by eliminating all the activities that are not postures or ambulation from the MIT dataset and by merging the different intensity levels of ambulatory activities such as *walking* and *running* into a single class. The column labeled as "Postures and Ambulation" in Appendix A2 shows what activities are considered in this experiment and what intensity levels of *walking* and *running* are being merged. The activities to recognize are the following eight activities:

Evaluation Method	All	Postures	Ambulation	Unknown
Subject dependent	$98.4 \pm 0.8$	98.9±1.9	95.4±3.9	-
without Unknown class		(0.1±0.1)	(0.4±0.4)	
Random guessing: 12.4%				
Subject dependent	$91.4 \pm 1.7$	93.8±5.1	87.9±5.4	$89.9\pm2.7$
with unknown class		(0.3±0.2)	(1.1±0.4)	$(7.1 \pm 1.9)$
Random guessing: 11.1%				
Subject independent	92.9 ± 3.9	96.4±8.9	85.3±15.5	-
without unknown class		(0.3±0.9)	(2.1±2.2)	
Random guessing: 12.5%				
Subject independent	$78.5 \pm 8.8$	71.8±22.7	68.4±24.2	$87.4 \pm 4.0$
with unknown class		(0.7±0.6)	(1.8±1.1)	$(24.2 \pm 3.7)$
Random guessing: 11.1%				

Table 5-42: True positive and false positive rate (shown in parenthesis) for the C4.5 classifier while recognizing postures and ambulation without intensity levels during subject dependent and independent evaluation. The feature set used is the *invariant reduced* feature set computed per axis over windows of 5.6s in length over all the seven accelerometers. The random guess probability for all activities shown in this table is 12.4% when the unknown class is not included and 11.1% when it is included.

	Subject Dependent			Subject Independent		
Class	TP Rate	FP Rate	F-Measure	TP Rate	FP Rate	F-Measure
Lying down	$99.9\pm0.3$	$0.1 \pm 0.1$	$99.8 \pm 0.3$	$100.0\pm0.0$	$0.2 \pm 0.9$	$99.6 \pm 1.9$
Standing	$97.5\pm3.9$	$0.2 \pm 0.2$	$96.5 \pm 3.7$	$96.3 \pm 9.6$	$0.7 \pm 2.3$	$92.7 \pm 15.5$
Sitting	$99.0\pm1.3$	$0.1 \pm 0.1$	$99.2\pm0.8$	$90.9 \pm 21.7$	$0.3 \pm 0.5$	$92.3 \pm 19.3$
kneeling	$99.0\pm2.0$	$0.0 \pm 0.1$	$99.1 \pm 1.6$	$98.4 \pm 4.3$	$0.0 \pm 0.1$	$98.6\pm2.5$
Walking	$98.9\pm0.9$	$0.7 \pm 0.7$	$99.0\pm0.8$	$93.2 \pm 6.5$	$4.9 \pm 3.5$	$93.3 \pm 4.4$
Running	$99.2 \pm 1.4$	$0.2 \pm 0.2$	$98.8 \pm 1.2$	$85.6 \pm 23.7$	$1.6 \pm 2.8$	$85.6\pm22.5$
Stairs - Ascend stairs	$93.1 \pm 5.1$	$0.3 \pm 0.2$	$93.4 \pm 3.9$	$94.4 \pm 6.4$	$0.7 \pm 0.9$	$91.5\pm6.6$
Stairs - Descend stairs	$90.6 \pm 8.4$	$0.4 \pm 0.3$	$91.4 \pm 6.8$	$68.0\pm25.6$	$1.3 \pm 1.5$	$68.5\pm24.4$

Table 5-43: Performance per activity obtained for the C4.5 classifier when recognizing postures and ambulation without intensity levels and without the *unknown* class during subject dependent and independent evaluation. The feature set used is the *invariant reduced* feature set computed per axis over windows of 5.6s in length over all seven accelerometers. The random guess probability for all activities shown in this table is 12.4%.

Lying down, standing, sitting, kneeling, walking, running, ascending stairs, and descending stairs. These eight activities produce a random guessing probability of 12.5%. The parameters used in the activity recognition algorithm are the C4.5 classifier, the *invariant reduced* feature set computed per axis, and a sliding window length of 5.6s.

Table 5-42 presents the true positive and false positive rate per activity category when the *unknown* class is included and when it is not. It is clear that the performance obtained while recognizing only postures and ambulation without intensity levels is excellent.

The best overall accuracy of 98% is obtained during subject dependent training when the unknown class is not used. The performance obtained during subject independent training is 92.9% when the *unknown* class is not used and 78.5% when it is used. This large decrease in performance (-14.4%) obtained when including the *unknown* class is due to the fact that the *unknown* class has unlabeled examples of *sitting*, *standing*, and *walking*. As a result, when the *unknown* class is included, it is confused with these activities resulting in the decreased performance observed.

Table 5-43 shows the true positive rate, false positive rate, and F-Measure per activity during subject dependent and independent training when the *unknown* activity is not used. The performance per activity during subject dependent training is excellent, ranging



Figure 5-37: Confusion matrices for the C4.5 classifier when recognizing postures and ambulation without intensity levels and without considering the *unknown* class during subject dependent and independent evaluation. The feature set used is the *invariant reduced* feature set computed per axis over windows of 5.6s in length over all accelerometers. The maximum number of errors in a given cell of a confusion matrix is 26 and 168 for subject dependent and independent training respectively.

Sensor Combination	All	Postures	Ambulation
All sensors	$98.4 \pm 0.8$	98.9±1.9	95.4±3.9
		(0.1±0.1)	(0.4±0.4)
Hip + DWrist + DFoot	98.1 ± 1.1	98.8±2.4	94.5±4.9
-		(0.1±0.1)	(0.5±0.5)
Hip + DWrist	$97.7 \pm 1.0$	98.2±2.8	93.0±5.0
		(0.1±0.2)	(0.6±0.5)
Hip + DFoot	$98.1 \pm 1.1$	98.6±2.7	94.3±5.4
		(0.1±0.2)	(0.5±0.5)
DWrist + DThigh	$97.0 \pm 0.6$	99.0±1.5	90.2±5.2
		(0.1±0.1)	(0.8±0.5)
DWrist + DFoot	$96.4 \pm 1.4$	97.5±2.9	88.1±6.1
		(0.2±0.2)	(1.0±0.6)
Hip	$97.36 \pm 1.2$	97.4±3.8	92.1±6.3
		(0.1±0.2)	(0.7±0.5)
DWrist	$94.3 \pm 1.9$	94.2±6.3	84.7±7.6
		(0.3±0.4)	(1.5±0.8)
DFoot	$95.6\pm1.8$	95.9±5.7	86.4±6.5
		(0.3±0.3)	(1.1±0.7)
DUpperArm	$96.0 \pm 2.4$	92.9±8.0	91.1±6.7
		(0.4±0.4)	(0.9±0.8)
DThigh	$96.6 \pm 0.9$	98.8±2.0	89.1±5.6
		$(0.1\pm0.2)$	$(0.9\pm0.5)$

Table 5-44: Subject dependent performance over different combination of sensors for the C4.5 classifier when recognizing postures and ambulation without intensity levels and without the *unknown* class. The feature set used is the *invariant reduced* feature set computed per axis over windows of 5.6s in length. The random guess probability for all activities shown in this table is 12.5%.

Sensor Combination	All	Postures	Ambulation
All sensors	$92.9 \pm 3.9$	96.4±8.9	85.3±15.5
		(0.3±0.9)	(2.1±2.2)
Hip + DWrist + DFoot	$86.5\pm5.6$	88.7±17.9	74.0±23.3
-		(1.0±1.7)	(3.5±3.5)
Hip + DWrist	$81.9\pm6.5$	59.6±27.3	73.6±20.2
-		(2.5±3.8)	(3.2±2.4)
Hip + DFoot	$83.6\pm7.2$	78.5±24.8	69.5±26.1
		(1.6±2.0)	(3.7±3.6)
DWrist + DThigh	$84.9 \pm 9.4$	73.4±32.2	76.8±18.5
-		(2.0±3.1)	(2.7±3.0)
DWrist + DFoot	$85.5 \pm 8.4$	90.7±15.6	71.3±22.0
		(1.0±1.8)	(3.7±3.7)
Hip	$81.0 \pm 5.9$	55.5±29.1	73.5±20.8
-		(2.9±3.6)	(3.1±2.3)
DWrist	$79.3 \pm 8.4$	58.5±28.4	73.0±19.1
		(2.8±3.3)	(3.6±3.3)
DFoot	$77.9 \pm 10.6$	75.7±27.0	66.8±23.6
		(2.4±2.4)	(4.6±4.1)
DUpperArm	86.5 ± 4.3	60.8±20.9	85.9±15.7
		(2.1±2.0)	(2.1±2.5)
DThigh	$79.9\pm9.4$	68.4±31.7	72.3±18.8
		$(3.1\pm3.2)$	$(3.2\pm3.5)$

Table 5-45: Subject independent performance over different combination of sensors for the C4.5 classifier when recognizing postures and ambulation without intensity levels and without the *unknown* class. The feature set used is the *invariant reduced* feature set computed per axis over windows of 5.6s in length. The random guess probability for all activities shown in this table is 12.5%.



Figure 5-38: True Positive rate per sensor combination using the C4.5 classifier when features are computed per axis over windows of 5.6s during subject dependent evaluation. The grayscale image is scaled so that the maximum true positive rate of 99.0% is represented by white and the minimum of 84.7% by black. In other words, poor areas of performance are shown in black.





from 90 to 99.9%. The activity with lowest performance (90%) is *descending stairs* because it is being confused with *walking* and *ascending stairs* due to the motion similarity (see Figure 5-37). Other activities being confused during subject dependent evaluation are *walking* with *running*, *descending stairs* with *ascending stairs*, *and standing* with *sitting*.

During subject independent training, the true positive per class is higher than 93.2% for all the activities except for *running* (85.6%) and *descending stairs* (68%). *Running* and *descending stairs* also have a relatively large false positive rate of 1.6 and 1.4% respectively. This is because *running* is being confused with *walking*, and *descending stairs* is being confused with *walking* and *ascending stairs*. In fact, the *walking* activity has a false positive rate of 4.9% because this class is being predicted when the *running* and *descending stairs* activities are performed. Confusion among ambulatory activities is expected, particularly during subject independent training due to the motion similarity among the activities.

Table 5-44 and Table 5-45 present the true positive and false positive rate when different sensor combinations are used to recognize activities during subject dependent and independent training. Figure 5-38 and Figure 5-39 also illustrate the true positive rate as a grayscale image normalized to highlight good performance in white and poor performance in black. For subject dependent training, the sensor combinations with higher performance are Hip+DFoot, Hip+DWrist+DFoot, and Hip+DWrist. The combinations Hip+DFoot and Hip+DWrist+DFoot have the same overall accuracy of 98.1% and very similar performance per activity category. Again, one sensor at the hip and one sensor either at the upper or lower body achieves a performance similar to the one obtained using the Hip+DWrist+DFoot sensor combination. During subject dependent training, all the sensor combinations explored achieve an overall accuracy greater than 94.3%. When the performance of single sensors is analyzed during subject dependent training, the sensor that achieves the best overall accuracy is the sensor at the hip (97.3%), followed by the sensor at the thigh (96.6%), and the sensor at the dominant upper arm (96%). Consequently, the best single sensor to utilize during subject dependent training is the Hip sensor. The sensor at the thigh, achieves a slightly higher performance (+1.4%) than the one at the hip in recognizing postures.

During subject independent training, the sensor combinations with best performance Hip+DWrist+DFoot, DUpperArm, and DWrist+DFoot. are The combinations Hip+DWrist+DFoot, DUpperArm have the same overall accuracy but the Hip+DWrist+DFoot combination shows a higher performance for postures (+20%) while the DUpperArm sensor presents a higher performance for ambulation (+12%). Thus, the additional sensors allow better discrimination among postures. One possible explanation of why the DUpperArm sensor has a higher performance in recognizing ambulation is that it experiences less random variations during these activities than the sensors DWrist and DFoot. During the data collections, it was observed that sensors at the wrists and at the feet generally experience more variation in the accelerometer signals due to the higher degree of motion available at these body parts. For example, while sitting, subjects fidget the feet in random patterns. This was also observed during the bicep curls and bench weight lifting activities where subjects impatiently fidget feet while experiencing physically intense effort. The accelerometers at the wrists also experience a high degree of variability because hands are constantly in motion, particularly during unconstrained

<b>Evaluation Method</b>	All	Lying down	Standing	Sitting	Kneeling	Unknown
Subject dependent	$99.3 \pm 0.7$	$99.9\pm0.2$	$97.7 \pm 3.2$	$98.6 \pm 2.7$	$98.6 \pm 2.8$	-
without unknown class		$(0.5 \pm 0.8)$	(0.1±0.3)	$(0.3 \pm 0.5)$	$(0.1 \pm 0.2)$	
Random guessing: 25%						
Subject dependent	$97.6\pm0.9$	$99.5 \pm 1.1$	$83.4 \pm 12.4$	$86.6 \pm 11.7$	$95.4 \pm 5.7$	$98.2\pm0.7$
with unknown class		$(0.4 \pm 0.3)$	$(0.5 \pm 0.3)$	$(0.5 \pm 0.4)$	$(0.1 \pm 0.2)$	$(4.1 \pm 2.5)$
Random guessing: 20%						
Subject independent	$98.0 \pm 4.9$	$100.0\pm0.0$	$98.1 \pm 8.5$	$95.9 \pm 17.7$	$100.0 \pm 0.0$	-
without unknown class		$(1.4 \pm 6.0)$	$(0.8 \pm 2.9)$	$(0.9 \pm 3.0)$	$(0.0 \pm 0.0)$	
Random guessing: 25%						
Subject independent	$93.4 \pm 2.8$	$90.7 \pm 18.4$	$68.5 \pm 24.8$	$36.2 \pm 40.2$	$94.8 \pm 7.6$	$97.4 \pm 1.8$
with unknown class		$(0.5 \pm 1.0)$	$(0.7 \pm 0.6)$	$(1.1 \pm 1.0)$	$(0.4 \pm 1.4)$	$(16.5 \pm 10.2)$
Random guessing: 20%						

Table 5-46: True positive rate and false positive rate (shown in parenthesis) per activity for the C4.5 classifier when recognizing postures in a subject dependent and independent manner. The feature set used is the *invariant reduced* feature set computed per axis over windows of 5.6s in length over all seven accelerometers. The random guess probability is 25% when the *unknown* class is not included and 20% when it is included.

activities such as household cleaning. The relatively poor performance of these sensors can be seen from Figure 5-39 as dark areas for the ambulation category. The best single sensors to use to recognizing postures and ambulation in a subject independent manner are DUpperArm, Hip, and DThigh. The DUpperArm sensor is better for recognizing ambulation, while the DThigh sensor is better at recognizing postures. The sensor at the hip presents average performance (with respect to DUpperArm and DThigh) in recognizing postures and ambulation.

In conclusion, a good combination to use for recognizing postures and ambulation without intensity levels during subject dependent training is Hip+DWrist. The best single sensor is the Hip sensor. During subject independent training, the best sensor combination to use is either DWrist+DFoot or Hip+DWrist+DFoot. The single best sensor to use is sensor at the dominant upper arm (DUpperArm). Thus, when only postures and ambulation need to be recognized, two sensors are enough to achieve good discrimination performance among these activities.

#### 5.4.9.6 How Well can Postures be Recognized?

This section explores how well can postures be recognized using the final implementation of the activity recognition algorithm while different combination of sensors are used. This is performed by eliminating all the activities that are not postures from the MIT dataset and by merging the activities *sitting fidgeting feet and legs* and *sitting fidgeting hands and arms* with the *sitting* activity. The column labeled as "Postures" in Appendix A2 shows the activities that were included in this experiment. The total number of activities to recognize is then four, giving a random guess probability of 25% when the unknown class is not included and 20% when it is included.

Table 5-46 presents the total accuracy, as well as the true positive and false positive rate per activity obtained from this experiment when the *unknown* class is included and when it is not. The overall accuracy obtained during subject dependent and independent training is excellent, ranging from 93 to 99%. Obviously, this is the result of the limited number of activities being recognized (4). The lowest performance is observed for



Figure 5-40: Confusion matrices of the C4.5 classifier when recognizing postures without the *unknown* class during (a) subject dependent and (b) independent training. The feature set used is the *invariant reduced* feature set is computed per axis over windows of 5.6s in length over all the sensors. The maximum number of errors in a given cell of a confusion matrix is 5 and 19 for subject dependent and independent training respectively.

Sensor Combination	All	Lying down	Standing	Sitting	Kneeling
All sensors	$99.3\pm0.7$	$99.9\pm0.2$	$97.7 \pm 3.2$	$98.6\pm2.7$	$98.6\pm2.8$
		$(0.5 \pm 0.8)$	(0.1±0.3)	$(0.3 \pm 0.5)$	$(0.1 \pm 0.2)$
Hip + DWrist + DFoot	$99.1\pm0.9$	$99.9\pm0.4$	$97.2\pm3.8$	$98.6 \pm 3.1$	$98.0\pm2.9$
		$(0.2 \pm 0.6)$	$(0.3 \pm 0.5)$	$(0.3 \pm 0.5)$	$(0.2 \pm 0.3)$
Hip + DWrist	$98.8\pm0.7$	$99.9\pm0.3$	$97.7\pm3.5$	$97.7\pm2.9$	$95.8\pm4.6$
		$(0.5 \pm 0.8)$	$(0.6 \pm 0.7)$	$(0.3 \pm 0.5)$	$(0.2 \pm 0.4)$
Hip + Foot	$99.2\pm0.8$	$99.9\pm0.3$	$97.41 \pm 4.2$	$98.8\pm2.6$	$98.0\pm2.9$
		$(0.3 \pm 0.7)$	$(0.3 \pm 0.5)$	$(0.3 \pm 0.5)$	$(0.0 \pm 0.21)$
DWrist + DThigh	$98.9\pm0.7$	$99.9 \pm 0.2$	$95.80 \pm 4.6$	$97.9 \pm 2.3$	$97.9 \pm 2.4$
_		$(0.8 \pm 0.9)$	$(0.3 \pm 0.3)$	$(0.4 \pm 0.4)$	$(0.2 \pm 0.4)$
DWrist + DFoot	$99.0\pm0.7$	$99.9\pm0.4$	$97.27 \pm 3.4$	$98.6 \pm 2.71$	$97.1 \pm 3.6$
		$(1.0 \pm 1.4)$	$(0.2 \pm 0.4)$	$(0.2 \pm 0.4)$	$(0.1 \pm 0.3)$
Hip	$98.6\pm0.9$	$99.9 \pm 0.3$	$95.81 \pm 4.9$	$97.9\pm2.9$	$95.8 \pm 4.6$
-		$(0.5 \pm 0.8)$	$(0.7 \pm 0.6)$	$(0.4 \pm 0.5)$	$(0.3 \pm 0.6)$
DWrist	$97.7 \pm 1.7$	$99.9\pm0.2$	$94.18\pm7.2$	$95.7 \pm 5.1$	$91.8\pm8.9$
		$(1.2 \pm 1.2)$	$(1.0 \pm 1.3)$	$(0.2 \pm 0.5)$	$(0.8 \pm 0.9)$
DFoot	$98.0 \pm 2.4$	99.4 ± 1.6	$97.19 \pm 5.3$	$97.9 \pm 4.4$	$91.8 \pm 12.0$
		$(2.5 \pm 3.9)$	$(0.4 \pm 0.7)$	$(0.5 \pm 0.8)$	( 0.3 ± 1.0)
DUpperArm	$96.7\pm2.7$	$99.9\pm0.2$	$89.9 \pm 10.1$	$95.4\pm7.8$	$88.5\pm10.6$
		$(0.6 \pm 0.8)$	$(1.4 \pm 1.7)$	$(0.5 \pm 0.9)$	(1.5 ± 1.3)
DThigh	$99.0\pm0.7$	$99.9\pm0.2$	$95.9\pm4.8$	$98.4 \pm 2.2$	$98.1 \pm 2.4$
-		$(0.4 \pm 0.7)$	$(0.2 \pm 0.3)$	$(0.4 \pm 0.4)$	$(0.3 \pm 0.5)$

Table 5-47: True positive and false positive rate (shown in parenthesis) during subject dependent training for the C4.5 classifier when recognizing postures without the *unknown* class over different sensor combinations. The feature set used is the *invariant reduced* feature set computed per axis over windows of 5.6s in length.

Sensor Combination	All	Lying down	Standing	Sitting	Kneeling
All sensors	$98.0\pm4.9$	$100.0 \pm 0.0$	$98.1 \pm 8.5$	95.9 ± 17.7	$100.0 \pm 0.0$
		$(1.4 \pm 6.0)$	$(0.8 \pm 2.9)$	$(0.9 \pm 3.0)$	$(0.0 \pm 0.0)$
Hip + DWrist + DFoot	$90.6 \pm 13.9$	$95.8 \pm 12.3$	80.1 ± 35.9	$88.5 \pm 30.1$	$94.0 \pm 22.9$
_		$(1.9 \pm 7.4)$	$(3.3 \pm 5.9)$	$(4.3 \pm 6.4)$	$(3.0 \pm 8.4)$
Hip + DWrist	$79.4 \pm 13.5$	$97.5 \pm 11.0$	$51.6\pm40.5$	$65.3 \pm 43.2$	$50.9 \pm 45.4$
_		$(5.0 \pm 22.4)$	$(6.3 \pm 6.3)$	$(6.5 \pm 14.8)$	$(11.2 \pm 5.8)$
Hip + Foot	$83.3 \pm 13.9$	$95.0 \pm 15.1$	$64.3 \pm 46.8$	$59.8 \pm 49.2$	$100.0\pm0.0$
_		$(1.6 \pm 7.1)$	$(9.1 \pm 6.8)$	$(9.8 \pm 6.5)$	$(3.9 \pm 11.2)$
DWrist + DThigh	$82.1 \pm 12.4$	$90.5 \pm 17.0$	$67.9 \pm 37.9$	$83.9 \pm 32.0$	$65.1 \pm 32.4$
_		$(4.5 \pm 11.1)$	$(4.8 \pm 5.5)$	$(10.1 \pm 13.7)$	$(4.9 \pm 5.1)$
DWrist + DFoot	$92.4 \pm 10.5$	$97.1 \pm 11.0$	$86.7 \pm 30.2$	$84.9 \pm 31.7$	$94.0\pm22.9$
		$(1.9 \pm 7.4)$	$(3.7 \pm 5.9)$	$(2.7 \pm 5.1)$	$(1.9 \pm 7.4)$
Hip	$71.4 \pm 11.6$	$96.0 \pm 12.2$	$31.5 \pm 35.9$	$38.7\pm40.9$	$35.1 \pm 38.9$
		$(5.0 \pm 22.4)$	$(13.7 \pm 10.2)$	$(8.6 \pm 8.0)$	$(13.0 \pm 10.7)$
DWrist	$66.9 \pm 18.4$	$79.4\pm23.4$	$44.3 \pm 34.4$	$70.1 \pm 37.8$	$43.5 \pm 37.2$
		$(9.1 \pm 20.7)$	$(14.0 \pm 14.9)$	$(14.2 \pm 16.2)$	$(11.1 \pm 11.5)$
DFoot	$75.9 \pm 14.0$	$84.8\pm23.4$	$65.4 \pm 43.04$	$64.2 \pm 46.3$	$82.0\pm27.0$
		$(7.0 \pm 10.7)$	$(8.6 \pm 6.9)$	$(7.6 \pm 6.6)$	$(11.9 \pm 17.0)$
DUpperArm	$75.4 \pm 9.0$	$96.5 \pm 11.4$	$37.9 \pm 31.4$	$50.7 \pm 38.3$	$41.3 \pm 23.9$
		$(6.8 \pm 17.1)$	$(9.3 \pm 9.9)$	$(7.0 \pm 8.0)$	$(10.7 \pm 7.3)$
DThigh	$69.3 \pm 22.9$	85.9 ± 22.6	$54.2 \pm 37.9$	$56.3 \pm 37.3$	$60.7 \pm 39.7$
-		$(14.1 \pm 12.6)$	$(4.2 \pm 5.6)$	$(19.4 \pm 26.0)$	$(6.4 \pm 4.8)$

Table 5-48: True positive and false positive rate (shown in parenthesis) during subject independent training for the C4.5 classifier when recognizing postures without the *unknown* class over different sensor combinations. The feature set used is the *invariant reduced* feature set computed per axis over windows of 5.6s in length.



Figure 5-41: True positive rate per sensor combination during subject dependent evaluation using the C4.5 classifier when features are computed per axis over windows of 5.6s. The grayscale image is scaled so that the maximum true positive rate of 99.9% is represented by white and the minimum of 88.5% by black. In other words, poor areas of performance are shown in black.





*standing* (97-98%) and *sitting* (98-95%) during subject dependent and independent training when the *unknown* class is not used. When the *unknown* activity is incorporated, the performance drops dramatically for *sitting* and *standing* during subject independent training. This is because subjects were mainly *standing* and *sitting* while resting during the unlabeled periods of time of the data collection (*unknown* activity).

The confusion matrices for subject dependent and independent training shown in Figure 5-40 illustrate that the activities most confused are *standing* with *sitting*, and *sitting* with *kneeling*. The confusion matrix for subject independent training also shows a relatively high degree of confusion between *lying down* and *sitting*. These confusions however, are not large enough to affect the recognition performance as shown in Table 5-46.

Table 5-47 and Table 5-48 present the true positive and false positive rate obtained when different sensor combinations are used. Figure 5-41 and Figure 5-42 show the true positive rate as a grayscale image normalized to highlight the differences in performance per class. The best performance is represented by the color white and the worse by the color black. Consequently, poor areas of performance can be identified by dark regions in the image.

These tables show an excellent performance for all the sensor combinations explored during subject dependent training. For example, all sensor combinations using two or more sensors have a true positive rate greater or equal than 98.8%. The best performing sensor combinations are Hip+DFoot, Hip+DWrist+DFoot, and DWrist+DFoot. The maximum difference in performance between these sensor combinations is just 0.2%. Again, the combination of a sensor at the hip and a sensor at the lower body is performing as well as the combination of a sensor at the hip, a sensor in the upper body and a sensor in the lower body. The single sensors with higher performance for subject dependent training in decreasing order are DThigh, Hip, and DFoot. The DThigh sensor is slightly better (+2.3%) than the Hip sensor at recognizing *kneeling*. The sensor DFoot is slightly better in recognizing *standing* (+1.4%) than the DThigh and DFoot sensors. The worse performing sensor is the DUpperArm sensor with an overall true positive rate of 96.7%. It seems that during subject dependent training, all the sensor combinations explored can be used to recognize postures with excellent results.

For subject independent training, the best sensor combinations in decreasing order are DWrist+DFoot, Hip+DWrist+DFoot, and Hip+DFoot. It is interesting to note that the two sensor combinations with higher performance include the sensors DWrist and DFoot. When either of these sensors is not included in a sensor combination, the performance drops approximately 9%. Consequently, during subject independent training, the best sensor combination to use to recognize postures is DWrist+DFoot. The single sensors with best overall performance during subject independent training are DFoot (75.9%), DUpperArm (75.4%), and Hip (71%). The decrease in performance with respect to the DWrist+DFoot combination when these single sensors are used is around 17%. When the performance per posture is inspected, it is found that the sensor DFoot also outperforms the other single sensors in almost all categories. The DUpperArm sensor only outperforms all the other single sensors in recognizing *lying down* (+11.7%) while the DWrist sensor outperforms all the other single sensors in recognizing *sitting*. The confusion matrix for the DFoot sensor shows that the postures being confused the most are *standing* and *sitting*. This is expected since the orientation of this sensor is very similar or identical

		Subject Dependent			Subject Independent		
Activities to	Random	Total	TPR Range	FPR Range	Total	TPR Range	FPR Range
recognize	Guess	Accuracy	_		Accuracy	_	_
All (51)	1.9%	$87.9\pm2.0$	80 - 96	0.1 - 0.4	$50.6 \pm 5.1$	34 - 47	0.5 - 1.3
All with no	3.2%	$91.4 \pm 1.6$	80 - 98	0.1 - 0.5	$72.0 \pm 5.7$	44 - 84	0.4 - 1.2
intensities (31)							
Postures,	9%	$96.5 \pm 1.1$	93 - 97	0.1 - 2.4	$81.3 \pm 4.7$	64 - 91	0.5 - 13
ambulation and							
two MET							
intensity							
categories (11)							
Postures and	12.5%	$98.4 \pm 0.8$	95 - 98	0.1 - 0.4	$92.9 \pm 3.9$	85 - 96	0.3 - 2.1
Ambulation with							
no intensity (8)							
Postures (4)	25%	$99.3 \pm 0.7$	98 - 100	0.1 - 0.5	$98.0 \pm 2.8$	96 - 100	0 - 1.4

Table 5-49: Random guess, total accuracy, and ranges for the true positive (TPR) and false positive rates (FPR) obtained over all activity categories when recognizing different sets of activities from the MIT dataset in a subject dependent and independent manner without including the *unknown* class.

during both activities. The DFoot sensor can better recognize *sitting* over *standing* when the foot's orientation with respect to ground changes (e.g. foot is raised from the floor in an angle) or when feet are moved in distinctive patterns during *sitting* such when fidgeting feet.

In conclusion, if subject dependent training is to be used, any of the sensor combinations explored can produce an excellent performance in recognizing postures

(TPR>96.7%). During subject independent training, the best sensor combination to use is the DWrist+DFoot. This sensor combination produces a true positive rate of 92.4% with a maximum false positive rate of 3.7% per posture.

### 5.4.9.7 Summary of Results

The results presented in this section indicate that the highest total accuracy obtained while recognizing all the 51 activities contained in the MIT dataset (without the *unknown* class) using the final implementation of the activity recognition algorithm is 87.9% for subject dependent training and 50.6% for subject independent training. The performance of 50.6% obtained during subject dependent training is low, but it represents a substantial improvement over the performance obtained by random guessing (2%). As the number of activities to recognize is decreased either by excluding or merging activities, total accuracy and performance per class increases as shown in Table 5-49. The table also illustrates that excellent recognition results can be obtained using subject dependent training. During subject independent training, a reasonable performance of 72% can be obtained if discrimination among the intensity levels of activities is not required.

When different sensor combinations are analyzed to find the best performing ones with most convenient locations, it is found that the best combination to use is Hip+DWrist+DFoot for both, subject dependent and independent training. This is because this sensor combination is able to detect upper body motion, lower body motion, and overall body motion at the hip. Table 5-50 shows the three best sensor combinations and the three best single sensors to utilize while recognizing several sets of activities during subject dependent and independent training. From the table, it can be concluded

	Subject De	pendent	Subject Inde	ependent
Activities to recognize	Sensor	Single sensors	Sensor combinations	Single sensors
(Number of activities)	combinations with	with higher	with higher	with higher
	higher performance	performance	performance	performance
All (51)	Hip + DWrist +	Hip	Hip + DWrist + DFoot	DUpperArm
	DFoot	DFoot	DWrist + DThigh	DWrist
	Hip + DWrist	DThigh	DWrist + DFoot	Hip
	Hip + DFoot	-		-
All with No intensities (31)	Hip + DWrist +	Hip	Hip + DWrist + DFoot	DUpperArm
	DFoot	DFoot	DŴrist + DThigh	DWrist
	Hip + DWrist	DThigh	DWrist + DFoot	Hip
	Hip + DFoot	-		*
Postures, ambulation and two	Hip + DFoot	Hip	DWrist + DThigh	DUpperArm
MET intensity categories (11)	Hip + DWrist +	DThigh	Hip + DWrist + DFoot	Hip
	DFoot	DUpperArm	DUpperArm	DWrist
	DHip			
Postures and Ambulation with no	Hip + DFoot	Hip	Hip + DWrist + DFoot	DUpperArm
intensity (8)	Hip + DWrist +	DThigh	DUpperArm	Hip
	DFoot	DUpperArm	DWrist + DFoot	DThigh
	Hip + DWrist			· ·
Postures (4)	Hip + DFoot	DThigh	DWrist + DFoot	DFoot
	Hip + DWrist +	Hip	Hip + DWrist + DFoot	DUpperArm
	DFoot	DFoot	Hip + DFoot	Hip
	DWrist + DFoot		-	-

Table 5-50: The three sensor combinations and the three single sensors with higher performance while recognizing different sets of activities from the MIT dataset using the final implementation of the activity recognition algorithm. The performance is higher for sensor combinations shown at the top and lower for the sensor combinations shown at the bottom.

that during subject dependent training, a sensor located at the hip and another sensor located either at the dominant wrist (Hip+DWrist) or the dominant foot (Hip+DFoot) achieves reasonable performance. This is because upper body and lower body activity induce changes in the acceleration at the hip that allows some degree of discrimination among upper body and lower body activities.

During subject independent training the best performing sensor combinations are the ones utilizing a sensor located at the dominant wrist to detect upper body motion and Another sensor located either at the dominant thigh or foot to detect lower body motion (DWrist+DThigh or DWrist+DFoot). Finally, when the number of activities to recognize is decreased, the number of sensors can also be decreased with relatively small decreases in performance. From Table 5-50, it can also be seen that the best single sensors to use while recognizing activities in a subject dependent manner are Hip, DThigh, and DFoot. For subject independent training they are DUpperArm, DWrist, and Hip. The good performance of the DUpperArm sensor might be explained by the fact that most activities contained in the MIT dataset include a high degree of upper body motion and ambulatory activities in the case of household activities.

# 5.5 How Does the Activity Recognition Algorithm and its Interactive Training Interface Perform in Real-Time?

This section presents exploratory results of evaluating the real-time implementation of the activity recognition system as well as a user interface designed to train the algorithm interactively. The results presented are based on a short study performed with five participants that trained and tested the performance of the activity recognition algorithm in real-time at MIT.

The final activity recognition algorithm and a user interface that allows users to interactively train the algorithm were implemented in Java and executed on a laptop computer with a 1GHz Intel core microprocessor running the Windows Vista operating system. The only difference in the implementation of the algorithm with respect the parameters found in this work was the utilization of a slightly shorter sliding window of 4.2s in length (instead of a window length of 5.6s). This is because the sampling frequency per sensor was increased in practice from 45Hz to 60Hz when the number of sensors was reduced from seven to three [199]. The new window length was the closest one to the original (5.6s) permitted by the constraint of using a number of samples in a power of two (e.g. 256) to efficiently estimate the fast Fourier transform algorithm.

Five participants, three males and two females, from the author's research group were recruited to participate in this short study and provided with the instructions presented in Appendix A9. Three of the participants were familiar with the activity recognition research being carried out at the lab, and one of them had taken several courses in machine learning and pattern classification. The two other participants had a background in architecture and were not familiar with the inner workings of the algorithms or technical aspects of the research. Participants were asked to "train a computer program to recognize 10 physical activities, exercises, postures, or activities done in a particular posture of their choice". Participants were also instructed to wear three MITes wireless accelerometers [199] at the hip, dominant wrist, and dominant foot. Figure 5-43 presents an image indicating the placement of the wireless accelerometers on the body during this short study. Once the accelerometers were attached to the body, participants were asked to type in 10 physical activities, exercises, postures, or activities done in a particular posture that they wanted the computer program to recognize and that could be executed continuously for 2 minutes. Once the activities were entered, the application guided the participant to provide examples of the activities specified by performing each of them for two minutes continuously. The application showed the activity the participant needed to perform and indicated training progress using a counter that reached zero once the participant had completed two minutes worth of examples for a particular activity, checking for signal loss and ensuring that all examples are of good signal quality. Finally, once the training phase was finished, the application trained the activity recognition algorithm on the data provided and started recognizing the activities entered in real-time. After training the activity models, the application also tested the performance over the training examples collected using 10-fold cross-validation and recorded the results in a file for later analysis. The cross-validation evaluation over the activity examples collected took from 2.5 to 4.4 seconds, depending on the complexity of the activities provided. Participants were finally instructed to evaluate the performance of the algorithm by reexecuting the activities as many times as they wished. They were also encouraged to experiment and suggest ideas on how the training or recognition of activities could be improved in future versions of the system.

#### 5.5.1 Real-Time and 10 Fold Cross-Validation Performance

Table 5-51 presents the activities specified by the participants, the total accuracy per subject, and the range of the true positive and false positive rates obtained per activity


Figure 5-43: Placement of the three wireless accelerometers during the real-time evaluation experiment. The sensors were worn at the hip, dominant wrist, and dominant foot, right on top of the shoe laces.

using 10 fold cross-validation. Appendix A9 presents the performance measures per activity and the confusion matrix for each subject. The table shows that participants provided a wide set of activities, ranging from postures (e.g. *sitting still*) to complex activities involving a high degree of variability in motion such as *Taekwondo forms*. The total accuracy per subject presented in the table was high, consistent with the results presented in Section 5.4.8.1, ranging from 78.9% to 91.7%. In general, the algorithm performed extremely well on most of the activities provided by participants, reflecting the high accuracies obtained for Subject 3 because he specified several activities involving very similar upper body motion such as *throwing, bowling*, and *tennis serve*; these were confused several times during testing.

Table 5-52 lists the activities confused most often when participants tested the performance of the activity recognition algorithm in real-time. In general, the algorithm had problems differentiating activities with similar postures and upper body motion (e.g. hand movement). This is because the motion of the upper limbs is relatively unconstrained and the algorithms seem to have difficulty differentiating random motion patterns that occur occasionally from the distinctive patterns associated with an activity. For example, Table 5-51 and Table 5-52 show that the worse performance was obtained for Subject 3 because three of the activities he wanted to recognize (*throwing, bowling, and tennis serve*) involved very similar high energy motion on the hands. These activities were also difficult to recognize because they are disjointed and not strongly periodic. For example, during *tennis serve*, the subject executed the motion walking some steps forward and had to return to his original position after completing the example, which introduced examples of *walking* in the sequences. The same situation was observed for the *throwing* and *bowling*, activities. Another problem experienced during these activities (*throwing, bowling, and tennis serve*) was the higher loss of wireless signals due to the

high energy motion involved and the position of the hands during these activities. Another example of activities being confused by similar posture or upper body motion can be observed in Table 5-52 for Subjects 4 and 5. For example, drawing on paper and talking on the phone (while sitting) were confused with eating because all involved the sitting position and upper body motion. During drawing, the subject brought the pencil to her mouth several times, which looks similar to eating. When talking on the phone, the subject frequently switched the phone from hand to hand while he performed gestures with the opposite arm (as when explaining something). For subject 5, the activities confused were knitting with filing nails, filing nails with applying cream, and wash dish with wash hands. Again, all these activities involve similar posture and movement of the upper arms. Knitting and filing nails were also confused because they involve motion at the fingers that is poorly detected by the accelerometer at the wrist. One interesting observation related to the confusion between *filing nails* and *applying cream* observed during real-time testing and later confirmed by inspecting the C4.5 classifier, was that during training, the C4.5 classifier attempted to find any difference between the activities that did not involve the use of the upper arms. For example, *applying cream* (sitting on a chair) was not detected if the subject did not raise her dominant leg from the floor to apply cream on it. In general, the algorithm performed best recognizing activities involving characteristic motion of a particular limb, a particular speed of execution, or repetitive execution style, as expected.

### 5.5.2 Training Usability

Subjects reported being more conscious of time when they provided examples of the activities to recognize for two minutes. They expressed that at first they thought that two minutes was a short time but that during the study it felt much longer. Obviously some activities indeed *did* take longer to train because data was not collected when wireless signals were. However, subjects commented on training length even during training of activities where the accelerometer signal was of excellent quality. Participants suggested that the training process would be more amenable and realistic if examples could be collected when activities are executed in free-living. Two participants suggested using a mobile phone application to allowed people to collect activity examples by simply specifying the amount of data to be collected (e.g. in minutes) at the press of a button. Data would be collected right after the timer expired. Participants were members of the House\_*n* research group and their familiarity with wearable devices, particularly mobile phone technologies was high. Therefore, it is not surprising that they suggested mobile phone based solutions to the activity recognition interactive training problem.

Participants were more tolerant of the errors committed by the system when they made sense to the subjects. For example when *wash hands* and *wash dishes* were being confused, one subject with a background in architecture with no programming experience declared "this makes sense; this computer is wrong but pretty smart". The same subject, as well as others, also expressed a desire for being able to "fix" the recognition algorithm by providing more training examples for activities being poorly recognized or confused. Nevertheless, participants showed some level of frustration when the activity being recognized made no sense to them. For example, during the confusion between *talking on* 

Subject	Activiti	ies performed	Total Accuracy (%)	True Positive Range (%)	False Positive Range (%)
1	Bouncing on a ball Waving hand Shaking my leg Taekwondo Form #1 Side stretch	Jumping jacks Punching as I walk forward Lifting dumbbells Riding a bike Playing the drums	89.6	89.3 - 94.8	0.8 - 1.0
2	Walking Sitting still Scratching head Carrying box Washing dishes	Shaking hands Tossing ball in air Typing Talking on phone	91.7	84.5 - 98.2	0.4 - 0.17
3	Throwing Bowling Bouncing Typing Stepping	Stretching arm Walking Tennis serve Stretching legs Bending	78.9	70.7 - 93.2	1.3 – 3.8
4	Walk Type in computer Washing window Drawing in paper Wiping surface	Talking on the phone Sweeping Combing my hair Hammering a nail Eating	89.3	74.1 – 94.8	0.6 - 2.1
5	Walk Bicep curls Stretching Applying cream Brushing teeth	Wash dish Knitting Wash hands Filing nails Play piano	85.2	77.6 – 94.8	0.6 – 2.7

Table 5-51: Performance obtained by recognizing participant's activities using a C4.5 classifier with the *invariant reduced* feature set computed per axis over window lengths of 5.6s in length. Performance was measured using 10-fold cross-validation. The random guess probability is 10% for all activities shown in this table.

Subject	ect Activities Most Often Confused		Brief Explanation		
-	Activity Performed	Activity Confused With			
1	Playing the drums	Shaking my leg	Participant shook leg rhythmically when playing the drum.		
1	Taekwondo Form #1	Punching as I walk forward	One of the sequences in the form involved punching walking forward.		
2	Typing	Sitting	Sitting recognized as typing when subject typed slowly.		
3	Throwing Bowling Tennis serve	Throwing Bowling Tennis serve	Activities often confused due to their similarity in upper body limb motion.		
3	Bending	Stretching legs	The lower body limb motion was very similar for these activities.		
4	Drawing on paper	Eating	These activities were confused due to their similar posture (sitting) and upper body motion. The algorithm usually defaulted to eating when drawing and talking on the phone were performed		
4	Talking on the phone	Eating	performed.		
5	Knitting	Filing nails	Similar posture of overall body (sitting), posture of hands, and high motion on the fingers poorly detected by the accelerometer at the wrist.		
5	Filing nails	Applying cream	Same posture (sitting) and motion upper body motion similarity.		
5	Wash dish	Wash hands	Similar body posture (standing) and motion at the upper body (hands).		

Table 5-52: Activities confused the most found when participants tested the performance of the activity recognition algorithm in real-time.

*the phone* and *eating*, one of the participants with no familiarity with computer programming, machine learning or pattern classification techniques lightheartedly expressed, "stupid computer, does it look to you like I am eating? Maybe you are hungry". The confusion between the activities did indeed make sense to the algorithm designer because of the postures and upper limb body motion involved, but the participant did not understand at all why the confusion was happening.

In summary, the exploratory study confirms that a real-time implementation of the final activity recognition algorithm proposed in Section 5.4 can do well detecting activities with a modest amount of training data. In this case, cross-validation accuracies per subject ranged from 78.9% to 91.7%. In general, the algorithm performed best recognizing activities involving characteristic motion of a particular limb, a particular speed of execution, or repetitive execution style. Not surprisingly, the algorithm has more difficulties when trying to discriminate activities involving similar posture and similar motion, particularly at the upper body. This is most likely because the motion of upper limbs is so unconstrained in everyday life that during the short execution of some activities it is perceived as noise by the recognition algorithm. The study also suggests that better tools are necessary to allow people to (1) collect data during free-living conditions and to (2) fix the algorithms when activities are either being poorly recognized or confused with others. Both are challenging problems to solve and test. This exploratory study also suggests, however, that excellent results can be obtained using subject dependent training even when relatively small amounts of training data per activity is available. The challenge, therefore, is to create the necessary user interface applications to allow end-users who are unfamiliar with sensors or machine learning to easily and interactively provide the training data in a manner that is not perceived as intrusive, tedious, or boring. This will require the development of tools that acquire training data in fun, entertaining, and educational ways, perhaps by interleaving activity recognition applications with other applications, such as games.

### 5.6 Energy Expenditure Algorithm Experiments

This section presents a set of experiments to determine an effective regression algorithm, feature set, sliding window length, feature computation method, and signal filtering techniques amenable for real-time estimation of energy expenditure. The section also presents experiments to measure the impact of adding sensors worn on various body locations and the value of adding a wireless heart rate sensor when estimating energy expenditure. The experiments are organized so that each answers a relevant question about the algorithm parameters incrementally, starting from the most restrictive parameters (e.g. regression algorithm, feature set) to the least restrictive parameters (sensor modality and location). Where tradeoffs must be made, settings that might permit real-time performance are preferred.

### 5.6.1 Overview

All the experiments presented in this section are evaluated utilizing the MIT energy expenditure dataset (see Section 4.6.4). This data set is particularly challenging for energy estimation for the following reasons: (1) it contains energy expenditure data for a large number of activities (52) collected from 16 subjects, (2) it includes energy expenditure data for 26 activities with different intensity levels such as *walking* at different speeds and inclinations, *running* at different speeds, *cycling* at different speeds and resistance levels, and *rowing* at different resistance levels, (3) it incorporates energy expenditure data for 18 household activities containing examples of the unconstrained motion found in everyday life, and (4) it includes data for an activity labeled as *garbage* class or *unknown* activity that contains the energy expenditure data for all the time periods with no associated activity labels during the data collection. The energy expenditure of the garbage class is difficult to estimate because it contains data for periods when participants were resting after an exercise session and, consequently, the heart rate and energy expenditure data sometimes have high readings but there is no accelerometer motion associated with these readings.

Currently, equipment to collect energy expenditure data of free-living individuals is bulky and expensive. Therefore, it is unlikely that the equipment would be available to a new user of an affordable, consumer-based EE estimation system – the goal of this work. Therefore, in this section results are only presented for subject independent training. As in the previous section, the results presented in this section will also be clustered by activity categories that are helpful while analyzing the results. These activity categories are *postures*, *ambulation*, *exercise*, *resistance exercise*, and *household* activities. Appendix A2 presents a table indicating what activities are included in each of the aforementioned categories.

Finally, all the experiments presented in this section utilize same signal preprocessing and segmentation strategies, described in Sections 5.4.2 and 5.4.3.

Activity	MIT Dataset	Boston University Dataset	Difference between MIT
	METS (Mean ± Stu)	METS (Mean ± Stu)	readings
Lying down	0.9 ± 0.1	$0.9 \pm 0.1$	0.0
Standing	$1.0 \pm 0.2$	$0.9 \pm 0.0$	0.1
Sitting	$1.0 \pm 0.2$	$0.9 \pm 0.0$	0.1
Walking - Treadmill 3mph 0	$3.4 \pm 0.4$	$3.2 \pm 0.1$	0.2
Walking - Treadmill 3mph 6	$4.9 \pm 0.4$	$4.9 \pm 0.2$	0.0
Walking - Treadmill 3mph 9	$5.7 \pm 0.6$	$5.7 \pm 0.5$	0.0
Running - Treadmill 5mph 0	$7.0 \pm 0.8$	$6.9 \pm 1.3$	0.1
Cycling - 100 rpm 1.0 kg	$5.8 \pm 0.8$	$5.7 \pm 1.3$	0.1
Cycling - 60 rpm 1.0 kg	$3.3 \pm 0.5$	$3.1 \pm 0.6$	0.2
Cycling - 80rpm 1.5 kg	$5.6 \pm 0.7$	$5.9 \pm 2.5$	-0.3
Calisthenics - Bicep curls 5lb	$1.4 \pm 0.5$	$2.5 \pm 0.3$	-1.1
Calisthenics - Crunches	$1.4 \pm 1.2$	$1.8\pm0.0$	-0.4
Calisthenics - Sit ups	$4.3 \pm 1.2$	$0.6 \pm 0.0$	3.7
Unknown Activity	$2.4 \pm 1.0$	$2.2 \pm 0.3$	0.2

Table 5-53: Average energy expenditure readings in METs for 14 activities (mean and standard deviation) collected using the Cosmed K4b2 indirect calorimeter from 16 subjects at MIT and the Parvo Medics indirect calorimeter from 2 subjects at the Boston Medical Center

#### 5.6.2 MIT Energy Expenditure Dataset Validation

The quality of the energy expenditure data collected from an indirect calorimeter depends on the brand of device, the proper calibration of the equipment before the data collection, and the proper attachment of the face mask required to collect the  $O_2$  and  $CO_2$  readings. Consequently, collecting good quality energy expenditure data can be a challenging task. The goal of this section is to validate the quality of the energy expenditure data collected for this work by comparing it to the quality of energy expenditure data collected for similar activities by researchers at another institution utilizing an indirect calorimeter of different brand.

The energy expenditure data collected for this work was measured using the Cosmed K4b2 portable indirect calorimeter [125]. The K4b2 indirect calorimeter has been found to be a valid instrument to measure energy expenditure as compared with the Douglas bag in a wide range of activities and intensities [208]. However, the quality of the data collected can suffer from improper calibration of the equipment or improper attachment of the face mask to the face of the participant.

To validate that the portable K4b2 indirect calorimeter used in this work was providing quality data, the energy expenditure data collected using this indirect calorimeter for 14 activities was compared against the energy expenditure data collected for these same activities by researchers at the Boston Medical Center utilizing the Parvo Medics TrueOne 2400 stationary metabolic measurement system [126]. Before performing the comparison, the data collected using the K4b2 indirect calorimeter was filtered using a 15s running average filter to reduce noisy readings. Table 5-53 presents the average number of METs obtained per activity for the data collected at MIT and the Boston Medical Center. The duration of each activity is between 3-4mins for both datasets, except for physically demanding activities such as *crunches, sit-ups* and *bicep curls* that sometimes have durations of less than one minute. Non-steady state energy expenditure readings were not eliminated for any of the activities in both datasets.

Visual inspection of Table 5-53 reveals that the difference in average METs readings per activity is near zero for most activities. This indicates that the data collected at both

institutions is comparable in quality. The only two activities that show a large difference are *bicep curls* and *sit-ups*. *Bicep curls* present a lower value for the MIT dataset perhaps because a bicep curls chair that allowed reclining the elbows was used during the MIT data collection. During the Boston Medical Center data collection, the subjects executed bicep curls without reclining the elbows on any surface. The extremely low value associated with *sit-ups* of 0.6 METs for the Boston Medical Center dataset suggests either a problem while collecting the data for this activity (e.g. air escaping from collector tube) or that the short duration of the examples for this activity (<40s) was not enough to reach steady state. Finally, it can be observed from the table that the standard deviation over the MIT dataset is higher than the one observed for the Boston Medical Center dataset. This is because the MET values were computed over 16 subjects for the MIT dataset and only over two subjects for the Boston Medical Center dataset.

In conclusion, from Table 5-53, we can see that the energy expenditure data collected for most activities is very similar; suggesting that the quality of the data collected using the Cosmed K4b2 indirect calorimeter is reasonable for the experiments performed in this work. Prior work utilizing the Cosmed K4b2 indirect calorimeter to measure energy expenditure such as the one performed by Bassett et al. [200] has measured standard deviations ranging from 0.31 to 1.58 MET for a set of 20 activities including household and exercise activities with moderate intensities. The work by Strath et al. [174] found standard deviations between 0.4 to 1.1MET over 14 lifestyle activities, and the work by Crouter et al. [34] standard deviations between 0.13 and 1.63MET for 18 exercise and lifestyle activates. Therefore, the standard deviations obtained for the MIT energy expenditure dataset during this analysis ranging between 0.1 and 1.2MET are within the limits reported in prior work.

### 5.6.1 How Well Has Energy Expenditure been Estimated From Accelerometer and Heart Rate Data in the Past?

This section discusses some of the most recent results in the area of energy expenditure estimation from accelerometer and heart rate data. The objective is to give the reader some background to better interpret the results later presented in this thesis. In addition, this section defines what would be considered to be a significant result in this work with respect to prior work. There is a large body of research in estimating energy expenditure from accelerometer and heart rate data. As a result, this section discusses only some relevant results published after year 2000.

Presently, the two state-of-the-art algorithms to estimate energy expenditure from a single accelerometer placed at the hip are the work by Crouter et al. [34] and Rothney [152, 181]. Crouter et al. [34] discriminates between sedentary, ambulatory, and lifestyle activities using the coefficient of variation computed over windows of 10s in length and applies two different regression equations for ambulatory and lifestyle activities. When sedentary activities are detected, a constant MET value of 1 is predicted. The subject independent evaluation of this algorithm presented by the authors indicates that energy expenditure can be predicted with a correlation coefficient (r) of 0.96, a standard error of the estimate (SEE) of 0.73MET, and a maximum absolute error deviation (MAED) of 0.75MET (with respect to ground truth energy expenditure). The algorithm was evaluated over 17 activities collected from 20 subjects. These results are strong with respect to prior

work; nevertheless, the performance measures presented are computed over mean values per activity instead of over prediction values obtained over minute-by-minute windows. In other words, error measures are computed by comparing the mean value of the predictions for each activity and the mean value of the energy expenditure measured using the indirect calorimeter for each activity. This smoothes out possible over and under predictions that could have happened during minute by minute estimates. This study also excluded the data collected for the *cycling* activity from analysis because the Actigraph activity monitor placed at the hip was not able to record any activity counts for this activity. This reveals one of the main problems of predicting energy expenditure from accelerometers at the hip: poor estimation of energy expenditure for upper body and non-ambulatory lower body motion. Analysis of performance per activity for this study showed that the largest errors occurred for activities involving upper body motion such as basketball, racquetball, vacuuming, and mowing the lawn. This is consistent with other studies that have found that accelerometers mounted at the hip significantly underestimate lower body activities such as cycling and sliding [33]. The work by Crouter et al. also explored the performance of commonly employed regression equations such as the Freedson equation [145], the Swartz equation [148], and the Heldenman equation[146] in estimating energy expenditure. The study found that these equations overestimate energy expenditure for activities with associated energy expenditures of less than 2METs and underestimate the energy expenditure for activities with energy expenditure values greater than 2MET.

The work by Rothney [152, 181] presents an algorithm that estimates energy expenditure using an artificial neural network (ANN) trained with the following features computed from a single accelerometer at the hip over one minute windows: Power spectral density, number of signal peaks, inter-quartile range, and coefficient of variation. The ANN is trained from nearly 24 hours of data collected from 81 participants. This approach achieves a correlation coefficient of r=0.93 and a root mean squared error (RMSE) of 0.47kcal/min. The difference in total daily energy expenditure achieved is  $21\pm115$  kcal/day. This compares favorably with the measurement intervals of 100-250kcal/day required for sustainable weight loss interventions. This work also explored the performance of the IDEEA device [41] and an Actigraph at the hip using the Freedson equation in estimating energy expenditure. The results obtained were r=0.92, RMSE=0.59 kcal/min for the IDEEA monitor and r=0.9, RMSE=0.74 kcal/day for the Actigraph located at the hip. Thus, the improvement in the correlation coefficient achieved by this method ranges between 0.01 and 0.03 units and between 0.12 and 0.27 kcal/min for the RMSE error.

Prior work by Basset et al. [200] found that the Caltrac, Kenz, Yamax SW-701, and CSA activity monitors (using different regression equations) underestimate energy expenditure for *lifestyle* activities such as *ironing*, *cooking*, *washing dishes* and overestimate energy expenditure for *walking*. The same work also found that energy expenditure was best predicted using the Hendelman's regression equation computed from the counts generated by a single CSA activity monitor placed at the hip over 28 activities. The mean error score (criterion - prediction) obtained was 0.05MET. However, this evaluation measure does not accurately reflect estimation error since underestimations (predicted<real) and overestimations (predicted>real) might cancel out during the averaging of error scores giving a false impression of low overall error. The

mean absolute error (MAE) or the root mean squared error (RMSE) that computes the absolute value of the error score or the squared of the error score before averaging would have produced a more reliable estimate of the real error. One disadvantage of the Hendelman's equation is that it predicts 2.9 MET by default when no motion is present so it would likely overestimate energy expenditure over the course of a normal day [200]. The mean error scores obtained were 0.97 MET for the work energy theorem [32] from a CSA monitor, 0.47 MET for the Freedson's equation [145] using a CSA monitor, 0.83 MET using the Caltrac, 0.96MET using Kenz (select 2), and 1.12 MET using the Yamax device. The correlation coefficients between these sensors and energy expenditure readings obtained were r=0.62 for work energy theorem and CSA; r=0.32 for Freedson's equation and CSA; r=0.58 for Caltrac; r=0.55 for Kenz; and r=0.49 for Yamax. The authors explain that these relatively low correlations were obtained because the activities explored included a wide variety of moderate intensity activities. The authors also point out that previous studies analyzing only treadmill *walking* and *jogging* have reported correlations between 0.80 and 0.92 [145, 146, 178, 231-233].

Prior work has also explored the combination of acceleration and heart rate data to predict energy expenditure. For example, The work by Haskell et al. [48] found that the correlation coefficient increased from r=0.83 to r=0.9 (+0.07units) when an accelerometer at the arm was incorporated to heart rate data during subject dependent estimation of energy expenditure. This result was obtained over data collected from 19 participants performing a wide range of activities in a laboratory setting. The same work found that the difference in the correlation coefficient between subject independent and subject dependent estimation of energy expenditure was only 0.09units (r=0.94 for subject dependent and r=0.85 for subject independent).

The work by Strath et al. [83] explored the estimation of energy expenditure by applying two different subject dependent heart rate regression equations depending on upper body or lower body motion as detected by two accelerometer placed at the wrist and thigh (HR+M technique). The model was trained over arm and leg activities (treadmill waking and arm ergometry) collected from 30 participants at a laboratory and tested over lifestyle activities performed for 15min each by the same individuals. It was found that a linear regression equation trained over heart rate data alone (subject dependent training) achieved a correlation coefficient of r=0.81, and on average overestimated energy expenditure by 0.4 MET. When two different heart rate regression equations were applied depending on upper or lower body motion, the correlation coefficient increased to r=0.9. This combination was also found to not significantly over or under estimate energy expenditure. Follow up validation work performed also by Strath et al. [174] found that energy expenditure can be predicted in a subject dependent manner with r=0.79 and SEE=0.76 MET using the heart rate flex method (FlexHR). Other work (e.g. [234, 235]) has also found correlation coefficients between 0.54 and 0.98 using the Flex HR method (subject dependent training). In contrast, Strath et al. [174] found that the HR+M technique achieved a performance of r=0.9 and SEE=0.55MET. This work also found that the FlexHR method overestimated energy expenditure over most activities while the HR+M technique did not. Thus, the difference in performance found in this work was 0.11 units for the correlation coefficient and 0.21MET for the SEE. These results were obtained from data collected from 10 participants performing physical tasks in a field setting over 6 hours.

Most recent work by Strath et al. 2005 [49] compared the performance of two methods that combine acceleration and heart rate data: The HR+M technique [83] (previously discussed) and the branched model [175]. This work evaluated the performance of these techniques in a subject dependent and independent manner, and compared their performance with the one obtained using a single accelerometer at the hip. The subject dependent results obtained were r=0.9, RMSE=0.64 MET for the HR+M technique and r=0.86, RMSE=0.67 MET for the Branched model. During subject independent evaluation, the performance found was r=0.81, RMSE=1.07 MET for the HR+M technique, r =0.86 and RMSE=0.76 MET for the Branched model, and r=0.64 and RMSE=1.22 METs when energy expenditure is predicted using the Freedson equation from a single accelerometer at the hip. Thus, the difference in performance between subject dependent and independent training for the HR+M technique is 0.09units for the correlation coefficient, and 0.43 MET for the RMSE. For the branched model, the difference is 0 for the correlation coefficient and 0.09 for the RMSE. The difference in performance between the branched model (best performing) and a single accelerometer at the hip was 0.22 for the correlation coefficient and 0.46 for the RMSE during subject independent evaluation.

The work by Swartz et al. [148] explored the performance obtained while estimating energy expenditure by combining a CSA accelerometer at the hip and another one at the dominant wrist. The data used in the evaluation was collected from 70 participants performing one to six activities from categories such as yardwork, housework, family care, occupation, recreation and conditioning. Linear regression equations were created for the sensor at the hip, the wrist, and the hip and wrist combination. The results obtained were r=0.56 and SEE= 1.1MET for the hip sensor, r=0.18 and SEE=1.38MET for the wrist sensor, and r=0.58 and SEE=1.14MET for the combination hip and wrist. The work found that the variance explained by the accelerometer at the wrist was only 5% and the variance explained by the accelerometer at the hip was 31.7%. Consequently, the authors concluded that the addition of the accelerometer at the wrist was not worth the extra cost and data analysis involved. Unfortunately, it is unclear from this work if energy expenditure is estimated in a subject dependent or independent manner. Furthermore, the result obtained in this work contradict prior findings by Melanson and Freedson [233]. These authors estimated energy expenditure using three accelerometers at the hip, wrist, and ankle during *walking* and *jogging* and found that activity counts significantly correlated with energy expenditure, no matter what accelerometer is used to estimate energy expenditure. The correlation coefficients found ranged from 0.66 to 0.81. The authors also found that the combination hip and wrist or wrist and ankle produced the highest correlation coefficient of r=0.94.

Finally, recent work [96] validating the performance of the SenseWear Pro armband in estimating energy expenditure in a subject independent manner found performances of r=0.77 and percent difference of  $6.9 \pm 8.5\%$  for walking, r=0.28 and percent difference of  $28.9 \pm 13.5\%$  for cycle ergometry, r = 0.63 and percent difference of  $17.7 \pm 11.8\%$  for stepping exercise, and r = 0.74 and percent difference:  $21.7 \pm 8.7\%$  for arm ergometry using data collected from 40 participants. Thus the lowest correlation coefficient and largest percentage error was obtained for cycling ergometry. This is because from its location at the upper arm, this armband presents difficulties detecting non-ambulatory

Work	Approach	Features	Sensors	Num	Results
2006 [34]	Use single accelerometer at hip but create activity dependent regression models Classify activities into walking/running, lifestyle, and sedentary and apply different regression models to each.	Activity counts, coefficient of variation	One uniaxial accelerometer at hip	17/20 3hrs per subject	Subject independent Maximum absolute error deviation MAED = 0.75 Within 0.75 of Measured METs. r = 0.96, SEE = 0.73 MET.
2007 [152]	Use single accelerometer at hip but extract features over signal and use non-linear regression Use a neural network to map features computed over the accelerometer data into energy expenditure.	Peak intensity, IQI, min CV, low Energy, and moderate to vigorous energy.	2-axis accelerometer at hip	12/81 24hrs per subject	Subject independent Neural network: r=0.93, RMSE=0.47 kcal/min Difference in TEE of 21±115 kcal/day IDEEA monitor: r=0.92, RMSE=0.59 kcal/min Actigraph at hip, Freedson equation r=0.9, RMSE=0.74 kcal/day
2005 [49]	Combine accelerometer and heart rate data and use different regression equations for upper or lower body motion. Estimate energy expenditure using two models: arm-leg HR+M and branched model	Activity counts, HR above resting HR	Three accelerometers at hip, wrist, and thigh and a chest heart rate monitor	2/10 6hours per subject	Subject dependent Arm-leg- heart rate + M: r = 0.9, RMSE = 0.64 MET Branched model: r=0.86, RMSE = 0.67 MET Subject independent Arm-leg- heart rate + M: r = 0.81, RMSE = 1.07 MET Branched model: r = 0.86, RMSE = 0.76 MET hip accelerometer alone using Freedson Eq: r = 0.64, RMSE = 1.22 METs
2004 [96]	Combine multiple sensor types at single body location Estimate energy expenditure from five wearable sensors located at the dominant arm over the triceps muscle using proprietary regression formulas.	Proprietary features extracted from the five sensors	2-axis acc, heat flux, galvanic skin response, skin temperature, and near body temperature.	4/40 1.6 hours per subject	Subject independent walking r=0.77, percent difference: 6.9 $\pm$ 8.5% cycle ergometry r=0.28, percent difference: 28.9 $\pm$ 13.5% stepping exercise r = 0.63, percent difference: 17.7 $\pm$ 11.8% arm ergometer r = 0.74, percent difference: 21.7 $\pm$ 8.7%
2001 [83]	Apply to different heart rate regression equations depending on upper body or lower body motion as detected by an accelerometer at the wrist and an accelerometer at the thigh.	Activity counts, beats per minute	Three CSA Actigraphs at the arm, thigh, and hip.	2/30	Subject independent hip accelerometer alone using Freedson Eq: r=0.73, average underestimation of 1.1MET Subject dependent Heart rate: r=0.81, average overestimation of 0.4MET Heart rate and Arm + thigh accelerometers r=0.9, No significant over or under prediction

Table 5-54: Summary of some relevant results in energy expenditure estimation from accelerometer and heart rate data after year 2001. The numbers in the column labeled as Num indicate Number of activities/number of subjects.

lower body activity. As discussed in Section 3.2.3, the armband utilizes five sensors to estimate energy expenditure from its location at the dominant upper arm: A 2-axis

accelerometer, heat flux, galvanic skin response, skin temperature, and near body temperature.

Table 5-54 presents a summary of some of the most recent results (after 2001) obtained while estimating energy expenditure from accelerometers and heart rate monitors.

In summary, it can be concluded from prior work that in general, subject dependent regression models perform better than subject independent models [49, 152]. However, the difference in performance is relatively small. For example, the difference in performance between subject dependent and independent training for two state of the art methods that combine accelerometer and heart rate data ranges from 0 and 0.09 for the correlation coefficient and between 0.09 and 0.43MET for the RMSE [49]. The difference in performance between the branched model (state-of-the-art algorithm that combines accelerometer and heart rate data) and a single accelerometer at the hip was 0.22 for the correlation coefficient and 0.46 for the RMSE during subject independent evaluation [49].

When energy expenditure is estimated from a single accelerometer at the hip, the state of the art results are the ones obtained by Crouter et al.: r=0.96, SEE=0.73MET, and MAED=0.75MET obtained over 17 activities. Prior algorithms have achieved correlation coefficients between 0.80 and 0.92 for treadmill *walking* and *jogging* activities [145, 146, 178, 231-233]. When the number of activities is increased, the correlation coefficient decreases according. For example, the work in [200] found correlation coefficients between 0.32 and 0.62 using four different brands of activity monitors placed at the hip.

Finally, the performance of utilizing several accelerometers to improve energy expenditure estimation is controversial. The work by Swartz et al. [148] found that the addition of an accelerometer at the wrist to an accelerometer at the hip improved the correlation coefficient only 0.02units and SEE by 0.04MET. Conversely, the work by Melanson and Freedson [233] found that activity counts significantly correlated with energy expenditure, no matter what accelerometer was used. The authors also found that the sensor combination hip and wrist or wrist and ankle produced the highest correlation coefficient of r=0.94. The analysis was performed on data collected from 15 subjects performing treadmill *walking* and *running* activities at different grades.

Therefore, in this work, it would be considered a significant result if energy expenditure for 52 activities can be estimated with a correlation coefficient between 0.80 and 0.92. This range is the best one obtained for *jogging* and *walking* using a single accelerometer at the hip and using heart rate data in a subject independent manner. A strong result would also be to obtain a RMSE between 0.64 and 0.86, the lowest RMSE obtained by combining accelerometer and heart rate data (HR+M and branched model) and using a single accelerometer at the hip (RMSE~0.73) as evaluated over less than 18 activities. An excellent result would also be to corroborate that the utilization of several accelerometers improves energy expenditure estimation, as one might intuitively expect at least for activities involving upper body and lower body motion.

### **5.6.2** How Well Can Energy Expenditure be Estimated Using Leading Algorithms from Prior Work?

This section explores the performance of several energy expenditure estimation algorithms that can be considered either state-of-the-art or popular by the medical community during research studies. These algorithms include the Crouter et al. 2006 [34] Actigraph based algorithm, estimation of energy expenditure using multivariable linear regression, estimation of energy expenditure using the Compendium of Physical Activities assuming the activities performed are known (e.g. from questionnaires or diaries), and estimation of energy expenditure using activity dependent regression models such as multivariable linear regression and non-linear regression when the activities performed are also known. The section also compares the performance of the different approaches to establish a performance baseline as a way to compare results obtained in upcoming sections where a new energy expenditure estimation algorithm is proposed.

# 5.6.2.1 How Well Can Energy Expenditure be Estimated Using a State-of the-Art Actigraph-Based Algorithm?

There exist several algorithms that have been developed to estimate energy expenditure from the acceleration counts (per minute) generated by the Actigraph activity monitor [32] (uniaxial accelerometer). Most of these algorithms assume that the device is worn at the hip (e.g. [145, 146, 148, 200]) and predict energy expenditure over one minute intervals. The current state-the-art Actigraph-based algorithm is the recently published 2regression method proposed by Crouter et al. [34]. The main idea behind this algorithm is to classify activities into three categories before estimating energy expenditure: (1) sedentary activities, (2) ambulatory activities such as walking and running, and (3) *lifestyle* activities. Once the activities are recognized, different regression models are applied for each activity type to estimate energy expenditure. The algorithm recognizes sedentary activities by simply setting a threshold over the acceleration counts. Once they are recognized, they are assigned an energy expenditure equivalent to 1MET. Ambulatory activities such as *walking* and *running* are differentiated from *lifestyle* activities by setting a threshold over the coefficient of variation (CV) as computed over 10s windows. If *walking* and/or *running* are detected, a linear regression model is applied to estimate their energy expenditure; otherwise, an exponential regression model is applied to estimate the energy expenditure associated with *lifestyle* activities. The pseudo code for this algorithm is presented in Figure 5-44.

This section explores how well energy expenditure can be estimated utilizing this algorithm by running it over the Actigraph data collected at the hip during the data collections performed for this work. The Actigraph data was originally collected using a 1s epoch (window) and thus, it had to be converted to 10s epochs as required by this algorithm. The algorithm was re-implemented in Java in order to present energy expenditure estimation results per activity. In order to validate the proper implementation of the algorithm in Java, its output (overall energy expenditure estimate) was compared against the output generated by the original implementation of this algorithm included in the Actigraph activity monitor software [32].

if (counts/min 
$$\leq$$
 50) EE\_METS = 1.0,  
else if (counts/min > 50){  
if (CV == 0 or CV > 10)  
EE\_METS = 2.330519 + (0.001646 × counts/min -  
[1.2017 × 10<sup>-7</sup> × counts/min<sup>2</sup>]+[3.3779 × 10<sup>-12</sup> × counts/min<sup>3</sup>]  
(R<sup>2</sup> = 0.854; SEE = 0.940)  
else if (CV/10s  $\leq$  10)  
EE\_METS = 2.379833 × exp(0.00013529 × counts/min)  
(R<sup>2</sup> = 0.701; SEE 0.275)

Figure 5-44: Pseudo code for the 2-regression Crouter et al. [34] Actigraph-based energy expenditure estimation algorithm. CV stands for coefficient of variation and is computed by dividing the standard deviation by the mean over a window of 10s.

Table 5-55 presents the performance over all the activities and Table 5-56 the performance per activity category when estimating energy expenditure using the Crouter et al. algorithm over the MIT energy expenditure dataset. From Table 5-55, it can be seen that the coefficient of correlation is 0.36 (relatively low) and that the RMSE is 2.66 METs. Table 5-56 shows that the root mean squared error (RMSE) is lower for postures and household activities and higher for resistance exercise, all exercise in general, and lower body activities. The algorithm generates small errors for postures because they are easily recognized by simple thresholding of the acceleration counts and assigned a MET value of one. Household activities have a small error because most of them involve either lifestyle activities or ambulation that the algorithm differentiates well using the coefficient of variation to then apply the corresponding regression model. A high error was expected for exercise and resistance exercise activities for two reasons: (1) accelerometers have difficulties recognizing the effort and resistance load associated with some of these activities (e.g. rowing at 30spm at different resistance levels), and (2) most activities in these categories include either lower body (e.g. cycling) or upper body motion (e.g. *bench weight lifting* and *bicep curls*) that is difficult to measure properly from an single accelerometer (Actigraph) located at the hip. In fact, the lower body category also presents a high error due to the poor detection of motion at the legs during cycling. When Actigraph vs. Cosmed energy expenditure is plotted (as shown in Appendix B1), it can be observed that the energy expenditure predicted for activities involving upper or lower body motion such as *cycling*, *bench weight lifting*, and *bicep curls* is 1MET (energy expenditure associated with sedentary activities). This is because these activities are confused with sedentary activities by the algorithm due to the inability of the accelerometer at the hip to capture these motions. Figure 5-45 presents the energy expenditure estimated for subject MIT-018 during the (a) cycling activity and (b) the bicep curls activity using the Crouter et al. algorithm from a single Actigraph accelerometer at the hip. The important thing to notice from these Figures is that the

Error Measures over all activities	Error Measures Values in METs
Correlation Coefficient (r)	$0.36\pm0.09$
Root Mean Square Error (RMSE)	$2.66\pm0.62$
Mean Absolute Error (MAE)	$1.94\pm0.45$
Maximum absolute error Deviation (MAED)	$6.94 \pm 1.77$

Table 5-55: Error statistics for estimating energy expenditure for 52 activities using the 2-regression Crouter et al. Actigraph-based algorithm with respect to the energy expenditure measured using the Cosmed K4b2 indirect calorimeter for the MIT energy expenditure dataset (n=16).

Activity	RMSE		
category	(MAE)		
Postures	0.6±0.2 (0.5±0.2)		
Ambulation	2.4±0.5 (2.4±0.5)		
Exercise	2.8±0.8 (2.8±0.8)		
Resistance	3.4±0.7 (3.4±0.7)		
Household	1.0±0.3 (1.0±0.3)		
Upper Body	1.2±0.3 (1.2±0.3)		
Lower Body	2.7±0.5 (2.7±0.5)		

Table 5-56: Root Mean Squared Error and Mean Absolute Error (shown in parenthesis) per activity category when estimating energy expenditure using the 2-regression Crouter et al. Actigraph-based algorithm. Errors are computed with respect to the energy expenditure measured using the Cosmed K4b2 indirect calorimeter for the MIT energy expenditure dataset (n=16).

Error Measures over all activities	Error Measures Values in METs
Correlation Coefficient (r)	$0.40 \pm 0.14$
Root Mean Square Error (RMSE)	$3.23\pm0.39$
Mean Absolute Error (MAE)	$2.45\pm0.18$
Maximum absolute error Deviation (MAED)	$7.01 \pm 2.27$

Table 5-57: Performance over all activities when estimating energy expenditure for 11 activities using the 2-regression Crouter et al. algorithm over the Boston University dataset (n=2).

Activity	RMSE	MAE	MAED
Lying down	$0.24\pm0.02$	$0.17\pm0.02$	$0.66\pm0.11$
Standing	$0.10\pm0.00$	$0.08\pm0.01$	$0.18\pm0.02$
Sitting	$0.11\pm0.02$	$0.10\pm0.02$	$0.15\pm0.05$
Walking - Treadmill 3mph 0	$4.53\pm0.41$	$4.52\pm0.40$	$5.05\pm0.50$
Walking - Treadmill 3mph 6	$3.30\pm0.95$	$3.26\pm0.96$	$4.11 \pm 1.00$
Walking - Treadmill 3mph 9	$2.62 \pm 1.03$	$2.54 \pm 1.03$	$3.56 \pm 1.20$
Running - Treadmill 5mph 0	$3.16\pm0.74$	$2.89 \pm 0.77$	$5.03 \pm 0.96$
Cycling - 60 rpm 1.0 kg	$3.06\pm0.83$	$3.05\pm0.82$	$3.28\pm0.95$
Cycling - 80rpm 1.5 kg	$5.65 \pm 1.70$	$5.63 \pm 1.72$	$6.00 \pm 1.58$
Cycling - 100 rpm 1.0 kg	$5.46 \pm 1.18$	$5.38 \pm 1.11$	$5.99 \pm 1.44$
Unknown	$3.31 \pm 0.50$	$2.77 \pm 0.23$	$6.71 \pm 2.70$

Table 5-58: Performance per activity when estimating energy expenditure using the 2-regression Crouter et al. algorithm over the Boston University dataset (n=2). RMSE stands for root mean squared error, MAE for mean absolute error, and MAED for maximum absolute error deviation.



Figure 5-45 Graphical representation of the energy expenditure estimated for the (a) cycling activity and (b) bicep curls activity using the Crouter et al. method from a single Actigraph placed at the hip for subject MIT-018. The label 'Prediced' corresponds to the estimated energy expenditure and the label 'Cosmed' to the ground truth energy expenditure data collected from the Cosmed K4b2 indirect calorimeter.

inability of a single accelerometer at the hip to measure lower body and upper body activity produces underestimations of energy expenditure for the *cycling* and *bicep curls* activities. Obviously, any accelerometer placed at the hip might generate readings for intense upper body or lower body activity if it is not properly attached; nevertheless, estimates of energy expenditure from these readings would be unreliable because they could be generated by any type of activity and would most likely bias the energy expenditure predictions for other activities if a single regression model is utilized for all activities.

When the errors per activity are analyzed for the Crouter et al. algorithm over the MIT energy expenditure dataset as shown in Appendix B1, it is found that the highest maximum absolute error deviations happen for the following activities: *bench weight lifting* different weight loads (2.2-2.32 METs), *sit-ups* (4.5), *cycling* at different resistance levels (5.6-6.6), *rowing* at different resistance levels (5.5-6.5), *ascending stairs* (4.3), *walking* on a treadmill at different inclinations (4.2-5.9), and for the *unknown* activity (5.4). All of these activities include changes in resistance or work load effort or upper body motion that is poorly detected by an accelerometer at the hip. The *unknown* class has a high MAED because it consisted primarily of resting episodes after exhausting exercise where the subject was *standing* or *sitting*. Consequently, the MET readings where high during these periods of time because subjects had just finished the exercise routine but the amount of motion from the accelerometer was uncharacteristically low.

The Crouter et al. algorithm was also run over the Boston Medical Center dataset (n=2). Table 5-57 and Table 5-58 present the results. From these tables, it can be observed that the coefficient of correlation of 0.4 is very similar to the one obtained over the MIT dataset (0.36). Table 5-57 shows that the RMSE obtained over this dataset is higher (+0.57MET) than the one obtained over the MIT dataset. This increased error is the result of the poorer performance for cycling activities as compared with the MIT dataset and the higher impact of this error over the reduced set of activities in this dataset (11). The Actigraph readings for cycling were close to zero for all the activity duration (see the Actigraph vs. Parvo indirect calorimeter plots in Appendix B1) As a result, cycling was confused with sedentary activities and assigned a MET value of one. If one compares the plots of energy expenditure predictions from the Actigraph for both datasets, one finds that the Actigraph readings are not zero during cycling for some MIT data collections (e.g. subject MIT-004 in Appendix B1). The reason is that the Actigraph was located inside a pouch on a belt during the MIT data collections that oscillated more during the cycling activity than the Actigraph used in the Boston Medical Center data collection. This Actigraph was tightly attached using a belt clip and Velcro.

In conclusion, the prediction of energy expenditure using the Crouter et al. method from an accelerometer at the hip performs poorly for activities involving primarily upper body or lower body motion and activities involving load or resistance effort. One potential problem with the Crouter algorithm, as well as with other standard EE estimation algorithms used in the medical community, is the prediction of energy expenditure over one minute intervals. This relatively long window of time may be a problem because (1) EE for short duration activities (<1min) might be poorly predicted and (2) this window introduces a relatively long real-time prediction delay that might be restrictive for some real-time medical interventions.

### 5.6.2.2 How Well Can Energy Expenditure be Estimated Using Simple Linear Regression Algorithms?

This section explores how well energy expenditure can be estimated using simple regression algorithms in order to obtain a baseline on performance. Table 5-59 presents the form of the simple regression equations explored in this section to estimate energy expenditure. The equations in the table are presented in increasing order of complexity. The general procedure followed in this section is to first present the form of the energy expenditure prediction equation (from Table 5-59), explain the intuition behind the form of the equation being assumed, and then to learn the equation coefficients from the energy expenditure data contained in the MIT energy expenditure dataset. The coefficients of the formulas are found by minimizing the least-squares error with respect to the data in the MIT energy expenditure dataset. This is achieved by utilizing the LSQCURVEFIT function in MATLAB [227]. This function is able find the coefficients that best fit the linear or non-linear equation provided in the least-squares sense with respect to the training data provided (MIT energy expenditure dataset). The value of the initial guess used for all the coefficients was one.

Equation (a) is the simplest one in Table 5-59 and predicts energy expenditure by simply multiplying the overall motion experienced by the body (ACTotalAbsArea) by a constant. Appendix A3 provides an explanation of the ACTotalAbsArea feature, as well as the other features used in the equations shown in Table 5-59 (ACAbsArea, ACTotalSF, and ACSF). The intuition behind equation (a) is that energy expenditure should be proportional to the overall amount of motion (acceleration) experienced by the body. The proportionality constant simply scales overall motion to match the ground truth energy expenditure values. Table 5-60 presents the coefficients of equation (a) as learned from the data and Table 5-61 presents the performance of predicting energy expenditure over the data in the MIT dataset using subject independent training. Table 5-61 shows that the overall RMSE is 1.75 METs, and that the correlation coefficient is relatively high 0.68 (maximum possible is one). The improvement in performance over Crouter et al. is 0.32 units for the correlation coefficient and 0.91MET for the RMSE. This is a considerable improvement and is achieved mostly due to the utilization of more sensors (7) that are capable of measuring upper body and lower body motion. The table also shows that the lowest RMSE errors are obtained for the household activity category. This is because most activities in the household category include motion patterns that are easy to detect from accelerometers (e.g. ambulation). The household activities that show the highest errors are activities involving effort due to resistance or work load such as *carrying* groceries, gardening, weeding, scrubbing a surface, and washing windows whose RMSE error oscillates around 1 MET. The activity categories with higher RMSE errors are exercise and resistance exercise. This is also because some activities in this category involve different resistance levels or work loads such as cycling and rowing at different resistance levels.

Equation	Form of prediction equation assumed
(a)	$METs = C_1 \cdot ACTotalAbsArea$
(b)	$METs = C_1 \cdot ACTotalAbsArea + C_2$
(c)	$METs = C_1 \cdot ACAbsArea_{Hip} + C_2 \cdot ACAbsArea_{NDWrist} +$
	$C_3 \cdot ACAbsArea_{DWrist} + C_4 \cdot ACAbsArea_{DFoot} +$
	$C_5$ ·ACAbsArea <sub>NDFoot</sub> + $C_6$ ·ACAbsArea <sub>DThigh</sub> +
	$C_7$ ·ACAbsArea <sub>DUpperArm</sub> + $C_8$
(d)	$METs = C_1 \cdot ACTotalSF + C_2$
	where
	$ACTotalSF = 0.578 \cdot ACAbsArea_{Hip} + 0.05 \cdot ACAbsArea_{NDWrist} + 0.05 $
	$0.05 \cdot ACAbsArea_{DWrist} + 0.161 \cdot ACAbsArea_{DFoot} +$
	0.161 ·ACAbsArea <sub>NDFoot</sub>
(e)	$METs = C_1 \cdot ACAbsArea_{Hip} +$
	$C_2$ ·ACAbsArea <sub>NDWrist</sub> + $C_3$ ·ACAbsArea <sub>DWrist</sub> +
	$C_4$ ·ACAbsArea <sub>DFoot</sub> + $C_5$ ·ACAbsArea <sub>NDFoot</sub>
	$+ C_{6}$

Table 5-59: Form of the simple regression equations explored to estimate energy expenditure from the MIT energy expenditure dataset. Appendix A3 explains how *ACTotalAbsArea* is computed.

Equation	Activities	Prediction equation learned from data
(a)	All	$METs = 0.021 \cdot ACT otalAbsArea$
(b)	All	$METs = 0.013 \cdot ACT otalAbsArea + 1.58$
(c)	All	$METs = -0.026 \cdot ACAbsArea_{Hip} + 0.012 \cdot ACAbsArea_{NDWrist} +$
		$0.019 \cdot ACAbsArea_{DWrist} + 0.019 \cdot ACAbsArea_{DFoot} +$
		0.015:ACAbsArea <sub>NDFoot</sub> + 0.024:ACAbsArea <sub>DThigh</sub> +
		$0.017 \cdot ACAbsArea_{DUpperArm} + 1.49$
(d)	All	$METs = 0.080 \cdot ACT otalSF + 1.83$
(e)	All	$METs = -0.0086 \cdot ACAbsArea_{Hip} +$
		0.015 · ACAbsArea <sub>NDWrist</sub> + 0.025 · ACAbsArea <sub>DWrist</sub> +
		$0.021 \cdot ACAbsArea_{DFoot} + 0.021 \cdot ACAbsArea_{NDFoot}$
		+ 1.53
(e)	Accelerometer	$METs = 0.022 \cdot ACAbsArea_{Hip} +$
	Recognizable	0.020·ACAbsArea <sub>NDWrist</sub> + 0.017·ACAbsArea <sub>DWrist</sub> +
	Activities	$0.0097 \cdot ACAbsArea_{DFoot} + 0.014 \cdot ACAbsArea_{NDFoot}$
		+ 1.34
(e)	Postures and	$METs = 0.029 \cdot ACAbsArea_{Hip} +$
	Ambulation	$0.018 \cdot ACAbsArea_{NDWrist} + 0.012 \cdot ACAbsArea_{DWrist} +$
		$0.009 \cdot ACAbsArea_{DFoot} + 0.019 \cdot ACAbsArea_{NDFoot}$
		+0.88

Table 5-60: Regression equations and their coefficients learned from the energy expenditure data contained in the MIT energy expenditure dataset.

Equation	Correlation	All	Postures	Ambulation	Exercise	Resistance	Household
(a)	$0.68\pm0.06$	$1.75 \pm 0.38$	0.8±0.2	1.4±0.5	2.1±0.8	1.6±0.6	0.9±0.2
		$(1.33 \pm 0.27)$	(0.7±0.2)	(1.3±0.5)	(2.0±0.8)	(1.5±0.6)	(0.8±0.2)
(b)	$0.68\pm0.06$	$1.36\pm0.31$	0.8±0.2	1.0±0.4	1.5±0.7	1.3±0.6	0.7±0.2
		$(1.03 \pm 0.17)$	(0.7±0.2)	(0.9±0.4)	(1.4±0.7)	(1.3±0.6)	(0.6±0.2)
(c)	$0.68\pm0.06$	$1.36\pm0.30$	0.7±0.2	1.1±0.4	1.5±0.7	1.4±0.6	0.7±0.2
		$(1.02 \pm 0.17)$	(0.7±0.2)	(1.0±0.4)	(1.4±0.7)	(1.3±0.6)	(0.6±0.2)
(d)	$0.65\pm0.07$	$1.41 \pm 0.33$	0.9±0.2	1.1±0.4	1.5±0.7	1.3±0.6	0.7±0.2
		$(1.07 \pm 0.18)$	(0.9±0.2)	(1.0±0.4)	(1.4±0.7)	(1.2±0.6)	(0.7±0.2)
(e)	$0.67\pm0.05$	$1.37 \pm 0.30$	0.7±0.2	1.1±0.4	1.5±0.7	1.4±0.6	0.7±0.2
		$(1.03 \pm 0.17)$	$(0.7\pm0.2)$	$(1.0\pm0.4)$	$(1.4\pm0.7)$	$(1.3\pm0.6)$	$(0.7\pm0.2)$

Table 5-61: Correlation coefficient (r), root mean squared error, and mean absolute error (shown in parenthesis) obtained when estimating energy expenditure using the regression formulas shown in Table 5-58.



Figure 5-46: Root mean squared error (RMSE) when predicting energy expenditure using the equations in Table 5-59 shown as a grayscale image. The image is scaled to present the lowest RMSE of 0.36 in white and the highest of 3.7 in black. The X axis corresponds to the regression equation used to predict energy expenditure from Table 5-59.

The poor performance over these activities can also be seen in Figure 5-46. Figure 5-46 presents the RMSE as a grayscale image scaled to highlight differences in performance per activity. The image represents a RMSE of 0.36 MET in white and a RMSE of 3.7 MET in black. From this figure, it can also be seen that the RMSE of *running* at different speeds is high. This is because equation (a) lacks a constant or coefficient representing the DC offset of energy expenditure. When the performance per activity is inspected for the postures category, it is found that they all have RMSE errors close 1MET. This suggests that the energy predicted for postures by equation (a) is zero. This intuitively makes sense, since equation (a) does not have a coefficient representing the DC offset of the energy expenditure signal. As a result, since postures involve almost no motion, the *ACTotalAbsArea* feature is zero and the predicted energy is zero for these activities.

In response to the shortcomings of equation (a), a coefficient representing the DC offset of the energy expenditure signal was added resulting in equation (b). From Table 5-61 and Figure 5-46 it can be observed that the performance over postures, running at different speeds, bench weight lifting, cycling and rowing has been improved. Table 5-61 shows that the performance over the ambulation and household activities has also been improved. However, the table deceptively shows that the performance for postures has not changed. When the performance per activity is inspected (see Appendix B2 for detailed results per activity), it is found that the performance for standing, sitting, and kneeling has been improved, but the performance over sitting fidgeting hands arms and sitting fidgeting legs has increased. As a result, the overall RMSE over the postures category remains the same as for equation (a). The value of the DC coefficient (1.58 MET) for equation (b) shown in Table 5-60 implies that the energy expenditure associated with sedentary activities (with MET values close to 1) will be always overestimated. This is because if there is no motion, the minimum prediction would be a value of 1.58 MET. This might be because a single regression model is utilized to estimate energy expenditure and thus, it has to optimize the coefficients for all activities simultaneously. As a result, this model is forced to increase the DC coefficient to produce better estimates for physically demanding activities that produced high values of energy expenditure such as cycling, bicep curls, and bench weight lifting.

Equation (c) in Table 5-59 improves over equation (b) by estimating energy expenditure multiplying the motion (acceleration) experienced by each body segment (with an accelerometer attached to it) by a constant. This equation also includes a coefficient representing the DC offset of the energy expenditure signal so that energy expenditure for sedentary activities can be better predicted. These coefficients can be thought as weights that can be tuned by the regression model during learning to better represent the individual contribution of each body segment to energy expenditure. Inspection of the results per activity from Figure 5-46 and Appendix B2 indicates that the RMSE slightly redistributes over all activities. As a result, Table 5-61 indicates that the overall RMSE and RMSE per activity category remain practically unchanged. Thus there is no clear advantage of using equation (c) over equation (b). In fact, equation (b) should be preferred over equation (c) given its lower computational requirements when only overall body motion (e.g. *ACTotalAbsArea* or *ACAbsArea* feature) is utilized to estimate energy expenditure. This might be because the overall amount of motion is the same when it is summed over all body segments or analyzed per body segment.

Another simple technique that can be used to predict energy expenditure derived from the work energy theorem is to compute the force associated with each body segment and multiply it by a scaling constant to obtain energy expenditure. Segmental force can be estimated by multiplying the motion or acceleration experienced at each body segment (ACAbsArea) by the approximate mass of the body segment. Equation (d) performs exactly this operation by obtaining the body segment masses from the Depmster's body segment model [236]. Equation (d) also includes a constant representing the DC offset of the energy expenditure signal. When the performance of equation (d) is evaluated, it is found that the overall coefficient of correlation decreases by 0.03 and overall RMSE decreases 0.05 METs as shown in Table 5-58. At first glance, this result is counter intuitive since it makes perfect sense to think that the energy expenditure associated with manipulating a body segment is directly proportional to the mass of that body segment. When the results are inspected per activity, it is found that energy expenditure prediction worsens for sedentary activities such as *writing*, *typing*, *playing video games*, *sitting*, and *standing* and for some resistance activities such as *rowing* and *cycling*. Performance decreases for sedentary activities because the DC offset learned using equation (d) is 1.83 and will always over predict the energy expenditure associated with sedentary activities. The only activities for which performance improves between 0.1 and 0.57 METs by the use of equation (d) are activities involving motion at the upper limbs such as *bicep curls*, *sitting fidgeting hands arms, carrying groceries, scrubbing a surface, and wiping/dusting*. This might be a result of equation (d) better capturing energy expenditure at the upper body due to the weighing of the accelerometer signal by the mass of the arms.

In order to better understand why equation (d) does not improve performance over equation (c), the coefficients representing the mass associated with each body segment were learned from the MIT dataset instead of being specified a priori using the Dempter's body segment model. This corresponds to equation (e) in Table 5-59. Ideally, when the coefficients are learned from the data, they should reflect the mass distribution of the body segments. For example, the coefficient for the accelerometer at the hip should be larger than the coefficient for the accelerometers at the wrists (arms) and feet (legs). Similarly, the coefficient for the accelerometers at the feet (legs) should also be larger than the coefficient for the accelerometers at the wrists (arms). Table 5-60 presents the coefficients learned from all the activities contained in the MIT EE dataset for equation (e). It is clear that not all coefficient values reflect the mass distribution of the body segments. For example, the coefficient of the accelerometer at the hip has a negative value and the coefficients for the wrists are not equal. The coefficient at the dominant wrist is also larger than the coefficient for both feet. One explanation for the distribution of values obtained for the coefficients is that most activities in the MIT EE dataset include different levels of resistance or work load effort associated with upper body and lower body extremities. For example, the high values of energy expenditure obtained during bench weight lifting, which primarily involve upper body motion, could distort (increase) the coefficient values associated with the wrists. A similar situation would happen during cycling at different resistance levels, leading to distorted values for the coefficients of all body segments. If this explanation is correct, that would indicate that energy expenditure is strongly depend on the activity being performed and that energy expenditure predicted for any activity involving the sightless amount of effort or resistance work using equations similar to equation (d) would be off. This might be one reason why the Crouter et al. regression model has been shown to estimate energy expenditure well since it utilizes activity dependent regression models for sedentary, ambulation and lifestyle activities.

To test if activities involving resistance or work load effort are distorting the coefficient values learned for the body segments in equation (e), they were eliminated from the training data. Appendix B3 shows what activities were preserved in the training data under the column labeled as "Accelerometer recognizable". Table 5-60 presents the coefficients learned for equation (e) using the "Accelerometer recognizable" activities set. It can be seen that now the hip has a positive coefficient larger than the coefficient for the wrists (arms) and feet (legs). Moreover, the coefficient for the DWrist is close to the NDWrist and the DFoot coefficient is close to the NDFoot coefficient.

However, the coefficient for the wrists (arms) is larger than the coefficients for the legs. This is because some activities contained in the "Accelerometer recognizable" category either contain some degree of effort or high motion associated with upper body activity such as *doing dishes, gardening, weeding, wiping/dusting* and *ironing*. In a final attempt to improve the distribution of the coefficient values equation (e) was re-trained using only postures and ambulation activities. The activities included in this category are also shown in Appendix B3 under the column labeled as "Posture and Ambulation". The final row of Table 5-60 present the final coefficients learned from this activity category. It can be seen that the distribution of the weights indeed improved since the coefficient for the hip increased, and the coefficient at the NDFoot also increased now being larger than the coefficients at the wrists (arms). The value of the coefficient representing the DC offset of the energy expenditure signal also improved since now is closer to 1MET, the ideal value for sedentary activities.

In summary, energy expenditure can be estimated using simple multivariable linear regression algorithms such as equation (c) obtaining a final correlation coefficient of 0.68 and a root mean squared error of 1.36MET. The incorporation of a coefficient representing the DC offset of the energy expenditure signal improves the estimate for sedentary activities such as postures, and other activities such as cycling and running. Finally, energy expenditure cannot be predicted reliably utilizing equation (d) (based on the work energy theorem) because different activities include different levels of upper body and lower body effort. Consequently, the importance weight or coefficient for each limb is dependent on the activity being performed. For example, if *bench weight lifting* is being performed, the acceleration at the upper limbs needs to be weighted more to predict energy expenditure reliably. Similarly, during *cycling* at high resistances, the weight for the lower limbs needs to be increased according. Nonetheless, during activities such as walking and running, the weight for upper body and lower body limbs could potentially be very close, thus contributing equally to overall energy expenditure. The results obtained using equation (c) are better than the ones obtained using the Crouter et al. algorithm. Obviously, this is mostly due to the utilization of seven sensors that better capture overall, upper body and lower body motion. The upcoming sections will investigate how well can energy expenditure can be estimated by creating regression models that depend on the activity being performed. In this way, the regression coefficients for the models used can have optimal values depending on the activity being performed.

## 5.6.2.3 How Well Can Energy Expenditure be Estimated Using the Compendium of Physical Activities?

One of the standard methods that the medical community utilizes to estimate energy expenditure during free-living is to collect information about subjects' activities using direct observation or self-report and later rate their energy expenditure according to the Compendium of Physical Activities [122]. The Compendium of Physical Activities consists of a list of the most common everyday physical activities and their associated average energy expenditure in metabolic equivalents (METs). Given the popularity of this method, this section explores how well it can estimate energy expenditure over the

Error Measures	<b>Comparable Activities</b>	Closest Activities
Total Correlation Coefficient	$0.86 \pm 0.10$	$0.80\pm0.09$
Total Root Mean Square Error	$1.27 \pm 0.33$	$1.61\pm0.41$
Total Mean Absolute Error	$0.85 \pm 0.21$	$1.12\pm0.28$
Maximum Absolute Deviation	$4.17\pm0.86$	$5.57 \pm 1.33$

Table 5-62: Performance of estimating energy expenditure using the Compendium of Physical Activities over the *comparable* and *closest* activity sets. Energy expenditure was predicted over windows of one minute in length.

Activity Set	Total Number of Activities	Postures	Ambulation	Exercise	Resistance	Household
Comparable	29	$0.2\pm0.1$	1.4±0.5	2.1±0.7	2.1±0.4	0.9±0.3
		$(0.2\pm0.1)$	(1.3±0.5)	(2.0±0.8)	(2.0±0.4)	(0.9±0.3)
Closest	52	0.3±0.2	1.5±0.4	2.3±0.7	2.0±0.5	0.9±0.3
		(0.3±0.2)	(1.4±0.4)	(2.2±0.7)	(2.0±0.6)	(0.9±0.3)

Table 5-63: Root mean squared error and mean absolute error (shown in parenthesis) per activity category when estimating energy expenditure using the Compendium of Physical Activities over the *comparable* and *closest* activity sets.

MIT dataset assuming that the activities performed are known from the activity labels collected during the data collections.

This experiment is performed over two sets of activities: (1) the *comparable* set of activities and (2) the *closest* set of activities. The *comparable* set of activities includes only the activities in the MIT dataset that are directly comparable with those found in the Compendium of Physical Activities. In other words, this set of activities completely matches the physical activity description, speed of execution, and/or resistance level found in the Compendium of Physical Activities. The *closest* set of activities consists of all the activities contained in the MIT dataset and their corresponding closest match found in the compendium. For example, *cycling at 80rmp* at a moderate intensity level (resistance level of 7 in a Precor C846 recumbent stationary bicycle) was not found in the Compendium. However, the closest activities contained in the MIT dataset, their corresponding closest activities from the Compendium, and whether or not they are included in the *comparable* activity set.

Table 5-62 presents the performance of estimating energy expenditure using the Compendium of Physical Activities over the *comparable* and *closest* activities sets assuming the activity being performed is known. In order to be able to compare the estimation performance with the results obtained using the Crouter et al. Actigraph-based algorithm, energy expenditure is predicted over windows of one minute in length. Table 5-62 shows that the correlation coefficient and RMSE error obtained over both activity sets are higher than the results obtained using the Crouter et al. Actigraphs-based algorithm (r=0.4, RMSE=3.23). As expected, performance is slightly better over the *comparable* activities set because it contains 23 fewer activities than the *closest* activity set. The maximum absolute error deviation of 4.2 METs and 5.6 METs for the *comparable* and *closet* activity sets, respectively, is also lower than the one obtained using the Crouter et al. method (7.0 METs). The largest decrease in RMSE error with respect to the Crouter et al. method was observed for the following activity categories: postures (-0.4 METs), ambulation (-1 MET), exercise (-0.7 METs), and resistance exercise (-1.3 MET).

When the RMSE per activity is analyzed from Appendix B5, it is found that the Compendium tends to overestimate the energy expenditure associated with some household activities such as gardening, making the bed, scrubbing a surface, weeding and vacuuming. These activities are highly variable and unconstrained since they can be performed with differently styles, intensities and objects (e.g. industrial vacuum cleaner vs. portable vacuum cleaner). Therefore, their energy expenditure value is difficult to predict from just the average value listed in the Compendium of Physical Activities. The Compendium of Physical Activities also overestimates energy expenditure for some physically demanding activities such as ascending stairs, sit-ups, and running at 6mph. This is because the activity examples collected for these activities have individual durations of less than a minute, thus preventing energy expenditure from reaching steady state (which is the average value listed in the Compendium of Physical Activities). This effect can also be seen in Appendix B5, which presents plots of the energy expenditure estimated using the Compendium of Physical Activities vs. the measurements using the Cosmed K4b2 indirect calorimeter. Finally, energy expenditure is better predicted for activities that reached steady state during the data collection such as postures and ambulation (walking and running).

In summary, if the activity performed is known, energy expenditure can be estimated well by simply predicting the average METs value associated with an activity. Obviously, the quality of the estimate depends on how close the average METs value utilized on the predictions and the truth energy expenditure of the activity of interest are. The main disadvantages of applying this method to estimate energy expenditure are: (1) the type and duration of the physical activities performed needs to be known in advance and (2) the physical activities performed need to be listed in the Compendium of Physical Activities.

## 5.6.2.4 Estimating Energy Expenditure Using One Linear Regression Model per Activity.

The results presented in previous sections and some recent work [34, 96] suggest that energy expenditure can be better estimated if the activities being performed are known. For example, Section 5.6.2.2 indicated that the optimal value for the coefficients of a regression equation used to estimate energy expenditure depend on the activity being performed. For example, during the execution of *bicep curls*, it might be necessary to assign a greater weight to the regression coefficients representing the acceleration at upper extremities. Conversely, during *cycling*, it might be necessary to assign a greater weight to the coefficients representing the acceleration at the lower limbs. One way to achieve this it to model energy expenditure utilizing independent regression models whose coefficients can be tuned depending on the activities being performed. In fact, Section 5.6.2.3 showed that a good performance in energy expenditure estimation with respect to state-of-the-art Actigraph algorithms can be achieved by simply predicting the average energy expenditure value for each activity. As a result, this section explores how well energy expenditure can be estimated if it is modeled using one independent multivariable linear regression model per activity.

Error Measures	Results
Total Correlation Coefficient	$0.87\pm0.10$
Total Root Mean Square Error	$1.02\pm0.45$
Total Mean Absolute Error	$0.63 \pm 0.19$
Maximum Absolute Error Deviation	$4.21 \pm 2.89$

Table 5-64: Performance obtained over the MIT energy expenditure dataset while estimating energy expenditure using one multivariable linear regression model per activity. The regression models per activity were trained using the *ACAbsArea* feature computed per sensor over windows of one minute in length. Performance is measured with respect to energy expenditure data collected from the Cosmed K4b2 indirect calorimeter

Activity Category	RMSE (MAE)
Postures	0.3±0.2 (0.3±0.2)
Ambulation	0.9±0.7 (0.8±0.7)
Exercise	1.1±0.9 (1.1±0.8)
Resistance	0.8±0.5 (0.8±0.5)
Household	0.5±0.3 (0.4±0.3)
Upper Body	0.4±0.3 (0.4±0.2)
Lower Body	0.9±0.5 (0.8±0.5)

Table 5-65: Root mean squared error and mean absolute error (shown in parenthesis) obtained over the MIT energy expenditure dataset when estimating energy expenditure using one multivariable linear regression model per activity. The regression models per activity were trained using the *ACAbsArea* feature computed per sensor over windows of one minute in length. Performance is measured with respect to energy expenditure data collected from the Cosmed K4b2 indirect calorimeter

Table 5-64 presents the performance of predicting energy expenditure over the MIT dataset using one multivariable linear regression model per activity. For simplicity, the activities performed are assumed to be known from the labels and timestamps recorded during the data collections. Performance is measured using the ACAbsArea feature, one of the most commonly employed features when estimating energy expenditure in prior work. Appendix A3 explains how the ACAbsArea feature is computed. Energy is estimated over windows of one minute in length to facilitate comparison with prior work. Table 5-64 shows that the coefficient of correlation (0.87) and RMSE error (1.02)obtained are the lowest achieved so far. This indicates that performance is clearly improved when activity dependent regression models are used. Appendix B7 presents plots of energy expenditure estimated versus measured using the Cosmed K4b2 indirect calorimeter for two subjects. The plots show that energy expenditure estimated closely follows energy expenditure measured except for activities involving different levels of resistance or work load. This is because the effort associated with these activities is not captured well from accelerometers since the motion signature of activities (e.g. speed of execution or motion patterns involved) does not change or is not detectable in a subject independent manner.

Table 5-65 shows the performance per activity category obtained using one linear regression model per activity. The best performance is obtained over postures and household activities and the worse over exercise, resistance exercise, and lower body activities, as found in previous sections. Lower body activities present a relatively high RMSE because this category consists mainly of the *cycling* activities that contain different levels of resistance work not detectable from accelerometers. Overall, the performance per activity category is also the highest found so far.

When the performance per activity is inspected from Appendix B7, it is found that the activities with lowest RMSE are *bench weight lifting* (0.16 MET), *doing dishes* (0.16 MET), and *watching TV* (0.18 MET). The activities with higher RMSE are calisthenics *crunches* and *sit-ups, descending stairs*, and *running* on a treadmill at different speeds. The maximum absolute error deviation is also the highest for these activities. As explained before, error is high for *crunches* and *sit-ups* because they were executed for short periods of time (<1min) and steady-state energy expenditure was never reached. The high error in *descending stairs* can be explained also by its short duration (<1min) and by the fact that it was executed almost immediately after *ascending stairs*. As a result, the indirect calorimeter readings were still high (from ascending stairs) when this activity did not justify the high energy expenditure values observed. Finally, the poor performance over the *running* activities can be explained by the different physical fitness level of individuals. For example, energy expenditure observed during running was higher for obese participants than for lean and physically fit participants.

In conclusion, the results presented in this section indicate that the use of activity dependent regression models improve energy expenditure estimation considerably. One obvious concern with the applicability of method in practice is that activities have to be first recognized in order to apply the corresponding regression models. Section 5.6.11.2 will later explore how well energy expenditure can be estimated in practice using this method by recognizing activities using the algorithm implemented in Section 5.4.9.

### 5.6.2.5 How Well Can Energy Expenditure be Estimated Using One Non-Linear Regression Model per Activity?

Following up on the preceding section, this section explores the performance of estimating energy expenditure using one non-linear regression model per activity when the activities performed are assumed to be known. The non-linear regression model used is a M5' model tree trained using the *ACAbsArea* feature computed per sensor over windows of one minute in length. Appendix A3 explains how the *ACAbsArea* feature is computed. The parameters of the M5' model tree used are shown in Table 5-70. The main objective of this section is to measure the improvement in performance with respect to one linear regression model per activity.

Table 5-67 presents the error measures obtained when estimating energy expenditure using one non-linear regression model per activity over the MIT energy expenditure dataset. The table shows that performance is improved for all error measures. For example, the correlation coefficient improves +0.04 units and RMSE improves 0.14 MET with respect to one linear regression model per activity. The largest improvement in performance of 0.85 MET is observed for the maximum absolute error deviation.

Table 5-67 illustrates the root mean squared error and mean absolute error per activity category. It can be seen that RMSE error is uniformly distributed for the household and upper body activity categories (0.4 MET) and for the ambulation, resistance exercise and lower body activity categories (0.8 MET). RMSE error is higher (1 MET) for exercise activities because this category contains more activities than the other categories (n=19). The only RMSE error that does not improve with respect to Table 5-65 and practically

Error Measures	Results
Total Correlation Coefficient	$0.91 \pm 0.04$
Total Root Mean Square Error	$0.88 \pm 0.25$
Total Mean Absolute Error	$0.59 \pm 0.15$
Maximum Absolute Deviation	$3.36 \pm 1.18$

Table 5-66: Performance obtained over the MIT energy expenditure dataset when estimating energy expenditure using one non-linear regression model (M5' model tree) per activity. The regression models per activity were trained using the *ACAbsArea* feature computed per sensor over windows of one minute in length. Performance is measured with respect to energy expenditure data measured using the Cosmed K4b2 indirect calorimeter

Activity Category	RMSE (MAE)
Postures	0.2±0.2 (0.2±0.2)
Ambulation	0.8±0.4 (0.7±0.4)
Exercise	1.0±0.6 (1.0±0.6)
Resistance	0.8±0.5 (0.7±0.5)
Household	0.4±0.3 (0.4±0.2)
Upper Body	0.4±0.3 (0.4±0.2)
Lower Body	$0.8\pm0.4$ (0.8±0.4)

Table 5-67: Root mean squared error and mean absolute error (shown in parenthesis) obtained over the MIT energy expenditure dataset when estimating energy expenditure using one non-linear regression model (M5' model tree) per activity. The regression models per activity were trained using the *ACAbsArea* feature computed per sensor over windows of one minute in length. Performance is measured with respect to energy expenditure data measured using the Cosmed K4b2 indirect calorimeter

remains unchanged is the error for the resistance exercise category. Apparently, this error is already as low as it can be given that energy expenditure associated with different resistance level and work load effort is poorly detected using accelerometers.

When the RMSE is inspected per activity and compared to the one obtained using one linear regression model per activity, it is found that it slightly improves for most activities and that improvement is higher for *crunches* (-1.6 MET), *ascending stairs* (-0.9 MET), and *bicep curls moderate* (-0.44 MET). The maximum absolute error deviation for these activities also improves substantially (0.9-1.7 MET). Intuitively this improvement makes sense, since the energy expenditure associated with these activities increases

exponentially over time, and is better captured using non-linear regression models. The same improvements in RMSE are observed when visualizing the energy expenditure plots presented in Appendix B8.

In summary, utilizing one non-linear regression model per activity improves performance considerably over one linear regression model per activity for some activities. Knowledge of which activities are improved the most might allow the application of non-linear regression models only for those activities that benefit from it, thus, successfully reducing computational complexity since linear models would still be applied for most activities.

#### 5.6.2.6 Summary of Baseline Results

In summary, the results presented in previous sections indicate that overall performance and performance per activity are improved the most when the activities performed are known. Section 5.6.2.3 illustrated that excellent results with respect to state-of-the-art

Error Measures	Crouter et al. Actigraph	Compendium Comparable Activities	Compendium Closest Activities	Linear Regression	One Linear Regression model per activity	One non-linear regression model per activity
Total Correlation Coefficient	$0.36\pm0.09$	$0.86\pm0.10$	$0.80\pm0.09$	$0.73\pm0.06$	$0.87\pm0.10$	$0.91\pm0.04$
Total Root Mean Square Error	$2.66 \pm 0.62$	$1.27\pm0.33$	$1.61\pm0.41$	1.28 ± 0. 29	$1.02\pm0.45$	$0.88\pm0.25$
Total Mean Absolute Error	$1.94 \pm 0.45$	$0.85\pm0.21$	$1.12\pm0.28$	$0.95\pm0.16$	$0.63\pm0.19$	$0.59\pm0.15$
Maximum Absolute Deviation	6.94 ± 1.77	4.17 ± 0.86	5.57 ± 1.33	4.12 ± 1.21	4.21 ± 2.89	3.36 ± 1.18

Table 5-68: Comparison of the performance obtained using the energy expenditure estimation methods explored in this section. All methods estimate energy expenditure over sliding windows of one minute in length. The methods that utilize linear or non-linear regression are trained using the *ACAbsArea* feature computed per sensor.

Activity	Crouter	Compendium	Compendium	Linear	One Linear	One Non-linear
Category	et al.	Comparable	Closest	Regression	Regression	regression model
	Actigraph	Activities	Activities		model per	per activity
					activity	
Postures	0.6±0.2	0.2±0.1	0.3±0.2	0.6±0.2	0.3±0.2	0.2±0.2
	(0.5±0.2)	$(0.2\pm0.1)$	(0.3±0.2)	(0.6±0.2)	(0.3±0.2)	(0.2±0.2)
Ambulation	2.4±0.5	1.4±0.5	1.5±0.4	1.0±0.5	0.9±0.7	0.8±0.4
	(2.4±0.5)	(1.3±0.5)	(1.4±0.4)	(1.0±0.5)	(0.8±0.7)	(0.7±0.4)
Exercise	2.8±0.8	2.1±0.7	2.3±0.7	1.5±0.8	1.1±0.9	1.0±0.6
	(2.8±0.8)	(2.0±0.8)	(2.2±0.7)	(1.4±0.7)	(1.1±0.8)	(1.0±0.6)
Resistance	3.4±0.7	2.1±0.4	2.0±0.5	1.3±0.6	0.8±0.5	0.8±0.5
	(3.4±0.7)	(2.0±0.4)	(2.0±0.6)	(1.3±0.6)	(0.8±0.5)	(0.7±0.5)
Household	1.0±0.3	0.9±0.3	0.9±0.3	0.7±0.3	0.5±0.3	0.4±0.3
	(1.0±0.3)	(0.9±0.3)	(0.9±0.3)	(0.6±0.3)	(0.4±0.3)	(0.4±0.2)
Upper Body	1.2±0.3	0.8±0.3	1.3±0.3	0.7±0.3	0.4±0.3	0.4±0.3
	(1.2±0.3)	(0.8±0.3)	(1.3±0.3)	(0.7±0.3)	(0.4±0.2)	(0.4±0.2)
Lower Body	2.7±0.5	-	1.3±0.5	$1.6\pm0.8$	0.9±0.5	$0.8\pm0.4$
	(2.7±0.5)		(1.3±0.5)	(1.6±0.8)	(0.8±0.5)	$(0.8\pm0.4)$

Table 5-69: Root mean squared error and mean absolute error (shown in parenthesis) obtained using the different energy expenditure estimation methods explored in this section. All methods estimate energy expenditure over sliding windows of one minute in length. The methods that utilize linear or non-linear regression are trained using the *ACAbsArea* feature computed per sensor.

Actigraph-based algorithms can be obtained by just predicting the average energy expenditure associated with each activity from the Compendium of Physical Activities. The results obtained with this method are also competitive with the performance obtained using multivariable linear regression and the *ACAbsArea* feature computed over all the seven accelerometer sensors. Section 5.6.8 will later explore the impact of reducing the number of sensor while estimating energy expenditure using multivariable linear regression. Clearly, another advantage of estimating energy expenditure by recognizing activities is that the knowledge of the activities performed is important information that can be used for medical purposes or in just-in-time interventions to motivate increases in physical activity levels.

Energy expenditure estimation can be further improved by applying linear and nonlinear activity dependent models. The performance of both methods improves over a single multivariable linear regression model applied to all activities at once. In particular, one non-linear regression model per activity improves performance for activities whose energy expenditure patterns are non-linear such as *ascending stairs* and *crunches*. These results hold when there is enough data to train the linear or non-linear regression models per activity as in the analysis presented previously, since only one feature *ACAbsArea* is utilized. Results might be different; for example, if more features are used to estimate energy expenditure and the amounts of data available are insufficient to train the large number of linear and/or non-linear regression models (as later found in Section 5.6.3).

Table 5-68 and Table 5-69 present a comparison of the performance obtained using the different methods explored in this section to estimate energy expenditure. Table 5-68 shows that the Crouter et al. Actigraph-based energy expenditure estimation method presents the worse performance. The most likely reason for this poor performance is that this method predicts energy expenditure from just one accelerometer located at the hip (Actigraph) while the methods that use linear or non-linear regression utilize the data from the best case scenario of seven accelerometers located at different body segments. Using one single accelerometer at the hip to estimate energy expenditure (as most medical research studies do) has the disadvantage of poorly detecting energy expenditure associated with upper body and lower body activity. In fact, most of the energy expenditure predicted from an accelerometer at the hip comes from ambulatory activities. Table 5-69 clearly shows that the Crouter et al. method has the worst performance in estimating energy expenditure for upper body and lower body activities. For example, the Crouter et al. method presents a RMSE of 6.4 MET for cycling hard at 80rmp while multivariable linear regression using all sensors presents an error of just 2.9 MET, mostly due to the inability of the accelerometers to capture information that can differentiate cycling at different resistance levels. Similarly, the Crouter et al. method presents a RMSE error of ~2.0 MET for bicep curls moderate and hard while linear regression presents an error between 0.2-0.5 MET for these activities. Energy expenditure algorithms capable of detecting upper body and lower body activity are likely to be superior in performance to algorithms based on single accelerometers located at the hip.

The following sections will perform systematic experiments in increasing order of complexity to determine the most suitable parameters for the energy expenditure estimation algorithm developed in this work with the ultimate goal of achieving real-time performance.

#### 5.6.3 Can Fast Run-Time Regression Algorithms Produce Acceptable Performance?

This section compares the performance of regression algorithms with fast prediction times that are amenable for real-time performance with other regression algorithms with longer training and prediction times. The goal of the section is to identify the regression algorithm that provides the best trade-off between computational complexity and performance.

The algorithms compared in this section are multivariable linear regression, M5' regression and model trees [237, 238], and epsilon support vector regression ( $\epsilon$ -SVR) [239, 240]. The performance of multivariable linear regression, M5' regression trees, and M5' model trees was evaluated using the Weka toolkit [226], and the performance of epsilon support vector regression was evaluated using the LibSVM library for support vector machines [241]. Table 5-70 presents a brief description of each regression algorithm as well as the parameters used for each of them during the experiments.

Regression	Description	Parameters	Ref
Algorithm Linear Regression (LR)	A regression algorithm that models the statistical relationship between predictor ( <i>x</i> ) and predicted ( <i>y</i> ) variables employing the following linear equation: $y = \sum_{i=1}^{n} w_i \cdot x_i + w_0$ The coefficients <i>w</i> are found by minimizing the sum of the square of the residuals ( <i>y<sub>i</sub></i> - <i>x<sub>i</sub></i> ) or least square error with respect to a set of training data points.	Multivariable linear regression using the M5 attribute selection method (based on the Akaike information criterion), elimination of collinear attributes, and a Ridge parameter value of 1.0E-8.	
M5' Regression Trees (RT)	A non-linear regression algorithm that builds tree-based models very similar to decision trees but that have numerical values representing the predicted variable at their leaves instead of predicted class values.	Regression tree built with the M5' algorithm using pruning, a minimum of four instances per leaf, and smoothed predictions.	[237] [238]
M5' Model Trees (MT)	A non-linear regression algorithm that builds tree-based models very similar to regression trees but that have multivariate linear models at their leaves instead of class values. As a result, predicted variables are approximated by piecewise linear functions. This algorithm can handle problems with higher dimensionality than MARS [242] and generate smaller and more accurate trees than CART [243].	Model tree built with the M5' algorithm using pruning, a minimum of four instances per leaf, and smoothed predictions.	[237] [238]
Epsilon Support Vector Regression (&SVR)	A regression algorithm based on support vector machines [216] that, when used in combination with a radial basis function kernel, is able to learn complex non-linear relationships between predictor and predicted variables.	Epsilon support vector regression using a radial basis function kernel, the shrinking heuristics, and parameters cost ( <i>C</i> ), gamma ( $\gamma$ ), and epsilon ( $\varepsilon$ ) found by performing a grid search over the parameter space. The optimal values found for these parameters were: <i>ACAbsArea</i> feature <i>C</i> =64, $\gamma$ =1.0, and $\varepsilon$ =0.5 <i>MaxAcceleration</i> feature set <i>C</i> =8.0, $\gamma$ =0.0625, $\varepsilon$ =0.0625	[239] [240]

Table 5-70: Brief description of the regression algorithms explored in this section and their parameters. Ref stands for reference and points to literature describing the algorithms in full detail.

	Feature Set	LR	RT	MT	ε-SVR
Total training time	ACAbsArea	0.71s	211.5s	285s	176s
(Average time per instance)		(0.03ms)	(8.1ms)	(11ms)	(7.4ms)
Total prediction time	ACAbsArea	1.0s	0.9s	1.0s	8.5s
(Average time per instance)		(0.5ms)	(0.5ms)	(0.5ms)	(4.0ms)
Total training time	MaxAcceleration	285.8s	528s	1387s	4227s
(Average time per instance)		(12.064ms)	(22.32ms)	(58.5ms)	(178.4ms)
Total prediction time	MaxAcceleration	1.5s	1.4s	4.4s	82.8s
(Average time per instance)		(0.7ms)	(0.7ms)	(2.1ms)	(38.7ms)

Table 5-71: Total training and prediction times in seconds obtained during the first round of subject independent training for the linear and non-linear regression algorithms explored in this work. The total number of instances learned is 25, 832.

In this work, support vector regression (SVR) was chosen over multilayer neural networks regression for four reasons. (1) Unlike neural networks, SVR does not suffer from local minima problems since training involves solving a linearly constrained quadratic programming problem that always has a unique and globally optimal solution [239]. (2) SVR has fewer parameters to optimize (e.g. C,  $\gamma$ , and  $\varepsilon$  when the radial basis function kernel is used) than neural networks (e.g. number of network layers, number of nodes per layer, learning rate, momentum, form of transfer function, number of epoch,

etc). Furthermore, there exist well-tested procedures to systematically find the optimal values for SVR parameters when the radial basis function kernel is used [244]. (3) SVR has capacity control of the regression function learned, which means that the complexity of the solution found is the minimal required to solve the problem at hand. (4) Finally, SVR has been found to be resistant to overfitting and to the curse of dimensionality [239].

The performance of each algorithm is analyzed in two extreme computational complexity conditions: (1) a best case scenario where energy expenditure is predicted from just one feature per sensor (*ACAbsArea*), and (2) a worse-case scenario where energy expenditure is predicted using all the accelerometer-based features computed per sensor (*MaxAcceleration* feature set). Features are computed over sliding windows of 5.6s in length.

Table 5-71 presents the total training and prediction times for each of the algorithms explored in seconds as well as the average training and prediction times per example (instance) in milliseconds. First, it can be observed that the training and prediction times are lower for the ACAbsArea feature set than for the MaxAcceleration feature set. This is because the ACAbsArea feature set has a vector size of only 7 while the MaxAcceleration feature set has a vector size of 247. The table shows that the algorithms with longer training times are model trees and  $\varepsilon$ -SVR. This is because model trees have to learn the tree structure from the data using the M5' algorithm and also one multivariable linear regression model (over subsets of the features) per leaf node. For E-SVR, training involves the solution of a quadratic optimization problem that is computationally expensive ( O(nNSV + NSV) were *n* is number of training examples and NSV number of support vectors [245]) and that is why it presents the long training times observed in Table 5-71. From the table, it can also be observed that the prediction times for linear regression, regression trees, and model trees are considerably lower than prediction times for  $\varepsilon$ -SVR. This is because prediction in linear regression only involves summations and multiplications over the features. For regression trees, it mainly involves the evaluation of fast if-then clauses over several feature values that depend on the structure of the tree learned. Once a leaf node is reached, the prediction made is a simple constant representing the average value of the predicted variable for the training examples that fall under that branch of the tree. In model trees, prediction also involves the evaluation of several if-then clauses over several feature values but it also requires the evaluation of the multivariable linear regression model at the leaf node. On the contrary, prediction in  $\varepsilon$ -SVR involves dot products over the number of support vectors found during training (subset of training data points that represent the final solution). As a result, prediction time depends on the number of support vectors found for a given problem. For the ACAbsArea feature, the number of support vectors found consists of 38% of the total number of training data points (25,832). For the MaxAcceleration set, the number of support vectors found is 30% of the total training examples. This large number of support vectors lead to the long prediction times observed in Table 5-71. The large number of support vectors found also suggests that energy expenditure prediction over the MIT EE dataset is difficult. In summary, from Table 5-71, we can conclude that the regression algorithms with best prediction times in increasing order are regression trees, linear regression, model trees, and support vector regression. The ordering of the algorithms with respect to training times in increasing order is: linear regression, regression trees,

Algorithm	Correlation	All	Postures	Ambulation	Exercise	Resistance	Household
	Coefficient						
LR	$0.68\pm0.06$	$1.36\pm0.30$	0.7±0.2	1.1±0.4	1.5±0.7	1.4±0.6	0.7±0.2
		$(1.02 \pm 0.17)$	(0.7±0.2)	(1.0±0.4)	(1.4±0.7)	(1.3±0.6)	(0.6±0.2)
RT	$0.73\pm0.07$	$1.25\pm0.29$	0.6±0.3	1.1±0.5	1.3±0.7	1.1±0.6	0.6±0.3
		$(0.88 \pm 0.18)$	(0.5±0.2)	(1.0±0.5)	(1.1±0.7)	(1.0±0.5)	(0.5±0.2)
MT	$0.72 \pm 0.08$	$1.24\pm0.29$	0.6±0.3	1.1±0.4	1.3±0.7	1.1±0.6	0.6±0.3
		$(0.86 \pm 0.19)$	(0.5±0.2)	(0.9±0.4)	(1.2±0.7)	(1.0±0.6)	(0.5±0.2)
ε-SVR	$0.74 \pm 0.06$	$1.28\pm0.29$	0.5±0.2	1.1±0.5	1.3±0.7	1.1±0.6	0.6±0.3
		$(0.89 \pm 0.21)$	(0.4±0.2)	(1.0±0.5)	(1.2±0.7)	(1.0±0.6)	(0.5±0.3)

Table 5-72: Root mean squared error and mean absolute error (shown in parenthesis) per activity category when estimating energy expenditure using different regression algorithms and the *ACAbsArea* feature computed per sensor over sliding windows of 5.6s in length.

Algorithm	Correlation Coefficient	All	Postures	Ambulation	Exercise	Resistance	Household
	Coefficient	1.24 0.20	0.5.0.0	1205	1205	11.0.6	0.5.0.2
LR	$0.74 \pm 0.10$	$1.24 \pm 0.30$	$0.7\pm0.3$	$1.2\pm0.5$	$1.2\pm0.7$	$1.1\pm0.6$	$0.7\pm0.3$
		$(0.91 \pm 0.20)$	(0.6±0.3)	(1.0±0.5)	(1.1±0.7)	(1.0±0.6)	(0.6±0.3)
RT	$0.75\pm0.06$	$1.21 \pm 0.31$	0.6±0.3	1.1±0.4	1.2±0.7	1.1±0.6	0.6±0.2
		$(0.86 \pm 0.20)$	(0.5±0.3)	(0.9±0.4)	(1.1±0.7)	(1.0±0.5)	(0.5±0.2)
MT	$0.64 \pm 0.17$	$1.56\pm0.67$	1.0±1.5	1.2±0.5	1.3±0.7	1.1±0.6	0.8±0.5
		$(0.94 \pm 0.24)$	(0.7±0.7)	(1.0±0.5)	(1.2±0.7)	(1.0±0.5)	(0.6±0.4)
ε-SVR	$0.80\pm0.06$	$1.10\pm0.27$	0.5±0.2	0.9±0.4	1.1±0.6	0.9±0.5	0.6±0.2
		$(0.78 \pm 0.17)$	(0.4+0.2)	(0.8+0.4)	(1.0+0.6)	(0.8+0.5)	(0.5+0.2)

Table 5-73: Root mean squared error and mean absolute error (shown in parenthesis) per activity category when estimating energy expenditure using different regression algorithms and all the accelerometer-based features (*MaxAcceleration*) computed per sensor over sliding windows of 5.6s in length.

model trees, and support vector regression. Given that training time is not too important in energy expenditure prediction (since subjects cannot perform subject dependent training due to the lack of equipment), the best algorithms to use in increasing order of desirability based on runtime are: linear regression, regression trees, and model trees.

Table 5-72 presents the root mean squared error and mean absolute error per activity category found using the regression algorithms explored when the ACAbsArea feature is computed per sensor over windows of 5.6s in length. Surprisingly, the table shows that the performance of linear regression is very close to the performance of all the other more complex non-linear regression algorithms. One possible explanation for this is that the complexity of energy expenditure prediction in the MIT energy expenditure dataset is so high that linear regression performs as good as the other algorithms due to the unavailability of enough amounts of training data. In other words, due to the relatively small amount of training data available (6 hours of data for 16 subjects), regression trees, model trees and  $\varepsilon$ -SVR may be unable to learn models that outperform linear regression considerably. Nevertheless, Table 5-72 still shows that the performance of regression trees, model trees, and  $\varepsilon$ -SVR is higher than the performance of multivariable linear regression. For example, the improvement achieved by  $\varepsilon$ -SVR over LR is +0.4 units for the coefficient of correlation, and +0.08 MET for the root mean squared error. Obviously, the extra computation incurred by  $\varepsilon$ -SVR does not justify the epsilon improvement in performance obtained over this dataset. As observed in previous sections, Table 5-72 also shows that the activity categories whose energy expenditure is easier to estimate are postures and household activities. This is because they consist primarily of activities with sedentary and light levels of energy expenditure whose motion is easy to detect from



Figure 5-47: Root mean squared error per activity as a gray scale image obtained when estimating energy expenditure with different linear and non-linear regression algorithms using all the accelerometer-based features computed per sensor over sliding windows of 5.6s in length. The image has been scaled to show the lowest RMSE of 0.2MET in white and the largest of 2.3MET in black.

accelerometers. Some household activities such as *scrubbing a surface*, *carrying groceries*, and *washing windows* do involve some level of effort not detectable from accelerometers. The level of effort, however, is substantially lower than that found during gymnasium activities. The table also shows that the activity categories that present the higher RMSE are exercise and resistance exercise. This is because they involve high levels of energy expenditure in most cases due to different levels of resistance level or work load that is not detectable from accelerometers. Some of these activities include *walking* at different incline grades, *rowing* and *cycling* at different resistance levels, *bench weight lifting*, and *bicep curls*.

Table 5-73 presents the same results as Table 5-72 but when the *MaxAcceleration* feature set is used. In general, the table presents the same tendencies found in Table 5-72.

RMSE is lower for the postures and household activity categories and higher for exercise and resistance exercise activities. Nonetheless, the table shows that the performance of model trees suffers with the increase in the number of features (7 to 247). The performance of model trees is worse than the performance for linear regression because there is not enough training data available at the leaf nodes of the tree to train the associated multivariable linear regression models. Regression trees on the contrary does not suffer from this phenomenon because they predict energy expenditure by estimating average EE values over the set of examples falling under

Each leaf node of the tree and these values (averages) can be learned well even from small number of examples.

The performance of  $\varepsilon$ -SVR on the other hand, is the highest in the table reflecting well the fact that  $\varepsilon$ -SVR is usually little affected by the curse of dimensionality (decrease in performance due to the increase of predictor variables) [239]. Performance per activity for the different regression algorithms can be found in Appendix B9. Figure 5-47 presents the root mean square error per activity for all the regression algorithms as a gray scale image normalized to show the lowest RMSE of 0.2 MET in white and the largest of 2.3 MET in black. Overall, the figure shows that all algorithms have difficulties predicting energy expenditure for the same activities. These activities are activities involving different intensity levels due to changes in speed, resistance level, or work load, activities involving high levels of energy expenditure, and activities of short duration. Figure 5-47 also illustrates the best overall performance of  $\varepsilon$ -SVR and the poor performance of the model tree with respect to the performance of linear regression.

In conclusion, the results presented in this section indicate that two good choices of algorithms to use when predicting energy expenditure are multivariable linear regression and regression trees. This is because these algorithms have fast prediction times and their performance is close to the best performance obtained using epsilon support vector regression. The advantage of regression trees over multivariable linear regression is that regression trees are able to learn non-linear relationships in the data. The disadvantage is that its training time is considerably longer than the training time for multivariable linear regression, particularly when a small number of features are used. The small differences found in this work between linear and non-linear regression algorithms do not improve performance substantially over linear regression techniques [246]. However, the main reason suspected for this little difference is unavailability of enough training data.

#### **5.6.4 Does Band-Pass Filtering of the Acceleration Signals Improve Performance?**

One procedure commonly applied in prior work during the estimation of energy expenditure from accelerometer data is to band-pass filter the raw acceleration signals between 0.1 and 20Hz (e.g. [32, 47, 153, 200]). The rationale behind this is that low pass filtering the signal at 20Hz eliminates signal components generated by non-human motion and high frequency noise and that high pass filtering at 0.1Hz eliminates posture information in the form of static acceleration (DC signal component) that does not contribute to energy expenditure considerably. However, to the best of the author's knowledge, no work has evaluated the impact in performance of band-pass filtering the accelerometer signals. As this section investigates the difference in performance obtained

	ACAbsAre	ea Feature	ACFFTPeaks Feature		
Error Measures	Without Band-pass Filtering	With Band-pass Filtering	Without Band-pass Filtering	With Band-pass Filtering	
Correlation Coefficient	$0.56\pm0.11$	$0.68\pm0.06$	$0.68\pm0.16$	$0.72\pm0.07$	
Root Mean Square Error	$1.66\pm0.50$	$1.36\pm0.30$	$1.34\pm0.34$	$1.28\pm0.30$	
Mean Absolute Error	$1.32\pm0.40$	$1.02\pm0.17$	$1.01\pm0.24$	$0.93\pm0.17$	
Maximum Absolute Error Deviation	$5.55 \pm 1.16$	$5.02 \pm 1.08$	$5.48 \pm 1.32$	$5.48 \pm 1.10$	

Table 5-74	: Perfo	ormance ob	tained whil	le estima	ting ener	gy expe	enditu	re usi	ng multi	variable li	near
regression	when	band-pass	filtering is	applied	and wh	en it is	s not.	The	features	computed	are
ACAbsArea and ACFFTPeaks computed per sensor over sliding windows of 5.6s in length.											

when band-pass filtering is applied and when it is not during the estimation of energy expenditure.

The experiments performed in this section utilize multivariable linear regression to measure the impact of applying and not applying the band-pass filter during the estimation of energy expenditure. The band-pass filter applied is a Chebyshev infinite impulse response filter designed using the parameters shown in Table 5-2. Performance is measured using two features: *ACAbsArea* and *ACFFTPeaks*. This is because the quality of these features is likely to be affected by the application of the band-pass filter.

Table 5-74 shows the performance of applying and not applying the filter while estimating energy expenditure using multivariable linear regression and the *ACAbsArea* and *ACFFTPeaks* features computed per sensor over windows of 5.6s in length. Performance per activity is shown in Appendix B10. The table clearly shows that the performance is higher when the filter is applied. In fact, the improvement in performance observed is more substantial than the improvements observed using more complex non-linear regression algorithms (see Section 5.6.3). For example, RMSE increased +0.3 MET for the *ACAbsArea* feature and +0.06 MET for the *ACFFTPeaks* feature. Performance improves for the *ACAbsArea* feature because band-pass filtering smoothes the signals this reducing the amount of noise. Similarly, performance improves for the *ACFFTPeaks* feature because high frequency noise is reduced leading to more consistent estimations of the frequency components of the signal. Performance improves less for the *ACFFTPeaks* feature because it is less dependent on the magnitude of the accelerometer signal (since it does not include the DC component of the signal), and the magnitude of the accelerometer signal is the most affected by the application of the filter.

In summary, the results presented in Table 5-74 indicate that band-pass filtering the signal significantly improves the performance of energy expenditure estimation. As a result, all experiments presented in this thesis utilize band-pass filtering between frequencies of 0.1 and 20Hz. One potential problem with band-pass filtering the acceleration signals is that potentially useful information about postures (static acceleration) is lost. It might be possible that this information is indeed necessary to better estimate energy expenditure since different postures have different energy expenditure costs [122] (e.g. 1 MET for *lying down* vs. 1.2 MET for *standing*) even when they involve no motion. Furthermore, energy expenditure can add up for postures executed over long periods of time. As a result, Section 5.6.7 will later explore the performance of energy expenditure estimation when features that capture posture information are used and when they are not.
### 5.6.5 Can Feature Computation Per Axis Improve Performance Over Feature Computation per Sensor?

Most prior work in energy expenditure estimation utilizes features computed over signals summarizing the overall motion (acceleration) experienced by a particular body segment. For example, the overall motion experienced by a triaxial accelerometer at the hip is summarized by computing the sum (x+y+z) or the signal vector magnitude over the accelerometer axes  $(\sqrt{x^2 + y^2 + z^2})$ . The rationale behind this is that overall motion per body segment is sufficient to estimate energy expenditure well since energy expended is directly proportional to the amount of motion per body segment. One advantage of this approach, referred as feature computation per sensor in this work, is that the amount of computation is reduced since features are computed over a single signal instead of over three different signals (x, y, and z). Nevertheless, some prior work [33, 151, 160, 247] [218] also suggests that analyzing motion over individual axis (e.g. x, y, or z), particularly the vertical component of acceleration, can improve energy expenditure estimates. As a result, this section investigates the difference in performance obtained when estimating energy expenditure using features computed over signals summarizing overall motion per sensor and using features computed over individual acceleration axis (referred as feature computation per axis).

In the experiments presented in this section, feature computation per sensor is performed by first summarizing the motion experienced at each sensor by summing the acceleration values at each individual axis (x+y+z) sample by sample. Feature computation per axis is performed by computing the features of interest over each individual axis (x, y, and z). Thus, feature vectors corresponding to feature computation per axis are three times longer that feature vectors for feature computation per sensor.

The performance of feature computation per sensor and per axis is measured by estimating energy expenditure in two extreme conditions: (1) when few simple features are used to estimate energy expenditure (*ACAbsArea*), and (2) when all the accelerometer-based features are used to estimate energy expenditure. The performance over both feature sets is also analyzed using a linear regression algorithm (multivariable linear regression) and a non-linear regression algorithm (M5' model trees). All features are computed over sliding windows of 5.6s in length. As will be explained later, this is the final window length selected for estimating energy expenditure in Section 5.6.6.

Table 5-75 presents the performance obtained over the MIT energy expenditure dataset when energy expenditure is estimated using multivariable linear regression and the *ACAbsArea* feature computed per sensor and per axis. Results per activity can be found in Appendix B11. The table shows that feature computation per axis improves the coefficient of correlation 0.03 units and RMSE 0.07 METs. The improvement obtained in RMSE is distributed across the postures, exercise, and resistance exercise activity categories. When the RMSE per activity is inspected (see Figure 5-48) it is found that performance is slightly better (-0.2 to -0.8 MET) for activities involving postures and upper body motion such as *writing*, *typing*, *wiping/dusting*, *playing video games*, *doing dishes*, and *sitting fidgeting hands and arms*, *and bicep curls*. At the same time, the performance over *bench weight lifting* (another activity involving upper body motion) slightly decreased between 0.1 and 0.3 MET.

Activity Category	Linear Regression Per Sensor	Linear Regression Per Axis	Change in RMSE Performance (Per Axis – Per Sensor)
<b>Correlation Coefficient</b>	$0.68\pm0.06$	$0.71 \pm 0.13$	-
All	$1.36 \pm 0.30 \; (1.02 \pm 0.17)$	$1.29 \pm 0.29 \; (0.92 \pm 0.18)$	0.07 Improvement
Postures	0.7±0.2 (0.7±0.2)	0.5±0.2 (0.4±0.2)	0.2 Improvement
Ambulation	1.1±0.4 (1.0±0.4)	1.1±0.4 (1.0±0.4)	0 Unchanged
Exercise	1.5±0.7 (1.4±0.7)	1.3±0.8 (1.2±0.8)	0.2 Improvement
Resistance Exercise	1.4±0.6 (1.3±0.6)	1.2±0.7 (1.1±0.7)	0.2 Improvement
Household	0.7±0.2 (0.6±0.2)	0.7±0.3 (0.6±0.3)	0 Unchanged

Table 5-75: Correlation coefficient, root mean squared error, and mean absolute error (shown in parenthesis) per activity category when estimating energy expenditure with multivariable linear regression using the *ACAbsArea* feature computed per sensor and per axis over windows of 5.6s in length.

Activity Category	M5' Model Tree Per Sensor	M5' Model Tree Per Axis	Change in RMSE Performance (Per Axis – Per Sensor)
Correlation Coefficient	$0.72\pm0.08$	$0.70 \pm 0.15$	-
All	1.24±0.29 (0.86±0.19)	$1.34\pm0.43~(0.92\pm0.23)$	0.1 Decline
Postures	0.6±0.3 (0.5±0.2)	0.5±0.2 (0.4±0.2)	0.1 Improvement
Ambulation	1.1±0.4 (0.9±0.4)	1.1±0.4 (1.0±0.4)	0 Unchanged
Exercise	1.3±0.7 (1.2±0.7)	1.3±0.9 (1.2±0.8)	0 Unchanged
Resistance Exercise	1.1±0.6 (1.0±0.6)	1.3±0.8 (1.1±0.7)	0.2 Decline
Household	0.6+0.3 (0.5+0.2)	$0.7\pm0.3$ (0.6±0.3)	0.1 Decline

Table 5-76: Correlation coefficient, root mean squared error, and mean absolute error (shown in parenthesis) per activity category when estimating energy expenditure with a M5' model tree using the *ACAbsArea* feature set computed per sensor and per axis over windows of 5.6s in length.

Activity Category	Activity Category Linear Regression		Change in RMSE
	Per Sensor	Per Axis	Performance
			(Per Axis – Per Sensor)
<b>Correlation Coefficient</b>	$0.74 \pm 0.10$	$0.73\pm0.12$	-
All	$1.24 \pm 0.30 \; (0.91 \pm 0.20)$	$1.27 \pm 0.26 \; (0.93 \pm 0.19)$	0.03 Decline
Postures	0.7±0.3 (0.6±0.3)	0.6±0.3 (0.5±0.3)	0.1 Improvement
Ambulation	1.2±0.5 (1.0±0.5)	1.1±0.5 (0.9±0.4)	0.1 Improvement
Exercise	1.2±0.7 (1.1±0.7)	1.2±0.7 (1.1±0.7)	0 Unchanged
Resistance Exercise	1.1±0.6 (1.0±0.6)	1.1±0.6 (1.0±0.6)	0 Unchanged
Household	0.7±0.3 (0.6±0.3)	0.8±0.4 (0.7±0.4)	0.1 Decline

Table 5-77: Correlation coefficient, root mean squared error, and mean absolute error (shown in parenthesis) per activity category when estimating energy expenditure with multivariable linear regression using the *MaxAcceleration* feature set computed per sensor and per axis over windows of 5.6s in length.

Activity Category	M5' Model Tree Per Sensor	M5' Model Tree Per Axis	Change in RMSE Performance (Per Axis – Per Sensor)
<b>Correlation Coefficient</b>	$0.64 \pm 0.17$	$0.64 \pm 0.12$	-
All	$1.56 \pm 0.67 \ (0.94 \pm 0.24)$	$1.46 \pm 0.23 \ (1.02 \pm 0.16)$	0.1 Improvement
Postures	1.0±1.5 (0.7±0.7)	0.6±0.3 (0.5±0.3)	0.4 Improvement
Ambulation	1.2±0.5 (1.0±0.5)	1.3±0.7 (1.2±0.6)	0.1 Decline
Exercise	1.3±0.7 (1.2±0.7)	1.5±0.8 (1.3±0.7)	0.2 Decline
Resistance Exercise	1.1±0.6 (1.0±0.5)	1.3±0.6 (1.2±0.5)	0.2 Decline
Household	0.8±0.5 (0.6±0.4)	0.9±0.5 (0.8±0.4)	0.1 Decline

Table 5-78: Correlation coefficient, root mean squared error, and mean absolute error (shown in parenthesis) per activity category when estimating energy expenditure with the M5' model tree using the *MaxAcceleration* feature set computed per sensor and per axis over windows of 5.6s in length.



Figure 5-48: Root mean squared error represented as a gray scale image when energy expenditure is estimated using multivariable linear regression using the ACAbsArea feature computed per sensor and per axis over windows of 5.6s in length. The image is scaled to show the lowest RMSE of 0.3 MET in white and the largest RMSE of 3.1 in black.

It seems that analyzing motion per sensor improves energy expenditure for upper body activities except for *bench weight lifting*. One possible reason for this is that the motion of *bench weight lifting* is so periodic that it is well captured using feature computation per sensor and using feature computation per axis only reduces the amount of training data available to train each predictor variable.

Table 5-76 presents the performance of estimating energy expenditure using an M5' model tree and the *ACAbsArea* feature computed per sensor and per axis over windows of 5.6s in length. The first thing to notice is that the performance obtained during feature computation per sensor is better than the performance obtained during feature computation per axis when multivariable linear regression is used. This is expected since model trees are more complex regression algorithms capable of learning non-linear

relationships in the data. However, when the performance per sensor and per axis are compared for the M5' model tree, it is found that feature computation per axis has a lower performance than feature computation per sensor. One possible explanation is that feature computation per axis produces a three-fold increase in the vector size of the features and consequently, the training data available at the leaf nodes of the M5' tree is insufficient to learn the multivariable linear regression models associated with each leaf node. Table 5-77 and Table 5-78 also illustrate the decline in performance of feature computation per axis when the number of features used is large. Given the amount of training data available in this work, it is unclear if feature computation per axis will improve performance over feature computation per sensor. The only improvement observed in this section was obtained with the *ACAbsArea* feature, which has a very low feature vector size of 7.

In conclusion, the results presented in this section suggest that when the number of features is low, feature computation per axis slightly improves performance over feature computation per sensor. The improvement in performance observed using the *ACAbsArea* feature and multivariable linear regression was +0.03 units for the correlation coefficient and 0.07 MET for the RMSE. Even so, when the number of features used is large and/or the regression algorithm used has large training data requirements, feature computation per axis damages performance by increasing the total number of features by a factor of three when not enough data is available for training. For these reasons, from this point on, feature computation per sensor will be used when presenting results in upcoming sections.

## 5.6.6 What is the Optimal Sliding Window Length to Use?

Most existing algorithms that estimate energy expenditure from accelerometer data generate estimates over sliding windows of one minute in length. This relatively long window length (also known as the epoch) was originally used due to hardware limitations in storage capacity and battery life of the first off-the-shelf accelerometers that appeared in the market. Nowadays, however, these limitations no longer exist. It may make sense to reduce the size of this window for two reasons: (1) shorter windows will allow shorter real-time estimation delays that would also enable faster interventions at the point of decision, and (2) motion can vary considerably over one minute windows, so reducing the window length might improve estimation of energy expenditure. As a result, this section explores the impact in performance of varying the length of the sliding window during the estimation of energy expenditure.

The length of the sliding window also affects the quality or resolution of some of the features used in this work such as the FFT transformation and the Pearson's correlation coefficients. Usually, the longer the window length, the better these features can be estimated. On the other hand, the longer the window length, the longer the end-user of an energy expenditure estimation system has to wait for a recognition result. Consequently, this section determines the most appropriate window length by measuring the performance of multivariable linear regression and M5'model trees over the two features whose quality is most likely to vary with the window length: FFT peaks (*ACFFTPeaks*) and Pearson correlation coefficients (*ACCorr*). As a baseline, the performance is also computed utilizing the *ACAbsArea* feature. Linear regression and model trees are used in

this experiment because they represent the linear and non-linear regression algorithms with best performance over all features as found in Section 5.6.3.

The features will be first computed over window lengths ranging from 1.4 to 91 seconds (64 to 2048 accelerometer samples) using linear regression and feature computation per sensor, and later on a reduced set of window lengths using the M5' model tree algorithm. This procedure is followed to minimize the time and number of experiments to run while maximizing interpretability of results. Window lengths shorter than 1.4s were not considered because they are intuitively too short to capture the repetitive motion patterns found in some periodic activities such as *walking* slowly at 2mph. Similarly, window lengths longer than 91s were not considered due to the extremely long real-time classification delay they introduce into the system.

The lengths of the sliding windows explored in this section are constrained to be a power of two by the algorithms required to compute the Fourier and Wavelet transformations efficiently. Table 5-11 shows the window lengths explored in this section in number of acceleration samples and corresponding time in seconds assuming a sampling rate of 45Hz.

Figure 5-49 presents the plot of the root mean squared error obtained while varying the window length from 1.4s to 91s. Appendix B12 presents the same information in a tabular manner. The plot indicates that the RMSE error decreases as the window length is increased when both the *ACAbsArea* and *FFTCorr* feature sets are used. For example, RMSE decreases -0.22 MET when the window length is increased from 1.4 to 91s for the *ACAbsArea* feature and -0.16 MET for the *FFTCorr* feature set.

This modest improvement might be due to the increased smoothing of the values for the *ACAbsArea* feature and for the ground truth energy expenditure signals. Another possible reason for the improvement is that the quality or resolution for the *ACFFTPeaks* and *ACCorr* features increases as longer window lengths are used. Figure 5-49 also illustrates that the RMSE for the ambulation and exercise categories remains practically unchanged up to a window length of 22.7s for the *ACAbsArea* feature and up to a window length of 44.5s for the *FFTCorr* feature set. Figure 5-49 shows that the RMSE for the postures category increases at window lengths of 11.3s and 45.5s, but later decreases at a window length of 91s. Plot Figure 5-49a on the contrary, illustrates that the RMSE for postures indeed decreases as the window length is increased, most likely due to the smoothing effect of longer windows.

When the RMSE is analyzed per activity for the *FFTCorr* feature set (see Figure 5-51 and Appendix B12), it is found that it changes little among all activities up to a window length of 11.3s. Once this window is reached, RMSE for all activities decreases between 0.1-0.2 METs at a window of 11.3s and between 0.1-0.6 METs at a final window length of 90s. One possible reason why the improvements are so minimal is that the estimated and ground truth energy expenditure values follow one another independently of the window length utilized. In other words, the error depends on the difference between the estimated and ground truth energy expenditure signals so changing how these signals are partitioned does not impact error significantly. One of the main problems of using long windows; however, is that performance decreases for physically intense activities of short duration such as *ascending stairs* and *descending stairs*. This can be seen in Figure 5-51 where the error is the maximum (black area) for these activities at a window length of 90s. The examples collected for these activities have individual durations of less than a



Figure 5-49: Root mean squared error obtained when estimating energy expenditure in a subject independent manner over the MIT dataset using multivariable linear regression, the *ACAbsArea* and *FFTCorr* (*ACFFTPeaks+ACCorr*) feature sets computed per sensor over windows of varying lengths.



Figure 5-50: Root mean squared error obtained when estimating energy expenditure in a subject independent manner over the MIT dataset using a M5' model trees, the *ACAbsArea* and *FFTCorr* (*ACFFTPeaks+ACCorr*) feature sets computed per sensor over windows of varying lengths.



Figure 5-51: Root mean squared error (RMSE) represented as a grayscale image when estimating energy expenditure using multivariable linear regression using the *FFTCorr* feature set computed per sensor over sliding windows of varying length. The image is scaled to show the largest RMSE of 2.2MET in black and the lowest RMSE of 0.3MET in white.

minute, and are followed by resting periods. As a result, their energy expenditure is poorly detected with a window length of 90s because energy expenditure associated with *ascending stairs* is averaged with the energy expenditure associated with resting. This effect can also be seen for *running at 6mph* in Figure 5-51 since its error increases after a window length of 11.3s. This is because most examples for this physically demanding activity have durations of less than 1.5min. Another problem is that at large window lengths, the number of training examples per activity is reduced considerably thus impacting performance. Figure 5-51 also shows that RMSE decreases between -0.1 and -0.6 METs for household activities.

Figure 5-50 presents plots of the root mean square error per activity category when the M5' model tree is used to predict energy expenditure using the ACAbsArea and FFTCorr

features over varying window lengths. Overall, the plots in Figure 5-50 indicate that the RMSE also decreases as the window length is increased. The RMSE is slightly lower (-0.14 MET for *ACAbsArea* and -0.03 for the *FFTCorr* feature set) than the one obtained using linear regression because model trees can capture non-linear relationships in the data. However, one might question if the improvement obtained justifies the utilization of this more complex algorithm. As the window is increased from 5.6s to 91s, overall RMSE decreases 0.27 MET for the *ACAbsArea* feature and 0.23 MET for the *FFTCorr* feature set. The improvement obtained is little and the activity categories that present the higher decrease in error are ambulation (0.3-0.5 MET) and exercise (0.3-0.4 MET). The category that experiences the least decrease in error (0.0 MET for *ACAbsArea* and 0.1 MET for *FFTCorr*) is household activities.

In summary, the results presented in this section suggest that in general, root mean squared error decreases as the window length increases but this decrease in RMSE is modest (e.g. between 0.16 and 0.27 MET when length is increased from 1.4s to 91s). RMSE error decreases as the window length increases because the feature values and the ground truth energy expenditure values are smoothed out as the window length is increased, particularly non-steady state intervals. It also increases because the quality or resolution of some features (e.g. ACFFTPeaks and ACCorr) also increases as the window length increases. As a result, small windows can be utilized to estimate energy expenditure without affecting error considerably. This is an intuitive result because the most important information to estimate energy expenditure from accelerometer data is overall amount of motion, which can be estimated well even when small windows are used. The results also indicate than long window lengths increase RMSE in short duration activities such as *ascending stairs*, *descending stairs*, and physically demanding activities (of short duration) such as running at 6mph. This is because their energy expenditure data is averaged with the energy expenditure data associated with activities preceding or following these short duration activities. As a result, a window length of 5.6s will be used to estimate energy expenditure for the remaining of this work. This window length allows prediction of energy expenditure over short duration activities, minimizes energy expenditure estimation lag (during real-time implementations), and it has shown good performance with respect to windows of shorter duration. Finally, this is the same window length used to recognize activities in Section 5.4.6. Using the same window length to recognize activities and estimate energy expenditure simultaneously reduces computational complexity since features do not have to be computed twice over different window lengths (assuming some features are used in both algorithms).

# 5.6.7 Which Features Improve Energy Expenditure Estimation and Maintain Computational Cost Low?

Most prior work in estimating energy expenditure from accelerometer data utilizes raw accelerometer values summed over windows of one minute in length as the predictor variables (features) in the regression equations. However, recent work [152, 181] suggests that more complex features computed over the accelerometer signal such as the inter-quartile interval, skew, kurtosis, and frequency domain energy combined with non-linear regression algorithms can improve energy expenditure estimation. The intuition behind this is that more complex features might capture important motion information

that would be otherwise lost if simple sums of accelerometer values over one minute windows are used. As a result, this section investigates the performance of estimating energy expenditure using combinations of a large set of 41 features. The main goal of the section is to identify the subset of features that maximizes performance when considering the computational requirements of the features. To the best of the author knowledge, no prior work has analyzed the impact in performance of such a large set of features in estimating energy expenditure.

This section first evaluates the performance over individual features to identify those with higher performance. Later, the performance over combinations of individual features with high performance is evaluated to determine the best subset of features to use. All experiments performed in this section utilize feature computation per sensor, sliding windows of 5.6s in length (window length selected in the previous Section), and multivariable linear regression to estimate energy expenditure in a subject independent manner. As shown in Section 5.6.3, the performance obtained using multivariable linear regression algorithms, at least over the dataset explored in this work. The experiments are performed in a best case scenario where all the seven accelerometers are used for estimating energy expenditure. Section 5.6.8 will later analyze what is the best combination of sensors to use and where should they be worn.

A complete list of all the features explored in this section is shown in Appendix A3. These features consist on a superset of features used in prior work to recognize activities that have shown good performance as well as some new features not explored before. Table 5-12 presents a list of the features explored and a brief explanation of the information they attempt to capture. Features are computed after preprocessing the acceleration signals to better differentiate between motion information and posture information. The features intended to capture motion information are computed over the accelerometer signals after applying a band-pass filter between the frequencies of 0.1 and 20Hz. This preprocessing has two goals: (1) eliminate the DC or static component of the acceleration signal due to the orientation of the sensors with respect to ground (posture information) and (2) filter high frequency noise and motion not generated by the human body. The features intended to capture posture information are computed over the accelerometer signals after low-pass filtering them at a cut-off frequency of 1Hz. This has the purpose of eliminating most of the signal information due to body motion and preserving the information due to static acceleration or posture. Features that capture motion information start with the prefix "AC" and those that capture posture information start with the prefix "DC".

Table 5-79 presents the performance obtained while individual features are used to estimate energy expenditure in a subject independent manner using multivariable linear regression. The features shown in Table 5-79 are ordered to present the ones with higher correlation coefficients at the top and the ones with lower correlation coefficients at the bottom. Table 5-79 shows that the best performing features is *ACFFTPeaks* with a correlation coefficient of 0.72 and a RMSE of 1.28. This is a good result because this feature does not strongly depend on the magnitude of the accelerometer signal. Invariance to the magnitude of the accelerometer signal is important because it can vary with changes in the location and orientation of the sensors (during installation on the body) and with hardware differences across accelerometers. The *ACFFTPeaks* feature

	Features subsets	Correlation	All	Postures	Ambulation	Exercise	Resistance	Househol
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$	(Number of	coefficient for	Activities				Exercise	d
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	features)	All						
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	ACFFTPeaks (70)	0.72 + 0.07	$1.28 \pm 0.30$	0.6+0.2	1.2+0.5	1.4+0.8	1.2+0.7	0.7+0.2
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$			$(0.93 \pm 0.17)$	(0.6±0.2)	(1.1±0.5)	$(1.3\pm0.8)$	(1.1±0.7)	(0.6±0.2)
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	ACFFTCoeff (889)	$0.72\pm0.06$	$1.34 \pm 0.25$	0.6±0.2	1.4±0.6	1.4±0.8	1.2±0.7	0.6±0.2
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$			$(0.97 \pm 0.17)$	(0.5±0.2)	(1.2±0.6)	(1.3±0.8)	(1.0±0.7)	(0.5±0.2)
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	ACIQR (7)	$0.70\pm0.06$	$1.32\pm0.31$	0.7±0.2	1.1±0.4	$1.5\pm0.7$	1.3±0.6	0.7±0.2
$\begin{array}{c} \mathrm{ACg}(7) & 0.69 \pm 0.06 & 1.33 \pm 0.30 & 0.7\pm 0.2 & 1.0\pm 0.4 & 1.5\pm 0.7 & 1.3\pm 0.6 & 0.7\pm 0.2 \\ \mathrm{(0.9\pm 0.16)} & (1.4\pm 0.7) & (1.2\pm 0.6) & (0.5\pm 0.2) \\ \mathrm{(1.0\pm 0.7)} & (1.3\pm 0.5) & 0.7\pm 0.2 & (0.9\pm 0.4) & (1.4\pm 0.7) & (1.3\pm 0.6) & (0.5\pm 0.2) \\ \mathrm{(1.0\pm 0.7)} & (1.3\pm 0.5) & 0.7\pm 0.2 & (1.0\pm 0.4) & 1.5\pm 0.7 & 1.4\pm 0.6 & 0.7\pm 0.2 \\ \mathrm{(1.0\pm 0.7)} & (1.3\pm 0.6) & 0.0\pm 0.2 & (1.0\pm 0.4) & (1.4\pm 0.7) & (1.3\pm 0.6) & (0.6\pm 0.2) \\ \mathrm{(1.0\pm 0.4)} & (1.0\pm 0.2) & (0.7\pm 0.2) & (1.0\pm 0.4) & (1.4\pm 0.7) & 1.4\pm 0.6 & 0.7\pm 0.2 \\ \mathrm{(1.0\pm 0.4)} & (1.0\pm 0.4) & (1.0\pm 0.4) & (1.4\pm 0.7) & (1.3\pm 0.6) & (0.6\pm 0.2) \\ \mathrm{(1.0\pm 0.4)} & (1.0\pm 0.4) & (1.0\pm 0.4) & (1.4\pm 0.7) & (1.3\pm 0.6) & (0.6\pm 0.2) \\ \mathrm{(1.0\pm 0.4)} & (1.0\pm 0.4) & (1.0\pm 0.4) & (1.4\pm 0.7) & (1.3\pm 0.6) & (0.6\pm 0.2) \\ \mathrm{(1.0\pm 0.4)} & (1.0\pm 0.4) & (1.0\pm 0.4) & (1.4\pm 0.7) & (1.2\pm 0.5) & 0.8\pm 0.2 \\ \mathrm{(1.0\pm 0.4)} & (1.0\pm 0.4) & (1.0\pm 0.4) & (1.4\pm 0.7) & (1.2\pm 0.5) & 0.8\pm 0.2 \\ \mathrm{(1.0\pm 0.4)} & (1.0\pm 0.4) & (1.0\pm 0.4) & (1.0\pm 0.4) & (1.0\pm 0.5) & (0.7\pm 0.2) \\ \mathrm{(1.0\pm 0.4)} & (1.0\pm 0.4) & (1.0\pm 0.4) & (1.0\pm 0.4) & (1.0\pm 0.5) & (0.7\pm 0.2) \\ \mathrm{(1.0\pm 0.4)} & (1.0\pm 0.4) & (1.0\pm 0.4) & (1.0\pm 0.4) & (1.0\pm 0.5) & (1.0\pm 0.6) & (0.7\pm 0.2) \\ \mathrm{(1.0\pm 0.4)} & (1.0\pm 0.4) \\ \mathrm{(1.0\pm 0.7)} & (1.0\pm 0.4) & (1.0\pm 0$			$(0.98 \pm 0.16)$	(0.7±0.2)	(1.0±0.4)	$(1.4\pm0.7)$	$(1.2\pm0.6)$	$(0.6\pm0.2)$
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	ACQ3 (7)	$0.69 \pm 0.06$	$1.33 \pm 0.30$	$0.7\pm0.2$	$1.0\pm0.4$	$1.5\pm0.7$	$1.3\pm0.6$	$0.7\pm0.2$
AC IotalAbsArea         0.68 $\pm$ 0.00         1.5 $\pm$ 0.1         0.8 $\pm$ 0.1         0.8 $\pm$ 0.1         0.8 $\pm$ 0.0         0.7 $\pm$ 0.2         1.1 $\pm$ 0.4         1.5 $\pm$ 0.7         1.3 $\pm$ 0.0         0.7 $\pm$ 0.2           ACAbsArea(7)         0.68 $\pm$ 0.06         1.3 $\pm$ 0.3         0.7 $\pm$ 0.2         1.1 $\pm$ 0.4         1.5 $\pm$ 0.7         1.4 $\pm$ 0.6         0.7 $\pm$ 0.2           ACMCR (7)         0.68 $\pm$ 0.06         1.3 $\pm$ 0.3         0.7 $\pm$ 0.2         1.1 $\pm$ 0.4         1.5 $\pm$ 0.7         1.4 $\pm$ 0.6         0.7 $\pm$ 0.2           ACMCR (7)         0.68 $\pm$ 0.06         1.37 $\pm$ 0.32         0.7 $\pm$ 0.4         1.4 $\pm$ 0.7         1.2 $\pm$ 0.5         0.8 $\pm$ 0.2           ACMOdVigEnergy         0.68 $\pm$ 0.09         1.37 $\pm$ 0.32         0.7 $\pm$ 0.4         1.4 $\pm$ 0.7         1.2 $\pm$ 0.6         0.7 $\pm$ 0.2           ACSF (5)         0.67 $\pm$ 0.05         1.37 $\pm$ 0.30         0.9 $\pm$ 0.2         1.2 $\pm$ 0.4         1.4 $\pm$ 0.7         1.2 $\pm$ 0.6         0.7 $\pm$ 0.2           ACTotalSF (1)         0.65 $\pm$ 0.07         1.41 $\pm$ 0.33         0.9 $\pm$ 0.2         1.1 $\pm$ 0.4         1.5 $\pm$ 0.7         1.3 $\pm$ 0.6         0.7 $\pm$ 0.2           ACTotalSF (1)         0.65 $\pm$ 0.07         1.41 $\pm$ 0.33         0.9 $\pm$ 0.2         1.1 $\pm$ 0.4         1.5 $\pm$ 0.7         1.3 $\pm$ 0.6         0.		0.69 . 0.06	$(0.99 \pm 0.16)$	$(0.7\pm0.2)$	$(0.9\pm0.4)$	$(1.4\pm0.7)$	$(1.2\pm0.6)$	$(0.6\pm0.2)$
$ \begin{array}{c} 10 \\ ACAbsArea(7) \\ ACAbsArea(7) \\ ACAbsArea(7) \\ ACAbsArea(7) \\ ACAbsArea(7) \\ ACAbsArea(7) \\ ACAbsMean (7) \\ 0.68 \pm 0.06 \\ 1.35 \pm 0.30 \\ (1.0 \pm 0.2) \\ (0.7\pm 0.2) \\ (1.0 \pm 0.2) \\ (0.7\pm 0.2) \\ (1.0 \pm 0.4) \\ (1.4 \pm 0.7) \\ (1.4 \pm 0.7) \\ (1.4 \pm 0.7) \\ (1.4 \pm 0.6) \\ (1.4 \pm 0.7) \\ (1.4 \pm 0.6) \\ (1.7 \pm 0.18) \\ (0.7 \pm 0.2) \\ (1.0 \pm 0.17) \\ (1.4 \pm 0.7) \\ (1.4 \pm 0.7) \\ (1.4 \pm 0.7) \\ (1.4 \pm 0.7) \\ (1.4 \pm 0.6) \\ (1.7 \pm 0.18) \\ (0.7 \pm 0.2) \\ (1.0 \pm 0.17) \\ (1.4 \pm 0.7) \\ (1.4 \pm 0.7) \\ (1.4 \pm 0.7) \\ (1.4 \pm 0.6) \\ (1.7 \pm 0.18) \\ (0.7 \pm 0.2) \\ (1.0 \pm 0.17) \\ (1.4 \pm 0.7) \\ (1.4 \pm 0.6) \\ (0.7 \pm 0.2) \\ (1.1 \pm 0.18) \\ (0.8 \pm 0.3) \\ (1.1 \pm 0.18) \\ (0.8 \pm 0.3) \\ (1.1 \pm 0.18) \\ (0.8 \pm 0.3) \\ (1.1 \pm 0.4) \\ (1.5 \pm 0.7) \\ (1.4 \pm 0.6) \\ (0.8 \pm 0.3) \\ (1.4 \pm 0.7) \\ (1$	ACTOTALADSAFea	$0.08 \pm 0.00$	$1.30 \pm 0.31$ (1.02 ± 0.17)	$0.8\pm0.2$	$1.0\pm0.4$	$1.5\pm0.7$	$1.3\pm0.0$	$0.7\pm0.2$
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$	(1) ACAbsArea $(7)$	$0.68 \pm 0.06$	$(1.03 \pm 0.17)$ 1.36 ± 0.30	$(0.7\pm0.2)$	$(0.9\pm0.4)$	$(1.4\pm0.7)$ 1.5+0.7	$(1.3\pm0.0)$	$(0.0\pm0.2)$
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	ACAUSAICa(7)	$0.03 \pm 0.00$	$(1.0 \pm 0.30)$	$(0.7\pm0.2)$	(1.0+0.4)	(1.4+0.7)	$(1.3\pm0.6)$	$(0.6\pm0.2)$
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$	ACAbsMean (7)	$0.68 \pm 0.06$	$1.6 \pm 0.2)$ 1.36 ± 0.30	0.7±0.2	1 1+0 4	15+07	1 4+0 6	$0.0\pm0.2$
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $		0.00 ± 0.00	$(1.02 \pm 0.17)$	$(0.7\pm0.2)$	$(1.0\pm0.4)$	$(1.4\pm0.7)$	$(1.3\pm0.6)$	$(0.6\pm0.2)$
	ACMCR (7)	$0.68\pm0.06$	$1.37 \pm 0.32$	0.7±0.4	1.4±0.5	1.4±0.7	1.2±0.5	0.8±0.2
$ \begin{array}{c cccc} ACMod VigEnergy \\ (7) & (1.02 \pm 0.19) & (0.8 \pm 0.2) & (1.2 \pm 0.4) & (1.4 \pm 0.7) & (1.1 \pm 0.6) & (0.7 \pm 0.2) \\ (1.02 \pm 0.19) & (0.8 \pm 0.2) & (1.0 \pm 0.4) & (1.3 \pm 0.7) & (1.1 \pm 0.6) & (0.7 \pm 0.2) \\ ACSF (5) & 0.67 \pm 0.05 & 1.37 \pm 0.30 & 0.7 \pm 0.2 & 1.1 \pm 0.4 & 1.5 \pm 0.7 & 1.4 \pm 0.6 & (0.7 \pm 0.2) \\ ACTotalSF (1) & 0.65 \pm 0.07 & 1.1 \pm 1.03 & 0.9 \pm 0.2 & (1.1 \pm 0.4 & 1.5 \pm 0.7) & 1.3 \pm 0.6 & (0.7 \pm 0.2) \\ ACBandEnergy (7) & 0.64 \pm 0.06 & 1.46 \pm 0.31 & 0.8 \pm 0.3 & (1.1 \pm 0.4 & 1.5 \pm 0.7) & 1.3 \pm 0.6 & (0.7 \pm 0.2) \\ ACBandEnergy (7) & 0.64 \pm 0.07 & 1.45 \pm 0.31 & 0.8 \pm 0.3 & 1.1 \pm 0.4 & 1.5 \pm 0.7 & 1.3 \pm 0.6 & (0.8 \pm 0.3) \\ (1.10 \pm 0.17) & (0.7 \pm 0.2) & (0.9 \pm 0.4) & (1.5 \pm 0.7) & (1.3 \pm 0.6) & (0.8 \pm 0.3) \\ ACRange (7) & 0.64 \pm 0.07 & 1.45 \pm 0.31 & 0.8 \pm 0.3 & 1.1 \pm 0.4 & 1.5 \pm 0.7 & 1.3 \pm 0.6 & (0.8 \pm 0.3) \\ ACRange (7) & 0.63 \pm 0.07 & 1.46 \pm 0.31 & 0.8 \pm 0.3 & (1.1 \pm 0.4 & 1.5 \pm 0.7) & 1.3 \pm 0.6 & (0.8 \pm 0.3) \\ ACRange (7) & 0.63 \pm 0.07 & 1.46 \pm 0.31 & 0.8 \pm 0.2 & 1.1 \pm 0.4 & 1.5 \pm 0.7 & 1.3 \pm 0.6 & 0.8 \pm 0.2 \\ (1.11 \pm 0.18) & (0.8 \pm 0.2) & (1.1 \pm 0.4 & 1.5 \pm 0.7) & 1.3 \pm 0.5 & 0.8 \pm 0.2 \\ ACVar (7) & 0.60 \pm 0.06 & 1.50 \pm 0.32 & 1.1 \pm 0.2 & 1.1 \pm 0.4 & 1.5 \pm 0.7 & 1.5 \pm 0.6 & 0.8 \pm 0.2 \\ ACVar (7) & 0.60 \pm 0.10 & 1.52 \pm 0.35 & 1.0 \pm 0.2 & 1.1 \pm 0.4 & 1.5 \pm 0.7 & 1.5 \pm 0.6 & 0.8 \pm 0.2 \\ (1.17 \pm 0.18) & (1.1 \pm 0.2) & (1.0 \pm 0.5) & (1.6 \pm 0.7) & 1.4 \pm 0.6 & (0.7 \pm 0.2) \\ ACLowEnergy (7) & 0.60 \pm 0.10 & 1.52 \pm 0.35 & 1.0 \pm 0.2 & 1.3 \pm 0.5 & 1.6 \pm 0.7 & 1.4 \pm 0.6 & (0.8 \pm 0.2) \\ (1.20 \pm 0.10) & 1.52 \pm 0.35 & 1.0 \pm 0.2 & 1.3 \pm 0.5 & 1.6 \pm 0.7 & 1.4 \pm 0.6 & (0.8 \pm 0.2) \\ ACLowEnergy (7) & 0.50 \pm 0.08 & 1.66 \pm 0.32 & 1.1 \pm 0.2 & 1.1 \pm 0.5 & 1.7 \pm 0.6 & (1.5 \pm 0.7) & (1.4 \pm 0.6) & (0.7 \pm 0.2) \\ ACLowEnergy (7) & 0.50 \pm 0.08 & 1.66 \pm 0.32 & 1.4 \pm 0.3 & 1.2 \pm 0.5 & 1.8 \pm 0.8 & 1.6 \pm 0.6 & 1.1 \pm 0.4 \\ (7) & (1.27 \pm 0.19) & (0.9 \pm 0.3) & (1.5 \pm 0.5) & (1.5 \pm 0.7) & (1.4 \pm 0.6) & (0.7 \pm 0.2) \\ ACLowEnergy (7) & 0.50 \pm 0.08 & 1.66 \pm 0.32 & 1.4 \pm 0.3 & 1.2 \pm 0.5 & 1.4 \pm 0.6 & (1.1 \pm 0.4 & 1.5 \pm 0.7) & (1.4 \pm 0.6) & (0.$			$(1.03 \pm 0.20)$	(0.6±0.4)	(1.3±0.5)	(1.2±0.7)	(1.1±0.5)	(0.7±0.2)
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	ACModVigEnergy	$0.68\pm0.09$	$1.37\pm0.36$	0.9±0.2	1.2±0.4	1.4±0.7	1.2±0.6	0.7±0.2
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	(7)		$(1.02 \pm 0.19)$	(0.8±0.2)	(1.0±0.4)	(1.3±0.7)	(1.1±0.6)	(0.7±0.2)
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	ACSF (5)	$0.67\pm0.05$	$1.37\pm0.30$	0.7±0.2	1.1±0.4	$1.5\pm0.7$	$1.4\pm0.6$	0.7±0.2
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$			$(1.03 \pm 0.17)$	$(0.7\pm0.2)$	$(1.0\pm0.4)$	$(1.4\pm0.7)$	$(1.3\pm0.6)$	$(0.7\pm0.2)$
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	ACTotalSF(1)	$0.65 \pm 0.07$	$1.41 \pm 0.33$	0.9±0.2	1.1±0.4	$1.5\pm0.7$	1.3±0.6	0.7±0.2
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $		0.54 0.05	$(1.07 \pm 0.18)$	$(0.9\pm0.2)$	(1.0±0.4)	$(1.4\pm0.7)$	$(1.2\pm0.6)$	$(0.7\pm0.2)$
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	ACBandEnergy (7)	$0.64 \pm 0.06$	$1.46 \pm 0.31$	$0.8\pm0.3$	$1.1\pm0.4$	$1.6\pm0.7$	$1.3\pm0.6$	$0.9\pm0.3$
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$	$\Lambda$ CD = $\pi = \pi (7)$	0.64 + 0.07	$(1.10 \pm 0.17)$	$(0.7\pm0.3)$	$(0.9\pm0.4)$	$(1.5\pm0.7)$	$(1.3\pm0.6)$	$(0.8\pm0.3)$
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	ACKange(7)	$0.04 \pm 0.07$	$1.45 \pm 0.31$ (1.11 + 0.18)	$0.8\pm0.3$ (0.8±0.3)	$1.1\pm0.4$ (1.0+0.4)	$1.5\pm0.7$ (1.4±0.7)	$1.4\pm0.0$ (1.3±0.6)	$0.8\pm0.3$ (0.7+0.2)
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	$\Delta CEntropy (7)$	$0.63 \pm 0.07$	$(1.11 \pm 0.13)$ 1.46 ± 0.34	$(0.8\pm0.3)$	$(1.0\pm0.4)$	$(1.4\pm0.7)$ 1.5+0.7	$(1.3\pm0.0)$	$(0.7\pm0.2)$
ACVar (7) $0.60 \pm 0.06$ $1.50 \pm 0.32$ $(1.1\pm 0.2)$ $(1.1\pm 0.5)$ $(1.2\pm 0.7)$ $(1.5\pm 0.6)$ $(0.8\pm 0.2)$ ACLowEnergy (7) $0.60 \pm 0.10$ $1.52 \pm 0.35$ $1.0\pm 0.2$ $1.3\pm 0.5$ $1.6\pm 0.7$ $1.4\pm 0.6$ $(0.7\pm 0.2)$ ACDomFreqRatio $0.53 \pm 0.11$ $1.58 \pm 0.31$ $1.0\pm 0.2$ $(1.2\pm 0.5)$ $(1.5\pm 0.7)$ $(1.3\pm 0.6)$ $(0.7\pm 0.2)$ ACDomFreqRatio $0.53 \pm 0.11$ $1.58 \pm 0.31$ $1.0\pm 0.3$ $1.2\pm 0.5$ $1.8\pm 0.8$ $1.6\pm 0.6$ $1.1\pm 0.4$ (7) $0.50 \pm 0.08$ $1.66 \pm 0.34$ $1.4\pm 0.4$ $1.3\pm 0.5$ $1.8\pm 0.8$ $1.6\pm 0.6$ $0.9\pm 0.3$ ACEnergy (7) $0.50 \pm 0.08$ $1.66 \pm 0.34$ $1.4\pm 0.4$ $1.3\pm 0.5$ $1.8\pm 0.8$ $1.6\pm 0.6$ $0.9\pm 0.3$ DCPostureDist $0.42 \pm 0.16$ $1.75 \pm 0.30$ $1.3\pm 0.5$ $1.6\pm 0.6$ $1.8\pm 0.9$ $1.4\pm 0.8$ $1.1\pm 0.6$ (21) $0.39 \pm 0.16$ $1.66 \pm 0.32$ $1.4\pm 0.4$ $1.5\pm 0.7$ $1.8\pm 0.9$ $1.6\pm 0.7$ $1.0\pm 0.3$ ACCorr (21) $0.39 \pm 0.16$ $1.66 \pm 0.32$ $1.4\pm 0.4$ $1.5\pm 0.7$ $1.8\pm 0.9$ $1.6\pm 0.7$ $1.0\pm 0.3$ ACPitch (7) $0.22 \pm 0.18$ $1.85 \pm 0.36$ $1.8\pm 0.4$ $1.7\pm 0.5$ $1.9\pm 0.7$ $1.6\pm 0.6$ $1.9\pm 0.3$ DCMean (7) $0.22 \pm 0.18$ $1.85 \pm 0.36$ $1.8\pm 0.4$ $1.7\pm 0.6$ $2.1\pm 0.8$ $1.6\pm 0.6$ $1.1\pm 0.4$ DCArea (7) $0.22 \pm 0.18$ $1.85 \pm 0.36$ $1.8\pm 0.3$ $1.6\pm 0.5$ $(1.2\pm 0.8)$ $(1.6\pm 0.6)$ $(1.1\pm 0.4)$ <	ACLINOPY (7)	0.05 ± 0.07	$(1.11 \pm 0.21)$	$(0.8\pm0.2)$	(1.0+0.4)	(1.4+0.7)	$(1.2\pm0.5)$	(0.7+0.2)
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	ACVar(7)	$0.60 \pm 0.06$	$1.50 \pm 0.32$	1.1+0.2	1.1+0.5	1.7+0.7	1.5+0.6	0.8+0.2
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$			$(1.17 \pm 0.18)$	(1.1±0.2)	(1.0±0.5)	(1.6±0.7)	(1.4±0.6)	$(0.7\pm0.2)$
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	ACLowEnergy (7)	$0.60 \pm 0.10$	$1.52 \pm 0.35$	1.0±0.2	1.3±0.5	1.6±0.7	1.4±0.6	0.8±0.2
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$			$(1.17 \pm 0.21)$	(1.0±0.2)	(1.2±0.5)	(1.5±0.7)	(1.3±0.6)	(0.7±0.2)
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	ACDomFreqRatio	$0.53\pm0.11$	$1.58\pm0.31$	1.0±0.3	1.2±0.5	1.8±0.8	1.6±0.6	1.1±0.4
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$	(7)		$(1.27 \pm 0.19)$	(0.9±0.3)	(1.1±0.5)	$(1.7\pm0.8)$	(1.5±0.6)	(1.0±0.3)
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$	ACEnergy (7)	$0.50 \pm 0.08$	$1.66 \pm 0.34$	1.4±0.4	1.3±0.5	1.8±0.8	1.6±0.6	0.9±0.3
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	DCD D	0.40 0.16	$(1.33 \pm 0.20)$	$(1.3\pm0.3)$	(1.2±0.5)	$(1.8\pm0.8)$	$(1.6\pm0.6)$	$(0.9\pm0.3)$
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$	DCPostureDist	$0.42 \pm 0.16$	$1.75 \pm 0.30$	$1.3\pm0.5$	$1.6\pm0.6$	$1.8\pm0.9$	$1.4\pm0.8$	$1.1\pm0.6$
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$	(21)	$0.20 \pm 0.16$	$(1.55 \pm 0.18)$ 1.66 ± 0.22	$(1.3\pm0.3)$	$(1.3\pm0.0)$ 1.5±0.7	$(1.7\pm0.9)$	$(1.4\pm0.8)$ 1.6±0.7	$(1.1\pm0.0)$ 1.0±0.2
ACPitch (7) $(1.51 \pm 0.01)$ $(1.52 \pm 0.10)$ $(1.52 \pm 0.5)$ $1.9 \pm 0.7$ $1.6 \pm 0.6$ $1.0 \pm 0.3$ DCMean (7) $0.22 \pm 0.18$ $1.85 \pm 0.36$ $1.8 \pm 0.4$ $1.7 \pm 0.6$ $2.1 \pm 0.8$ $1.6 \pm 0.6$ $1.1 \pm 0.4$ DCArea (7) $0.22 \pm 0.18$ $1.85 \pm 0.36$ $1.8 \pm 0.4$ $1.7 \pm 0.6$ $2.1 \pm 0.8$ $1.6 \pm 0.6$ $1.1 \pm 0.4$ DCArea (7) $0.22 \pm 0.18$ $1.85 \pm 0.36$ $1.8 \pm 0.4$ $1.7 \pm 0.6$ $2.1 \pm 0.8$ $1.6 \pm 0.6$ $1.1 \pm 0.4$ DCTotalMean (7) $0.21 \pm 0.19$ $1.82 \pm 0.36$ $1.8 \pm 0.3$ $1.6 \pm 0.5$ $2.1 \pm 0.8$ $1.7 \pm 0.6$ $1.1 \pm 0.3$ MCTotalMean (7) $0.21 \pm 0.19$ $1.82 \pm 0.36$ $1.8 \pm 0.3$ $1.6 \pm 0.5$ $2.1 \pm 0.8$ $1.7 \pm 0.6$ $1.1 \pm 0.3$ MCTotalMean (7) $0.17 \pm 0.08$ $1.83 \pm 0.33$ $1.6 \pm 0.5$ $(2.0 \pm 0.8)$ $(1.6 \pm 0.6)$ $(1.0 \pm 0.3)$ ACSkew (7	ACCOIL(21)	0.39 ± 0.10	$(1.31 \pm 0.19)$	$(1.3\pm0.4)$	$(1.4\pm0.7)$	$(1.7\pm0.9)$	$(1.5\pm0.7)$	$(0.9\pm0.2)$
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	ACPitch (7)	$0.26 \pm 0.10$	$(1.31 \pm 0.17)$ 1 78 ± 0.31	2 2+0 5	1.4±0.7)	19+07	1.6±0.6	10+03
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	ner iten (/)	0.20 ± 0.10	$(1.41 \pm 0.20)$	$(2.0\pm0.5)$	$(1.5\pm0.5)$	$(1.8\pm0.7)$	$(1.5\pm0.6)$	$(0.9\pm0.3)$
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	DCMean (7)	$0.22 \pm 0.18$	$1.85 \pm 0.36$	1.8±0.4	1.7±0.6	2.1±0.8	1.6±0.6	1.1±0.4
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $			$(1.51 \pm 0.22)$	(1.8±0.5)	(1.7±0.6)	(2.0±0.8)	(1.6±0.6)	(1.1±0.4)
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	DCArea (7)	$0.22\pm0.18$	$1.85 \pm 0.36$	1.8±0.4	1.7±0.6	2.1±0.8	1.6±0.6	1.1±0.4
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$			$(1.51 \pm 0.22)$	(1.8±0.5)	(1.7±0.6)	(2.0±0.8)	(1.6±0.6)	(1.1±0.4)
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	DCTotalMean (7)	$0.21\pm0.19$	$1.82\pm0.36$	1.8±0.3	1.6±0.5	2.1±0.8	1.7±0.6	1.1±0.3
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$			$(1.48 \pm 0.21)$	(1.8±0.3)	(1.5±0.5)	(2.0±0.8)	(1.6±0.6)	(1.0±0.3)
ACKur (7) $(1.49 \pm 0.20)$ $(1.6\pm 0.3)$ $(1.6\pm 0.5)$ $(2.0\pm 0.8)$ $(1.6\pm 0.6)$ $(1.0\pm 0.3)$ ACKur (7) $0.11 \pm 0.04$ $1.83 \pm 0.33$ $1.8\pm 0.2$ $1.7\pm 0.5$ $2.1\pm 0.8$ $1.7\pm 0.6$ $1.0\pm 0.3$ $(1.49 \pm 0.21)$ $(1.8\pm 0.3)$ $(1.6\pm 0.5)$ $(2.0\pm 0.8)$ $(1.6\pm 0.6)$ $(1.0\pm 0.3)$ $ACAbsCV(7)$ $0.13 \pm 0.06$ $1.83 \pm 0.33$ $1.8\pm 0.3$ $1.7\pm 0.5$ $2.1\pm 0.8$ $1.7\pm 0.6$ $1.0\pm 0.3$ $ACAbsCV(7)$ $0.13 \pm 0.06$ $1.83 \pm 0.33$ $1.8\pm 0.3$ $1.7\pm 0.5$ $2.1\pm 0.8$ $1.7\pm 0.6$ $1.0\pm 0.3$ $(1.49 \pm 0.22)$ $(1.8\pm 0.3)$ $(1.6\pm 0.5)$ $(2.0\pm 0.8)$ $(1.6\pm 0.6)$ $(1.0\pm 0.3)$	ACSkew (7)	$0.17 \pm 0.08$	$1.83 \pm 0.33$	1.6±0.3	1.7±0.5	2.1±0.8	1.7±0.6	1.1±0.3
ACKur (7) $0.11 \pm 0.04$ $1.83 \pm 0.35$ $1.8 \pm 0.2$ $1.7 \pm 0.5$ $2.1 \pm 0.8$ $1.7 \pm 0.6$ $1.0 \pm 0.3$ ACAbs CV (7) $0.13 \pm 0.06$ $1.83 \pm 0.33$ $1.8 \pm 0.3$ $1.7 \pm 0.5$ $2.1 \pm 0.8$ $1.7 \pm 0.6$ $1.0 \pm 0.3$ ACAbs CV (7) $0.13 \pm 0.06$ $1.83 \pm 0.33$ $1.8 \pm 0.3$ $1.7 \pm 0.5$ $2.1 \pm 0.8$ $1.7 \pm 0.6$ $1.0 \pm 0.3$ ACAbs CV (7) $0.13 \pm 0.06$ $1.83 \pm 0.33$ $1.8 \pm 0.3$ $1.7 \pm 0.5$ $2.1 \pm 0.8$ $1.7 \pm 0.6$ $1.0 \pm 0.3$		0.11 0.04	$(1.49 \pm 0.20)$	$(1.6\pm0.3)$	(1.6±0.5)	$(2.0\pm0.8)$	(1.6±0.6)	$(1.0\pm0.3)$
$(1.49 \pm 0.21)$ $(1.0\pm 0.5)$ $(1.0\pm 0.5)$ $(1.0\pm 0.5)$ $(1.0\pm 0.5)$ $(1.0\pm 0.5)$ ACAbsCV(7) $0.13 \pm 0.06$ $1.83 \pm 0.33$ $1.8\pm 0.3$ $1.7\pm 0.5$ $2.1\pm 0.8$ $1.7\pm 0.6$ $1.0\pm 0.3$ $(1.49 \pm 0.22)$ $(1.8\pm 0.3)$ $(1.6\pm 0.5)$ $(2.0\pm 0.8)$ $(1.6\pm 0.6)$ $(1.0\pm 0.3)$	ACKur (/)	$0.11 \pm 0.04$	$1.83 \pm 0.33$ (1.49 ± 0.21)	$1.8\pm0.2$	$1./\pm0.5$	$(2.1\pm0.8)$	$1./\pm0.6$	$1.0\pm0.3$ (1.0±0.3)
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$	ACAbsCV(7)	$0.13 \pm 0.06$	$(1.49 \pm 0.21)$ 1.83 ± 0.32	(1.0±0.3) 1 8±0 3	(1.0±0.3) 1 7±0 5	(2.0±0.0) 2 1±0 8	(1.0±0.0) 1 7±0.6	$(1.0\pm0.3)$ 1.0±0.3
	nenosev (/)	0.15 ± 0.00	$(1.49 \pm 0.22)$	$(1.8\pm0.3)$	$(1.6\pm0.5)$	$(2.0\pm0.8)$	(1.6±0.6)	$(1.0\pm0.3)$

Table 5-79: Root mean squared error and mean absolute error (shown in parenthesis) when estimating energy expenditure in a subject independent manner using linear regression over individual features computed per sensor over windows of 5.6s in length.



Figure 5-52: Root mean squared error per activity category obtained when estimating energy expenditure over individual features computed per sensor over windows of 5.6s in length using multivariable linear regression. The grayscale image is scaled to show the lowest RMSE of 0.6MET in white and the largest RMSE of 2.2MET in black. In other words, poor areas of performance are shown as dark regions in the image.

was not expected to perform too well on the postures activity category because the first FFT coefficient representing the DC offset of the signal (posture information) is not used in this feature. However, this feature is able to predict energy expenditure for these sedentary activities by detecting very low frequency motion associated with these activities (0.1-3Hz) as observed from the variables included in the regression models learned. One disadvantage of this feature is its relatively high computational complexity of  $O(n\log n)$ .

The difference in RMSE values between the best and worst performing features in Table 5-79 is 0.55 MET reflects the difficulty of predicting energy expenditure over the MIT energy expenditure dataset. This dataset contains data for 51 activities collected at a gym and at a residential home. Furthermore, some of these activities include different intensity levels due to both changes in the speed of execution of the activities and changes in resistance level or work load. The dataset also contains examples of rapid changes in energy expenditure such as when someone walks upstairs (rapid increase in energy expenditure) for one minute and later rests for another minute (decreases in energy expenditure) repeated continuously up to three times in a row.

Moreover, unlabeled periods of time (*unknown* activity) when subjects rest after exhaustive exercise sessions were not eliminated from the dataset. Finally, periods of

time where energy expenditure does not reach steady state were also not eliminated from the dataset. From Table 5-79, a more reliable indicator of the performance of individual features appears to be the correlation coefficient since the difference in this value for the best and worst performing activities is 0.59. The difference between the best and worst possible correlation coefficients is one.

Interestingly, the ACFFTCoeff feature expected to have a poor performance due to its large size (889); nonetheless, it is the second best performing feature. This is because the linear regression algorithm used performs feature selection using the M5 method and therefore was able to successfully find a subset of the 889 features that provided good performance. As observed by inspecting the variables included in the regression model learned. Table 5-79 also shows that the inter-quartile range feature (ACIQR) is the third best performing feature. This feature achieves a better performance than the ACTotalAbsArea and the ACAbsArea features, the two features most used to predict energy expenditure in prior work. The problem with the ACIQR feature is its high computational complexity since all acceleration values inside a given window need to be ordered in ascending order before this feature can be computed  $(O(n^2))$  if bubble, insertion, selection, and shell sorts are utilized and  $O(n \log n)$  if quick sort is utilized). Clearly, the improvement in performance over the ACAbsArea feature of only 0.02MET does not justify the extra computation incurred. The performance of the ACAbsArea and ACTotalAbsArea features is similar because they both capture similar information on overall body motion. It seems that computing body motion per limb (ACAbsArea) does not improve over computing overall body motion (ACTotalAbsArea), perhaps because the regression coefficients need to be optimized for all activities at once. The ACAbsArea feature might provide an advantage if different regression models are used to estimate energy expenditure over different sets of activities. When the performance of features capturing motion energy is compared (ACEnergy, ACBandEnergy, ACModVigEnergy, ACLowEnergy), it is found that the best performing features are ACModVigEnergy and ACBandEnergy. This is likely because most activities in the MIT energy expenditure dataset include moderate to vigorous motion that may be better captured by the ACModVigEnergy feature since it computes energy over the frequency range for these motion intensities (0.71 - 10 Hz). Similarly, the *ACBandEnergy* feature computes energy over a frequency range that overlaps with the range of moderate and vigorous motion (0.3-3.5Hz).

Figure 5-52 presents the root mean squared as a gray scale image that highlights the difference in performance among activity categories. From this figure, it can be seen that the best performing features achieve the lowest RMSE error for the postures and household activity categories (white areas located at the top of Figure 5-52). This is because these activities have low levels of energy expenditure as compared to the other activity categories (ambulation, exercise and resistance exercise activities). It is true that some household activities involve some levels of effort (e.g. *scrubbing a surface*) or work load (e.g. *carrying groceries*) but the energy expenditure associated with these activities is low compared to the energy expenditure values found during exercise or resistance exercise activities performed at the gym. As a result, the RMSE for household activities is lower than the one found in the exercise and resistance exercise activities categories. The high RMSE error associated with exercise and resistance exercise activities can be observed in Figure 5-52 as darker regions for these categories, particularly for the

Information cantured by the features	Features						
mormation captured by the leatures	Features						
Measures of body posture	DCPostureDist, DCArea, DCMean, and DCTotalMean.						
Measures of motion shape	ACTotalAbsArea, ACAbsArea, ACAbsMean, ACO3, ACO1, ACO2,						
L	ACTotalSVM, ACEntropy, ACSkew, and ACKur.						
Measures of motion variation	ACIQR, ACRange, ACVar, and ACAbsCV,						
Measures of motion spectral content	ACFFTPeaks, ACFFTCoeff, and FWTCoeff.						
Measures of motion energy	ACModVigEnergy, ACBandEnergy, ACLowEnergy, and ACEnergy.						
Measures of motion periodicity	ACMCR, ACPitch, and ACDomFreqRatio.						
Measures of motion similarity across body	ACCorr						
segments							
Measures of force employed per body segment	ACSF and ACTotalSF.						

Table 5-80: Ordering of the features according to their individual performance and computational requirements (decreasing order of usefulness from left to right) clustered based on the information they attempt to capture from the accelerometer signals.

Features subsets	Correlation	All	Postures	Ambulation	Exercise	Resistance	Household
		Activities				Exercise	
All Features:	$0.74 \pm 0.10$	$1.24\pm0.30$	0.7±0.3	1.2±0.5	1.2±0.7	1.1±0.6	0.7±0.3
MaxAcceleration		$(0.91 \pm 0.20)$	(0.6±0.3)	(1.0±0.5)	(1.1±0.7)	(1.0±0.6)	(0.6±0.3)
Fast to compute:	$0.72\pm0.11$	$1.27\pm0.27$	0.7±0.4	1.2±0.5	1.2±0.7	1.1±0.6	0.8±0.5
ACAbsArea,		$(0.94 \pm 0.19)$	$(0.7\pm0.4)$	(1.1±0.5)	(1.1±0.7)	(1.0±0.6)	$(0.7\pm0.5)$
DCArea, ACVar,							
ACRange, ACMCR							
Invariant reduced	$0.72\pm0.11$	$1.28\pm0.29$	0.7±0.3	1.2±0.5	1.3±0.8	1.2±0.6	$0.8\pm0.4$
DCPostureDist,		$(0.95 \pm 0.19)$	$(0.6\pm0.3)$	(1.1±0.5)	$(1.2\pm0.8)$	(1.1±0.6)	$(0.7\pm0.4)$
ACVar,							
ACBandEnergy,							
ACFFTPeaks,							
S1:	$0.73 \pm 0.07$	$1.27 \pm 0.29$	$0.6\pm0.2$	$1.2\pm0.5$	$1.4\pm0.8$	$1.2\pm0.6$	$0.7\pm0.2$
ACFFTPeaks.		$(0.92 \pm 0.17)$	$(0.5\pm0.2)$	$(1.1\pm0.5)$	$(1.3\pm0.8)$	$(1.1\pm0.6)$	$(0.6\pm0.2)$
ACAbsArea							
S2:	$0.75 \pm 0.06$	$1.23 \pm 0.29$	$0.7\pm0.3$	$1.1\pm0.5$	1.3±0.7	1.1±0.6	$0.7\pm0.2$
ACFFTPeaks.		$(0.89 \pm 0.18)$	$(0.6\pm 0.3)$	$(1.0\pm0.5)$	$(1.2\pm0.7)$	$(1.0\pm0.6)$	$(0.6\pm0.2)$
ACEntropy							
ACMCR							
ACModVigEnergy							
S3:	$0.74 \pm 0.06$	$1.24 \pm 0.28$	0.7±0.3	$1.2\pm0.5$	$1.3\pm0.8$	1.1±0.6	$0.7 \pm 0.2$
ACFFTPeaks.		$(0.89 \pm 0.17)$	$(0.6\pm0.3)$	$(1.0\pm0.5)$	$(1.2\pm0.7)$	$(1.0\pm0.6)$	$(0.6\pm0.2)$
ACMCR							
ACModVigEnergy							
ACFFTPeaks	$0.72 \pm 0.07$	$1.28 \pm 0.30$	$0.6\pm0.2$	$1.2\pm0.5$	$1.4\pm0.8$	$1.2\pm0.7$	$0.7\pm0.2$
		$(0.93 \pm 0.17)$	$(0.6\pm0.2)$	$(1.1\pm0.5)$	(1.3±0.8)	$(1.1\pm0.7)$	$(0.6\pm0.2)$
ACAbsArea	$0.68\pm0.06$	$1.36\pm0.30$	$0.7 \pm 0.2$	$1.1\pm0.4$	$1.5\pm0.7$	$1.4\pm0.6$	$0.7 \pm 0.2$
		$(1.0 \pm 0.2)$	$(0.7\pm0.2)$	$(1.0\pm0.4)$	$(1.4\pm0.7)$	(1.3±0.6)	$(0.6\pm 0.2)$

Table 5-81: Root mean squared error and mean absolute error (shown in parenthesis) for the six subsets of accelerometer-based features with highest performance found when estimating energy expenditure in a subject independent manner using linear regression over the MIT dataset. Features are computed per sensor over windows of 5.6s in length.

features located at the bottom of the figure (poor performing features). Figure 5-52 highlights that the ACPitch feature has a very high RMSE error for the postures category. This is expected since this feature captures the periodicity of motion (fundamental frequency) and motion is minimal for sedentary activities.

Once the performance over individual features was found, they were ordered in decreasing order of usefulness by taking into account their computational requirements. Table 5-80 presents the final ordering of the features according to their individual performance (from Table 5-79) and computational requirements clustered based on the information they attempt to capture. The ordering of features presented in this table was

then used to measure the performance over different combinations of features. These features are more invariant to the magnitude of the accelerometer signal than other features and showed high discrimination during activity recognition tasks. The *fast to compute* feature set also includes high discriminant features for activity recognition but that are fast to compute such as the *ACAbsArea*, *DCArea*, *ACVar*, *ACRange*, and *ACMCR*. Table 5-81 also presents the performance for a set of features labeled as *MaxAcceleration* set which contains all the accelerometer-based features.

This set provides a baseline for the best performance that can be obtained when all the accelerometer-based features are used to estimate energy expenditure. Performance per activity for these feature sets is presented in Appendix B13.

Table 5-81 presents the feature sets with highest performance found when estimating energy expenditure in a subject independent manner over the MIT energy expenditure dataset. The feature sets labeled as *fast to compute* and *invariant reduced* correspond to two of the sets of features found with higher performance during the activity recognition experiments in Section 5.4.7. As a reminder to the reader, the *invariant reduced* feature set consists on the *DCPostureDist*, *ACVar*, *ACBandEnergy*, *ACFFTPeaks* features.

From Table 5-81, we can observe that the feature set with highest correlation coefficient (0.75) and lowest RMSE is the one containing the features ACFFTPeaks, ACMCR, and ACModVigEnergy. This set is the best performing because the features ACFFTPeaks and ACMCR capture the frequencies at which body segments are being moved. Intuitively, it makes sense that higher energy expenditure will be associated with rapid or high frequency motion. The ACModVigEnergy feature captures the energy associated with the body limb movement over the moderate and vigorous motion levels. As explained before, most activities contained in the MIT dataset involve either moderate or vigorous motion, so this feature outperforms the other features measuring motion energy (ACBandEnergy, ACLowEnergy, and ACEnergy). Overall, the performance using all six subsets of features is similar and very close. For example, the difference between the best and worse correlation coefficient is just 0.03 and the difference between the best and worse RMSE is only 0.05 MET for the feature combinations. One possible explanation is that the unavailability of enough training data due to the utilization of a large number of features (since they are computed over all seven sensors) prevents the differences from being larger.

To better understand the difference in performance among these feature sets, the RMSE per activity was analyzed. Figure 5-53 shows RMSE as a gray scale image scaled to highlight the difference in performance per activity. The image shows the lowest RMSE of 0.2 MET in white and the largest of 2.2 MET in black. From this gray scale image, it can be observed that as expected, the feature set with lowest RMSE per activity is the *MaxAcceleration* set. The second subset with lowest RMSE per activity is the set labeled as S3 including the *ACFFTPeaks*, *ACMCR*, and *ACModVigEnergy* features. This is because the fast to compute feature shows slightly higher RMSE for *walking around block*, *gardening* and *weeding* perhaps because it does not contain the *ACFFTPeaks* feature set shows a slightly higher RMSE for *running at 6mph* than the S3 feature set. The fact that the S3 feature set is the highest performing one is good news since it includes only features that are relatively invariant to the magnitude of the accelerometer signal. Nevertheless, this feature set improves the correlation coefficient only 0.02 units



Figure 5-53: Root mean squared error per activity as a gray scale image when multivariable linear regression is used to estimate energy expenditure over several subset of accelerometer-based features computed per sensor over windows of 5.6s in length. The image is scaled to show the lowest RMSE error of 0.2MET in white and the largest of 2.2MET in black. Subset labeled as S1 corresponds to the *ACFFTPeaks* and *ACAbsArea* features, S2 corresponds to the *ACFFTPeaks*, *ACEntropy*, *ACMRC*, and *ACModVigEnergy*, and the S3 to the *ACFFTPeaks*, *ACMRC*, and *ACModVigEnergy* features.

and overall RMSE only 0.04MET over the performance obtained using the *ACFFTPeaks* feature alone.

In conclusion, the best single feature with best performance for estimating energy expenditure is the *ACFFTPeaks* feature. This is because this feature captures the frequency of motion of the body segments, and intuitively, the more rapid the motion (higher frequency), the higher energy expenditure will be. This feature improves the correlation coefficient +0.04 units and overall RMSE 0.08 MET over the *ACAbsArea* feature, which is the feature normally used to predict energy expenditure in prior work. However, the main advantage of the *ACFFTPeaks* feature over the *ACAbsArea* is that is

more invariant to the magnitude of the accelerometer signal. Unfortunately, this feature is not as computationally efficient as one would hope since it computational complexity is O(nlogn). Nevertheless, the utilization of only the five FFT coefficients with larger magnitude (peaks) per accelerometer signal effectively reduces the number of features (or predictor variables) from 889 to only 70 when all seven sensors and windows of 256 samples in length are used. Given the small differences found for the highest performing feature combinations, it is necessary to explore the performance of these feature sets over subsets of accelerometers to better understand their importance. Consequently, the next section explores the performance of the five highest performing feature sets over subsets of accelerometers.

### 5.6.8 How Well can Energy Expenditure Be Estimated by Computing the Highest Performing Features Over Subsets of Accelerometers?

The previous section identified the best performing subsets of accelerometer-based features when they were computed over all seven accelerometers. However; in practice, wearing so many accelerometers would be difficult and intrusive. Consequently, it is necessary to explore the performance of these features when the number of accelerometers is reduced. This section explores this question by evaluating five of the highest performing sets of accelerometer-based features found in the previous section in a subject independent manner over eleven subsets of accelerometers. The five feature sets explored in this section are (1) *ACAbsArea*, (2) *ACFFTPeaks*, (3) *ACFFTPeaks* + *ACModVigEnergy* + *ACMCR*, (4) the *fast to compute* feature set, and (5) the *invariant reduced* feature sets. For a detailed description of the features included in the *fast to compute* and *invariant reduced* feature sets refer to Section 5.4.7. Energy expenditure is estimated by utilizing a single multivariable linear regression model trained using features compute per sensor over windows of 5.6s in length over the different sensor combinations.

Table 5-82 through Table 5-86 present the results obtained using the five feature sets computed over eleven sensor combinations. From these tables, it can be seen that in general, feature sets containing multiple features other than the ACAbsArea feature improve performance, particularly when single sensors are used to estimate energy expenditure. For example, the performance of the ACFFTPeaks + ACModVigEnergy + ACMCR feature set over different subsets of accelerometers is higher than the performance obtained using single features such as the ACAbsArea and the ACFFTPeaks. For instance, the ACFFTPeaks + ACModVigEnergy + ACMCR feature set achieves a correlation coefficient of 0.70 and a RMSE of 1.34MET over the hip sensor, while the ACFFTPeaks feature alone only achieves a correlation coefficient of 0.49 and a RMSE of 1.62MET. Similarly, the ACAbsArea feature alone achieves a correlation coefficient of 0.61 and a RMSE of 1.62MET over the hip sensor. The performance of the ACFFTPeaks + ACModVigEnergy + ACMCR feature set is also higher over the other single accelerometers (DWrist, DFoot, DThigh, and DUpperArm) than when the ACFFTPeaks or ACAbsArea features are used alone. This indicates that computing features that capture additional information other than overall amount of motion usually captured by the ACAbsArea feature (often used by the medical community) indeed improves energy

Sensor	Correlation	All	Postures	Ambulation	Exercise	Resistance	Household
Combination						Exercise	
All sensors	$0.68\pm0.06$	$1.36\pm0.30$	0.7±0.2	1.1±0.4	1.5±0.7	1.4±0.6	0.7±0.2
		$(1.0 \pm 0.2)$	(0.7±0.2)	(1.0±0.4)	(1.4±0.7)	(1.3±0.6)	(0.6±0.2)
Hip + DWrist +	$0.67\pm0.05$	$1.38\pm0.29$	0.8±0.2	1.1±0.4	1.5±0.7	1.4±0.6	0.8±0.3
DFoot		$(1.04 \pm 0.15)$	(0.7±0.2)	(1.0±0.4)	(1.4±0.7)	(1.3±0.6)	(0.7±0.2)
Hip + DWrist	$0.61\pm0.07$	$1.49 \pm 0.33$	1.0±0.2	1.1±0.5	1.7±0.8	1.4±0.6	0.8±0.2
-		$(1.16 \pm 0.19)$	(1.0±0.2)	(1.0±0.5)	(1.6±0.7)	(1.4±0.6)	(0.8±0.2)
Hip + DFoot	$0.64\pm0.08$	$1.43 \pm 0.33$	1.0±0.2	1.1±0.4	1.5±0.7	1.3±0.6	0.7±0.2
-		$(1.08 \pm 0.17)$	(0.9±0.2)	(1.0±0.4)	(1.4±0.7)	(1.2±0.6)	(0.7±0.2)
DWrist + DThigh	$0.65\pm0.08$	$1.42 \pm 0.32$	0.9±0.3	1.1±0.4	1.5±0.7	1.4±0.6	0.8±0.2
_		$(1.09 \pm 0.17)$	(0.9±0.3)	(1.0±0.4)	(1.5±0.7)	(1.3±0.6)	(0.7±0.2)
DWrist + DFoot	$0.67\pm0.05$	$1.38\pm0.29$	0.8±0.2	1.1±0.4	1.5±0.7	1.4±0.6	0.8±0.3
		$(1.04 \pm 0.15)$	(0.7±0.2)	(1.0±0.4)	(1.4±0.7)	(1.3±0.6)	(0.7±0.2)
Hip	$0.61\pm0.08$	$1.51 \pm 0.34$	1.1±0.2	1.1±0.5	1.6±0.8	1.4±0.6	0.8±0.2
		$(1.16 \pm 0.20)$	(1.1±0.2)	(1.0±0.5)	(1.5±0.8)	(1.3±0.6)	(0.7±0.2)
DWrist	$0.45\pm0.15$	$1.66\pm0.30$	1.2±0.3	1.4±0.5	2.0±0.8	1.8±0.7	1.2±0.3
		$(1.37 \pm 0.20)$	(1.2±0.3)	(1.3±0.5)	(1.9±0.8)	(1.8±0.7)	(1.1±0.3)
DFoot	$0.63\pm0.08$	$1.45 \pm 0.33$	1.0±0.2	1.2±0.5	1.5±0.7	1.3±0.6	0.8±0.2
		$(1.10 \pm 0.17)$	(1.0±0.2)	(1.1±0.5)	(1.4±0.7)	(1.2±0.6)	(0.7±0.2)
DUpperArm	$0.54 \pm 0.11$	$1.56 \pm 0.31$	1.0±0.3	1.1±0.5	1.8±0.8	1.6±0.6	1.1±0.4
		$(1.25 \pm 0.19)$	(0.9±0.3)	(1.0±0.5)	(1.7±0.8)	(1.5±0.6)	(1.0±0.3)
DThigh	$0.63 \pm 0.09$	$1.44 \pm 0.34$	1.1±0.3	1.1±0.4	1.5±0.7	1.4±0.6	0.7±0.2
		$(1.11 \pm 0.19)$	(1.1±0.3)	$(1.0\pm0.4)$	(1.4±0.7)	(1.3±0.6)	(0.7±0.2)

Table 5-82: Root mean squared error and mean absolute error (shown in parenthesis) obtained when estimating energy expenditure in a subject independent manner using linear regression and the *ACAbsArea* feature computed per sensor using windows of 5.6s in length over different subsets of accelerometers. Energy expenditure is estimated for the 51 activities contained in the MIT energy expenditure dataset.

Sensor	Correlation	All	Postures	Ambulation	Exercise	Resistance	Household
Combination						Exercise	
All sensors	$0.72\pm0.07$	$1.28\pm0.30$	0.6±0.2	1.2±0.5	$1.4\pm0.8$	1.2±0.7	0.7±0.2
		$(0.93 \pm 0.17)$	(0.6±0.2)	(1.1±0.5)	(1.3±0.8)	(1.1±0.7)	(0.6±0.2)
Hip + DWrist +	$0.72\pm0.06$	$1.28\pm0.31$	0.6±0.2	1.1±0.5	1.4±0.8	1.2±0.6	0.7±0.3
DFoot		$(0.93 \pm 0.18)$	(0.5±0.2)	(1.0±0.5)	(1.3±0.8)	(1.1±0.6)	(0.6±0.2)
Hip + DWrist	$0.51\pm0.09$	$1.59\pm0.31$	1.2±0.2	1.3±0.5	1.9±0.8	1.7±0.6	0.9±0.3
		$(1.28 \pm 0.20)$	(1.1±0.3)	(1.2±0.5)	(1.8±0.8)	(1.6±0.6)	(0.9±0.3)
Hip + DFoot	$0.71\pm0.06$	$1.29\pm0.30$	0.6±0.2	1.1±0.5	1.4±0.8	1.2±0.6	0.7±0.2
		$(0.94 \pm 0.18)$	(0.6±0.2)	(1.0±0.5)	(1.3±0.8)	(1.1±0.6)	(0.6±0.2)
DWrist + DThigh	$0.66\pm0.06$	$1.39\pm0.29$	0.7±0.2	1.1±0.4	1.5±0.8	$1.4\pm0.7$	0.8±0.3
		$(1.02 \pm 0.15)$	(0.6±0.2)	(1.0±0.4)	(1.4±0.7)	(1.3±0.7)	(0.7±0.2)
DWrist + DFoot	$0.71\pm0.07$	$1.29\pm0.31$	0.6±0.2	1.1±0.4	1.4±0.8	1.3±0.7	0.7±0.3
		$(0.93 \pm 0.17)$	(0.5±0.2)	(1.0±0.4)	(1.3±0.8)	(1.1±0.7)	(0.6±0.2)
Hip	$0.49\pm0.10$	$1.62\pm0.32$	1.3±0.2	1.3±0.5	1.9±0.8	1.7±0.6	0.9±0.2
		$(1.31 \pm 0.22)$	(1.2±0.2)	(1.2±0.5)	(1.9±0.8)	(1.6±0.6)	(0.8±0.2)
DWrist	$0.44 \pm 0.15$	$1.66\pm0.29$	1.2±0.3	1.4±0.5	2.0±0.8	1.9±0.7	1.1±0.3
		$(1.36 \pm 0.20)$	(1.2±0.3)	(1.3±0.5)	(1.9±0.8)	(1.8±0.7)	(1.0±0.3)
DFoot	$0.70\pm0.07$	$1.31\pm0.30$	0.7±0.2	1.1±0.5	$1.4\pm0.8$	1.3±0.7	0.7±0.3
		$(0.96 \pm 0.17)$	(0.6±0.2)	(1.0±0.5)	(1.3±0.8)	(1.2±0.7)	(0.6±0.2)
DUpperArm	$0.66 \pm 0.08$	$1.41 \pm 0.34$	0.8±0.3	1.1±0.4	1.5±0.7	1.3±0.6	0.7±0.2
		$(1.05 \pm 0.20)$	(0.8±0.3)	(1.0±0.4)	(1.4±0.7)	(1.2±0.6)	(0.6±0.2)
DThigh	$0.61\pm0.09$	$1.49\pm0.35$	1.0±0.2	1.2±0.5	1.6±0.8	1.3±0.6	0.8±0.2
		$(1.11 \pm 0.18)$	(0.9±0.2)	(1.1±0.5)	(1.5±0.8)	(1.2±0.6)	(0.7±0.2)

Table 5-83: Root mean squared error and mean absolute error (shown in parenthesis) obtained when estimating energy expenditure in a subject independent manner using linear regression and the *ACFFTPeaks* feature computed per sensor using windows of 5.6s in length over different subsets of accelerometers. Energy expenditure is estimated for the 51 activities contained in the MIT energy expenditure dataset.

Sensor	Correlation	All	Postures	Ambulation	Exercise	Resistance	Household
Combination						Exercise	
All sensors	$0.74\pm0.06$	$1.24\pm0.28$	0.7±0.3	1.2±0.5	1.3±0.8	1.1±0.6	0.7±0.2
		$(0.89 \pm 0.17)$	(0.6±0.3)	(1.0±0.5)	(1.2±0.7)	(1.0±0.6)	(0.6±0.2)
Hip + DWrist +	$0.73\pm0.05$	$1.25\pm0.29$	0.5±0.2	1.1±0.4	1.3±0.7	1.2±0.6	0.7±0.2
DFoot		$(0.91 \pm 0.18)$	(0.5±0.2)	(1.0±0.4)	(1.2±0.7)	(1.1±0.6)	(0.6±0.2)
Hip + DWrist	$0.70\pm0.06$	$1.33\pm0.29$	0.6±0.2	1.2±0.5	1.4±0.7	1.3±0.6	0.8±0.3
		$(1.01 \pm 0.20)$	(0.5±0.2)	(1.1±0.5)	(1.3±0.7)	(1.2±0.6)	(0.7±0.2)
Hip + DFoot	$0.73\pm0.05$	$1.25\pm0.28$	0.6±0.2	1.2±0.4	1.3±0.7	1.2±0.6	0.7±0.2
-		$(0.91 \pm 0.18)$	(0.5±0.2)	(1.0±0.4)	(1.2±0.7)	(1.1±0.6)	(0.6±0.2)
DWrist + DThigh	$0.71\pm0.06$	$1.32\pm0.29$	0.7±0.4	1.1±0.4	$1.4\pm0.8$	1.2±0.7	0.7±0.2
		$(0.96 \pm 0.17)$	(0.7±0.4)	(1.0±0.4)	(1.3±0.7)	(1.1±0.6)	(0.6±0.2)
DWrist + DFoot	$0.72\pm0.07$	$1.28\pm0.30$	0.6±0.2	1.1±0.4	1.4±0.8	1.2±0.7	0.7±0.2
		$(0.92 \pm 0.17)$	(0.5±0.2)	(1.0±0.4)	(1.2±0.8)	(1.1±0.6)	(0.6±0.2)
Hip	$0.70\pm0.06$	$1.34\pm0.29$	0.6±0.2	1.3±0.5	1.4±0.7	1.3±0.6	0.8±0.2
		$(1.02 \pm 0.20)$	(0.6±0.2)	(1.2±0.5)	(1.3±0.7)	(1.1±0.6)	(0.7±0.2)
DWrist	$0.53\pm0.10$	$1.58\pm0.31$	1.0±0.3	1.3±0.4	1.9±0.8	1.7±0.7	1.1±0.3
		$(1.26 \pm 0.20)$	(0.9±0.3)	(1.2±0.4)	(1.8±0.8)	(1.6±0.7)	(1.0±0.3)
DFoot	$0.71\pm0.06$	$1.29\pm0.29$	0.6±0.2	1.1±0.4	$1.4\pm0.8$	1.2±0.7	0.7±0.2
		$(0.94 \pm 0.17)$	(0.6±0.2)	(1.0±0.4)	(1.3±0.8)	(1.1±0.7)	(0.6±0.2)
DUpperArm	$0.68 \pm 0.08$	$1.36 \pm 0.34$	0.6±0.2	1.1±0.4	1.4±0.7	1.3±0.6	0.8±0.2
		$(1.02 \pm 0.19)$	(0.6±0.3)	(1.0±0.4)	(1.3±0.7)	(1.2±0.6)	(0.7±0.2)
DThigh	$0.66\pm0.08$	$1.40 \pm 0.33$	0.9±0.4	1.2±0.5	1.4±0.7	1.2±0.6	0.7±0.2
		$(1.03 \pm 0.19)$	(0.9+0.4)	(1.1+0.5)	(1.3+0.7)	(1.1+0.6)	(0.6+0.2)

Table 5-84: Root mean squared error and mean absolute error (shown in parenthesis) obtained when estimating energy expenditure in a subject independent manner using linear regression and the ACFFTPeaks + ACModVigEnergy + ACMCR features computed per sensor using windows of 5.6s in length over different subsets of accelerometers. Energy expenditure is estimated for the 51 activities contained in the MIT energy expenditure dataset.

Sensor	Correlation	All	Postures	Ambulation	Exercise	Resistance	Household
Combination						Exercise	
All sensors	$0.72\pm0.11$	$1.27\pm0.27$	0.7±0.4	1.2±0.5	$1.2\pm0.7$	1.1±0.6	0.8±0.5
		$(0.94 \pm 0.19)$	(0.7±0.4)	(1.1±0.5)	(1.1±0.7)	(1.0±0.6)	(0.7±0.5)
Hip + DWrist +	$0.69\pm0.12$	$1.33\pm0.33$	0.6±0.3	1.2±0.5	1.3±0.7	1.2±0.6	0.9±0.6
DFoot		$(0.99 \pm 0.24)$	(0.5±0.3)	(1.1±0.5)	(1.2±0.7)	(1.1±0.6)	(0.8±0.6)
Hip + DWrist	$0.69\pm0.06$	$1.37\pm0.28$	0.6±0.2	1.2±0.5	1.3±0.7	1.2±0.6	0.9±0.3
-		$(1.04 \pm 0.18)$	(0.5±0.2)	(1.1±0.5)	(1.3±0.6)	(1.1±0.6)	(0.8±0.3)
Hip + DFoot	$0.66 \pm 0.13$	$1.38\pm0.35$	0.7±0.3	1.3±0.6	1.3±0.7	1.1±0.6	0.9±0.6
-		$(1.03 \pm 0.24)$	(0.7±0.3)	(1.2±0.6)	(1.2±0.7)	(1.0±0.6)	(0.8±0.6)
DWrist + DThigh	$0.72\pm0.06$	$1.28\pm0.25$	0.9±0.5	1.1±0.4	1.2±0.6	1.1±0.6	0.7±0.3
_		$(0.95 \pm 0.15)$	(0.8±0.5)	(1.0±0.4)	(1.1±0.6)	(1.0±0.5)	(0.6±0.3)
DWrist + DFoot	$0.69 \pm 0.10$	$1.33 \pm 0.33$	0.7±0.3	1.1±0.5	1.3±0.7	1.2±0.7	0.9±0.6
		$(0.99 \pm 0.23)$	(0.6±0.3)	(1.0±0.5)	(1.2±0.7)	(1.2±0.6)	(0.8±0.6)
Hip	$0.67\pm0.07$	$1.40 \pm 0.31$	0.7±0.2	1.3±0.5	1.4±0.7	1.2±0.6	0.8±0.2
-		$(1.06 \pm 0.19)$	(0.7±0.2)	(1.2±0.5)	(1.3±0.6)	(1.1±0.5)	(0.7±0.2)
DWrist	$0.49 \pm 0.13$	$1.63 \pm 0.31$	1.1±0.4	1.3±0.5	2.0±0.8	1.8±0.7	1.2±0.4
		$(1.32 \pm 0.20)$	(1.1±0.4)	(1.2±0.5)	(1.9±0.8)	(1.7±0.7)	(1.1±0.4)
DFoot	$0.63\pm0.10$	$1.42\pm0.35$	0.9±0.3	1.3±0.5	1.4±0.7	1.2±0.6	0.9±0.4
		$(1.07 \pm 0.20)$	(0.9±0.3)	(1.2±0.5)	(1.3±0.7)	(1.1±0.6)	(0.8±0.4)
DUpperArm	$0.60\pm0.07$	$1.48\pm0.31$	0.7±0.3	1.1±0.4	1.6±0.8	1.5±0.7	1.1±0.3
		$(1.14 \pm 0.19)$	(0.7±0.3)	(1.0±0.4)	(1.5±0.8)	(1.4±0.7)	(1.0±0.3)
DThigh	$0.69\pm0.09$	$1.32\pm0.28$	1.0±0.5	1.1±0.4	1.2±0.6	1.1±0.5	0.7±0.3
-		$(0.99 \pm 0.15)$	$(1.0\pm0.5)$	$(1.0\pm0.4)$	$(1.1\pm0.6)$	$(1.0\pm0.5)$	$(0.6\pm0.2)$

Table 5-85: Root mean squared error and mean absolute error (shown in parenthesis) obtained when estimating energy expenditure in a subject independent manner using linear regression and the *Fast to compute* feature set computed per sensor using windows of 5.6s in length over different subsets of accelerometers. Energy expenditure is estimated for the 51 activities contained in the MIT energy expenditure dataset.

Sensor Combination	Correlation	All	Postures	Ambulation	Exercise	Resistance	Household
						Exercise	
All sensors	$0.72\pm0.11$	$1.28\pm0.29$	0.7±0.3	1.2±0.5	1.3±0.8	1.2±0.6	0.8±0.4
		$(0.95 \pm 0.19)$	(0.6±0.3)	(1.1±0.5)	(1.2±0.8)	(1.1±0.6)	$(0.7\pm0.4)$
Hip + DWrist + DFoot	$0.70\pm0.11$	$1.31 \pm 0.34$	0.6±0.3	1.2±0.5	1.4±0.8	1.2±0.7	0.8±0.5
		$(0.97 \pm 0.23)$	(0.5±0.3)	(1.0±0.5)	(1.3±0.8)	(1.1±0.6)	(0.7±0.5)
Hip + DWrist	$0.58\pm0.08$	$1.53\pm0.28$	1.0±0.4	1.3±0.4	$1.8\pm0.8$	1.5±0.6	1.0±0.3
		$(1.20 \pm 0.18)$	(1.0±0.4)	(1.1±0.5)	(1.7±0.8)	(1.5±0.6)	(0.9±0.3)
Hip + DFoot	$0.71 \pm 0.11$	$1.29\pm0.32$	0.6±0.2	1.2±0.5	1.4±0.8	1.2±0.6	0.8±0.5
		$(0.96 \pm 0.22)$	(0.5±0.2)	(1.0±0.5)	(1.3±0.7)	(1.1±0.6)	(0.7±0.5)
DWrist + DThigh	$0.68\pm0.05$	$1.35\pm0.29$	0.8±0.3	1.1±0.4	1.5±0.7	1.3±0.6	0.7±0.3
		$(1.00 \pm 0.17)$	(0.7±0.3)	(1.0±0.4)	(1.4±0.7)	(1.2±0.6)	(0.6±0.3)
DWrist + DFoot	$0.69\pm0.12$	$1.32\pm0.36$	0.6±0.3	1.1±0.5	1.4±0.8	1.3±0.7	0.8±0.5
		$(0.97 \pm 0.23)$	(0.5±0.3)	(1.0±0.5)	(1.3±0.8)	(1.2±0.7)	(0.7±0.5)
Hip	$0.55\pm0.09$	$1.55\pm0.28$	1.0±0.3	1.3±0.5	$1.8\pm0.8$	1.6±0.6	1.0±0.3
		$(1.24 \pm 0.18)$	(1.0±0.3)	(1.2±0.5)	(1.7±0.8)	(1.5±0.6)	(0.9±0.3)
DWrist	$0.48\pm0.15$	$1.62\pm0.32$	1.2±0.3	1.2±0.5	1.9±0.8	1.7±0.7	1.0±0.3
		$(1.30 \pm 0.21)$	(1.2±0.3)	(1.2±0.5)	(1.9±0.8)	(1.7±0.7)	(1.0±0.3)
DFoot	$0.69\pm0.11$	$1.32\pm0.34$	0.7±0.3	1.1±0.5	1.4±0.8	1.3±0.7	0.8±0.4
		$(0.97 \pm 0.21)$	(0.6±0.3)	(1.0±0.5)	(1.3±0.8)	(1.2±0.7)	(0.7±0.4)
DUpperArm	$0.66\pm0.08$	$1.39\pm0.33$	0.7±0.2	1.1±0.5	1.5±0.8	1.3±0.6	0.8±0.2
		$(1.04 \pm 0.19)$	(0.7±0.2)	(1.0±0.5)	(1.4±0.7)	(1.2±0.6)	(0.7±0.2)
DThigh	$0.64\pm0.09$	$1.42 \pm 0.33$	0.9±0.3	1.2±0.5	1.5±0.7	1.3±0.6	0.8±0.3
		$(1.07 \pm 0.18)$	(0.9+0.3)	(1.1+0.5)	(14+07)	(1.2+0.6)	(0.7+0.3)

Table 5-86: Root mean squared error and mean absolute error (shown in parenthesis) obtained when estimating energy expenditure in a subject independent manner using linear regression and the *Invariant reduced* feature set computed per sensor using windows of 5.6s in length over different subsets of accelerometers. Energy expenditure is estimated for the 51 activities contained in the MIT energy expenditure dataset.

expenditure estimation. This effect was perhaps not observed in the previous section because of the relatively limited amount of training data available with respect to the large number of features being computed (since features were computed over all seven accelerometers). This effect produced by computing a large number of features when training data is limited can also be observed by comparing Table 5-86 and Table 5-84. These tables show that the overall performance of the *invariant reduced* feature set is slightly lower than the performance of the ACFFTPeaks + ACModVigEnergy + ACMCR feature set. This is because the invariant reduced feature set contains more features (ACFFTPeaks, ACBandEnergy, ACVar, and DCPostureDist) and consequently, the training data available to train the regression model (containing more variables) is not enough. The result that multiple features achieve higher performance over single sensors and combinations of sensors than single features (e.g. ACAbsArea) confirms recent findings by Rothney [152, 181]. This work also found that extracting multiple features such as the coefficient of variation, inter-quartile range, and power spectral density improves the performance in energy expenditure estimation when a single accelerometer at the hip is used. The work presented in this section goes a step further by showing, as one might expect, that multiple features also improve overall performance and performance per activity when multiple sensors are used provided there is enough training data to train the regression models. In this paragraph, 'multiple features' refer to features other than the ACAbsArea feature that capture motion information other than overall amount of motion that helps in estimating energy expenditure better.

From Table 5-82 through Table 5-86 it can also be observed that overall, the sensor combinations with higher performance in decreasing order are All Sensors, Hip + DWrist

+ DFoot, and DWrist + DFoot. As explained in previous sections, the Hip + DWrist + DFoot sensor combination is able to capture upper body, lower body, and overall body motion well enough to achieve a performance similar to the one obtained using all seven accelerometers. Interestingly, the sensor combination DWrist + DFoot also achieves a performance that is very close to the one obtained using all the seven sensors. For example, the difference in the correlation coefficient between all sensors and the DWrist + DFoot sensor combination is between 0.01 and 0.02 for all tables. The difference for the RMSE is between 0.01 and 0.04MET. One possible explanation for the good performance of this sensor combination is that it effectively detects upper body and lower body motion since one sensor at the wrist and another at the foot are used. At the same time, overall body motion due to activities such as ambulation is also picked up at these two sensors locations. This is an important result because it can be argued that sensors at the wrist and at the foot are easier to wear than sensors at the hip, particularly for women (e.g. women wearing dresses). The accelerometer at the wrist can also be embedded in devices already worn at this location such as wrist watches and the sensor at the foot in shoe pods. The sensor combinations Hip + DWrist, Hip + DFoot, and DWrist + DThigh also achieve a competitive performance (with respect to the all sensors combination) although their ordering with respect to performance depends on the features being computed. The result obtained in this section are in agreement with prior work by Melanson and Freedson [233] who found that the sensor combination hip + wrist or wrist + ankle (foot in our case) produced a correlation coefficient of r=0.94 during *walking* and *jogging*. Obviously, the correlation coefficients obtained in this work are lower (r between 0.51 and 0.72) because energy expenditure is estimated for 52 activities some of them containing different levels of resistance work and work load.

Overall, the sensor combinations that present the lowest performance in the tables (Table 5-82 through Table 5-86) are the single accelerometer at the wrist (DWrist), and the single accelerometer at the hip (Hip). For example, when the wrist sensor is used, the difference in the correlation coefficient with respect to the all sensors combination is 0.23 units and the difference in RMSE is 0.3MET for the ACAbsArea feature (Table 5-82). When the ACFFTPeaks feature is used, the difference in the correlation coefficient for this sensor again, with respect to the all sensors combination, is 0.24 units and 0.3 MET for the RMSE. The differences are reduced slightly when the ACFFTPeaks + ACModVigEnergy + ACMCR is used. The poor performance of the DWrist sensor can be explained by the fact that the wrist presented the highest motion variability (as observed during the data collections) for most activities, particularly during household activities since the RMSE is higher over these categories for the wrist sensor. The low performance of the DUpperArm sensor can be explained by the fact that it has difficulties detecting upper body and non-ambulatory lower body motion. For example, when the performance per activity is analyzed for the Hip sensor for the five feature sets, it is found that the activities with lowest performance are activities involving lower body motion such as cycling at different speeds and resistance levels and rowing at different resistance levels. For these activities, the RMSE ranged between 1 and 3.5MET. Upper body activities such as *bench weight lifting*, *bicep curls*, and *wiping a surface* also presented the lowest performance. Appendix B15 presents the performance per activity for some of the highest performing features computed over the accelerometers at the hip, dominant wrist, and dominant foot.

Sensor	Correlation	All	Postures	Ambulation	Exercise	Resistance	Household
Combination						Exercise	
ACFFTPeaks +	$0.73 \pm 0.05$	$1.25 \pm 0.29$	0.5±0.2	1.1±0.4	1.3±0.7	1.2±0.6	0.7±0.2
ACModVigEnergy +		$(0.91 \pm 0.18)$	(0.5±0.2)	(1.0±0.4)	(1.2±0.7)	(1.1±0.6)	(0.6±0.2)
ACMCR							
ACFFTPeaks	$0.72 \pm 0.06$	$1.28\pm0.31$	0.6±0.2	1.1±0.5	1.4±0.8	1.2±0.6	0.7±0.3
		$(0.93 \pm 0.18)$	(0.5±0.2)	(1.0±0.5)	(1.3±0.8)	(1.1±0.6)	(0.6±0.2)
Invariant Reduced	$0.70 \pm 0.11$	$1.31\pm0.34$	0.6±0.3	1.2±0.5	1.4±0.8	1.2±0.7	0.8±0.5
		$(0.97 \pm 0.23)$	(0.5±0.3)	(1.0±0.5)	(1.3±0.8)	(1.1±0.6)	(0.7±0.5)
Fast to compute	$0.69 \pm 0.12$	$1.33\pm0.33$	0.6±0.3	1.2±0.5	1.3±0.7	1.2±0.6	0.9±0.6
_		$(0.99 \pm 0.24)$	(0.5±0.3)	(1.1±0.5)	(1.2±0.7)	(1.1±0.6)	(0.8±0.6)
ACAbsArea	$0.67\pm0.05$	$1.38 \pm 0.29$	0.8±0.2	1.1±0.4	1.5±0.7	1.4±0.6	0.8±0.3
		$(1.04 \pm 0.15)$	(0.7±0.2)	(1.0±0.4)	(1.4±0.7)	(1.3±0.6)	(0.7±0.2)

Table 5-87: Root mean squared error and mean absolute error (shown in parenthesis) obtained when estimating energy expenditure in a subject independent manner using linear regression and the five highest performing feature sets computed per sensor using windows of 5.6s in length over the accelerometers at the hip, dominant wrist, and dominant foot. Energy expenditure is estimated for the 51 activities contained in the MIT energy expenditure dataset.

Table 5-87 presents a summary of the results obtained using the five feature sets explored over the accelerometers located at the hip, dominant wrist, and dominant foot. This sensor combination (Hip+DWrist+DFoot) was the highest performing with respect to the performance obtained using all seven accelerometers. From the table, it can be seen that the performance of the *ACFFTPeaks* + *ACModVigEnergy* + *ACMCR* is the highest, followed by the performing feature sets is r=+0.01 and RMSE = -0.03MET. As a result, it can be argued that even when the *ACFFTPeaks* + ACModVigEnergy + *ACMCR*, the *ACFFTPeaks* feature achieves a close performance but with lower computational requirements. As a result, the final implementation of the energy expenditure algorithm will be utilizing the *ACFFTPeaks* feature to estimate energy expenditure. Performance will also be measured using the *ACAbsArea* feature given its low computational requirements. The final implementation of the EE estimation algorithm is discussed later in Section 5.6.11.

#### 5.6.9 Does Heart Rate Data Improve Energy Expenditure Estimation?

In previous sections, the lowest performance while estimating energy expenditure was obtained for activities involving different levels of resistance or work load effort. Examples of these activities include *cycling* and *rowing* at different resistance levels, performing *bicep curls* and *bench weight lifting* with different weights (at the same speed of motion), *walking on a treadmill* at different inclination grades and walking *carrying groceries*. This is because accelerometers have difficulties detecting the extra effort associated with these activities because they can only measure amount of motion (acceleration). Heart rate on the other hand, can detect changes in effort and resistance load because it has a linear relationship with energy expenditure during moderate and vigorous activities [47, 48, 81]. As a result, this section explores if combining accelerometer and heart rate data improves estimation of energy expenditure.

This section first evaluates the performance over individual heart rate features (*HRMean*, *HRAboveRest*, *ScaledHR*, *HRVar*, and *HRTrend*) to identify the single features

Heart Rate Features	Algorithm	Correlation Coefficient	All	Postures	Ambulation	Exercise	Resistance Exercise	Household
ScaledHR (1)	LR	$0.83\pm0.09$	$\begin{array}{c} 1.01 \ \pm 0.3 \\ (0.8 \pm 0.2) \end{array}$	0.5±0.3 (0.5±0.3)	0.9±0.5 (0.9±0.4)	1.3±0.7 (1.3±0.7)	1.2±0.6 (1.1±0.6)	0.5±0.3 (0.5±0.2)
HRAboveRest (1)	LR	$0.83 \pm 0.10$	$\begin{array}{c} 1.1 \pm 0.4 \\ (0.9 \pm 0.3) \end{array}$	0.5±0.3 (0.5±0.3)	1.0±0.6 (1.0±0.6)	1.5±0.8 (1.4±0.8)	1.3±0.7 (1.2±0.7)	0.5±0.3 (0.5±0.2)
HRMean (1)	LR	$0.82 \pm 0.09$	$1.3 \pm 0.4$ (1.0 ± 0.4)	0.8±0.4 (0.8±0.4)	1.2±0.7 (1.2±0.7)	1.6±0.9 (1.5±0.9)	1.3±0.8 (1.3±0.8)	0.6±0.3 (0.6±0.3)
HRVar (1)	LR	$0.08 \pm 0.05$	$\begin{array}{c} 1.8 \pm 0.3 \\ (1.5 \pm 0.2) \end{array}$	1.7±0.2 (1.7±0.2)	1.8±0.5 (1.7±0.5)	2.1±0.8 (2.0±0.8)	1.7±0.6 (1.6±0.6)	1.0±0.3 (1.0±0.3)
HRTrend (1)	LR	$0.04 \pm 0.05$	$\begin{array}{c} 1.8 \pm 0.3 \\ (0.0 \pm 0.0) \end{array}$	1.8±0.2 (1.7±0.2)	1.8±0.5 (1.7±0.5)	2.1±0.8 (2.0±0.8)	1.7±0.6 (1.6±0.6)	1.0±0.3 (1.0±0.3)
ScaledHR (1)	MT	$0.84 \pm 0.09$	$\begin{array}{c} 1.0 \pm 0.3 \\ (0.8 \pm 0.3) \end{array}$	0.5±0.3 (0.5±0.3)	1.0±0.5 (0.9±0.5)	1.3±0.8 (1.2±0.8)	1.2±0.7 (1.1±0.6)	0.5±0.2 (0.5±0.2)

Table 5-88: Root mean squared error and mean absolute error (shown in parenthesis) obtained while estimating energy expenditure in a subject independent manner over the MIT EE dataset using multivariable linear regression and individual heart rate features computed over windows of 5.6s in length. LR stands for linear regression and MT for model trees (M5').

with highest performance. Appendix A3 provides an explanation of how these features are computed and what information they attempt to capture. Once the heart rate feature with highest performance is identified, it is incorporated to the best set of accelerometerbased features found in Section 5.4.7. Energy expenditure is then estimated using multivariable linear regression with feature computation per sensor over sliding windows of 5.6s in length for both, accelerometer and heart rate data. The heart rate signal is preprocessed to reduce noise by applying a 15s running average filter. When the accelerometer data is included, the heart rate window length extends from the end of the acceleration window backwards in time as shown in Figure 5-24. Finally, heart rate windows and their associated acceleration window. All results presented in this section are evaluated using subject independent training unless otherwise indicated.

Table 5-88 presents the root mean squared error and mean absolute error (shown in parenthesis) obtained over individual heart rate features. It can be seen that the heart rate feature with higher overall performance is ScaledHR. The correlation coefficient obtained is 0.83 (out of a max of 1) and the overall RMSE error is 1.0MET when linear regression is used. The activity categories with higher RMSE are exercise and resistance exercise activities with RMSE values between 1.2 and 1.3MET respectively for the ScaledHR feature. This is because most of the activities contained in these categories involve resistance or work load effort. Heart rate can detect changes in resistance or work load effort during moderate and vigorous activities [47, 48, 81]; however, the main challenge in estimating energy expenditure for these activities is inter-individual differences due to fitness level and body composition (e.g. weight and fat mass). This is a problem because two subjects performing the same activity but with different fitness level will produce different heart rate readings. For example, more fit individuals tend to have lower heart rates than non-fit individuals [47] when they perform the same activity. This is why the lowest performance is obtained for the exercise and resistance exercise categories. Table 5-88 also shows that the performance of the *ScaledHR* feature is slightly higher than the one obtained for the HRAboveRest feature. This is because the ScaledHR feature normalizes heart rate readings to fall between 0 and 1 for resting heart rate and heart rate while running on a treadmill at 5mph for each subject.

Heart Rate Features	Correlation Coefficient	All	Postures	Ambulation	Exercise	Resistance Exercise	Household
ScaledHR	$0.83\pm0.09$	$0.92\pm0.29$	0.5±0.3	0.9±0.4	1.2±0.6	1.1±0.5	0.5±0.2
(1)		$(0.71 \pm 0.22)$	(0.5±0.3)	(0.8±0.4)	(1.2±0.6)	(1.1±0.5)	(0.4±0.2)
HRAboveRest	$0.83\pm0.10$	$0.93\pm0.30$	0.5±0.3	0.9±0.4	1.2±0.6	1.1±0.5	0.5±0.2
(1)		$(0.71 \pm 0.23)$	(0.5±0.3)	(0.8±0.4)	(1.2±0.6)	(1.1±0.5)	(0.4±0.2)

Table 5-89: Root mean squared error and mean absolute error (shown in parenthesis) obtained while estimating energy expenditure in a subject dependent manner over the MIT dataset using multivariable linear regression and individual heart rate features computed over windows of 5.6s in length.

This normalization helps in reducing inter-individual variations in heart rate because two individuals with different heart rate readings would be performing in the same intensity zone with respect to resting and sub-maximal heart rate (running at 5mph). Even when the overall difference in performance between the ScaledHR and HRAboveRest features is tiny (0.09MET in overall RMSE); performance per activity category is improved between 0.1 and 0.2MET for the ambulation, exercise, and resistance exercise categories. These are the activity categories that one would expect to benefit the most from the heart rate data normalization. One possible explanation of why the difference in performance is not larger is that heart rate while running on a treadmill at 5mph was used as a substitute for maximum heart rate during the normalization procedure. The measurement of maximal heart rate requires individuals to perform physically intense exercise (e.g. running on a treadmill) until heart rate reaches a maximum and no longer increases. Thus, this test is inconvenient in practice because it requires individuals to perform maximal physical effort. Furthermore, some individuals suffering from physical conditions might not even be able to perform such a test. Table 5-88 also shows that the features with lowest performance are HRTrend and HRVar.

This is because *HRVar* measures the variance over the heart rate data and this value is similar for most activities: high at the beginning of an activity due to non-steady state conditions and low towards the end of activities due to steady-state conditions. Similarly, *HRTrend* tells if heart rate is increasing (slope>0), decreasing (slope<0), or in steady state over time (slope~0). Thus, most activities present the same trends in this value: positive at the beginning of activities when energy expenditure is increasing and near zero towards the end of the activity when energy expenditure reaches steady-state condition. The last row of Table 5-88 presents the performance of the *ScaledHR* feature when a M5' model tree is used to estimate energy expenditure. The performance of the M5' model tree was tested over the best performing heart rate feature because this regression algorithm is able to capture non-linear relationships in the data and thus, could improve energy expenditure estimation. From the table, it can be observed that the performance indeed increases as expected, but the increase is so small (+0.01 for both, RMSE and r) that it does not justify the additional complexity of running this non-linear regression algorithm. One reason for this small improvement could be the unavailability of enough training examples at the leaf nodes of the model tree to allow adequate training of the linear regression models. For example, the model trees generated during the evaluation of the algorithm contained between 43 and 76 leaf nodes. This means that between 43 and 76 multivariable linear regression models have to be trained over each leaf node in order to predict energy expenditure. This partition of the available training data into 43-76

Features	Correlation	All	Postures	Ambulation	Exercise	Resistance	Household
	( <b>r</b> )					Exercise	
ScaledHR	0.83 0.09	$1.01 \pm 0.3$	0.5±0.3	0.9±0.5	1.3±0.7	$1.2\pm0.6$	0.5±0.3
		$(0.8\pm0.2)$	(0.5±0.3)	$(0.9\pm0.4)$	(1.3±0.7)	(1.1±0.6)	(0.5±0.2)
ScaledHR +	$0.83 \pm 0.09$	$1.01 \pm .32$	0.5±0.3	0.9±0.5	1.3±0.7	1.2±0.6	0.5±0.3
Weight		$(0.7 \pm 0.2)$	(0.5±0.3)	$(0.9\pm0.5)$	(1.3±0.7)	(1.1±0.6)	(0.5±0.2)
ScaledHR +	$0.83\pm0.09$	$1.01\pm0.32$	0.5±0.3	0.9±0.5	1.3±0.7	1.2±0.6	0.5±0.3
FatPercent		$(0.8\pm0.2)$	(0.5±0.3)	(0.9±0.5)	(1.3±0.7)	(1.1±0.6)	(0.5±0.3)
ScaledHR +	$0.83\pm0.10$	$1.07\pm0.36$	0.6±0.3	1.0±0.6	1.5±0.8	1.3±0.7	0.5±0.3
FitnessIndex		$(0.8 \pm 0.3)$	(0.5±0.3)	(0.9±0.5)	(1.4±0.8)	(1.2±0.7)	(0.5±0.3)
ScaledHR +	$0.83 \pm 0.09$	$1.06 \pm 0.34$	0.5±0.3	1.0±0.5	$1.4\pm0.8$	1.2±0.7	0.6±0.3
FitnessIndex +		$(0.8 \pm 0.3)$	(0.5±0.3)	(0.9±0.5)	(1.3±0.8)	(1.2±0.7)	(0.5±0.3)
Weight + Height +							
Age + Gender +							
FatPercent							

Table 5-90: Root mean squared error and mean absolute error (shown in parenthesis) obtained while combining features that attempt to describe the fitness level of an individual with heart rate features (*ScaledHR*) during subject independent evaluation of energy expenditure using linear regression. The target activities were the 51 activities contained in the MIT energy expenditure dataset.

nodes limits the amount of data available for training at each leaf node.

When plots are generated for subject independent estimation of energy expenditure using the ScaledHR feature (see Appendix B14 for an example) it is found that heart rate overestimates energy expenditure of upper body activities involving resistance exercise such as *bicep curls* and *bench weight lifting* and for postures such as *standing*, *kneeling*, and sitting (including sitting fidgeting feet and legs and hands and arms). Heart rate also underestimates the energy expenditure for most household activities (e.g. *sweeping*, *mopping* and *vacuuming*), for walking and for exercise activities such as *running*, *cycling* and *rowing* at different speeds and resistance levels. The worse overestimation of energy expenditure was observed for the *bicep curls* and *bench weightlifting* activities. One explanation for this is that these activities were performed after the *cycling hard* activity, one of the most physically demanding activities for which data was collected, so heart rate might have been still altered (with a high value) when these activities were performed even after participants rested for at least 5min before the activities. This highlights one of the problems of utilizing heart rate data to estimate energy expenditure: Heart rate lags physical activity and remains altered once a physically demanding activity has being finished. The time heart rate remains altered is subject dependent (e.g. depending on the fitness level of individuals).

Table 5-89 presents the results for subject dependent estimation of energy expenditure using the two highest performing heart rate features: *ScaledHR* and *HRAboveRest*. The first thing to observe is that the performance does not increase substantially with respect to the performance obtained during subject independent evaluation. For example, the increase in performance in RMSE with respect to subject independent training for the *ScaledHR* feature is +0.09MET and +0.17MET for the *HRAboveRest* feature. The coefficient of correlation remains practically unchanged. One possible explanation is that intra-individual variations in heart rate are as difficult to overcome during the regression procedure as inter-individual variations due to fitness level, age and gender. These results are in agreement with the results obtained by Haskell et al. [48] where a small difference of only 0.09units in the correlation coefficient was found between subject dependent and independent estimation of energy expenditure using heart rate data.

One possibility to reduce the differences in heart rate readings due to differences in fitness levels of individuals would be to include features that directly or indirectly capture the fitness level of individuals. Therefore, an experiment was run to determine if the addition of these features improves subject independent estimation of energy expenditure when the highest performing heart rate feature is used (ScaledHR). The features used in this experiment to directly or indirectly capture the fitness level of individuals are Weight, Height, Age, Gender, FatPercent, and FitnessIndex. These features are described in detail in Appendix A3. Table 5-90 presents the results of this experiment. In short, the table shows that adding these features does not improve energy expenditure estimation. In fact, performance is slightly degraded when some of these features are incorporated as observed for the FitnessIndex feature. The incorporation of the Weight figure was not expected to improve energy expenditure estimation because energy expenditure is being predicted in METs, a measurement unit that already normalizes energy expenditure with respect to body weight [246]. In other words, MET units normalize energy expenditure so that it is identical for a slim subject and an obese subject performing the same activity. Obviously, one assumption is that the mechanical efficiency of both individuals is the same. One possible explanation for why the incorporation of all these features does not improve energy expenditure estimation is that the number of subjects (n=16) included in the MIT energy expenditure dataset is not large enough so that relationships between energy expenditure and subject's physical characteristics or fitness level are captured. Another explanation is that these features are practically constants for each subject and consequently, some of them are eliminated by the M5 feature selection algorithm used by the linear regression algorithm employed as confirmed in practice by observing the resulting regression models. When the M5 feature selection algorithm is turned off, the same results were observed: No significant improvements in performance for most features and slight decreases in performance when all these features are incorporated at once and when the FitnessIndex feature is incorporated. Again, this might be due to the relatively limited number of subjects included in the dataset used.

Prior work has successfully incorporated subject's characteristics such as body mass, height and gender in the regression equations to compensate for inter-individual variations in energy expenditure [145, 246, 248]. Although body mass and height have been shown to improve energy expenditure estimates, gender has been found to not impact energy expenditure at least when body mass and speed are held constant during walking activities [246, 249, 250]. Appendix B14 presents the results per activity for energy expenditure estimation when the *ScaledHR* feature and the *ScaledHR* + *ACFFTPeaks* feature are utilized.

# 5.6.10 How Well can Energy Expenditure be Estimated by Combining Acceleration and Heart Rate Data?

In this section, the highest performing heart rate feature found in the previous section (*ScaledHR*) is incorporated to the highest performing set of acceleration-based features found in Section 5.4.7 (*invariant reduced* feature set). Table 5-91 presents the results of combining both feature sets using linear regression during subject independent evaluation. Both feature sets are computed over windows of 5.6s in length and the accelerometer features are computed per sensor over all the seven accelerometers.

Features subsets	Correlation	All	Postures	Ambulation	Exercise	Resistance	Household
						Exercise	
ScaledHR	$0.84\pm0.09$	$1.01 \pm 0.3$	0.5±0.3	0.9±0.5	1.3±0.7	1.2±0.6	0.5±0.3
		$(0.8 \pm 0.2)$	(0.5±0.3)	(0.9±0.4)	(1.3±0.7)	(1.1±0.6)	(0.5±0.2)
ACFFTPeaks +	$0.74\pm0.06$	$1.24\pm0.28$	0.7±0.3	1.2±0.5	1.3±0.8	1.1±0.6	0.7±0.2
ACModVigEnergy		$(0.9 \pm 0.17)$	(0.6±0.3)	(1.0±0.5)	(1.2±0.7)	(1.0±0.6)	(0.6±0.2)
+ACMCR							
ACFFTPeaks +	$0.88\pm0.05$	$0.88 \pm 0.20$	0.5±0.3	0.8±0.4	1.0±0.6	0.8±0.5	0.5±0.2
ACModVigEnergy		$(0.7 \pm 0.14)$	(0.5±0.3)	(0.7±0.4)	(0.9±0.6)	(0.8±0.5)	(0.4±0.2)
+ACMCR +							
ScaledHR							
ACFFTPeaks	$0.72\pm0.07$	$1.28\pm0.30$	0.6±0.2	1.2±0.5	1.4±0.8	1.2±0.7	0.7±0.2
		$(0.9 \pm 0.2)$	(0.6±0.2)	(1.1±0.5)	(1.3±0.8)	(1.1±0.7)	(0.6±0.2)
ACFFTPeaks +	$0.88\pm0.05$	$0.89\pm0.21$	0.5±0.2	0.8±0.4	1.1±0.6	0.9±0.5	0.5±0.2
ScaledHR		$(0.7 \pm 0.2)$	(0.4±0.2)	(0.7±0.4)	(1.0±0.6)	(0.8±0.5)	(0.4±0.2)
ACAbsArea	$0.68\pm0.06$	$1.36\pm0.30$	0.7±0.2	1.1±0.4	1.5±0.7	1.4±0.6	0.7±0.2
		$(1.0 \pm 0.2)$	(0.7±0.2)	(1.0±0.4)	(1.4±0.7)	(1.3±0.6)	(0.6±0.2)
ACAbsArea +	$0.87\pm0.06$	$0.90\pm0.24$	0.4±0.2	0.7±0.4	1.1±0.6	1.0±0.6	0.5±0.3
ScaledHR		$(0.7 \pm 0.2)$	(0.4±0.2)	(0.6±0.3)	(1.1±0.6)	(0.9±0.6)	(0.4±0.2)
Invariant Reduced	$0.72 \pm 0.11$	$1.28\pm0.29$	0.7±0.3	1.2±0.5	1.3±0.8	1.2±0.6	0.8±0.4
		$(0.9 \pm 0.2)$	(0.6±0.3)	(1.1±0.5)	(1.2±0.8)	(1.1±0.6)	(0.7±0.4)
Invariant Reduced	$0.88\pm0.05$	$0.90 \pm 0.22$	0.5±0.3	0.8±0.4	1.1±0.6	0.9±0.5	0.5±0.2
+ ScaledHR		$(0.7 \pm 0.2)$	(0.5±0.3)	(0.7±0.4)	(1.0±0.6)	(0.8±0.5)	(0.4±0.2)
Fast to compute	$0.72 \pm 0.11$	$1.27 \pm 0.27$	0.7±0.4	1.2±0.5	1.2±0.7	1.1±0.6	0.8±0.5
-		$(0.9 \pm 0.2)$	(0.7±0.4)	(1.1±0.5)	(1.1±0.7)	(1.0±0.6)	(0.7±0.5)
Fast to compute +	$0.88\pm0.05$	$0.89\pm0.20$	0.5±0.3	0.8±0.4	1.0±0.6	0.9±0.5	0.5±0.3
ScaledHR		$(0.7 \pm 0.2)$	(0.5±0.3)	(0.7±0.4)	(0.9±0.6)	(0.8±0.5)	(0.5±0.3)

Table 5-91: Root mean squared error and mean absolute error (shown in parenthesis) obtained while estimating energy expenditure using linear regression when the most discriminating accelerometer (*invariant reduced* feature set) and heart rate feature (*ScaledHR*) are used and computed per sensor over all the seven accelerometers. Energy expenditure was predicted over the 51 activities contained in the MIT energy expenditure dataset. The definition of the *Invariant* and *fast to compute* feature sets can be found in Section 5.4.7. Energy expenditure is evaluated in a subject independent manner.

The first result that is highlighted by Table 5-91 is that the performance of estimating energy expenditure using the *ScaledHR* feature alone is higher than the one obtained by utilizing any of the highest performing accelerometer-based features alone. For instance, the ScaledHR feature alone achieves a correlation coefficient +0.14 units higher than the ACFFTPeaks feature alone. The ScaledHR feature also achieves a RMSE 0.27MET lower than the ACFFTPeaks feature alone. Similarly, the ScaledHR feature improves the correlation coefficient +0.16units and the RMSE in +0.35MET over the performance of the ACAbsArea feature alone. One explanation for this result is that most of the activities in the MIT energy expenditure dataset have different intensity levels due to varying resistance work or load effort. Thus, their energy expenditure is better estimated using heart rate data. It is important to remember that when the accelerometer-based features are used to estimate energy expenditure, a single multivariable linear regression model is used to model the energy expenditure associated with all the activities performed. Section 5.6.11.2 will explore if activity dependent regression models improve energy expenditure estimation over the utilization of single regression model for all activities based on accelerometer features.

Even though the performance of the *ScaledHR* feature is superior to the performance of accelerometer-based features, when both feature sets are combined, overall performance and performance per activity is improved. For example, it can be seen from Table 5-91 that when the *ScaledHR* feature is incorporated, the correlation coefficient improves

+0.16 for the ACFFTPeaks feature, +0.19 for the ACAbsArea feature, +0.16 for the invariant reduced and fast to compute feature sets. The RMSE improves 0.38MET for the ACFFTPeaks feature, 0.46MET for the ACAbsArea feature, and 0.38MET for the invariant reduced and fast to compute feature sets. Thus, the single feature that benefits the most with the incorporation of the ScaledHR feature is the ACAbsArea feature. Interestingly, the improvement over the *invariant reduced* and *fast to compute* feature sets when the ScaledHR feature is incorporated is almost identical. This is because the performance of these feature sets is almost already identical before the incorporation of the heart rate feature as can be seen in Table 5-91. In general, although all activity categories benefit from the incorporation of the ScaledHR feature, the categories that benefit the most are ambulation (+0.4MET in RMSE on average), exercise activities (+0.2-0.4MET) and resistance exercise activities (+0.2-0.4MET). This is because they incorporate more activities that involve different intensity levels due to changes in resistance work or load effort. Finally, from Table 5-91, it can be concluded that the best feature combinations to utilize when estimating energy expenditure from seven accelerometers in this work) are *ACFFTPeaks+ScaledHR* (at least and ACAbsArea+ScaledHR. This is because their performance is very close to the performance obtained using the invariant reduced and fast to compute feature sets but their computational requirements are lower (less number of accelerometer-based features).

Table 5-92 presents the performance of combining heart rate and accelerometer-based features when two subsets of accelerometers are used: (a) three sensors worn at the hip, dominant wrist, and dominant foot and (b) a single sensor worn at the hip. It can be observed that the performance utilizing three sensors at the hip, dominant wrist, and dominant foot is very close to the performance obtained when all seven sensors are used. This is because three sensors at these locations are able to capture upper body, lower body, and overall body motion well enough to produce estimates close to the ones obtained using all the accelerometers. This is explained in detail in Section 5.4.7.

From Table 5-92, it can also be seen that the best two feature combination to utilize when estimating energy expenditure are still the ACFFTPeaks + ScaledHR and ACAbsArea + ScaledHR. This is because their performance per activity category is either almost identical or slightly higher to the one obtained using the same features but computed over all seven accelerometers. For example, the correlation coefficients remain practically unchanged, and the RMSE increases +0.01MET for the ACFFTPeaks feature when only three sensors are used, and the RMSE increases +0.1MET for the exercise category when the ACAbsArea feature is computed over three sensors only (hip, dominant wrist, and dominant foot). The slight improvement in performance is due to the reduction in the number of predictor variables (since features are computed over fewer sensors). In general, the incorporation of accelerometer-based features to the ScaledHR feature improves performance only slightly. For example, the correlation coefficient improves between 0.02 and 0.03 units and the RMSE decreases between 0.03 and 0.1MET when the features computed over the accelerometer at the hip are incorporated to the ScaledHR feature. Similarly, the correlation coefficient improves between 0.03 and 0.04 units and the RMSE decreases between 0.11 and 0.13 MET when the features computed over the accelerometers at the hip, dominant wrist, and dominant foot are incorporated to the ScaledHR feature. Finally, the improvement in RMSE per activity

Features subsets	Correlation	All	Postures	Ambulation	Exercise	Resistance Exercise	Household
ACFFTPeaks + ScaledHR Hip	$0.84 \pm 0.08$	$\begin{array}{c} 0.98 \pm 0.28 \\ (0.8 \pm 0.2) \end{array}$	0.6±0.3 (0.5±0.3)	0.9±0.4 (0.9±0.4)	1.2±0.7 (1.1±0.7)	1.1±0.6 (1.1±0.6)	0.6±0.2 (0.5±0.2)
ACFFTPeaks + ScaledHR Hip + DWrist + DFoot	$0.88 \pm 0.05$	$\begin{array}{c} 0.88 \pm 0.22 \\ (0.7 \pm 0.2) \end{array}$	0.5±0.2 (0.4±0.2)	0.8±0.3 (0.7±0.3)	1.1±0.6 (1.0±0.6)	0.9±0.5 (0.9±0.5)	0.5±0.2 (0.4±0.2)
ACAbsAreas + ScaledHR Hip	$0.86 \pm 0.07$	$\begin{array}{c} 0.95 \pm 0.28 \\ (0.7 \pm 0.2) \end{array}$	0.4±0.2 (0.4±0.2)	0.8±0.4 (0.7±0.4)	1.2±0.7 (1.2±0.7)	1.1±0.6 (1.0±0.6)	0.5±0.2 (0.4±0.2)
ACAbsAreas + ScaledHR Hip + DWrist + DFoot	0.87 ± 0.06	$\begin{array}{c} 0.90 \pm 0.25 \\ (0.7 \pm 0.2) \end{array}$	0.4±0.2 (0.4±0.2)	0.7±0.3 (0.6±0.3)	1.2±0.6 (1.1±0.6)	1.0±0.6 (1.0±0.6)	0.5±0.2 (0.4±0.2)
Invariant Reduced + ScaledHR Hip	$0.86 \pm 0.08$	$\begin{array}{c} 0.94 \pm 0.27 \\ (0.7 \pm 0.2) \end{array}$	0.6±0.3 (0.5±0.3)	0.9±0.4 (0.8±0.4)	1.2±0.7 (1.1±0.7)	1.0±0.6 (0.9±0.6)	0.5±0.2 (0.5±0.2)
Invariant Reduced + ScaledHR Hip + DWrist + DFoot	0.88 ± 0.05	$\begin{array}{c} 0.88 \pm 0.23 \\ (0.7 \pm 0.2) \end{array}$	0.5±0.3 (0.4±0.3)	0.8±0.4 (0.7±0.3)	1.1±0.6 (1.0±0.6)	0.9±0.5 (0.8±0.5)	0.5±0.3 (0.5±0.2)
Fast to compute + ScaledHR Hip	$0.87\pm0.06$	$\begin{array}{c} 0.91 \pm 0.25 \\ (0.7 \pm 0.2) \end{array}$	0.4±0.3 (0.4±0.3)	0.8±0.4 (0.7±0.4)	1.1±0.6 (1.1±0.6)	1.0±0.6 (0.9±0.5)	0.5±0.2 (0.4±0.2)
Fast to compute + ScaledHR Hip + DWrist + DFoot	0.88 ± 0.05	$\begin{array}{c} 0.89 \pm 0.22 \\ (0.7 \pm 0.2) \end{array}$	0.4±0.3 (0.4±0.3)	0.8±0.4 (0.7±0.4)	1.1±0.6 (1.0±0.6)	0.9±0.5 (0.9±0.5)	0.5±0.3 (0.5±0.3)

Table 5-92: Root mean squared error and mean absolute error (shown in parenthesis) obtained while estimating energy expenditure in a subject independent manner using linear regression when the most discriminating accelerometer (*invariant reduced* feature set) and heart rate (*ScaledHR*) features are used. Features are computed per sensor over two sets of accelerometers: (a) hip, dominant wrist, and dominant foot, and (b) hip. Energy expenditure was predicted over the 51 activities contained in the MIT energy expenditure dataset.

category ranges between 0 and 0.3MET when accelerometer-based features are incorporated to the *ScaledHR* feature. The activity category that benefits the least from the incorporation of accelerometer-based features is the household category. The performance per activity for ScaledHR+ACFFTPeaks features computed over the accelerometers at the hip, dominant wrist, and dominant foot can be found in Appendix B14.

# 5.6.11 How Well Can Energy Expenditure be Estimated Using the Selected Window Length, Feature Set, and Signal Preprocessing Techniques?

This section evaluates the performance of the final implementation of the energy expenditure estimation algorithm using the set of parameters incrementally selected in the previous sections. These parameters consist on the *ACFFTPeaks* feature set, feature computation per sensor over sliding windows of 5.6s in length, and the multivariable linear regression algorithm. The algorithm computes features over the accelerometers

Evaluation Method	Correlation	All	Postures	Ambulation	Exercise	Resistance	Household
						Exercise	
ACAbsArea	$0.67\pm0.05$	$1.38\pm0.29$	0.8±0.2	1.1±0.4	1.5±0.7	1.4±0.6	0.8±0.3
Subject independent		$(1.04 \pm 0.15)$	(0.7±0.2)	(1.0±0.4)	(1.4±0.7)	(1.3±0.6)	(0.7±0.2)
ACFFTPeaks	$0.72\pm0.06$	$1.28\pm0.31$	0.6±0.2	1.1±0.5	$1.4\pm0.8$	1.2±0.6	0.7±0.3
Subject independent		$(0.93 \pm 0.18)$	(0.5±0.2)	(1.0±0.5)	(1.3±0.8)	(1.1±0.6)	(0.6±0.2)
ACFFTPeaks +	$0.73\pm0.05$	$1.25\pm0.29$	0.5±0.2	1.1±0.4	1.3±0.7	1.2±0.6	0.7±0.2
ACModVigEnergy +		$(0.91 \pm 0.18)$	(0.5±0.2)	(1.0±0.4)	(1.2±0.7)	(1.1±0.6)	(0.6±0.2)
ACMCR							
Subject independent							
Fast to Compute	$0.69\pm0.12$	$1.33\pm0.33$	0.6±0.3	1.2±0.5	1.3±0.7	$1.2\pm0.6$	0.9±0.6
Subject Independent		$(0.99 \pm 0.24)$	(0.5±0.3)	(1.1±0.5)	(1.2±0.7)	(1.1±0.6)	$(0.8\pm0.6)$
Invariant Reduced	$0.70\pm0.11$	$1.31\pm0.34$	0.6±0.3	1.2±0.5	$1.4\pm0.8$	$1.2\pm0.7$	0.8±0.5
Subject Independent		$(0.97 \pm 0.23)$	(0.5±0.3)	(1.0±0.5)	(1.3±0.8)	(1.1±0.6)	(0.7±0.5)
ACAbsArea	$0.70\pm0.05$	$1.24\pm0.31$	0.8±0.4	1.0±0.4	1.3±0.7	1.2±0.6	0.7±0.3
Subject dependent		$(0.93 \pm 0.25)$	(0.8±0.4)	$(0.9\pm0.4)$	(1.2±0.7)	(1.2±0.6)	(0.6±0.3)
ACFFTPeaks	$0.79\pm0.04$	$1.06\pm0.24$	0.7±0.3	1.0±0.3	$1.0\pm0.4$	0.9±0.4	0.6±0.3
Subject dependent		$(0.78 \pm 0.19)$	(0.6±0.3)	(0.8±0.3)	(0.9±0.4)	(0.8±0.3)	(0.5±0.2)
ACFFTPeaks +	$0.80\pm0.04$	$1.02\pm0.23$	0.6±0.3	0.9±0.3	0.9±0.4	0.9±0.3	0.6±0.2
ACModVigEnergy +		$(0.75 \pm 0.18)$	(0.5±0.3)	(0.8±0.2)	(0.8±0.3)	(0.8±0.3)	(0.5±0.2)
ACMCR							
Subject dependent							
Fast to Compute	$0.79\pm0.05$	$1.06\pm0.24$	0.7±0.3	0.9±0.3	$1.0\pm0.4$	$0.9\pm0.4$	0.6±0.3
Subject dependent		$(0.79 \pm 0.19)$	(0.6±0.3)	(0.8±0.3)	(0.8±0.4)	(0.8±0.3)	(0.5±0.2)
Invariant Reduced	$0.81 \pm 0.04$	$1.01 \pm 0.23$	0.7±0.3	0.9±0.3	0.9±0.4	0.8±0.3	0.6±0.3
Subject dependent		$(0.75 \pm 0.18)$	$(0.6\pm 0.3)$	$(0.8\pm0.2)$	$(0.8\pm0.3)$	$(0.7\pm0.3)$	$(0.5\pm0.2)$

Table 5-93: Root mean squared error and mean absolute error (shown in parenthesis) obtained by estimating energy expenditure in a subject independent manner using linear regression and different feature sets computed per sensor over windows of 5.6s in length over three accelerometers located at the hip, dominant wrist, and dominant foot.

located at the hip, dominant wrist, and dominant foot. The section starts by presents a summary of how well can energy expenditure be estimated in a subject independent and dependent manner using the five feature sets with highest performance as found in Section 5.6.7. Later, the section analyzes how well can energy expenditure be estimated by utilizing activity dependent regression models using the final set of features in a subject dependent and independent manner. During this evaluation, the 51 activities contained in the MIT dataset are recognized using the final implementation of the activity recognition algorithm (see Section 5.4.9). The section also explores how well can energy expenditure be predicted when the number of activities to recognize is reduced to a set of 11 activities that would be useful to recognize during medical studies in practice. Finally, the section briefly explores how well can energy expenditure be predicted when the mean energy expenditure value for each activity is predicted after recognizing activities.

## 5.6.11.1 How Well Can Energy Expenditure be Estimated Using Linear Regression?

Table 5-93 presents a summary of how well can energy expenditure be predicted in a subject dependent and independent manner using the five feature sets with higher performance from Section 5.6.7. During this analysis, features are computed per sensor over windows of 5.6s in length using the accelerometers located at the hip, dominant wrist, and dominant foot. Energy expenditure is estimated using multivariable linear regression. The table shows that during subject independent evaluation, the feature set that achieves the highest performance is the *ACFFTPeaks* + *ACModVigEnergy* +

ACMCR feature set (r=0.73, RMSE=1.25). The improvement obtained with respect to the ACAbsArea feature is +0.06units in the correlation coefficient and 0.13MET for the RMSE. The table also shows that the performance of the ACAbsArea feature is lower than any of the other feature combinations. This confirms, as found in the previous sections, that computing features that capture information other than overall amount of motion (captured by ACAbsArea) improves energy expenditure estimation. From Table 5-93 it can also be seen that the performance of the fast to compute and invariant reduced feature sets is lower than for the one obtained for the ACFFTPeaks and the ACFFTPeaks + ACModVigEnergy + ACMCR feature sets. As explained previously, this is because the number of predictor variables increases and there is insufficient amount of data to train the regression models (now containing a larger set of variables).

The performance of the *ACFFTPeaks* feature set is slightly lower than the performance of the *ACFFTPeaks* + *ACModVigEnergy* + *ACMCR* feature combination but it requires fewer computations since the *ACModVigEnergy* and the *ACMCR* features do not have to be computed. The decrease in the correlation coefficient is just 0.01units and the increase in the RMSE is 0.03MET with respect to the *ACFFTPeaks* + *ACModVigEnergy* + *ACMCR* feature set when the *ACFFTPeaks* feature is utilized. The difference in performance between estimating energy expenditure using the *ScaledHR* feature and the *ACFFTPeaks* feature set during subject independent evaluation is 0.11units for the correlation coefficient and 0.27MET for the RMSE.

During subject dependent evaluation, it can be seen that the feature sets with highest performance in decreasing order are: the invariant reduced feature set (r=0.81, RMSE=1.01), followed by the ACFFTPeaks + ACModVigEnergy + ACMCR (r=0.80, RMSE=1.02), the fast to compute (r=0.79, RMSE=1.06) and the ACFFTPeaks (r=0.79, RMSE=1.06) feature set. One possible reason why the invariant reduced feature set achieves the highest performance despite its larger vector size (predictor variables) is that since there are no inter-individual variations in motion during subject dependent training (there is less variability per activity in the training data), the regression model is able to capture motion characteristics well for each activity despite the limited amount of training data. It can also be observed that all feature sets improve performance over the ACAbsArea feature during both, subject dependent and independent training. When the performance of the ACFFTPeaks feature is compared to the one obtained using the invariant reduced feature set, it is found that the difference in the correlation coefficient is 0.02units and 0.05MET for the RMSE. Moreover, the performance achieved by any of the feature sets but the ACAbsArea feature is close to the one obtained using the ScaledHR feature during subject independent evaluation (r=0.83, RMSE=1.01). This is a good result, since it can be argued that wearing accelerometers longitudinally is more comfortable than wearing heart rate monitors placed on the chest. Unfortunately, subject dependent energy expenditure data is presently difficult to collect due to the high cost of the equipment (e.g. indirect calorimeter) and expertise required to operate it.

Thus, from the results shown in Table 5-93, it can be argued that the *ACFFTPeaks* feature set presents a good compromise between subject dependent and independent performance and computational complexity with respect to the other features shown in the table.

## 5.6.11.2 How Well Can Energy Expenditure be Estimated Using One Linear Regression Model Per Activity?

This section explores the performance of utilizing activity dependent regression models during the estimation of energy expenditure when the activities performed are recognized using the activity recognition algorithm implemented in Section 5.4.9. Specifically; in this section, activities being performed are recognized using the C4.5 decision tree classifier trained over features computed per sensor over windows of 5.6s in length over different accelerometer combinations. This activity recognition algorithm is evaluated over the two highest performing feature sets found for activity recognition: (1) the *fast to compute* feature set and (2) the *invariant reduced* feature set. For a detailed description of the features included in these feature sets refer to Section 5.4.7. Once activities are recognized using this algorithm, activity dependent linear regression models are used to estimate energy expenditure. These multivariable linear regression models are trained in a subject independent manner over the energy expenditure data collected for each activity. No analysis is presented in this section for subject dependent estimation of energy expenditure due to the unavailability (for most people) of the necessary equipment to collect energy expenditure data during free-living.

Table 5-94 presents the performance of estimating energy expenditure by (1) recognizing the 51 activities contained in the MIT energy expenditure dataset in a subject independent manner using the algorithm implemented in Section 5.4.9 and (2) applying a multivariable linear regression model per activity trained in a subject independent manner. Activities are recognized using the *fast to compute* feature set and energy expenditure is estimated using the ACAbsArea feature. Both feature sets are computed per sensor over windows of 5.6s in length over the different sensor combinations. The objective of analyzing this feature pair (fast to compute, ACAbsArea) is to determine how well energy expenditure can be estimated using features that are not as computationally intensive as the other feature sets explored in this work. Table 5-94 shows that the highest overall performance (r=0.82, RMSE=1.12) is obtained using all the seven accelerometers. The second highest performing sensor combination is Hip+DWrist+DFoot, followed by the DWrist+DFoot and by the Hip+DWrist sensor combination. The performance of the DWrist+DFoot combination is almost identical to the one obtained using the Hip+DWrist+DFoot sensor combination as highlighted in Section 5.6.8. Between these two sensor combinations, there is only a 0.01MET difference in RMSE for the postures and ambulation categories. The performance obtained for the Hip+DWrist+DFoot sensor combination (r=0.80, RMSE=1.17) using activity dependent regression models is higher than the performance obtained using a single multivariable linear regression trained over the ACAbsArea feature (r=0.67, RMSE=1.38). This indicates that activity dependent regression models indeed improve energy expenditure estimation, even when a large set of activities is recognized in a subject independent manner. It is interesting to note that energy expenditure can be predicted well even when the overall accuracy of recognizing 51 activities (without the unknown class) in a subject independent manner using the fast to compute feature set is only  $\sim$ 45% (see Section 5.4.9). One possible explanation is that even when the activity recognition algorithm performs misclassifications, these misclassifications correspond to activities that are similar (in motion patterns) to the real

Sensor Combination	Correlation	All	Postures	Ambulation	Exercise	Resistance	Household
						Exercise	
All sensors	$0.82\pm0.07$	$1.12\pm0.26$	0.5±0.3	1.3±0.4	1.2±0.6	1.1±0.5	0.7±0.4
		$(0.78 \pm 0.21)$	(0.4±0.3)	(1.1±0.4)	(1.1±0.6)	(1.0±0.4)	(0.6±0.3)
Hip + DWrist +	$0.80\pm0.09$	$1.17\pm0.35$	0.4±0.3	1.2±0.5	1.3±0.7	1.2±0.6	0.7±0.4
DFoot		$(0.79 \pm 0.21)$	(0.4±0.3)	(1.0±0.5)	(1.1±0.7)	(1.0±0.6)	(0.6±0.3)
Hip + DWrist	$0.78\pm0.05$	$1.21\pm0.24$	0.5±0.3	1.3±0.6	1.2±0.7	1.1±0.6	0.8±0.5
		$(0.82 \pm 0.18)$	(0.4±0.2)	(1.1±0.5)	(1.1±0.7)	(1.0±0.6)	$(0.6\pm0.4)$
Hip + DFoot	$0.80\pm0.08$	$1.20\pm0.38$	0.5±0.3	1.3±0.6	1.2±0.7	1.1±0.6	0.8±0.5
		$(0.83 \pm 0.25)$	(0.4±0.2)	(1.1±0.5)	(1.1±0.7)	(1.0±0.6)	$(0.6\pm0.4)$
DWrist + DThigh	$0.78\pm0.07$	$1.21\pm0.24$	0.7±0.5	1.2±0.4	1.3±0.6	1.2±0.5	0.7±0.3
		$(0.83 \pm 0.18)$	(0.5±0.4)	(1.1±0.4)	(1.1±0.6)	(1.1±0.5)	(0.5±0.2)
DWrist + DFoot	$0.80\pm0.06$	$1.18\pm0.28$	0.5±0.3	1.3±0.5	1.3±0.7	1.2±0.6	0.7±0.4
		$(0.81 \pm 0.20)$	(0.4±0.3)	(1.1±0.4)	(1.1±0.6)	(1.1±0.5)	(0.6±0.3)
Hip	$0.71\pm0.07$	$1.40\pm0.28$	0.4±0.3	1.5±0.5	1.5±0.7	1.3±0.6	1.0±0.4
		$(0.96 \pm 0.20)$	(0.4±0.2)	(1.3±0.5)	(1.3±0.6)	(1.1±0.6)	(0.7±0.3)
DWrist	$0.46\pm0.14$	$1.85\pm0.28$	1.1±0.6	1.8±0.6	2.1±0.8	1.9±0.6	1.4±0.5
		$(1.32 \pm 0.24)$	(0.8±0.4)	(1.6±0.6)	(1.8±0.8)	(1.6±0.6)	(1.1±0.4)
DFoot	$0.75\pm0.07$	$1.31\pm0.29$	0.7±0.4	1.5±0.6	1.4±0.7	1.2±0.5	0.9±0.4
		$(0.94 \pm 0.20)$	(0.6±0.3)	(1.3±0.5)	(1.2±0.6)	(1.1±0.5)	(0.7±0.3)
DUpperArm	$0.54\pm0.09$	$1.76\pm0.35$	0.8±0.6	1.6±0.5	2.0±0.8	1.9±0.7	1.3±0.5
		$(1.23 \pm 0.29)$	$(0.6\pm0.4)$	(1.4±0.5)	(1.7±0.8)	(1.6±0.7)	(1.0±0.4)
DThigh	$0.76\pm0.06$	$1.29 \pm 0.23$	0.8±0.4	1.2±0.4	1.4±0.7	1.3±0.6	0.8±0.3
		$(0.88 \pm 0.17)$	$(0.6\pm0.4)$	$(1.1\pm0.4)$	$(1.2\pm0.7)$	$(1.1\pm0.6)$	(0.6+0.2)

Table 5-94: Root mean squared error and mean absolute error (shown in parenthesis) obtained by estimating energy expenditure by first recognizing 51 activities in a subject independent manner and then applying a multivariable linear regression model per activity trained in a subject independent manner. The *fast to compute* feature set is used to recognize activities and the *ACAbsArea* feature to estimate energy expenditure. Both feature sets are computed per sensor over windows of 5.6s in length.

Sensor Combination	Correlation	All	Postures	Ambulation	Exercise	Resistance	Household
						Exercise	
All sensors	$0.77\pm0.07$	$1.30\pm0.28$	0.6±0.4	1.4±0.6	1.4±0.6	1.3±0.6	0.8±0.5
		$(0.85 \pm 0.21)$	(0.4±0.3)	(1.2±0.5)	(1.1±0.6)	(1.1±0.5)	(0.6±0.3)
Hip + DWrist +	$0.77 \pm 0.11$	$1.31\pm0.40$	0.6±0.4	1.5±0.7	1.4±0.8	1.3±0.6	0.9±0.5
DFoot		$(0.87 \pm 0.25)$	(0.5±0.3)	(1.3±0.6)	(1.2±0.7)	(1.0±0.5)	(0.6±0.3)
Hip + DWrist	$0.75\pm0.08$	$1.29 \pm 0.33$	0.8±1.1	1.3±0.5	1.3±0.6	1.2±0.5	0.9±0.5
-		$(0.87 \pm 0.21)$	(0.5±0.6)	(1.1±0.5)	(1.1±0.5)	(1.0±0.4)	(0.7±0.3)
Hip + DFoot	$0.78\pm0.12$	$1.26\pm0.42$	0.7±0.5	1.4±0.6	1.3±0.7	1.2±0.6	0.9±0.6
_		$(0.85 \pm 0.27)$	(0.5±0.4)	(1.2±0.6)	(1.1±0.7)	(1.0±0.6)	(0.6±0.4)
DWrist + DThigh	$0.79\pm0.07$	$1.21 \pm 0.27$	0.6±0.6	1.2±0.4	1.3±0.6	1.2±0.6	0.7±0.3
		$(0.81 \pm 0.17)$	(0.5±0.5)	(1.0±0.4)	(1.1±0.6)	(1.0±0.5)	(0.6±0.2)
DWrist + DFoot	$0.77 \pm 0.11$	$1.31\pm0.43$	0.6±0.5	1.4±0.6	1.3±0.7	1.3±0.7	0.9±0.7
		$(0.86 \pm 0.25)$	(0.5±0.4)	(1.2±0.6)	(1.1±0.6)	(1.1±0.5)	(0.7±0.5)
Hip	$0.77\pm0.07$	$1.22\pm0.20$	0.5±0.3	1.4±0.5	1.3±0.5	1.2±0.4	0.9±0.4
		$(0.85 \pm 0.17)$	(0.4±0.2)	(1.1±0.5)	(1.2±0.5)	$(1.0\pm0.4)$	(0.6±0.3)
DWrist	$0.60\pm0.12$	$1.59\pm0.29$	0.9±0.5	1.5±0.6	1.7±0.8	1.6±0.6	1.2±0.5
		$(1.08 \pm 0.22)$	(0.7±0.4)	(1.3±0.6)	(1.5±0.8)	(1.3±0.6)	(0.9±0.4)
DFoot	$0.74 \pm 0.16$	$1.45\pm0.78$	0.6±0.5	1.3±0.5	1.4±0.9	1.4±1.0	1.1±1.2
		$(0.90 \pm 0.31)$	(0.5±0.4)	(1.1±0.4)	(1.2±0.7)	(1.2±0.8)	(0.8±0.7)
DUpperArm	$0.78\pm0.04$	$1.25 \pm 0.31$	0.5±0.4	1.3±0.5	1.3±0.7	1.3±0.6	0.7±0.3
		$(0.83 \pm 0.19)$	(0.4±0.3)	(1.1±0.5)	(1.2±0.6)	(1.1±0.5)	(0.6±0.2)
DThigh	$0.77 \pm 0.05$	$1.25 \pm 0.26$	0.7±0.4	1.3±0.4	1.4±0.6	1.3±0.5	0.8±0.3
		$(0.87 \pm 0.18)$	(0.5±0.4)	(1.1±0.4)	(1.2±0.6)	(1.1±0.5)	(0.6±0.2)

Table 5-95: Root mean squared error and mean absolute error (shown in parenthesis) obtained by estimating energy expenditure by first recognizing 51 activities in a subject independent manner and then applying a multivariable linear regression model per activity trained in a subject independent manner. The *invariant reduced* feature set is used to recognize activities and the *ACFFTPeaks* feature to estimate energy expenditure. Both feature sets are computed per sensor over windows of 5.6s in length.

activity being performed and thus, the energy expenditure estimated is close to the one of the target activity.

The performance of the Hip+DWrist+DFoot sensor combination (r=0.80, RMSE=1.17) is similar to the performance obtained using a single linear regression model and the *fast to compute* feature set (r=0.80, RMSE=1.18). This is because in Table 5-94, energy expenditure is predicted using only the *ACAbsArea* feature but the improved performance obtained is mainly due to the use of activity dependent regression models. This suggests that energy expenditure could be further improved by including features other than the *ACAbsArea* feature (e.g. *fast to compute* feature set) provided there is enough training data to train the activity dependent models.

Table 5-95 presents the performance of estimating energy expenditure also utilizing activity dependent linear regression models; however, the 51 activities contained in the MIT energy expenditure dataset are now recognized using the *invariant reduced* feature set and energy expenditure is predicted using the ACFFTPeaks feature. Activities and energy expenditure are evaluated in a subject independent manner. Both feature sets are again computed per sensor over windows of 5.6s in length over the different sensor combinations. The objective of evaluating the performance over these feature sets is to (1) test how better recognition of activities (from the *invariant reduced* feature set) improves energy expenditure estimation and (2) how a more complex feature set (ACFFTPeaks) improves energy expenditure estimation. However, a quick inspection of Table 5-95 reveals that the correlation coefficients and RMSE obtained have a lower performance than the one obtained in Table 5-94. The reason for this is that there is less training data available to train each of the linear regression models due to the utilization of 51 activity dependent regression models. This is mainly due to the relatively large feature vector size (70 features per sensor) of the ACFFTPeaks feature with respect to the training data available per activity. This is also highlighted by the fact that performance increases as the number of sensors is decreased for some sensor combinations in Table 5-95. For example, the performance of the single sensor at the hip is r=0.77 and RMSE=1.22 while the performance over the Hip+DWrist+DFoot is r=0.77 and RMSE=1.31. For this reason, feature combinations incorporating more features than the ACFFTPeaks feature set are not evaluated during the analysis of energy expenditure estimation using activity dependent models. This does not mean that incorporating more features would not improve energy expenditure estimation. Estimating energy expenditure using the *fast to compute* and ACFFTPeaks feature sets have already been shown to improve performance when a single linear regression model is used to estimate energy expenditure (Section 5.6.11.1). This only means that the amount of data contained in the MIT energy expenditure dataset is insufficient to evaluate the performance over larger number of features when activity dependent regression models are used to estimate energy expenditure. Nevertheless, even when training data is insufficient, Table 5-95 still shows that overall performance and performance per activity for the sensor combinations all sensors, Hip+DWrist+DFoot, DWrist+DFoot, and DWrist+DThigh is similar as observed in previous sections.

Table 5-96 presents the results obtained when energy expenditure is estimated by recognizing the 51 activities in a subject dependent manner and when activity dependent linear regression models trained in a subject independent manner are used. Activities are recognized using the C4.5 decision tree classifier trained using the *fast to compute* feature

Sensor Combination	Correlation	All	Postures	Ambulation	Exercise	Resistance	Household
						Exercise	
All sensors	$0.88\pm0.05$	$0.98 \pm 0.27$	0.3±0.2	1.0±0.4	1.2±0.6	0.9±0.5	0.5±0.2
		$(0.65 \pm 0.16)$	(0.2±0.1)	(0.8±0.4)	(1.1±0.6)	(0.8±0.4)	(0.4±0.2)
Hip + DWrist +	$0.89\pm0.04$	$0.93\pm0.28$	0.3±0.2	0.9±0.3	1.1±0.6	0.9±0.5	0.5±0.2
DFoot		$(0.61 \pm 0.15)$	(0.2±0.1)	(0.7±0.3)	(1.0±0.6)	(0.8±0.5)	(0.4±0.2)
Hip + DWrist	$0.88\pm0.04$	$0.91\pm0.21$	0.3±0.2	0.9±0.3	1.1±0.6	0.9±0.4	0.6±0.2
		$(0.60 \pm 0.13)$	$(0.2\pm0.1)$	(0.7±0.3)	(0.9±0.5)	(0.8±0.4)	(0.5±0.2)
Hip + DFoot	$0.88\pm0.06$	$0.93\pm0.31$	0.3±0.2	0.9±0.3	1.1±0.6	1.0±0.5	0.5±0.2
		$(0.62 \pm 0.16)$	$(0.2\pm0.1)$	(0.7±0.3)	(1.0±0.6)	(0.8±0.5)	(0.4±0.2)
DWrist + DThigh	$0.89\pm0.03$	$0.88\pm0.18$	0.4±0.3	0.9±0.3	1.0±0.5	0.9±0.4	0.6±0.2
		$(0.60 \pm 0.13)$	(0.3±0.2)	(0.8±0.3)	(0.9±0.5)	(0.8±0.4)	(0.5±0.2)
DWrist + DFoot	$0.89\pm0.03$	$0.91\pm0.22$	0.4±0.3	0.9±0.3	1.0±0.6	0.9±0.5	0.5±0.2
		$(0.61 \pm 0.13)$	(0.3±0.2)	(0.8±0.3)	(0.9±0.5)	$(0.8\pm0.4)$	(0.5±0.2)
Нір	$0.87\pm0.05$	$0.96\pm0.24$	0.4±0.2	0.9±0.3	1.1±0.5	1.0±0.5	0.7±0.3
		$(0.64 \pm 0.14)$	(0.3±0.2)	(0.8±0.3)	(0.9±0.5)	$(0.8\pm0.4)$	(0.5±0.2)
DWrist	$0.81\pm0.07$	$1.12\pm0.23$	0.7±0.4	1.1±0.4	1.2±0.6	1.1±0.5	0.9±0.4
		$(0.74 \pm 0.15)$	(0.4±0.2)	(0.9±0.3)	(1.0±0.5)	$(0.9\pm0.4)$	(0.7±0.3)
DFoot	$0.87\pm0.04$	$0.97\pm0.26$	0.4±0.2	1.1±0.4	1.1±0.6	1.0±0.5	0.6±0.2
		$(0.66 \pm 0.16)$	(0.3±0.1)	(0.9±0.3)	(0.9±0.5)	(0.9±0.4)	(0.5±0.2)
DUpperArm	$0.82\pm0.04$	$1.08\pm0.22$	0.5±0.3	1.1±0.4	1.2±0.5	1.2±0.5	0.8±0.3
		$(0.72 \pm 0.15)$	(0.4±0.2)	(0.9±0.3)	(1.0±0.5)	(0.9±0.4)	(0.6±0.3)
DThigh	$0.88\pm0.03$	$0.91 \pm 0.21$	0.4±0.3	1.0±0.4	1.0±0.5	0.9±0.4	0.6±0.2
		$(0.62 \pm 0.14)$	$(0.3\pm0.2)$	$(0.8\pm0.3)$	$(0.9\pm0.5)$	$(0.8\pm0.4)$	$(0.5\pm0.2)$

Table 5-96: Root mean squared error and mean absolute error (shown in parenthesis) obtained by estimating energy expenditure by first recognizing 51 activities in a subject dependent manner and then applying a multivariable linear regression model per activity trained in a subject independent manner. The *fast to compute* feature set is used to recognize activities and the *ACAbsArea* feature to estimate energy expenditure. Both feature sets are computed per sensor over windows of 5.6s in length.

Sensor Combination	Correlation	All	Postures	Ambulation	Exercise	Resistance	Household
						Exercise	
All sensors	$0.82 \pm 0.08$	$1.21 \pm 0.41$	0.4±0.2	1.1±0.6	1.4±0.8	1.1±0.5	0.7±0.3
		$(0.70 \pm 0.17)$	(0.3±0.1)	(0.9±0.4)	(1.1±0.6)	(0.9±0.4)	(0.5±0.2)
Hip + DWrist +	$0.88\pm0.03$	$0.99\pm0.27$	0.4±0.2	1.0±0.4	1.2±0.6	1.0±0.5	0.6±0.2
DFoot		$(0.66 \pm 0.18)$	(0.3±0.2)	(0.8±0.4)	(1.0±0.5)	(0.8±0.4)	(0.5±0.2)
Hip + DWrist	$0.88\pm0.04$	$0.93 \pm 0.20$	0.4±0.3	1.0±0.4	1.1±0.5	0.9±0.4	0.6±0.2
		$(0.63 \pm 0.16)$	(0.3±0.2)	(0.8±0.3)	(0.9±0.4)	(0.8±0.4)	(0.5±0.2)
Hip + DFoot	$0.88\pm0.03$	$0.98\pm0.28$	0.3±0.2	1.0±0.4	1.2±0.6	1.0±0.5	0.6±0.2
_		$(0.65 \pm 0.17)$	(0.3±0.2)	$(0.8\pm0.4)$	(1.0±0.5)	(0.8±0.4)	(0.5±0.2)
DWrist + DThigh	$0.88\pm0.03$	$0.93\pm0.19$	0.3±0.2	0.9±0.3	1.1±0.6	1.0±0.5	0.6±0.2
		$(0.61 \pm 0.12)$	(0.2±0.2)	(0.8±0.3)	(0.9±0.5)	(0.8±0.4)	(0.5±0.2)
DWrist + DFoot	$0.86\pm0.06$	$1.05\pm0.44$	0.3±0.2	1.0±0.4	1.2±0.7	1.1±0.6	0.6±0.3
		$(0.65 \pm 0.15)$	(0.2±0.1)	(0.8±0.3)	(1.0±0.6)	(0.9±0.5)	(0.5±0.2)
Нір	$0.88\pm0.04$	$0.91\pm0.21$	0.4±0.3	1.0±0.4	1.0±0.5	0.9±0.4	0.7±0.3
		$(0.62 \pm 0.15)$	(0.3±0.2)	$(0.8\pm0.4)$	(0.9±0.4)	(0.7±0.3)	(0.5±0.2)
DWrist	$0.83\pm0.05$	$1.07\pm0.21$	0.6±0.4	1.0±0.3	1.2±0.5	1.1±0.4	0.9±0.4
		$(0.70 \pm 0.14)$	(0.4±0.2)	(0.8±0.3)	(1.0±0.4)	$(0.9\pm0.4)$	(0.6±0.3)
DFoot	$0.85\pm0.09$	$1.18\pm0.93$	0.3±0.2	1.0±0.4	1.3±1.0	1.2±0.9	0.6±0.3
		$(0.67 \pm 0.17)$	(0.3±0.1)	(0.8±0.3)	(1.0±0.7)	(0.9±0.6)	(0.5±0.2)
DUpperArm	$0.86\pm0.09$	$1.02\pm0.45$	0.4±0.2	1.0±0.3	1.2±0.7	0.9±0.4	0.6±0.2
		$(0.64 \pm 0.18)$	(0.3±0.2)	(0.8±0.3)	(1.0±0.7)	(0.8±0.4)	(0.5±0.2)
DThigh	$0.87\pm0.04$	$0.96 \pm 0.25$	0.3±0.2	1.0±0.3	1.1±0.5	1.0±0.5	0.7±0.3
		$(0.65 \pm 0.15)$	(0.3±0.2)	(0.8±0.3)	$(0.9\pm0.5)$	(0.9±0.4)	(0.5±0.2)

Table 5-97: Root mean squared error and mean absolute error (shown in parenthesis) obtained by estimating energy expenditure by first recognizing 51 activities in a subject dependent manner and then applying a multivariable linear regression model per activity trained in a subject independent manner. The *invariant reduced* feature set is used to recognize activities and the *ACFFTPeaks* feature to estimate energy expenditure. Both feature sets are computed per sensor over windows of 5.6s in length.

set. The linear regression models are trained using the ACAbsArea feature. Both features are computed per sensor over non-overlapping sliding windows of 5.6s in length. The table shows that overall performance and performance per activity is higher than when activities are recognized in a subject independent manner. This is an expected result since the overall accuracy obtained when the 51 activities are recognized in a subject dependent manner using the *fast to compute* feature set is ~80% (+35% higher). From Table 5-96, it can also be observed that the feature sets with higher performance are DWrist+DThigh (r=0.89, RMSE=0.88), DWrist+DFoot (r=0.89, RMSE=0.91), and Hip+DWrist (r=0.88, RMSE=0.91). Again, the fact that performance increases as the number of sensors decreases (contrary to the behavior observed when a single linear regression model is used) indicates that the amount of data available is insufficient to evaluate performance when a large number of features are computed (over several sensors). It would be expected that given more training data, the performance of the Hip+DWrist+DFoot and all sensors combinations would be higher than the one obtained for the DWrist+DThigh sensor combination. Still, the results obtained using the DWrist+DThigh sensor combination (r=0.89, RMSE=0.88) are outstanding since they are higher than the results obtained using heart rate data (ScaledHR feature) using subject dependent (r=0.83, RMSE=0.92) and independent evaluation (r=0.83, RMSE=1.01). Furthermore, the performance of the DWrist+DThigh sensor combination is higher than the best results obtained using a single linear regression model trained over the fast to compute feature set during subject independent evaluation (r=0.80, RMSE-1.18). The performance of the DWrist+DThigh combination is also close to the performance obtained during subject dependent evaluation of a single regression model trained using the ACFFTPeaks feature (r=0.87, RMSE=0.86). This confirms that activity dependent regression models indeed improve energy expenditure estimation.

When the performance of the Hip+DWrist+DFoot sensor combination is analyzed per activity during the use activity dependent regression models using the *invariant reduced* feature set (r=0.88, RMSE=0.99) and a single regression model (r=0.80, RMSE=1.18), it is found that the performance over the ambulation and resistance exercise categories increases 0.3MET when activity dependent models are used. One important advantage of estimating energy expenditure by utilizing activity dependent regression models is that 51 activities are simultaneously recognized with an overall accuracy of 45% during subject independent training and ~80% during subject dependent training. This could allow for the development of health related interventions that trigger based on context and/or physical activity level. Obviously, one disadvantage of such approach if subject dependent training is utilized is that end-users need to provide examples of the activities to recognize. Nonetheless, the amount of training data required can be as low as 2mins as discussed in Section 5.6.11.

Table 5-97 presents the results obtained when activities are recognized using the *invariant reduced* feature set and when energy expenditure is predicted using the *ACFFTPeaks* feature set. The table shows that the performance of the all sensors and the Hip+DWrist+DFoot sensor combinations is lower than the one obtained in Table 5-96. This is because the number of features increased from one feature per sensor (*ACAbsArea*) to ten features per sensor (*ACFFTPeaks*). The table also shows that the performance of all single sensors is higher than the performance obtained using the all sensors combination. This contradicts the results obtained when a single regression model
is used and intuition, since it has been already show that the computation of more features and the utilization of more sensors improve energy expenditure estimation. This indicates again that the available training data is insufficient to learn the large number of regression coefficients (features) during the utilization of activity dependent regression models. During this situation, the sensor at the hip achieves the highest performance overall (r=0.88, RMSE=0.91) and per activity. It is important to note that the performance of this sensor is higher than the performance for the same sensor in Table 5-96 (r=0.87, RMSE=0.96). This corroborates the fact that the inclusions of features that better capture motion signatures improve energy expenditure estimation.

In summary, this section has shown that the utilization of activity dependent linear regression models in estimation of energy expenditure indeed improves performance. In fact, energy expenditure can be estimated with higher performance than when heart rate data alone is utilized during subject independent evaluation. The performance obtained using activity dependent models is also close to the performance obtained using heart rate data alone during subject dependent evaluation. The section has also shown that the incorporation of features others than the *ACAbsArea* improve energy expenditure estimation. These results hold when there is enough training data to train the activity dependent regression models or the single regression models containing large number of predictor variables (features). Finally, another finding was that although activity dependent linear regression models achieve higher performance (with respect to the use of single models); they require more data during training.

# 5.6.11.3 How well Can Energy Expenditure be Estimated if the Number of Activities to Recognize is Reduced?

One of the main problems found in the previous section was that the amount of training data available per activity was not sufficient to reliably test the performance of dependent regression models when a large number of features are utilized during the recognition of 51 activities. The number of features can be large either because several features are computed or because a large number of sensors is used (e.g. seven). As a result, this section explores if reducing the number of activities being recognized (and thus, the number of regression models to train) from 51 to 11 improves energy expenditure estimation. The activities being recognized in this section are the same as the ones recognized in Section 5.4.9.4. These 11 activities are lying down, standing, sitting, kneeling, walking at 2mph and 3mph, running at 4, 5, and 6mph, and the moderate and vigorous MET intensity categories. Appendix A2 shows a detailed list of the activities that were merged into the *moderate* and *vigorous* intensity categories according to their associated number of METs from the Compendium of Physical Activities [122]. This set of activities is explored because most daily energy expenditure is spent in sedentary or light activities such as postures and ambulation. Moreover, when medical interventions are designed to foster an increase in physical activity levels, it is important to know if the target population is exercising at *moderate* or *vigorous* intensity levels. If they are, there may be no need for an intervention. However, if they are not, it might be important to know what activities are being performed (e.g. postures, ambulation type and intensity) to plan the intervention according.

Sensor	Correlation	All	Postures	Ambulation	Exercise	Resistance	Household
Combination						Exercise	
Fast to compute	$0.79\pm0.08$	$1.26\pm0.22$	0.5±0.3	1.4±0.6	$1.4\pm0.7$	1.2±0.5	0.9±0.3
All sensors		$(0.91 \pm 0.14)$	(0.4±0.2)	(1.3±0.6)	(1.3±0.7)	(1.1±0.5)	(0.7±0.3)
Fast to compute	$0.77\pm0.07$	$1.28\pm0.25$	0.5±0.2	1.3±0.6	1.4±0.7	1.3±0.6	1.0±0.4
Hip + DWrist +		$(0.92 \pm 0.16)$	(0.4±0.2)	(1.2±0.6)	(1.2±0.6)	(1.1±0.5)	(0.8±0.3)
DFoot							
Fast to compute	$0.71\pm0.07$	$1.43 \pm 0.23$	0.8±0.4	1.4±0.5	1.5±0.6	1.3±0.6	1.1±0.4
Hip		$(1.06 \pm 0.16)$	(0.6±0.4)	(1.2±0.5)	(1.3±0.6)	(1.2±0.5)	(0.9±0.4)
Invariant reduced	$0.78\pm0.12$	$1.24 \pm 0.31$	0.6±0.3	1.4±0.6	1.3±0.7	1.2±0.6	0.9±0.5
All sensors		$(0.86 \pm 0.21)$	(0.4±0.3)	(1.2±0.6)	(1.1±0.6)	(1.0±0.6)	(0.7±0.4)
Invariant reduced	$0.79\pm0.09$	$1.23 \pm 0.34$	0.5±0.2	1.4±0.7	1.4±0.7	1.3±0.6	0.9±0.5
Hip + DWrist +		$(0.86 \pm 0.19)$	(0.4±0.2)	(1.2±0.7)	(1.2±0.7)	(1.1±0.5)	(0.7±0.3)
DFoot							
Invariant reduced	$0.72\pm0.09$	$1.33 \pm 0.17$	0.7±0.4	1.3±0.5	1.4±0.6	1.3±0.5	1.1±0.4
Hip		$(0.96 \pm 0.13)$	(0.5±0.3)	(1.1±0.5)	(1.2±0.6)	(1.1±0.4)	(0.9±0.3)

Table 5-98: Root mean squared error and mean absolute error (shown in parenthesis) obtained when energy expenditure is estimated by first recognizing 11 activities in a subject independent manner and activity dependent linear regression models trained in a subject independent manner are applied. When activities are recognized using the *fast to compute* feature set, energy expenditure is predicted using the *ACAbsArea* feature. When the *invariant reduced* feature set is used to recognize activities, energy expenditure is predicted using the *ACFFTPeaks* feature.

Sensor	Correlation	All	Postures	Ambulation	Exercise	Resistance	Household
Combination						Exercise	
Fast to compute	$0.80\pm0.07$	$1.22\pm0.24$	0.3±0.2	1.3±0.5	1.3±0.7	1.2±0.6	0.9±0.3
All sensors		$(0.87 \pm 0.15)$	(0.3±0.1)	(1.2±0.5)	(1.2±0.7)	(1.0±0.6)	(0.7±0.3)
Fast to compute	$0.81\pm0.05$	$1.18\pm0.24$	0.3±0.2	1.2±0.4	1.3±0.6	1.1±0.6	0.9±0.4
Hip + DWrist +		$(0.84 \pm 0.15)$	(0.2±0.2)	(1.1±0.4)	(1.2±0.6)	(1.0±0.6)	(0.7±0.3)
DFoot							
Fast to compute	$0.79\pm0.07$	$1.21 \pm 0.25$	0.4±0.2	1.2±0.4	1.3±0.6	1.2±0.6	1.0±0.4
Hip		$(0.87 \pm 0.17)$	(0.3±0.2)	(1.0±0.4)	(1.2±0.6)	(1.1±0.5)	(0.8±0.4)
Invariant reduced	$0.81\pm0.08$	$1.15\pm0.22$	0.4±0.2	1.3±0.5	1.2±0.5	1.1±0.6	0.9±0.5
All sensors		$(0.78 \pm 0.16)$	(0.3±0.1)	(1.1±0.5)	(1.0±0.5)	(0.9±0.5)	(0.7±0.4)
Invariant reduced	$0.84\pm0.05$	$1.07\pm0.20$	0.4±0.2	1.2±0.5	1.2±0.5	1.0±0.5	0.8±0.4
Hip + DWrist +		$(0.77 \pm 0.15)$	(0.3±0.1)	(1.0±0.4)	(1.0±0.5)	(0.9±0.4)	(0.7±0.3)
DFoot							
Invariant reduced	$0.76\pm0.14$	$1.24\pm0.21$	0.4±0.3	1.1±0.5	1.3±0.6	1.2±0.5	1.1±0.6
Hip		$(0.85 \pm 0.13)$	(0.3±0.2)	$(0.9\pm0.4)$	(1.1±0.5)	(1.0±0.4)	(0.9±0.5)

Table 5-99: Root mean squared error and mean absolute error (shown in parenthesis) obtained when energy expenditure is estimated by first recognizing 11 activities in a subject dependent manner and activity dependent linear regression models trained in a subject independent manner are applied. When activities are recognized using the *fast to compute* feature set, energy expenditure is predicted using the *ACAbsArea* feature. When the *invariant reduced* feature set is used to recognize activities, energy expenditure is predicted using the *ACFFTPeaks* feature

Table 5-98 presents the results obtained when energy expenditure is predicted by first recognizing 11 activities in a subject independent manner and then applying activity dependent linear regression models. Activities are recognized using the C4.5 decision tree classifier trained over two sets of features: the *fast to compute* feature set and the *invariant reduced* feature set. Energy expenditure is estimated using the *ACAbsArea* feature when activities are recognized using the *fast to compute* feature set and using the *ACFFTPeaks* feature set when activities are recognized using the *invariant reduced* feature sets are computed per sensor over windows of 5.6s in length. Although the number of activities being recognized is 11, energy expenditure estimation performance is tested over the 51 activities contained in the MIT energy expenditure dataset.

Table 5-98 shows that the performance obtained using the *fast to compute* feature set is lower than the performance obtained in Table 5-94 when the 51 activities were recognized. For example, in Table 5-98 the correlation coefficient ranges between 0.71 and 0.79 and RMSE between 1.26 and 1.43 while for Table 5-94 the correlation

coefficient ranges between 0.71 and 0.82 and the RMSE between 1.12 and 1.40 for the fast to compute feature set. Performance is also slightly lower for the sensor at the hip in Table 5-98 for the *invariant reduced* feature set (r=0.72, RMSE=1.33) than the one obtained in Table 5-95 (r=0.77, RMSE=1.22). This indicates that energy expenditure is not improved by reducing the number of activities to recognize from 51 to 11. In fact, overall performance and performance per activity using the *fast to compute* and *invariant* reduced feature sets slightly degrades for most activities when the number of activities to recognize is reduced suggesting that energy expenditure better estimated as more activities are recognized. This is an intuitive result because as the number of activities to recognize increases, there are more linear regression models available to perform fine tuned energy expenditure predictions per activity. Nevertheless, Table 5-98 also shows that the performance obtained using the *invariant reduced* feature set slightly increases for the sensor combinations all sensors and Hip+DWrist+DFoot when the number of activities is reduced from 51 to 11. For example, the correlation coefficient in Table 5-98 ranges between 0.78 and 0.79 and RMSE between 1.23 and 1.24 for these sensor combinations while in Table 5-95 the correlation coefficient is 0.77 and the RMSE ranges between 1.31 and 1.32. Given the relatively large vector size of the invariant reduced feature set (13 features per sensor), the increase in performance is most likely caused due to an increase in the amount of training data available to train the reduced number of linear regression models for each activity. One would expect that given the availability of more training data (e.g. more subjects in the MIT energy expenditure dataset), overall performance and performance per activity would degrade as fewer numbers of activities are recognized as shown for the *fast to compute* feature set.

Table 5-99 presents the performance obtained when energy expenditure is predicted by first recognizing 11 activities in a subject dependent manner and then activity dependent linear regression models trained in a subject independent manner are applied. The feature sets explored in this table are the same as the ones explored in Table 5-98. Table 5-99 shows that the highest performance (r=0.84, RMSE=1.07) is obtained for the *invariant* reduced feature set computed over the accelerometers at the hip, dominant wrist, and dominant foot. The difference in performance of this feature/sensor combination with respect to subject independent recognition of activities is +0.05 for the correlation coefficient and -0.17MET for the RMSE. The fact that the *invariant reduced* feature set computed over the sensors at the hip, wrist and foot achieved the highest overall performance and performance per activity in Table 5-98 and Table 5-99, suggests that this is a good feature/sensor combination to use. This feature/sensor combination also achieved either the best performance or a performance very close to the highest one in Table 5-94 through Table 5-97. The good performance of the *fast to compute* feature set over the accelerometers at the hip, dominant wrist, and dominant foot is also considered a strong result since the computational requirements of this feature set are lower than the ones required by the invariant reduced feature set.

One obvious disadvantage of applying activity dependent regression models to estimate energy expenditure is that misclassifications generated by the activity

recognition algorithm could affect energy expenditure estimation. Some misclassification errors might not impact energy expenditure considerably since they would consist on activities similar (in the motion pattern) to the real activity being performed. For example, running at 4mph could be recognized when someone is running at 5mph. Consequently, the regression model for *running at 4mph* would be applied for *running at* 5mph. Although incorrect, energy expenditure estimates would be close since these activities are closely related. In other scenarios where the activity being performed is completely different from the ones on which the activity recognition algorithm was trained, energy expenditure prediction could be off. For example, *dancing* could be confused with running if the activity recognizer is only trained on postures and ambulatory activities. In this scenario, the energy expenditure predictions would be most likely significantly off from the true values associated with the real activity being performed. Obviously, the easiest way to handle this situation is to train the activity recognizer over a large set of mutually exclusive activities (the ones most likely to be performed by individuals). Another possibility to handle this scenario is to train the activity recognition algorithm over an 'unknown' activity. Then, when the activity being performed considerably differs from the ones available during training, the unknown activity could be predicted and a single generic linear regression model trained over a large variety of activities can be applied. A different technique to address this problem could be to mark periods of time where the unknown activity is recognized for extended periods of time so that individuals can be latter prompted for the label of the true activity performed (thus improving activity labeling and energy expenditure estimates). During the activity dependent algorithms presented in this section, the unknown class was not incorporated in the regression models because a relatively large set of mutually exclusive activities was being recognized (51 and 11 activities).

# 5.6.11.4 How Well Can Energy Expenditure be Estimated by Predicting Mean Values per Activity?

One of the main problems found when energy expenditure was estimated using linear regression models per activities in the previous sections was the limited amount of training data available to train the regression models in part due to the utilization of a large set of features. Therefore, this section explores the performance of estimating energy expenditure when no features are used. In other words, this section explores how well can energy expenditure be estimated by just predicting the average energy expenditure value associated with each activity. This has the main advantage of requiring very little training data since the average energy expenditure value per activity can be computed over a few set of examples.

Table 5-100 illustrates the results obtained when energy expenditure is estimated by first recognizing the activity being performed (out of a set of 51) in a subject independent manner and then predicting the average MET value associated with each activity recognized. The feature sets used to recognize activities are the *fast to compute* and the *invariant reduced* computed per sensor over windows of 5.6s in length. Average MET values for each activity are computed in a subject independent manner. As expected, the table shows that the best performance is obtained using the all sensors combination,

Sensor Combination	Correlation	All	Postures	Ambulation	Exercise	Resistance	Household
						Exercise	
Fast to compute	$0.83\pm0.05$	$1.09\pm0.28$	0.5±0.3	1.2±0.5	1.2±0.6	1.1±0.5	0.7±0.4
All sensors		$(0.76 \pm 0.20)$	(0.4±0.3)	(1.1±0.4)	(1.0±0.6)	(0.9±0.4)	(0.6±0.3)
Fast to compute	$0.81\pm0.07$	$1.12\pm0.30$	0.5±0.3	1.3±0.5	1.2±0.7	1.1±0.5	0.7±0.4
Hip + DWrist + DFoot		$(0.78 \pm 0.20)$	(0.4±0.3)	(1.1±0.5)	(1.1±0.6)	(1.0±0.5)	(0.6±0.3)
Fast to compute	$0.71\pm0.07$	$1.41 \pm 0.29$	0.5±0.3	1.5±0.5	1.5±0.7	1.4±0.6	1.0±0.4
Hip		$(0.98 \pm 0.21)$	(0.4±0.2)	(1.3±0.5)	(1.3±0.6)	(1.2±0.6)	(0.8±0.3)
Invariant reduced	$0.82\pm0.04$	$1.12 \pm 0.24$	0.5±0.3	1.2±0.4	1.2±0.6	1.1±0.5	0.7±0.3
All sensors		$(0.77 \pm 0.17)$	(0.4±0.2)	(1.0±0.4)	(1.1±0.6)	(1.0±0.5)	(0.5±0.2)
Invariant reduced	$0.80\pm0.08$	$1.15 \pm 0.31$	0.5±0.3	1.3±0.6	1.2±0.7	1.1±0.5	0.7±0.3
Hip + DWrist + DFoot		$(0.80 \pm 0.23)$	$(0.4\pm0.2)$	(1.2±0.5)	(1.0±0.6)	(1.0±0.5)	(0.6±0.3)
Invariant reduced	$0.75\pm0.06$	$1.27\pm0.27$	0.5±0.3	1.3±0.5	1.4±0.6	1.2±0.5	0.9±0.4
Hip		$(0.87 \pm 0.19)$	$(0.4\pm0.2)$	(1.1±0.4)	(1.2±0.6)	(1.1±0.5)	(0.6±0.3)

Table 5-100: Root mean squared error and mean absolute error (shown in parenthesis) obtained when energy expenditure is estimated by first recognizing the activity being performed (out of a set of 51) in a subject independent manner and then predicting the average MET value associated with the activity recognized. The feature sets used to recognize activities are the *fast to compute* and the *invariant reduced* computed per sensor over windows of 5.6s in length. Average MET values for each activity are computed in a subject independent manner.

Sensor Combination	Correlation	All	Postures	Ambulation	Exercise	Resistance Exercise	Household
Fast to compute	$0.90\pm0.04$	$0.83 \pm 0.23$	0.3±0.2	0.8±0.3	1.0±0.5	0.9±0.4	0.5±0.2
All sensors		$(0.57 \pm 0.14)$	$(0.2\pm0.1)$	(0.7±0.3)	(0.9±0.5)	(0.8±0.4)	(0.5±0.2)
Fast to compute	$0.90\pm0.03$	$0.84 \pm 0.23$	0.3±0.2	0.9±0.3	1.0±0.5	0.9±0.4	0.5±0.2
Hip + DWrist + DFoot		$(0.58 \pm 0.14)$	$(0.2\pm0.1)$	(0.7±0.3)	(0.9±0.5)	(0.8±0.4)	(0.5±0.2)
Fast to compute	$0.87\pm0.05$	$0.93 \pm 0.21$	0.4±0.2	0.9±0.3	1.0±0.5	0.9±0.4	0.7±0.3
Hip		$(0.63 \pm 0.14)$	(0.3±0.2)	(0.8±0.3)	(0.9±0.5)	(0.8±0.4)	(0.6±0.2)
Invariant reduced	$0.90\pm0.03$	$0.83 \pm 0.23$	0.3±0.2	0.8±0.3	1.0±0.5	0.8±0.4	0.5±0.2
All sensors		$(0.57 \pm 0.14)$	(0.2±0.2)	(0.7±0.3)	(0.9±0.5)	(0.7±0.4)	(0.5±0.2)
Invariant reduced	$0.90\pm0.04$	$0.84 \pm 0.23$	0.3±0.2	0.8±0.3	1.0±0.5	0.9±0.4	0.6±0.2
Hip + DWrist + DFoot		$(0.58 \pm 0.14)$	$(0.2\pm0.1)$	(0.7±0.3)	(0.9±0.5)	(0.8±0.4)	(0.5±0.2)
Invariant reduced	$0.88\pm0.04$	$0.90 \pm 0.22$	0.4±0.2	0.9±0.3	1.0±0.5	0.9±0.4	0.7±0.3
Hip		$(0.61 \pm 0.14)$	$(0.3\pm0.2)$	$(0.7\pm0.3)$	$(0.9\pm0.5)$	$(0.8\pm0.4)$	$(0.5\pm0.2)$

Table 5-101: Root mean squared error and mean absolute error (shown in parenthesis) obtained when energy expenditure is estimated by first recognizing the activity being performed (out of a set of 51) in a subject dependent manner and then predicting the average MET value associated with the activity recognized. The feature sets used to recognize activities are the *fast to compute* and the *invariant reduced* computed per sensor over windows of 5.6s in length. Average MET values for each activity are computed in a subject independent manner.

followed by the Hip+DWrist+DFoot and Hip sensor combinations. This is because the activity recognition algorithm recognizes activities better as the number of sensors is increased. Table 5-101 also follows the same trend during subject dependent evaluation, although the difference in performance when the number of sensors is reduced is lower (as found in Section 5.4.7 for subject dependent recognition of activities). In Table 5-100, the Hip+DWrist+DFoot sensor combination achieves a performance of r=0.81, RMSE=1.12 using the *fast to compute* feature set and a performance of r=80, RMSE=1.15 using the *invariant reduced* feature set. These results are better than the ones obtained when activity dependent linear regression models are used to estimate energy expenditure using the *fast to compute* (r=0.80, RMSE=1.17) and *invariant reduced* (r=0.77, RMSE=1.31) feature sets. The performance using the *ACAbsArea* feature (vector size of 3) is +0.01 units higher in the correlation coefficient and -0.05MET lower in the RMSE when energy expenditure is estimated by predicting the average METs values

associated with each activity than when activity dependent linear regression models are utilized. Although this improvement is small, the fact that performance did not decrease indicates that the utilization of activity dependent regression models is more important than the complexity of the models used to estimate energy expenditure. In other words, if activities can be reliably recognized, energy expenditure can be predicted well by only predicting the average number of METs associated with each activity.

Table 5-101 presents the results obtained when activities are recognized in a subject dependent manner. The results obtained for the *fast to compute* (r=0.9 and RMSE=0.84) and invariant reduced (r=0.90, RMSE=0.84) feature sets are higher than the ones obtained using subject independent recognition of activities. This is expected since activities are better recognized in a subject dependent manner. These results presented in Table 5-101 are the highest results obtained for energy expenditure estimation so far. These results improve the correlation coefficient in +0.21 units and RMSE in 0.48 METs with respect to the use of a single linear regression model. Performance per activity category is also improved between 0.3 and 0.4MET over the utilization of a single linear regression model. In fact, these results also improve the correlation coefficient ~+0.1 units and the RMSE ~0.2MET with respect to subject dependent estimation of energy expenditure using a single linear regression model. These results are outstanding since it means that energy expenditure can be estimated better by recognizing activities than by collecting subject specific energy expenditure data using expensive and intrusive laboratory equipment (indirect calorimeter). These results obtained in this section also improve the correlation coefficient +0.07 units and the RMSE in 0.08 MET with respect to subject dependent estimation of energy expenditure using the ScaledHR feature. This is also outstanding since subject dependent estimation of energy expenditure using heart rate data is currently considered one of the most accurate methods available to estimate energy expenditure. The main disadvantages of subject dependent estimation of energy expenditure using heart rate data has been the need of subject specific calibration of regression equations using data collected from an indirect calorimeter in laboratory settings and the need to wear an uncomfortable heart rate monitor at the chest.

One disadvantage of estimating energy expenditure by recognizing activities and predicting average MET values per activity in practice is that steady-state energy expenditure values are predicted even when activities are performed for a short time. This is a problem particularly for physically demanding activities such as *ascending stairs* or *sit-ups* where energy expenditure increases constantly over time until steady state energy expenditure is reached. For example, if someone *ascends stairs* for a short time (e.g. the short stairs located at the entry door of a building), the energy expenditure predicted for this short duration activity would be close to the energy expenditure of that activity at steady state (as if the activity was performed over a long period of time). Consequently, energy expenditure would be overestimated for physically demanding activities performed for short durations of time. Chapter 6 discusses how this situation might be overcome in future work.

#### 5.6.11.5 Summary of Results

Table 5-102 presents a brief summary of the results obtained for subject independent estimation of energy expenditure using the algorithms implemented in this chapter. In

Method	Feature	Correlation	All	Postures	Ambulation	Exercise	Resistance Exercise	Household
Single	East to	$0.69 \pm 0.12$	$1.33 \pm 0.33$	0.6+0.3	1 2+0 5	1 3+0 7	1 2+0 6	0.9+0.6
IR	compute	$0.09 \pm 0.12$	$(0.99 \pm 0.33)$	$(0.5\pm0.3)$	$(1.1\pm0.5)$	$(1.2\pm0.7)$	$(1.1\pm0.6)$	$(0.8\pm0.6)$
Single	Invariant	$0.70 \pm 0.11$	$131 \pm 0.24$	$0.6\pm0.3$	1 2+0 5	$1.2\pm0.7$	12+0.7	0.8±0.5
LR	reduced	0.70 ± 0.11	$(0.97 \pm 0.23)$	$(0.5\pm0.3)$	(1.0+0.5)	(1.3+0.8)	(1.1+0.6)	(0.7+0.5)
Single	ScaledHR	$0.83 \pm 0.09$	$102 \pm 0.22$	0.5+0.3	0.9+0.5	1 3+0 7	12+06	0.5+0.3
LR	Statuarit	0100 - 0109	$(0.8\pm0.2)$	$(0.5\pm0.3)$	$(0.9\pm0.4)$	$(1.3\pm0.7)$	$(1.1\pm0.6)$	$(0.5\pm0.2)$
Single	Fast to	$0.88 \pm 0.05$	$0.89 \pm 0.22$	0.4+0.3	0.8+0.4	1.1+0.6	0.9+0.5	0.5+0.3
LR	compute		$(0.7 \pm 0.2)$	$(0.4\pm0.3)$	$(0.7\pm0.4)$	(1.0±0.6)	$(0.9\pm0.5)$	$(0.5\pm0.3)$
	ScaledHR		· · · · ·	Ň,	``´´´		× /	
Single	Invariant	$0.88\pm0.05$	$0.88 \pm 0.23$	0.5±0.3	0.8±0.4	1.1±0.6	0.9±0.5	0.5±0.3
LR	reduced		$(0.7 \pm 0.2)$	(0.4±0.3)	(0.7±0.3)	(1.0±0.6)	(0.8±0.5)	(0.5±0.2)
	ScaledHR							
51 activities	Fast to	$0.80\pm0.09$	$1.17\pm0.35$	0.4±0.3	1.2±0.5	1.3±0.7	1.2±0.6	0.7±0.4
ARSI LR	compute		$(0.79 \pm 0.21)$	(0.4±0.3)	(1.0±0.5)	(1.1±0.7)	(1.0±0.6)	(0.6±0.3)
51 activities	Invariant	$0.77 \pm 0.11$	$1.31 \pm 0.40$	0.6±0.4	1.5±0.7	$1.4\pm0.8$	1.3±0.6	0.9±0.5
ARSI LR	reduced		$(0.87 \pm 0.25)$	(0.5±0.3)	(1.3±0.6)	(1.2±0.7)	(1.0±0.5)	(0.6±0.3)
51 activities	Fast to	$0.89\pm0.04$	$0.93\pm0.28$	0.3±0.2	0.9±0.3	1.1±0.6	0.9±0.5	0.5±0.2
ARSD LR	compute		$(0.61 \pm 0.15)$	(0.2±0.1)	(0.7±0.3)	(1.0±0.6)	(0.8±0.5)	(0.4±0.2)
51 activities	Invariant	$0.88\pm0.03$	$0.99\pm0.27$	0.4±0.2	$1.0\pm0.4$	$1.2\pm0.6$	1.0±0.5	0.6±0.2
ARSD LR	reduced		$(0.66 \pm 0.18)$	(0.3±0.2)	(0.8±0.4)	(1.0±0.5)	(0.8±0.4)	(0.5±0.2)
11 activities	Fast to	$0.77\pm0.07$	$1.28\pm0.25$	0.5±0.2	1.3±0.6	$1.4\pm0.7$	1.3±0.6	1.0±0.4
ARSI LR	compute		$(0.92 \pm 0.16)$	(0.4±0.2)	(1.2±0.6)	(1.2±0.6)	(1.1±0.5)	(0.8±0.3)
11 activities	Invariant	$0.79\pm0.09$	$1.23\pm0.34$	0.5±0.2	$1.4\pm0.7$	$1.4\pm0.7$	1.3±0.6	0.9±0.5
ARSI LR	reduced		$(0.86 \pm 0.19)$	$(0.4\pm0.2)$	(1.2±0.7)	(1.2±0.7)	(1.1±0.5)	(0.7±0.3)
11 activities	Fast to	$0.81\pm0.05$	$1.18\pm0.24$	0.3±0.2	$1.2\pm0.4$	1.3±0.6	1.1±0.6	0.9±0.4
ARSD LR	compute		$(0.84 \pm 0.15)$	(0.2±0.2)	(1.1±0.4)	(1.2±0.6)	(1.0±0.6)	(0.7±0.3)
11 activities	Invariant	$0.84\pm0.05$	$1.07\pm0.20$	$0.4\pm0.2$	$1.2\pm0.5$	$1.2\pm0.5$	$1.0\pm0.5$	$0.8\pm0.4$
ARSD LR	reduced		$(0.77 \pm 0.15)$	$(0.3\pm0.1)$	(1.0±0.4)	(1.0±0.5)	$(0.9\pm0.4)$	(0.7±0.3)
51 activities	Fast to	$0.81\pm0.07$	$1.12\pm0.30$	0.5±0.3	1.3±0.5	$1.2\pm0.7$	1.1±0.5	0.7±0.4
ARSI mean	compute		$(0.78 \pm 0.20)$	(0.4±0.3)	(1.1±0.5)	(1.1±0.6)	(1.0±0.5)	(0.6±0.3)
51 activities	Invariant	$0.80\pm0.08$	$1.15\pm0.31$	0.5±0.3	1.3±0.6	$1.2\pm0.7$	$1.1\pm0.5$	0.7±0.3
ARSI Mean	reduced		$(0.80 \pm 0.23)$	$(0.4\pm0.2)$	$(1.2\pm0.5)$	$(1.0\pm0.6)$	(1.0±0.5)	(0.6±0.3)
51 activities	Fast to	$0.90 \pm 0.03$	$0.84 \pm 0.23$	0.3±0.2	0.9±0.3	1.0±0.5	0.9±0.4	0.5±0.2
ARSD Mean	compute		$(0.58 \pm 0.14)$	$(0.2\pm0.1)$	(0.7±0.3)	$(0.9\pm0.5)$	$(0.8\pm0.4)$	(0.5±0.2)
51 activities	Invariant	$0.90 \pm 0.04$	$0.84 \pm 0.23$	0.3±0.2	0.8±0.3	$1.0\pm0.5$	$0.9\pm0.4$	0.6±0.2
ARSD Mean	reduced		$(0.58 \pm 0.14)$	$(0.2\pm0.1)$	(0.7±0.3)	$(0.9\pm0.5)$	$(0.8\pm0.4)$	$(0.5\pm0.2)$

Table 5-102: Root mean squared error and mean absolute error (shown in parenthesis) obtained when estimating energy expenditure using different methods and two sets of features (*fast to compute* and *invariant reduced*) computed over the accelerometers located at the hip, dominant wrist, and dominant ankle. ARSI stands for subject independent recognition of activities, ARSD for subject dependent recognition of activities, and LR for linear regression. Energy expenditure is always predicted in a subject independent manner.

Table 5-102 'LR' stands for multivariable linear regression, 'ARSI' for subject independent recognition of activities, 'ARSD' for subject dependent recognition of activities, and 'Mean' for estimation of energy expenditure by predicting average MET values per activity. From the table, it can be seen that energy expenditure can be best predicted by first recognizing activities in a subject dependent manner and then predicting the average energy expenditure associated with each activity in METs (51 activities ARSD Mean). This technique improves the correlation coefficient +0.01units and RMSE 0.12MET over the utilization of linear regression models per activity (51 activities ARSD LR) using either the *fast to compute* or *invariant reduced* feature sets. This is an important result since highlights the fact that activity recognition is more important than the complexity of the model used to estimate energy expenditure. This is also supported by the fact that performance does not improve when the number of activities to recognize is reduced from 51 to 11 during subject dependent and independent

recognition of activities (11 activities ARSD and ARSI LR). This, suggests that energy expenditure is better predicted when more activity dependent regression models are used. The table also shows that subject dependent recognition of 51 activities (and the use of activity dependent regression models) outperforms energy expenditure estimation using heart rate data (*ScaledHR*) using a single linear regression model. This is also an important result since chest strap heart rate monitors, the most commonly available type of heart rate monitor, are uncomfortable to wear and can cause skin irritation when used longitudinally. Finally, estimation of energy expenditure by recognition of activities has the advantage of providing contextual information that might be valuable for real-time interventions designed to foster increases in physical activity levels.

This section also found that given enough training data, energy expenditure estimation improves as the number of wearable sensors used is increased. This is an intuitive result since more sensors are able to better capture the motion characteristics of individuals. Nevertheless, it was also found that a performance close to the one obtained using all the seven accelerometers (explored in this work) can be obtained using just three sensors located at the hip, dominant wrist, and dominant foot or just two sensors worn at the dominant wrist and dominant foot. These sets of sensors are able to capture upper body, lower body and overall body motion well to produce good estimates of energy expenditure for most of the activities explored in this work (51). In general, when single sensors are used to estimate energy expenditure, the highest performance is obtained from sensor located at the foot, hip, and thigh, and the worse performance is obtained from the sensor located at the dominant wrist. The sensor at the dominant wrist obtains the worse performance perhaps due to the high motion variability experienced at the wrists during most activities. This section also found that given sufficient amount of training data, the computation of features that capture additional information than overall amount of motion (ACAbsArea feature) such as the ACFFTPeaks, ACModVigEnergy, and the ACMCR improve energy expenditure estimates, particularly when single sensors are used to estimate energy expenditure. Finally, the section found that when activity dependent models are not used, energy expenditure is best estimated using heart rate data alone and that the combination of accelerometer and heart rate data improves energy expenditure estimation (although only slightly over the dataset utilized in this analysis).

#### 5.6.12 How Well Can Time Spent in Physical Activity Level be Recognized?

This section explores how well the energy expenditure estimation algorithms explored in this section can recognize time spent in *sedentary*, *light*, *moderate*, *vigorous*, and *very vigorous* physical activity. This is achieved by thresholding the energy expenditure estimates provided by the algorithms using the thresholds shown in Table 5-103. These MET thresholds were obtained from the study performed by Cradock et al. [251]. These threshold values are commonly utilized by the medical community during research studies. To test the recognition accuracy, predicted MET intensity levels (computed by thresholding the estimated energy expenditure) were compared against the ground truth physical activity levels obtained by thresholding the energy expenditure readings collected using the Cosmed K4b2 [125] indirect calorimeter.

Physical Activity Level	MET Range
Sedentary	≤ 1.5
Light	> 1.5 and < 3.0
Moderate	$\geq$ 3.0 and $\leq$ 6
Vigorous	$> 6 \text{ and } \le 9$
Very Vigorous	> 9

Table 5-103: Classificat	ion of physical	activity	level into	sedentary,	light,	moderate,	vigorous,	and
very vigorous by thresho	lding the MET	intensity	v value.					

Method	Feature set	Sedentary (%)	Light (%)	Moderate (%)	Vigorous (%)	Very Vigorous (%)	Total (%)
Single LR	Fast to compute	31.9	59.7	82.8	30.4	0.0	57.9
Single LR	Invariant reduced	43.3	60.3	80.5	32.5	0.0	60.5
Single LR	ScaledHR	53.7	71.1	76.2	29.9	38.7	65.6
Single LR	Fast to compute ScaledHR	58.0	65.5	85.9	49.7	0.0	70.0
Single LR	Invariant reduced ScaledHR	60.4	71.3	80.2	50.8	29.3	70.1
51 activities ARSI LR	Fast to compute	78.4	65.1	64.8	41.3	0.0	65.9
51 activities ARSI LR	Invariant reduced	77.4	65.9	60.3	44.8	2.7	64.7
51 activities ARSD LR	Fast to compute	81.2	73.6	75.1	46.2	0.0	73.3
51 activities ARSD LR	Invariant reduced	81.6	71.9	71.2	50.0	13.7	71.9
11 activities ARSI LR	Fast to compute	73.0	54.5	76.6	30.4	0.0	64.2
11 activities ARSI LR	Invariant reduced	72.3	58.6	72.5	39.0	8.0	64.8
11 activities ARSD LR	Fast to compute	69.6	61.3	81.4	38.4	0.0	67.9
11 activities ARSD LR	Invariant reduced	68.1	64.5	80.0	44.4	6.7	68.7
51 activities ARSI mean	Fast to compute	75.5	62.0	68.9	39.5	0.0	65.4
51 activities ARSI Mean	Invariant reduced	74.5	65.0	69.0	36.0	0.0	65.9
51 activities ARSD Mean	Fast to compute	77.6	72.2	79.6	44.7	0.0	73.3
51 activities ARSD Mean	Invariant reduced	77.5	73.2	79.6	44.1	0.0	73.6

Table 5-104: Accuracy in recognizing time spent sedentary, light, moderate, vigorous, and very vigorous physical activity using the different energy expenditure estimation algorithms explored in this section when accelerometers at the hip, dominant wrist, and dominant foot are utilized.

Table 5-104 presents the accuracy of recognizing the different physical activity levels using the algorithms explored in this section. It can be seen that the highest overall performance is obtained by recognizing 51 activities in a subject dependent manner using the *invariant reduced* feature set and predicting the average MET value associated with each activity (51 Activities ARSD Mean). This algorithm achieves an overall recognition accuracy of 73.6%. It is outstanding that this algorithm achieves a higher overall performance than when the *ScaledHR* feature is utilized (65.6%) and that when the *ScaledHR* feature is combined with the *fast to compute* (70%) and *invariant reduced* 

(70.1%) feature sets using a single linear regression model.

This algorithm (51 Activities ARSD Mean) also achieves a relatively good performance over most physical activity levels (ranging from 44.1% to 79.6%) except for the very vigorous intensity level (accuracy of zero). This intensity level is only recognized relatively well using a single linear regression model trained over the ScaledHR feature (38.7% accuracy). In fact, the recognition accuracy for this intensity level is only different from zero when the ScaledHR feature or the invariant reduced feature sets are utilized in combination with linear regression models to estimate energy expenditure. This is because when this intensity level is reached, it is confused with the vigorous and moderate intensity levels. The reason is that energy expenditure reached the very vigorous intensity level mostly during the cycling hard at 80rpm activity, which involves resistance work or load effort. Thus, energy expenditure for this activity is under predicted when most accelerometer-based features were utilized. The only accelerometerbased feature that is able to recognize the very vigorous intensity level is the invariant reduced feature set because it combines the ACFFTPeaks and ACBandEnergy features that detect some of the extra physical effort involved during the cycling hard at 80rmp activity. Heart rate data on the other hand (ScaledHR feature), is able to better estimate energy expenditure for the cycling hard at 80rmp activity and thus, able to better recognize the very vigorous intensity level. The very vigorous intensity level was only reached during 0.4% of the time (75 windows out of 17746) during the data collections.

The second highest performance in Table 5-104 is obtained when 51 activities are recognized in a subject dependent manner and activity dependent linear regression models (trained in a subject independent manner) are utilized to estimate energy expenditure. In This scenario, the *fast to compute* feature set achieves a higher accuracy than the *invariant reduced* feature set (+1.4%) but the *invariant reduced* feature set is able to recognize the very vigorous intensity level to some extent (13.7%). An important advantage of the fast to compute feature set over the invariant reduced feature set is that its computational complexity is significantly lower since the ACFFTPeaks feature is not computed (the fast Fourier transformation does not need to be computed). The difference in performance between these algorithms (51 Activities ARSD) and the one obtained using a single linear regression model trained over the same accelerometer-based features (fast to compute and invariant reduced) ranges between +11.4% and +15.4%. The difference in performance decreases to lie between +4.2% and +8% (with respect to a single linear regression model) when activities are recognized in a subject independent manner (51 Activities ARSI). Utilizing a linear regression model per activity improves recognition of some physical activity levels over prediction of mean values per activity. For example, the *sedentary* and *vigorous* intensity levels are estimated with an accuracy +4% higher than when mean MET values are predicted for each activity recognized. This is because a linear regression is able to better model the energy expenditure associated with these intensity levels.

Finally from Table 5-104, it can be seen that the lowest recognition accuracy is obtained using a single linear regression model trained over the *fast to compute* and *invariant reduced* feature sets (57.9 - 60.5%). Therefore, the main result that can be drawn from Table 5-104 is that activity dependent regression models also improve recognition of time spent in different physical activity levels, particularly when activities are recognized in a subject dependent manner.

## **6** Conclusions

This chapter presents a summary of the major findings and results presented in this work, including a summary of the primary contributions and a discussion on how well the design goals stated in Section 5.1 were met. In addition, some areas for future research are identified.

### **6.1 Summary of Results**

Presently, the most accurate technologies to measure energy expenditure are only suitable for laboratory settings (e.g. room or portable indirect calorimeters) due to their cost, intrusiveness, and complexity. Other technologies more amenable for free-living such as paper and electronic diaries are burdensome and time consuming. Accelerometers offer a promising strategy to recognize physical activity and estimate energy expenditure in freeliving people. However, existing accelerometer-based devices provide little or no contextual information (activity information) or have difficulties detecting upper or nonambulatory lower body activity or do not make the data available in real-time. As a result, the goal of the work presented in this thesis was to develop algorithms based on wireless accelerometers to (1) recognize activity type, intensity, and duration and (2) estimate energy expenditure while achieving a reasonable real-time performance from sensors worn at convenient locations. This work explored the trade-offs that needed to be made in order to achieve these goals.

#### 6.1.1 Activity Recognition

There exists a large body of prior work in recognizing activities from accelerometer data. However, most of this work relies on researchers as subjects or evaluates the algorithms implemented over few subjects (often less than five). Furthermore, researchers often explore a fixed set of activities (often less than 15) and select the algorithm parameters (e.g. classifier, feature set, window length, etc) as well as the type, number and placement of the sensors using common sense depending on the activities being explored. Finally, most prior work has been evaluated off-line and few real-time implementations of the algorithms presented exist (see section 3 for a full discussion of prior work and a discussion of some of the exceptions to the statements made in this paragraph).

The main contributions of this work to the area of activity recognition with respect to prior work are: (1) to explore the recognition of 52 activities collected from 20 non-researchers (120hrs of data) at a gymnasium and a residential home, (2) recognize the activity type and intensity, (3) perform a set of systematic experiments to determine the algorithm parameters, value of accelerometers vs. heart rate, location and number of sensors, and (4) to perform a proof of viability of a real-time system to recognize arbitrary activities.

The final activity recognition algorithm implemented in this work uses three accelerometers located at the hip, dominant wrist, and dominant foot and consists on the

following steps: (1) segmentation of the accelerometer data into 5.6s overlapping windows. (2) Interpolation of the signals over each window using cubic splines interpolation to compensate for values lost during wireless transmission. (3) Digital filtering of each acceleration axis using a band-pass filter (0.1-20Hz) and low-pass filter (1Hz) to separate motion from posture information. (4) The computation of features over each band-pass filtered acceleration axis such as the variance to capture the signal variability, the top five peaks of the FFT coefficients to capture the frequency or periodicity of motion, the energy between 0.3-3.5Hz to capture the intensity of activity, and the distances between each low-pass filtered axis (e.g. xz, xy, yz) to capture posture information. (5) The training of a C4.5 decision tree classifier using the features computed to create a set of rules to discriminate among the activities of interest. This algorithm was found by performing a set of systematic experiments evaluated in a subject dependent and independent manner and targeted to identify the set of parameters that allowed reasonable real-time performance, ease-of-use, and comfort of usage. The final feature set selected maximized performance and reduced dependency of the algorithm on small variations in the placement of the sensors on the body.

The final activity recognition algorithm presented achieved a subject independent accuracy of 50.6% and a subject dependent accuracy of 87.9% over a set of 51 activities. The large percent change of 73% from subject independent to subject dependent evaluation indicates that activities are easier to recognize if subjects provide examples of the activities. In general, it was found that the activities with higher performance were postures and exercises and the activities with poorer performance were household and resistance activities. Postures were relatively easy to recognize because they do not involve too much variability. Some exercises were relatively easy to recognize because they involved the use of a particular limb or a particular speed of execution. Household activities were more difficult to recognize due to their high variability (e.g. making the bed or wiping/dusting) and because they often involve sequences of postures and ambulation. Resistance activities presented a poorer performance because accelerometers have difficulties detecting resistance or work load effort. The activities confused most often were household activities, activities involving different intensity levels, and activities involving similar upper body motion. Activities with different intensity levels were confused because some intensity levels involved changes in resistance or work load that was poorly detectable from accelerometers. Activities with similar upper body motion were confused due to the high variability of motion found at the wrists during everyday life.

Due to the significantly higher performance (73% percent difference) of subject dependent training vs. subject independent training over 51 activities, experiments were performed to determine the amount of data required to achieve reasonable performance during subject dependent training. After partitioning each activity example into 75% training data and 25% testing data, and training the C4.5 classifier with varying amounts of training data, it was found that 60% of the training data achieved an overall accuracy of 76% over 51 activities. At this percentage of training data, most activities were 2 minutes in length except for physically demanding activities which were less than a minute in length. This indicated that a reasonable performance could be obtained by just providing 2 minutes of data per activity even when the number of activities to recognize was large (51).

The performance of the final activity recognition algorithm was also tested over subsets of activities that might be important to recognize during medical research studies or ubiquitous computing applications (e.g. 31, 11, 8 and 4 activities). It was found that if recognition of the intensity of activities is not important (e.g. if the intensity levels are merged into a single class), total overall accuracies of 72% and 91.4% can be obtained during subject independent and dependent evaluation respectively. This is an interesting result because it indicates that a total of 31 activities can be recognized with a subject independent accuracy of 72% in practice. The analysis also indicated that postures, ambulation (including intensity levels), and two MET intensity levels (moderate and vigorous) could be recognized with accuracies of 96.5% and 81.3% during subject dependent and independent evaluation respectively. Finally, four postures can be recognized with accuracies over 98% during both subject dependent and independent evaluation respectively.

When the performance of 11 subsets of sensors out of a total of seven sensors was explored, it was found that the best performing subset during both, subject dependent and independent training was the combination hip + dominant wrist + dominant foot. This sensor combination was able to capture upper, lower, and overall body motion that was important to recognize the 51 activities of interest. In general, when the number of activities to recognize was decreased, the number of sensors could also be decreased with relatively small decreases in performance. During subject dependent evaluation, it was found that a sensor located at the hip and another sensor located either at the dominant wrist (Hip+DWrist) or the dominant foot (Hip+DFoot) achieved reasonable performance (with respect to the use of all seven accelerometers). This is because upper body and lower body activity induced changes in the acceleration at the hip that allowed some degree of discrimination among upper body and lower body activities during subject dependent training. The highest performing single sensors during subject dependent training were the sensor at the hip, and the sensors at the lower body (foot and thigh). During subject independent evaluation, the best performing sensor combinations were those using a sensor located at the dominant wrist to detect upper body motion and another sensor located either at the dominant thigh or the dominant foot to detect lower body motion (DWrist+DThigh or DWrist+DFoot). The best performing single sensors during subject independent evaluation were DUpperArm, DWrist, and Hip. The good performance of the DUpperArm sensor might be explained by the fact that most activities contained in the MIT dataset included a high degree of upper body motion and ambulatory activities in the case of household activities.

It was also found that the length of the sliding window to utilize strongly depends on the activities to recognize. For example, postures are better recognized using short duration windows (<5.6s) and household activities with longer windows (e.g. >22s). This is because the variability of motion in postures is low and the variability in motion in household activities is high. The main disadvantage found for the use of long windows was low performance over short duration activities (e.g. physically demanding activities) and long real-time classification delays. Even though the window length depends on the activity to recognize, it is computationally expensive to have different window length for each activity since features have to be recomputed for each window of different length. Consequently, a single sliding window of 5.6s in length was selected in this work to maximize performance over most activities, reduce classification delay, and to improve performance over short duration activities.

Heart rate data was not used in the final implementation of the activity recognition algorithm because the performance obtained with the highest performing heart rate feature alone (ScaledHR) was 38.4% and 13.8% during subject dependent and independent evaluation respectively. This performance was considerably lower than the one obtained using the highest performing accelerometer-based features. When the highest performing heart rate feature (ScaledHR) was added to the highest performing accelerometer-based feature set (invariant reduced), it is found that overall accuracy increased only between 2-4% during both subject dependent and independent evaluation. As expected, the activities for which recognition improved most substantially with the incorporation of heart rate data were those involving resistance or work load effort (e.g. cycling, rowing, bicep curls, etc). Several reasons explain the poor performance of heart rate data. First, heart rate lags physical activity and remains altered once an activity has ended. This produces errors at the beginning and at the end of each activity. Moreover, activities for which heart rate data constantly increases as the activity is performed (e.g. physically demanding activities such as *ascending stairs*) do not present a single heart rate value characteristic of that activity. Finally, heart rate presents inter-individual variations due to fitness level (e.g. more physically fit individuals tend to have lower heart rate readings) and intra-individual variations due to emotional state, nicotine consumption, and even environmental temperature.

Finally, the activity recognition algorithm was implemented on a laptop computer and tested in real-time during a short pilot study. Five participants were asked to (1) wear three accelerometers at the hip, dominant wrist, and dominant foot, (2) to type in 10 physical activities, exercises, postures, or activities of their choice that they wanted the system to recognize, (3) to provide 2min of data for each activity, and (4) to test the performance of the recognition algorithm in real-time. The participants provided a variety of complex activities such as Taekwondo forms, scratching head, bowling, tennis serve, hammering a nail, applying cream, knitting, filing nails and drawing on a piece of paper. The cross-validation accuracy obtained during the study for each participant ranged from 79% to 92%. During the study, participants expressed the desire to provide the training data during free-living (and consequently not over a single tedious session of repeating an activity for 2min) and the desire to 'fix' the algorithms in real time when they did not perform as expected by providing more training examples of the problematic activities.

#### **6.1.2 Energy Expenditure Estimation**

The two most popular approaches to estimate energy expenditure employed by the medical community are the use of linear regression equations based on accelerometer data and the use of the Compendium of Physical Activities [122]. The use of linear regression equations based on accelerometer data consists on the following steps: (1) Collect accelerometer and ground truth energy expenditure data from a subject wearing an indirect calorimeter and a single accelerometer at the hip, (2) segment the accelerometer and ground truth data into one minute windows, (3) sum the accelerometer values (or counts) over the one minute intervals and compute the mean value over the

energy expenditure windows, and (4) create a single linear regression model to map the acceleration sums into the mean ground truth energy expenditure values. The Compendium of Physical Activities can be used to estimate energy expenditure by collecting information about the activity type being performed and its duration (e.g. using diaries). Once this information is collected, it can be converted to energy expenditure by using the mean energy expenditure values listed for a variety of activities in the Compendium of Physical Activities.

Presently, the state-of-the-art algorithm to estimate energy expenditure from an accelerometer at the hip is the work by Crouter et al [34]. This algorithm first classifies activities into *sedentary, ambulatory* and *lifestyle* activities and then applies different regression models depending on the type of activity detected. For example, if *sedentary* activities are detected, a MET value of one is predicted; if *ambulatory* activities are detected, an exponential regression model is applied. This algorithm achieves a correlation coefficient of 0.96, a maximum absolute error deviation of 0.75MET, and a RMSE of 0.73MET when evaluated over a set of 17 activities collected from 20 participants (3 hours per participant). However, the authors excluded the *cycling* activity from analysis because it did not generate any sensor readings at the accelerometer at the hip.

The work presented in this thesis extends the work by Crouter et al. by exploring the use of 51 activity-dependent regression models, the use of seven accelerometers and subsets of them, the exploration of 41 features computed over the raw accelerometer data, the use of shorter window lengths (less than a minute long), and the use of non-linear regression models in the estimation of energy expenditure. Finally, the algorithms implemented are evaluated over data collected from 16 individuals between 18 and 40 years old performing 52 activities at two locations: A gymnasium and a residential home.

After running the Crouter et al. algorithm over the dataset collected for this work, it was found that its performance (r=0.4, RMSE=2.7) on 52 activities was considerably lower than the one reported by the authors (r=0.92, RMSE=0.73) for 17 activities. One of the reasons for the poor performance obtained was the inability of the single accelerometer at the hip to capture upper body (e.g. *bicep curls*) and non-ambulatory lower body activity (e.g. cycling), because the sensor readings at the hip for these activities were mostly zero. When information about activity type and duration from the labels collected was converted to energy expenditure using the Compendium of Physical Activities, it was found that the performance obtained (r=0.9, RMSE=1.27) represented an improvement of ~100% for the correlation coefficient and ~50% for the RMSE with respect to the Crouter et al. algorithm. These results indicated that knowledge of the activities being performed was very important since this technique only predicted the average energy expenditure values listed in the Compendium of Physical Activities for each activity. In order to obtain a baseline on performance, a single linear regression model was created to estimate energy expenditure from the accelerometer sums computed over one minute windows for each of the seven accelerometers. The results obtained (r=0.73, RMSE=1.4) were also considerably higher than those obtained by Crouter et al. For example, the improvement over the correlation coefficient was 82% and over the RMSE was 48%. The improvements obtained were mainly due to the use of seven accelerometers located at different body segments that captured upper, lower, and overall body motion more fully than a single accelerometer at the hip. When activity-

dependent linear regression models were applied to estimate energy expenditure assuming the activity being performed was known, it was found that the performance obtained (r=0.87, RMSE=1) was also considerably higher than the performance obtained for the Crouter et al algorithm. For example, the improvement was 117% for the correlation coefficient and 63% for the RMSE. The improvement achieved over the use of a single linear regression model was 19% for the correlation coefficient and 28.5% for the RMSE. Activity-dependent regression models helped by allowing regression coefficients to be tuned for each activity instead of being globally optimized for all activities at once. Finally, activity dependent non-linear regression models (M5' model trees) were applied to estimate energy expenditure over the dataset assuming the activities performed were also known. The results obtained (r=0.91, RMSE=0.88) represented an improvement of 127% for the correlation coefficient and 67% for the RMSE with respect to the one obtained by Crouter et al. The improvement obtained over activity-dependent linear regression models was 5% for the correlation coefficient and 12% for the RMSE. The improvement over the use of a single linear regression model was 25% for the correlation coefficient and 37% for the RMSE.

In general, the lowest performance in energy expenditure estimation was obtained for activities involving resistance work or load effort. This is because accelerometers have difficulties detecting resistance work or load effort. The activities with highest performance were postures and household activities. This is because energy expenditure changed little for these activities with respect to the changes observed for physically intense activities such as gymnasium activities. The use of a single linear regression model trained over seven accelerometers overestimated energy expenditure for sedentary activities. This is because the regression coefficient representing the DC offset of the energy expenditure signal was increased during training to compensate for the high energy expenditure observed for physically intense activities involving resistance work (e.g. cycling). This single linear regression model also estimated energy expenditure well for upper body and lower body activity due to the use of additional (six) accelerometers at different body locations. The Compendium of Physical Activities was found to overestimate energy expenditure for household activities and short duration activities. This is because the energy expenditure value listed in the Compendium of Physical Activities is the one measured during steady-state conditions. Thus, energy expenditure is overestimated for household and short duration physically demanding activities because steady-state energy expenditure was not reached. The Compendium of Physical Activities was found to estimate energy expenditure better for activities that reached steady-state during the data collections such as *walking*, *running*, *cycling*, and *rowing*.

In summary, the first set of experiments on energy expenditure estimation indicated that accelerometers at the upper and lower body as well as activity-dependent regression models improved energy expenditure performance. A new set of systematic experiments was performed to explore if the number of accelerometers could be reduced, if heart rate data improved performance (over accelerometer data), and if the use of the activity recognition algorithm previously implemented could produced reasonable performance (with respect to the assumption of known activity). This new set of experiments as will be later explained in detail indicated that reasonable performance could be achieved by the use of only three accelerometers located at the hip, dominant wrist, and dominant foot. It was also found that the top five peaks of the FFT coefficients used as features and computed over sliding windows of 5.6s in length achieved a good compromise between performance and computational complexity.

When eleven subsets of the seven accelerometers were analyzed to find the most comfortable combination with highest performance, it was found that the combination of hip, dominant wrist, and dominant foot achieved the highest performance. The decrease in performance with respect to the use of all seven accelerometers was less than 3% for both the correlation coefficient and the RMSE during subject independent evaluation. The second best performing sensor subset was the combination of a sensor at the dominant wrist and another sensor at the dominant foot. Using this subset, the decrease in performance for both the correlation coefficient and the RMSE with respect to all seven accelerometers was approximately 3%. This is an important result since it can be argued that sensors at the wrist and at the foot are easier to wear since they can be embedded in convenient devices such as wristwatches and shoe pods. When the performance of single sensors was analyzed, it was found that the best single sensors to use depend on the activities with highest energy expenditure levels in the dataset. For example, in this work, the activity with highest energy expenditure was cycling hard. As a result, the highest performing single sensors were the sensors worn at the lower body: The sensor at the dominant foot and the sensor at the dominant thigh.

When the activity recognition algorithm was used to estimate energy expenditure using activity dependent regression models, it was found that the performance for subject independent recognition of activities was r=0.77, RMSE=1.31. Performance for subject dependent recognition of activities was r=0.88, RMSE=0.99. The performance obtained using subject dependent recognition of activities was close to the performance obtained when the activities were assumed to be known (<2%). This was an interesting result since the accuracy of recognizing activities in a subject dependent manner was only ~80%. The reason why performance was high despite this non-perfect accuracy was that the classification errors performed by the recognition algorithm involved activities similar in their motion patterns and thus, with similar energy expenditure values. As a result, even when errors were made, activity-dependent regression models similar to the ones of the 'true' activity being performed were applied thus achieving estimates close to the true values expected.

Experiments using heart rate data to estimate energy expenditure showed that the best single feature to use was the *ScaledHR* feature. This feature consisted of the number of heart beats per minute normalized to lie between zero and one for resting and running on a treadmill at 5mph. This feature achieved a performance of r=0.84, RMSE=1.01 over the dataset explored. This performance was better than any of the results obtained using accelerometer-based features. However, the performance obtained using subject dependent recognition of activities and activity-dependent linear regression models was as good (r=0.88, RMSE=0.99) as the one obtained using the *ScaledHR* feature alone. This indicates that activity dependent regression models improve energy expenditure estimation considerably and can compensate for the use of uncomfortable sensors such as chest strap heart rate monitors.

The work presented here also suggests that features that capture information other than overall amount of motion (*ACAbsArea* feature) improve performance, particularly when single sensors are used to estimate energy expenditure. For example, estimating energy expenditure using the top five FFT peaks, the band energy, and the mean crossing rate

improved r by 13% and RMSE by 21% over the use of the overall amount of motion feature computed from a single accelerometer at the hip. This work also found that the addition of heart rate data to the best accelerometer-based feature (top five FFT peaks) improved the correlation coefficient 22% and the RMSE 31%.

Finally, this work explored the performance of estimating energy expenditure by first recognizing the activities being performed using the implemented algorithm and then predicting the average energy expenditure value per activity as computed from the data collected. The performance of this technique during subject dependent recognition of activities was the highest obtained in this work (r=0.90, RMSE=0.84). This technique achieved an improvement in the correlation coefficient between 2-4% and an improvement on the RMSE between 12-15% with respect to the use of activity-dependent linear regression models. The improvement is modest, but the fact that performance did not degrade when such a relatively simple technique is used to estimate energy expenditure is notable. Estimating energy expenditure by recognizing activities and then predicting mean energy expenditure values per activity has some limitations. For example, this technique is likely to overestimate energy expenditure for short duration activities, particularly physically demanding ones. Steady-state energy expenditure might not be reached in this case. Another potential problem is that misclassifications could affect the energy expenditure estimates. For example, if the activity recognition algorithm is not trained to recognize *dancing* but it is trained to recognize *running*, *dancing* might be confused with *running* and, consequently, the mean energy expenditure for *running* would be predicted for *dancing*. These energy expenditure estimates could be significantly off, thus affecting the overall performance of energy expenditure estimation. Other classification errors might not affect energy expenditure considerably. Confusing running at 6mph with running at 5mph, for example, would produce energy expenditure estimates that although incorrect would still be close to the true energy expenditure value. One possible improvement might be to train the activity recognition algorithm on a larger set of mutually exclusive activities, the ones that end-users of the system are most likely to perform over the course of a day. Another possibility is to train the activity recognition algorithm to recognize an unknown activity. Then, when this unknown activity is recognized, a generic regression model can be applied to estimate energy expenditure. Another option is to ask the user at the end of the day for the real label of the activity performed when the *unknown* activity was recognized to better estimate energy expenditure and improve activity labeling. Finally, another potential problem is that activity recognition algorithms tend to generate spurious classifications while people transition between activities. Thus, it is necessary to filter these spurious classifications, perhaps utilizing information about how people transition between activities to reduce the errors introduced in the energy expenditure estimates.

#### **6.2 Primary Contributions**

In summary, the main contributions of this work to the field of activity recognition from wearable sensors are:

- 1. The recognition of 52 activities and subsets of these activities on data collected from 20 non-researchers. A dataset larger and more complex than the ones used in prior work to the best of the Author's knowledge.
- 2. The recognition of not only the type of activity but also the intensity of some activities.
- 3. The exploration of subject dependent recognition of activities using 2min of data as a promising strategy to recognize activities in practice.
- 4. The presentation of results that indicate that three sensors at hip, dominant wrist, and dominant foot offer a good compromise to recognize the 52 activities explored in this work, as well as arbitrary activities provided by users.
- 5. The presentation of results that indicate that acceptable activity recognition performance can be obtained without using heart rate data.
- 6. The proof of viability of a real-time system than can be trained to recognize arbitrary activities.
- 7. The presentation of an activity recognition algorithm that is amenable for realtime performance in low-processing power devices such as mobile phones.
- 8. The exploration of the impact of different parameters of the activity recognition algorithm such as the type of classifier (four types of classifiers explored), feature set (41 feature types explored), feature computation technique (e.g. per sensor vs. per axis), sliding window length (varied from 1s to 91s), signal processing techniques (e.g. digital filtering applied, interpolation technique), and sensor subsets required (eleven subsets of seven sensors explored).

The main contributions to the field of energy expenditure estimation from wearable sensors are:

- 1. The estimation of energy expenditure for 52 activities and some subsets of these activities on data collected from 20 non-researchers. A dataset larger and more complex than the ones used in prior work to the best of the Author's knowledge.
- 2. The presentation of results that indicate that activity-dependent models improve energy expenditure estimation performance over the use of single linear regression models trained over accelerometer or heart rate data alone. Furthermore, activity-dependent regression models achieved a performance similar to the one obtained by combining the highest performing accelerometer and heart rate features explored in this work when single regression models are used (e.g. differences in r of 0.04 and in RMSE of 0.02 were found).
- 3. The presentation of experiments that demonstrate that when activity-dependent models are used, prediction of mean energy expenditure values per activity outperforms the use of linear regression models per activity in the dataset explored in this work.
- 4. The presentation of results that strongly suggest that heart rate data outperforms the use of accelerometer data during the estimation of energy expenditure, at least in the dataset explored in this work.
- 5. The presentation of results that indicate that features that capture information additional to overall amount of motion (e.g. FFT peaks, frequency domain energy,

mean crossing rate) improve energy expenditure performance, particularly when single sensors are used to estimate energy expenditure.

- 6. The presentation of results that indicate that sensors at the dominant wrist and dominant foot are important to measure upper body and non-ambulatory lower body activity and corresponding energy expenditure. Reasonable performance can be obtained by using only these two sensors and no sensor at the hip.
- 7. The exploration of the impact of different parameters of the energy expenditure algorithm such as the type of regressor (four types of regressors explored), feature set (41 feature types explored), feature computation technique (e.g. per sensor vs. per axis), sliding window length (varied from 1s to 91s), signal processing techniques (e.g. digital filtering applied, interpolation technique), and sensor subsets required (eleven sensor subsets explored).

### **6.3 Revisiting the Design Goals**

This section briefly revisits the design goals stated at the beginning of this work and/or the evaluation measures neglected in most prior work to identify which ones were met and which ones were not met.

#### • Complexity of the activities to recognize:

- *Number of activities to recognize*: This work explored the recognition and energy expenditure estimation of a set of 52 diverse activities, a number that to the best of the author's knowledge is larger than any number of activities explored in prior work.
- *Complexity of the types of activities to recognize*: The 52 activities explored in this work contained 26 activities with different intensity levels due to different speeds of execution and resistance work, 18 household activities containing examples of unconstrained motion in a residential home, and activities involving upper, lower, and overall body motion. This work also evaluated the recognition performance over subsets of the 52 activities to identify how reducing the number of activities to recognize impacts performance.
- Complexity of the training data collected for the activities: The training data for this work was collected from 20 participants executing a total of 52 activities at two locations: A gymnasium and a residential home. The activities at the gymnasium were relatively constrained due to the use of gymnasium equipment such as a treadmill and stationary machines (e.g. cycling and rowing machines). However, the data collection at the residential home was less constrained since participants were allowed to perform the activities as they pleased.

#### • Training data requirements of an algorithm

 Subject independent recognition of activities: This work showed that a subject independent accuracy of 51% can be achieved over 52 activities. In contrast, the accuracy obtained using subject dependent training was 80%. However, if the intensity of physical activities does not need to be recognized, a subject independent accuracy of 72% can be achieved.

• Amount of training data required for subject dependent recognition of activities: Given that the performance of subject dependent training consistently outperformed the performance of subject independent training, experiments were performed to determine the minimum amount of training data required to achieve a reasonable performance. It was found that 2 minutes of data per activity achieved a total accuracy of 72% over 52 activities. Experiments using this amount of training data per activity were also performed in real-time where five individuals provided a list of 10 arbitrary activities that they wanted the system to recognize. The 10-fold cross-validation accuracies obtained during the short study ranged from 79% to 90%.

#### • Sensor data requirements for the algorithm

- *Number of sensors required to recognize activities*: This work explored the performance of eleven subsets of seven accelerometers to identify the best performing subset with most convenient locations. It was found that three sensors at the hip, dominant wrist, and dominant foot provided the highest performance during both activity recognition and estimation of energy expenditure (with respect to the use of all seven accelerometers).
- Intrusiveness of the sensors required to recognize activities: This work compared the performance of accelerometer data and heart rate data during the recognition of activities and estimation of energy expenditure. It was found that heart rate data performs poorly during activity recognition (e.g. accuracies of 14% during subject independent evaluation). During energy expenditure estimation, heart rate data performs well (e.g. r=0.84, RMSE=1.01) and better than accelerometer-based features. However, a performance of r=0.88, RMSE=0.99 can be obtained using activity dependent regression models based on accelerometer data alone. As a result, good performance can be obtained during both, activity recognition and estimation of energy expenditure without using heart rate data. This was an important result since heart rate monitors are presently uncomfortable to wear and intrusive.
- Location of the sensors required to recognize activities: Only three accelerometers at the hip, dominant wrist, and dominant foot were found to achieve reasonable performance during activity recognition and estimation of energy expenditure. For example, decreases in performance of less than 3% were observed during energy expenditure estimation and between 2-8% during recognition of activities using these three sensors with respect to the use of all seven accelerometers. Sensors at these locations could be conveniently embedded in objects such as wristwatches, shoe pods, belt clips or simply put inside the pocket in the case of the sensor at the hip.

#### • Usability factors imposed by the algorithm

• *Real-time recognition of activities*: The algorithm presented in this work to recognize activities achieved real-time performance on a 1GHz laptop

computer. The energy expenditure algorithm presented is also amenable for real-time performance due to its lightweight computational requirements. Presently the House\_*n* research group at MIT is working on implementing both algorithms for existing mobile phones.

• *Real-time recognition delay*: The algorithms presented to recognize activities and estimate energy expenditure have an overall real-time delay of 5.6 seconds. This delay is short enough to allow just-in-time interventions and real-time behavioral feedback.

In summary, it has been shown that the algorithms presented in this work offer a reasonable compromise between overall performance as evaluated using standard evaluation measures and evaluation measures often neglected in prior work such as complexity of the activities to recognize, training data requirements of the algorithms, sensor data requirements of the algorithms, and usability factors imposed by the algorithm.

## 6.4 Unresolved Issues for Long-Term Deployment

This section briefly discusses some of the unresolved issues for long-term deployment of activity recognition and energy expenditure estimation algorithms based on mobile phones during free-living.

#### 6.4.1 Hardware Limitations for Mobile Interventions

There exist some hardware limitations that need to be overcome in order to exploit mobile phone based interventions using activity recognition and energy expenditure estimation algorithms in practice. For example, sensors need to be built in small form factors so that they can be worn comfortably and unobtrusively over the course of a day. Existing semiconductor manufacturing technologies already allow the production of wireless sensor nodes that are as small as 6.6mm<sup>3</sup> [252]. However, two remaining challenges are (1) how to power the sensors continuously since batteries are presently inefficient and have relatively large form factors (with respect to the size of the sensor nodes that are as tiny as the sensor nodes themselves. Thus, it seems that a practical strategy would be to create sensors nodes that are relatively small (e.g. the size of an American quarter coin) with rechargeable batteries that last for at least a day. In this way, sensors could be embedded in objects such as wrist watches and shoe pods that people could recharge every day after use.

#### 6.4.2 Usability Issues and Impact on Algorithms

There also exist some usability challenges that need to be overcome in order to make mobile phone based interventions using activity recognition and energy expenditure estimation algorithms a reality in practice. For example, if subject dependent training is

used to recognize activities, intuitive and easy-to-use user interfaces would be required to allow individuals to interactively train and test the performance of the algorithms in realtime. It would also be necessary to come up with strategies to allow individuals to provide the necessary training data when they are actually engaged in the activities of interest during free living instead of providing simulated examples of the activities of interest over a single long session. Furthermore, it is expected that the activity recognition algorithms would inevitably make mistakes during the recognition of activities in free-living. Some possible reasons for these mistakes are insufficient training data for a given activity, motion similarities with other activities being recognized, or changes in the patterns of activity execution over time. Thus, there would be a need for strategies to allow individuals to 'fix' the recognition algorithm when it does not perform as expected. One possible strategy would be to allow individuals to provide more training examples for the problematic activities. Another strategy, although more challenging, would be to allow individuals to directly modify the activity models used by the algorithms to recognize activities. The challenge here would be to create activity recognition algorithms that utilize models that are easy to interpret by end-users with no technical background.

### **6.5 Future Research Directions**

This section discusses some research directions for future work based on some of the findings of this work and some of the limitations of the algorithms presented.

#### 6.5.1 Activity Recognition

Some future research directions in activity recognition are:

- Create intuitive and easy-to-use user interfaces to allow individuals train and test the performance of activity recognition algorithms in real-time.
- Create strategies and user interfaces to allow individuals to 'fix' the recognition algorithms when they do not perform as expected. This could be achieved by allowing individuals to provide more training examples for the problematic activities or by allowing individuals to directly modify the activity models utilized by the recognition algorithm.
- Use hierarchical models to recognize highly complex and variable activities such as household activities. For example, since household activities include sequences of postures and ambulation in their execution, it might be beneficial to first have a classifier to recognize postures and ambulation and later use the output of this classifier to better identify household activities. In this scenario, *wiping/dusting* might be better recognized by detecting sequences of postures, ambulation, and wiping dusting patterns executed during the activity duration.
- Use activity-dependent window lengths to improve the performance over activities of interest. For example, longer window lengths could be used to better capture the variability found in complex household activities. This technique is

obviously more computationally expensive but might bring benefits for activities that are challenging to recognize such as household activities.

- Repeat some or all of the experiments performed in this work over data collected from a given set of individuals over different days. This would allow testing the performance of the recognition algorithms when the position and orientation of the sensors changes slightly from day to day.
- Include hand gesture recognition as a high level feature in activity recognition. This strategy might be helpful because the wrists presented high motion variability during the data collections. For example, the household activity *wiping/dusting* might be easier to recognize if a hand gesture recognition algorithm first recognizes *high energy circular motion* and then the recognize *high energy circular motion* and then the recognize *wiping/dusting*.

#### 6.5.2 Energy Expenditure Estimation

Some future research directions in the area of energy expenditure estimation are:

- Incorporate information about the duration of activities when estimating energy expenditure. This technique might be useful for better estimating the energy expenditure associated with activities whose energy expenditure increases constantly over time. For example, the energy expenditure for *ascending stairs* might be better estimated if information about when the activity started is incorporated. In this way, the algorithm could predict that energy expenditure is expected to increase linearly if *ascending stairs* has been executed for less than 10 minutes but would reach steady state (a constant value) after being executed for longer than 10 minutes.
- Filter the energy expenditure estimates using information about how energy expenditure transitions from activity to activity. As discussed previously, activity recognition algorithms might generate spurious classifications while transitioning from one activity to another. These spurious transitions might be filtered by incorporating information on how people transition from activity to activity.
- Repeat some or all of the experiments presented in this work over data collected from more individuals, ideally more than one hundred and when energy expenditure is recorded continuously over 24 hours.

# **Appendix A1: Performance Measures for Activity Recognition**

Performance Measure	Description	Computed by
Confusion Matrix	Also known as contingency table. It is a square matrix that shows the correctly and incorrectly classified instances ("confusions") The columns of the matrix correspond to the predictions or classifications done by the model and the rows correspond to the actual classification or "ground truth" labels. Most performance measures presented in this table can be computed from the confusion matrix.	Increase the count at each corresponding matrix cell by one whenever a new classification is made available using the pair of values (predicted, actual). The matrix is given in the format: confusionMatrix[predicted][actual] confusionMatrix[columns][rows] An element of a matrix is represented as $x_{column,row} = x_{i j}$
Total Accuracy (Success rate)	It is the fraction of correct classifications and is computed as the sum over the diagonal elements of the confusion matrix divided by total number of instances or examples in the matrix. All other instances outside the diagonal are incorrect classifications. Accuracy puts equal weight on relevant and irrelevant classes, and it is often the case that there are more irrelevant classes than relevant ones.	$\frac{\displaystyle\sum_{i=1}^N x_{ii}}{\displaystyle\sum_{i=1}^N \sum_{j=1}^N x_{ij}}$
True Positive rate (Recall or sensitivity)	The proportion of examples which were classified as class x, among all examples which truly have class x, i.e. how much part of the class was captured. It is equivalent to Recall. In the confusion matrix, this is the diagonal element divided by the sum over the relevant row. Measure widely used in information retrieval that represents the fraction of relevant documents that are retrieved. Pr[retrieved relevant]. You can get a perfect recall (but low precision) by retrieving all docs for all queries!	$\frac{x_{ii}}{\displaystyle\sum_{j=1}^N x_{ij}}$
False Positive Rate	Is the proportion of examples which were classified as class x, but belong to a different class, among all examples which are not of class x. In the matrix, this is the column sum of class x minus the diagonal element, divided by the rows sums of all other classes.	$\frac{\displaystyle \sum_{j=1}^{N} x_{ij} - x_{ii}}{\displaystyle \sum_{i=1}^{N} \sum_{j=1}^{N} x_{ij} - \sum_{j=1}^{N} x_{ij}}$
F-Measure	It is a combined measure of precision and recall computed as the harmonic average over precision and recall.	$\frac{2 \cdot presion_i \cdot recall_i}{presion_i + recall_i}$

 Table A1-1: Performance measures utilized to evaluate activity recognition algorithms in this work.

 All performance measures are multiplied by 100 to be expressed as percentages.

**Appendix A2: Clustering of Activities for Presentation of Results** 

Activity (Number of METs from the Commondium of physical	Static	Ambulation	Exercise Activity	Resistance	House-hold
(Number of NIE1's from the Compendium of physical activities)	rostures			Activity	
Bench weight lifting – hard (6.0)			V	V	
Bench weight lifting – light (3.0)			N	7	
Bench weight lifting – moderate $(>3.0)$			N	1	
Bicen curls – hard (6 0)			N	2	
Bicep curls light (3.0)			2	2	
Bicep curls $=$ inglit (5.0) Bicep curls $=$ moderate (>3.0)			2	2	
Calisthenics Crunches (8.0)			2	v	
Calisthenics – Crunches (8.0)			2		
Cualing Cuala hard Cuala 20rpm (>2.0)					
Cycling - Cycle hard - Cycle 801pm (>8.0)	-		N	N	
Cycling - Cycle light - Cycle 1001plii (7.0)			N		
Cycling - Cycle light - Cycle 80rpm (5.5)			N		
Cycling - Cycle light - Cycle 80rpm (5.5)			N	N	
Cycling - Cycle moderate - Cycle 80rpm (8.0)	1		N	N	
Lying down (1.0)	N		1	1	
Rowing - Rowing hard - Rowing 30spm (8.5)			N	N	
Rowing - Rowing light - Rowing 30spm (3.5)			N	N	
Rowing - Rowing moderate - Rowing 30spm (7.0)		,	N	V	
Running - Treadmill 4mph - Treadmill 0 (5.0)		V	N		
Running - Treadmill 5mph - Treadmill 0 (8.0)		V	V		
Running - Treadmill 6mph - Treadmill 0 (10.0)		V	V		
Sitting (1.0)	V				
Sitting - Fidget feet legs (1.0)					
Sitting - Fidget hands arms (1.0)					
Stairs - Ascend stairs (8.0)		$\checkmark$		$\checkmark$	
Stairs - Descend stairs (3.0)		$\checkmark$			
Standing (1.2)					
Walking - Treadmill 2mph - Treadmill 0 (2.5)		$\checkmark$			
Walking - Treadmill 3mph - Treadmill 0 (3.3)		$\checkmark$			
Walking - Treadmill 3 mph - Treadmill 3 - light (>3.3)		$\checkmark$		$\checkmark$	
Walking - Treadmill 3mph - Treadmill 6 - moderate (>3.3)		$\checkmark$		$\checkmark$	
Walking - Treadmill 3mph - Treadmill 9 - hard (>3.3)		$\checkmark$		$\checkmark$	
Kneeling (1.0)					
Unknown (N/A)					
Carrying groceries (3.0)		$\checkmark$		$\checkmark$	
Doing dishes (2.3)					
Gardening (4.0)					
Ironing (2.3)					
Making the bed (2.0)					
Mopping (3.5)					
Playing videogames (1.5)					
Scrubbing a surface (3.8)					
Stacking groceries (2.5)					
Sweeping (3.3)	1				√.
Typing (1.8)					۰ ۷
Vacuuming (3.5)		1		İ	, V
Walking around block (3.5)		V	1	1	, v
Washing windows (3.0)	1	, v		V	V V
Watching TV (1 0)	+			, v	v V
Weeding (4 5)	+				1
Wining/Dusting (2.5)					V
Writing (1.8)	+				1
Taking out trash (2.5)			1		v v

Table A2-1: The 51 activities contained in the MIT dataset and their categorization according to increasing order of classification complexity: (1) postures, (2) postures and ambulation, (3) exercise activities, and (4) household activities. Activities that are included in the garbage class are represented with G and activities eliminated from the dataset with E. Activities including in the training process are marked with a check mark.

Activity (Number of METs from the Compendium of physical activities)	Upper Body	Lower Body	Postures and Ambulation	Postures and Ambulation with METs	All Activities with No Intensity
				Intensity	
Bench weight lifting – hard (6.0)	$\checkmark$			Moderate	$\sqrt{(\text{Bench})}$
Bench weight lifting – light (3.0)	$\checkmark$			Moderate	√(Bench)
Bench weight lifting – moderate (>3.0)	$\checkmark$			Moderate	$\sqrt{(\text{Bench})}$
Bicep curls – hard (6.0)	$\checkmark$			Moderate	√(Biceps)
Bicep curls – light (3.0)	$\checkmark$			Moderate	√(Biceps)
Bicep curls – moderate (>3.0)	$\checkmark$			Moderate	√(Biceps)
Calisthenics – Crunches (8.0)				Vigorous	V
Calisthenics - Sit ups (8.0)				Vigorous	V
Cycling - Cycle hard - Cycle 80rpm (>8.0)				Vigorous	√(Cycling)
Cycling - Cycle light - Cycle 100rpm (7.0)				Vigorous	√(Cycling)
Cycling - Cycle light - Cycle 60rpm (3.0)				Moderate	$\sqrt{(Cycling)}$
Cycling - Cycle light - Cycle 80rpm (5.5)				Moderate	$\sqrt{(Cycling)}$
Cycling - Cycle moderate - Cycle 80rpm (8.0)				Vigorous	√(Cycling)
Lying down (1.0)			√	√	V
Rowing - Rowing hard - Rowing 30spm (8.5)				Vigorous	√(Rowing)
Rowing - Rowing light - Rowing 30spm (3.5)				Moderate	√(Rowing)
Rowing - Rowing moderate - Rowing 30spm (7.0)				Vigorous	$\sqrt{(Rowing)}$
Running - Treadmill 4mph - Treadmill 0 (5.0)			$\sqrt{(\mathbf{R}_{11})}$	√ √	$\sqrt{(\text{Running})}$
Running - Treadmill 5mph - Treadmill 0 (8.0)			$\sqrt{(Running)}$	v V	$\sqrt{(\text{Running})}$
Running - Treadmill 6 mph - Treadmill 0 (10.0)			$\sqrt{(Running)}$	1	$\sqrt{(\text{Running})}$
Sitting (1.0)			$\sqrt{(\text{Kutting})}$	√ (Sitting)	√ (Kutting)
Sitting - Fidget feet legs (1.0)		V	$\sqrt{\text{(Sitting)}}$	$\sqrt{(\text{Sitting})}$	$\sqrt{(Sitting)}$
Sitting - Fidget hands arms (1.0)	1	v	√ (Sitting)	$\sqrt{(\text{Sitting})}$	√ (Sitting)
Stairs - Ascend stairs (8.0)	v	-	v (Sitting)	Vigorous	v (Sitting)
Stairs - Descend stairs (3.0)		-	1	Moderate	N
Standing (1.2)		-	1	Nioderate	2
Walking - Treadmill 2mph - Treadmill () (2.5)			v (welling)	2	) (welling)
Walking - Treadmill 3mph - Treadmill 0 (2.3)			V (walking)	v (walking 2)	(walking)
Walking Treadmill 3mph Treadmill 3 light		-	V (walking)	V (walking 3)	v (walking)
(>3.3)			(waiking)	(waiking 5)	(waiking)
Walking - Treadmill 3mph - Treadmill 6 - moderate (>3.3)			√(walking)	√(walking 3)	√(walking)
Walking - Treadmill 3mph - Treadmill 9 - hard (>3.3)			$\sqrt{(\text{walking})}$	$\sqrt{(\text{walking 3})}$	$\sqrt{(\text{walking })}$
Kneeling (1.0)			$\checkmark$		V
Unknown (N/A)					
Carrying groceries (3.0)					√(walking)
Doing dishes (2.3)	$\checkmark$				V
Gardening (4.0)				Moderate	V
Ironing (2.3)	$\checkmark$				V
Making the bed (2.0)					V
Mopping (3.5)				Moderate	V
Playing videogames (1.5)	$\checkmark$				V
Scrubbing a surface (3.8)	$\checkmark$			Moderate	V
Stacking groceries (2.5)					V
Sweeping (3.3)				Moderate	V
Typing (1.8)	√				V
Vacuuming (3.5)				Moderate	N
Walking around block (3.5)	1		1		√(walking)
Washing windows (3.0)	$\checkmark$		1		. √
Watching TV (1.0)			1		V
Weeding (4.5)	1		1	Moderate	V
Wiping/Dusting (2.5)		1			V
Writing (1.8)	V		1	İ	v V
Taking out trash (2.5)		1	1		2

Table A2-2: The 51 activities contained in the MIT dataset and their categorization according to increasing order of classification complexity: (1) postures, (2) postures and ambulation, (3) exercise activities, and (4) household activities. Activities that are included in the garbage class are represented with G and activities eliminated from the dataset with E. Activities including in the training process are marked with a check mark.

# **Appendix A3: The 41 Features Explored in This Work**

Features	Abbreviation	Description
Measures of body posture		
• Mean	DCMean	Mean or average over the signal window after low-pass filtering the
		acceleration signal at 1Hz. This is a measure of the DC value or the static
		component of acceleration that changes with body posture.
• Mean over all	DCTotalMean	Same as DCMean but computed over the summation of all the acceleration
acceleration axis		signals over all axis and accelerometers available (sensors). This feature
		the acceleration signals
• Area under signal	DCArea	The area under the signal simply computed by summing the acceleration
• Area under signar	DCAICa	samples contained in a given window
		total samples per window
		$DCArea = \sum_{n=1}^{n} \sum_{n=1}^{n} acceleration$
		$DCMed = \sum_{i=1}^{n} acceleration_i$
	DCDostureDist	l=1
Mean distances     hetween evic	DCPostuleDist	The differences between the mean values of the A-1, A-Z, and 1-Z
between axis		the sensor with respect to ground or body posture information. The feature
		is computed after low-pass filtering the acceleration signals at 1Hz
Measures of motion shape		
Mean of absolute signal	ACAbsMean	Mean or average over the absolute value of the band-pass filtered (0.1-
value		20Hz) accelerometer signals. Acceleration can have positive and negative
		values, so computing the absolute value guarantees the mean will not be
		zero for perfect oscillatory motion with equal positive and negative
		acceleration magnitudes.
Cumulative sum over	ACAbsArea	The area under the absolute value of the signal computed by simply
absolute signal value		summing the accelerometer samples inside a given window. The sum is
		Computed after band-pass filtering (0.1-20Hz) the accelerometer signals.
		absolute value guarantees the area will not be zero for perfect oscillatory
		motion with equal positive and negative acceleration magnitudes.
		windowLengh
		$ACAbsArea = \sum_{a}  a  +  a  +  a $
		$\sum_{i=1}^{n \text{ current PerSensor}}  \mathcal{U}X_i  +  \mathcal{U}Y_i  +  \mathcal{U}Z_i $
		t-1 windowLanth
		$\Lambda CAbsArag = \sum_{minuter git} a$
		$ACADSATEU_{PerAxis} - \sum_{i=1}^{n}  u_{Axis_i} $
	ACT at al A ha A rea	l=1
<ul> <li>Iotal Cumulative sum over absolute signals</li> </ul>	ACTOLAIADSATEA	over all axis and accelerometers available (sensors). This feature captures
value		the overall motion experienced by the human body as experienced by all the
value		accelerometers worn.
		numberSenors
		$ACT otalAbsArea = \sum ACAbsArea$
		t-1 numbarAris
		$\Lambda CT$ at all has $\Lambda rag = \sum_{namericans} \Lambda CA has \Lambda rag$
		$ACTOINIADSATEN = \sum_{i=1}^{n} ACADSATEN _{PerAxis_i}$
		i=1
<ul> <li>Total signal vector</li> </ul>	ACTotalSVM	The average value of the signal vector magnitude of all the accelerometer
<ul> <li>Total signal vector</li> <li>magnitude</li> </ul>	ACTOLAISVIN	samples for a given window. This feature is computed after hand-pass
mugintude		filtering (0.1-20Hz) the accelerometer signals. The Signal vector magnitude
		for all samples at a given time is computed using the following formula:
		total number of ares
		$SVM = \frac{\sum_{i=1}^{N} \sum_{j=1}^{N} \sum_{i=1}^{N} \sum_{i=1}^{N} \sum_{i=1}^{N} \sum_{j=1}^$
		$\sum u_i$
_		V <i>i</i> =1
Entropy	ACEntropy	Measures the degree of "disorder" in the band-pass filtered (0.1-20Hz)
		window low the
		$\mu = \sum_{i=1}^{winaow_length} 1_{2} + 1_{2}$
		$\Pi = - \sum_{v \in \log_2(v)} V \cdot \log_2(v)$
1		Where v corresponds to the normalized values of the FFT magnitude.

		mag(i)		
		$v = \frac{\sigma(v)}{window \ length}$		
		$\sum_{i=1}^{n} mag(i)$		
		$\sum_{i=1}^{mag(i)}$		
		<i>l</i> -1		
		The final value of the entropy is normalized to fall in the range [0, 1] by dividing each entropy value by the entropy of the uniform distribution.		
Skewness (signal	ACSkew	It is a measure of the "peakedness" of the accelerometer signal over a given		
moment)		infrequent extreme deviations as opposed to frequent modestly-sized		
		deviations. It is computed over the band-pass filtered acceleration using the		
		following formula:		
		$\sqrt{n}\sum^{n}(x_{\cdot}-\overline{x})^{3}$		
		$\binom{n}{2}$		
		$\sum (x_i - \overline{x})^2$		
		$\left( \sum_{i=1}^{l} \left( i \right) \right)$		
Kurtosis (signal	ACKur	It is a measure of the "peakedness" of the accelerometer signal over a given		
moment)		window or a measure of its relative flatness as compared to the normal		
		distribution. It is computed over the band-pass filtered (0.1-20Hz)		
		accelerometer signals using the following formula:		
		$n\sum_{n=1}^{n}(x-\overline{x})^4$		
		$n \sum_{i=1}^{n} (x_i - x_i)$		
		$\frac{l=1}{\sqrt{2}-3}$		
		$\left(\sum_{n=1}^{n} (1-z)^{2}\right)^{2}$		
		$\sum (x_i - x)$		
		( <i>i</i> =1 )		
• Quartiles (first, second,	ACQ1	Quartiles are computed by partitioning the accelerometer signal over a		
and third)	ACQ2 ACQ3	given window into four quarters each containing 25% of the data ( $Q_{1}=25$ , $Q_{2}=50$ and $Q_{3}=75\%$ ). This is achieved by sorting the signal values		
	11020	according to increasing magnitude and finding the values at 25%, 50% and		
		75% of the window length. The values measure the distribution of the		
		accelerometer magnitude across the window. The quartiles are computed		
Magunos of motion variation		over the band-pass filtered (0.1-20Hz) accelerometer signals.		
Variance	ACVar	The variance of the accelerometer signal computed over a given window. It		
• variance		is computed after band-pass filtering (0.1-20Hz) the accelerometer signals.		
Coefficient of variation	ACAbsCV	Computed as the ratio of the standard deviation and the mean over each		
over the absolute value		signal window multiplied by 100. This measures the dispersion of the		
of the signal		acceleration signal. Acceleration is band-pass filtered (0.1-20Hz) before		
<ul> <li>Inter quartile range</li> </ul>	ACIOR	Computed as the difference between quartiles three and one (Q3-Q1). This		
inter quartie range	nongii	value describes the dispersion of the acceleration signal. This feature is		
		computed after band-pass filtering (0.1-20Hz) the accelerometer signals.		
Range or maximum	ACRange	Difference between the maximum and minimum values of the		
signal amplitude		acceleration or maximum motion inside the window. Accelerometer signals		
		are band-pass filtered (0.1-20Hz) before computing this feature.		
Measures of motion spectral content				
• Fast Fourier transform (FFT) coefficients	ACFFTCoeff	The coefficients of the fast Fourier transform computed over the		
		accelerometer signal for a given window. The signal is first band-pass		
		There (0.1-20Hz) and the first coefficient (DC component) is not utilized.		
		minus one. It is a measure of the magnitudes of the frequencies contained		
		within the accelerometer signal.		
Fast Fourier transform	ACFFTPeaks	The X number of frequencies with maximum magnitudes computed over		
(FFT) peaks		the FFT coefficients. These consist on X number of frequency, magnitude		
		pairs (treq, mag). In this work, X=5. It is a measure of the X number of largest frequencies present in the signal and their magnitudes. This feature		
		does not include the DC component of the signal or magnitude at frequency		
		zero. Before computing this feature, signals are band-pass filtered (0.1-		

		20Hz).			
Fast wavelet transform (FWT) coefficients	ACFWTCoeff	The coefficients of the Haar wavelet transform computed over the accelerometer signal for a given window. There are as many coefficients as signal samples in the window. The coefficients are a measure of the frequencies contained in the accelerometer signals at different time scale resolutions. Accelerometer signals are band-pass filtered (0.1-20Hz) before computing this feature. The coefficient corresponding to the DC component of the signals is not included.			
• Total apargy	ACEnergy	It is computed as follows from the EET coefficient magnitudes:			
• Total energy	ACEllergy	Note that the index $i$ starts from 1 and not from zero to avoid computing the energy of the DC component. The index also goes to only half of the window since the EET components after half of the window are redundant.			
		This is the total energy contained in all the frequencies. The FFT coefficients are computed over the band-pass filtered (0.1-20Hz) accelerometer signals.			
• Activity band energy (0.3-3.5Hz)	ACBandEnergy	Computed as the sum of the energy contained between frequencies of $0.3 - 3.5$ Hz. This has been found to be the major energy band for human activities in previous work. This feature does not include the DC component of the signal. The energy is computed from the FFT coefficients computed over the band-pass filtered (0.1-20Hz) accelerometer signals.			
<ul> <li>Energy of low intensity physical activity (0 – 0.7Hz)</li> </ul>	ACLowEnergy	Computed as the sum of the energy contained between frequencies of $0 - 0.7$ Hz. This has been found to be Associated with sedentary activities in previous work. This energy does include the DC component of the signal.			
<ul> <li>Energy of moderate to vigorous physical activity (0.71 – 10 Hz)</li> </ul>	ACModVigEnergy	Computed as the sum of the energy contained between frequencies of 0.71 - 10Hz. This has been found to be associated with locomotion and high intensity activities in previous work			
Heart rate mean	HRMean	This feature is the average value of the heart rate data in beats-per-minute over the heart rate data window length.			
Heart rate above resting heart rate	HRAboveRest	Computed by subtracting the resting heart rate (RHR) from the mean heart rate value (in bmp) over each window. It measures the intensity associated with an activity. The value will be near zero for sedentary activities and will increase with the activity intensity.			
• Heart rate normalized to lie between the range resting heart rate (RHR) and the mean heart rate measured while running at 5mph on a treadmill.	ScaledHR	Computed by scaling the mean heart rate value (in bmp) over each window to lie between the range $0 - 1$ using the resting heart rate and the mean heart rate value obtained while subjects run at 5mph on a treadmill. It measures the intensity associated with an activity.			
The variance of the heart rate data.	HRVar	This feature is simply the variance computed over the heart rate data window.			
physical activity intensity					
Heart rate trend line	HRTrend	Slope of the regression line computed over the heart rate signal window. This feature captures if hear rate is increasing, decreasing or in steady-state over time.			
Measures of motion periodicity					
• Pitch	ACPitch	Computed as the magnitude of the second peak of the autocorrelation function after $Rx$ (0). It measures the magnitude of the fundamental frequency contained in the acceleration signal. Accelerometer signals are band-pass filtered (0.1-20Hz) before computing this feature.			
Ratio of energy in dominant frequency	ACDomFreqRatio	Computed by dividing the magnitude of the FFT coefficient with largest magnitude by the sum of the coefficients in all other frequencies. It measures how much of the signal is dominated by a particular frequency. The FFT is computed over the band-pass filtered (0.1-20Hz) accelerometer signals.			
Mean crossing rate	ACMCR	Number of times the acceleration signal crosses its mean value over the window. It measures how fast the acceleration signal changes over time. The acceleration signal is band-pass filtered (0.1-20Hz) before computing this feature.			
Measures of motion similarity across body					

segments					
Pearson's Correlation     Coefficients	ACCorr	The correlation coefficients among each pair of acceleration signals over a given window. The correlation coefficients are computed according to the following formula:			
		$\operatorname{cov}(i, j)$			
		$R(i, j) = \frac{(j, j)}{\sqrt{\operatorname{cov}(i, i) \cdot \operatorname{cov}(j, j)}}$			
		Cov stands for the covariance function. The correlation coefficients measure the linear dependence between two acceleration signals. The coefficient lies between $[-1, +1]$ were the sign indicates the correlation direction or the orientation of the line. Two acceleration signals that are identical would give a correlation coefficient of one. In simple terms, the coefficients characterize how similar is the simultaneous motion of limbs.			
Measure of force per body					
segment					
Segmental Force	ACSF	Computed by multiplying the ACAbsArea feature for the accelerometer at the hip, wrists, and feet by the segment mass per limb obtained from the Depmster's body segment model [236]. This feature consists of five coefficients given by the following formulas:			
		$ACSF_{Trunk} = ACAbsArea_{Hip} \cdot BodyTrunkMass$			
		$ACSF_{DWrist} = ACAbsArea_{DWrist} \cdot ArmMass$			
		$ACSF_{NDWrist} = ACAbsArea_{NDWrist} \cdot ArmMass$			
		$ACSF_{DFoot} = ACAbsArea_{DFoot} \cdot LegMass$			
		$ACSF_{NDFoot} = ACAbsArea_{NDFoot} \cdot LegMass$			
		Were the prefix "D" stands for dominant and "ND" stands for non dominant.			
Total Segmental Force	ACTotalSF	ACTotalSF is the sum of the five coefficients of the segmental force.			
-		$ACTotalSF = ACSF_{Trunk} + ACSF_{DArm} + ACSF_{NDArm}$			
		$+ ACSF_{DFoot} + ACSF_{NDFoot}$			
Measures of subject characteristics					
• Gender	Gender	A number specifying the gender of the subject: 0- Male, 1-Female			
• Age	Age	A number specifying the age of the subject.			
Height	Height	A number specifying the height of the subject in meters			
Weight	Weight	A number specifying the body weight of the subject in Kg			
Measures of subject fitness					
• BMI	BMI	Computed by dividing body weight of a subject (in Kg) by the squared of its height (in meters)			
• Fat percentage	FatPercent	A number representing the percentage of fat or non-muscle tissue in the subject.			
Fitness Index	FitnessIndex	This feature is computed by dividing the number of steps per minute by the average heart rate value (in beats-per-minute) per minute of a subject running on a treadmill at 5mph. This is an approximation of the EitnessIndex utilized in [230]			

 Table A3-1: List of features explored in this work to (1) recognize activities and (2) estimate energy expenditure from wearable accelerometers and a heart rate monitor. The list is a superset of most features found in previous work. Features starting with the prefix "AC" were computed over the accelerometer signals after applying a band-pass filter between the frequencies of 0.1 and 20Hz. Features starting with the prefix "DC" were computed over the accelerometer signals after applying a low-pass filter with a cutoff frequency of 1Hz.

# **Appendix A4: Amount of Data in the MIT Dataset**

Activity	Amount of training time (in minutes)	Average amount of training data per subject (in minutes)	Number of Training Examples when using a window of 5.6s
Bench weight lifting – Light	30.4	1.52	326
Bench weight lifting – Moderate	21.8	1.09	234
Bench weight lifting – Hard	11.5	0.575	124
Bicen curls – Light	31.7	1.585	340
Bicen curls – Moderate	21.8	1.09	234
Bicen curls – Hard	21.3	1.065	229
Calisthenics Crunches	19.6	0.98	211
Calisthenics Sit ups	25.0	1.25	268
Cycling Cycle 100rpm (15mph 120.4W) – Light	45.0	2 25	483
Cycling Cycle 60rpm (8 9mph 66 9W) – Light	54.0	2.23	579
Cycling Cycle 80rpm (11 2mph 100 4W) – Light	52.9	2 645	567
Cycling Cycle 80rpm – Moderate	46.2	2:045	496
Cycling Cycle 80rpm – Hard	34.8	1 74	373
Lying down	149.1	7.455	1 598
Rowing 30spm – Light	50.4	2 52	540
Rowing 30spm Moderate	12.7	2.52	458
Rowing 30spm Hard	35.0	1.75	438
Rowing Jospin – Hard	33.0	2 295	512
Running Treadmill 5mph Treadmill 0	47.7	2.365	J12 478
Running Treadmill Smph Treadmill 0	21.1	2.23	478
Sitting	29.4	1.555	205
Sitting Eident fast lass	20.4	1.42	208
Sitting Fidget herds arms	27.8	1.39	298
Situng Fluget hands arms	27.8	1.39	298
Stairs Ascend stairs	40.4	2.32	498
Stairs Descend stairs	42.7	2.135	458
Standing Wellting Treedmill 2mph Treedmill 0	29.9	1.495	521
Walking Treadmill 2mph Treadmill 0	57.1	2.855	612
Walking Treadmill Sinph Treadmill 0	56.3	2.623	606
Walking Treadmill Sinph Treadmill 6 Madageta	50.2	2.81	603
Walking Treadmill 3mph Treadmill 6 – Moderate	57.4	2.87	010
Walking Treadmill Smph Treadmill 9 – Hard	30.5	2.825	000
Kneeling	29.4	1.47	515
Carrying groceries	50.4	2.52	541
Doing dishes	54.9	2.745	589
Gardening	52.2	1.01	345
Ironing Malaina tha had	57.8	2.89	620
Making the Bed	33.7	2.085	570
Mopping	43.1	2.155	462
Playing videogames	58.9	2.945	632
Scrubbing a surface	39.2	1.96	421
Stacking groceries	38.4	1.92	412
Sweeping	42.9	2.145	460
Typing	60.9	3.045	653
Vacuuming	44.4	2.22	476
Walking around block	53.1	2.655	569
Washing windows	48.90	2.445	524
Watching TV	59.9	2.995	642
Weeding	30.1	1.505	323
Wiping/Dusting	52.1	2.605	559
Writing	61.5	3.075	659
Taking out trash	42.3	2.115	454
Unknown	938.1	46.905	10.052

Table A4-1: Amount of training data available (or duration of each activity) for the 52 activities contained in the MIT dataset and explored in this work.
## **Appendix A5: Sliding Window Length for Activity Recognition**

Activity	1.4s	2.8s	5.6s	11.3s	22.7s	45.5s	91.0s
Category							
All	58.20±7.32	62.55±7.58	65.51±6.33	66.83±6.24	67.09±5.81	60.06±6.58	23.31±10.11
Postures	51.37±9.98	54.57±11.99	53.10±14.09	52.52±16.48	45.42±22.41	18.97±11.75	12.39±3.58
	(0.78±0.24)	(0.73±0.27)	(0.67±0.33)	(0.68±0.39)	$(0.82 \pm 0.49)$	(1.13±0.69)	(2.45±1.29)
Ambulation	60.80±10.92	68.76±11.39	73.36±11.30	76.86±14.88	75.85±17.95	68.09±24.47	6.45±15.05
	(0.81±0.26)	(0.66±0.26)	(0.57±0.28)	(0.52±0.28)	(0.57±0.36)	(0.88±0.59)	$(2.79 \pm 1.61)$
Exercise	66.35±13.33	70.28±12.85	73.31±14.06	74.58±16.44	69.74±23.43	58.01±31.93	5.18±10.99
	(0.55±0.21)	(0.48±0.21)	(0.44±0.23)	(0.43±0.26)	(0.51±0.38)	$(0.84 \pm 0.59)$	(2.94±1.53)
Resistance	52.11±14.46	57.92±14.91	62.68±15.71	66.18±18.16	64.39±23.95	52.38±31.56	$4.49 \pm 10.90$
Exercise	(0.85±0.29)	(0.74±0.30)	(0.65±0.32)	(0.62±0.33)	(0.69±0.47)	(0.96±0.63)	(3.02±1.68)
Household	39.49±11.86	44.18±13.78	48.79±14.98	52.05±17.93	56.00±21.76	50.71±29.45	6.63±14.03
	(1.15±0.44)	(1.07±0.45)	(0.99±0.46)	(0.97±0.52)	$(0.98 \pm 0.61)$	(1.17±0.84)	(3.24±2.10)

Table A5-1: Performance of the C4.5 decision tree classifier using the ACAbsArea feature over window lengths varying from 1.4 to 45.5 seconds during subject dependent evaluation.

Activity	1.4s	2.8s	5.6s	11.3s	22.7s	45.5s	91.0s
Category							
All	$33.62 \pm 4.04$	$35.00 \pm 4.27$	$36.82 \pm 5.51$	$38.32\pm5.19$	$39.40 \pm 5.34$	$43.46 \pm 6.17$	$42.31\pm7.26$
Postures	18.63±11.54	19.62±13.19	21.19±15.12	23.97±18.12	26.54±20.04	35.70±29.14	51.06±42.66
	(0.94±0.39)	(0.88±0.42)	(0.87±0.51)	(0.84±0.65)	(0.94±0.65)	(0.93±0.78)	(1.64±1.38)
Ambulation	23.33±18.34	23.85±22.03	26.48±22.65	28.33±26.35	34.58±31.13	39.51±34.98	39.40±35.13
	(1.14±0.72)	(1.07±0.84)	(1.09±0.95)	$(1.05 \pm 1.02)$	(1.23±1.16)	(1.51±1.39)	$(1.82 \pm 1.64)$
Exercise	20.00±19.44	20.18±21.34	22.80±24.12	25.96±28.60	28.01±26.29	35.96±31.07	41.42±35.31
	(0.61±0.50)	(0.61±0.57)	(0.64±0.69)	(0.73±0.74)	(0.96±0.94)	(1.25±1.13)	(2.18±1.51)
Resistance	13.04±13.39	13.50±15.85	14.96±17.99	16.32±22.00	18.86±22.58	27.19±26.95	33.29±37.79
Exercise	(0.85±0.56)	(0.79±0.61)	(0.79±0.70)	(0.82±0.76)	(1.08±0.96)	(1.29±1.09)	(2.42±1.65)
Household	12.53±10.02	14.68±12.36	16.62±14.49	18.92±18.33	22.05±22.58	29.17±28.09	35.00±37.20
	(1.13+0.49)	(1.12+0.58)	(1.11+0.65)	(1.08+0.74)	$(1.15 \pm 0.80)$	$(1.25 \pm 1.00)$	(1.87+1.48)

 Table A5-2: Performance of the C4.5 decision tree classifier using the ACAbsArea feature over window lengths varying from 1.4 to 45.5 seconds during subject independent evaluation

Activity	1.4s	2.8s	5.6s	11.3s	22.7s	45.5s	91.0s
Category							
All	$56.82 \pm 7.51$	$62.58 \pm 6.89$	$66.36 \pm 6.34$	$66.33 \pm 7.44$	$64.81 \pm 6.29$	$55.20\pm7.38$	$21.34 \pm 9.04$
Postures	44.39±9.97	51.52±10.68	55.00±14.08	49.54±16.28	45.21±21.83	26.58±22.71	13.25±3.10
	(1.01±0.29)	(0.85±0.32)	(0.75±0.32)	(0.79±0.41)	(0.93±0.59)	(1.41±0.97)	(2.24±1.24)
Ambulation	65.38±9.81	71.24±10.13	75.79±10.99	80.02±13.98	76.72±20.43	65.97±30.36	4.50±11.39
	(0.71±0.24)	(0.58±0.22)	(0.48±0.23)	(0.44±0.27)	(0.50±0.38)	(0.87±0.71)	$(2.46 \pm 1.49)$
Exercise	65.60±11.51	71.40±12.27	75.82±11.41	74.42±16.69	67.82±20.62	51.26±32.96	4.56±10.00
	(0.54±0.19)	(0.45±0.19)	(0.39±0.17)	(0.43±0.26)	(0.54±0.39)	$(0.89 \pm 0.66)$	(2.95±1.54)
Resistance	53.64±12.74	60.97±13.99	67.62±13.12	67.31±17.08	63.19±20.99	48.92±30.27	4.55±10.74
Exercise	(0.80±0.25)	(0.68±0.25)	(0.56±0.23)	(0.58±0.31)	$(0.64 \pm 0.44)$	(1.01±0.72)	(2.96±1.67)
Household	38.73±12.04	45.74±12.89	49.98±15.48	52.13±18.16	52.80±22.81	43.44±30.38	7.11±14.73
	(1.21±0.47)	(1.07±0.43)	(0.99±0.46)	(1.00±0.55)	(1.04±0.65)	(1.41±0.99)	(3.23±1.84)

 Table A5-3: Performance of the C4.5 decision tree classifier using the ACFFTPeaks + ACCorr

 features over window lengths varying from 1.4 to 45.5 seconds. Subject dependent evaluation

Activity	1.4s	2.8s	5.6s	11.3s	22.7s	45.5s	91.0s
Category							
All	$34.74\pm3.79$	$38.16\pm3.76$	$41.23\pm5.04$	$43.38 \pm 4.71$	$45.32\pm6.23$	$46.40\pm6.46$	$45.38 \pm 8.50$
Postures	18.56±11.48	22.63±13.62	25.60±17.39	28.97±20.94	30.65±22.44	42.39±29.68	55.05±44.27
	(0.98±0.34)	(0.90±0.37)	$(0.84 \pm 0.46)$	$(0.80\pm0.48)$	(0.90±0.61)	$(0.94 \pm 0.84)$	(1.56±1.34)
Ambulation	22.77±18.39	25.17±21.08	28.56±23.79	31.08±28.16	35.50±33.85	38.11±33.75	46.01±32.08
	(1.04±0.67)	(0.98±0.69)	$(1.02\pm0.84)$	(1.00±0.94)	$(1.07 \pm 1.10)$	$(1.49 \pm 1.43)$	(1.85±1.63)
Exercise	21.03±19.45	23.48±22.09	27.54±25.82	31.91±27.31	38.24±33.74	47.40±35.60	48.75±39.75
	(0.59±0.46)	(0.54±0.53)	(0.60±0.58)	(0.75±0.79)	(0.95±0.98)	(1.02±0.98)	$(2.09 \pm 1.58)$
Resistance	15.28±14.76	16.49±15.82	20.01±20.92	22.46±22.00	29.45±28.91	39.02±33.13	35.82±40.85
Exercise	(0.79±0.53)	(0.75±0.56)	(0.79±0.63)	$(0.92 \pm 0.88)$	(1.07±0.95)	(1.23±1.05)	(2.20±1.51)
Household	15.33±11.30	17.78±13.56	20.26±16.85	22.52±20.31	25.61±23.10	30.93±28.60	43.28±38.52
	(1.17±0.45)	(1.13±0.51)	(1.10±0.63)	(1.06±0.70)	(1.03±0.70)	(1.23±0.96)	$(1.76 \pm 1.41)$

 Table A5-4: Performance of the C4.5 decision tree classifier using the ACFFTPeaks + ACCorr

 features over window lengths varying from 1.4 to 45.5 seconds. Subject independent evaluation

Activity	1.4s	5.6s	11.3s	45.5s
Category				
All	$58.22 \pm 3.43$	$67.44 \pm 2.04$	$67.85 \pm 2.96$	$52.42 \pm 4.28$
Postures	47.1±9.2	59.7±12.1	53.4±15.0	23.6±18.6
	(0.9±0.2)	(0.6±0.2)	(0.7±0.3)	(1.1±0.7)
Ambulation	70.7±9.0	78.8±10.8	81.2±12.4	64.3±27.8
	(0.5±0.1)	(0.3±0.2)	(0.3±0.2)	(0.7±0.5)
Exercise	69.3±11.2	78.4±11.4	78.0±14.3	49.9±31.6
	(0.4±0.1)	(0.3±0.1)	(0.3±0.2)	(0.8±0.5)
Resistance	59.2±12.9	72.1±12.6	72.5±16.3	46.6±31.0
Exercise	(0.6±0.2)	(0.4±0.2)	(0.4±0.2)	(0.9±0.6)
Household	45.6±10.8	55.5±13.1	57.3±16.6	45.6±26.1
	(0.9±0.3)	(0.7±0.3)	(0.7±0.3)	(1.1±0.8)

Table A5-5: Performance of the C4.5 decision tree classifier using the ACFFTPeaks + ACCorr features over window lengths varying from 1.4 to 45.5 seconds. Subject dependent evaluation feature computation per axis

Activity	1.4s	5.6s	11.3s	45.5s
Category				
All	$38.01 \pm 2.94$	$44.36\pm3.98$	$47.47 \pm 4.74$	$51.84 \pm 5.23$
Postures	29.4±11.0	37.9±16.0	40.4±20.0	51.9±33.7
	(1.0±0.3)	(0.8±0.4)	(0.8±0.4)	(0.9±0.8)
Ambulation	32.3±17.0	38.3±24.9	43.0±28.1	53.8±34.1
	(0.9±0.5)	(0.8±0.6)	(0.8±0.7)	$(1.0\pm1.0)$
Exercise	34.6±17.7	38.0±24.9	42.8±27.9	55.4±32.9
	(0.6±0.4)	(0.6±0.5)	(0.6±0.5)	$(0.8\pm0.8)$
Resistance	25.6±14.3	29.1±21.6	32.9±26.9	43.1±34.0
Exercise	(0.8±0.4)	(0.8±0.5)	(0.8±0.6)	(1.0±0.8)
Household	28.0±11.1	35.3±16.4	38.6±19.2	47.4±29.1
	$(1.1\pm0.3)$	$(1.0\pm0.4)$	$(0.9\pm0.5)$	$(1.0\pm0.8)$

Table A5-6: Performance of the C4.5 decision tree classifier using the ACFFTPeaks + ACCorr features over window lengths varying from 1.4 to 45.5 seconds. Subject independent evaluation feature computation per axis

	True Positive Rate								
Class	1.4s	2.8s	5.6s	11.3s	22.7s	44.5s	90s		
Bench weight lifting - hard	$23.02 \pm 12.84$	$33.82 \pm 13.76$	$44.76 \pm 11.77$	$46.64 \pm 15.05$	$17.50 \pm 15.41$	$0.00\pm0.00$	$0.00 \pm 0.00$		
Bench weight lifting - light	$45.47 \pm 13.49$	$58.23 \pm 18.42$	$61.40 \pm 11.16$	$58.81 \pm 25.34$	$48.05 \pm 15.60$	$54.55 \pm 37.34$	$0.00 \pm 0.00$		
Bench weight lifting - moderate	$33.33 \pm 9.45$	$35.42 \pm 16.06$	$45.17 \pm 12.02$	$35.64 \pm 22.76$	$27.80 \pm 15.39$	$15.62 \pm 24.57$	$0.00 \pm 0.00$		
Bicep curls - hard	$51.09 \pm 9.67$	66.89 ± 12.80	$73.60 \pm 11.74$	$74.58 \pm 12.11$	$65.13 \pm 15.64$	$45.83 \pm 42.49$	$16.67 \pm 25.82$		
Bicep curls - light	$61.45 \pm 18.03$	$70.45 \pm 16.73$	$75.95 \pm 12.55$	$76.74 \pm 17.48$	$81.07 \pm 17.01$	$59.17 \pm 32.97$	8.33 ± 20.41		
Bicep curls - moderate	$53.30 \pm 14.01$	$62.28 \pm 12.54$	$68.43 \pm 9.00$	$65.20 \pm 22.12$	$60.83 \pm 27.88$	$41.67 \pm 42.72$	$0.00 \pm 0.00$		
Calisthenics - Crunches	$81.81 \pm 11.20$	$84.25 \pm 8.37$	$89.95 \pm 8.47$	83.57 ± 9.03	$76.29 \pm 23.73$	$33.33 \pm 42.16$	$0.00 \pm 0.00$		
Calisthenics - Sit ups	$88.42 \pm 4.47$	$92.58 \pm 2.91$	$91.54 \pm 5.36$	91.79 ± 8.37	$85.60 \pm 16.73$	$69.05 \pm 41.31$	$0.00 \pm 0.00$		
Cycling - Cycle hard - Cycle 80rpm	$67.52 \pm 19.58$	71.68 ± 19.30	$76.86 \pm 19.49$	$70.25 \pm 26.34$	$62.95 \pm 25.45$	$36.54 \pm 30.53$	$4.55 \pm 15.08$		
Cycling - Cycle light - Cycle 100rpm	$90.51 \pm 9.95$	$94.13 \pm 7.82$	$94.50 \pm 6.00$	$92.45 \pm 15.79$	$84.12 \pm 25.90$	$72.22 \pm 31.29$	$7.69 \pm 18.78$		
Cycling - Cycle light - Cycle 60rpm	$85.24 \pm 9.12$	$85.82 \pm 10.58$	$86.58 \pm 8.39$	82.52 ± 13.29	$78.96 \pm 17.36$	$63.89 \pm 30.65$	$12.50 \pm 22.36$		
Cycling - Cycle light - Cycle 80rpm	$80.26 \pm 10.21$	$84.41 \pm 8.91$	$85.00 \pm 10.30$	$80.97 \pm 14.58$	$75.35 \pm 18.37$	$56.02 \pm 25.21$	$11.54 \pm 21.93$		
Cycling - Cycle moderate - Cycle 80rpm	$67.70 \pm 11.08$	$70.99 \pm 9.41$	$72.47 \pm 17.26$	$68.59 \pm 14.40$	$65.65 \pm 25.89$	$44.67 \pm 41.48$	$0.00 \pm 0.00$		
Lying down	$56.09 \pm 11.60$	69.99 ± 12.01	$78.19 \pm 7.99$	76.51 ± 13.29	$75.48 \pm 15.92$	$83.62 \pm 16.49$	$79.50 \pm 18.62$		
Rowing - Rowing hard - Rowing 30spm	53.36 ± 12.69	$63.38 \pm 14.88$	$70.82 \pm 15.50$	$73.97 \pm 16.07$	$70.66 \pm 27.95$	$64.39 \pm 31.64$	$4.55 \pm 15.08$		
Rowing - Rowing light - Rowing 30spm	$60.11 \pm 14.95$	$65.62 \pm 19.65$	$79.74 \pm 11.45$	$78.76 \pm 13.96$	$81.34 \pm 14.59$	$62.75 \pm 31.61$	$13.64 \pm 32.33$		
Rowing - Rowing moderate - Rowing	$51.09 \pm 14.29$	$56.35 \pm 15.74$	$64.68 \pm 15.67$	$66.14 \pm 20.51$	$57.67 \pm 26.06$	$44.64 \pm 35.75$	$0.00 \pm 0.00$		
30spm									
Running - Treadmill 4mph - Treadmill 0	86.66 ± 5.69	$89.73 \pm 6.31$	$88.02 \pm 9.16$	$92.78 \pm 10.01$	$85.85 \pm 15.57$	$68.85 \pm 27.64$	$7.14 \pm 18.16$		
Running - Treadmill 5mph - Treadmill 0	$85.05 \pm 6.55$	$87.34 \pm 6.03$	$86.69 \pm 9.12$	$88.75 \pm 10.25$	$88.39 \pm 10.71$	$66.67 \pm 40.37$	$0.00\pm0.00$		
Running - Treadmill 6mph - Treadmill 0	$81.02 \pm 11.40$	$83.14 \pm 12.88$	$84.33 \pm 12.39$	$85.74 \pm 29.60$	$75.34 \pm 36.46$	$74.07 \pm 36.43$	$0.00\pm0.00$		
Sitting	$16.61 \pm 9.43$	$28.95 \pm 9.94$	$35.94 \pm 12.81$	$24.26 \pm 17.77$	$18.13 \pm 22.43$	$7.69 \pm 18.78$	$0.00 \pm 0.00$		
Sitting - Fidget feet legs	$85.14 \pm 12.10$	$81.55 \pm 13.41$	$83.54 \pm 12.62$	$80.38 \pm 15.29$	$70.24 \pm 18.97$	$27.78 \pm 24.96$	$0.00\pm0.00$		
Sitting - Fidget hands arms	$73.30 \pm 13.66$	76.60 ± 12.35	$71.50\pm20.82$	$69.48 \pm 19.28$	$61.43 \pm 25.38$	$31.25 \pm 45.81$	$0.00 \pm 0.00$		
Stairs - Ascend stairs	$66.48 \pm 6.68$	$70.60 \pm 9.76$	$80.80\pm7.18$	$84.46 \pm 12.18$	$76.54 \pm 19.74$	$80.00 \pm 21.08$	$0.00 \pm 0.00$		
Stairs - Descend stairs	$57.20 \pm 9.59$	67.71 ± 8.39	74.46 ± 13.39	81.77 ± 10.58	$58.33 \pm 27.32$	51.11 ± 35.34	$0.00 \pm 0.00$		
Standing	$17.34 \pm 7.93$	$24.47 \pm 8.13$	$30.31 \pm 10.62$	$21.69 \pm 13.80$	$18.22 \pm 15.42$	$4.17 \pm 14.43$	$0.00 \pm 0.00$		
Walking - Treadmill 2mph - Treadmill 0	$80.57 \pm 7.02$	$83.05 \pm 7.43$	$86.03 \pm 7.53$	$85.26 \pm 12.22$	$90.48 \pm 9.09$	$80.98 \pm 25.08$	$4.55 \pm 15.08$		
Walking - Treadmill 3mph - Treadmill 0	$56.91 \pm 9.80$	$65.24 \pm 9.04$	$66.34 \pm 9.43$	$75.22 \pm 14.23$	$79.57 \pm 20.81$	$68.82 \pm 32.95$	$16.67 \pm 30.86$		
Walking - Treadmill 3mph - Treadmill 3	$45.08 \pm 11.57$	$55.63 \pm 11.98$	$64.65 \pm 11.78$	$69.63 \pm 12.19$	$73.47 \pm 23.09$	$61.88 \pm 23.17$	$3.33 \pm 12.91$		
- light									
Walking - Treadmill 3mph - Treadmill 6	$45.88 \pm 10.49$	$51.71 \pm 11.66$	$63.51 \pm 13.31$	$65.82 \pm 15.00$	$72.86 \pm 16.53$	$50.20 \pm 32.19$	$3.85 \pm 13.87$		
- moderate									
Walking - Treadmill 3mph - Treadmill 9	$61.01 \pm 12.73$	$67.26 \pm 11.23$	$73.09 \pm 9.73$	$74.04 \pm 15.34$	$77.38 \pm 24.11$	$54.02 \pm 27.82$	$3.85 \pm 13.87$		
- hard	1001 010								
kneeling	$1/.86 \pm 5.12$	$27.55 \pm 8.26$	30.51 ± 19.63	24.91 ± 18.24	27.78 ± 32.85	5.00 ± 15.81	$0.00 \pm 0.00$		
unknown	$61.60 \pm 11.79$	65.53 ± 11.37	66.87 ± 11.96	66.16 ± 12.26	65.08 ± 10.59	$61.56 \pm 17.00$	50.74 ± 25.93		
Carrying groceries	$62.73 \pm 13.83$	/1.13 ± 15.05	/1.4/ ± 16.9/	80.31 ± 13.31	70.66 ± 20.54	$69.33 \pm 34.04$	$11.54 \pm 21.93$		
Doing dishes	35.16 ± 9.70	42.74 ± 14.08	49.39 ± 17.85	$53.19 \pm 13.78$	$54.05 \pm 25.81$	$50.00 \pm 35.92$	$20.83 \pm 33.43$		
Gardening	$39.03 \pm 14.88$	47.83 ± 15.37	49.61 ± 25.36	$54.00 \pm 27.24$	$50.22 \pm 27.20$	48.50 ± 35.20	7.14 ± 18.90		
Ironing Making the head	38.05 ± 11.31	$47.39 \pm 10.98$	$51.36 \pm 1/.58$	$50.73 \pm 14.10$	$55.11 \pm 21./1$	53.96 ± 38.80	$10.00 \pm 28.03$		
Making the bed	$26.72 \pm 7.47$	$33.14 \pm 11.86$	38.80 ± 11.30	$43.20 \pm 15.81$	$48.45 \pm 18.14$	55.78 ± 33.82	$24.24 \pm 36.03$		
Mopping	35.06 ± 12.82	38.49 ± 14.67	48.30 ± 18.09	46.86 ± 19.69	49.11 ± 28.15	35.14 ± 23.99	18.18 ± 33.71		
Playing videogames	47.47 ± 18.31	57.92 ± 19.37	$57.26 \pm 18.84$	$57.35 \pm 21.35$	45.67 ± 26.95	$32.56 \pm 27.17$	$6.94 \pm 16.60$		
Scrubbing a surface	38.43 ± 12.70	42.81 ± 13.73	44.95 ± 11.72	49.13 ± 16.32	57.15 ± 26.85	$44.03 \pm 14.80$	7.14 ± 18.90		
Stacking groceries	$26.56 \pm 12.49$	35.90 ± 15.09	$44.07 \pm 18.35$	$48.73 \pm 22.44$	52.45 ± 19.09	$42.42 \pm 33.63$	$0.00 \pm 0.00$		
Sweeping	30.98 ± 15.94	38.98 ± 17.74	$41.68 \pm 15.35$	49.56 ± 19.89	49.94 ± 23.95	$20.00 \pm 20.37$	$0.00 \pm 0.00$		
1 yping	$68.82 \pm 14.65$	/5.33 ± 13.09	/3./1±13./8	/3.99 ± 15.70	69.98 ± 26.60	$47.55 \pm 26.76$	3.85 ± 13.87		
Vacuuming	33.44 ± 11.06	37.27 ± 8.86	$42.25 \pm 9.14$	48.95 ± 14.58	$65.38 \pm 20.83$	$29.10 \pm 29.12$	$11.11 \pm 33.33$		
Walking around block	55.96 ± 12.37	02.3/±11./9	/0.14 ± 11.88	/0.45 ± 12.81	/1.80 ± 21.21	$05.6/\pm 28.20$	$3.03 \pm 10.05$		
Washing Windows	30.01 ± 13.75	41.55 ± 13.88	44.79 ± 19.20	$41.00 \pm 18.70$	$52.58 \pm 28.50$	39.36 ± 30.29	$0.00 \pm 0.00$		
Waading	$21.55 \pm 0.60$	$25.75 \pm 7.70$	$35.39 \pm 13.21$	$28.25 \pm 10.43$	$19.81 \pm 12.63$	$15.31 \pm 21.15$	$0.00 \pm 0.00$		
Weeding	$28.75 \pm 13.56$	38.//±8.3/	$41.19 \pm 15.53$	41.51 ± 25.56	41.45 ± 19.89	50.57 ± 29.22	11.11 ± 19.25		
Wiping/Dusting	$24.00 \pm 10.61$	51.91 ± 14.03	$30.40 \pm 1/.16$	58.81 ± 21.55	$41.08 \pm 20.73$	39.1/±30.2/	$0.00 \pm 0.00$		
writing	00.93 ± 12.36	/3.08 ± 12.32	/3.36 ± 12.55	09.12 ± 19.71	05.22 ± 25.63	45.45 ± 54.61	$0.00 \pm 0.00$		
taking out trash	$18.72 \pm 5.06$	23.78 ± 7.85	$50.53 \pm 8.48$	56.57 ± 16.29	$46.80 \pm 22.96$	$44.23 \pm 34.33$	$0.00 \pm 0.00$		

Table A5-7: True positive rate when training classifiers using the *FFTCorr* feature and subject dependent evaluation

				False Positive R	ate		
Class	1.4s	2.8s	5.6s	11.3s	22.7s	44.5s	90s
Bench weight lifting - hard	$0.92 \pm 0.15$	$0.80 \pm 0.18$	$0.60 \pm 0.09$	$0.62 \pm 0.30$	$0.88 \pm 0.34$	$0.91 \pm 0.47$	$3.82 \pm 1.62$
Bench weight lifting - light	$0.85 \pm 0.34$	$0.74 \pm 0.35$	$0.71 \pm 0.27$	$0.73 \pm 0.54$	$1.06 \pm 0.57$	$1.13 \pm 0.77$	$3.61 \pm 2.41$
Bench weight lifting - moderate	$0.96 \pm 0.26$	$0.81 \pm 0.31$	$0.67 \pm 0.28$	$0.86 \pm 0.42$	$1.06 \pm 0.63$	$1.56 \pm 0.78$	$3.68 \pm 2.11$
Bicep curls - hard	$0.76 \pm 0.27$	$0.57 \pm 0.26$	$0.36 \pm 0.13$	$0.42 \pm 0.29$	$0.31 \pm 0.19$	$1.00 \pm 0.73$	$3.10 \pm 1.63$
Bicep curls - light	$0.72 \pm 0.31$	$0.55 \pm 0.31$	$0.47 \pm 0.30$	$0.51 \pm 0.27$	$0.60 \pm 0.37$	$0.93 \pm 1.15$	$3.38 \pm 1.62$
Bicep curls - moderate	$0.80 \pm 0.27$	$0.66 \pm 0.29$	$0.66 \pm 0.30$	$0.54 \pm 0.35$	$0.58 \pm 0.26$	$0.81 \pm 0.37$	$2.24 \pm 1.15$
Calisthenics - Crunches	$0.29 \pm 0.16$	$0.17 \pm 0.08$	$0.20\pm0.08$	$0.25 \pm 0.16$	$0.23 \pm 0.32$	$0.79 \pm 0.76$	$1.79 \pm 0.72$
Calisthenics - Sit ups	$0.19 \pm 0.11$	$0.16 \pm 0.05$	$0.10 \pm 0.11$	$0.15 \pm 0.14$	$0.28 \pm 0.37$	$0.38 \pm 0.39$	$3.33 \pm 1.91$
Cycling - Cycle hard - Cycle 80rpm	$0.37\pm0.12$	$0.37 \pm 0.14$	$0.34\pm0.15$	$0.37\pm0.27$	$0.60 \pm 0.65$	$1.34 \pm 0.73$	3.91 ± 1.67
Cycling - Cycle light - Cycle 100rpm	$0.11 \pm 0.05$	$0.11 \pm 0.07$	$0.12 \pm 0.08$	$0.13 \pm 0.12$	$0.16 \pm 0.24$	$0.28 \pm 0.38$	$3.33 \pm 1.66$
Cycling - Cycle light - Cycle 60rpm	$0.26 \pm 0.14$	$0.25 \pm 0.13$	$0.32 \pm 0.19$	$0.41 \pm 0.30$	$0.57 \pm 0.50$	$0.91 \pm 0.86$	$3.30 \pm 1.51$
Cycling - Cycle light - Cycle 80rpm	$0.34 \pm 0.17$	$0.33 \pm 0.14$	$0.34 \pm 0.20$	$0.42 \pm 0.26$	$0.45 \pm 0.45$	$0.76 \pm 0.85$	$2.56 \pm 1.44$
Cycling - Cycle moderate - Cycle 80rpm	$0.58 \pm 0.14$	$0.46 \pm 0.16$	$0.38 \pm 0.16$	$0.55 \pm 0.27$	$0.65 \pm 0.51$	$1.00 \pm 0.61$	$1.77 \pm 0.40$
Lying down	$2.16 \pm 0.76$	$1.75 \pm 0.78$	$1.32 \pm 0.56$	$1.31 \pm 0.64$	$1.57 \pm 0.92$	$1.97 \pm 1.86$	$2.04 \pm 1.69$
Rowing - Rowing hard - Rowing 30spm	$0.78 \pm 0.22$	$0.66 \pm 0.24$	$0.46 \pm 0.13$	$0.51 \pm 0.26$	$0.68 \pm 0.42$	$0.61 \pm 0.60$	$2.90 \pm 1.75$
Rowing - Rowing light - Rowing 30spm	$0.69 \pm 0.21$	$0.51 \pm 0.27$	$0.42 \pm 0.19$	$0.44 \pm 0.23$	$0.62 \pm 0.47$	$0.98 \pm 0.58$	$2.95 \pm 2.10$
Rowing - Rowing moderate - Rowing 30spm	$0.84 \pm 0.16$	$0.69 \pm 0.20$	$0.57 \pm 0.21$	$0.65 \pm 0.28$	$0.57 \pm 0.29$	$1.17 \pm 0.62$	$2.96 \pm 1.64$
Running - Treadmill 4mph - Treadmill 0	$0.29 \pm 0.17$	$0.23 \pm 0.13$	$0.25 \pm 0.16$	$0.21 \pm 0.14$	$0.30 \pm 0.23$	$0.82 \pm 0.68$	$3.09 \pm 2.25$
Running - Treadmill 5mph - Treadmill 0	$0.29 \pm 0.14$	$0.23 \pm 0.13$	$0.22 \pm 0.10$	$0.27 \pm 0.19$	$0.40 \pm 0.30$	$0.78 \pm 0.71$	$2.97 \pm 1.52$
Running - Treadmill 6mph - Treadmill 0	$0.21 \pm 0.13$	$0.22 \pm 0.12$	$0.20 \pm 0.12$	$0.18 \pm 0.14$	$0.27 \pm 0.23$	$0.75 \pm 0.56$	$1.30 \pm 0.13$
Sitting	$1.22 \pm 0.22$	$0.96 \pm 0.30$	$1.03 \pm 0.37$	$1.07 \pm 0.44$	$1.19 \pm 0.72$	$1.35 \pm 0.54$	$3.76 \pm 2.19$
Sitting - Fidget feet legs	$0.22 \pm 0.19$	$0.24 \pm 0.23$	$0.25 \pm 0.20$	$0.21 \pm 0.17$	$0.39 \pm 0.39$	$1.38 \pm 1.14$	$2.18 \pm 1.38$
Sitting - Fidget hands arms	$0.37 \pm 0.23$	$0.33 \pm 0.19$	$0.39 \pm 0.32$	$0.41 \pm 0.28$	$0.46 \pm 0.38$	$1.17 \pm 0.65$	$1.95 \pm 1.02$
Stairs - Ascend stairs	$0.66 \pm 0.16$	$0.51 \pm 0.18$	$0.29 \pm 0.11$	$0.28 \pm 0.20$	$0.31 \pm 0.26$	$0.53 \pm 0.54$	$0.00 \pm 0.00$
Stairs - Descend stairs	$0.74 \pm 0.23$	$0.56 \pm 0.26$	$0.45 \pm 0.28$	$0.34 \pm 0.20$	$0.55 \pm 0.34$	$0.63 \pm 0.65$	$0.00 \pm 0.00$
Standing	$1.13 \pm 0.17$	$0.96 \pm 0.23$	$0.82 \pm 0.25$	$1.01 \pm 0.48$	$1.02 \pm 0.37$	$1.53 \pm 1.13$	$1.66 \pm 0.51$
Walking - Treadmill 2mph - Treadmill 0	$0.45 \pm 0.18$	$0.38 \pm 0.12$	$0.38 \pm 0.19$	$0.35 \pm 0.30$	$0.52 \pm 0.44$	$0.73 \pm 0.54$	$2.04 \pm 0.87$
Walking - Treadmill 3mph - Treadmill 0	$1.01 \pm 0.23$	$0.78 \pm 0.25$	$0.68 \pm 0.21$	$0.50 \pm 0.24$	$0.57 \pm 0.45$	$1.02 \pm 0.78$	$3.74 \pm 2.77$
Walking - Treadmill 3mph - Treadmill 3 - light	$1.23 \pm 0.34$	$1.08 \pm 0.30$	$0.85 \pm 0.31$	0.78 ± 0.29	$0.60 \pm 0.46$	$1.28 \pm 0.69$	$3.84 \pm 3.40$
Walking - Treadmill 3mph - Treadmill 6 -	$1.18\pm0.33$	$1.02\pm0.32$	$0.80\pm0.27$	$0.74\pm0.21$	$0.77\pm0.54$	$1.65\pm0.89$	$4.47\pm2.67$
Walking - Treadmill 3mph - Treadmill 9 -	$0.82 \pm 0.29$	$0.68 \pm 0.29$	$0.56 \pm 0.30$	$0.51 \pm 0.31$	$0.46 \pm 0.38$	$0.92 \pm 0.87$	2.68 ± 1.38
hard							
kneeling	$0.95 \pm 0.18$	$0.87 \pm 0.22$	$0.69 \pm 0.24$	$0.76 \pm 0.44$	$0.97 \pm 0.77$	$1.05 \pm 0.52$	$1.86 \pm 0.65$
unknown	$16.64 \pm 4.12$	$14.86 \pm 4.51$	$13.68 \pm 3.73$	$12.67 \pm 4.58$	$11.49 \pm 4.80$	$8.77 \pm 3.32$	$10.08 \pm 6.54$
Carrying groceries	$0.71 \pm 0.29$	$0.54 \pm 0.24$	$0.50\pm0.28$	$0.51 \pm 0.45$	$0.50 \pm 0.37$	$0.58 \pm 0.83$	$2.97 \pm 1.42$
Doing dishes	$1.46\pm0.50$	$1.27 \pm 0.50$	$1.24 \pm 0.79$	$1.14 \pm 0.51$	$0.94 \pm 0.69$	$1.58 \pm 1.23$	$3.43 \pm 2.39$
Gardening	$1.29 \pm 0.57$	$1.10 \pm 0.39$	$0.99 \pm 0.53$	$0.96 \pm 0.64$	$0.78 \pm 0.66$	$1.24 \pm 0.75$	$2.80\pm2.18$
Ironing	$1.48 \pm 0.47$	$1.18 \pm 0.31$	$1.12 \pm 0.39$	$1.00 \pm 0.46$	$1.01 \pm 0.53$	$1.25 \pm 0.78$	$4.19 \pm 3.21$
Making the bed	$1.63 \pm 0.47$	$1.41 \pm 0.45$	$1.19 \pm 0.31$	$1.26\pm0.52$	$1.26 \pm 0.90$	$1.77 \pm 1.44$	$2.49 \pm 1.42$
Mopping	$1.22\pm0.36$	$1.19 \pm 0.44$	$1.06\pm0.41$	$1.20\pm0.62$	$1.09 \pm 0.75$	$1.28\pm0.55$	$4.04 \pm 2.74$
Playing videogames	$1.25\pm0.51$	$1.01\pm0.46$	$0.96 \pm 0.43$	$1.13\pm0.70$	$1.37 \pm 0.91$	$1.57 \pm 1.21$	$3.12 \pm 1.54$
Scrubbing a surface	$1.10 \pm 0.37$	$0.98 \pm 0.41$	$1.07 \pm 0.51$	$1.06\pm0.55$	$0.78 \pm 0.53$	$1.02 \pm 0.85$	$2.40 \pm 1.24$
Stacking groceries	$1.29 \pm 0.45$	$1.20 \pm 0.54$	$0.87 \pm 0.45$	$1.07 \pm 0.76$	$0.83 \pm 0.54$	$0.86 \pm 0.74$	$0.00 \pm 0.00$
Sweeping	$1.30 \pm 0.54$	$1.22 \pm 0.50$	$1.13 \pm 0.65$	$1.07 \pm 0.68$	$1.13 \pm 0.70$	$1.72 \pm 0.93$	$2.58 \pm 1.64$
Typing	0.61 ± 0.32	0.56 ± 0.32	0.53 ± 0.27	$0.86 \pm 0.54$	0.85 ± 0.51	$1.45 \pm 0.92$	3.44 ± 1.48
Vacuuming	$1.29 \pm 0.56$	$1.22 \pm 0.52$	$1.09 \pm 0.44$	$1.09 \pm 0.50$	$0.90 \pm 0.54$	$1.92 \pm 1.17$	$3.53 \pm 1.53$
Walking around block	$0.88\pm0.40$	$0.72 \pm 0.32$	$0.63 \pm 0.39$	$0.58 \pm 0.53$	$0.74 \pm 0.50$	$0.77 \pm 0.75$	$2.40 \pm 1.53$
Washing windows	$1.19 \pm 0.42$	$1.18 \pm 0.40$	$1.03 \pm 0.46$	$0.99 \pm 0.39$	$0.89 \pm 0.68$	$1.01 \pm 0.95$	$2.36 \pm 1.58$
Watching TV	$1.78 \pm 0.56$	$1.49 \pm 0.41$	$1.40 \pm 0.57$	$1.44 \pm 0.64$	$1.82 \pm 0.73$	2.10 ± 1.18	6.17 ± 3.33
Weeding	$1.03 \pm 0.57$	$1.00 \pm 0.51$	$1.15 \pm 0.67$	$1.13 \pm 0.78$	$1.37 \pm 0.74$	$1.53 \pm 0.92$	3.36 ± 1.33
Wiping/Dusting	$1.35 \pm 0.46$	$1.22 \pm 0.52$	$1.14 \pm 0.44$	0.99 ± 0.52	$1.07 \pm 0.62$	$1.71 \pm 1.05$	3.62 ± 1.84
Writing	$0.71 \pm 0.40$	$0.60 \pm 0.37$	$0.57 \pm 0.36$	$0.54 \pm 0.33$	0.81 ± 0.62	$1.65 \pm 1.44$	3.78 ± 1.53
taking out trash	$1.30 \pm 0.52$	$1.18 \pm 0.45$	$1.14 \pm 0.45$	$0.96 \pm 0.38$	$1.38 \pm 0.73$	$1.32 \pm 1.06$	$3.94 \pm 2.46$

Table A5-8: False positive rate when training the C4.5 classifier using the *FFTCorr* feature and subject dependent evaluation

				F-Measure			
Class	1.4s	2.8s	5.6s	11.3s	22.7s	44.5s	90s
Bench weight lifting - hard	$22.04 \pm 12.49$	$32.13 \pm 13.54$	43.83 ± 10.66	44.96 ± 13.51	$18.52 \pm 16.73$	$0.00 \pm 0.00$	$0.91 \pm 0.47$
Bench weight lifting - light	45.13 ± 13.54	56.92 ± 16.46	$58.83 \pm 11.25$	56.53 ± 24.45	$45.99 \pm 15.42$	46.55 ± 28.62	$1.13 \pm 0.77$
Bench weight lifting - moderate	$32.25 \pm 9.22$	35.87 ± 15.46	$45.86 \pm 11.33$	34.88 ± 21.95	$28.30 \pm 15.78$	$14.13 \pm 19.98$	$1.56 \pm 0.78$
Bicep curls - hard	$51.42 \pm 10.61$	66.17 ± 12.31	$74.65 \pm 10.51$	$74.81 \pm 10.87$	$70.13 \pm 10.34$	$44.46 \pm 41.46$	$1.00 \pm 0.73$
Bicep curls - light	$61.41 \pm 16.88$	70.60 ± 15.96	$76.12 \pm 11.80$	75.11 ± 14.58	76.64 ± 16.79	57.55 ± 32.70	0.93 ± 1.15
Bicep curls - moderate	$52.44 \pm 13.85$	$61.36 \pm 12.00$	65.65 ± 9.75	65.11 ± 22.11	$60.29 \pm 25.38$	38.93 ± 37.27	0.81 ± 0.37
Calisthenics - Crunches	80.00 ± 11.54	85.57 ± 6.24	$87.76 \pm 6.68$	82.44 ± 8.87	77.77 ± 20.16	36.67 ± 44.57	$0.79 \pm 0.76$
Calisthenics - Sit ups	$88.01 \pm 5.11$	90.91 ± 2.71	92.45 ± 6.19	91.25 ± 6.46	84.90 ± 14.76	67.35 ± 36.39	0.38 ± 0.39
Cycling - Cycle hard - Cycle 80rpm	67.39 ± 19.51	70.32 ± 19.57	73.08 ± 22.06	69.27 ± 24.99	63.89 ± 28.33	32.65 ± 27.82	$1.34 \pm 0.73$
Cycling - Cycle light - Cycle 100rpm	$90.86 \pm 9.14$	93.17 ± 7.50	92.96 ± 5.87	90.70 ± 15.44	$84.08 \pm 24.58$	74.76 ± 29.35	$0.28 \pm 0.38$
Cycling - Cycle light - Cycle 60rpm	$85.01 \pm 9.09$	85.65 ± 9.71	84.88 ± 8.39	80.75 ± 12.43	$76.56 \pm 17.08$	$60.40 \pm 27.92$	0.91 ± 0.86
Cycling - Cycle light - Cycle 80rpm	80.61 ± 10.38	83.20 ± 8.67	83.45 ± 10.05	79.10 ± 12.56	75.24 ± 16.93	$57.88 \pm 24.34$	$0.76 \pm 0.85$
Cycling - Cycle moderate - Cycle 80rpm	66.68 ± 10.48	71.41 ± 9.44	73.37 ± 14.09	67.94 ± 12.28	$64.14 \pm 23.38$	41.37 ± 36.79	$1.00 \pm 0.61$
Lying down	$55.62 \pm 11.02$	$68.08 \pm 11.02$	76.25 ± 7.15	75.14 ± 12.32	73.06 ± 13.22	77.54 ± 15.96	$1.97 \pm 1.86$
Rowing - Rowing hard - Rowing 30spm	$53.00 \pm 12.96$	$61.93 \pm 14.61$	$70.04 \pm 13.46$	$71.10 \pm 15.70$	$65.81 \pm 27.46$	$62.45 \pm 28.44$	$0.61 \pm 0.60$
Rowing - Rowing light - Rowing 30spm	$58.74 \pm 15.50$	$65.79 \pm 20.39$	$78.20 \pm 10.92$	77.53 ± 12.33	76.31 ± 15.39	$57.17 \pm 27.00$	$0.98 \pm 0.58$
Rowing - Rowing moderate - Rowing 30spm	$50.84 \pm 13.71$	$56.95 \pm 15.18$	$65.06 \pm 14.12$	$64.11 \pm 17.47$	$59.26 \pm 22.65$	$39.78 \pm 29.54$	$1.17 \pm 0.62$
Running - Treadmill 4mph - Treadmill 0	$85.95 \pm 6.40$	88.93 ± 5.69	$87.62 \pm 7.78$	$90.72 \pm 7.74$	85.38 ± 12.69	$64.12 \pm 26.74$	$0.82 \pm 0.68$
Running - Treadmill 5mph - Treadmill 0	$84.53 \pm 6.48$	87.13 + 5.88	86.96 + 6.36	87.37 + 7.53	84.42 + 9.80	60.77 + 36.15	$0.78 \pm 0.71$
Running - Treadmill 6mph - Treadmill 0	80.38 + 10.81	81.38 + 12.67	83.87 + 11.91	$83.49 \pm 27.50$	73.78 + 34.78	66.09 + 32.41	$0.75 \pm 0.56$
Sitting	$15.83 \pm 9.20$	$28.64 \pm 10.01$	$33.67 \pm 12.73$	$23.43 \pm 17.64$	$17.64 \pm 21.45$	$5.64 \pm 13.84$	$1.35 \pm 0.54$
Sitting - Fidget feet legs	84.11 ± 12.73	$81.67 \pm 14.28$	$82.25 \pm 12.04$	81.19 ± 14.55	$71.62 \pm 20.41$	$22.57 \pm 21.32$	$1.38 \pm 1.14$
Sitting - Fidget hands arms	$72.81 \pm 14.48$	$75.70 \pm 12.48$	$71.04 \pm 19.63$	$68.77 \pm 17.52$	$61.58 \pm 24.05$	$27.50 \pm 41.32$	$1.17 \pm 0.65$
Stairs - Ascend stairs	$64.81 \pm 6.90$	$70.41 \pm 8.81$	81.26 ± 7.35	83.12 ± 10.43	76.65 ± 13.58	$76.32 \pm 18.61$	$0.53 \pm 0.54$
Stairs - Descend stairs	$56.57 \pm 9.74$	$67.07 \pm 9.27$	$73.25 \pm 13.71$	$80.24 \pm 9.40$	$56.84 \pm 23.33$	52.86 ± 35.75	$0.63 \pm 0.65$
Standing	$16.49 \pm 7.05$	$24.37 \pm 7.88$	$31.13 \pm 10.99$	$20.85 \pm 12.95$	$17.55 \pm 13.47$	$2.38 \pm 8.25$	$1.53 \pm 1.13$
Walking - Treadmill 2mph - Treadmill 0	$79.88 \pm 6.34$	$82.65 \pm 5.70$	84.32 ± 7.43	84.27 ± 11.88	85.25 ± 9.03	$74.72 \pm 21.79$	$0.73 \pm 0.54$
Walking - Treadmill 3mph - Treadmill 0	$55.36 \pm 9.27$	$64.34 \pm 8.77$	66.75 ± 8.27	75.09 ± 12.47	$76.93 \pm 19.64$	$63.44 \pm 28.95$	$1.02 \pm 0.78$
Walking - Treadmill 3mph - Treadmill 3 -	44.14 ± 10.83	$53.56 \pm 10.82$	$62.98 \pm 12.06$	66.97 ± 11.06	$72.84 \pm 21.00$	57.13 ± 20.27	1.28 ± 0.69
Walking - Treadmill 3mph - Treadmill 6 -	45.62 ± 10.73	51.81 ± 11.61	$62.89 \pm 11.94$	65.18 ± 12.53	$70.65 \pm 16.67$	$45.30\pm28.06$	$1.65\pm0.89$
moderate							
Walking - Treadmill 3mph - Treadmill 9 - hard	61.17 ± 12.71	67.63 ± 10.91	73.60 ± 9.76	74.45 ± 14.31	77.07 ± 21.50	55.31 ± 28.62	0.92 ± 0.87
kneeling	$18.26\pm4.87$	$27.91 \pm 8.79$	$32.07 \pm 19.31$	$26.04 \pm 16.96$	$25.87 \pm 28.24$	$4.00 \pm 12.65$	$1.05 \pm 0.52$
unknown	$61.11 \pm 11.71$	$65.03 \pm 11.42$	$66.55 \pm 11.85$	$66.60 \pm 12.41$	$66.27 \pm 10.91$	$64.00 \pm 15.57$	8.77 ± 3.32
Carrying groceries	63.18 ± 13.94	$71.46 \pm 14.42$	$72.27 \pm 14.45$	79.04 ± 11.71	$71.77 \pm 18.16$	70.29 ± 31.85	$0.58 \pm 0.83$
Doing dishes	$34.77 \pm 9.56$	42.11 ± 13.33	$48.35 \pm 17.76$	52.31 ± 12.13	$55.08 \pm 23.04$	46.09 ± 31.83	$1.58 \pm 1.23$
Gardening	$38.62 \pm 14.49$	47.07 ± 13.92	$49.55 \pm 24.36$	53.75 ± 25.79	$52.54 \pm 24.56$	43.59 ± 31.17	$1.24 \pm 0.75$
Ironing	$37.37 \pm 10.69$	$47.34 \pm 10.81$	$50.85 \pm 16.92$	51.99 ± 13.23	$55.33 \pm 21.05$	49.64 ± 35.09	$1.25 \pm 0.78$
Making the bed	$26.81 \pm 7.35$	33.83 ± 11.86	$40.08 \pm 10.42$	$43.23 \pm 14.10$	$48.34 \pm 15.35$	$50.46 \pm 31.31$	$1.77 \pm 1.44$
Mopping	$35.57 \pm 12.44$	$38.88 \pm 14.59$	$47.28 \pm 14.88$	$45.67 \pm 18.91$	$47.52 \pm 26.08$	35.39 ± 23.64	$1.28 \pm 0.55$
Playing videogames	47.93 ± 17.53	$58.15 \pm 18.49$	$58.07 \pm 17.35$	56.91 ± 20.26	$45.32 \pm 25.72$	33.08 ± 24.86	$1.57 \pm 1.21$
Scrubbing a surface	39.39 ± 13.06	$44.25 \pm 14.04$	$45.38 \pm 11.74$	48.88 ± 16.21	57.33 ± 22.66	$46.95 \pm 18.95$	$1.02 \pm 0.85$
Stacking groceries	$27.12 \pm 12.93$	35.96 ± 15.66	$45.89 \pm 18.37$	$48.20 \pm 23.36$	$52.74 \pm 17.19$	$42.77 \pm 31.12$	$0.86 \pm 0.74$
Sweeping	31.77 ± 15.75	$38.72 \pm 16.64$	$42.82 \pm 15.98$	50.25 ± 19.56	$49.52 \pm 24.15$	$20.94 \pm 19.59$	$1.72 \pm 0.93$
Typing	$70.18 \pm 13.93$	$75.28 \pm 11.97$	74.71 ± 10.99	$70.23 \pm 14.75$	$66.44 \pm 23.84$	44.69 ± 22.03	$1.45 \pm 0.92$
Vacuuming	34.49 ± 11.31	$38.51 \pm 9.56$	43.69 ± 9.55	48.59 ± 12.71	63.06 ± 19.20	$28.05 \pm 26.21$	$1.92 \pm 1.17$
Walking around block	$57.24 \pm 11.86$	$63.95 \pm 11.35$	$70.90 \pm 11.74$	$76.25 \pm 13.14$	$69.79 \pm 16.97$	66.31 ± 26.57	$0.77 \pm 0.75$
Washing windows	$38.17 \pm 13.45$	$42.20 \pm 12.54$	$46.39 \pm 18.15$	$43.46 \pm 18.80$	$52.63 \pm 26.29$	42.66 ± 29.87	$1.01 \pm 0.95$
Watching TV	$22.06 \pm 6.22$	$27.34 \pm 7.64$	$36.71 \pm 13.04$	29.82 ± 15.31	$20.42 \pm 12.77$	$13.82 \pm 18.18$	$2.10 \pm 1.18$
Weeding	31.11 ± 14.15	$42.04 \pm 7.89$	$42.92 \pm 13.66$	$41.57 \pm 20.50$	$40.34 \pm 18.72$	44.67 ± 26.59	$1.53 \pm 0.92$
Wiping/Dusting	$26.43 \pm 11.40$	$33.69 \pm 13.60$	$38.21 \pm 16.96$	$41.40 \pm 20.44$	$44.33 \pm 21.22$	$36.38 \pm 28.72$	$1.71 \pm 1.05$
Writing	$68.15 \pm 11.62$	73.65 ± 11.87	$74.04 \pm 12.55$	$70.98 \pm 17.07$	$65.05 \pm 22.36$	$41.54 \pm 32.10$	$1.65 \pm 1.44$
taking out trash	$20.58 \pm 5.14$	$26.14 \pm 9.06$	$32.68 \pm 8.28$	39.27 ± 17.10	$43.33 \pm 19.05$	$40.78 \pm 26.82$	$1.32 \pm 1.06$

Table A5-9: F-Measure when training the C4.5 classifier using the *FFTCorr* feature and subject dependent evaluation

	True Positive Rate									
Class	1.4s	2.8s	5.6s	11.3s	22.7s	44.5s	90s			
Bench weight lifting - hard	$4.76 \pm 8.96$	$2.27 \pm 3.39$	$3.06 \pm 5.22$	$3.33 \pm 7.67$	9.09 ± 30.15	8.33 ± 20.41	$0.00 \pm 0.00$			
Bench weight lifting - light	13.33 ± 13.19	$15.47 \pm 16.26$	$14.82 \pm 21.56$	$19.38 \pm 26.61$	$13.24 \pm 16.93$	$23.72 \pm 32.25$	$25.00 \pm 46.29$			
Bench weight lifting - moderate	$8.46 \pm 9.39$	7.31 ± 9.24	$7.35 \pm 11.40$	$14.32 \pm 17.90$	8.35 ± 18.05	$18.75 \pm 32.20$	$30.00 \pm 44.72$			
Bicep curls - hard	$11.09 \pm 14.57$	$15.21 \pm 19.95$	$8.92 \pm 13.49$	$36.17 \pm 28.00$	$40.67 \pm 39.34$	$53.57 \pm 50.89$	$33.33 \pm 51.64$			
Bicep curls - light	$17.94 \pm 19.31$	$19.76 \pm 18.93$	$24.05 \pm 19.87$	$32.17 \pm 22.70$	$56.25 \pm 37.10$	$55.56 \pm 47.14$	$60.00 \pm 54.77$			
Bicep curls - moderate	9.91 ± 13.51	$15.47 \pm 20.52$	$23.42 \pm 24.31$	$17.42 \pm 19.61$	$53.65 \pm 42.83$	83.33 ± 27.89	$60.00 \pm 54.77$			
Calisthenics - Crunches	$10.02 \pm 21.58$	$9.80 \pm 22.69$	$11.97 \pm 24.76$	$25.41 \pm 39.89$	$24.53 \pm 38.18$	$8.33 \pm 18.00$	$0.00 \pm 0.00$			
Calisthenics - Sit ups	$38.08 \pm 40.32$	38.99 ± 44.24	$56.88 \pm 42.13$	$55.05 \pm 42.00$	$67.90 \pm 36.95$	$72.22 \pm 37.27$	$60.00 \pm 54.77$			
Cycling - Cycle hard - Cycle 80rpm	$11.97 \pm 17.72$	9.70 ± 12.03	$14.99 \pm 23.31$	$9.88 \pm 17.62$	$16.87 \pm 22.44$	$19.64 \pm 30.42$	$12.50 \pm 25.00$			
Cycling - Cycle light - Cycle 100rpm	$54.94 \pm 32.78$	$62.65 \pm 36.84$	$62.95 \pm 36.29$	76.13 ± 33.34	$74.00 \pm 39.06$	$90.62 \pm 27.20$	$100.00 \pm 0.00$			
Cycling - Cycle light - Cycle 60rpm	$41.28 \pm 23.31$	$54.69 \pm 33.90$	$55.76 \pm 34.56$	$56.34 \pm 34.97$	$68.51 \pm 34.54$	$76.04 \pm 33.18$	$90.62 \pm 27.20$			
Cycling - Cycle light - Cycle 80rpm	$28.59 \pm 25.29$	$35.86 \pm 26.29$	$39.78 \pm 37.10$	$33.50 \pm 27.93$	$70.34 \pm 32.29$	$76.28 \pm 31.52$	$50.00 \pm 47.14$			
Cycling - Cycle moderate - Cycle	$11.46\pm11.02$	$9.65 \pm 9.92$	$23.52 \pm 24.73$	$25.38 \pm 28.38$	$19.34 \pm 29.14$	$40.52 \pm 34.54$	$40.91 \pm 49.08$			
80rpm										
Lying down	50.32 ± 12.94	66.09 ± 13.63	71.70 ± 22.54	74.29 ± 18.31	74.77 ± 21.69	78.13 ± 23.39	85.83 ± 18.95			
Rowing - Rowing hard - Rowing 30spm	$14.20 \pm 14.71$	15.48 ± 18.90	21.60 ± 28.91	7.45 ± 11.09	15.75 ± 29.33	47.92 ± 39.28	50.00 ± 57.74			
Rowing - Rowing light - Rowing 30spm	19.74 ± 15.36	19.88 ± 19.30	24.69 ± 28.96	44.71 ± 28.11	35.13 ± 37.71	55.60 ± 41.32	$44.44\pm52.70$			
Rowing - Rowing moderate - Rowing 30spm	17.82 ± 14.75	16.54 ± 16.79	18.44 ± 21.65	20.34 ± 24.00	20.00 ± 26.66	22.50 ± 38.10	$50.00\pm53.45$			
Running - Treadmill 4mph - Treadmill 0	23.61 ± 22.22	25.14 ± 28.59	26.24 ± 25.02	34.38 ± 36.23	47.05 ± 41.91	39.26 ± 46.97	59.38 ± 49.05			
Running - Treadmill 5mph - Treadmill 0	35.23 ± 27.58	41.13 ± 31.06	36.32 ± 32.86	37.71 ± 31.66	40.84 ± 42.17	57.58 ± 42.40	80.00 ± 42.16			
Running - Treadmill 6mph - Treadmill 0	27.17 ± 24.05	31.17 ± 30.90	48.54 ± 34.51	57.28 ± 41.23	45.04 ± 46.28	50.91 ± 45.44	80.00 ± 44.72			
Sitting	$5.63 \pm 5.02$	$3.71 \pm 3.96$	$5.39 \pm 6.43$	$6.36 \pm 9.06$	$12.78 \pm 16.89$	$14.91 \pm 22.84$	$50.00 \pm 52.70$			
Sitting - Fidget feet legs	$30.01 \pm 26.41$	$34.74 \pm 31.53$	$32.64 \pm 30.28$	$38.23 \pm 34.32$	$51.18 \pm 38.06$	$70.51 \pm 37.98$	$66.67 \pm 50.00$			
Sitting - Fidget hands arms	$12.08 \pm 15.13$	$14.95 \pm 17.17$	$17.11 \pm 17.52$	$24.32 \pm 32.51$	$10.00 \pm 18.47$	$52.94 \pm 37.38$	$71.43 \pm 48.80$			
Stairs - Ascend stairs	$38.06 \pm 26.24$	$40.30 \pm 27.87$	$51.20 \pm 33.87$	$50.11 \pm 38.40$	$59.99 \pm 36.03$	$80.95 \pm 28.39$	$0.00 \pm 0.00$			
Stairs - Descend stairs	$27.04 \pm 19.53$	$31.46 \pm 24.64$	$43.87 \pm 31.20$	35.37 ± 34.17	47.69 ± 43.33	$52.38 \pm 38.60$	$0.00 \pm 0.00$			
Standing	$5.99 \pm 4.49$	$7.70 \pm 6.69$	$10.99 \pm 10.76$	$17.13 \pm 14.83$	$17.67 \pm 19.01$	$20.00 \pm 31.62$	$20.00 \pm 44.72$			
Walking - Treadmill 2mph - Treadmill 0	35.99 ± 24.64	40.26 ± 25.69	41.44 ± 31.78	47.56 ± 33.72	47.94 ± 33.92	$58.53 \pm 40.96$	92.31 ± 18.78			
Walking - Treadmill 3mph - Treadmill 0	14.50 ± 10.57	16.00 ± 15.32	17.37 ± 15.06	17.87 ± 16.41	25.30 ± 33.88	13.44 ± 21.35	33.33 ± 32.57			
Walking - Treadmill 3mph - Treadmill 3 - light	9.62 ± 8.03	10.97 ± 7.31	14.76 ± 12.87	15.03 ± 16.90	22.58 ± 21.79	16.57 ± 21.33	35.00 ± 41.16			
Walking - Treadmill 3mph - Treadmill 6 - moderate	9.20 ± 8.67	8.64 ± 9.16	11.09 ± 12.23	9.07 ± 14.42	13.64 ± 23.83	20.83 ± 31.98	33.33 ± 35.36			
Walking - Treadmill 3mph - Treadmill 9 - hard	14.07 ± 15.59	12.54 ± 13.52	7.89 ± 10.05	15.51 ± 17.40	21.01 ± 25.65	19.89 ± 29.76	16.67 ± 32.57			
kneeling	7.31 ± 4.89	$8.59 \pm 8.77$	$15.76 \pm 16.83$	$13.52 \pm 16.60$	$17.50 \pm 20.49$	$17.86 \pm 24.86$	36.36 ± 50.45			
unknown	$53.22 \pm 3.07$	55.76 ± 3.25	$58.70 \pm 3.78$	$61.93 \pm 3.28$	64.78 ± 6.01	64.43 ± 9.70	$64.22 \pm 16.37$			
Carrying groceries	$22.19 \pm 19.46$	$27.95 \pm 23.54$	$28.58 \pm 27.65$	$28.99 \pm 31.32$	31.49 ± 31.18	$23.44 \pm 25.11$	$57.14 \pm 47.46$			
Doing dishes	$15.93 \pm 9.71$	$20.22 \pm 15.50$	$23.43 \pm 17.88$	$28.29 \pm 23.93$	35.16 ± 27.79	$41.93 \pm 32.05$	67.86 ± 37.25			
Gardening	$10.22 \pm 12.14$	$11.38 \pm 13.47$	$14.90 \pm 17.12$	$18.00 \pm 22.72$	$28.80 \pm 34.42$	$32.50 \pm 42.51$	$22.73 \pm 34.38$			
Ironing	$16.90 \pm 10.93$	$22.42 \pm 13.51$	$28.44 \pm 18.43$	$27.78 \pm 24.98$	$28.89 \pm 25.60$	$30.75 \pm 30.19$	$50.00 \pm 39.22$			
Making the bed	$15.40 \pm 10.44$	$18.30 \pm 13.25$	$27.20 \pm 21.74$	$33.72 \pm 23.63$	46.77 ± 29.91	$60.94 \pm 32.68$	$55.95 \pm 39.01$			
Mopping	$14.57 \pm 13.10$	$17.57 \pm 15.61$	$18.76 \pm 18.05$	$20.28 \pm 19.94$	$21.65 \pm 22.57$	$35.00 \pm 25.74$	$50.00 \pm 42.26$			
Playing videogames	$14.83 \pm 10.22$	$15.89 \pm 11.79$	$14.14 \pm 11.79$	$15.12 \pm 17.33$	$12.19 \pm 12.88$	$23.43 \pm 30.82$	$39.29 \pm 40.09$			
Scrubbing a surface	$8.60 \pm 8.40$	$10.29 \pm 10.28$	$13.69 \pm 15.93$	$15.09 \pm 17.47$	$18.22\pm23.98$	$25.83 \pm 28.04$	$35.00 \pm 41.16$			
Stacking groceries	$8.08 \pm 7.98$	$11.30 \pm 11.57$	$13.90 \pm 14.41$	$17.90 \pm 19.83$	$21.32\pm20.75$	$35.56 \pm 36.66$	$0.00\pm0.00$			
Sweeping	$7.92 \pm 7.46$	$10.70 \pm 9.83$	$12.48 \pm 13.13$	$17.86 \pm 16.66$	$19.60 \pm 18.20$	$29.90 \pm 28.71$	33.33 ± 44.38			
Typing	$38.47 \pm 20.02$	$41.90 \pm 22.07$	$40.92 \pm 24.32$	$41.22 \pm 27.83$	29.31 ± 28.33	$29.44 \pm 22.69$	$26.92 \pm 38.81$			
Vacuuming	$12.39 \pm 10.80$	$13.12 \pm 12.19$	$13.20 \pm 12.67$	$15.53 \pm 15.25$	$21.91 \pm 20.50$	$22.13 \pm 27.78$	29.17 ± 39.65			
Walking around block	$16.55 \pm 14.14$	16.46 ± 15.31	$15.47 \pm 18.44$	24.06 ± 26.03	23.45 ± 26.25	23.53 ± 32.73	65.00 ± 41.16			
Washing windows	$12.59 \pm 9.86$	13.86 ± 11.85	22.02 ± 19.33	21.50 ± 17.89	$22.70 \pm 20.00$	35.00 ± 33.83	$46.43 \pm 41.44$			
Watching TV	7.89 ± 4.76	9.79 ± 6.79	$13.6/\pm 8.36$	14.56 ± 9.20	$10.61 \pm 11.15$	$10.58 \pm 12.80$	32.29 ± 34.68			
Weeding Wining (Depting	3.01 ± 3.38	$2.74 \pm 3.54$	3.49 ± 5.61	3.97 ± 6.48	4.82 ± 7.77	8.33 ± 16.14	$42.86 \pm 44.99$			
White a	$10.42 \pm 7.56$	$11.54 \pm 8.14$	10.95 ± 14.74	20.2/±16.2/	52.20 ± 20.37	45.75 ± 26.19	$08.18 \pm 40.22$			
writing toking out trash	$44.00 \pm 27.30$	$49.1/\pm 30.43$ 5.60 ± 5.72	$31.37 \pm 33.21$ $5.55 \pm 6.45$	46.19 ± 34.8/ 9.19 ± 11.50	$3/.30 \pm 35.38$ 12.80 ± 16.70	$32.13 \pm 33.8/$ 19 44 $\pm 22.21$	$40.88 \pm 38.00$ $45.00 \pm 42.78$			

 Table A5-10: True positive rate when training classifiers using the FFTCorr feature and subject independent evaluation

	False Positive Rate								
Class	1.4s	2.8s	5.6s	11.3s	22.7s	44.5s	90s		
Bench weight lifting - hard	$0.38\pm0.22$	$0.23 \pm 0.14$	$0.25 \pm 0.18$	$0.26 \pm 0.17$	$0.60 \pm 0.50$	$0.89 \pm 0.35$	$1.97 \pm 0.57$		
Bench weight lifting - light	$0.73 \pm 0.29$	$0.70 \pm 0.41$	$0.57 \pm 0.42$	$0.64 \pm 0.58$	$0.77 \pm 0.62$	$0.80 \pm 0.40$	$1.66 \pm 1.11$		
Bench weight lifting - moderate	$0.54 \pm 0.22$	$0.47 \pm 0.25$	$0.54 \pm 0.38$	$0.44 \pm 0.29$	$0.67 \pm 0.62$	$0.81 \pm 0.49$	$2.43 \pm 1.11$		
Bicep curls - hard	$0.54 \pm 0.51$	$0.64 \pm 0.63$	$0.53 \pm 0.53$	$0.99 \pm 0.84$	$0.85 \pm 0.83$	$1.27 \pm 0.77$	$2.41 \pm 0.96$		
Bicep curls - light	$0.54 \pm 0.55$	$0.52 \pm 0.46$	$0.53 \pm 0.43$	$1.54 \pm 1.77$	$1.04 \pm 1.18$	$0.77 \pm 1.07$	$2.95 \pm 1.76$		
Bicep curls - moderate	$0.46 \pm 0.49$	$0.46 \pm 0.54$	$0.99 \pm 0.77$	$0.88 \pm 0.55$	$1.94 \pm 1.62$	$0.68 \pm 0.65$	$2.96 \pm 3.52$		
Calisthenics - Crunches	$0.25 \pm 0.11$	$0.27 \pm 0.32$	$0.23 \pm 0.13$	$0.48 \pm 0.90$	$0.37 \pm 0.49$	$0.98 \pm 0.73$	$1.86 \pm 1.11$		
Calisthenics - Sit ups	$0.16 \pm 0.08$	$0.13 \pm 0.16$	$0.16 \pm 0.21$	$0.16 \pm 0.18$	$0.33 \pm 0.74$	$0.34 \pm 0.56$	$1.05 \pm 1.01$		
Cycling - Cycle hard - Cycle 80rpm	$0.63 \pm 0.52$	$0.72 \pm 0.79$	$0.84 \pm 0.73$	$0.67 \pm 0.51$	$1.13 \pm 0.94$	$1.27 \pm 1.02$	$1.68 \pm 0.26$		
Cycling - Cycle light - Cycle 100rpm	$0.48 \pm 0.51$	$0.31 \pm 0.40$	$0.52 \pm 1.06$	$0.58 \pm 1.37$	$0.40 \pm 0.85$	$0.22 \pm 0.46$	$0.46 \pm 1.26$		
Cycling - Cycle light - Cycle 60rpm	$0.61 \pm 0.56$	$0.52 \pm 0.69$	$0.32 \pm 0.20$	$0.30 \pm 0.24$	0.99 ± 2.16	$0.80 \pm 1.40$	$2.02 \pm 3.03$		
Cycling - Cycle light - Cycle 80rpm	$0.89 \pm 0.91$	$0.93 \pm 1.00$	$0.56 \pm 0.53$	$1.00 \pm 1.21$	$1.46 \pm 1.41$	$1.16 \pm 1.03$	$2.68 \pm 2.41$		
Cycling - Cycle moderate - Cycle 80rpm	$0.68 \pm 0.33$	$0.63 \pm 0.35$	$0.91 \pm 0.72$	$1.39 \pm 1.33$	$0.78 \pm 0.60$	$0.84 \pm 0.72$	$2.50 \pm 2.14$		
Lying down	$2.33 \pm 0.86$	$1.86 \pm 0.79$	$1.62 \pm 0.93$	$1.46 \pm 0.85$	$1.86 \pm 1.07$	$1.15 \pm 1.25$	$0.98 \pm 1.12$		
Rowing - Rowing hard - Rowing 30spm	$0.72 \pm 0.53$	$0.67 \pm 0.70$	$0.76 \pm 0.96$	$0.71 \pm 0.88$	$1.15 \pm 0.97$	$1.66 \pm 1.26$	$2.61 \pm 2.03$		
Rowing - Rowing light - Rowing 30spm	$0.82 \pm 0.58$	$0.72 \pm 0.81$	$0.82 \pm 0.85$	$0.92 \pm 0.85$	$1.51 \pm 1.16$	$1.21 \pm 1.75$	$2.19 \pm 1.23$		
Rowing - Rowing moderate - Rowing 30spm	$0.87 \pm 0.65$	$0.69 \pm 0.65$	$0.71 \pm 0.67$	$0.84 \pm 0.65$	$1.02 \pm 0.98$	$1.72 \pm 1.22$	$2.42 \pm 0.98$		
Running - Treadmill 4mph - Treadmill 0	$0.64 \pm 0.62$	$0.41 \pm 0.45$	$0.75 \pm 0.74$	$0.87 \pm 0.99$	$1.06 \pm 1.20$	$1.63 \pm 2.53$	$3.34 \pm 3.55$		
Running - Treadmill 5mph - Treadmill 0	$0.76 \pm 0.63$	$0.78 \pm 0.73$	$0.58 \pm 0.43$	$0.85 \pm 0.90$	$0.70 \pm 0.74$	$1.11 \pm 0.87$	$1.37 \pm 1.37$		
Running - Treadmill 6mph - Treadmill 0	$0.47 \pm 0.51$	$0.52 \pm 0.59$	$0.82 \pm 1.03$	$0.71 \pm 0.75$	$1.20 \pm 1.00$	$1.31 \pm 1.28$	$1.18 \pm 0.70$		
Sitting	$0.94 \pm 0.32$	$0.87 \pm 0.25$	$0.81 \pm 0.37$	$0.87 \pm 0.55$	$0.89 \pm 0.61$	$0.90 \pm 0.62$	$1.72 \pm 1.67$		
Sitting - Fidget feet legs	$0.30 \pm 0.20$	$0.33 \pm 0.19$	$0.45 \pm 0.39$	$0.34 \pm 0.31$	$0.36 \pm 0.40$	$0.45 \pm 0.62$	$0.75 \pm 1.40$		
Sitting - Fidget hands arms	$0.52 \pm 0.18$	$0.54 \pm 0.35$	$0.55 \pm 0.45$	$0.60 \pm 0.42$	$0.57 \pm 0.50$	$0.79 \pm 1.18$	$1.16 \pm 1.22$		
Stairs - Ascend stairs	$1.05 \pm 0.56$	$0.85 \pm 0.34$	$0.55 \pm 0.34$	$0.53 \pm 0.59$	$0.37 \pm 0.46$	$0.51 \pm 0.78$	$0.00 \pm 0.00$		
Stairs - Descend stairs	$1.40 \pm 0.86$	$1.04 \pm 0.77$	$0.76 \pm 0.58$	$0.59 \pm 0.54$	$0.83 \pm 1.24$	$0.67 \pm 0.83$	$0.00 \pm 0.00$		
Standing	$0.91 \pm 0.26$	$0.90 \pm 0.29$	$0.82 \pm 0.28$	$0.82 \pm 0.40$	$0.89 \pm 0.54$	$1.11 \pm 0.67$	$2.58 \pm 1.32$		
Walking - Treadmill 2mph - Treadmill 0	$0.87 \pm 0.73$	$0.74 \pm 0.89$	$0.92 \pm 1.55$	$0.61 \pm 0.42$	$0.72 \pm 0.97$	$1.66 \pm 1.80$	$1.62 \pm 1.93$		
Walking - Treadmill 3mph - Treadmill 0	$1.21 \pm 0.61$	$1.43 \pm 0.79$	$1.28 \pm 0.84$	$1.31 \pm 1.06$	$1.55 \pm 1.56$	$1.91 \pm 1.17$	$3.04 \pm 1.99$		
Walking - Treadmill 3mph - Treadmill 3 - light	$1.00 \pm 0.58$	$1.20 \pm 0.50$	$1.23 \pm 0.70$	$1.34 \pm 1.17$	$1.68 \pm 1.26$	$1.71 \pm 1.65$	$2.86 \pm 2.48$		
Walking - Treadmill 3mph - Treadmill 6 - moderate	$1.02 \pm 0.49$	$1.05 \pm 0.62$	$1.08 \pm 0.62$	$1.03 \pm 0.65$	$1.32 \pm 1.45$	$1.82 \pm 1.77$	$3.05 \pm 2.29$		
Walking - Treadmill 3mph - Treadmill 9 - hard	$1.12 \pm 1.03$	$0.73 \pm 0.39$	$1.06 \pm 0.76$	$0.94 \pm 0.96$	$0.90 \pm 0.78$	$1.91 \pm 1.32$	$1.95 \pm 1.78$		
kneeling	$0.87 \pm 0.24$	$0.88 \pm 0.36$	$0.81 \pm 0.34$	$0.70 \pm 0.35$	$0.84 \pm 0.53$	$1.23 \pm 0.72$	$2.17 \pm 1.32$		
unknown	$27.80 \pm 4.75$	$27.03 \pm 4.00$	24.11 ± 5.33	$22.04 \pm 6.30$	$18.79 \pm 5.48$	$14.61 \pm 6.51$	$10.17 \pm 4.50$		
Carrying groceries	$1.26 \pm 0.62$	$1.19 \pm 0.97$	$1.43 \pm 1.20$	$1.50 \pm 2.15$	$0.87 \pm 1.14$	$1.98 \pm 1.88$	$2.29 \pm 1.63$		
Doing dishes	$1.53 \pm 0.40$	$1.48 \pm 0.45$	$1.17 \pm 0.52$	$1.17 \pm 0.68$	$1.06 \pm 0.72$	$0.98 \pm 0.91$	$1.29 \pm 1.40$		
Gardening	$0.86\pm0.38$	$0.78 \pm 0.40$	$0.72 \pm 0.44$	$0.63 \pm 0.43$	$0.92 \pm 0.60$	$1.12 \pm 1.38$	$2.55 \pm 1.56$		
Ironing	$1.40 \pm 0.48$	$1.34 \pm 0.50$	$1.33 \pm 0.62$	$1.15 \pm 0.70$	$1.24 \pm 0.64$	$1.42 \pm 1.32$	$1.88 \pm 0.96$		
Making the bed	$1.60 \pm 0.66$	$1.49 \pm 0.60$	$1.27 \pm 0.77$	$1.04 \pm 0.79$	$0.87 \pm 0.73$	$0.41 \pm 0.36$	$1.41 \pm 1.44$		
Mopping	$1.13 \pm 0.46$	$1.19 \pm 0.70$	$1.57 \pm 1.47$	$0.86 \pm 0.48$	$0.96 \pm 0.44$	$0.70 \pm 0.68$	$1.31 \pm 0.94$		
Playing videogames	$1.37 \pm 0.56$	$1.32 \pm 0.58$	$1.38 \pm 0.67$	$1.51 \pm 0.67$	$1.64 \pm 0.83$	$2.05 \pm 1.07$	$2.51 \pm 2.33$		
Scrubbing a surface	$0.94 \pm 0.47$	$0.94 \pm 0.44$	$0.95 \pm 0.58$	$0.85 \pm 0.52$	$1.21 \pm 1.03$	$1.43 \pm 0.79$	$2.39 \pm 2.41$		
Stacking groceries	$1.12 \pm 0.42$	$0.96 \pm 0.37$	$0.90 \pm 0.34$	$0.94 \pm 0.63$	$0.59 \pm 0.45$	$1.18 \pm 0.85$	$2.37 \pm 0.51$		
Sweeping	$1.05 \pm 0.37$	$1.04 \pm 0.40$	$0.94 \pm 0.38$	$0.88 \pm 0.50$	$0.96 \pm 0.61$	$0.89 \pm 0.66$	$1.67 \pm 1.68$		
Typing	$0.95 \pm 0.43$	$0.86 \pm 0.40$	$0.87 \pm 0.61$	$0.90 \pm 0.67$	$1.34 \pm 0.92$	$1.12 \pm 0.98$	$2.65 \pm 2.49$		
Vacuuming	$0.95 \pm 0.31$	$0.89 \pm 0.29$	$0.86 \pm 0.31$	$0.77 \pm 0.38$	$0.79 \pm 0.48$	$1.54 \pm 1.07$	$1.22 \pm 1.10$		
Walking around block	$1.72 \pm 0.77$	$1.87 \pm 1.29$	$1.74 \pm 1.32$	$1.71 \pm 1.11$	$1.63 \pm 1.42$	$1.60 \pm 1.23$	$1.48 \pm 1.81$		
Washing windows	$0.96 \pm 0.40$	$1.02 \pm 0.45$	$0.92 \pm 0.47$	$1.01 \pm 0.75$	$1.12 \pm 0.66$	$1.13 \pm 0.77$	$1.06 \pm 0.91$		
Watching TV	$1.55 \pm 0.37$	$1.46\pm0.35$	$1.41 \pm 0.53$	$1.59\pm0.54$	$1.68 \pm 0.74$	$1.56 \pm 1.09$	$2.82 \pm 1.61$		
Weeding	$0.67 \pm 0.22$	$0.63 \pm 0.22$	$0.63 \pm 0.29$	$0.69 \pm 0.37$	$0.78 \pm 0.56$	$0.95 \pm 0.50$	$1.72 \pm 1.26$		
Wiping/Dusting	$1.16 \pm 0.50$	$1.14 \pm 0.49$	$1.04 \pm 0.53$	$1.10 \pm 0.73$	$0.72 \pm 0.66$	$1.24 \pm 0.95$	$1.77 \pm 1.43$		
Writing	$0.78\pm0.38$	$0.70\pm0.46$	$0.62 \pm 0.41$	$0.79\pm0.55$	$0.74 \pm 0.52$	$1.00\pm0.75$	$1.05 \pm 1.23$		
taking out trash	$0.96 \pm 0.42$	$0.96 \pm 0.32$	$0.96 \pm 0.41$	$0.88 \pm 0.49$	$0.66 \pm 0.41$	$1.23 \pm 0.75$	$0.72 \pm 1.16$		

 Table A5-11: False positive rate when training the C4.5 classifier using the FFTCorr feature and subject independent evaluation

				F-Measure			
Class	1.4s	2.8s	5.6s	11.3s	22.7s	44.5s	90s
Bench weight lifting - hard	$5.26 \pm 9.21$	$3.24 \pm 4.89$	$3.78 \pm 5.85$	$4.48 \pm 10.49$	$4.78 \pm 15.87$	$8.33 \pm 20.41$	$0.00 \pm 0.00$
Bench weight lifting - light	$14.84 \pm 13.74$	$16.47 \pm 16.85$	15.19 ± 18.11	$17.50 \pm 20.92$	13.86 ± 17.78	21.58 ± 25.79	25.00 ± 46.29
Bench weight lifting - moderate	$9.48 \pm 10.07$	$8.40 \pm 10.20$	$7.82 \pm 10.41$	$14.94 \pm 16.67$	8.95 ± 20.21	$15.83 \pm 24.58$	$20.00 \pm 27.39$
Bicep curls - hard	$11.65 \pm 15.04$	$14.83 \pm 18.13$	$10.30 \pm 15.29$	34.56 ± 24.33	37.19 ± 33.83	$43.10 \pm 42.17$	$17.78 \pm 28.80$
Bicep curls - light	19.14 + 20.86	$21.96 \pm 20.14$	27.31 + 20.99	32.49 + 21.18	50.85 + 33.34	$56.30 \pm 46.80$	36.00 + 40.99
Bicep curls - moderate	$10.45 \pm 13.54$	$15.70 \pm 19.13$	20.81 + 20.30	$17.09 \pm 17.02$	38 57 + 28 61	71 39 + 19 68	$32.38 \pm 33.40$
Calisthenics - Crunches	$11.67 \pm 23.95$	$10.55 \pm 23.91$	14 11 + 27 55	$26.61 \pm 40.31$	26 27 + 36 76	$11.67 \pm 24.91$	$0.00 \pm 0.00$
Calisthenics - Sit ups	$40.68 \pm 42.28$	$39.74 \pm 44.63$	$58.82 \pm 41.34$	57.68 ± 40.96	$67.80 \pm 35.72$	$72.78 \pm 35.20$	$53.33 \pm 50.55$
Cycling - Cycle bard - Cycle 80rpm	$12.45 \pm 14.01$	$9.00 \pm 9.42$	$13.70 \pm 18.82$	970 + 16.82	15 87 ± 19 24	19.68 ± 26.58	12 50 ± 25 00
Cycling - Cycle light - Cycle 100rpm	$12.43 \pm 14.01$ 56.02 ± 30.53	$5.00 \pm 5.42$	$13.70 \pm 10.02$ 63.65 ± 35.07	$74.60 \pm 33.44$	73 26 ± 30 10	17.00 ± 20.00 87.80 ± 26.14	$12.30 \pm 23.00$ $03.33 \pm 17.73$
Cycling Cycle light Cycle 1001pm	14 21 ± 22 25	$55.05 \pm 22.71$	58 52 ± 21 05	50 50 ± 33.44	$62.72 \pm 20.57$	71.81 ± 21.22	70.21 ± 20.45
Cycling - Cycle light - Cycle 001pm	44.21 ± 22.23	$35.05 \pm 32.71$	28 01 ± 20 65	$39.30 \pm 32.72$	$02.72 \pm 30.37$	71.81 ± 31.23	70.21 ± 29.45
Cycling - Cycle light - Cycle sorphi	$27.95 \pm 10.34$	$33.00 \pm 23.30$	$38.91 \pm 30.03$	$32.30 \pm 20.04$	$33.10 \pm 24.31$	$05.00 \pm 24.04$	39.07 ± 41.33
Cycling - Cycle moderate - Cycle	$15.51 \pm 12.41$	$11.74 \pm 11.40$	22.07 ± 19.81	20.34 ± 19.04	$17.75 \pm 21.76$	38.00 ± 31.09	$50.50 \pm 57.15$
Solphi	40.08 - 10.07	(2.57 . 10.21	(7.22 . 19.05	71 (5 + 12.05	(2.02 + 16.62	77.14 - 16.50	95.00 . 11.07
Lying down	49.28 ± 12.27	63.57 ± 10.21	67.33 ± 18.95	/1.05 ± 13.25	68.98 ± 16.62	77.14 ± 16.59	85.08 ± 11.97
Rowing - Rowing liald - Rowing Sospin	14.89 ± 13.28	$13.32 \pm 10.24$	$21.36 \pm 24.34$	8.51 ± 10.49	$15.34 \pm 21.82$	38.55 ± 20.51	37.30 ± 47.87
Rowing - Rowing light - Rowing 30spm	$21.08 \pm 13.74$	$21.46 \pm 18.60$	$22.95 \pm 20.93$	42.58 ± 21.45	26.95 ± 25.37	48.51 ± 33.67	$29.26 \pm 35.50$
30spm	$18.22 \pm 13.47$	$17.07 \pm 16.44$	18.19 ± 16.19	$20.76 \pm 22.43$	$17.57 \pm 25.25$	$14.72 \pm 26.77$	29.17 ± 31.81
Running - Treadmill 4mph - Treadmill	26.30 + 23.28	$28.56 \pm 29.77$	28.85 + 26.52	32.09 + 31.30	$41.69 \pm 38.01$	30.07 + 37.14	$43.50 \pm 43.81$
0							
Running - Treadmill 5mph - Treadmill	35.27 ± 26.26	$38.79 \pm 26.44$	$35.43 \pm 28.82$	36.64 ± 26.62	38.37 ± 36.24	47.97 ± 34.18	64.00 ± 39.40
0							
Running - Treadmill 6mph - Treadmill	$25.51 \pm 21.02$	$27.23 \pm 25.98$	$39.38 \pm 27.49$	$44.56 \pm 34.19$	$30.25 \pm 29.91$	$42.53 \pm 37.20$	$60.00 \pm 36.51$
0							
Sitting	$6.35 \pm 5.86$	$4.15 \pm 4.50$	6.38 ± 7.54	$6.98 \pm 10.02$	$14.25 \pm 19.43$	$16.24 \pm 26.18$	$41.67 \pm 46.65$
Sitting - Fidget feet legs	34.75 ± 29.29	37.59 ± 31.36	35.60 ± 30.93	$40.85 \pm 35.50$	53.32 ± 36.59	67.69 ± 35.55	$66.67 \pm 50.00$
Sitting - Fidget hands arms	$14.09 \pm 15.73$	$16.83 \pm 17.77$	$19.86 \pm 18.69$	$23.08 \pm 26.86$	$10.86 \pm 18.78$	$48.57 \pm 35.61$	$57.14 \pm 44.99$
Stairs - Ascend stairs	$36.55 \pm 23.17$	$38.96 \pm 25.34$	$50.05 \pm 31.73$	$50.03 \pm 35.55$	$60.24 \pm 34.55$	$75.77 \pm 21.33$	$0.00 \pm 0.00$
Stairs - Descend stairs	$24.96 \pm 18.25$	$31.08 \pm 24.03$	$42.66 \pm 29.01$	$35.63 \pm 32.42$	$44.21 \pm 40.89$	$53.03 \pm 33.60$	$0.00 \pm 0.00$
Standing	$6.75 \pm 5.06$	$8.52 \pm 7.62$	$12.01 \pm 11.24$	$18.55 \pm 15.56$	$17.09 \pm 17.51$	$17.00 \pm 29.26$	$13.33 \pm 29.81$
Walking - Treadmill 2mph - Treadmill	37.46 ± 23.50	42.21 ± 23.93	42.06 ± 31.50	48.48 ± 32.21	49.18 ± 31.87	48.87 ± 34.18	76.96 ± 17.77
Walking - Treadmill 3mph - Treadmill	15.67 ± 10.78	15.16 ± 11.38	17.84 ± 13.07	18.08 ± 15.62	20.02 ± 21.80	11.23 ± 17.50	26.55 ± 25.25
0							
Walking - Treadmill 3mph - Treadmill	$10.65 \pm 8.07$	$12.12\pm7.04$	$14.97 \pm 11.45$	$14.05 \pm 14.23$	19.73 ± 15.95	$16.62\pm19.89$	$26.19\pm34.36$
Walking Treadmill 2mph Treadmill	$10.44 \pm 8.72$	$0.52 \pm 8.06$	$11.41 \pm 11.27$	9 42 + 11 71	$10.05 \pm 16.92$	$15.54 \pm 24.26$	$25.02 \pm 26.50$
waiking - Treadmin Shiph - Treadmin	10.44 ± 0.72	$9.32 \pm 0.90$	11.41 ± 11.27	0.45 ± 11.71	10.95 ± 10.82	13.34 ± 24.20	$25.95 \pm 20.50$
Walking Traadmill 2mph Traadmill	14 27 + 12 78	$14.04 \pm 14.20$	$0.20 \pm 10.05$	$16.10 \pm 14.75$	20.82 + 22.42	18.06 + 28.08	17 78 + 22 22
9 - hard	14.27 ± 12.70	14.74 ± 14.37	J.50 ± 10.75	10.10 ± 14.75	20.02 ± 25.42	10.70 ± 20.70	17.76 ± 52.55
kneeling	8 06 + 5 36	8 93 + 8 70	15 50 + 15 06	13.96 + 16.48	$15.57 \pm 18.00$	12 78 + 18 67	25 76 + 37 54
unknown	48 31 + 7 29	$50.37 \pm 7.72$	$13.37 \pm 13.70$ 53.66 ± 7.62	$15.70 \pm 10.48$ 56.70 ± 8.43	59.62 ± 8.81	$60.08 \pm 11.10$	50 22 ± 16 11
Carrying groceries	$-70.31 \pm 7.27$ 22.07 ± 18.11	$26.78 \pm 21.90$	$24.02 \pm 20.60$	20 16 + 27 64	$35.02 \pm 0.01$	$23.31 \pm 25.24$	$37.22 \pm 10.11$ $42.41 \pm 35.47$
Carrying grocenes	$16.19 \pm 0.79$	$20.76 \pm 21.69$	$24.92 \pm 20.09$	$29.10 \pm 27.04$	$35.46 \pm 32.00$	$23.31 \pm 23.34$	42.41 ± 33.47
Condening	11.76 + 12.04	12.06 ± 14.09	$23.80 \pm 10.73$	$29.30 \pm 21.90$	$30.33 \pm 27.18$	43.40 ± 31.97	$00.46 \pm 33.30$
Januaring	17.05 + 11.12	$15.00 \pm 13.55$	$17.00 \pm 20.37$	$20.43 \pm 23.44$	$27.62 \pm 52.44$	21.09 ± 30.22	20.91 ± 33.00
Molving the hed	17.95 ± 11.12	23.40 ± 13.49	26.40 ± 10.78	20.07 ± 22.93	45 91 ± 22.34	$51.20 \pm 20.40$	44.70 ± 33.04
Making the bed	15.05 ± 9.71	$17.53 \pm 11.50$	23.28 ± 18.03	$35.12 \pm 22.32$	43.81 ± 23.87	04.39 ± 26.36	33.81 ± 33.01
Mopping	$15.25 \pm 13.38$	10.87 ± 13.90	17.00 ± 15.82	21.98 ± 21.09	$23.32 \pm 22.50$	38.14 ± 25.05	44.44 ± 50.00
Playing videogames	10.55 ± 11.11	$17.86 \pm 12.70$	$15.75 \pm 12.30$	15.79 ± 16.52	13.00 ± 13.31	18.47 ± 22.26	$32.38 \pm 31.34$
Scrubbing a surface	$10.02 \pm 9.48$	$11.55 \pm 10.93$	14.39 ± 15.05	16.72 ± 17.86	18.75 ± 22.80	$24.04 \pm 26.30$	35.67 ± 41.93
Stacking groceries	9.10 ± 9.00	$12.22 \pm 11.82$	15.26 ± 15.84	17.73 ± 19.19	$24.90 \pm 24.60$	32.86 ± 33.23	$0.00 \pm 0.00$
Sweeping	9.01 ± 8.61	$11.50 \pm 10.69$	14.4/±14.95	19.69 ± 17.77	20.80 ± 19.02	30.25 ± 25.98	54.72 ± 45.20
1 yping	39.82 ± 19.47	45.68 ± 22.16	42.25 ± 22.76	41.80 ± 25.67	28.31 ± 23.48	33.84 ± 23.70	29.49 ± 40.91
Vacuuming	$14.27 \pm 12.27$	15.19 ± 13.27	$16.03 \pm 15.12$	$18.52 \pm 17.82$	$25.93 \pm 23.01$	$20.95 \pm 24.44$	30.56 ± 41.34
Walking around block	$15.89 \pm 12.33$	$15.22 \pm 12.67$	$15.80 \pm 17.86$	$22.49 \pm 20.77$	$23.37 \pm 23.98$	$22.76 \pm 28.40$	63.33 ± 38.26
Washing windows	$14.70 \pm 11.49$	15.41 ± 12.91	$23.60 \pm 19.90$	$23.95 \pm 19.08$	$25.35 \pm 20.46$	33.95 ± 30.83	46.43 ± 39.86
Watching TV	$9.09 \pm 5.71$	$10.97 \pm 7.39$	$15.43 \pm 9.96$	$15.99 \pm 10.41$	$11.68 \pm 12.63$	$12.87 \pm 15.37$	$29.17 \pm 28.97$
Weeding	$3.75 \pm 4.41$	$3.72 \pm 4.72$	$4.72 \pm 7.62$	$5.01 \pm 8.06$	$6.67 \pm 10.76$	8.35 ± 15.70	37.62 ± 36.25
Wiping/Dusting	$11.64 \pm 8.40$	$12.64 \pm 8.72$	$18.17 \pm 15.22$	$22.24 \pm 16.89$	35.98 ± 28.59	44.31 ± 26.72	$51.86 \pm 37.37$
Writing	$45.29 \pm 25.82$	$49.81 \pm 29.15$	$52.07 \pm 30.12$	46.78 ± 32.37	$56.42 \pm 32.71$	48.89 ± 30.76	47.71 ± 37.86
taking out trash	552 + 485	696 + 653	627 + 711	9 35 + 11 55	1458 + 1915	20.94 + 26.24	$50.00 \pm 45.13$

 Taking out ta

## **Appendix A6: Feature Selection for Activity Recognition**

Features subsets	All	Postures	Ambulation	Exercise	Resistance	Household
(Number of features)	Activities				Exercise	
ACFFT Peaks (70)	$41.59 \pm 4.31$	27.19±19.59	28.19±22.73	28.16±24.28	20.11±18.42	19.77±16.57
		(0.82±0.47)	(0.97±0.69)	(0.60±0.56)	(0.82±0.64)	(1.05±0.59)
ACAbsMean	$38.14 \pm 4.95$	23.3±15.0	26.4±24.5	21.4±23.7	14.2±17.8	$16.8 \pm 15.0$
		(0.8±0.4)	(1.1±1.0)	(0.7±0.7)	(0.8±0.7)	(1.0±0.6)
ACIQR (7)	$37.92 \pm 4.46$	21.86±13.79	25.52±20.29	26.16±25.03	16.19±17.98	15.36±13.75
		(0.88±0.49)	(1.13±0.86)	(0.69±0.66)	(0.89±0.75)	$(1.05\pm0.61)$
ACQ3 (7)	$37.01 \pm 4.47$	20.13±12.54	24.08±19.07	22.59±22.73	13.84±16.16	$14.24 \pm 12.58$
		(0.85±0.46)	(1.10±0.78)	(0.63±0.56)	$(0.86 \pm 0.64)$	(1.09±0.59)
ACAbsArea(7)	$36.82\pm5.51$	21.19±15.12	26.48±22.65	22.80±24.12	14.96±17.99	16.62±14.49
		$(0.87 \pm 0.51)$	$(1.09\pm0.95)$	$(0.64 \pm 0.69)$	$(0.79\pm0.70)$	$(1.11\pm0.65)$
ACFFTCoeff (889)	$36.76 \pm 3.78$	23.15±14.79	24.25±18.60	25.45±21.07	17.88±16.13	$14.24 \pm 11.90$
		$(0.93\pm0.41)$	$(1.09\pm0.67)$	$(0.67 \pm 0.49)$	$(0.89\pm0.60)$	$(1.18\pm0.52)$
ACVar (7)	$34.81 \pm 4.49$	$18.64 \pm 3.40$	25.84±23.77	$20.80 \pm 24.37$	15.61±19.44	$9.97 \pm 10.01$
		$(2.20\pm0.78)$	$(1.07 \pm 1.02)$	$(0.62 \pm 0.60)$	$(0.78\pm0.76)$	$(0.72\pm0.52)$
ACPitch (7)	$34.79 \pm 6.91$	7.00±9.76	18.42±16.88	$21.65 \pm 16.70$	13.45±11.48	$2.64 \pm 3.16$
	2440 455	(0.30±0.20)	(1.19±0.84)	$(0.71\pm0.62)$	$(0.84\pm0.56)$	$(0.37\pm0.22)$
ACBandEnergy (7)	$34.19 \pm 4.55$	18.47±5.62	$20.60 \pm 17.14$	16.93±16.53	9.00±10.20	7.84±7.66
	24.14 4.61	$(1.00\pm0.46)$	$(1.22\pm0.72)$	$(0.71\pm0.54)$	$(0.8/\pm0.52)$	$(0.88\pm0.45)$
ACSF (5)	$34.14 \pm 4.61$	$20.50\pm14.35$	$24.90\pm22.44$	$18.98 \pm 20.63$	$12.90\pm15.44$	$13.22\pm12.16$
	22.04 . 2.42	$(0.86\pm0.53)$	$(1.10\pm0.93)$	$(0./8\pm0.//)$	$(0.89\pm0.73)$	$(1.10\pm0.61)$
ACRange(7)	$32.94 \pm 3.43$	$20.55 \pm 12.28$	$18.6/\pm1/.41$	$18.28 \pm 18.85$	12.5/±13.88	$13.51\pm11.77$
$\mathbf{A} \mathbf{C} \mathbf{T}_{-1} \mathbf{A} \mathbf{b}_{-1}	22.49 + 6.72	$(0.92\pm0.51)$	$(1.19\pm0.90)$	$(0.70\pm0.54)$	$(0.92\pm0.63)$	$(1.06\pm0.53)$
ACTOTALADSAFEA (1)	$32.48 \pm 6.73$	$14.24\pm5.03$	$22.33\pm13.80$	$11.0/\pm 0.51$	$2.87\pm2.97$	$2.79\pm2.95$
ACT-+-18E (1)	22.46 + 7.92	$(0.73\pm0.39)$	$(1.25\pm0.05)$	$(0.52\pm0.51)$	$(0.69\pm0.28)$	$(0.41\pm0.24)$
ACTOTALSF (1)	$32.40 \pm 7.82$	$13.30\pm4.03$	$20.03\pm10.10$ (1.24±0.72)	$10.19\pm7.43$	$3.40\pm3.30$	$1.00\pm1.10$
ACMCD (7)	21 75 + 2 22	$(1.03\pm0.34)$	$(1.24\pm0.73)$	$(0.41\pm0.24)$	$(0.00\pm0.30)$	$(0.23\pm0.17)$
ACMICK (7)	$51.75 \pm 5.52$	$(1.24\pm0.45)$	$(1.32\pm0.78)$	$(0.79\pm0.54)$	$9.04 \pm 10.00$	$(0.96\pm0.47)$
ACL owEnergy (7)	$31.0 \pm 5.6$	$(1.24\pm0.43)$ 12 80+8 43	$(1.52\pm0.78)$ 15.7+12.5	$(0.7)\pm0.54)$	91+87	1 6+2 2
ACLOWLINITGY (7)	$51.0 \pm 5.0$	$(1.63\pm0.70)$	(1.3+0.7)	(0.7+0.5)	$(0.7\pm0.4)$	(0.4+0.2)
ACModVigEnergy (7)	$30.80 \pm 4.88$	9.69+8.16	$20.88 \pm 17.01$	20.81+15.18	$12 12 \pm 10.63$	3 13+3 50
richiod (igEnergy (i))	50.00 ± 1.00	(1.25+0.52)	(1.22+0.73)	(0.74+0.50)	(0.90+0.52)	(0.79+0.32)
DCTotalMean	29.25 + 7.54	4.7+5.4	2.4+4.2	1.6+2.5	0.9+1.7	0.0+0.0
		$(0.8\pm1.0)$	(0.5±0.5)	$(0.5\pm0.3)$	$(0.4\pm0.2)$	$(0.2\pm0.2)$
ACDomFreqRatio (7)	$28.87 \pm 4.65$	13.67±4.97	11.49±11.29	10.16±9.02	6.20±6.08	5.13±5.43
1 ()		(1.06±0.46)	(1.33±0.69)	(0.84±0.45)	(0.91±0.38)	(0.87±0.38)
DCPostureDist	$28.38 \pm 5.48$	33.0±22.0	18.7±25.0	18.8±27.2	14.8±23.6	15.0±19.3
		(0.4±0.5)	(1.0±1.1)	(0.8±1.1)	(0.9±1.2)	(1.2±1.3)
ACEnergy (7)	$27.99 \pm 4.93$	7.75±8.02	19.95±18.78	14.54±13.89	9.11±10.62	3.59±4.19
		(1.25±0.51)	(1.21±0.96)	$(0.84 \pm 0.53)$	(0.97±0.57)	$(0.82 \pm 0.44)$
ACEntropy (7)	$27.89 \pm 6.98$	$11.44 \pm 4.42$	14.52±13.71	6.61±6.72	3.80±4.66	$1.78 \pm 2.25$
		(1.41±0.68)	(1.28±0.77)	(0.63±0.36)	(0.75±0.39)	(0.51±0.29)
ACAbsCV (7)	$27.56 \pm 2.96$	13.83±7.57	14.03±13.76	$14.28 \pm 16.91$	10.22±11.93	$9.64 \pm 8.27$
		$(1.07\pm0.44)$	(1.24±0.74)	(0.86±0.59)	$(1.01\pm0.64)$	(1.16±0.47)
ACFWTCoeff ()	$26.61 \pm 3.47$	$17.10 \pm 10.08$	$14.86 \pm 11.00$	$16.68 \pm 14.38$	11.59±10.36	$6.94 \pm 5.87$
		$(1.08\pm0.44)$	$(1.31\pm0.50)$	$(0.87 \pm 0.42)$	$(1.07\pm0.46)$	$(1.35\pm0.42)$
DCMean (7)	$24.69 \pm 6.81$	14.41±15.85	8.60±15.94	$12.80 \pm 21.77$	$7.40 \pm 14.17$	7.04±11.37
		$(0.66 \pm 1.14)$	$(0.94 \pm 1.22)$	$(0.79 \pm 1.01)$	$(0.94 \pm 1.28)$	$(1.26\pm1.52)$
DCArea (7)	$24.53 \pm 6.63$	12.23±16.16	8.87±17.00	12.52±21.85	7.75±14.77	7.72±13.16
1.00 (01)	04.14 4.65	(0.57±0.78)	$(0.99\pm1.42)$	$(0.86\pm1.11)$	$(0.90\pm1.10)$	$(1.22\pm1.38)$
ACCorr (21)	$24.11 \pm 4.29$	9.19±4.11	$14.02\pm17.11$	11.90±16.34	10.47±13.85	6.98±7.09
ACKee (7)	01.01 + 2.25	$(1.35\pm0.56)$	(0.94±0.83)	$(0.75\pm0.74)$	$(0.82\pm0.75)$	(1.36±0.60)
ACKur (/)	$21.81 \pm 3.36$	9.74±4.54	/.92±8.3/	$(1.25\pm10.05)$	$5.82\pm/.91$	$5.5/\pm 5.08$
A (CCL (7)	17.15 - 0.47	$(1.31\pm0.40)$	$(1.29\pm0.57)$	(1.05±0.51)	$(1.09\pm0.52)$	$(1.12\pm0.39)$
AUSKEW (7)	$17.15 \pm 2.47$	$0.30\pm 3.91$	$9.18 \pm 10.83$ (1.26 $\pm 0.67$ )	$4.80\pm0.03$	$3.97\pm 3.28$ (1.15±0.42)	$2.79\pm 3.28$
		$(1.42\pm0.40)$	$(1.20\pm0.07)$	$(1.09\pm0.40)$	$(1.15\pm0.42)$	$(1.14\pm0.37)$

Table A6-1: True positive rate per activity category while evaluating different subsets of features using the C4.5 decision tree classifier using a sliding window of 5.6s and evaluating the results in a subject independent manner. False positive rate is shown in parenthesis.

Features subsets (Number of features)	All Activities	Postures	Ambulation	Exercise	Resistance Exercise	Household
ACEnergy ACEntropy	$26.01 \pm 3.05$	10 /3+8 60	10 26+20 10	14 13+14 28	0.78+11.78	5 03+5 20
ACEhergy, ACEhiropy	$20.01 \pm 3.95$	$(1.22\pm0.67)$	$(1.22\pm0.08)$	$(0.85\pm0.54)$	$9.78\pm11.78$	$(1.10\pm0.52)$
ACIOP ACO1 ACO2 ACO2	$26.51 \pm 4.41$	$(1.33\pm0.07)$	$(1.23\pm0.98)$	$(0.05\pm0.04)$	$(0.98\pm0.39)$	$(1.10\pm0.52)$
ACIQK, ACQ1, ACQ2, ACQ3	$50.51 \pm 4.41$	$(0.02\pm0.48)$	$(1.00\pm0.82)$	$23.04\pm23.07$	$(0.88\pm0.67)$	$(1.18\pm0.65)$
ACAboAroo ACAroo ACVor	20.97 + 6.44	$(0.93\pm0.48)$	$(1.09\pm0.03)$	$(0.00\pm0.30)$	$(0.88\pm0.07)$	$(1.18\pm0.03)$
ACADSAFea, ACAFea, ACVar	$39.87 \pm 0.44$	$40.40\pm31.33$	$20.81\pm22.99$	$22.10\pm 20.84$	$15.95\pm21.81$	$19.20\pm 21.03$
	40.14 5.05	$(0.48\pm0.62)$	$(1.13\pm1.21)$	$(0.70\pm0.82)$	$(0.95\pm1.08)$	$(0.97\pm0.91)$
ACAbsArea, ACArea	$40.14 \pm 5.95$	$39.78\pm31.23$	$22.48\pm24.87$	24.56±29.00	$16.39\pm23.79$	$18.70\pm20.97$
	40.40.4.57	$(0.49\pm0.00)$	$(1.02\pm1.14)$	$(0.70\pm0.83)$	$(0.91\pm1.04)$	$(0.92\pm0.84)$
ACAbsArea, ACVar,	$40.40 \pm 4.57$	$23.40\pm17.36$	28.80±24.20	25.70±25.13	18.36±19.97	19.90±17.28
ACLowEnergy, AC Range,		$(0.82\pm0.43)$	$(0.98\pm0.88)$	$(0.6/\pm 0.66)$	$(0.85\pm0.74)$	$(1.04\pm0.57)$
ACMCR	44.00 6.04	10.05.00.15	24.52.25.50	0.1.50.05.0.C	15.00.00.00	
ACAbsArea, ACArea, ACVar,	$41.00 \pm 6.01$	40.26±33.45	24.53±25.78	24./2±2/.86	17.93±23.98	20.73±22.07
ACRange, ACMCR, ACIQR		$(0.53\pm0.74)$	(0.91±0.85)	(0.66±0.77)	$(0.81\pm0.84)$	$(0.92\pm0.73)$
ACFFTPeaks, ACCorr	$41.23 \pm 5.04$	25.60±17.39	28.56±23.79	$27.54 \pm 25.82$	$20.01 \pm 20.92$	$20.26 \pm 16.85$
		$(0.84 \pm 0.46)$	$(1.02\pm0.84)$	$(0.60\pm0.58)$	$(0.79 \pm 0.63)$	$(1.10\pm0.63)$
ACAbsArea, ACArea, ACVar,	$41.69 \pm 6.58$	41.55±33.26	$24.44 \pm 24.35$	23.66±28.15	$16.89 \pm 22.20$	$21.78 \pm 23.35$
ACRange, ACMCR		$(0.45 \pm 0.60)$	$(0.97 \pm 0.99)$	$(0.70\pm0.76)$	$(0.89\pm0.90)$	$(0.94 \pm 0.80)$
ACVar, ACEnergy,	$41.70 \pm 4.74$	26.65±19.30	30.71±23.68	26.29±25.13	$18.96 \pm 20.44$	20.63±16.92
ACEntropy, ACFFTPeaks,		$(0.82 \pm 0.43)$	(0.98±0.95)	$(0.59 \pm 0.61)$	(0.73±0.63)	$(1.05\pm0.57)$
ACCorr						
ACVar, ACEnergy,	$41.73 \pm 4.19$	27.52±19.16	30.48±22.91	28.50±23.22	20.44±19.05	20.38±16.31
ACEntropy, ACFFTPeaks		(0.79±0.42)	(0.97±0.68)	(0.62±0.55)	(0.82±0.61)	(1.06±0.56)
ACVar, ACFFTPeaks	$41.82 \pm 4.22$	27.19±19.46	29.04±23.39	27.94±24.71	19.44±19.00	19.96±16.20
		$(0.79 \pm 0.43)$	$(1.00\pm0.75)$	$(0.60 \pm 0.56)$	(0.81±0.66)	$(1.06\pm0.59)$
ACVar.AC FFTPeaks.	$41.83 \pm 4.35$	27.52±20.01	29.86±23.93	27.85±24.39	19.72±18.40	20.08±16.16
ACBandEnergy		$(0.78\pm0.41)$	$(1.01\pm0.74)$	$(0.63 \pm 0.59)$	$(0.84 \pm 0.67)$	$(1.04\pm0.57)$
ACVar. ACBandEnergy.	41.82 + 5.28	27.06+17.97	30.40+24.72	28.61+26.69	20.13+21.35	20.96+17.09
ACEntropy, ACEFTPeaks.		(0.83+0.42)	(0.96+0.83)	(0.63+0.64)	(0.77+0.69)	(1.07+0.61)
ACCorr		(0.00-00.02)	(00,020000)	(0.00-0.0.0)	(011120105)	()
ACVar, ACEntropy,	$42.14 \pm 4.16$	28.19±19.51	30.70±23.61	28.05±23.92	20.29±19.75	20.96±17.10
ACBandEnergy, ACFFTPeaks		$(0.77\pm0.42)$	$(0.98 \pm 0.68)$	$(0.61\pm0.55)$	$(0.82\pm0.63)$	$(1.05\pm0.58)$
ACFETPeaks ACVar ACIOR	42.21 + 4.74	27.66+19.14	27 65+23 94	27 53+25 45	20 53+19 99	21.08+17.80
		$(0.76\pm0.42)$	$(1.02\pm0.78)$	(0.65+0.61)	(0.86+0.65)	(1.01+0.51)
ACVar ACEETPeaks	$4230 \pm 418$	27 38+18 33	28 96+23 36	26 85+24 63	19 79+19 26	21.00+17.39
ACBandEnergy ACIOR	12.50 ± 1.10	$(0.80\pm0.41)$	(1.02+0.79)	$(0.63\pm0.61)$	$(0.84 \pm 0.65)$	(1.01+0.52)
ACAbsArea ACArea ACVar	$42.85 \pm 6.56$	39 20+31 82	24 25+24 83	26 57+29 94	1938+2479	20 58+22 33
ACRange ACPitch	42.05 ± 0.50	$(0.46\pm0.58)$	$(0.99 \pm 1.04)$	$(0.69\pm0.76)$	$(0.86\pm0.90)$	$(0.93\pm0.79)$
ACAbsArea ACArea ACVar	$42.94 \pm 5.61$	$(0.40\pm0.50)$ 38.92+31.03	$(0.75 \pm 1.04)$	$26.01 \pm 0.10$	$(0.00\pm0.00)$ 19 26+22 89	$(0.93\pm0.19)$
ACEnergy ACEntropy	42.94 ± 5.01	$(0.48\pm0.64)$	$(0.92\pm0.88)$	$(0.63\pm0.71)$	$(0.84\pm0.83)$	$(0.94\pm0.75)$
ACEET Peaks ACCorr		(0.40±0.04)	$(0.92\pm0.00)$	(0.05±0.71)	(0.04±0.05)	$(0.94\pm0.75)$
ACRange ACMCR ACPitch						
All features but heart rate based	$43.12 \pm 6.70$	44 56+34 74	27 98+25 55	2671+2878	18 72+23 77	21 36+21 81
and subject characteristics	45.12 ± 0.70	$(0.45\pm0.50)$	$(0.96 \pm 1.04)$	$(0.62\pm0.73)$	$(0.85\pm0.93)$	$(1.00\pm0.92)$
ACAbsArea ACArea	$1124 \pm 6.88$	$(0.45\pm0.50)$	$(0.90\pm1.04)$	$(0.02\pm0.73)$ 28 70+30 03	$(0.05\pm0.05)$	$(1.00\pm0.02)$
ACRandEnargy ACEETDooles	44.24 ± 0.88	$(0.48\pm0.60)$	$(1.02 \pm 1.08)$	$(0.60\pm0.76)$	$(0.00\pm0.00)$	$(0.02\pm0.82)$
ACVar		$(0.48\pm0.00)$	$(1.02\pm1.08)$	$(0.09\pm0.70)$	$(0.90\pm0.99)$	$(0.93\pm0.82)$
ACAbsArss ACArss ACVer	44 20 + 6 58	11 66+25 12	29 59+26 62	27 95+29 62	10.84+22.67	22 70+22 78
ACAUSAIea, ACAiea, ACVai,	$44.39 \pm 0.36$	$(0.52\pm0.59)$	$(0.03\pm0.04)$	$(0.61\pm0.67)$	$(0.78\pm0.77)$	$(0.94\pm0.80)$
ACEntropy ACEETDooks		$(0.32\pm0.39)$	$(0.93\pm0.94)$	$(0.01\pm0.07)$	$(0.78\pm0.77)$	$(0.94\pm0.00)$
ACCorr						
	11 10 + 6 56	44.09 24.00	27 26 27 06	27.70+20.10	21 18 24 00	22.05 22.67
ACAUSAICa, ACAICa,	44.49 ± 0.30	(0.51, 0.65)	$27.30\pm 27.00$	$21.19\pm 29.10$	$21.10\pm24.90$	$22.93\pm22.07$
ACAL-AREA ACA	44.65 . 6.94	$(0.31\pm0.03)$	$(0.96\pm0.91)$	$(0.04\pm0.71)$	$(0.03\pm0.09)$	$(0.92\pm0.79)$
ACAbsArea, ACArea, ACVar,	$44.05 \pm 6.84$	45.82±34.22	28.17±25.93	28.23±29.25	$21.22\pm24.32$	23.07±22.18
ACEntropy, ACFFTPeaks,		$(0.4/\pm0.58)$	$(0.96\pm1.02)$	$(0.66\pm0.73)$	(0.86±0.89)	$(0.90\pm0.75)$
ACBandEnergy,	44.72 5.05	44.05.00.03	20.00 21.75	20 60 20 01	20.20.20.75	
ACAbsArea, ACArea, ACVar,	$44.72 \pm 6.82$	44.85±32.86	29.89±26.52	30.60±28.91	20.38±23.76	22.33±22.47
ACFFTPeaks, ACCorr,		$(0.4/\pm0.60)$	(0.96±0.80)	$(0.63\pm0.70)$	$(0.86\pm0.86)$	(0.99±0.84)
ACKange, ACMICR, ACPitch,	44.02 5.15	10.04.07.07	20.07.25.55	07.40.00.15	20.00.27.01	22.02.22.7-
All Features but heart rate	$44.82 \pm 6.16$	42.96±35.25	29.07±25.78	27.40±30.12	20.08±25.04	23.83±22.75
based	1	$(0.44 \pm 0.59)$	$(0.95 \pm 1.03)$	$(0.59\pm0.70)$	$(0.82\pm0.88)$	(0.92+0.73)

Table A6-2: True positive rate per activity category while evaluating different subsets of features using the C4.5 decision tree classifier, a sliding window of 5.6s and subject independent training. False positive rate is shown in parenthesis.

	True Positive Rate					
Activity	AllButHR	ACAbsArea,	ACAbsArea	Total Invariant	Invariant	
		DCArea,	DCArea		Reduced	
		ACVar,				
		ACRange,				
Bench weight lifting - hard	87.4 + 8.5	89.5 + 7.6	91.1 + 5.6	85.4 + 11.4	88.7 + 10.1	
Bench weight lifting - light	91.1 + 8.8	$90.7 \pm 11.8$	$92.0 \pm 10.1$	91.5 + 11.3	93.2 + 10.4	
Bench weight lifting - moderate	87.9 + 17.4	91.1 + 9.9	88.4 + 12.5	89.5 + 12.9	88.2 + 17.1	
Bicep curls - hard	92.4 + 7.1	91.4 + 8.5	91.5 + 7.4	90.4 + 6.8	92.2 + 6.8	
Bicep curls - light	$94.5 \pm 6.5$	$94.6 \pm 3.6$	$94.0 \pm 2.9$	$94.4 \pm 5.3$	$95.1 \pm 4.4$	
Bicep curls - moderate	85.0 + 11.1	91.5 + 4.9	92.1 + 5.5	86.9 + 9.9	90.5 + 6.0	
Calisthenics - Crunches	95.0 + 2.4	$92.0 \pm 5.8$	90.9 + 8.1	94.0 + 6.4	96.5 + 3.2	
Calisthenics - Sit ups	91.4 + 2.9	93.1 + 4.3	92.8 + 4.2	90.0 + 4.9	91.2 + 5.3	
Cycling - Cycle hard - Cycle 80rpm	80.8 + 22.5	76.9 + 21.7	79.6 + 22.4	83.1 + 15.7	79.6 + 22.0	
Cycling - Cycle light - Cycle 100rpm	$94.1 \pm 8.1$	$93.7 \pm 7.6$	$92.7 \pm 11.8$	$91.9 \pm 16.1$	$93.3 \pm 12.0$	
Cycling - Cycle light - Cycle 60rpm	$89.9 \pm 6.8$	90.4 + 6.8	91.9 + 6.8	89.6 + 6.8	91.6 + 7.8	
Cycling - Cycle light - Cycle 80rpm	90.4 + 7.3	91.0 + 9.3	$91.5 \pm 5.4$	91.2 + 9.4	88.3 + 12.4	
Cycling - Cycle moderate - Cycle 80rpm	81.6 + 10.0	$79.6 \pm 10.0$	81.0 + 10.2	81.4 + 8.5	80.6 + 11.9	
Lying down	$98.8 \pm 1.4$	98.8 + 1.5	98.7 + 1.6	$98.7 \pm 1.1$	$99.0 \pm 0.9$	
Rowing - Rowing hard - Rowing 30spm	77.2 + 18.4	79.6 + 16.1	79.5 + 16.0	76.5 + 18.7	79.6 + 13.1	
Rowing - Rowing light - Rowing 30spm	83.3 + 11.2	80.6 + 11.6	$82.5 \pm 11.1$	83.1 + 9.5	83.1 + 9.1	
Rowing - Rowing moderate - Rowing 30spm	74.6 + 16.6	74.1 + 14.2	76.6 + 14.6	$77.4 \pm 13.5$	77.1 + 16.1	
Running - Treadmill 4mph - Treadmill 0	91.4 + 6.0	88.4 + 8.7	89.4 + 8.2	$91.2 \pm 6.3$	89.6 + 7.8	
Running - Treadmill 5mph - Treadmill 0	86.8 + 9.6	87.5 + 7.6	$88.9 \pm 6.4$	87.6+8.3	86.5 + 7.9	
Running - Treadmill 6mph - Treadmill 0	83.1 + 11.2	87.4 + 9.3	$86.6 \pm 10.2$	83.2 + 9.0	$85.6 \pm 7.2$	
Sitting	87.0 + 7.9	$90.9 \pm 7.3$	$91.6 \pm 6.1$	887+7.5	$92.5 \pm 5.8$	
Sitting - Fidget feet legs	87.8 + 7.6	90.0 + 9.5	90.0 + 8.6	$90.6 \pm 10.2$	88.2 + 12.4	
Sitting - Fidget hands arms	84.7 + 11.2	89.2 + 7.4	$89.8 \pm 4.2$	837+132	$85.8 \pm 10.4$	
Stairs - Ascend stairs	842+65	$80.3 \pm 10.1$	$82.9 \pm 7.4$	$869 \pm 76$	845 + 84	
Stairs - Descend stairs	$80.6 \pm 8.2$	$78.2 \pm 10.1$	77.4 + 8.2	$78.8 \pm 9.2$	$764 \pm 138$	
Standing	$85.0 \pm 0.2$	$87.2 \pm 10.9$	$87.3 \pm 10.1$	$825 \pm 102$	$82.0 \pm 11.2$	
Walking - Treadmill 2mph - Treadmill 0	88.1 + 5.2	88.6 + 4.8	89.7 + 4.6	$90.5 \pm 10.2$	89.7 + 4.1	
Walking - Treadmill 3mph - Treadmill 0	78.0 + 7.8	77.0 + 8.8	$73.3 \pm 11.0$	75.4 + 10.4	74.4 + 10.0	
Walking - Treadmill 3mph - Treadmill 3 -	$70.8 \pm 12.4$	$67.4 \pm 13.2$	$65.5 \pm 11.6$	$68.6 \pm 13.8$	$65.4 \pm 13.5$	
light						
Walking - Treadmill 3mph - Treadmill 6 - moderate	69.8 ± 14.1	$70.8 \pm 15.8$	$71.2 \pm 11.0$	69.3 ± 15.3	$70.1 \pm 11.4$	
Walking - Treadmill 3mph - Treadmill 9 - hard	80.2 ± 11.0	81.2 ± 10.4	$78.9 \pm 10.8$	77.6 ± 11.4	$80.4\pm9.9$	
kneeling	93.0 ± 7.6	$94.7 \pm 4.7$	$94.4 \pm 4.8$	92.1 ± 6.9	$95.2 \pm 5.8$	
unknown	$76.8 \pm 8.2$	$77.9 \pm 8.2$	$79.1 \pm 8.5$	$77.0 \pm 8.5$	$76.1 \pm 9.2$	
Carrying groceries	$83.8 \pm 9.7$	$84.0 \pm 9.4$	$84.9 \pm 9.3$	$83.4 \pm 9.8$	82.8 ± 11.9	
Doing dishes	77.5 ± 12.2	$81.8 \pm 10.0$	$82.3 \pm 9.3$	$76.2 \pm 15.2$	$77.2 \pm 15.9$	
Gardening	$74.8 \pm 14.5$	$75.6 \pm 19.5$	$78.6 \pm 17.2$	$73.4 \pm 20.6$	$73.1 \pm 21.5$	
Ironing	$77.4 \pm 10.2$	$79.6 \pm 9.8$	$82.2 \pm 11.3$	$77.4 \pm 11.8$	$80.8 \pm 8.5$	
Making the bed	53.3 ± 11.6	57.6 ± 11.7	$59.6 \pm 11.5$	$49.8 \pm 13.3$	$54.3 \pm 12.3$	
Mopping	$62.3 \pm 12.5$	$65.9 \pm 14.1$	$66.7 \pm 14.6$	$63.0 \pm 13.1$	$59.8 \pm 14.5$	
Playing videogames	$93.2 \pm 5.7$	$94.3 \pm 4.0$	93.1 ± 5.0	$93.4 \pm 6.2$	$93.2 \pm 5.1$	
Scrubbing a surface	$77.0 \pm 17.5$	$77.8 \pm 13.5$	$79.5 \pm 15.4$	$75.2 \pm 14.1$	$78.2 \pm 14.5$	
Stacking groceries	57.8 ± 13.6	$61.3 \pm 16.7$	$63.0 \pm 12.8$	$59.0 \pm 10.3$	$64.8 \pm 12.9$	
Sweeping	$54.7 \pm 13.2$	$56.2 \pm 11.6$	$60.5 \pm 11.3$	$53.8 \pm 16.7$	$60.4 \pm 13.6$	
Typing	$94.8 \pm 3.5$	$94.1 \pm 5.7$	$95.0 \pm 4.3$	$95.2 \pm 3.9$	$95.0 \pm 4.0$	
Vacuuming	$65.2 \pm 13.8$	$67.6 \pm 7.4$	$62.2 \pm 10.7$	63.1 ± 11.2	$63.9 \pm 13.9$	
Walking around block	$82.2 \pm 7.2$	$82.0 \pm 8.2$	$82.9 \pm 8.2$	$85.6 \pm 7.3$	$83.5 \pm 7.4$	
Washing windows	$60.4 \pm 10.4$	63.3 ± 13.1	$64.9 \pm 9.7$	59.6 ± 11.5	$59.6 \pm 8.6$	
Watching TV	$93.0 \pm 6.8$	$92.8 \pm 6.5$	$94.6 \pm 4.6$	93.1 ± 6.3	$93.2 \pm 5.6$	
Weeding	76.6 + 13.2	75.8 + 9.6	75.3 + 7.5	$71.5 \pm 14.8$	$71.8 \pm 11.5$	
Wiping/Dusting	54.5 + 14.9	$57.1 \pm 15.0$	56.5 + 15.8	56.2 + 14.4	59.4 ± 15.5	
Writing	93.9 + 4.1	96.5 + 2.4	96.4 + 3.3	94.0 + 5.8	95.7 + 3.4	
taking out trash	$44.8 \pm 11.3$	48.2 ± 13.8	52.3 ± 10.9	$44.1 \pm 12.0$	47.9 ± 11.4	

Table A6-3: True positive rate when evaluating the four highest performing feature subsets computed per sensor using the C4.5 classifier in a subject dependent manner.

	False Positive Rate					
Activity	AllButHR	AbsCumSum, CumSum, Var,	ACAbsArea DCArea	Total Invariant	Invariant Reduced	
Bench weight lifting - hard	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.1 \pm 0.2$	
Bench weight lifting - light	$0.1 \pm 0.1$	$0.1 \pm 0.1$ 0.1 + 0.1	$0.1 \pm 0.1$	$0.1 \pm 0.1$ 0.1 + 0.1	$0.1 \pm 0.2$ 0.1 ± 0.1	
Bench weight lifting - moderate	$0.1 \pm 0.1$ 0 2 + 0 2	$0.1 \pm 0.1$ 0.1 + 0.1	$0.1 \pm 0.0$ 0.2 + 0.1	$0.1 \pm 0.1$ 0.2 + 0.2	$0.1 \pm 0.1$ 0.2 + 0.2	
Bicep curls - hard	$0.2 \pm 0.2$	$0.1 \pm 0.1$ 0.2 + 0.1	$0.2 \pm 0.1$ 0.2 + 0.1	$0.2 \pm 0.2$ 0.2 + 0.1	$0.2 \pm 0.2$	
Bicep curls - light	$0.3 \pm 0.2$ 0.2 + 0.1	$0.2 \pm 0.1$ 0.1 + 0.1	$0.2 \pm 0.1$ 0.1 + 0.1	$0.2 \pm 0.1$ 0.2 + 0.1	$0.1 \pm 0.1$	
Bicep curls - moderate	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.1 \pm 0.1$ $0.2 \pm 0.1$	$0.2 \pm 0.1$	$0.2 \pm 0.2$	
Calisthenics - Crunches	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.0 \pm 0.1$	$0.1 \pm 0.0$	$0.1 \pm 0.1$	
Calisthenics - Sit ups	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.2 \pm 0.1$	$0.1 \pm 0.0$	
Cycling - Cycle hard - Cycle 80rpm	$0.2 \pm 0.1$	$0.3 \pm 0.2$	$0.2 \pm 0.1$	$0.2 \pm 0.1$	$0.2 \pm 0.1$	
Cycling - Cycle light - Cycle 100rpm	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	
Cycling - Cycle light - Cycle 60rpm	$0.2 \pm 0.2$	$0.2 \pm 0.1$	$0.2 \pm 0.1$	$0.2 \pm 0.1$	$0.2 \pm 0.2$	
Cycling - Cycle light - Cycle 80rpm	$0.2 \pm 0.1$	$0.3 \pm 0.2$	$0.2 \pm 0.2$	$0.3 \pm 0.2$	$0.3 \pm 0.2$	
Cycling - Cycle moderate - Cycle 80rpm	$0.2 \pm 0.1$	$0.3 \pm 0.2$	$0.3 \pm 0.1$	$0.3 \pm 0.2$	$0.3 \pm 0.2$	
Lying down	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	
Rowing - Rowing hard - Rowing 30spm	$0.3 \pm 0.2$	$0.4 \pm 0.2$	$0.3 \pm 0.2$	$0.4 \pm 0.2$	$0.4 \pm 0.2$	
Rowing - Rowing light - Rowing 30spm	$0.4 \pm 0.2$	$0.3 \pm 0.2$	$0.3 \pm 0.2$	$0.3 \pm 0.1$	$0.3 \pm 0.1$	
Rowing - Rowing moderate - Rowing 30spm	$0.4 \pm 0.3$	$0.5 \pm 0.3$	$0.5 \pm 0.3$	$0.4 \pm 0.2$	$0.4 \pm 0.2$	
Running - Treadmill 4mph - Treadmill 0	$0.2 \pm 0.1$	$0.2 \pm 0.1$	$0.2 \pm 0.1$	$0.2\pm0.2$	$0.2 \pm 0.1$	
Running - Treadmill 5mph - Treadmill 0	$0.2 \pm 0.1$	$0.2 \pm 0.1$	$0.2 \pm 0.1$	$0.2 \pm 0.1$	$0.2 \pm 0.1$	
Running - Treadmill 6mph - Treadmill 0	$0.2 \pm 0.1$	$0.2 \pm 0.1$	$0.2 \pm 0.1$	$0.2 \pm 0.1$	$0.2 \pm 0.1$	
Sitting	$0.2\pm0.2$	$0.2 \pm 0.2$	$0.1 \pm 0.1$	$0.2 \pm 0.1$	$0.2 \pm 0.1$	
Sitting - Fidget feet legs	$0.2 \pm 0.2$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.2 \pm 0.2$	
Sitting - Fidget hands arms	$0.3 \pm 0.3$	$0.2 \pm 0.1$	$0.2 \pm 0.1$	$0.2 \pm 0.1$	$0.2 \pm 0.1$	
Stairs - Ascend stairs	$0.3 \pm 0.2$	$0.4 \pm 0.2$	$0.3 \pm 0.2$	$0.3 \pm 0.1$	$0.3 \pm 0.2$	
Stairs - Descend stairs	$0.3 \pm 0.2$	$0.3 \pm 0.2$	$0.3 \pm 0.2$	$0.3 \pm 0.2$	$0.5 \pm 0.2$	
Standing	$0.2 \pm 0.1$	$0.2 \pm 0.2$	$0.2 \pm 0.1$	$0.2 \pm 0.1$	$0.2 \pm 0.1$	
Walking - Treadmill 2mph - Treadmill 0	$0.3 \pm 0.2$	$0.3 \pm 0.2$	$0.2 \pm 0.2$	$0.3 \pm 0.2$	$0.2 \pm 0.1$	
Walking - Treadmill 3mph - Treadmill 0	$0.5 \pm 0.2$	$0.6 \pm 0.2$	$0.5 \pm 0.2$	$0.5 \pm 0.2$	$0.6 \pm 0.3$	
light	$0.7 \pm 0.2$	0.8 ± 0.3	$0.9 \pm 0.4$	0.7 ± 0.3	0.8 ± 0.3	
Walking - Treadmill 3mph - Treadmill 6 - moderate	$0.6 \pm 0.2$	0.7 ± 0.3	0.6 ± 0.4	0.7 ± 0.3	$0.7 \pm 0.3$	
Walking - Treadmill 3mph - Treadmill 9 - hard	$0.4 \pm 0.3$	$0.4 \pm 0.2$	$0.4 \pm 0.2$	$0.4 \pm 0.2$	$0.4 \pm 0.2$	
kneeling	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.2 \pm 0.2$	$0.2 \pm 0.1$	
unknown	$9.0 \pm 2.0$	$8.0 \pm 1.8$	$8.0 \pm 1.9$	$8.6 \pm 2.2$	$8.4 \pm 2.1$	
Carrying groceries	$0.3 \pm 0.2$	$0.3 \pm 0.2$	$0.3 \pm 0.2$	$0.3 \pm 0.2$	$0.3 \pm 0.2$	
Doing dishes	$0.5 \pm 0.3$	$0.5 \pm 0.3$	$0.4 \pm 0.3$	$0.5 \pm 0.4$	$0.5 \pm 0.3$	
Gardening	$0.5 \pm 0.4$	$0.4 \pm 0.4$	$0.4 \pm 0.3$	$0.6 \pm 0.3$	$0.4 \pm 0.3$	
Ironing Making the heat	$0.5 \pm 0.2$	$0.5 \pm 0.2$	$0.4 \pm 0.3$	$0.5 \pm 0.3$	$0.5 \pm 0.3$	
Making the bed	$1.1 \pm 0.5$	$1.0 \pm 0.4$	$0.9 \pm 0.4$	$1.0 \pm 0.4$	$1.0 \pm 0.4$	
Mopping Diaving video comes	$0.8 \pm 0.4$	$0.7 \pm 0.4$	$0.7 \pm 0.4$	$0.9 \pm 0.3$	$0.8 \pm 0.5$	
Playing videogames	$0.2 \pm 0.2$	$0.2 \pm 0.1$	$0.2 \pm 0.1$	$0.2 \pm 0.2$	$0.2 \pm 0.2$	
Studding groceries	$0.4 \pm 0.2$	$0.4 \pm 0.3$	$0.4 \pm 0.3$	$0.4 \pm 0.3$	$0.5 \pm 0.3$	
Stacking groceries	$0.7 \pm 0.5$	$0.8 \pm 0.7$	$0.5 \pm 0.5$	$0.0 \pm 0.4$	$0.7 \pm 0.0$	
Typing	$0.8 \pm 0.4$	$0.8 \pm 0.4$	$0.8 \pm 0.4$	$0.8 \pm 0.4$	$0.7 \pm 0.3$	
Vacuuming	$0.1 \pm 0.1$	$0.1 \pm 0.1$ 0.6 ± 0.4	$0.1 \pm 0.1$ 0.7 ± 0.4	$0.1 \pm 0.1$ 0.6 ± 0.3	$0.2 \pm 0.2$	
Walking around block	$0.0 \pm 0.3$	$0.0 \pm 0.4$	$0.7 \pm 0.4$	$0.0 \pm 0.3$	$0.7 \pm 0.4$	
Washing windows	$0.4 \pm 0.2$ 0.9 + 0.4	$0.4 \pm 0.2$ 0.7 + 0.4	$0.4 \pm 0.2$ 0.7 + 0.3	$0.4 \pm 0.3$ 0.9 + 0.4	$0.4 \pm 0.2$ 0.7 + 0.4	
Watching TV	$0.7 \pm 0.4$ 0.3 ± 0.2	$0.7 \pm 0.4$ 0.2 + 0.1	$0.7 \pm 0.3$ 0.2 + 0.2	$0.7 \pm 0.4$ 0.2 + 0.1	$0.7 \pm 0.4$ 0.2 + 0.1	
Weeding	$0.5 \pm 0.2$ 0.6 + 0.5	$0.2 \pm 0.1$ 0.5 + 0.3	$0.2 \pm 0.2$ 0.4 + 0.2	$0.2 \pm 0.1$ 0.5 + 0.3	$0.2 \pm 0.1$ 0.6 + 0.4	
Wining/Dusting	$0.0 \pm 0.3$ 0.8 + 0.4	$0.5 \pm 0.3$ 0.8 + 0.2	$0.7 \pm 0.2$ 0.8 ± 0.4	$0.5 \pm 0.3$ 0.8 + 0.4	$0.0 \pm 0.4$ 0.8 + 0.4	
Writing	$0.0 \pm 0.4$ $0.1 \pm 0.1$	$0.0 \pm 0.2$ 0.2 + 0.2	$0.2 \pm 0.4$	$0.0 \pm 0.4$ $0.2 \pm 0.2$	$0.2 \pm 0.4$	
taking out trash	$0.9 \pm 0.2$	$0.8 \pm 0.2$	$0.8 \pm 0.2$	$1.0 \pm 0.4$	$0.9 \pm 0.3$	
9			0.0 = 0.0		0.7 = 0.0	

Table A6-4: False positive rate when evaluating the four of the highest performing feature subsets computed per sensor using the C4.5 classifier in a subject dependent manner.

			F-Measure		
Activity	AllButHR	ACAbsArea,	ACAbsArea	Total Invariant	Invariant
		DCArea,	DCArea		Reduced
		ACVar,			
		ACRange,			
Bench weight lifting - hard	90.5 + 7.1	91.0 + 7.9	$90.8 \pm 5.6$	87.2 + 10.8	F-Measure
Bench weight lifting - light	$90.8 \pm 9.1$	$90.6 \pm 10.3$	91.7 + 7.4	$90.8 \pm 11.2$	88.3 + 9.4
Bench weight lifting - moderate	88.0 + 13.1	$90.0 \pm 10.3$	869+130	893+118	919 + 98
Bicen curls - hard	$89.2 \pm 8.9$	$90.1 \pm 9.2$ $90.8 \pm 7.5$	$90.9 \pm 15.0$	$90.1 \pm 6.0$	$85.8 \pm 15.4$
Bicep curls - light	$93.2 \pm 0.9$	$94.4 \pm 2.9$	$93.8 \pm 3.1$	$93.6 \pm 4.5$	$90.8 \pm 6.3$
Bicep curls - moderate	$\frac{93.3 \pm 9.0}{88.2 \pm 8.5}$	$91.3 \pm 4.1$	$91.2 \pm 3.1$	93.0 ± 4.3	$95.0 \pm 0.3$
Calisthenics - Crunches	$93.1 \pm 3.4$	$97.3 \pm 4.1$ $97.7 \pm 4.9$	$93.3 \pm 6.3$	$94.0 \pm 7.5$	$93.1 \pm 4.1$ 887 ± 60
Calisthenics - Sit ups	$919 \pm 36$	$94.0 \pm 2.9$	$94.1 \pm 2.5$	$90.0 \pm 5.6$	$94.2 \pm 4.4$
Cycling - Cycle hard - Cycle 80rpm	$79.9 \pm 3.0$	$76.1 \pm 2.7$	$78.7 \pm 2.5$	82.1 + 15.7	$94.2 \pm 4.4$ $91.9 \pm 3.9$
Cycling - Cycle light - Cycle 100rpm	$93.6 \pm 6.0$	$93.6 \pm 6.1$	$92.0 \pm 11.2$	$91.8 \pm 11.7$	$79.0 \pm 20.9$
Cycling - Cycle light - Cycle 60rpm	$93.0 \pm 0.0$	$93.0 \pm 0.1$	$90.5 \pm 5.2$	$91.0 \pm 11.7$	$0.0 \pm 20.0$
Cycling - Cycle light - Cycle 80rpm	$89.7 \pm 5.9$ $89.1 \pm 6.2$	$89.0 \pm 3.2$	$90.3 \pm 5.2$	$88.1 \pm 3.9$	$92.7 \pm 9.1$
Cycling Cycle moderate Cycle 80rpm	$89.1 \pm 0.2$	$33.5 \pm 10.0$	$89.4 \pm 0.3$	$00.4 \pm 0.3$	$89.7 \pm 0.9$
L ving down	$02.0 \pm 0.4$	$79.4 \pm 9.9$	$08.3 \pm 9.0$	$00.0 \pm 0.0$	$80.3 \pm 10.0$ $80.0 \pm 10.2$
Powing Powing hard Powing 30cpm	$90.3 \pm 1.1$	$90.1 \pm 1.0$ 78.2 ± 15.7	$90.2 \pm 1.2$ 70.7 ± 14.0	$96.1 \pm 1.2$ 76.2 ± 10.2	$0.0 \pm 10.2$
Rowing - Rowing light Rowing 30spm	$77.0 \pm 10.9$	$70.2 \pm 13.7$	$79.7 \pm 14.0$	$70.2 \pm 19.3$	$90.3 \pm 1.1$ 79.2 ± 12.9
Rowing - Rowing nght - Rowing Sospin	$82.0 \pm 10.3$	$30.9 \pm 10.3$	02.7 ± 9.0	$03.0 \pm 0.3$	$70.3 \pm 13.0$
Rowing - Rowing moderate - Rowing Sospin	$74.0 \pm 10.0$	$73.1 \pm 14.0$	$73.4 \pm 15.0$	$70.2 \pm 12.8$	$63.0 \pm 6.0$
Running - Treadmill Smph - Treadmill 0	$90.0 \pm 4.9$	$88.4 \pm 7.0$	$88.8 \pm 0.0$	$89.4 \pm 5.0$	$70.0 \pm 14.8$
Running - Treadmill Smph - Treadmill 0	$80.8 \pm 7.3$	$88.0 \pm 5.8$	$88.0 \pm 4.7$	$87.0 \pm 0.8$	$88.7 \pm 0.8$
Running - Treadmill ompn - Treadmill 0	$83.6 \pm 10.4$	$85.9 \pm 9.0$	$84.3 \pm 11.8$	$84.3 \pm 9.8$	$86.3 \pm 6.3$
Sitting Eident fast lass	$86.8 \pm 7.4$	$90.2 \pm 7.8$	$91.3 \pm 5.2$	$86.8 \pm 7.8$	$86.1 \pm 7.4$
Sitting - Fidget feet legs	$86.8 \pm 8.1$	$90.3 \pm 8.3$	89.6 ± 7.5	$90.9 \pm 8.9$	$88.9 \pm 5.4$
Sitting - Flaget hands arms	83.0 ± 12.4	$88.0 \pm 6.4$	$89.4 \pm 6.1$	$84.0 \pm 10.4$	88.1 ± 10.4
Stairs - Ascend stairs	$82.5 \pm 6.8$	79.1 ± 9.2	$81.5 \pm 7.5$	84.4 ± 7.3	$86.3 \pm 9.5$
Stars - Descend stars	$80.1 \pm 8.0$	$78.2 \pm 8.3$	/8.8 ± /.3	/8.8 ± 8.5	$82.8 \pm 7.8$
Standing Well-ing Tree desill Sweet Tree desill 0	$85.5 \pm 6.1$	$86.4 \pm 9.8$	86.6 ± 9.7	83.1 ± 9.5	$74.4 \pm 10.8$
Walking - Treadmill 2mph - Treadmill 0	86.7±4.9	88.1 ± 4.4	$89.2 \pm 4.5$	87.7 ± 5.0	83.4 ± 9.4
Walking - Treadmill 3mph - Treadmill 0	$7/.4 \pm 8.1$	$74.9 \pm 7.7$	$73.5 \pm 10.8$	/5.4 ± 9.3	$89.1 \pm 3.1$
light	69.2 ± 11.1	$66.2 \pm 11.0$	$63.5 \pm 10.9$	67.0 ± 12.0	/2.7 ± 9.4
Walking - Treadmill 3mph - Treadmill 6 - moderate	$69.4 \pm 10.7$	69.4 ± 13.3	$71.0\pm10.5$	$68.6 \pm 14.9$	64.1 ± 11.2
Walking - Treadmill 3mph - Treadmill 9 - hard	$79.8 \pm 10.3$	80.7 ± 8.1	$79.3\pm9.4$	$77.8 \pm 10.8$	68.5 ± 11.9
kneeling	918+61	$93.0 \pm 5.1$	931 + 49	898+75	797+86
unknown	$77.0 \pm 8.3$	$78.6 \pm 8.0$	79.4 + 8.2	77.4 + 8.8	$91.7 \pm 5.6$
Carrying groceries	83.4 + 9.4	$83.8 \pm 9.0$	849+91	83.2 + 9.5	77.1 + 9.3
Doing dishes	$77.9 \pm 11.2$	81.1 + 9.2	$82.0 \pm 7.1$	$76.2 \pm 14.4$	83.3 + 11.4
Gardening	74.5 + 14.4	76.2 + 18.6	78.1 + 14.8	71.9 + 20.3	$76.8 \pm 14.1$
Ironing	77.6 + 9.4	$79.2 \pm 9.1$	$82.9 \pm 10.4$	$77.4 \pm 10.0$	73.9 + 21.0
Making the bed	$53.4 \pm 11.6$	$56.9 \pm 10.8$	$59.4 \pm 10.3$	$51.1 \pm 12.5$	80.1 + 8.0
Mopping	$61.5 \pm 11.9$	$65.1 \pm 10.0$	$65.9 \pm 14.5$	$61.0 \pm 12.8$	$55.0 \pm 11.4$
Plaving videogames	$921 \pm 50$	$931 \pm 35$	$93.2 \pm 4.1$	$921 \pm 63$	$60.3 \pm 11.4$
Scrubbing a surface	$769 \pm 156$	$78.4 \pm 12.8$	$795 \pm 138$	$76.0 \pm 13.9$	$923 \pm 43$
Stacking groceries	$595 \pm 13.0$	$60.7 \pm 16.2$	$65.1 \pm 10.9$	$61.6 \pm 10.8$	$77.4 \pm 13.2$
Sweeping	$57.3 \pm 13.3$	$57.9 \pm 10.2$	$61.3 \pm 10.9$	$56.2 \pm 15.4$	$64.8 \pm 14.0$
Typing	9/9 + 23	$94.0 \pm 11.1$	$95.0 \pm 3.2$	$95.0 \pm 2.9$	$61.7 \pm 12.6$
Vacuuming	$57.5 \pm 2.5$	$57.0 \pm 7.4$ 68.6 + 8.2	$63.8 \pm 11.2$	$5.0 \pm 2.9$ 65.4 + 10.4	94.4 + 3.0
Walking around block	$83.2 \pm 5.6$	$81.9 \pm 6.6$	$82.0 \pm 11.2$	84.8 + 6.4	$57.7 \pm 3.0$
Washing windows	$60.5 \pm 10.3$	$64.5 \pm 10.0$	$65.1 \pm 10.8$	$60.2 \pm 0.4$	$87.8 \pm 6.0$
Watching TV	$916 \pm 50$	$97.4 \pm 5.1$	$93.1 \pm 10.0$	$97.2 \pm 11.2$	$62.0 \pm 0.0$ $62.0 \pm 0.3$
Weeding	$75.0 \pm 3.9$	$72.4 \pm 3.1$ 766 ± 83	$77.6 \pm 5.4$	$72.2 \pm 3.2$ $73.4 \pm 13.0$	$02.0 \pm 9.3$ $03.0 \pm 4.0$
Wining/Dusting	$15.1 \pm 11.0$ 568 ± 14.0	$70.0 \pm 0.3$	$77.0 \pm 3.4$ 57.0 $\pm 14.7$	7.5.4 ± 13.0 57.5 ± 12.5	$73.0 \pm 4.0$ $72.8 \pm 0.5$
Writing	$30.0 \pm 14.9$ $01.2 \pm 2.1$	$37.0 \pm 13.4$ 05.3 $\pm$ 2.9	$37.7 \pm 14.7$ 05.2 ± 2.6	$37.3 \pm 13.3$ 03.2 ± 5.2	$12.0 \pm 9.3$
taking out trash	$77.2 \pm 3.1$	$50.3 \pm 2.0$	$53.2 \pm 3.0$ 53.0 ± 0.0	$75.2 \pm 5.5$	$00.4 \pm 14.3$ $04.5 \pm 3.0$
out trush	TI.J ± 10.9	50.5±12.4	JJ.J ± 7.7	<b>¬</b> J.J ± 12.J	) J.U

Table A6-5: F-Measure rate when evaluating the four of the highest performing feature subsets computed per sensor using the C4.5 classifier in a subject dependent manner.

	True Positive Rate					
Activity	AllButHR	ACAbsArea,	ACAbsArea	Total Invariant	Invariant	
		DCArea,	DCArea		Reduced	
		ACVar,				
		AC Range,				
Bench weight lifting - hard	16.7 + 29.2	9.5 + 22.0	10.9 + 24.0	7.2 + 13.5	134 + 295	
Bench weight lifting - light	$15.6 \pm 28.2$	35.6 + 39.4	$29.7 \pm 40.6$	18.0 + 29.6	25.2 + 33.7	
Bench weight lifting - moderate	$11.6 \pm 18.5$	$37 \pm 63$	14+46	$39 \pm 131$	13 + 29	
Bicep curls - hard	19.9 + 37.1	$24.8 \pm 33.0$	244 + 322	$21.5 \pm 34.1$	$1.3 \pm 2.9$ 128 + 287	
Bicen curls - light	$19.9 \pm 37.1$ $18.8 \pm 27.3$	$17.3 \pm 28.2$	148 + 248	$23.3 \pm 31.8$	$26.6 \pm 35.7$	
Bicen curls - moderate	$10.0 \pm 27.3$ $12.2 \pm 26.7$	$17.9 \pm 20.2$ $12.9 \pm 27.4$	$23.6 \pm 36.0$	$23.3 \pm 31.0$ 21 4 + 32 4	$17.8 \pm 26.8$	
Calisthenics - Crunches	$12.2 \pm 20.7$ 16.0 + 31.9	$12.9 \pm 27.4$ 15.0 + 31.7	$164 \pm 30.6$	$15.8 \pm 29.6$	$24.7 \pm 20.0$	
Calisthenics - Sit ups	$10.0 \pm 51.9$	$34.2 \pm 30.7$	$26.5 \pm 30.5$	$13.0 \pm 27.0$ $17.4 \pm 45.3$	$42.8 \pm 44.7$	
Cycling - Cycle hard - Cycle 80rpm	$18.0 \pm 20.0$	$10.7 \pm 22.1$	$6.1 \pm 11.3$	$+7.4 \pm +3.3$ 10.2 ± 22.2	$42.0 \pm 44.7$	
Cycling - Cycle light - Cycle 100rpm	$18.0 \pm 29.9$ 70.6 + 37.7	$10.7 \pm 22.1$ 63.6 + 37.9	74.2 + 31.4	$10.2 \pm 22.2$ 70.3 ± 36.6	$69.7 \pm 40.8$	
Cycling - Cycle light - Cycle 60rpm	$39.6 \pm 36.2$	$33.1 \pm 37.3$	$74.2 \pm 31.4$ $34.1 \pm 36.5$	$16.3 \pm 30.0$	$50.6 \pm 38.9$	
Cycling Cycle light Cycle 80rpm	$39.0 \pm 30.2$	$33.1 \pm 32.3$	$12.0 \pm 22.2$	$40.3 \pm 37.9$	$30.0 \pm 38.9$	
Cycling - Cycle light - Cycle solphi	$43.0 \pm 39.9$	$20.1 \pm 29.4$	$13.9 \pm 22.2$ 17.6 ± 20.4	$44.0 \pm 30.2$	$42.3 \pm 39.0$	
Lying - Cycle moderate - Cycle sorpin	$16.1 \pm 19.4$	$10.2 \pm 12.2$	$17.0 \pm 29.4$	$12.7 \pm 19.1$	$24.2 \pm 35.2$	
Lying down	$76.9 \pm 34.3$	$75.0 \pm 30.1$	$79.1 \pm 29.1$	$73.2 \pm 30.1$	$70.0 \pm 37.8$	
Rowing - Rowing light Rowing 30spin	$19.5 \pm 25.5$	$17.0 \pm 20.0$	$24.0 \pm 33.1$	$10.0 \pm 22.0$	$24.2 \pm 29.9$	
Rowing - Rowing light - Rowing Josphi	$24.3 \pm 23.7$	$15.7 \pm 17.5$	$16.9 \pm 19.3$	$30.0 \pm 29.5$	$23.2 \pm 29.1$	
Rowing - Rowing moderate - Rowing Sospin	$18.7 \pm 21.5$	$17.1 \pm 30.0$	$10.4 \pm 29.8$	$22.7 \pm 28.0$	$15.9 \pm 18.0$	
Running - Treadmill Smph. Treadmill 0	$28.2 \pm 30.7$	$26.6 \pm 35.6$	$28.7 \pm 36.6$	$29.0 \pm 31.0$	$28.5 \pm 39.9$	
Running - Treadmill Smph - Treadmill 0	$49.0 \pm 34.9$	$4/.3 \pm 31.5$	$35.8 \pm 34.7$	$43.4 \pm 36.8$	$44.6 \pm 29.3$	
Running - Treadmitt ompn - Treadmitt 0	$38.3 \pm 31.0$	$34.5 \pm 38.2$	$51.2 \pm 34.6$	$42.8 \pm 36.4$	$48.7 \pm 34.8$	
Sitting	$15.3 \pm 26.4$	$27.8 \pm 38.9$	$22.7 \pm 34.0$	$24.6 \pm 35.9$	$15.1 \pm 33.7$	
Sitting - Fidget feet legs	$28.8 \pm 36.8$	$24.1 \pm 27.3$	$19.8 \pm 26.2$	$22.1 \pm 24.0$	$29.1 \pm 31.9$	
Sitting - Fidget hands arms	$28.5 \pm 32.2$	$16.3 \pm 25.3$	$14.1 \pm 25.1$	$23.7 \pm 28.7$	$24.1 \pm 28.2$	
Stairs - Ascend stairs	$49.4 \pm 33.6$	$37.7 \pm 29.6$	$23.4 \pm 31.4$	$51.0 \pm 34.7$	$51.3 \pm 36.4$	
Stairs - Descend stairs	$39.7 \pm 30.7$	$24.4 \pm 24.0$	$16.1 \pm 19.3$	$42.4 \pm 29.8$	$28.2 \pm 25.2$	
Standing	$41.3 \pm 38.6$	$37.0 \pm 33.1$	$36.5 \pm 34.4$	$41.3 \pm 34.8$	$40.3 \pm 36.4$	
Walking - Treadmill 2mph - Treadmill 0	$45.6 \pm 34.2$	$42.0 \pm 34.5$	$41.5 \pm 37.4$	$42.9 \pm 34.4$	$48.3 \pm 31.5$	
Walking - Treadmill 3mph - Treadmill 0	$24.3 \pm 25.4$	9.0 ± 13.8	$8.6 \pm 17.8$	$14.2 \pm 19.2$	$13.1 \pm 18.0$	
Walking - Treadmill 3mph - Treadmill 3 - light	$10.1 \pm 12.6$	$12.8 \pm 14.5$	$5.6 \pm 9.2$	$20.2 \pm 23.7$	$14.2 \pm 16.4$	
Walking - Treadmill 3mph - Treadmill 6 - moderate	$10.5 \pm 14.9$	$12.2 \pm 18.0$	$18.2\pm25.9$	$11.7\pm13.0$	$11.3\pm18.6$	
Walking - Treadmill 3mph - Treadmill 9 - hard	15.1 ± 23.2	9.1 ± 11.1	9.8 ± 13.7	8.0 ± 15.6	$12.3\pm18.9$	
kneeling	66.9 + 43.1	68.6 + 38.8	66.5 + 38.6	57.6 + 42.0	53.8 + 45.8	
unknown	64.0 + 5.2	65.1 + 6.0	65.1 + 6.4	63.6 + 4.9	62.4 + 4.8	
Carrying groceries	$19.8 \pm 20.3$	$16.6 \pm 17.8$	18.3 + 22.8	184 + 17.6	21.6 + 25.8	
Doing dishes	29.3 + 28.4	31.6 + 31.2	27.7 + 25.3	32.0 + 31.4	35.7 + 29.6	
Gardening	14.4 + 20.8	14.7 + 20.9	7.6 + 13.0	$17.3 \pm 21.7$	12.1 + 19.3	
Ironing	37.1 + 31.5	34.0 + 31.6	27.7 + 28.9	39.6 + 29.8	33.1 + 33.6	
Making the bed	$26.0 \pm 17.8$	$25.8 \pm 18.8$	$24.6 \pm 20.0$	$27.0 \pm 19.2$	$28.5 \pm 20.2$	
Mopping	24.2 + 21.5	21.4 + 21.0	22.7 + 21.8	22.6 + 22.7	24.2 + 24.9	
Playing videogames	$29.2 \pm 34.7$	$16.9 \pm 26.8$	$17.3 \pm 30.5$	$24.1 \pm 30.8$	$18.8 \pm 28.9$	
Scrubbing a surface	$13.9 \pm 17.8$	$11.8 \pm 14.9$	$11.6 \pm 14.5$	$13.6 \pm 21.8$	$10.3 \pm 13.4$	
Stacking groceries	$11.5 \pm 12.0$	$13.9 \pm 14.9$	$9.0 \pm 12.0$	$14.5 \pm 14.0$	$14.4 \pm 14.9$	
Sweeping	$16.5 \pm 17.9$	$12.8 \pm 17.9$	$12.6 \pm 17.3$	$14.5 \pm 14.7$	$13.1 \pm 15.8$	
Typing	$49.2 \pm 37.5$	$37.4 \pm 33.9$	$36.1 \pm 35.2$	$45.0 \pm 36.2$	$50.0 \pm 37.3$	
Vacuuming	$23.0 \pm 21.7$	$25.1 \pm 23.4$	$15.0 \pm 16.6$	$26.5 \pm 26.8$	$22.0 \pm 20.8$	
Walking around block	$18.9 \pm 18.0$	$21.0 \pm 23.6$	$12.6 \pm 15.2$	$17.6 \pm 16.5$	$17.1 \pm 17.2$	
Washing windows	$22.4 \pm 19.2$	20.3 + 20.6	$20.0 \pm 18.3$	22.9 + 20.9	$26.4 \pm 24.3$	
Watching TV	$20.1 \pm 30.1$	17.4 + 25.2	15.8 + 29.1	14.9 + 26.3	$18.7 \pm 29.8$	
Weeding	4.5 + 7.8	9.8 + 23.5	3.8 + 7.3	5.4 + 5.8	6.8 + 13.6	
Wiping/Dusting	21.6 + 20.1	21.7 + 21.8	17.9 + 18.3	22.5 + 20.2	22.7 + 21.6	
Writing	51.1 + 40.0	42.3 + 37.2	39.4 + 37.4	43.2 + 37.0	49.8 + 38.6	
taking out trash	$10.1 \pm 10.4$	9.1 ± 10.2	8.6 ± 8.3	$10.5 \pm 12.6$	8.6±8.8	

Table A6-6: True positive rate when evaluating the four of the highest performing feature subsets computed per sensor using the C4.5 classifier in a subject independent manner.

	False Positive Rate					
Activity	AllButHR	ACAbsArea,	ACAbsArea	Total Invariant	Invariant	
		DCArea,	DCArea		Reduced	
		ACVar, ACRange				
		ACMCR				
Bench weight lifting - hard	$0.6 \pm 0.6$	$0.4 \pm 0.4$	$0.2 \pm 0.2$	$0.4 \pm 0.4$	$0.5 \pm 0.6$	
Bench weight lifting - light	$0.6 \pm 0.7$	$0.8 \pm 0.9$	$0.7 \pm 0.9$	$0.6 \pm 0.7$	$0.6 \pm 0.8$	
Bench weight lifting - moderate	$0.6 \pm 0.6$	$0.3 \pm 0.4$	$0.2 \pm 0.3$	$0.4 \pm 0.4$	$0.4 \pm 0.4$	
Bicep curls - hard	$0.8 \pm 1.1$	$0.7 \pm 0.7$	$1.0 \pm 0.8$	$0.7 \pm 0.7$	$0.4 \pm 0.4$	
Bicep curls - light	$0.9 \pm 1.2$	$1.2 \pm 1.1$	$0.8 \pm 0.9$	$0.8 \pm 1.1$	$1.0 \pm 1.1$	
Bicep curls - moderate	$0.5 \pm 0.6$	$1.1 \pm 0.9$	$1.2 \pm 1.1$	$1.5 \pm 0.9$	$1.0 \pm 0.9$	
Calisthenics - Crunches	$0.2 \pm 0.4$	$0.3 \pm 0.6$	$0.2 \pm 0.4$	$0.4 \pm 0.6$	$0.5 \pm 0.8$	
Calisthenics - Sit ups	$0.1 \pm 0.1$	$0.2 \pm 0.4$	$0.4 \pm 0.4$	$0.2 \pm 0.2$	$0.3 \pm 0.5$	
Cycling - Cycle hard - Cycle 80rpm	$1.0 \pm 1.1$	$0.7 \pm 0.7$	$1.0 \pm 1.1$	$0.8 \pm 0.8$	$0.7 \pm 0.6$	
Cycling - Cycle light - Cycle 100rpm	$0.1 \pm 0.2$	$0.6 \pm 1.2$	$0.7 \pm 1.4$	$0.1 \pm 0.2$	$0.1 \pm 0.1$	
Cycling - Cycle light - Cycle 60rpm	$0.2 \pm 0.2$	$0.5 \pm 0.4$	$0.4 \pm 0.4$	$0.3 \pm 0.3$	$0.3 \pm 0.2$	
Cycling - Cycle light - Cycle 80rpm	$1.1 \pm 1.3$	$1.2 \pm 1.2$	$1.2 \pm 1.3$	$1.3 \pm 1.2$	$1.7 \pm 1.6$	
Cycling - Cycle moderate - Cycle 80rpm	$0.8 \pm 0.9$	$1.1 \pm 0.8$	$0.9 \pm 1.3$	$0.8 \pm 0.8$	$1.1 \pm 1.2$	
Lying down	$0.7 \pm 1.1$	$0.8 \pm 1.5$	$1.0 \pm 1.7$	$0.8 \pm 1.2$	$0.8 \pm 1.3$	
Rowing - Rowing hard - Rowing 30spm	$0.7 \pm 0.8$	$0.9 \pm 1.0$	$0.9 \pm 1.2$	$0.7 \pm 0.8$	$1.3 \pm 1.3$	
Rowing - Rowing light - Rowing 30spm	$0.7 \pm 0.7$	$0.8 \pm 1.1$	$0.7 \pm 0.9$	$1.0 \pm 0.9$	$0.5 \pm 0.7$	
Rowing - Rowing moderate - Rowing 30spm	$0.8\pm0.8$	$0.9 \pm 1.1$	$0.8 \pm 1.2$	$0.8 \pm 0.7$	$0.9 \pm 0.8$	
Running - Treadmill 4mph - Treadmill 0	$0.7 \pm 0.9$	$0.4 \pm 0.6$	$0.4 \pm 0.5$	$0.4 \pm 0.3$	$0.5 \pm 0.4$	
Running - Treadmill 5mph - Treadmill 0	$0.7\pm0.6$	$0.8 \pm 0.7$	$0.8\pm0.6$	$1.0 \pm 0.8$	$0.8 \pm 0.6$	
Running - Treadmill 6mph - Treadmill 0	$0.3\pm0.5$	$0.4 \pm 0.5$	$0.8 \pm 0.8$	$0.7 \pm 1.0$	$1.0 \pm 1.4$	
Sitting	$0.7\pm0.9$	$0.6 \pm 0.7$	$0.8 \pm 0.7$	$0.6 \pm 0.7$	$1.1 \pm 2.4$	
Sitting - Fidget feet legs	$0.3 \pm 0.3$	$0.4 \pm 0.4$	$0.2 \pm 0.3$	$0.4 \pm 0.5$	$0.3 \pm 0.4$	
Sitting - Fidget hands arms	$0.5\pm0.8$	$0.4 \pm 0.4$	$0.4 \pm 0.3$	$0.3 \pm 0.3$	$0.4 \pm 0.3$	
Stairs - Ascend stairs	$0.6 \pm 0.5$	$0.7 \pm 0.5$	$1.1 \pm 1.3$	$0.5 \pm 0.2$	$0.7 \pm 0.8$	
Stairs - Descend stairs	$0.9\pm0.9$	$1.1 \pm 1.1$	$1.2 \pm 1.2$	$0.9 \pm 0.8$	$1.0 \pm 0.5$	
Standing	$0.4 \pm 0.3$	$0.4 \pm 0.5$	$0.4 \pm 0.4$	$0.4 \pm 0.3$	$0.4 \pm 0.3$	
Walking - Treadmill 2mph - Treadmill 0	$0.8 \pm 1.4$	$1.0 \pm 1.4$	$0.7\pm0.8$	$0.6\pm0.8$	$0.8\pm0.7$	
Walking - Treadmill 3mph - Treadmill 0	$1.4 \pm 1.4$	$1.0 \pm 1.1$	$1.1 \pm 1.5$	$1.5 \pm 1.2$	$1.1 \pm 0.8$	
Walking - Treadmill 3mph - Treadmill 3 -	$1.0 \pm 0.9$	$1.2 \pm 1.1$	$0.8\pm0.8$	$1.9 \pm 1.3$	$1.2 \pm 1.1$	
light			15.01	10.00	10.17	
Walking - Treadmill 3mph - Treadmill 6 -	$0.8 \pm 0.8$	$1.1 \pm 1.0$	$1.7 \pm 2.1$	$1.0 \pm 0.9$	$1.2 \pm 1.5$	
Walking - Treadmill 3mph - Treadmill 9 -	$0.9 \pm 1.2$	$0.6 \pm 0.5$	$0.8 \pm 1.0$	07+10	$0.8 \pm 0.8$	
hard	0.7 ± 1.2	0.0 ± 0.5	0.0 ± 1.0	0.7 ± 1.0	0.0 ± 0.0	
kneeling	$0.1\pm0.1$	$0.1\pm0.2$	$0.2\pm0.2$	$0.2\pm0.2$	$0.2\pm0.2$	
unknown	30.0 ±	$32.8\pm7.5$	$34.9\pm8.7$	$27.2\pm5.9$	$27.8\pm6.6$	
	8.0					
Carrying groceries	$1.3\pm1.4$	$1.3 \pm 1.4$	$0.9\pm0.7$	$1.0\pm0.8$	$1.2 \pm 1.4$	
Doing dishes	$0.7\pm0.5$	$0.7\pm0.6$	$0.7 \pm 0.5$	$1.0 \pm 0.7$	$1.0 \pm 0.8$	
Gardening	$0.6 \pm 0.5$	$0.6 \pm 0.5$	$0.8 \pm 0.6$	$0.7 \pm 0.5$	$0.7 \pm 0.7$	
Ironing	$0.8\pm0.5$	$0.7\pm0.6$	$0.7 \pm 0.5$	$0.8\pm0.6$	$0.8\pm0.8$	
Making the bed	$1.2\pm0.7$	$1.1 \pm 0.5$	$1.1 \pm 0.5$	$1.2 \pm 0.7$	$1.2 \pm 0.7$	
Mopping	$0.9\pm0.5$	$0.8\pm0.6$	$0.7 \pm 0.5$	$1.0 \pm 1.0$	$1.0 \pm 1.1$	
Playing videogames	$1.2 \pm 1.3$	$1.0\pm0.8$	$0.9 \pm 1.0$	$1.2 \pm 1.4$	$1.2 \pm 1.5$	
Scrubbing a surface	$0.8\pm0.8$	$1.0 \pm 1.1$	$0.8 \pm 1.0$	$0.8\pm0.8$	$0.9\pm0.9$	
Stacking groceries	$0.9\pm0.5$	$0.8\pm0.6$	$1.1 \pm 0.9$	$0.8 \pm 0.4$	$0.7 \pm 0.5$	
Sweeping	$0.8\pm0.6$	$0.8\pm0.5$	$0.7\pm0.5$	$0.9\pm0.4$	$0.8\pm0.4$	
Typing	$0.6 \pm 0.6$	$1.0 \pm 1.3$	$1.0 \pm 1.2$	$0.7 \pm 0.7$	$0.6 \pm 0.6$	
Vacuuming	$0.7\pm0.5$	$0.7 \pm 0.4$	$0.8 \pm 0.5$	$0.7 \pm 0.3$	0.7 ± 0.3	
Walking around block	$1.8 \pm 1.6$	$2.2 \pm 2.0$	$1.9 \pm 2.4$	$1.8 \pm 2.2$	$1.9 \pm 1.4$	
Washing windows	$0.9 \pm 0.5$	$1.2 \pm 1.5$	$1.4 \pm 1.5$	$1.1 \pm 1.2$	$1.0 \pm 0.8$	
Watching TV	$1.0 \pm 1.0$	$1.2 \pm 1.3$	$1.2 \pm 1.3$	$1.5 \pm 1.2$	$1.3 \pm 1.4$	
Weeding	$0.7 \pm 0.5$	$0.4 \pm 0.3$	$0.5 \pm 0.4$	$0.6 \pm 0.5$	$0.7 \pm 0.6$	
Wiping/Dusting	$0.9\pm0.5$	$1.0 \pm 0.8$	$1.0 \pm 1.0$	$0.8 \pm 0.6$	$1.0 \pm 0.5$	
Writing	$0.6 \pm 0.8$	$0.4 \pm 0.4$	$0.4 \pm 0.5$	$1.1 \pm 1.3$	$0.9 \pm 1.0$	
taking out trash	$1.0 \pm 0.5$	$0.9 \pm 0.4$	$0.8 \pm 0.6$	$1.0 \pm 0.5$	$0.9 \pm 0.3$	

Table A6-7: False positive rate when evaluating the four of the highest performing feature subsets computed per sensor using the C4.5 classifier in a subject independent manner.

			F-Measure		
Activity	AllButHR	ACAbsArea,	ACAbsArea	Total Invariant	Invariant
-		DCArea,	DCArea		Reduced
		ACVar,			
		ACRange,			
Danch weight lifting hand	12.0 + 10.0	10.1 + 21.9	11.0 + 00.2	7.0 + 12.5	0.8 + 20.7
Bench weight lifting - hard	$13.0 \pm 18.8$	$10.1 \pm 21.8$	$11.2 \pm 22.3$	$7.9 \pm 13.5$	$9.8 \pm 20.7$
Bench weight lifting - light	$14.6 \pm 21.4$	$30.3 \pm 30.2$	$23.6 \pm 28.3$	$16.6 \pm 23.7$	$25.0 \pm 29.5$
Bench weight lifting - moderate	$11.3 \pm 16.2$	$6.0 \pm 10.2$	$2.2 \pm 7.3$	$3.4 \pm 11.0$	$2.1 \pm 5.1$
Bicep curls - hard	$12.8 \pm 20.1$	$19.6 \pm 23.3$	$18.5 \pm 21.1$	$16.1 \pm 24.7$	$10.5 \pm 21.8$
Bicep curls - light	$17.8 \pm 19.2$	$13.0 \pm 18.2$	$15.6 \pm 26.8$	$21.4 \pm 24.0$	$22.4 \pm 23.5$
Bicep curls - moderate	$10.1 \pm 19.2$	$9.7 \pm 16.6$	$16.0 \pm 21.8$	$16.0 \pm 19.3$	$14.2 \pm 20.1$
Calisthenics - Crunches	$18.2 \pm 35.5$	$16.6 \pm 32.0$	$19.9 \pm 33.8$	$17.2 \pm 31.7$	$24.5 \pm 37.6$
Calisthenics - Sit ups	$48.8 \pm 44.2$	$36.9 \pm 41.7$	$27.2\pm37.3$	$49.0\pm45.0$	$44.0\pm44.4$
Cycling - Cycle hard - Cycle 80rpm	$10.2\pm12.6$	$8.1\pm15.3$	$7.6 \pm 13.4$	$7.8 \pm 14.4$	$5.6 \pm 7.1$
Cycling - Cycle light - Cycle 100rpm	$73.4\pm36.4$	$61.2 \pm 33.4$	$70.4\pm27.5$	$73.5 \pm 33.5$	$71.1 \pm 39.1$
Cycling - Cycle light - Cycle 60rpm	$44.3 \pm 37.1$	$36.4 \pm 32.6$	$36.2 \pm 34.9$	$49.4 \pm 37.2$	$53.0\pm37.5$
Cycling - Cycle light - Cycle 80rpm	$35.6 \pm 31.1$	$16.3 \pm 22.2$	$13.4 \pm 20.2$	$35.7 \pm 24.9$	$30.2 \pm 25.8$
Cycling - Cycle moderate - Cycle 80rpm	$16.0 \pm 15.7$	$10.7 \pm 11.6$	$14.3 \pm 18.2$	$12.2 \pm 14.4$	$17.9 \pm 22.0$
Lying down	$75.3 \pm 31.2$	$73.6 \pm 31.4$	$76.7 \pm 23.9$	$71.2 \pm 31.9$	$68.3 \pm 33.4$
Rowing - Rowing hard - Rowing 30spm	$19.5 \pm 22.7$	$16.4 \pm 14.7$	$19.3 \pm 20.7$	$15.1 \pm 17.4$	$19.4 \pm 20.0$
Rowing - Rowing light - Rowing 30spm	$24.8 \pm 21.4$	$16.2 \pm 15.8$	$17.0 \pm 17.8$	$27.6 \pm 25.8$	$23.5 \pm 23.7$
Rowing - Rowing moderate - Rowing 30spm	18.6 + 19.1	$12.9 \pm 18.3$	11.7 + 16.4	21.5 + 23.3	15.0 + 14.9
Running - Treadmill 4mph - Treadmill 0	$30.7 \pm 29.8$	27.0 + 33.9	30.2 + 35.0	324 + 320	26.8 + 34.5
Running - Treadmill 5mph - Treadmill 0	$46.7 \pm 29.0$	$44.8 \pm 26.8$	$33.0 \pm 28.0$	$36.8 \pm 26.8$	$44.1 \pm 25.9$
Running - Treadmill 6mph - Treadmill 0	$40.2 \pm 20.3$	$\frac{44.0 \pm 20.0}{21.4 \pm 28.3}$	$33.0 \pm 28.0$	$36.0 \pm 26.8$	$37.3 \pm 20.2$
Sitting	$30.9 \pm 20.0$	$31.4 \pm 20.3$	$44.0 \pm 26.0$	$30.0 \pm 20.8$	$37.3 \pm 20.2$
Sitting Fidget feet legs	$12.3 \pm 19.4$ 20.7 ± 29.1	$24.7 \pm 34.4$ $27.4 \pm 28.1$	$16.2 \pm 20.8$ 25.0 ± 21.2	$21.0 \pm 29.4$ $27.2 \pm 27.7$	$12.1 \pm 20.4$
Sitting - Fidget herde arms	$30.7 \pm 38.1$	$27.4 \pm 20.1$	$23.0 \pm 31.2$	$21.2 \pm 21.1$	$33.1 \pm 33.2$
Sitting - Fluget hands arms	$30.3 \pm 31.3$	$18.8 \pm 27.2$	$14.0 \pm 25.0$	$20.8 \pm 28.4$	$27.7 \pm 29.4$
Stairs - Ascend stairs	$49.6 \pm 32.1$	$38.6 \pm 27.3$	$20.2 \pm 25.8$	$50.5 \pm 31.2$	$49.3 \pm 33.0$
Stairs - Descend stairs	$38.5 \pm 29.0$	$22.5 \pm 18.7$	$14.3 \pm 15.9$	$40.4 \pm 27.7$	$27.3 \pm 22.1$
Standing	$39.9 \pm 34.3$	$39.3 \pm 32.6$	38.6 ± 34.1	42.6 ± 34.7	$39.8 \pm 34.5$
Walking - Treadmill 2mph - Treadmill 0	$45.4 \pm 32.6$	$39.8 \pm 32.6$	$40.6 \pm 34.9$	$44.6 \pm 32.2$	$49.0 \pm 27.7$
Walking - Treadmill 3mph - Treadmill 0	$21.0 \pm 20.0$	8.7 ± 11.6	$6.9 \pm 12.8$	$13.0 \pm 15.1$	$12.2 \pm 14.0$
Walking - Treadmill 3mph - Treadmill 3 - light	$10.0 \pm 10.1$	$12.5 \pm 11.6$	5.6 ± 8.3	$14.9 \pm 13.8$	$13.5 \pm 13.2$
Walking - Treadmill 3mph - Treadmill 6 - moderate	11.1 ± 12.6	11.1 ± 12.1	13.7 ± 15.6	$12.2 \pm 11.0$	$10.8\pm12.0$
Walking - Treadmill 3mph - Treadmill 9 -	$12.6\pm16.2$	$11.1 \pm 11.8$	$10.5\pm15.3$	$8.0 \pm 10.5$	$12.7\pm15.2$
kneeling	657 + 122	$60.0 \pm 37.0$	665+366	$58.6 \pm 41.3$	$53.0 \pm 45.2$
unknown	$54.0 \pm 8.0$	$52.0 \pm 37.0$	$52.0 \pm 8.2$	$55.0 \pm 41.3$	$53.9 \pm 43.2$
Carrying grocories	$34.0 \pm 0.9$	$33.2 \pm 0.4$	$32.0 \pm 0.2$	$33.2 \pm 9.0$	$34.0 \pm 0.9$
Doing dishes	$21.3 \pm 20.7$	$10.1 \pm 19.1$	$20.4 \pm 24.0$	$21.1 \pm 19.3$	$21.3 \pm 24.4$
Cardening	$26.6 \pm 23.0$	$31.9 \pm 20.7$	$29.3 \pm 24.2$	$50.4 \pm 23.9$	$33.3 \pm 24.3$
Gardening	$16.3 \pm 22.9$	$16.3 \pm 22.3$	9.0 ± 15.1	$18.5 \pm 22.7$	$13.0 \pm 21.2$
Ironing	$37.9 \pm 29.7$	$35.4 \pm 30.8$	$29.4 \pm 27.5$	$40.6 \pm 28.4$	$33.1 \pm 30.4$
Making the bed	$25.6 \pm 17.6$	$25.6 \pm 17.2$	$24.7 \pm 17.5$	$26.7 \pm 17.9$	$28.0 \pm 18.2$
Mopping	$24.7 \pm 21.3$	$23.0 \pm 21.1$	$25.0 \pm 23.4$	$21.3 \pm 20.6$	$23.2 \pm 23.1$
Playing videogames	$27.7 \pm 31.0$	$17.5 \pm 25.8$	$16.1 \pm 27.1$	$22.5 \pm 27.4$	$17.6 \pm 25.6$
Scrubbing a surface	$14.9 \pm 18.6$	$13.3 \pm 15.6$	$13.4 \pm 15.1$	$14.9 \pm 21.1$	$12.1 \pm 15.6$
Stacking groceries	$13.5 \pm 14.0$	$16.0 \pm 16.9$	$10.1 \pm 12.6$	$16.6\pm16.0$	$16.8\pm16.8$
Sweeping	$17.9 \pm 19.0$	$14.0 \pm 16.8$	$14.3 \pm 18.3$	$16.3 \pm 16.6$	$15.3 \pm 17.6$
Typing	$50.1\pm35.5$	$38.0\pm33.4$	$35.4 \pm 32.2$	$45.4\pm35.6$	$50.6 \pm 35.4$
Vacuuming	$25.9\pm23.9$	$27.8\pm25.2$	$17.2\pm17.6$	$28.2\pm26.7$	$25.5 \pm 22.5$
Walking around block	$18.9 \pm 16.3$	$20.1\pm23.2$	$13.0 \pm 14.0$	$19.1 \pm 17.8$	$17.8 \pm 18.4$
Washing windows	$24.7\pm21.1$	$19.9 \pm 17.3$	$19.1\pm16.8$	$23.5\pm21.1$	$28.3\pm24.3$
Watching TV	$19.1 \pm 24.9$	$18.2\pm23.6$	$15.4\pm26.4$	$15.1\pm26.0$	$19.3\pm29.6$
Weeding	$6.0 \pm 10.1$	$7.0 \pm 11.8$	$5.6 \pm 10.7$	$7.2 \pm 7.7$	$6.6 \pm 11.0$
Wiping/Dusting	$23.7 \pm 21.1$	$21.9 \pm 19.9$	$19.3 \pm 17.6$	$24.7 \pm 22.0$	$23.9 \pm 22.6$
Writing	$51.6 \pm 38.4$	$44.6 \pm 35.2$	$41.7 \pm 35.9$	$38.9 \pm 31.3$	$47.6 \pm 35.6$
taking out trash	$11.5 \pm 11.7$	$10.4 \pm 11.3$	$10.6 \pm 9.9$	$11.7 \pm 13.6$	$10.2 \pm 9.9$

Table A6-8: F-Measure when evaluating the four of the highest performing feature subsets computed per sensor using the C4.5 classifier in a subject independent manner.

	True Positive Rate					
Activity	AllButHR	ACAbsArea,	ACAbsArea	Total Invariant	Invariant	
		DCArea,	DCArea		Reduced	
		ACVar,				
		ACRange,				
Bench weight lifting - hard	80.6 ± 10.3	873±85	80.2 + 7.2	$90.3 \pm 7.5$	$90.8 \pm 10.5$	
Bonch weight lifting light	$0.0 \pm 10.3$	$07.3 \pm 0.3$	$09.2 \pm 7.2$	$90.3 \pm 7.3$	$90.8 \pm 10.3$	
Bench weight hitting - light	$93.0 \pm 0.4$	$90.3 \pm 15.3$	$92.9 \pm 8.1$	$91.2 \pm 9.4$	$93.8 \pm 1.0$	
Bench weight lifting - moderate	84.9 ± 16.4	85.8 ± 12.7	8/.0±11.0	86.4 ± 13.7	89.1 ± 11.5	
Bicep curis - hard	90.9 ± 12.0	$91.3 \pm 11.9$	$91.4 \pm 10.2$	91.6 ± 8.1	88.4 ± 14.5	
Bicep curls - light	89.7 ± 8.8	$88.6 \pm 10.6$	89.3 ± 8.7	89.1 ± 9.7	88.8 ± 9.6	
Bicep curls - moderate	$91.5 \pm 8.5$	$90.8 \pm 8.3$	$91.0 \pm 8.6$	$89.5 \pm 9.9$	$89.8 \pm 8.7$	
Calisthenics - Crunches	$91.9 \pm 5.2$	$95.4 \pm 3.8$	$95.5 \pm 3.7$	$93.0 \pm 5.4$	$95.2 \pm 5.5$	
Calisthenics - Sit ups	$92.3 \pm 4.8$	$95.0 \pm 4.3$	$94.3 \pm 5.3$	$93.2 \pm 5.0$	$95.2 \pm 4.7$	
Cycling - Cycle hard - Cycle 80rpm	$83.5 \pm 18.9$	84.0 ± 13.2	$85.8 \pm 13.6$	$85.4 \pm 12.8$	$86.8 \pm 13.4$	
Cycling - Cycle light - Cycle 100rpm	$95.8 \pm 5.4$	$94.2 \pm 5.4$	$93.4 \pm 7.2$	$94.9 \pm 5.8$	$94.6 \pm 5.1$	
Cycling - Cycle light - Cycle 60rpm	$88.6\pm7.3$	$91.2\pm5.6$	$92.3 \pm 5.4$	$91.1\pm4.8$	$92.4\pm4.9$	
Cycling - Cycle light - Cycle 80rpm	$92.6\pm5.5$	$91.8\pm5.6$	$93.1\pm5.6$	$92.9\pm5.7$	$91.7\pm8.7$	
Cycling - Cycle moderate - Cycle 80rpm	$86.2 \pm 8.2$	$83.8\pm8.9$	$85.6\pm8.9$	$85.0 \pm 8.4$	$85.7 \pm 7.1$	
Lying down	$99.5 \pm 0.8$	$99.2 \pm 1.4$	$99.2 \pm 1.2$	$99.6 \pm 0.6$	$99.2 \pm 1.0$	
Rowing - Rowing hard - Rowing 30spm	81.6 ± 15.2	$80.8 \pm 16.0$	$83.2 \pm 13.4$	$82.8 \pm 13.0$	85.3 ± 12.2	
Rowing - Rowing light - Rowing 30spm	$86.5 \pm 9.3$	$84.5 \pm 12.5$	$86.9 \pm 9.0$	$82.8 \pm 10.3$	83.6 ± 11.8	
Rowing - Rowing moderate - Rowing 30spm	$81.6 \pm 15.8$	$78.2 \pm 14.6$	$79.6 \pm 13.4$	$80.1 \pm 14.9$	$81.4 \pm 11.5$	
Running - Treadmill 4mph - Treadmill 0	88.8 + 8.7	92.0 + 7.6	88.9 + 8.6	89.8 + 10.2	90.0 + 8.5	
Running - Treadmill 5mph - Treadmill 0	87.7 + 6.1	87.3 + 8.6	88.5 + 6.9	86.9 + 5.7	86.6 + 8.6	
Running - Treadmill 6mph - Treadmill 0	$87.2 \pm 15.9$	$86.8 \pm 13.2$	$90.4 \pm 8.0$	$79.8 \pm 19.3$	$78.9 \pm 19.2$	
Sitting	$924 \pm 66$	$91.4 \pm 5.8$	$92.0 \pm 6.7$	$89.6 \pm 7.9$	$93.9 \pm 19.2$	
Sitting - Fidget feet legs	$91.2 \pm 7.3$	$91.4 \pm 3.0$	$94.6 \pm 6.1$	$0.0 \pm 7.9$	$91.0 \pm 9.1$	
Sitting - Fidget hands arms	$91.2 \pm 7.3$	$91.9 \pm 7.0$ $91.7 \pm 6.4$	$94.0 \pm 0.1$ $92.4 \pm 6.5$	$90.1 \pm 9.9$ 83.2 ± 12.4	$91.0 \pm 9.1$ $91.6 \pm 7.5$	
Stairs Accand stairs	$86.9 \pm 6.9$	$91.7 \pm 0.4$ 86.1 ± 0.5	$92.4 \pm 0.3$ $97.2 \pm 7.0$	$85.2 \pm 12.4$	$91.0 \pm 7.3$ $92.7 \pm 9.2$	
Stairs - Ascend stairs	$03.4 \pm 3.0$	$80.1 \pm 9.3$	07.2 ± 7.0	$00.1 \pm 9.1$	$02.7 \pm 0.2$	
Standing	$83.1 \pm 7.3$	$83.3 \pm 8.0$	$63.6 \pm 3.0$	$04.0 \pm 0.1$	$03.0 \pm 1.1$	
Standing Welling Tree deails 2 meth. Tree deails 0	$88.4 \pm 5.8$	$89.9 \pm 6.8$	$90.7 \pm 0.1$	$85.2 \pm 15.9$	87.4 ± 9.9	
waiking - Treadmill 2mph - Treadmill 0	89.1 ± 4.5	$91.2 \pm 5.0$	$92.0 \pm 6.3$	$90.9 \pm 4.9$	92.1 ± 5.0	
Walking - Treadmill 3mph - Treadmill 0	82.6 ± 6.9	84.2 ± 7.6	83.0 ± 8.5	/9.1 ± /.2	80.2 ± 8.9	
Walking - Treadmill 3mph - Treadmill 3 - light	76.4 ± 11.4	75.5 ± 11.9	79.1 ± 10.2	74.4 ± 10.0	70.0 ± 15.0	
Walking - Treadmill 3mph - Treadmill 6 - moderate	$76.8 \pm 13.6$	74.8 ± 11.9	78.5 ± 11.1	$75.2 \pm 12.0$	72.4 ± 11.7	
Walking - Treadmill 3mph - Treadmill 9 - hard	83.0 ± 12.4	86.3 ± 10.3	85.3 ± 9.2	82.9 ± 11.0	81.4 ± 12.4	
kneeling	$93.0 \pm 5.9$	$96.2 \pm 4.0$	$95.3 \pm 4.0$	$94.4 \pm 5.0$	$95.2 \pm 3.9$	
unknown	$74.8 \pm 5.5$	$75.6 \pm 5.2$	$76.4 \pm 5.9$	$73.5 \pm 7.0$	$73.6 \pm 6.4$	
Carrying groceries	86.6 + 7.4	87.2 + 8.7	88.8 + 6.8	84.9 + 10.6	87.0 + 10.8	
Doing dishes	$81.5 \pm 10.9$	83.1 + 9.7	83.8 + 7.5	$75.3 \pm 10.8$	$80.4 \pm 11.0$	
Gardening	$80.8 \pm 10.4$	$80.9 \pm 14.8$	$81.0 \pm 10.7$	$74.5 \pm 13.6$ $74.5 \pm 13.4$	74.1 + 12.4	
Ironing	$79.4 \pm 10.2$	819 + 96	841+86	$77.0 \pm 12.8$	77.6 + 13.1	
Making the bed	$54.5 \pm 13.4$	$585 \pm 138$	$59.4 \pm 11.3$	$54.8 \pm 9.0$	$53.0 \pm 10.2$	
Monning	$54.5 \pm 13.4$ $62.0 \pm 12.1$	$55.5 \pm 13.0$	$57.4 \pm 11.5$ $67.7 \pm 10.9$	$60.2 \pm 12.8$	$60.2 \pm 13.9$	
Playing videogames	$02.0 \pm 12.1$	$05.0 \pm 13.0$	$07.7 \pm 10.9$	$00.2 \pm 12.0$	$00.2 \pm 13.9$	
Scrubbing a surface	$90.1 \pm 3.0$ 70.2 ± 12.0	$93.4 \pm 3.0$ 70.5 ± 15.9	$93.3 \pm 4.0$ 80.0 ± 14.5	$94.0 \pm 0.2$	$95.1 \pm 5.0$ 70.0 ± 14.4	
Stacking groceries	$79.3 \pm 12.9$	$79.3 \pm 15.8$	$64.6 \pm 14.5$	$77.1 \pm 14.0$	$79.0 \pm 14.4$	
Stacking grocenes	$36.2 \pm 13.2$	$61.3 \pm 13.3$	$64.0 \pm 14.0$	$39.0 \pm 13.0$	$38.4 \pm 14.0$	
Tuning	$00.0 \pm 13.7$	$00.0 \pm 13.1$	$03.0 \pm 14.4$	$02.4 \pm 14.0$	$04.7 \pm 14.8$	
i ypnig Voormeine	$90.8 \pm 4.4$	$90.0 \pm 3.4$	$90.3 \pm 3.8$	$90.1 \pm 3.9$	$90.7 \pm 4.1$	
vacuuming Wallaine assessed block	/1.6 ± 9.2	$/1.4 \pm 10.4$	$70.0 \pm 10.6$	$64.6 \pm 10.4$	$6/.9 \pm 11.5$	
Walking around block	84.2 ± 9.9	85.3 ± 9.9	$85.2 \pm 8.2$	80.6 ± 9.1	83.8 ± 9.0	
wasning windows	$66.9 \pm 8.4$	$68.6 \pm 10.4$	$6/.0 \pm 10.1$	$62.9 \pm 11.4$	$63.8 \pm 8.8$	
watching TV	$91.0 \pm 6.9$	$94.0 \pm 4.7$	$92.2 \pm 6.6$	$92.9 \pm 4.3$	$94.5 \pm 5.8$	
Weeding	$72.1 \pm 16.5$	$78.9 \pm 10.3$	$76.5 \pm 12.4$	$68.0 \pm 14.6$	$69.7 \pm 14.2$	
Wiping/Dusting	$63.7 \pm 13.2$	$65.5 \pm 13.0$	$66.8 \pm 13.5$	$57.3 \pm 13.2$	$60.0 \pm 13.6$	
Writing	$95.6 \pm 3.9$	$96.5 \pm 3.6$	$96.1 \pm 3.8$	$96.7 \pm 3.7$	$96.9 \pm 2.6$	
taking out trash	$45.8 \pm 11.7$	$56.1 \pm 13.2$	$59.2 \pm 9.8$	$48.4 \pm 11.4$	$53.9 \pm 12.8$	

Table A6-9: True positive rate when evaluating the four of the highest performing feature subsets computed per axis using the C4.5 classifier in a subject dependent manner.

	False Positive Rate					
Activity	AllButHR	ACAbsArea,	ACAbsArea	Total Invariant	Invariant	
		DCArea,	DCArea		Reduced	
		ACVar,				
		ACRange,				
Bench weight lifting - hard	$0.1 \pm 0.1$	ACMUR $0.1 \pm 0.1$	0.1 + 0.1	0.1 + 0.1	$0.1 \pm 0.1$	
Bench weight lifting light	$0.1 \pm 0.1$					
Bench weight lifting - nght	$0.1 \pm 0.1$					
Bench weight lifting - moderate	$0.1 \pm 0.1$	$0.1 \pm 0.0$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	
Bicep curis - hard	$0.2 \pm 0.1$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	
Bicep curls - light	$0.1 \pm 0.1$	$0.2 \pm 0.1$	$0.2 \pm 0.2$	$0.2 \pm 0.1$	$0.2 \pm 0.2$	
Bicep curls - moderate	$0.2 \pm 0.2$	$0.2 \pm 0.1$	$0.2 \pm 0.1$	$0.2 \pm 0.2$	$0.2 \pm 0.2$	
Calisthenics - Crunches	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.1 \pm 0.0$	
Calisthenics - Sit ups	$0.1 \pm 0.1$					
Cycling - Cycle hard - Cycle 80rpm	$0.2 \pm 0.1$					
Cycling - Cycle light - Cycle 100rpm	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.1 \pm 0.0$	
Cycling - Cycle light - Cycle 60rpm	$0.1 \pm 0.1$	$0.2 \pm 0.1$	$0.1 \pm 0.1$	$0.2 \pm 0.1$	$0.2 \pm 0.1$	
Cycling - Cycle light - Cycle 80rpm	$0.2 \pm 0.1$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.2 \pm 0.1$	$0.1 \pm 0.1$	
Cycling - Cycle moderate - Cycle 80rpm	$0.2 \pm 0.1$					
Lying down	$0.1 \pm 0.1$					
Rowing - Rowing hard - Rowing 30spm	$0.2 \pm 0.1$	$0.3 \pm 0.2$	$0.2 \pm 0.2$	$0.2 \pm 0.2$	$0.2 \pm 0.2$	
Rowing - Rowing light - Rowing 30spm	$0.2 \pm 0.1$					
Rowing - Rowing moderate - Rowing 30spm	$0.3 \pm 0.2$	$0.3 \pm 0.2$	$0.3 \pm 0.2$	$0.4 \pm 0.2$	$0.3 \pm 0.2$	
Running - Treadmill 4mph - Treadmill 0	$0.2 \pm 0.1$					
Running - Treadmill 5mph - Treadmill 0	$0.2 \pm 0.1$					
Running - Treadmill 6mph - Treadmill 0	$0.1 \pm 0.1$	$0.1 \pm 0.0$	$0.1 \pm 0.1$	$0.2 \pm 0.1$ 0.2 + 0.1	$0.2 \pm 0.1$	
Sitting	$0.1 \pm 0.1$	$0.1 \pm 0.0$ 0.1 + 0.1	$0.1 \pm 0.1$	$0.2 \pm 0.1$ 0.1 ± 0.1	$0.2 \pm 0.1$ 0.1 ± 0.1	
Sitting - Fidget feet legs	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.1 \pm 0.1$ 0.1 + 0.1	
Sitting - Fidget hands arms	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.2 \pm 0.1$	$0.1 \pm 0.1$	
Stairs Accand stairs	$0.2 \pm 0.1$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.2 \pm 0.2$	$0.1 \pm 0.1$	
Stairs - Ascend stairs	$0.2 \pm 0.1$	$0.2 \pm 0.2$	$0.2 \pm 0.1$	$0.2 \pm 0.1$	$0.3 \pm 0.2$	
Standing	$0.2 \pm 0.2$	$0.2 \pm 0.1$	$0.2 \pm 0.1$	$0.2 \pm 0.1$	$0.2 \pm 0.1$	
Standing Welling Tree deails 2 mells Tree deails 0	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.1 \pm 0.0$	$0.2 \pm 0.1$	$0.1 \pm 0.1$	
waiking - Treadmill 2mpn - Treadmill 0	$0.2 \pm 0.1$					
Walking - Treadmill 3mph - Treadmill 0	$0.3 \pm 0.2$	$0.4 \pm 0.1$	$0.3 \pm 0.2$	$0.3 \pm 0.1$	$0.4 \pm 0.2$	
Walking - Treadmill 3mph - Treadmill 3 - light	$0.4 \pm 0.2$	$0.5 \pm 0.2$	$0.4 \pm 0.2$	$0.5 \pm 0.1$	0.6 ± 0.2	
Walking - Treadmill 3mph - Treadmill 6 - moderate	$0.4 \pm 0.2$	$0.4 \pm 0.2$	$0.4 \pm 0.2$	$0.4 \pm 0.2$	$0.5 \pm 0.2$	
Walking - Treadmill 3mph - Treadmill 9 - hard	$0.3 \pm 0.1$	$0.3 \pm 0.1$	$0.2 \pm 0.1$	$0.3 \pm 0.2$	$0.3 \pm 0.2$	
kneeling	$0.1 \pm 0.1$	$0.1 \pm 0.0$	$0.1 \pm 0.0$	$0.0 \pm 0.1$	$0.1 \pm 0.0$	
unknown	$7.6 \pm 1.4$	$7.3 \pm 1.4$	$7.2 \pm 1.3$	$8.2 \pm 1.4$	$7.6 \pm 1.5$	
Carrying groceries	$0.2 \pm 0.1$	$0.2 \pm 0.1$	$0.2 \pm 0.1$	$0.2 \pm 0.1$	$0.2 \pm 0.2$	
Doing dishes	$0.3 \pm 0.2$	0.4 + 0.2	$0.2 \pm 0.2$	0.4 + 0.2	0.4 + 0.2	
Gardening	$0.3 \pm 0.2$	$0.3 \pm 0.2$	$0.3 \pm 0.1$	$0.4 \pm 0.2$	$0.4 \pm 0.2$	
Ironing	$0.4 \pm 0.2$	$0.3 \pm 0.2$	$0.3 \pm 0.2$	$0.5 \pm 0.2$	$0.4 \pm 0.2$	
Making the bed	$0.8 \pm 0.3$	$0.7 \pm 0.3$	$0.7 \pm 0.3$	$0.8 \pm 0.3$	$0.8 \pm 0.2$	
Mopping	$0.0 \pm 0.0$	$0.6 \pm 0.2$	$0.6 \pm 0.2$	$0.0 \pm 0.0$ 0.7 ± 0.2	$0.0 \pm 0.2$ 0.7 ± 0.3	
Playing videogames	$0.1 \pm 0.1$	$0.0 \pm 0.2$ 0.1 + 0.1	$0.0 \pm 0.2$ 0.1 + 0.1	$0.1 \pm 0.2$ 0.1 ± 0.1	$0.1 \pm 0.0$ 0.1 + 0.1	
Scrubbing a surface	$0.1 \pm 0.1$	$0.1 \pm 0.1$ 0.2 ± 0.1	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.1 \pm 0.1$ 0.3 ± 0.2	
Stacking groceries	$0.5 \pm 0.2$	$0.2 \pm 0.1$ 0.5 ± 0.3	$0.5 \pm 0.2$	$0.5 \pm 0.2$	$0.5 \pm 0.2$ 0.6 ± 0.3	
Sweeping	$0.6 \pm 0.3$	$0.5 \pm 0.3$	$0.0 \pm 0.3$	$0.0 \pm 0.2$	$0.0 \pm 0.3$	
Typing	$0.0 \pm 0.4$ 0.1 ± 0.0	$0.0 \pm 0.3$ 0.1 ± 0.1	$0.0 \pm 0.3$ 0.1 ± 0.1	$0.7 \pm 0.3$ 0.1 ± 0.1	$0.0 \pm 0.3$	
Vacuuming	$0.1 \pm 0.0$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	
Walking around block	$0.0 \pm 0.2$	$0.3 \pm 0.2$	$0.3 \pm 0.2$	$0.0 \pm 0.2$	$0.3 \pm 0.3$	
Washing windows	$0.5 \pm 0.2$	$0.2 \pm 0.2$	$0.2 \pm 0.1$	$0.5 \pm 0.1$	$0.5 \pm 0.1$	
Watching TV	$0.5 \pm 0.1$	$0.5 \pm 0.2$	$0.5 \pm 0.2$	$0.6 \pm 0.2$	$0.6 \pm 0.2$	
Watching IV	$0.2 \pm 0.1$	$0.2 \pm 0.1$	$0.2 \pm 0.1$	$0.2 \pm 0.1$	$0.1 \pm 0.0$	
weeding	$0.4 \pm 0.2$	$0.4 \pm 0.2$	$0.3 \pm 0.2$	$0.4 \pm 0.2$	$0.4 \pm 0.2$	
wiping/Dusting	$0.6 \pm 0.3$	$0.5 \pm 0.2$	$0.6 \pm 0.2$	$0.7 \pm 0.4$	$0.8 \pm 0.4$	
Writing	$0.1 \pm 0.1$					
taking out trash	$0.7 \pm 0.2$	$0.7 \pm 0.3$	$0.6 \pm 0.2$	$0.7 \pm 0.3$	$0.7 \pm 0.2$	

Table A6-10: False positive rate when evaluating the four of the highest performing feature subsets computed per axis using the C4.5 classifier in a subject dependent manner.

			F-Measure		
Activity	AllButHR	ACAbsArea,	ACAbsArea	Total Invariant	Invariant
		DCArea,	DCArea		Reduced
		ACVar,			
		ACRange,			
Bench weight lifting - hard	89.8 + 10.6	887+7.6	89.7 + 6.3	89.1 + 8.0	91.1 + 9.9
Bench weight lifting - light	$90.7 \pm 7.8$	88 2 + 13 6	$91.0 \pm 8.4$	$89.6 \pm 9.3$	$91.1 \pm 9.9$ $91.1 \pm 9.2$
Bench weight lifting - moderate	84.7 + 13.3	$86.4 \pm 11.2$	87.7 + 9.5	84.9 + 12.4	88.6 + 10.3
Bicep curls - hard	$88.9 \pm 12.2$	$90.5 \pm 10.8$	91.2 + 8.2	90.3 + 8.5	87.6 + 15.2
Bicep curls - light	90.2 + 7.3	88.8 + 10.1	89.0 + 8.4	88.8 + 8.9	87.7 + 9.1
Bicep curls - moderate	90.6 + 8.8	89.6 + 7.6	90.1 + 8.1	88.6 + 9.4	88.1 + 8.6
Calisthenics - Crunches	$92.5 \pm 4.4$	95.4 + 2.9	94.7 + 2.9	92.5 + 4.3	94.9 + 3.1
Calisthenics - Sit ups	$92.0 \pm 4.2$	$92.8 \pm 5.1$	$92.6 \pm 5.5$	$92.3 \pm 4.2$	$93.2 \pm 5.0$
Cycling - Cycle hard - Cycle 80rpm	81.9 ± 18.6	82.6 ± 12.5	83.0 ± 12.9	84.0 ± 13.7	84.9 ± 13.4
Cycling - Cycle light - Cycle 100rpm	$94.3 \pm 5.3$	$93.3 \pm 5.6$	$93.5 \pm 6.5$	$94.2 \pm 6.0$	$93.6 \pm 6.4$
Cycling - Cycle light - Cycle 60rpm	$89.7 \pm 5.0$	$90.9 \pm 4.8$	$91.9 \pm 4.4$	$90.2 \pm 3.3$	$90.8 \pm 4.6$
Cycling - Cycle light - Cycle 80rpm	91.6 ± 5.3	$91.4 \pm 5.1$	$92.2 \pm 4.6$	91.6 ± 4.3	$91.6 \pm 6.5$
Cycling - Cycle moderate - Cycle 80rpm	86.3 ± 7.2	$84.3 \pm 6.8$	85.6 ± 7.1	$85.2 \pm 7.0$	$84.8 \pm 6.8$
Lying down	$99.0 \pm 0.8$	$98.6 \pm 1.4$	$98.7 \pm 1.0$	$98.9 \pm 0.8$	$98.8 \pm 1.0$
Rowing - Rowing hard - Rowing 30spm	81.4 ± 14.3	$79.6 \pm 16.0$	83.0 ± 13.0	$82.2 \pm 12.4$	$84.2 \pm 11.3$
Rowing - Rowing light - Rowing 30spm	$85.8 \pm 8.0$	$84.7 \pm 11.5$	$86.5 \pm 7.9$	$84.2 \pm 8.7$	$83.7 \pm 10.0$
Rowing - Rowing moderate - Rowing 30spm	80.1 ± 15.2	$78.0 \pm 14.2$	$79.3 \pm 13.2$	$78.3 \pm 14.4$	$79.8 \pm 11.5$
Running - Treadmill 4mph - Treadmill 0	$88.5 \pm 6.4$	$89.9 \pm 6.7$	$88.5 \pm 6.2$	$89.2 \pm 7.7$	$88.7 \pm 7.0$
Running - Treadmill 5mph - Treadmill 0	87.2 ± 5.1	$87.6 \pm 6.6$	88.8 ± 5.2	86.1 ± 5.7	87.3 ± 6.1
Running - Treadmill 6mph - Treadmill 0	86.1 ± 13.3	$84.6 \pm 12.4$	$89.0 \pm 7.4$	$80.4 \pm 17.3$	$79.8 \pm 17.3$
Sitting	$89.6 \pm 6.4$	$89.3 \pm 6.4$	$90.9 \pm 5.5$	$88.5 \pm 7.6$	$90.5 \pm 4.6$
Sitting - Fidget feet legs	$90.4 \pm 6.0$	$92.3 \pm 5.6$	$94.0 \pm 4.9$	$88.4 \pm 9.7$	$89.2 \pm 6.8$
Sitting - Fidget hands arms	$85.8 \pm 8.4$	91.3 ± 5.6	90.9 ± 4.3	82.4 ± 11.6	89.2 ± 7.3
Stairs - Ascend stairs	$85.4 \pm 5.8$	84.3 ± 8.2	$86.2 \pm 7.0$	$84.9 \pm 8.1$	$81.6 \pm 7.2$
Stairs - Descend stairs	83.1 ± 7.4	$85.2 \pm 6.5$	85.7 ± 5.7	$84.2 \pm 7.5$	$82.9 \pm 7.2$
Standing	$87.4 \pm 6.2$	$90.0 \pm 5.2$	$91.0 \pm 4.2$	83.6 ± 13.3	$87.1 \pm 8.8$
Walking - Treadmill 2mph - Treadmill 0	88.4 ± 3.9	89.8 ± 4.2	90.6 ± 5.6	89.9 ± 3.9	90.5 ± 4.1
Walking - Treadmill 3mph - Treadmill 0	$82.5 \pm 6.1$	$81.8 \pm 7.1$	$82.3 \pm 8.1$	$79.9 \pm 6.3$	$79.6 \pm 8.5$
Walking - Treadmill 3mph - Treadmill 3 -	$75.5\pm9.6$	74.4 ± 11.2	77.1 ± 9.1	$73.0\pm7.8$	68.7 ± 13.5
Walking - Treadmill 3mph - Treadmill 6 - moderate	76.9 ± 12.4	74.5 ± 11.1	77.7 ± 10.0	74.6 ± 11.1	71.6 ± 11.0
Walking - Treadmill 3mph - Treadmill 9 -	83.3 ± 10.7	85.5 ± 9.2	85.2 ± 8.4	83.5 ± 10.3	81.4 ± 11.5
kneeling	$92.5 \pm 4.6$	951+31	93.9 + 3.7	$94.6 \pm 5.0$	937+37
unknown	$75.3 \pm 1.0$	$762 \pm 52$	$768 \pm 52$	$737 \pm 69$	$745 \pm 60$
Carrying groceries	86.4 + 7.4	$86.9 \pm 7.2$	$88.7 \pm 5.8$	$85.2 \pm 9.4$	87.4 + 9.9
Doing dishes	$81.4 \pm 10.0$	$82.2 \pm 9.0$	$84.9 \pm 6.2$	76.4 + 9.3	78.8 + 9.6
Gardening	$80.3 \pm 10.2$	$80.3 \pm 15.3$	$79.9 \pm 10.0$	73.7 + 12.9	$75.5 \pm 12.8$
Ironing	79.4 + 9.3	82.1 + 8.7	83.1 + 8.3	$75.9 \pm 11.4$	$78.0 \pm 12.0$
Making the bed	$54.8 \pm 13.2$	59.0 + 13.4	$59.7 \pm 10.8$	55.0 + 8.8	54.3 + 8.8
Mopping	62.2 + 12.4	66.0 + 11.6	$67.8 \pm 10.6$	61.2 + 11.9	60.0 + 14.0
Plaving videogames	$95.2 \pm 3.3$	952 + 27	948 + 32	94.1 + 4.5	94.3 + 4.1
Scrubbing a surface	$80.4 \pm 11.1$	82.2 + 12.6	81.3 + 13.1	78.4 + 14.5	$80.9 \pm 13.7$
Stacking groceries	$59.5 \pm 12.9$	$62.9 \pm 16.0$	$64.2 \pm 13.5$	$60.2 \pm 12.9$	$59.5 \pm 14.9$
Sweeping	$65.7 \pm 12.5$	$66.7 \pm 13.0$	$67.0 \pm 14.0$	$62.7 \pm 13.2$	$65.4 \pm 13.2$
Typing	96.5 + 3.0	95.9 + 2.9	95.9 + 3.2	94.7 + 2.8	95.4 + 3.2
Vacuuming	$70.8 \pm 9.6$	$72.7 \pm 10.1$	$71.3 \pm 10.5$	65.9 + 9.2	$69.4 \pm 11.2$
Walking around block	83.9 + 9.7	85.9 + 8.7	86.5 + 6.9	81.3 + 8.2	84.4 + 6.7
Washing windows	69.1 + 7.3	70.6 + 9.4	69.7 + 9.0	64.9 + 9.7	65.7 + 7.4
Watching TV	91.0 + 6.4	92.1 + 4.9	91.8 + 5.5	92.3 + 4.2	93.5 + 4.2
Weeding	73.8 + 14.8	77.6 + 10.6	78.1 + 10.5	70.7 + 13.7	71.5 + 13.4
Wiping/Dusting	63.8 + 12.8	$67.0 \pm 10.0$	$67.0 \pm 10.5$	58.7 + 14.3	60.0 + 14.2
Writing	94.8 + 3.8	95.2 + 2.6	95.5 + 2.6	95.5 + 3.8	95.6 + 2.3
taking out trash	48.9 ± 11.3	56.9 ± 12.1	$60.6 \pm 8.9$	50.1 ± 11.3	54.5 ± 10.9

Table A6-11: F-Measure rate when evaluating the four of the highest performing feature subsets computed per axis using the C4.5 classifier in a subject dependent manner.

			True Positive Ra	te	
Activity	AllButHR	ACAbsArea,	ACAbsArea	Total Invariant	Invariant
		DCArea,	DCArea		Reduced
		ACVar,			
		ACKange,			
Bench weight lifting - hard	12.8 + 31.2	21.2 + 35.5	27.5 + 36.0	4.8 + 9.5	14.1 + 27.0
Bench weight lifting - light	$39.5 \pm 39.0$	$38.2 \pm 38.6$	$59.6 \pm 39.3$	$283 \pm 296$	285 + 324
Bench weight lifting - moderate	26.3 + 34.3	$29.2 \pm 35.9$	27.3 + 33.1	$25.6 \pm 31.3$	$31.9 \pm 40.8$
Bicep curls - hard	44.8 + 34.7	324 + 394	$27.5 \pm 35.1$ 28.1 + 38.5	$30.8 \pm 32.0$	$31.3 \pm 10.0$ $31.3 \pm 37.4$
Bicen curls - light	$31.8 \pm 39.1$	$25.6 \pm 35.2$	$30.2 \pm 38.0$	$25.0 \pm 32.0$	$19.0 \pm 29.2$
Bicep curls - moderate	$22.0 \pm 31.7$	$20.0 \pm 30.2$ $20.2 \pm 30.9$	$36.6 \pm 34.7$	$28.3 \pm 34.6$	28.4 + 33.8
Calisthenics - Crunches	$55.1 \pm 40.0$	$58.4 \pm 40.7$	$60.0 \pm 31.1$	57.2 + 37.9	48.2 + 38.1
Calisthenics - Sit ups	$74.6 \pm 34.9$	$70.6 \pm 36.7$	$53.9 \pm 42.5$	$\frac{37.2 \pm 37.9}{87.5 \pm 22.8}$	$66.9 \pm 36.2$
Cycling - Cycle hard - Cycle 80rpm	$74.0 \pm 34.9$ 25.7 + 27.0	$18.7 \pm 30.7$	$7.1 \pm 9.6$	$19.1 \pm 22.0$	$30.8 \pm 32.4$
Cycling - Cycle light - Cycle 100rpm	$67.1 \pm 27.0$	$10.7 \pm 32.2$ 897 + 197	$7.4 \pm 7.0$ 85.4 + 22.4	$17.1 \pm 20.3$ 65.6 + 34.2	$30.8 \pm 32.4$ 85 3 + 25 2
Cycling - Cycle light - Cycle 60rpm	$63.4 \pm 34.6$	$64.0 \pm 38.8$	$58.8 \pm 36.7$	$72.2 \pm 27.3$	$70.3 \pm 23.2$
Cycling - Cycle light - Cycle 80rpm	$28.8 \pm 27.6$	$34.0 \pm 30.0$	$30.0 \pm 30.7$	$72.2 \pm 27.3$ 22.6 + 30.6	$70.3 \pm 35.7$ $34.9 \pm 36.3$
Cycling - Cycle moderate - Cycle 80rpm	$26.0 \pm 27.0$	$34.4 \pm 37.1$	$14.0 \pm 43.0$ 21.0 ± 23.3	$22.0 \pm 30.0$ 28.8 ± 25.7	$34.7 \pm 30.3$
L ving down	$20.7 \pm 20.2$ 88.4 ± 25.6	$22.4 \pm 27.3$	$21.9 \pm 25.3$ 85.2 ± 26.8	$26.6 \pm 23.7$ 83.0 ± 28.1	$21.4 \pm 13.2$ 82.0 ± 23.7
Powing Powing hard Powing 30cpm	$33.4 \pm 23.0$	$30.9 \pm 16.0$	$33.2 \pm 20.8$	$33.0 \pm 23.1$	$32.9 \pm 23.7$ 21.6 ± 20.2
Rowing - Rowing light - Rowing 30spm	$29.0 \pm 29.3$	$57.2 \pm 30.7$	$32.0 \pm 41.4$	$30.2 \pm 33.9$	$21.0 \pm 30.2$ $22.2 \pm 20.4$
Rowing - Rowing moderate - Rowing 30spm	$41.9 \pm 40.0$ 22.0 ± 35.7	$33.0 \pm 33.0$ 8 1 ± 15 3	$42.3 \pm 30.4$ 10.6 ± 37.2	$20.0 \pm 29.0$ 28.7 ± 30.7	$32.2 \pm 29.4$ 22.1 ± 23.4
Running - Treadmill 4mph - Treadmill 0	$22.9 \pm 33.7$	$535 \pm 416$	$19.0 \pm 37.2$	$26.7 \pm 30.7$	$22.1 \pm 23.4$ 50 4 ± 38 7
Running Treadmill 5mph Treadmill 0	$44.8 \pm 28.8$	$35.5 \pm 41.0$	$41.0 \pm 41.3$ 55.2 ± 25.0	$44.0 \pm 40.4$	$30.4 \pm 30.7$
Running Treadmill 6mph Treadmill 0	$40.7 \pm 34.7$ 51.0 ± 25.5	$40.0 \pm 39.0$ 51.4 ± 25.1	$53.2 \pm 33.0$	$30.1 \pm 32.0$ $42.0 \pm 27.2$	$42.7 \pm 32.4$
Sitting	$51.0 \pm 35.3$	$51.4 \pm 35.1$	$03.0 \pm 30.1$	$42.0 \pm 37.2$	$49.3 \pm 34.1$
Sitting Eidget feet logs	$04.2 \pm 39.9$	$01.0 \pm 41.2$	$47.5 \pm 41.2$	$37.6 \pm 44.2$	$50.7 \pm 40.7$
Sitting - Fidget herds arms	$43.2 \pm 32.8$	$40.3 \pm 39.3$	$40.0 \pm 38.8$	$44.4 \pm 33.2$	$31.0 \pm 30.9$
Stuing - Fluget hands arms	$48.5 \pm 30.2$	$45.3 \pm 32.0$	$40.3 \pm 35.0$	$51.1 \pm 30.5$	$42.8 \pm 28.0$
Stairs - Ascend stairs	$60.3 \pm 21.0$	$44.1 \pm 21.8$	$37.4 \pm 22.0$	$59.9 \pm 27.0$	$60.3 \pm 22.3$
Stants - Descend stans	$69.9 \pm 27.3$	$50.5 \pm 20.2$	$47.8 \pm 29.8$	$03.3 \pm 23.0$	$50.7 \pm 28.0$
Walking Troodmill 2mph Troodmill 0	$03.4 \pm 32.1$	$49.6 \pm 37.2$	$39.0 \pm 30.3$	$04.0 \pm 27.3$	$44.0 \pm 30.0$
Walking Treadmill 2mph Treadmill 0	$30.9 \pm 31.2$	$43.0 \pm 30.3$	$46.9 \pm 32.4$	$34.2 \pm 29.8$	$34.4 \pm 27.3$
Walking Treadmill 2mph Treadmill 2	$23.4 \pm 24.0$	$0.3 \pm 11.4$	$10.7 \pm 19.0$	$21.3 \pm 22.4$	$17.1 \pm 10.3$
light	$18.8 \pm 20.4$	$11.1 \pm 17.8$	$9.8 \pm 12.2$	$10.0 \pm 17.8$	$18.1 \pm 28.0$
Walking - Treadmill 3mph - Treadmill 6 - moderate	16.1 ± 23.8	15.6 ± 22.4	15.0 ± 23.2	19.3 ± 21.8	$14.3 \pm 16.4$
Walking - Treadmill 3mph - Treadmill 9 -	$22.0\pm24.6$	$22.5\pm24.3$	$28.2\pm25.7$	$28.9\pm25.7$	21.1 ± 24.1
kneeling	89.5 + 21.5	93.8 + 7.3	88.0 + 16.6	74.1 + 40.7	74 5 + 37 4
unknown	$65.4 \pm 6.6$	$64.0 \pm 6.4$	$62.6 \pm 7.5$	$63.8 \pm 5.3$	61.7 + 4.6
Carrying groceries	55.5 + 26.5	33.8 + 23.9	34.5 + 29.5	49.1 + 26.1	52.2 + 29.0
Doing dishes	45.2 + 30.0	$53.6 \pm 26.1$	54.1 + 22.9	48.0 + 24.2	44.1 + 23.1
Gardening	14.0 + 21.4	19.1 + 22.4	14.1 + 17.9	15.2 + 19.0	11.8 + 14.8
Ironing	51.3 + 28.5	51.9 + 27.5	56.7 + 26.4	43.8 + 30.6	45.5 + 27.6
Making the bed	$37.8 \pm 15.7$	44.4 + 17.9	40.5 + 20.6	$32.1 \pm 12.5$	$36.8 \pm 15.7$
Mopping	$31.5 \pm 20.0$	$31.3 \pm 16.5$	$26.5 \pm 17.7$	$30.9 \pm 11.9$	31.1 + 9.4
Plaving videogames	49.7 + 39.6	46.5 + 43.9	$45.3 \pm 44.1$	362 + 39.8	47.1 + 43.2
Scrubbing a surface	$367 \pm 263$	$385 \pm 197$	$40.1 \pm 25.7$	$34.3 \pm 28.2$	$31.1 \pm 27.4$
Stacking groceries	34.4 + 15.4	$32.3 \pm 15.3$	$28.7 \pm 17.2$	$25.8 \pm 14.9$	$29.2 \pm 16.4$
Sweeping	$33.8 \pm 21.0$	$31.6 \pm 18.1$	$30.2 \pm 18.4$	$29.8 \pm 17.5$	$27.9 \pm 13.1$
Typing	$53.0 \pm 21.0$ 60.0 + 37.6	$70.8 \pm 27.6$	$59.2 \pm 10.4$	$29.0 \pm 15.3$ 58.8 + 36.2	$64.0 \pm 33.0$
Vacuuming	$41.8 \pm 21.2$	$44.5 \pm 10.1$	$38.4 \pm 10.0$	$40.5 \pm 30.2$	$43.2 \pm 33.9$
Walking around block	$\frac{41.0 \pm 21.2}{29.4 \pm 13.7}$	$44.3 \pm 19.1$ 25.4 + 18.7	$30.4 \pm 19.9$ 23.8 + 19.1	$40.3 \pm 21.0$ 26.0 ± 17.0	$-43.2 \pm 24.0$ 23.6 ± 16.3
Washing windows	$367 \pm 21.6$	$25.4 \pm 10.7$ $35.4 \pm 17.8$	$25.0 \pm 17.1$ $36.7 \pm 18.2$	$20.0 \pm 17.0$ 38.2 + 20.0	$40.7 \pm 20.8$
Watching TV	$35.7 \pm 21.0$ $35.6 \pm 30.5$	$35.7 \pm 17.0$ $35.5 \pm 38.7$	$20.7 \pm 10.2$	$39.2 \pm 20.9$	$41.7 \pm 20.0$
Weeding	$10.6 \pm 14.0$	$20.6 \pm 25.9$	$157 \pm 12.7$	$14.3 \pm 92.1$	$145 \pm 257$
Wining/Dusting	$10.0 \pm 14.0$ 39.9 + 10.5	$20.0 \pm 23.0$ 38.9 + 17.4	$13.7 \pm 10.0$ $39.2 \pm 17.7$	$14.3 \pm 23.1$ 33.8 + 18.6	$14.5 \pm 25.7$ $36.4 \pm 10.6$
Writing	$56.8 \pm 12.0$	71.5 + 34.2	$67.0 \pm 10.8$	$65.8 \pm 35.0$	$63.1 \pm 37.3$
taking out trash	$185 \pm 113$	$71.5 \pm 54.5$ 20.1 + 10.0	$18.4 \pm 12.2$	$18.2 \pm 33.3$	$160 \pm 100$
uning out trash	$10.3 \pm 11.3$	20.1 ± 10.9	10.4 ± 12.2	$10.2 \pm 12.1$	10.0 ± 10.0

Table A6-12: True positive rate when evaluating the four of the highest performing feature subsets computed per axis using the C4.5 classifier in a subject independent manner.

	False Positive Rate				
Activity	AllButHR	ACAbsArea,	ACAbsArea	Total Invariant	Invariant
		DCArea,	DCArea		Reduced
		ACVar,			
		ACKange,			
Bench weight lifting - hard	$0.3 \pm 0.4$	$0.4 \pm 0.5$	$0.2 \pm 0.3$	$0.4 \pm 0.5$	$0.3 \pm 0.5$
Bench weight lifting - light	$0.5 \pm 0.7$	$0.7 \pm 0.8$	$0.6 \pm 0.6$	$0.5 \pm 0.5$	$0.6 \pm 0.7$
Bench weight lifting - moderate	$0.4 \pm 0.5$	$0.6 \pm 0.5$	$0.4 \pm 0.4$	$0.6 \pm 0.5$	$0.8 \pm 0.8$
Bicep curls - hard	$1.4 \pm 0.8$	$1.0 \pm 1.2$	$0.8 \pm 0.9$	$0.9 \pm 0.9$	$1.1 \pm 1.0$
Bicep curls - light	$0.9 \pm 1.0$	$0.8 \pm 0.9$	$0.5 \pm 0.6$	$0.5 \pm 0.6$	$0.6 \pm 0.6$
Bicep curls - moderate	$0.5\pm0.5$	$1.0 \pm 1.0$	$1.1 \pm 1.0$	$0.8 \pm 0.8$	$0.9\pm0.8$
Calisthenics - Crunches	$0.1 \pm 0.2$	$0.1 \pm 0.1$	$0.1 \pm 0.2$	$0.4 \pm 0.8$	$0.3 \pm 0.5$
Calisthenics - Sit ups	$0.2 \pm 0.3$	$0.2 \pm 0.1$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.1 \pm 0.1$
Cycling - Cycle hard - Cycle 80rpm	$0.9\pm0.7$	$0.8 \pm 0.9$	$0.4 \pm 0.3$	$0.9\pm0.9$	$1.1 \pm 1.1$
Cycling - Cycle light - Cycle 100rpm	$0.1\pm0.1$	$0.1\pm0.1$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.3 \pm 0.5$
Cycling - Cycle light - Cycle 60rpm	$0.1\pm0.1$	$0.3\pm0.2$	$0.3 \pm 0.4$	$0.2 \pm 0.3$	$0.2 \pm 0.2$
Cycling - Cycle light - Cycle 80rpm	$0.9 \pm 1.0$	$1.2 \pm 1.2$	$1.2 \pm 1.1$	$0.6\pm0.8$	$0.9 \pm 1.0$
Cycling - Cycle moderate - Cycle 80rpm	$0.9\pm0.7$	$0.6\pm0.7$	$0.7 \pm 1.0$	$1.0 \pm 0.8$	$0.7\pm0.6$
Lying down	$0.1\pm0.1$	$0.2\pm0.3$	$0.2\pm0.3$	$0.4 \pm 0.6$	$0.3 \pm 0.4$
Rowing - Rowing hard - Rowing 30spm	$0.6\pm0.6$	$0.8\pm0.9$	$0.8\pm0.8$	$1.0 \pm 1.1$	$0.6\pm0.8$
Rowing - Rowing light - Rowing 30spm	$1.2 \pm 1.1$	$1.6 \pm 1.0$	$1.4 \pm 1.4$	$0.8\pm0.8$	$0.9\pm0.8$
Rowing - Rowing moderate - Rowing 30spm	$0.9 \pm 1.0$	$0.4 \pm 0.5$	$0.6 \pm 1.0$	$1.1 \pm 0.9$	$1.0 \pm 1.0$
Running - Treadmill 4mph - Treadmill 0	$0.6\pm0.6$	$0.9 \pm 1.5$	$0.3 \pm 0.4$	$0.4 \pm 0.3$	$0.7 \pm 0.6$
Running - Treadmill 5mph - Treadmill 0	$0.5 \pm 0.6$	$0.6 \pm 0.5$	$0.6 \pm 0.6$	$0.7 \pm 0.6$	$0.6 \pm 0.4$
Running - Treadmill 6mph - Treadmill 0	$0.3 \pm 0.4$	$0.5 \pm 0.6$	$0.6 \pm 0.8$	$0.4 \pm 0.4$	$0.5 \pm 0.6$
Sitting	$0.5 \pm 0.8$	$0.6 \pm 0.9$	$0.9 \pm 1.0$	$0.6 \pm 0.8$	$0.6 \pm 1.0$
Sitting - Fidget feet legs	$0.4 \pm 0.6$	$0.3 \pm 0.4$	$0.4 \pm 0.5$	$0.3 \pm 0.3$	$0.3 \pm 0.2$
Sitting - Fidget hands arms	$0.2 \pm 0.2$	$0.4 \pm 0.5$	$0.3 \pm 0.3$	$0.4 \pm 0.2$	$0.5 \pm 0.5$
Stairs - Ascend stairs	$0.3 \pm 0.1$	$1.0 \pm 0.9$	$0.7 \pm 0.5$	$0.4 \pm 0.3$	$0.5 \pm 0.4$
Stairs - Descend stairs	$0.3 \pm 0.2$	$0.5 \pm 0.4$	$0.6 \pm 0.4$	$0.9 \pm 1.4$	$0.5 \pm 0.3$
Standing	$0.4 \pm 0.4$	$0.2 \pm 0.2$	$0.3 \pm 0.2$	$0.3 \pm 0.3$	$0.3 \pm 0.3$
Walking - Treadmill 2mph - Treadmill 0	$0.4 \pm 0.5$	$0.3 \pm 0.2$	$0.3 \pm 0.2$	$0.5 \pm 0.5$	$0.5 \pm 0.2$
Walking - Treadmill 3mph - Treadmill 0	$1.0 \pm 1.0$	$0.6 \pm 0.4$	$1.0 \pm 0.8$	$0.8 \pm 0.5$	$1.0 \pm 0.6$
waiking - Treadmill Smpn - Treadmill S -	$0.9 \pm 0.6$	$0.7 \pm 0.7$	$0.7 \pm 0.6$	$0.6 \pm 0.4$	$1.3 \pm 1.2$
Walking - Treadmill 3mph - Treadmill 6 -	$1.2 \pm 1.2$	1.4 ± 1.3	$1.2 \pm 1.0$	$1.2 \pm 1.0$	$0.9\pm0.8$
Walking - Treadmill 3mph - Treadmill 9 -	$0.8\pm0.6$	$1.1 \pm 0.8$	1.0 ± 1.2	$1.1 \pm 1.0$	$1.2 \pm 1.2$
kneeling	$0.1 \pm 0.1$	$0.2 \pm 0.3$	$0.1 \pm 0.1$	$0.2 \pm 0.6$	$0.2 \pm 0.5$
unknown	$24.0 \pm$	$0.2 \pm 0.3$ 235 ± 67	$0.1 \pm 0.1$ 25.1 + 5.8	$24.8 \pm 6.6$	$0.2 \pm 0.3$ 24 2 + 5 4
	7.2	23.5 ± 0.7	$25.1 \pm 5.0$	$24.0 \pm 0.0$	$24.2 \pm 3.4$
Carrying groceries	$1.3 \pm 1.7$	$1.2 \pm 1.3$	$1.1 \pm 0.8$	$0.9 \pm 1.1$	$0.8 \pm 0.8$
Doing dishes	$0.7\pm0.3$	$0.6 \pm 0.3$	$0.5 \pm 0.3$	$0.9 \pm 0.5$	$0.8 \pm 0.3$
Gardening	$0.6\pm0.5$	$0.7\pm0.7$	$0.6 \pm 0.5$	$0.8\pm0.4$	$0.8\pm0.4$
Ironing	$0.8\pm0.6$	$0.8\pm0.4$	$0.7\pm0.4$	$0.7\pm0.4$	$0.8\pm0.4$
Making the bed	$1.2\pm0.5$	$1.2\pm0.8$	$1.3\pm0.8$	$1.3\pm0.8$	$1.3 \pm 0.7$
Mopping	$1.1 \pm 0.4$	$1.1\pm0.4$	$1.1\pm0.5$	$1.2 \pm 0.4$	$1.1 \pm 0.5$
Playing videogames	$0.5 \pm 0.7$	$0.8\pm0.8$	$1.0 \pm 1.0$	$0.7 \pm 0.9$	$0.7\pm0.9$
Scrubbing a surface	$0.9\pm0.7$	$1.3 \pm 1.0$	$1.2 \pm 1.1$	$1.0 \pm 0.6$	$1.0 \pm 0.7$
Stacking groceries	$0.9\pm0.4$	$0.9\pm0.3$	$0.9\pm0.5$	$0.8 \pm 0.4$	$0.8\pm0.3$
Sweeping	$1.0 \pm 0.4$	$1.0\pm0.6$	$1.2\pm0.8$	$0.9 \pm 0.3$	$1.1 \pm 0.4$
Typing	$0.3\pm0.5$	$0.3 \pm 0.4$	$0.4 \pm 0.5$	$0.5 \pm 0.6$	$0.5 \pm 0.7$
Vacuuming	$0.6\pm0.2$	$0.6 \pm 0.3$	$0.9 \pm 0.4$	$0.7 \pm 0.3$	$0.8 \pm 0.3$
Walking around block	$2.2 \pm 2.4$	$2.0 \pm 1.5$	$2.4 \pm 2.2$	$1.9 \pm 1.7$	$2.0 \pm 2.3$
Washing windows	$1.0 \pm 0.4$	$0.9 \pm 0.5$	$0.9 \pm 0.3$	$0.9 \pm 0.3$	$0.8 \pm 0.5$
Watching TV	$1.2 \pm 2.3$	$1.1 \pm 1.2$	$1.0 \pm 1.2$	0.9 ± 1.0	$0.6 \pm 0.5$
Weeding	$0.6 \pm 0.5$	$0.5 \pm 0.3$	$0.6 \pm 0.4$	$0.8 \pm 0.5$	$0.7 \pm 0.5$
Wiping/Dusting	$0.8 \pm 0.4$	$0.9 \pm 0.5$	$0.9 \pm 0.4$	$0.9 \pm 0.5$	$0.9 \pm 0.4$
Writing	$0.4 \pm 0.6$	$0.4 \pm 0.4$	$0.6 \pm 1.1$	$0.3 \pm 0.3$	$0.3 \pm 0.4$
taking out trash	$1.1 \pm 0.5$	$1.0 \pm 0.4$	$1.1 \pm 0.5$	$1.0 \pm 0.3$	$1.0 \pm 0.4$

Table A6-13: False positive rate when evaluating the four of the highest performing feature subsets computed per axis using the C4.5 classifier in a subject independent manner.

	F-Measure				
Activity	AllButHR	ACAbsArea,	ACAbsArea	Total Invariant	Invariant
		DCArea,	DCArea		Reduced
		ACVar,			
		ACRange,			
Bench weight lifting - hard	80+160	$18.2 \pm 21.2$	20.5 ± 25.8	$5.4 \pm 11.4$	$12.2 \pm 17.2$
Bonch weight lifting light	$8.0 \pm 10.9$	$16.2 \pm 31.2$	$29.3 \pm 33.8$	$3.4 \pm 11.4$	$12.2 \pm 17.2$
Bench weight lifting mederate	$34.8 \pm 31.0$	$30.0 \pm 27.0$	$49.3 \pm 29.3$	$20.8 \pm 22.4$	$20.8 \pm 27.4$
Bench weight inting - moderate	$19.8 \pm 22.8$	$21.3 \pm 21.5$	$24.8 \pm 25.8$	$19.2 \pm 18.7$	$20.0 \pm 20.3$
Bicep curis - nard	$32.3 \pm 20.1$	$19.2 \pm 21.6$	$19.5 \pm 23.4$	$24.4 \pm 21.0$	$23.5 \pm 27.7$
Bicep curis - light	$22.3 \pm 21.9$	$20.6 \pm 22.7$	$28.3 \pm 32.2$	$24.1 \pm 20.9$	$17.3 \pm 23.2$
Bicep curls - moderate	$21.2 \pm 23.8$	$13.4 \pm 16.6$	$27.1 \pm 20.3$	$21.2 \pm 20.0$	$21.2 \pm 20.1$
Calisthenics - Crunches	$57.7 \pm 37.7$	$61.1 \pm 39.0$	$64.2 \pm 37.8$	$54.2 \pm 35.7$	$49.3 \pm 35.2$
Calisthenics - Sit ups	$74.4 \pm 31.3$	$70.1 \pm 32.2$	$56.5 \pm 41.2$	$87.8 \pm 22.5$	$70.0 \pm 34.9$
Cycling - Cycle hard - Cycle 80rpm	$20.0 \pm 18.9$	$14.7 \pm 21.2$	$9.2 \pm 11.1$	$14.6 \pm 16.6$	$21.1 \pm 18.5$
Cycling - Cycle light - Cycle 100rpm	$72.1 \pm 31.9$	$90.0 \pm 14.2$	$87.1 \pm 17.4$	$71.7 \pm 29.3$	$82.6 \pm 20.6$
Cycling - Cycle light - Cycle 60rpm	$68.0\pm32.6$	$64.5 \pm 34.5$	$61.3 \pm 33.2$	$74.3 \pm 23.6$	$72.0 \pm 32.5$
Cycling - Cycle light - Cycle 80rpm	$27.6\pm22.2$	$24.4\pm22.6$	$29.5\pm27.4$	$22.2\pm21.0$	$29.4 \pm 27.3$
Cycling - Cycle moderate - Cycle 80rpm	$23.5\pm17.9$	$21.8\pm22.4$	$20.2\pm16.7$	$24.2\pm17.8$	$22.9 \pm 13.9$
Lying down	$90.3 \pm 22.2$	$92.7 \pm 12.0$	$87.5 \pm 23.8$	$84.1 \pm 24.4$	86.1 ± 16.9
Rowing - Rowing hard - Rowing 30spm	$27.3 \pm 23.0$	$32.1 \pm 28.4$	$26.1 \pm 33.1$	$23.4 \pm 18.6$	$18.6 \pm 18.4$
Rowing - Rowing light - Rowing 30spm	$27.8 \pm 23.5$	$37.6 \pm 17.2$	$28.3 \pm 15.8$	$22.3 \pm 19.7$	$28.3 \pm 17.9$
Rowing - Rowing moderate - Rowing 30spm	$15.8 \pm 21.8$	8.1 ± 13.6	$12.4 \pm 21.9$	$22.4 \pm 20.6$	$19.1 \pm 16.0$
Running - Treadmill 4mph - Treadmill 0	$46.1 \pm 28.2$	$46.9 \pm 37.1$	$41.4 \pm 36.4$	$45.0 \pm 37.9$	$47.9 \pm 35.8$
Running - Treadmill 5mph - Treadmill 0	$46.8 \pm 30.5$	$42.6 \pm 30.6$	$53.0 \pm 31.4$	$34.6 \pm 25.1$	$40.8 \pm 25.4$
Running - Treadmill 6mph - Treadmill 0	52.2 + 35.9	45.3 + 32.6	52.7 + 30.2	37.0 + 28.3	44.5 + 25.4
Sitting	$55.8 \pm 35.0$	532 + 369	38.4 + 36.0	$345 \pm 406$	$27.3 \pm 36.7$
Sitting - Fidget feet legs	$49.0 \pm 32.0$	$47.8 \pm 36.2$	$39.6 \pm 34.5$	$47.1 \pm 33.4$	$50.9 \pm 32.0$
Sitting - Fidget hands arms	$49.0 \pm 32.9$ 50.2 + 31.6	$47.0 \pm 30.2$	$33.0 \pm 34.5$	$\frac{47.1 \pm 33.4}{50.5 \pm 24.4}$	$30.9 \pm 32.0$
Stairs - Ascend stairs	$50.2 \pm 31.0$	$40.3 \pm 31.1$ $41.2 \pm 17.9$	$41.0 \pm 32.0$ 20.2 ± 20.4	$50.3 \pm 24.4$	$41.3 \pm 24.0$ 65.7 ± 19.1
Stairs - Descend stairs	$60.0 \pm 22.6$	$41.2 \pm 17.8$ 51.2 ± 24.1	$39.3 \pm 20.4$	$59.8 \pm 24.8$	$03.7 \pm 10.1$ 51.4 ± 24.5
Standing	$09.0 \pm 23.0$	$51.3 \pm 24.1$ 50.1 ± 25.2	$43.0 \pm 24.0$ 57 4 ± 22 5	$50.7 \pm 25.4$	$31.4 \pm 24.3$
Walking Treadmill 2mph Treadmill 0	$00.3 \pm 29.3$	$30.1 \pm 33.2$	$57.4 \pm 55.5$	$02.4 \pm 20.2$	$43.7 \pm 34.4$
Walking - Treadmill 2mph - Treadmill 0	$34.3 \pm 20.4$	$46.0 \pm 29.1$	$33.8 \pm 28.2$	$33.7 \pm 27.8$	$30.9 \pm 24.0$
Walking - Treadmin Sinph - Treadmin 0	$20.7 \pm 17.3$	$10.3 \pm 13.3$	$10.7 \pm 18.4$	$21.0 \pm 20.4$	$17.3 \pm 15.4$
light	17.8 ± 15.6	$11.0 \pm 11.6$	$10.9 \pm 12.2$	$17.2 \pm 16.8$	13.7 ± 16.3
Walking - Treadmill 3mph - Treadmill 6 - moderate	$12.2 \pm 14.2$	$11.8 \pm 14.6$	$11.9 \pm 16.0$	$15.8 \pm 14.3$	$14.5 \pm 15.8$
Walking - Treadmill 3mph - Treadmill 9 - hard	$21.9\pm20.0$	$22.7\pm20.4$	$27.0\pm24.5$	27.4 ± 21.3	$19.2\pm18.6$
kneeling	88.1 + 18.2	89.6 + 9.9	88.9 + 11.4	71.1 + 39.6	71.2 + 35.6
unknown	54.7 + 7.1	54.2 + 7.2	52.0 + 7.4	53.2 + 6.9	52.2 + 6.2
Carrying groceries	$50.0 \pm 22.2$	$32.6 \pm 22.8$	$32.7 \pm 23.8$	$48.3 \pm 21.8$	$52.0 \pm 25.9$
Doing dishes	44.7 + 26.1	$53.9 \pm 20.7$	57.1 + 18.5	464 + 204	$43.6 \pm 19.3$
Gardening	$14.7 \pm 20.1$ 14.8 + 21.7	$21.3 \pm 20.7$	$167 \pm 205$	$16.1 \pm 20.1$ $16.5 \pm 19.5$	$14.1 \pm 17.3$
Ironing	$50.3 \pm 24.9$	$51.3 \pm 22.0$	$55.9 \pm 21.8$	$43.7 \pm 27.1$	$45.1 \pm 23.9$
Making the bed	$35.8 \pm 13.1$	$31.3 \pm 23.0$ $41.8 \pm 13.6$	$37.2 \pm 14.8$	$43.7 \pm 27.1$ 31.3 + 10.5	$34.8 \pm 11.7$
Monning	$30.7 \pm 15.1$	$41.0 \pm 15.0$ $31.3 \pm 14.6$	$37.2 \pm 14.0$ 26.3 ± 15.5	$31.3 \pm 10.3$ $31.0 \pm 10.4$	$31.8 \pm 8.3$
Playing videogames	$30.7 \pm 13.0$	$31.3 \pm 14.0$ $42.2 \pm 28.1$	$20.3 \pm 13.3$ 20.2 ± 27.2	$31.0 \pm 10.4$	$31.0 \pm 0.3$
Scrubbing a surface	$40.4 \pm 34.3$	$43.2 \pm 30.1$	$39.3 \pm 37.2$	$34.0 \pm 33.4$	$43.9 \pm 37.4$
Steaking grospring	$30.3 \pm 23.4$	$30.2 \pm 17.0$	$30.4 \pm 21.4$	$33.2 \pm 24.0$	$30.1 \pm 23.0$
Stacking grocenes	$33.4 \pm 14.4$	$32.7 \pm 13.0$	$28.0 \pm 13.8$	$27.1 \pm 14.3$	$31.1 \pm 10.0$
Typing	$34.3 \pm 1/./$	$33.0 \pm 13.3$	$31.3 \pm 1/.4$	$32.0 \pm 14.2$	$29.8 \pm 13.1$
i ypnig Ve energine	$62.4 \pm 36.4$	$13.9 \pm 23.5$	$60.8 \pm 34.8$	$58.9 \pm 31.8$	$03.4 \pm 29.1$
vacuuming Wallsing agent hlagh	$44./\pm 18.0$	$4/.3 \pm 16.4$	$39.9 \pm 19.3$	42.2 ± 19.5	$43.7 \pm 21.9$
Walking around block	$2/.2 \pm 11.6$	$20.9 \pm 13.7$	$21.5 \pm 14.6$	$23.2 \pm 15.5$	$23.7 \pm 16.3$
wasning windows	3/.6 ± 22.1	37.6 ± 18.9	38.5 ± 19.0	39.6 ± 21.0	$42.9 \pm 21.4$
Watching TV	$31.7 \pm 33.6$	$32.8 \pm 33.6$	$19.9 \pm 29.0$	$36.8 \pm 36.4$	$41.2 \pm 37.8$
Weeding	$12.5 \pm 15.1$	$20.0 \pm 20.3$	$18.7 \pm 20.2$	$14.8 \pm 19.9$	$14.7 \pm 22.9$
Wiping/Dusting	$40.7 \pm 18.8$	$39.8 \pm 16.1$	$39.6 \pm 16.0$	$34.6 \pm 17.2$	37.1 ± 16.4
Writing	$56.0 \pm 41.0$	$70.5 \pm 28.6$	$62.8 \pm 38.4$	$66.5 \pm 34.1$	$64.0 \pm 34.3$
taking out trash	$19.1\pm10.6$	$21.6 \pm 11.5$	$19.7 \pm 12.9$	$20.0\pm13.0$	$17.3\pm10.3$

Table A6-14: F-Measure when evaluating the four of the highest performing feature subsets computed per axis using the C4.5 classifier in a subject independent manner.

	CHERT CHERT	
	我我它口里学说就不好我心思会我会不可怜得关军怎!人! 一 马知口戏的更穿孔主力我主题机口 护城于两年	
	4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4	
	4 0 1 1 1 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	
	доооооооооооооооооооооооооооооооооооо	
	н н 1 н н н н н н н н н н н н н н н н н	
	00000000000000000000000000000000000000	
	800 10000000000000000000000000000000000	
	10000000000000000000000000000000000000	
	на 2 або с с с с с с с с с с с с с с с с с с с	Lock
	8 0 1 1 1 1 1 1 2 1 2 1 2 1 2 1 2 1 2 1 2	d d b ws ash ash
		= be froc a stin stin
		ing d the d the d the d the d the d the d t
	4 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	den cing pin pin cing kin kin kin cing cing cing
		Gar Ircc Mop Pla Scr Pla Scr Pla Vo Vac Vac Vac Vac Vac Vac Vac Vac Vac Vac
		A REALX RECORDED
	и о о о о о о о о о о о о о о о о о о о	44
	日 日	ighi
s		
¥	· · · · · · · · · · · · · · · · · · ·	
D	0000000000000000000000000000000000000	
Ξ	< o o o o o o o o o o o o o o o o i o o o o o o o a de se de la la de se o o o o o o o o de o o o o o de o o o o	11_( eadr eadr eadr eadr
ΙE		
s s	く 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	ph
¥,	00000000000000000000000000000000000000	1 1 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2
CI	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	imil function in the set of the s
E	2000 1210 1210 1210 1210 1210 1210 1210	ill_ idge scendd read read read scend read
Γ	、 、 、 、 、 、 、 、 、 、 、 、 、 、	Sher T TTT Peser T
N	* « + ~ + ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~	Ing Ing Ing Ing Ing Ing Ing Own Own Own Own Own
Y X	**************************************	tunn sittt sittt sittt alkankankan alkankankan alkankankan alkan alka
ЕУ	20000000000000000000000000000000000000	
	H 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	oba (  <mark>b</mark> >u < axx m < ax
	асорионорости и соста со со со со со со со со со со со со со	
	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
	N 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
		u dis
	400000000iiiiiiiiiiiiiiiiiiiiiiiiiiiii	m pm spm iill
	00000000000000000000000000000000000000	100 100 100 100 100 100 100 100 100 100
	20000000000000000000000000000000000000	int lerai cole cole ring Kong vin
	00000000000000000000000000000000000000	Pheno s ministration s ministration s s s s s s s s s s s s s s s s s s s
	# 0 0 0 0 0 9 0 0 0 0 0 0 0 0 0 0 0 0 0	ig ig ht ht erai erai erai
	0 0 0 0 0 0 9 0 0 0 0 0 0 0 0 0 0 0 0 0	ftin ftin ftin ard light light har high har high fin d
	H 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	11111111111111111111111111111111111111
	8 4 1 N C C C N 8 8 C C C C C C C C C C C C C	ight http://www.ss. Rowworks. CCVS.
	H O O O H S S O O O O O O O O O O O O O	weeken weeke
	м <del>4</del> 4 м <del>4</del> 4 м 2 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	ench liceph lice
	4 4 40 4 4 4 40 9 6 4 4 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
	4 # # N 1 # # N 1 # N	КШООЫЕЮНЬХНЖХОРОК

Figure A6-1: Confusion Matrix for C4.5 Classifier using the *ACAbsArea*, *DCArea*, *ACVar*, *ACRange*, *and ACMCR* feature set computed per sensor and evaluated in a subject independent manner.

	C M H S H Z A L M A M A M A M A M A M A M A M A M A	
	ゆきごうそうゆせまびだいがおののみえきますびが火王ジーペート キシコネシモロバシのドロロン	
	риососочничессникососсикройноёнаёкаёкойсоё	
	жоноосососокимости и осторовой и колососоки в осторости осторости и осторости и осторости осторости осторости о	
	10120100000000000000000000000000000000	
	9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	
	00000000000000000000000000000000000000	
	10000000000000000000000000000000000000	
	1 0 0 0 0 0 0 0 0 1 1 1 0 0 0 0 0 0 0 0	id surfi surfi seri ows ows ows ows ows ows ows ows ows of bio to to to to to to to to to to to to to
		t_triv
		ing diff diff diff diff diff diff diff dif
	риороноки се се се се коросоно се се се се се се се се се се се се се	rden ppin ppin rubb aackii shir trubb fiin fiir fiin fiin
	そりりつりりょうしゅうののこうのがいつつつつのののののののののののののののののののであるのののです。 うちょう ひょう しょう うちょう ひょう しょう うちょう しょう ひょう しょう ひょう しょう ひょう しょう ひょう しょう ひょう ひょう ひょう ひょう ひょう ひょう ひょう ひょう ひょう ひ	A Maran Cara Maran Cara Maran Mar
	елелососососососососсоссоссососососососо	
	1048 - Kouou - Kubia garaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaa	
	1 + 1 + 1 + 1 + 1 + 1 + 1 + 1 + 1 + 1 +	++
	00000000000000000000000000000000000000	> ligh
s	A 47 47 47 47 47 47 47 47 47 47 47 47 44 44	
<pre>V</pre>	· • • • • • • • • • • • • • • • • • • •	
Е	90000000000000000000000000000000000000	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1
F	<pre>0000000000001000000000000000000000000</pre>	read read read read read
S	- 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
s		mph mph mph mph
ΓV	моссосомосоомосоосоосоосоосоосоосоосоосо	s lilili staaet -5
U	0 0 0 0 0 2 0 0 0 0 0 0 0 0 0 0 0 0 0 0	- 6mul 9et 9et admi. ddmi. ddmi. ddmi. ddmi. ddmi. rie rie rie
E	XX 00000000000000000000000000000000000	Tread rindo
Ч		
Μ		ning tring kking kking ng com king ng com ng com ng com ng com ng com ng com ng com ng com ng com ng com ng com ng com ng com ng com ng com com com com com com com com com com
X	осичессососососососсилали соссернососсилосос и посос п	Run Sit Sit Sit Sit Sit Sit Sit Sit Sit Sit
E	+ · · · · · · · · · · · · · · · · · · ·	^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^
	· · · · · · · · · · · · · · · · · · ·	NHDVRXYD-/
	ни 000000000000000000000000000000000000	
	и 	
	н н н н н н н н н н н н н н н н н н н	
	и 6 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
	2 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	e e
	H 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	rpm 1 ospi
		00000 0000 00000 00000 00000 00000 00000
	x o o o o o o o o o o o o o o o o o o o	ate e=10 g_30 ng_3 ng_3 reac
	роосоосоосоосоосоосоосоосоосоосоосоосоос	oder voi voi voi voi voi voi voi voi
	+ • • • • • • • • • • • • • • • • • • •	pt and constraints
		ing a d b d b d b d b d c c c c c c c c c c c c c
	PD0004400000004400000000000000000000000	ifitt: ift: ligh mode cru cru cru e ligh e ligh e ligh e ligh e ligh e ligh e ligh e ligh e ligh e ligh e ligh e ligh e f e ligh e f e f e f e f e f e f e f e f e f e
	#0000/#0000000000000000000000000000000	htt htt htt htt htt htt htt htt htt htt
		eeigl eeigl uurls uurls eenic own own cown cown cown cown cown cown
	0 жйёроо 4 о о о о о о о о о о о о о о о о о	2h_w 2h_w 2h_w 2h_w 2h_m 2h_m 2h_m 2h_m 2h_m 2h_m 2h_m 2h_m
	a (1) (1) (1) (1) (1) (1) (1) (1) (1) (1)	Benk Benk Berce Berce Califico Califico Califico Califico
	ж <sup>1</sup> 1800000000000000000000000000000000000	$\uparrow \uparrow $
		4 H O L D L H D L H D L H D L D L H D L D L H D L H D L H D L H D L H D L H D L H D L H D L H D L H D L H D L H

Figure A6-2: Confusion Matrix for C4.5 Classifier using the *ACFFTPeaks* feature computed per sensor and evaluated in a subject independent manner.



Figure A6-3: Confusion Matrix for C4.5 Classifier using the ACAbsArea and DCArea features computed per sensor and evaluated in a subject independent manner.



Figure A6-4: Confusion Matrix for C4.5 Classifier using the *MaxAcceleration* feature set computed per sensor and evaluated in a subject independent manner.

	LEBAL LAN-G-RO	
	みおごの変更のほよびた上述がつかられまていか得火を歩く、 「 みわこののまゆれょうをょれれっかびをませ	
	1 1 1 1 1 1 1 1 1 1 1 1 1 1	
	* ° % % ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° °	
	x o a o o o o o o o o o o o o o o o o o	
	цоососоцосоцононососць с <mark>1</mark> чососось 8 с с 6 ойс 4 ойс во 8 с с 6 ойс 4 ойс в с 8 с с 6 ойс 8 с с 6 ойс в с 8 с с	
	а а а а а а а а а а а а а а а а а а а	
	00000000000000000000000000000000000000	
	1 1 1 1 1 1 1 1 1 1 1 1 1 1	
	8 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
	40 × × × × × × × × × × × × × × × × × × ×	к С к С с
	x 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	l lifame srie srie ish
	00000000000000000000000000000000000000	ting trace
	4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4	out Tar
	H + 4 + 6 0 0 0 0 1 0 0 0 0 0 0 0 0 0 0 0 0 0 0	Meni Ping Aling Aling Aling Aling Aling
	10000000000000000000000000000000000000	Gar I Tan Mak Mak Mop Scri Scri Vak Vak Mat Mat Mat Mat Tak Tak
	00000000000000000000000000000000000000	<u>^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ </u>
	н и и и и и и и и и и и и и и и и и и и	じゅそうれょうメールはっちびょうち
	20000000000000000000000000000000000000	
	10/01/01/01/01/00/00000000000000000000	ght rd
	нонноож ноосоооооооооооооооооооооооооооо	- <sup>11</sup> - <sup>1</sup>
A S	4 4 4 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	
D	н н о и о о о о о о о о о о о о о о о о	a
E	000000000000000000000000000000000000000	admi admin admin admin admin admin
F	,	Tree Tree Tree Tree
S I		
A S		3 app
Г		s 11 111 stait
С		admi get get admi admi erie erie
LΕ	X 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	TTREE ESCERT
Р	хороноороророссососонарорососососоранироранориороро 0 0 0 н вн	
M	>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>	nin ttin ttin ttin kkin kkin kkin ng_c
X /	200000000000000000040000400004000044000 	Run Sitt Sitt Sitt Sta Sta Mal Wall Wall Wall Khe Khe Cark
Е		<u>^ ġ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^</u>
	N 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	NADARXHD-/
	00000000000000000000000000000000000000	
	00000000000000000000000000000000000000	
	**************************************	
	00000000000000000000000000000000000000	
	4 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	n pm pm spm spm sign
	X00000040940900000000000000000000000000	1000 1000 1000 1000 100 100 100 100 100
	, 000000000000000000000000000000000000	ht erat cole rve ving
	H 0 0 0 0 0 0 8 8 8 1 2 8 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	multiplication in the second s
		99 9 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1
	и осососососососососососососососососо 1 с с с с с с с с с с с с с с с с с с с	ttin ght nodert noder ligiligi ligiligi ligiligi ligiligi ligili
	8 2 1 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	and a state of the
	1441 1441 1441 1441 1441 1441 1441 144	ght ght ls ls ls cyc cyc cyc cyc cyc cyc cyc cyc cyc cy
	1 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	weile downer curr
	а а а а а а а а а а а а а а а а а а а	nch nch nch nch nch nch nch nch nch nch
	р м и и и и и и и и и и и и и и и и и и и	Been Been Beiting Cyver Row Row Row Row
	4 N 4 U N 8 U 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
1		NACHERORNOAHZZOHOR

Figure A6-5: Confusion Matrix for C4.5 Classifier using the *ACAbsArea*, *DCArea*, and *ACFFTPeaks* computed per sensor and evaluated in a subject independent manner.

	LEBAL LANHOHRO	
	10000000000000000000000000000000000000	
	а со со со со со со со со со со со со со	
	н со со со со со со со со со со со со со	
	80000000000000000000000000000000000000	
	00000000000000000000000000000000000000	
	алариана состания с с с с с с с с с с с с с с с с с с с	
	сооссоссоссоссосси чесь чесь чесь соссоссоссоссос с с с с с с с с с с с	
	マロンション・マン・マン・マン・マン・マン・マン・マン・マン・マン・マン・マン・マン・マン	
	нчичовоооооооооооооооооооооооооооооооооо	a X
	20000000000000000000000000000000000000	h bloc
	ино колоно о о о о о о о о о о о о о о о о о	bed sur sur cer ng ng ng
	4 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	a vide grou Trv astind
	4 + 10 + 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	nin a c bind bind bind bind bind a c bind bind c c bind c bind c c bind c c bind c c bind c c bind c c bind c c bind c c bind c c c bind c c c c c c c c c c c c c
	80000000000000000000000000000000000000	kini kini ppi ppi kini kini kini kini ki
	ноооооооооооооооооооооооооооооооооооо	
	* * 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
	доо4000004000ч00000000000000000000000000	
	0.0000000000000000000000000000000000000	
	······································	hard
s	235 235 235 235 235 234 234 234 244 244 244 255 255 255 255 255 255 25	1 III E4,
V	. 000000000000000000000000000000000000	
ED	1 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	11 11 11 11 11 11 11 11 11 11
Ξ	• • • • • • • • • • • • • • • • • • •	ead ead ead ead
Ξ	- 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
ss	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	phi- phi- phi- phi- phi- phi- phi- phi-
Y	и 1000000000000000000000000000000000000	
CI	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	fer function for the formula f
Ξ	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	read ill scendge scendge read read scend s
5 L	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	810 1 1 1 1 0 6 8 8 8 9 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1
Ā	околино о о о о о о о о о о о о о о о о о о	Ving diding Ving diding Ving
Y	ооооооооооооооооооооооооооооооооооооо	g itt: itt: tai tai tai tai tai tai tai tai tai tai
EX	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
	+ 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	oba (
	9 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
	a = = = = = = = = = = = = = = = = = = =	
	4 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
	x 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	u uc
	20000000000000000000000000000000000000	n n n n 30sp 11_(
		001 001 001 001 002 002 002 002 002 002
	X 0 0 0 0 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	rate lelle Cycles Kowij Trea
		oden cycle cycle cycle cycle
		The second
		ing http://www. ightun bhtard ightun ight ight ight ight ight ight ight ight
		idi p p p p p p p p p p p p p p p p p p
	N 10 0 0 1 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	httl httl httl httl httl httl httl httl
	****	eigl eigl uurls uurls eenid eenid enid
	0 m 0 m 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	zh.w zh.w zh.w zh.w zh.w zh.m zh.m zh.m zh.m zh.m zh.m zh.m zh.m
	ш <u>г</u> ййооооооооооооооооооооооооооооооооооо	BBBERK BBBERK BBBERK BBBERK CCCCCCCCCCCCCCCCCCCCCCCCCCCCCCCCCCCC
	καίδο ο ο ο ο ο ο ο ο ο ο ο ο ο ο ο ο ο ο	$\begin{array}{c} \begin{array}{c} \begin{array}{c} \end{array}{} \end{array} \\ \end{array} \\ \begin{array}{c} \end{array} \\ \end{array} \\ \end{array} \\ \begin{array}{c} \end{array} \\ \end{array} \\ \end{array} \\ \begin{array}{c} \end{array} \\ \end{array} \\ \end{array} \\ \begin{array}{c} \end{array} \\ \end{array} \\ \end{array} \\ \begin{array}{c} \end{array} \\ \end{array} \\ \end{array} \\ \begin{array}{c} \end{array} \\ \end{array} \\ \end{array} \\ \begin{array}{c} \end{array} \\ \end{array} \\ \end{array} \\ \begin{array}{c} \end{array} \\ \end{array} \\ \end{array} \\ \begin{array}{c} \end{array} \\ \end{array} \\ \end{array} \\ \begin{array}{c} \end{array} \\ \end{array} \\ \end{array} \\ \begin{array}{c} \end{array} \\ \end{array} \\ \end{array} \\ \begin{array}{c} \end{array} \\ \end{array} \\ \end{array} \\ \begin{array}{c} \end{array} \\ \end{array} \\ \end{array} \\ \begin{array}{c} \end{array} \\ \end{array} \\ \end{array} \\ \end{array} \\ \begin{array}{c} \end{array} \\ \end{array} \\ \end{array} \\ \end{array} \\ \begin{array}{c} \end{array} \\ \end{array} \\ \end{array} \\ \end{array} \\ \end{array} \\ \begin{array}{c} \end{array} \\ \end{array} \\ \end{array} \\ \end{array} \\ \end{array} \\ \end{array} \\ \end{array} \\ \end{array} \\ \end{array} \\ \end{array} $
	14 <del>4</del> 47	R R R R R R R R R R R R R R R R R R R

Figure A6-6: Confusion Matrix for C4.5 Classifier using the *ACAbsArea*, *DCArea*, *ACRange*, *ACVar*, *ACFFTPeaks*, *ACCorr*, *ACMCR*, and *ACPitch* computed per sensor evaluated in a subject independent manner.



Figure A6-7: Confusion Matrix for C4.5 Classifier using the *ACAbsArea*, *DCArea*, *ACVar*, *ACRange*, *and ACMCR* feature set computed per sensor and evaluated in a subject independent manner.

	LEBYL LAN-G-NO	
	みおじひますのはそうだんだおりかられるエリマがパナポートーへ 下 みわらはちまなねょうだえ 5 ちゅわはチョン	
	воососоносососоносоновиосососбеггчёное шёсия чо ч гоё	
	2 - & 1 - & 2 - &	
	· · · · · · · · · · · · · · · · · · ·	
	8 1 9 1 9 1 9 1 9 1 9 1 9 1 9 1 9 1 9 1	
	60 00000000000000000000000000000000000	
	20100000000000000000000000000000000000	
		ock ng
	и и 1 а о о о о о о о о о о о о о о о о о о	ame. rfa. sh
	ин и и и и и и и и и и и и и и и и и и и	bed leog coce ing tra
		ng vic gggg wir Dust out
	и соинососоносоколоси сосокосокососи сососи и сосососи и сососокоси и сососокоси и сосососи и сососи и сососи и с и сососососи и сососокоси и сосососи и сосососи и сосососи и сососокоси и сосососи и сосососи и сососи и сосос и сососососи и сососокоси и сосососи и сосососи и сосососи и сососокоси и сосососи и сосососи и сососи и сосос и сососососи и сососокоси и сосососи и сосососи и сосососи и сососокоси и сосососи и сосососи и сосососи и сосо и сосососи и сососокоси и сососокоси и сососокоси и сососокоси и сососокоси и сосососи и сосососи и сососокоси и сосососи и сососокоси и сососи и сососи и сососокоси и сососи и сососи и сососи и сососи и сососи и сососи и с и сососи и br>и сососи и с и сососи и сосо	ing ing ing ing ing ing ing ing
	80400004000400000000000000000000000000	Sard Makin Makin Macuruy Macuruy Matcash Mipin Mipin Mipin
	40000000000000000000000000000000000000	<u>^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ </u>
	ососососососососососососососососососос	L G H G U H U K U F K U
	90000000000000000000000000000000000000	
	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	ght cd
	00000000000000000000000000000000000000	
ΑS	1130 1130	
a	0 1 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
Е	* • • • • • • • • • • • • • • • • • • •	idministration in the second s
F	<pre></pre>	Trea Trea Trea Trea
I S	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	a a a a a a a a a a a a a a a a a a a
s	/0000000000000000000000000000000000000	amph amph amph amph amph amph amph amph
ΓV		philippine
с		and shaked and shaked
Ξ		Treason Fiddo Friddo Treason T
Р		
Μ		rying ry
ΧV	5 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	Run Sit Sit Sit Sit Sta Sta Sta Mal Wal Wal Wal Wal Wal Wal Unk
Э	N	· · · · · · · · · · · · · · · · · · ·
	* • • • • • • • • • • • • • • • • • • •	obe il <u>, &gt;u <u u="" variante<=""></u></u>
	然り 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
	00000000000000000000000000000000000000	
	······································	
	x0000000000000000000000000000000000000	
	и и и и и и и и и и и и и и и и и и и	E E
		pm pm spm ill.
	жоосососоцо4чщоосососончосниесосокисососоночно 	e 60r 100p 1330s 100r 100r 130s
	»	ant erat Cole Rowing Rowing
	но о о о о о о о о о о о о о о о о о о	Phart Phart Cyc. Cyc. Phart Cyc. Phart
	77 000100000000000000000000000000000000	- and a state
	анонано <sup>й</sup> иоооооооноооооноооообоонооообоонооонооноо	ttink ttink ght rd ght lit ligh harc mode filligh harc illigh harc illigh harc illigh harc
		A P P P P P P P P P P P P P P P P P P P
	жоосуща со со	ght ght ght lls cycc cyc cyc cyc cyc cyc cyc cyc cyc c
		wei wei cur cur cur e e e e e e
	10 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	Ach Ach Ach Ach Ach Ach Ach Ach Ach Ach
	и и и и и и и и и и и и и и и и и и и	Ben Beiter Beite
	4 0 H H U 4 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
		N NOLEHOLHOLHORHORHOR

Figure A6-8: Confusion Matrix for C4.5 Classifier using the *ACFFTPeaks* feature computed per sensor and evaluated in a subject independent manner.

	LEBAC LAN-G-MO	
	武長でしますのはまいだよ対対のりの我の不いで読みてきく!( ト きあつけやすうはようだよれれつかびするち	
	1 1 1 1 1 1 1 1 1 1 1 1 1 1	
	8 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
	2 2 2 2 2 2 2 2 2 2 2 2 2 2	
	24 24 24 24 21 24 21 24 21 21 21 21 21 21 21 22 21 21 21 22 22	
	$\begin{smallmatrix} & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & $	
	00000000000000000000000000000000000000	
	4112 4112	
	10000000000000000000000000000000000000	
		ock ssee
	23 23 23 23 23 23 23 23 23 23	d urfa ws ash ash
	44000000000000000000000000000000000000	deo deo roccircoc vdo ndo tin
	4 1 1 1 1 1 1 1 1 1 1 1 1 1	outt
		den hing ping cking king ching ching ing
	90000000000000000000000000000000000000	Gar Mop Scar Scar Scar Scar Mop Mat Scar Mar Mar Mar Mar Mar Scar
	10000000000000000000000000000000000000	^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^
	80000000000000000000000000000000000000	00404-1240202404
	2000 2000 2000 2000 2000 2000 2000 200	
	40000000000000000000000000000000000000	ght rd
		A 11 4
S V	a 103 103 103 103 103 103 103 103	
Q		a', a'a'a'a'
Ε		admi admi admi admi admi
Ξ.		
S		
A S		3 mpi
Г		s lilli
C	M N N N N N N N N N N N N N N N N N N N	adm get adm adm eric eric
LE	X = = = = = = = = = = = = = = = = = = =	Tree cree cree
Ч	20070000000000000000000000000000000000	
M	× × × × × × × × × × × × × × × × × × ×	nin ttin ttin kin kin kin kin kin now how
X	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	Run Sit Sit Sit Sit Sit Mal Wal Wal Wal Wal Uuwku Kne Carkton Coir
Ξ	τ 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	^
	N 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	ODP (  ) > - < - GRXSCANN
	жароороороороооообренееского чар 1000 нарооооооооосоос 	
	₩₽₽₽₩₽₽₽₽₽₽₽₽₩₽₽₽₽₽₽₽₽₽₽₽₩₽₽₽₽₽₽₽₽₽₽₽	
	• • • • • • • • • • • • • • • • • • •	
	M 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
	x	u u dis
	₩ 0 0 0 0 0 4 0 m 4 0 4 10 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	ran ran ran ran ran ran ran ran ran ran
	жааа алаа алаа алаа алаа алаа алаа алаа	eadro 2000 2000 2000 2000 2000 2000 2000 200
	4 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	ht bt ccle ccle rre Non Ron
	н 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	har nod Cyc Row Cyc
	H H H H H H H H H H H H H H H H H H H	99
		ttin ttin ght noder node
	7 215 215 11 11 11 11 0 0 0 0 0 0 0 0 0 0 0 0 0	
	800 100 100 100 100 100 100 100 100 100	ght ght ght lls lls cyc cyc cyc cyc cyc cyc cyc cyc cyc cy
		wei wei dog curr
	215 215 15 15 15 15 15 15 15 15 15 15 15 15 1	Ach Ach Ach Ach Ach Ach Ach Ach Ach Ach
	ли предоктово с с с с с с с с с с с с с с с с с с	Beneral Strain S
	4 4 1 4 4 0 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7	
		NHORMORNORHZZOHOR

Figure A6-9: Confusion Matrix for C4.5 Classifier using the *ACAbsArea and DCArea* features computed per sensor and evaluated in a subject independent manner.



Figure A6-10: Confusion Matrix for C4.5 Classifier using the *MaxAcceleration* feature set computed per sensor and evaluated in a subject independent manner.

	C M H C H Z H C M H C M H C M H C M H C M H C M H C M H C M H C M H C M H C M H C M H C M H C M H C M H C M H C	
	みちつひはずられまえたが対してなれるでいず減火でのエペート で きわっちゃそうれょうたもmmopkkk	
	10000000000000000000000000000000000000	
	0 8 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
	1 1 1 1 1 1 1 1 1 1 1 1 1 1	
	00000000000000000000000000000000000000	
	# 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
	и в в в в в в в в в в в в в в в в в в в	. X
	жаааааааааааааааааааааааааааааааааааа	h bloo
	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	eoga sur sur cer dows tras
	4 0 0 1 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	d vid grad aro win ust
		enir ng di di di di di di di di di di di di di
		ard lakin lopp copp copp copp copp copp copp acculation
	4 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
	● 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	ひもま ダね う リメー 加れ つ ひ ひょうち
	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	ght rd
	н н н н н н н н н н н н н н н н н н н	
A S	123 123 123 14 14 14 14 155 14 155 14 155 14 155 155	
D	· · · · · · · · · · · · · · · · · · ·	a aaaa a
ΙE		admi admi admi admi
F		Tree 3
s	×0000000000000000000000000000000000000	ини ини ини ини ини ини ини ини ини ини
ΥS	 ≜au	3322 1 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2
T	й	e dillite
3	ноноссость соссать сос И И И И И И И И И И И И И И И И И И И	addin addin
Г	х 1	Pescint Truck and the second s
M P	манана со со со со со со со со со со со со со	ng ng ss dis dis dis dis
A	и и и и и и и и и и и и и и и и и и и	1 1 1 1 1 1 1 1 1 1 1 1 1 1
Ξ		N N N N N N N N N N N N N N N N N N N
	4 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	CDB 1
	4 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	
	$\begin{smallmatrix} \mathbf{a} \\ \mathbf{a} $	
	000004000000000000000000000000000000000	
		e <sup>E</sup> o,o
	0 /	spm 11111
	жооччооооооооооооооооооооооооооооооооо	e 100pn 100pn 30sp 30sp 30sp 30sp 10g
		Cost Cost Cost Cost Cost Cost Cost Cost
	но о о о о о о о о о о о о о о о о о о	hard model herovo herovo herovo
	жноосособоссососососососососососососососо	Amatri sititi si alla si a
	10000000000000000000000000000000000000	ttink ttink ght rd ght rrunk rrunk hard hard hard illigh hard illigh hard illigh hard illigh thard thard illigh thard illi
	**************************************	
		ght ght ght ls ls ls cyc cyc cyc cyc cyc cyc cyc cyc cyc cy
	11200000000000000000000000000000000000	weele weele weele weele weele weele weele weele weele weele weele weele weele weele weele weele weele weele we
		nnch nch nch nch nch nch nch nch nch nch
		R. S. S. C. C. C. C. B. F. F. B.
	× 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	

Figure A6-11: Confusion Matrix for C4.5 Classifier using the *ACAbsArea*, *DCArea*, and *ACFFTPeaks* computed per sensor and evaluated in a subject independent manner.

	O M H O H V A L O H O H O H O H O H O H O H O H O H O	
	しのののののこのののののののののののののでののののののののののののののでものののののののの	
	8 <sup>0</sup> 4 00 00000000000000000000000000000000	
	۵ ۵ ۵ م	
	б. н нн Ф В	
	10000000000000000000000000000000000000	~
		loc
	× × × × × × × × × × × × × × × × × × ×	d surf surf seri ng ng rash
		=_be jrocc trv ccur stin c_tr
	90000000000000000000000000000000000000	fing d the d
		den phing phing ching ching ching ching ching
	20000000000000000000000000000000000000	Markov Mop Mop Mop Mop Markov Markov Markov Markov Markov Markov Markov Markov Markov Markov Markov Markov Markov Markov Mop Mop Mop Mop Mop Mop Mop Mop Mop Mop
	90000000000000000000000000000000000000	<u>^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ </u>
	4 8000000000000000000000000000000000000	A S R A D D D D F L L L L L D D H D D D D D D D D D D D D
	10000000000000000000000000000000000000	
	лооососсоссоссоссоссоссоссоссос ф - 4 и 8 и но н и 4 о о о е и 9 и 1 и 1 и 4 о о о е и 4 и 8 и но н и 4 о о о е и 4 и 8 и 1 и 1 и 4 о о о е и 4 и 8 и 1 и 1 и 4 о о о е и 4 и 8 и 1 и 1 и 1 и 4 о о о е и и 4 и 8 и 1 и 1 и 1 и 1 и 1 и 1 и 1 и 1 и 1	ght rd
	ФО000000000000000000000000000000000000	
A S	1222 132 132 133 133 133 133 133 133 133	
0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
Ξ	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	dm i dm i dm i dm i dm i dm i dm i dm i
ΕI	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	rea 111
SI	- 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
Ś	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	mph mph mph mph mph mph mph mph mph mph
P 7	и 1 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	1 3 3 3 2 1 2
5	0 0 0 1 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	dmil 6 mil 1 1 mil 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1
Ξ	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	illi ildg cendo reac reac socei
P.L	8 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	The second secon
Σ	80 H 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	ring ring ring ring ring ring
V	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	unn itt itt tai tai tai tai tai kai kai kai kai kai kai kai kai kai k
EX	H 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
	4 4 9 1 9 1 9 1 9 1 9 1 9 1 9 1 9 1	oba √   → − − − − − − − − − − − − − − − − − −
	$\begin{smallmatrix} & A \\ A$	
	4 6 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	
	00000444400000000000000000000000000000	
	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
	и и и и и и и и и и и и и и и и и и и	
	м м м м м м м м м м м м м м м м м м м	e e
	x o o o o o o o o o o o o o o o o o o o	rpm 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
	8 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	rpm rpm sspm sspm sspm sspm sspm sspm ss
	x	1 30 80 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1
	4 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	dera dera vcle vcle vcle vcle r
	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	ph Ph Ph
	а а а а а а а а а а а а а а	igigi igigi ichte i i i i i i i i i i i i i i i i i i i
	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	ftir ftir ftir ftir ftir ftir har har har har har har har har har ha
	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	adaga eeeee
	M + 0 0 4 N 6 0 + 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	ght ght ght sins cyver c
	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	weil weil weil weil weil weil weil
	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	nch nch nch nch ncep ncep ncep ncep ncep ncep ncep ncep
	м # 4 С 8 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	Been Beiting Brith
	4 6 0 8 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
		"HOUNHQIHUXUXOUOK

Figure A6-12: Confusion Matrix for C4.5 Classifier using the ACAbsAreas, ACRange, ACVar, ACFFTPeaks, ACCorr, ACMCR, and ACPitch computed per sensor and evaluated in a subject independent manner.
# **Appendix A7: Activity Recognition Using Heart Rate Data**

	0 × - 0 - × - 0 - × 0	
	丸谷じひまずの日まびれ北対りやなお点でワママメンホートーペーで ユロロはのぎびねょうをうれれらかべます	
	19日(19日)(19日)(19日)(19日)(19日)(19日)(19日)(1	
	R 0 0 0 0 0 0 0 0 4 約0 0 0 0 0 0 0 0 0 0	
	9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	
	911000000000000000000040040400000000000	
	200000040000000000000000000000000000000	
	Earlounau00000000000000000000000000000000000	, , , , , , , , , , , , , , , , , , ,
		ace es loc)
		ed surf cerif ng b ng ng rash
		e be grou TV t_t_t
	Точчиванооовимаровововоаченина и маниранирановимановиов М. М. М. М. М. М. М. М. М. М. М. М. М. М	19 19 19 19 10 10 10 10 10 10 10 10 10 10 10
	роооооооооооооооооооооооооооооооооооо	oni ayin ayin ayin ackin bin kin kin kin kin
	そののですすのののであるのののするののないのであるのでので、またのののの話をの話がのままでののです。 またまたで、「「」」」では、「」」」	T A MARANA A A A A A A A A A A A A A A A A A
	●18ヵヵ120010000000000000000000000000000000	о өч өд н п х н ё д о д б я ю
	2000 000 000 000 000 000 000 000 000 00	
	00000000000000000000000000000000000000	t a
	10040000000000000000000000000000000000	harc
s	「「「」」」」」」」」」」」」」」」」」」」」」」」」」」」」」」」」」」	
	· 。。。。。。。。。。。。。。。。。。。。。。。。。。。。。。。。。。。。	
Ξ	90000000000000000000000000000000000000	dmil dmil dmil dmili dmil dmil dmil
FI	<pre>&lt; \$</pre>	treaa aaa Treaa aa
S I	- 1 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	
s	、しばはしはは、4000%がのでののの00000000%がのかのではのはのためのかがののようです。	amph amph amph amph amph amph amph amph
LA	-084000800444404648495568555555555555555555555555555555	s 1 1111 staine
U	10000000000000000000000000000000000000	admi gget hi admi admi admi admi admi admi admi a
E	X48048884480688048449484844848448494484044840	Tree Fride Tree Free Free
-	本日日日日日日日日日日日日日日日日日日日日日日日日日日日日日日日日日日日日	
M	>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>	nin tin tin kin kin kin kin kin kin dir seliu tin tin tin tin tin tin tin tin tin tin
x	20000000000000000000000000000000000000	Rur Rur Sitt Rur Ster Ster Wal Wal Wal Wal Van Car Car Car Car Car Car Car Car Car Car
ы	HBH400004801004044481000000000000000000000	
	00000000000000000000000000000000000000	
	ティット ちゅうしゅう (1) - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 -	
	н н с с с с с с с с с с с с с с с с с с	
	8 8 8 8 7 8 7 8 7 8 8 8 8 8 8 9 8 8 7 9 8 9 8	e
	и 200004100410000000000000000000000000000	nd nd nd nd nd nd nd nd nd nd nd nd nd n
	。 また また また で、 で、 で、 で、 で、 で、 で、 で、 で、 で、	crpm 00rpm 00rpm 330spm 05pm 05pm 05pm
	*8;5;1;1;4,8,9,9,9,9,9,9,9,9,9,9,9,9,9,9,9,9,9,9,	real real real 3
	20000000000000000000000000000000000000	Province of the stand
	и и и и и и и и и и и и и и и и и и и	
	00000000000000000000000000000000000000	ing d d bh bh bh t unch t unch t unch t unch t ight t ight l 1 4 r u bh t u u b h d d d d d d d d d d d d d d d d d d
	PURSURSCACHAADD44000000000000000000000000000000000	diministration of the second s
	NG6643602 N N 64 L 90 N N 04 M 0000 N N 00 6 N N 90 0000 N N 000 4 000000 H N N H	httl httl httl httl httl httl httl httl
	おろ畑の路路は41002280001400000000000000000000000000000	Feigle Fe
	しっぽが回っ心えるのまちの目ののつちのもののののううちちょうものできょうないのできょうしょう しょうちょう ちょうちょう ちょうちょう ちょうちょう ちょうちょう ちょうちょう ちょうちょう ちょうちょう しょうしょう	ch.v. ch.v. ep.c.v.v. listh ling lling lling lling dlling lling lling dlling lling lling lling lling
	最近的認知時点であったちはロワーチャーキューロックを回路であったこのです。 いうこう キュリのよう ちょうしょう しょう こう	Ben Ben Ber Bic Bic Cal: Cyc Cyc Cyc Rowi Rowi Rowi Rowi Runr
	«Параии»	^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^ ^
		A R O R G R R R R R R R R R R R R R R R R

Figure A7-1: Confusion Matrix for C4.5 Classifier using the *ScaledHR* feature and evaluated in a subject dependent manner without the garbage class (unknown) when the feature is computed over windows of 5.6s.



Figure A7-2: Confusion Matrix for C4.5 Classifier using the *ScaledHR* feature and evaluated in a subject independent manner without the garbage class (unknown) when the feature is computed over windows of 5.6s.

	0	ы	-	Ċ	-	z	r J		Г	v	<b>n</b> 13		2	
														1
	4 8 0 0 8	64 (B 14	нья	122	0 4 0	នេះគេ៖	N N N N N	N /	<  - a	, d o i		4 11 3	часоруна	
		040	000	000	0 1 0	0000	2 8 8 9 1 1 7 8 8 9 7	0 0 1 0 0		12 25	35 14 35 14	6 m c	11 11 11 11 12 13 13 13 14 12 12 12 12 12 12 12 12 12 12 12 12 12	
	400000	000	000	000	000	000	0000	0000	0040		5 0 0 1	0 0 ž		
	800000	400	000	000	000	000,	1000		0004	38	9 9 9 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3	5 1 G	8-12-28-18	
	A00000	000	000	0 11 0	0 r r	000,	e	0000	0014	61 B	* 2 * 0 *	52 0	1 2 3 7 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4	
	0 11 10 10 00	000	000	008	000	0003	4 0 0 0			: ° :	54007	o o ;	22400	
	E04101	0 0 0	000	000	000	000,	101010	r H O O		10 22	28 28 68 68	28	349 0 25 1 349 0 2	
	E00000	000	000	000	000	n o o o	5 0 5 C	58 58 79	8001	001	0 4 0 0 0	000	312 0 0 0 1 31 0 0	
	100100	0 4 -	000	000	004	0000	1010	0 0 0 0	0000	12	28 28 13	6 F 0	415 11 25 32 32 32 33 33 33	C K C K C K C K
	201 10 10 10 00	000	008	000	000	0008	19800	0000	0000	401	90009	1 8 3	314 93400	d urfa erie d bl ash
		000	000	000	01-0		100120	0000	000-	1°:	12 127 127 127	48 251	23 23 23 23 24 25 25 25 25 25 25 25 25 25 25 25 25 25	a stin stin truc
	440884	° ° ;	000	000	15 20	0000	0044	0000	0004	900	122602	35	0.501.508	out la la la la la la la la la la la la la
	200000	04-	000	000	004		4784	0000	0040	161	28 16 28 16 298 16	26	50\$15405	nin nin vyin vyin ving king chij nin ving ting
	004000	000		003	000						10000	00°	2005	Irc Mor Pla Scr Scr Scr Scr Mar War War War War War tak
	00000-			000	040	- 200		0 9 0 0		2 - 1	14506	121		
		400	000	000	000	000-		*000	000	- 61	2 H 0 M M	48°	480444000	0 0 H 0 H 9 H 9 H 10 H 0 H 0 H 0
	0000m0	*00		000	E 10 H	0000				9 9	8 H C H N	840 H - 6	0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 -	
	200-00				~ • • • •	,	00 0 0 0	1000		12 12	0 0 0	8 297	N N N N N N N N N N N N N N N N N N N	igh.
s	*****				r 4 4	n e e	0048	0 8 5 5 1	6 - 0 2	000		0 11 0	9000000	
•				000				0000			0 4 0 0 0			
D S	100000		0 10			<u>n</u> o o o	0-10-1	02 1 0 02 1 0	010 01 01 01	000	04000		000000000	111 ja 111 ja 111 ja 111 ja 111 ja
Ξ	< 0 0 0 0 0	000	0 10 0	000	000		0000	1210	1 0 0 1	000	0 0 0 0 0		10010030	eadd eadd eadd eadd
=	-00000	000	0 1 0	0 1 0	000	r o o d	0000	15400	5 17 º 5	000			04000000	
s	/00000	000	000	0 1 0	000	a o o o	0000	147	00010	000		000	1360000	uph
<b>V</b>	-00000	00-	000		000	8000	23 12 0 3	516 71	8 9 0 S	000	0 1 1 4 4 0	0 11 0	N 800 H H O 8	1 - 5 - 5 - 5 - 5 - 5 - 5 - 5 - 5 - 5 - 5
5	10 0 0 0 0	000	000	000	000	0000	14 10 11 10	354	0 0 0 0	000	N 0 1 1 0	0 11 -	9 4 9 4 9 4 0 H	in in in the second sec
ы	× 0 0 0 0 0	000	000	400	ei 10 ei	0400	320 2 0 1 0	24 24 20	17 2	0 - 0	0 1 1 1 0 0	m r- c	31008101	scence sc
1	X 0 0 0 0 0	0 8 6	000	000	0 ji 4	0000	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	4100	8 8 0 8	000	04000	n n c	8800140	
Σ	800HNN	0 8 0	000	000	0 9 8	0000	3611	4000		5.43	09 - 18 e	9 <del>4</del> 8	001100990	rng rng rng rng rng rng rng rng rng rng
X	>	0 10 0	000	000	000	000-	286	0 9 0 0	0000	0 -1 -		0 11 11	00040000	Statttun Saunn Saunn Sautttun Sauttun Soorkes
E	200m00		000	000	000	0000	1200	0 0 0 0 0 0 0 0		0 10 1	04084	008	1008000	· · · · · · · · · · · · · · · · · · ·
						1988								STD ALANYS/C-
	#00000					981 981		0040	0 0 0 0				0.0000000	
	000000					0000 8000			0000				00000000	
	A 0 0 0 0 H	n 4 C		000	55 17 19 19 19 19		00004				00400		00004000	
	000000	moc		000	82 57 59 62 59 59					000			00400000	
	X0 + # 0 0	040	000	005	000	0000	10000	0000		000	00040		00050000	
	Σοοοοο	000	129	222 177 0 20	000	0000		0000		000		000		
	400000	000	133	221	007	0000	0000	00 11 0	0 11 0 0	000	0 1 0 0 0			an Drpm Box Spm Jspm ill
	*00000	0 % 0	10	- 90	000	0000	* • • • •			000		0 1 0		eadmin. 1000 1000 1000 1000 1000 1000 1000 10
	500000	000	557 9	5 4 0	000	0040	0000	0004	0 11 0 0	000		0 - 0		rd ght vcle cy wing Tro
	носссо	000	187	96 176 0	000	0000		0000	0000	001	00004	000		phene CCC
	#00000	0 F 998	000	000	6 6 F	0000	N H O O	0000	0000	000	0 10 10 10 10 10 10 10 10 10 10 10 10 10	* o c	100H0000	land the second
	804168	65 5 F	000	001	10 10	0000	00400	0000	0000			o o -	00444400	ftii ffii ffii ffii dender fii fii fii fii fii fii fii fii fii fi
	#00085	m o o	000	004	0 4 4		0400	0000		000	00400	000	0040000	contraction of the second seco
	800095	136	000	000	000	0000		0000				- 0 0		Rown C.C. C.
	1981 1982 1981	0 0 F	000	000	m 0 %	0000	00400	0000				N O C	0000000000	hge current of the second of t
	140 C	26 8 28 28	000	000	000	0000		0000	0000	- 21	00000		00400000	encl encl ficer ficer alis alis vycli vycli vycli vycli vycli un hin vycli i encl alis encli alis encli encli alis encli alis encli alis encli alis encli alis encli alis encli alis encli alis encli alis encli alis encli alis encli alis encli encli encli encli encli encli alis encli encoli encli encli encli encli encli encli encli encli
	1476 1476 1476 1476 1476	m ex c					0000					0 10 0		
	41.089.00	000		000	000	0000	, 4000	0000		000				NOTO NALKUHHGRADO WA

Figure A7-3: Confusion Matrix for C4.5 Classifier using the Invariant reduced and ScaledHR features evaluated in a subject dependent manner without the garbage class (unknown) when the all the features are computed over windows of 5.6s.

	0 M - 0 - Z 4 J 4 M M J					
	きりつりつつこうこうようのようのこうのののののちろえてかておりつつても目のものこちちゅの用いてまてとろうま おおしむををはだい がんしかのか ひみのていい 特別シンシントイーン ホンゥウのようひょうれ オロル かつどう					
	そうりつうのののこうののここののので、こうのころのののことをなって、ほうのこののここのので、					
	фраароороосоососсинанаюаароосикаефг/косбилефер/алег и иал в ина е и е ни иен					
	00040000000000000000000000000000000000					
		~				
		loc				
		ogan surf cerif ng rash				
		e be grou t tu t tindo				
		a d b b d d d d d d d d d d d d d d d d				
	THEERSHY CROWN COUNCESS CONTRACTION CONTRACTICON CONTRACTION CONTRACTICON C	pinn pinn pinn pinn pinn pinn pinn pinn				
	13/2008/24408/001240/2000000/200000000000000000000000	Irc Mop Plas Scr Scr Scr Scr Mop Na Scr Mor La Mut La Mut La Mut				
		<u>^</u> <u>^</u> <u>^</u> <u>^</u> <u>^</u> <u>^</u> <u>^</u> <u>^</u> <u>^</u> <u>^</u>				
		D D H D Z H D X H E Z O D D H D				
	0 0 0 1 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0					
	000200400020000000000000000000000000000	ght rd				
	わっよるり 0 0 1 0 1 0 1 0 1 0 0 0 0 0 0 0 0 1 1 2 0 0 0 0	ha li				
A S	<pre>s = 1 = 2 = 2 = 2 = 2 = 2 = 2 = 2 = 2 = 2</pre>					
a	· 0004444000400000000619884664800066488265					
Ξ	1111 1111004040000000000000000000000000	비명 전 전 전 전 전 전 전 전 전 전 전 전 전 전 전 전 전 전 전				
F	< 001000440004801000000000000000000000000					
I S	- 9 8 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9					
s	1 1 1 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	mph mph mph mph mph				
Γ	1 1 2 1 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	111_1 111_1 111_3 111_3 111_3 111_3 111_3 111_3 111_3 111_3 111_3 111_3 111_3 111_5 1111_5 1111_5 111_5 111_5 111_5 111_5 111_5 111_5 111_5				
U	202200000000000000000000000000000000000	da da da da da da da da da da da da da d				
ы	メュ は ゅ ほ ぎょう べ ご は ご う う さ る る く く く こ く く ご う ひ ご う む う ご う う う う う う う う う う う う う う う	rrea rea cendo rea rea rea rea rea rea				
P L	X 0 0 0 8 4 4 8 8 6 6 6 6 7 8 9 9 7 0 0 7 0 0 9 9 4 7 7 8 6 7 8 9 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7					
Σ	1999年1999年1999年1999年1999年1999年1999年199	children childr				
Y	▶ 0 4 4 8 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	unn ittt ittt ttai ttai ttai ttai ttai tai				
EX	00000000000000000000000000000000000000	~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~				
,	1 0 0 0 0 1 0 0 0 0 0 0 0 0 0 0 0 0 0 0	NHDVRXYN-/				
	10000000000000000000000000000000000000					
	秋日~~日1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1					
	04801888999941409845889990091008680400000404040000000000000000000000					
	4 n n n n n n n n n n n n n n n n n n n					
	00010000000000000000000000000000000000					
	200 110 110 110 110 110 110 110 110 110					
	X888894488998787878888800008409409400440004484844000000	e di eso				
	1 2 1 1 2 1 2 2 2 2 2 2 2 2 2 2 2 2 2 2	Da Crpm spm 0 spm 1 il				
	и али али али али али али али али али ал	ead wwing cole				
	9 8 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	rd fight /cle/ /cle/ /cle/ /cle/ /cle/ /cle/				
	てののこの50回回回の100回回日の111000004041416464000111160004010111000001	Philipsi 2017				
	第2500010000~10000000000000000000000000000	dg dg hht hht era era era				
	「このこれのことのないない」ののここので、このではないないので、なっている」で、「このころの」で、「この」で、「この」、「この」、「この」、「この」、「」、「」、「」、「」、「」、」、「」、	Ftin Ftin Situn Jiggert Par Jiggert Par Jiggert Ftin Jiggert Ftin Ftin Ftin Ftin Ftin Ftin Ftin Fti				
	「本本のこのないないないです。」「「「」」」」」」」」」」」」」」」」」」」」」」」」」」」」」」」」」	and a second sec				
	第44441995325555555555555555555555555555555555	ghti ghti ls: Cycycy Rowi Rowi Rowi Rowi				
	10000000000000000000000000000000000000	wei wei dowr dowr dowr dowr dowr dowr dowr dowr				
	0.0.4.0.0.0.0.0.0.0.0.0.0.0.0.0.0.0.0.0	hoch hoch cep cep cep cep cep cep cep cep cep cep				
	* • • • • • • • • • • • • • • • • • • •	Ber Ber Ber Bitc Bitc Cal Bitc Cal Bitc Cal Frow Row Row Row Row Row				
	С 0 1 1 0 1 1 1 1 1 0 0 1 4 4 0 1 1 1 1 1	111111				
		KUUUHFQHHJMHXOFQK				

Figure A7-4: Confusion Matrix for C4.5 Classifier using the *Invariant reduced* and *ScaledHR* features evaluated in a subject independent manner without the garbage class (*unknown*) when the all the features are computed over windows of 5.6s.

	Subject Dependent			Subject Independent			
Activity	TP Rate	FP Rate	F-Measure	TP Rate	FP Rate	F-Measure	
Bench weight lifting - hard	$8.74 \pm 9.83$	$1.00 \pm 0.50$	$8.85 \pm 9.93$	$0.00 \pm 0.00$	$0.57 \pm 0.28$	$0.00 \pm 0.00$	
Bench weight lifting - light	$29.60 \pm 23.90$	$1.86 \pm 1.06$	$25.84 \pm 18.34$	$2.15 \pm 3.45$	$1.77 \pm 0.59$	$2.27 \pm 3.55$	
Bench weight lifting - moderate	$12.32 \pm 12.36$	$1.10 \pm 0.66$	$13.80 \pm 12.81$	$0.82 \pm 2.21$	$1.13 \pm 0.59$	$1.01 \pm 2.55$	
Bicep curls - hard	$38.33 \pm 28.41$	$1.59 \pm 0.86$	35.79 ± 27.13	$2.59 \pm 4.31$	$0.94 \pm 0.43$	$3.35 \pm 5.29$	
Bicep curls - light	$28.19 \pm 17.48$	$2.42 \pm 1.59$	$24.88 \pm 14.40$	$1.13 \pm 2.17$	$1.70 \pm 0.65$	$1.44 \pm 2.79$	
Bicep curls - moderate	$16.58 \pm 10.40$	$1.55 \pm 0.53$	$17.36 \pm 10.34$	1.00 + 2.40	$0.89 \pm 0.35$	1.43 + 3.33	
Calisthenics - Crunches	$27.19 \pm 30.80$	$1.00 \pm 0.00$ $1.11 \pm 0.57$	$24.94 \pm 26.87$	$1.00 \pm 2.10$ $1.02 \pm 2.12$	$0.82 \pm 0.41$	$1.13 \pm 0.00$ $1.37 \pm 0.99$	
Calisthenics - Sit ups	$28.04 \pm 27.09$	$1.27 \pm 0.85$	$26.36 \pm 19.83$	1.02 = 2.12 1.88 + 3.23	$1.12 \pm 0.97$	$2.74 \pm 4.55$	
Cycling - Cycle hard - Cycle 80rpm	$65.06 \pm 33.80$	$0.70 \pm 0.87$	$65.11 \pm 30.76$	$1.00 \pm 3.09$ $1.07 \pm 3.09$	$0.78 \pm 0.30$	$1.51 \pm 4.14$	
Cycling - Cycle light - Cycle 100rpm	$3850 \pm 2639$	$1.06 \pm 0.65$	$39.47 \pm 24.51$	$9.90 \pm 7.38$	$2.15 \pm 1.22$	$9.92 \pm 7.72$	
Cycling - Cycle light - Cycle 60rpm	$66.59 \pm 20.05$	$2.00 \pm 0.87$	$54.94 \pm 20.76$	8 67 ± 7.81	$3.05 \pm 1.22$	$7.74 \pm 7.12$	
Cycling Cycle light Cycle 80rpm	$44.42 \pm 20.08$	$1.72 \pm 0.88$	$34.94 \pm 20.10$	$6.07 \pm 7.01$	$3.03 \pm 1.00$ 2.79 ± 1.56	$6.18 \pm 4.83$	
Cycling - Cycle light - Cycle 801pm	$44.42 \pm 20.08$	$1.72 \pm 0.00$ $1.70 \pm 1.17$	$41.07 \pm 10.17$	$0.90 \pm 5.41$	$2.79 \pm 1.30$ 1.47 ± 0.84	$0.18 \pm 4.83$	
80rpm	$40.00 \pm 27.38$	1.70 ± 1.17	$40.55 \pm 25.60$	$4.34 \pm 3.40$	$1.47 \pm 0.04$	$5.04 \pm 0.01$	
L ving down	87 53 + 9 39	3 12 + 3 84	78 36 + 15 96	88 44 + 19 64	4 16 + 4 18	73 02 + 19 34	
Rowing - Rowing hard - Rowing	$4450 \pm 3483$	$0.92 \pm 0.74$	$44 \ 43 \ + \ 33 \ 97$	$0.98 \pm 2.29$	$1.34 \pm 1.05$	$133 \pm 314$	
30spm	11.50 ± 51.05	0.72 ± 0.74	11.15 ± 55.57	0.90 ± 2.29	1.54 ± 1.05	1.55 ± 5.14	
Rowing - Rowing light - Rowing	$36.24\pm25.92$	$1.46\pm0.74$	$34.81\pm20.36$	6.84 ± 12.09	1.81 ± 1.38	6.68 ± 10.34	
Rowing - Rowing moderate - Rowing	47.04 + 17.90	1 69 + 1 06	43 86 + 15 72	$10.25 \pm 19.92$	178 + 148	8 24 + 11 61	
30snm	47.04 ± 17.90	1.09 ± 1.00	+5.00 ± 15.72	10.25 ± 19.92	1.70 ± 1.40	0.24 ± 11.01	
Running - Treadmill 4mph -	29.05 + 19.31	$1.25 \pm 0.74$	32.02 + 17.16	$625 \pm 593$	$1.64 \pm 0.79$	$7.00 \pm 6.49$	
Treadmill 0							
Running - Treadmill 5mph -	$56.20 \pm 26.75$	$1.23\pm0.75$	$52.95 \pm 22.92$	$53.61 \pm 20.51$	$2.80 \pm 2.61$	$44.32 \pm 20.07$	
Treadmill 0							
Running - Treadmill 6mph -	$64.14 \pm 37.51$	$0.64\pm0.67$	$62.05 \pm 35.34$	$52.88 \pm 32.31$	$1.98 \pm 2.91$	$44.39 \pm 27.77$	
Treadmill 0		0.00 0.50	00.05 01.50	2.00 4.00	1.05 0.55	2.50 5.01	
Sitting	$22.22 \pm 22.02$	$0.89 \pm 0.50$	$23.05 \pm 21.58$	$2.08 \pm 4.39$	$1.06 \pm 0.65$	$2.79 \pm 5.94$	
Sitting - Fidget feet legs	$31.44 \pm 20.30$	$1.10 \pm 0.42$	$30.83 \pm 18.09$	$3.28 \pm 4.08$	$1.38 \pm 0.86$	$4.17 \pm 5.48$	
Sitting - Fidget hands arms	$33.48 \pm 30.64$	$1.66 \pm 0.78$	$27.06 \pm 22.24$	$1.40 \pm 3.57$	$1.30 \pm 0.67$	$1.89 \pm 4.91$	
Stairs - Ascend stairs	$2.89 \pm 3.92$	$0.96 \pm 0.40$	3.78 ± 4.95	$2.65 \pm 3.61$	$1.58 \pm 0.57$	$3.12 \pm 4.30$	
Stairs - Descend stairs	$11.35 \pm 8.14$	$1.12 \pm 0.67$	$13.84 \pm 9.98$	$2.12 \pm 2.91$	$1.79 \pm 0.68$	$2.14 \pm 2.97$	
Standing	$24.87 \pm 25.85$	$1.13 \pm 0.43$	$23.62 \pm 22.84$	$0.73 \pm 1.77$	$1.30 \pm 0.49$	$0.90 \pm 2.21$	
Walking - Treadmill 2mph -	$41.60 \pm 33.11$	$2.24 \pm 0.88$	$36.44 \pm 28.34$	$10.75 \pm 11.59$	$3.24 \pm 1.77$	$9.02 \pm 8.62$	
Treadmill 0							
Walking - Treadmill 3mph -	$55.93 \pm 20.29$	$2.37 \pm 0.65$	$46.54 \pm 17.53$	$7.78 \pm 6.45$	$3.17 \pm 1.42$	$7.14 \pm 5.39$	
Treadmill 0							
Walking - Treadmill 3mph -	$54.20 \pm 29.34$	$2.38 \pm 1.02$	$44.60 \pm 22.21$	$7.23 \pm 5.85$	$2.83 \pm 1.08$	$7.66 \pm 7.01$	
Treadmill 3 - light							
Walking - Treadmill 3mph -	$51.67 \pm 21.17$	$2.25 \pm 1.07$	$45.30 \pm 17.21$	$11.68 \pm 9.66$	$2.50 \pm 1.35$	$12.00 \pm 9.51$	
Treadmill 6 - moderate							
Walking - Treadmill 3mph -	$47.08 \pm 25.48$	$2.18 \pm 1.42$	$42.06 \pm 19.64$	$9.71 \pm 6.19$	$2.51 \pm 1.26$	$10.15 \pm 6.34$	
Treadmill 9 - hard							
kneeling	$26.51 \pm 25.41$	$0.99 \pm 0.40$	$26.63 \pm 21.69$	$1.43 \pm 2.25$	$1.05 \pm 0.48$	$1.93 \pm 3.06$	
Carrying groceries	$27.43 \pm 19.63$	$2.08 \pm 0.96$	$27.99 \pm 20.18$	$2.55 \pm 3.57$	$1.96 \pm 0.70$	$2.88 \pm 4.13$	
Doing dishes	$24.80 \pm 15.94$	$2.86 \pm 1.48$	$22.35 \pm 12.92$	$4.28 \pm 4.48$	$2.94 \pm 1.43$	$3.94 \pm 4.18$	
Gardening	$19.26 \pm 14.88$	$2.11 \pm 1.54$	$19.37 \pm 13.44$	$1.75 \pm 3.08$	$1.04 \pm 0.55$	$2.59 \pm 4.52$	
Ironing	$36.28 \pm 24.77$	$2.79 \pm 0.69$	$31.43 \pm 20.39$	$11.46 \pm 12.58$	$3.00 \pm 1.88$	$9.83 \pm 10.56$	
Making the bed	$12.76 \pm 8.64$	$2.28 \pm 0.90$	$13.59 \pm 8.69$	$2.56 \pm 3.02$	$1.94 \pm 0.76$	$3.11 \pm 3.78$	
Mopping	$21.19 \pm 11.49$	$2.11 \pm 0.81$	$20.70 \pm 9.09$	$2.36 \pm 4.83$	$1.55 \pm 0.75$	$2.89 \pm 5.59$	
Playing videogames	$34.88 \pm 24.73$	$2.12 \pm 1.66$	$33.02 \pm 21.44$	$11.26 \pm 10.44$	$2.07 \pm 1.24$	$12.64 \pm 11.92$	
Scrubbing a surface	$11.54 \pm 8.40$	$2.14\pm0.88$	$11.36 \pm 6.76$	$2.21 \pm 2.88$	$1.19 \pm 0.62$	$2.83 \pm 3.68$	
Stacking groceries	$14.96 \pm 10.26$	$1.71 \pm 0.70$	$15.48\pm9.19$	$1.71 \pm 2.40$	$1.27 \pm 0.63$	$2.01 \pm 2.74$	
Sweeping	$18.52\pm15.97$	$2.08\pm0.83$	$17.95 \pm 14.13$	$3.59\pm3.81$	$1.46\pm0.79$	$4.04 \pm 4.30$	
Typing	$30.93 \pm 24.48$	$1.60 \pm 0.93$	$31.80 \pm 21.01$	$6.97 \pm 6.18$	$2.02 \pm 1.21$	$8.12 \pm 7.04$	
Vacuuming	$20.80\pm16.28$	$2.17 \pm 1.25$	$20.42 \pm 15.47$	$2.30\pm3.12$	$1.66 \pm 1.24$	$2.75\pm3.62$	
Walking around block	$27.40 \pm 21.53$	$2.04\pm0.85$	$27.26\pm20.99$	$2.76\pm3.33$	$1.52\pm0.68$	$3.18\pm3.69$	
Washing windows	$14.70\pm8.78$	$1.78\pm0.82$	$16.72\pm9.25$	$1.80\pm2.58$	$1.38\pm0.52$	$2.36\pm3.39$	
Watching TV	$27.34\pm30.35$	$1.70 \pm 1.60$	$26.21 \pm 28.00$	$5.25\pm5.99$	$1.31\pm0.83$	$7.10\pm8.35$	
Weeding	$14.25 \pm 16.77$	$1.48 \pm 1.02$	$14.34 \pm 13.67$	$0.39\pm0.92$	$0.65 \pm 0.30$	$0.61 \pm 1.46$	
Wiping/Dusting	$22.09 \pm 16.10$	$2.36 \pm 1.69$	$20.85 \pm 11.81$	$4.29 \pm 4.08$	$1.99 \pm 1.15$	$4.51\pm3.91$	
Writing	$48.00\pm31.74$	$2.18 \pm 1.18$	$40.73\pm23.94$	$12.52\pm12.30$	$2.55 \pm 1.17$	$12.80\pm12.75$	
taking out trash	$1675 \pm 11.93$	$1.79 \pm 1.20$	$17.89 \pm 10.75$	$0.28 \pm 0.85$	$0.82 \pm 0.45$	$0.41 \pm 1.28$	

Table A7-1: Subject dependent and subject independent performance results using the C4.5 and the *ScaledHR* feature without considering the garbage class (*unknown* activity) when the feature is computed over windows of 5.6s.

	Subject Dependent			Subject Independent			
Activity	TP Rate	FP Rate	F-Measure	TP Rate	FP Rate	F-Measure	
Bench weight lifting - hard	8.1 ± 9.2	$1.0 \pm 0.5$	$8.4 \pm 9.4$	$0.5 \pm 1.6$	$0.7 \pm 0.4$	$0.6 \pm 2.0$	
Bench weight lifting - light	$28.9 \pm 22.2$	$1.8 \pm 1.0$	$25.6 \pm 17.1$	$3.5 \pm 5.0$	$1.9 \pm 0.9$	$3.2 \pm 4.5$	
Bench weight lifting - moderate	$12.0 \pm 12.2$	$1.0 \pm 0.6$	$12.8 \pm 12.0$	$0.7 \pm 1.6$	$1.0 \pm 0.4$	0.8 ± 2.1	
Bicep curls - hard	$34.3 \pm 29.2$	$1.7 \pm 0.9$	$32.4 \pm 28.6$	$5.7 \pm 20.0$	$1.0 \pm 0.5$	$5.1 \pm 16.0$	
Bicep curls - light	$30.4 \pm 18.4$	$2.6 \pm 1.5$	$25.8 \pm 15.3$	$5.2 \pm 7.4$	$1.9 \pm 0.8$	$5.7 \pm 8.4$	
Bicep curls - moderate	$16.7 \pm 10.8$	$1.5 \pm 0.5$	$17.8 \pm 10.6$	$2.8 \pm 5.6$	$1.1 \pm 0.5$	$3.3 \pm 6.2$	
Calisthenics - Crunches	$23.8 \pm 22.7$	$1.3 \pm 1.1$	$22.6 \pm 21.7$	$0.6 \pm 1.6$	$0.9 \pm 0.6$	$0.7 \pm 1.8$	
Calisthenics - Sit ups	$28.4 \pm 27.9$	$1.3 \pm 0.8$	$26.6 \pm 20.2$	$2.9 \pm 3.3$	$1.2 \pm 0.5$	$3.3 \pm 3.6$	
Cycling - Cycle hard - Cycle 80rpm	65.1 ± 33.8	$0.7 \pm 0.9$	$64.7 \pm 30.8$	$1.3 \pm 3.5$	$1.1 \pm 0.5$	$1.6 \pm 4.3$	
Cycling - Cycle light - Cycle 100rpm	$38.9 \pm 26.2$	$1.1 \pm 0.7$	$39.6 \pm 24.3$	$8.4 \pm 7.0$	$2.4 \pm 1.6$	$8.0 \pm 7.2$	
Cycling - Cycle light - Cycle 60rpm	$64.0 \pm 26.1$	$2.1 \pm 0.9$	$52.5 \pm 20.7$	$6.0 \pm 6.4$	$3.2 \pm 1.2$	$5.2 \pm 6.3$	
Cycling - Cycle light - Cycle 80rpm	$45.3 \pm 19.0$	$1.8 \pm 0.9$	$42.1 \pm 15.6$	$5.3 \pm 5.7$	$2.8 \pm 1.3$	$5.0 \pm 5.4$	
Cycling - Cycle moderate - Cycle 80rpm	$46.0 \pm 27.1$	$1.7 \pm 1.2$	$40.9 \pm 23.7$	$3.7 \pm 4.4$	$1.7 \pm 0.8$	$4.2 \pm 5.1$	
Lying down	$87.0\pm9.6$	$3.1 \pm 3.6$	$78.2 \pm 16.7$	$90.6 \pm 21.3$	$4.3 \pm 4.4$	$73.4 \pm 21.7$	
Rowing - Rowing hard - Rowing 30spm	$44.5 \pm 34.8$	$0.9 \pm 0.7$	$44.6 \pm 33.9$	$5.7 \pm 8.3$	$1.7 \pm 1.1$	$6.2 \pm 8.4$	
Rowing - Rowing light - Rowing 30spm	$35.8\pm26.0$	$1.4 \pm 0.8$	$34.7\pm20.6$	$6.9\pm 6.0$	$2.8 \pm 1.0$	$6.3 \pm 5.0$	
Rowing - Rowing moderate - Rowing	$46.6\pm18.2$	$1.7 \pm 1.1$	$43.5\pm15.9$	$6.4 \pm 4.8$	$2.1 \pm 1.0$	$6.6 \pm 5.4$	
30spm							
Running - Treadmill 4mph - Treadmill 0	$28.0\pm20.2$	$1.3 \pm 0.7$	$30.6 \pm 18.3$	$3.6 \pm 4.1$	$1.7\pm0.8$	$4.3 \pm 5.1$	
Running - Treadmill 5mph - Treadmill 0	$56.2\pm26.5$	$1.2\pm0.8$	$53.0\pm22.8$	$7.7 \pm 10.0$	$1.8 \pm 1.2$	$7.2 \pm 9.1$	
Running - Treadmill 6mph - Treadmill 0	$63.9\pm37.5$	$0.6\pm0.6$	$62.0 \pm 35.2$	$14.2\pm29.8$	$1.2 \pm 0.9$	$6.9 \pm 10.2$	
Sitting	$20.9\pm22.4$	$0.9 \pm 0.5$	$22.6\pm22.5$	$2.2 \pm 3.0$	$0.9 \pm 0.5$	$2.9 \pm 4.0$	
Sitting - Fidget feet legs	$33.1 \pm 21.8$	$1.2 \pm 0.4$	$30.6 \pm 19.6$	$3.6 \pm 4.7$	$1.3 \pm 0.7$	$4.3 \pm 5.5$	
Sitting - Fidget hands arms	$36.0 \pm 27.7$	$1.6 \pm 0.8$	$30.2 \pm 19.5$	$4.7 \pm 6.6$	$1.5 \pm 1.0$	$5.8 \pm 8.8$	
Stairs - Ascend stairs	$1.5 \pm 2.8$	$1.0 \pm 0.5$	$2.1 \pm 4.0$	$2.6 \pm 3.2$	$1.6 \pm 0.6$	$3.1 \pm 3.8$	
Stairs - Descend stairs	$11.6 \pm 8.5$	$1.3 \pm 0.8$	$13.7 \pm 10.2$	$3.5 \pm 3.4$	$2.0 \pm 0.6$	$3.8 \pm 3.8$	
Standing	$22.1 \pm 24.9$	$1.1 \pm 0.4$	$21.4 \pm 22.0$	$1.9 \pm 3.2$	$1.2 \pm 0.5$	$2.0 \pm 3.5$	
Walking - Treadmill 2mph - Treadmill 0	$45.1 \pm 30.2$	$2.3 \pm 0.8$	$38.7 \pm 26.3$	$6.8 \pm 6.0$	$3.3 \pm 1.8$	$7.0 \pm 6.9$	
Walking - Treadmill 3mph - Treadmill 0	$48.5 \pm 26.0$	$2.4 \pm 0.7$	$41.0 \pm 22.6$	$7.6 \pm 9.1$	$3.1 \pm 1.1$	$7.1 \pm 7.5$	
Walking - Treadmill 3mph - Treadmill 3 - light	57.4 ± 28.1	2.3 ± 1.0	47.8 ± 21.6	4.9 ± 5.5	2.9 ± 1.2	5.1 ± 6.9	
Walking - Treadmill 3mph - Treadmill 6 - moderate	51.6 ± 21.4	2.2 ± 1.0	45.3 ± 17.5	$3.2 \pm 3.2$	2.5 ± 1.1	3.3 ± 3.5	
Walking - Treadmill 3mph - Treadmill 9 -	$46.9\pm25.4$	$2.2 \pm 1.4$	$42.0\pm19.6$	$7.0\pm 6.6$	2.5 ± 1.3	$6.8\pm 6.3$	
nard	26.4 + 22.6	0.0 + 0.2	27.2 . 20.0	24+20	00.05	22 + 40	
Comprise procession	$20.4 \pm 23.0$	$0.9 \pm 0.3$	$27.3 \pm 20.0$	$2.4 \pm 3.0$	$0.9 \pm 0.5$	$3.2 \pm 4.0$	
Carrying groceries	$35.0 \pm 17.4$	$2.1 \pm 0.8$	$31.9 \pm 10.5$	$1.8 \pm 2.0$	$2.0 \pm 0.5$	$2.2 \pm 3.2$	
Condening	$23.0 \pm 10.3$	$2.0 \pm 1.4$	$22.1 \pm 14.0$	$3.4 \pm 3.7$	$2.2 \pm 1.0$	$5.4 \pm 5.4$	
Ironing	$16.0 \pm 14.0$ 37.0 ± 24.0	$2.1 \pm 1.3$ 27 ± 0.9	$10.4 \pm 13.0$ $32.5 \pm 10.0$	$0.0 \pm 1.2$	$0.0 \pm 0.4$	$0.9 \pm 1.9$	
Making the hed	$103 \pm 62$	$2.7 \pm 0.9$	$11.0 \pm 6.4$	$10.0 \pm 11.0$	$3.0 \pm 1.0$ 1.8 ± 0.9	$9.2 \pm 11.0$	
Monning	$10.5 \pm 0.2$ 19.1 + 12.3	$2.4 \pm 1.0$ 2.1 ± 0.9	$19.0 \pm 10.8$	$0.3 \pm 1.0$ 2 2 + 3 5	$1.0 \pm 0.9$ 1.4 ± 0.6	$0.3 \pm 1.0$ 27 + 4.2	
Playing videogames	$17.1 \pm 12.3$ $33.6 \pm 21.8$	$2.1 \pm 0.9$	$19.0 \pm 10.0$ $31.8 \pm 10.0$	$9.7 \pm 11.1$	$1.4 \pm 0.0$	$2.7 \pm 4.2$ 11.5 ± 13.0	
Scrubbing a surface	118+96	$2.3 \pm 1.0$ 2.2 + 1.0	$11.3 \pm 7.4$	$0.3 \pm 1.1$	$1.0 \pm 0.5$	$0.5 \pm 10.0$	
Stacking groceries	$11.0 \pm 9.0$ 14.3 + 9.8	$1.2 \pm 1.0$ 1.7 ± 0.8	$11.3 \pm 7.4$ 15.0 ± 9.5	$1.8 \pm 3.0$	$1.0 \pm 0.5$ 1.0 ± 0.4	$26 \pm 44$	
Sweeping	18.1 + 13.1	$2.0 \pm 0.8$	$18.0 \pm 9.0$	$39 \pm 51$	$1.0 \pm 0.4$ 1.5 ± 0.8	$4.6 \pm 5.7$	
Typing	$10.1 \pm 15.1$ $33.0 \pm 25.9$	$1.7 \pm 0.0$	$10.2 \pm 13.2$ 33.0 + 22.6	$10.6 \pm 11.3$	$1.5 \pm 0.0$ 2.5 + 1.1	$10.7 \pm 11.3$	
Vacuuming	$19.1 \pm 16.4$	$1.7 \pm 1.1$ 2.0 + 1.2	$19.1 \pm 15.4$	$22 \pm 28$	$2.3 \pm 1.1$ 1 3 ± 0 7	$29 \pm 37$	
Walking around block	$30.6 \pm 21.1$	$2.0 \pm 1.2$ 2.2 ± 0.8	$29.3 \pm 20.0$	$48 \pm 52$	$1.5 \pm 0.7$ 2.0 ± 0.8	$57 \pm 65$	
Washing windows	11 3 + 8 3	$1.6 \pm 0.8$	135+92	15+26	12+07	18+29	
Watching TV	269 + 292	$1.0 \pm 0.0$	27.1 + 27.7	81+104	$1.2 \pm 0.7$ $1.2 \pm 0.6$	$10.4 \pm 13.2$	
Weeding	$137 \pm 164$	$1.7 \pm 1.0$ $1.7 \pm 1.0$	$13.0 \pm 12.4$	$0.4 \pm 10.4$	$0.6 \pm 0.4$	$0.6 \pm 1.9$	
Wining/Dusting	$20.6 \pm 14.7$	$2.4 \pm 1.6$	19.9 + 11.6	46+43	19+12	$55 \pm 52$	
Writing	46.6 + 31.8	$2.0 \pm 1.0$ $2.0 \pm 1.2$	40.6 + 24.3	10.7 + 8.1	3.1 + 1.6	10.3 + 8.7	
taking out trash	$14.3 \pm 9.5$	$2.0 \pm 1.1$	15.4 ± 9.9	0.7 ± 1.3	$1.1 \pm 0.7$	$1.1 \pm 2.0$	

Table A7-2: Subject dependent and subject independent performance results using the C4.5 and the *HRAboveRest* feature without considering the garbage class (*unknown* activity) when the feature is computed over windows of 5.6s.

Activity	ScaledHR	ScaledHR +	ScaledHR +	ScaledHR +
		Weight	FitnessIndex	FatPercent
Bench weight lifting - hard	$0.00 \pm 0.00$	$0.0 \pm 0.0$	$1.8 \pm 4.1$	$1.8 \pm 2.7$
Bench weight lifting - light	$2.15 \pm 3.45$	$2.2 \pm 7.9$	$2.9 \pm 8.1$	$2.9 \pm 5.4$
Bench weight lifting - moderate	$0.82 \pm 2.21$	$0.5 \pm 1.5$	$3.4 \pm 8.0$	$2.1 \pm 4.2$
Bicep curls - hard	$2.59 \pm 4.31$	$6.0 \pm 10.6$	$5.7 \pm 16.1$	$1.3 \pm 3.0$
Bicep curls - light	$1.13 \pm 2.17$	$2.3 \pm 3.7$	$1.0 \pm 2.2$	$0.3 \pm 1.0$
Bicep curls - moderate	$1.00 \pm 2.40$	$1.2 \pm 2.4$	$1.2 \pm 2.4$	$0.0 \pm 0.0$
Calisthenics - Crunches	$1.02 \pm 2.12$	$6.4 \pm 13.4$	$5.1 \pm 13.2$	$1.0 \pm 3.1$
Calisthenics - Sit ups	$1.88 \pm 3.23$	$4.7 \pm 9.6$	$10.4 \pm 16.6$	$3.6 \pm 8.6$
Cycling - Cycle hard - Cycle 80rpm	$1.07 \pm 3.09$	$0.4 \pm 1.1$	$15.4 \pm 37.6$	$0.5 \pm 1.8$
Cycling - Cycle light - Cycle 100rpm	$9.90 \pm 7.38$	$12.3 \pm 21.5$	$14.6 \pm 22.0$	$7.8 \pm 12.8$
Cycling - Cycle light - Cycle 60rpm	$8.67 \pm 7.81$	$5.9 \pm 15.1$	$1.7 \pm 4.7$	$8.1 \pm 17.0$
Cycling - Cycle light - Cycle 80rpm	$6.96 \pm 5.41$	10.7 + 17.3	6.8 + 16.4	9.7 + 18.2
Cycling - Cycle moderate - Cycle 80rpm	434 + 540	2.7 + 7.1	36+62	44 + 114
Lying down	88 44 + 19 64	$\frac{2.7}{85.1} \pm 21.0$	84 2 + 21 8	858+227
Rowing - Rowing hard - Rowing 30spm	0.98 + 2.29	$19.4 \pm 35.0$	$11.0 \pm 26.0$	14 + 31
Rowing - Rowing light - Rowing 30spm	6.84 ± 12.09	$24 \pm 39$	54+168	$9.3 \pm 21.1$
Rowing - Rowing moderate - Rowing 30spm	10 25 + 19 92	$15.2 \pm 20.2$	137+232	7.8 ± 19.2
Running - Treadmill 4mph - Treadmill 0	6 25 + 5 93	$3.0 \pm 5.2$	56+124	31+69
Running Treadmill 5mph Treadmill 0	$53.61 \pm 20.51$	$3.0 \pm 3.2$	40.6 ± 28.0	$\frac{5.1 \pm 0.9}{45.0 \pm 20.0}$
Running - Treadmill 6mph - Treadmill 0	52 88 + 32 31	$41.3 \pm 27.3$ $42.8 \pm 40.3$	$40.0 \pm 28.0$ $47.9 \pm 47.8$	$43.0 \pm 29.0$ $32.7 \pm 27.6$
Sitting	$2.08 \pm 4.30$	$9.0 \pm 15.1$	$9.7 \pm 18.1$	9.6 ± 18.1
Sitting Eidget feet lags	$2.00 \pm 4.09$	$7.0 \pm 15.1$ $7.0 \pm 16.1$	$9.7 \pm 10.1$ 12.4 ± 21.1	$9.0 \pm 10.1$
Sitting - Fidget hands arms	5.20 ± 4.00	$7.0 \pm 10.1$	12.4 ± 21.1	$10.7 \pm 20.3$
Stuling - Fluget hands arms	$1.40 \pm 5.57$	9.9 ± 21.9	8.9 ± 20.8	$0.3 \pm 14.4$
Stairs - Ascend stairs	$2.05 \pm 3.01$	$1.3 \pm 2.3$	$3.0 \pm 4.0$	2.9 ± 5.0
Stairs - Descend stairs	$2.12 \pm 2.91$	$6.4 \pm 9.4$	$4.2 \pm 4.8$	$1.8 \pm 2.5$
Standing	$0.73 \pm 1.77$	$1.8 \pm 3.0$	2.4 ± 3.4	$0.3 \pm 1.2$
Walking - Treadmill 2mph - Treadmill 0	10.75 ± 11.59	$13.2 \pm 20.1$	$11.2 \pm 18.4$	$14.0 \pm 24.0$
Walking - Treadmill 3mph - Treadmill 0	7.78 ± 6.45	$17.6 \pm 26.3$	9.2 ± 16.9	$5.1 \pm 8.2$
Walking - Treadmill 3mph - Treadmill 3 - light	7.23 ± 5.85	$14.0 \pm 26.2$	20.6 ± 26.9	$10.1 \pm 14.2$
Walking - Treadmill 3mph - Treadmill 6 - moderate	$11.68 \pm 9.66$	$11.1 \pm 16.6$	$10.3 \pm 21.2$	$6.3 \pm 14.1$
Walking - Treadmill 3mph - Treadmill 9 - hard	9.71 ± 6.19	$6.4 \pm 8.6$	7.9 ± 18.2	4.8 ± 13.0
kneeling	$1.43 \pm 2.25$	$1.5 \pm 3.8$	$5.5 \pm 8.1$	$2.9 \pm 9.1$
Carrying groceries	$2.55 \pm 3.57$	$6.2 \pm 12.3$	$4.9 \pm 9.7$	$7.0 \pm 12.9$
Doing dishes	$4.28 \pm 4.48$	$3.6 \pm 9.0$	$4.5 \pm 8.6$	$4.7 \pm 10.5$
Gardening	$1.75 \pm 3.08$	$2.2 \pm 5.1$	$4.0 \pm 10.0$	0.8 ± 1.7
Ironing	$11.46 \pm 12.58$	$8.3 \pm 15.8$	8.8 ± 12.8	$4.2 \pm 8.6$
Making the bed	$2.56 \pm 3.02$	$3.5 \pm 4.2$	$5.6 \pm 8.6$	$3.3 \pm 5.1$
Mopping	$2.36 \pm 4.83$	$2.0 \pm 3.4$	$4.6 \pm 9.8$	$2.7 \pm 5.4$
Playing videogames	$11.26 \pm 10.44$	$14.7 \pm 20.2$	$22.1 \pm 24.6$	$12.4 \pm 18.5$
Scrubbing a surface	$2.21 \pm 2.88$	$2.2 \pm 2.5$	$2.1 \pm 2.8$	$6.0 \pm 12.8$
Stacking groceries	$1.71 \pm 2.40$	$2.7 \pm 4.4$	$1.1 \pm 1.9$	$1.4 \pm 2.1$
Sweeping	$3.59 \pm 3.81$	$2.6\pm5.6$	$5.2 \pm 6.4$	$2.2 \pm 3.6$
Typing	$6.97 \pm 6.18$	$6.8\pm10.5$	$11.2 \pm 16.4$	$10.6 \pm 17.3$
Vacuuming	$2.30\pm3.12$	$4.8\pm7.2$	$6.7\pm8.6$	$2.6\pm3.8$
Walking around block	$2.76\pm3.33$	$9.8\pm22.6$	$4.5\pm6.7$	$1.5 \pm 2.8$
Washing windows	$1.80\pm2.58$	$1.3 \pm 2.4$	$2.2\pm3.6$	$2.1 \pm 3.0$
Watching TV	$5.25 \pm 5.99$	$5.0\pm7.6$	$13.7\pm22.9$	$4.0 \pm 5.8$
Weeding	$0.39\pm0.92$	$2.1\pm 6.5$	$2.2\pm4.9$	$0.4 \pm 1.3$
Wiping/Dusting	$4.29 \pm 4.08$	$5.2 \pm 7.4$	$10.9 \pm 17.5$	$2.3 \pm 6.4$
Writing	$12.52 \pm 12.30$	$13.0 \pm 21.3$	$4.3 \pm 6.7$	$10.6 \pm 18.0$
taking out trash	$0.28 \pm 0.85$	$0.7 \pm 1.8$	$0.2 \pm 0.8$	$1.4 \pm 5.0$

Table A7-3: Performance obtained when adding the Weight, FatPercent, and FitnessIndex to the ScaledHR feature during subject independent evaluation using the C4.5 classifier. Features are computed over windows of 5.6s in length and the activities to recognize are the 51 activities included in the MIT dataset, without including the *unknown* class.

Class	TP Rate	FP Rate	<b>F-Measure</b>
Bicep curls - hard	$46.8\pm35.0$	$36.7\pm27.5$	$38.6\pm25.5$
Bicep curls - light	$46.7\pm39.5$	$34.1 \pm 32.9$	$39.0\pm27.8$
Bicep curls - moderate	$26.9\pm37.1$	$35.5\pm32.0$	$20.8\pm25.8$

Table A7-4: Performance while recognizing only among the intensity levels of bicep curls in a subject independent manner using the C4.5 Decision tree classifier and the ScaledHR feature computed over windows of 5.6s.

Class	TP Rate	FP Rate	F-Measure
Bench weight lifting - hard	$28.3\pm21.1$	$14.1\pm17.5$	$36.2\pm26.3$
Bench weight lifting - light	$78.9\pm35.4$	$63.8\pm39.4$	$59.9\pm26.7$
Bench weight lifting - moderate	$48.7 \pm 17.5$	$60.1\pm26.8$	$34.6\pm8.2$

Table A7-5: Performance while recognizing only among the intensity levels of bench weight lifting in a subject independent manner using the C4.5 Decision tree classifier and the ScaledHR feature computed over windows of 5.6s.

Class	TP Rate	FP Rate	<b>F-Measure</b>
Rowing - Rowing hard - Rowing 30spm	$17.4 \pm 26.7$	$31.8\pm22.3$	$15.2 \pm 17.9$
Rowing - Rowing light - Rowing 30spm	$44.7 \pm 29.4$	$40.8\pm30.0$	$38.6 \pm 19.9$
Rowing - Rowing moderate - Rowing 30spm	$21.3\pm21.9$	$34.2\pm23.7$	$20.9 \pm 15.0$

Table A7-6: Performance while recognizing only among the intensity levels of rowing in a subject independent manner using the C4.5 Decision tree classifier and the ScaledHR feature computed over windows of 5.6s.

Class	TP Rate	FP Rate	<b>F-Measure</b>
Walking - Treadmill 3mph - Treadmill 0	$56.0\pm42.3$	$10.7\pm9.2$	$50.4\pm33.9$
Walking - Treadmill 3mph - Treadmill 3 - light	$44.1\pm29.3$	$27.9 \pm 11.7$	$36.9\pm21.4$
Walking - Treadmill 3mph - Treadmill 6 - moderate	$28.6\pm22.5$	$20.0\pm12.7$	$28.6\pm20.9$
Walking - Treadmill 3mph - Treadmill 9 - hard	$56.5 \pm 43.1$	$21.9 \pm 18.2$	$41.0 \pm 27.2$

Table A7-7: Performance while recognizing only among the intensity levels of walking at 3mph in a subject independent manner using the C4.5 Decision tree classifier and the ScaledHR feature computed over windows of 5.6s.

Activity Category	5.6	11.3	22.7	44.5
All	$38.4\pm7.8$	$35.5\pm8.0$	$32.0\pm6.7$	$25.0\pm5.6$
Postures	37.7±22.3	32.1±20.5	28.2±22.8	19.8±15.5
	(1.5±1.0)	(1.6±1.1)	(1.7±1.2)	$(2.1\pm1.4)$
Ambulation	39.0±22.2	37.8±22.9	33.7±22.7	27.0±28.0
	(1.7±0.8)	(1.9±0.9)	(1.9±1.1)	(2.4±1.2)
Exercise	38.3±23.9	35.6±24.6	31.0±26.1	22.6±26.1
	(1.4±0.8)	(1.5±0.9)	(1.7±1.0)	(2.0±1.2)
Resistance	34.2±20.6	31.0±22.1	27.3±22.8	21.2±24.5
Exercise	(1.6±0.9)	(1.7±0.9)	$(1.8\pm1.1)$	(2.4±1.3)
Household	24.0±18.0	21.3±19.3	19.2±20.2	13.7±19.5
	(2.0±1.1)	(2.2±1.2)	(2.4±1.6)	(2.8±1.8)

Table A7-8: True positive rate and false positive rate (shown in parenthesis) obtained utilizing the C4.5 classifier when the *ScaledHR* feature is computed over varying window lengths as evaluated with subject dependent training. The target activities are the 51 activities contained in the MIT dataset without including the *unknown* class.

Activity Category	5.6	11.3	22.7	44.5
All	7.7±1.9	$7.6 \pm 1.9$	$7.6 \pm 3.2$	$7.2 \pm 2.9$
Postures	9.7±9.4	5.2±6.5	5.1±5.2	5.8±4.4
	(3.8±1.2)	(3.1±1.3)	(3.0±1.5)	(3.0±1.6)
Ambulation	5.0±7.1	5.5±7.9	4.9±9.6	6.9±10.6
	(1.7±0.9)	(2.4±1.3)	(2.7±1.6)	(3.1±1.9)
Exercise	8.6±11.1	9.7±12.7	7.3±13.9	7.1±14.7
	(1.9±0.9)	(2.7±1.2)	(2.7±1.4)	(2.9±1.8)
Resistance	8.6±10.4	9.6±11.6	8.1±14.0	9.1±15.0
Exercise	(1.9±0.9)	(2.6±1.3)	(2.7±1.5)	(3.2±1.9)
Household	4.0±6.0	4.5±7.4	4.8±9.4	2.9±8.2
	(1.5±0.8)	(2.3±1.5)	(2.5±1.8)	(3.1±2.0)

Table A7-9: True positive rate and false positive rate (shown in parenthesis) obtained utilizing the C4.5 classifier when the *HRVar* feature is computed over varying window lengths as evaluated with subject dependent training. The target activities are the 51 activities contained in the MIT dataset without including the *unknown* class.

Activity Category	5.6	11.3	22.7	44.5
All	$8.2 \pm 1.8$	$7.62 \pm 1.65$	$6.28 \pm 2.81$	$5.87 \pm 2.96$
Postures	10.3±8.4	6.1±8.5	4.7±5.2	5.5±7.3
	(3.8±1.0)	(3.0±1.4)	(3.3±1.7)	(3.7±2.0)
Ambulation	4.9±7.0	7.0±7.9	6.1±11.1	6.8±9.9
	(1.7±0.8)	(2.6±1.3)	(2.5±1.4)	(2.9±2.2)
Exercise	8.8±11.0	9.7±13.2	5.9±10.6	5.2±10.3
	(1.9±0.9)	(2.6±1.2)	(2.9±1.4)	(2.9±1.9)
Resistance	9.2±11.3	11.3±13.3	7.6±11.7	8.0±11.9
Exercise	(1.9±0.9)	(2.6±1.3)	(2.7±1.4)	(3.1±2.0)
Household	4.0±6.0	4.4±6.6	3.3±7.3	2.9±6.0
	(1.5±0.8)	(2.3±1.5)	(2.4±1.7)	(3.1±2.1)

Table A7-10: True positive rate and false positive rate (shown in parenthesis) obtained utilizing the C4.5 classifier when the *HRTrend* feature is computed over varying window lengths as evaluated with subject dependent training. The target activities are the 51 activities contained in the MIT dataset without including the *unknown* class.

Activity Category	5.6	11.3	22.7	44.5
All	$13.8 \pm 3.2$	$13.5 \pm 3.4$	$15.7 \pm 3.7$	$15.95 \pm 3.39$
Postures	16.2±5.9	16.7±7.7	17.7±9.4	17.9±9.1
	(1.7±1.2)	(1.7±1.4)	(1.7±1.5)	(1.9±1.6)
Ambulation	14.2±9.3	13.7±10.7	16.0±13.1	17.9±18.8
	(2.3±1.3)	(2.3±1.4)	(2.3±1.5)	(2.4±1.7)
Exercise	9.0±7.5	8.6±8.0	10.8±10.3	10.6±12.2
	(1.6±1.0)	(1.6±1.1)	(1.6±1.1)	(2.0±1.4)
Resistance	4.1±5.2	4.1±6.1	5.2±7.9	6.5±10.1
Exercise	(1.7±0.8)	(1.7±0.9)	(1.7±1.0)	(2.0±1.2)
Household	4.3±4.9	4.3±5.4	5.6±8.9	5.9±10.6
	(1.7+0.9)	(1.7+1.0)	(1.8+1.1)	(2.1+1.3)

Table A7-11: True positive rate and false positive rate (shown in parenthesis) obtained utilizing the C4.5 classifier when the *ScaledHR* feature is computed over varying window lengths as evaluated with subject independent training. The target activities are the 51 activities contained in the MIT dataset without including the *unknown* class.

Activity Category	5.6	11.3	22.7	44.5
All	$6.4 \pm 1.3$	$5.2 \pm 1.6$	$4.5 \pm 1.9$	$4.8 \pm 2.4$
Postures	11.9±4.2	9.8±5.1	7.8±5.9	6.0±6.9
	(7.7±1.8)	(6.5±1.9)	(4.3±1.5)	(3.4±1.6)
Ambulation	2.2±2.4	1.8±3.0	2.0±4.7	4.5±8.5
	(1.2±0.6)	(1.6±0.8)	(2.0±1.0)	(2.3±1.3)
Exercise	2.7±4.1	1.1±2.5	1.2±3.1	1.2±3.9
	(1.2±0.6)	(1.4±0.6)	(1.7±0.8)	(2.0±1.1)
Resistance	3.1±4.3	1.6±3.0	1.9±4.7	3.1±6.0
Exercise	(1.4±0.6)	(1.4±0.6)	(1.8±0.9)	(2.2±1.2)
Household	1.2±1.8	1.1±2.3	0.9±2.3	1.8±5.0
	(1.0±0.4)	(1.1±0.6)	(1.5±0.8)	(2.0±1.1)

Table A7-12: True positive rate and false positive rate (shown in parenthesis) obtained utilizing the C4.5 classifier when the HRVar feature is computed over varying window lengths as evaluated with subject independent training. The target activities are the 51 activities contained in the MIT dataset without including the *unknown* class.

Activity Category	5.6	11.3	22.7	44.5
All	$7.6 \pm 1.2$	$5.20 \pm 1.12$	$4.65 \pm 1.15$	$5.78 \pm 2.36$
Postures	13.0±4.5	7.7±3.5	6.6±4.0	6.8±2.8
	(7.0±2.2)	(5.1±1.0)	(4.5±0.9)	$(4.9 \pm 1.5)$
Ambulation	5.2±4.2	3.7±4.0	3.8±5.4	5.8±9.1
	(1.4±0.6)	(1.8±0.7)	(1.9±1.0)	(2.1±1.3)
Exercise	2.7±3.2	1.1±2.0	$1.2\pm2.8$	$1.2\pm4.1$
	(1.2±0.6)	(1.5±0.6)	(1.8±0.9)	$(1.9\pm1.1)$
Resistance	4.4±3.9	2.5±3.0	3.0±4.5	4.1±6.5
Exercise	(1.5±0.7)	(1.7±0.7)	(1.9±1.0)	(2.0±1.2)
Household	$1.8 \pm 2.2$	$1.8\pm2.8$	1.3±2.7	1.7±4.6
	(1.1±0.4)	(1.3±0.5)	(1.5±0.8)	$(1.8\pm1.1)$

Table A7-13: True positive rate and false positive rate (shown in parenthesis) obtained utilizing the C4.5 classifier when the *HRTrend* feature is computed over varying window lengths as evaluated with subject independent training. The target activities are the 51 activities contained in the MIT dataset without including the *unknown* class.

Activity	ctivity True Positive Rate						
	Subject	Dependent	Subject	et Independent			
	Invariant	Invariant Reduced	Invariant	Invariant Reduced			
	Reduced	+ ScaledHR	Reduced	+ ScaledHR			
Bench weight lifting - hard	$93.3\pm8.3$	$93.3 \pm 8.3$	$14.5 \pm 22.4$	$13.1 \pm 25.2$			
Bench weight lifting - light	$93.6 \pm 9.6$	$93.6 \pm 9.6$	$38.8\pm37.0$	$34.0 \pm 35.4$			
Bench weight lifting - moderate	91.3 ± 10.9	$91.3 \pm 10.9$	$28.4 \pm 36.4$	$24.8 \pm 33.7$			
Bicep curls - hard	94.1 ± 10.0	94.1 ± 10.0	$43.0 \pm 40.5$	$39.9 \pm 39.4$			
Bicep curls - light	$91.9 \pm 8.1$	$91.9 \pm 8.1$	$20.5 \pm 31.7$	$27.0 \pm 36.4$			
Bicep curls - moderate	$91.6 \pm 9.9$	$91.8\pm9.8$	$21.8\pm30.8$	$17.7 \pm 26.6$			
Calisthenics - Crunches	$96.2 \pm 4.4$	$96.2 \pm 4.4$	$66.6 \pm 38.4$	$66.6 \pm 38.4$			
Calisthenics - Sit ups	$96.6 \pm 3.1$	$96.6 \pm 3.1$	$80.5\pm35.0$	$81.4 \pm 34.1$			
Cycling - Cycle hard - Cycle 80rpm	$88.4 \pm 8.7$	$92.9 \pm 7.7$	$33.4\pm31.9$	$35.8 \pm 31.2$			
Cycling - Cycle light - Cycle 100rpm	$98.1 \pm 2.3$	$98.1 \pm 2.3$	$97.0\pm5.8$	$97.6 \pm 6.1$			
Cycling - Cycle light - Cycle 60rpm	$99.5 \pm 1.2$	$99.5 \pm 1.2$	$87.6\pm25.2$	$87.7\pm25.3$			
Cycling - Cycle light - Cycle 80rpm	$97.5\pm3.2$	$97.5 \pm 3.2$	$41.6\pm39.1$	$44.2\pm41.5$			
Cycling - Cycle moderate - Cycle 80rpm	$92.6\pm5.2$	$94.2 \pm 5.3$	$32.6\pm32.0$	$31.5\pm29.6$			
Lying down	$99.9 \pm 0.3$	$99.9 \pm 0.3$	$95.9 \pm 12.3$	96.0 ± 12.3			
Rowing - Rowing hard - Rowing 30spm	$85.2\pm14.2$	$88.1 \pm 14.6$	$34.2\pm33.5$	$27.2\pm30.8$			
Rowing - Rowing light - Rowing 30spm	$88.9 \pm 10.3$	$90.0\pm9.6$	$41.9\pm32.3$	$41.3 \pm 35.0$			
Rowing - Rowing moderate - Rowing 30spm	$81.4\pm13.7$	$83.8\pm11.9$	$24.0\pm23.6$	$31.7 \pm 29.4$			
Running - Treadmill 4mph - Treadmill 0	$97.7\pm2.9$	$97.7 \pm 2.9$	$57.4 \pm 39.6$	$55.4\pm39.2$			
Running - Treadmill 5mph - Treadmill 0	$93.7\pm4.8$	$93.7 \pm 5.2$	$56.8\pm29.1$	$68.0 \pm 25.4$			
Running - Treadmill 6mph - Treadmill 0	$88.4 \pm 15.1$	$88.6 \pm 15.2$	$64.0\pm36.0$	$56.1\pm36.2$			
Sitting	$97.4 \pm 4.1$	$97.4 \pm 4.1$	$58.1\pm43.0$	$57.9 \pm 43.3$			
Sitting - Fidget feet legs	$95.3\pm5.3$	$95.3\pm5.3$	$66.7\pm30.2$	$67.7 \pm 31.1$			
Sitting - Fidget hands arms	$93.4\pm8.0$	$93.7\pm8.1$	$57.8\pm32.6$	$51.9\pm34.9$			
Stairs - Ascend stairs	$91.1\pm6.7$	$90.3 \pm 7.2$	$71.8\pm28.6$	$72.7\pm29.5$			
Stairs - Descend stairs	$90.7 \pm 6.6$	$90.7 \pm 6.6$	$59.6\pm26.4$	$57.9 \pm 29.3$			
Standing	$97.9 \pm 3.2$	$97.9 \pm 3.2$	$86.3 \pm 21.1$	86.3 ± 21.1			
Walking - Treadmill 2mph - Treadmill 0	$96.8 \pm 4.7$	$96.8 \pm 4.7$	$70.9\pm25.5$	$70.5 \pm 23.8$			
Walking - Treadmill 3mph - Treadmill 0	$84.5 \pm 10.2$	$91.5 \pm 7.2$	$20.1 \pm 23.7$	$27.5 \pm 20.4$			
Walking - Treadmill 3mph - Treadmill 3 - light	$76.2 \pm 14.6$	$88.3 \pm 10.9$	$24.4\pm25.8$	$23.8 \pm 26.7$			
Walking - Treadmill 3mph - Treadmill 6 - moderate	$77.0 \pm 10.3$	$89.5 \pm 8.2$	$14.8\pm18.8$	$35.5 \pm 27.4$			
Walking - Treadmill 3mph - Treadmill 9 - hard	$88.4 \pm 9.5$	$92.1 \pm 5.0$	$27.7 \pm 24.8$	$56.4 \pm 35.9$			
kneeling	$97.3 \pm 3.7$	$97.3 \pm 3.7$	$97.3 \pm 5.0$	$97.3 \pm 5.0$			
Carrying groceries	$91.1 \pm 7.6$	$91.6 \pm 7.5$	$56.7\pm27.9$	$58.3 \pm 29.7$			
Doing dishes	85.1 ± 7.3	$85.7 \pm 6.7$	$55.9 \pm 25.4$	$57.2 \pm 24.0$			
Gardening	$79.6 \pm 11.8$	$79.9 \pm 11.4$	$20.7 \pm 25.9$	$20.3 \pm 25.0$			
Ironing	$85.4 \pm 6.3$	$86.4 \pm 5.1$	$53.3 \pm 28.4$	$53.2 \pm 28.7$			
Making the bed	$62.2 \pm 10.8$	$68.1 \pm 10.1$	$40.5 \pm 18.3$	39.4 ± 17.8			
Mopping	$72.4 \pm 11.8$	$73.0 \pm 12.1$	$33.3 \pm 18.4$	$32.3 \pm 14.6$			
Playing videogames	99.0 ± 2.3	99.0 ± 2.3	$64.4 \pm 42.2$	63.8 ± 42.0			
Scrubbing a surface	85.0 ± 12.9	85.5 ± 11.7	$40.0 \pm 35.4$	$40.3 \pm 35.2$			
Stacking groceries	$71.0 \pm 16.0$	$74.0 \pm 13.4$	$33.8 \pm 20.9$	33.8 ± 21.3			
Sweeping	$68.3 \pm 15.2$	$70.6 \pm 14.5$	$34.4 \pm 20.6$	$32.4 \pm 20.6$			
Typing	98.3 ± 2.4	$98.3 \pm 2.4$	$75.0 \pm 29.4$	$71.5 \pm 33.5$			
Vacuuming	74.5 ± 10.1	76.0 ± 8.6	53.4 ± 23.9	$54.2 \pm 22.8$			
Walking around block	90.4 ± 7.7	90.0 ± 7.4	34.9 ± 18.6	$40.0 \pm 21.7$			
Washing windows	71.2 ± 9.4	72.9 ± 8.9	$44.2 \pm 21.5$	$43.9 \pm 21.0$			
Watching TV	98.2 ± 2.5	98.2 ± 2.5	48.2 ± 43.6	$45.8 \pm 44.0$			
Weeding	$75.2 \pm 11.0$	$75.2 \pm 10.4$	$15.5 \pm 19.1$	$14.8 \pm 20.4$			
Wiping/Dusting	$69.4 \pm 13.9$	$74.0 \pm 12.1$	$39.4 \pm 21.6$	$38.6 \pm 21.0$			
Writing	97.3 ± 3.0	97.3 ± 3.0	$61.7 \pm 40.6$	$61./ \pm 40.6$			
taking out trash	$62.2 \pm 13.6$	$65.4 \pm 14.9$	$23.0 \pm 14.1$	$25.3 \pm 15.2$			

Table A7-14: True positive rate obtained when adding the *ScaledHR* feature to *the invariant reduced* feature set using the C4.5 classifier. The sliding window length for accelerometer data and heart rate is 5.6s. The activities to recognize are the 51 contained in the MIT dataset without including the *unknown* class.

Activity	False Positive Rate						
	Subject De	pendent	Subject In	dependent			
	Invariant Reduced	Invariant Reduced + ScaledHR	Invariant Reduced	Invariant Reduced + ScaledHR			
Bench weight lifting - hard	$0.1 \pm 0.1$	0.1 + 0.1	$0.7 \pm 0.8$	$0.7 \pm 0.8$			
Bench weight lifting - light	$0.1 \pm 0.1$	0.1 ± 0.1	$0.8 \pm 0.8$	$0.7 \pm 0.8$ 0.8 ± 0.8			
Bench weight lifting - moderate	$0.1 \pm 0.2$	$0.1 \pm 0.1$	11+13	1 1 + 1 3			
Bicen curls - hard	0.1 + 0.2	$0.1 \pm 0.2$	1.6 + 1.2	$1.6 \pm 1.3$			
Bicep curls - light	$0.1 \pm 0.1$	0.1 ± 0.1	0.6 ± 0.9	$0.8 \pm 1.1$			
Bicep curls - moderate	$0.2 \pm 0.2$	$0.2 \pm 0.2$	$1.1 \pm 1.1$	$0.9 \pm 1.0$			
Calisthenics - Crunches	$0.0 \pm 0.0$	$0.0 \pm 0.0$	$0.6 \pm 1.2$	$0.6 \pm 1.2$			
Calisthenics - Sit ups	$0.1 \pm 0.2$	$0.1 \pm 0.1$	$0.2 \pm 0.2$	$0.2 \pm 0.2$			
Cycling - Cycle hard - Cycle 80rpm	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$1.1 \pm 1.4$	$1.1 \pm 1.2$			
Cycling - Cycle light - Cycle 100rpm	$0.0 \pm 0.0$	$0.0 \pm 0.0$	$0.1 \pm 0.4$	$0.1 \pm 0.4$			
Cycling - Cycle light - Cycle 60rpm	$0.0 \pm 0.0$	$0.0 \pm 0.0$	$0.1 \pm 0.2$	$0.1 \pm 0.2$			
Cycling - Cycle light - Cycle 80rpm	0.1 ± 0.1	$0.1 \pm 0.1$	$1.3 \pm 1.4$	$1.2 \pm 1.2$			
Cycling - Cycle moderate - Cycle 80rpm	$0.2 \pm 0.1$	$0.1 \pm 0.1$	$1.2 \pm 1.0$	$1.2 \pm 0.9$			
Lying down	$0.0 \pm 0.1$	$0.0 \pm 0.1$	$0.3 \pm 0.6$	$0.3 \pm 0.7$			
Rowing - Rowing hard - Rowing 30spm	$0.2 \pm 0.2$	$0.2 \pm 0.2$	$1.1 \pm 0.8$	$0.7 \pm 0.6$			
Rowing - Rowing light - Rowing 30spm	$0.2 \pm 0.2$	$0.2 \pm 0.2$	$1.8 \pm 1.2$	$1.7 \pm 1.3$			
Rowing - Rowing moderate - Rowing 30spm	$0.4 \pm 0.3$	$0.3 \pm 0.3$	$1.0 \pm 0.9$	$1.2 \pm 1.1$			
Running - Treadmill 4mph - Treadmill 0	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.8 \pm 1.0$	$0.7 \pm 1.0$			
Running - Treadmill 5mph - Treadmill 0	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.9 \pm 0.7$	$1.0 \pm 1.0$			
Running - Treadmill 6mph - Treadmill 0	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.7 \pm 0.8$	$0.3 \pm 0.4$			
Sitting	$0.1 \pm 0.1$	$0.0 \pm 0.1$	$1.3 \pm 1.2$	$1.3 \pm 1.2$			
Sitting - Fidget feet legs	$0.0 \pm 0.1$	$0.0 \pm 0.1$	$0.2 \pm 0.2$	$0.2 \pm 0.2$			
Sitting - Fidget hands arms	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.9 \pm 0.9$	$0.9 \pm 0.9$			
Stairs - Ascend stairs	$0.2 \pm 0.2$	$0.2\pm0.2$	$0.5 \pm 0.5$	$0.7 \pm 1.0$			
Stairs - Descend stairs	$0.2 \pm 0.2$	$0.1 \pm 0.1$	$0.7 \pm 0.8$	$0.8\pm0.8$			
Standing	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.5\pm0.9$	$0.5\pm0.9$			
Walking - Treadmill 2mph - Treadmill 0	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.6\pm0.9$	$0.7 \pm 1.4$			
Walking - Treadmill 3mph - Treadmill 0	$0.4 \pm 0.2$	$0.2 \pm 0.2$	$1.2 \pm 1.0$	$1.2 \pm 1.0$			
Walking - Treadmill 3mph - Treadmill 3 - light	$0.6 \pm 0.3$	$0.3 \pm 0.2$	$1.6\pm0.9$	$1.0\pm0.8$			
Walking - Treadmill 3mph - Treadmill 6 - moderate	$0.5 \pm 0.3$	$0.3 \pm 0.2$	$1.5 \pm 1.0$	$1.4 \pm 1.1$			
Walking - Treadmill 3mph - Treadmill 9 - hard	$0.3 \pm 0.2$	$0.1 \pm 0.1$	$1.5 \pm 1.5$	$1.5 \pm 1.6$			
kneeling	$0.0 \pm 0.1$	$0.1 \pm 0.1$	$0.0 \pm 0.1$	$0.1 \pm 0.1$			
Carrying groceries	$0.2 \pm 0.2$	$0.2 \pm 0.2$	$0.9 \pm 0.6$	$1.0 \pm 0.9$			
Doing dishes	$0.4 \pm 0.1$	$0.4 \pm 0.2$	$1.2 \pm 0.9$	$1.2 \pm 1.0$			
Gardening	$0.4 \pm 0.3$	$0.4 \pm 0.2$	$1.0 \pm 0.8$	$1.1 \pm 0.8$			
Ironing	$0.4 \pm 0.3$	$0.4 \pm 0.2$	$1.3 \pm 0.8$	$1.3 \pm 0.8$			
Making the bed	$0.8 \pm 0.4$	$0.7 \pm 0.3$	$2.0 \pm 0.9$	$2.0 \pm 0.9$			
Mopping	0.9 ± 0.3	0.8 ± 0.3	$1.8 \pm 0.9$	$1.8 \pm 0.9$			
Playing videogames	$0.0 \pm 0.0$	$0.0 \pm 0.0$	$1.2 \pm 1.2$	$1.2 \pm 1.1$			
Scrubbing a surface	$0.3 \pm 0.3$	$0.3 \pm 0.2$	$1.5 \pm 0.8$	$1.5 \pm 0.9$			
Stacking groceries	$0.6 \pm 0.3$	$0.5 \pm 0.3$	$1.2 \pm 0.8$	$1.2 \pm 0.9$			
Sweeping	$0.7 \pm 0.3$	$0.7 \pm 0.3$	$1.5 \pm 0.4$	$1.5 \pm 0.5$			
Typing	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.8 \pm 1.0$	$0.8 \pm 1.0$			
Vacuuming	$0.5 \pm 0.3$	$0.5 \pm 0.2$	$1.1 \pm 0.6$	$1.1 \pm 0.6$			
Walking around block	$0.2 \pm 0.1$	$0.2 \pm 0.1$	$3.1 \pm 2.5$	$1.7 \pm 1.1$			
wasning windows	0.6 ± 0.2	0.5 ± 0.2	$1.2 \pm 0.7$	$1.2 \pm 0.7$			
watching I v	$0.0 \pm 0.0$	$0.0 \pm 0.0$	$1.5 \pm 1.6$	$1.5 \pm 1.6$			
Weeding	0.6 ± 0.3	$0.5 \pm 0.2$	$1.1 \pm 0.7$	$1.0 \pm 0.6$			
wiping/Dusting	$0.8 \pm 0.3$	0.6 ± 0.3	$1.5 \pm 0.6$	$1.5 \pm 0.6$			
writing tabing out track	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.3 \pm 0.4$	$0.3 \pm 0.4$			
taking out trasn	$0.7 \pm 0.3$	$0.0 \pm 0.3$	$1.2 \pm 0.4$	$1.2 \pm 0.4$			

Table A7-15: False positive rate obtained when adding the *ScaledHR* feature to *the invariant reduced* feature set using the C4.5 classifier. The sliding window length for accelerometer data and heart rate is 5.6s. The activities to recognize are the 51 contained in the MIT dataset without including the *unknown* class.

Activity	F-Measure						
	Subject I	Dependent	Subject In	dependent			
	Invariant Reduced	Invariant Reduced	Invariant Reduced	Invariant Reduced			
		+ ScaledHR		+ ScaledHR			
Bench weight lifting - hard	$93.5\pm8.2$	$93.5\pm8.2$	$12.8\pm18.5$	$9.3\pm15.4$			
Bench weight lifting - light	$94.1\pm8.4$	$94.1 \pm 8.4$	$33.5\pm26.2$	$29.7 \pm 23.1$			
Bench weight lifting - moderate	$91.1 \pm 11.4$	$91.1 \pm 11.4$	$18.7\pm22.1$	$16.5 \pm 17.4$			
Bicep curls - hard	$92.4 \pm 12.7$	$92.4 \pm 12.7$	$28.3\pm26.9$	$26.8\pm26.3$			
Bicep curls - light	$92.8\pm 6.6$	$92.9\pm6.5$	$18.5\pm20.0$	$22.3\pm22.3$			
Bicep curls - moderate	$91.2\pm9.3$	$91.3\pm9.3$	$17.5 \pm 20.2$	$15.1\pm18.8$			
Calisthenics - Crunches	$96.8 \pm 3.2$	$97.0 \pm 3.0$	$63.8\pm38.2$	$64.2\pm38.0$			
Calisthenics - Sit ups	$94.8 \pm 5.3$	$95.4 \pm 3.8$	$77.6\pm32.8$	$79.6\pm32.3$			
Cycling - Cycle hard - Cycle 80rpm	$89.5 \pm 7.8$	93.1 ± 7.4	$27.6\pm22.8$	$31.5 \pm 26.0$			
Cycling - Cycle light - Cycle 100rpm	$98.2 \pm 1.9$	$98.2 \pm 1.9$	$95.6 \pm 9.5$	$95.2 \pm 8.7$			
Cycling - Cycle light - Cycle 60rpm	$98.8 \pm 1.4$	$98.8 \pm 1.4$	$89.2 \pm 21.8$	$89.3 \pm 21.9$			
Cycling - Cycle light - Cycle 80rpm	$97.4\pm2.5$	$97.4\pm2.5$	$33.8\pm26.7$	$36.4 \pm 31.0$			
Cycling - Cycle moderate - Cycle 80rpm	$92.2 \pm 5.2$	$94.6 \pm 4.0$	$27.7 \pm 24.3$	$26.7 \pm 20.0$			
Lying down	$99.7 \pm 0.7$	$99.7 \pm 0.7$	$95.8 \pm 8.6$	$95.8 \pm 8.7$			
Rowing - Rowing hard - Rowing 30spm	$85.0 \pm 13.6$	87.4 ± 14.1	$30.7 \pm 24.5$	$26.5 \pm 22.9$			
Rowing - Rowing light - Rowing 30spm	$89.1 \pm 10.4$	$89.9\pm9.8$	$31.0 \pm 17.2$	$31.0 \pm 20.5$			
Rowing - Rowing moderate - Rowing 30spm	$80.8 \pm 13.9$	84.0 ± 12.2	$23.2 \pm 17.7$	$27.0 \pm 22.0$			
Running - Treadmill 4mph - Treadmill 0	$96.5 \pm 2.6$	$96.5 \pm 2.6$	$53.1 \pm 36.5$	$54.1 \pm 38.2$			
Running - Treadmill 5mph - Treadmill 0	$93.7 \pm 3.9$	$93.8 \pm 4.1$	$54.5 \pm 21.0$	$62.3 \pm 24.2$			
Running - Treadmill 6mph - Treadmill 0	$90.0 \pm 12.0$	$90.2 \pm 12.2$	$54.8 \pm 26.0$	$57.5 \pm 34.1$			
Sitting	$96.6 \pm 3.3$	$96.8 \pm 3.0$	$45.9 \pm 34.6$	$45.3 \pm 34.8$			
Sitting - Fidget feet legs	$95.8 \pm 4.4$	$95.8 \pm 4.4$	$70.3 \pm 25.0$	$70.7 \pm 25.0$			
Sitting - Fidget hands arms	$91.9 \pm 6.2$	$92.0 \pm 6.2$	$50.1 \pm 28.1$	$45.8 \pm 29.4$			
Stairs - Ascend stairs	$91.4 \pm 5.7$	$90.8 \pm 5.9$	$68.9 \pm 25.1$	$67.2 \pm 25.6$			
Stairs - Descend stairs	$90.8 \pm 7.0$	$91.1 \pm 6.4$	$58.5 \pm 22.0$	$55.4 \pm 24.5$			
Standing	$97.0 \pm 3.2$	$96.9 \pm 3.4$	$81.6 \pm 20.9$	81.3 ± 20.9			
Walking - Treadmill 2mph - Treadmill 0	$95.9 \pm 4.0$	$95.9 \pm 4.0$	$70.1 \pm 21.1$	$70.6 \pm 20.2$			
Walking - Treadmill 3mph - Treadmill 0	$83.8 \pm 8.4$	$91.2 \pm 7.0$	$19.0 \pm 19.1$	$28.1 \pm 18.6$			
Walking - Treadmill 3mph - Treadmill 3 - light	$75.4 \pm 13.8$	$87.6 \pm 10.4$	$20.6 \pm 17.2$	$24.7 \pm 24.4$			
Walking - Treadmill 3mph - Treadmill 6 - moderate	$78.0 \pm 9.8$	$88.5 \pm 7.0$	$13.5 \pm 13.7$	$33.6 \pm 23.1$			
Walking - Treadmill 3mph - Treadmill 9 - hard	$88.4 \pm 7.8$	$93.7 \pm 3.7$	$24.8 \pm 15.4$	$47.8 \pm 29.7$			
kneeling	$96.6 \pm 3.2$	$96.5 \pm 3.7$	$96.7 \pm 3.5$	$96.5 \pm 3.8$			
Carrying groceries	90.5 ± 8.3	90.8 ± 8.1	56.1 ± 23.8	57.3 ± 27.3			
Doing dishes	$85.2 \pm 5.6$	85.1 ± 5.5	$52.4 \pm 20.6$	53.9 ± 19.4			
Gardening	79.9 ± 11.7	80.5 ± 10.9	$23.1 \pm 25.1$	$22.8 \pm 24.2$			
Ironing	$84.4 \pm 6.6$	85.6 ± 5.6	$50.5 \pm 23.9$	$50.5 \pm 24.1$			
Making the bed	64.2 ± 9.9	$69.2 \pm 10.4$	$36.0 \pm 14.9$	35.6 ± 15.2			
Mopping	$69.6 \pm 10.1$	$71.1 \pm 10.0$	$31.0 \pm 14.1$	30.6 ± 12.0			
Playing videogames	98.9 ± 1.5	98.9 ± 1.5	$58.9 \pm 37.8$	$57.0 \pm 36.7$			
Scrubbing a surface	85.9 ± 12.1	86.4 ± 11.2	35.9 ± 27.7	35.9 ± 27.7			
Stacking groceries	71.1 ± 14.6	$73.9 \pm 12.5$	34.0 ± 19.0	33.5 ± 18.4			
Sweeping	$68.8 \pm 12.4$	70.1 ± 11.6	32.8 ± 17.5	31.2 ± 17.9			
Typing	97.2 ± 2.6	97.4 ± 2.0	/0.8 ± 2/.3	$67.9 \pm 30.5$			
vacuuming	75.9 ± 9.7	1/1.8 ± 1.9	$51.8 \pm 18.8$	$53.0 \pm 17.6$			
Walking around block	91.6 ± 5.6	91.6 ± 5.1	29.4 ± 16.0	3/.8 ± 16.8			
Washing windows	$72.9 \pm 9.0$	74.6 ± 7.9	$44.8 \pm 20.7$	$45.3 \pm 20.8$			
Watching TV	$98.6 \pm 1.6$	98.6 ± 1.6	$42.4 \pm 37.6$	$39.1 \pm 36.6$			
Weeding	/4.8 ± 8.6	75.9 ± 8.8	$1/.1 \pm 1/.6$	$16.4 \pm 18.3$			
wiping/Dusting	68./±12.6	/3.8 ± 11.4	37.6 ± 18.5	37.0 ± 18.0			
Writing	$96.3 \pm 3.1$	$96.3 \pm 3.1$	63.8 ± 38.9	63.9 ± 38.9			
taking out trash	$93.5 \pm 8.2$	$67.3 \pm 13.5$	$25.7 \pm 15.3$	$26.9 \pm 15.8$			

Table A7-16: F-Measure obtained when adding the *ScaledHR* feature to *the invariant reduced* feature set using the C4.5 classifier. The sliding window length for accelerometer data and heart rate is 5.6s. The activities to recognize are the 51 contained in the MIT dataset without including the *unknown* class.

# **Appendix A8: Activity Recognition – Final Performance**

Activity	Sul	bject Depend	lent	Sub	ject Indepen	dent
· ·	TP Rate	FP Rate	F-Measure	TP Rate	FP Rate	F-Measure
Bench weight lifting - hard	$93.3 \pm 8.3$	$0.1 \pm 0.1$	$93.5 \pm 8.2$	$14.5 \pm 22.4$	$0.7 \pm 0.8$	$12.8 \pm 18.5$
Bench weight lifting - light	$93.6 \pm 9.6$	$0.1 \pm 0.1$	$94.1 \pm 8.4$	$38.8 \pm 37.0$	$0.8 \pm 0.8$	$33.5 \pm 26.2$
Bench weight lifting - moderate	$91.3 \pm 10.9$	$0.1 \pm 0.2$	91.1 ± 11.4	$28.4 \pm 36.4$	$1.1 \pm 1.3$	$18.7 \pm 22.1$
Bicep curls - hard	$94.1 \pm 10.0$	$0.1 \pm 0.2$	92.4 ± 12.7	$43.0 \pm 40.5$	$1.6 \pm 1.2$	$28.3 \pm 26.9$
Bicep curls - light	$91.9 \pm 8.1$	$0.1 \pm 0.1$	$92.8 \pm 6.6$	$20.5 \pm 31.7$	$0.6 \pm 0.9$	$18.5 \pm 20.0$
Bicep curls - moderate	$91.6 \pm 9.9$	$0.2 \pm 0.2$	$91.2 \pm 9.3$	$21.8 \pm 30.8$	$1.1 \pm 1.1$	$17.5 \pm 20.2$
Calisthenics - Crunches	$96.2 \pm 4.4$	$0.0\pm0.0$	$96.8 \pm 3.2$	$66.6 \pm 38.4$	$0.6 \pm 1.2$	$63.8 \pm 38.2$
Calisthenics - Sit ups	$96.6 \pm 3.1$	$0.1 \pm 0.2$	$94.8 \pm 5.3$	$80.5 \pm 35.0$	$0.2 \pm 0.2$	$77.6 \pm 32.8$
Cycling - Cycle hard - Cycle 80rpm	$88.4 \pm 8.7$	$0.1 \pm 0.1$	$89.5 \pm 7.8$	33.4 ± 31.9	$1.1 \pm 1.4$	$27.6 \pm 22.8$
Cycling - Cycle light - Cycle 100rpm	$98.1 \pm 2.3$	$0.0\pm0.0$	$98.2 \pm 1.9$	$97.0 \pm 5.8$	$0.1 \pm 0.4$	$95.6 \pm 9.5$
Cycling - Cycle light - Cycle 60rpm	$99.5 \pm 1.2$	$0.0\pm0.0$	$98.8 \pm 1.4$	$87.6 \pm 25.2$	$0.1 \pm 0.2$	$89.2 \pm 21.8$
Cycling - Cycle light - Cycle 80rpm	$97.5 \pm 3.2$	$0.1 \pm 0.1$	$97.4 \pm 2.5$	$41.6 \pm 39.1$	$1.3 \pm 1.4$	$33.8 \pm 26.7$
Cycling - Cycle moderate - Cycle 80rpm	$92.6 \pm 5.2$	$0.2 \pm 0.1$	$92.2 \pm 5.2$	$32.6 \pm 32.0$	$1.2 \pm 1.0$	$27.7 \pm 24.3$
Lying down	$99.9 \pm 0.3$	$0.0 \pm 0.1$	$99.7 \pm 0.7$	95.9 ± 12.3	$0.3 \pm 0.6$	$95.8 \pm 8.6$
Rowing - Rowing hard - Rowing 30spm	$85.2 \pm 14.2$	$0.2 \pm 0.2$	85.0 ± 13.6	$34.2 \pm 33.5$	$1.1 \pm 0.8$	$30.7 \pm 24.5$
Rowing - Rowing light - Rowing 30spm	$88.9 \pm 10.3$	$0.2 \pm 0.2$	89.1 ± 10.4	$41.9 \pm 32.3$	$1.8 \pm 1.2$	$31.0 \pm 17.2$
Rowing - Rowing moderate - Rowing 30spm	$81.4 \pm 13.7$	$0.4 \pm 0.3$	80.8 ± 13.9	$24.0 \pm 23.6$	$1.0 \pm 0.9$	$23.2 \pm 17.7$
Running - Treadmill 4mph - Treadmill 0	$97.7 \pm 2.9$	$0.1 \pm 0.1$	$96.5 \pm 2.6$	$57.4 \pm 39.6$	$0.8 \pm 1.0$	$53.1 \pm 36.5$
Running - Treadmill 5mph - Treadmill 0	93.7 + 4.8	$0.1 \pm 0.1$	93.7 + 3.9	56.8 + 29.1	$0.9 \pm 0.7$	54.5 + 21.0
Running - Treadmill 6mph - Treadmill 0	88.4 ± 15.1	$0.1 \pm 0.1$	$90.0 \pm 12.0$	$64.0 \pm 36.0$	$0.7 \pm 0.8$	$54.8 \pm 26.0$
Sitting	97.4 + 4.1	$0.1 \pm 0.1$	96.6 + 3.3	58.1 + 43.0	$1.3 \pm 1.2$	45.9 + 34.6
Sitting - Fidget feet legs	95.3 + 5.3	$0.0 \pm 0.1$	$95.8 \pm 4.4$	66.7 + 30.2	$0.2 \pm 0.2$	70.3 + 25.0
Sitting - Fidget hands arms	934 + 80	$0.1 \pm 0.1$	$919 \pm 62$	57.8 + 32.6	$0.9 \pm 0.9$	$50.1 \pm 28.1$
Stairs - Ascend stairs	$91.1 \pm 6.7$	$0.2 \pm 0.2$	91.4 + 5.7	$71.8 \pm 28.6$	$0.5 \pm 0.5$	68.9 + 25.1
Stairs - Descend stairs	$90.7 \pm 6.6$	$0.2 \pm 0.2$	$90.8 \pm 7.0$	59.6 + 26.4	$0.7 \pm 0.8$	585 + 220
Standing	$97.9 \pm 3.2$	$0.1 \pm 0.1$	$97.0 \pm 3.2$	863+211	$0.7 \pm 0.0$ 0.5 ± 0.9	$81.6 \pm 20.9$
Walking - Treadmill 2mph - Treadmill ()	$968 \pm 47$	$0.1 \pm 0.1$	$959 \pm 40$	$70.9 \pm 25.5$	$0.5 \pm 0.9$	$70.1 \pm 20.9$
Walking - Treadmill 3mph - Treadmill 0	$845 \pm 102$	$0.1 \pm 0.1$	838+84	$20.1 \pm 23.3$	$1.2 \pm 1.0$	$19.0 \pm 19.1$
Walking - Treadmill 3mph - Treadmill 3 - light	$762 \pm 10.2$	$0.4 \pm 0.2$ 0.6 + 0.3	$754 \pm 138$	244 + 258	$1.2 \pm 1.0$ 1 6 ± 0.9	$20.6 \pm 17.2$
Walking - Treadmill 3mph - Treadmill 6 - moderate	$77.0 \pm 10.3$	$0.0 \pm 0.0$ 0.5 ± 0.3	$78.0 \pm 9.8$	$148 \pm 188$	$1.0 \pm 0.9$ $1.5 \pm 1.0$	$135 \pm 137$
Walking - Treadmill 3mph - Treadmill 9 - hard	$88.4 \pm 9.5$	$0.3 \pm 0.3$	$88.4 \pm 7.8$	$27.7 \pm 24.8$	$1.5 \pm 1.0$ 1.5 ± 1.5	$24.8 \pm 15.4$
kneeling	$97.3 \pm 3.7$	$0.0 \pm 0.2$ 0.0 + 0.1	$96.6 \pm 3.2$	$97.3 \pm 5.0$	$1.5 \pm 1.5$ 0.0 + 0.1	$967 \pm 35$
Carrying groceries	$91.1 \pm 7.6$	$0.0 \pm 0.1$	$90.5 \pm 8.3$	$567 \pm 3.0$	$0.0 \pm 0.1$	$56.1 \pm 23.8$
Doing dishes	$91.1 \pm 7.0$ 85.1 + 7.3	$0.2 \pm 0.2$	$90.3 \pm 0.5$ 85.2 + 5.6	$55.9 \pm 25.4$	$1.2 \pm 0.9$	$50.1 \pm 25.0$ $52.4 \pm 20.6$
Gardening	$79.6 \pm 11.8$	$0.4 \pm 0.1$ 0.4 ± 0.3	$79.9 \pm 11.7$	$20.7 \pm 25.4$	$1.2 \pm 0.9$ $1.0 \pm 0.8$	$32.4 \pm 20.0$ 23.1 + 25.1
Ironing	$75.0 \pm 11.0$ 85.4 + 6.3	$0.4 \pm 0.3$	84.4 + 6.6	$53.3 \pm 28.4$	$1.0 \pm 0.0$ $1.3 \pm 0.8$	$50.5 \pm 23.1$
Making the hed	$62.4 \pm 0.3$	$0.4 \pm 0.3$	$64.4 \pm 0.0$	$33.3 \pm 28.4$	$1.3 \pm 0.8$	$36.0 \pm 14.9$
Manning the bed	$02.2 \pm 10.3$ 72.4 ± 11.8	$0.3 \pm 0.4$	$69.6 \pm 10.1$	$40.3 \pm 18.3$	$2.0 \pm 0.9$	$30.0 \pm 14.9$
Playing videogemes	$72.4 \pm 11.0$	$0.9 \pm 0.3$	$09.0 \pm 10.1$	$33.3 \pm 10.4$	$1.0 \pm 0.9$	$31.0 \pm 14.1$ 58.0 ± 27.8
Scrubbing a surface	$99.0 \pm 2.3$ 85.0 ± 12.9	$0.0 \pm 0.0$	96.9 ± 1.3	$40.0 \pm 35.4$	$1.2 \pm 1.2$ $1.5 \pm 0.8$	$36.9 \pm 37.8$ $35.9 \pm 27.7$
Scrubbing a surface	$63.0 \pm 12.9$	$0.3 \pm 0.3$	$63.9 \pm 12.1$	$40.0 \pm 33.4$	$1.3 \pm 0.8$	$33.9 \pm 27.7$
Stacking groceries	$71.0 \pm 10.0$	$0.0 \pm 0.3$	$71.1 \pm 14.0$	$33.8 \pm 20.9$	$1.2 \pm 0.8$	$34.0 \pm 19.0$
Turking	$08.3 \pm 13.2$	$0.7 \pm 0.3$	$08.8 \pm 12.4$	$34.4 \pm 20.6$	$1.3 \pm 0.4$	$32.8 \pm 17.3$
Typing	$98.3 \pm 2.4$	$0.1 \pm 0.1$	97.2 ± 2.6	$75.0 \pm 29.4$	$0.8 \pm 1.0$	$70.8 \pm 27.3$
Vacuuming Waltring around block	$74.5 \pm 10.1$	$0.5 \pm 0.3$	/5.9 ± 9.7	$55.4 \pm 25.9$	$1.1 \pm 0.6$	$51.8 \pm 18.8$
Walking around block	90.4 ± 7.7	$0.2 \pm 0.1$	$91.6 \pm 5.6$	$34.9 \pm 18.0$	$3.1 \pm 2.5$	$29.4 \pm 16.0$
wasning windows	/1.2 ± 9.4	$0.6 \pm 0.2$	12.9 ± 9.0	$44.2 \pm 21.5$	$1.2 \pm 0.7$	$44.8 \pm 20.7$
Watching I V	$98.2 \pm 2.5$	$0.0 \pm 0.0$	98.6 ± 1.6	$48.2 \pm 43.6$	$1.5 \pm 1.6$	$42.4 \pm 37.6$
Weeding	$/5.2 \pm 11.0$	$0.6 \pm 0.3$	/4.8 ± 8.6	$15.5 \pm 19.1$	$1.1 \pm 0.7$	$17.1 \pm 17.6$
wiping/Dusting	$69.4 \pm 13.9$	$0.8 \pm 0.3$	08./±12.6	59.4 ± 21.6	$1.5 \pm 0.6$	37.6 ± 18.5
Writing	$97.3 \pm 3.0$	$0.1 \pm 0.1$	$96.3 \pm 3.1$	$61./ \pm 40.6$	$0.3 \pm 0.4$	$63.8 \pm 38.9$
taking out trash	$02.2 \pm 13.6$	$0.7 \pm 0.3$	$04.4 \pm 12.4$	$23.0 \pm 14.1$	$1.2 \pm 0.4$	$23.7 \pm 13.3$

Table A8-1: Performance of the C4.5 classifier in recognizing the 51 activities in the MIT dataset (without the *unknown* class) using all the accelerometers (7) and the *invariant reduced* feature set over windows of 5.6s in length during subject dependent and independent evaluation.

Activity	Sul	piect Depend	lent	Sub	iect Indepen	dent
· ·	TP Rate	FP Rate	F-Measure	TP Rate	FP Rate	F-Measure
Bench weight lifting - hard	$90.8 \pm 9.3$	$0.1 \pm 0.1$	90.5 ± 7.2	$16.9 \pm 33.0$	$0.8 \pm 0.9$	$10.6 \pm 15.3$
Bench weight lifting - light	$93.5 \pm 7.7$	$0.1 \pm 0.2$	$92.2 \pm 7.9$	$41.3 \pm 41.9$	$1.1 \pm 1.3$	$31.2 \pm 28.3$
Bench weight lifting - moderate	$83.4 \pm 18.3$	$0.1 \pm 0.1$	85.4 ± 12.8	$22.0 \pm 33.1$	$0.7 \pm 0.7$	$15.4 \pm 20.8$
Bicen curls - hard	90.0 + 8.9	$0.2 \pm 0.2$	88.6 + 10.1	39.2 + 32.4	$1.7 \pm 1.3$	$27.0 \pm 18.0$
Bicen curls - light	911+88	$0.2 \pm 0.2$	$91.5 \pm 8.8$	32.4 + 37.4	14 + 14	$253 \pm 205$
Bicep curls - moderate	887+93	$0.2 \pm 0.2$	887+89	$162 \pm 191$	$0.7 \pm 0.6$	$174 \pm 176$
Calisthenics - Crunches	$961 \pm 51$	$0.0 \pm 0.0$	969 + 34	51.1 + 38.0	$0.7 \pm 0.8$	534 + 369
Calisthenics - Sit ups	$92.7 \pm 5.7$	$0.0 \pm 0.0$ 0.1 + 0.1	933 + 59	$341 \pm 337$	$0.6 \pm 0.8$	362 + 332
Cycling - Cycle hard - Cycle 80rpm	88.1 + 8.3	$0.2 \pm 0.2$	87.3 + 9.4	19.6 + 28.4	$1.3 \pm 1.3$	$16.8 \pm 21.2$
Cycling - Cycle light - Cycle 100rpm	$98.1 \pm 2.3$	$0.0 \pm 0.1$	$97.3 \pm 3.6$	$90.1 \pm 19.5$	$0.2 \pm 0.2$	88 1 + 18 5
Cycling - Cycle light - Cycle 60rpm	$98.1 \pm 3.4$	$0.0 \pm 0.1$ 0.0 + 0.1	$97.8 \pm 2.6$	$815 \pm 236$	$0.2 \pm 0.2$ 0.4 + 0.4	$80.0 \pm 16.0$
Cycling - Cycle light - Cycle 80rpm	$95.9 \pm 4.9$	$0.0 \pm 0.1$	$96.0 \pm 4.7$	$43.9 \pm 33.3$	$1.2 \pm 1.2$	$39.3 \pm 25.1$
Cycling Cycle moderate Cycle S0rpm	$91.2 \pm 6.2$	$0.1 \pm 0.1$	91.0 ± 5.9	$357 \pm 33.3$	$1.2 \pm 1.2$ 1.2 + 1.0	$30.6 \pm 18.5$
Lying down	$91.2 \pm 0.2$	$0.2 \pm 0.1$	$91.0 \pm 5.9$	$33.7 \pm 27.7$	$1.2 \pm 1.0$	$30.0 \pm 18.3$
Powing Powing hard Powing 20cpm	$99.9 \pm 0.4$ 81.0 ± 12.4	$0.0 \pm 0.0$	$99.6 \pm 0.4$	$89.0 \pm 20.9$	$0.3 \pm 0.8$	$90.8 \pm 13.3$
Rowing - Rowing light Rowing 30spm	$81.9 \pm 12.4$ $82.7 \pm 10.7$	$0.4 \pm 0.2$	$80.3 \pm 11.8$	$13.0 \pm 23.8$ 25.8 ± 22.0	$1.1 \pm 0.0$ $1.4 \pm 1.2$	$11.3 \pm 17.4$ 28.1 ± 25.2
Rowing - Rowing nght - Rowing Sospin	$32.7 \pm 10.7$ 75.5 ± 12.0	$0.3 \pm 0.2$	$34.4 \pm 10.0$ 74.2 ± 12.2	$33.8 \pm 32.9$	$1.4 \pm 1.2$ 1.1 + 1.0	$26.1 \pm 23.3$ $27.1 \pm 18.2$
Rowing - Rowing moderate - Rowing Jospin	$75.5 \pm 12.9$	$0.0 \pm 0.3$	$74.3 \pm 12.2$	$26.2 \pm 22.8$	$1.1 \pm 1.0$ 1.2 + 1.2	$27.1 \pm 10.3$ $26.9 \pm 25.9$
Running - Treadmill 4mph - Treadmill 0	$90.4 \pm 4.4$	$0.1 \pm 0.0$	$90.7 \pm 2.8$	$41.0 \pm 42.0$ 55.0 ± 22.1	$1.5 \pm 1.5$	$30.0 \pm 33.0$
Running - Treadmill Smph - Treadmill 0	$95.2 \pm 4.2$ 80.0 ± 17.2	$0.1 \pm 0.1$	$94.3 \pm 5.2$ 00.2 ± 12.8	$33.0 \pm 32.1$	$1.4 \pm 0.9$	$40.3 \pm 21.3$ $42.5 \pm 27.6$
Running - Headmin ompil - Headmin o	$07.9 \pm 17.2$	$0.1 \pm 0.1$	$90.3 \pm 13.8$	$47.3 \pm 33.2$	$0.0 \pm 0.0$	$42.3 \pm 27.0$
Sitting Eident fant lann	$97.0 \pm 4.8$	$0.1 \pm 0.1$	$93.0 \pm 3.4$	$60.9 \pm 39.8$	$1.2 \pm 1.3$	$31.3 \pm 33.9$
Sitting - Fidget leet legs	$95.8 \pm 1.2$	$0.1 \pm 0.1$	$93.0 \pm 0.2$	$36.4 \pm 27.9$	$0.5 \pm 0.6$	$37.2 \pm 22.4$
Sitting - Fidget hands arms	$95.6 \pm 5.1$	$0.1 \pm 0.1$	$94.7 \pm 4.4$	$38.2 \pm 32.7$	$0.6 \pm 0.7$	$36.8 \pm 28.5$
Stairs - Ascend stairs	$90.2 \pm 7.0$	$0.1 \pm 0.1$	$91.1 \pm 5.9$	$55.3 \pm 26.5$	$1.0 \pm 0.7$	$51.7 \pm 21.2$
Stairs - Descend stairs	$88.4 \pm 6.2$	$0.2 \pm 0.1$	$89.4 \pm 5.6$	$50.7 \pm 24.5$	$0.9 \pm 0.7$	$50.2 \pm 22.5$
Standing	95.8 ± 4.8	$0.0 \pm 0.1$	$96.2 \pm 3.6$	/3.1 ± 36.9	$0.7 \pm 1.3$	$6/.6 \pm 34.6$
Walking - Treadmill 2mph - Treadmill 0	$97.4 \pm 4.2$	$0.1 \pm 0.1$	96.4 ± 2.6	$69.8 \pm 31.1$	$0.8 \pm 1.3$	6/./±2/.5
Walking - Treadmill 3mph - Treadmill 0	$81.8 \pm 10.6$	$0.4 \pm 0.3$	82.3 ± 9.0	19.4 ± 22.7	$1.3 \pm 0.8$	18.3 ± 18.9
Walking - Treadmill 3mph - Treadmill 3 - light	$73.4 \pm 12.2$	$0.7 \pm 0.3$	$72.3 \pm 11.2$	$22.1 \pm 24.2$	$1.6 \pm 1.5$	$18.0 \pm 15.3$
Walking - Treadmill 3mph - Treadmill 6 - moderate	74.6 ± 13.7	$0.6 \pm 0.3$	$73.5 \pm 12.0$	$15.4 \pm 19.4$	$1.3 \pm 1.1$	$14.6 \pm 16.7$
Walking - Treadmill 3mph - Treadmill 9 - hard	83.4 ± 9.0	$0.3 \pm 0.2$	84.7 ± 8.1	$16.1 \pm 23.2$	$1.0 \pm 0.9$	$15.6 \pm 20.3$
kneeling	$97.4 \pm 2.9$	$0.0 \pm 0.0$	$96.9 \pm 2.2$	$81.6 \pm 35.4$	$0.6 \pm 1.2$	$77.3 \pm 33.9$
Carrying groceries	$93.4 \pm 7.0$	$0.3 \pm 0.2$	$90.9 \pm 8.1$	$57.8 \pm 31.3$	$1.3 \pm 1.5$	$53.6 \pm 26.5$
Doing dishes	$81.0 \pm 9.0$	$0.5 \pm 0.2$	$80.2 \pm 7.5$	$47.1 \pm 23.0$	$1.3 \pm 0.6$	$44.1 \pm 18.2$
Gardening	$80.1 \pm 12.2$	$0.5 \pm 0.3$	$78.4 \pm 11.4$	$19.4 \pm 22.0$	$1.0 \pm 0.9$	$20.5 \pm 20.3$
Ironing	$82.9 \pm 8.5$	$0.5 \pm 0.3$	$82.5 \pm 8.2$	$42.0 \pm 25.0$	$1.4 \pm 0.9$	$39.9 \pm 18.9$
Making the bed	$61.0 \pm 10.0$	$1.0 \pm 0.4$	$61.3 \pm 9.4$	$38.1 \pm 19.4$	$2.4 \pm 2.3$	$34.0 \pm 17.7$
Mopping	$68.8 \pm 13.7$	$0.9 \pm 0.4$	$66.5 \pm 12.8$	$29.5 \pm 12.0$	$2.0 \pm 1.3$	$28.1 \pm 9.8$
Playing videogames	$98.8 \pm 1.9$	$0.1 \pm 0.1$	$98.2 \pm 2.9$	$64.6 \pm 38.9$	$1.3 \pm 1.6$	$58.0 \pm 32.7$
Scrubbing a surface	$84.0 \pm 14.3$	$0.3 \pm 0.3$	$85.0 \pm 14.0$	$41.6 \pm 28.9$	$1.7 \pm 1.3$	$37.9 \pm 25.2$
Stacking groceries	$64.0 \pm 17.0$	$0.7 \pm 0.4$	$64.1 \pm 16.6$	$35.6 \pm 15.5$	$1.4 \pm 0.5$	$34.2 \pm 13.7$
Sweeping	$66.2 \pm 16.7$	$0.8 \pm 0.4$	$66.9 \pm 15.5$	$32.5 \pm 16.9$	$1.3 \pm 0.5$	$33.2 \pm 15.2$
Typing	$98.5 \pm 2.5$	$0.0 \pm 0.1$	$98.1 \pm 2.2$	$75.2 \pm 31.6$	$0.5 \pm 0.8$	$75.2 \pm 29.9$
Vacuuming	$73.5 \pm 11.4$	$0.6 \pm 0.3$	$74.7\pm10.8$	$52.3\pm22.3$	$1.1 \pm 0.6$	$51.6 \pm 19.2$
Walking around block	$90.0\pm7.0$	$0.2 \pm 0.2$	$90.4 \pm 6.0$	$24.6 \pm 15.2$	$2.8 \pm 2.3$	$21.4\pm12.6$
Washing windows	$69.1 \pm 11.6$	$0.6 \pm 0.2$	$71.7\pm10.1$	$45.0\pm24.1$	$1.3 \pm 0.7$	$44.2\pm24.0$
Watching TV	$97.5 \pm 3.1$	$0.0 \pm 0.0$	$98.3 \pm 1.7$	$52.7\pm43.1$	$0.6 \pm 0.5$	$49.5\pm38.2$
Weeding	$7\overline{3.5\pm15.5}$	$0.6 \pm 0.2$	$74.0\pm13.7$	$14.8\pm23.6$	$1.2 \pm 1.0$	$11.2 \pm 13.7$
Wiping/Dusting	$66.6 \pm 9.6$	$0.7 \pm 0.4$	$68.6 \pm 11.6$	$38.2\pm22.5$	$1.3\pm0.8$	$37.3 \pm 20.0$
Writing	$97.7 \pm 2.8$	$0.1 \pm 0.1$	$96.7 \pm 2.4$	$77.0 \pm 29.1$	$0.8 \pm 1.4$	$73.5 \pm 27.9$
taking out trash	$59.9 \pm 14.7$	$0.8 \pm 0.4$	$61.3 \pm 14.0$	$25.9 \pm 16.8$	$1.2 \pm 0.5$	$27.7 \pm 17.2$

Table A8-2: Performance of the C4.5 classifier in recognizing the 51 activities in the MIT dataset (without the *unknown* class) using the accelerometers hip, dominant wrist, and dominant foot and the *invariant reduced* feature set over windows of 5.6s in length during subject dependent and independent evaluation.

Activity	Sul	oject Depend	lent	Sub	ject Indepen	dent
· ·	TP Rate	FP Rate	F-Measure	TP Rate	FP Rate	F-Measure
Bench weight lifting - hard	$80.8 \pm 26.8$	$0.1 \pm 0.1$	82.5 ± 25.5	$5.8 \pm 8.6$	$0.4 \pm 0.4$	$6.9 \pm 11.0$
Bench weight lifting - light	90.5 ± 11.6	$0.2 \pm 0.2$	89.1 ± 12.4	$47.3 \pm 30.3$	$0.9 \pm 0.8$	$42.0 \pm 23.6$
Bench weight lifting - moderate	$80.6 \pm 24.5$	$0.2 \pm 0.2$	$81.0 \pm 23.5$	$23.3 \pm 29.2$	$0.7 \pm 0.6$	$18.7 \pm 19.3$
Bicep curls - hard	$82.1 \pm 18.6$	$0.3 \pm 0.2$	$82.1 \pm 17.1$	$39.6 \pm 21.9$	$1.3 \pm 1.1$	$36.1 \pm 19.7$
Bicep curls - light	$86.4 \pm 12.0$	$0.3 \pm 0.3$	86.2 ± 11.9	$32.5 \pm 30.9$	$0.9 \pm 0.9$	$29.2 \pm 24.6$
Bicep curls - moderate	$80.1 \pm 16.1$	$0.4 \pm 0.4$	$79.9 \pm 16.4$	$25.8 \pm 29.0$	$1.1 \pm 0.9$	$23.0 \pm 18.0$
Calisthenics - Crunches	$96.6 \pm 4.2$	$0.1 \pm 0.1$	$96.0 \pm 3.3$	$51.0 \pm 40.8$	$0.8 \pm 0.9$	$48.9 \pm 37.1$
Calisthenics - Sit ups	91.7 + 5.7	$0.1 \pm 0.1$	$91.6 \pm 5.2$	33.5 + 31.9	$0.7 \pm 1.2$	35.6 + 31.6
Cycling - Cycle hard - Cycle 80rpm	83.3 + 16.5	$0.3 \pm 0.2$	81.9 + 17.9	7.7 + 9.6	$0.9 \pm 1.2$	$9.7 \pm 11.0$
Cycling - Cycle light - Cycle 100rpm	$91.9 \pm 18.3$	$0.1 \pm 0.1$	92.2 + 17.7	579 + 296	12 + 20	$53.6 \pm 28.8$
Cycling - Cycle light - Cycle 60rpm	$96.2 \pm 4.2$	$0.1 \pm 0.2$	$95.0 \pm 4.1$	$64.2 \pm 32.3$	$0.8 \pm 0.7$	$59.3 \pm 25.3$
Cycling - Cycle light - Cycle 80rpm	$93.0 \pm 8.2$	$0.1 \pm 0.2$	933 + 83	32.7 + 29.7	19 + 13	247 + 191
Cycling - Cycle moderate - Cycle 80rpm	83 2 + 11 6	$0.1 \pm 0.2$	848+95	$23.1 \pm 20.6$	$1.2 \pm 0.8$	$21.6 \pm 18.8$
Lying down	$99.9 \pm 0.4$	$0.0 \pm 0.0$	$99.7 \pm 0.5$	$\frac{26.1}{86.4} \pm 21.8$	$0.6 \pm 1.0$	$87.7 \pm 14.0$
Rowing - Rowing hard - Rowing 30spm	$79.3 \pm 10.6$	$0.0 \pm 0.0$	$78.9 \pm 11.3$	$16.8 \pm 19.1$	$0.0 \pm 1.0$ 0.9 ± 0.7	$17.6 \pm 19.4$
Rowing - Rowing light - Rowing 30spm	$81.2 \pm 9.7$	$0.4 \pm 0.3$	81.2 + 9.5	$38.2 \pm 26.1$	$1.7 \pm 1.0$	$31.1 \pm 17.1$
Rowing - Rowing moderate - Rowing 30spm	$72.3 \pm 12.4$	$0.4 \pm 0.3$	$71.9 \pm 10.8$	$28.8 \pm 22.0$	$1.7 \pm 1.0$ $1.4 \pm 0.9$	$25.4 \pm 15.4$
Running - Treadmill 4mph - Treadmill 0	$95.6 \pm 3.7$	$0.0 \pm 0.2$ 0.1 + 0.1	953 + 32	$31.8 \pm 32.5$	$1.4 \pm 0.5$ 1.2 + 1.5	$29.5 \pm 30.6$
Running - Treadmill Smph - Treadmill 0	$91.8 \pm 5.6$	$0.1 \pm 0.1$	$91.2 \pm 5.2$	$51.0 \pm 32.0$ $55.5 \pm 35.9$	$1.2 \pm 1.3$ 1.1 + 0.8	$47.7 \pm 30.0$
Running - Treadmill 6mph - Treadmill 0	$83.3 \pm 25.0$	$0.2 \pm 0.2$	$83.1 \pm 24.2$	$57.6 \pm 34.6$	$1.1 \pm 0.0$ 1 1 + 1 2	$46.3 \pm 27.2$
Sitting	$07.3 \pm 3.6$	$0.1 \pm 0.1$	$96.3 \pm 3.4$	$42.5 \pm 42.8$	$1.1 \pm 1.2$	$34.2 \pm 33.0$
Sitting Fidget feet lage	$97.3 \pm 3.0$	$0.1 \pm 0.1$	$90.3 \pm 3.4$ 03.1 ± 3.0	$42.3 \pm 42.3$	$1.1 \pm 1.5$ $1.3 \pm 1.5$	$34.2 \pm 33.9$ $35.4 \pm 32.8$
Sitting Fidget hands arms	$94.3 \pm 4.3$	$0.1 \pm 0.1$	$93.1 \pm 3.9$	$40.1 \pm 35.0$	$1.3 \pm 1.3$	$35.4 \pm 32.8$
Stilling - Fluget Halids arms	$91.9 \pm 0.1$	$0.1 \pm 0.1$	$91.0 \pm 7.2$	$33.0 \pm 29.9$	$0.8 \pm 0.7$	$33.1 \pm 26.1$
Stairs - Ascend stairs	$86.2 \pm 7.4$	$0.2 \pm 0.1$	$87.3 \pm 0.0$	$30.8 \pm 33.0$	$1.4 \pm 1.6$	$47.0 \pm 31.0$
Staris - Descend stars	$87.2 \pm 7.0$	$0.2 \pm 0.2$	$87.3 \pm 0.8$	$31.5 \pm 27.6$	$1.4 \pm 1.0$	$44.8 \pm 22.0$ 24.0 ± 27.0
Walking Treadmill 2mph Treadmill 0	$95.0 \pm 4.4$	$0.1 \pm 0.1$	$95.5 \pm 4.1$	$40.0 \pm 42.0$ 56.4 ± 20.7	$0.8 \pm 0.7$	$34.9 \pm 37.0$ 51.5 ± 26.8
Walking - Treadmill 2mph - Treadmill 0	$90.2 \pm 3.0$	$0.1 \pm 0.1$	93.0 ± 2.7	$30.4 \pm 30.7$	$1.1 \pm 1.0$	$31.3 \pm 20.8$
Walking - Treadmill 2mph - Treadmill 2	$62.0 \pm 9.2$	$0.4 \pm 0.3$	$62.3 \pm 9.5$	$13.0 \pm 10.4$	$1.5 \pm 1.2$	$13.1 \pm 13.4$ 12.7 ± 12.6
Walking - Treadmill 2mph - Treadmill 6 - moderate	$72.1 \pm 12.0$	$0.0 \pm 0.3$	$71.0 \pm 11.0$ $74.2 \pm 11.1$	$12.1 \pm 14.3$	$1.1 \pm 0.7$	$12.7 \pm 13.0$ $17.2 \pm 17.0$
Walking - Ireadmill 3mph - Ireadmill 0 - moderate	$75.2 \pm 12.2$	$0.6 \pm 0.3$	$74.2 \pm 11.1$	$19.1 \pm 22.0$	$1.2 \pm 1.3$	$17.3 \pm 17.0$
waiking - meadinin Shiph - meadinin 9 - hard	$05.2 \pm 9.1$	$0.3 \pm 0.2$	$04.6 \pm 4.4$	$23.7 \pm 23.9$	$1.0 \pm 1.3$	$20.4 \pm 19.0$
Comming	$93.4 \pm 3.0$	$0.1 \pm 0.1$	$94.0 \pm 4.4$	$51.6 \pm 53.4$	$1.2 \pm 1.1$	$20.3 \pm 20.7$
Carrying groceries	$90.2 \pm 9.1$	$0.3 \pm 0.2$	88./±8.0	$52.6 \pm 32.0$	$1.3 \pm 1.1$	$49.5 \pm 25.7$
Doing disnes	$77.8 \pm 11.3$	$0.6 \pm 0.3$	$77.5 \pm 11.7$	$42.8 \pm 24.3$	$1.0 \pm 1.0$	$38.4 \pm 18.2$
Gardening	$74.4 \pm 17.6$	$0.6 \pm 0.3$	$73.5 \pm 16.3$	$22.7 \pm 25.2$	$1.0 \pm 0.6$	$23.4 \pm 23.2$
Ironing Mating the hel	78.1 ± 11.0	$0.6 \pm 0.3$	$77.1 \pm 11.5$	$44.3 \pm 22.0$	$1.4 \pm 0.7$	$42.8 \pm 18.0$
Making the bed	$55.3 \pm 10.0$	$1.0 \pm 0.3$	$50.3 \pm 8.7$	$35.0 \pm 15.8$	$2.0 \pm 1.1$	$32.1 \pm 12.5$
Mopping	$60.5 \pm 12.6$	$1.0 \pm 0.4$	$59.6 \pm 11.6$	$51.8 \pm 14.4$	$2.1 \pm 1.0$	$29.4 \pm 12.5$
Playing videogames	$98.6 \pm 2.9$	$0.0 \pm 0.1$	$98.4 \pm 2.5$	$57.7 \pm 38.1$	$0.9 \pm 1.4$	$56.2 \pm 36.2$
Scrubbing a surface	$73.8 \pm 11.3$	$0.6 \pm 0.3$	$74.4 \pm 11.8$	$37.4 \pm 19.6$	$1.3 \pm 0.7$	$36.6 \pm 16.2$
Stacking groceries	$56.7 \pm 18.0$	$0.8 \pm 0.4$	57.9 ± 18.1	$30.2 \pm 17.0$	$1.6 \pm 0.9$	$27.2 \pm 12.9$
Sweeping	$63.9 \pm 16.9$	$0.9 \pm 0.5$	62.7 ± 14.8	33.8 ± 15.9	$1.4 \pm 0.6$	35.0 ± 15.1
Typing	$98.0 \pm 2.3$	$0.1 \pm 0.1$	97.6 ± 2.4	84.8 ± 25.6	$0.4 \pm 0.7$	84.5 ± 23.0
Vacuuming	$72.9 \pm 13.0$	$0.6 \pm 0.3$	$/3.0 \pm 12.1$	$51.2 \pm 18.5$	$1.0 \pm 0.5$	$51.6 \pm 14.0$
Walking around block	$90.1 \pm 6.0$	$0.2 \pm 0.2$	$90.2 \pm 5.6$	$30.8 \pm 21.2$	$2.7 \pm 2.4$	27.1 ± 17.5
Washing windows	$65.2 \pm 8.6$	$0.7 \pm 0.3$	67.8 ± 8.7	$43.0 \pm 20.4$	$1.2 \pm 0.6$	43.2 ± 19.8
watching I V	$96.2 \pm 4.3$	$0.0 \pm 0.1$	$97.0 \pm 3.1$	$61.6 \pm 42.2$	$1.3 \pm 1.6$	$52.8 \pm 36.4$
Weeding	$69.3 \pm 13.3$	$0.7 \pm 0.3$	$69.2 \pm 12.6$	$15.4 \pm 23.3$	$1.0 \pm 0.4$	$12.6 \pm 13.6$
Wiping/Dusting	$60.6 \pm 16.0$	$0.7 \pm 0.3$	$63.0 \pm 14.0$	$35.2 \pm 17.0$	$1.4 \pm 0.6$	$35.3 \pm 15.9$
Writing	97.9 ± 3.2	$0.1 \pm 0.1$	97.0 ± 2.9	$74.2 \pm 37.5$	$0.5 \pm 0.7$	$70.9 \pm 35.0$
taking out trash	$52.1 \pm 11.6$	$1.0 \pm 0.3$	$53.3 \pm 11.0$	$24.4 \pm 13.0$	$1.4 \pm 0.5$	$25.5 \pm 13.2$

Table A8-3: Performance of the C4.5 classifier in recognizing the 51 activities in the MIT dataset (without the *unknown* class) using the accelerometers at the hip and wrist, and the *invariant reduced* feature set over windows of 5.6s in length during subject dependent and independent evaluation.

Activity	Su	oject Depend	lent	Sub	ject Indepen	dent
ν. ·	TP Rate	FP Rate	F-Measure	TP Rate	FP Rate	F-Measure
Bench weight lifting - hard	$91.3 \pm 8.8$	$0.1 \pm 0.1$	$91.6 \pm 7.9$	$17.7 \pm 30.5$	$0.8 \pm 0.6$	$11.5 \pm 19.7$
Bench weight lifting - light	$92.5 \pm 10.8$	$0.1 \pm 0.2$	$91.2 \pm 11.3$	$21.1 \pm 31.1$	$1.1 \pm 1.1$	$18.8 \pm 23.7$
Bench weight lifting - moderate	87.1 ± 15.3	$0.2 \pm 0.1$	87.3 ± 12.7	$11.5 \pm 28.7$	$0.5 \pm 0.4$	$8.0 \pm 17.4$
Bicep curls - hard	91.5 ± 7.3	$0.2 \pm 0.2$	$90.0 \pm 8.8$	$22.1 \pm 33.8$	$1.2 \pm 1.4$	$14.8 \pm 19.7$
Bicep curls - light	$90.0 \pm 10.0$	$0.2 \pm 0.2$	$90.9 \pm 9.0$	33.5 ± 32.1	$1.1 \pm 1.5$	$32.1 \pm 24.1$
Bicep curls - moderate	$89.0 \pm 8.6$	$0.2 \pm 0.2$	$88.7 \pm 8.8$	$42.1 \pm 41.0$	$1.3 \pm 1.3$	$30.4 \pm 25.6$
Calisthenics - Crunches	$97.4 \pm 3.6$	$0.0 \pm 0.0$	$97.5 \pm 2.8$	$53.2 \pm 40.0$	$0.4 \pm 0.7$	$56.4 \pm 39.0$
Calisthenics - Sit ups	$92.1 \pm 6.6$	$0.1 \pm 0.1$	$92.4 \pm 5.7$	$42.6 \pm 32.4$	$0.6 \pm 0.4$	44.7 ± 31.6
Cycling - Cycle hard - Cycle 80rpm	$86.9 \pm 9.3$	$0.2 \pm 0.2$	$86.6 \pm 7.8$	$15.9 \pm 25.1$	$1.1 \pm 1.0$	$14.0 \pm 18.7$
Cycling - Cycle light - Cycle 100rpm	$97.8 \pm 3.2$	$0.0 \pm 0.0$	$97.0 \pm 3.7$	$90.4 \pm 20.0$	$0.2 \pm 0.3$	89.4 ± 15.5
Cycling - Cycle light - Cycle 60rpm	$98.3 \pm 3.3$	$0.0 \pm 0.1$	$97.9 \pm 2.9$	$80.9 \pm 22.8$	$0.3 \pm 0.4$	$81.1 \pm 16.4$
Cycling - Cycle light - Cycle 80rpm	$95.6 \pm 4.6$	$0.1 \pm 0.1$	$95.8 \pm 4.4$	$48.5 \pm 38.1$	$1.3 \pm 1.2$	$41.0 \pm 27.4$
Cycling - Cycle moderate - Cycle 80rpm	$89.2 \pm 6.9$	$0.2 \pm 0.1$	$89.7 \pm 6.1$	$36.9 \pm 32.0$	$1.4 \pm 1.1$	$31.1 \pm 24.6$
Lying down	$100.0 \pm 0.2$	$0.0 \pm 0.0$	$99.9 \pm 0.3$	$89.5 \pm 20.4$	$0.6 \pm 1.0$	89.7 ± 13.2
Rowing - Rowing hard - Rowing 30spm	79.0 ± 17.5	$0.4 \pm 0.3$	77.0 ± 16.6	$21.7 \pm 23.5$	$1.0 \pm 0.9$	$18.1 \pm 17.1$
Rowing - Rowing light - Rowing 30spm	$81.9 \pm 9.4$	$0.3 \pm 0.2$	$83.6 \pm 9.6$	$33.4 \pm 31.9$	$1.2 \pm 1.1$	$29.6 \pm 23.7$
Rowing - Rowing moderate - Rowing 30spm	$72.3 \pm 17.5$	$0.5 \pm 0.3$	72.1 ± 15.2	$30.1 \pm 32.2$	$1.4 \pm 1.2$	$23.7 \pm 17.7$
Running - Treadmill 4mph - Treadmill 0	97.1 + 4.0	$0.0 \pm 0.1$	97.2 + 3.0	40.2 + 42.2	$1.8 \pm 2.1$	33.9 + 34.1
Running - Treadmill 5mph - Treadmill 0	95.5 + 3.8	$0.1 \pm 0.1$	95.2 + 3.0	52.6 + 30.6	$1.2 \pm 0.9$	46.5 + 21.4
Running - Treadmill 6mph - Treadmill 0	$93.2 \pm 7.1$	$0.1 \pm 0.1$	$93.4 \pm 4.9$	$48.8 \pm 33.8$	$0.6 \pm 0.5$	$46.8 \pm 27.3$
Sitting	97.5 + 3.9	$0.1 \pm 0.1$	95.9 + 5.7	25.8 + 37.0	$1.2 \pm 1.5$	22.8 + 33.8
Sitting - Fidget feet legs	943+55	$0.1 \pm 0.1$	942 + 57	60.9 + 30.7	$0.4 \pm 0.5$	61.1 + 24.3
Sitting - Fidget hands arms	$961 \pm 47$	$0.1 \pm 0.1$	$94.6 \pm 4.2$	$50.0 \pm 32.0$	$0.9 \pm 0.8$	$45.0 \pm 27.8$
Stairs - Ascend stairs	$90.1 \pm 4.7$ $90.5 \pm 7.4$	$0.1 \pm 0.1$ 0.2 + 0.1	$90.4 \pm 6.2$	$435 \pm 259$	$13 \pm 11$	$39.4 \pm 22.6$
Stairs - Descend stairs	885+74	$0.2 \pm 0.1$	$889 \pm 70$	442 + 223	$1.0 \pm 0.9$	432 + 202
Standing	$95.3 \pm 5.6$	$0.2 \pm 0.2$	$95.5 \pm 4.1$	$46.9 \pm 41.6$	$0.9 \pm 0.8$	$38.8 \pm 29.4$
Walking - Treadmill 2mph - Treadmill ()	$97.1 \pm 3.0$	$0.1 \pm 0.1$	$95.6 \pm 3.3$	$53.8 \pm 35.6$	$0.7 \pm 0.0$	$56.0 \pm 29.4$ 56.1 + 32.0
Walking - Treadmill 3mph - Treadmill 0	$79.9 \pm 10.3$	$0.1 \pm 0.1$ 0.4 + 0.3	80.6 + 10.1	$245 \pm 25.0$	$1.7 \pm 1.6$	$20.1 \pm 32.0$ $20.1 \pm 17.2$
Walking - Treadmill 3mph - Treadmill 3 - light	$70.7 \pm 10.9$	$0.4 \pm 0.3$ 0.7 + 0.3	$70.6 \pm 14.2$	$20.1 \pm 28.6$	$1.7 \pm 1.0$ 14 + 13	$154 \pm 18.6$
Walking - Treadmill 3mph - Treadmill 6 - moderate	$75.1 \pm 13.2$	$0.7 \pm 0.3$	$70.0 \pm 14.2$ $72.9 \pm 12.7$	$13.3 \pm 21.4$	$1.4 \pm 1.3$ 1.1 + 1.2	$12.7 \pm 10.0$ 12.7 ± 14.8
Walking - Treadmill 3mph - Treadmill 9 - hard	81.9 ± 11.0	$0.7 \pm 0.3$	$83.4 \pm 10.0$	$13.5 \pm 21.4$ 21.9 ± 26.1	$1.1 \pm 1.2$ 15 ± 13	$12.7 \pm 14.0$ 19.7 + 19.5
kneeling	$967 \pm 31$	$0.0 \pm 0.2$	$964 \pm 30$	$72.0 \pm 39.8$	$1.0 \pm 1.0$	$65.3 \pm 38.7$
Carrying groceries	$89.1 \pm 10.4$	$0.0 \pm 0.0$	88.6 + 9.5	44.4 + 27.2	$1.0 \pm 1.0$ 1 8 + 1 4	$40.2 \pm 20.1$
Doing dishes	$80.9 \pm 12.0$	$0.5 \pm 0.2$	$79.4 \pm 11.1$	$38.5 \pm 24.5$	$1.0 \pm 1.4$ $1.2 \pm 0.7$	$37.5 \pm 22.6$
Gardening	$77.0 \pm 10.3$	$0.0 \pm 0.3$	$76.1 \pm 11.1$	$14.7 \pm 20.4$	$1.2 \pm 0.7$ 1 4 + 1 2	$15.8 \pm 20.7$
Ironing	82 1 + 8 8	$0.0 \pm 0.3$	$81.3 \pm 9.4$	$42.0 \pm 28.4$	$1.4 \pm 1.2$ $1.6 \pm 0.9$	$15.0 \pm 20.7$ 37.1 + 23.1
Making the bed	58.8 + 9.6	$0.3 \pm 0.4$	$585 \pm 9.4$	$36.7 \pm 15.8$	$1.0 \pm 0.9$	$33.6 \pm 15.0$
Manning the bed	$53.8 \pm 9.0$	$1.1 \pm 0.4$ $1.0 \pm 0.4$	$50.5 \pm 0.0$	$30.7 \pm 13.8$	$2.2 \pm 1.7$ $1.7 \pm 0.6$	$35.0 \pm 15.0$
Deving videogemes	$07.2 \pm 4.7$	$1.0 \pm 0.4$	$02.9 \pm 13.9$	$20.4 \pm 11.2$ 25.0 ± 26.1	$1.7 \pm 0.0$ 1.0 ± 2.1	$20.1 \pm 10.4$
Scrubbing a surface	$97.2 \pm 4.7$ $82.0 \pm 16.2$	$0.1 \pm 0.1$	$97.3 \pm 3.0$ $83.1 \pm 15.3$	$33.9 \pm 30.1$	$1.9 \pm 2.1$ $1.4 \pm 1.8$	$30.0 \pm 30.3$
Steaking grocories	$61.0 \pm 10.2$	$0.4 \pm 0.3$	$61.8 \pm 16.6$	$33.3 \pm 23.0$	$1.4 \pm 1.0$	$37.0 \pm 27.2$
Stacking groceries	$01.9 \pm 10.0$	$0.8 \pm 0.4$	$01.0 \pm 10.0$	$22.3 \pm 13.3$	$1.3 \pm 0.0$	$21.4 \pm 12.7$
Turing	$37.3 \pm 10.0$	$0.9 \pm 0.4$	$36.6 \pm 10.0$	$20.9 \pm 14.0$	$2.1 \pm 1.4$	$27.8 \pm 13.2$
Typing Vocumenta	$90.3 \pm 4.0$	$0.1 \pm 0.1$	$93.0 \pm 4.4$	$34.5 \pm 32.8$	$1.2 \pm 0.8$	$49.0 \pm 27.9$
Walking around block	$07.7 \pm 14.9$	$0.0 \pm 0.3$	$0/.2 \pm 14.8$ $975 \pm 70$	$40.3 \pm 20.1$	$1.0 \pm 0.8$ $2.0 \pm 1.2$	$30.0 \pm 1/.1$
Washing windows	$67.0 \pm 7.9$	$0.5 \pm 0.2$ 0.7 ± 0.2	$0/.3 \pm 1.8$	$20.1 \pm 20.4$	$2.0 \pm 1.2$ $1.4 \pm 0.7$	$1/.9 \pm 13.0$ $24.7 \pm 16.2$
Watching TV	$05.2 \pm 15.2$	$0.7 \pm 0.3$	$07.3 \pm 12.0$	$34.4 \pm 10.9$	$1.4 \pm 0.7$	$34.7 \pm 10.3$
Watching IV	90.8 ± 4.5	$0.1 \pm 0.1$	$9/.1 \pm 4.2$	$31.3 \pm 34.2$	$1.2 \pm 1.4$	$33.9 \pm 27.3$
Wining/Dusting	$12.2 \pm 10.5$	$0.0 \pm 0.4$	$12.3 \pm 13.2$	$10.7 \pm 23.8$	$1.4 \pm 1.2$	$13.0 \pm 15.0$
Wiking	$00.3 \pm 12.9$	$0.0 \pm 0.4$	$02.7 \pm 13.7$	$20.0 \pm 13.3$	$1.3 \pm 0.7$	$21.3 \pm 13.7$
writing talring out treak	$90.9 \pm 4.2$	$0.1 \pm 0.1$	$90.4 \pm 3.0$	$34.8 \pm 40.8$	$0.8 \pm 1.0$	$31.9 \pm 31.3$
Taking out uash	$1.7.9 \pm 14.9$	0.9 ± 0.4				

Table A8-4: Performance of the C4.5 classifier in recognizing the 51 activities in the MIT dataset (without the *unknown* class) using the accelerometers at the hip and foot, and the *invariant reduced* feature set over windows of 5.6s in length during subject dependent and independent evaluation.

Activity	Sul	Subject Dependent			Subject Independent			
Acuvity	TP Rate	FP Rate	F-Measure	TP Rate	FP Rate	F-Measure		
Bench weight lifting - hard	758 + 272	$0.2 \pm 0.2$	77 3 + 27 3	65 + 149	$0.9 \pm 1.1$	$54 \pm 107$		
Bench weight lifting - light	$87.8 \pm 13.3$	$0.1 \pm 0.2$	$88.0 \pm 13.6$	$29.0 \pm 38.1$	$0.8 \pm 0.7$	$25.1 \pm 28.9$		
Bench weight lifting - moderate	$83.7 \pm 18.8$	$0.1 \pm 0.2$ 0.2 + 0.3	82.2 + 20.4	$188 \pm 261$	$0.0 \pm 0.1$	192 + 228		
Bicen curls - hard	$82.1 \pm 19.0$	$0.2 \pm 0.2$	$81.0 \pm 16.6$	343 + 342	$13 \pm 13$	257 + 232		
Bicep curls - light	$85.4 \pm 12.9$	$0.3 \pm 0.2$	$85.6 \pm 11.4$	$35.4 \pm 35.0$	$1.3 \pm 1.3$	$27.8 \pm 21.5$		
Bicep curls - moderate	$80.9 \pm 12.9$	$0.5 \pm 0.5$	$80.7 \pm 14.2$	99 + 148	$1.5 \pm 1.1$ 0.6 ± 0.4	$11.7 \pm 13.3$		
Calisthenics Crunches	$968 \pm 60$	$0.4 \pm 0.4$	$96.1 \pm 14.2$	$53.8 \pm 40.1$	$0.6 \pm 1.0$	$53.8 \pm 36.8$		
Calisthenics - Sit ups	$90.0 \pm 0.0$	$0.1 \pm 0.1$	$90.4 \pm 4.9$ 87.2 + 10.9	$183 \pm 242$	$0.0 \pm 1.0$ 0.7 ± 0.3	$20.3 \pm 25.2$		
Cycling - Cycle hard - Cycle 80rpm	$78.4 \pm 20.5$	$0.2 \pm 0.2$	$78.1 \pm 10.9$	$95 \pm 161$	$1.0 \pm 1.1$	$95 \pm 160$		
Cycling - Cycle light - Cycle 100rpm	$92.3 \pm 18.4$	$0.5 \pm 0.2$	$92.7 \pm 17.3$	$5.5 \pm 10.1$	$1.0 \pm 1.1$ $1.2 \pm 2.3$	$56.1 \pm 35.6$		
Cycling - Cycle light - Cycle f00ipm	$965 \pm 40$	$0.1 \pm 0.1$	$95.4 \pm 4.6$	$64.4 \pm 28.6$	$1.2 \pm 2.3$ 0.9 ± 0.7	$59.6 \pm 21.5$		
Cycling Cycle light Cycle 80rpm	$93.3 \pm 7.2$	$0.1 \pm 0.1$	$93.4 \pm 4.0$	$40.5 \pm 33.4$	$1.8 \pm 1.7$	$31.8 \pm 25.7$		
Cycling Cycle moderate Cycle S0rpm	93.3 ± 7.2	$0.1 \pm 0.1$	93.0 ± 0.7	$-40.3 \pm 33.4$ $24.4 \pm 28.1$	$1.0 \pm 1.7$	$31.0 \pm 23.7$ 21.3 $\pm$ 22.1		
Lying down	$99.7 \pm 0.5$	$0.3 \pm 0.2$	$99.6 \pm 0.6$	$24.4 \pm 20.1$ 88.9 + 20.5	$1.2 \pm 1.1$ 0.6 + 1.5	$21.3 \pm 22.1$ 89.6 + 16.5		
Powing Powing bard Powing 30cpm	$74.6 \pm 15.9$	$0.0 \pm 0.0$	$73.8 \pm 15.4$	$30.9 \pm 20.3$ $31.0 \pm 27.0$	$0.0 \pm 1.3$	$39.0 \pm 10.3$ 26.9 ± 20.2		
Rowing - Rowing light - Rowing 30spm	$74.0 \pm 13.9$	$0.3 \pm 0.3$	$75.8 \pm 15.4$ 70.5 ± 11.5	$31.9 \pm 27.9$	$1.1 \pm 0.8$ $1.5 \pm 1.8$	$20.9 \pm 20.2$		
Rowing - Rowing nght - Rowing Sospin	$78.9 \pm 12.3$	$0.4 \pm 0.2$	$79.3 \pm 11.3$	$42.2 \pm 30.3$	$1.3 \pm 1.8$ 1.0 ± 0.8	$30.0 \pm 19.3$		
Rowing - Rowing moderate - Rowing Jospin	$06.8 \pm 10.0$	$0.0 \pm 0.3$	$06.5 \pm 14.4$	$23.3 \pm 22.1$	$1.0 \pm 0.8$	$24.4 \pm 17.0$ $26.0 \pm 20.2$		
Running - Treadmill 5mph Treadmill 0	$90.7 \pm 3.3$	$0.1 \pm 0.1$	$90.3 \pm 2.2$ 02.2 ± 4.2	$39.0 \pm 37.7$	$1.1 \pm 0.9$	$30.0 \pm 29.2$		
Running - Treadmill 6mph - Treadmill 0	$93.0 \pm 4.2$	$0.1 \pm 0.1$	$93.3 \pm 4.2$	$60.8 \pm 22.2$	$1.0 \pm 1.1$	$51.6 \pm 20.8$		
Sitting	$07.4 \pm 12.0$	$0.1 \pm 0.1$	$00.9 \pm 12.2$	$00.8 \pm 33.2$	$0.0 \pm 0.7$	$31.0 \pm 20.1$		
Sitting Eident fast lass	$93.9 \pm 3.8$	$0.1 \pm 0.1$	$94.3 \pm 3.0$	$29.9 \pm 33.9$	$1.0 \pm 0.8$	$26.2 \pm 29.3$		
Sitting - Fidget leet legs	$94.3 \pm 3.9$	$0.1 \pm 0.1$	$95.2 \pm 4.0$	$31.4 \pm 29.1$	$1.2 \pm 0.9$	$20.0 \pm 22.7$		
Sitting - Flaget nands arms	$92.9 \pm 7.5$	$0.1 \pm 0.1$	$91.5 \pm 0.2$	$48.8 \pm 32.7$	$0.6 \pm 0.4$	$47.9 \pm 29.1$		
Stairs - Ascend stairs	$89.2 \pm 8.1$	$0.2 \pm 0.1$	$88.2 \pm 8.1$	$53.0 \pm 27.5$	$0.9 \pm 0.4$	$49.6 \pm 23.4$		
Stairs - Descend stairs	$80.3 \pm 8.2$	$0.2 \pm 0.2$	$80.5 \pm 7.4$	$44.3 \pm 22.2$	$1.8 \pm 1.5$	$37.9 \pm 20.1$		
Standing	$92.2 \pm 10.2$	$0.1 \pm 0.2$	91.9 ± 9.3	$19.4 \pm 23.8$	$1.5 \pm 0.9$	$14.7 \pm 14.0$		
Walking - Treadmill 2mph - Treadmill 0	$95.7 \pm 4.0$	$0.1 \pm 0.2$	$94.9 \pm 4.0$	$50.6 \pm 33.2$	$0.8 \pm 0.7$	$50.3 \pm 50.0$		
Walking - Treadmill Smph - Treadmill 0	$63.4 \pm 6.7$	$0.4 \pm 0.2$	$82.8 \pm 8.9$	$14.2 \pm 19.1$	$1.7 \pm 1.3$	$11.7 \pm 14.1$ $14.9 \pm 12.5$		
Walking - Treadmill 3mph - Treadmill 3 - light	$71.2 \pm 13.3$	$0.6 \pm 0.3$	$72.0 \pm 12.7$	$15.2 \pm 15.1$	$1.5 \pm 1.2$	$14.8 \pm 13.5$		
Walking - Treadmill 3mph - Treadmill 6 - moderate	$75.6 \pm 12.1$	$0.7 \pm 0.4$	$73.7 \pm 12.2$	$20.6 \pm 24.5$	$1.4 \pm 1.2$	$17.5 \pm 10.8$		
Walking - Treadmill 3mph - Treadmill 9 - hard	83.8 ± 11.2	$0.3 \pm 0.2$	$84.6 \pm 10.2$	$20.5 \pm 25.0$	$1.5 \pm 1.2$	$18.2 \pm 19.5$		
Kneeling	$93.7 \pm 6.6$	$0.1 \pm 0.1$	$93.5 \pm 6.1$	$22.4 \pm 24.3$	$1.3 \pm 1.2$	$19.4 \pm 19.9$		
Carrying groceries	85.2 ± 12.0	$0.4 \pm 0.3$	84.9 ± 12.2	$38.0 \pm 22.5$	$1.6 \pm 1.2$	$35.5 \pm 18.0$		
Doing dishes	$72.7 \pm 12.1$	$0.7 \pm 0.3$	$71.8 \pm 10.9$	$32.2 \pm 20.0$	$1.6 \pm 0.8$	$30.3 \pm 17.9$		
Gardening	$65.0 \pm 16.8$	$0.8 \pm 0.2$	$64.4 \pm 14.7$	$15.8 \pm 18.7$	$1.3 \pm 0.7$	$15.9 \pm 17.0$		
Ironing	70.9 ± 11.5	$0.8 \pm 0.4$	70.6 ± 11.2	$32.1 \pm 16.3$	$1.7 \pm 1.1$	$30.9 \pm 13.8$		
Making the bed	$55.2 \pm 9.4$	$1.1 \pm 0.3$	55.0 ± 8.5	$34.1 \pm 12.2$	$2.2 \pm 1.4$	$30.7 \pm 11.5$		
Mopping	$53.5 \pm 14.8$	$1.2 \pm 0.4$	53.1 ± 13.8	$22.6 \pm 11.3$	$2.0 \pm 0.7$	$21.6 \pm 9.1$		
Playing videogames	$96.6 \pm 5.2$	$0.1 \pm 0.1$	$96.4 \pm 4.4$	$27.4 \pm 35.6$	$1.7 \pm 1.3$	$23.7 \pm 31.2$		
Scrubbing a surface	65.8 ± 19.6	$0.8 \pm 0.4$	65.9 ± 18.9	$29.2 \pm 16.6$	$1.8 \pm 1.0$	$27.0 \pm 14.4$		
Stacking groceries	$49.5 \pm 15.1$	$1.0 \pm 0.3$	$49.2 \pm 14.2$	$19.5 \pm 12.2$	$2.0 \pm 0.8$	$17.0 \pm 10.0$		
Sweeping	$52.8 \pm 17.4$	$1.0 \pm 0.4$	$53.2 \pm 15.8$	$27.3 \pm 14.8$	$1.9 \pm 0.5$	$26.3 \pm 14.5$		
Typing	$94.7 \pm 6.3$	$0.2 \pm 0.2$	$93.6 \pm 5.7$	$44.0 \pm 35.4$	$0.8 \pm 0.4$	$44.9 \pm 31.8$		
Vacuuming	$59.3 \pm 17.9$	$1.0 \pm 0.5$	$58.9 \pm 17.5$	$31.8 \pm 19.7$	$1.8 \pm 0.7$	$29.4 \pm 15.8$		
Walking around block	87.4 ± 7.7	$0.3 \pm 0.2$	86.9 ± 7.1	$26.9 \pm 24.9$	$2.5 \pm 1.9$	$21.4 \pm 18.0$		
Washing windows	$46.1 \pm 13.9$	$1.1 \pm 0.4$	$48.2 \pm 13.5$	$21.2 \pm 13.2$	$1.7 \pm 0.7$	$22.0 \pm 13.5$		
Watching TV	$94.8 \pm 5.2$	$0.1 \pm 0.1$	$96.0 \pm 3.8$	$46.8 \pm 40.8$	$1.3 \pm 1.2$	$40.5 \pm 34.4$		
Weeding	$60.1\pm11.9$	$0.8\pm0.2$	$62.1 \pm 9.9$	$8.5 \pm 9.8$	$1.0 \pm 0.7$	$10.3 \pm 11.6$		
Wiping/Dusting	$53.9 \pm 17.7$	$1.0 \pm 0.5$	54.7 ± 17.5	$21.6 \pm 13.4$	$1.5 \pm 0.5$	$22.1 \pm 12.0$		
Writing	$94.3 \pm 6.9$	$0.1 \pm 0.2$	$94.4 \pm 5.8$	$37.7 \pm 30.8$	$1.8 \pm 1.8$	$32.8 \pm 24.6$		
taking out trash	44.0 + 10.9	10+04	$46.0 \pm 10.2$	$195 \pm 116$	15+04	$20.7 \pm 11.6$		

Table A8-5: Performance of the C4.5 classifier in recognizing the 51 activities in the MIT dataset (without the *unknown* class) using the accelerometers at the hip, and the *invariant reduced* feature set over windows of 5.6s in length during subject dependent and independent evaluation.

Activity	Sul	oject Depend	lent	Subject Independent				
•	TP Rate	FP Rate	F-Measure	TP Rate	FP Rate	F-Measure		
Bench weight lifting - hard	$83.9 \pm 14.5$	$0.2 \pm 0.3$	82.0 ± 16.6	$19.3 \pm 26.9$	$0.5 \pm 0.6$	$16.4 \pm 19.2$		
Bench weight lifting - light	89.6 ± 13.0	$0.2 \pm 0.2$	85.9 ± 13.6	$36.5 \pm 25.1$	$0.7 \pm 0.6$	$37.0 \pm 22.7$		
Bench weight lifting - moderate	$81.1 \pm 16.4$	$0.2 \pm 0.1$	81.3 ± 15.1	$14.1 \pm 16.6$	$1.0 \pm 0.8$	$11.1 \pm 12.8$		
Bicep curls - hard	$85.8 \pm 13.8$	$0.2 \pm 0.2$	$86.2 \pm 12.4$	$27.4 \pm 27.1$	$1.0 \pm 1.0$	$24.3 \pm 20.3$		
Bicep curls - light	$86.1 \pm 10.8$	$0.3 \pm 0.2$	86.5 ± 10.3	$30.1 \pm 31.1$	$1.2 \pm 1.4$	$26.2 \pm 18.4$		
Bicep curls - moderate	$84.8 \pm 11.7$	$0.3 \pm 0.2$	$84.0 \pm 10.6$	$27.2 \pm 26.9$	$1.1 \pm 0.9$	$24.0 \pm 19.5$		
Calisthenics - Crunches	$95.6 \pm 5.1$	$0.1 \pm 0.1$	$94.8 \pm 4.9$	$20.9 \pm 21.3$	$0.7 \pm 0.7$	$23.0 \pm 21.9$		
Calisthenics - Sit ups	$95.2 \pm 4.0$	$0.1 \pm 0.1$	$93.2 \pm 4.6$	$36.8 \pm 33.7$	$0.7 \pm 0.6$	$37.6 \pm 33.0$		
Cycling - Cycle hard - Cycle 80rpm	$82.0 \pm 14.9$	$0.3 \pm 0.2$	$82.2 \pm 12.9$	$30.0 \pm 32.2$	$1.3 \pm 1.0$	$23.2 \pm 19.4$		
Cycling - Cycle light - Cycle 100rpm	$98.5 \pm 2.7$	$0.0 \pm 0.1$	$97.6 \pm 2.6$	$84.8 \pm 27.8$	$0.5 \pm 1.0$	$79.2 \pm 30.4$		
Cycling - Cycle light - Cycle 60rpm	$96.5 \pm 4.4$	$0.1 \pm 0.1$	$96.7 \pm 3.2$	$80.4 \pm 25.5$	$0.4 \pm 0.4$	$79.8 \pm 20.0$		
Cycling - Cycle light - Cycle 80rpm	$91.3 \pm 8.2$	$0.2 \pm 0.2$	$91.2 \pm 8.2$	$38.9 \pm 26.4$	$1.0 \pm 1.0$	$38.9 \pm 17.3$		
Cycling - Cycle moderate - Cycle 80rpm	$84.0 \pm 12.8$	$0.3 \pm 0.2$	$84.0 \pm 12.0$	$33.0 \pm 24.7$	$1.0 \pm 0.8$	$31.5 \pm 17.5$		
Lying down	$99.9 \pm 0.3$	$0.0 \pm 0.0$	$997 \pm 04$	$91.6 \pm 18.4$	$0.5 \pm 1.0$	912 + 139		
Rowing - Rowing hard - Rowing 30spm	$74.3 \pm 18.9$	$0.5 \pm 0.4$	$72.6 \pm 19.0$	26.0 + 28.5	$1.2 \pm 1.1$	22.2 + 18.2		
Rowing - Rowing light - Rowing 30spm	77.8 + 15.6	$0.4 \pm 0.3$	77.9 + 15.2	40.3 + 30.9	1.4 + 1.1	33.4 + 20.4		
Rowing - Rowing moderate - Rowing 30spm	$66.5 \pm 16.1$	$0.6 \pm 0.3$	$67.0 \pm 15.8$	22.4 + 29.4	$1.1 \pm 1.1$	19.2 + 18.6		
Running - Treadmill 4mph - Treadmill 0	$93.0 \pm 5.8$	$0.1 \pm 0.1$	93.2 + 4.4	27.9 + 31.2	$1.3 \pm 1.4$	29.8 + 31.3		
Running - Treadmill 5mph - Treadmill 0	$92.1 \pm 6.7$	$0.1 \pm 0.1$	92.6 + 4.4	45.4 + 32.7	$1.3 \pm 0.9$	39.7 + 24.0		
Running - Treadmill 6mph - Treadmill 0	$93.8 \pm 5.0$	$0.1 \pm 0.1$	$93.3 \pm 5.8$	$52.6 \pm 36.1$	$0.9 \pm 0.9$	$40.3 \pm 26.4$		
Sitting	965 + 35	$0.1 \pm 0.1$	960 + 41	27 1 + 38 8	12+12	249 + 352		
Sitting - Fidget feet legs	$91.6 \pm 7.2$	$0.1 \pm 0.1$	$92.0 \pm 5.2$	48.8 + 32.0	$0.6 \pm 0.6$	48.6 + 25.0		
Sitting - Fidget hands arms	$934 \pm 61$	$0.1 \pm 0.1$	$92.0 \pm 5.4$	44 2 + 32 6	$0.6 \pm 0.3$	43 2 + 27 7		
Stairs - Ascend stairs	$77.8 \pm 11.8$	$0.1 \pm 0.1$	777+112	$56.1 \pm 22.3$	$1.0 \pm 0.4$	$524 \pm 189$		
Stairs - Descend stairs	$764 \pm 11.8$	$0.1 \pm 0.2$ 0.5 + 0.2	$75.3 \pm 10.6$	439 + 210	$1.0 \pm 0.1$ $1.2 \pm 0.8$	$40.6 \pm 17.8$		
Standing	$958 \pm 51$	$0.0 \pm 0.2$ 0.1 + 0.1	951 + 39	$77.9 \pm 32.1$	$0.5 \pm 0.9$	$71.8 \pm 29.9$		
Walking - Treadmill 2mph - Treadmill 0	$962 \pm 34$	$0.1 \pm 0.1$	$95.2 \pm 3.1$	$66.0 \pm 35.2$	$0.9 \pm 0.9$	643 + 314		
Walking - Treadmill 3mph - Treadmill 0	$67.3 \pm 12.6$	$0.7 \pm 0.3$	$67.7 \pm 11.2$	$20.6 \pm 24.0$	$14 \pm 12$	$179 \pm 169$		
Walking - Treadmill 3mph - Treadmill 3 - light	$59.6 \pm 15.4$	$1.0 \pm 0.5$	$58.0 \pm 15.4$	157 + 157	13+09	$160 \pm 140$		
Walking - Treadmill 3mph - Treadmill 6 - moderate	$564 \pm 12.0$	$1.0 \pm 0.3$	$56.9 \pm 11.2$	155 + 173	13 + 11	$14.6 \pm 14.0$		
Walking - Treadmill 3mph - Treadmill 9 - hard	$70.8 \pm 11.8$	$0.6 \pm 0.3$	$71.8 \pm 11.4$	$14.1 \pm 14.3$	$1.4 \pm 1.1$	$13.7 \pm 12.8$		
kneeling	$97.9 \pm 2.4$	$0.0 \pm 0.0$	$98.2 \pm 1.9$	$87.4 \pm 23.2$	$0.5 \pm 1.0$	$83.7 \pm 24.8$		
Carrying groceries	$90.8 \pm 6.7$	$0.3 \pm 0.2$	89.7 + 7.3	60.3 + 29.7	$1.2 \pm 0.8$	57.2 + 26.1		
Doing dishes	73.3 + 11.6	$0.6 \pm 0.3$	$73.0 \pm 10.5$	42.5 + 22.4	$1.5 \pm 0.7$	$38.2 \pm 17.1$		
Gardening	$73.6 \pm 11.9$	$0.7 \pm 0.3$	$72.5 \pm 11.0$	$22.9 \pm 21.9$	$1.2 \pm 1.1$	$24.4 \pm 22.8$		
Ironing	$80.8 \pm 9.3$	$0.6 \pm 0.3$	$79.2 \pm 9.5$	$47.4 \pm 27.9$	$1.6 \pm 0.9$	$42.9 \pm 21.2$		
Making the bed	$47.2 \pm 8.7$	$1.3 \pm 0.4$	$47.9 \pm 9.3$	$33.0 \pm 19.0$	$2.0 \pm 1.2$	$28.8 \pm 16.3$		
Mopping	$57.9 \pm 18.3$	$1.1 \pm 0.4$	$57.2 \pm 17.4$	$26.3 \pm 12.1$	$2.0 \pm 1.2$	$24.9 \pm 11.6$		
Playing videogames	$97.0 \pm 4.2$	$0.1 \pm 0.1$	$97.2 \pm 3.5$	$55.4 \pm 41.3$	$1.3 \pm 1.4$	$48.3 \pm 35.2$		
Scrubbing a surface	$78.6 \pm 19.5$	$0.5 \pm 0.4$	$78.7 \pm 18.5$	$41.8 \pm 29.4$	$1.7 \pm 1.8$	$36.8 \pm 22.9$		
Stacking groceries	$65.0 \pm 17.7$	$0.6 \pm 0.3$	$66.2 \pm 16.9$	31.1 ± 16.7	$1.9 \pm 1.2$	$28.7 \pm 15.6$		
Sweeping	$62.3 \pm 12.7$	$1.0 \pm 0.4$	$62.5 \pm 13.0$	33.7 ± 16.3	$1.5 \pm 0.7$	$32.7 \pm 13.5$		
Typing	$97.7 \pm 2.7$	$0.1 \pm 0.1$	$97.6 \pm 2.4$	$77.4 \pm 27.3$	$0.6 \pm 1.0$	$76.2 \pm 25.0$		
Vacuuming	$69.4 \pm 10.5$	$0.7 \pm 0.3$	$70.2 \pm 10.0$	$45.4 \pm 27.4$	$1.1 \pm 0.6$	$44.0 \pm 23.4$		
Walking around block	83.8 ± 11.8	$0.4 \pm 0.3$	$84.0 \pm 10.4$	$37.2 \pm 20.6$	$2.8 \pm 1.6$	$29.6 \pm 14.8$		
Washing windows	$64.7 \pm 12.1$	$0.7 \pm 0.3$	$66.5 \pm 10.7$	$40.6 \pm 24.0$	$1.2 \pm 0.7$	$42.1 \pm 24.6$		
Watching TV	$96.8 \pm 2.6$	$0.1 \pm 0.1$	96.6 ± 3.0	$30.9 \pm 30.5$	$1.5 \pm 1.1$	31.4 ± 28.5		
Weeding	59.3 ± 22.6	$0.8 \pm 0.4$	$61.0 \pm 20.5$	$10.1 \pm 16.2$	$1.5 \pm 1.7$	$10.0 \pm 13.3$		
Wiping/Dusting	56.9 ± 13.7	$0.9 \pm 0.4$	58.8 ± 13.3	38.2 ± 17.4	$1.3 \pm 0.6$	38.6 ± 16.8		
Writing	$96.8 \pm 3.4$	$0.1 \pm 0.1$	$95.6 \pm 2.9$	$67.9 \pm 35.4$	$0.9 \pm 1.6$	$66.3 \pm 35.0$		
taking out trash	$52.4 \pm 11.5$	$1.0 \pm 0.4$	53.9 ± 12.3	$21.6 \pm 14.2$	$1.3 \pm 0.6$	$23.5 \pm 14.4$		

Table A8-6: Performance of the C4.5 classifier in recognizing the 51 activities in the MIT dataset (without the *unknown* class) using the accelerometers at the dominant wrist and dominant foot, and the *invariant reduced* feature set over windows of 5.6s in length during subject dependent and independent evaluation.

Activity	Sul	oject Depend	lent	Subject Independent				
	TP Rate	FP Rate	F-Measure	TP Rate	FP Rate	F-Measure		
Bench weight lifting - hard	87.5 ± 12.9	$0.2\pm0.2$	84.7 ± 12.5	$8.0 \pm 23.4$	$0.4 \pm 0.6$	$5.6 \pm 11.5$		
Bench weight lifting - light	$93.5 \pm 8.2$	$0.1 \pm 0.2$	$93.3 \pm 8.3$	$41.5 \pm 38.4$	$0.8\pm0.7$	$37.2 \pm 31.6$		
Bench weight lifting - moderate	$84.8 \pm 13.6$	$0.1 \pm 0.1$	$85.5 \pm 12.5$	$29.0 \pm 26.5$	$0.9 \pm 1.0$	$23.6 \pm 16.6$		
Bicep curls - hard	$85.7 \pm 11.6$	$0.2 \pm 0.2$	$86.4 \pm 10.7$	$30.7 \pm 34.6$	$1.1 \pm 1.1$	$25.6 \pm 26.2$		
Bicep curls - light	$86.2 \pm 9.4$	$0.3 \pm 0.3$	$86.4 \pm 9.8$	$36.4 \pm 35.9$	$1.1 \pm 1.1$	$30.5 \pm 24.5$		
Bicep curls - moderate	$81.0 \pm 14.7$	$0.4 \pm 0.3$	$80.7 \pm 13.7$	$22.6 \pm 29.7$	$1.2 \pm 1.0$	$18.1 \pm 18.7$		
Calisthenics - Crunches	$98.2 \pm 3.7$	$0.0 \pm 0.0$	$98.0 \pm 2.0$	$65.2 \pm 37.1$	$0.5 \pm 0.9$	$62.4 \pm 35.4$		
Calisthenics - Sit ups	$94.6 \pm 4.0$	$0.1 \pm 0.1$	$93.4 \pm 3.2$	$61.9 \pm 32.0$	$0.3 \pm 0.3$	$63.4 \pm 29.3$		
Cycling - Cycle hard - Cycle 80rpm	$81.9 \pm 10.8$	$0.3 \pm 0.2$	$81.4 \pm 11.2$	$27.0 \pm 29.0$	$0.7 \pm 0.7$	$25.5 \pm 26.3$		
Cycling - Cycle light - Cycle 100rpm	98.3 + 3.9	$0.0 \pm 0.0$	97.7 + 3.7	$90.4 \pm 17.8$	$0.1 \pm 0.4$	$92.0 \pm 12.3$		
Cycling - Cycle light - Cycle 60rpm	$99.2 \pm 1.5$	$0.0 \pm 0.0$	$98.6 \pm 1.4$	$91.8 \pm 11.8$	$0.2 \pm 0.3$	91.4 + 9.1		
Cycling - Cycle light - Cycle 80rpm	$94.1 \pm 6.4$	$0.1 \pm 0.1$	$93.9 \pm 6.0$	50.4 + 34.8	$1.3 \pm 1.0$	41.9 + 25.1		
Cycling - Cycle moderate - Cycle 80rpm	$834 \pm 100$	$0.3 \pm 0.2$	84 4 + 8 6	340 + 275	$13 \pm 0.7$	$29.0 \pm 16.3$		
Lying down	$99.9 \pm 0.4$	$0.0 \pm 0.0$	$99.6 \pm 0.4$	$83.0 \pm 27.6$	$1.0 \pm 0.7$ $1.1 \pm 1.5$	$\frac{29.6}{80.6} \pm \frac{10.9}{23.8}$		
Rowing - Rowing hard - Rowing 30spm	$76.8 \pm 14.9$	$0.0 \pm 0.0$	76.6 + 14.3	$35.0 \pm 27.0$ $35.1 \pm 25.5$	$1.1 \pm 1.5$ $1.2 \pm 0.9$	$32.2 \pm 19.5$		
Rowing - Rowing light - Rowing 30spm	$83.0 \pm 12.5$	$0.4 \pm 0.3$	$82.9 \pm 12.4$	$42.6 \pm 31.8$	$1.2 \pm 0.9$ 1.6 ± 1.3	$32.2 \pm 19.3$ $33.8 \pm 19.3$		
Rowing - Rowing moderate - Rowing 30spm	$71.4 \pm 15.2$	$0.4 \pm 0.3$	$71.6 \pm 15.2$	$27.5 \pm 28.2$	$1.0 \pm 1.0$ 1 1 + 1 1	$245 \pm 19.0$		
Running - Treadmill 4mph - Treadmill 0	$91.8 \pm 6.7$	$0.0 \pm 0.3$	$91.4 \pm 5.2$	$27.3 \pm 20.2$ 53.2 + 38.9	$0.7 \pm 1.0$	$53.4 \pm 37.2$		
Running - Treadmill 5mph - Treadmill 0	$91.0 \pm 0.7$ 89.2 + 7.9	$0.2 \pm 0.1$	$91.4 \pm 5.3$ 89.4 ± 6.7	$68.0 \pm 26.3$	$1.0 \pm 0.9$	$53.4 \pm 37.2$		
Running - Treadmill 6mph - Treadmill 0	$81.6 \pm 19.7$	$0.2 \pm 0.1$ 0.1 + 0.1	$83.4 \pm 17.5$	$52.0 \pm 20.3$	$1.0 \pm 0.5$	$49.2 \pm 21.7$		
Sitting	$965 \pm 4.7$	$0.1 \pm 0.1$	$05.4 \pm 17.5$	$32.0 \pm 31.3$	$1.2 \pm 1.2$	$47.2 \pm 20.3$		
Sitting Eidget feet loge	$90.3 \pm 4.7$	$0.1 \pm 0.1$	$93.9 \pm 4.1$ 02.4 ± 7.5	$43.8 \pm 44.3$ 50.4 ± 24.2	$1.2 \pm 1.2$	$33.0 \pm 34.0$ 52.7 ± 20.0		
Sitting - Fidget hands arms	$93.1 \pm 9.9$	$0.1 \pm 0.1$	$93.4 \pm 7.3$	$30.4 \pm 34.2$	$0.4 \pm 0.3$	$32.7 \pm 29.9$		
Sitting - Fluget hands arms	$95.4 \pm 8.0$	$0.1 \pm 0.2$	$92.0 \pm 7.4$	$37.2 \pm 28.2$	$0.7 \pm 0.7$	$50.8 \pm 25.8$		
Stairs - Ascend stairs	$63.1 \pm 7.3$	$0.4 \pm 0.2$	$63.3 \pm 7.6$	$02.7 \pm 25.1$	$1.0 \pm 0.7$	$30.0 \pm 10.7$		
Stalls - Descend stalls	19.8 ± 1.8	$0.3 \pm 0.2$	$80.3 \pm 8.2$	$46.3 \pm 24.4$	$0.7 \pm 0.4$	$48.0 \pm 20.1$		
Standing Welling Treedwill Quark Treedwill O	$95.0 \pm 7.4$	$0.1 \pm 0.1$	$94.3 \pm 7.6$	$68.8 \pm 34.8$	$0.6 \pm 0.8$	$62.7 \pm 31.0$		
Walking - Headmin 2mph - Headmin 0	$94.0 \pm 4.3$	$0.2 \pm 0.1$	$94.2 \pm 3.0$	$75.3 \pm 22.8$	$0.3 \pm 0.7$	$73.3 \pm 16.0$		
Walking - Treadmill Smph - Treadmill 0	$77.3 \pm 13.7$	$0.6 \pm 0.2$	$76.2 \pm 12.0$	$29.3 \pm 24.2$	$2.0 \pm 1.4$	$24.3 \pm 16.0$		
Waiking - Treadmill Smph - Treadmill S - light	$60.6 \pm 14.4$	$0.8 \pm 0.3$	$61.3 \pm 14.2$	$11.2 \pm 12.2$	$1.0 \pm 0.6$	$12.4 \pm 11.8$		
Walking - Treadmill 3mph - Treadmill 6 - moderate	$6/.9 \pm 13.6$	$0.8 \pm 0.4$	$66.5 \pm 13.2$	$1/.1 \pm 13.4$	$1.7 \pm 1.0$	$16.4 \pm 10.8$		
Walking - Treadmill 3mph - Treadmill 9 - hard	/8.8 ± 11.3	$0.5 \pm 0.3$	/9.1 ± 10.6	$23.2 \pm 25.7$	$1.2 \pm 0.9$	$21.7 \pm 21.4$		
kneeling	$95.4 \pm 4.1$	$0.1 \pm 0.1$	$94.5 \pm 3.6$	$49.4 \pm 36.0$	$0.5 \pm 0.5$	49.4 ± 30.8		
Carrying groceries	87.6 ± 9.8	$0.3 \pm 0.2$	87.8 ± 8.0	$53.4 \pm 30.1$	$1.0 \pm 0.5$	$52.9 \pm 26.6$		
Doing dishes	74.1 ± 16.1	$0.6 \pm 0.4$	73.8 ± 14.0	$51.1 \pm 23.2$	$1.5 \pm 1.1$	46.3 ± 19.0		
Gardening	/5.9 ± 14./	$0.6 \pm 0.4$	$75.4 \pm 15.5$	$19.9 \pm 19.5$	$0.8 \pm 0.4$	$22.8 \pm 21.0$		
Ironing	74.3 ± 15.4	$0.7 \pm 0.4$	72.9 ± 14.3	$53.3 \pm 24.3$	$1.4 \pm 0.7$	$49.3 \pm 16.5$		
Making the bed	$51.1 \pm 12.2$	$1.2 \pm 0.4$	$51.0 \pm 11.8$	$34.0 \pm 13.1$	$2.1 \pm 0.9$	$30.6 \pm 11.2$		
Mopping	53.0 ± 13.3	$1.2 \pm 0.3$	$51.8 \pm 11.8$	31.4 ± 13.9	1.9 ± 0.6	$29.2 \pm 11.2$		
Playing videogames	$99.1 \pm 1.6$	$0.0 \pm 0.0$	$99.2 \pm 1.2$	$42.0 \pm 36.8$	$1.1 \pm 1.2$	$41.1 \pm 33.7$		
Scrubbing a surface	68.1 ± 16.5	$0.8 \pm 0.3$	67.6 ± 14.7	$29.0 \pm 16.5$	$1.8 \pm 0.9$	$27.4 \pm 14.6$		
Stacking groceries	$56.1 \pm 15.0$	$0.8 \pm 0.3$	$57.0 \pm 15.2$	$36.9 \pm 18.6$	$1.4 \pm 0.8$	$34.6 \pm 15.0$		
Sweeping	$56.9 \pm 11.0$	$0.9 \pm 0.5$	$58.9 \pm 11.8$	$31.4 \pm 12.9$	$1.5 \pm 0.6$	$32.1 \pm 11.9$		
Typing	$99.2 \pm 1.2$	$0.0\pm0.0$	$98.9 \pm 0.8$	$77.8 \pm 30.4$	$0.4 \pm 0.7$	$77.5 \pm 27.6$		
Vacuuming	$71.2 \pm 12.6$	$0.6 \pm 0.2$	$72.6 \pm 10.6$	$48.4 \pm 17.6$	$1.1 \pm 0.5$	$49.1 \pm 13.2$		
Walking around block	$83.8 \pm 10.1$	$0.4 \pm 0.2$	$84.4 \pm 8.7$	$35.7 \pm 17.1$	$2.6 \pm 2.2$	$32.3 \pm 16.3$		
Washing windows	$60.8\pm10.9$	$0.8\pm0.4$	$63.1 \pm 11.2$	$41.3 \pm 20.9$	$1.5 \pm 1.3$	$40.8 \pm 22.4$		
Watching TV	$98.5 \pm 1.9$	$0.0\pm0.0$	$98.8 \pm 1.3$	$27.5 \pm 30.1$	$2.4 \pm 2.5$	$23.8 \pm 25.4$		
Weeding	$62.4 \pm 15.2$	$0.8\pm0.5$	$63.9 \pm 15.1$	$8.6\pm9.2$	$1.2\pm0.8$	$10.3 \pm 10.3$		
Wiping/Dusting	$55.6 \pm 14.1$	$1.1 \pm 0.4$	$55.0 \pm 14.3$	$35.5 \pm 18.6$	$1.3\pm0.6$	$35.4 \pm 16.8$		
Writing	$98.1\pm2.8$	$0.1\pm0.1$	$97.5\pm2.2$	$58.8\pm38.6$	$0.6\pm0.6$	$58.3\pm34.4$		
taking out trash	$49.1 \pm 13.2$	$1.0 \pm 0.3$	$50.5 \pm 12.8$	$28.1 \pm 16.6$	$1.4 \pm 0.4$	$28.0 \pm 14.3$		

Table A8-7: Performance of the C4.5 classifier in recognizing the 51 activities in the MIT dataset (without the *unknown* class) using the accelerometers at the dominant wrist and dominant thigh, and the *invariant reduced* feature set over windows of 5.6s in length during subject dependent and independent evaluation.

Activity	Su	biect Depend	lent	Subject Independent				
	TP Rate	FP Rate	F-Measure	TP Rate	FP Rate	F-Measure		
Bench weight lifting - hard	$50.0 \pm 17.9$	$0.4 \pm 0.2$	$52.2 \pm 17.8$	$15.5 \pm 15.4$	$0.7 \pm 0.7$	$14.0 \pm 14.0$		
Bench weight lifting - light	$71.6 \pm 15.0$	$0.5 \pm 0.3$	$71.0 \pm 15.1$	$40.1 \pm 24.2$	$0.8 \pm 0.6$	$41.0 \pm 20.7$		
Bench weight lifting - moderate	$59.3 \pm 21.0$	$0.6 \pm 0.3$	$57.4 \pm 19.2$	$22.4 \pm 20.1$	$0.7 \pm 0.4$	$20.1 \pm 16.4$		
Bicep curls - hard	$71.2 \pm 14.5$	$0.6 \pm 0.3$	69.9 ± 14.8	$37.4 \pm 28.0$	$1.4 \pm 0.8$	30.1 ± 19.6		
Bicep curls - light	$68.5 \pm 14.1$	$0.6 \pm 0.3$	69.3 ± 13.8	33.7 ± 21.1	$1.2 \pm 0.6$	$32.4 \pm 16.7$		
Bicep curls - moderate	$62.4 \pm 14.4$	$0.7 \pm 0.3$	$62.7 \pm 13.0$	$22.4 \pm 15.0$	$0.9 \pm 0.7$	$24.6 \pm 12.6$		
Calisthenics - Crunches	$87.0 \pm 9.9$	$0.2 \pm 0.2$	$87.0 \pm 9.4$	$35.6 \pm 33.8$	$0.8 \pm 0.9$	$35.8 \pm 32.0$		
Calisthenics - Sit ups	$90.7 \pm 5.9$	$0.2 \pm 0.2$	$88.2 \pm 6.9$	$40.0 \pm 31.2$	$0.8 \pm 1.2$	$38.5 \pm 30.2$		
Cycling - Cycle hard - Cycle 80rpm	$67.2 \pm 24.5$	$0.6 \pm 0.4$	$65.5 \pm 24.6$	$15.8 \pm 21.7$	$1.3 \pm 0.8$	$13.7 \pm 16.3$		
Cycling - Cycle light - Cycle 100rpm	$80.2 \pm 10.8$	$0.4 \pm 0.2$	$79.4 \pm 10.8$	$47.9 \pm 27.2$	$1.1 \pm 1.1$	45.1 ± 24.0		
Cycling - Cycle light - Cycle 60rpm	$84.4 \pm 12.6$	$0.4 \pm 0.3$	82.5 ± 12.4	$57.2 \pm 25.6$	$1.2 \pm 0.8$	$51.0 \pm 21.3$		
Cycling - Cycle light - Cycle 80rpm	$77.1 \pm 19.9$	$0.5 \pm 0.5$	$76.6 \pm 20.2$	$25.4 \pm 20.6$	$1.7 \pm 1.2$	$24.1 \pm 19.2$		
Cycling - Cycle moderate - Cycle 80rpm	$65.7 \pm 16.3$	$0.7 \pm 0.4$	$65.1 \pm 16.5$	$16.4 \pm 15.0$	$1.7 \pm 0.7$	$15.0 \pm 12.8$		
Lying down	$99.4 \pm 0.9$	$0.0 \pm 0.1$	$99.3 \pm 0.8$	57.8 + 37.8	$1.2 \pm 1.4$	59.6 + 37.1		
Rowing - Rowing hard - Rowing 30spm	$66.5 \pm 16.5$	$0.6 \pm 0.3$	$65.7 \pm 15.8$	$31.6 \pm 26.0$	$1.1 \pm 0.7$	$27.1 \pm 16.3$		
Rowing - Rowing light - Rowing 30spm	71.1 + 13.2	$0.7 \pm 0.4$	$69.4 \pm 13.5$	$35.1 \pm 17.7$	$1.2 \pm 0.8$	$34.6 \pm 11.8$		
Rowing - Rowing moderate - Rowing 30spm	$563 \pm 133$	$0.8 \pm 0.2$	$569 \pm 12.7$	$255 \pm 173$	11 + 10	$25.6 \pm 11.2$		
Running - Treadmill 4mph - Treadmill 0	877+75	$0.0 \pm 0.2$ 0.4 + 0.3	853+95	$29.0 \pm 17.0$ $29.1 \pm 27.3$	$1.1 \pm 1.0$ $1.5 \pm 1.5$	$28.4 \pm 26.4$		
Running - Treadmill 5mph - Treadmill 0	88.6 ± 6.7	$0.1 \pm 0.2$ 0.3 ± 0.2	87.1 + 7.5	$40.5 \pm 33.1$	$0.9 \pm 0.6$	$37.3 \pm 25.2$		
Running - Treadmill 6mph - Treadmill 0	84 3 + 13 4	$0.3 \pm 0.2$ 0.2 + 0.1	84 1 + 11 2	$30.8 \pm 28.1$	$0.5 \pm 0.8$	$37.3 \pm 25.2$ $32.2 \pm 26.7$		
Sitting	$925 \pm 69$	$0.1 \pm 0.1$	$919 \pm 67$	$29.2 \pm 36.8$	$1.6 \pm 1.6$	$23.5 \pm 27.8$		
Sitting - Fidget feet legs	$88.7 \pm 10.9$	$0.1 \pm 0.1$	88 3 + 11 8	$27.6 \pm 31.8$	$0.8 \pm 0.8$	$27.2 \pm 27.0$		
Sitting - Fidget hands arms	$81.5 \pm 17.1$	$0.2 \pm 0.2$	$80.7 \pm 13.8$	$27.0 \pm 31.0$ $28.0 \pm 23.4$	$0.8 \pm 0.0$	$27.2 \pm 20.3$ 28.9 + 20.3		
Stairs - Ascend stairs	$70.5 \pm 17.1$	$0.2 \pm 0.1$ 0.7 + 0.3	$67.7 \pm 13.6$	$42.6 \pm 29.6$	$0.0 \pm 0.7$	$20.9 \pm 20.3$ 38.8 + 24.4		
Stairs - Descend stairs	$62.5 \pm 12.4$	$0.7 \pm 0.3$	$67.7 \pm 13.0$ $62.6 \pm 11.2$	$41.4 \pm 24.9$	$1.1 \pm 0.5$ $1.5 \pm 1.0$	$34.7 \pm 18.1$		
Standing	$02.3 \pm 12.4$ $02.3 \pm 7.0$	$0.0 \pm 0.2$	$02.0 \pm 11.2$ $01.6 \pm 7.7$	$-71.4 \pm 24.9$ 26.8 ± 31.7	$1.5 \pm 1.0$ 0.8 ± 0.7	$34.7 \pm 10.1$ $24.1 \pm 26.5$		
Walking Treadmill 2mph Treadmill 0	$90.4 \pm 6.4$	$0.1 \pm 0.1$	91.0 ± 7.7	$20.0 \pm 31.7$ 51.3 ± 20.7	$0.0 \pm 0.7$	$40.3 \pm 23.8$		
Walking Treadmill 3mph Treadmill 0	$50.4 \pm 0.4$	$0.5 \pm 0.2$	$61.0 \pm 0.0$	$31.3 \pm 29.7$ 21.1 ± 21.5	$1.1 \pm 1.5$ 1.6 ± 1.0	$10.1 \pm 16.4$		
Walking - Treadmill 3mph - Treadmill 3 - light	$51.8 \pm 12.9$	$1.2 \pm 0.3$	$50.7 \pm 14.5$	99 + 193	$1.0 \pm 1.0$ 1 1 + 1 5	$\frac{19.1 \pm 10.4}{8.4 \pm 10.4}$		
Walking Treadmill 3mph Treadmill 6 moderate	$51.0 \pm 13.0$ $52.6 \pm 13.8$	$1.2 \pm 0.4$ 1 1 + 0 3	$50.7 \pm 17.5$	$12.0 \pm 19.3$	$1.1 \pm 1.5$ $1.3 \pm 0.9$	$11.3 \pm 12.1$		
Walking Treadmill 3mph Treadmill 0 hard	$52.0 \pm 13.8$	$1.1 \pm 0.3$	$52.4 \pm 12.0$	$12.0 \pm 10.1$ 26.5 ± 27.1	$1.5 \pm 0.9$	$11.3 \pm 12.1$ $22.4 \pm 18.2$		
kneeling	$00.0 \pm 12.7$	$0.7 \pm 0.3$	$00.0 \pm 12.8$	$20.3 \pm 27.1$ $42.4 \pm 37.0$	$1.3 \pm 1.2$ 1.0 ± 1.0	$22.4 \pm 10.2$ $32.6 \pm 25.2$		
Corruing	$91.2 \pm 0.1$ 85.2 ± 11.4	$0.1 \pm 0.1$	$91.9 \pm 7.0$ $84.0 \pm 0.4$	$42.4 \pm 37.0$	$1.0 \pm 1.0$ 1.0 ± 0.8	$52.0 \pm 23.2$		
Doing dishos	$35.2 \pm 11.4$	$0.4 \pm 0.3$	$34.0 \pm 9.4$	$30.8 \pm 35.7$	$1.0 \pm 0.8$	$34.4 \pm 32.0$		
Gardening	$48.0 \pm 14.0$ 52.6 ± 19.4	$1.3 \pm 0.4$	$47.7 \pm 14.7$ $53.4 \pm 18.4$	$37.8 \pm 10.1$ 27.6 ± 25.5	$1.8 \pm 0.8$ $1.0 \pm 0.4$	$34.4 \pm 12.8$ 28.9 ± 25.5		
Ironing	$52.0 \pm 19.4$	$0.9 \pm 0.4$	$55.4 \pm 18.4$	$27.0 \pm 23.3$ $27.2 \pm 12.2$	$1.0 \pm 0.4$	$26.9 \pm 23.3$		
Making the had	$37.6 \pm 13.6$ 25.2 ± 0.4	$1.2 \pm 0.3$	$30.9 \pm 13.7$	$27.3 \pm 13.2$ 22.6 ± 12.5	$2.2 \pm 2.0$	$20.0 \pm 11.0$ 22.1 ± 12.2		
Manning	$33.3 \pm 9.4$	$1.7 \pm 0.3$	$34.9 \pm 0.0$	$23.0 \pm 13.3$	$1.9 \pm 0.8$	$22.1 \pm 12.2$		
Mopping Disving video comes	$41.3 \pm 10.8$	$1.4 \pm 0.3$	$41.1 \pm 13.3$	$22.9 \pm 10.0$	$2.2 \pm 1.3$	$21.1 \pm 0.3$		
Playing videogames	$95.9 \pm 4.7$	$0.1 \pm 0.2$	$95.0 \pm 4.8$	$39.7 \pm 37.8$	$1.1 \pm 1.3$	$34.8 \pm 31.9$		
Scrubbing a surface	$47.2 \pm 15.0$	$1.1 \pm 0.4$	$46.7 \pm 13.3$	$20.0 \pm 18.4$	$1.7 \pm 1.0$	$25.2 \pm 10.3$		
Stacking groceries	$45.2 \pm 18.1$	$1.0 \pm 0.5$	$46.0 \pm 19.1$	$33.4 \pm 18.0$	$1.7 \pm 1.0$	$31.3 \pm 15.8$		
Sweeping	$52.4 \pm 14.1$	$1.1 \pm 0.5$	$52.9 \pm 12.8$	$29.6 \pm 13.5$	$1.6 \pm 0.6$	$29.8 \pm 13.4$		
Typing	$96.1 \pm 3.3$	$0.1 \pm 0.1$	96.2 ± 2.8	$72.4 \pm 31.9$	$0.4 \pm 0.6$	$74.6 \pm 31.0$		
Vacuuming Wall-ing angend black	$55.6 \pm 14.2$	$0.8 \pm 0.3$	$57.9 \pm 12.3$	$49.0 \pm 16.4$	$1.2 \pm 0.4$	$48.5 \pm 12.1$		
Washing around block	$78.1 \pm 13.5$	$0.5 \pm 0.3$	/9.1 ± 12.3	$28.9 \pm 20.6$	$2.5 \pm 1.9$	$25.1 \pm 10.1$		
Washing windows	$56.8 \pm 13.8$	$0.8 \pm 0.3$	59.6 ± 12.8	$39.8 \pm 21.0$	$1.3 \pm 1.0$	$41.1 \pm 21.5$		
watching IV	92.4 ± 7.5	$0.1 \pm 0.2$	93.7 ± 6.4	$26.5 \pm 21.9$	$2.8 \pm 2.2$	$21.9 \pm 17.9$		
Weeding	$41.3 \pm 16.1$	$1.3 \pm 0.4$	$41.6 \pm 14.9$	$5.0 \pm 6.8$	$1.0 \pm 0.5$	$6.4 \pm 8.6$		
Wiping/Dusting	43.6 ± 16.9	$1.2 \pm 0.4$	$45.4 \pm 16.2$	$25.3 \pm 13.8$	$1.5 \pm 0.5$	$26.1 \pm 13.1$		
Writing	$93.4 \pm 4.6$	$0.2 \pm 0.1$	92.6 ± 4.4	$70.3 \pm 34.5$	$1.1 \pm 2.2$	$66.6 \pm 32.9$		
taking out trash	$134.6 \pm 14.4$	$1.2 \pm 0.3$	$35.8 \pm 13.6$	$15.0 \pm 11.3$	$1.4 \pm 0.4$	$158 \pm 110$		

Table A8-8: Performance of the C4.5 classifier in recognizing the 51 activities in the MIT dataset (without the *unknown* class) using the accelerometers at the dominant wrist, and the *invariant* reduced feature set over windows of 5.6s in length during subject dependent and independent evaluation.

Activity	Sul	bject Depend	lent	Subject Independent				
	TP Rate	FP Rate	F-Measure	TP Rate	FP Rate	F-Measure		
Bench weight lifting - hard	$82.4 \pm 19.5$	$0.2 \pm 0.2$	82.5 ± 17.9	$9.3 \pm 14.5$	$0.5 \pm 0.4$	$8.6 \pm 12.9$		
Bench weight lifting - light	$84.2 \pm 15.3$	$0.2 \pm 0.2$	$83.5 \pm 14.8$	$20.1 \pm 19.3$	$1.0 \pm 0.5$	$19.3 \pm 17.2$		
Bench weight lifting - moderate	$80.2 \pm 17.8$	$0.2 \pm 0.2$	$80.9 \pm 16.8$	$15.7 \pm 15.6$	$0.9 \pm 0.6$	$14.6 \pm 14.2$		
Bicep curls - hard	$86.4 \pm 12.2$	$0.3 \pm 0.2$	85.8 ± 11.5	$23.3 \pm 29.3$	$1.4 \pm 1.0$	$16.6 \pm 18.2$		
Bicep curls - light	$81.0 \pm 15.6$	$0.3 \pm 0.3$	82.1 ± 15.6	$24.2 \pm 25.6$	$1.3 \pm 1.1$	$20.9 \pm 16.9$		
Bicep curls - moderate	$82.4 \pm 15.1$	$0.4 \pm 0.3$	81.2 ± 13.8	$17.9 \pm 21.0$	$1.0 \pm 1.0$	$16.6 \pm 16.1$		
Calisthenics - Crunches	$91.8 \pm 9.1$	$0.2 \pm 0.2$	$90.8 \pm 8.4$	$6.1 \pm 9.7$	$1.4 \pm 0.9$	$5.4 \pm 7.2$		
Calisthenics - Sit ups	$93.9 \pm 7.0$	$0.2 \pm 0.2$	$91.3 \pm 8.0$	$26.2 \pm 25.4$	$1.2 \pm 1.8$	$24.2 \pm 20.9$		
Cycling - Cycle hard - Cycle 80rpm	$79.9 \pm 14.7$	$0.3 \pm 0.3$	$80.0 \pm 12.5$	$18.7 \pm 23.3$	$0.9 \pm 0.8$	$17.6 \pm 18.7$		
Cycling - Cycle light - Cycle 100rpm	$98.7 \pm 2.5$	$0.0 \pm 0.0$	$97.8 \pm 2.1$	93.3 ± 17.3	$0.2 \pm 0.2$	90.5 ± 13.4		
Cycling - Cycle light - Cycle 60rpm	$97.4 \pm 3.5$	$0.0 \pm 0.1$	97.7 ± 2.4	$87.4 \pm 20.5$	$0.2 \pm 0.3$	87.4 ± 15.1		
Cycling - Cycle light - Cycle 80rpm	$86.4 \pm 10.5$	$0.3 \pm 0.2$	$86.2 \pm 9.6$	$39.2 \pm 29.4$	$1.2 \pm 1.0$	$37.3 \pm 23.4$		
Cycling - Cycle moderate - Cycle 80rpm	76.1 + 13.8	$0.4 \pm 0.3$	76.4 + 13.1	45.2 + 23.9	$1.7 \pm 1.0$	$37.0 \pm 11.7$		
Lying down	$99.2 \pm 1.7$	$0.1 \pm 0.1$	$99.0 \pm 1.7$	85.4 + 22.5	$0.9 \pm 1.3$	84.9 + 15.9		
Rowing - Rowing hard - Rowing 30spm	64.2 + 25.3	$0.6 \pm 0.4$	64.4 + 24.6	21.6 + 28.7	$1.0 \pm 1.3$	$18.8 \pm 19.5$		
Rowing - Rowing light - Rowing 30spm	74.3 + 15.6	$0.6 \pm 0.4$	73.2 + 15.7	34.8 + 32.0	$1.5 \pm 1.0$	29.8 + 23.1		
Rowing - Rowing moderate - Rowing 30spm	$637 \pm 190$	$0.8 \pm 0.5$	63 3 + 18 6	$240 \pm 238$	$1.0 \pm 0.8$	$23.8 \pm 21.9$		
Running - Treadmill 4mph - Treadmill 0	947 + 41	$0.0 \pm 0.0$ 0.1 + 0.1	941 + 31	292 + 236	$1.6 \pm 0.0$	$28.2 \pm 22.5$		
Running - Treadmill 5mph - Treadmill 0	$90.0 \pm 6.1$	$0.1 \pm 0.1$ 0.2 ± 0.1	$89.8 \pm 4.6$	$42.2 \pm 23.0$ $42.2 \pm 23.0$	$1.0 \pm 0.8$	$42.6 \pm 22.9$		
Running - Treadmill 6mph - Treadmill 0	$90.6 \pm 6.8$	$0.2 \pm 0.1$ 0.1 + 0.1	$914 \pm 53$	$62.0 \pm 29.8$	$0.8 \pm 0.8$	$51.7 \pm 26.7$		
Sitting	$94.3 \pm 7.7$	$0.1 \pm 0.1$	$93.8 \pm 7.8$	$22.0 \pm 25.0$ $22.4 \pm 36.0$	$14 \pm 17$	$18.0 \pm 25.1$		
Sitting - Fidget feet legs	$89.1 \pm 9.4$	$0.1 \pm 0.1$	883+95	$50.3 \pm 31.5$	$0.5 \pm 0.6$	$51.7 \pm 27.6$		
Sitting - Fidget hands arms	$90.7 \pm 9.4$	$0.2 \pm 0.2$	89.8 + 8.6	$36.6 \pm 25.0$	$1.1 \pm 0.7$	$31.7 \pm 27.0$ $31.3 \pm 20.2$		
Stairs - Ascend stairs	$71.6 \pm 12.1$	$0.2 \pm 0.1$ 0.5 ± 0.2	$72.9 \pm 10.8$	$30.0 \pm 23.0$ 41.9 + 24.6	$1.1 \pm 0.7$ $1.3 \pm 0.8$	$36.8 \pm 18.5$		
Stairs - Descend stairs	$73.7 \pm 9.0$	$0.5 \pm 0.2$	$71.5 \pm 9.0$	$37.1 \pm 15.7$	$1.5 \pm 0.0$ 1 4 ± 0.8	$33.9 \pm 15.2$		
Standing	$92.7 \pm 9.6$	$0.0 \pm 0.2$	$923 \pm 84$	$26.6 \pm 40.4$	$1.4 \pm 0.0$	$33.9 \pm 10.2$ 21.1 + 30.1		
Walking - Treadmill 2mph - Treadmill ()	$96.0 \pm 3.2$	$0.1 \pm 0.1$	$94.9 \pm 3.2$	$47.2 \pm 20.9$	$1.3 \pm 1.1$ $1.2 \pm 2.4$	$50.4 \pm 25.3$		
Walking Treadmill 3mph Treadmill 0	$58.7 \pm 15.1$	$0.2 \pm 0.1$	$58.1 \pm 14.5$	$\frac{47.2 \pm 29.9}{20.4 \pm 16.5}$	$1.2 \pm 2.4$ $1.7 \pm 1.2$	18 0 ± 13 3		
Walking - Treadmill 3mph - Treadmill 3 - light	$49.2 \pm 15.1$	$1.0 \pm 0.4$ 1 3 ± 0.4	$47.9 \pm 15.2$	$17.7 \pm 16.2$	$1.7 \pm 1.2$ $1.4 \pm 0.9$	$17.3 \pm 12.0$		
Walking - Treadmill 3mph - Treadmill 6 - moderate	$49.2 \pm 13.3$	$1.5 \pm 0.4$ 1 1 + 0 4	$51.0 \pm 13.2$	$11.7 \pm 10.2$ 11.9 + 10.1	$1.4 \pm 0.9$ 1.3 ± 0.8	$17.3 \pm 12.0$ $12.2 \pm 9.4$		
Walking - Treadmill 3mph - Treadmill 9 - hard	$57.9 \pm 20.3$	$1.1 \pm 0.4$ 0.9 ± 0.4	$51.0 \pm 14.0$ 58.6 ± 19.2	$11.9 \pm 10.1$ $13.6 \pm 13.4$	$1.3 \pm 0.6$	$12.2 \pm 9.4$ $13.8 \pm 12.8$		
kneeling	$93.9 \pm 20.3$	$0.9 \pm 0.4$	$95.0 \pm 19.2$	$13.0 \pm 13.4$	$1.3 \pm 0.0$ $1.7 \pm 2.8$	$13.0 \pm 12.0$ 52.4 ± 36.5		
Corrying groceries	$93.9 \pm 9.0$ 70.1 ± 10.8	$0.0 \pm 0.1$	$95.0 \pm 0.1$	$41.4 \pm 27.3$	$1.7 \pm 2.8$ $1.5 \pm 0.8$	$32.4 \pm 30.3$		
Doing dishes	$79.1 \pm 10.8$ 72.0 ± 12.6	$0.3 \pm 0.3$	$79.1 \pm 11.0$ $72.0 \pm 12.0$	$41.4 \pm 27.3$ 21.1 $\pm$ 17.4	$1.3 \pm 0.8$ 1.7 ± 1.0	$39.0 \pm 23.4$		
Gardening	$73.0 \pm 13.0$	$0.7 \pm 0.4$	$72.0 \pm 13.0$	$31.1 \pm 17.4$ 11.8 ± 12.2	$1.7 \pm 1.0$ $1.7 \pm 1.6$	$26.6 \pm 14.8$ 12.0 ± 13.3		
Ironing	$08.3 \pm 13.0$	$0.7 \pm 0.4$	$06.2 \pm 14.7$ 75.0 ± 12.0	$11.0 \pm 12.2$ 28.2 ± 20.4	$1.7 \pm 1.0$ $1.8 \pm 0.8$	$12.0 \pm 13.3$		
Making the had	$17.2 \pm 12.9$	$0.7 \pm 0.4$	$73.9 \pm 13.0$	$28.3 \pm 20.4$	$1.0 \pm 0.0$	$20.8 \pm 10.0$ 22.5 ± 12.0		
Manning	$45.3 \pm 14.1$	$1.3 \pm 0.3$	$44.3 \pm 13.4$	$23.9 \pm 14.3$	$1.9 \pm 0.8$	$22.3 \pm 12.0$		
Mopping Disving video comes	$43.3 \pm 20.7$	$1.4 \pm 0.0$	$43.2 \pm 21.1$	$16.7 \pm 11.7$	$2.0 \pm 0.0$	$17.0 \pm 10.8$		
Playing videoganies	$\frac{67.6 \pm 12.1}{74.6 \pm 24.6}$	$0.5 \pm 0.5$	$\frac{67.7 \pm 11.6}{74.2 \pm 24.7}$	$12.2 \pm 17.1$	$1.5 \pm 0.9$	$13.2 \pm 17.3$		
Scrubbing a surface	$74.0 \pm 24.0$	$0.0 \pm 0.0$	$74.5 \pm 24.7$	$33.8 \pm 29.3$	$2.1 \pm 1.0$	$26.2 \pm 23.0$		
Stacking groceries	$46.3 \pm 19.1$	$1.1 \pm 0.5$	$46.0 \pm 18.8$	$11.2 \pm 7.9$	$1.7 \pm 0.8$	$10.8 \pm 7.4$		
Sweeping	$39.4 \pm 10.3$	$1.5 \pm 0.5$	$39.3 \pm 15.0$	$15.6 \pm 8.0$	$2.0 \pm 0.5$	$15.8 \pm 8.1$		
1 yping	$90.3 \pm 9.0$	$0.3 \pm 0.3$	89.9 ± 8.6	$41.0 \pm 29.7$	$1.4 \pm 1.0$	$38.0 \pm 21.8$		
Vacuuming	$43.5 \pm 20.2$	$1.3 \pm 0.5$	$44.2 \pm 19.6$	$18.4 \pm 13.6$	$1.7 \pm 0.7$	$18.2 \pm 10.7$		
Walking around block	/1./±15.0	$0.6 \pm 0.4$	$72.8 \pm 14.5$	$26.8 \pm 12.2$	$2.4 \pm 1.3$	$23.9 \pm 9.8$		
wasning windows	$53.1 \pm 14.8$	$1.1 \pm 0.4$	$55.8 \pm 13.8$	$25.5 \pm 16.0$	$2.0 \pm 1.7$	$25.8 \pm 16.2$		
Watching I V	8/.1 ± 15.0	$0.4 \pm 0.3$	86.5 ± 13.5	$24.5 \pm 28.4$	$1.9 \pm 1.2$	$21.4 \pm 23.5$		
Weeding	$51.5 \pm 22.7$	$0.9 \pm 0.6$	$54.6 \pm 23.1$	$9.4 \pm 22.6$	$1.2 \pm 0.9$	$5.4 \pm 7.4$		
wiping/Dusting	$44.0 \pm 12.2$	$1.1 \pm 0.4$	46.4 ± 12.6	16.0 ± 9.4	$1.7 \pm 0.8$	$10.2 \pm 8.7$		
Writing	$92.2 \pm 7.7$	$0.2 \pm 0.2$	$91.8 \pm 7.6$	$43.2 \pm 27.1$	$1.5 \pm 1.6$	$41.9 \pm 24.5$		
taking out trash	$+43.2 \pm 13.9$	$1.1 \pm 0.4$	$43.9 \pm 14.7$	$9.4 \pm 1.2$	$1.2 \pm 0.5$	$11.0 \pm 7.8$		

Table A8-9: Performance of the C4.5 classifier in recognizing the 51 activities in the MIT dataset (without the *unknown* class) using the accelerometers at the dominant foot, and the *invariant reduced* feature set over windows of 5.6s in length during subject dependent and independent evaluation.

Activity	Su	biect Depend	ent	Subject Independent				
	TP Rate	FP Rate	F-Measure	TP Rate	FP Rate	F-Measure		
Bench weight lifting - hard	$74.9 \pm 21.7$	$0.3 \pm 0.3$	$74.4 \pm 22.0$	$0.3 \pm 1.2$	$0.7 \pm 1.0$	$0.2 \pm 0.8$		
Bench weight lifting - light	$91.0 \pm 11.4$	$0.1 \pm 0.1$	$91.3 \pm 9.7$	$11.8 \pm 19.2$	$1.5 \pm 1.3$	$10.1 \pm 18.0$		
Bench weight lifting - moderate	80.5 ± 17.3	$0.2 \pm 0.2$	82.9 ± 16.0	$4.8 \pm 11.7$	$1.8 \pm 3.0$	$2.3 \pm 4.3$		
Bicep curls - hard	$80.0 \pm 19.7$	$0.3 \pm 0.2$	$80.7 \pm 19.0$	$10.0 \pm 23.1$	$0.9 \pm 1.4$	$7.9 \pm 15.2$		
Bicep curls - light	$86.2 \pm 12.8$	$0.3 \pm 0.3$	85.7 ± 12.5	$16.2 \pm 26.3$	$1.4 \pm 1.2$	$14.6 \pm 21.2$		
Bicep curls - moderate	$81.1 \pm 14.2$	$0.4 \pm 0.4$	$80.4 \pm 14.8$	$20.0 \pm 31.2$	$1.5 \pm 1.4$	$14.4 \pm 19.8$		
Calisthenics - Crunches	$98.0 \pm 2.6$	$0.0 \pm 0.0$	$97.6 \pm 2.2$	$63.0 \pm 41.2$	$0.9 \pm 2.2$	$65.0 \pm 41.4$		
Calisthenics - Sit ups	$94.0 \pm 7.5$	$0.1 \pm 0.1$	$93.1 \pm 6.4$	$80.0 \pm 27.8$	$0.2 \pm 0.3$	$80.0 \pm 26.7$		
Cycling - Cycle hard - Cycle 80rpm	$73.2 \pm 14.3$	$0.4 \pm 0.3$	$73.2 \pm 14.8$	$21.4 \pm 33.6$	$1.0 \pm 1.0$	$15.8 \pm 20.6$		
Cycling - Cycle light - Cycle 100rpm	$98.4 \pm 3.9$	$0.0 \pm 0.0$	$97.4 \pm 3.7$	$90.4 \pm 17.4$	$0.3 \pm 0.8$	$88.6 \pm 17.0$		
Cycling - Cycle light - Cycle 60rpm	$99.2 \pm 1.5$	$0.1 \pm 0.0$	$98.2 \pm 1.5$	$91.9 \pm 22.0$	$0.1 \pm 0.2$	$91.9\pm20.8$		
Cycling - Cycle light - Cycle 80rpm	$92.8 \pm 6.8$	$0.2 \pm 0.2$	$92.3 \pm 7.8$	$50.2 \pm 42.3$	$1.4 \pm 1.4$	$37.5 \pm 29.2$		
Cycling - Cycle moderate - Cycle 80rpm	$76.5 \pm 14.5$	$0.4 \pm 0.3$	$76.9 \pm 14.0$	$42.9 \pm 35.6$	$1.4 \pm 1.0$	32.3 ± 21.1		
Lying down	$99.3 \pm 1.2$	$0.1 \pm 0.1$	$99.1 \pm 1.4$	33.4 ± 27.4	$2.2 \pm 2.2$	$36.7 \pm 28.5$		
Rowing - Rowing hard - Rowing 30spm	72.0 + 18.7	$0.5 \pm 0.3$	71.6 + 17.8	31.9 ± 31.3	$1.0 \pm 0.9$	$27.4 \pm 22.4$		
Rowing - Rowing light - Rowing 30spm	$80.7 \pm 13.2$	0.5 + 0.3	79.4 + 13.8	36.3 ± 30.2	$1.1 \pm 1.0$	33.0 ± 23.1		
Rowing - Rowing moderate - Rowing 30spm	64.5 + 19.2	0.7 + 0.4	65.1 + 19.1	24.6 ± 30.7	$1.1 \pm 1.3$	$21.4 \pm 22.8$		
Running - Treadmill 4mph - Treadmill 0	93.4 + 5.6	$0.2 \pm 0.1$	92.8 + 4.4	$50.4 \pm 35.4$	$0.7 \pm 0.7$	50.5 ± 33.2		
Running - Treadmill 5mph - Treadmill 0	88.3 + 9.5	0.2 + 0.2	89.2 + 8.6	$61.2 \pm 26.7$	$0.8 \pm 0.4$	$58.2 \pm 19.5$		
Running - Treadmill 6mph - Treadmill 0	85.7 + 11.9	0.2 + 0.2	85.3 + 11.6	59.6 ± 37.1	$0.6 \pm 0.7$	$52.2 \pm 33.0$		
Sitting	94.3 + 7.4	$0.1 \pm 0.2$	93.1 + 7.9	$3.5 \pm 5.4$	$1.8 \pm 1.4$	$3.2 \pm 5.6$		
Sitting - Fidget feet legs	$922 \pm 58$	$0.1 \pm 0.1$	$935 \pm 48$	41.8 + 32.9	$0.7 \pm 0.6$	41.4 + 32.3		
Sitting - Fidget hands arms	$90.8 \pm 11.6$	$0.1 \pm 0.1$ 0.1 + 0.2	$90.2 \pm 11.0$	$4.8 \pm 11.0$	$1.9 \pm 1.9$	$2.7 \pm 5.4$		
Stairs - Ascend stairs	$90.0 \pm 11.0$ 81 9 + 8 7	$0.1 \pm 0.2$ $0.4 \pm 0.2$	$90.2 \pm 11.0$ 81 1 + 8 1	54.4 + 28.2	$0.9 \pm 0.6$	52.8 + 22.8		
Stairs - Descend stairs	77.4 + 8.2	$0.4 \pm 0.2$ 0.4 + 0.2	77.9 + 8.5	40.0 + 18.7	$1.2 \pm 0.6$	38.1 + 16.7		
Standing	$895 \pm 84$	$0.1 \pm 0.2$	881+86	19.7 + 22.8	$1.0 \pm 0.6$	$183 \pm 180$		
Walking - Treadmill 2mph - Treadmill 0	949 + 48	$0.2 \pm 0.2$	$93.0 \pm 4.3$	65.8 + 28.2	$0.6 \pm 0.4$	65.8 + 22.9		
Walking - Treadmill 3mph - Treadmill 0	$71.0 \pm 15.5$	$0.2 \pm 0.1$ 0.8 + 0.4	$69.2 \pm 15.7$	20.2 + 19.3	1.4 + 1.0	$18.8 \pm 13.4$		
Walking - Treadmill 3mph - Treadmill 3 - light	$562 \pm 172$	$1.0 \pm 0.1$	$55.9 \pm 17.5$	18.9 + 13.5	$1.2 \pm 0.9$	$19.8 \pm 11.8$		
Walking - Treadmill 3mph - Treadmill 6 - moderate	$50.2 \pm 17.2$ 58 3 + 15 6	$1.0 \pm 0.4$ 0.9 + 0.4	$58.5 \pm 17.9$	$17.0 \pm 16.0$	1.2 = 0.9 1.4 + 1.0	$162 \pm 120$		
Walking - Treadmill 3mph - Treadmill 9 - hard	742 + 162	$0.5 \pm 0.4$ 0.6 + 0.4	$74.0 \pm 15.6$	$18.3 \pm 16.0$	$1.4 \pm 1.0$	$18.3 \pm 13.6$		
kneeling	922 + 82	$0.0 \pm 0.1$	$90.1 \pm 9.1$	12.9 + 23.0	$1.0 \pm 0.8$	13.2 + 23.3		
Carrying groceries	$64.8 \pm 17.3$	$0.2 \pm 0.2$ 0.7 + 0.4	$664 \pm 177$	29.8 + 27.0	$2.3 \pm 1.2$	24.0 + 18.1		
Doing dishes	$67.6 \pm 16.1$	$0.7 \pm 0.4$ 0.9 ± 0.6	$65.8 \pm 16.3$	$22.5 \pm 13.4$	$3.0 \pm 1.2$	$17.6 \pm 9.2$		
Gardening	$60.0 \pm 18.5$	$1.0 \pm 0.5$	$58.6 \pm 19.0$	67 + 73	$2.0 \pm 1.3$	7.0 + 8.2		
Ironing	$63.0 \pm 14.9$	$1.0 \pm 0.0$ $1.0 \pm 0.4$	$62.7 \pm 14.9$	293 + 189	$2.6 \pm 1.6$	244 + 138		
Making the bed	$43.6 \pm 13.1$	$1.0 \pm 0.4$ $1.6 \pm 0.5$	$42.4 \pm 12.7$	$23.8 \pm 10.9$	$2.0 \pm 0.6$	22.7 + 8.9		
Monning	$43.0 \pm 13.1$	$1.0 \pm 0.3$ $1.6 \pm 0.4$	$42.4 \pm 12.7$ $42.4 \pm 11.5$	$23.0 \pm 11.1$ $23.4 \pm 15.3$	$1.8 \pm 0.7$	$22.3 \pm 11.1$		
Plaving videogames	$97.8 \pm 11.5$	$1.0 \pm 0.4$	$968 \pm 4.7$	19.1 + 31.5	22 + 21	154 + 209		
Scrubbing a surface	$57.3 \pm 4.0$ 60.7 ± 20.8	$0.1 \pm 0.1$ 0.8 + 0.4	$50.0 \pm 4.7$	$18.6 \pm 11.8$	2.2 = 2.1 2.7 + 1.9	$15.1 \pm 2019$ 15.3 + 9.1		
Stacking groceries	$39.1 \pm 18.3$	$1.2 \pm 0.4$	$30.0 \pm 19.3$	$14.8 \pm 8.8$	$1.7 \pm 0.6$	$13.5 \pm 9.1$ 14.5 ± 9.0		
Sweening	$29.8 \pm 14.0$	$1.2 \pm 0.5$	$31.1 \pm 13.8$	$138 \pm 98$	$23 \pm 0.9$	$12.8 \pm 7.5$		
Typing	$27.3 \pm 14.0$ 03.2 ± 11.7	$1.3 \pm 0.3$	$925 \pm 11.7$	$20.0 \pm 35.6$	$1.4 \pm 1.6$	$17.5 \pm 28.0$		
Vacuuming	$33.2 \pm 11.7$	$0.2 \pm 0.4$ $1.4 \pm 0.4$	$32.3 \pm 11.7$	$20.0 \pm 35.0$ 21.9 + 10.2	$1.4 \pm 1.0$ $1.6 \pm 0.5$	$17.3 \pm 20.0$ 22.3 + 9.5		
Walking around block	$33.5 \pm 10.0$	$1.4 \pm 0.4$	$39.4 \pm 11.0$	$21.9 \pm 10.2$ $24.9 \pm 19.4$	$1.0 \pm 0.3$ $3.0 \pm 2.4$	$22.3 \pm 7.3$		
Washing windows	$40.6 \pm 12.2$	$0.7 \pm 0.4$ 13 $\pm 0.4$	$110 \pm 122$	18.4 + 11.8	$1.0 \pm 2.4$ $1.8 \pm 0.7$	$18.9 \pm 12.8$		
Watching TV	$+0.0 \pm 12.3$	$1.3 \pm 0.4$ 0.1 ± 0.1	$+1.7 \pm 12.3$	$5.4 \pm 13.9$	$1.0 \pm 0.7$ $1.4 \pm 1.4$	$5.4 \pm 10.1$		
Weeding	$30.7 \pm 3.3$	$0.1 \pm 0.1$	$90.3 \pm 3.0$ 52.0 $\pm 12.5$	$5.4 \pm 13.9$ $67 \pm 7.8$	$1.4 \pm 1.4$ $1.6 \pm 1.4$	$3.4 \pm 10.1$ 8 3 + 10 4		
Wining/Ducting	$+9.2 \pm 13.0$ 25.9 $\pm$ 15.2	$0.9 \pm 0.4$ 1.2 ± 0.2	$32.7 \pm 13.3$ $27.2 \pm 14.7$	$0.7 \pm 7.0$ 12.2 ± 8.2	$1.0 \pm 1.4$ $1.6 \pm 0.9$	$0.3 \pm 10.4$ 13.3 $\pm$ 8.6		
Writing	$33.0 \pm 13.3$	$1.3 \pm 0.3$	$31.2 \pm 14.7$ $02.8 \pm 0.6$	$12.2 \pm 0.2$ 11+26	$1.0 \pm 0.0$ $2.6 \pm 2.9$	$13.3 \pm 0.0$ $1.4 \pm 3.2$		
taking out trach	$7+.2 \pm 7.9$ $13.8 \pm 14.0$	$0.2 \pm 0.3$ 11 ± 0.2	$73.0 \pm 7.0$	$1.1 \pm 2.0$ $17.9 \pm 12.1$	$1.0 \pm 2.0$ $1.7 \pm 0.8$	$1.4 \pm 3.2$ 18.2 + 11.8		

Table A8-10: Performance of the C4.5 classifier in recognizing the 51 activities in the MIT dataset (without the *unknown* class) using the accelerometers at the dominant thigh, and the *invariant reduced* feature set over windows of 5.6s in length during subject dependent and independent evaluation.

Activity	Sul	oject Depend	lent	Subject Independent				
, , , , , , , , , , , , , , , , , , ,	TP Rate	FP Rate	F-Measure	TP Rate	FP Rate	F-Measure		
Bench weight lifting - hard	$46.2 \pm 20.2$	$0.5 \pm 0.2$	$45.2 \pm 18.9$	$7.4 \pm 11.5$	$0.4 \pm 0.3$	8.5 ± 13.6		
Bench weight lifting - light	$70.4 \pm 18.6$	$0.5 \pm 0.3$	68.1 ± 16.9	$44.4 \pm 22.1$	$1.4 \pm 1.4$	38.7 ± 21.1		
Bench weight lifting - moderate	$50.0 \pm 18.6$	$0.6 \pm 0.3$	$50.7 \pm 19.0$	$13.0 \pm 12.8$	$1.1 \pm 1.1$	$13.2 \pm 13.4$		
Bicep curls - hard	$71.8 \pm 18.9$	$0.5 \pm 0.4$	$71.0 \pm 18.0$	$14.4 \pm 19.0$	$1.0 \pm 0.9$	$13.3 \pm 17.0$		
Bicep curls - light	69.5 + 14.6	$0.6 \pm 0.3$	$69.5 \pm 14.2$	16.6 + 19.6	$1.8 \pm 1.9$	14.7 + 15.9		
Bicep curls - moderate	64.9 + 17.2	$0.7 \pm 0.4$	$64.9 \pm 17.5$	14.0 + 18.3	$1.3 \pm 1.0$	12.1 + 13.0		
Calisthenics - Crunches	92.0 + 8.1	$0.1 \pm 0.1$	92.8 + 7.0	54.9 + 42.6	$0.5 \pm 0.6$	52.0 + 38.6		
Calisthenics - Sit ups	$94.5 \pm 5.0$	$0.1 \pm 0.1$	93.7 + 4.5	87.0 + 24.1	$0.2 \pm 0.4$	85.2 + 22.8		
Cycling - Cycle hard - Cycle 80rpm	78.2 + 17.4	$0.2 \pm 0.1$	79.2 + 14.7	37.9 + 36.3	$1.2 \pm 1.1$	27.3 + 24.6		
Cycling - Cycle light - Cycle 100rpm	$97.8 \pm 4.1$	$0.0 \pm 0.0$	$97.8 \pm 2.8$	981 + 44	$0.1 \pm 0.3$	$965 \pm 62$		
Cycling - Cycle light - Cycle 60rpm	98.4 + 2.4	$0.0 \pm 0.1$	$98.0 \pm 1.9$	95.2 + 7.3	$0.1 \pm 0.2$	$95.0 \pm 5.6$		
Cycling - Cycle light - Cycle 80rpm	$934 \pm 69$	$0.2 \pm 0.2$	$93.1 \pm 7.2$	33.6 + 35.2	11 + 12	$30.0 \pm 26.3$		
Cycling - Cycle moderate - Cycle 80rpm	$85.9 \pm 11.5$	$0.2 \pm 0.2$	845+119	40.1 + 34.5	$1.2 \pm 1.0$	$33.3 \pm 21.8$		
Lying down	$99.9 \pm 0.4$	$0.5 \pm 0.5$	$99.5 \pm 0.8$	72.4 + 24.8	$0.7 \pm 1.3$	$77.6 \pm 15.1$		
Rowing - Rowing hard - Rowing 30spm	$67.8 \pm 17.1$	$0.1 \pm 0.1$	$67.0 \pm 16.3$	$72.4 \pm 24.0$ 29.3 + 19.6	$1.1 \pm 0.8$	$77.0 \pm 13.1$ 28.1 + 14.5		
Rowing - Rowing light - Rowing 30spm	$72.0 \pm 17.1$	$0.0 \pm 0.3$	$70.7 \pm 10.5$	$29.3 \pm 19.0$ 50.2 + 20.3	$1.1 \pm 0.0$ $1.6 \pm 0.7$	$42.8 \pm 13.7$		
Rowing - Rowing moderate - Rowing 30spm	$72.0 \pm 10.2$ 65.1 + 14.9	$0.0 \pm 0.3$	$64.5 \pm 13.5$	$15.0 \pm 11.2$	$0.8 \pm 0.6$	$17.9 \pm 10.7$		
Running - Treadmill 4mph - Treadmill 0	$96.2 \pm 4.6$	$0.1 \pm 0.3$	$955 \pm 45$	$13.0 \pm 11.2$ 52.5 + 33.0	$1.2 \pm 1.3$	$46.9 \pm 27.6$		
Running Treadmill 5mph Treadmill 0	$95.0 \pm 5.6$	$0.1 \pm 0.1$	$94.7 \pm 4.5$	$52.3 \pm 33.0$ 55.2 ± 32.6	$1.2 \pm 1.3$	$51.7 \pm 27.0$		
Running - Treadmill 6mph - Treadmill 0	$93.0 \pm 5.0$ $94.0 \pm 6.4$	$0.1 \pm 0.1$	$94.7 \pm 4.3$ $94.3 \pm 4.9$	$55.2 \pm 32.0$	$0.9 \pm 0.7$	$56.9 \pm 29.6$		
Citting	$99.4 \pm 14.2$	$0.1 \pm 0.1$	$97.2 \pm 12.2$	$16.1 \pm 19.1$	$0.0 \pm 0.5$	$14.7 \pm 15.4$		
Sitting Fidget feet legs	$90.4 \pm 14.2$	$0.2 \pm 0.2$	$87.3 \pm 13.2$ $87.7 \pm 7.4$	$10.1 \pm 10.1$	$1.1 \pm 0.0$	$14.7 \pm 13.4$ 64.4 ± 28.7		
Sitting - Fidget hands arms	90.4 ± 7.7	$0.2 \pm 0.1$	07.7 ± 7.4	$06.1 \pm 34.0$	$0.4 \pm 0.3$	$04.4 \pm 20.7$		
Sitting - Fluget findus arms	$19.4 \pm 12.9$	$0.3 \pm 0.2$	$19.7 \pm 13.0$	$24.7 \pm 22.4$	$1.3 \pm 1.3$	$22.0 \pm 20.0$		
Stairs - Ascend stairs	$80.0 \pm 10.5$	$0.3 \pm 0.2$	$84.8 \pm 9.8$	$76.9 \pm 23.7$	$0.6 \pm 0.4$	$73.4 \pm 21.3$		
Stalls - Descend stalls	$80.2 \pm 12.0$	$0.3 \pm 0.2$	$80.2 \pm 12.1$	$62.0 \pm 27.4$	$1.0 \pm 1.0$	$30.9 \pm 23.8$		
Standing Wellsing Transforill Querk, Transforill O	$84.7 \pm 13.4$	$0.2 \pm 0.2$	$83.0 \pm 12.8$	$28.8 \pm 26.0$	$0.9 \pm 0.8$	$26.9 \pm 19.8$		
Walking - Treadmin 2mph - Treadmin 0	$92.0 \pm 3.4$	$0.2 \pm 0.2$	$91.9 \pm 3.0$	$67.0 \pm 23.1$	$0.8 \pm 1.2$	$00.3 \pm 22.8$		
Walking - Treadmill 3mph - Treadmill 0	$75.7 \pm 14.0$	$0.6 \pm 0.4$	$74.9 \pm 14.1$	$17.6 \pm 17.9$	$1.3 \pm 1.0$	$1/.5 \pm 1/.4$		
waiking - Treadmit Smph - Treadmit S - light	$67.0 \pm 15.0$	$0.8 \pm 0.4$	$65.5 \pm 14.5$	$13.2 \pm 16.1$	$1.3 \pm 0.9$	$12.4 \pm 12.2$		
Walking - Treadmill 3mph - Treadmill 6 - moderate	$64.3 \pm 19.1$	$0.8 \pm 0.3$	$64.2 \pm 17.6$	$20.6 \pm 16.4$	$1.6 \pm 1.0$	$19.9 \pm 13.6$		
Walking - Treadmill 3mph - Treadmill 9 - hard	/8.9 ± 11.2	$0.4 \pm 0.3$	80.1 ± 11.5	$30.0 \pm 22.3$	$1.1 \pm 0.8$	31.1 ± 18.9		
kneeling	85.3 ± 10.0	$0.3 \pm 0.2$	83.4 ± 11.6	$25.4 \pm 20.0$	$1.3 \pm 0.9$	$23.7 \pm 19.0$		
Carrying groceries	76.4 ± 9.1	$0.6 \pm 0.3$	75.9 ± 9.7	$25.7 \pm 19.7$	$2.0 \pm 1.1$	$24.2 \pm 17.3$		
Doing dishes	$58.7 \pm 15.0$	$1.0 \pm 0.3$	$57.8 \pm 13.0$	$28.2 \pm 12.2$	$2.2 \pm 1.1$	$25.6 \pm 11.0$		
Gardening	$54.0 \pm 17.1$	$1.2 \pm 0.5$	$52.7 \pm 18.4$	$15.8 \pm 15.0$	$1.3 \pm 0.7$	$16.9 \pm 15.8$		
Ironing	$64.4 \pm 12.3$	$1.0 \pm 0.4$	$62.9 \pm 11.7$	$37.2 \pm 14.6$	$1.5 \pm 0.6$	$37.5 \pm 11.5$		
Making the bed	$38.3 \pm 10.2$	$1.5 \pm 0.4$	$38.8 \pm 10.0$	$29.3 \pm 11.8$	$2.0 \pm 0.8$	$27.2 \pm 9.0$		
Mopping	$47.6 \pm 13.7$	$1.4 \pm 0.4$	46.1 ± 13.8	$26.0 \pm 11.5$	$1.9 \pm 0.5$	$25.0 \pm 10.3$		
Playing videogames	$94.1 \pm 6.8$	$0.2 \pm 0.2$	$93.4 \pm 6.6$	$23.4 \pm 33.1$	$1.9 \pm 1.4$	$19.9 \pm 23.9$		
Scrubbing a surface	$53.2 \pm 16.8$	$1.0 \pm 0.4$	$53.7 \pm 16.2$	$21.2 \pm 13.9$	$2.1 \pm 1.2$	$20.1 \pm 12.2$		
Stacking groceries	$45.4 \pm 14.6$	$1.0 \pm 0.4$	$46.2 \pm 13.9$	$19.6 \pm 9.7$	$1.5 \pm 0.7$	$19.8 \pm 9.5$		
Sweeping	$41.6 \pm 16.5$	$1.4 \pm 0.4$	$41.6 \pm 15.9$	$17.6 \pm 5.3$	$1.9 \pm 0.4$	$17.7 \pm 5.3$		
Typing	$89.6 \pm 13.3$	$0.3 \pm 0.4$	89.6 ± 12.1	$14.9 \pm 26.9$	$2.0 \pm 1.3$	$11.7 \pm 17.9$		
Vacuuming	53.0 ± 13.3	$0.9 \pm 0.4$	55.6 ± 13.4	35.8 ± 12.2	$1.6 \pm 0.9$	$35.6 \pm 11.4$		
Walking around block	$74.0 \pm 11.1$	$0.6 \pm 0.2$	$74.6\pm9.9$	$21.0 \pm 10.7$	$3.0 \pm 1.8$	$17.2 \pm 8.4$		
Washing windows	$48.1\pm10.6$	$1.1 \pm 0.4$	$50.2 \pm 11.5$	$24.8 \pm 18.4$	$1.8\pm1.0$	$23.2\pm14.6$		
Watching TV	$93.8 \pm 5.5$	$0.2 \pm 0.2$	$93.2 \pm 6.1$	8.0 ± 12.7	$1.7 \pm 1.2$	$9.2 \pm 15.0$		
Weeding	$46.7 \pm 15.1$	$0.9\pm0.2$	$49.8 \pm 15.5$	$16.6 \pm 23.2$	$0.9\pm0.3$	$14.6\pm14.6$		
Wiping/Dusting	$43.9\pm13.6$	$1.2 \pm 0.4$	$45.1 \pm 13.0$	$22.1 \pm 9.3$	$1.5 \pm 0.7$	$23.3 \pm 9.4$		
Writing	$90.3\pm13.4$	$0.2\pm0.2$	$90.8\pm11.8$	$47.7\pm37.2$	$1.7 \pm 1.6$	$40.0\pm29.3$		
taking out trash	$42.5 \pm 16.0$	$1.1 \pm 0.4$	$43.9 \pm 15.4$	$21.7 \pm 13.6$	$1.4 \pm 0.5$	$22.6 \pm 12.7$		

Table A8-11: Performance of the C4.5 classifier in recognizing the 51 activities in the MIT dataset (without the *unknown* class) using the accelerometers at the dominant upper arm, and the *invariant reduced* feature set over windows of 5.6s in length during subject dependent and independent evaluation.



Figure A8-1: Confusion Matrix for C4.5 Classifier using the *invariant reduced* feature evaluated in a subject dependent manner when features are computed per axis over windows of 5.6s. The activities to recognize are the 51 activities contained in the MIT dataset without including the unknown class.

	0 M - 0 - 7 A A A A A A A A A A A A A A A A A A	
	ねきごりますのはまずれたおおののなれるモワマダメンターペーペート みつじきゅうかん うちょ かれつかびます	
	800000400000004000004004000000000000000	
	013 0 H [100 H ]300 H 200 H 000 H 000 H 000 H 000 H 000 H 00 H 00 H 0 H	
	ан состания с с с с с с с с с с с с с с с с с с с	
	0 H 4 H H 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
	0 1 1 1 1 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
	а с с с с с с с с с с с с с	
		ock ock
	0,000,000,000,000,000,000,000,000,000,	game urfa erie sah
	00110000000000000000000000000000000000	- bee deoc rocc rocc tine
	00000000000000000000000000000000000000	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1
	2000004+00000004000+010 1000004+0000000000	ping ying ving ckirk king king ting ting
	80400000000000000000000000000000000000	LICO MADA MADA Plan Mal Mal Marta Muripe
	#+++++++++++++++++++++++++++++++++++++	<u> </u>
	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	00404-0×-E40Q0+0
		ight
s		
v	· · · · · · · · · · · · · · · · · · ·	
Ω	100000000000000000000000000000000000000	
IE	、000000000000000000000000000000000000	eadm eadm eadm eadm eadm
ΙF		
s		phone source sou
V	-00000010001000000000000000000000000000	ann de la company ann de la co
L L	й иоиоооооооооооооооооооооооооооооооооо	in in it is the initial in the initial is the initi
	х х х х х х х х х х х х х х х х х х х	ead dge dge cead ead cer cer
Ч		B B C C C C C C C C C C C C C C C C C C
M	のまちゅうくののであるのののののののののののののののののののののののののののののののです。 2011年9月1日の100000000000000000000000000000000000	ng ng ng ng ng ng ng ng ng ng ng ng ng n
Ā	00000000000000000000000000000000000000	unni ittti ittti ittti ittti ttti itttti ittti ittti ittti ittti ittti ittti ittti ittti ittti ittti i
ЕХ	1 1 1 1 1 1 1 1 1 1 1 1 1 1	
	7 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	してした。 「」」」」」」」」」」」」」」」」」」」」」」」」」」」」」」」」」」」」
	м коососсоссоссоссоссоссими и ч косссссссссссссссссссссссссссссссссссс	
	00000000000000000000000000000000000000	
	0,0,0 9,0000000000000000000000000000000	
	00000000000000000000000000000000000000	
		e E.o
		orpa 11_1_1_11_11_11_11_11_11_11_11_11_11_11
		orpa 1001 1001 1001 1001 1001 1001 1001 10
	роососсобрана с мала с с с с с с с с с с с с с с с с с с	Trate Tree Tree
	но о о о о о о о о о о о о о о о о о о	Cycl Cycl node Cycl Cycl add
	目の 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	44pp/ 44pp/ 44pp/ 44pp/ 4p
	аран каланаларын ороосоосоосоосоо оружана осоосоосоосоосоосоосоосоосоосоосоосоосо	ting ting ting ting ting ting ting ting
	росс80,80000000004000000000000000000000000	liff liff han noc liff liff liff liff liff liff liff lif
	N N N N N N N N N N N N N N N N N N N	Jht ls ls ls ls ls ls ls ls ls ls ls ls ls
	90000000000000000000000000000000000000	weis weis curl dowr dowr dowr dowr dowr
	0 6 6 6 6 7 6 7 6 7 6 7 6 7 0 0 0 0 0 0	ach sep sep lint ding ding ding ding
		Ben Ber Bic Bic Cal Bic Cyc Cyc Cyc Cyc Cyc Cyc Cyc Roww R Row R Row R Row
		$\begin{array}{c} \uparrow \uparrow \uparrow \uparrow \uparrow \uparrow \uparrow \uparrow \uparrow \uparrow \uparrow \uparrow \uparrow \uparrow \uparrow \uparrow \uparrow \uparrow \uparrow$
		K H U U H H U H U H U H U H U H U H U H

Figure A8-2: Confusion Matrix for C4.5 Classifier using the *invariant reduced* feature evaluated in a subject independent manner when features are computed per axis over windows of 5.6s. The activities to recognize are the 51 activities contained in the MIT dataset without including the *unknown* class.

#### All Activities without Discriminating Intensity Levels

Activity	Sub	ject Depend	dent	Subject Independent					
	TP Rate	FP Rate	F-Measure	TP Rate	FP Rate	<b>F-Measure</b>			
Bench weight lifting	$96.2 \pm 7.4$	$0.1 \pm 0.1$	$96.7 \pm 4.7$	$79.4 \pm 27.4$	$0.6 \pm 0.9$	$78.4\pm24.3$			
Bicep curls	$99.3 \pm 0.9$	$0.1 \pm 0.0$	$99.1 \pm 0.6$	$81.1 \pm 32.3$	$0.4 \pm 0.6$	$81.8\pm30.2$			
Calisthenics - Crunches	$95.5 \pm 5.5$	$0.0 \pm 0.1$	$96.1 \pm 3.8$	$65.6 \pm 34.8$	$0.8 \pm 1.2$	$62.5 \pm 31.4$			
Calisthenics - Sit ups	$95.3 \pm 4.5$	$0.1 \pm 0.1$	$94.4 \pm 4.2$	$76.2 \pm 32.2$	$0.3 \pm 0.4$	$74.4 \pm 29.9$			
Cycling	$99.8 \pm 0.4$	$0.0\pm0.0$	$99.7 \pm 0.3$	$98.3 \pm 2.8$	$0.4 \pm 1.1$	$97.4 \pm 4.2$			
Lying down	$100.0 \pm 0.2$	$0.0\pm0.0$	$99.8 \pm 0.4$	$95.9 \pm 12.6$	$0.3 \pm 0.6$	$95.8 \pm 8.8$			
Rowing	$99.7 \pm 0.5$	$0.0\pm0.0$	$99.6 \pm 0.6$	$89.8 \pm 21.0$	$0.0 \pm 0.1$	$92.5 \pm 17.2$			
Running	99.2 ± 1.3	$0.1 \pm 0.1$	$99.0 \pm 1.0$	$83.2 \pm 24.6$	$0.5 \pm 0.8$	$84.3 \pm 23.1$			
Sitting	$97.7 \pm 2.2$	$0.1 \pm 0.1$	$97.6 \pm 1.8$	$51.0 \pm 26.3$	$1.8 \pm 1.8$	$50.1 \pm 19.0$			
Stairs - Ascend stairs	$91.6 \pm 6.0$	$0.2 \pm 0.2$	$90.4 \pm 5.7$	$79.2 \pm 23.6$	$0.4 \pm 0.4$	$76.3\pm20.6$			
Stairs - Descend stairs	$89.3 \pm 9.7$	$0.2 \pm 0.2$	$88.7\pm8.2$	$60.8 \pm 25.5$	$0.6 \pm 0.6$	$60.4 \pm 22.7$			
Standing	$94.2 \pm 5.5$	$0.0 \pm 0.1$	$95.2 \pm 3.9$	$93.4 \pm 8.8$	$0.2 \pm 0.3$	$89.6 \pm 9.1$			
Walking	$97.4 \pm 2.1$	$0.5 \pm 0.4$	$97.4 \pm 2.0$	$90.2 \pm 7.6$	$1.6 \pm 1.2$	$90.7 \pm 5.6$			
kneeling	97.1 ± 3.6	$0.0 \pm 0.1$	$96.8 \pm 3.3$	$95.4 \pm 6.1$	$0.1 \pm 0.3$	$93.7 \pm 8.5$			
Doing dishes	87.3 ± 5.5	$0.4 \pm 0.2$	$86.5 \pm 4.3$	$59.0 \pm 28.5$	$1.2 \pm 0.7$	$52.2 \pm 20.9$			
Gardening	$81.8\pm7.4$	$0.4 \pm 0.2$	$82.2\pm8.8$	$14.1\pm20.8$	$1.1 \pm 1.0$	$14.8 \pm 18.4$			
Ironing	$85.1 \pm 7.8$	$0.4 \pm 0.2$	$85.6\pm7.2$	$48.6\pm28.6$	$1.0 \pm 0.4$	$48.0\pm24.3$			
Making the bed	$61.0\pm11.3$	$0.9 \pm 0.3$	$62.2\pm9.0$	$38.4 \pm 16.3$	$1.6 \pm 0.8$	$37.1 \pm 14.6$			
Mopping	$67.4 \pm 13.9$	$0.8 \pm 0.4$	$66.6 \pm 13.6$	$36.8 \pm 14.0$	$1.9 \pm 1.0$	$33.8\pm9.2$			
Playing videogames	$99.3 \pm 2.3$	$0.0\pm0.0$	$98.9 \pm 1.3$	$60.3\pm45.0$	$1.3 \pm 1.5$	$54.2\pm40.6$			
Scrubbing a surface	$84.2 \pm 11.7$	$0.4 \pm 0.3$	$83.6 \pm 12.1$	$39.7 \pm 33.1$	$1.5 \pm 1.7$	$35.4\pm26.7$			
Stacking groceries	$67.1 \pm 17.3$	$0.6 \pm 0.2$	$67.0 \pm 15.3$	$33.5 \pm 21.6$	$0.9 \pm 0.4$	$34.6 \pm 19.5$			
Sweeping	$66.0 \pm 16.7$	$0.7 \pm 0.3$	$66.7 \pm 13.5$	$34.9 \pm 19.4$	$1.5 \pm 0.6$	$34.4 \pm 17.6$			
Typing	$97.8 \pm 2.2$	$0.0 \pm 0.1$	$98.2 \pm 2.2$	$66.3 \pm 37.4$	$0.6 \pm 0.6$	$64.9\pm34.8$			
Vacuuming	$77.3 \pm 10.9$	$0.6 \pm 0.2$	$77.0\pm8.8$	$58.1 \pm 23.2$	$1.0 \pm 0.5$	$56.2 \pm 19.7$			
Washing windows	$70.9\pm9.6$	$0.6 \pm 0.3$	$72.0\pm9.2$	$47.4 \pm 22.2$	$1.3 \pm 0.9$	$46.2\pm21.3$			
Watching TV	$99.0 \pm 2.2$	$0.1 \pm 0.1$	$98.5 \pm 1.9$	$43.6\pm43.2$	$1.1 \pm 1.0$	$40.0\pm39.4$			
Weeding	$74.4 \pm 11.2$	$0.6 \pm 0.3$	$74.8\pm9.9$	$17.6 \pm 24.5$	$1.1 \pm 0.6$	$14.3 \pm 15.4$			
Wiping/Dusting	$70.0 \pm 11.3$	$0.7 \pm 0.3$	$70.0 \pm 9.9$	$39.6 \pm 21.2$	$1.5 \pm 0.9$	$37.4 \pm 17.8$			
Writing	$98.0\pm2.7$	$0.1 \pm 0.1$	$97.1 \pm 2.5$	$76.4 \pm 25.4$	$1.2 \pm 1.9$	$71.3\pm28.5$			
taking out trash	$64.9 \pm 10.7$	$0.8 \pm 0.4$	$65.4 \pm 10.4$	$26.1 \pm 15.0$	$1.2 \pm 0.5$	$28.3 \pm 15.2$			

Table A8-12: Subject dependent and independent performance of the C4.5 classifier in recognizing the activities contained in the MIT dataset without discriminating among the intensity level of activities (31 activities in total) and without including the *unknown* class. The feature set utilized is the *invariant reduced* feature set computed per axis over windows of 5.6s in length using all the accelerometers (7) available.



Figure A8-3: Confusion Matrix for C4.5 Classifier using the *invariant reduced* feature evaluated in a subject dependent manner when features are computed per axis over windows of 5.6s. The activities to recognize are the activities contained in the MIT dataset without discriminating among intensity levels of an activity (31 activities in total) and without including the unknown class.

	0	ч	-	•	Ċ	-		Z	v	Г			Г	v	<u>م</u>	9	ы	Г									
ASSIFIED AS	0 8 0 5 12 3 2 1 1 0 2 9 1 1 2 1 A	4 1 4 1 0 0 0 1 0 0 0 0 0 1 B 1 3 4 0 2 0 0 1 20 12 0 0 0 0 C R	0 6 4 0 0 10 19 1 3 1 1 0 0 0 1 D 0 0 1 0 0 0 0 0 0 0 0 0 0 0 E	0 0 0 56 1 0 0 0 0 0 23 0 0 0 F			0 26 4 0 0 8 0 4 0 0 3 7 0 18 K 8 0 1 1 0 2 0 3 0 6 1 0 0 0 1 L	0 24 10 0 1 0 6 0 21 5 0 7 2 0 40 M N 0 1 0 0 6 0 0 0 0 0 0 0 1 N	18 7 18 1 29 9 6 0 4 33 12 7 56 2 23 0 7 8 5 1 102 1 27 0 11 19 3 79 28 0 6 P	3 32 20 2 12 9 7 5 25 20 16 13 47 6 11 0 2 20 20 20 20 25 20 16 13 47 6 11 0 1 20 20 20 20 20 20 20 20 20 20 20 20 20	7 71 27 28 0 20 91 91 0 49 18 0 11 40 0 32 8	1 0 0 484 0 0 1 7 0 0 30 1 0 108 0 T 1 19 7 5 295 5 5 0 10 74 0 46 63 9 18 U	:0 32 114 0 22 220 32 0 20 43 0 9 20 0 14 V 8 77 110 10 23 48 268 0 31 16 0 23 36 0 44 W	7 0 0 1 0 0 524 0 1 61 0 0 119 0 X 2 25 54 1 21 19 39 0 443 6 0 38 25 0 14 X A	16 18 17 2 48 23 9 3 11 374 1 29 58 0 15 2 1 0 0 139 2 1 0 11 0 1 345 1 0 74 0 [ R	7 27 27 2 60 8 32 3 21 25 9 89 21 5 9 1 19 18 30 0 57 15 33 0 25 77 3 8 306 1 31	2 2 2 4 12 5 2 2 7 2 7 2 7 2 7 2 7 2 7 2 7 2 7 2		stairs U -> Scrubbing a surface V -> Starking groceries	W -> Sweeping	X -> Typing	Y -> Vacuuming 7 -> Washing windows	[ -> Watching TV	/ -> Weeding	] -> Wiping/Dusting ^ -> Writing	-> taking out trash	
ЕХАМЕ	0 N 0 V 0 0	0 0 0 0 0 0	1 0 0 0 13 0	0 0 0 12 0		21 0 41 0	341 1 89 0 0 402 0 0	106 0 4567 1 0 0 0 399	0 2 0 0	0 8 0	1 T C C C	0 7 0 7 0 0 7 0 0 7 0 0 7 0 7 0 7 0 7 0	1 2 0 0	11 15 1 9	2 2 6 1 0 17 1 1	3 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		K -> Stairs L -> Standi	M -> Walkir	N -> kneeli	Doing - Carder	Q -> Ironir	R -> Making	S -> Moppir T -> Plavir	4 74 9 8 8	
	3 H I J	0 0 1 0	L 0 28 1	0 0 0	0 1374 0	0 0 1 481	0 1 0 0 1 0	1 139 3 49	0 14 0 14 0 14 0	20 20	100	0 152	0 0 0 0 0 0	27	0 148 0	1 0 20	- 0 - - 0 -	•									
	8 C D E 1 0 C	9 133 0 1 0 6 294 17 0 1 2	3 1 353 5 1 0 0 2850 0	0 7 0 0 2043 ( 3 15 10 0 0 0 1625			0 0 1 3 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 93 0 0 1 0 93 0	000000000000000000000000000000000000000	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	+ o o + o o • • • • •		1 1 21 0 1 0 1 3 0 0 0	00	3 6 1 0 0 1 1 0 64 0	0 0 0 0 0 0		, , ,	uch_weight_lifting cen_curls	listhenics - Crunches	listhenics - Sit ups	cling ing down	wing	nning	tting airs - Ascend stairs		
	A 1 837 16	82 134 69 1	m ) 0 0	~ ; N 0	00	00	00	00	190	нc	001	nο	00	00		00	500		A -> Be R -> Ri	Ca - Ca	D -> Ca		G - Ro	H -> Ru			

Figure A8-4: Confusion Matrix for C4.5 Classifier using the *invariant reduced* feature evaluated in a subject independent manner when features are computed per axis over windows of 5.6s. The activities to recognize are the activities contained in the MIT dataset without discriminating among intensity levels of an activity (31 activities in total) and without including the unknown class.

### **Postures and Ambulatory Motions with MET Intensity Levels**

Activity	Sub	ject Depen	dent	Sub	ject Indepen	dent
	TP Rate	FP Rate	<b>F-Measure</b>	TP Rate	FP Rate	<b>F-Measure</b>
Lying down	$99.9\pm0.3$	$0.0\pm0.0$	$99.7\pm0.3$	$99.3 \pm 1.3$	$0.1\pm0.4$	$99.1 \pm 1.9$
Moderate	$96.4 \pm 1.7$	$2.4 \pm 1.2$	$96.4 \pm 1.4$	$86.8\pm5.4$	$13.0\pm5.4$	$83.8\pm3.6$
Running - Treadmill 4mph - Treadmill 0	$97.7 \pm 2.3$	$0.1 \pm 0.1$	$97.0\pm2.6$	$53.0\pm36.1$	$1.9 \pm 2.3$	$47.9\pm30.6$
Running - Treadmill 5mph - Treadmill 0	$94.7 \pm 3.3$	$0.1 \pm 0.1$	$94.9\pm2.8$	$52.8 \pm 34.1$	$1.4 \pm 1.1$	$47.5\pm26.2$
Running - Treadmill 6mph - Treadmill 0	$91.0\pm13.2$	$0.1 \pm 0.1$	$91.9 \pm 10.2$	$51.3\pm40.0$	$0.8 \pm 0.9$	$45.0\pm33.4$
Sitting	$96.1 \pm 3.9$	$0.2 \pm 0.2$	$96.2\pm3.2$	$73.7 \pm 19.3$	$1.6 \pm 1.8$	$73.3\pm16.8$
Standing	$93.3 \pm 7.0$	$0.1 \pm 0.1$	$94.5\pm6.0$	$94.2 \pm 12.1$	$0.2 \pm 0.7$	$92.8\pm13.6$
Vigorous	$93.3\pm3.5$	$1.3 \pm 0.6$	$93.8\pm2.9$	$66.0\pm15.4$	$5.3 \pm 2.9$	$69.2\pm10.7$
Walking - Treadmill 2mph - Treadmill 0	$97.8 \pm 2.4$	$0.2 \pm 0.1$	$96.4\pm2.8$	$75.8 \pm 34.2$	$0.3 \pm 0.4$	$77.7\pm29.5$
Walking - Treadmill 3mph	$99.0 \pm 1.2$	$0.2 \pm 0.1$	$98.9\pm0.8$	89.0 ± 23.3	$0.7 \pm 1.0$	$89.6\pm20.5$
kneeling	$96.6 \pm 3.9$	$0.0 \pm 0.1$	$97.3 \pm 3.2$	$97.9 \pm 4.4$	$0.1 \pm 0.2$	$97.4 \pm 5.1$

Table A8-13: Subject dependent and independent performance of the C4.5 classifier in recognizing postures and ambulation including METs intensity levels (11 activities in total) and without including the *unknown* class. The feature set utilized is the *invariant reduced* feature set computed per axis over windows of 5.6s in length using all the accelerometers (7) available.



Figure A8-5: Confusion Matrix for C4.5 Classifier using the *invariant reduced* feature evaluated in a subject dependent manner when features are computed per axis over windows of 5.6s. The activities to recognize are postures and ambulation including METs intensity levels (11 activities in total) and without including the unknown class.

	EXAMPLE CLASSIFIED AS													
												I		
A	в	С	D	E	F	G	н	I	J	K				
2118	4	0	0	0	8	0	2	0	0	0	A	G		
6	7697	0	0	0	319	7	777	17	31	8	в	I		
0	21	308	167	32	0	0	14	0	71	0	С	Ν		
0	1	132	297	127	0	0	4	0	11	0	D	А		
0	6	59	118	217	0	0	4	0	1	0	Е			
17	235	0	0	0	940	41	33	0	0	7	F	L		
0	12	0	0	0	9	405	4	0	0	0	G			
0	1330	38	1	0	21	0	2876	16	8	0	н	L		
0	104	0	0	0	0	0	55	543	20	0	I	Α		
0	63	155	1	0	0	0	75	32	2536	0	J	ъ		
1	6	0	0	0	1	1	0	0	0	410	к	Б		
												Е		
												L		
7 -> T	ting dow	n					-> 0+->	ding				_		
B -> M	B -> Moderate H -> Vigurous													
C -> R	unning -	Treadm	ill 4mpl	n - Trea	admill (	) I	-> Wall	king -	Treadmil	1 2mph ·	-			
D -> R	unning	Treadm	ill_5mpl	nTrea	admill_(	)	Treadmi	11_0						
E -> R	E -> RunningTreadmill_6mphTreadmill_0 J -> WalkingTreadmill_3mph													
Ŀ' −> S:	ıtting					K	> knee	e⊥ing						

Figure A8-6: Confusion Matrix for C4.5 Classifier using the *invariant reduced* feature evaluated in a subject dependent manner when features are computed per axis over windows of 5.6s. The activities to recognize are postures and ambulation including METs intensity levels (11 activities in total) and without including the unknown class.

### **Postures and Ambulatory Motions**

Activity	Su	bject Depen	dent	Subject Independent					
	TP Rate	FP Rate	<b>F-Measure</b>	TP Rate	FP Rate	F-Measure			
Lying down	$99.9\pm0.3$	$0.1 \pm 0.1$	$99.8\pm0.3$	$100.0\pm0.0$	$0.2\pm0.9$	$99.6 \pm 1.9$			
Running	$99.2 \pm 1.4$	$0.2\pm0.2$	$98.8 \pm 1.2$	$85.6\pm23.7$	$1.6 \pm 2.8$	$85.6\pm22.5$			
Sitting	$99.0 \pm 1.3$	$0.1 \pm 0.1$	$99.2\pm0.8$	$90.9\pm21.7$	$0.3 \pm 0.5$	$92.3 \pm 19.3$			
Stairs - Ascend stairs	$93.1 \pm 5.1$	$0.3 \pm 0.2$	$93.4 \pm 3.9$	$94.4\pm6.4$	$0.7\pm0.9$	$91.5\pm6.6$			
Stairs - Descend stairs	$90.6 \pm 8.4$	$0.4 \pm 0.3$	$91.4 \pm 6.8$	$68.0\pm25.6$	$1.3 \pm 1.5$	$68.5\pm24.4$			
Standing	$97.5\pm3.9$	$0.2\pm0.2$	$96.5 \pm 3.7$	$96.3\pm9.6$	$0.7 \pm 2.3$	$92.7 \pm 15.5$			
Walking	$98.9\pm0.9$	$0.7\pm0.7$	$99.0\pm0.8$	$93.2\pm6.5$	$4.9 \pm 3.5$	$93.3\pm4.4$			
kneeling	$99.0 \pm 2.0$	$0.0 \pm 0.1$	99.1 ± 1.6	$98.4 \pm 4.3$	$0.0 \pm 0.1$	$98.6\pm2.5$			

Table A8-14: Performance of the C4.5 classifier when recognizing postures and ambulation using the *invariant reduced* feature set computed per axis over windows of 5.6s in length during subject dependent and independent training. All sensors (7) were utilized and the *unknown* class was not included in this experiment.

EXAMPLE CLASSIFIED AS										
									F	
									1	
A	в	с	D	Е	F	G	н		C	
2129	0	1	0	0	2	0	0	А	1	
0	1578	0	0	1	0	11	0	в	N	
1	0	1261	0	5	5	0	1	С	Δ	
1	0	1	552	19	7	12	0	D		
0	4	1	15	493	3	26	2	Е	1	
0	0	5	2	2	419	2	0	F		
2	19	1	17	13	3	5019	1	G	I	
2	0	0	2	0	0	0	415	н	Α	
									I	
									1	
A -> Lying_down E -> StairsDescend_stairs										
B -> Running F -> Standing										
D -> Stairs - Ascend stairs H -> kneeling										

Figure A8-7: Confusion Matrix for C4.5 Classifier using the *invariant reduced* feature evaluated in a subject dependent manner when features are computed per axis over windows of 5.6s. The activities to recognize are postures and ambulation (9 activities in total) and without including the unknown class.

EXAMPLE CLASSIFIED AS									
									R
									I
А	в	с	D	Е	F	G	н		G
2132	0	0	0	0	0	0	0	A	Ι
0	1425	0	1	1	0	163	0	в	Ν
19	0	1160	1	0	83	7	3	С	А
0	2	0	545	17	0	28	0	D	т
0	0	0	24	378	0	142	0	Е	Ľ
0	0	15	0	0	414	1	0	F	
0	168	4	47	129	0	4725	2	G	L
0	0	6	0	0	0	1	412	н	Α
									в
									Е
									L
A -> Lying_down B -> Running C -> Sitting D -> StairsAscend_stairs H -> kneeling									rs

Figure A8-8: Confusion Matrix for C4.5 Classifier using the *invariant reduced* feature evaluated in a subject independent manner when features are computed per axis over windows of 5.6s. The activities to recognize are postures and ambulation (9 activities in total) and without including the unknown class.

#### Postures

Activity		Subject Depend	ent	Subject Independent			
	TP Rate	FP Rate	F-Measure	TP Rate	FP Rate	F-Measure	
Lying down	$99.96 \pm 0.20$	$0.48\pm0.76$	$99.83 \pm 0.23$	$100.00\pm0.00$	$1.35\pm6.03$	$99.59 \pm 1.82$	
Sitting	$98.56 \pm 2.74$	$0.30\pm0.45$	$98.23 \pm 2.50$	95.93 ± 17.73	$0.94\pm3.00$	$95.01 \pm 16.96$	
Standing	$97.65 \pm 3.22$	$0.17\pm0.30$	$98.22 \pm 2.16$	$98.10 \pm 8.52$	$0.80\pm2.94$	$96.83 \pm 8.58$	
kneeling	$98.55 \pm 2.75$	$0.10\pm0.24$	$98.91 \pm 2.00$	$100.00\pm0.00$	$0.00\pm0.00$	$100.00\pm0.00$	

Table A8-15: Performance per class while recognizing postures using the C4.5 classifier using the invariant reduced feature set computed per axis over windows of 5.6s in length. The unknown class is not considered.
	ЕХАМ	PLE CI	LASSI	FIED	A S	0	,
						P	
						ĸ	
						1	
	A	в	С	D		G	
	2131	0	1	0	Δ	I	
	1	425	1	1	л Ъ	Ν	
	1	423	4	1	Б	А	
	3	5	420	2	С		
	2	4	0	413	D	L	
						L	
						А	
						в	
						5	
						E	
						L	
 7 _ `	Tuing	down	1	C ->	gtanding		_
л -/ р \	· LyIng_	~			braching		
	> SILLIN	9		D ->	KNEETING		

Figure A8-9: Confusion Matrix for C4.5 Classifier using the *invariant reduced* feature evaluated in a subject dependent manner when features are computed per axis over windows of 5.6s. The activities to recognize are postures (4 activities in total) and without including the *unknown* class.

r							
		EXAN	MPLE	CLASS	IFIED	) A S	0
							R
							I
							G
		A	в	С	D		I
		2132	0	0	0	А	Ν
		17	395	19	0	в	Α
		0	8	422	0	С	L
		0	19	5	395	D	
							B
							E
							L
	Δ -	> Luina	down		C ->	Standir	n a
	л <i>-</i> ,	< цутид > сітті				brandlir	19
	р-,	/ SILLI	ug		ע- ע		iy

Figure A8-10: Confusion Matrix for C4.5 Classifier using the *invariant reduced* feature evaluated in a subject independent manner when features are computed per axis over windows of 5.6s. The activities to recognize are postures (4 activities in total) and without including the *unknown* class.

## **Appendix A9: Real-Time Interactive Training Study**

# **Recognizing 10 activities of your choice in real-time using three wireless accelerometers**

During this short study, you are provided with a software application that will allow you to train a computer program to recognize 10 physical activities, exercises, postures, or activities done in a particular poster of your choice. The computer program will recognize the activities you provide by sensing the motion at your hip, dominant wrist, and dominant foot by means of three small wireless sensors that you will be provided with during this study.

The procedure you need to follow to train the computer program to recognize your activities is simple:

- 1. Wear the tiny sensor at the following locations: Hip, dominant wrist, and dominant foot.
- 2. Type in 10 physical activities, exercises, postures, or activities done in a particular poster you want the computer program to recognize that can be executed continuously for 2 minutes.
- 3. Provide examples of the activities you specified by performing them for two minutes each as indicated by the application. The application will show on the screen the activities you need to perform one at a time, and a counter that indicates your progress. The counter will reach zero once you have finished the training process for a particular activity.

Once you finish the training process, you will hear a message informing you of the successful termination of this phase. The computer program will then start recognizing the activities you provided on the fly. Please feel free to evaluate the performance of the algorithm by executing the activities as many times as you wish. Feel free to experiment and suggest ideas on how the training or recognition of activities can be improved in future versions.

Please be aware that the computer program cannot collect activity examples if the motion signals coming from the sensors are lost due to body blocking or environmental noise. As a result, you might feel at times that the counter on the screen does not decrease as fast as expected. This because signals are not being received, so please be patient.

Subject One		
Correctly Classified Instances Incorrectly Classified Instances Kappa statistic Mean absolute error Root mean squared error Relative absolute error Root relative squared error Maximum absolute error deviation Total Number of Instances	518 89.6194 % 60 10.3806 % 0.8847 0.0263 0.1394 14.5864 % 46.4629 % 0 578	
TP Rate         FP Rate         Precision         Recall           0.893         0.017         0.847         0.893           0.948         0.008         0.932         0.944           0.914         0.006         0.946         0.917           0.776         0.025         0.776         0.777           0.845         0.019         0.831         0.844           0.948         0.01         0.917         0.944           0.948         0.01         0.917         0.944           0.948         0.01         0.917         0.944           0.948         0.01         0.917         0.949           0.897         0.006         0.929         0.899           0.931         0.01         0.915         0.933           0.914         0.008         0.93         0.91	I         F-Measure         ROC Area         Class           13         0.87         0.946         bouncing_on_a_ball           18         0.94         0.976         waving_my_hand_to_say_hello           14         0.93         0.969         shaking_my_leg           15         0.838         0.935         side_stretch           18         0.932         0.976         jumping_jacks           17         0.912         0.955         punching_as_I_walk_forward           17         0.923         0.964         riding_abike           14         0.922         0.966         playing_the_drums	
=== Confusion Matrix ===		
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	<pre>classified as * bouncing_on_a_ball * waving_my_hand_to_say_hello * shaking_my_leg * Taekwondo_Form_#1 * side_stretch * jumping_jacks * jumping_as_I_walk_forward * lifting_dumbells * riding_a bike * playing_the_drums</pre>	
Subject Two		
Correctly Classified Instances Incorrectly Classified Instances Kappa statistic Mean absolute error Root mean squared error Relative absolute error Root relative squared error Maximum absolute error deviation Total Number of Instances	477 91.7308 % 43 8.2692 % 0.907 0.0222 0.1285 11.2626 % 40.8777 % 0 520	
=== Detailed Accuracy By Class ===		
TP Rate         FP Rate         Precision         Recall           0.982         0.006         0.948         0.982           0.948         0.009         0.932         0.944           0.931         0.017         0.871         0.933           0.897         0.009         0.929         0.897           0.914         0.019         0.855         0.91           0.931         0.011         0.915         0.93           0.845         0.013         0.891         0.84           0.914         0.004         0.964         0.91	I         F-Measure         ROC Area         Class           12         0.965         0.998         walking           18         0.94         0.987         sitting_still           10         0.9         0.975         scratching_hfead           17         0.912         0.961         carrying_box           17         0.929         0.954         washing_dishes           14         0.883         0.951         shaking_hands           11         0.923         0.966         tossing_ball_in_air           15         0.867         0.928         typing           14         0.938         0.985         talkin_gon_phone	
=== Confusion Matrix ===		
a       b       c       d       e       f       g       h       i       < class	ussified as using tting_still tratching_hfead using_dishes taking_hands sssing_ball_in_air ping ukin_gon_phone	

### Subject 3

Correctly Classified Instances Incorrectly Classified Instance Kappa statistic Mean absolute error Root mean squared error Relative absolute error Maximum absolute error deviatio Total Number of Instances	456 122 0.7655 0.046 0.1953 25.5448 % 65.111 % 0 578	78.8927 21.1073	ş Ş	
=== Detailed Accuracy By Class				
TP Rate         FP Rate         Precision           0.768         0.013         0.86           0.776         0.038         0.692           0.931         0.013         0.885           0.828         0.025         0.787           0.81         0.025         0.783           0.845         0.013         0.885           0.845         0.013         0.875           0.707         0.033         0.683           0.638         0.033         0.685           0.776         0.015         0.849	Recall         F-Measure           0.768         0.811           0.776         0.732           0.931         0.908           0.828         0.807           0.81         0.797           0.81         0.797           0.845         0.86           0.707         0.695           0.638         0.661           0.776         0.811	ROC Area Cl 0.89 t 0.887 b 0.962 b 0.915 t 0.934 s 0.906 s 0.927 w 0.873 T 0.818 s 0.901 b	ass hrowing owling ouncing yping tepping treching_arm alking ennis_serve treching_legs ending	
=== Confusion Matrix ===				
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	<pre>&lt; classified as   a = throwing   b = bowling   c = bouncing   d = typing   f = strephing   f = strephing_arm   g = walking   h = Tennis_serve   i = strephing_leg   j = bending</pre>	s		
Subject 4				
Correctly Classified Instances Incorrectly Classified Instance Kappa statistic Mean absolute error Root mean squared error Relative absolute error Root relative squared error Maximum absolute error deviatio Total Number of Instances	516 62 0.8808 0.0242 0.1358 13.4529 % 45.2635 % n 0 578	89.2734 10.7266	49 49	
=== Detailed Accuracy By Class				
TP Rate         FP Rate         Precision           0.929         0.008         0.929           0.931         0.01         0.915           0.741         0.021         0.796           0.948         0.013         0.887           0.931         0.01         0.915           0.948         0.013         0.887           0.931         0.01         0.915           0.914         0.012         0.898           0.897         0.013         0.881           0.897         0.01         0.912           0.862         0.017         0.847           0.879         0.006         0.944	Recall         F-Measure           0.929         0.929           0.931         0.923           0.741         0.768           0.948         0.917           0.931         0.923           0.917         0.931           0.918         0.917           0.931         0.923           0.914         0.906           0.897         0.889           0.897         0.904           0.862         0.855           0.879         0.911	ROC Area         Cl           0.981         w           0.987         w           0.987         w           0.973         d           0.977         w           0.962         0.961           0.953         h           0.961         e	ass alk ype_in_computer ashing_window rawing_in_paper iping_surface alking_on_the_phone sweeping ombing_my_hair ammering_a_nail ating	
=== Confusion Matrix ===				
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	<pre>&lt; classified as   a = walk   b = type_in_compu   c = washing_windo   d = drawing_in_pa   e = wiping_surfac.</pre>	ter w per e		

### Subject 5

Correctly Classified Instances Incorrectly Classified Instances Kappa statistic Mean absolute error Root mean squared error Relative absolute error Root relative squared error Maximum absolute error deviation Total Number of Instances			n	493 85 0.8366 0.0348 0.1625 19.3456 % 54.1615 % 0 578	85.2 14.7	941 % 059 %
=== Detai	led Accura	cy By Class				
TP Rate 0.946 0.879 0.776 0.862 0.793 0.776 0.879 0.862 0.81 0.948	FP Rate 0.011 0.027 0.027 0.023 0.023 0.015 0.01 0.012 0.006	Precision 0.898 0.785 0.763 0.806 0.793 0.849 0.911 0.909 0.887 0.948	Recall 0.946 0.879 0.776 0.862 0.793 0.776 0.879 0.862 0.81 0.948	F-Measure 0.922 0.829 0.769 0.833 0.793 0.811 0.895 0.885 0.885 0.847 0.948	ROC Area 0.98 0.952 0.875 0.936 0.913 0.898 0.938 0.938 0.919 0.98	Class walk bicep_curls stretching applying_cream brushing_theet wash_dish knitting wash_hands filing_nails play_piano
=== Confu	sion Matri:	x ===				
a b c 53 0 2 3 51 2 0 7 45 0 1 2 1 2 3 1 1 1 1 0 1 0 1 1 0 2 2 0 0 0 0	d     e     f       2     0     0       2     1     1       3     0     2       50     1     1       3     46     1       1     7     45       1     1     3       2     0     1       2     0     2       1     0     2	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	<pre>&lt; cl   a = w   b = b   c = s   d = a   e = b   f = w   g = k   h = w   i = f   j = p</pre>	assified as valk vicep_curls stretching upplying_creas rrushing_thee vash_dish mitting mash_hands filing_nails vlay_piano	n E	

Table A9-1. Performance obtained by recognizing participant's activities using a C4.5 classifier with the *invariant reduced* feature set computed per axis over window lengths of 5.6s in length. Performance was measured using 10 Fold Cross-validation.

**Appendix A10: Boston Medical Center Data Collection Protocol**  In this protocol, two participants performed the script of activities shown in the table below. Each activity was performed one at a time for three to four minutes while participants wore the same set of sensors utilized in the MIT protocol. The indirect calorimeter utilized in this study was the Parvo Medics TrueOne 2400 metabolic measurement system [126].

Activity Performed
Lying quietly on back, arms at side
Sitting in chair
Standing hands at sides, feet shoulder-width apart
Walking on TM, 3 mph, 0% grade
Walking on TM, 3 mph, 4% grade
Walking on TM, 3 mph, 8% grade
Jog on treadmill, 5 mph, 0%
(optional activity depending on subject's capacity)
Cycle ergometer (legs), 60 RPM, "light" resistance
Cycle ergometer (legs), 100 RPM, "light" resistance
Cycle ergometer (legs), 60 RPM, "hard" resistance
Carry box with both hands in front of subject (elbows bent at 90
degrees) and walk at 2 mph on treadmill
Jumping Jacks
Perform sit-ups with knees bent and hands behind head-15 in
one minute or as many as comfortable
Perform push-ups from the knees—15 in one minute or as many
as comfortable
Perform biceps curl with both arms (3 or 5 pound weight)—15
curls in one minute or as many as comfortable

Table A10-1: Boston Medical Center data collection protocol. The activities listed in the table were performed in the order indicated starting from the one at the top of the table and ending with the one at the bottom of the table.

**Appendix A11: Stanford and Boston University Data Collection Protocols** 

## Script for Laboratory/Gymnasium Activities

### Session #1

- > Each activity was performed for a period of two minutes unless otherwise specifically noted.
- The subject's exertion perception on the RPE scale was annotated for each activity involving different activity levels.

Activity	Annotation Guide
Lying quietly on back, arms at side	Lying Down
	n/a
	n/a
Sitting in chair very quietly—hands on knees, feet flat on	Sitting
floor	n/a
	n/a
Standing very quietly, hands at sides, feet shoulder-width	Standing
apart	n/a
•	n/a
Getting ready and annotation	
Walking on Treadmill (TM), 2 miles per hour (mph), 0%	Walking
grade	Treadmill 2 mph
	Treadmill 0%
Walking on TM, 3 mph, 0% grade	Walking
	Treadmill 3 mph
	Treadmill 0%
Walking on TM, 4 mph, 0% grade	Walking
	Treadmill 4 mph
	Treadmill 0%
Getting ready and annotation	
Walking on TM, 3 mph, 4% grade	Walking
	Treadmill 3 mph
	Treadmill 3%
Walking on TM, 3 mph, 8% grade	Walking
	Treadmill 3 mph
	Treadmill 6%
Walking on TM, 3 mph, 12% grade	Walking
	Treadmill 3 mph
	Treadmill 9%
Getting ready and annotation	
Jog on treadmill, 5 mph, 0%	Running
(optional activity depending on subject's capacity)	Treadmill 5 mph
	Treadmill 0%

### Bout A

Break—end of *Bout A* 

- Subjects rested for approximately five minutes and prepared for the cycling bout (*Bout B*)
- Before the beginning of *Bout B*, the light, moderate and hard resistance levels were determined for each subject using 80 rpm on the cycle ergometer (legs).
- Note: a rowing machine may be used interchangeably with the arm cycle ergometer and will be indicated by the annotation

Bout	В
------	---

Activity	Annotation Guide
Cycle ergometer (legs), 60 RPM, "light" resistance	Cycling
	Cycle light
	Cycle 60 rpm
Cycle ergometer (legs), 80 RPM, "light" resistance	Cycling
	Cycle moderate
	Cycle 80 rpm
Cycle ergometer (legs), 100 RPM, "light" resistance	Cycling
	Cycle light
	Cycle 100 rpm
Sit in chair and fidget with feet and legs	Sitting
	Fidget feet legs
	n/a
Cycle ergometer (legs), 80 RPM, "light" resistance	Cycling
	Cycle light
	Cycle 80 rpm
Cycle ergometer (legs), 80 RPM, "moderate" resistance	Cycling
	Cycle moderate
	Cycle 80 rpm
Cycle ergometer (legs), 80 RPM, "hard" resistance	Cycling
	Cycle hard
	Cycle 80 rpm
Sit in chair and fidget with hands and arms	Sitting
	Fidget hands arms
	n/a
Cycle ergometer (arms), 60 RPM, 25 watts	Rowing
OR	Rowing light
Rowing machine, 30 strokes per minute (SPM), "light"	Cycle 60 rpm (*Rowing 30 spm)
resistance	
Cycle ergometer (arms), 75 RPM, 25 watts	Rowing
OR	Rowing moderate
Rowing machine, 30 SPM, "moderate" resistance	Cycle 75 rpm (*Rowing 30 spm)
Cycle ergometer (arms), 90 RPM, 25 watts	Rowing
OR	Rowing hard
Rowing machine, 30 SPM, "heavy" resistance	Cycle 90 rpm (*Rowing 30 spm)

\*Use annotations in parentheses () for the rowing machine where appropriate

End of Session #1

### Session #2

- > Research assistant carried the laptop for data recording within 10 feet of the subject.
- For activities involving the use of a box, a box or similar object (i.e.—plate weight) weighing approximately five pounds was utilized for activities in *Bout A*.

### Bout A

Activity	Annotation Guide
Move box from floor to table—subject picks up box from	Move weight
floor, turns 90 degrees, takes 2-3 steps to place box on table,	Move wt low
wait 1 second, then return box to the original position	n/a
(Note: the table should be about waist high on the subject)	
Move box from floor to high shelf—subject picks up box	Move weight
from floor, turns 90 degrees, takes 2-3 steps to place box on	Move wt high
shelf, wait 1 second, then return box to the original position	n/a
(Note: the shelf should be at least should high on the subject)	
Stand at table and move box from one side of person to the	Move weight
other side of the person while standing in one place	Move wt side
(Note: sit box down for 2 seconds between moves)	n/a
Carry box with both hands in front of subject (elbows bent at	Move weight
90 degrees) and walk at 2 mph on treadmill	Carry wt 2mph
	n/a

### Bout B

Activity	Annotation Guide
Stair climbing—go up and down three flights of stairs for a	Stairs
period of two minutes, annotate "ascending" and	Ascend stairs OR descend stairs
"descending" appropriately	n/a
(Please note the number of stairs in the "flight" used)	
Rest	
Stair climbing—go up and down three flights of stairs for a	Stairs
period of two minutes, annotate "ascending" and	Ascend stairs OR descend stairs
"descending" appropriately	n/a
(Please note the number of stairs in the "flight" used)	
Rest and move to area for calisthenics	
Perform sit-ups with knees bent and hands behind head—15	Calisthenics
in one minute or as many as comfortable	Sit ups
	n/a
Rest	
Perform push-ups from the knees—15 in one minute or as	Calisthenics
many as comfortable	Push ups
	n/a
Rest	
Jumping jacks—20 in one minute or as many as comfortable	Calisthenics
	n/a
	Jumping jacks
	n/a
Rest	
Perform biceps curl with both arms (3 or 5 pound weight)—	Calisthenics
15 curls in one minute or as many as comfortable	Bicep curl
-	n/a

End of Session #2

Once data collection was complete, subjects removed the heart rate strap/monitor and accelerometers and performed and end of study interview.

### Script for Home Cleaning Activities

During this data collection, participants performed the activities listed in the order shown in the table below, one at a time. Participants were also allowed to complete activities at their own pace and for however long they wished. The research assistant; however, keep track of the length of the activities and suggest moving on to another activity for those taking longer than 15 minutes. This was only a general rule and was left up to the discretion of the research assistant.

### List of Activities

- stacking groceries—to be simulated by un-stacking several shelves into a bag, then restacking them
- doing dishes/putting away dishes
- folding and stacking laundry
- making beds
- emptying trashcans
- scrubbing toilet and/or bathtub or shower
- dusting
- vacuuming
- washing windows
- sweeping
- scrubbing the floor
- mopping

Activity	Description of Activity
U activity	Unknown activity: performing an activity not otherwise specified below.
Dishes	Cleaning dishes by hand
Bed	Actions involved in making a bed, including removing sheets/blankets from a bed, putting sheets/blankets onto a bed, stuffing pillowcases, and carrying sheets/blankets to/from a
	bed.
Scrub	Making quick and rapid movements with the arm. This involves more intensity than wiping/dusting.
Stack	Includes stacking supplies or linens on a shelf.
Sweep	Using a broom to sweep a floor.
Trashcans	Emptying trashcans or carrying a trashcan or trash bag.
Vacuum	Includes vacuuming and also moving a vacuum cleaner to a different location.
Wipe/dust	Similar to scrub, but with shorter and slower arm movements. Less intense than cleaning
	a toilet.
Мор	Using a mop to clean a floor. Does not include carrying a mop.
Laundry	Folding laundry
Shifting objects	ONLY includes extended periods of moving objects around (greater than 5 seconds)

# **Appendix A12: MIT Data Collection Protocol**

The following data collection protocol was used during each of the two data collection sessions that took take place at (1) the MIT Zesiger Sports and Fitness Center and (2) at the PlaceLab instrumented residential home.

### Data collection protocol

- 1. The participant's demographic, medical, and physical activity information was recorded.
- 2. Seven wireless accelerometers (Onbody MITes) were placed on the participant at the feet (on shoe laces), wrists, hip (non-dominant side), dominant upper arm (near bicep muscle), and dominant thigh.
- 3. The participant was asked to wear the polar chest strap heart rate monitor and to place the heart rate MITes transceiver in his/her pocket.
- 4. The participant was asked to wear the bodybugg armband on his/her dominant upper arm.
- 5. Two Actigraph activity monitors were then firmly attached to the participant at the belt around the dominant side of the hip (on the outer edge of the hip bone) and another at the dominant wrist.
- 6. The K4b2 portable indirect calorimeter was installed on the participant. The participant was instructed not to talk and to breathe mainly through the nose during the experiment to increase the accuracy of the  $VO_2$  and  $VCO_2$  measurements.
- 7. The participant's resting metabolic rate (RMR) and resting heart rate were measured by having the participant lie down for a period of 5 minutes.
- 8. The participant was asked to perform a script of activities depending o the location of the data collection. The two scripts utilized during the MIT Zesiger Sports and Fitness Center and PlaceLab instrumented residential home are presented below.

# Activity Script for the MIT Zesiger Sports and Fitness Center Data Collection

Activity	Description of Activity
Lying down (5min)	Lying quietly on back, arms at side
Sitting*	Sitting in chair very quietly-hands on knees, feet flat on floor
Sitting fidgeting* hands and	Sit in chair and fidget with hands and arms
arms	
Standing still*	Standing very quietly, hands at sides, feet shoulder-width apart
Kneeling*	Supporting the body on your knees, both knees on the floor.
Walking	Walking on a treadmill at the specified speed and inclination
Walking upstairs	Walk upstairs three flights of stairs for a period of two minutes, annotate "ascending" and
	"descending" appropriately
Walking downstairs	Walk downstairs three flights of stairs for a period of two minutes, annotate "ascending" and
	"descending" appropriately. (Please note the number of stairs in the "flight" used)
Running	Running on a treadmill at the specified speed and inclination
Cycle ergometer	Cycling on a leg ergometer at the specified speed.
Rowing	
Sit-ups	Perform sit-ups with knees bent and hands behind head. 15 in one minute or as many as
	comfortable.
Crunches	Perform crunches. 15 in one minute or as many as comfortable
Biceps curls	Perform biceps curl with both arms using the specified weight. 15 curls in one minute or as many
	as comfortable.
Weight lifting	Perform bench weight lifting with both arms using the specified weight. 15 repetitions in one
	minute or as many as comfortable.

 Table A12-1: Description of the activities performed at MIT Zesiger Sports and Fitness center.

 (\*Data collected for 2mins only)

This session was divided into several activity periods of 20-30 minutes, separated by 5 minute rest periods. Even though this data collection lasted for about three hours, each activity was only performed for three to four minutes and there were pauses between activities to allow the participant's to have some rest. The participants were asked to perform the activities in the order listed, one at a time for three to four minutes or for as long as was comfortable, if the activity was physically demanding. The participants were also asked to remove the sensors and put them on at least two times during the data collections. In this way, slight variations in orientation due to sensor installation were captured in each dataset. If the participant found any of the activities too physically demanding the activity was no longer performed. The participant's heart rate was also monitored during the data collection to assure it did not exceed the participant's maximum estimated peak exercise heart rate. A member of the study team quietly annotated the activities and let the participant know when it was time to pause or start the next activity.

Activity	Annotation guide			
	Category	Intensity	Difficulty	
Lying down (5 minutes)	Lying down	n/a	n/a	
Sitting (2min)	Sitting	n/a	n/a	
Sit in chair and fidget with feet and legs (2min)	Sitting	Fidget feet legs	n/a	
Sitting fidgeting hands and arms (2min)	Sitting	Fidget hands arms	n/a	
Standing still (2min)	Standing still	n/a	n/a	
Kneeling (2min)	Kneeling	n/a	n/a	

Bout A

Break-end of Bout A

Allow subject to rest for approximately five minutes and get ready for *Bout B* 

#### Bout B

Walking on TM, 2 mph, 0% grade	Walking	Treadmill 2 mph	Treadmill 0%
Walking on TM, 3 mph, 0% grade	Walking	Treadmill 3 mph	Treadmill 0%
Walking on TM, 3 mph, 3% grade "light"	Walking	Treadmill 3 mph	Treadmill 3%
Walking on TM, 3 mph, 6% grade "moderate"	Walking	Treadmill 3 mph	Treadmill 6%
Walking on TM, 3 mph, 9% grade "hard"	Walking	Treadmill 3 mph	Treadmill 9%
Walking upstairs	Stairs	Ascend stairs	n/a
Walking downstairs	Stairs	Descend stairs	n/a
Running on TM, 4 mph, 0% grade	Running	Treadmill 4 mph	Treadmill 0%
Running on TM, 5 mph, 0% grade	Running	Treadmill 5 mph	Treadmill 0%
(optional depending on subject)			
Running on TM, 6 mph, 0% grade	Running	Treadmill 6 mph	Treadmill 0%
(optional depending on subject)			

#### Break—end of Bout B

- Allow subject to rest for approximately five minutes and get ready for *Bout C*
- Before beginning *Bout C*, determine light, moderate, and hard resistance for subject using 80 rpm on the cycle ergometer (legs). Document resistances used in "Notes" section of electronic data when data is merged/cleaned.
- Note: a rowing machine may be used interchangeably with the arm cycle ergometer and will be indicated by the annotation

|--|

Activity	Annotation guide			
	Category	Intensity	Difficulty	
Cycle ergometer (legs), 60 RPM, "light" resistance	Cycling	Cycle light (2 R)	Cycle 60 rpm	
Cycle ergometer (legs), 80 RPM, "light" resistance	Cycling	Cycle light (2 R)	Cycle 80 rpm	
Cycle ergometer (legs), 100 RPM, "light" resistance	Cycling	Cycle light (2 R)	Cycle 100 rpm	
Cycle ergometer (legs), 80 RPM, <b>"moderate"</b> resistance	Cycling	Cycle moderate (7 R)	Cycle 80 rpm	
Cycle ergometer (legs), 80 RPM, "hard" resistance	Cycling	Cycle hard (13 R)	Cycle 80 rpm	

\*Use annotations in parentheses () for the rowing machine where appropriate

### Break—end of *Bout C*

- Allow subject to rest for approximately five minutes and get ready for *Bout D*
- Before beginning *Bout D*, determine light, moderate, and hard resistance for subject using 80 rpm on the cycle ergometer (arms). Document resistances used in "Notes" section of electronic data when data is merged/cleaned.
- Note: a rowing machine may be used interchangeably with the arm cycle ergometer and will be indicated by the annotation

#### Bout D

Cycle ergometer (arms), 60 RPM, 25 watts	Rowing	Rowing light	Cycle 60 rpm
OR		(2 R)	(*Rowing 30 spm)
Rowing machine, 30 strokes per minute (SPM),			
"light" resistance			
Cycle ergometer (arms), 75 RPM, 25 watts	Rowing	Rowing moderate	Cycle 75 rpm
OR		(5 R)	(*Rowing 30 spm)
Rowing machine, 30 SPM, "moderate" resistance			
Cycle ergometer (arms), 90 RPM, 25 watts	Rowing	Rowing hard	Cycle 90 rpm
OR		(8 R)	(*Rowing 30 spm)
Rowing machine, 30 SPM, "hard" resistance			

### Break—end of *Bout D*

- Allow subject to rest for approximately five minutes and get ready for *Bout D*
- Before beginning *Bout E*, determine light, moderate, and hard weights for subject for "Bicep curls" and "Bench weight lifting". Document weights used in "Notes" section of electronic data when data is merged/cleaned.

Bout E			
Activity	Annotation guide		
	Category	Intensity	Difficulty
Sit-ups	Calisthenics	Sit ups	n/a
Crunches	Calisthenics	Push ups	n/a
Bicep curls "light"	Resistance	Bicep curl light (2	n/a
		Lb)	
Bicep curls "moderate"	Resistance	Bicep curl moderate	n/a
		(5 Lb)	
Bicep curls "hard"	Resistance	Bicep curl hard (8	n/a
		Lb)	
Bench weight lifting "light"	Resistance	Bench light (2 Lb)	n/a
Bench weight lifting "moderate"	Resistance	Bench moderate	n/a
		(7 Lb)	
Bench weight lifting "hard"	Resistance	Bench hard (17 Lb)	n/a

The *walking* and *running* activities were collected on a Precor C956 treadmill, the *cycling* activities on a Precor C846 recumbent stationary bicycle, and the *rowing* activities on a Concept2 PM2 Rowing machine.



Table A12-2: Stationary exercise equipment utilized during the MIT gymnasium data collections.

### Activity Script for the PlaceLab Residential Home Data Collection

Activity	Description of Activity
Lying down	Lying down on a bed with hands on sides
Watching TV	Watching TV while sitting on a couch
Playing	Playing video games while sitting on a couch
videogames	
Typing	Typing on a computer while sitting.
Writing	Handwriting on a piece of paper while sitting.
Making the bed	Actions involved in making a bed, including removing sheets/blankets from a
	bed, putting sheets/blankets onto a bed, stuffing pillowcases, and carrying
	sheets/blankets to/from a bed.
Taking out trash	Emptying trashcans or carrying a trashcan or trash bag.
Walking around	Walking at normal speed around the PlaceLab block
block	
Carrying groceries	Carrying grocery bags either with one or two hands. The weight of the bags
	should be greater than 2kg.
Stacking groceries	Includes stacking supplies or linens on a shelf.
Ironing	Ironing clothes by hand using a hand ironing machine while standing up or
	sitting.
Doing dishes	Washing dishes by hand.
Wiping/dusting	Similar to scrub, but with shorter and slower arm movements. Less intense than
	cleaning a toilet.
Sweeping	Using a broom to sweep a floor. Does not include carrying a broom.
Mopping	Using a mop to clean a floor. Does not include carrying a mop.
Vacuuming	Includes vacuuming floor and also moving a vacuum cleaner to a different
	location.
Washing windows	Washing a window by scrubbing while standing up.
Scrubbing a	Making quick and rapid movements with the arm. This involves more intensity
surface	than wiping/dusting such as cleaning the bath tub.
Weeding	Weeding grass at PlaceLab patio
Gardening	Gardening at PlaceLab patio

 Table A12-3: Description of the activities performed at the PlaceLab.

During this data collection, the participant was asked to perform the activities in the order listed, one at a time for three to four minutes. A member of the study team quietly annotated the activities using a laptop computer. The participant was also asked to remove the sensors and put them on at least two times during the data collection. In this way, slight variations in orientation due to sensor installation will be captured in each dataset.

## **Appendix A13: Activity Recognition Using Different Classifiers**

Activity		True Pos	itive Rate	
	NN	NB	LogitBoost	C4.5
Bench weight lifting - hard	$7.49 \pm 13.72$	$6.35 \pm 15.14$	$5.15 \pm 10.60$	$16.66 \pm 29.15$
Bench weight lifting - light	$20.07 \pm 20.51$	$27.56 \pm 29.41$	$37.68 \pm 37.08$	$15.56 \pm 28.17$
Bench weight lifting - moderate	8.71 ± 11.20	$14.25 \pm 22.02$	$12.61 \pm 21.93$	$11.64 \pm 18.54$
Bicep curls - hard	$24.50 \pm 24.14$	$31.10 \pm 40.89$	$14.89 \pm 25.14$	$19.88 \pm 37.06$
Bicep curls - light	24.49 + 16.65	10.22 + 14.82	49.48 + 33.58	18.75 + 27.29
Bicen curls - moderate	19.43 + 21.12	8 69 + 12.85	28.11 + 31.30	12.19 + 26.74
Calisthenics - Crunches	42.41 + 44.87	19.20 + 34.53	48.65 + 42.82	15.96 + 31.90
Calisthenics - Sit ups	69.76 + 37.44	$63.16 \pm 46.40$	68.77 + 39.12	44.79 + 42.87
Cycling - Cycle hard - Cycle 80rnm	$11.62 \pm 12.17$	$38.14 \pm 37.45$	1674 + 27.66	$17.97 \pm 29.91$
Cycling - Cycle light - Cycle 100rpm	$65.05 \pm 39.10$	78 87 + 34 77	$79.70 \pm 35.49$	$70.55 \pm 37.69$
Cycling - Cycle light - Cycle 60rpm	$62.05 \pm 39.10$ $62.35 \pm 29.28$	$74.81 \pm 35.76$	$74.04 \pm 29.94$	$39.57 \pm 36.19$
Cycling - Cycle light - Cycle 80rpm	$33.84 \pm 25.04$	$35.22 \pm 35.70$	$19.07 \pm 27.09$	$\frac{39.97 \pm 30.19}{42.97 \pm 39.93}$
Cycling Cycle moderate Cycle S0rpm	$33.04 \pm 23.04$	$672 \pm 9.04$	$\frac{49.42 \pm 37.08}{28.94 \pm 25.49}$	$42.77 \pm 39.93$
Lying down	$27.40 \pm 17.83$	$0.72 \pm 9.04$	$20.94 \pm 20.49$	$10.14 \pm 19.43$ 76.00 ± 24.28
Powing Powing hard Powing 20cpm	$03.30 \pm 14.32$ 27.40 ± 24.62	$95.03 \pm 8.71$	$37.40 \pm 24.04$ $32.40 \pm 40.70$	$10.90 \pm 34.28$
Rowing - Rowing light - Rowing Josphi	$27.40 \pm 24.02$	$32.11 \pm 40.42$	$32.49 \pm 40.79$	$19.49 \pm 23.34$
Rowing - Rowing light - Rowing Sospin	$29.70 \pm 23.39$	$30.82 \pm 33.18$	$30.90 \pm 34.98$	$24.27 \pm 23.73$
Rowing - Rowing moderate - Rowing Sospm	$20.00 \pm 14.08$	$10.32 \pm 29.88$	$21.96 \pm 33.29$	$18.72 \pm 21.51$
Running - Treadmill 4mpn - Treadmill 0	$33.10 \pm 29.07$	$39.80 \pm 46.32$	$53.67 \pm 40.94$	$28.20 \pm 30.71$
Running - Treadmill Smpn - Treadmill 0	$00.31 \pm 33.33$	$63.54 \pm 41.47$	$61.53 \pm 37.93$	$48.97 \pm 34.91$
Running - Treadmill 6mph - Treadmill 0	$46.11 \pm 42.27$	$62.90 \pm 47.34$	$57.85 \pm 37.82$	$38.31 \pm 30.97$
Sitting	$4.68 \pm 6.65$	$20.97 \pm 17.24$	$26.27 \pm 35.41$	$15.29 \pm 26.44$
Sitting - Fidget feet legs	$33.05 \pm 33.93$	$55.90 \pm 41.30$	$42.83 \pm 39.44$	$28.83 \pm 36.85$
Sitting - Fidget hands arms	$26.00 \pm 25.91$	$48.64 \pm 43.71$	$42.17 \pm 35.81$	$28.51 \pm 32.18$
Stairs - Ascend stairs	$48.07 \pm 38.29$	$63.44 \pm 38.83$	$53.41 \pm 37.71$	$49.43 \pm 33.57$
Stairs - Descend stairs	$53.34 \pm 35.25$	$55.70 \pm 34.50$	$48.98 \pm 35.71$	$39.71 \pm 30.67$
Standing	$22.05 \pm 14.14$	$4.84 \pm 9.25$	$48.37 \pm 35.76$	$41.26 \pm 38.65$
Walking - Treadmill 2mph - Treadmill 0	$49.94 \pm 35.46$	$71.81 \pm 37.44$	$64.46 \pm 33.33$	$45.61 \pm 34.16$
Walking - Treadmill 3mph - Treadmill 0	$13.71 \pm 13.78$	$43.30 \pm 38.47$	$29.00 \pm 23.39$	$24.28 \pm 25.45$
Walking - Treadmill 3mph - Treadmill 3 - light	$8.14 \pm 8.56$	$18.61 \pm 25.83$	$13.24 \pm 22.25$	$10.06 \pm 12.60$
Walking - Treadmill 3mph - Treadmill 6 - moderate	$16.89 \pm 19.18$	$31.50 \pm 34.97$	$12.05 \pm 12.89$	$10.52 \pm 14.89$
Walking - Treadmill 3mph - Treadmill 9 - hard	$27.46 \pm 26.66$	$21.07 \pm 30.95$	$26.17 \pm 31.53$	$15.11 \pm 23.20$
kneeling	$14.01 \pm 12.05$	$13.68 \pm 9.79$	$70.32 \pm 36.57$	$66.94 \pm 43.12$
unknown	$52.78 \pm 5.16$	$5.89 \pm 2.42$	$76.89 \pm 4.84$	$63.97 \pm 5.18$
Carrying groceries	$27.08 \pm 24.99$	$37.92 \pm 29.05$	$27.48\pm30.35$	$19.75 \pm 20.28$
Doing dishes	$31.75 \pm 19.94$	$39.19 \pm 29.22$	$45.58 \pm 25.90$	$29.34\pm28.43$
Gardening	$16.33 \pm 18.81$	$28.14 \pm 32.12$	$25.89\pm30.41$	$14.44\pm20.82$
Ironing	$44.71 \pm 25.93$	$57.39 \pm 35.17$	$37.69 \pm 32.58$	$37.07 \pm 31.48$
Making the bed	$40.23 \pm 29.15$	$28.24 \pm 21.62$	$31.61 \pm 22.27$	$25.95 \pm 17.78$
Mopping	$28.25 \pm 25.94$	$22.88 \pm 24.33$	$25.02 \pm 24.17$	$24.21 \pm 21.46$
Playing videogames	$12.01 \pm 10.77$	$11.14 \pm 19.14$	$28.20 \pm 30.41$	$29.18 \pm 34.66$
Scrubbing a surface	$14.72 \pm 16.70$	$18.16 \pm 21.13$	$20.10 \pm 27.82$	$13.91 \pm 17.77$
Stacking groceries	$18.28 \pm 21.20$	$23.04 \pm 25.00$	$13.93 \pm 13.66$	$11.49 \pm 12.03$
Sweeping	$20.03 \pm 20.50$	$30.99 \pm 30.60$	$18.83 \pm 16.22$	$16.52 \pm 17.90$
Typing	$39.10 \pm 20.13$	$34.84 \pm 31.44$	$53.33 \pm 35.33$	$49.25 \pm 37.47$
Vacuuming	28.43 + 25.22	40.81 + 37.09	30.54 + 25.49	23.05 + 21.70
Walking around block	22.61 + 20.55	26.91 + 26.45	19.04 + 22.89	18.94 + 17.99
Washing windows	2545 + 21.92	17.07 + 17.67	27.49 + 26.54	22.42 + 19.23
Watching TV	7.25 + 5.85	5.32 + 4.81	21.67 + 27.42	20.12 + 30.06
Weeding	5.72 + 9.12	$12.18 \pm 18.22$	9.55 + 24.66	4.51 + 7.78
Wining/Dusting	31.64 + 23.52	20.89 + 17.93	30.43 + 26.30	$21.58 \pm 20.09$
Writing	$25.75 \pm 14.98$	56.85 + 36.69	45.84 + 34.44	51 11 + 39 97
taking out trash	15.81 + 16.03	13.51 + 15.59	2.69 + 3.67	10.09 + 10.42

Table A13-1: True positive rate when training classifiers using the *MaxAccelerationSet1* feature set computed per sensor evaluated in a subject independent manner.

Activity		False Pos	sitive Rate	
	NN	NB	LogitBoost	C4.5
Bench weight lifting - hard	$0.38 \pm 0.26$	$0.23 \pm 0.20$	$0.26 \pm 0.19$	0.64 + 0.64
Bench weight lifting - light	$0.85 \pm 0.76$	$0.92 \pm 1.01$	$0.66 \pm 0.67$	$0.56 \pm 0.68$
Bench weight lifting - moderate	$0.51 \pm 0.39$	$1.25 \pm 0.76$	$0.49 \pm 0.91$	$0.57 \pm 0.58$
Bicen curls - hard	$0.77 \pm 0.55$	1.37 + 1.49	$0.84 \pm 1.00$	$0.83 \pm 1.07$
Bicen curls - light	$0.57 \pm 0.51$	$0.38 \pm 0.50$	$2.00 \pm 1.26$	$0.90 \pm 1.20$
Bicep curls - moderate	$0.75 \pm 0.57$	$0.56 \pm 0.55$	$0.83 \pm 0.76$	$0.50 \pm 0.20$ $0.51 \pm 0.63$
Calisthenics - Crunches	$0.03 \pm 0.03$	$1.16 \pm 0.72$	$0.03 \pm 0.70$ $0.12 \pm 0.22$	$0.31 \pm 0.03$ $0.22 \pm 0.38$
Calisthenics - Sit ups	$0.03 \pm 0.05$	$0.20 \pm 0.39$	$0.12 \pm 0.22$ $0.05 \pm 0.08$	$0.22 \pm 0.33$ $0.12 \pm 0.13$
Cycling - Cycle hard - Cycle 80rpm	$0.02 \pm 0.00$ 0.50 + 0.40	$1.91 \pm 1.28$	$0.03 \pm 0.00$ $0.92 \pm 1.08$	$0.12 \pm 0.13$ 0.98 + 1.14
Cycling - Cycle light - Cycle 100rpm	$0.76 \pm 1.41$	$0.56 \pm 1.23$	$0.14 \pm 0.21$	$0.12 \pm 0.18$
Cycling - Cycle light - Cycle 60rpm	$0.88 \pm 0.64$	$0.50 \pm 1.22$ 0.51 + 0.51	$0.24 \pm 0.19$	$0.12 \pm 0.18$ 0.17 + 0.18
Cycling - Cycle light - Cycle 80rpm	$1.24 \pm 1.09$	$0.51 \pm 0.51$	$1.10 \pm 1.18$	$1.08 \pm 1.31$
Cycling - Cycle moderate - Cycle 80rpm	$1.24 \pm 0.75$	$0.01 \pm 0.02$ 0.37 ± 0.40	$0.80 \pm 0.83$	$0.76 \pm 0.87$
Lying down	$1.20 \pm 0.79$ $1.44 \pm 1.00$	$4.84 \pm 1.91$	$0.00 \pm 0.03$	$0.70 \pm 0.07$
Rowing - Rowing hard - Rowing 30spm	$1.44 \pm 1.00$ $1.09 \pm 0.77$	$1.04 \pm 1.01$	$1.08 \pm 1.16$	$0.00 \pm 1.10$ 0.67 ± 0.81
Rowing - Rowing light - Rowing 30spm	$1.09 \pm 0.77$ 0.79 ± 0.69	$0.91 \pm 1.99$	$1.03 \pm 1.10$ $1.07 \pm 1.10$	$0.07 \pm 0.01$
Rowing - Rowing moderate - Rowing 30spm	$0.77 \pm 0.05$	$0.51 \pm 1.09$ 0.54 ± 0.99	$1.07 \pm 1.10$ 0.88 ± 0.95	$0.70 \pm 0.07$ 0.77 ± 0.82
Running - Treadmill (mph - Treadmill ()	$0.74 \pm 0.05$	$0.34 \pm 0.09$	$0.03 \pm 0.03$	$0.77 \pm 0.02$
Running Treadmill 5mph Treadmill 0	$0.40 \pm 0.49$ 1.02 ± 0.79	$0.75 \pm 1.10$ 1 14 ± 0.07	$0.37 \pm 0.07$	$0.71 \pm 0.00$
Running - Treadmill 6mph - Treadmill 0	$1.02 \pm 0.79$	$1.14 \pm 0.97$	$0.83 \pm 0.89$	$0.00 \pm 0.03$
Sitting	$0.33 \pm 0.30$	$0.64 \pm 1.02$	$0.37 \pm 0.30$	$0.33 \pm 0.47$
Sitting Eidget feet legs	$0.30 \pm 0.17$	$2.47 \pm 1.03$ 1.24 ± 0.02	$0.34 \pm 0.79$	$0.09 \pm 0.08$
Sitting - Fidget leet legs	$0.24 \pm 0.22$	$1.34 \pm 0.93$	$0.10 \pm 0.09$	$0.20 \pm 0.20$
Sitting - Fluget fiands arms	$0.32 \pm 0.14$	$1.38 \pm 0.93$	$0.20 \pm 0.33$	$0.31 \pm 0.84$
Stalls - Ascellu stalls	$0.32 \pm 0.24$	$1.27 \pm 0.70$ 1.22 ± 0.87	$0.53 \pm 0.39$	$0.05 \pm 0.04$
Stants - Descend stans	$0.30 \pm 0.24$	$1.32 \pm 0.87$	$0.38 \pm 0.33$	$0.91 \pm 0.90$
Walking Transferill 2mmh Transferill 0	$0.01 \pm 0.44$	$0.30 \pm 0.27$	$0.29 \pm 0.33$	$0.37 \pm 0.34$
Walking - Treadmill 2mph - Treadmill 0	$0.79 \pm 1.34$	$1.03 \pm 1.40$	$0.90 \pm 1.93$	$0.83 \pm 1.43$
Walking - Treadmill 2mph - Treadmill 2 - light	$0.80 \pm 0.02$	$2.41 \pm 1.08$	$1.50 \pm 1.21$	$1.40 \pm 1.40$
Walking - Headmin Shiph - Headmin S - light	$1.08 \pm 0.81$	$1.24 \pm 1.20$	$1.12 \pm 1.29$	$1.03 \pm 0.94$
Walking - Treadmill 2mph - Treadmill 0 - Inoderate	$1.39 \pm 1.02$	$2.02 \pm 1.82$	$0.90 \pm 0.73$	$0.83 \pm 0.78$
waiking - freadmin Smph - freadmin 9 - nard	$1.07 \pm 0.92$	$0.73 \pm 1.02$	$1.45 \pm 1.72$	$0.94 \pm 1.22$
kileelilig	$0.33 \pm 0.24$	$0.76 \pm 0.40$	$0.15 \pm 0.21$	$0.14 \pm 0.12$
	$16.16 \pm 3.33$	$2.02 \pm 2.01$	$25.64 \pm 9.67$	$30.04 \pm 7.98$
Carrying grocenes	$1.42 \pm 1.21$	$0.72 \pm 0.73$	$1.60 \pm 2.55$	$1.31 \pm 1.43$
Doing disnes	$1.00 \pm 0.57$	$2.08 \pm 1.30$	$0.50 \pm 0.39$	$0.74 \pm 0.50$
Gardening	$0.55 \pm 0.35$	$2.06 \pm 1.22$	$0.42 \pm 0.48$	$0.59 \pm 0.46$
Ironing	$2.65 \pm 1.35$	$2.08 \pm 1.47$	$0.58 \pm 0.50$	$0.78 \pm 0.52$
Making the bed	$1.37 \pm 0.85$	$0.81 \pm 0.55$	$0.83 \pm 0.67$	$1.23 \pm 0.70$
Mopping	$0.74 \pm 0.46$	$0.78 \pm 0.61$	$0.61 \pm 0.78$	$0.86 \pm 0.49$
Playing videogames	$1.54 \pm 0.63$	$0.97 \pm 0.80$	$0.61 \pm 0.87$	$1.20 \pm 1.33$
Scrubbing a surface	$0.84 \pm 0.72$	$2.19 \pm 1.80$	$0.55 \pm 0.50$	$0.84 \pm 0.83$
Stacking groceries	$0.67 \pm 0.54$	$1.33 \pm 0.69$	$0.46 \pm 0.36$	$0.92 \pm 0.46$
Sweeping	$1.17 \pm 0.59$	$1.81 \pm 1.11$	$0.25 \pm 0.20$	$0.85 \pm 0.64$
Typing	$2.67 \pm 1.03$	$0.74 \pm 0.54$	$0.58 \pm 0.67$	$0.61 \pm 0.62$
Vacuuming	$1.00 \pm 0.48$	$3.44 \pm 1.56$	$0.37 \pm 0.64$	$0.70 \pm 0.52$
Walking around block	$1.94 \pm 2.34$	$1.02 \pm 0.70$	$1.08 \pm 1.01$	$1./5 \pm 1.65$
Washing windows	$0.87 \pm 0.48$	$1.46 \pm 0.74$	$0.59 \pm 0.44$	$0.92 \pm 0.48$
Watching TV	$1.19 \pm 0.37$	$1.00 \pm 0.73$	$0.99 \pm 1.26$	$1.00 \pm 1.01$
Weeding	$0.50 \pm 0.31$	$2.32 \pm 1.44$	$0.66 \pm 1.21$	$0.70 \pm 0.49$
Wiping/Dusting	$1.79 \pm 0.81$	$0.89 \pm 0.58$	$0.59 \pm 0.59$	$0.88 \pm 0.50$
Writing	$1.82 \pm 0.73$	$1.58 \pm 1.25$	$0.58 \pm 0.58$	$0.64 \pm 0.84$
taking out trash	$1.02 \pm 0.49$	$2.04 \pm 0.64$	$0.15 \pm 0.08$	$0.96 \pm 0.53$

Table A13-2: False positive rate obtained when training different classifiers using the *MaxAcceleration* feature set computed per sensor in a subject independent manner.

NN         NB         LogiBox         C4.5           Bench weight lifting - hard         8.9 + 13.12         7.05 + 15.01         5.75 ± 10.77         11.22 ± 18.84           Bench weight lifting - moderate         9.77 ± 11.74         9.71 ± 12.91         11.38 ± 17.30         11.28 ± 10.20           Bicep curls - hard         23.76 ± 21.60         18.29 ± 22.43         13.08 ± 17.56         12.82 ± 20.10           Bicep curls - noderate         18.91 ± 19.46         9.46 ± 11.93         23.00 ± 23.01         10.71 ± 19.18           Calishtenics - noderate         18.91 ± 19.46         9.46 ± 11.93         23.00 ± 23.07         10.07 ± 19.18           Calishtenics - Stups         74.77 ± 37.20         62.23 ± 45.90         72.66 ± 37.37         48.81 ± 44.17           Cycling - Cycle hight - Cycle 007pm         62.84 ± 73.66         74.23 ± 35.78         80.20 ± 35.06         73.43 ± 3.638           Cycling - Cycle hight - Cycle 807pm         24.07 ± 11.69         9.39 ± 12.15         75.40 ± 73.41         44.29 ± 37.10           Cycling - Cycle modrat - Cycle 807pm         24.07 ± 11.69         9.39 ± 12.15         72.44 ± 2.16 ± 2.26.6           Rowing - Rowing Indar - Rowing 30spm         24.36 ± 17.48         31.14 ± 2.26.1 ± 2.17 ± 32.48         19.43         14.92 ± 2.43         14.14 ± 2.24.92 ± 3.41.54         14.42 ± 2.53.19     <	Activity		F-Me	asure	
Bench weight lifting - Ind $8.59 + 13.12$ $7.05 \pm 15.01$ $5.75 \pm 10.01$ $5.75 \pm 10.71$ $13.02 \pm 18.48$ Bench weight lifting - moderate $9.77 \pm 11.74$ $9.71 \pm 12.91$ $11.38 \pm 17.39$ $11.28 \pm 16.20$ Bicep curds - hard $23.76 \pm 21.60$ $18.29 \pm 22.43$ $13.08 \pm 17.59$ $11.28 \pm 16.20$ Bicep curds - Ind $23.76 \pm 21.60$ $18.29 \pm 22.43$ $13.08 \pm 17.59$ $11.28 \pm 16.20$ Bicep curds - moderate $18.91 \pm 19.46$ $9.46 \pm 11.93$ $23.00 \pm 23.07 \pm 13.07$ $11.07 \pm 19.18$ Calisthenics - Crunches $46.22 \pm 46.19$ $13.66 \pm 23.76$ $52.63 \pm 41.17$ $11.53 \pm 54.6$ Calisthenics - Sit ups $74.77 \pm 37.20$ $62.23 \pm 45.99$ $72.66 \pm 37.73$ $48.18 \pm 44.17$ Cycling - Cycle light - Cycle 000pm $62.24 \pm 23.66$ $73.42 \pm 35.86$ $73.44 \pm 32.43$ Cycling - Cycle light - Cycle 000pm $56.21 \pm 24.38$ $68.71 \pm 31.13$ $75.40 \pm 27.41$ $44.29 \pm 27.10$ Cycling - Cycle light - Cycle 000pm $62.44 \pm 37.66$ $73.22 \pm 28.43$ $14.59 \pm 28.63$ $35.55 \pm 31.09$ Cycling - Cycle light - Cycle 000pm $62.44 \pm 9.09$ $65.01 \pm 7.79$ $82.61 \pm 22.18$ $75.30 \pm 31.19$ Rowing Rowing Bidt - Rowing 30spm $30.31 \pm 22.11$ $32.5 \pm 24.43$ $13.42 \pm 25.99$ $24.84 \pm 21.45$ Rowing Rowing Bidt - Rowing 30spm $33.4 \pm 22.14$ $13.42 \pm 25.99$ $14.84 \pm 21.45$ Rowing Rowing Bidt - Rowing 30spm $33.44 \pm 23.142$ $24.60 \pm 15.258 \pm 30.01 \pm 28.79$ Rowing Rowing Bidt - Rowing 30spm $33.44 \pm 20.142$ $14.29 \pm 20.45$ <		NN	NB	LogitBoost	C4.5
$\begin{split} \hline Bench weight lifting - inght = 19.42 \pm 19.83 24.61 \pm 22.00 35.03 \pm 12.2 14.60 \pm 21.42 \\ Bench weight lifting - moderate = 9.77 \pm 11.74 9.71 \pm 12.91 13.88 \pm 17.39 \pm 11.28 \pm 16.20 \\ Bicep curls - hard = 23.76 \pm 21.60 18.29 \pm 22.43 13.08 \pm 17.56 12.82 \pm 20.10 \\ Bicep curls - noderate = 18.91 \pm 19.46 9.46 \pm 11.93 2.50 \pm 23.07 \pm 23.07 \pm 10.07 \pm 19.18 \\ Calishtenics - Crunches = 46.28 \pm 46.19 13.66 \pm 23.76 52.63 \pm 41.74 18.15 \pm 35.46 \\ Calishtenics - Stups = 74.77 \pm 37.20 62.23 \pm 45.99 - 72.66 \pm 37.37 48.81 \pm 44.17 \\ Cycling - Cycle hight - Cycle 80rpm = 13.40 \pm 12.77 \pm 23.49 \pm 21.93 13.58 \pm 17.73 10.15 \pm 12.56 \\ Cycling - Cycle hight - Cycle 100pm = 62.84 \pm 37.66 74.23 \pm 35.78 80.20 \pm 35.06 73.43 \pm 36.38 \\ Cycling - Cycle light - Cycle 100pm = 24.97 \pm 11.69 9.39 \pm 12.15 = 77.55 \pm 19.33 15.97 \pm 15.68 \\ Lying down = 0.64.8 \pm 9.69 65.01 \pm 7.77 + 82.61 \pm 22.18 75.30 \pm 31.59 \pm 31.05 \\ Cycling - Cycle light - Cycle 80rpm = 24.97 \pm 11.69 9.39 \pm 12.15 = 27.55 \pm 19.33 15.57 \pm 15.68 \\ Lying down = 0.64.8 \pm 9.69 65.01 \pm 1.77 + 82.61 \pm 2.1.67 \pm 25.47 19.46 \pm 22.66 \\ Rowing - Rowing Indr - Rowing 30spm = 23.46 \pm 15.34 14.89 \pm 22.10 16.53 \pm 21.53 \pm 13.09 \pm 21.45 \\ Rowing - Rowing Indr - Rowing 30spm = 23.46 \pm 15.34 14.89 \pm 22.10 16.53 \pm 21.53 \pm 13.69 \pm 23.57 \\ Rowing - Rowing Indr - Rowing 30spm = 23.46 \pm 15.34 14.89 \pm 22.10 16.53 \pm 21.53 \pm 13.59 \pm 24.81 \pm 21.45 \\ Rowing - Rowing Indr - Rowing 30spm = 23.46 \pm 15.34 14.89 \pm 22.10 16.53 \pm 21.53 \pm 13.57 \pm 31.29 \pm 24.81 \pm 21.45 \\ Rowing - Rowing Indr - Rowing 30spm = 23.46 \pm 15.34 14.35 \pm 30.90 \pm 12.57 \pm 13.58 \pm 30.79 \pm 22.57 \\ Rowing - Rowing Indr - Rowing 30spm = 23.46 \pm 15.34 14.35 \pm 30.09 \pm 12.53 \pm 31.99 \\ Running - Treadmill 0.51.8 \pm 32.00 \pm 33.74 \pm 31.35 \pm 30.75 \pm 30.04 48.21 \pm 26.54 \\ Running - Treadmill 0.51.8 \pm 32.00 \pm 33.74 \pm 31.43 \pm 32.90 \pm 12.54 \pm 20.44 \\ Running - Treadmill 0.51.53 \pm 30.79 \pm 32.55 \pm 30.75 \pm 30.04 48.21 \pm 26.54 \\ Stars - Doex dtairs = 55.29 \pm 34.47 \pm 36.46 \pm 27.52 \pm 30.75 \pm 30.34 \pm 39.99 \\ 10.41 \pm 20.55 \pm 30.01 \pm 33.14 \pm 20.61 \\ Stairs - Stars = 10.$	Bench weight lifting - hard	8.59 ± 13.12	$7.05 \pm 15.01$	$5.75 \pm 10.77$	$13.02 \pm 18.84$
Bench weight lifting - moderate $9.77 + 11.74$ $9.71 + 12.91$ $11.38 \pm 17.35$ $11.28 \pm 16.20$ Bicep curls - Indu $23.76 \pm 21.60$ $18.29 \pm 22.43$ $13.08 \pm 17.56$ $12.82 \pm 20.10$ Bicep curls - Inductate $18.91 \pm 19.46$ $9.46 \pm 11.93$ $23.00 \pm 23.00 \pm 23.00 \pm 23.00 \pm 12.07$ $10.07 \pm 19.18$ Bicep curls - Inductate $18.91 \pm 19.46$ $9.46 \pm 11.93$ $23.00 \pm 23.00 \pm 23.00 \pm 12.07$ $10.07 \pm 19.18$ Calisthenics - Crunches $46.28 \pm 46.19$ $13.66 \pm 23.76$ $52.63 \pm 41.74$ $18.15 \pm 35.46$ Calisthenics - Sit ups $74.77 \pm 37.20$ $62.23 \pm 45.99$ $72.66 \pm 37.37$ $48.18 \pm 48.17$ Cycling - Cycle lapt - Cycle 800pm $56.21 \pm 24.88$ $68.71 \pm 31.13$ $75.40 \pm 27.14$ $44.29 \pm 37.61$ Cycling - Cycle light - Cycle 600pm $56.21 \pm 24.88$ $68.71 \pm 31.13$ $75.40 \pm 27.14$ $44.29 \pm 37.16$ Cycling - Cycle light - Cycle 800pm $24.97 \pm 11.60$ $9.39 \pm 12.15$ $27.55 \pm 19.32$ $53.51 \pm 31.09$ Cycling - Cycle Inght - Cycle 800pm $24.36 \pm 17.48$ $31.21 \pm 22.46$ $11.67 \pm 23.74$ $19.46 \pm 22.66$ Rowing - Rowing Inder - Rowing 30.50m $23.34 \pm 15.34$ $51.42 \pm 25.99$ $12.48 \pm 12.45$ Rowing - Rowing Inder - Rowing 30.50m $23.46 \pm 15.34$ $11.42 \pm 25.99$ $12.48 \pm 32.40$ Running - Treadmill Mph - Treadmill 0 $51.38 \pm 20.01$ $51.23 \pm 31.51$ $55.74 \pm 32.00$ Running - Treadmill Smph - Treadmill 0 $51.38 \pm 20.01$ $53.44 \pm 20.04$ $33.04 \pm 20.84$ Stining - Fidget fact lgs $36.40 \pm 32.74$ $43.66 \pm$	Bench weight lifting - light	$19.42 \pm 19.83$	$24.61 \pm 22.60$	$35.03 \pm 31.22$	$14.60 \pm 21.42$
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Bench weight lifting - moderate	9.77 ± 11.74	9.71 ± 12.91	$11.38 \pm 17.39$	$11.28 \pm 16.20$
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Bicep curls - hard	$23.76 \pm 21.60$	$18.29 \pm 22.43$	$13.08 \pm 17.56$	$12.82 \pm 20.10$
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Bicep curls - light	$27.98 \pm 19.13$	$11.90 \pm 14.64$	$32.97 \pm 17.10$	$17.80 \pm 19.20$
	Bicep curls - moderate	$18.91 \pm 19.46$	$9.46 \pm 11.93$	$23.00 \pm 23.07$	$10.07 \pm 19.18$
$\begin{array}{c} \label{eq:constraints} \hline $$ 14.77 \pm 37.20$ $$ 62.23 \pm 45.99$ $$ 72.66 \pm 37.37$ $$ 48.81 \pm 44.17$ $$ Cycle hight - Cycle Bopt - Cycle hight - Cycle Bopt - Display 13.40 \pm 12.77$ $$ 23.49 \pm 21.93$ $$ 13.585$ $$ 17.73$ $$ 10.15 \pm 12.56$ $$ Cycle hight - Cycle Bopt - Display 13.40 \pm 12.77$ $$ 12.89 \pm 24.39$ $$ 11.31$ $$ 75.40$ $$ 11.51 \pm 12.56$ $$ 12.57$ $$ 80.20 \pm 35.06$ $$ 73.43 \pm 36.38$ $$ Cycle hight - Cycle Bopt - Display 10.00 \pm 21.79$ $$ 12.89 \pm 24.34$ $$ 11.59 \pm 24.80$ $$ 15.59 \pm 31.09$ $$ 21.79$ $$ 12.05$ $$ 15.55$ \pm 31.09$ Cycle hight - Cycle Bopt - Display - Cycle Bopt - Cycle Bopt - Display - Cycle Bopt - Cycle Bopt - Display - Cycle Bopt - Cycle Bopt - Display - Cycle Bopt - Display - Cycle Bopt - Display - Cycle Bopt - Display - Cycle Bopt - Cycle Bopt - Cycle Bopt - Display - Cycle Bopt - Display - Cycle Bopt - Display - Display - Cy$	Calisthenics - Crunches	$46.28 \pm 46.19$	$13.66 \pm 23.76$	$52.63 \pm 41.74$	$18.15 \pm 35.46$
$\begin{array}{c} \hline Cycling - Cycle hard - Cycle 80rpm \\ \hline 13.40 \pm 12.77 \\ \hline 23.49 \pm 21.93 \\ \hline 13.58 \pm 17.73 \\ \hline 10.15 \pm 12.56 \\ \hline Cycling - Cycle light - Cycle 80rpm \\ \hline 62.84 \pm 37.66 \\ \hline 74.23 \pm 35.78 \\ \hline 80.20 \pm 35.06 \\ \hline 73.43 \pm 36.38 \\ \hline 68.71 \pm 31.13 \\ \hline 75.40 \pm 27.41 \\ \hline 44.29 \pm 37.10 \\ \hline Cycling - Cycle light - Cycle 80rpm \\ \hline 24.97 \pm 11.69 \\ \hline 9.39 \pm 12.15 \\ \hline 27.55 \pm 19.33 \\ \hline 15.97 \pm 15.68 \\ \hline Lying down \\ \hline 66.48 \pm 9.69 \\ \hline 65.01 \pm 7.79 \\ \hline 82.61 \pm 22.43 \\ \hline 15.28 \pm 28.43 \\ \hline 15.41 \pm 22.18 \\ \hline 75.30 \pm 31.19 \\ \hline Cycling - Cycle moderate - Cycle 80rpm \\ \hline 24.39 \pm 11.69 \\ \hline 9.39 \pm 12.15 \\ \hline 27.55 \pm 19.33 \\ \hline 15.97 \pm 15.68 \\ \hline Lying down \\ \hline Rowing - Rowing fund - Rowing 30spm \\ \hline 23.46 \pm 17.48 \\ \hline 31.21 \pm 22.46 \\ \hline 21.67 \pm 25.47 \\ \hline 19.46 \pm 22.65 \\ \hline Rowing - Rowing funderate - Rowing 30spm \\ \hline 23.46 \pm 15.34 \\ \hline 14.89 \pm 22.10 \\ \hline 16.53 \pm 21.53 \\ \hline 18.42 \pm 22.58 \\ \hline Rowing - Treadmill Mmph - Treadmill 0 \\ \hline 31.38 \pm 26.01 \\ \hline 51.23 \pm 31.51 \\ \hline 57.74 \pm 32.00 \\ \hline 46.21 \pm 26.53 \\ \hline Running - Treadmill 0 \\ \hline 41.36 \pm 32.92 \\ \hline 45.10 \pm 35.42 \\ \hline 50.73 \pm 30.04 \\ \hline 82.9 \\ \hline 13.46 \pm 10.77 \\ \hline 33.88 \pm 27.06 \\ \hline 50.41 \pm 77.98 \\ \hline 51.23 \pm 31.51 \\ \hline 57.74 \pm 32.00 \\ \hline 46.21 \pm 26.53 \\ \hline Running - Treadmill 0 \\ \hline 41.36 \pm 32.92 \\ \hline 45.10 \pm 33.44 \\ \hline 41.75 \pm 30.90 \\ \hline 47.34 \pm 39.98 \\ \hline 30.68 \pm 38.11 \\ \hline 51.116 \\ \hline 51.116 \\ \hline 51.116 \\ \hline 51.116 \\ \hline 51.116 \\ \hline 51.116 \\ \hline 51.23 \pm 51.45 \\ \hline 51.23 \pm 51.45 \\ \hline 51.23 \pm 51.45 \\ \hline 51.23 \pm 51.45 \\ \hline 51.23 \pm 51.45 \\ \hline 51.23 \pm 51.45 \\ \hline 51.23 \pm 51.45 \\ \hline 51.23 \pm 51.45 \\ \hline 51.23 \pm 51.45 \\ \hline 51.23 \pm 51.45 \\ \hline 51.23 \pm 51.45 \\ \hline 51.23 \pm 51.45 \\ \hline 51.23 \pm 51.45 \\ \hline 51.23 \pm 51.45 \\ \hline 51.23 \pm 51.45 \\ \hline 51.23 \pm 51.45 \\ \hline 51.23 \pm 51.45 \\ \hline 51.23 \pm 51.45 \\ \hline 51.23 \pm 51.47 \\ \hline 51.25$	Calisthenics - Sit ups	74.77 + 37.20	62.23 + 45.99	72.66 + 37.37	48.81 + 44.17
$\begin{array}{c} \hline Cycling - Cycle light - Cycle 100 pm \\ \hline 62.84 \pm 37.66 \\ 74.23 \pm 35.78 \\ 80.20 \pm 35.06 \\ 73.43 \pm 36.38 \\ Cycling - Cycle light - Cycle 60 pm \\ 73.43 \pm 36.38 \\ Cycling - Cycle light - Cycle 80 pm \\ 10.0 \pm 21.79 \\ 24.97 \pm 11.69 \\ 93.9 \pm 12.15 \\ 27.55 \pm 19.33 \\ 15.97 \pm 15.68 \\ Lying down \\ Rowing hard - Rowing 30 spm \\ 24.36 \pm 17.48 \\ 31.21 \pm 22.48 \\ 24.24 \\ 21.67 \pm 22.48 \\ 75.30 \pm 31.55 \\ 22.48 \\$	Cycling - Cycle hard - Cycle 80rpm	$13.40 \pm 12.77$	$23.49 \pm 21.93$	$13.58 \pm 17.73$	$10.15 \pm 12.56$
$\begin{array}{c} \hline Cycling - Cycle light - Cycle 60pm \\ \hline Solution 21, 79 \\ \hline Cycling - Cycle indert - Cycle 80pm \\ \hline Solution 21, 79 \\ \hline Solution 21, 79 \\ \hline Solution 21, 79 \\ \hline Solution 21, 79 \\ \hline Solution 21, 79 \\ \hline Solution 21, 79 \\ \hline Solution 21, 79 \\ \hline Solution 21, 79 \\ \hline Solution 21, 79 \\ \hline Solution 21, 79 \\ \hline Solution 21, 79 \\ \hline Solution 21, 79 \\ \hline Solution 21, 79 \\ \hline Solution 21, 79 \\ \hline Solution 21, 79 \\ \hline Solution 21, 70 \\ $	Cycling - Cycle light - Cycle 100rpm	62.84 + 37.66	74.23 + 35.78	80.20 + 35.06	73.43 + 36.38
$\begin{array}{c} \hline Cycling - Cycle light - Cycle 80rpm \\ 24.97 \pm 11.69 \\ 9.39 \pm 12.15 \\ 27.55 \pm 19.31 \\ 15.97 \pm 15.68 \\ 15.97 \pm 11.69 \\ 23.9 \pm 28.43 \\ 41.59 \pm 28.05 \\ 35.55 \pm 31.09 \\ Cycling - Cycle moderate - Cycle 80rpm \\ 24.97 \pm 11.69 \\ 9.39 \pm 12.15 \\ 27.79 \\ 82.61 \pm 22.18 \\ 75.30 \\ 22.11 \\ 75.28 \\ 22.12 \\ 15.28 \\ 22.12 \\ 15.28 \\ 22.12 \\ 15.28 \\ 22.12 \\ 15.28 \\ 22.12 \\ 15.28 \\ 22.12 \\ 15.28 \\ 22.10 \\ 15.55 \\ 21.55 \\ 15.25 \\ 21.55 \\ 14.2 \\ 25.99 \\ 24.81 \\ 21.42 \\ 25.74 \\ 23.90 \\ 47.34 \\ 23.98 \\ 30.06 \\ 23.84 \\ 25.92 \\ 45.10 \\ 23.84 \\ 25.90 \\ 47.34 \\ 23.99 \\ 30.64 \\ 23.84 \\ 23.90 \\ 47.34 \\ 23.98 \\ 30.69 \\ 23.84 \\ 23.90 \\ 47.34 \\ 23.99 \\ 30.64 \\ 23.84 \\ 23.90 \\ 47.34 \\ 23.99 \\ 30.64 \\ 23.84 \\ 23.90 \\ 47.34 \\ 23.99 \\ 30.64 \\ 23.84 \\ 23.90 \\ 47.34 \\ 23.99 \\ 30.29 \\ 23.126 \\ 51.30 \\ 51.30 \\ 51.47 \\ 40.64 \\ 23.30 \\ 30.29 \\ 43.31 \\ 24.82 \\ 49.64 \\ 43.30 \\ 39.90 \\ 24.31 \\ 24.31 \\ 24.31 \\ 24.31 \\ 24.31 \\ 24.31 \\ 24.42 \\ 24.44 \\ 24.44 \\ 24.44 \\ 24.44 \\ 24.44 \\ 24.44 \\ 24.44 \\ 24.44 \\ 24.44 \\ 24.44 \\ 24.44 \\ 24.44 \\ 24.44 \\ 24.44 \\ 24.44 \\ 24.44 \\ 24.44 \\ 24.46 \\ 24.24 \\ 24.44 \\ 2$	Cycling - Cycle light - Cycle 60rpm	56.21 + 24.38	68.71 + 31.13	75.40 + 27.41	44.29 + 37.10
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Cycling - Cycle light - Cycle 80rpm	31.00 + 21.79	32.89 + 28.43	41.59 + 28.05	35.55 + 31.09
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Cycling - Cycle moderate - Cycle 80rpm	$24.97 \pm 11.69$	$9.39 \pm 12.15$	$27.55 \pm 19.33$	$15.97 \pm 15.68$
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$	Lying down	66.48 + 9.69	65.01 + 7.79	82.61 + 22.18	75.30 + 31.19
RowingRowi	Rowing - Rowing hard - Rowing 30spm	$2436 \pm 1748$	3121 + 2246	$21.67 \pm 25.47$	1946 + 2266
Rowing - Rowing moderate - Rowing 30spm $23.46 \pm 15.34$ $14.89 \pm 22.10$ $16.53 \pm 21.53$ $18.61 \pm 19.06$ Running - Treadmill 4mph - Treadmill 0 $36.99 \pm 29.94$ $33.88 \pm 37.66$ $50.64 \pm 33.89$ $30.71 \pm 29.80$ Running - Treadmill 6mph - Treadmill 0 $41.36 \pm 32.92$ $45.10 \pm 35.42$ $50.75 \pm 30.04$ $38.91 \pm 28.79$ Sitting - Treadmill 6mph - Treadmill 0 $41.36 \pm 32.92$ $45.10 \pm 35.42$ $50.75 \pm 30.04$ $38.91 \pm 28.79$ Sitting - Fidget feet legs $36.40 \pm 33.74$ $41.35 \pm 10.900$ $47.34 \pm 39.98$ $30.68 \pm 38.11$ Sitting - Fidget feet legs $36.40 \pm 33.74$ $41.35 \pm 30.900$ $47.34 \pm 39.98$ $30.68 \pm 38.11$ Sitting - Stairs - Ascend stairs $48.35 \pm 34.79$ $49.65 \pm 28.54$ $49.64 \pm 32.006$ Stairs - Descend stairs $55.29 \pm 34.47$ $43.68 \pm 27.22$ $50.19 \pm 33.78$ $38.48 \pm 29.002$ Standing $24.23 \pm 15.46$ $6.68 \pm 12.78$ $49.64 \pm 3.30$ $39.90 \pm 34.33$ Walking - Treadmill 2mph - Treadmill 0 $15.59 \pm 13.25$ $30.15 \pm 26.38$ $26.14 \pm 17.12$ $20.95 \pm 20.02$ Walking - Treadmill 3mph - Treadmill 6 - moderate $14.68 \pm 12.66$ $19.89 \pm 19.69$ $13.01 \pm 12.33$ $11.08 \pm 12.60$ Walking - Treadmill 3mph - Treadmill 9 - hard $25.52 \pm 27.27$ $18.79 \pm 22.57$ $19.63 \pm 17.78$ $12.58 \pm 16.20$ I cardmill 3mph - Treadmill 9 - hard $25.52 \pm 27.27$ $18.79 \pm 22.57$ $19.63 \pm 17.78$ $12.58 \pm 16.20$ Kalking - Treadmill 3mph - Treadmill 9 - hard $25.52 \pm 27.27$ $18.79 \pm 22.57$ $10.61 \pm 17.78$ $12.54 $	Rowing - Rowing light - Rowing 30spm	30.31 + 22.11	32.82 + 24.43	31.42 + 25.99	24.81 + 21.45
IteratingIteratingIteratingIteratingIteratingIteratingIteratingIteratingIteratingIteratingRunningTreadmill omphTreadmill 0 $51.38 \pm 20.01$ $51.23 \pm 31.51$ $55.74 \pm 32.00$ $46.21 \pm 26.53$ RunningTreadmill omphTreadmill 0 $41.36 \pm 32.92$ $45.10 \pm 35.42$ $50.75 \pm 30.04$ $38.91 \pm 28.79$ Sitting $674 \pm 9.40$ $13.46 \pm 11.07$ $23.58 \pm 29.90$ $12.27 \pm 19.43$ SittingFidget feel legs $36.40 \pm 33.74$ $41.35 \pm 30.90$ $47.34 \pm 39.98$ $30.68 \pm 38.11$ SittingFidget hands arms $30.16 \pm 26.00$ $33.46 \pm 30.84$ $46.15 \pm 35.80$ $30.29 \pm 31.26$ StairsAscend stairs $55.29 \pm 34.47$ $43.66 \pm 27.22$ $50.19 \pm 33.78$ $38.48 \pm 29.02$ StairsAscend stairs $55.29 \pm 34.47$ $43.66 \pm 27.22$ $50.19 \pm 33.78$ $38.48 \pm 29.02$ StairsAscend stairs $55.29 \pm 34.47$ $43.66 \pm 27.22$ $50.19 \pm 33.78$ $38.48 \pm 29.02$ StainsTreadmill OmphTreadmill 0 $45.59 \pm 13.25$ $30.15 \pm 26.38$ $26.14 \pm 17.12$ $20.95 \pm 20.02$ WalkingTreadmill 3mphTreadmill 0 $15.59 \pm 13.25$ $30.15 \pm 26.38$ $26.14 \pm 17.12$ $20.95 \pm 20.02$ WalkingTreadmill 3mphTreadmill 6-moderate $14.68 \pm 12.66$ $19.89 \pm 19.60$ $10.01 \pm 10.12$ WalkingTreadmill 3mphTreadmill 3 $-11.64$ $25.52 \pm 22.72$ $18.79 \pm 22.57$ $19.63 \pm 17.78$ $12.58 \pm 16.20$ Walking </td <td>Rowing - Rowing moderate - Rowing 30spm</td> <td><math>23.46 \pm 15.34</math></td> <td><math>14.89 \pm 22.10</math></td> <td>1653 + 2153</td> <td><math>18.61 \pm 19.06</math></td>	Rowing - Rowing moderate - Rowing 30spm	$23.46 \pm 15.34$	$14.89 \pm 22.10$	1653 + 2153	$18.61 \pm 19.06$
RunningTreadmill 0 $51.32 \pm 20.01$ $51.32 \pm 31.51$ $55.74 \pm 32.00$ $46.21 \pm 26.53$ RunningTreadmill 0 $41.36 \pm 32.92$ $45.10 \pm 35.42$ $50.75 \pm 30.04$ $38.91 \pm 28.79$ Sitting $6.74 \pm 9.40$ $13.46 \pm 11.07$ $23.58 \pm 29.90$ $12.27 \pm 19.43$ Sitting - Fidget feet legs $36.40 \pm 33.74$ $41.35 \pm 30.90$ $47.34 \pm 39.98$ $30.68 \pm 38.11$ Sitting - Fidget hands arms $30.16 \pm 26.00$ $33.46 \pm 30.84$ $46.15 \pm 35.80$ $30.29 \pm 31.26$ Stairs - Ascend stairs $48.35 \pm 34.79$ $49.65 \pm 28.54$ $54.29 \pm 36.42$ $49.64 \pm 32.06$ Stairs - Descend stairs $55.29 \pm 34.47$ $43.68 \pm 27.22$ $50.19 \pm 33.78$ $83.48 \pm 29.02$ Standing24.23 \pm 15.46 $6.68 \pm 12.78$ $49.64 \pm 33.30$ $39.90 \pm 34.33$ Walking - Treadmill 3mph - Treadmill 0 $48.18 \pm 31.74$ $61.77 \pm 31.54$ $60.80 \pm 28.68$ $45.42 \pm 32.64$ Walking - Treadmill 3mph - Treadmill 0 $45.59 \pm 13.25$ $30.15 \pm 26.32$ $26.14 \pm 17.12$ $20.95 \pm 20.02$ Walking - Treadmill 3mph - Treadmill 0 $45.52 \pm 22.72$ $18.79 \pm 12.79$ $13.01 \pm 12.33$ $11.08 \pm 12.60$ Walking - Treadmill 3mph - Treadmill 0 $-13.32$ $13.30 \pm 11.44$ $73.36 \pm 35.83$ $65.00 \pm 10.02$ $53.97 \pm 8.82$ Unknown $53.10 \pm 6.33$ $10.48 \pm 3.85$ $65.60 \pm 10.02$ $53.97 \pm 8.82$ $55.92 \pm 12.70$ $13.21 \pm 13.54$ $13.44 \pm 20.68$ Doing disks $28.38 \pm 17.79$ $29.18 \pm 21.95$ $50.50 \pm 23.69$ $28.82 \pm 25.60$ Gardening <t< td=""><td>Running - Treadmill 4mph - Treadmill 0</td><td><math>36.99 \pm 29.94</math></td><td><math>33.88 \pm 37.66</math></td><td>50.64 + 35.89</td><td><math>30.71 \pm 29.80</math></td></t<>	Running - Treadmill 4mph - Treadmill 0	$36.99 \pm 29.94$	$33.88 \pm 37.66$	50.64 + 35.89	$30.71 \pm 29.80$
Running - Treadmill Gruph - Treadmill 0 $41.36 \pm 32.92$ $45.10 \pm 33.42$ $50.75 \pm 30.04$ $38.91 \pm 28.79$ Sitting $6.74 \pm 9.40$ $13.46 \pm 11.07$ $23.58 \pm 29.90$ $12.27 \pm 19.43$ Sitting - Fidget feet legs $36.40 \pm 33.74$ $41.35 \pm 30.90$ $47.34 \pm 39.98$ $30.68 \pm 38.11$ Sitting - Fidget hands arms $30.16 \pm 26.00$ $33.46 \pm 30.84$ $46.15 \pm 35.80$ $30.29 \pm 31.26$ Stairs - Ascend stairs $48.35 \pm 34.79$ $49.65 \pm 28.54$ $54.29 \pm 36.42$ $49.64 \pm 32.06$ Stairs - Descend stairs $55.29 \pm 34.47$ $43.68 \pm 12.78$ $49.64 \pm 32.06$ Starding $24.23 \pm 15.46$ $6.68 \pm 12.78$ $49.64 \pm 32.06$ Starding $24.23 \pm 15.46$ $6.68 \pm 12.78$ $49.64 \pm 32.06$ Walking - Treadmill Omph - Treadmill 0 $48.18 \pm 31.74$ $61.77 \pm 31.54$ $60.80 \pm 28.68$ Walking - Treadmill 3mph - Treadmill 0 $15.59 \pm 13.25$ $30.15 \pm 26.38$ $26.14 \pm 17.12$ $20.95 \pm 20.02$ Walking - Treadmill 3mph - Treadmill 0 - moderate $14.68 \pm 12.66$ $19.89 \pm 19.69$ $13.01 \pm 12.33$ $11.00 \pm 12.60$ Walking - Treadmill 3mph - Treadmill 9 - hard $25.52 \pm 22.72$ $18.79 \pm 22.57$ $19.63 \pm 17.78$ $12.58 \pm 16.20$ Inceling $15.96 \pm 13.32$ $15.30 \pm 11.44$ $73.36 \pm 35.83$ $65.70 \pm 42.25$ unknown $53.10 \pm 6.33$ $10.48 \pm 38.56$ $56.60 \pm 10.02$ $53.79 \pm 28.88$ Carrying groceries $27.40 \pm 26.54$ $39.80 \pm 28.13$ $28.01 \pm 28.84$ $27.72 \pm 24.94$ Making the bed $34.31 \pm 24.03$ $30.08 \pm 1$	Running - Treadmill 5mph - Treadmill 0	$50.39 \pm 29.91$ 51 38 + 26 01	$51.00 \pm 31.00$ $51.23 \pm 31.51$	$55.01 \pm 33.09$ $55.74 \pm 32.00$	$46.21 \pm 29.00$
Kinning Treadmint on pin 1 readmint of $47.80 \pm 9.22$ $67.32 \pm 20.32 $	Running - Treadmill 6mph - Treadmill 0	$41.36 \pm 32.92$	$45.10 \pm 31.51$	$50.75 \pm 30.04$	$38.91 \pm 28.79$
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Sitting	$674 \pm 940$	$\frac{43.10 \pm 33.42}{13.46 \pm 11.07}$	$23.58 \pm 29.90$	$12.27 \pm 10.43$
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Sitting - Fidget feet legs	$36.40 \pm 33.74$	$13.40 \pm 11.07$ $41.35 \pm 30.90$	1734 + 39.98	$12.27 \pm 17.43$ 30.68 + 38.11
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$	Sitting - Fidget hands arms	$30.40 \pm 35.74$ $30.16 \pm 26.00$	$41.35 \pm 30.90$ 33.46 + 30.84	$47.34 \pm 37.90$	$30.00 \pm 30.11$ $30.29 \pm 31.26$
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Stairs Ascend stairs	$30.10 \pm 20.00$	$33.40 \pm 30.84$	$40.13 \pm 35.80$ 54.29 ± 36.42	$30.29 \pm 31.20$
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Stairs - Ascend stairs	$48.33 \pm 34.79$ 55 29 + 34 47	$43.03 \pm 28.34$ $43.68 \pm 27.22$	$54.29 \pm 30.42$ 50 19 + 33 78	$\frac{49.04 \pm 32.00}{38.48 \pm 29.02}$
Standing $22.52 \pm 15.40$ $45.03 \pm 12.74$ $47.04 \pm 35.35$ $45.24 \pm 32.54$ Walking - Treadmill 2mph - Treadmill 0 $48.18 \pm 31.74$ $61.77 \pm 31.54$ $60.80 \pm 28.68$ $45.42 \pm 32.54$ Walking - Treadmill 3mph - Treadmill 3 - light $8.75 \pm 8.02$ $14.60 \pm 16.07$ $10.79 \pm 14.36$ $10.01 \pm 10.12$ Walking - Treadmill 3mph - Treadmill 6 - moderate $14.68 \pm 12.66$ $19.89 \pm 19.69$ $13.01 \pm 12.33$ $11.08 \pm 12.60$ Walking - Treadmill 3mph - Treadmill 9 - hard $25.52 \pm 22.72$ $18.79 \pm 22.57$ $19.63 \pm 17.78$ $12.58 \pm 16.20$ Kneeling $15.96 \pm 13.32$ $15.30 \pm 11.44$ $73.36 \pm 35.83$ $65.70 \pm 42.25$ unknown $53.10 \pm 6.33$ $10.48 \pm 3.85$ $65.60 \pm 10.02$ $53.97 \pm 8.88$ Carrying groceries $27.40 \pm 26.54$ $39.80 \pm 28.13$ $28.01 \pm 28.14$ $21.34 \pm 20.68$ Doing dishes $28.38 \pm 17.79$ $29.18 \pm 21.95$ $50.50 \pm 23.69$ $28.82 \pm 25.60$ Gardening $19.05 \pm 22.00$ $19.53 \pm 22.36$ $27.67 \pm 30.44$ $16.27 \pm 22.88$ Ironing $32.52 \pm 19.51$ $42.52 \pm 40.64$ $40.06 \pm 31.37$ $37.94 \pm 29.72$ Making the bed $34.31 \pm 24.03$ $30.08 \pm 19.99$ $32.94 \pm 21.00$ $25.64 \pm 17.56$ Mopping $28.57 \pm 24.48$ $22.86 \pm 23.06$ $27.02 \pm 24.96$ $24.69 \pm 21.27$ Playing videogames $12.57 \pm 10.52$ $10.85 \pm 15.89$ $31.55 \pm 28.34$ $27.73 \pm 30.98$ Stacking groceries $20.35 \pm 22.41$ $19.88 \pm 20.07$ $17.52 \pm 16.40$ $13.52 \pm 14.04$ Sweeping $19$	Standing	$33.27 \pm 34.47$ 24 23 + 15 46	$6.68 \pm 12.78$	$49.64 \pm 33.70$	$30.40 \pm 27.02$ $30.00 \pm 34.33$
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Walking - Treadmill 2mph - Treadmill ()	$48.18 \pm 31.74$	$61.77 \pm 31.54$	$49.04 \pm 33.30$ 60.80 + 28.68	$35.00 \pm 34.33$
Walking - Treadmill 3mph - Treadmill 3 - light $15.39 \pm 15.32 \pm 130.51$ $20.53 \pm 120.32$ $20.54 \pm 17.12$ $20.57 \pm 120.32$ Walking - Treadmill 3mph - Treadmill 6 - moderate $14.68 \pm 12.66$ $118.09 \pm 19.69$ $13.01 \pm 12.33$ $11.08 \pm 12.60$ Walking - Treadmill 3mph - Treadmill 9 - hard $25.52 \pm 22.72$ $18.79 \pm 22.57$ $19.63 \pm 17.78$ $12.58 \pm 16.20$ kneeling $15.96 \pm 13.32$ $15.30 \pm 11.44$ $73.36 \pm 35.83$ $65.70 \pm 42.25$ uknown $53.10 \pm 6.33$ $10.48 \pm 3.85$ $65.60 \pm 10.02$ $53.97 \pm 8.88$ Carrying groceries $27.40 \pm 26.54$ $39.80 \pm 28.13$ $28.01 \pm 28.14$ $21.34 \pm 20.68$ Doing dishes $28.38 \pm 17.79$ $29.18 \pm 21.95$ $50.50 \pm 23.69$ $28.82 \pm 25.60$ Gardening $19.05 \pm 22.00$ $19.53 \pm 22.36$ $27.67 \pm 30.44$ $16.27 \pm 22.88$ Ironing $32.52 \pm 19.51$ $42.52 \pm 24.06$ $40.06 \pm 31.37$ $37.94 \pm 29.72$ Making the bed $34.31 \pm 24.03$ $30.08 \pm 19.99$ $32.94 \pm 21.00$ $25.64 \pm 17.56$ Mopping $28.57 \pm 24.48$ $22.86 \pm 23.06$ $27.02 \pm 24.96$ $24.69 \pm 21.27$ Playing videogames $12.57 \pm 10.52$ $10.85 \pm 15.89$ $31.55 \pm 28.34$ $27.73 \pm 30.98$ Scrubbing a surface $16.90 \pm 18.41$ $14.06 \pm 15.91$ $20.88 \pm 26.06$ $14.89 \pm 18.59$ Stacking groceries $20.35 \pm 22.41$ $19.88 \pm 20.07$ $17.52 \pm 16.40$ $13.52 \pm 14.04$ Sweeping $19.15 \pm 19.40$ $22.90 \pm 22.75$ $25.39 \pm 19.23$ $17.86 \pm 19.00$ Typing $30.27 \pm 16$	Walking - Treadmill 3mph - Treadmill 0	$40.10 \pm 31.74$ 15 59 + 13 25	$30.15 \pm 26.38$	$26.14 \pm 17.12$	$\frac{49.42 \pm 32.04}{20.95 \pm 20.02}$
Walking - Treadmin 3mph - Treadmin 5 - ngint $0.73 \pm 0.30 \pm 10.30 \pm 10.30 \pm 10.31 \pm 12.33$ $10.60 \pm 10.72 \pm 10.30 \pm 10.31 \pm 12.33$ Walking - Treadmill 3mph - Treadmill 9 - hard $25.52 \pm 22.72$ $18.79 \pm 22.57$ $19.63 \pm 17.78$ $12.58 \pm 16.20$ kneeling $15.96 \pm 13.32 \pm 22.72$ $18.79 \pm 22.57$ $19.63 \pm 17.78$ $12.58 \pm 16.20$ kneeling $15.96 \pm 13.32 \pm 15.30 \pm 11.44$ $73.36 \pm 35.83 = 65.70 \pm 42.25$ unknown $53.10 \pm 6.33 = 10.48 \pm 3.85 = 65.60 \pm 10.02 = 53.97 \pm 8.88$ Carrying groceries $27.40 \pm 26.54 = 39.80 \pm 28.13 = 28.01 \pm 28.14 = 21.34 \pm 20.68$ Doing dishes $28.38 \pm 17.79 = 29.18 \pm 21.95 = 50.50 \pm 23.69 = 28.82 \pm 25.60$ Gardening $19.05 \pm 22.00 = 19.53 \pm 22.36 = 27.67 \pm 30.44 = 16.27 \pm 22.88$ Ironing $32.52 \pm 19.51 = 42.52 \pm 24.06 = 40.06 \pm 31.37 = 37.94 \pm 29.72$ Making the bed $34.31 \pm 24.03 = 30.08 \pm 19.99 = 32.94 \pm 21.00 = 25.64 \pm 17.56$ Mopping $28.57 \pm 24.48 = 22.86 \pm 23.06 = 27.02 \pm 24.96 = 24.69 \pm 21.27$ Playing videogames $12.57 \pm 10.52 = 10.85 \pm 15.89 = 31.55 \pm 28.34 = 27.73 \pm 30.98$ Scrubbing a surface $16.90 \pm 18.41 = 14.06 \pm 15.91 = 20.88 \pm 26.06 = 14.89 \pm 18.59$ Stacking groceries $20.35 \pm 22.21 = 27.77 \pm 23.41 = 38.12 \pm 27.47 = 25.90 \pm 23.87 = 25.90 \pm 23.87 = 27.90 \pm 23.91 = 27.47 = 25.90 \pm 23.87 = 27.91 \pm 23.91 = 27.47 = 25.90 \pm 23.87 = 27.91 \pm 24.91 = 24.9$	Walking - Treadmill 3mph - Treadmill 3 - light	$8.75 \pm 8.02$	$14.60 \pm 16.07$	$10.79 \pm 17.12$	$10.01 \pm 10.12$
Walking - Treadmill o Finderate14.06 $\pm$ 12.0015.05 $\pm$ 12.0015.05 $\pm$ 12.0015.05 $\pm$ 12.00Walking - Treadmill 3mph - Treadmill 9 - hard25.52 $\pm$ 22.7218.79 $\pm$ 22.5719.63 $\pm$ 17.7812.58 $\pm$ 16.20kneeling15.96 $\pm$ 13.3215.30 $\pm$ 11.4473.36 $\pm$ 35.8365.70 $\pm$ 42.25unknown53.10 $\pm$ 6.3310.48 $\pm$ 3.8565.60 $\pm$ 10.0253.97 $\pm$ 8.88Carrying groceries27.40 $\pm$ 26.5439.80 $\pm$ 28.1328.01 $\pm$ 28.1421.34 $\pm$ 20.68Doing dishes28.38 $\pm$ 17.7929.18 $\pm$ 21.9550.50 $\pm$ 23.6928.82 $\pm$ 25.60Gardening19.05 $\pm$ 22.0019.53 $\pm$ 22.3627.67 $\pm$ 30.4416.27 $\pm$ 22.88Ironing32.52 $\pm$ 19.5142.52 $\pm$ 24.0640.06 $\pm$ 31.3737.94 $\pm$ 29.72Making the bed34.31 $\pm$ 24.0330.08 $\pm$ 19.9932.94 $\pm$ 21.0025.64 $\pm$ 17.56Mopping28.57 $\pm$ 24.4822.86 $\pm$ 23.0627.02 $\pm$ 24.9624.69 $\pm$ 21.27Playing videogames12.57 $\pm$ 10.5210.85 $\pm$ 15.8931.55 $\pm$ 28.3427.73 $\pm$ 30.98Scrubbing a surface16.90 $\pm$ 18.4114.06 $\pm$ 15.9120.88 $\pm$ 26.0614.89 $\pm$ 18.59Stacking groceries20.35 $\pm$ 22.4119.88 $\pm$ 20.0717.52 $\pm$ 16.4013.52 $\pm$ 14.04Sweeping19.15 $\pm$ 19.4022.90 $\pm$ 22.7525.39 $\pm$ 19.2317.86 $\pm$ 19.00Typing30.27 $\pm$ 16.0736.56 $\pm$ 30.1653.72 $\pm$ 34.3750.10 $\pm$ 35.50Vacuuming29.33 $\pm$ 25.2124.77 $\pm$ 23.4138.12 $\pm$ 27.4725.90	Walking Treadmill 3mph Treadmill 6 moderate	$14.68 \pm 12.66$	$14.00 \pm 10.07$ 10.80 ± 10.60	$10.77 \pm 14.30$ 13.01 ± 12.33	$10.01 \pm 10.12$ 11.08 ± 12.60
Waking = Treadmin 9 Final $2.32 \pm 2.32 \pm 22.32$ $18.09 \pm 22.37 \pm 19.05 \pm 117.18$ $12.38 \pm 10.26$ kneeling $15.96 \pm 13.32$ $15.30 \pm 11.44$ $73.36 \pm 35.83$ $65.70 \pm 42.25$ unknown $53.10 \pm 6.33$ $10.48 \pm 3.85$ $65.60 \pm 10.02$ $53.97 \pm 8.88$ Carrying groceries $27.40 \pm 26.54$ $39.80 \pm 28.13$ $28.01 \pm 28.14$ $21.34 \pm 20.68$ Doing dishes $28.38 \pm 17.79$ $29.18 \pm 21.95$ $50.50 \pm 23.69$ $28.82 \pm 25.60$ Gardening $19.05 \pm 22.00$ $19.53 \pm 22.36$ $27.67 \pm 30.44$ $16.27 \pm 22.88$ Ironing $32.52 \pm 19.51$ $42.52 \pm 24.06$ $40.06 \pm 31.37$ $37.94 \pm 29.72$ Making the bed $34.31 \pm 24.03$ $30.08 \pm 19.99$ $32.94 \pm 21.00$ $25.64 \pm 17.56$ Mopping $28.57 \pm 24.48$ $22.86 \pm 23.06$ $27.02 \pm 24.96$ $24.69 \pm 21.27$ Playing videogames $12.57 \pm 10.52$ $10.85 \pm 15.89$ $31.55 \pm 28.34$ $27.73 \pm 30.98$ Scrubbing a surface $16.90 \pm 18.41$ $14.06 \pm 15.91$ $20.88 \pm 26.06$ $14.89 \pm 18.59$ Stacking groceries $20.35 \pm 22.41$ $19.88 \pm 20.07$ $17.52 \pm 16.40$ $13.52 \pm 14.04$ Sweeping $19.15 \pm 19.40$ $22.90 \pm 22.75$ $25.39 \pm 19.23$ $17.86 \pm 19.00$ Typing $30.27 \pm 16.07$ $36.56 \pm 30.16$ $53.72 \pm 34.37$ $50.10 \pm 35.50$ Vacuming $29.33 \pm 25.21$ $24.77 \pm 23.41$ $38.12 \pm 27.47$ $25.90 \pm 23.87$ Walking around block $22.41 \pm 19.53$ $27.07 \pm 22.82$ $18.76 \pm 18.25$ $18.92 \pm 16.26$ Washi	Walking Treadmill 3mph Treadmill 0 hard	$14.00 \pm 12.00$ 25.52 ± 22.72	$19.09 \pm 19.09$ 18.70 ± 22.57	$10.63 \pm 17.78$	$11.00 \pm 12.00$ 12.58 ± 16.20
Interning137.52137.53137.54137.54137.54137.54137.54147.17unknown53.10 $\pm 6.33$ 10.483.8565.60 $\pm 10.02$ 53.97 $\pm 8.88$ Carrying groceries27.40 $\pm 26.54$ 39.80 $\pm 28.13$ 28.01 $\pm 28.14$ 21.34 $\pm 20.68$ Doing dishes28.38 $\pm 17.79$ 29.18 $\pm 21.95$ 50.50 $\pm 23.69$ 28.82 $\pm 25.60$ Gardening19.05 $\pm 22.00$ 19.53 $\pm 22.36$ 27.67 $\pm 30.44$ 16.27 $\pm 22.88$ Ironing32.52 $\pm 19.51$ 42.52 $\pm 24.06$ 40.06 $\pm 31.37$ 37.94 $\pm 29.72$ Making the bed34.31 $\pm 24.03$ 30.08 $\pm 19.99$ 32.94 $\pm 21.00$ 25.64 $\pm 17.56$ Mopping28.57 $\pm 24.48$ 22.86 $\pm 23.06$ 27.02 $\pm 24.96$ 24.69 $\pm 21.27$ Playing videogames12.57 $\pm 10.52$ 10.85 $\pm 15.89$ 31.55 $\pm 28.34$ 27.73 $\pm 30.98$ Scrubbing a surface16.90 $\pm 18.41$ 14.06 $\pm 15.91$ 20.88 $\pm 26.06$ 14.89 $\pm 18.59$ Stacking groceries20.35 $\pm 22.41$ 19.88 $\pm 20.07$ 17.52 $\pm 16.40$ 13.52 $\pm 14.04$ Sweeping19.15 $\pm 19.40$ 22.90 $\pm 22.75$ 25.39 $\pm 19.23$ 17.86 $\pm 19.00$ Typing30.27 $\pm 16.07$ 36.56 $\pm 30.16$ 53.72 $\pm 34.37$ 50.10 $\pm 35.50$ Vacuming29.33 $\pm 25.21$ 24.77 $\pm 23.41$ 38.12 $\pm 27.47$ 25.90 $\pm 23.87$ Walking around block22.41 $\pm 19.53$ 27.07 $\pm 22.82$ 18.76 $\pm 18.25$ 18.92 $\pm 16.26$ Washing windows26.70 $\pm 20.89$ 15.69 $\pm 14.76$ 30.23 $\pm 26.31$ 24.71 $\pm 21.09$ <td>kneeling</td> <td><math>15.96 \pm 13.32</math></td> <td><math>15.79 \pm 22.57</math> 15.30 + 11.44</td> <td><math>73.36 \pm 35.83</math></td> <td><math>12.30 \pm 10.20</math> 65 70 + 42 25</td>	kneeling	$15.96 \pm 13.32$	$15.79 \pm 22.57$ 15.30 + 11.44	$73.36 \pm 35.83$	$12.30 \pm 10.20$ 65 70 + 42 25
Inition $35.10 \pm 0.53$ $10.40 \pm 1.05$ $30.00 \pm 10.02$ $35.71 \pm 0.03$ Carrying groceries $27.40 \pm 26.54$ $39.80 \pm 28.13$ $28.01 \pm 21.14$ $21.34 \pm 20.68$ Doing dishes $28.38 \pm 17.79$ $29.18 \pm 21.95$ $50.50 \pm 23.69$ $28.82 \pm 25.60$ Gardening $19.05 \pm 22.00$ $19.53 \pm 22.36$ $27.67 \pm 30.44$ $16.27 \pm 22.88$ Ironing $22.52 \pm 19.51$ $42.52 \pm 24.06$ $40.06 \pm 31.37$ $37.94 \pm 29.72$ Making the bed $34.31 \pm 24.03$ $30.08 \pm 19.99$ $32.94 \pm 21.00$ $25.64 \pm 17.56$ Mopping $28.57 \pm 24.48$ $22.86 \pm 23.06$ $27.02 \pm 24.96$ $24.69 \pm 21.27$ Playing videogames $12.57 \pm 10.52$ $10.85 \pm 15.89$ $31.55 \pm 28.34$ $27.73 \pm 30.98$ Scrubbing a surface $16.90 \pm 18.41$ $14.06 \pm 15.91$ $20.88 \pm 26.06$ $14.89 \pm 18.59$ Stacking groceries $20.35 \pm 22.41$ $19.88 \pm 20.07$ $17.52 \pm 16.40$ $13.52 \pm 14.04$ Sweeping $19.15 \pm 19.40$ $22.90 \pm 22.75$ $25.39 \pm 19.23$ $17.86 \pm 19.00$ Typing $30.27 \pm 16.07$ $36.56 \pm 30.16$ $53.72 \pm 34.37$ $50.10 \pm 35.50$ Vacuuming $29.33 \pm 25.21$ $24.77 \pm 23.41$ $38.12 \pm 27.47$ $25.90 \pm 23.87$ Walking around block $22.41 \pm 19.53$ $27.07 \pm 22.82$ $18.76 \pm 18.25$ $18.92 \pm 16.26$ Washing windows $26.70 \pm 20.89$ $15.69 \pm 14.76$ $30.23 \pm 26.31$ $24.71 \pm 21.09$ Watching TV $9.03 \pm 7.11$ $7.30 \pm 6.82$ $23.12 \pm 26.29$ $19.12 \pm 24.91$ Weeding <t< td=""><td>unknown</td><td><math>13.90 \pm 13.32</math> 53.10 + 6.33</td><td><math>10.48 \pm 3.85</math></td><td><math>75.50 \pm 35.03</math></td><td><math>53.07 \pm 8.88</math></td></t<>	unknown	$13.90 \pm 13.32$ 53.10 + 6.33	$10.48 \pm 3.85$	$75.50 \pm 35.03$	$53.07 \pm 8.88$
$\begin{array}{c} 21.30 \pm 20.34 \pm 20.34 \pm 20.34 \pm 20.31 \pm 20.34 \pm 20.34 \pm 20.31 \pm 20.34 \pm 20.36 \pm 20.34 \pm 20.34 \pm 20.36 \pm 20.34 \pm 20.34 \pm 20.36 \pm 20.34 \pm 20.36 \pm 20.34 \pm 20.36 \pm 20.37 \pm 20.37 \pm 20.38 \pm 20.66 \pm 14.39 \pm 18.59 \pm 20.36 \pm 20.37 \pm 20.37 \pm 20.37 \pm 20.37 \pm 20.37 \pm 20.37 \pm 20.37 \pm 20.38 \pm 20.37 \pm 20.37 \pm 20.38 \pm 20.37 \pm 20.38 \pm 20.37 \pm 20.38 \pm 20.37 \pm 20.38 \pm 20.36 \pm$	Carrying groceries	$33.10 \pm 0.33$	$39.80 \pm 28.13$	$28.01 \pm 28.14$	$33.77 \pm 0.00$
Doing disines $20.30 \pm 17.77$ $20.718 \pm 21.73$ $30.30 \pm 25.07$ $28.52 \pm 23.50$ Gardening $19.05 \pm 22.00$ $19.53 \pm 22.36$ $27.67 \pm 30.44$ $16.27 \pm 22.88$ Ironing $32.52 \pm 19.51$ $42.52 \pm 24.06$ $40.06 \pm 31.37$ $37.94 \pm 29.72$ Making the bed $34.31 \pm 24.03$ $30.08 \pm 19.99$ $32.94 \pm 21.00$ $25.64 \pm 17.56$ Mopping $28.57 \pm 24.48$ $22.86 \pm 23.06$ $27.02 \pm 24.96$ $24.69 \pm 21.27$ Playing videogames $12.57 \pm 10.52$ $10.85 \pm 15.89$ $31.55 \pm 28.34$ $27.73 \pm 30.98$ Scrubbing a surface $16.90 \pm 18.41$ $14.06 \pm 15.91$ $20.88 \pm 26.06$ $14.89 \pm 18.59$ Stacking groceries $20.35 \pm 22.41$ $19.88 \pm 20.07$ $17.52 \pm 16.40$ $13.52 \pm 14.04$ Sweeping $19.15 \pm 19.40$ $22.90 \pm 22.75$ $25.39 \pm 19.23$ $17.86 \pm 19.00$ Typing $30.27 \pm 16.07$ $36.56 \pm 30.16$ $53.72 \pm 34.37$ $50.10 \pm 35.50$ Vacuuming $29.33 \pm 25.21$ $24.77 \pm 23.41$ $38.12 \pm 27.47$ $25.90 \pm 23.87$ Walking around block $22.41 \pm 19.53$ $27.07 \pm 22.82$ $18.76 \pm 18.25$ $18.92 \pm 16.26$ Washing windows $26.70 \pm 20.89$ $15.69 \pm 14.76$ $30.23 \pm 26.31$ $24.71 \pm 21.09$ Watching TV $9.03 \pm 7.11$ $7.30 \pm 6.82$ $23.12 \pm 26.29$ $19.12 \pm 24.91$ Weeding $7.69 \pm 11.34$ $9.18 \pm 14.05$ $5.92 \pm 8.06$ $5.95 \pm 10.07$ Witing $26.12 \pm 19.21$ $23.20 \pm 19.28$ $33.86 \pm 27.83$ $23.66 \pm 21.09$ Witting $24.01 \pm 1$	Doing dishes	$27.40 \pm 20.34$ 28.38 + 17.79	$29.18 \pm 21.05$	$20.01 \pm 20.14$ 50 50 + 23 69	$21.34 \pm 20.00$ 28.82 + 25.60
Gateling17.53 $\pm$ 22.3017.53 $\pm$ 22.3017.53 $\pm$ 22.3016.27 $\pm$ 22.406Ironing $32.52 \pm 19.51$ $42.52 \pm 24.06$ $40.06 \pm 31.37$ $37.94 \pm 29.72$ Making the bed $34.31 \pm 24.03$ $30.08 \pm 19.99$ $32.94 \pm 21.00$ $25.64 \pm 17.56$ Mopping $28.57 \pm 24.48$ $22.86 \pm 23.06$ $27.02 \pm 24.96$ $24.69 \pm 21.27$ Playing videogames $12.57 \pm 10.52$ $10.85 \pm 15.89$ $31.55 \pm 28.34$ $27.73 \pm 30.98$ Scrubbing a surface $16.90 \pm 18.41$ $14.06 \pm 15.91$ $20.88 \pm 26.06$ $14.89 \pm 18.59$ Stacking groceries $20.35 \pm 22.41$ $19.88 \pm 20.07$ $17.52 \pm 16.40$ $13.52 \pm 14.04$ Sweeping $19.15 \pm 19.40$ $22.90 \pm 22.75$ $25.39 \pm 19.23$ $17.86 \pm 19.00$ Typing $30.27 \pm 16.07$ $36.56 \pm 30.16$ $53.72 \pm 34.37$ $50.10 \pm 35.50$ Vacuuming $29.33 \pm 25.21$ $24.77 \pm 23.41$ $38.12 \pm 27.47$ $25.90 \pm 23.87$ Walking around block $22.41 \pm 19.53$ $27.07 \pm 22.82$ $18.76 \pm 18.25$ $18.92 \pm 16.26$ Washing windows $26.70 \pm 20.89$ $15.69 \pm 14.76$ $30.23 \pm 26.31$ $24.71 \pm 21.09$ Watching TV $9.03 \pm 7.11$ $7.30 \pm 6.82$ $23.12 \pm 26.29$ $19.12 \pm 24.91$ Weeding $7.69 \pm 11.34$ $9.18 \pm 14.05$ $5.92 \pm 8.06$ $5.95 \pm 10.07$ Witing $24.01 \pm 13.37$ $44.62 \pm 27.94$ $47.73 \pm 33.54$ $51.55 \pm 38.41$ Upting/Dusting $24.01 \pm 13.37$ $44.62 \pm 27.94$ $47.73 \pm 33.54$ $51.55 \pm 38.41$	Gardening	$19.05 \pm 22.00$	$19.53 \pm 21.95$	$27.67 \pm 20.07$	$16.27 \pm 23.00$
Initing $32.32 \pm 17.31$ $42.52 \pm 24.00$ $30.00 \pm 13.37$ $37.4 \pm 27.72$ Making the bed $34.31 \pm 24.03$ $30.08 \pm 19.99$ $32.94 \pm 21.00$ $25.64 \pm 17.56$ Mopping $28.57 \pm 24.48$ $22.86 \pm 23.06$ $27.02 \pm 24.96$ $24.69 \pm 21.27$ Playing videogames $12.57 \pm 10.52$ $10.85 \pm 15.89$ $31.55 \pm 28.34$ $27.73 \pm 30.98$ Scrubbing a surface $16.90 \pm 18.41$ $14.06 \pm 15.91$ $20.88 \pm 26.06$ $14.89 \pm 18.59$ Stacking groceries $20.35 \pm 22.41$ $19.88 \pm 20.07$ $17.52 \pm 16.40$ $13.52 \pm 14.04$ Sweeping $19.15 \pm 19.40$ $22.90 \pm 22.75$ $25.39 \pm 19.23$ $17.86 \pm 19.00$ Typing $30.27 \pm 16.07$ $36.56 \pm 30.16$ $53.72 \pm 34.37$ $50.10 \pm 35.50$ Vacuuming $29.33 \pm 25.21$ $24.77 \pm 23.41$ $38.12 \pm 27.47$ $25.90 \pm 23.87$ Walking around block $22.41 \pm 19.53$ $27.07 \pm 22.82$ $18.76 \pm 18.25$ $18.92 \pm 16.26$ Washing windows $26.70 \pm 20.89$ $15.69 \pm 14.76$ $30.23 \pm 26.31$ $24.71 \pm 21.09$ Watching TV $9.03 \pm 7.11$ $7.30 \pm 6.82$ $23.12 \pm 26.29$ $19.12 \pm 24.91$ Weeding $7.69 \pm 11.34$ $9.18 \pm 14.05$ $5.92 \pm 8.06$ $5.95 \pm 10.07$ Witing $24.01 \pm 13.37$ $44.62 \pm 27.94$ $47.73 \pm 33.54$ $51.55 \pm 38.41$ Utering $24.01 \pm 13.37$ $44.62 \pm 27.94$ $47.73 \pm 33.54$ $51.55 \pm 38.41$	Ironing	$17.03 \pm 22.00$ 32.52 + 19.51	$17.53 \pm 22.50$ $42.52 \pm 24.06$	$40.06 \pm 31.37$	$10.27 \pm 22.00$ 37.04 + 20.72
Making ine odd $34.31 \pm 24.03$ $36.03 \pm 19.39$ $32.34 \pm 21.00$ $23.04 \pm 17.30$ Mopping $28.57 \pm 24.48$ $22.86 \pm 23.06$ $27.02 \pm 24.96$ $24.69 \pm 21.27$ Playing videogames $12.57 \pm 10.52$ $10.85 \pm 15.89$ $31.55 \pm 28.34$ $27.73 \pm 30.98$ Scrubbing a surface $16.90 \pm 18.41$ $14.06 \pm 15.91$ $20.88 \pm 26.06$ $14.89 \pm 18.59$ Stacking groceries $20.35 \pm 22.41$ $19.88 \pm 20.07$ $17.52 \pm 16.40$ $13.52 \pm 14.04$ Sweeping $19.15 \pm 19.40$ $22.90 \pm 22.75$ $25.39 \pm 19.23$ $17.86 \pm 19.00$ Typing $30.27 \pm 16.07$ $36.56 \pm 30.16$ $53.72 \pm 34.37$ $50.10 \pm 35.50$ Vacuuming $29.33 \pm 25.21$ $24.77 \pm 23.41$ $38.12 \pm 27.47$ $25.90 \pm 23.87$ Walking around block $22.41 \pm 19.53$ $27.07 \pm 22.82$ $18.76 \pm 18.25$ $18.92 \pm 16.26$ Washing windows $26.70 \pm 20.89$ $15.69 \pm 14.76$ $30.23 \pm 26.31$ $24.71 \pm 21.09$ Watching TV $9.03 \pm 7.11$ $7.30 \pm 6.82$ $23.12 \pm 26.29$ $19.12 \pm 24.91$ Weeding $7.69 \pm 11.34$ $9.18 \pm 14.05$ $5.92 \pm 8.06$ $5.95 \pm 10.07$ Witing $24.01 \pm 13.37$ $44.62 \pm 27.94$ $47.73 \pm 33.54$ $51.55 \pm 38.41$ Utring $24.01 \pm 13.37$ $44.62 \pm 27.94$ $47.73 \pm 33.54$ $51.55 \pm 38.41$	Making the hed	$32.32 \pm 17.31$ $34.31 \pm 24.03$	$42.32 \pm 24.00$ $30.08 \pm 10.00$	$40.00 \pm 31.37$ 32.04 ± 21.00	$37.94 \pm 29.12$ 25.64 ± 17.56
Nopping $22.37 \pm 24.43$ $22.30 \pm 23.30$ $24.02 \pm 24.30$ $24.237$ Playing videogames $12.57 \pm 10.52$ $10.85 \pm 15.89$ $31.55 \pm 28.34$ $27.73 \pm 30.98$ Scrubbing a surface $16.90 \pm 18.41$ $14.06 \pm 15.91$ $20.88 \pm 26.06$ $14.89 \pm 18.59$ Stacking groceries $20.35 \pm 22.41$ $19.88 \pm 20.07$ $17.52 \pm 16.40$ $13.52 \pm 14.04$ Sweeping $19.15 \pm 19.40$ $22.90 \pm 22.75$ $25.39 \pm 19.23$ $17.86 \pm 19.00$ Typing $30.27 \pm 16.07$ $36.56 \pm 30.16$ $53.72 \pm 34.37$ $50.10 \pm 35.50$ Vacuuming $29.33 \pm 25.21$ $24.77 \pm 23.41$ $38.12 \pm 27.47$ $25.90 \pm 23.87$ Walking around block $22.41 \pm 19.53$ $27.07 \pm 22.82$ $18.76 \pm 18.25$ $18.92 \pm 16.26$ Washing windows $26.70 \pm 20.89$ $15.69 \pm 14.76$ $30.23 \pm 6.31$ $24.71 \pm 21.09$ Watching TV $9.03 \pm 7.11$ $7.30 \pm 6.82$ $23.12 \pm 26.29$ $19.12 \pm 24.91$ Weeding $7.69 \pm 11.34$ $9.18 \pm 14.05$ $5.92 \pm 8.06$ $5.95 \pm 10.07$ Wiping/Dusting $24.01 \pm 13.37$ $44.62 \pm 27.94$ $47.73 \pm 33.54$ $51.55 \pm 38.41$ Uking out trach $16.71 \pm 15.99$ $11.06 \pm 12.11$ $4.66 \pm 6.29$ $11.52 \pm 11.70$	Monning	$34.51 \pm 24.03$ 28 57 + 24.48	$30.03 \pm 19.99$ 22.86 + 23.06	$32.94 \pm 21.00$ 27.02 + 24.96	$23.04 \pm 17.30$ $24.69 \pm 21.27$
Table Videgames $12.57 \pm 10.52$ $10.53 \pm 10.53 \pm 12.54$ $27.73 \pm 30.78$ Scrubbing a surface $16.90 \pm 18.41$ $14.06 \pm 15.91$ $20.88 \pm 26.06$ $14.89 \pm 18.59$ Stacking groceries $20.35 \pm 22.41$ $19.88 \pm 20.07$ $17.52 \pm 16.40$ $13.52 \pm 14.04$ Sweeping $19.15 \pm 19.40$ $22.90 \pm 22.75$ $25.39 \pm 19.23$ $17.86 \pm 19.00$ Typing $30.27 \pm 16.07$ $36.56 \pm 30.16$ $53.72 \pm 34.37$ $50.10 \pm 35.50$ Vacuuming $29.33 \pm 25.21$ $24.77 \pm 23.41$ $38.12 \pm 27.47$ $25.90 \pm 23.87$ Walking around block $22.41 \pm 19.53$ $27.07 \pm 22.82$ $18.76 \pm 18.92 \pm 16.26$ Washing windows $26.70 \pm 20.89$ $15.69 \pm 14.76$ $30.23 \pm 26.31$ $24.71 \pm 21.09$ Watching TV $9.03 \pm 7.11$ $7.30 \pm 6.82$ $23.12 \pm 26.29$ $19.12 \pm 24.91$ Weeding $7.69 \pm 11.34$ $9.18 \pm 14.05$ $5.92 \pm 8.06$ $5.95 \pm 10.07$ Wiping/Dusting $24.01 \pm 13.37$ $44.62 \pm 27.94$ $47.73 \pm 33.54$ $51.55 \pm 38.41$ Ukring cut trach $16.71 \pm 15.99$ $11.06 \pm 12.11$ $4.66 \pm 6.29$ $11.52 \pm 11.70$	Playing videogames	$12.57 \pm 24.48$ $12.57 \pm 10.52$	$10.85 \pm 15.89$	$21.02 \pm 24.00$ 31.55 + 28.34	$27.73 \pm 30.98$
Stacking groceries $10.90 \pm 10.41$ $14.00 \pm 13.91$ $20.03 \pm 20.00$ $14.39 \pm 18.59$ Stacking groceries $20.35 \pm 22.41$ $19.88 \pm 20.07$ $17.52 \pm 16.40$ $13.52 \pm 14.04$ Sweeping $19.15 \pm 19.40$ $22.90 \pm 22.75$ $25.39 \pm 19.23$ $17.86 \pm 19.00$ Typing $30.27 \pm 16.07$ $36.56 \pm 30.16$ $53.72 \pm 34.37$ $50.10 \pm 35.50$ Vacuuming $29.33 \pm 25.21$ $24.77 \pm 23.41$ $38.12 \pm 27.47$ $25.90 \pm 23.87$ Walking around block $22.41 \pm 19.53$ $27.07 \pm 22.82$ $18.76 \pm 18.25$ $18.92 \pm 16.26$ Washing windows $26.70 \pm 20.89$ $15.69 \pm 14.76$ $30.23 \pm 26.31$ $24.71 \pm 21.09$ Watching TV $9.03 \pm 7.11$ $7.30 \pm 6.82$ $23.12 \pm 66.29$ $19.12 \pm 24.91$ Weeding $7.69 \pm 11.34$ $9.18 \pm 14.05$ $5.92 \pm 8.06$ $5.95 \pm 10.07$ Wiping/Dusting $26.12 \pm 19.21$ $23.20 \pm 19.28$ $33.86 \pm 27.83$ $23.66 \pm 21.09$ Writing $24.01 \pm 13.37$ $44.62 \pm 27.94$ $47.73 \pm 33.54$ $51.55 \pm 38.41$	Scrubbing a surface	$12.57 \pm 10.52$ 16.00 ± 18.41	$10.05 \pm 15.07$ $14.06 \pm 15.01$	$20.88 \pm 26.04$	$1/1.00 \pm 10.00$
Sweeping $19.35 \pm 22.41$ $19.38 \pm 20.07$ $17.32 \pm 10.40$ $13.32 \pm 14.04$ Sweeping $19.15 \pm 19.40$ $22.90 \pm 22.75$ $25.39 \pm 19.23$ $17.86 \pm 19.00$ Typing $30.27 \pm 16.07$ $36.56 \pm 30.16$ $53.72 \pm 34.37$ $50.10 \pm 35.50$ Vacuuming $29.33 \pm 25.21$ $24.77 \pm 23.41$ $38.12 \pm 27.47$ $25.90 \pm 23.87$ Walking around block $22.41 \pm 19.53$ $27.07 \pm 22.82$ $18.76 \pm 18.92 \pm 16.26$ Washing windows $26.70 \pm 20.89$ $15.69 \pm 14.76$ $30.23 \pm 26.31$ $24.71 \pm 21.09$ Watching TV $9.03 \pm 7.11$ $7.30 \pm 6.82$ $23.12 \pm 66.29$ $19.12 \pm 24.91$ Weeding $7.69 \pm 11.34$ $9.18 \pm 14.05$ $5.92 \pm 8.06$ $5.95 \pm 10.07$ Wiping/Dusting $24.01 \pm 13.37$ $44.62 \pm 27.94$ $47.73 \pm 33.54$ $51.55 \pm 38.41$ taking out trach $16.71 \pm 15.99$ $11.06 \pm 12.11$ $4.66 \pm 6.29$ $11.52 \pm 11.70$	Stacking groceries	$10.90 \pm 13.41$ 20.35 ± 22.41	$14.00 \pm 13.91$ 19.88 $\pm 20.07$	$20.83 \pm 20.00$ 17.52 ± 16.40	$14.89 \pm 18.39$ 13.52 ± 14.04
Sweeping19.13 $\pm$ 19.4022.90 $\pm$ 22.1323.33 $\pm$ 19.2317.80 $\pm$ 19.00Typing $30.27 \pm 16.07$ $36.56 \pm 30.16$ $53.72 \pm 34.37$ $50.10 \pm 35.50$ Vacuuming $29.33 \pm 25.21$ $24.77 \pm 23.41$ $38.12 \pm 27.47$ $25.90 \pm 23.87$ Walking around block $22.41 \pm 19.53$ $27.07 \pm 22.82$ $18.76 \pm 18.92 \pm 16.26$ Washing windows $26.70 \pm 20.89$ $15.69 \pm 14.76$ $30.23 \pm 26.31$ $24.71 \pm 21.09$ Watching TV $9.03 \pm 7.11$ $7.30 \pm 6.82$ $23.12 \pm 26.29$ $19.12 \pm 24.91$ Weeding $7.69 \pm 11.34$ $9.18 \pm 14.05$ $5.92 \pm 8.06$ $5.95 \pm 10.07$ Wiping/Dusting $26.12 \pm 19.21$ $23.20 \pm 19.28$ $33.86 \pm 27.83$ $23.66 \pm 21.09$ Writing $24.01 \pm 13.37$ $44.62 \pm 27.94$ $47.73 \pm 33.54$ $51.55 \pm 38.41$ taking out trach $16.71 \pm 15.99$ $11.06 \pm 12.11$ $4.66 \pm 6.29$ $11.52 \pm 11.70$	Sweening	$20.33 \pm 22.41$ 10.15 ± 10.40	$19.88 \pm 20.07$ 22.90 ± 22.75	$17.32 \pm 10.40$ 25.30 ± 10.23	$13.32 \pm 14.04$ 17.86 ± 10.00
Typing $36.27 \pm 10.07$ $36.30 \pm 36.16$ $35.72 \pm 34.37$ $30.10 \pm 35.30$ Vacuuming $29.33 \pm 25.21$ $24.77 \pm 23.41$ $38.12 \pm 27.47$ $25.90 \pm 23.87$ Walking around block $22.41 \pm 19.53$ $27.07 \pm 22.82$ $18.76 \pm 18.25$ $18.92 \pm 16.26$ Washing windows $26.70 \pm 20.89$ $15.69 \pm 14.76$ $30.23 \pm 26.31$ $24.71 \pm 21.09$ Watching TV $9.03 \pm 7.11$ $7.30 \pm 6.82$ $23.12 \pm 26.29$ $19.12 \pm 24.91$ Weeding $7.69 \pm 11.34$ $9.18 \pm 14.05$ $5.92 \pm 8.06$ $5.95 \pm 10.07$ Wiping/Dusting $26.12 \pm 19.21$ $23.20 \pm 19.28$ $33.86 \pm 27.83$ $23.66 \pm 21.09$ Writing $24.01 \pm 13.37$ $44.62 \pm 27.94$ $47.73 \pm 33.54$ $51.55 \pm 38.41$ taking out trach $16.71 \pm 15.99$ $11.06 \pm 12.11$ $4.66 \pm 6.29$ $11.52 \pm 11.70$	Tuning	$19.13 \pm 19.40$ 20.27 ± 16.07	$22.90 \pm 22.13$	$23.39 \pm 19.23$ 52.72 ± 24.27	$17.80 \pm 19.00$ 50.10 ± 25.50
Vacuuming $23.33 \pm 23.21$ $24.77 \pm 23.41$ $36.12 \pm 27.47$ $25.96 \pm 23.87$ Walking around block $22.41 \pm 19.53$ $27.07 \pm 22.82$ $18.76 \pm 18.25$ $18.92 \pm 16.26$ Washing windows $26.70 \pm 20.89$ $15.69 \pm 14.76$ $30.23 \pm 26.31$ $24.71 \pm 21.09$ Watching TV $9.03 \pm 7.11$ $7.30 \pm 6.82$ $23.12 \pm 26.29$ $19.12 \pm 24.91$ Weeding $7.69 \pm 11.34$ $9.18 \pm 14.05$ $5.92 \pm 8.06$ $5.95 \pm 10.07$ Wiping/Dusting $26.12 \pm 19.21$ $23.20 \pm 19.28$ $33.86 \pm 27.83$ $23.66 \pm 21.09$ Writing $24.01 \pm 13.37$ $44.62 \pm 27.94$ $47.73 \pm 33.54$ $51.55 \pm 38.41$ taking out trach $16.71 \pm 15.99$ $11.06 \pm 12.11$ $4.66 \pm 6.29$ $11.52 \pm 11.70$	Vacuuming	$30.27 \pm 10.07$ 20.33 $\pm 25.21$	$30.30 \pm 30.10$ $24.77 \pm 23.41$	$33.72 \pm 34.37$ 38.12 ± 27.47	$30.10 \pm 33.30$ 25.00 ± 23.87
Warking around block $22.41 \pm 19.33$ $21.07 \pm 22.32$ $16.70 \pm 10.23$ $16.79 \pm 10.20$ Washing windows $26.70 \pm 20.89$ $15.69 \pm 14.76$ $30.23 \pm 26.31$ $24.71 \pm 21.09$ Watching TV $9.03 \pm 7.11$ $7.30 \pm 6.82$ $23.12 \pm 26.29$ $19.12 \pm 24.91$ Weeding $7.69 \pm 11.34$ $9.18 \pm 14.05$ $5.92 \pm 8.06$ $5.95 \pm 10.07$ Wiping/Dusting $26.12 \pm 19.21$ $23.20 \pm 19.28$ $33.86 \pm 27.83$ $23.66 \pm 21.09$ Writing $24.01 \pm 13.37$ $44.62 \pm 27.94$ $47.73 \pm 33.54$ $51.55 \pm 38.41$ taking out trach $16.71 \pm 15.99$ $11.06 \pm 12.11$ $4.66 \pm 6.29$ $11.52 \pm 11.70$	Walking around block	$27.33 \pm 23.21$ 22.41 $\pm$ 10.52	$24.77 \pm 23.41$ 27.07 $\pm 22.92$	$18.12 \pm 21.41$	$23.90 \pm 23.07$ 18.02 ± 16.26
Washing windows $20.70 \pm 20.39$ $13.09 \pm 14.76$ $50.23 \pm 20.31$ $24.71 \pm 21.09$ Watching TV $9.03 \pm 7.11$ $7.30 \pm 6.82$ $23.12 \pm 26.29$ $19.12 \pm 24.91$ Weeding $7.69 \pm 11.34$ $9.18 \pm 14.05$ $5.92 \pm 8.06$ $5.95 \pm 10.07$ Wiping/Dusting $26.12 \pm 19.21$ $23.20 \pm 19.28$ $33.86 \pm 27.83$ $23.66 \pm 21.09$ Writing $24.01 \pm 13.37$ $44.62 \pm 27.94$ $47.73 \pm 33.54$ $51.55 \pm 38.41$ taking out trach $16.71 \pm 15.99$ $11.06 \pm 12.11$ $4.66 \pm 6.29$ $11.52 \pm 11.70$	Washing windows	$22.41 \pm 19.33$	$27.07 \pm 22.02$ 15.60 ± 14.76	$16.70 \pm 16.23$ $20.22 \pm 26.21$	$16.92 \pm 10.20$ 24.71 ± 21.00
Watching IV $7.50 \pm 0.32$ $2.12 \pm 20.29$ $19.12 \pm 24.91$ Weeding $7.69 \pm 11.34$ $9.18 \pm 14.05$ $5.92 \pm 8.06$ $5.95 \pm 10.07$ Wiping/Dusting $26.12 \pm 19.21$ $23.20 \pm 19.28$ $33.86 \pm 27.83$ $23.66 \pm 21.09$ Writing $24.01 \pm 13.37$ $44.62 \pm 27.94$ $47.73 \pm 33.54$ $51.55 \pm 38.41$ taking out track $16.71 \pm 15.99$ $11.06 \pm 12.11$ $4.66 \pm 6.29$ $11.52 \pm 11.70$	Watching TV	$20.70 \pm 20.09$ $0.03 \pm 7.11$	$13.07 \pm 14.70$ 7 30 ± 6.82	$30.23 \pm 20.31$ 23.12 $\pm$ 26.20	$24.71 \pm 21.09$ 10.12 $\pm 24.01$
weeding $7.09 \pm 11.34$ $9.16 \pm 14.05$ $3.92 \pm 8.00$ $3.92 \pm 10.07$ Wiping/Dusting $26.12 \pm 19.21$ $23.20 \pm 19.28$ $33.86 \pm 27.83$ $23.66 \pm 21.09$ Writing $24.01 \pm 13.37$ $44.62 \pm 27.94$ $47.73 \pm 33.54$ $51.55 \pm 38.41$ taking out trach $16.71 \pm 15.99$ $11.06 \pm 12.11$ $4.66 \pm 6.29$ $11.52 \pm 11.70$	Weeding	$7.03 \pm 7.11$ $7.60 \pm 11.24$	$7.50 \pm 0.62$ 0.18 $\pm 14.05$	$23.12 \pm 20.29$ 5.02 ± 8.06	$17.12 \pm 24.71$ 5.05 $\pm 10.07$
wrping busing $25.20 \pm 17.21$ $25.20 \pm 17.26$ $35.00 \pm 27.05$ $25.00 \pm 27.05$ Writing $24.01 \pm 13.37$ $44.62 \pm 27.94$ $47.73 \pm 33.54$ $51.55 \pm 38.41$ taking out trach $16.71 \pm 15.90$ $11.06 \pm 12.11$ $4.66 \pm 6.20$ $11.52 \pm 11.70$	Wining/Dusting	$7.07 \pm 11.34$ 26.12 ± 10.21	$7.10 \pm 14.03$ $73.20 \pm 10.20$	$3.32 \pm 0.00$ 33.86 $\pm$ 27.82	$3.35 \pm 10.07$ 23.66 ± 21.00
writing $24.01 \pm 13.57$ $44.02 \pm 27.94$ $47.75 \pm 53.54$ $51.53 \pm 56.41$ taking out trash       16.71 + 15.00       11.06 + 12.11 $A.66 \pm 6.20$ 11.52 \pm 11.70	Writing	$20.12 \pm 19.21$ $24.01 \pm 13.37$	$23.20 \pm 19.20$ 119.20	$33.00 \pm 21.03$ $17.73 \pm 33.54$	$23.00 \pm 21.09$ 51 55 $\pm$ 38 /1
	taking out trash	$1671 \pm 15.37$	$\frac{1106 \pm 21.94}{1106 \pm 1211}$	$\frac{1}{4} \frac{1}{66} + \frac{5}{6} \frac{3}{20}$	$11.55 \pm 50.41$ 11.52 + 11.70

Table A13-3: F-Measure obtained when training different classifiers using the *MaxAcceleration* feature set computed per sensor in a subject independent manner.

**Appendix A14: Description of the 52 Activities Contained in the MIT Dataset** 

Activity	Intensity Level	Collection Place	Description
Bench weight lifting - Light		Gym	Sitting leaning back lifting 0.9Kg (2Lb) weight with both hands
Bench weight lifting – Moderate	V	Gym	Sitting leaning back lifting 3.1Kg (7Lb) weight with both hands
Bench weight lifting – Hard	V	Gym	Sitting leaning back lifting 7.7Kg (17Lb) weight with both hands
Bicep curls – Light	V	Gvm	Sitting leaning forward with 0.9Kg (2Lb) weight on each hand
Bicep curls – Moderate	v V	Gvm	Sitting leaning forward with 2.2Kg (5Lb) weight on each hand
Bicep curls – Hard	V	Gym	Sitting leaning forward with 3 6Kg (8Lb) weight on each hand
Calisthenics Crunches	-	Gym	Shiring fourning for ward what brong (620) weight on each hand
Calisthenics Sit ups	-	Gym	Sit-ups using body as weight
Cycling Cycle 100rpm (15mph, 120.4W) – Light	V	Gym	Speed 100rmp, resistance of 2
Cycling Cycle 60rpm (8.9mph, 66.9W) – Light	V	Gym	Speed 60rmp, resistance of 2
Cycling Cycle 80rpm (11.2mph, 100.4W) – Light	V	Gvm	Speed 80rpm, resistance of 2
Cycling Cycle 80rpm – Moderate	N	Gym	Speed 80rmp, resistance of 7
Cycling Cycle 80rpm – Hard	V	Gym	Speed 80rmp, resistance of 13
Lying down	-	Gym	L ving down still
Rowing 30spm – Light	V	Gym	Speed 30spm, resistance of 2
Rowing 30spm – Moderate	V	Gym	Speed 30spm, resistance of 5
Rowing 30spm – Hard	N	Gym	Speed 30spm, resistance of 8
Running Treadmill 4mph Treadmill 0	N	Gym	Speed 4mph 0% incline
Running Treadmill 5mph Treadmill 0	2	Gym	Speed 5mph, 0% incline
Running Treadmill 6mph Treadmill 0	2	Gym	Speed 5mph, 0% incline
Sitting	2	Gym	Sitting still hands on thighs
Sitting Fidget feet legs	N	Gym	Sitting still, hands on thighs
Sitting Fidget hands arms	N	Gym	Unstairs 4 floors continuously
Stuffig Fluget hands arms	v	Gym	downstairs 4 floors continuously
Stairs Ascend stairs	-	Gym	downstairs 4 moors continuously
Standing	-	Gym	Standing still
Walking Treadmill 2mph Treadmill 0	2	Gym	Speed 2mph .0% incline
Walking Treadmill 3mph Treadmill 0	2	Gym	Speed 3mph, 0% incline
Walking Treadmill 3mph Treadmill 3 Light	2	Gym	Speed 3mph, 3% incline
Walking Treadmill 3mph Treadmill 6	2	Gym	Speed 3mph, 5% incline
Moderate	v	Gym	Speed Shiph, 0% herine
Walking Treadmill 3mph Treadmill 9 – Hard	V	Gvm	Speed 3mph, 9% incline
Kneeling	-	PlaceLab	Kneeling still
Carrying groceries	V	PlaceLab	Walking normal speed with one 3Kg bag on each hand
Doing dishes	-	PlaceLab	Doing dishes while standing
Gardening	-	PlaceLab	Planting seeds with small shovel on the floor
Ironing	-	PlaceLab	Ironing while standing using a handheld iron
Making the bed	-	PlaceLab	
Mopping	-	PlaceLab	Mopping using a wet mop
Playing videogames	-	PlaceLab	Handheld Nintendo DS
Scrubbing a surface	-	PlaceLab	Scrubbing bathtub
Stacking groceries	-	PlaceLab	From bag on floor to shelf
Sweeping	-	PlaceLab	
Typing	-	PlaceLab	Typing on a computer while sitting
Vacuuming	-	PlaceLab	
Walking around block	-	PlaceLab	Normal walking speed
Washing windows	-	PlaceLab	Standing washing glass door
Watching TV	-	PlaceLab	Watching and changing channel
Weeding	-	PlaceLab	Using bare hands
Wiping/Dusting	-	PlaceLab	Using Clorox wipes
Writing	-	PlaceLab	17 1.1
1 aking out trash	-	PlaceLab	IKg trash bag
UNKNOWN		Both	Any unlabeled time period during the data collections

 Table A14-1: Description of the 52 activities contained in the MIT dataset.

**Appendix A15: Feature Computation per Sensor vs. Feature Computation per Axis** 

True positive rate Using Subject Dependent Evaluation					
Activity	NB	NB	C4.5	C4.5	
	Per Sensor	Per Axis	Per Sensor	Per Axis	
Bench weight lifting - hard	$71.7\pm19.8$	$79.2\pm20.8$	$93.1\pm9.1$	$93.6\pm8.2$	
Bench weight lifting - light	$87.7 \pm 11.8$	$91.9\pm6.1$	$93.3\pm10.1$	$97.0\pm4.3$	
Bench weight lifting - moderate	$76.9 \pm 24.2$	$77.9 \pm 23.7$	$92.1 \pm 10.3$	$93.9 \pm 10.2$	
Bicep curls - hard	$93.4 \pm 6.5$	$91.7 \pm 9.3$	$92.8 \pm 5.3$	$93.2 \pm 11.7$	
Bicep curls - light	$86.6 \pm 16.4$	$90.4 \pm 7.3$	$96.4 \pm 4.9$	$92.9\pm8.8$	
Bicep curls - moderate	$83.9 \pm 6.0$	$87.9 \pm 8.3$	$91.5 \pm 6.4$	$92.8 \pm 10.9$	
Calisthenics - Crunches	$97.9 \pm 3.0$	$96.4 \pm 3.8$	$94.2 \pm 2.7$	$95.8 \pm 3.3$	
Calisthenics - Sit ups	97.0 ± 2.3	$97.2 \pm 2.5$	$92.4 \pm 6.1$	$95.4 \pm 5.2$	
Cycling - Cycle hard - Cycle 80rpm	81.7 ± 17.6	86.2 ± 17.7	$91.6 \pm 8.8$	$89.6 \pm 8.6$	
Cycling - Cycle light - Cycle 100rpm	$99.8 \pm 0.8$	$99.9 \pm 0.6$	$97.7 \pm 4.2$	$98.6 \pm 2.0$	
Cycling - Cycle light - Cycle 60rpm	$97.5 \pm 2.6$	$98.7 \pm 1.9$	$97.8 \pm 2.5$	$99.0 \pm 1.5$	
Cycling - Cycle light - Cycle 80rpm	$97.2 \pm 5.0$	$97.6 \pm 3.6$	$94.2 \pm 6.3$	$95.8 \pm 5.4$	
Cycling - Cycle moderate - Cycle 80rpm	90.7 + 6.9	95.8 + 4.7	88.8 + 9.7	92.8 + 6.4	
Lving down	$98.9 \pm 1.8$	$98.0 \pm 1.9$	$100.0 \pm 0.0$	$99.9 \pm 0.3$	
Rowing - Rowing hard - Rowing 30spm	84.2 + 12.2	84.8 + 16.8	84.0 + 13.7	83.4 + 13.4	
Rowing - Rowing light - Rowing 30spm	90.4 + 8.9	$93.4 \pm 6.3$	86.7 + 10.9	91.8 + 7.2	
Rowing - Rowing moderate - Rowing 30spm	85.5 + 11.7	$864 \pm 10.7$	77.0 + 14.5	82.1 + 14.3	
Running - Treadmill 4mph - Treadmill 0	$99.6 \pm 1.1$	992 + 14	984 + 24	$97.0 \pm 3.6$	
Running - Treadmill 5mph - Treadmill 0	98.4 + 2.9	$99.0 \pm 2.7$	$951 \pm 36$	$942 \pm 40$	
Running - Treadmill 6mph - Treadmill 0	89.2 + 15.3	$92.6 \pm 20.6$	887+133	$91.2 \pm 1.0$ 91.3 + 14.4	
Sitting	$92.6 \pm 4.3$	$84.3 \pm 8.8$	$968 \pm 29$	$95.6 \pm 4.4$	
Sitting - Fidget feet legs	$96.2 \pm 4.3$	$97.0 \pm 3.0$	$95.6 \pm 7.6$	$94.4 \pm 6.8$	
Sitting - Fidget hands arms	$96.6 \pm 4.3$	$95.2 \pm 4.8$	$91.8 \pm 1.0$	$935 \pm 51$	
Stairs - Ascend stairs	$95.0 \pm 4.3$	$97.6 \pm 2.0$	$94.0 \pm 4.2$ 89.2 + 7.0	$93.3 \pm 3.1$ $93.3 \pm 4.4$	
Stairs - Descend stairs	$91.6 \pm 5.3$	$967 \pm 32$	$88.7 \pm 9.1$	$92.0 \pm 6.1$	
Standing	$91.0 \pm 5.3$	$90.7 \pm 3.2$ 863 + 102	$95.0 \pm 6.0$	$96.1 \pm 5.9$	
Walking - Treadmill 2mph - Treadmill ()	$97.9 \pm 7.1$	$97.5 \pm 3.2$	$91.0 \pm 0.0$	$97.6 \pm 2.8$	
Walking - Treadmill 3mph - Treadmill 0	$97.9 \pm 2.3$	$91.2 \pm 5.2$	$94.9 \pm 4.4$ 817+95	$97.0 \pm 2.0$ $87.4 \pm 8.0$	
Walking Treadmill 3mph Treadmill 3 light	$83.2 \pm 0.0$	$91.2 \pm 5.4$	$72.8 \pm 16.0$	$87.4 \pm 8.0$	
Walking Treadmill 3mph Treadmill 6 moderate	$83.9 \pm 7.7$	$90.0 \pm 0.8$	$72.8 \pm 10.0$ $74.0 \pm 9.4$	$84.5 \pm 11.1$	
Walking Treadmill 3mph Treadmill 0 hard	85.0 ± 0.5	$03.1 \pm 6.8$	$74.0 \pm 7.4$ 84.1 ± 0.4	$90.3 \pm 6.7$	
kneeling	$03.5 \pm 7.5$	$93.1 \pm 0.0$	$04.1 \pm 7.4$	$96.1 \pm 3.7$	
Carrying groceries	$92.0 \pm 4.3$	$97.1 \pm 4.4$	$90.9 \pm 2.8$	$90.1 \pm 5.7$	
Doing dishes	$91.7 \pm 0.0$ 85.6 ± 6.4	$97.1 \pm 4.4$ $91.4 \pm 5.0$	$90.1 \pm 3.1$ $85.4 \pm 7.4$	$91.8 \pm 3.8$ $87.8 \pm 8.6$	
Gardening	$83.0 \pm 0.4$	$91.4 \pm 3.0$	$0.3.4 \pm 1.4$	$87.8 \pm 8.0$	
Ironing	$84.0 \pm 13.2$	$86.0 \pm 9.0$	$84.9 \pm 10.8$	$84.9 \pm 9.0$	
Making the hed	$61.9 \pm 9.4$	$87.2 \pm 0.0$ 78.7 ± 12.4	$63.1 \pm 7.7$	$67.7 \pm 12.2$	
Manning	$09.9 \pm 10.4$	$76.7 \pm 12.4$ $91.2 \pm 10.2$	$04.9 \pm 11.1$	$07.7 \pm 12.2$ 72.8 ± 12.0	
Mopping Disving video games	$73.4 \pm 12.3$	$01.2 \pm 10.2$	$04.0 \pm 14.0$	$12.8 \pm 13.0$	
Playing videogames	$94.5 \pm 3.3$	$95.2 \pm 3.9$	$96.9 \pm 2.1$	$98.2 \pm 2.2$	
Scrubbing a surface	$81.2 \pm 13.0$	$88.0 \pm 7.4$	$80.5 \pm 12.9$	$88.0 \pm 8.0$	
Stacking groceries	$74.7 \pm 10.4$	$82.1 \pm 9.6$	$69.4 \pm 16.1$	$74.0 \pm 11.6$	
Sweeping	$70.4 \pm 15.1$	$74.6 \pm 12.1$	$64.9 \pm 19.9$	$73.4 \pm 14.0$	
Typing	$95.2 \pm 3.7$	$95.5 \pm 2.8$	$9/.0 \pm 3.9$	$98.4 \pm 2.3$	
vacuuming	$76.9 \pm 8.8$	$80.7 \pm 7.6$	$78.9 \pm 10.8$	$1/.9 \pm 8.2$	
Walking around block	$93.4 \pm 0.2$	$90.1 \pm 2.7$	$90.8 \pm 0.6$	$89.1 \pm /.4$	
washing windows	$1/.2 \pm 13.6$	$\frac{8}{.5 \pm 8.0}$	$04.4 \pm /.8$	$13.8 \pm 9.3$	
watching I V	$91.5 \pm 4.4$	89.6±6./	$9/.2 \pm 4.4$	$97.7 \pm 2.6$	
Weeding	$80.4 \pm 13.1$	82.0 ± 8.7	$81./\pm 13.6$	$80.5 \pm 8.4$	
wiping/Dusting	$62.4 \pm 13.1$	$/5.2 \pm 11.6$	$62.2 \pm 14.9$	$68.4 \pm 14.8$	
writing	$95.2 \pm 4.4$	$94.7 \pm 3.6$	$98.5 \pm 1.8$	$97.2 \pm 2.3$	
taking out trash	$70.6 \pm 10.9$	/6./±8.9	$60.3 \pm 15.7$	$68.8 \pm 11.7$	

Table A15-1: True positive rate obtained using the naïve Bayes (NB) and C4.5 classifiers using the *MaxAccelerationSet1* feature set computed per sensor and per axis during subject dependent evaluation without the *unknown* class.

False positive rate Using Subject dependent Evaluation					
Activity	NB	NB	C4.5	C4.5	
	Per Sensor	Per Axis	Per Sensor	Per Axis	
Bench weight lifting - hard	$0.2 \pm 0.2$	$0.2 \pm 0.2$	$0.1 \pm 0.1$	$0.0\pm0.1$	
Bench weight lifting - light	$0.2 \pm 0.2$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	
Bench weight lifting - moderate	$0.5 \pm 0.5$	$0.1 \pm 0.1$	$0.2 \pm 0.2$	$0.1 \pm 0.1$	
Bicep curls - hard	$0.2 \pm 0.1$	$0.1 \pm 0.1$	$0.2 \pm 0.2$	$0.2 \pm 0.2$	
Bicep curls - light	$0.2 \pm 0.2$	$0.2 \pm 0.2$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	
Bicep curls - moderate	$0.4 \pm 0.6$	$0.2 \pm 0.2$	$0.2 \pm 0.1$	$0.1 \pm 0.2$	
Calisthenics - Crunches	$0.0 \pm 0.0$	$0.0 \pm 0.0$	$0.1 \pm 0.1$	$0.0 \pm 0.1$	
Calisthenics - Sit ups	$0.0 \pm 0.1$	$0.0 \pm 0.0$	$0.1 \pm 0.1$	$0.1 \pm 0.0$	
Cycling - Cycle hard - Cycle 80rpm	$0.2 \pm 0.2$	$0.1 \pm 0.1$	$0.2 \pm 0.2$	$0.1 \pm 0.1$	
Cycling - Cycle light - Cycle 100rpm	0.0 + 0.0	0.0 + 0.0	$0.1 \pm 0.1$	0.0 + 0.0	
Cycling - Cycle light - Cycle 60rpm	$0.0 \pm 0.0$	$0.0 \pm 0.0$	$0.1 \pm 0.1$	$0.0 \pm 0.1$	
Cycling - Cycle light - Cycle 80rpm	$0.0 \pm 0.0$ 0.1 + 0.2	$0.0 \pm 0.0$	$0.2 \pm 0.2$	$0.0 \pm 0.1$	
Cycling - Cycle moderate - Cycle 80rnm	$0.1 \pm 0.2$ 0.3 + 0.2	$0.0 \pm 0.0$ 0.1 + 0.1	$0.2 \pm 0.2$	$0.1 \pm 0.1$	
Lying down	$0.0 \pm 0.2$ 0.0 + 0.1	$0.1 \pm 0.1$	$0.2 \pm 0.2$	$0.2 \pm 0.2$	
Bowing - Rowing hard - Rowing 30spm	$0.0 \pm 0.1$ 0.3 ± 0.2	$0.2 \pm 0.2$	$0.0 \pm 0.0$	$0.0 \pm 0.0$	
Powing Powing light Powing 30cpm	$0.3 \pm 0.2$	$0.2 \pm 0.2$	$0.4 \pm 0.3$	$0.3 \pm 0.2$	
Rowing - Rowing light - Rowing Sospin	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.5 \pm 0.2$	$0.2 \pm 0.2$	
Rowing - Rowing moderate - Rowing Jospin	$0.4 \pm 0.2$	$0.3 \pm 0.3$	$0.5 \pm 0.3$	$0.3 \pm 0.2$	
Running - Treadmill 5mph - Treadmill 0	$0.0 \pm 0.0$	$0.0 \pm 0.0$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	
Running - Treadmin Sinph - Treadmin 0	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	
Running - Treadmill omph - Treadmill 0	$0.0 \pm 0.0$	$0.0 \pm 0.0$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	
Sitting	$0.0 \pm 0.1$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	
Sitting - Fidget feet legs	$0.0 \pm 0.1$	$0.0 \pm 0.1$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	
Sitting - Fidget hands arms	$0.1 \pm 0.2$	$0.0 \pm 0.1$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	
Stairs - Ascend stairs	$0.0 \pm 0.1$	$0.0 \pm 0.0$	$0.2 \pm 0.2$	$0.1 \pm 0.1$	
Stairs - Descend stairs	$0.2 \pm 0.2$	$0.0 \pm 0.1$	$0.3 \pm 0.2$	$0.1 \pm 0.1$	
Standing	$0.0 \pm 0.0$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	
Walking - Treadmill 2mph - Treadmill 0	$0.0 \pm 0.0$	$0.0 \pm 0.0$	$0.2 \pm 0.2$	$0.1 \pm 0.1$	
Walking - Treadmill 3mph - Treadmill 0	$0.3 \pm 0.1$	$0.1 \pm 0.1$	$0.5 \pm 0.4$	$0.2 \pm 0.2$	
Walking - Treadmill 3mph - Treadmill 3 - light	$0.5 \pm 0.3$	$0.3 \pm 0.2$	$0.8 \pm 0.4$	$0.4 \pm 0.2$	
Walking - Treadmill 3mph - Treadmill 6 - moderate	$0.7 \pm 0.5$	$0.2 \pm 0.2$	$0.8 \pm 0.4$	$0.4 \pm 0.2$	
Walking - Treadmill 3mph - Treadmill 9 - hard	$0.3 \pm 0.2$	$0.2 \pm 0.1$	$0.5 \pm 0.4$	$0.2 \pm 0.2$	
kneeling	$0.0 \pm 0.1$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.0 \pm 0.1$	
Carrying groceries	$0.2 \pm 0.2$	$0.1 \pm 0.1$	$0.3 \pm 0.2$	$0.2 \pm 0.2$	
Doing dishes	$0.5 \pm 0.4$	$0.2 \pm 0.2$	$0.6 \pm 0.5$	$0.3 \pm 0.2$	
Gardening	$0.9\pm0.9$	$0.5 \pm 0.6$	$0.5 \pm 0.6$	$0.3 \pm 0.2$	
Ironing	$0.3 \pm 0.2$	$0.2 \pm 0.2$	$0.5 \pm 0.2$	$0.4 \pm 0.2$	
Making the bed	$1.1 \pm 0.4$	$0.8 \pm 0.3$	$1.2 \pm 0.7$	$0.8\pm0.4$	
Mopping	$0.6 \pm 0.5$	$0.5 \pm 0.4$	$1.2 \pm 0.7$	$0.7\pm0.3$	
Playing videogames	$0.1 \pm 0.2$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.0\pm0.0$	
Scrubbing a surface	$0.9 \pm 0.7$	$0.5 \pm 0.5$	$0.5 \pm 0.4$	$0.3 \pm 0.3$	
Stacking groceries	$0.6 \pm 0.7$	$0.2 \pm 0.2$	$0.8 \pm 0.7$	$0.5 \pm 0.3$	
Sweeping	$0.9\pm0.8$	$0.6 \pm 0.4$	$0.9 \pm 0.6$	$0.7 \pm 0.3$	
Typing	$0.1 \pm 0.2$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	
Vacuuming	$0.3 \pm 0.2$	$0.1 \pm 0.1$	$0.6 \pm 0.4$	$0.5 \pm 0.2$	
Walking around block	$0.3 \pm 0.2$	$0.1 \pm 0.1$	$0.3 \pm 0.3$	$0.2 \pm 0.1$	
Washing windows	$1.9 \pm 1.0$	$1.1 \pm 0.6$	$1.0 \pm 0.3$	$0.6 \pm 0.3$	
Watching TV	$0.2 \pm 0.3$	$0.2 \pm 0.1$	$0.0 \pm 0.1$	$0.1 \pm 0.0$	
Weeding	$0.9 \pm 0.6$	$0.6 \pm 0.4$	$0.5 \pm 0.4$	$0.4 \pm 0.3$	
Wiping/Dusting	$0.8 \pm 0.4$	$0.4 \pm 0.4$	$1.0 \pm 0.4$	$0.7 \pm 0.4$	
Writing	$0.1 \pm 0.2$	$0.0 \pm 0.1$	$0.1 \pm 0.1$	$0.1 \pm 0.1$	
taking out trash	$1.4 \pm 0.7$	0.8 + 0.4	$1.2 \pm 0.4$	$0.7 \pm 0.4$	

Table A15-2: False positive rate obtained using the naïve Bayes (NB) and C4.5 classifiers using the *MaxAccelerationSet1* feature set computed per sensor and per axis during subject dependent evaluation without the *unknown* class.

True positive rate Using Subject Independent Evaluation					
Activity	NB	NB	C4.5	C4.5	
	Per Sensor	Per Axis	Per Sensor	Per Axis	
Bench weight lifting - hard	$9.2 \pm 17.7$	$11.3 \pm 18.4$	$4.2 \pm 9.4$	$13.1 \pm 25.8$	
Bench weight lifting - light	$35.4 \pm 28.4$	$24.4 \pm 19.2$	$33.6 \pm 32.4$	$41.5 \pm 35.1$	
Bench weight lifting - moderate	$15.8 \pm 22.7$	59.1 ± 32.2	$11.6 \pm 20.5$	$30.2 \pm 35.7$	
Bicep curls - hard	$56.0 \pm 40.1$	$15.4 \pm 18.6$	$32.1 \pm 40.2$	$35.1 \pm 36.9$	
Bicep curls - light	$13.6 \pm 15.3$	$15.8 \pm 17.6$	$36.3 \pm 39.0$	$50.7 \pm 43.4$	
Bicep curls - moderate	$11.6 \pm 13.8$	$73.7 \pm 25.4$	$8.5 \pm 21.9$	$30.5 \pm 36.2$	
Calisthenics - Crunches	$22.9 \pm 37.1$	$72.0 \pm 37.9$	$20.1 \pm 35.2$	$72.7 \pm 40.9$	
Calisthenics - Sit ups	$68.0 \pm 44.8$	$86.8 \pm 30.2$	$44.7 \pm 41.0$	$74.3 \pm 32.7$	
Cycling - Cycle hard - Cycle 80rpm	$41.5 \pm 36.6$	$34.8 \pm 29.3$	$8.5 \pm 7.9$	$35.0 \pm 36.7$	
Cycling - Cycle light - Cycle 100rpm	88.7 ± 22.9	$97.0 \pm 10.5$	$95.7 \pm 8.0$	$98.8 \pm 4.5$	
Cycling - Cycle light - Cycle 60rpm	$79.3 \pm 31.6$	$93.8 \pm 20.1$	$85.7 \pm 27.0$	$93.4 \pm 15.5$	
Cycling - Cycle light - Cycle 80rpm	$38.2 \pm 34.1$	$64.4 \pm 39.2$	$58.5 \pm 37.6$	$38.9 \pm 37.0$	
Cycling - Cycle moderate - Cycle 80rpm	9.2 + 9.4	29.0 + 24.0	30.3 + 30.6	32.4 + 26.4	
Lying down	95.9 + 8.0	$94.8 \pm 10.7$	84.5 + 27.0	92.0 + 18.5	
Rowing - Rowing hard - Rowing 30spm	66.6 + 33.2	$55.6 \pm 40.3$	27.0 + 27.9	27.8 + 32.2	
Rowing - Rowing light - Rowing 30spm	41.7 + 34.6	49.2 + 33.8	41.1 + 36.3	43.8 + 35.4	
Rowing - Rowing moderate - Rowing 30spm	19.1 + 32.7	$16.0 \pm 20.4$	24.3 + 22.6	29.0 + 34.0	
Running - Treadmill 4mph - Treadmill 0	44.2 + 46.8	$72.6 \pm 39.5$	$42.7 \pm 40.6$	$61.0 \pm 35.5$	
Running - Treadmill 5mph - Treadmill 0	$67.5 \pm 39.4$	662 + 397	$77.7 \pm 25.9$	$67.0 \pm 35.5$ $62.4 \pm 38.0$	
Running - Treadmill 6mph - Treadmill 0	$72.6 \pm 43.0$	$87.2 \pm 37.1$	$61.2 \pm 37.3$	$72.4 \pm 30.0$	
Sitting	$21.0 \pm 43.0$	$30.9 \pm 9.4$	$43.0 \pm 40.6$	$72.4 \pm 34.1$ 73.6 ± 38.1	
Sitting - Fidget feet legs	$60.2 \pm 17.3$	$30.7 \pm 7.4$ 88.2 + 19.9	$39.1 \pm 45.0$	$75.0 \pm 30.1$	
Sitting Fidget hands arms	$51.1 \pm 43.1$	$70.4 \pm 27.5$	$37.1 \pm 43.4$ $37.0 \pm 30.1$	$60.3 \pm 40.5$	
Stairs - Ascend stairs	$51.1 \pm 43.1$ $63.9 \pm 38.6$	$19.4 \pm 27.3$ 83 3 + 23 8	$57.9 \pm 39.1$	$67.0 \pm 27.6$	
Stairs - Descend stairs	$57.0 \pm 34.0$	$78.9 \pm 25.0$	$37.2 \pm 35.4$ $44.4 \pm 26.7$	$69.1 \pm 24.0$	
Standing	$11.3 \pm 12.2$	$76.9 \pm 23.4$ 84.9 + 12.1	$\frac{44.4 \pm 20.7}{82.4 \pm 34.0}$	$90.1 \pm 24.7$	
Walking - Treadmill 2mph - Treadmill 0	$79.9 \pm 30.2$	$91.9 \pm 12.1$	$60.2 \pm 39.6$	$50.1 \pm 22.9$	
Walking - Treadmill 3mph - Treadmill 0	$13.7 \pm 30.2$	$15.9 \pm 12.3$	$16.8 \pm 20.3$	$18.4 \pm 21.9$	
Walking - Treadmill 3mph - Treadmill 3 - light	$19.7 \pm 36.3$	$43.9 \pm 37.9$	$10.3 \pm 20.3$ 10.7 + 12.1	$10.4 \pm 21.9$ 23.2 + 23.6	
Walking - Treadmill 3mph - Treadmill 6 - moderate	$10.7 \pm 20.1$ 31.6 + 35.2	$20.0 \pm 20.4$ 38 4 + 25 4	$13.9 \pm 12.1$	$31.2 \pm 29.0$	
Walking - Treadmill 3mph - Treadmill 9 - hard	$23.3 \pm 31.8$	$64.6 \pm 31.8$	$19.7 \pm 15.9$	$31.2 \pm 29.7$ 22.2 + 28.5	
kneeling	$16.1 \pm 9.0$	$56.1 \pm 22.7$	$17.7 \pm 25.0$ 65.2 + 44.3	$95.3 \pm 7.4$	
Carrying groceries	$10.4 \pm 7.0$	$73.3 \pm 27.6$	$03.2 \pm 44.3$ 23.2 + 19.6	$565 \pm 7.4$	
Doing dishes	$41.9 \pm 27.3$	$75.5 \pm 21.0$	$42.7 \pm 31.0$	$56.1 \pm 20.0$	
Gardening	$40.4 \pm 23.9$	$41.6 \pm 34.4$	$\frac{160 \pm 220}{160 \pm 220}$	$30.1 \pm 30.3$	
Ironing	$53.6 \pm 33.3$	$72.4 \pm 16.9$	$10.9 \pm 22.0$ 54.0 ± 31.8	$20.0 \pm 24.2$	
Making the bed	$07.0 \pm 27.4$ 31.6 ± 20.7	$72.4 \pm 10.9$	$34.9 \pm 31.0$ $37.8 \pm 22.5$	$46.6 \pm 21.4$	
Making the bed	$31.0 \pm 20.7$	$39.2 \pm 19.9$	$37.0 \pm 22.3$	$40.0 \pm 21.4$ 26.7 ± 19.9	
Playing videogames	$33.0 \pm 21.4$ 123 ± 104	$40.4 \pm 22.4$ $30.0 \pm 25.2$	$20.3 \pm 23.7$ $33.4 \pm 36.1$	$30.7 \pm 10.0$ 77.8 ± 30.0	
Flaying videogames	$12.3 \pm 19.4$	$30.0 \pm 23.2$ 54.5 ± 20.2	$33.4 \pm 30.1$	$77.8 \pm 30.9$	
Scrubbling a surface	$19.3 \pm 21.0$	$34.3 \pm 30.2$	$20.1 \pm 22.7$	$41.3 \pm 30.7$	
Stacking glocenes	$21.3 \pm 23.0$	$50.1 \pm 27.1$	$23.4 \pm 23.1$	$37.3 \pm 10.8$	
Turing	$41.3 \pm 26.0$	$37.7 \pm 14.9$	$21.1 \pm 22.0$	$41.2 \pm 17.4$	
Vacuuming	$30.7 \pm 31.2$	$47.0 \pm 30.0$	$30.2 \pm 30.4$	$04.0 \pm 29.4$	
Vacuuning Wellting ground block	$40.3 \pm 30.0$	$0/.0 \pm 21.4$	$37.4 \pm 31.1$	$30.0 \pm 20.3$	
Washing windows	$28.3 \pm 20.7$	$33.3 \pm 20.3$	$20.2 \pm 21.9$	$43.1 \pm 23.2$	
Washing Windows	$1/.5 \pm 18.1$	$50.1 \pm 26.8$	$23.8 \pm 21.8$	$39.6 \pm 21.5$	
Wading	$3.2 \pm 4.8$	$14.3 \pm 0.8$	$23.0 \pm 29.3$	$00.1 \pm 37.9$	
Weeding	$12.9 \pm 20.2$	$23.3 \pm 21.5$	$1/.8 \pm 2/.0$	$23.0 \pm 27.8$	
wiping/Dusting	$21.3 \pm 15.0$	$38.9 \pm 20.4$	$34.9 \pm 18.7$	$45.4 \pm 19.6$	
withing	$30.9 \pm 30.8$	$\frac{61.7 \pm 25.4}{22.5 \pm 20.2}$	$33.0 \pm 39.5$	$65.5 \pm 29.2$	
Taking our mash	1 119+129	1 - 1/2 + 700	1 199 + 164	1 / (9 + 1) 4	

Table A15-3: True positive rate obtained using the naïve Bayes (NB) and C4.5 classifiers using the *MaxAccelerationSet1* feature set computed per sensor and per axis during subject independent evaluation without the *unknown* class.

False positive rate Using Subject independent Evaluation					
Activity	NB	NB	C4.5	C4.5	
	Per Sensor	Per Axis	Per Sensor	Per Axis	
Bench weight lifting - hard	$0.3 \pm 0.3$	$0.3 \pm 0.3$	$0.8 \pm 0.7$	$0.6 \pm 0.7$	
Bench weight lifting - light	$1.1 \pm 1.2$	$0.4 \pm 0.4$	$1.4 \pm 1.5$	$1.2 \pm 1.3$	
Bench weight lifting - moderate	$0.8 \pm 0.8$	$1.2 \pm 0.7$	$0.8 \pm 0.7$	$1.1 \pm 1.2$	
Bicep curls - hard	$3.0 \pm 1.8$	$0.3 \pm 0.4$	$1.8 \pm 1.6$	$1.4 \pm 1.3$	
Bicep curls - light	$0.6 \pm 0.8$	$0.4 \pm 0.4$	$1.5 \pm 1.8$	$1.6 \pm 1.4$	
Bicep curls - moderate	$0.8 \pm 0.7$	$2.5 \pm 0.9$	$0.8 \pm 0.9$	$1.2 \pm 1.0$	
Calisthenics - Crunches	$0.5 \pm 0.5$	$0.2 \pm 0.2$	$0.3 \pm 0.3$	$0.4 \pm 1.0$	
Calisthenics - Sit ups	$0.3 \pm 0.8$	$0.1 \pm 0.3$	$0.2 \pm 0.4$	$0.2 \pm 0.4$	
Cycling - Cycle hard - Cycle 80rpm	2.7 + 1.7	1.0 + 1.2	$1.0 \pm 0.8$	1.4 + 1.3	
Cycling - Cycle light - Cycle 100rpm	0.6 + 1.4	$0.2 \pm 0.6$	$0.0 \pm 0.1$	$0.0 \pm 0.1$	
Cycling - Cycle light - Cycle 60rpm	$0.2 \pm 0.4$	$0.0 \pm 0.2$	$0.3 \pm 0.3$	$0.1 \pm 0.2$	
Cycling - Cycle light - Cycle 80rpm	$0.2 \pm 0.1$	$1.1 \pm 1.0$	$2.1 \pm 2.0$	1.4 + 1.4	
Cycling - Cycle moderate - Cycle 80rpm	$0.7 \pm 0.9$	$1.1 \pm 1.1$	$1.8 \pm 1.7$	$1.1 \pm 0.8$	
Lying down	$5.9 \pm 2.6$	$4.1 \pm 1.6$	$1.9 \pm 2.5$	$0.0 \pm 0.0$	
Rowing - Rowing hard - Rowing 30spm	$3.2 \pm 2.0$ $3.2 \pm 1.5$	$1.1 \pm 1.0$ $1.8 \pm 1.2$	31+70	11 + 13	
Rowing - Rowing light - Rowing 30spm	$12 \pm 1.5$	$1.0 \pm 1.2$ $1.4 \pm 1.3$	$19 \pm 1.0$	$1.1 \pm 1.3$ 1 8 + 1 2	
Rowing - Rowing moderate - Rowing 30spm	$1.2 \pm 1.3$ 0.8 + 1.3	$1.4 \pm 1.3$ 0.5 ± 0.7	$1.9 \pm 1.4$ 1 2 + 1 1	$1.0 \pm 1.2$ 1.2 + 1.1	
Running - Treadmill (mph - Treadmill ()	$0.0 \pm 1.5$ 0.8 ± 1.5	$0.9 \pm 0.7$	$1.2 \pm 1.1$ 1.6 + 3.8	$1.2 \pm 1.1$ 0.6 ± 0.8	
Running - Treadmill 5mph - Treadmill 0	$0.0 \pm 1.3$	$0.0 \pm 1.1$	$1.0 \pm 3.0$ $1.3 \pm 1.0$	$0.0 \pm 0.0$	
Running - Treadmill 6mph - Treadmill 0	$1.5 \pm 1.5$ $1.1 \pm 1.4$	$0.7 \pm 0.8$	$1.5 \pm 1.0$	$0.3 \pm 0.9$	
Sitting	$1.1 \pm 1.4$ $1.2 \pm 0.9$	$0.0 \pm 1.0$	$0.3 \pm 0.8$ 11 ± 1.8	$0.5 \pm 0.0$	
Sitting Fidget feet lags	$1.2 \pm 0.9$	$0.0 \pm 0.3$	$1.1 \pm 1.0$ 0.2 ± 0.2	$0.9 \pm 1.2$	
Sitting - Fidget hands arms	$0.3 \pm 0.3$	$0.3 \pm 0.3$	$0.3 \pm 0.2$	$0.2 \pm 0.3$	
Stulig - Fluget fiands arms	$0.8 \pm 0.8$	$0.7 \pm 0.7$	$0.0 \pm 0.0$	$0.4 \pm 0.3$	
Stairs - Ascenu stairs	$0.0 \pm 0.0$	$0.3 \pm 0.4$	$0.8 \pm 0.3$	$0.4 \pm 0.3$	
Staris - Descend stars	$1.0 \pm 0.9$	$0.0 \pm 0.3$	$1.2 \pm 0.8$	$0.0 \pm 0.4$	
Waltring Treadmill 2mmh Treadmill 0	$0.0 \pm 0.0$	$0.0 \pm 0.1$	$0.2 \pm 0.0$	$0.3 \pm 0.3$	
Walking - Treadmill 2mph - Treadmill 0	$0.8 \pm 2.0$	$0.4 \pm 1.3$	$1.0 \pm 1.5$	$0.9 \pm 1.7$	
waiking - Treadmill Smpn - Treadmill 0	$3.0 \pm 2.3$	$1.5 \pm 1.4$	$1.4 \pm 1.4$	$1.5 \pm 1.4$	
Walking - Treadmill 3mph - Treadmill 5 - light	$1.0 \pm 1.0$	$1.0 \pm 0.9$	$1.3 \pm 1.3$	$1.5 \pm 1.0$	
Walking - Headmin Sinph - Headmin 6 - moderate	$2.0 \pm 2.4$	$1.0 \pm 1.3$	$1.7 \pm 1.3$	$2.4 \pm 1.8$	
waiking - Treadmill Smpn - Treadmill 9 - hard	$0.9 \pm 1.3$	$1.3 \pm 2.1$	$1.1 \pm 1.0$	$1.2 \pm 1.4$	
kneeling	$0.2 \pm 0.2$	$0.1 \pm 0.2$	$0.3 \pm 0.4$	$0.1 \pm 0.2$	
Carrying groceries	$1.0 \pm 1.0$	$0.4 \pm 0.4$	$2.8 \pm 3.7$	$1.2 \pm 1.0$	
Doing disnes	$2.0 \pm 1.5$	$0.8 \pm 0.6$	$1.2 \pm 1.2$	$0.7 \pm 0.6$	
Gardening	$2.2 \pm 1.6$	$1.4 \pm 0.6$	$1.0 \pm 0.8$	$1.2 \pm 1.0$	
Ironing	$1.8 \pm 1.0$	$0.8 \pm 0.0$	$1.4 \pm 1.4$	$1.1 \pm 1.0$	
Making the bed	$0.9 \pm 0.8$	$0.8 \pm 0.9$	$1.7 \pm 0.9$	$1.3 \pm 0.6$	
Mopping	$0.9 \pm 0.7$	$1.0 \pm 0.7$	$1.5 \pm 1.0$	$1.6 \pm 0.8$	
Playing videogames	$0.9 \pm 1.2$	$0.8 \pm 0.9$	$2.1 \pm 3.0$	$1.1 \pm 1.2$	
Scrubbing a surface	$1.8 \pm 2.2$	$1.7 \pm 2.7$	$2.1 \pm 2.2$	$1.4 \pm 0.8$	
Stacking groceries	$0.9 \pm 0.8$	$0.8 \pm 0.4$	$1.5 \pm 1.0$	$1.2 \pm 0.5$	
Sweeping	$2.1 \pm 1.2$	$1.4 \pm 0.5$	$1.0 \pm 0.7$	$1.5 \pm 0.4$	
Typing	$0.6 \pm 0.6$	$0.1 \pm 0.1$	$1.0 \pm 1.3$	$0.6 \pm 1.0$	
Vacuuming	$1.4 \pm 1.3$	$0.9 \pm 0.6$	$0.8 \pm 0.8$	$0.8 \pm 0.4$	
Walking around block	$0.8 \pm 0.8$	$1.1 \pm 1.2$	$2.6 \pm 1.9$	$2.2 \pm 2.1$	
Washing windows	$0.9 \pm 0.8$	$0.9 \pm 0.6$	$2.2 \pm 2.5$	$1.3 \pm 0.9$	
Watching TV	$0.8 \pm 0.8$	$0.6 \pm 0.4$	$1.9 \pm 2.3$	$1.2 \pm 1.9$	
Weeding	$1.0 \pm 1.0$	$1.0 \pm 0.7$	$1.0 \pm 0.6$	$0.9 \pm 0.6$	
Wiping/Dusting	$0.9 \pm 0.6$	$0.7 \pm 0.4$	$1.4 \pm 0.7$	$1.4 \pm 0.7$	
Writing	$1.8 \pm 1.7$	$1.0 \pm 0.7$	$1.4 \pm 1.4$	$0.5\pm0.8$	
taking out trash	$1.2 \pm 0.9$	$1.3 \pm 0.7$	1.6 + 1.2	$1.7 \pm 0.6$	

Table A15-4: True positive rate obtained using the naïve Bayes (NB) and C4.5 classifiers using the *MaxAccelerationSet1* feature set computed per sensor and per axis during subject independent evaluation without the *unknown* class.

# **Appendix A16: Summary of Recent Work in Activity Recognition from Accelerometer Data**

Ref	Approach	Activities Explored	Sensors Utilized	Results	Data collection
[38]	C4.5 decision tree classifier with the following features: Mean, energy, entropy, and correlation.	Total: 20 Postures, ambulation, exercise and household activities such as vacuuming eating, scrubbing and brushing teeth	Five biaxial accelerometers at dominant wrist, ankle, hip and non-dominant upper arm, and thigh.	Subject dependent: 10-fold cross- validation 71.5% accuracy Subject independent: 84.3% accuracy Accelerometers at thigh and wrist achieve ~80% accuracy	Twenty subjects under laboratory and semi- naturalistic conditions.
[37]	Combine discriminative and generative algorithms (Adaboost + HMMs) to improve discrimination and smoothness Features: linear and Mel-scale FFT coefficients, cepstral coefficients, entropy, band-pass filter coefficients, integrals, means and variances.	Total: 10 Sitting Standing Walking Jogging Walking upstairs Walking downstairs Riding bike Riding car elevator up elevator down	Eight sensors: One triaxial accelerometer, Microphone, IR light and visible light sensors, temperature, barometric pressure, humidity and Compass. All sensors were mounted at the shoulder strap of a backpack.	Trained in 80% of data, tested on 20%. Accuracy: 95% Precision: 98% Recall: 84%	Two subjects under naturalistic conditions
[116]	Naïve Bayes classifier combined with the means and variance as features.	Total: 8 Sitting Standing Walking Ascending stairs Descending stairs Writing, typing Shaking hands	Twelve sensors in total: Ankles, knees, hip (both sides), wrists, elbows, and shoulders.	Subject dependent using all sensors: 70-95% accuracy Lower body activity using sensors at hip and ankle ~80—92% accuracy Upper body activity using sensors at shoulder and wrist: ~40-95% accuracy	Number of subjects unknown, probably one. Data collection conditions also not reported. 18.7 minutes of data.
[98]	Handcrafted dynamic Bayesian networks combined with Rao- Blackwellized particle filters	Total: 8 walking, running, upstairs, downstairs, or driving a vehicle	Eight sensors: One triaxial accelerometer, Microphone, IR light and visible light sensors, temperature, barometric pressure, humidity and Compass. All sensors were mounted at the shoulder strap of a backpack.	Performance of DBN and particle filter very similar to results obtained using simple HMMs. 64% accuracy for DBN + particle filter with GPS data. 72% accuracy for HMMs without GPS	Number of subjects not specified. Data collected in naturalistic conditions. 2 hours of data
[99]	Handcrafted dynamic Bayesian networks in a hierarchy combined with Rao- Blackwellized particle filters	Total: 8 stationary, walking, running, driving vehicle, up-stairs, down stairs, situation assessment from cover, incapacitated	Eight sensors: One triaxial accelerometer, Microphone, IR light and visible light sensors, temperature, barometric pressure, humidity and Compass. All sensors were mounted at the shoulder strap of a backpack.	Evaluation approach not described in detail. Total Accuracy 77% Adaboost 86% DBN + particle filter	Eight subjects collecting data under naturalistic conditions. Approximately 30min of data per subject.

[253]	Adaboost with decision stumps in combination with the following features: mean, variance, and energy in different frequency bands.	Total: 14 Running Crawling Lying down Weapon up Kneeling Driving Walking Sitting Ascending stairs Descending stairs Situation assessment Shaking hands Opening door Standing still	Six triaxial accelerometers at the wrists, thigh, hip, chest, and at the soldier's machine gun.	Subject independent evaluation 78.8% accuracy with null misclassifications 90.3% accuracy over periods of activity	3.3 hours of data from soldiers under naturalistic
[101]	Handcrafted HMMs using Gaussian distributions as observation nodes	Total: 9 Saw, drill, screw, hammer, sand, file, drawer, vice, and clap.	Three triaxial accelerometers and two Microphones Accelerometers were mounted at wrists and forearm, microphones at dominant wrist and neck, near the chest.	Subject dependent leave-one-out cross- validation 93.3% total accuracy	One subject under laboratory conditions
[103]	Handcrafted HMMs with mixture of Gaussians as observation nodes using the following features: Mean, variance, number of peaks, and mean amplitude of the peaks.	Total: 9 hammer, saw, file, drill, sand, grind, screwed, vise, drawer, null	3 triaxial axis accelerometer 2 microphones Wrists+upper arm(acc) Wrist+chest (microphone)	Subject independent 44.5% total accuracy, 46.2% of false positives	One subject under laboratory conditions
[254]	Self-organizing maps (SOM)	Total: 10 Lying Kneeling Sitting Standing Walking Running Climbing stairs Descending stairs Bicycling jumping	20 biaxial accelerometers 90 Ball switch Both legs	93% (SOM) self organizing maps	Number of subjects not specified. Data collected at a laboratory.
[40]	Decision tables, decision trees, support vector machines (SVMs), nearest- neighbor, and naïve Bayes classifiers individually and in different meta- classifier configurations such as boosting, bagging, stacking, and plurality voting. Combined with the following features: mean, standard deviation, energy, and correlation	Total: 8 standing walking running upstairs downstairs sit-ups vacuuming brushing teeth	One triaxial accelerometer at the hip (pelvic region).	Subject independent evaluation using leave-one-subject-out Boosted SVMs: 73% accuracy Naïve Bayes: 64% accuracy Subject dependent evaluation using crossvalidation Plurality voting: 99.6% accuracy Naive Bayes: 98.9% accuracy	2 subjects under data collection conditions not specified.

[119]	Analysis of the performance of different features and window lengths using K-means clustering	Total: 6 walking standing jogging skipping hopping riding a bus	One triaxial accelerometer mounted on a the shoulder strap of a backpack	Priority was not recognition accuracy. Main result: there is no single feature or window length optimal for all	Two subjects under naturalistic conditions (200 minutes of data)
				activities. The choice of features and window length depends on the activities to recognize	
[39]	HMMs with Gaussian observation vectors using the following features: means and variances.	Total: 8 Sit-down, Run, Squat, Walk, Stand, Crawl, Lay down, Hand movements	Three triaxial accelerometers placed at the hip, dominant wrist, and chest.	Subject dependent 9- fold crossvalidation 62% accuracy using single accelerometer at chest. 92% accuracy using sensor combination wrist+chest+hip	Three subjects under data collection conditions not specified. 90s of data per activity.

Table A16-1: Summary of recent work in the area of recognizing physical activities from accelerometer data. L stands for data collected in laboratory settings and N for data collected under naturalistic conditions. Ref stands for bibliographical reference to the work.

**Appendix B1: Estimating Energy Expenditure Using the 2-Regression Crouter Algorithm**
Activity	DMSE	MAE	MAED
Bench weight lifting _ hard	$2.06 \pm 0.22$	$2.05 \pm 0.21$	$2.21 \pm 0.28$
Bench weight lifting - light	$1.87 \pm 0.54$	$1.86 \pm 0.53$	$2.21 \pm 0.20$ 2.05 ± 0.69
Bench weight lifting - moderate	$2.09 \pm 0.63$	$2.08 \pm 0.63$	$2.03 \pm 0.09$ 2.32 ± 0.68
Bicen curls - hard	$1.47 \pm 0.05$	$1.47 \pm 0.05$	$1.55 \pm 0.00$
Bicep curls - light	$1.47 \pm 0.31$ 1.81 ± 0.32	$1.47 \pm 0.31$ $1.80 \pm 0.31$	$1.55 \pm 0.57$ 1.99 ± 0.48
Bicep curls - moderate	$1.01 \pm 0.32$ 1.53 ± 0.38	$1.00 \pm 0.01$ 1.51 ± 0.36	$1.99 \pm 0.40$ 1.68 ± 0.50
Calisthenics - Crunches	$1.33 \pm 0.30$ 1.72 + 1.29	$1.51 \pm 0.30$ $1.62 \pm 1.31$	$1.00 \pm 0.50$ 2.15 + 1.45
Calisthenics Sit ups	$1.72 \pm 1.27$ $3.01 \pm 0.82$	$1.02 \pm 1.31$ $3.84 \pm 0.88$	$2.13 \pm 1.43$
Cycling Cycle hard Cycle 20rpm	$5.91 \pm 0.82$	$5.84 \pm 0.88$	$4.57 \pm 0.55$
Cycling Cycle light Cycle 801pm	$0.40 \pm 0.09$	$0.40 \pm 0.09$	$0.01 \pm 0.72$
Cycling - Cycle light - Cycle f001pm	$1.28 \pm 0.30$ 0.51 ± 0.34	$1.23 \pm 0.38$	$1.38 \pm 0.01$
Cycling - Cycle light - Cycle 001pm	$0.51 \pm 0.54$	$0.49 \pm 0.33$	$1.28 \pm 0.57$
Cycling - Cycle light - Cycle 801phi	$1.10 \pm 0.34$ 5.48 ± 0.70	$1.08 \pm 0.33$ 5.48 ± 0.70	$1.20 \pm 0.37$
Lying down	$3.48 \pm 0.79$	$3.46 \pm 0.79$	$3.04 \pm 0.03$
Dynig down	$0.17 \pm 0.08$	$0.13 \pm 0.08$	$0.29 \pm 0.12$
Rowing - Rowing hard - Rowing Suspm	$6.15 \pm 1.97$	$6.12 \pm 1.94$	$6.57 \pm 2.19$
Rowing - Rowing light - Rowing Suspm	$5.14 \pm 1.54$	$5.11 \pm 1.51$	$5.30 \pm 1.78$
Rowing - Rowing moderate - Rowing 30spm	$5.96 \pm 1.85$	$5.93 \pm 1.84$	$6.38 \pm 2.01$
Running - Treadmin 4mph - Treadmin 0	$0.88 \pm 0.45$	$0.81 \pm 0.44$	$1.10 \pm 0.38$
Running - Treadmill Smph - Treadmill 0	$1.19 \pm 0.69$	$1.14 \pm 0.72$	$1.4/\pm 0.76$
Running - Treadmill 6mph - Treadmill 0	$2.40 \pm 0.99$	$2.38 \pm 0.99$	$2.66 \pm 1.02$
Sitting	$0.21 \pm 0.14$	$0.19 \pm 0.14$	$0.26 \pm 0.18$
Sitting - Fidget feet legs	$1.28 \pm 0.27$	$1.27 \pm 0.27$	$1.36 \pm 0.27$
Sitting - Fidget hands arms	$1.16 \pm 0.22$	$1.16 \pm 0.22$	$1.21 \pm 0.24$
Stairs - Ascend stairs	$4.14 \pm 0.30$	$4.13 \pm 0.30$	$4.37 \pm 0.24$
Stairs - Descend stairs	$0.92 \pm 0.30$	$0.88 \pm 0.34$	$1.07 \pm 0.32$
Standing	$0.24 \pm 0.17$	$0.23 \pm 0.17$	$0.28 \pm 0.18$
Walking - Treadmill 2mph - Treadmill 0	$0.29 \pm 0.30$	$0.27 \pm 0.31$	$0.37 \pm 0.33$
Walking - Treadmill 3mph - Treadmill 0	$0.30 \pm 0.26$	$0.28 \pm 0.26$	$0.39 \pm 0.29$
Walking - Treadmill 3mph - Treadmill 3 - light	$4.08 \pm 0.38$	$4.08 \pm 0.38$	$4.23 \pm 0.36$
Walking - Treadmill 3mph - Treadmill 6 - moderate	$4.84 \pm 0.43$	$4.84 \pm 0.43$	$5.03 \pm 0.43$
Walking - Treadmill 3mph - Treadmill 9 - hard	$5.69 \pm 0.57$	$5.69 \pm 0.57$	$5.86 \pm 0.60$
Kneeling	$0.29 \pm 0.13$	$0.26 \pm 0.11$	$0.38 \pm 0.18$
Unknown	$2.72 \pm 0.55$	$2.44 \pm 0.53$	$5.41 \pm 1.17$
Carrying groceries	$3.34 \pm 0.75$	$3.32 \pm 0.75$	$3.60 \pm 0.85$
Doing dishes	$0.66 \pm 0.25$	$0.66 \pm 0.25$	$0.75 \pm 0.25$
Gardening	$1.79 \pm 0.41$	$1.78 \pm 0.40$	$1.99 \pm 0.41$
Ironing	$0.80 \pm 0.25$	$0.77 \pm 0.27$	$0.97 \pm 0.22$
Making the bed	$1.35 \pm 0.58$	$1.23 \pm 0.55$	$1.75 \pm 0.69$
Mopping	$0.84 \pm 0.38$	$0.80 \pm 0.40$	$1.05 \pm 0.41$
Playing videogames	$0.99 \pm 0.14$	$0.99 \pm 0.13$	$1.08 \pm 0.21$
Scrubbing a surface	$1.23 \pm 0.47$	$1.21 \pm 0.47$	$1.47 \pm 0.55$
Stacking groceries	$0.42 \pm 0.38$	$0.41 \pm 0.38$	$0.47 \pm 0.37$
Sweeping	$0.82 \pm 0.27$	$0.77 \pm 0.28$	$1.06 \pm 0.32$
Typing	$0.75 \pm 0.14$	$0.74 \pm 0.15$	$0.84 \pm 0.15$
Vacuuming	$1.07 \pm 0.43$	$1.05 \pm 0.46$	$1.27 \pm 0.41$
Walking around block	$0.84\pm0.36$	$0.80 \pm 0.35$	$1.08 \pm 0.46$
Washing windows	$0.79\pm0.35$	$0.77\pm0.37$	$0.96\pm0.40$
Watching TV	$0.14\pm0.07$	$0.12\pm0.07$	$0.20 \pm 0.10$
Weeding	$2.17\pm0.43$	$2.13\pm0.47$	$2.50\pm0.43$
Wiping/Dusting	$0.50\pm0.21$	$0.46\pm0.22$	$0.66 \pm 0.23$
Writing	$0.79\pm0.18$	$0.79\pm0.18$	$0.86\pm0.17$
Taking out trash	$0.42 \pm 0.23$	$0.37 \pm 0.18$	$0.59 \pm 0.34$

Table B1-1: Error statistics per activity while estimating energy expenditure using the 2-regression Crouter Actigraph-based algorithm with respect to the measurements obtained using the Cosmed K4b2 indirect calorimeter for the MIT energy expenditure dataset. Energy expenditure was computed over one minute sliding windows. RMSE stands for root mean squared error, MAE for mean absolute error, and MAED for maximum absolute error deviation.



Figure B1-1. Plot of energy expenditure (in METS) computed from the Actigraph using the 2-regression Crouter Method and the Cosmed K4b2 Indirect Calorimeter for Subject MIT-001 over one minute windows.



Figure B1-2. Plot of energy expenditure (in METS) computed from the Actigraph using the 2-regression Crouter Method and the Cosmed K4b2 Indirect Calorimeter for Subject MIT-003 over one minute windows.



Figure B1-3: Plot of energy expenditure (in METS) computed from the Actigraph using the 2-regression Crouter Method and the Cosmed K4b2 Indirect Calorimeter for Subject MIT-004 over one minute windows.



Figure B1-4: Plot of energy expenditure (in METS) computed from the Actigraph using the 2-regression Crouter Method and the Cosmed K4b2 Indirect Calorimeter for Subject MIT-018 over one minute windows.



Figure B1-5: Comparison of Parvo indirect calorimeter data and energy expenditure estimated using the 2-regression Crouter algorithm from an Actigraph worn at the hip for subject BU-001.Energ expenditure was computed over one minute sliding windows.



Figure B1-6: Comparison of Parvo indirect calorimeter data and energy expenditure estimated using the 2-regression Crouter algorithm from an Actigraph worn at the hip for subject BU-003.

**Appendix B2: Estimating Energy Expenditure Using Simple Regression Algorithms** 

Activity	Root Mean Squared Error				
	(a)	(b)	(c)	(d)	(e)
Bench weight lifting - light	$1.06 \pm 0.46$	$0.54 \pm 0.22$	$0.54 \pm 0.21$	$0.54 \pm 0.24$	$0.52 \pm 0.25$
Bench weight lifting - moderate	$1.15 \pm 0.52$	$0.57 \pm 0.22$	$0.56 \pm 0.25$	$0.57 \pm 0.22$	$0.54 \pm 0.25$
Bench weight lifting - hard	$1.21 \pm 0.30$	$0.38 \pm 0.23$	$0.39 \pm 0.25$	$0.36 \pm 0.21$	$0.36 \pm 0.24$
Bicep curls - light	$0.51 \pm 0.26$	$1.02 \pm 0.29$	$1.09 \pm 0.37$	$0.58 \pm 0.24$	$1.31 \pm 0.44$
Bicep curls - moderate	$0.50 \pm 0.16$	$1.23 \pm 0.29$	$1.32 \pm 0.32$	$0.75 \pm 0.27$	$1.56 \pm 0.35$
Bicep curls - hard	$0.70 \pm 0.15$	$1.29 \pm 0.52$	$1.38 \pm 0.57$	$0.86 \pm 0.36$	$1.60 \pm 0.64$
Calisthenics - Crunches	$1.13 \pm 0.45$	$1.38 \pm 0.63$	$1.40\pm0.68$	$1.26 \pm 0.64$	$1.34 \pm 0.57$
Calisthenics - Sit ups	$1.88 \pm 0.91$	$1.46 \pm 0.65$	$1.48 \pm 0.67$	$1.68 \pm 0.76$	$1.79 \pm 0.91$
Cycling - Cycle light - Cycle 100rpm	$1.85 \pm 0.93$	$1.73 \pm 0.85$	$1.54 \pm 0.83$	$1.73 \pm 0.84$	$1.54 \pm 0.84$
Cycling - Cycle light - Cycle 60rpm	$2.12\pm0.58$	$1.03 \pm 0.48$	$1.04 \pm 0.46$	$0.92 \pm 0.43$	$1.05 \pm 0.49$
Cycling - Cycle light - Cycle 80rpm	$2.06\pm0.92$	$1.41 \pm 0.72$	$1.39 \pm 0.63$	$1.34 \pm 0.66$	$1.31 \pm 0.70$
Cycling - Cycle moderate - Cycle 80rpm	$3.12 \pm 0.93$	$2.46 \pm 0.79$	$2.26 \pm 0.93$	$2.43 \pm 0.76$	$2.37 \pm 0.85$
Cycling - Cycle hard - Cycle 80rpm	$3.56 \pm 1.25$	$3.01 \pm 0.91$	$2.83 \pm 1.06$	$2.99 \pm 0.81$	$2.95 \pm 0.94$
Lying down	$0.84 \pm 0.16$	$0.73 \pm 0.16$	$0.62 \pm 0.15$	$0.96 \pm 0.16$	$0.64 \pm 0.15$
Rowing - Rowing light - Rowing 30spm	$2.72 \pm 1.16$	$2.09 \pm 1.28$	$2.12 \pm 1.31$	$2.39 \pm 1.40$	$2.05 \pm 1.31$
Rowing - Rowing moderate - Rowing 30spm	$3.45 \pm 1.43$	$2.83 \pm 1.58$	$2.85 \pm 1.60$	$3.18 \pm 1.73$	$2.77 \pm 1.62$
Rowing - Rowing hard - Rowing 30spm	$3.68 \pm 1.60$	$3.05 \pm 1.72$	$3.07 \pm 1.75$	$3.39 \pm 1.83$	$2.99 \pm 1.75$
Running - Treadmill 4mph - Treadmill 0	$2.01 \pm 0.99$	$1.26 \pm 0.51$	$1.22 \pm 0.50$	$1.34 \pm 0.68$	$1.17 \pm 0.39$
Running - Treadmill 5mph - Treadmill 0	$2.85\pm0.92$	$1.05 \pm 0.53$	$0.98 \pm 0.53$	$1.13 \pm 0.56$	$0.96 \pm 0.42$
Running - Treadmill 6mph - Treadmill 0	$3.50 \pm 1.01$	$1.14 \pm 0.72$	$1.05 \pm 0.66$	$1.25 \pm 0.83$	$1.01 \pm 0.54$
Sitting	$0.94 \pm 0.27$	$0.65 \pm 0.17$	$0.56 \pm 0.15$	$0.84 \pm 0.21$	$0.58 \pm 0.15$
Sitting - Fidget feet legs	$0.43 \pm 0.25$	$0.97 \pm 0.24$	$1.05 \pm 0.33$	$1.07 \pm 0.22$	$1.04 \pm 0.29$
Sitting - Fidget hands arms	$0.47 \pm 0.23$	$0.98 \pm 0.26$	$1.00 \pm 0.35$	$0.91 \pm 0.20$	$0.99 \pm 0.30$
Stairs - Ascend stairs	$0.92 \pm 0.18$	$0.84 \pm 0.21$	$0.87 \pm 0.23$	$0.84 \pm 0.20$	$0.87 \pm 0.21$
Stairs - Descend stairs	$1.43 \pm 0.39$	$1.38\pm0.28$	$1.35 \pm 0.30$	$1.42 \pm 0.27$	$1.37 \pm 0.29$
Standing	$0.93 \pm 0.26$	$0.67 \pm 0.16$	$0.58 \pm 0.13$	$0.86 \pm 0.20$	$0.58 \pm 0.14$
Walking - Treadmill 2mph - Treadmill 0	$0.50 \pm 0.29$	$0.76\pm0.26$	$0.84\pm0.35$	$0.87 \pm 0.29$	$0.89\pm0.38$
Walking - Treadmill 3mph - Treadmill 0	$0.87 \pm 0.45$	$0.84 \pm 0.35$	$0.97 \pm 0.42$	$0.97 \pm 0.39$	$1.06 \pm 0.41$
Walking - Treadmill 3mph - Treadmill 3 - light	$0.48 \pm 0.24$	$0.43 \pm 0.17$	$0.57 \pm 0.28$	$0.47 \pm 0.21$	$0.60 \pm 0.30$
Walking - Treadmill 3mph - Treadmill 6 - moderate	$0.78\pm0.38$	$0.75 \pm 0.37$	$0.73 \pm 0.42$	$0.66 \pm 0.31$	$0.63 \pm 0.40$
Walking - Treadmill 3mph - Treadmill 9 - hard	$1.49 \pm 0.70$	$1.50 \pm 0.63$	$1.45 \pm 0.72$	$1.36 \pm 0.62$	$1.31 \pm 0.71$
kneeling	$1.00 \pm 0.21$	$0.57 \pm 0.20$	$0.48 \pm 0.19$	$0.77 \pm 0.21$	$0.49 \pm 0.19$
unknown	$1.99 \pm 0.50$	$1.52 \pm 0.37$	$1.53 \pm 0.35$	$1.53 \pm 0.38$	$1.54 \pm 0.35$
Carrying groceries	$0.98 \pm 0.32$	$0.97 \pm 0.24$	$1.14 \pm 0.29$	$0.97 \pm 0.27$	$1.20 \pm 0.33$
Doing dishes	$0.69 \pm 0.14$	$0.61 \pm 0.16$	$0.60 \pm 0.16$	$0.50 \pm 0.22$	$0.63 \pm 0.15$
Gardening	$1.05 \pm 0.36$	$0.52 \pm 0.28$	$0.56 \pm 0.34$	$0.48 \pm 0.18$	$0.54 \pm 0.28$
Ironing	$0.83 \pm 0.29$	0.66 ± 0.18	$0.62 \pm 0.17$	$0.67 \pm 0.22$	$0.62 \pm 0.18$
Making the bed	$1.09 \pm 0.30$	0.88 ± 0.29	$0.93 \pm 0.30$	$0.90 \pm 0.31$	$0.94 \pm 0.32$
Mopping	$0.90 \pm 0.33$	$0.54 \pm 0.21$	$0.58 \pm 0.22$	$0.52 \pm 0.29$	$0.55 \pm 0.24$
Playing videogames	$0.76 \pm 0.18$	0.79 ± 0.17	$0.68 \pm 0.17$	$1.00 \pm 0.17$	$0.71 \pm 0.16$
Scrubbing a surface	$1.01 \pm 0.25$	$0.54 \pm 0.16$	$0.61 \pm 0.18$	$0.52 \pm 0.19$	$0.61 \pm 0.19$
Stacking groceries	$0.58 \pm 0.22$	0.85 ± 0.26	$0.78 \pm 0.28$	$0.73 \pm 0.27$	$0.72 \pm 0.25$
Sweeping	$0.81 \pm 0.20$	$0.52 \pm 0.23$	$0.56 \pm 0.30$	$0.47 \pm 0.15$	$0.55 \pm 0.28$
Typing	$0.82 \pm 0.13$	$0.72 \pm 0.13$	$0.63 \pm 0.12$	$0.89 \pm 0.14$	$0.66 \pm 0.12$
Vacuuming	$1.00 \pm 0.37$	$0.49 \pm 0.22$	$0.54 \pm 0.26$	$0.48 \pm 0.21$	$0.52 \pm 0.24$
Walking around block	$1.43 \pm 0.25$	1.43 ± 0.25	$1.63 \pm 0.29$	1.41 ± 0.28	1.67 ± 0.30
Washing windows	0.90 ± 0.2/	$0.55 \pm 0.13$	$0.62 \pm 0.20$	$0.51 \pm 0.21$	$0.62 \pm 0.23$
Watching TV	$0.79 \pm 0.13$	$0.78 \pm 0.14$	$0.67 \pm 0.14$	$0.99 \pm 0.15$	$0.69 \pm 0.13$
Weeding	$1.13 \pm 0.34$	$0.56 \pm 0.23$	$0.57 \pm 0.31$	0.57 ± 0.19	0.57 ± 0.29
Wiping/Dusting	0.71 ± 0.19	$0.65 \pm 0.22$	$0.70 \pm 0.29$	$0.54 \pm 0.17$	$0.68 \pm 0.21$
Writing	0.78 ± 0.18	$0.75 \pm 0.19$	$0.65 \pm 0.19$	0.95 ± 0.19	$0.67 \pm 0.18$
taking out trash	$0.76 \pm 0.13$	$0.73 \pm 0.19$	$0.78 \pm 0.20$	$0.72 \pm 0.17$	$0.81 \pm 0.21$

Table B2-1: Root mean squared error per activity when estimating energy expenditure using the simple linear regression equations (a), (b), (c), (d), and (e).

Error Measures	Results
Total Correlation Coefficient	$0.73 \pm 0.06$
Total root Mean Square Error	$1.28 \pm 0.29$
Total mean Absolute Error	$0.95\pm0.16$
Total relative Absolute Error	$63.63 \pm 5.04$
Total root Relative Squared Error	69.73 ± 5.27
Maximum absolute Deviation	$4.12 \pm 1.21$

Table B2-2: Performance of estimating energy expenditure using multivariable linear regression and the *ACAbsArea* feature set computed per sensor over one minute windows with respect to energy expenditure measured using the Cosmed K4b2 indirect calorimeter for the MIT dataset.

Energy expenditure (METs) = $-0.0043 * absarea 1T +$
0.0013 * absarea 4T +
0.0024 * absarea 7T +
0.0025 * absarea 8T +
0.0024 * absarea 11T +
0.0026 * absarea 14T +
0.0019 * absarea 17T +
1.3746

Table B2-3: Prediction equation learned using multivariable linear regression and the *ACAbsArea* feature set computed per sensor over one minute windows from the MIT energy expenditure dataset.

Activity	RMSE	MAE	MAED
Bench weight lifting - hard	$0.27 \pm 0.18$	$0.24 \pm 0.16$	$0.34 \pm 0.25$
Bench weight lifting - light	$0.58 \pm 0.19$	$0.54 \pm 0.20$	$0.70 \pm 0.23$
Bench weight lifting - moderate	$0.53 \pm 0.28$	$0.49 \pm 0.27$	$0.63 \pm 0.35$
Bicep curls - hard	$1.41 \pm 0.63$	$1.40 \pm 0.64$	$1.53 \pm 0.64$
Bicep curls - light	$1.10 \pm 0.41$	$1.09 \pm 0.44$	$1.25 \pm 0.32$
Bicep curls - moderate	$1.40 \pm 0.37$	$1.38 \pm 0.40$	$1.55 \pm 0.26$
Calisthenics - Crunches	$1.31 \pm 0.85$	$1.17 \pm 0.80$	$1.62 \pm 1.00$
Calisthenics - Sit ups	$1.40 \pm 0.79$	$1.17 \pm 0.65$	$1.88 \pm 1.06$
Cycling - Cycle hard - Cycle 80rpm	$2.92 \pm 1.19$	$2.90 \pm 1.20$	$3.19 \pm 1.22$
Cycling - Cycle light - Cycle 100rpm	$1.50 \pm 0.78$	$1.43 \pm 0.82$	$1.82 \pm 0.79$
Cycling - Cycle light - Cycle 60rpm	$1.05 \pm 0.51$	$1.03 \pm 0.53$	$1.19 \pm 0.50$
Cycling - Cycle light - Cycle 80rpm	$1.36 \pm 0.62$	$1.32 \pm 0.64$	$1.59 \pm 0.65$
Cycling - Cycle moderate - Cycle 80rpm	2.04 + 1.06	2.01 + 1.04	2.24 + 1.19
Lying down	$0.49 \pm 0.15$	$0.48 \pm 0.16$	$0.61 \pm 0.14$
Rowing - Rowing hard - Rowing 30spm	$2.94 \pm 1.70$	$2.84 \pm 1.64$	$3.38 \pm 1.99$
Rowing - Rowing light - Rowing 30spm	$1.95 \pm 1.29$	$1.82 \pm 1.19$	$2.37 \pm 1.56$
Rowing - Rowing moderate - Rowing 30spm	$2.73 \pm 1.55$	$2.62 \pm 1.54$	$3.21 \pm 1.77$
Running - Treadmill 4mph - Treadmill 0	$1.19 \pm 0.56$	$1.06 \pm 0.60$	$1.63 \pm 0.68$
Running - Treadmill 5mph - Treadmill 0	$1.00 \pm 0.63$	$0.93 \pm 0.63$	$1.00 \pm 0.00$ $1.28 \pm 0.79$
Running - Treadmill 6mph - Treadmill 0	$0.92 \pm 0.76$	$0.92 \pm 0.02$ $0.91 \pm 0.76$	$1.07 \pm 0.83$
Sitting	$0.92 \pm 0.10$ 0.43 + 0.17	$0.91 \pm 0.10$ $0.42 \pm 0.17$	$0.50 \pm 0.16$
Sitting - Fidget feet legs	$1.01 \pm 0.17$	$1.00 \pm 0.35$	$1.09 \pm 0.10$
Sitting - Fidget hands arms	$0.96 \pm 0.38$	$0.95 \pm 0.39$	$1.05 \pm 0.11$ $1.05 \pm 0.39$
Stairs - Ascend stairs	$0.50 \pm 0.30$ $0.60 \pm 0.34$	$0.55 \pm 0.35$ 0.54 ± 0.34	$0.83 \pm 0.39$
Stairs - Descend stairs	$1.11 \pm 0.36$	$1.08 \pm 0.34$	$1.05 \pm 0.05$
Standing	$0.45 \pm 0.14$	$0.43 \pm 0.13$	$0.55 \pm 0.18$
Walking - Treadmill 2mph - Treadmill 0	$0.49 \pm 0.14$ 0.88 ± 0.41	$0.45 \pm 0.13$ 0.85 ± 0.42	$1.11 \pm 0.39$
Walking - Treadmill 3mph - Treadmill 0	$1.08 \pm 0.49$	$1.05 \pm 0.12$	$1.11 \pm 0.59$ $1.32 \pm 0.56$
Walking - Treadmill 3mph - Treadmill 3 - light	$0.66 \pm 0.37$	$0.63 \pm 0.38$	$0.88 \pm 0.45$
Walking - Treadmill 3mph - Treadmill 6 - moderate	$0.60 \pm 0.37$ 0.67 ± 0.42	$0.63 \pm 0.50$	$0.85 \pm 0.44$
Walking - Treadmill 3mph - Treadmill 9 - hard	$1.28 \pm 0.79$	$1.05 \pm 0.13$	$149 \pm 0.90$
Kneeling	$0.33 \pm 0.18$	$0.31 \pm 0.19$	$0.39 \pm 0.19$
Unknown	$1.31 \pm 0.33$	$1.02 \pm 0.24$	$3.26 \pm 0.87$
Carrying groceries	$1.31 \pm 0.33$ 1.26 ± 0.37	$1.02 \pm 0.21$ $1.17 \pm 0.40$	$1.20 \pm 0.07$ $1.70 \pm 0.45$
Doing dishes	$0.56 \pm 0.18$	$0.55 \pm 0.18$	$0.65 \pm 0.20$
Gardening	$0.50 \pm 0.10$ 0.51 + 0.41	$0.33 \pm 0.10$ 0.47 + 0.41	$0.05 \pm 0.20$ 0.65 ± 0.44
Ironing	$0.51 \pm 0.11$ 0.55 ± 0.18	$0.11 \pm 0.11$ 0.51 ± 0.18	$0.03 \pm 0.11$ $0.72 \pm 0.27$
Making the bed	$0.83 \pm 0.10$ 0.84 + 0.36	$0.31 \pm 0.10$ $0.70 \pm 0.30$	$1.28 \pm 0.59$
Monning	$0.50 \pm 0.30$	$0.76 \pm 0.33$	$0.70 \pm 0.35$
Plaving videogames	$0.50 \pm 0.21$ 0.55 ± 0.17	$0.10 \pm 0.23$ 0.54 ± 0.18	$0.63 \pm 0.17$
Scrubbing a surface	$0.55 \pm 0.17$ 0.48 ± 0.23	$0.37 \pm 0.10$ $0.42 \pm 0.22$	$0.03 \pm 0.17$ 0.73 ± 0.31
Stacking groceries	$0.40 \pm 0.23$	$0.42 \pm 0.22$ 0.63 ± 0.32	$0.79 \pm 0.31$ 0.79 ± 0.30
Sweening	$0.00 \pm 0.30$ $0.47 \pm 0.28$	$0.03 \pm 0.32$ 0.41 ± 0.24	$0.77 \pm 0.30$ $0.72 \pm 0.44$
Typing	$0.47 \pm 0.20$	$0.41 \pm 0.24$ 0.50 ± 0.12	$0.72 \pm 0.44$ 0.62 ± 0.14
Vacuuming	$0.32 \pm 0.12$ 0.47 + 0.32	$0.30 \pm 0.12$ $0.44 \pm 0.30$	$0.62 \pm 0.14$ 0.63 + 0.42
Walking around block	$1.78 \pm 0.32$	$1.71 \pm 0.30$	$2.03 \pm 0.42$ $2.24 \pm 0.33$
Washing windows	$0.58 \pm 0.27$	$0.51 \pm 0.30$	$0.80 \pm 0.42$
Watching TV	$0.50 \pm 0.20$ 0.53 + 0.14	$0.51 \pm 0.27$ 0.52 + 0.14	$0.63 \pm 0.42$
Weeding	$0.05 \pm 0.14$ $0.46 \pm 0.30$	$0.32 \pm 0.14$ 0.39 + 0.26	$0.63 \pm 0.17$ $0.64 \pm 0.44$
Wining/Dusting	$0.69 \pm 0.30$	$0.57 \pm 0.20$ 0.64 + 0.33	$0.04 \pm 0.44$ 0.91 + 0.53
Writing	$0.57 \pm 0.57$ 0.54 + 0.19	$0.54 \pm 0.55$	$0.51 \pm 0.55$ $0.61 \pm 0.18$
Taking out trash	$0.65 \pm 0.19$	$0.60 \pm 0.19$	$0.90 \pm 0.29$
Tutting out aution	5.05 ± 0.25	0.00 ± 0.20	5.70 - 0.27

Table B2-4: Performance per class while predicting energy expenditure using multivariable linear regression and the *ACAbsArea* feature set computed per sensor over one minute windows for the MIT energy expenditure dataset.

**Appendix B3: Activities Utilized to Predict Energy Expenditure using Equation (e)** 

Activity	Accelerometer	Posture
	Recognizable	and Ambulation
Bench weight lifting – Light		Ambulation
Bench weight lifting – Moderate		
Bench weight lifting – Hard		
Bicep curls – Light		
Bicep curls – Moderate		
Bicep curls – Hard		
Calisthenics Crunches		
Calisthenics Sit ups		
Cycling Cycle 100rpm (15mph, 120.4W) – Light		
Cycling Cycle 60rpm (8.9mph, 66.9W) – Light		
Cycling Cycle 80rpm (11.2mph, 100.4W) - Light		
Cycling Cycle 80rpm – Moderate		
Cycling Cycle 80rpm – Hard		
Lying down	$\checkmark$	$\checkmark$
Rowing 30spm – Light		
Rowing 30spm – Moderate		
Rowing 30spm – Hard		
Running Treadmill 4mph Treadmill 0	$\checkmark$	$\checkmark$
Running Treadmill 5mph Treadmill 0	$\checkmark$	$\checkmark$
Running Treadmill 6mph Treadmill 0	$\checkmark$	$\checkmark$
Sitting		
Sitting Fidget feet legs		
Sitting Fidget hands arms		
Stairs Ascend stairs		
Stairs Descend stairs		
Standing	$\checkmark$	$\checkmark$
Walking Treadmill 2mph Treadmill 0	$\checkmark$	$\checkmark$
Walking Treadmill 3mph Treadmill 0	$\checkmark$	$\checkmark$
Walking Treadmill 3mph Treadmill 3 - Light		
Walking Treadmill 3mph Treadmill 6 - Moderate		
Walking Treadmill 3mph Treadmill 9 - Hard		
Kneeling	V	
Carrying groceries		
Doing dishes	V	
Gardening	V	
Ironing	V	
Making the bed	V	
Mopping		
Playing videogames	V	
Scrubbing a surface		
Stacking groceries		
Sweeping	1	
Typing	N	N
Vacuuming	1	1
Walking around block	N	N
Washing windows	1	
watching 1 V	N	
Wining Desting	N	
wiping/Dusting	N,	,
writing	V	N
I aking out trash	2	

Table B3-1: Activities utilized to learn the coefficients of equation (e). The column 'Accelerometer Recognizable' includes only activities that can be recognized from accelerometer data and consequently, those activities that do not contain resistance work or load effort. The column 'Posture and Ambulation' includes only activities involving posture or ambulation.

**Appendix B4: Comparison of Activities Contained in the MIT Energy Expenditure Dataset and the Ones Found in the Compendium of Physical Activities** 

Gymnasium Dataset Activity	Closest Compendium Activity	METs	Comparable Activity
Lying down	Lying in bed awake, listening to music (not talking or reading	1.0	
Sitting	Sitting quietly and watching television	1.0	
Sitting - Fidget feet legs	Sitting quietly and watching television	1.0	
Sitting - Fidget hands arms	Sitting - knitting, sewing, lt. wrapping (presents)	1.5	
Standing	Standing quietly (standing in a line)	1.2	
Kneeling	Kneeling in church/at home (praying)	1.0	
Walking - Treadmill 2mph - Treadmill 0%	Walking, 2.0 mph, level, slow pace, firm surface	2.5	
Walking - Treadmill 3mph - Treadmill 0%	Walking, 3.0 mph, level, moderate pace, firm surface	3.3	
Walking - Treadmill 3mph - Treadmill 3% - light	Walking, 3.5 mph, uphill	6.0	
Walking - Treadmill 3mph - Treadmill 6% - moderate	Walking, 3.5 mph, uphill	6.0	
Walking - Treadmill 3mph - Treadmill 9% - hard	Walking, 3.5 mph, uphill	6.0	
Stairs - Ascend stairs	Upstairs, using or climbing up ladder	8.0	
Stairs - Descend stairs	Downstairs	3.0	
Running - Treadmill 4mph - Treadmill 0%	Walking, 4.0 mph, level, firm surface, very brisk pace	5.0	$\checkmark$
Running - Treadmill 5mph - Treadmill 0%	Running, 5 mph (12 min/mile)	8.0	
Running - Treadmill 6mph - Treadmill 0%	Running, 6 mph (10 min/mile)	10.0	V
Cycling - Cycle light - Cycle 60rpm (8.9mph or 66.9W)	Bicycling, stationary, 50watt, very light effort	3.0	V
Cycling - Cycle hard - Cycle 80rpm (11.2mph or 100W)	Bicycling, stationary, 100watts, light effort	5.5	V
Cycling - Cycle light - Cycle 100rpm (15mph or 140W)	Bicycling, stationary, 150watts, moderate effort	7.0	$\checkmark$
Cycling - Cycle moderate - Cycle 80rpm	Bicycling, stationary, 100watts, light effort	5.5	
Cycling - Cycle hard - Cycle 80rpm	Bicycling, stationary, 100watts, light effort	5.5	
Rowing - Rowing light - Rowing 30spm	Rowing, stationary, 50 watts, light effort	3.5	
Rowing - Rowing moderate - Rowing 30spm	Rowing, stationary, 100 watts, moderate effort	7.0	
Rowing - Rowing hard - Rowing 30spm	Rowing, stationary, 150 watts, vigorous effort	8.5	
Bench weight lifting - hard	Weight lifting (free weight, nautilus or universal- type), power lifting or body building, vigorous effort	6.0	
Bench weight lifting - moderate	Weight lifting (free, nautilus or universal-type), light or moderate effort, light workout, general	3.0	
Bench weight lifting - light	Weight lifting (free, nautilus or universal-type), light or moderate effort, light workout, general	3.0	
Bicep curls - hard	Weight lifting (free, nautilus or universal-type), light or moderate effort, light workout, general	3.0	
Bicep curls - moderate	Weight lifting (free, nautilus or universal-type), light or moderate effort, light workout, general	3.0	
Bicep curls - light	Weight lifting (free, nautilus or universal-type), light or moderate effort, light workout, general	3.0	
Calisthenics - Crunches	Calisthenics (e.g. pushups, sit-ups, pull-ups, jumping jacks), heavy, vigorous effort	8.0	
Calisthenics - Sit ups	Calisthenics (e.g. pushups, sit-ups, pull-ups, iumping jacks), heavy, vigorous effort	8.0	V

Table B4-1: Mapping between gymnasium dataset activities for which data was collected and the activities listed in the Compendium of Physical Activities. The METs column indicates energy expenditure in METs from the Compendium and the column "Comparable Activity" indicates if the Compendium activity was used for comparison in the energy expenditure experiments.

Home Dataset Activity	Closest Compendium Activity Description	METs	Comparable Activity
Lying down	Lying in bed awake, listening to music (not talking or reading	1.0	V
Writing	Sitting - writing, desk work, typing	1.8	
Typing	Sitting - writing, desk work, typing	1.8	
Watching TV	Lying quietly, watching television	1.0	
Playing videogames	Sitting - card playing, playing board games	1.5	
Making the bed	Making bed	2.0	
Taking out trash	Cleaning, light (dusting, straightening up, changing linen, carrying out trash	2.5	V
Ironing	Ironing	2.3	
Doing dishes	Wash dishes - standing or in general (not broken into stand/walk components	2.3	
Wiping/Dusting	Cleaning, light (dusting, straightening up, changing linen, carrying out trash)	2.5	V
Walking around block	Walking for pleasure	3.5	
Carrying groceries (one 3kg bag on each hand)	Walking 2.5mph slowly and carrying light objects less than 25Lb	3.0	
Stacking groceries	Putting away groceries (e.g. carrying groceries, shopping without a grocery cart), carrying packages.	2.5	N
Sweeping	Carpet sweeping, sweeping floors	3.3	
Mopping	Mopping	3.5	
Vacuuming	Vacuuming	3.5	V
Washing windows	Cleaning, heavy or major (e.g. wash car, wash windows, clean garage), vigorous effort	3.0	
Weeding	Weeding, cultivating garden	4.5	
Gardening	Gardening, general	4.0	
Scrubbing a surface	Scrubbing floors, on hands and knees,	3.8	

Table B4-2: Mapping between household dataset activities for which data was collected and the activities listed in the Compendium of Physical Activities. The METs column indicates energy expenditure in METs from the Compendium and the column "Comparable Activity" indicates if the Compendium activity was used for comparison in the energy expenditure experiments.

**Appendix B5: Estimating Energy Expenditure Using the Compendium of Physical Activities** 

Activity	RMSE	MAE	MAED
Calisthenics - Sit ups	$3.91 \pm 0.82$	$3.84\pm0.88$	$4.57 \pm 0.55$
Lying down	$0.17\pm0.08$	$0.15\pm0.08$	$0.29 \pm 0.12$
Running - Treadmill 4mph - Treadmill 0	$0.88\pm0.45$	$0.81 \pm 0.44$	$1.16 \pm 0.58$
Running - Treadmill 5mph - Treadmill 0	$1.19\pm0.69$	$1.14\pm0.72$	$1.47\pm0.76$
Running - Treadmill 6mph - Treadmill 0	$2.40 \pm 0.99$	$2.38 \pm 0.99$	$2.66 \pm 1.02$
Sitting	$0.21 \pm 0.14$	$0.19 \pm 0.14$	$0.26 \pm 0.18$
Stairs - Ascend stairs	$4.14 \pm 0.30$	$4.13 \pm 0.30$	$4.37 \pm 0.24$
Stairs - Descend stairs	$0.92\pm0.30$	$0.88\pm0.34$	$1.07\pm0.32$
Standing	$0.24 \pm 0.17$	$0.23 \pm 0.17$	$0.28 \pm 0.18$
Walking - Treadmill 2mph - Treadmill 0	$0.29\pm0.30$	$0.27 \pm 0.31$	$0.37\pm0.33$
Walking - Treadmill 3mph - Treadmill 0	$0.30 \pm 0.26$	$0.28 \pm 0.26$	$0.39 \pm 0.29$
kneeling	$0.29 \pm 0.13$	$0.26 \pm 0.11$	$0.38 \pm 0.18$
Doing dishes	$0.66 \pm 0.25$	$0.66 \pm 0.25$	$0.75 \pm 0.25$
Gardening	$1.79 \pm 0.41$	$1.78 \pm 0.40$	$1.99 \pm 0.41$
Ironing	$0.80 \pm 0.25$	$0.77 \pm 0.27$	$0.97 \pm 0.22$
Making the bed	$1.35 \pm 0.58$	$1.23\pm0.55$	$1.75 \pm 0.69$
Mopping	$0.84 \pm 0.38$	$0.80\pm0.40$	$1.05 \pm 0.41$
Scrubbing a surface	$1.23 \pm 0.47$	$1.21 \pm 0.47$	$1.47 \pm 0.55$
Stacking groceries	$0.42\pm0.38$	$0.41 \pm 0.38$	$0.47 \pm 0.37$
Sweeping	$0.82 \pm 0.27$	$0.77\pm0.28$	$1.06 \pm 0.32$
Typing	$0.75 \pm 0.14$	$0.74\pm0.15$	$0.84\pm0.15$
Vacuuming	$1.07 \pm 0.43$	$1.05 \pm 0.46$	$1.27 \pm 0.41$
Walking around block	$0.84 \pm 0.36$	$0.80 \pm 0.35$	$1.08 \pm 0.46$
Washing windows	$0.79 \pm 0.35$	$0.77 \pm 0.37$	$0.96 \pm 0.40$
Watching TV	$0.14 \pm 0.07$	$0.12\pm0.07$	$0.20\pm0.10$
Weeding	$2.17 \pm 0.43$	$2.13 \pm 0.47$	$2.50 \pm 0.43$
Wiping/Dusting	$0.50 \pm 0.21$	$0.46 \pm 0.22$	$0.66 \pm 0.23$
Writing	$0.79 \pm 0.18$	$0.79 \pm 0.18$	$0.86 \pm 0.17$
taking out trash	$0.42 \pm 0.23$	$0.37 \pm 0.18$	$0.59 \pm 0.34$

Table B5-1: Results of estimating energy expenditure using only the *comparable* Compendium activities over the MIT energy expenditure dataset assuming activity is known. MAE stands for mean absolute error, RMSE for root mean squared error, MAED for maximum absolute error deviation over a one minute window.

Activity	RMSE	MAE	MAED
Bench weight lifting - hard	$4.08 \pm 0.14$	$4.08 \pm 0.14$	$4.16 \pm 0.22$
Bench weight lifting - light	$1.30 \pm 0.50$	$1.27 \pm 0.52$	$1.43 \pm 0.44$
Bench weight lifting - moderate	$1.17 \pm 0.40$	$1.12 \pm 0.46$	$1.31 \pm 0.37$
Bicep curls - hard	$4.55 \pm 0.49$	$4.55 \pm 0.49$	$4.65 \pm 0.44$
Bicep curls - light	$1.20 \pm 0.26$	$1.19 \pm 0.27$	$1.36 \pm 0.26$
Bicep curls - moderate	$1.42 \pm 0.36$	$1.40 \pm 0.38$	$1.56 \pm 0.21$
Calisthenics - Crunches	$6.22 \pm 1.37$	$6.18 \pm 1.41$	$6.68 \pm 1.22$
Calisthenics - Sit ups	$4.39 \pm 0.75$	$4.26 \pm 0.85$	$5.44 \pm 0.56$
Cycling - Cycle hard - Cycle 80rpm	$1.74 \pm 0.68$	$1.71 \pm 0.69$	$2.10 \pm 0.70$
Cycling - Cycle light - Cycle 100rpm	$1.45 \pm 0.59$	$1.39 \pm 0.60$	$1.89 \pm 0.67$
Cycling - Cycle light - Cycle 60rpm	$0.50 \pm 0.31$	$0.48 \pm 0.32$	$0.61 \pm 0.31$
Cycling - Cycle light - Cycle 80rpm	$1.22 \pm 0.53$	$1.19 \pm 0.55$	$1.53 \pm 0.57$
Cycling - Cycle moderate - Cycle 80rpm	$2.74 \pm 0.82$	$2.70 \pm 0.86$	$3.05 \pm 0.76$
Lying down	$0.17 \pm 0.08$	$0.15 \pm 0.08$	$0.29 \pm 0.15$
Rowing - Rowing hard - Rowing 30spm	$2.92 \pm 1.35$	$2.81 \pm 1.47$	$3.45 \pm 1.31$
Rowing - Rowing light - Rowing 30spm	$1.62 \pm 1.39$	$1.46 \pm 1.29$	$2.06 \pm 1.61$
Rowing - Rowing moderate - Rowing 30spm	$2.09 \pm 0.88$	$1.95 \pm 0.96$	$2.62 \pm 0.91$
Running - Treadmill 4mph - Treadmill 0	$1.03 \pm 0.31$	$0.91 \pm 0.28$	$1.42 \pm 0.48$
Running - Treadmill 5mph - Treadmill 0	$1.38 \pm 0.68$	$1.29 \pm 0.71$	$1.85 \pm 0.81$
Running - Treadmill 6mph - Treadmill 0	$2.51 \pm 0.93$	$2.46 \pm 0.93$	$2.89 \pm 1.17$
Sitting	$0.26 \pm 0.20$	$0.23 \pm 0.19$	$0.34 \pm 0.25$
Sitting - Fidget feet legs	$0.20 \pm 0.20$ 0.30 + 0.20	$0.29 \pm 0.19$ 0.29 + 0.21	$0.34 \pm 0.20$
Sitting - Fidget hands arms	$0.36 \pm 0.17$	$0.25 \pm 0.21$ 0.35 ± 0.17	$0.01 \pm 0.120$ $0.42 \pm 0.19$
Stairs - Ascend stairs	$5.09 \pm 0.26$	$5.08 \pm 0.26$	$5.42 \pm 0.15$
Stairs - Descend stairs	$1.09 \pm 0.20$ $1.22 \pm 0.42$	$1.17 \pm 0.39$	$141 \pm 0.23$
Standing	$0.25 \pm 0.12$	$0.22 \pm 0.17$	$0.32 \pm 0.25$
Walking - Treadmill 2mph - Treadmill 0	$0.23 \pm 0.19$ 0.32 + 0.28	$0.22 \pm 0.17$ $0.29 \pm 0.29$	$0.52 \pm 0.20$ 0.45 + 0.30
Walking - Treadmill 3mph - Treadmill 0	$0.35 \pm 0.23$	$0.23 \pm 0.23$ 0.31 + 0.23	$0.49 \pm 0.26$
Walking - Treadmill 3mph - Treadmill 3 - light	$2.00 \pm 0.38$	$1.99 \pm 0.38$	$2.28 \pm 0.41$
Walking - Treadmill 3mph - Treadmill 6 - moderate	$1.27 \pm 0.42$	$1.25 \pm 0.43$	$1.59 \pm 0.45$
Walking - Treadmill 3mph - Treadmill 9 - hard	$0.64 \pm 0.28$	$0.60 \pm 0.30$	$0.87 \pm 0.33$
kneeling	$0.32 \pm 0.16$	$0.30 \pm 0.16$	$0.41 \pm 0.19$
Carrying groceries	$0.74 \pm 0.38$	$0.70 \pm 0.38$	$0.96 \pm 0.47$
Doing dishes	$0.64 \pm 0.26$	$0.64 \pm 0.26$	$0.70 \pm 0.28$
Gardening	$1.79 \pm 0.44$	$1.79 \pm 0.44$	$1.96 \pm 0.46$
Ironing	$0.82 \pm 0.26$	$0.78 \pm 0.28$	$0.98 \pm 0.30$
Making the bed	$1.23 \pm 0.55$	$1.09 \pm 0.49$	$1.63 \pm 0.67$
Mopping	$0.88 \pm 0.40$	$0.84 \pm 0.41$	$1.10 \pm 0.44$
Plaving videogames	$0.51 \pm 0.14$	$0.50 \pm 0.15$	$0.57 \pm 0.12$
Scrubbing a surface	$1.34 \pm 0.41$	$1.31 \pm 0.41$	$1.63 \pm 0.42$
Stacking groceries	$0.50 \pm 0.35$	$0.47 \pm 0.37$	$0.58 \pm 0.33$
Sweeping	$0.85 \pm 0.27$	$0.79 \pm 0.29$	$1.12 \pm 0.33$
Typing	$0.69 \pm 0.14$	$0.67 \pm 0.14$	$0.82 \pm 0.16$
Vacuuming	$1.06 \pm 0.39$	$1.03 \pm 0.41$	$1.26 \pm 0.39$
Walking around block	$0.98 \pm 0.34$	$0.90 \pm 0.36$	$1.33 \pm 0.44$
Washing windows	$0.79 \pm 0.35$	$0.77 \pm 0.35$	$0.94 \pm 0.37$
Watching TV	$0.16 \pm 0.05$	$0.13 \pm 0.05$	$0.24 \pm 0.07$
Weeding	$2.19 \pm 0.38$	$2.15 \pm 0.41$	$2.53 \pm 0.51$
Wiping/Dusting	$0.51 \pm 0.23$	$0.49 \pm 0.24$	$0.64 \pm 0.26$
Writing	$0.78 \pm 0.18$	0.77 ± 0.19	0.85 ± 0.18
taking out trash	$0.43 \pm 0.22$	$0.38 \pm 0.20$	0.64 ± 0.29

Table B5-2: Results of estimating energy expenditure using the *closest* Compendium activities over the MIT energy expenditure dataset assuming activity is known. MAE stands for mean absolute error, RMSE for root mean squared error, and MAED for maximum absolute error deviation.



Figure B5-1: Plot of energy expenditure estimated using the Compendium of Physical Activities vs. the one measured using the Cosmed K4b2 Indirect Calorimeter for subject MIT-001. The activities shown are only the directly comparable to the compendium Activities (*comparable* set).



Figure B5-2: Plot of energy expenditure estimated using the Compendium of Physical Activities vs. the one measured using the Cosmed K4b2 Indirect Calorimeter for subject MIT-003. The activities shown are only the directly comparable to the compendium Activities (*comparable* set).



Figure B5-3: Plot of energy expenditure estimated using the Compendium of Physical Activities vs. the one measured using the Cosmed K4b2 Indirect Calorimeter for subject MIT-004. The activities shown are only the directly comparable to the compendium Activities (*comparable* set).



Figure B5-4: Plot of energy expenditure estimated using the Compendium of Physical Activities vs. the one measured using the Cosmed K4b2 Indirect Calorimeter for subject MIT-018. The activities shown are only the directly comparable to the compendium Activities (*comparable* set).



Figure B5-4: Plot of energy expenditure estimated using the Compendium of Physical Activities vs. the one measured using the Cosmed K4b2 indirect calorimeter for subject MIT-001. The activities shown are the closest with respect to the compendium activities (*closest* set).



Figure B5-5. Plot of energy expenditure estimated using the Compendium of Physical Activities vs. the one measured using the Cosmed K4b2 indirect calorimeter for subject MIT-004. The activities shown are the closest with respect to the compendium activities (*closest* set).



Figure B5-6: Plot of energy expenditure estimated using the Compendium of Physical Activities vs. the one measured using the Cosmed K4b2 indirect calorimeter for subject MIT-018. The activities shown are the closest with respect to the compendium activities (*closest* set).

Appendix B6: Estimating Energy Expenditure Using Linear Regression and the ACAbsArea Feature Computed over One Minute Windows.

Activity	RMSE	MAE	MAED
Bench_weight_liftinghard	$0.27\pm0.18$	$0.24\pm0.16$	$0.34\pm0.25$
Bench_weight_liftinglight	$0.58 \pm 0.19$	$0.54 \pm 0.20$	$0.70 \pm 0.23$
Bench_weight_liftingmoderate	$0.53 \pm 0.28$	$0.49 \pm 0.27$	$0.63 \pm 0.35$
Bicep_curlshard	$1.41 \pm 0.63$	$1.40 \pm 0.64$	$1.53 \pm 0.64$
Bicep curls - light	$1.10 \pm 0.41$	$1.09 \pm 0.44$	$1.25 \pm 0.32$
Bicep curls - moderate	$1.40 \pm 0.37$	$1.38 \pm 0.40$	$1.55 \pm 0.26$
Calisthenics - Crunches	$1.31 \pm 0.85$	$1.17 \pm 0.80$	1.62 + 1.00
Calisthenics - Sit ups	$1.40 \pm 0.79$	$1.17 \pm 0.65$	1.88 + 1.06
Cycling - Cycle hard - Cycle 80rpm	$2.92 \pm 1.19$	$2.90 \pm 1.20$	$3.19 \pm 1.22$
Cycling - Cycle light - Cycle 100rpm	$1.50 \pm 0.78$	$1.43 \pm 0.82$	$1.82 \pm 0.79$
Cycling - Cycle light - Cycle 60rpm	$1.05 \pm 0.51$	$1.03 \pm 0.53$	$1.19 \pm 0.50$
Cycling - Cycle light - Cycle 80rpm	$1.36 \pm 0.62$	$1.32 \pm 0.64$	$1.59 \pm 0.65$
Cycling - Cycle moderate - Cycle 80rpm	$2.04 \pm 1.06$	$2.01 \pm 1.04$	$2.24 \pm 1.19$
Lying down	$0.49 \pm 0.15$	$0.48 \pm 0.16$	$0.61 \pm 0.14$
Rowing - Rowing hard - Rowing 30spm	$2.94 \pm 1.70$	$2.84 \pm 1.64$	$3.38 \pm 1.09$
Rowing - Rowing light - Rowing 30spm	$1.95 \pm 1.70$	$1.82 \pm 1.04$	$2.30 \pm 1.00$
RowingRowing_ngitRowing_Jospin	$1.73 \pm 1.27$	$1.02 \pm 1.17$	$2.37 \pm 1.30$
RowingKowing_indefateKowing_Sospin	$2.73 \pm 1.55$ 1 19 ± 0.56	$2.02 \pm 1.54$	$3.21 \pm 1.77$ 1.63 ± 0.68
RunningIteadmin_4mpnIteadmin_0_	$1.19 \pm 0.30$ 1.00 ± 0.62	$1.00 \pm 0.00$	$1.03 \pm 0.08$ 1.28 ± 0.70
RunningIteadminShiphIteadmin_0_	$1.00 \pm 0.03$	$0.93 \pm 0.03$	$1.20 \pm 0.79$ $1.07 \pm 0.82$
Sitting	$0.92 \pm 0.70$	$0.91 \pm 0.70$	$1.07 \pm 0.83$
Sitting Fident fact land	$0.43 \pm 0.17$	$0.42 \pm 0.17$	$0.30 \pm 0.10$
Sitting Fidget hands arms	$1.01 \pm 0.30$	$1.00 \pm 0.33$	$1.09 \pm 0.41$
StungFluget_hands_arms	$0.96 \pm 0.38$	$0.93 \pm 0.39$	$1.03 \pm 0.39$
StairsAscend_stairs	$0.60 \pm 0.34$	$0.54 \pm 0.34$	$0.83 \pm 0.39$
StairsDescend_stairs	$1.11 \pm 0.36$	$1.08 \pm 0.33$	$1.25 \pm 0.43$
Standing	$0.45 \pm 0.14$	$0.43 \pm 0.13$	$0.55 \pm 0.18$
waikingIreadmiii_2mpnIreadmiii_0_	$0.88 \pm 0.41$	$0.85 \pm 0.42$	$1.11 \pm 0.39$
waikingIreadmiii_3mpnIreadmiii_0_	$1.08 \pm 0.49$	$1.05 \pm 0.50$	$1.32 \pm 0.36$
WalkingIreadmill_3mphIreadmill_3light	$0.66 \pm 0.37$	$0.63 \pm 0.38$	$0.88 \pm 0.45$
WalkingIreadmill_3mpnIreadmill_6moderate	$0.67 \pm 0.42$	$0.63 \pm 0.43$	$0.85 \pm 0.44$
WalkingTreadmill_3mphTreadmill_9hard	$1.28 \pm 0.79$	$1.25 \pm 0.78$	$1.49 \pm 0.90$
kneeling	$0.33 \pm 0.18$	$0.31 \pm 0.19$	$0.39 \pm 0.19$
Unknown	$1.31 \pm 0.33$	$1.02 \pm 0.24$	$3.26 \pm 0.87$
Carrying_groceries	$1.26 \pm 0.37$	$1.1/\pm0.40$	$1.70 \pm 0.45$
Doing_dishes	$0.56 \pm 0.18$	$0.55 \pm 0.18$	$0.65 \pm 0.20$
Gardening	$0.51 \pm 0.41$	$0.47 \pm 0.41$	$0.65 \pm 0.44$
Ironing	$0.55 \pm 0.18$	$0.51 \pm 0.18$	$0.72 \pm 0.27$
Making_the_bed	$0.84 \pm 0.36$	$0.70 \pm 0.30$	$1.28 \pm 0.59$
Mopping	$0.50 \pm 0.24$	$0.46 \pm 0.23$	$0.70 \pm 0.35$
Playing_videogames	$0.55 \pm 0.17$	$0.54 \pm 0.18$	$0.63 \pm 0.17$
Scrubbing_a_surface	$0.48 \pm 0.23$	$0.42 \pm 0.22$	$0.73 \pm 0.31$
Stacking_groceries	$0.66 \pm 0.30$	$0.63 \pm 0.32$	$0.79 \pm 0.30$
Sweeping	$0.47 \pm 0.28$	$0.41 \pm 0.24$	$0.72 \pm 0.44$
Typing	$0.52 \pm 0.12$	$0.50 \pm 0.12$	$0.62 \pm 0.14$
Vacuuming	$0.47 \pm 0.32$	$0.44 \pm 0.30$	$0.63 \pm 0.42$
Walking_around_block	$1.78 \pm 0.27$	$1.71 \pm 0.30$	$2.24 \pm 0.33$
Washing_windows	$0.58 \pm 0.28$	$0.51 \pm 0.27$	$0.80 \pm 0.42$
Watching_TV	$0.53\pm0.14$	$0.52\pm0.14$	$0.63\pm0.17$
Weeding	$0.46\pm0.30$	$0.39\pm0.26$	$0.64 \pm 0.44$
Wiping/Dusting	$0.69 \pm 0.37$	$0.64 \pm 0.33$	$0.91\pm0.53$
Writing	$0.54\pm0.19$	$0.53\pm0.19$	$0.61\pm0.18$
taking_out_trash	$0.65 \pm 0.23$	$0.60 \pm 0.23$	$0.90\pm0.29$

Table B6-1: Performance per activity obtained when estimating energy expenditure using a multivariable linear model per activity over the MIT dataset. The models are trained using the *ACAbsArea* feature computed per sensor over one minute sliding windows.



Figure B6-1. Plot of energy expenditure (in METS) computed using linear regression over the sum of the areas over all axis per sensor (7) and the Cosmed K4b2 Indirect Calorimeter for Subject MIT-001 over one minute windows.



Figure B6-2. Plot of energy expenditure (in METS) computed using linear regression over the sum of the areas over all axis per sensor (7) and the Cosmed K4b2 Indirect Calorimeter for Subject MIT-004 over one minute windows.



Figure B6-3. Plot of energy expenditure (in METS) computed using linear regression over the sum of the areas over all axis per sensor (7) and the Cosmed K4b2 Indirect Calorimeter for Subject MIT-018 over one minute windows.

**Appendix B7: Estimating Energy Expenditure Using One Linear Regression Model Per Known Activity** 

Activity	RMSE	MAE	MAED
Bench weight lifting - hard	0.16 ± 0.16	$0.14 \pm 0.14$	0.21 ± 0.20
Bench weight lifting - light	$1.01 \pm 0.74$	$0.94 \pm 0.66$	$1.25 \pm 0.98$
Bench weight lifting - moderate	$1.07 \pm 0.49$	$0.96 \pm 0.39$	$1.37 \pm 0.73$
Bicep curls - hard	$0.31 \pm 0.15$	$0.28 \pm 0.16$	$0.40 \pm 0.17$
Bicep curls - light	$0.62 \pm 0.40$	$0.50 \pm 0.26$	$1.05 \pm 0.97$
Bicep_curlsmoderate	$0.80 \pm 0.47$	$0.79 \pm 0.46$	$0.93 \pm 0.56$
CalisthenicsCrunches	$3.27 \pm 5.09$	$3.13 \pm 5.11$	$3.76 \pm 5.39$
CalisthenicsSit_ups	$1.54 \pm 0.56$	$1.39 \pm 0.52$	$2.03 \pm 0.79$
CyclingCycle_hardCycle_80rpm	$1.24 \pm 0.46$	$1.17 \pm 0.47$	$1.53\pm0.50$
CyclingCycle_lightCycle_100rpm	$1.07 \pm 0.57$	$1.02 \pm 0.55$	$1.37 \pm 0.73$
CyclingCycle_lightCycle_60rpm	$0.64 \pm 0.59$	$0.62 \pm 0.59$	$0.77 \pm 0.66$
CyclingCycle_lightCycle_80rpm	$0.87 \pm 0.59$	$0.82 \pm 0.60$	$1.10\pm0.67$
CyclingCycle_moderateCycle_80rpm	$1.14 \pm 0.60$	$1.07 \pm 0.59$	$1.36 \pm 0.66$
Lying_down	$0.21 \pm 0.09$	$0.18 \pm 0.09$	$0.34\pm0.16$
RowingRowing_hardRowing_30spm	$1.49 \pm 0.72$	$1.38 \pm 0.73$	$1.89\pm0.83$
RowingRowing_lightRowing_30spm	$1.51 \pm 1.13$	$1.40 \pm 1.11$	$1.89 \pm 1.21$
RowingRowing_moderateRowing_30spm	$1.49 \pm 1.18$	$1.42 \pm 1.19$	$1.87 \pm 1.30$
RunningTreadmill_4mphTreadmill_0_	$1.10 \pm 0.64$	$0.97 \pm 0.60$	$1.52\pm0.88$
RunningTreadmill_5mphTreadmill_0_	$1.16 \pm 0.61$	$1.05 \pm 0.62$	$1.54 \pm 0.74$
RunningTreadmill_6mphTreadmill_0_	$1.29 \pm 1.06$	$1.23 \pm 1.05$	$1.51 \pm 1.24$
Sitting	$0.26 \pm 0.22$	$0.25 \pm 0.21$	$0.31\pm0.28$
SittingFidget_feet_legs	$0.34 \pm 0.12$	$0.32 \pm 0.12$	$0.40 \pm 0.15$
SittingFidget_hands_arms	$0.22 \pm 0.15$	$0.21 \pm 0.15$	$0.27\pm0.18$
StairsAscend_stairs	$0.31 \pm 0.15$	$0.27 \pm 0.13$	$0.46 \pm 0.25$
StairsDescend_stairs	$2.46 \pm 2.66$	$2.39 \pm 2.67$	$2.81 \pm 2.75$
Standing	$0.30 \pm 0.28$	$0.27 \pm 0.26$	$0.40 \pm 0.35$
WalkingTreadmill_2mphTreadmill_0_	$0.49 \pm 0.36$	$0.45 \pm 0.37$	$0.65 \pm 0.38$
WalkingTreadmill_3mphTreadmill_0_	$0.57 \pm 0.49$	$0.53 \pm 0.50$	$0.75 \pm 0.56$
WalkingTreadmill_3mphTreadmill_3light	$0.62 \pm 0.49$	$0.57 \pm 0.51$	$0.81 \pm 0.54$
WalkingTreadmill_3mphTreadmill_6moderate	$0.70\pm0.54$	$0.65\pm0.54$	$0.94\pm0.63$
WalkingTreadmill_3mphTreadmill_9hard	$0.85\pm0.65$	$0.82\pm0.66$	$1.07\pm0.72$
kneeling	$0.33\pm0.26$	$0.30\pm0.26$	$0.40\pm0.30$
Unknown	$1.33 \pm 0.34$	$1.02\pm0.25$	$3.11\pm0.90$
Carrying_groceries	$0.57\pm0.25$	$0.52\pm0.26$	$0.76\pm0.33$
Doing_dishes	$0.16\pm0.12$	$0.14\pm0.12$	$0.24\pm0.17$
Gardening	$0.80\pm0.63$	$0.77\pm0.60$	$1.03\pm0.88$
Ironing	$0.34\pm0.22$	$0.29\pm0.17$	$0.50\pm0.36$
Making_the_bed	$1.02\pm0.62$	$0.91\pm0.60$	$1.42\pm0.73$
Mopping	$0.55\pm0.36$	$0.50\pm0.33$	$0.74\pm0.52$
Playing_videogames	$0.21\pm0.10$	$0.19\pm0.09$	$0.31\pm0.16$
Scrubbing_a_surface	$0.48\pm0.22$	$0.39\pm0.17$	$0.72\pm0.35$
Stacking_groceries	$0.59\pm0.23$	$0.55\pm0.26$	$0.71\pm0.19$
Sweeping	$0.35\pm0.12$	$0.30\pm0.11$	$0.52\pm0.19$
Typing	$0.22\pm0.10$	$0.19\pm0.08$	$0.32\pm0.18$
Vacuuming	$0.60\pm0.50$	$0.54\pm0.49$	$0.79\pm0.60$
Walking_around_block	$0.60\pm0.23$	$0.52\pm0.20$	$0.83\pm0.34$
Washing_windows	$0.42\pm0.25$	$0.39\pm0.23$	$0.57\pm0.34$
Watching_TV	$0.\overline{18\pm0.07}$	$0.\overline{16\pm0.07}$	$0.\overline{24\pm0.10}$
Weeding	$0.\overline{65\pm0.64}$	$0.\overline{60}\pm0.62$	$0.\overline{87}\pm0.77$
Wiping/Dusting	$0.46\pm0.27$	$0.43 \pm 0.25$	$0.63 \pm 0.37$
Writing	$0.27 \pm 0.17$	$0.25\pm0.16$	$0.35 \pm 0.25$
taking out trash	$0.37 \pm 0.11$	$0.33 \pm 0.11$	$0.53 \pm 0.14$

taking\_out\_trash $0.37 \pm 0.11$  $0.33 \pm 0.11$  $0.53 \pm 0.14$ Table B7-1: Performance per activity when estimating energy expenditure using a multivariablelinear model per activity over the MIT dataset. The models are trained using the ACAbsArea featurecomputed per sensor over one minute sliding windows.



Figure B7-1. Plot of energy expenditure (in METS) estimated using one linear regression model per activity vs. energy expenditure measured using the Cosmed K4b2 Indirect Calorimeter for Subject MIT-004. Energy is predicted using the *ACAbsArea* feature (7) computed per sensor over one minute windows when activities are assumed to be known.



Figure B7-2. Plot of energy expenditure (in METS) estimated using one linear regression model per activity vs. energy expenditure measured using the Cosmed K4b2 Indirect Calorimeter for Subject MIT-018. Energy is predicted using the *ACAbsArea* feature (7) computed per sensor over one minute windows when activities are assumed to be known.

**Appendix B8: Estimating Energy Expenditure Using One Non-Linear Regression Model Per Known Activity**
Activity	DMCE	MAE	MAED
Activity Bonch weight lifting hard	$0.10 \pm 0.21$	<b>NIAE</b> $0.17 \pm 0.10$	$0.24 \pm 0.27$
Bench_weight_intingindu	$0.19 \pm 0.21$	$0.17 \pm 0.19$	$0.24 \pm 0.27$ 1.00 ± 0.81
Bench_weight_intinglight	$0.94 \pm 0.00$	$0.91 \pm 0.03$	$1.09 \pm 0.01$ 1.12 ± 0.72
Beach avela hard	$0.93 \pm 0.38$	$0.91 \pm 0.30$	$1.12 \pm 0.73$
Dicep_curisnaru	$0.48 \pm 0.29$	$0.40 \pm 0.29$	$0.38 \pm 0.34$
Bicep_curisingnt Bicep_curisingnt	$0.05 \pm 0.43$	$0.49 \pm 0.23$	$1.14 \pm 1.04$
Galisthanian Cransha	$0.30 \pm 0.29$	$0.32 \pm 0.24$	$0.44 \pm 0.57$
CalisthenicsCrunches	$1.04 \pm 1.15$	$1.45 \pm 1.03$	$2.11 \pm 1.51$
CaustinemicsSit_ups	$1.88 \pm 1.30$	$1.77 \pm 1.30$	$2.24 \pm 1.39$
CyclingCycle_nardCycle_80rpm	$1.12 \pm 0.23$	$1.04 \pm 0.25$	$1.51 \pm 0.55$
CyclingCycle_lightCycle_100rpm	$1.10 \pm 0.52$	$1.05 \pm 0.50$	$1.41 \pm 0.05$
Cycling_Cycle_light_Cycle_001pm	$0.02 \pm 0.47$	$0.00 \pm 0.49$	$0.74 \pm 0.48$ 1.00 ± 0.54
Cycling_Cycle_lightCycle_801pm	$0.78 \pm 0.48$	$0.73 \pm 0.49$	$1.00 \pm 0.34$
CyclingCycle_moderateCycle_80rpm	$1.07 \pm 0.38$	$1.01 \pm 0.38$	$1.28 \pm 0.03$
Lyng_down Barring Barring hand Barring 20ann	$0.21 \pm 0.08$	$0.18 \pm 0.08$	$0.33 \pm 0.14$
RowingRowing_nardRowing_Suspin	$1.47 \pm 1.10$	$1.34 \pm 1.04$	$1.89 \pm 1.21$
RowingRowing_lightRowing_Suspin	$1.03 \pm 1.32$	$1.47 \pm 1.34$	$2.10 \pm 1.48$
RowingRowing_moderateRowing_30spm	$1.50 \pm 0.71$	$1.43 \pm 0.74$	$1.8/\pm 0.78$
RunningIreadmii_4mpnIreadmiii_0_	$1.02 \pm 0.60$	$0.89 \pm 0.54$	$1.43 \pm 0.85$
RunningIreadmill_SmphIreadmill_O_	$1.11 \pm 0.53$	$1.01 \pm 0.54$	$1.47 \pm 0.68$
RunningIreadmill_6mpnIreadmill_0_	$1.23 \pm 0.62$	$1.1/\pm0.60$	$1.45 \pm 0.75$
Sitting	$0.25 \pm 0.23$	$0.24 \pm 0.22$	$0.30 \pm 0.28$
SittingFidget_leet_legs	$0.31 \pm 0.12$	$0.29 \pm 0.12$	$0.37 \pm 0.13$
SittingFidget_nands_arms	$0.22 \pm 0.16$	$0.20 \pm 0.15$	$0.27 \pm 0.19$
StairsAscend_stairs	$0.36 \pm 0.21$	$0.31 \pm 0.19$	$0.53 \pm 0.30$
StairsDescend_stairs	$1.59 \pm 0.42$	$1.4/\pm 0.35$	$1.91 \pm 0.65$
Standing	$0.23 \pm 0.24$	$0.21 \pm 0.23$	$0.28 \pm 0.31$
WalkingIreadmill_2mphIreadmill_0_	$0.41 \pm 0.36$	$0.37 \pm 0.37$	$0.58 \pm 0.39$
WalkingIreadmill_3mphIreadmill_0_	$0.51 \pm 0.43$	$0.47 \pm 0.44$	$0.68 \pm 0.48$
WalkingIfeadmill_SmphIfeadmill_Slight	$0.48 \pm 0.43$	$0.43 \pm 0.44$	$0.06 \pm 0.50$
WalkingIfeadmill_3mphIfeadmill_6moderate	$0.07 \pm 0.53$	$0.62 \pm 0.54$	$0.88 \pm 0.58$
waikingireadmiii_3mpnireadmiii_9nard	$0.88 \pm 0.57$	$0.84 \pm 0.38$	$1.12 \pm 0.01$
Lineung	$0.28 \pm 0.21$	$0.20 \pm 0.20$	$0.33 \pm 0.27$
	$1.30 \pm 0.34$	$1.08 \pm 0.20$	$3.37 \pm 1.07$
Deine dieber	$0.51 \pm 0.26$	$0.46 \pm 0.27$	$0.70 \pm 0.32$
Doing_dishes	$0.15 \pm 0.09$	$0.14 \pm 0.09$	$0.23 \pm 0.12$
Gardening	$0.79 \pm 0.60$	$0.76 \pm 0.36$	$1.02 \pm 0.85$
Maling Maling	$0.34 \pm 0.23$	$0.29 \pm 0.18$	$0.30 \pm 0.39$
Making_ine_bed	$1.09 \pm 0.08$	$1.02 \pm 0.07$	$1.45 \pm 0.70$
Mopping	$0.03 \pm 0.36$	$0.57 \pm 0.32$	$0.85 \pm 0.51$
Playing_videogames	$0.21 \pm 0.10$	$0.18 \pm 0.09$	$0.31 \pm 0.17$
Strubbing_a_surface	$0.44 \pm 0.13$	$0.38 \pm 0.14$	$0.00 \pm 0.18$
Stacking_groceries	$0.53 \pm 0.31$	$0.48 \pm 0.31$	$0.05 \pm 0.35$
Sweeping	$0.33 \pm 0.11$	$0.28 \pm 0.10$	$0.49 \pm 0.17$
Typing Veguuming	$0.23 \pm 0.09$	$0.21 \pm 0.09$	$0.32 \pm 0.12$
Walting around block	$0.33 \pm 0.34$	$0.48 \pm 0.32$	$0.71 \pm 0.47$
Washing_alound_block	$0.05 \pm 0.28$	$0.34 \pm 0.23$	$0.88 \pm 0.42$
Watching_TV	$0.40 \pm 0.30$	$0.42 \pm 0.29$	$0.04 \pm 0.38$
Watching_1v	$0.10 \pm 0.07$ 0.48 ± 0.22	$0.13 \pm 0.00$	$0.22 \pm 0.09$
Wining/Dusting	$0.40 \pm 0.33$	$0.43 \pm 0.29$	$0.00 \pm 0.40$
Wiping/Dusting	$0.43 \pm 0.20$	$0.40 \pm 0.24$	$0.30 \pm 0.30$
taking out trash	$0.20 \pm 0.10$ 0.35 ± 0.13	$0.24 \pm 0.13$ 0.31 + 0.14	$0.33 \pm 0.24$ 0.50 ± 0.18

 Table B8-1: Performance per activity when estimating energy expenditure using a non-linear model (M5' model tree) per activity over the MIT dataset. The models are trained using the ACAbsArea feature computed per sensor over one minute sliding windows.



Figure B8-1: Plot of energy expenditure (in METS) estimated using one non-linear regression model per activity (M5' model tree) vs. energy expenditure measured using the Cosmed K4b2 Indirect Calorimeter for Subject MIT-004. Energy is predicted using the *ACAbsArea* feature (7) computed per sensor over one minute windows when activities are assumed to be known.



Figure B8-2: Plot of energy expenditure (in METS) estimated using one non-linear regression model per activity (M5' model tree) vs. energy expenditure measured using the Cosmed K4b2 Indirect Calorimeter for Subject MIT-018. Energy is predicted using the *ACAbsArea* feature (7) computed per sensor over one minute windows when activities are assumed to be known.

## **Appendix B9: Estimating Energy Expenditure Using Linear and Non-linear Regression Algorithms**

Activity	ID	рт	МТ	c-SVD
Bench weight lifting - hard	$0.6 \pm 0.4$	$0.6 \pm 0.3$	$0.6 \pm 0.3$	$0.7 \pm 0.4$
Bench weight lifting - light	$0.0 \pm 0.4$ 0.7 ± 0.3	$0.0 \pm 0.3$	$0.0 \pm 0.3$ 0.7 + 0.3	$0.7 \pm 0.4$
Bench weight lifting - moderate	$0.7 \pm 0.3$ 0.7 ± 0.4	$0.7 \pm 0.3$	$0.7 \pm 0.3$ $0.7 \pm 0.4$	$0.7 \pm 0.4$
Bicen curls - hard	$10 \pm 0.4$	$0.7 \pm 0.4$ 0.8 ± 0.4	$0.7 \pm 0.4$ 0.5 + 0.3	$0.7 \pm 0.5$
Bicep curls - light	$1.0 \pm 0.4$ 0.7 ± 0.2	$0.6 \pm 0.4$	$0.5 \pm 0.5$ 0.7 + 0.5	$0.5 \pm 0.5$
Bicen curls - moderate	$0.7 \pm 0.2$	$0.0 \pm 0.2$	$0.7 \pm 0.5$	$0.0 \pm 0.4$
Calisthenics - Crunches	$1.7 \pm 0.3$	$1.8 \pm 0.8$	$\frac{0.0 \pm 0.0}{2.0 \pm 1.0}$	$1.3 \pm 0.9$
Calisthenics - Sit ups	$1.7 \pm 0.9$	$1.0 \pm 0.0$	$1.0 \pm 1.0$	$1.3 \pm 0.9$
Cycling - Cycle hard - Cycle 80rpm	$1.5 \pm 0.4$ 18+10	$2.0 \pm 1.0$	$2.0 \pm 1.1$	$1.3 \pm 0.4$ 17+09
Cycling - Cycle light - Cycle 100rpm	$1.0 \pm 1.0$ $1.2 \pm 0.8$	$12 \pm 0.7$	11+04	$1.1 \pm 0.5$
Cycling - Cycle light - Cycle 60rpm	$0.8 \pm 0.5$	$0.7 \pm 0.4$	$0.8 \pm 0.5$	$0.7 \pm 0.4$
Cycling - Cycle light - Cycle 80rpm	$11 \pm 0.8$	$0.7 \pm 0.1$ 0.9 + 0.3	$12 \pm 0.5$	$0.7 \pm 0.1$ 0.9 ± 0.6
Cycling - Cycle moderate - Cycle 80rpm	$1.1 \pm 0.0$ $1.5 \pm 0.7$	$14 \pm 0.9$	$1.2 \pm 0.7$ $1.5 \pm 0.6$	$13 \pm 0.5$
Lying down	$0.4 \pm 0.7$	$0.2 \pm 0.1$	$1.0 \pm 0.0$ $1.0 \pm 2.8$	$0.3 \pm 0.1$
Rowing - Rowing hard - Rowing 30spm	$18 \pm 16$	$20 \pm 15$	$23 \pm 14$	14+12
Rowing - Rowing light - Rowing 30spm	$1.0 \pm 1.0$ 1.4 + 1.1	$1.6 \pm 1.3$	$1.8 \pm 0.9$	$1.1 \pm 1.2$ $1.3 \pm 0.8$
Rowing - Rowing moderate - Rowing 30spm	$1.7 \pm 1.1$	$21 \pm 1.6$	$22 \pm 13$	$1.3 \pm 0.0$ 1 4 + 1 2
Running - Treadmill 4mph - Treadmill 0	$1.7 \pm 1.3$ $1.3 \pm 0.7$	$1.2 \pm 0.6$	$1.3 \pm 0.7$	$1.0 \pm 0.5$
Running - Treadmill 5mph - Treadmill 0	$1.5 \pm 0.7$ $1.4 \pm 0.8$	$1.2 \pm 0.0$ $1.4 \pm 0.9$	$1.3 \pm 0.7$ $1.8 \pm 1.0$	$1.0 \pm 0.5$ 1 3 + 0 5
Running - Treadmill 6mph - Treadmill 0	$1.8 \pm 1.1$	$14 \pm 11$	$1.6 \pm 1.0$ $1.6 \pm 1.1$	$1.0 \pm 0.0$ $1.4 \pm 0.8$
Sitting	$0.7 \pm 0.2$	$0.6 \pm 0.2$	$1.0 \pm 1.1$ $1.9 \pm 4.7$	$0.5 \pm 0.2$
Sitting - Fidget feet legs	$1.3 \pm 0.5$	$1.2 \pm 0.9$	$1.3 \pm 0.6$	$0.7 \pm 0.3$
Sitting - Fidget hands arms	$0.8 \pm 0.4$	$0.6 \pm 0.2$	$0.6 \pm 0.3$	$0.7 \pm 0.3$ 0.5 ± 0.2
Stairs - Ascend stairs	$0.0 \pm 0.1$ 0.9 + 0.2	$0.0 \pm 0.2$ 0.9 + 0.2	$1.0 \pm 0.2$	$0.9 \pm 0.2$
Stairs - Descend stairs	$1.5 \pm 0.3$	$1.5 \pm 0.2$	$1.5 \pm 0.2$	14+0.5
Standing	$0.4 \pm 0.1$	$0.6 \pm 0.3$	$0.6 \pm 0.2$	$0.5 \pm 0.2$
Walking - Treadmill 2mph - Treadmill 0	$0.1 \pm 0.1$ 0.8 ± 0.5	$0.0 \pm 0.3$	$0.0 \pm 0.3$	$0.5 \pm 0.2$ 0.4 + 0.2
Walking - Treadmill 3mph - Treadmill 0	$10 \pm 0.6$	$0.0 \pm 0.3$ 0.8 ± 0.3	$0.0 \pm 0.3$ 0.8 + 0.2	$0.1 \pm 0.2$ 0.5 ± 0.2
Walking - Treadmill 3mph - Treadmill 3 - light	$0.8 \pm 0.5$	$0.7 \pm 0.2$	$0.8 \pm 0.2$	$0.5 \pm 0.2$
Walking - Treadmill 3mph - Treadmill 6 - moderate	$1.0 \pm 0.5$	$1.1 \pm 0.4$	$1.1 \pm 0.6$	$0.8 \pm 0.2$
Walking - Treadmill 3mph - Treadmill 9 - hard	$1.5 \pm 0.7$	$1.6 \pm 0.6$	$1.5 \pm 0.7$	$1.3 \pm 0.6$
kneeling	$0.5 \pm 0.2$	$0.5 \pm 0.2$	0.6 + 0.2	0.4 + 0.2
unknown	$1.5 \pm 0.3$	$1.5 \pm 0.4$	1.6 + 0.4	$1.5 \pm 0.4$
Carrying groceries	$0.9 \pm 0.2$	$0.8 \pm 0.2$	1.0 + 0.2	$0.8 \pm 0.3$
Doing dishes	$0.5 \pm 0.2$	$0.0 \pm 0.2$ 0.4 ± 0.2	$0.5 \pm 0.2$	$0.0 \pm 0.0$ 0.4 + 0.2
Gardening	$0.7 \pm 0.4$	$0.7 \pm 0.2$	0.6 + 0.2	$0.7 \pm 0.3$
Ironing	$0.5 \pm 0.2$	$0.6 \pm 0.2$	0.8 + 1.0	$0.5 \pm 0.2$
Making the bed	$1.0 \pm 0.4$	$1.0 \pm 0.4$	$1.2 \pm 0.5$	$0.9 \pm 0.3$
Mopping	$0.6 \pm 0.2$	$0.7 \pm 0.3$	0.8 + 0.3	$0.6 \pm 0.2$
Plaving videogames	$0.7 \pm 0.4$	$0.4 \pm 0.2$	1.1 + 1.9	0.4 + 0.2
Scrubbing a surface	$0.9 \pm 0.5$	$0.8 \pm 0.3$	$0.7 \pm 0.2$	$0.6 \pm 0.2$
Stacking groceries	$1.0 \pm 0.5$	$0.7 \pm 0.2$	$1.0 \pm 0.5$	$0.7 \pm 0.2$
Sweeping	$0.7 \pm 0.4$	$0.6 \pm 0.3$	$0.8 \pm 0.4$	$0.6 \pm 0.3$
Typing	$0.5 \pm 0.2$	$0.4 \pm 0.2$	0.4 + 0.2	$0.4 \pm 0.1$
Vacuuming	$0.5 \pm 0.1$	$0.5 \pm 0.2$	0.6 + 0.2	0.4 + 0.1
Walking around block	$1.1 \pm 0.5$	$1.0 \pm 0.2$	$1.3 \pm 0.9$	$0.9 \pm 0.5$
Washing windows	$0.7 \pm 0.4$	$0.6 \pm 0.3$	$0.8 \pm 0.5$	$0.5 \pm 0.2$
Watching TV	$0.7 \pm 0.5$	$0.4 \pm 0.2$	$1.1 \pm 2.0$	$0.4 \pm 0.2$
Weeding	$0.8 \pm 0.2$	$0.6 \pm 0.2$	$0.9 \pm 0.2$	$0.8 \pm 0.3$
Wiping/Dusting	$0.7 \pm 0.6$	$0.6 \pm 0.2$	$0.6 \pm 0.3$	$0.5 \pm 0.2$
Writing	$0.6 \pm 0.2$	$0.4 \pm 0.2$	$0.4 \pm 0.2$	$0.4 \pm 0.1$
taking out trash	$0.7 \pm 0.2$	$0.6 \pm 0.2$	$0.6 \pm 0.2$	$0.5 \pm 0.1$

Table B9-1: Root mean square error obtained while estimating energy expenditure using different linear and non-linear regression algorithms utilizing all the accelerometer-based features computed per sensor over sliding windows of 5.6s in length. LR stands for linear regression, RT for regression trees, MT for model trees and  $\varepsilon$ -SVR for epsilon support vector regression.

Activity	LR	RT	MT	E-SVR
Bench weight lifting - hard	$0.5 \pm 0.4$	$0.5 \pm 0.2$	$0.5 \pm 0.2$	$0.6 \pm 0.3$
Bench weight lifting - light	$0.6 \pm 0.3$	$0.6 \pm 0.3$	$0.6 \pm 0.3$	$0.6 \pm 0.4$
Bench weight lifting - moderate	$0.6 \pm 0.4$	$0.6 \pm 0.4$	$0.6 \pm 0.4$	$0.6 \pm 0.5$
Bicep curls - hard	$0.9 \pm 0.5$	$0.7 \pm 0.3$	$0.5 \pm 0.3$	$0.8 \pm 0.5$
Bicep curls - light	$0.6 \pm 0.2$	$0.5 \pm 0.2$	$0.5 \pm 0.4$	$0.6 \pm 0.3$
Bicep curls - moderate	$0.6 \pm 0.3$	$0.5 \pm 0.2$	$0.4 \pm 0.4$	$0.5 \pm 0.2$
Calisthenics - Crunches	$1.5 \pm 0.8$	$1.6 \pm 0.8$	$1.8 \pm 0.9$	$1.2 \pm 0.8$
Calisthenics - Sit ups	$1.2 \pm 0.3$	$1.4 \pm 0.6$	$1.5 \pm 0.5$	$1.1 \pm 0.4$
Cycling - Cycle hard - Cycle 80rpm	$1.7 \pm 1.0$	$1.9 \pm 1.0$	$1.9 \pm 1.0$	$1.6 \pm 0.9$
Cycling - Cycle light - Cycle 100rpm	$1.1 \pm 0.8$	$1.1 \pm 0.8$	$1.0 \pm 0.4$	$1.0 \pm 0.6$
Cycling - Cycle light - Cycle 60rpm	$0.7 \pm 0.5$	$0.6 \pm 0.4$	$0.8 \pm 0.5$	$0.6 \pm 0.4$
Cycling - Cycle light - Cycle 80rpm	$1.0 \pm 0.8$	$0.8 \pm 0.3$	$1.0 \pm 0.6$	$0.8 \pm 0.6$
Cycling - Cycle moderate - Cycle 80rpm	$1.4 \pm 0.7$	$1.3 \pm 0.9$	$1.3 \pm 0.5$	$1.2 \pm 0.5$
Lying down	$0.4 \pm 0.3$	$0.2 \pm 0.1$	$0.7 \pm 1.9$	$0.3 \pm 0.1$
Rowing - Rowing hard - Rowing 30spm	$1.7 \pm 1.5$	$1.8 \pm 1.5$	$2.1 \pm 1.4$	$1.3 \pm 1.2$
Rowing - Rowing light - Rowing 30spm	$1.2 \pm 1.0$	$1.4 \pm 1.1$	$1.6 \pm 0.9$	$1.1 \pm 0.8$
Rowing - Rowing moderate - Rowing 30spm	$1.6 \pm 1.5$	$1.9 \pm 1.6$	$1.9 \pm 1.2$	$1.3 \pm 1.2$
Running - Treadmill 4mph - Treadmill 0	$1.1 \pm 0.7$	$1.1 \pm 0.6$	$1.1 \pm 0.6$	$0.9 \pm 0.4$
Running - Treadmill 5mph - Treadmill 0	$1.2 \pm 0.8$	$1.2 \pm 1.0$	$1.4 \pm 0.9$	$1.2 \pm 0.5$
Running - Treadmill 6mph - Treadmill 0	$1.6 \pm 1.1$	$1.3 \pm 1.1$	$1.4 \pm 1.2$	$1.3 \pm 0.9$
Sitting	$0.6 \pm 0.2$	$0.4 \pm 0.2$	$0.7 \pm 1.0$	$0.4 \pm 0.2$
Sitting - Fidget feet legs	$1.2 \pm 0.5$	$1.0 \pm 0.8$	$1.1 \pm 0.5$	$0.6 \pm 0.3$
Sitting - Fidget hands arms	$0.7 \pm 0.4$	$0.5 \pm 0.2$	$0.5 \pm 0.3$	$0.4 \pm 0.2$
Stairs - Ascend stairs	$0.8 \pm 0.2$	$0.7 \pm 0.1$	$0.8 \pm 0.2$	$0.8 \pm 0.2$
Stairs - Descend stairs	$1.3 \pm 0.3$	$1.3 \pm 0.2$	$1.2 \pm 0.2$	$1.2 \pm 0.5$
Standing	$0.4 \pm 0.1$	$0.5 \pm 0.3$	$0.5 \pm 0.2$	$0.4 \pm 0.2$
Walking - Treadmill 2mph - Treadmill 0	$0.7 \pm 0.5$	$0.5 \pm 0.2$	$0.5 \pm 0.2$	$0.4 \pm 0.2$
Walking - Treadmill 3mph - Treadmill 0	$0.9 \pm 0.6$	$0.6 \pm 0.3$	$0.6 \pm 0.3$	$0.4 \pm 0.2$
Walking - Treadmill 3mph - Treadmill 3 - light	$0.7 \pm 0.5$	$0.6 \pm 0.2$	$0.7 \pm 0.3$	$0.4 \pm 0.1$
Walking - Treadmill 3mph - Treadmill 6 - moderate	$0.9 \pm 0.5$	$1.0 \pm 0.4$	$0.9 \pm 0.5$	$0.7\pm0.3$
Walking - Treadmill 3mph - Treadmill 9 - hard	$1.4 \pm 0.7$	$1.5 \pm 0.7$	$1.4 \pm 0.7$	$1.2 \pm 0.6$
kneeling	$0.4 \pm 0.2$	$0.4 \pm 0.2$	$0.4 \pm 0.1$	$0.4 \pm 0.2$
unknown	$1.2 \pm 0.2$	$1.1 \pm 0.3$	$1.2 \pm 0.3$	$1.1 \pm 0.3$
Carrying groceries	$0.7 \pm 0.2$	$0.7 \pm 0.2$	$0.8 \pm 0.1$	$0.7 \pm 0.3$
Doing dishes	$0.4 \pm 0.3$	$0.3 \pm 0.2$	$0.4 \pm 0.2$	$0.4 \pm 0.2$
Gardening	$0.6 \pm 0.4$	$0.6 \pm 0.2$	$0.6 \pm 0.2$	$0.6 \pm 0.3$
Ironing	$0.4 \pm 0.2$	$0.5 \pm 0.2$	$0.5 \pm 0.3$	$0.4 \pm 0.2$
Making the bed	$0.8 \pm 0.3$	$0.8 \pm 0.3$	$1.0 \pm 0.4$	$0.7 \pm 0.3$
Mopping	$0.5 \pm 0.2$	$0.6 \pm 0.3$	$0.6 \pm 0.3$	$0.5 \pm 0.2$
Playing videogames	$0.6 \pm 0.4$	$0.3 \pm 0.2$	$0.8 \pm 1.4$	$0.3 \pm 0.2$
Scrubbing a surface	$0.8 \pm 0.4$	$0.6 \pm 0.2$	$0.6 \pm 0.2$	$0.5\pm0.2$
Stacking groceries	$0.9\pm0.5$	$0.6 \pm 0.2$	$0.7 \pm 0.2$	$0.6 \pm 0.2$
Sweeping	$0.6 \pm 0.4$	$0.5 \pm 0.2$	$0.6 \pm 0.4$	$0.5\pm0.3$
Typing	$0.4 \pm 0.2$	$0.3 \pm 0.1$	$0.3 \pm 0.2$	$0.3 \pm 0.1$
Vacuuming	$0.4 \pm 0.1$	$0.4 \pm 0.2$	$0.5 \pm 0.2$	$0.4 \pm 0.1$
Walking around block	$1.0 \pm 0.5$	$0.8 \pm 0.2$	$0.9 \pm 0.2$	$0.8\pm0.5$
Washing windows	$0.6 \pm 0.3$	$0.5 \pm 0.2$	$0.6 \pm 0.3$	$0.4 \pm 0.2$
Watching TV	$0.6 \pm 0.5$	$0.3 \pm 0.2$	$0.8 \pm 1.5$	$0.4 \pm 0.2$
Weeding	$0.7\pm0.2$	$0.5\pm0.2$	$0.7\pm0.2$	$0.7\pm0.3$
Wiping/Dusting	$0.6 \pm 0.5$	$0.5 \pm 0.2$	$0.5\pm0.2$	$0.4 \pm 0.2$
Writing	$0.5 \pm 0.2$	$0.3 \pm 0.2$	$0.4 \pm 0.2$	$0.3 \pm 0.1$
taking out trash	$0.6 \pm 0.2$	$0.5 \pm 0.2$	$0.5 \pm 0.1$	0.4 + 0.1

 Laking out uasin
  $0.0 \pm 0.2 + 0.5 \pm 0.2 + 0.5 \pm 0.1 + 0.4 \pm 0.1$  

 Table B9-2: Mean absolute error obtained while estimating energy expenditure using different linear and non-linear regression algorithms utilizing all the accelerometer-based features computed per sensor over sliding windows of 5.6s in length.

Activity	LR	RT	MT	ε-SVR
Bench weight lifting - hard	$0.9 \pm 0.5$	$1.1 \pm 0.5$	$1.2 \pm 0.6$	$1.2 \pm 0.5$
Bench weight lifting - light	$1.4 \pm 0.5$	$1.2 \pm 0.4$	$1.4 \pm 0.5$	$1.2 \pm 0.5$
Bench weight lifting - moderate	$1.2 \pm 0.5$	$1.3 \pm 0.5$	$1.3 \pm 0.9$	$1.2 \pm 0.6$
Bicep curls - hard	$1.9 \pm 0.4$	$1.8 \pm 1.0$	$1.2 \pm 0.8$	$1.4 \pm 0.7$
Bicep curls - light	$1.7 \pm 0.4$	$1.9 \pm 1.0$	$2.6 \pm 1.6$	$1.3 \pm 1.0$
Bicep curls - moderate	$1.4 \pm 0.5$	$1.9 \pm 0.9$	$1.5 \pm 1.6$	$1.3 \pm 0.2$
Calisthenics - Crunches	$2.8 \pm 1.1$	$3.5 \pm 0.8$	$3.9 \pm 2.4$	$2.5 \pm 1.5$
Calisthenics - Sit ups	$2.4 \pm 0.7$	$2.6 \pm 0.7$	3.3 ± 1.3	$2.1 \pm 0.5$
Cycling - Cycle hard - Cycle 80rpm	$2.6 \pm 1.0$	$3.0 \pm 1.4$	$3.7 \pm 2.4$	$2.5 \pm 1.0$
Cycling - Cycle light - Cycle 100rpm	$2.1 \pm 1.1$	$2.0 \pm 0.9$	$2.3 \pm 0.9$	$2.0 \pm 0.7$
Cycling - Cycle light - Cycle 60rpm	$1.3 \pm 0.6$	$1.3 \pm 0.7$	$1.6 \pm 0.7$	$1.2 \pm 0.5$
Cycling - Cycle light - Cycle 80rpm	$1.9 \pm 1.0$	$1.8 \pm 0.6$	$2.6 \pm 2.1$	$1.5 \pm 0.7$
Cycling - Cycle moderate - Cycle 80rpm	$2.3 \pm 0.9$	$2.4 \pm 1.1$	$2.8 \pm 0.9$	$2.2 \pm 0.6$
Lying down	$1.0 \pm 0.4$	$0.5 \pm 0.2$	$1.8 \pm 4.3$	$0.7 \pm 0.2$
Rowing - Rowing hard - Rowing 30spm	$2.8 \pm 2.0$	$3.0 \pm 1.6$	$3.9 \pm 1.7$	$2.3 \pm 1.4$
Rowing - Rowing light - Rowing 30spm	$2.4 \pm 1.3$	$2.7 \pm 1.2$	$3.3 \pm 1.4$	$2.4 \pm 1.1$
Rowing - Rowing moderate - Rowing 30spm	$2.5 \pm 1.8$	$3.2 \pm 1.8$	$4.1 \pm 2.0$	$2.2 \pm 1.4$
Running - Treadmill 4mph - Treadmill 0	$2.6 \pm 1.2$	$2.4 \pm 0.9$	$3.1 \pm 1.7$	$2.1 \pm 1.0$
Running - Treadmill 5mph - Treadmill 0	$2.7 \pm 1.0$	$2.7 \pm 1.1$	$4.2 \pm 2.8$	$2.3 \pm 0.8$
Running - Treadmill 6mph - Treadmill 0	$3.3 \pm 1.3$	$2.2 \pm 1.3$	$3.0 \pm 1.8$	$2.2 \pm 1.0$
Sitting	$1.6 \pm 0.5$	$1.4 \pm 0.7$	$7.6 \pm 21.9$	$1.0 \pm 0.3$
Sitting - Fidget feet legs	$1.9 \pm 0.6$	$2.0 \pm 1.3$	$2.6 \pm 1.0$	$1.3 \pm 0.6$
Sitting - Fidget hands arms	$1.6 \pm 0.6$	$1.1 \pm 0.4$	$1.2 \pm 0.5$	$1.1 \pm 0.4$
Stairs - Ascend stairs	$1.8 \pm 0.4$	$2.0 \pm 0.4$	$2.2 \pm 0.4$	$2.0 \pm 0.5$
Stairs - Descend stairs	$2.6 \pm 0.4$	$2.9 \pm 0.5$	$3.1 \pm 0.7$	$2.6 \pm 0.8$
Standing	$1.0 \pm 0.2$	$1.4 \pm 0.7$	$1.4 \pm 0.8$	$1.0 \pm 0.4$
Walking - Treadmill 2mph - Treadmill 0	$1.5 \pm 0.6$	$1.4 \pm 0.6$	$1.5 \pm 0.8$	$0.9 \pm 0.3$
Walking - Treadmill 3mph - Treadmill 0	$1.8\pm0.6$	$1.6 \pm 0.5$	$1.8 \pm 0.4$	$1.0 \pm 0.3$
Walking - Treadmill 3mph - Treadmill 3 - light	$1.5 \pm 0.6$	$1.4 \pm 0.4$	$1.8 \pm 0.7$	$1.0 \pm 0.4$
Walking - Treadmill 3mph - Treadmill 6 - moderate	$1.8 \pm 0.7$	$1.8 \pm 0.5$	$2.1 \pm 0.8$	$1.4 \pm 0.4$
Walking - Treadmill 3mph - Treadmill 9 - hard	$2.3\pm0.8$	$2.6 \pm 0.8$	$2.8 \pm 1.1$	$2.0\pm0.8$
kneeling	$1.2 \pm 0.5$	$1.2 \pm 0.5$	$1.3 \pm 0.5$	$0.8 \pm 0.2$
unknown	$5.3\pm0.8$	$5.0 \pm 1.1$	$6.8\pm3.2$	$5.2 \pm 1.2$
Carrying groceries	$2.0\pm0.8$	$1.8 \pm 0.4$	$2.3\pm0.7$	$1.7\pm0.8$
Doing dishes	$1.0 \pm 0.5$	$0.9 \pm 0.4$	$1.1 \pm 0.4$	$0.8 \pm 0.2$
Gardening	$1.6 \pm 0.7$	$1.2 \pm 0.2$	$1.4 \pm 0.2$	$1.3 \pm 0.5$
Ironing	$1.2 \pm 0.4$	$1.3 \pm 0.5$	$3.1 \pm 6.8$	$1.1\pm0.6$
Making the bed	$2.1 \pm 1.0$	$2.7 \pm 1.3$	$3.2 \pm 1.9$	$2.0 \pm 0.7$
Mopping	$1.6\pm0.6$	$1.5 \pm 0.7$	$1.9\pm0.5$	$1.2 \pm 0.3$
Playing videogames	$1.5\pm0.6$	$1.0 \pm 0.5$	$1.9 \pm 2.8$	$0.9 \pm 0.5$
Scrubbing a surface	$2.2\pm1.3$	$2.2\pm0.9$	$1.8\pm0.7$	$1.4 \pm 0.5$
Stacking groceries	$2.1\pm0.9$	$1.9\pm0.6$	$3.0 \pm 2.4$	$1.4\pm0.4$
Sweeping	$1.5\pm0.5$	$1.5 \pm 0.7$	$2.0 \pm 1.2$	$1.3\pm0.5$
Typing	$1.1 \pm 0.4$	$0.9 \pm 0.4$	$0.9\pm0.5$	$0.9 \pm 0.4$
Vacuuming	$1.2 \pm 0.2$	$1.2 \pm 0.5$	$1.7\pm0.6$	$0.9 \pm 0.4$
Walking around block	$2.3 \pm 0.6$	$2.3 \pm 0.6$	$4.5 \pm 7.0$	$1.8 \pm 0.7$
Washing windows	$1.7\pm0.7$	$1.7 \pm 1.1$	$2.6 \pm 2.2$	$1.2 \pm 0.3$
Watching TV	$1.4 \pm 0.7$	$1.2\pm0.5$	$2.0 \pm 2.7$	$0.9 \pm 0.5$
Weeding	$1.7\pm0.7$	$1.2 \pm 0.6$	$1.9 \pm 0.8$	$1.5\pm0.5$
Wiping/Dusting	$1.6\pm0.9$	$1.3 \pm 0.4$	$1.5 \pm 0.7$	$1.1 \pm 0.4$
Writing	$1.2 \pm 0.4$	$1.0\pm0.6$	$1.0 \pm 0.5$	$0.8 \pm 0.3$
taking out trash	$1.6 \pm 0.4$	$1.6 \pm 0.4$	$1.7 \pm 0.4$	$1.2 \pm 0.3$

Table B9-3: Maximum absolute error deviation obtained while estimating energy expenditure using different linear and non-linear regression algorithms utilizing all the accelerometer-based features computed per sensor over sliding windows of 5.6s in length.

**Appendix B10: Estimating Energy Expenditure When Band-Pass Filtering the Accelerometer Signal and When Not** 

Gymnasium Activity	RMSE	RMSE	RMSE	RMSE
	ACAbsArea	ACAbsArea with	ACFFTPeaks	ACFF TPeaks
	Band-pass	Band-pass	Band-pass	Band-pass
	Filtering	Filtering	Filtering	Filtering
Bench weight lifting - hard	$0.89\pm0.69$	$0.39\pm0.25$	$0.53\pm0.27$	$0.54\pm0.28$
Bench weight lifting - light	$0.70\pm0.47$	$0.54\pm0.21$	$0.65\pm0.27$	$0.63\pm0.32$
Bench weight lifting - moderate	$0.81\pm0.59$	$0.56\pm0.25$	$0.64\pm0.38$	$0.64 \pm 0.41$
Bicep curls - hard	$1.02\pm0.37$	$1.38\pm0.57$	$1.47\pm0.65$	$1.44 \pm 0.74$
Bicep curls - light	$0.81 \pm 0.43$	$1.09 \pm 0.37$	$1.17\pm0.38$	$1.13 \pm 0.61$
Bicep curls - moderate	$0.70 \pm 0.43$	$1.32\pm0.32$	$1.37 \pm 0.41$	$1.24 \pm 0.56$
Calisthenics - Crunches	$1.19 \pm 0.54$	$1.40 \pm 0.68$	$1.44 \pm 0.63$	$1.56 \pm 0.66$
Calisthenics - Sit ups	$2.21 \pm 0.81$	$1.48 \pm 0.67$	$1.71 \pm 0.72$	$1.62 \pm 0.60$
Cycling - Cycle hard - Cycle 80rpm	$2.59 \pm 1.08$	$2.83 \pm 1.06$	$2.10 \pm 1.18$	$2.10 \pm 1.13$
Cycling - Cycle light - Cycle 100rpm	$1.82 \pm 0.88$	$1.54 \pm 0.83$	$1.33 \pm 0.62$	$1.22 \pm 0.52$
Cycling - Cycle light - Cycle 60rpm	$0.82 \pm 0.45$	$1.04 \pm 0.46$	$0.76 \pm 0.35$	$0.80 \pm 0.37$
Cycling - Cycle light - Cycle 80rpm	$1.32 \pm 0.67$	$1.39 \pm 0.63$	$1.20 \pm 0.61$	$1.24 \pm 0.68$
Cycling - Cycle moderate - Cycle 80rpm	$2.10 \pm 1.09$	$2.26 \pm 0.93$	$1.65 \pm 1.01$	$1.62 \pm 0.95$
Lying down	$2.11 \pm 1.41$	$0.62 \pm 0.15$	$0.58 \pm 0.72$	$0.35 \pm 0.06$
Rowing - Rowing hard - Rowing 30spm	$3.01 \pm 1.78$	$3.07 \pm 1.75$	$2.23 \pm 1.63$	$2.30 \pm 1.73$
Rowing - Rowing light - Rowing 30spm	$1.98 \pm 1.36$	$2.12 \pm 1.31$	$1.47 \pm 1.15$	$1.46 \pm 1.34$
Rowing - Rowing moderate - Rowing 30spm	$2.67 \pm 1.59$	$2.85 \pm 1.60$	$2.08 \pm 1.44$	$2.13 \pm 1.62$
Running - Treadmill 4mph - Treadmill 0	$1.14 \pm 0.45$	$1.22 \pm 0.50$	$1.38 \pm 0.56$	$1.42 \pm 0.70$
Running - Treadmill 5mph - Treadmill 0	$1.38 \pm 0.70$	$0.98 \pm 0.53$	$1.34 \pm 0.87$	$1.45 \pm 0.93$
Running - Treadmill 6mph - Treadmill 0	$1.37 \pm 0.75$	$1.05 \pm 0.67$	$2.02 \pm 1.51$	$2.03 \pm 1.54$
Sitting	$1.08 \pm 0.52$	$0.56 \pm 0.15$	$0.47 \pm 0.19$	$0.59 \pm 0.25$
Sitting - Fidget feet legs	$1.22 \pm 0.43$	$1.05 \pm 0.33$	$1.12 \pm 0.42$	$1.11 \pm 0.48$
Sitting - Fidget hands arms	$1.02 \pm 0.48$	$1.00 \pm 0.35$	$0.67 \pm 0.28$	$0.62 \pm 0.29$
Stairs - Ascend stairs	$0.93 \pm 0.24$	$0.87 \pm 0.23$	$0.87 \pm 0.21$	$0.91 \pm 0.27$
Stairs - Descend stairs	$1.30 \pm 0.37$	$1.35 \pm 0.30$	$1.44 \pm 0.32$	$1.44 \pm 0.35$
Standing	$1.26 \pm 0.31$	$0.58 \pm 0.13$	$0.60 \pm 0.17$	$0.56 \pm 0.12$
Walking - Treadmill 2mph - Treadmill 0	$0.66 \pm 0.31$	$0.84 \pm 0.35$	$0.77 \pm 0.29$	$0.78 \pm 0.30$
Walking - Treadmill 3mph - Treadmill 0	$0.68 \pm 0.31$	$0.97 \pm 0.42$	$0.87 \pm 0.44$	$0.89 \pm 0.40$
Walking - Treadmill 3mph - Treadmill 3 - light	$0.55 \pm 0.32$	$0.57 \pm 0.28$	$0.66 \pm 0.29$	$0.66 \pm 0.28$
Walking - Treadmill 3mph - Treadmill 6 - moderate	$1.13 \pm 0.45$	$0.73 \pm 0.43$	$0.97 \pm 0.44$	$0.96 \pm 0.39$
Walking - Treadmill 3mph - Treadmill 9 - hard	$1.93 \pm 0.63$	$1.45 \pm 0.72$	$1.59 \pm 0.64$	$1.64 \pm 0.62$
Kneeling	$2.02 \pm 0.53$	$0.48 \pm 0.19$	$0.44 \pm 0.15$	$0.56 \pm 0.18$
Unknown	$1.73 \pm 0.45$	$1.53 \pm 0.35$	$1.61 \pm 0.36$	$1.55 \pm 0.36$
Carrying groceries	$1.05 \pm 0.66$	$1.14 \pm 0.29$	$0.99 \pm 0.33$	$0.91 \pm 0.26$
Doing dishes	$1.20 \pm 1.12$	$0.60 \pm 0.16$	$0.77 \pm 0.51$	$0.53 \pm 0.09$
Gardening	$1.62 \pm 1.10$	$0.56 \pm 0.34$	$0.85 \pm 0.40$	$0.55 \pm 0.20$
Ironing	$1.11 \pm 0.88$	$0.62 \pm 0.17$	$0.92 \pm 0.36$	$0.60 \pm 0.15$
Making the bed	$1.19 \pm 0.72$	$0.93 \pm 0.30$	$1.01 \pm 0.41$	$0.91 \pm 0.31$
Mopping	$1.13 \pm 1.16$	$0.58 \pm 0.22$	$0.74 \pm 0.61$	$0.63 \pm 0.17$
Playing videogames	$1.93 \pm 1.77$	$0.68 \pm 0.17$	$0.92 \pm 0.98$	$0.46 \pm 0.13$
Scrubbing a surface	$1.10 \pm 0.64$	$0.61 \pm 0.18$	$1.00 \pm 0.45$	$0.71 \pm 0.27$
Stacking groceries	$1.25 \pm 1.34$	$0.78 \pm 0.28$	$1.10 \pm 0.85$	$1.08 \pm 0.52$
Sweeping	$1.0/\pm 1.3/$	$0.56 \pm 0.30$	$0.87 \pm 0.99$	$0.59 \pm 0.25$
1 yping	$1.41 \pm 0.98$	$0.65 \pm 0.12$	$0.71 \pm 0.34$	$0.4/\pm0.10$
Vacuuming	$0.92 \pm 0.82$	$0.54 \pm 0.26$	$0.60 \pm 0.34$	$0.50 \pm 0.20$
waiking around block	$1.39 \pm 1.01$	$1.63 \pm 0.29$	$1.39 \pm 0.64$	$1.22 \pm 0.33$
Washing Windows	$0.9/\pm 0./4$	$0.62 \pm 0.20$	$0.98 \pm 0.48$	$0.6/\pm0.22$
watching I V	$2.00 \pm 1.83$	$0.6/\pm 0.14$	$0.84 \pm 0.92$	$0.48 \pm 0.16$
Weeding Wining/Dusting	$1.8/\pm 1.1/$	$0.5 / \pm 0.31$	$0.83 \pm 0.50$	$0.04 \pm 0.22$
wiping/Dusting	$0.89 \pm 0.98$	$0.70 \pm 0.29$	$0.90 \pm 0.48$	$0.00 \pm 0.22$
Writing Taking out trach	$1.27 \pm 1.03$	$0.05 \pm 0.19$	$0.76 \pm 0.35$	$0.52 \pm 0.16$
Taking out trash	$1.04 \pm 0.77$	$0.78 \pm 0.20$	$0.97 \pm 0.33$	$0.79 \pm 0.23$

Table B-10: Performance of multivariable linear regression per activity in predicting energy expenditure when band-pass filtering is applied and not. The features utilized are the *ACAbsArea* feature and the *ACFFTPeaks* feature computed per sensor over windows of 5.6s in length.

## **Appendix B11: Feature Computation per Sensor vs. Feature Computation per Axis for Energy Expenditure Estimation**

Activity	Feature C	omputation	Per Sensor	Feature (	Computation	Per Axis
· ·	RMSE	MAE	MAED	RMSE	MAE	MAED
Bench weight lifting - hard	$0.4 \pm 0.2$	$0.3 \pm 0.2$	$0.8 \pm 0.5$	$0.6 \pm 0.5$	$0.6 \pm 0.5$	$1.1 \pm 0.6$
Bench weight lifting - light	$0.5 \pm 0.2$	$0.5 \pm 0.2$	$0.9 \pm 0.3$	$0.8 \pm 0.5$	$0.7 \pm 0.5$	$1.1 \pm 0.5$
Bench weight lifting - moderate	$0.6 \pm 0.2$	$0.5 \pm 0.2$	$0.8 \pm 0.3$	$0.7 \pm 0.5$	$0.7 \pm 0.5$	$1.1 \pm 0.5$
Bicep curls - hard	$1.4 \pm 0.6$	$1.4 \pm 0.6$	$1.8 \pm 0.6$	$1.0 \pm 0.9$	$1.0 \pm 0.9$	$1.4 \pm 1.0$
Bicep curls - light	$1.1 \pm 0.4$	$1.0 \pm 0.4$	$1.6 \pm 0.4$	$1.0 \pm 0.7$	$1.0 \pm 0.7$	$1.5 \pm 0.9$
Bicep curls - moderate	$1.3 \pm 0.3$	$1.3 \pm 0.3$	$1.8 \pm 0.2$	$1.0 \pm 0.8$	$0.9 \pm 0.8$	$1.3 \pm 0.8$
Calisthenics - Crunches	$1.4 \pm 0.7$	$1.2 \pm 0.7$	$2.1 \pm 0.8$	$1.6 \pm 0.8$	$1.5 \pm 0.8$	2.4 + 1.2
Calisthenics - Sit ups	$1.5 \pm 0.7$	$1.3 \pm 0.6$	$2.4 \pm 0.7$	$1.0 \pm 0.0$ $1.4 \pm 0.5$	$1.2 \pm 0.5$	$2.4 \pm 0.8$
Cycling - Cycle hard - Cycle 80rpm	$2.8 \pm 1.1$	$2.8 \pm 1.1$	$3.4 \pm 1.0$	$24 \pm 0.9$	$24 \pm 0.9$	$3.0 \pm 1.0$
Cycling - Cycle light - Cycle 100rpm	$1.5 \pm 0.8$	$1.5 \pm 0.9$	$2.1 \pm 0.8$	$15 \pm 16$	$13 \pm 16$	$2.3 \pm 1.0$
Cycling - Cycle light - Cycle 60mm	$1.0 \pm 0.5$	$1.0 \pm 0.5$	$1.3 \pm 0.4$	$0.6 \pm 0.3$	$0.6 \pm 0.3$	$1.0 \pm 0.3$
Cycling - Cycle light - Cycle 80rpm	$1.4 \pm 0.6$	$1.3 \pm 0.6$	$1.8 \pm 0.6$	$0.0 \pm 0.5$ 0.8 ± 0.6	$0.0 \pm 0.5$	$1.0 \pm 0.0$ $1.3 \pm 0.8$
Cycling - Cycle moderate - Cycle 80rpm	$2.3 \pm 0.9$	$2.2 \pm 0.9$	$2.8 \pm 1.0$	$1.7 \pm 0.9$	$1.6 \pm 0.9$	$1.3 \pm 0.0$ 2 3 + 1 0
Lying down	$0.6 \pm 0.2$	$0.6 \pm 0.2$	$0.8 \pm 0.1$	$1.7 \pm 0.2$	$1.0 \pm 0.2$	$2.5 \pm 1.0$
Rowing - Rowing hard - Rowing 30spm	$3.0 \pm 0.2$ $3.1 \pm 1.8$	$3.0 \pm 1.6$	$38 \pm 20$	$0.4 \pm 0.2$ 2 3 + 1 5	$0.4 \pm 0.2$ 2 2 + 1 4	$3.1 \pm 1.7$
Rowing - Rowing light Rowing 30spm	21+13	$2.0 \pm 1.2$	28+16	$2.5 \pm 1.5$ 1.5 ± 1.1	$1.3 \pm 1.9$	$3.1 \pm 1.7$
Rowing - Rowing nght - Rowing Sospin	$2.1 \pm 1.5$ 2.8 ± 1.6	$2.0 \pm 1.2$ 2.7 ± 1.6	$35 \pm 1.0$	$1.3 \pm 1.1$ 2.1 ± 1.4	$1.3 \pm 1.0$ 1.0 ± 1.4	$2.4 \pm 1.2$ 2.8 ± 1.7
Rowing - Rowing moderate - Rowing Sospin	$1.2 \pm 0.5$	$1.0 \pm 0.5$	$3.5 \pm 1.0$ 2.6 + 1.1	$2.1 \pm 1.4$ 1.2 ± 0.5	$1.9 \pm 1.4$ $1.0 \pm 0.5$	$2.8 \pm 1.7$
Running - Headmin 4mph - Headmin 0	$1.2 \pm 0.5$ $1.0 \pm 0.5$	$1.0 \pm 0.5$	$1.0 \pm 1.1$	$1.5 \pm 0.3$	$1.0 \pm 0.3$	$2.6 \pm 1.2$
Running - Treadmill Simpli - Treadmill 0	$1.0 \pm 0.3$	$0.9 \pm 0.3$	$1.9 \pm 1.1$ 1.9 ± 1.1	$1.2 \pm 0.7$	$1.0 \pm 0.0$	$2.1 \pm 0.8$
Running - Treadmin ompir - Treadmin 0	$1.0 \pm 0.7$	1.0 ± 0.7	1.8 ± 1.1	$1.3 \pm 0.8$	$1.4 \pm 0.8$	$2.4 \pm 1.4$
Sitting	$0.0 \pm 0.2$	$0.3 \pm 0.2$	$0.8 \pm 0.1$	$0.4 \pm 0.2$	$0.4 \pm 0.2$	$0.6 \pm 0.2$
Sitting - Flaget feet legs	1.0 ± 0.3	1.0 ± 0.3	1.4 ± 0.4	$0.7 \pm 0.4$	$0.7 \pm 0.4$	$1.0 \pm 0.4$
Sitting - Fidget hands arms	$1.0 \pm 0.4$	$1.0 \pm 0.4$	$1.4 \pm 0.3$	$0.4 \pm 0.2$	$0.4 \pm 0.2$	$0.7 \pm 0.3$
Stairs - Ascend stairs	$0.9 \pm 0.2$	$0.8 \pm 0.2$	$1.6 \pm 0.4$	$0.9 \pm 0.2$	$0.7 \pm 0.2$	$1.8 \pm 0.4$
Stairs - Descend stairs	$1.4 \pm 0.3$	$1.2 \pm 0.3$	$2.3 \pm 0.3$	$1.4 \pm 0.4$	$1.2 \pm 0.4$	$2.5 \pm 0.4$
Standing	$0.6 \pm 0.1$	$0.6 \pm 0.1$	$0.9 \pm 0.2$	$0.4 \pm 0.2$	$0.4 \pm 0.2$	$0.7 \pm 0.2$
Walking - Treadmill 2mph - Treadmill 0	$0.8 \pm 0.4$	$0.8 \pm 0.4$	$1.4 \pm 0.4$	$0.6 \pm 0.2$	$0.6 \pm 0.2$	$1.1 \pm 0.4$
Walking - Treadmill 3mph - Treadmill 0	$1.0 \pm 0.4$	$0.9 \pm 0.4$	$1.6 \pm 0.6$	$0.8 \pm 0.4$	$0.8 \pm 0.4$	$1.4 \pm 0.5$
Walking - Treadmill 3mph - Treadmill 3 - light	$0.6 \pm 0.3$	0.5 ± 0.3	$1.0 \pm 0.4$	$0.5 \pm 0.2$	$0.4 \pm 0.2$	$0.9 \pm 0.3$
Walking - Treadmill 3mph - Treadmill 6 - moderate	0.7 ± 0.4	0.7±0.4	1.1 ± 0.4	$0.9 \pm 0.4$	$0.8 \pm 0.4$	$1.3 \pm 0.5$
Walking - Treadmill 3mph - Treadmill 9 - hard	$1.4 \pm 0.7$	$1.4 \pm 0.7$	1.9 ± 0.8	$1.6 \pm 0.5$	$1.6 \pm 0.5$	$2.2 \pm 0.6$
kneeling	0.5 ± 0.2	$0.4 \pm 0.2$	0.9 ± 0.3	$0.4 \pm 0.2$	$0.3 \pm 0.2$	$0.7 \pm 0.3$
unknown	1.5 ± 0.4	$1.2 \pm 0.2$	4.9 ± 1.1	$1.6 \pm 0.4$	$1.2 \pm 0.3$	$5.4 \pm 1.3$
Carrying groceries	$1.1 \pm 0.3$	$1.0 \pm 0.3$	$2.2 \pm 0.6$	$0.9 \pm 0.4$	$0.8 \pm 0.4$	$2.1 \pm 0.7$
Doing dishes	0.6 ± 0.2	0.6 ± 0.2	$1.0 \pm 0.4$	$0.3 \pm 0.1$	$0.3 \pm 0.1$	$0.7 \pm 0.2$
Gardening	0.6 ± 0.3	0.5 ± 0.3	$1.1 \pm 0.5$	$0.6 \pm 0.2$	$0.5 \pm 0.2$	$1.2 \pm 0.3$
Ironing	0.6 ± 0.2	0.6 ± 0.2	$1.1 \pm 0.4$	$0.4 \pm 0.2$	$0.3 \pm 0.1$	$1.0 \pm 0.6$
Making the bed	$0.9 \pm 0.3$	$0.8 \pm 0.2$	$2.0 \pm 0.7$	$1.0 \pm 0.3$	$0.9 \pm 0.3$	$1.9 \pm 0.4$
Mopping	$0.6 \pm 0.2$	$0.5 \pm 0.2$	$1.3 \pm 0.5$	$0.9 \pm 0.6$	$0.8\pm0.6$	$1.5 \pm 0.8$
Playing videogames	$0.7 \pm 0.2$	$0.7 \pm 0.2$	$1.0 \pm 0.3$	$0.4 \pm 0.2$	$0.4 \pm 0.2$	$0.7 \pm 0.2$
Scrubbing a surface	$0.6 \pm 0.2$	$0.5 \pm 0.2$	$1.4 \pm 0.3$	$0.9 \pm 0.4$	$0.8 \pm 0.4$	$1.6 \pm 0.6$
Stacking groceries	$0.8 \pm 0.3$	$0.7 \pm 0.3$	$1.4 \pm 0.3$	$1.0 \pm 0.5$	$0.8 \pm 0.4$	$2.0 \pm 1.1$
Sweeping	$0.6 \pm 0.3$	$0.4 \pm 0.2$	$1.1 \pm 0.4$	$0.7 \pm 0.4$	$0.6 \pm 0.4$	$1.4 \pm 0.5$
Typing	$0.6 \pm 0.1$	$0.6 \pm 0.1$	$0.9 \pm 0.2$	$0.3 \pm 0.1$	$0.3 \pm 0.1$	$0.5\pm0.2$
Vacuuming	$0.5 \pm 0.3$	$0.5\pm0.3$	$1.2 \pm 0.4$	$0.8\pm0.6$	$0.7\pm0.6$	$1.5 \pm 0.8$
Walking around block	$1.6 \pm 0.3$	$1.5\pm0.3$	$2.7 \pm 0.3$	$1.4 \pm 0.4$	$1.2\pm0.5$	$2.5\pm0.6$
Washing windows	$0.6 \pm 0.2$	$0.5 \pm 0.2$	$1.3 \pm 0.4$	$0.6 \pm 0.4$	$0.6 \pm 0.3$	$1.3\pm0.6$
Watching TV	$0.7 \pm 0.1$	$0.6 \pm 0.1$	$1.0 \pm 0.3$	$0.5\pm0.2$	$0.4 \pm 0.2$	$0.8 \pm 0.2$
Weeding	$0.6 \pm 0.3$	$0.5 \pm 0.3$	$1.3 \pm 0.8$	$0.7 \pm 0.4$	$0.6 \pm 0.4$	$1.3 \pm 0.6$
Wiping/Dusting	$0.7 \pm 0.3$	$0.6\pm0.3$	$1.6 \pm 0.5$	$0.4 \pm 0.2$	$0.4 \pm 0.2$	$1.0 \pm 0.6$
Writing	$0.6 \pm 0.2$	$0.6 \pm 0.2$	$0.8 \pm 0.2$	$0.4 \pm 0.2$	$0.4 \pm 0.2$	$0.5 \pm 0.2$
taking out trash	$0.8 \pm 0.2$	$0.6 \pm 0.2$	$1.7 \pm 0.4$	$0.7 \pm 0.4$	$0.6 \pm 0.4$	$1.4 \pm 0.5$

Table B11-1: Performance obtained using multivariable linear regression for estimating energy expenditure using all the accelerometer-based features computed per sensor and per axis over sliding windows of 5.6s in length.

**Appendix B12: Window Length Selection for Energy Expenditure Estimation** 

Activity	1.4s	2.8s	5.6s	11.3s	22.7s	45.5s	91.0s
Category							
All	$1.38\pm0.30$	$1.37\pm0.30$	$1.36\pm0.30$	$1.34\pm0.30$	$1.32\pm0.30$	$1.27\pm0.27$	$1.16\pm0.24$
	$(1.04 \pm 0.17)$	$(1.03 \pm 0.17)$	$(1.02 \pm 0.17)$	$(1.01 \pm 0.17)$	$(0.98 \pm 0.17)$	$(0.9 \pm 0.16)$	$(0.87 \pm 0.15)$
Postures	0.8±0.2	0.7±0.2	0.7±0.2	0.7±0.2	0.6±0.2	0.6±0.2	0.5±0.2
	(0.7±0.2)	(0.7±0.2)	(0.7±0.2)	(0.7±0.2)	(0.6±0.2)	(0.6±0.2)	(0.5±0.2)
Ambulation	1.1±0.4	1.1±0.4	1.1±0.4	1.1±0.4	1.1±0.5	1.0±0.5	0.8±0.4
	(1.0±0.4)	(1.0±0.4)	(1.0±0.4)	(1.0±0.4)	(1.0±0.5)	(1.0±0.5)	(0.8±0.5)
Exercise	1.5±0.7	1.5±0.7	1.5±0.7	1.5±0.7	1.5±0.7	1.4±0.7	1.3±0.7
	(1.4±0.7)	(1.4±0.7)	(1.4±0.7)	(1.4±0.7)	(1.4±0.7)	(1.4±0.7)	(1.3±0.7)
Resistance	1.4±0.6	1.4±0.6	1.4±0.6	1.3±0.6	1.3±0.6	1.3±0.6	1.2±0.6
Exercise	(1.3±0.6)	(1.3±0.6)	(1.3±0.6)	(1.3±0.6)	(1.3±0.6)	(1.2±0.6)	(1.2±0.6)
Household	0.8±0.2	0.8±0.2	0.7±0.2	0.7±0.2	0.7±0.2	0.7±0.3	0.7±0.3
	(0.7±0.2)	$(0.7\pm0.2)$	$(0.6\pm0.2)$	$(0.6\pm0.2)$	$(0.6\pm0.2)$	$(0.6\pm 0.3)$	$(0.6\pm0.3)$

Table B12-1: Root mean squared error and mean absolute error (shown in parenthesis) obtained by estimating energy expenditure in a subject independent manner using multivariable linear regression and the *ACAbsArea* feature computed per sensor over window of varying lengths.

Activity	1.4s	2.8s	5.6s	11.3s	22.7s	45.5s	91.0s
Category							
All	$1.29\pm0.28$	$1.28\pm0.29$	$1.28\pm0.29$	$1.25 \pm 0.30$	$1.22 \pm 0.29$	$1.21 \pm 0.29$	$1.13 \pm 0.29$
	$(0.94 \pm 0.16)$	$(0.93 \pm 0.17)$	$(0.93 \pm 0.17)$	$(0.91 \pm 0.18)$	$(0.88 \pm 0.18)$	$(0.87 \pm 0.18)$	$(0.79 \pm 0.16)$
Postures	0.7±0.2	0.6±0.2	0.6±0.2	0.7±0.3	0.6±0.3	0.7±0.3	0.5±0.3
	(0.6±0.2)	(0.6±0.2)	(0.6±0.2)	(0.6±0.3)	(0.6±0.3)	(0.6±0.3)	(0.5±0.3)
Ambulation	1.2±0.5	1.2±0.5	1.2±0.5	1.2±0.6	1.2±0.7	1.2±0.7	1.0±0.6
	(1.1±0.5)	(1.1±0.5)	(1.1±0.5)	(1.0±0.6)	(1.1±0.7)	(1.1±0.7)	(0.9±0.6)
Exercise	1.3±0.7	1.3±0.7	1.3±0.7	1.3±0.8	1.3±0.8	1.3±0.9	1.2±0.9
	(1.2±0.7)	(1.2±0.7)	(1.2±0.7)	(1.2±0.8)	(1.2±0.8)	(1.2±0.9)	(1.2±0.9)
Resistance	1.2±0.6	1.2±0.6	1.2±0.6	1.1±0.7	1.1±0.7	1.1±0.7	1.0±0.7
Exercise	(1.1±0.6)	(1.0±0.6)	(1.0±0.6)	(1.0±0.7)	(1.0±0.7)	(1.0±0.7)	(0.9±0.7)
Household	0.8±0.2	0.7±0.2	0.7±0.2	0.6±0.2	0.6±0.2	0.6±0.3	0.4±0.3
	(0.6±0.2)	(0.6±0.2)	(0.6±0.2)	(0.5±0.2)	(0.5±0.2)	(0.5±0.2)	(0.4±0.2)

Table B12-2: Root mean squared error and mean absolute error (shown in parenthesis) obtained by estimating energy expenditure in a subject independent manner using multivariable linear regression and the *FFTCorr* (*ACFFTPeaks* + *ACCorr*) feature set computed per sensor over windows of varying lengths.

Activity	5.65	11.3s	22.78	45.55	91.0s
Category	2.05	11.05		10105	21005
All	$1.24 \pm 0.29$	$1.29\pm0.31$	$1.24 \pm 0.27$	$1.13 \pm 0.23$	$0.97\pm0.22$
	$(0.86 \pm 0.19)$	$(0.90 \pm 0.20)$	$(0.87 \pm 0.18)$	$(0.8 \pm 0.2)$	$(0.7 \pm 0.2)$
Postures	0.6±0.3	0.6±0.3	0.5±0.3	0.6±0.4	0.4±0.3
	(0.5±0.2)	(0.5±0.3)	(0.5±0.3)	(0.5±0.3)	(0.4±0.3)
Ambulation	1.1±0.4	1.2±0.5	1.2±0.6	1.1±0.5	0.8±0.5
	(0.9±0.4)	(1.1±0.5)	(1.1±0.6)	(1.0±0.5)	(0.8±0.5)
Exercise	1.3±0.7	1.3±0.7	1.2±0.7	1.2±0.6	0.9±0.6
	(1.2±0.7)	(1.2±0.7)	(1.1±0.7)	(1.1±0.6)	(0.9±0.6)
Resistance	1.1±0.6	1.1±0.6	1.1±0.6	1.0±0.5	0.8±0.5
Exercise	(1.0±0.6)	(1.0±0.6)	(1.0±0.6)	(0.9±0.5)	(0.7±0.5)
Household	0.6±0.3	0.6±0.3	0.6±0.3	0.6±0.3	0.6±0.3
	$(0.5\pm0.2)$	$(0.5\pm0.3)$	$(0.5\pm0.3)$	$(0.5\pm0.3)$	(0.6±0.3)

Table B12-3: Root mean squared error and mean absolute error (shown in parenthesis) obtained by estimating energy expenditure in a subject independent manner using M5' model trees and the *ACAbsArea* feature computed per sensor over windows of varying lengths

Activity	5.6s	11.3s	22.7s	45.5s	91.0s
Category					
All	$1.26\pm0.33$	$1.23\pm0.34$	$1.26\pm0.35$	$1.19\pm0.35$	$1.03\pm0.24$
	$(0.88 \pm 0.21)$	$(0.88 \pm 0.22)$	$(0.87 \pm 0.21)$	$(0.83 \pm 0.20)$	$(0.74 \pm 0.17)$
Postures	0.6±0.3	0.6±0.3	0.6±0.3	$0.5\pm0.4$	0.5±0.4
	(0.5±0.2)	(0.4±0.2)	(0.5±0.3)	(0.4±0.3)	(0.5±0.4)
Ambulation	1.3±0.5	1.2±0.4	1.2±0.7	1.2±0.6	0.8±0.5
	(1.1±0.5)	(1.0±0.4)	(1.1±0.7)	(1.1±0.6)	(0.8±0.5)
Exercise	1.3±0.7	1.3±0.7	1.3±0.9	$1.2\pm0.8$	1.0±0.8
	(1.1±0.7)	(1.1±0.6)	(1.2±0.9)	(1.1±0.8)	(1.0±0.8)
Resistance	1.1±0.5	1.1±0.6	1.1±0.7	1.0±0.7	0.9±0.7
Exercise	(1.0±0.5)	(0.9±0.6)	(1.0±0.7)	(0.9±0.6)	(0.8±0.7)
Household	0.6±0.2	0.7±0.3	0.6±0.3	0.6±0.3	0.5±0.3
	(0.5±0.2)	(0.5±0.2)	(0.5±0.2)	(0.5±0.3)	(0.5±0.3)

Table B12-4: Root mean squared error and mean absolute error (shown in parenthesis) obtained by estimating energy expenditure in a subject independent manner using M5' model trees and the *FFTCorr* (*ACFFTPeaks* + *ACCorr*) feature computed per sensor over windows of varying lengths.

Activity	Root Mean Squared Error						
	1.4s	2.8s	5.6s	11.3s	22.7s	44.5s	90s
Bench weight lifting - hard	$0.6 \pm 0.2$	$0.6 \pm 0.2$	$0.6 \pm 0.2$	$0.5 \pm 0.4$	$0.5 \pm 0.4$	$0.4 \pm 0.5$	$0.5 \pm 0.4$
Bench weight lifting - light	$0.6 \pm 0.2$	$0.6 \pm 0.3$	$0.6 \pm 0.3$	$0.7 \pm 0.4$	$0.6 \pm 0.4$	$0.7 \pm 0.5$	$0.8 \pm 0.5$
Bench weight lifting - moderate	$0.7 \pm 0.2$	$0.7 \pm 0.3$	$0.7 \pm 0.3$	$0.7 \pm 0.4$	$0.7 \pm 0.4$	$0.7 \pm 0.3$	$0.8 \pm 0.4$
Bicep curls - hard	$1.1 \pm 0.4$	$1.2 \pm 0.6$	$1.2 \pm 0.6$	$1.4 \pm 0.8$	$1.2 \pm 0.7$	$1.3 \pm 0.8$	$1.2 \pm 0.7$
Bicep curls - light	$0.9 \pm 0.3$	$0.9 \pm 0.3$	$0.9 \pm 0.3$	$1.0 \pm 0.6$	$0.9 \pm 0.5$	$1.3 \pm 0.8$	$1.2 \pm 0.9$
Bicep curls - moderate	$0.9 \pm 0.1$	$0.8 \pm 0.3$	$0.8 \pm 0.3$	$1.2 \pm 0.5$	$1.0 \pm 0.3$	$0.9 \pm 0.6$	$0.6 \pm 0.5$
Calisthenics - Crunches	$1.6 \pm 1.0$	$1.5 \pm 0.9$	$1.5 \pm 0.9$	$1.6 \pm 0.8$	$1.5 \pm 0.6$	$1.7 \pm 0.7$	$1.4 \pm 0.9$
Calisthenics - Sit ups	$1.3 \pm 0.4$	$1.3 \pm 0.5$	$1.3 \pm 0.5$	$1.2 \pm 0.3$	$1.2 \pm 0.3$	$1.2 \pm 0.5$	$0.9 \pm 0.3$
Cycling - Cycle hard - Cycle 80rpm	$2.0 \pm 1.1$	$2.0 \pm 1.1$	$2.0 \pm 1.1$	$1.9 \pm 1.1$	$1.9 \pm 1.0$	$1.9 \pm 1.1$	$1.7 \pm 1.3$
Cycling - Cycle light - Cycle 100rpm	$13 \pm 0.7$	$13 \pm 0.6$	$13 \pm 0.6$	$13 \pm 0.7$	$12 \pm 0.8$	$12 \pm 07$	$0.9 \pm 0.8$
Cycling - Cycle light - Cycle 60rpm	$0.8 \pm 0.3$	$0.8 \pm 0.4$	$0.8 \pm 0.4$	$0.9 \pm 0.4$	$0.9 \pm 0.4$	$1.0 \pm 0.4$	$1.0 \pm 0.6$
Cycling - Cycle light - Cycle 80rpm	$1.2 \pm 0.5$	$1.2 \pm 0.6$	$1.2 \pm 0.6$	$1.3 \pm 0.7$	$1.3 \pm 0.7$	$1.2 \pm 0.7$	$1.4 \pm 0.9$
Cycling - Cycle moderate - Cycle 80rpm	$1.5 \pm 0.9$	$1.6 \pm 0.9$	$1.6 \pm 0.9$	$1.6 \pm 0.9$	$1.5 \pm 0.9$	$1.4 \pm 0.9$	$1.3 \pm 1.1$
Lying down	$0.4 \pm 0.1$	$0.3 \pm 0.1$	$0.3 \pm 0.1$	$0.3 \pm 0.1$	$0.3 \pm 0.1$	$0.3 \pm 0.1$	$0.3 \pm 0.1$
Rowing - Rowing hard - Rowing 30spm	$22 \pm 18$	$21 \pm 1.8$	$21 \pm 1.8$	$19 \pm 19$	$18 \pm 17$	$1.7 \pm 1.8$	$15 \pm 0.1$
Rowing - Rowing light - Rowing 30spm	$15 \pm 13$	14 + 13	14 + 13	$1.9 \pm 1.9$ $1.4 \pm 1.3$	$1.0 \pm 1.7$ $1.3 \pm 1.2$	$1.7 \pm 1.0$ $1.3 \pm 1.2$	$1.0 \pm 1.0$ $1.0 \pm 1.0$
Rowing - Rowing moderate - Rowing 30spm	$2.0 \pm 1.6$	$2.0 \pm 1.7$	$2.0 \pm 1.7$	$1.1 \pm 1.0$ $1.8 \pm 1.7$	$1.7 \pm 1.6$	$1.0 \pm 1.2$ $1.7 \pm 1.8$	$1.0 \pm 1.0$ $1.5 \pm 1.5$
Running - Treadmill 4mph - Treadmill 0	$14 \pm 0.6$	$14 \pm 0.7$	$14 \pm 0.7$	$1.6 \pm 0.8$	$1.7 \pm 0.9$	$1.7 \pm 1.0$	$1.5 \pm 1.0$ $1.5 \pm 1.1$
Running - Treadmill 5mph - Treadmill 0	$1.1 \pm 0.6$	$14 \pm 0.8$	$1.1 \pm 0.17$ $1.4 \pm 0.8$	$1.0 \pm 0.0$ $1.4 \pm 1.0$	$1.7 \pm 0.9$	$1.7 \pm 1.1$ $1.6 \pm 1.4$	$1.8 \pm 1.6$
Running - Treadmill 6mph - Treadmill 0	$1.5 \pm 0.0$ $1.7 \pm 1.3$	$1.8 \pm 1.2$	$1.1 \pm 0.0$ $1.8 \pm 1.2$	$1.9 \pm 1.3$	$2.0 \pm 1.6$	$2.2 \pm 2.1$	$2.2 \pm 1.1$
Sitting	$0.5 \pm 0.1$	$0.6 \pm 0.2$	$0.6 \pm 0.2$	$0.8 \pm 0.3$	$0.8 \pm 0.4$	$0.9 \pm 0.4$	$0.7 \pm 0.4$
Sitting - Fidget feet legs	$12 \pm 0.4$	11 + 04	$11 \pm 04$	$10 \pm 0.5$	$0.0 \pm 0.1$	$0.9 \pm 0.6$	$0.7 \pm 0.1$
Sitting - Fidget hands arms	$0.8 \pm 0.4$	$0.7 \pm 0.4$	$0.7 \pm 0.4$	$0.6 \pm 0.3$	$0.5 \pm 0.3$	$0.5 \pm 0.3$	$0.7 \pm 0.0$ 0.5 + 0.4
Stairs - Ascend stairs	$10 \pm 0.1$	$10 \pm 0.2$	$10 \pm 0.2$	$0.8 \pm 0.2$	$0.0 \pm 0.0$ 0.7 + 0.3	$0.6 \pm 0.3$	$0.0 \pm -0.0$
Stairs - Descend stairs	$1.5 \pm 0.3$	$1.5 \pm 0.3$	$1.5 \pm 0.2$	$1.4 \pm 0.4$	$1.4 \pm 0.5$	$1.4 \pm 0.4$	$0.0 \pm -0.0$
Standing	$0.5 \pm 0.1$	$0.5 \pm 0.1$	$0.5 \pm 0.1$	$0.5 \pm 0.2$	$0.6 \pm 0.2$	$0.6 \pm 0.2$	$0.3 \pm 0.2$
Walking - Treadmill 2mph - Treadmill 0	$0.8 \pm 0.4$	$0.8 \pm 0.4$	$0.8 \pm 0.4$	$0.8 \pm 0.4$	$0.9 \pm 0.4$	$0.9 \pm 0.4$	$0.7 \pm 0.4$
Walking - Treadmill 3mph - Treadmill 0	$0.8 \pm 0.4$	$0.9 \pm 0.4$	$0.9 \pm 0.4$	$1.0 \pm 0.6$	$1.1 \pm 0.6$	$1.2 \pm 0.6$	$1.1 \pm 0.7$
Walking - Treadmill 3mph - Treadmill 3 - light	$0.7 \pm 0.4$	$0.7 \pm 0.3$	$0.7 \pm 0.3$	$0.7 \pm 0.5$	$0.8 \pm 0.6$	$0.8 \pm 0.5$	$0.8 \pm 0.4$
Walking - Treadmill 3mph - Treadmill 6 - moderate	$1.1 \pm 0.4$	$1.1 \pm 0.4$	$1.1 \pm 0.4$	$1.0 \pm 0.4$	$0.9 \pm 0.4$	$0.9 \pm 0.5$	$0.9 \pm 0.6$
Walking - Treadmill 3mph - Treadmill 9 - hard	$1.7 \pm 0.7$	$1.7 \pm 0.8$	$1.7 \pm 0.8$	$1.5 \pm 0.7$	$1.5 \pm 0.7$	$1.4 \pm 0.7$	$1.4 \pm 0.8$
kneeling	$0.6 \pm 0.2$	$0.7 \pm 0.2$	$0.7 \pm 0.2$	$0.7 \pm 0.3$	$0.8 \pm 0.3$	$0.8 \pm 0.4$	$0.6 \pm 0.4$
unknown	$1.6 \pm 0.4$	$1.6 \pm 0.4$	$1.6 \pm 0.4$	$1.5 \pm 0.4$	$1.5 \pm 0.4$	$1.4 \pm 0.4$	$1.4 \pm 0.6$
Carrying groceries	$1.0 \pm 0.2$	$0.9 \pm 0.2$	$0.9 \pm 0.2$	$0.8 \pm 0.2$	$0.9 \pm 0.3$	$0.9 \pm 0.3$	$0.6 \pm 0.4$
Doing dishes	$0.6 \pm 0.1$	$0.6 \pm 0.1$	$0.6 \pm 0.1$	$0.5 \pm 0.1$	$0.5 \pm 0.1$	$0.4 \pm 0.2$	$0.4 \pm 0.2$
Gardening	$0.6 \pm 0.2$	$0.7 \pm 0.1$	$0.7 \pm 0.1$	$0.6 \pm 0.1$	$0.6 \pm 0.1$	$0.6 \pm 0.2$	$0.5 \pm 0.2$
Ironing	$0.8 \pm 0.1$	$0.7 \pm 0.2$	$0.7 \pm 0.2$	$0.6 \pm 0.2$	$0.5 \pm 0.2$	$0.5 \pm 0.2$	$0.4 \pm 0.3$
Making the bed	$1.1 \pm 0.3$	$1.1 \pm 0.4$	$1.1 \pm 0.4$	$0.9 \pm 0.3$	$0.9 \pm 0.3$	$0.8 \pm 0.4$	$0.6 \pm 0.3$
Mopping	$0.7 \pm 0.2$	$0.6 \pm 0.2$	$0.6 \pm 0.2$	$0.6 \pm 0.2$	$0.6 \pm 0.3$	$0.6 \pm 0.4$	$0.4 \pm 0.3$
Playing videogames	$0.5 \pm 0.2$	$0.5\pm0.3$	$0.5 \pm 0.3$	$0.4 \pm 0.1$	$0.4 \pm 0.1$	$0.4 \pm 0.1$	$0.3 \pm 0.1$
Scrubbing a surface	$0.8 \pm 0.3$	$0.9\pm0.4$	$0.9 \pm 0.4$	$0.8 \pm 0.3$	$0.7 \pm 0.3$	$0.6 \pm 0.3$	$0.6 \pm 0.5$
Stacking groceries	$0.9 \pm 0.2$	$0.9\pm0.2$	$0.9 \pm 0.2$	$0.9 \pm 0.4$	$1.0 \pm 0.4$	$0.9 \pm 0.4$	$0.3 \pm 0.0$
Sweeping	$0.7 \pm 0.2$	$0.7 \pm 0.2$	$0.7 \pm 0.2$	$0.5 \pm 0.1$	$0.5 \pm 0.1$	$0.5 \pm 0.1$	$0.4 \pm 0.2$
Typing	$0.5 \pm 0.1$	$0.5 \pm 0.1$	$0.5 \pm 0.1$	$0.5 \pm 0.1$	$0.5 \pm 0.2$	$0.5 \pm 0.2$	$0.4 \pm 0.2$
Vacuuming	$0.7 \pm 0.2$	$0.6 \pm 0.2$	$0.6 \pm 0.2$	$0.5 \pm 0.3$	$0.5 \pm 0.3$	$0.5 \pm 0.3$	$0.4 \pm 0.4$
Walking around block	$1.3 \pm 0.3$	$1.2 \pm 0.2$	$1.2 \pm 0.2$	$1.0 \pm 0.3$	$0.9 \pm 0.3$	$0.8 \pm 0.4$	$0.6 \pm 0.3$
Washing windows	$0.8 \pm 0.2$	$0.8\pm0.3$	$0.8 \pm 0.3$	$0.7 \pm 0.3$	$0.6 \pm 0.3$	$0.5\pm0.3$	$0.4 \pm 0.4$
Watching TV	$0.5 \pm 0.2$	$0.5\pm0.2$	$0.5\pm0.2$	$0.5 \pm 0.2$	$0.5 \pm 0.2$	$0.5\pm0.2$	$0.4 \pm 0.2$
Weeding	$0.8 \pm 0.2$	$0.8 \pm 0.2$	$0.8\pm0.2$	$0.7\pm0.2$	$0.6 \pm 0.2$	$0.6 \pm 0.2$	$0.4 \pm 0.3$
Wiping/Dusting	$0.8 \pm 0.2$	$0.8\pm0.2$	$0.8 \pm 0.2$	$0.6 \pm 0.2$	$0.6 \pm 0.2$	$0.4 \pm 0.2$	$0.5 \pm 0.2$
Writing	$0.5 \pm 0.1$	$0.5\pm0.1$	$0.5\pm0.1$	$0.5\pm0.1$	$0.4 \pm 0.1$	$0.4 \pm 0.2$	$0.4 \pm 0.2$
taking out tool	00.00	0.0 . 0.1	0.0 + 0.1	06:00	06.00	05.00	05.00

taking out trash $0.8 \pm 0.2$  $0.8 \pm 0.1$  $0.8 \pm 0.1$  $0.6 \pm 0.2$  $0.6 \pm 0.2$  $0.5 \pm 0.2$  $0.5 \pm 0.2$ Table B12-5: Root mean squared error when predicting energy expenditure in a subject independent<br/>manner using multivariable linear regression with the *FFTCorr* feature computed per sensor over<br/>sliding windows of varying length.

## **Appendix B13: Feature Selection for Energy Expenditure Estimation**

Activity	All	Fast to	Invariant	ACFFTPeaks	ACFFTPeaks.	ACFFTPeaks.,
		compute	Reduced	ACADSATea	ACMCR,	ACModVigEnergy
					ACModVigEnerg	
Bench weight lifting - hard	$0.6 \pm 0.4$	$0.5 \pm 0.4$	$0.7 \pm 0.3$	$0.6 \pm 0.3$	$0.6 \pm 0.2$	$0.6 \pm 0.2$
Bench weight lifting - light	$0.7 \pm 0.3$	$0.6 \pm 0.3$	$0.7 \pm 0.2$	0.6 ± 0.3	$0.7 \pm 0.3$	$0.7 \pm 0.3$
Bench weight lifting - moderate	$0.7 \pm 0.4$	$0.5 \pm 0.2$	$0.7 \pm 0.3$	$0.6 \pm 0.4$	$0.6 \pm 0.4$	$0.6 \pm 0.4$
Bicep curls - hard	$1.0 \pm 0.4$	$1.1 \pm 0.5$	$1.3 \pm 0.6$	$1.4 \pm 0.6$	$1.3 \pm 0.6$	$1.2 \pm 0.6$
Bicep curls - light	$0.7 \pm 0.2$	$0.8\pm0.3$	$0.9 \pm 0.4$	$1.1 \pm 0.5$	$1.0 \pm 0.4$	$1.0 \pm 0.4$
Bicep curls - moderate	$0.7 \pm 0.3$	$1.0\pm0.2$	$1.0 \pm 0.4$	$1.1 \pm 0.4$	$1.0 \pm 0.5$	$1.0 \pm 0.4$
Calisthenics - Crunches	$1.7 \pm 0.9$	$1.6\pm0.8$	$1.8 \pm 0.9$	$1.6 \pm 0.7$	$1.6 \pm 0.6$	$1.6 \pm 0.6$
Calisthenics - Sit ups	$1.3\pm0.4$	$1.5\pm0.5$	$1.8\pm0.8$	$1.7 \pm 0.6$	$1.6\pm0.5$	$1.6\pm0.5$
Cycling - Cycle hard - Cycle 80rpm	$1.8 \pm 1.0$	$1.8\pm1.0$	$1.9 \pm 1.2$	$2.1 \pm 1.2$	$1.9\pm0.9$	$1.9 \pm 1.0$
Cycling - Cycle light - Cycle 100rpm	$1.2 \pm 0.8$	$1.1 \pm 0.5$	$1.3 \pm 0.6$	$1.3 \pm 0.6$	$1.2 \pm 0.6$	$1.2 \pm 0.6$
Cycling - Cycle light - Cycle 60rpm	$0.8 \pm 0.5$	$0.6 \pm 0.4$	$0.7 \pm 0.4$	$0.8 \pm 0.4$	$0.6 \pm 0.4$	$0.6 \pm 0.4$
Cycling - Cycle light - Cycle 80rpm	$1.1 \pm 0.8$	$0.9 \pm 0.6$	$1.2 \pm 0.6$	$1.2 \pm 0.6$	$1.0 \pm 0.7$	$1.0 \pm 0.7$
Cycling - Cycle moderate - Cycle 80rpm	$1.5 \pm 0.7$	$1.4 \pm 0.7$	$1.5 \pm 0.8$	$1.7 \pm 1.0$	$1.6 \pm 0.7$	$1.6 \pm 0.7$
Lying down	$0.4 \pm 0.2$	$0.6 \pm 0.4$	$0.5 \pm 0.3$	0.4 ± 0.1	0.3 ± 0.1	0.3 ± 0.1
Rowing - Rowing hard - Rowing 30spm	$1.8 \pm 1.6$	$2.0 \pm 1.6$	$2.0 \pm 1.6$	$2.2 \pm 1.7$	$2.1 \pm 1.7$	$2.1 \pm 1.7$
Rowing - Rowing light - Rowing 30spm	$1.4 \pm 1.1$	$1.3 \pm 1.1$	$1.4 \pm 1.2$	1.4 ± 1.3	1.4 ± 1.3	1.4 ± 1.3
Rowing - Rowing moderate - Rowing 30spm	$1.7 \pm 1.5$	$1.8 \pm 1.4$	$1.8 \pm 1.6$	$2.1 \pm 1.6$	$2.0 \pm 1.6$	$1.9 \pm 1.6$
Running - Treadmill 4mph - Treadmill 0	$1.3 \pm 0.7$	$1.3 \pm 0.7$	$1.4 \pm 0.6$	$1.4 \pm 0.8$	$1.2 \pm 0.6$	$1.2 \pm 0.6$
Running - Treadmill Smph - Treadmill 0	$1.4 \pm 0.8$	$1.4 \pm 0.8$	$1.5 \pm 0.9$	$1.5 \pm 0.9$	$1.3 \pm 0.7$	$1.4 \pm 0.8$
Running - Treadmill ompn - Treadmill 0	$1.8 \pm 1.1$	$1.8 \pm 0.8$	$2.0 \pm 1.3$	$2.1 \pm 1.5$	$1.8 \pm 1.3$	$1.9 \pm 1.4$
Sitting Fidget feet logs	$0.7 \pm 0.2$ 1.2 ± 0.5	$0.3 \pm 0.2$	$0.7 \pm 0.2$	$0.0 \pm 0.3$	$0.0 \pm 0.3$ 1.2 ± 0.5	$0.0 \pm 0.3$ 1.2 ± 0.5
Sitting - Fidget hands arms	$1.3 \pm 0.3$ 0.8 ± 0.4	$1.4 \pm 0.7$ 1.1 + 0.9	$1.1 \pm 0.4$ 0.8 ± 0.4	$1.0 \pm 0.4$ 0.6 ± 0.2	$1.2 \pm 0.3$ 0.7 + 0.6	$1.2 \pm 0.3$ 0.8 ± 0.6
Stairs - Ascend stairs	$0.0 \pm 0.4$ 0.9 ± 0.2	$1.1 \pm 0.2$	$0.0 \pm 0.4$	$0.0 \pm 0.2$ 0.9 ± 0.3	$0.7 \pm 0.0$ 0.9 ± 0.2	$0.0 \pm 0.0$
Stairs - Descend stairs	$1.5 \pm 0.2$	$1.5 \pm 0.2$	$15 \pm 0.2$	$15 \pm 0.3$	$1.4 \pm 0.3$	$1.4 \pm 0.3$
Standing	$0.4 \pm 0.1$	$0.3 \pm 0.2$	$0.4 \pm 0.1$	$0.6 \pm 0.1$	$0.5 \pm 0.1$	$0.5 \pm 0.1$
Walking - Treadmill 2mph - Treadmill 0	$0.8 \pm 0.5$	$0.9 \pm 0.2$	$0.6 \pm 0.3$	$0.7 \pm 0.3$	$0.8 \pm 0.4$	$0.8 \pm 0.4$
Walking - Treadmill 3mph - Treadmill 0	$1.0 \pm 0.6$	$1.0 \pm 0.5$	$0.8 \pm 0.4$	$0.9 \pm 0.4$	$1.0 \pm 0.4$	$1.0 \pm 0.5$
Walking - Treadmill 3mph - Treadmill 3 -	$0.8\pm0.5$	$0.6\pm0.4$	$0.6\pm0.3$	$0.7 \pm 0.3$	$0.7\pm0.3$	$0.7 \pm 0.3$
light						
Walking - Treadmill 3mph - Treadmill 6 -	$1.0 \pm 0.5$	$0.9 \pm 0.4$	$1.0 \pm 0.4$	$1.0 \pm 0.4$	$0.9 \pm 0.4$	$0.9 \pm 0.4$
moderate						
Walking - Treadmill 3mph - Treadmill 9 -	$1.5 \pm 0.7$	$1.5 \pm 0.7$	$1.6 \pm 0.6$	$1.6 \pm 0.6$	$1.6 \pm 0.7$	$1.6 \pm 0.7$
kneeling	$0.5 \pm 0.2$	$0.4 \pm 0.3$	$0.5 \pm 0.2$	$0.6 \pm 0.2$	$0.6 \pm 0.2$	$0.6 \pm 0.2$
unknown	$1.5 \pm 0.2$	$1.6 \pm 0.3$	$1.6 \pm 0.2$	$1.5 \pm 0.4$	$1.5 \pm 0.4$	$1.5 \pm 0.4$
Carrying groceries	$0.9 \pm 0.2$	$1.0 \pm 0.4$ $1.0 \pm 0.4$	$1.0 \pm 0.3$ $1.0 \pm 0.3$	$0.9 \pm 0.2$	$0.9 \pm 0.2$	$1.0 \pm 0.1$ $1.0 \pm 0.2$
Doing dishes	$0.5 \pm 0.2$ $0.5 \pm 0.3$	$0.5 \pm 0.6$	$0.5 \pm 0.4$	$0.5 \pm 0.1$	$0.4 \pm 0.1$	$0.5 \pm 0.1$
Gardening	$0.7 \pm 0.4$	$0.9 \pm 0.9$	$0.8 \pm 0.7$	$0.6 \pm 0.2$	$0.6 \pm 0.2$	$0.5 \pm 0.2$
Ironing	$0.5 \pm 0.2$	$0.5 \pm 0.3$	$0.5 \pm 0.2$	$0.6 \pm 0.2$	$0.5 \pm 0.2$	$0.5 \pm 0.2$
Making the bed	$1.0 \pm 0.4$	$1.0 \pm 0.5$	$1.0 \pm 0.4$	$0.9 \pm 0.3$	$0.9 \pm 0.3$	$0.9 \pm 0.3$
Mopping	$0.6 \pm 0.2$	$0.6 \pm 0.3$	$0.6 \pm 0.2$	$0.6 \pm 0.2$	$0.7 \pm 0.2$	$0.7 \pm 0.2$
Playing videogames	$0.7 \pm 0.4$	$0.8\pm0.6$	$0.8 \pm 0.5$	$0.4 \pm 0.1$	$0.4 \pm 0.1$	$0.4 \pm 0.1$
Scrubbing a surface	$0.9 \pm 0.5$	$0.9 \pm 0.6$	$0.9 \pm 0.5$	$0.7 \pm 0.4$	$0.7 \pm 0.2$	$0.7 \pm 0.2$
Stacking groceries	$1.0 \pm 0.5$	$1.0\pm0.6$	$1.1 \pm 0.6$	$1.1 \pm 0.5$	$1.0 \pm 0.3$	$1.0 \pm 0.4$
Sweeping	$0.7 \pm 0.4$	$0.7 \pm 0.7$	$0.7\pm0.6$	$0.6 \pm 0.2$	$0.7 \pm 0.2$	$0.6 \pm 0.2$
Typing	$0.5\pm0.2$	$0.4 \pm 0.3$	$0.6 \pm 0.3$	$0.4 \pm 0.1$	$0.4 \pm 0.1$	$0.4 \pm 0.1$
Vacuuming	$0.5 \pm 0.1$	$0.5\pm0.1$	$0.5 \pm 0.1$	$0.5 \pm 0.2$	$0.6 \pm 0.2$	$0.6 \pm 0.2$
Walking around block	$1.1 \pm 0.5$	$1.5\pm0.6$	$1.3 \pm 0.6$	$1.3 \pm 0.3$	$1.2 \pm 0.2$	$1.3 \pm 0.2$
Washing windows	$0.7 \pm 0.4$	$0.7\pm0.4$	$0.8\pm0.3$	$0.7 \pm 0.2$	$0.6 \pm 0.2$	$0.7 \pm 0.2$
Watching TV	$0.7 \pm 0.5$	$0.8\pm0.5$	$0.8\pm0.5$	$0.5 \pm 0.2$	$0.5\pm0.2$	$0.5\pm0.2$
Weeding	$0.8\pm0.2$	$1.0\pm0.5$	$0.8\pm0.4$	$0.7 \pm 0.2$	$0.8\pm0.2$	$0.7\pm0.2$
Wiping/Dusting	$0.7 \pm 0.6$	$0.8\pm0.8$	$0.7 \pm 0.4$	0.7 ± 0.3	$0.7 \pm 0.3$	$0.7 \pm 0.2$
Writing	0.6 ± 0.2	$0.5 \pm 0.4$	$0.6 \pm 0.3$	0.5 ± 0.2	$0.5 \pm 0.2$	$0.5 \pm 0.2$
taking out trash	$0.7 \pm 0.2$	$0.8 \pm 0.2$	$0.8 \pm 0.3$	$0.8 \pm 0.2$	$0.8 \pm 0.2$	$0.8 \pm 0.2$

Table B13-1: Root mean square error per activity obtained when estimating energy expenditure using multivariable linear regression and different subsets of features computed per sensor over windows of 5.6s in length.

Activity	All	Fast to	Invariant	ACFFTPeak	ACFFTPeaks.	ACFFTPeaks.
		comput	reduced	S A CALEARE	ACEntropy	ACMCR
		e		ACADSAFea	ACModVigEner	rov
					gy	-65
Bench weight lifting - hard	$0.5 \pm 0.4$	$0.5 \pm 0.3$	$0.6 \pm 0.2$	$0.5 \pm 0.2$	0.5 ± 0.2	$0.5 \pm 0.2$
Bench weight lifting - light	$0.6 \pm 0.3$	$0.5 \pm 0.2$	$0.6 \pm 0.2$	$0.6 \pm 0.3$	$0.6 \pm 0.3$	$0.6 \pm 0.3$
Bench weight lifting - moderate	$0.6 \pm 0.4$	$0.5 \pm 0.2$	$0.6 \pm 0.3$	$0.6 \pm 0.4$	$0.6 \pm 0.4$	$0.6 \pm 0.4$
Bicep curls - hard	$0.9\pm0.5$	$1.0 \pm 0.5$	$1.2\pm0.6$	$1.2 \pm 0.7$	$1.2\pm0.6$	$1.1 \pm 0.6$
Bicep curls - light	$0.6 \pm 0.2$	$0.7 \pm 0.3$	$0.8 \pm 0.4$	$1.0 \pm 0.5$	$0.9 \pm 0.4$	$0.8 \pm 0.4$
Bicep curls - moderate	$0.6 \pm 0.3$	$1.0 \pm 0.2$	$1.0 \pm 0.5$	$1.0 \pm 0.5$	$0.9\pm0.5$	$0.9 \pm 0.5$
Calisthenics - Crunches	$1.5\pm0.8$	$1.5\pm0.8$	$1.7 \pm 0.9$	$1.4 \pm 0.6$	$1.4 \pm 0.5$	$1.4 \pm 0.6$
Calisthenics - Sit ups	$1.2 \pm 0.3$	$1.3 \pm 0.5$	$1.6 \pm 0.7$	$1.4 \pm 0.5$	$1.4 \pm 0.5$	$1.4 \pm 0.5$
Cycling - Cycle hard - Cycle 80rpm	$1.7 \pm 1.0$	$1.7 \pm 1.0$	$1.9 \pm 1.2$	$2.0 \pm 1.2$	$1.8 \pm 1.0$	$1.8 \pm 1.0$
Cycling - Cycle light - Cycle 100rpm	$1.1\pm0.8$	$1.0\pm0.5$	$1.2 \pm 0.6$	$1.2\pm0.6$	$1.1\pm0.6$	$1.0 \pm 0.6$
Cycling - Cycle light - Cycle 60rpm	$0.7\pm0.5$	$0.6 \pm 0.4$	$0.7 \pm 0.4$	$0.7 \pm 0.4$	$0.6 \pm 0.4$	$0.6 \pm 0.4$
Cycling - Cycle light - Cycle 80rpm	$1.0\pm0.8$	$0.9\pm0.6$	$1.1 \pm 0.6$	$1.1 \pm 0.6$	$0.9\pm0.6$	$0.9 \pm 0.7$
Cycling - Cycle moderate - Cycle 80rpm	$1.4 \pm 0.7$	$1.3 \pm 0.7$	$1.4 \pm 0.8$	$1.6 \pm 1.0$	$1.4 \pm 0.7$	$1.4 \pm 0.7$
Lying down	$0.4 \pm 0.3$	$0.6 \pm 0.4$	$0.4 \pm 0.3$	$0.3 \pm 0.0$	$0.3 \pm 0.0$	$0.3 \pm 0.0$
Rowing - Rowing hard - Rowing 30spm	$1.7 \pm 1.5$	$1.8 \pm 1.5$	$1.8 \pm 1.6$	$2.1 \pm 1.6$	$2.0 \pm 1.6$	$1.9 \pm 1.6$
Rowing - Rowing light - Rowing 30spm	$1.2 \pm 1.0$	$1.2 \pm 1.0$	$1.2 \pm 1.1$	$1.3 \pm 1.2$	$1.2 \pm 1.2$	$1.2 \pm 1.2$
Rowing - Rowing moderate - Rowing 30spm	$1.6 \pm 1.5$	1.7 ± 1.4	$1.7 \pm 1.6$	$2.0\pm1.6$	$1.8 \pm 1.6$	$1.8 \pm 1.6$
Running - Treadmill 4mph - Treadmill 0	$1.1 \pm 0.7$	$1.1 \pm 0.7$	$1.2 \pm 0.6$	$1.2 \pm 0.7$	$1.0 \pm 0.5$	$1.0 \pm 0.5$
Running - Treadmill 5mph - Treadmill 0	$1.2 \pm 0.8$	$1.1 \pm 0.17$ 1 2 ± 0.8	$1.2 \pm 0.0$ $1.3 \pm 0.9$	$1.2 \pm 0.9$	$1.0 \pm 0.0$ $1.1 \pm 0.7$	$1.0 \pm 0.0$ $1.2 \pm 0.8$
Running - Treadmill 6mph - Treadmill 0	$1.6 \pm 1.1$	$1.6 \pm 0.8$	$1.8 \pm 1.3$	$1.9 \pm 1.6$	$1.6 \pm 1.3$	$1.7 \pm 1.4$
Sitting	$0.6 \pm 0.2$	$0.4 \pm 0.2$	$0.6 \pm 0.2$	$0.5 \pm 0.2$	$0.5 \pm 0.2$	$0.5 \pm 0.2$
Sitting - Fidget feet legs	$1.2 \pm 0.5$	$1.4 \pm 0.7$	$1.0 \pm 0.5$	$1.0 \pm 0.4$	$1.2 \pm 0.5$	$1.2 \pm 0.5$
Sitting - Fidget hands arms	$0.7 \pm 0.4$	$1.0 \pm 0.8$	$0.7 \pm 0.4$	$0.5 \pm 0.2$	$0.6 \pm 0.6$	$0.7 \pm 0.6$
Stairs - Ascend stairs	$0.8 \pm 0.2$	$0.7 \pm 0.2$	$0.8 \pm 0.2$	$0.8 \pm 0.2$	$0.8 \pm 0.2$	0.7 ± 0.2
Stairs - Descend stairs	$1.3 \pm 0.3$	$1.3 \pm 0.3$	$1.3 \pm 0.4$	$1.3 \pm 0.3$	$1.3 \pm 0.3$	$1.3 \pm 0.3$
Standing	$0.4 \pm 0.1$	$0.3 \pm 0.2$	$0.4 \pm 0.1$	$0.5 \pm 0.1$	$0.4 \pm 0.1$	$0.4 \pm 0.1$
Walking - Treadmill 2mph - Treadmill 0	$0.7 \pm 0.5$	$0.8 \pm 0.4$	$0.6 \pm 0.3$	$0.6 \pm 0.3$	$0.8 \pm 0.4$	$0.8 \pm 0.4$
Walking - Treadmill 3mph - Treadmill 0	$0.9 \pm 0.6$	$0.9 \pm 0.5$	$0.8 \pm 0.5$	$0.8 \pm 0.4$	$0.9 \pm 0.5$	$0.9 \pm 0.5$
Walking - Treadmill 3mph - Treadmill 3 -	$0.7\pm0.5$	$0.6 \pm 0.4$	$0.6 \pm 0.3$	$0.6\pm0.3$	$0.6 \pm 0.3$	$0.6\pm0.3$
Walking - Treadmill 3mph - Treadmill 6 -	$0.9\pm0.5$	$0.8 \pm 0.4$	$0.9\pm0.4$	$0.9\pm0.4$	$0.8\pm0.4$	$0.8\pm0.4$
Walking - Treadmill 3mph - Treadmill 9 -	$1.4 \pm 0.7$	$1.4 \pm 0.7$	1.6 ± 0.6	1.5 ± 0.6	$1.5 \pm 0.7$	1.5 ± 0.7
hard						
kneeling	$0.4 \pm 0.2$	$0.4 \pm 0.3$	$0.4 \pm 0.2$	$0.5 \pm 0.2$	$0.5 \pm 0.2$	$0.5 \pm 0.2$
unknown	$1.2 \pm 0.2$	$1.2 \pm 0.2$	$1.2 \pm 0.2$	$1.2 \pm 0.3$	$1.1 \pm 0.3$	$1.1 \pm 0.3$
Carrying groceries	$0.7 \pm 0.2$	$0.8 \pm 0.3$	$0.8 \pm 0.2$	$0.7 \pm 0.2$	$0.8 \pm 0.2$	$0.8 \pm 0.2$
Doing dishes	$0.4 \pm 0.3$	$0.4 \pm 0.5$	$0.4 \pm 0.4$	$0.4 \pm 0.1$	$0.4 \pm 0.1$	$0.4 \pm 0.1$
Gardening	$0.6 \pm 0.4$	0.8 ± 0.9	$0.8 \pm 0.7$	$0.4 \pm 0.1$	$0.4 \pm 0.1$	$0.4 \pm 0.1$
Ironing	$0.4 \pm 0.2$	$0.4 \pm 0.3$	$0.4 \pm 0.2$	$0.5 \pm 0.1$	$0.4 \pm 0.1$	$0.4 \pm 0.1$
Making the bed	$0.8 \pm 0.3$	$0.8 \pm 0.4$	$0.8 \pm 0.3$	$0.8 \pm 0.2$	$0.8 \pm 0.3$	$0.8 \pm 0.3$
Mopping	$0.5 \pm 0.2$	$0.5 \pm 0.2$	$0.5 \pm 0.2$	$0.5 \pm 0.2$	$0.6 \pm 0.2$	0.6 ± 0.2
Playing videogames	$0.6 \pm 0.4$	$0.8 \pm 0.6$	$0.7 \pm 0.5$	$0.4 \pm 0.1$	$0.3 \pm 0.1$	$0.3 \pm 0.1$
Scrubbing a surface	$0.8 \pm 0.4$	$0.8 \pm 0.5$	$0.8 \pm 0.5$	$0.6 \pm 0.3$	$0.5 \pm 0.2$	$0.6 \pm 0.2$
Stacking groceries	$0.9 \pm 0.5$	$0.9 \pm 0.6$	$0.9 \pm 0.6$	$1.0 \pm 0.4$	$0.9 \pm 0.3$	$0.9 \pm 0.4$
Sweeping	$0.6 \pm 0.4$	$0.6 \pm 0.7$	$0.6 \pm 0.6$	$0.5 \pm 0.2$	$0.5 \pm 0.2$	$0.5 \pm 0.2$
1 yping Version	$0.4 \pm 0.2$	$0.4 \pm 0.3$	$0.5 \pm 0.3$	$0.4 \pm 0.1$	$0.3 \pm 0.1$	$0.5 \pm 0.1$
Vacuuming Wallsing agreent blasts	$0.4 \pm 0.1$	$0.4 \pm 0.1$	$0.4 \pm 0.1$	$0.4 \pm 0.2$	$0.5 \pm 0.2$	$0.5 \pm 0.2$
Washing around block	$1.0 \pm 0.5$	$1.5 \pm 0.6$	$1.2 \pm 0.6$	$1.1 \pm 0.3$	$1.1 \pm 0.2$	$1.1 \pm 0.3$
Watching TV	$0.0 \pm 0.3$	$0.0 \pm 0.4$	$0.0 \pm 0.3$	$0.6 \pm 0.2$	$0.5 \pm 0.2$	$0.5 \pm 0.2$
Waading IV	$0.0 \pm 0.5$	$0.8 \pm 0.6$	$0.7 \pm 0.5$	$0.4 \pm 0.2$	$0.4 \pm 0.2$	$0.4 \pm 0.2$
Wining Dusting	$0.7 \pm 0.2$	$0.8 \pm 0.4$	$0.7 \pm 0.4$	$0.0 \pm 0.2$	$0.0 \pm 0.2$	$0.0 \pm 0.2$
Writing	$0.0 \pm 0.3$	$0.7 \pm 0.7$	$0.0 \pm 0.4$	$0.0 \pm 0.2$	$0.0 \pm 0.3$	$0.0 \pm 0.3$
taking out trash	$0.5 \pm 0.2$ 0.6 + 0.2	$0.5 \pm 0.4$ 0.6 + 0.2	$0.0 \pm 0.3$ 0.7 + 0.3	$0.4 \pm 0.2$ 0.7 + 0.2	$0.4 \pm 0.2$ 0.6 ± 0.2	$0.4 \pm 0.1$ 0.7 + 0.2

Table B13-2: Mean absolute error per activity obtained when estimating energy expenditure using multivariable linear regression and different subsets of features computed per sensor over windows of 5.6s in length.

Activity	All	Fast to	Invariant	ACFFTPeak	ACFFTPeaks.	ACFFTPeaks.
		comput	Reduced	s ACAbsArea	ACEntropy ACMCR.	ACMCR ACModVigEn
					ACModVigEne	ergy
Developments ht 1100 ments	0.0 . 0.5	10.00	12:04	11.04	rgy	11:04
Bench weight lifting light	$0.9 \pm 0.3$	$1.0 \pm 0.0$	$1.2 \pm 0.4$	$1.1 \pm 0.4$	$1.0 \pm 0.4$	$1.1 \pm 0.4$
Bench weight lifting - ngh	$1.4 \pm 0.3$	$1.1 \pm 0.4$	$1.3 \pm 0.4$	$1.2 \pm 0.4$	$1.5 \pm 0.5$	$1.5 \pm 0.4$
Bench weight lifting - moderate	$1.2 \pm 0.5$	$0.9 \pm 0.3$	$1.2 \pm 0.4$	$1.1 \pm 0.5$	$1.1 \pm 0.0$	$1.1 \pm 0.6$
Bicep curis - hard	$1.9 \pm 0.4$	$1.6 \pm 0.5$	$2.2 \pm 0.5$	$2.4 \pm 0.6$	$2.3 \pm 0.8$	$2.2 \pm 0.7$
Bicep curis - light	$1.7 \pm 0.4$	$1.4 \pm 0.5$	$1.8 \pm 0.5$	$2.1 \pm 0.5$	$2.0 \pm 0.6$	$1.9 \pm 0.6$
Calisthanias Crunchas	$1.4 \pm 0.5$	$1.5 \pm 0.2$ 2.4 ± 0.0	$1.9 \pm 0.0$ 2.0 ± 1.2	$2.1 \pm 0.0$ 2.8 ± 1.2	$1.9 \pm 0.0$ 2.8 ± 1.2	$1.9 \pm 0.0$ 2.8 ± 1.1
Calisthenics - Crunches	$2.6 \pm 1.1$	$2.4 \pm 0.9$	$2.9 \pm 1.2$	$2.0 \pm 1.2$	$2.0 \pm 1.2$	$2.0 \pm 1.1$
Custing Custs hard Custs 20mm	$2.4 \pm 0.7$	$2.4 \pm 0.7$	$2.9 \pm 0.0$	$2.6 \pm 0.0$	$2.0 \pm 0.0$	$2.7 \pm 0.0$
Cycling - Cycle light - Cycle 801pm	$2.0 \pm 1.0$	$2.4 \pm 1.2$ 1.8 ± 0.7	$2.0 \pm 1.3$ 2.1 ± 0.0	$2.6 \pm 1.3$ 2.1 ± 0.0	$2.0 \pm 1.1$ 2.2 ± 0.8	$2.7 \pm 1.1$ $2.1 \pm 0.8$
Cycling - Cycle light - Cycle Toorphi	$2.1 \pm 1.1$	$1.6 \pm 0.7$	$2.1 \pm 0.9$	$2.1 \pm 0.9$	$2.2 \pm 0.8$	$2.1 \pm 0.8$
Cycling - Cycle light - Cycle 60rpm	$1.3 \pm 0.0$ 1.0 + 1.0	$1.1 \pm 0.3$ $1.6 \pm 0.0$	$1.1 \pm 0.3$ $1.0 \pm 0.8$	$1.2 \pm 0.4$ 2.0 ± 0.0	$1.1 \pm 0.3$ 1.0 ± 1.1	$1.1 \pm 0.3$ $1.0 \pm 1.1$
Cycling - Cycle light - Cycle 801pm	$1.9 \pm 1.0$	$1.0 \pm 0.9$	$1.9 \pm 0.8$	$2.0 \pm 0.9$	$1.9 \pm 1.1$ 2.5 ± 0.0	$1.9 \pm 1.1$ 2.4 ± 0.0
Lying down	$2.3 \pm 0.9$	$2.1 \pm 0.9$	$2.2 \pm 1.0$ 1.2 ± 0.5	$2.3 \pm 1.1$	$2.3 \pm 0.9$	$2.4 \pm 0.9$
Powing Powing hard Powing 20cpm	$1.0 \pm 0.4$	$0.8 \pm 0.4$	$1.2 \pm 0.3$	$0.9 \pm 0.3$	$0.9 \pm 0.3$	$0.9 \pm 0.3$
Rowing - Rowing light Rowing 30spm	$2.6 \pm 2.0$ $2.4 \pm 1.3$	$2.9 \pm 1.0$ 2.2 ± 1.3	$2.8 \pm 1.9$ 2.2 ± 1.4	$3.2 \pm 2.0$ 23 ± 15	$3.1 \pm 2.0$ 2.4 ± 1.5	$3.1 \pm 2.0$ 2.3 ± 1.5
Rowing - Rowing nght - Rowing Josphi Rowing - Rowing moderate - Rowing 30spm	$2.4 \pm 1.3$ 25 ± 1.8	$2.2 \pm 1.3$ $2.5 \pm 1.7$	$2.2 \pm 1.4$ 25 ± 1.8	$2.3 \pm 1.3$ 2.9 ± 1.8	2.4 ± 1.5	$2.5 \pm 1.5$ 2.7 ± 1.8
Rowing - Rowing moderate - Rowing Sospin	$2.5 \pm 1.0$ 2.6 ± 1.2	$2.3 \pm 1.7$ 27 ± 1.3	$2.3 \pm 1.3$ 2.8 ± 1.3	$2.9 \pm 1.0$ 2.9 ± 1.6	$2.6 \pm 1.0$	$2.7 \pm 1.0$ 2.6 ± 1.4
Running - Treadmill 5mph - Treadmill 0	$2.0 \pm 1.2$ 2.7 ± 1.0	$2.7 \pm 1.3$ $2.4 \pm 1.1$	$2.8 \pm 1.3$ 2.8 ± 1.1	$2.9 \pm 1.0$ 3.0 ± 1.2	$2.0 \pm 1.4$ 2.8 ± 1.0	$2.0 \pm 1.4$ 2.0 ± 1.4
Running - Treadmill 6mph - Treadmill 0	$2.7 \pm 1.0$	$2.4 \pm 1.1$ 2.8 ± 1.0	$2.0 \pm 1.1$ $3.5 \pm 1.5$	$3.0 \pm 1.2$ $3.7 \pm 1.6$	$2.0 \pm 1.0$ $3.4 \pm 1.6$	$2.9 \pm 1.1$ $3.4 \pm 1.6$
Sitting	$3.3 \pm 1.3$ 16±05	$2.8 \pm 1.0$ 0.9 ± 0.2	$3.3 \pm 1.3$ 1 3 + 0 4	$3.7 \pm 1.0$ 1.2 ± 0.6	$3.4 \pm 1.0$ 1.2 ± 0.6	$3.4 \pm 1.0$ 1.2 ± 0.6
Sitting Fidget feet lags	$1.0 \pm 0.5$	$0.9 \pm 0.2$	$1.5 \pm 0.4$	$1.2 \pm 0.0$ 1.6 ± 0.5	$1.2 \pm 0.0$ 1.8 ± 0.6	$1.2 \pm 0.0$ 1.8 ± 0.6
Sitting Fidget hands arms	$1.9 \pm 0.0$	$2.0 \pm 0.7$ 17 ± 11	$1.0 \pm 0.4$ $1.4 \pm 0.5$	$1.0 \pm 0.3$ 1.2 ± 0.4	$1.6 \pm 0.0$ $1.4 \pm 0.8$	$1.8 \pm 0.0$ $1.4 \pm 0.0$
Staire Accord stairs	$1.0 \pm 0.0$	$1.7 \pm 1.1$ $1.8 \pm 0.3$	$1.4 \pm 0.3$ $1.8 \pm 0.4$	$1.2 \pm 0.4$ $1.7 \pm 0.4$	$1.4 \pm 0.0$ 1.8 ± 0.4	$1.4 \pm 0.9$ $1.8 \pm 0.4$
Stairs - Descend stairs	$1.0 \pm 0.4$ 2.6 ± 0.4	$1.0 \pm 0.5$ 2.6 ± 0.5	$2.6 \pm 0.4$	$1.7 \pm 0.4$ 2.5 ± 0.4	$1.0 \pm 0.4$ 2.3 ± 0.3	$1.0 \pm 0.4$ 2.3 ± 0.4
Standing	$1.0 \pm 0.4$	$0.5 \pm 0.3$	$0.9 \pm 0.3$	$\frac{2.3 \pm 0.4}{1.1 \pm 0.3}$	$2.3 \pm 0.3$ 1 1 + 0 3	$2.3 \pm 0.4$ 1 1 + 0 3
Walking - Treadmill 2mph - Treadmill 0	$1.0 \pm 0.2$ 15 ± 0.6	$1.6 \pm 0.5$	$13 \pm 0.4$	$1.1 \pm 0.5$ 1 3 ± 0 5	$1.1 \pm 0.5$ 1 5 ± 0 5	$1.1 \pm 0.5$ 1 5 ± 0 5
Walking - Treadmill 3mph - Treadmill 0	$1.5 \pm 0.6$	$1.0 \pm 0.5$ $1.7 \pm 0.6$	$1.5 \pm 0.4$ 15 ± 0.6	$1.5 \pm 0.5$ 1.7 ± 0.4	$1.5 \pm 0.5$ 1.7 ± 0.5	$1.5 \pm 0.5$ 1.7 ± 0.5
Walking - Treadmill 3mph - Treadmill 3 - light	$1.0 \pm 0.0$ $1.5 \pm 0.6$	$1.7 \pm 0.0$ $1.3 \pm 0.5$	$1.3 \pm 0.0$ $1.2 \pm 0.4$	$1.7 \pm 0.4$ $1.4 \pm 0.6$	$1.7 \pm 0.5$ $1.4 \pm 0.4$	$1.7 \pm 0.5$ $1.4 \pm 0.5$
Walking - Treadmill 3mph - Treadmill 6 -	$1.8 \pm 0.7$	$1.5 \pm 0.6$ $1.7 \pm 0.6$	$1.2 \pm 0.1$ $1.8 \pm 0.4$	$1.8 \pm 0.5$	$1.7 \pm 0.6$	$1.7 \pm 0.6$
moderate	1.0 = 0.7	1 = 0.0	1.0 = 0.11	110 = 010	117 = 010	117 = 010
Walking - Treadmill 3mph - Treadmill 9 - hard	$2.3\pm0.8$	$2.2\pm0.9$	$2.4\pm0.8$	$2.4\pm0.8$	$2.4 \pm 0.7$	$2.4\pm0.8$
kneeling	$1.2\pm0.5$	$0.8 \pm 0.3$	$1.0 \pm 0.3$	$1.2\pm0.3$	$1.2 \pm 0.4$	$1.2 \pm 0.4$
unknown	$5.3\pm0.8$	$5.1 \pm 1.0$	$5.0 \pm 1.0$	$5.0 \pm 1.1$	$5.0 \pm 1.1$	$5.0 \pm 1.1$
Carrying groceries	$2.0\pm0.8$	$2.0\pm0.9$	$2.0\pm0.8$	$2.0\pm0.6$	$2.0 \pm 0.6$	$2.1 \pm 0.6$
Doing dishes	$1.0 \pm 0.5$	$1.0 \pm 0.8$	$0.9 \pm 0.5$	$1.0 \pm 0.3$	$1.0 \pm 0.2$	$1.0 \pm 0.2$
Gardening	$1.6 \pm 0.7$	$1.7 \pm 1.1$	$1.6 \pm 1.0$	$1.3 \pm 0.3$	$1.3 \pm 0.5$	$1.2 \pm 0.4$
Ironing	$1.2 \pm 0.4$	$1.0 \pm 0.4$	$1.0 \pm 0.4$	$1.2 \pm 0.5$	$1.3 \pm 0.5$	$1.3 \pm 0.5$
Making the bed	$2.1 \pm 1.0$	$2.4 \pm 1.2$	$2.0 \pm 0.9$	$2.0 \pm 0.7$	$2.0 \pm 0.7$	$2.0 \pm 0.7$
Mopping	$1.6 \pm 0.6$	$1.4 \pm 0.5$	$1.5 \pm 0.7$	$1.6 \pm 0.4$	$1.5 \pm 0.3$	$1.4 \pm 0.3$
Playing videogames	$1.5 \pm 0.6$	$1.1 \pm 0.6$	$1.5 \pm 0.6$	$1.1 \pm 0.6$	$1.1 \pm 0.7$	$1.0 \pm 0.7$
Scrubbing a surface	$2.2 \pm 1.3$	$2.0 \pm 0.9$	$2.0 \pm 1.2$	$1.7 \pm 1.1$	$1.5 \pm 0.4$	$1.6 \pm 0.5$
Stacking groceries	$2.1 \pm 0.9$	$1.7 \pm 0.9$	$2.0 \pm 0.9$	$2.1 \pm 0.9$	$1.9 \pm 0.4$	$1.9 \pm 0.6$
Sweeping	$1.5 \pm 0.5$	$1.5 \pm 0.8$	$1.4 \pm 0.8$	$1.3 \pm 0.3$	$1.5 \pm 0.3$	$1.4 \pm 0.3$
Typing	$1.1 \pm 0.4$	$0.8 \pm 0.6$	$1.1 \pm 0.6$	$1.0 \pm 0.4$	$1.0 \pm 0.4$	$1.0 \pm 0.4$
Vacuuming	$1.2 \pm 0.2$	$1.2 \pm 0.2$	$1.1 \pm 0.4$	$1.1 \pm 0.4$	$1.4 \pm 0.2$	$1.3 \pm 0.3$
Walking around block	$2.3 \pm 0.6$	$2.7 \pm 0.7$	$2.4 \pm 0.7$	$2.4 \pm 0.6$	$2.3 \pm 0.4$	$2.4 \pm 0.4$
Washing windows	$1.7 \pm 0.7$	$1.8 \pm 0.9$	$1.6 \pm 0.6$	$1.6 \pm 0.5$	$1.5 \pm 0.3$	$1.5 \pm 0.3$
Watching TV	$1.4 \pm 0.7$	$1.1 \pm 0.7$	$1.3 \pm 0.6$	$0.9 \pm 0.2$	$1.0 \pm 0.3$	$1.0 \pm 0.2$
Weeding	$1.7 \pm 0.7$	$1.7 \pm 0.8$	$1.6 \pm 0.8$	$1.5 \pm 0.5$	$1.5 \pm 0.5$	$1.5 \pm 0.5$
Wiping/Dusting	$1.6 \pm 0.9$	$1.7 \pm 1.0$	$1.5 \pm 0.7$	$1.4 \pm 0.4$	$1.5 \pm 0.4$	$1.5 \pm 0.4$
Writing	$1.2 \pm 0.4$	$0.8 \pm 0.5$	$1.2 \pm 0.4$	$1.0 \pm 0.2$	$1.0 \pm 0.2$	$1.0 \pm 0.2$
taking out trash	$1.6 \pm 0.4$	$1.8 \pm 0.4$	$1.8 \pm 0.5$	$1.9 \pm 0.5$	$1.7 \pm 0.4$	$1.8 \pm 0.4$

Table B13-3: Maximum absolute error deviation per activity obtained when estimating energy expenditure using multivariable linear regression and different subsets of features computed per sensor over windows of 5.6s in length.

**Appendix B14: Energy Expenditure Estimation Using Heart Rate Data** 

Activity	RMSE	MAE	MAED
Bench weight lifting - hard	$1.93 \pm 1.22$	$1.86 \pm 1.19$	$249 \pm 1.65$
Bench weight lifting - light	$1.99 \pm 1.22$ $1.70 \pm 0.59$	$1.60 \pm 0.19$ 1.68 ± 0.58	$2.19 \pm 1.03$ 2 11 + 0 74
Bench weight lifting - moderate	$1.63 \pm 0.69$	$1.00 \pm 0.00$ $1.60 \pm 0.66$	$2.11 \pm 0.11$ 2 19 + 1 10
Bicen curls - hard	$1.03 \pm 0.03$	$1.00 \pm 0.00$ $1.96 \pm 0.72$	$2.19 \pm 1.10$ 2 39 ± 0.81
Bicep_curlslight	$1.57 \pm 0.72$ $1.65 \pm 0.63$	$1.90 \pm 0.72$ $1.62 \pm 0.64$	$1.95 \pm 0.61$
Bicep_curls - moderate	$1.03 \pm 0.03$ $1.83 \pm 0.62$	$1.02 \pm 0.01$ $1.80 \pm 0.65$	$2.35 \pm 0.68$
Calisthenics - Crunches	$1.05 \pm 0.02$ 1.86 ± 0.82	$1.00 \pm 0.05$ $1.77 \pm 0.75$	$2.39 \pm 0.00$ 2 39 ± 0.92
Calisthenics - Sit uns	$1.00 \pm 0.02$ $1.43 \pm 0.76$	$1.77 \pm 0.75$ $1.30 \pm 0.82$	$2.39 \pm 0.92$ 2.29 ± 0.78
Cycling - Cycle hard - Cycle 80rpm	$1.45 \pm 0.70$ 1.86 + 1.67	$1.30 \pm 0.02$ 1 79 + 1 67	$2.25 \pm 0.70$ 2 45 + 1 93
Cycling - Cycle light - Cycle 100rpm	$0.85 \pm 0.66$	$0.81 \pm 0.66$	$1.13 \pm 1.33$ $1.14 \pm 0.71$
Cycling - Cycle light - Cycle 60rpm	$0.62 \pm 0.00$ $0.48 \pm 0.26$	$0.01 \pm 0.00$ $0.46 \pm 0.26$	$0.75 \pm 0.33$
Cycling - Cycle light - Cycle 80rpm	$0.63 \pm 0.34$	$0.60 \pm 0.34$	$0.92 \pm 0.38$
Cycling - Cycle moderate - Cycle 80rpm	$1.34 \pm 1.07$	$1.22 \pm 1.06$	$220 \pm 150$
Lying down	$0.25 \pm 0.11$	$0.23 \pm 0.10$	$0.47 \pm 0.19$
Rowing - Rowing hard - Rowing 30spm	$0.23 \pm 0.11$ $0.92 \pm 0.45$	$0.23 \pm 0.10$ 0.84 + 0.45	$1.57 \pm 0.78$
Rowing - Rowing light - Rowing 30spm	$0.72 \pm 0.43$ 0.73 + 0.42	$0.64 \pm 0.43$ 0.66 ± 0.41	$1.57 \pm 0.78$ 1 19 + 0 52
Rowing - Rowing moderate - Rowing 30spm	$0.73 \pm 0.12$ 0.73 ± 0.37	$0.68 \pm 0.36$	$1.09 \pm 0.02$ $1.09 \pm 0.48$
Running - Treadmill 4mph - Treadmill 0	$0.73 \pm 0.57$	$0.00 \pm 0.50$ 0.86 ± 0.50	$1.09 \pm 0.40$ 1.61 ± 0.96
Running _ Treadmill 5mph _ Treadmill 0	$1.34 \pm 0.77$	$1.28 \pm 0.77$	$1.01 \pm 0.90$ $1.89 \pm 0.92$
Running _ Treadmill_Smph _ Treadmill_0	$1.54 \pm 0.77$ 1.63 ± 0.98	$1.20 \pm 0.77$ 1 55 ± 0.95	$1.09 \pm 0.92$ 2 29 + 1 41
Sitting	$0.54 \pm 0.18$	$0.41 \pm 0.14$	$1.27 \pm 0.48$
Sitting - Fidget feet legs	$0.34 \pm 0.10$	$0.41 \pm 0.14$ 0.30 ± 0.20	$1.27 \pm 0.40$
Sitting - Fidget hands arms	$0.34 \pm 0.21$ 0.49 ± 0.26	$0.30 \pm 0.20$ 0.45 ± 0.26	$0.01 \pm 0.03$
Stairs - Ascend stairs	$0.49 \pm 0.20$	$0.43 \pm 0.20$	$0.71 \pm 0.50$ 1 57 ± 0.68
Stairs - Descend stairs	$0.03 \pm 0.41$	$0.75 \pm 0.33$	$1.37 \pm 0.08$ $1.72 \pm 0.69$
Standing	$0.94 \pm 0.39$	$0.86 \pm 0.55$	$1.72 \pm 0.09$ $1.20 \pm 0.54$
Walking - Treadmill 2mph - Treadmill ()	$0.09 \pm 0.49$	$0.00 \pm 0.00$	$1.20 \pm 0.94$ $1.05 \pm 0.99$
Walking - Treadmill 3mph - Treadmill 0	$0.50 \pm 0.57$	$0.49 \pm 0.29$	$1.03 \pm 0.03$
Walking _ Treadmill 3mph _ Treadmill 3 _ light	$0.30 \pm 0.20$	$0.33 \pm 0.20$	$1.04 \pm 0.31$
Walking _ Treadmill 3mph _ Treadmill 6 _ moderate	$0.74 \pm 0.22$	$0.71 \pm 0.30$ 0.91 + 0.44	$1.04 \pm 0.031$ $1.47 \pm 0.73$
Walking _ Treadmill 3mph _ Treadmill 9 _ hard	$1.10 \pm 0.54$	$1.07 \pm 0.55$	$1.47 \pm 0.75$ $1.47 \pm 0.64$
kneeling	$1.10 \pm 0.34$ 0.65 ± 0.43	$0.58 \pm 0.45$	$1.47 \pm 0.04$ $1.03 \pm 0.49$
unknown	$1.11 \pm 0.38$	$0.90 \pm 0.49$ 0.89 + 0.30	$3.44 \pm 1.67$
Carrying groceries	$0.88 \pm 0.42$	$0.09 \pm 0.00$ $0.81 \pm 0.40$	$1.25 \pm 0.57$
Doing dishes	$0.00 \pm 0.42$ 0.45 ± 0.10	$0.01 \pm 0.40$ $0.40 \pm 0.10$	$1.23 \pm 0.37$
Gardening	$0.49 \pm 0.10$ 0.40 ± 0.16	$0.40 \pm 0.10$ 0.32 ± 0.15	$0.77 \pm 0.10$ 0.93 ± 0.36
Ironing	$0.40 \pm 0.10$	$0.32 \pm 0.15$ 0.39 + 0.15	$0.93 \pm 0.30$
Making the bed	$0.40 \pm 0.17$ 0.80 ± 0.35	$0.37 \pm 0.13$ 0.71 ± 0.32	$1.34 \pm 0.45$
Monning	$0.00 \pm 0.00$	$0.71 \pm 0.32$ 0.67 ± 0.39	$1.34 \pm 0.49$ 1.25 ± 0.70
Plaving videogames	$0.71 \pm 0.10$ $0.25 \pm 0.12$	$0.07 \pm 0.03$	$0.50 \pm 0.70$
Scrubbing a surface	$0.25 \pm 0.12$ 0.68 ± 0.33	$0.22 \pm 0.11$ 0.60 ± 0.30	$1.30 \pm 0.21$
Stacking groceries	$0.00 \pm 0.05$	$0.00 \pm 0.00$	$0.96 \pm 0.36$
Sweening	$0.54 \pm 0.10$ 0.57 ± 0.17	$0.49 \pm 0.15$ 0.50 ± 0.16	$1.07 \pm 0.42$
Typing	$0.37 \pm 0.17$	$0.30 \pm 0.10$ 0.27 ± 0.12	$1.07 \pm 0.42$ 0.60 ± 0.24
Vacuuming	$0.51 \pm 0.13$	$0.27 \pm 0.12$ 0.59 ± 0.47	$0.00 \pm 0.24$ 1 13 ± 0.66
Walking around block	$0.05 \pm 0.47$ 0.67 ± 0.19	$0.59 \pm 0.47$ 0.58 + 0.18	$1.13 \pm 0.00$ 1 18 + 0.29
Washing windows	$0.67 \pm 0.19$	$0.50 \pm 0.10$ 0.52 + 0.27	$1.10 \pm 0.29$ $1.26 \pm 0.53$
Watching TV	$0.05 \pm 0.00$	$0.32 \pm 0.27$ 0.21 + 0.13	$0.51 \pm 0.17$
Weeding	$0.23 \pm 0.12$ 0.58 + 0.27	$0.54 \pm 0.13$	$0.93 \pm 0.17$
Wining/Dusting	$0.42 \pm 0.27$	$0.34 \pm 0.27$ 0.36 ± 0.27	$0.81 \pm 0.33$
Writing	$0.42 \pm 0.24$ 0.30 ± 0.15	$0.30 \pm 0.20$ 0.28 + 0.15	$0.01 \pm 0.44$ $0.49 \pm 0.18$
taking out trash	$0.71 \pm 0.13$	$0.63 \pm 0.13$	$1.28 \pm 0.13$

 Table B14-1: Estimation of energy expenditure estimation using linear regression and the ScaledHR feature evaluated in a subject independent manner. The heart rate feature was computed over sliding windows of 5.6s in length.



Figure B14-1. Subject independent estimation of energy expenditure using linear regression and the *ScaledHR* feature computed over windows of 5.6s in length.

Activity	RMSE	MAE	MAED
Bench weight lifting - hard	$1.87 \pm 0.91$	$1.79 \pm 0.89$	$2.43 \pm 1.24$
Bench weight lifting - light	$1.70 \pm 0.64$	$1.69 \pm 0.63$	$2.10 \pm 0.82$
Bench_weight_liftingmoderate	$1.52 \pm 0.71$	$1.49 \pm 0.69$	$2.04 \pm 1.09$
Bicep curls - hard	$1.82 \pm 0.36$	$1.81 \pm 0.36$	$2.23 \pm 0.44$
Bicep curls - light	$1.63 \pm 0.49$	$1.61 \pm 0.49$	$1.95 \pm 0.52$
Bicep curls - moderate	$1.72 \pm 0.48$	$1.70 \pm 0.49$	$2.26 \pm 0.65$
Calisthenics - Crunches	$1.67 \pm 0.61$	$1.59 \pm 0.58$	$2.19 \pm 0.63$
Calisthenics - Sit ups	$1.30 \pm 0.45$	$1.14 \pm 0.52$	$2.27 \pm 0.47$
Cycling - Cycle hard - Cycle 80rpm	$1.67 \pm 1.46$	$1.58 \pm 1.45$	$2.29 \pm 1.71$
Cycling - Cycle light - Cycle 100rpm	$0.69 \pm 0.45$	$0.65 \pm 0.45$	$1.00 \pm 0.49$
Cycling - Cycle light - Cycle 60rpm	$0.40 \pm 0.24$	$0.37 \pm 0.25$	$0.69 \pm 0.33$
Cycling - Cycle light - Cycle 80rpm	$0.47 \pm 0.28$	$0.44 \pm 0.28$	$0.74 \pm 0.35$
Cycling - Cycle moderate - Cycle 80rpm	$1.17 \pm 0.98$	$1.07 \pm 0.95$	$1.92 \pm 1.49$
Lving down	$0.39 \pm 0.26$	$0.37 \pm 0.26$	$0.60 \pm 0.29$
Rowing - Rowing hard - Rowing 30spm	$0.83 \pm 0.39$	$0.71 \pm 0.40$	$1.60 \pm 0.86$
Rowing - Rowing light - Rowing 30spm	$0.66 \pm 0.33$	$0.57 \pm 0.33$	$1.16 \pm 0.44$
Rowing - Rowing moderate - Rowing 30spm	$0.69 \pm 0.42$	$0.62 \pm 0.43$	$1.15 \pm 0.50$
Running - Treadmill 4mph - Treadmill 0	$0.90 \pm 0.38$	$0.81 \pm 0.35$	$1.55 \pm 0.77$
Running - Treadmill 5mph - Treadmill 0	$1.25 \pm 0.65$	$1.20 \pm 0.64$	$1.79 \pm 0.82$
Running - Treadmill 6mph - Treadmill 0	$1.32 \pm 1.01$	$1.25 \pm 0.96$	$1.96 \pm 1.51$
Sitting	$0.56 \pm 0.25$	$0.46 \pm 0.26$	$1.25 \pm 0.46$
SittingFidget_feet_legs	$0.36 \pm 0.23$	$0.33 \pm 0.23$	$0.60 \pm 0.39$
Sitting - Fidget hands arms	$0.45 \pm 0.28$	$0.41 \pm 0.28$	$0.66 \pm 0.33$
Stairs - Ascend stairs	$0.79 \pm 0.35$	$0.68 \pm 0.32$	$1.50 \pm 0.58$
Stairs - Descend stairs	$0.91 \pm 0.36$	$0.76 \pm 0.31$	$1.65 \pm 0.58$
Standing	$0.81 \pm 0.49$	$0.78 \pm 0.50$	$1.13 \pm 0.58$
Walking - Treadmill 2mph - Treadmill 0	$0.62 \pm 0.31$	$0.56 \pm 0.24$	$1.10 \pm 0.85$
Walking - Treadmill 3mph - Treadmill 0	$0.65 \pm 0.20$	$0.62 \pm 0.20$	$0.97 \pm 0.29$
WalkingTreadmill_3mphTreadmill_3light	$0.75 \pm 0.23$	$0.72 \pm 0.24$	$1.05 \pm 0.24$
WalkingTreadmill_3mphTreadmill_6moderate	$0.96 \pm 0.33$	$0.91 \pm 0.34$	$1.51 \pm 0.82$
WalkingTreadmill_3mphTreadmill_9hard	$1.04 \pm 0.42$	$1.01 \pm 0.43$	$1.44 \pm 0.49$
kneeling	$0.68 \pm 0.48$	$0.64 \pm 0.49$	$1.01 \pm 0.51$
unknown	$1.03 \pm 0.33$	$0.83 \pm 0.26$	$3.33 \pm 1.64$
Carrying_groceries	$0.81 \pm 0.31$	$0.76 \pm 0.29$	$1.18 \pm 0.45$
Doing_dishes	$0.40 \pm 0.15$	$0.36 \pm 0.14$	$0.70 \pm 0.25$
Gardening	$0.35 \pm 0.14$	$0.28 \pm 0.14$	$0.84 \pm 0.32$
Ironing	$0.39 \pm 0.19$	$0.32 \pm 0.17$	$0.79 \pm 0.45$
Making_the_bed	$0.79 \pm 0.30$	$0.71 \pm 0.29$	$1.32 \pm 0.43$
Mopping	$0.67 \pm 0.33$	$0.60 \pm 0.32$	$1.16 \pm 0.58$
Playing_videogames	$0.22 \pm 0.17$	$0.19 \pm 0.16$	$0.41 \pm 0.20$
Scrubbing_a_surface	$0.57 \pm 0.29$	$0.49 \pm 0.26$	$1.19 \pm 0.72$
Stacking_groceries	$0.46 \pm 0.18$	$0.41 \pm 0.17$	$0.88 \pm 0.30$
Sweeping	$0.49 \pm 0.08$	$0.41 \pm 0.04$	$0.97 \pm 0.37$
Typing	$0.30 \pm 0.12$	$0.26 \pm 0.12$	$0.61 \pm 0.24$
Vacuuming	$0.60 \pm 0.47$	$0.55 \pm 0.47$	$1.07 \pm 0.73$
Walking_around_block	$0.67 \pm 0.15$	$0.59 \pm 0.16$	$1.13 \pm 0.22$
Washing_windows	$0.62 \pm 0.31$	$0.51 \pm 0.28$	$1.25 \pm 0.50$
Watching_TV	$0.23\pm0.15$	$0.19 \pm 0.13$	$0.49 \pm 0.26$
Weeding	$0.51 \pm 0.31$	$0.46 \pm 0.32$	$0.83\pm0.30$
Wiping/Dusting	$0.41 \pm 0.28$	$0.35\pm0.25$	$0.82\pm0.50$
Writing	$0.27\pm0.19$	$0.25 \pm 0.19$	$0.45 \pm 0.23$
taking out trash	$0.66 \pm 0.42$	$0.59 \pm 0.40$	$1.21 \pm 0.80$

Table B14-2: Estimation of energy expenditure using linear regression and the *ScaledHR* feature evaluated in a subject dependent manner. The heart rate feature was computed over sliding windows of 5.6s in length.

Activity	MAE	RMSE	MAED
Bench weight lifting - hard	$1.21 \pm 0.98$	$1.28 \pm 1.00$	$1.79 \pm 1.27$
Bench weight lifting - light	$1.02 \pm 0.53$	$1.06 \pm 0.53$	$1.52 \pm 0.65$
Bench weight lifting - moderate	$0.94 \pm 0.59$	$1.00 \pm 0.59$	$1.53 \pm 0.87$
Bicep curls - hard	$0.96 \pm 0.79$	$1.02 \pm 0.76$	$1.62 \pm 0.83$
Bicep curls - light	$0.72 \pm 0.44$	$0.80 \pm 0.42$	$1.40 \pm 0.50$
Bicep curls - moderate	$0.84 \pm 0.51$	$0.93 \pm 0.47$	$1.63 \pm 0.45$
Calisthenics - Crunches	$1.26 \pm 0.64$	$1.39 \pm 0.67$	$2.21 \pm 0.77$
Calisthenics - Sit ups	$1.22 \pm 0.31$	$1.33 \pm 0.33$	$2.13 \pm 0.58$
Cycling - Cycle hard - Cycle 80rpm	1.70 + 1.26	$1.78 \pm 0.00$	$2.46 \pm 1.38$
Cycling - Cycle light - Cycle 100rpm	$0.92 \pm 0.53$	$0.99 \pm 0.51$	$1.52 \pm 0.59$
Cycling - Cycle light - Cycle 60rpm	$0.46 \pm 0.33$	$0.50 \pm 0.32$	$0.84 \pm 0.40$
Cycling - Cycle light - Cycle 80rpm	$0.61 \pm 0.47$	$0.65 \pm 0.46$	$1.03 \pm 0.55$
Cycling - Cycle moderate - Cycle 80rpm	$1.14 \pm 0.90$	$1.26 \pm 0.90$	$2.18 \pm 1.28$
Lying down	$0.23 \pm 0.10$	$0.27 \pm 0.10$	$0.67 \pm 0.37$
Rowing - Rowing hard - Rowing 30spm	$1.08 \pm 0.70$	$1.18 \pm 0.70$	$1.88 \pm 0.87$
Rowing - Rowing light - Rowing 30spm	$0.82 \pm 0.55$	$0.91 \pm 0.59$	$1.00 \pm 0.07$ $1.44 \pm 0.82$
Rowing - Rowing moderate - Rowing 30spm	$0.96 \pm 0.54$	$1.02 \pm 0.57$	$1.50 \pm 0.77$
Running - Treadmill 4mph - Treadmill 0	$0.76 \pm 0.34$	$0.92 \pm 0.43$	$2.08 \pm 1.06$
Running - Treadmill 5mph - Treadmill 0	$0.70 \pm 0.51$ 0.94 + 0.43	$1.07 \pm 0.45$	$2.00 \pm 1.00$ 2 11 + 0 74
Running - Treadmill 6mph - Treadmill 0	$1.15 \pm 0.64$	$1.07 \pm 0.13$ $1.27 \pm 0.67$	$2.19 \pm 1.14$
Sitting	$0.34 \pm 0.14$	$0.43 \pm 0.19$	$0.94 \pm 0.46$
Sitting - Fidget feet legs	$0.31 \pm 0.11$ $0.42 \pm 0.17$	$0.15 \pm 0.19$ 0.46 ± 0.18	$0.91 \pm 0.10$ 0.81 + 0.33
Sitting - Fidget hands arms	$0.12 \pm 0.17$ 0.38 ± 0.15	$0.10 \pm 0.10$ $0.42 \pm 0.16$	$0.01 \pm 0.00$ $0.73 \pm 0.20$
Stairs - Ascend stairs	$0.30 \pm 0.19$ 0.72 + 0.30	$0.12 \pm 0.10$ $0.83 \pm 0.31$	$1.61 \pm 0.52$
Stairs - Descend stairs	$0.72 \pm 0.30$ 0.86 ± 0.30	$0.03 \pm 0.01$ 0.98 + 0.35	$1.01 \pm 0.52$ $1.72 \pm 0.56$
Standing	$0.00 \pm 0.00$	$0.90 \pm 0.33$ $0.62 \pm 0.34$	$1.00 \pm 0.38$
Walking - Treadmill 2mph - Treadmill 0	$0.37 \pm 0.35$ 0.37 + 0.26	$0.02 \pm 0.01$	$0.96 \pm 0.83$
Walking - Treadmill 3mph - Treadmill 0	$0.37 \pm 0.20$ 0.38 + 0.20	$0.45 \pm 0.20$	$0.90 \pm 0.03$
Walking - Treadmill 3mph - Treadmill 3 - light	$0.20 \pm 0.20$ $0.40 \pm 0.17$	$0.45 \pm 0.18$	$0.87 \pm 0.29$
Walking - Treadmill 3mph - Treadmill 6 - moderate	$0.57 \pm 0.27$	$0.64 \pm 0.27$	$1.24 \pm 0.42$
Walking - Treadmill 3mph - Treadmill 9 - hard	$0.87 \pm 0.27$ 0.84 + 0.48	$0.01 \pm 0.27$ $0.89 \pm 0.47$	$1.21 \pm 0.12$ $1.43 \pm 0.58$
kneeling	$0.47 \pm 0.34$	$0.52 \pm 0.33$	$0.90 \pm 0.37$
unknown	$0.82 \pm 0.21$	$1.04 \pm 0.27$	$3.51 \pm 1.30$
Carrying groceries	$0.55 \pm 0.28$	$0.62 \pm 0.31$	$1.14 \pm 0.52$
Doing dishes	$0.34 \pm 0.12$	$0.02 \pm 0.01$ $0.40 \pm 0.13$	$0.85 \pm 0.27$
Gardening	$0.34 \pm 0.16$	$0.40 \pm 0.18$	$0.89 \pm 0.38$
Ironing	$0.33 \pm 0.15$	$0.10 \pm 0.10$ $0.40 \pm 0.20$	$0.85 \pm 0.49$
Making the bed	$0.60 \pm 0.10$	$0.69 \pm 0.31$	$1.36 \pm 0.50$
Monning	$0.58 \pm 0.31$	$0.67 \pm 0.32$	$1.25 \pm 0.53$ $1.25 \pm 0.51$
Plaving videogames	$0.30 \pm 0.31$ $0.23 \pm 0.14$	$0.07 \pm 0.02$ $0.28 \pm 0.14$	$0.63 \pm 0.26$
Scrubbing a surface	$0.23 \pm 0.11$ 0.60 + 0.28	$0.20 \pm 0.11$	$1.45 \pm 0.54$
Stacking groceries	$0.00 \pm 0.20$ 0.51 + 0.22	$0.62 \pm 0.29$	$1.19 \pm 0.61$ $1.29 \pm 0.66$
Sweening	$0.31 \pm 0.22$ 0.47 + 0.17	$0.02 \pm 0.129$	$1.29 \pm 0.00$ $1.11 \pm 0.23$
Typing	$0.29 \pm 0.10$	$0.34 \pm 0.11$	$0.71 \pm 0.23$
Vacuuming	$0.54 \pm 0.39$	$0.61 \pm 0.38$	$1.22 \pm 0.46$
Walking around block	$0.46 \pm 0.15$	$0.56 \pm 0.17$	$1.18 \pm 0.35$
Washing windows	$0.10 \pm 0.10$ $0.50 \pm 0.21$	$0.60 \pm 0.17$ 0.61 + 0.23	$1.10 \pm 0.00$ $1.31 \pm 0.39$
Watching TV	$0.24 \pm 0.14$	$0.28 \pm 0.15$	$0.62 \pm 0.24$
Weeding	$0.51 \pm 0.14$	$0.58 \pm 0.29$	$0.97 \pm 0.42$
Wiping/Dusting	$0.34 \pm 0.17$	$0.42 \pm 0.20$	$0.90 \pm 0.35$
Writing	$0.25 \pm 0.17$	$0.29 \pm 0.12$	$0.54 \pm 0.20$
taking out trash	$0.50 \pm 0.34$	$0.58 \pm 0.35$	$1.15 \pm 0.61$

Table B14-3: Estimation of energy expenditure using linear regression and the *ScaledHR+ACFFTPeaks* feature evaluated in a subject independent manner. The features were computed over sliding windows of 5.6s in length.

**Appendix B15: Feature Selection over Subsets of Accelerometers for Energy Expenditure Estimation** 

Activity	RMSE	MAE	MAED
Bench_weight_liftinghard	$0.37 \pm 0.25$	$0.32 \pm 0.21$	$0.77 \pm 0.50$
Bench_weight_liftinglight	$0.53 \pm 0.25$	$0.48 \pm 0.25$	$0.88 \pm 0.35$
Bench_weight_liftingmoderate	$0.56 \pm 0.26$	$0.53 \pm 0.24$	$0.86 \pm 0.33$
Bicep_curlshard	$1.22 \pm 0.54$	$1.20 \pm 0.55$	$1.57 \pm 0.58$
Bicep_curlslight	$0.92 \pm 0.42$	$0.86 \pm 0.44$	$1.46 \pm 0.60$
Bicep curls - moderate	$1.13 \pm 0.35$	$1.08 \pm 0.35$	$1.63 \pm 0.28$
Calisthenics - Crunches	$1.31 \pm 0.42$	$1.17 \pm 0.42$	$2.07 \pm 0.70$
Calisthenics - Sit ups	$1.75 \pm 0.86$	$1.54 \pm 0.77$	$2.58 \pm 1.04$
Cycling - Cycle hard - Cycle 80rpm	$2.94 \pm 0.95$	$2.89 \pm 0.98$	$3.50 \pm 0.91$
CyclingCycle_lightCycle_100rpm	$1.60 \pm 0.89$	$1.51 \pm 0.93$	$2.24 \pm 0.85$
CyclingCycle_lightCycle_60rpm	$1.00 \pm 0.48$	$0.97 \pm 0.50$	$1.31 \pm 0.45$
CyclingCycle_lightCycle_80rpm	$1.33 \pm 0.72$	$1.28 \pm 0.74$	$1.72 \pm 0.75$
CyclingCycle_moderateCycle_80rpm	$2.36 \pm 0.87$	$2.29 \pm 0.87$	$2.94 \pm 0.84$
Lying_down	$0.70 \pm 0.16$	$0.69 \pm 0.16$	$0.90 \pm 0.13$
RowingRowing_hardRowing_30spm	$3.06 \pm 1.80$	$2.95 \pm 1.70$	$3.70 \pm 2.14$
Rowing - Rowing light - Rowing 30spm	$2.10 \pm 1.35$	$1.94 \pm 1.23$	$2.77 \pm 1.59$
Rowing - Rowing moderate - Rowing 30spm	$2.82 \pm 1.67$	$2.70 \pm 1.64$	$3.49 \pm 1.92$
Running - Treadmill 4mph - Treadmill 0	$1.19 \pm 0.48$	$0.97 \pm 0.43$	$2.68 \pm 1.25$
Running - Treadmill 5mph - Treadmill 0	$1.03 \pm 0.38$	$0.88 \pm 0.36$	$2.07 \pm 0.98$
Running - Treadmill 6mph - Treadmill 0	$0.99 \pm 0.49$	$0.90 \pm 0.47$	$1.79 \pm 1.04$
Sitting	$0.63 \pm 0.16$	$0.60 \pm 0.18$	$0.86 \pm 0.12$
Sitting - Fidget feet legs	$1.03 \pm 0.35$	$1.01 \pm 0.35$	$1.39 \pm 0.45$
Sitting - Fidget hands arms	$1.00 \pm 0.31$	$0.98 \pm 0.32$	$1.34 \pm 0.35$
Stairs - Ascend stairs	$0.85 \pm 0.22$	$0.73 \pm 0.21$	$1.66 \pm 0.40$
Stairs - Descend stairs	$1.39 \pm 0.29$	$1.27 \pm 0.30$	$2.34 \pm 0.27$
Standing	$0.64 \pm 0.15$	$0.61 \pm 0.16$	$0.90 \pm 0.19$
Walking - Treadmill 2mph - Treadmill 0	$0.87 \pm 0.43$	$0.83 \pm 0.45$	$1.49 \pm 0.53$
Walking - Treadmill 3mph - Treadmill 0	$1.00 \pm 0.48$	$0.94 \pm 0.50$	$1.64 \pm 0.68$
Walking - Treadmill 3mph - Treadmill 3 - light	$0.60 \pm 0.37$	$0.53 \pm 0.37$	$1.10 \pm 0.52$
Walking - Treadmill 3mph - Treadmill 6 - moderate	$0.73 \pm 0.41$	$0.67 \pm 0.42$	$1.16 \pm 0.45$
Walking - Treadmill 3mph - Treadmill 9 - hard	$1.42 \pm 0.73$	$1.38 \pm 0.73$	$1.89 \pm 0.79$
kneeling	$0.54 \pm 0.19$	$0.51 \pm 0.20$	$0.87 \pm 0.24$
unknown	$1.54 \pm 0.35$	$1.16 \pm 0.24$	$5.15 \pm 1.17$
Carrying_groceries	$1.12 \pm 0.40$	$0.97 \pm 0.40$	$2.23 \pm 0.81$
Doing dishes	$0.69 \pm 0.16$	$0.64 \pm 0.17$	$1.21 \pm 0.33$
Gardening	$0.62 \pm 0.41$	$0.54 \pm 0.39$	$1.17 \pm 0.65$
Ironing	$0.65 \pm 0.18$	$0.59 \pm 0.18$	$1.15 \pm 0.44$
Making the bed	$0.93 \pm 0.30$	$0.79 \pm 0.27$	$1.93 \pm 0.57$
Mopping	$0.59 \pm 0.22$	$0.51 \pm 0.22$	$1.31 \pm 0.47$
Plaving videogames	$0.76 \pm 0.16$	$0.75 \pm 0.17$	$1.05 \pm 0.32$
Scrubbing a surface	$0.86 \pm 0.27$	$0.71 \pm 0.24$	$1.91 \pm 0.55$
Stacking groceries	$0.84 \pm 0.29$	$0.77 \pm 0.31$	$1.43 \pm 0.39$
Sweeping	$0.55 \pm 0.28$	$0.45 \pm 0.23$	$1.18 \pm 0.44$
Typing	$0.71 \pm 0.14$	$0.68 \pm 0.14$	$1.00 \pm 0.21$
Vacuuming	$0.55 \pm 0.25$	$0.47 \pm 0.24$	$1.17 \pm 0.47$
Walking around block	$1.60 \pm 0.36$	$1.48 \pm 0.39$	$2.73 \pm 0.44$
Washing windows	$0.83 \pm 0.37$	0.71 ± 0.34	$1.67 \pm 0.77$
Watching TV	$0.75 \pm 0.15$	$0.74 \pm 0.15$	$1.04 \pm 0.34$
Weeding	$0.63 \pm 0.27$	$0.54 \pm 0.23$	$1.27 \pm 0.64$
Wiping/Dusting	$0.85 \pm 0.26$	$0.74 \pm 0.24$	$1.78 \pm 0.53$
Writing	$0.74 \pm 0.19$	$0.73 \pm 0.20$	$0.95 \pm 0.21$
taking out trash	$0.78 \pm 0.21$	$0.65 \pm 0.18$	$1.65 \pm 0.37$

Table B15-1: Energy expenditure estimation using linear regression and the *ACAbsArea* feature computed per sensor over the accelerometers at the hip, dominant wrist and dominant foot evaluated in a subject independent manner. The feature is computed over windows of 5.6s in length.

Activity	RMSE	MAE	MAED
Bench_weight_liftinghard	$0.46 \pm 0.21$	$0.39\pm0.18$	$0.94 \pm 0.34$
Bench_weight_liftinglight	$0.59 \pm 0.22$	$0.52\pm0.22$	$1.12 \pm 0.35$
Bench_weight_liftingmoderate	$0.62 \pm 0.30$	$0.56\pm0.28$	$1.06 \pm 0.44$
Bicep_curlshard	$1.54 \pm 0.74$	$1.45\pm0.77$	$2.38\pm0.75$
Bicep_curlslight	$1.22 \pm 0.55$	$1.11\pm0.57$	$2.12 \pm 0.53$
Bicep_curlsmoderate	$1.44 \pm 0.44$	$1.33 \pm 0.46$	$2.47 \pm 0.53$
CalisthenicsCrunches	$1.50 \pm 0.60$	$1.31\pm0.53$	$2.75 \pm 1.27$
CalisthenicsSit_ups	$1.77\pm0.88$	$1.56\pm0.79$	$2.75 \pm 1.07$
CyclingCycle_hardCycle_80rpm	$2.22 \pm 1.08$	$2.14 \pm 1.10$	$3.02 \pm 1.15$
CyclingCycle_lightCycle_100rpm	$1.26 \pm 0.71$	$1.16\pm0.72$	$2.09\pm0.91$
CyclingCycle_lightCycle_60rpm	$0.72\pm0.43$	$0.68\pm0.45$	$1.10\pm0.48$
CyclingCycle_lightCycle_80rpm	$1.09 \pm 0.67$	$1.00\pm0.68$	$1.83\pm0.87$
CyclingCycle_moderateCycle_80rpm	$1.76 \pm 0.93$	$1.66\pm0.92$	$2.56 \pm 1.08$
Lying_down	$0.37\pm0.08$	$0.31\pm0.08$	$0.86\pm0.37$
RowingRowing_hardRowing_30spm	$2.30 \pm 1.79$	$2.15\pm1.69$	$3.24\pm2.13$
RowingRowing_lightRowing_30spm	$1.49 \pm 1.32$	$1.35\pm1.22$	$2.27 \pm 1.64$
RowingRowing_moderateRowing_30spm	$2.13 \pm 1.66$	$2.01 \pm 1.65$	$2.88 \pm 1.85$
RunningTreadmill_4mphTreadmill_0_	$1.31\pm0.73$	$1.08\pm0.60$	$2.94 \pm 1.76$
RunningTreadmill_5mphTreadmill_0_	$1.51\pm0.65$	$1.27\pm0.64$	$3.23\pm0.95$
RunningTreadmill_6mphTreadmill_0_	$1.65 \pm 1.05$	$1.40\pm1.04$	$3.37 \pm 1.67$
Sitting	$0.51 \pm 0.20$	$0.45\pm0.19$	$0.91\pm0.32$
SittingFidget_feet_legs	$1.05\pm0.40$	$1.00\pm0.41$	$1.55\pm0.45$
SittingFidget_hands_arms	$0.71\pm0.32$	$0.65\pm0.33$	$1.19\pm0.43$
StairsAscend_stairs	$0.85\pm0.20$	$0.72\pm0.18$	$1.72\pm0.40$
StairsDescend_stairs	$1.48\pm0.38$	$1.31\pm0.35$	$2.66\pm0.52$
Standing	$0.49\pm0.14$	$0.43\pm0.15$	$0.92\pm0.19$
WalkingTreadmill_2mphTreadmill_0_	$0.72\pm0.32$	$0.64\pm0.32$	$1.41\pm0.48$
WalkingTreadmill_3mphTreadmill_0_	$0.82\pm0.36$	$0.72\pm0.35$	$1.67\pm0.67$
WalkingTreadmill_3mphTreadmill_3light	$0.65 \pm 0.29$	$0.55\pm0.29$	$1.36\pm0.43$
WalkingTreadmill_3mphTreadmill_6moderate	$0.97\pm0.45$	$0.87\pm0.47$	$1.85\pm0.59$
WalkingTreadmill_3mphTreadmill_9hard	$1.69\pm0.62$	$1.62\pm0.64$	$2.58\pm0.78$
kneeling	$0.51\pm0.15$	$0.44\pm0.15$	$1.00\pm0.24$
unknown	$1.53\pm0.37$	$1.16\pm0.26$	$5.22 \pm 1.10$
Carrying_groceries	$0.88 \pm 0.33$	$0.75\pm0.29$	$1.92\pm0.78$
Doing_dishes	$0.64\pm0.12$	$0.58\pm0.11$	$1.23\pm0.20$
Gardening	$0.57\pm0.24$	$0.50\pm0.22$	$1.15\pm0.35$
Ironing	$0.62\pm0.17$	$0.54\pm0.15$	$1.21\pm0.41$
Making_the_bed	$0.93\pm0.30$	$0.78\pm0.26$	$2.01\pm0.71$
Mopping	$0.63\pm0.19$	$0.52\pm0.17$	$1.57\pm0.50$
Playing_videogames	$0.55\pm0.18$	$0.49\pm0.19$	$1.06\pm0.32$
Scrubbing_a_surface	$0.76\pm0.26$	$0.62\pm0.21$	$1.84\pm0.62$
Stacking_groceries	$1.13\pm0.68$	$0.99\pm0.58$	$2.05\pm1.30$
Sweeping	$0.61\pm0.28$	$0.49\pm0.22$	$1.32\pm0.46$
Typing	$0.54\pm0.13$	$0.48\pm0.13$	$1.05\pm0.31$
Vacuuming	$0.52\pm0.16$	$0.43\pm0.15$	$1.18\pm0.34$
Walking_around_block	$1.11 \pm 0.33$	$0.95\pm0.32$	$2.33\pm0.54$
Washing_windows	$0.75\pm0.28$	$0.63\pm0.27$	$1.64\pm0.55$
Watching_TV	$0.55\pm0.21$	$0.50\pm0.21$	$1.02\pm0.34$
Weeding	$0.60\pm0.31$	$0.50\pm0.26$	$1.31\pm0.71$
Wiping/Dusting	$0.72\pm0.24$	$0.61\pm0.23$	$1.56\pm0.42$
Writing	$0.57\pm0.20$	$0.52\pm0.20$	$1.02\pm0.27$
taking out trash	$0.76 \pm 0.23$	$0.63 \pm 0.21$	$1.59 \pm 0.41$

Table B15-2: Energy expenditure estimation using linear regression and the *ACFFTPeaks* feature computed per sensor over the accelerometers at the hip, dominant wrist and dominant foot evaluated in a subject independent manner. The feature is computed over windows of 5.6s in length.

Activity	MAE	RMSE	MAED
Bench_weight_liftinghard	$0.42 \pm 0.24$	$0.49 \pm 0.27$	$0.92 \pm 0.45$
Bench_weight_liftinglight	$0.52 \pm 0.26$	$0.59 \pm 0.26$	$1.11 \pm 0.34$
Bench_weight_liftingmoderate	$0.57 \pm 0.40$	$0.64 \pm 0.44$	$1.11 \pm 0.58$
Bicep_curlshard	$1.36 \pm 0.63$	$1.47 \pm 0.60$	$2.37 \pm 0.67$
Bicep_curlslight	$1.05 \pm 0.55$	$1.17 \pm 0.54$	$2.08 \pm 0.50$
Bicep_curlsmoderate	$1.21 \pm 0.51$	$1.33 \pm 0.47$	$2.27 \pm 0.60$
CalisthenicsCrunches	$1.34 \pm 0.56$	$1.55 \pm 0.64$	$2.89 \pm 1.44$
CalisthenicsSit_ups	$1.55 \pm 0.77$	$1.75 \pm 0.87$	$2.81 \pm 1.06$
CyclingCycle_hardCycle_80rpm	$2.01\pm0.90$	$2.09 \pm 0.89$	$2.91 \pm 1.00$
CyclingCycle_lightCycle_100rpm	$1.02 \pm 0.50$	$1.14 \pm 0.49$	$2.04 \pm 0.63$
CyclingCycle_lightCycle_60rpm	$0.65 \pm 0.41$	$0.70 \pm 0.39$	$1.12 \pm 0.42$
CyclingCycle_lightCycle_80rpm	$0.90\pm0.62$	$1.00\pm0.61$	$1.81\pm0.86$
CyclingCycle_moderateCycle_80rpm	$1.52 \pm 0.76$	$1.63\pm0.78$	$2.47\pm0.92$
Lying_down	$0.28\pm0.06$	$0.34\pm0.07$	$0.87\pm0.47$
RowingRowing_hardRowing_30spm	$2.05 \pm 1.63$	$2.20 \pm 1.72$	3.11 ± 1.99
RowingRowing_lightRowing_30spm	$1.29 \pm 1.16$	$1.43 \pm 1.25$	$2.25 \pm 1.44$
RowingRowing_moderateRowing_30spm	$1.90 \pm 1.57$	$2.00 \pm 1.58$	$2.71 \pm 1.79$
RunningTreadmill_4mphTreadmill_0_	$0.98 \pm 0.51$	$1.18 \pm 0.61$	$2.58 \pm 1.42$
RunningTreadmill_5mphTreadmill_0_	$1.18 \pm 0.74$	$1.37 \pm 0.73$	$2.76 \pm 0.89$
RunningTreadmill_6mphTreadmill_0_	$1.51\pm0.90$	$1.73\pm0.90$	$3.36 \pm 1.50$
Sitting	$0.41\pm0.17$	$0.48 \pm 0.20$	$0.89\pm0.36$
SittingFidget_feet_legs	$0.95 \pm 0.39$	$1.00 \pm 0.38$	$1.50 \pm 0.47$
SittingFidget_hands_arms	$0.49 \pm 0.23$	$0.56 \pm 0.22$	$1.10 \pm 0.37$
StairsAscend_stairs	$0.72\pm0.18$	$0.85 \pm 0.18$	$1.67 \pm 0.34$
StairsDescend_stairs	$1.31 \pm 0.30$	$1.44 \pm 0.27$	$2.37\pm0.30$
Standing	$0.34 \pm 0.11$	$0.40 \pm 0.11$	$0.81 \pm 0.23$
WalkingTreadmill_2mphTreadmill_0_	$0.95\pm0.38$	$1.00 \pm 0.37$	$1.63 \pm 0.49$
WalkingTreadmill_3mphTreadmill_0_	$0.83\pm0.46$	$0.91 \pm 0.44$	$1.65 \pm 0.54$
WalkingTreadmill_3mphTreadmill_3light	$0.48\pm0.19$	$0.56\pm0.19$	$1.17 \pm 0.35$
WalkingTreadmill_3mphTreadmill_6moderate	$0.76\pm0.42$	$0.85\pm0.40$	$1.58\pm0.57$
WalkingTreadmill_3mphTreadmill_9hard	$1.55\pm0.65$	$1.61\pm0.64$	$2.34\pm0.79$
kneeling	$0.41\pm0.16$	$0.49\pm0.17$	$1.07\pm0.37$
unknown	$1.15\pm0.27$	$1.53\pm0.37$	$4.96 \pm 1.12$
Carrying_groceries	$0.82\pm0.17$	$0.97\pm0.18$	$2.05\pm0.55$
Doing_dishes	$0.39\pm0.08$	$0.46\pm0.08$	$0.98\pm0.15$
Gardening	$0.49\pm0.14$	$0.58\pm0.17$	$1.22\pm0.30$
Ironing	$0.41\pm0.13$	$0.50\pm0.18$	$1.11\pm0.46$
Making_the_bed	$0.79\pm0.29$	$0.93\pm0.31$	$1.99\pm0.75$
Mopping	$0.55\pm0.17$	$0.66\pm0.17$	$1.52\pm0.27$
Playing_videogames	$0.38\pm0.15$	$0.44\pm0.16$	$0.94\pm0.28$
Scrubbing_a_surface	$0.54\pm0.20$	$0.66\pm0.24$	$1.57\pm0.50$
Stacking_groceries	$0.93\pm0.53$	$1.07\pm0.61$	$2.07 \pm 1.21$
Sweeping	$0.51\pm0.21$	$0.63\pm0.26$	$1.36\pm0.39$
Typing	$0.40\pm0.12$	$0.47\pm0.13$	$1.03\pm0.34$
Vacuuming	$0.44\pm0.20$	$0.52\pm0.22$	$1.19\pm0.44$
Walking_around_block	$1.17\pm0.26$	$1.31\pm0.24$	$2.39\pm0.26$
Washing_windows	$0.54\pm0.22$	$0.67\pm0.23$	$1.55\pm0.49$
Watching_TV	$0.43 \pm 0.20$	$0.49 \pm 0.20$	$0.96 \pm 0.33$
Weeding	$0.48\pm0.\overline{23}$	$0.59 \pm 0.30$	$1.26 \pm 0.70$
Wiping/Dusting	$0.55\pm0.2\overline{3}$	$0.66 \pm 0.23$	$1.51 \pm 0.52$
Writing	$0.45\pm0.15$	$0.52\pm0.15$	$1.02\pm0.27$
taking_out_trash	$0.68 \pm 0.20$	$0.81 \pm 0.22$	$1.66 \pm 0.42$

Table B15-3: Energy expenditure estimation using linear regression and the ACFFTPeaks + ACModVigEnergy + ACMCR feature computed per sensor over the accelerometers at the hip, dominant wrist and dominant foot evaluated in a subject independent manner. The feature is computed over windows of 5.6s in length.

## **Appendix B16: MIT Energy Expenditure Dataset Statistics**

Activity	Energy Expenditure In METs
	(Mean ± Std)
Bench weight lifting - hard	$2.041 \pm 0.234$
Bench weight lifting - light	$1.859\pm0.5$
Bench weight lifting - moderate	$1.887 \pm 0.474$
Bicep curls - hard	$1.456 \pm 0.485$
Bicep curls - light	$1.852 \pm 0.35$
Bicep curls - moderate	$1.582 \pm 0.372$
Calisthenics - Crunches	$1.751 \pm 1.161$
Calisthenics - Sit ups	$3.788 \pm 0.808$
Cycling - Cycle hard - Cycle 80rpm	$5.719 \pm 1.569$
Cycling - Cycle light - Cycle 100rpm	$5.607 \pm 0.781$
Cycling - Cycle light - Cycle 60rpm	$3.372 \pm 0.438$
Cycling - Cycle light - Cycle 80rpm	$4.362 \pm 0.602$
Cycling - Cycle moderate - Cycle 80rpm	$5.4 \pm 0.72$
Lving down	$0.99 \pm 0.16$
Rowing - Rowing hard - Rowing 30spm	5.984 ± 1.831
Rowing - Rowing light - Rowing 30spm	$4.92 \pm 1.376$
Rowing - Rowing moderate - Rowing 30spm	5,776 + 1,798
Running - Treadmill 4mph - Treadmill 0	$4844 \pm 0.799$
Running - Treadmill 5mph - Treadmill 0	6,736 + 0,765
Running - Treadmill 6mph - Treadmill 0	$7352 \pm 0.000$
Sitting	$1139 \pm 0.264$
Sitting - Fidget feet legs	$1.155 \pm 0.264$ 1 254 + 0 245
Sitting - Fidget hands arms	$1.254 \pm 0.245$ 1 176 + 0 203
Stairs - Ascend stairs	$1.170 \pm 0.203$ 3 104 + 0.23
Stairs Descend stairs	$3.104 \pm 0.23$
Standing	$1.131 \pm 0.255$
Walking Treadmill 2mph Treadmill ()	$2.627 \pm 0.255$
Walking Treadmill 3mph Treadmill 0	$2.027 \pm 0.301$
Walking Treadmill 3mph Treadmill 3 light	$4.033 \pm 0.351$
Walking Treadmill 2mph Treadmill 6 moderate	$4.055 \pm 0.572$
Walking Treadmill 3mph Treadmill 9 hard	$4.77 \pm 0.431$
knooling	1 221 ± 0 222
unknown	$1.221 \pm 0.222$
	$2.000 \pm 0.501$
Daing dichas	$3.211 \pm 0.097$
Cordoning	$1.003 \pm 0.203$
Laning	$2.219 \pm 0.423$
Maline the head	$1.000 \pm 0.427$
Making the bed	$3.094 \pm 0.541$
Mopping	$2.746 \pm 0.521$
Playing videogames	$1 \pm 0.151$
Scrubbing a surface	$2.53 \pm 0.452$
Stacking groceries	$2.031 \pm 0.409$
Sweeping	$2.522 \pm 0.283$
Typing	$1.123 \pm 0.143$
Vacuuming	$2.53 \pm 0.534$
Walking around block	$2.777 \pm 0.51$
Washing windows	$2.418 \pm 0.619$
Watching TV	$0.985 \pm 0.129$
Weeding	$2.312 \pm 0.48$
Wiping/Dusting	$2.098 \pm 0.34$
Writing	$1.03 \pm 0.186$
taking out trash	$2.452 \pm 0.417$

Table B16-1: Average energy expenditure per activity in METs (Mean  $\pm$  STD) computed for all the subjects included in the MIT energy expenditure dataset. The average was computed over all the activity duration, without eliminating periods corresponding to non steady-state energy expenditure.

Activity	Energy Expenditure in
	Kcal/Min
	(Mean ± Std)
Bench weight lifting - hard	$2.88 \pm 0.515$
Bench weight lifting - light	$2.503 \pm 0.776$
Bench weight lifting - moderate	$2.581 \pm 0.638$
Bicep curls - hard	$1.908 \pm 0.696$
Bicep curls - light	$2.389 \pm 0.427$
Bicep curls - moderate	$2.054 \pm 0.489$
Calisthenics - Crunches	$2.314 \pm 1.551$
Calisthenics - Sit ups	$4.908 \pm 1.141$
Cycling - Cycle hard - Cycle 80rpm	$7.268 \pm 2.113$
Cycling - Cycle light - Cycle 100rpm	$6.967 \pm 1.008$
Cycling - Cycle light - Cycle 60rpm	$4.272 \pm 0.497$
Cycling - Cycle light - Cycle 80rpm	$5.519 \pm 0.621$
Cycling - Cycle moderate - Cycle 80rpm	$6.708 \pm 0.928$
Lying down	$1.292 \pm 0.285$
Rowing - Rowing hard - Rowing 30spm	$7.896 \pm 2.915$
Rowing - Rowing light - Rowing 30spm	$6.423 \pm 2.092$
Rowing - Rowing moderate - Rowing 30spm	7.575 ± 2.719
Running - Treadmill 4mph - Treadmill 0	$6.054 \pm 1.074$
Running - Treadmill 5mph - Treadmill 0	8.457 ± 1.366
Running - Treadmill 6mph - Treadmill 0	9.33 ± 2.023
Sitting	$1.478 \pm 0.442$
Sitting - Fidget feet legs	$1.626 \pm 0.431$
Sitting - Fidget hands arms	$1.524 \pm 0.371$
Stairs - Ascend stairs	$3.899 \pm 0.574$
Stairs - Descend stairs	$4.319 \pm 1.16$
Standing	$1.45 \pm 0.401$
Walking - Treadmill 2mph - Treadmill 0	$3.298 \pm 0.603$
Walking - Treadmill 3mph - Treadmill 0	$4.198\pm0.6$
Walking - Treadmill 3mph - Treadmill 3 - light	$5.048 \pm 0.658$
Walking - Treadmill 3mph - Treadmill 6 - moderate	$5.97 \pm 0.746$
Walking - Treadmill 3mph - Treadmill 9 - hard	$7.084 \pm 1.181$
kneeling	$1.566 \pm 0.389$
unknown	$3.468 \pm 0.806$
Carrying groceries	4.274 ± 1.229
Doing dishes	$2.199 \pm 0.483$
Gardening	$3.106 \pm 0.94$
Ironing	$2.077 \pm 0.437$
Making the bed	$3.981 \pm 0.927$
Mopping	$3.385 \pm 0.615$
Playing videogames	$1.329 \pm 0.24$
Scrubbing a surface	$3.09 \pm 0.528$
Stacking groceries	$2.598 \pm 0.637$
Sweeping	$3.13 \pm 0.516$
Typing	$1.493 \pm 0.35$
Vacuuming	$3.098 \pm 0.519$
Walking around block	$3.738 \pm 1.185$
Washing windows	$3.116 \pm 0.718$
Watching TV	$1.318 \pm 0.277$
Weeding	$3.221 \pm 0.995$
Wiping/Dusting	$2.685 \pm 0.532$
Writing	1.342 + 0.22
taking out trash	$3.144 \pm 0.712$

Table B16-2: Average energy expenditure per activity in Kcal/min (Mean  $\pm$  STD) computed for all the subjects included in the MIT energy expenditure dataset. The average was computed over all the activity duration, without eliminating periods corresponding to non steady-state energy expenditure.

Activity	Heart Data
Acuvity	in DDM
	(Mean + Std)
Bench weight lifting - hard	$105803 \pm 17434$
Bench weight lifting - light	105,219 + 14,674
Bench weight lifting - moderate	$104.009 \pm 15.293$
Bicen curls - hard	$103.279 \pm 18.848$
Bicep curls - light	$103.816 \pm 15.641$
Bicep curls - moderate	104.364 + 14.466
Calisthenics - Crunches	$105.112 \pm 15.706$
Calisthenics - Sit ups	$125.837 \pm 18.782$
Cycling - Cycle hard - Cycle 80rpm	$124.149 \pm 28.561$
Cycling - Cycle light - Cycle 100rpm	$144.068 \pm 27.564$
Cvcling - Cvcle light - Cvcle 60rpm	$110.288 \pm 19.285$
Cvcling - Cvcle light - Cvcle 80rpm	$125.54 \pm 25.041$
Cvcling - Cvcle moderate - Cvcle 80rpm	$129.489 \pm 26.728$
Lying down	64.764 + 10.693
Rowing - Rowing hard - Rowing 30spm	144.113 + 16.766
Rowing - Rowing light - Rowing 30spm	$132.423 \pm 20.585$
Rowing - Rowing moderate - Rowing 30spm	$141.078 \pm 17.359$
Running - Treadmill 4mph - Treadmill 0	$129.144 \pm 27.538$
Running - Treadmill 5mph - Treadmill 0	$150.453 \pm 24.984$
Running - Treadmill 6mph - Treadmill 0	$157.564 \pm 22.615$
Sitting	$74.005 \pm 13.449$
Sitting - Fidget feet legs	76.991 ± 14.33
Sitting - Fidget hands arms	$78.385 \pm 14.099$
Stairs - Ascend stairs	$115.256 \pm 21.955$
Stairs - Descend stairs	$107.139 \pm 22.107$
Standing	85.288 ± 14.404
Walking - Treadmill 2mph - Treadmill 0	$94.879 \pm 24.702$
Walking - Treadmill 3mph - Treadmill 0	$103.268 \pm 23.886$
Walking - Treadmill 3mph - Treadmill 3 - light	$112.353 \pm 24.231$
Walking - Treadmill 3mph - Treadmill 6 - moderate	$121.403 \pm 25.435$
Walking - Treadmill 3mph - Treadmill 9 - hard	$134.084 \pm 27$
kneeling	81.124 ± 12.448
unknown	$102.588 \pm 16.651$
Carrying groceries	$100.205 \pm 10.553$
Doing dishes	82.965 ± 14.384
Gardening	90.935 ± 8.824
Ironing	$80.428 \pm 13.641$
Making the bed	97.61 ± 14.814
Mopping	94.711 ± 12.615
Playing videogames	70.771 ± 10.447
Scrubbing a surface	$95.536 \pm 10.933$
Stacking groceries	$84.708 \pm 10.772$
Sweeping	$90.987 \pm 12.919$
Typing	69.467 ± 10.497
Vacuuming	$88.993 \pm 10.27$
Walking around block	$94.379 \pm 9.452$
Washing windows	89.432 ± 14.269
Watching TV	67.808 ± 11.203
Weeding	88.933 ± 12.831
Wiping/Dusting	83.843 ± 12.822
Writing	71.034 ± 10.211
taking out trash	92.181 + 11.946

Table B16-3: Average heart rate per activity in beats-per-minute (BPM) (Mean  $\pm$  STD) computed for all the subjects included in the MIT energy expenditure dataset. The average was computed over all the activity duration, without eliminating periods corresponding to non steady-state heart rate.

Activity	METs
Acuvity	(Mean + Std)
Bench weight lifting - hard	$2 122 \pm 0.51$
Bench weight lifting - light	$1.946 \pm 0.532$
Bench weight lifting - moderate	$1.940 \pm 0.532$ 1 967 + 0 623
Bicen curls hard	$1.907 \pm 0.023$
Bicep curls - hard	$1.465 \pm 0.308$
Bicep curis - light Bicep curis - moderate	$1.761 \pm 0.312$ $1.529 \pm 0.217$
Calisthanias Crunchas	$1.538 \pm 0.517$ 1.402 + 1.205
Calisthenics - Clutches	$1.493 \pm 1.203$
Cualing Cuale hard Cuale 20mm	$4.300 \pm 1.201$ 5.027 ± 1.528
Cycling - Cycle liald - Cycle Solphi	$5.927 \pm 1.328$
Cycling - Cycle light - Cycle Toorpin	$3.802 \pm 0.824$
Cycling - Cycle light - Cycle bolphi	$3.593 \pm 0.484$
Cycling - Cycle light - Cycle Sorpm	$4.544 \pm 0.62$
Lating - Cycle moderate - Cycle 80rpm	$5.655 \pm 0.737$
Lying down	$0.931 \pm 0.143$
Rowing - Rowing hard - Rowing 30spm	$6.412 \pm 2.157$
Rowing - Rowing light - Rowing 30spm	$5.458 \pm 1.701$
Rowing - Rowing moderate - Rowing 30spm	6.183 ± 1.958
Running - Treadmill 4mph - Treadmill 0	5.273 ± 0.959
Running - Treadmill 5mph - Treadmill 0	$7.035 \pm 0.776$
Running - Treadmill 6mph - Treadmill 0	$7.469 \pm 0.978$
Sitting	$1.076 \pm 0.241$
Sitting - Fidget feet legs	$1.274 \pm 0.26$
Sitting - Fidget hands arms	$1.145 \pm 0.213$
Stairs - Ascend stairs	$3.352 \pm 0.311$
Stairs - Descend stairs	$4.409 \pm 1.433$
Standing	$1.055 \pm 0.254$
Walking - Treadmill 2mph - Treadmill 0	$2.722\pm0.365$
Walking - Treadmill 3mph - Treadmill 0	$3.478 \pm 0.379$
Walking - Treadmill 3mph - Treadmill 3 - light	$4.157 \pm 0.373$
Walking - Treadmill 3mph - Treadmill 6 - moderate	$4.919\pm0.432$
Walking - Treadmill 3mph - Treadmill 9 - hard	$5.766 \pm 0.599$
kneeling	$1.177 \pm 0.24$
unknown	$2.484 \pm 1.016$
Carrying groceries	$3.427 \pm 0.755$
Doing dishes	$1.672 \pm 0.266$
Gardening	$2.271 \pm 0.405$
Ironing	$1.564 \pm 0.31$
Making the bed	$3.447 \pm 0.733$
Mopping	$2.894 \pm 0.618$
Playing videogames	$0.984 \pm 0.128$
Scrubbing a surface	$2.656 \pm 0.523$
Stacking groceries	$2.051 \pm 0.558$
Sweeping	$2.601 \pm 0.34$
Typing	$1.051 \pm 0.169$
Vacuuming	$2.517 \pm 0.533$
Walking around block	$3.086 \pm 0.607$
Washing windows	2.511 ± 0.661
Watching TV	$0.949 \pm 0.127$
Weeding	$2.486 \pm 0.584$
Wiping/Dusting	$2.202 \pm 0.432$
Writing	$1.022 \pm 0.182$
taking out trash	$2575 \pm 0.485$

Table B16-4: Average energy expenditure per activity in METs (Mean  $\pm$  STD) computed for all the subjects included in the MIT energy expenditure dataset. The average was computed over the activity duration after eliminating 33.3% (1/3) of the data at the beginning of the activity. In other words, non-steady state energy expenditure periods at the beginning of each activity were first eliminated.

Activity	Kcal/min
	(Mean ± Std)
Bench weight lifting - hard	$2.974 \pm 0.716$
Bench weight lifting - light	$2.606 \pm 0.763$
Bench weight lifting - moderate	$2.686 \pm 0.825$
Bicep curls - hard	$1.949 \pm 0.735$
Bicep curls - light	$2.299 \pm 0.414$
Bicep curls - moderate	$1.995 \pm 0.411$
Calisthenics - Crunches	$1.973 \pm 1.583$
Calisthenics - Sit ups	$5.602 \pm 1.705$
Cycling - Cycle hard - Cycle 80rpm	$7.526 \pm 2.09$
Cycling - Cycle light - Cycle 100rpm	$7.289 \pm 1.09$
Cycling - Cycle light - Cycle 60rpm	$4.293 \pm 0.498$
Cycling - Cycle light - Cycle 80rpm	$5.75 \pm 0.632$
Cycling - Cycle moderate - Cycle 80rpm	$7.009 \pm 0.824$
Lying down	$1.212 \pm 0.246$
Mask off	$2.075 \pm 1.319$
Rowing - Rowing hard - Rowing 30spm	$8.446 \pm 3.289$
Rowing - Rowing light - Rowing 30spm	$7.137 \pm 2.55$
Rowing - Rowing moderate - Rowing 30spm	$8.118 \pm 2.985$
Running - Treadmill 4mph - Treadmill 0	$6.57 \pm 1.149$
Running - Treadmill 5mph - Treadmill 0	$8.842 \pm 1.456$
Running - Treadmill 6mph - Treadmill 0	$9.481 \pm 2.043$
Sitting	$1.402 \pm 0.436$
Sitting - Fidget feet legs	$1.652 \pm 0.442$
Sitting - Fidget hands arms	$1.482 \pm 0.363$
Stairs - Ascend stairs	$4.191 \pm 0.472$
Stairs - Descend stairs	$5.615 \pm 2.17$
Standing	$1.353 \pm 0.398$
Walking - Treadmill 2mph - Treadmill 0	$3.413 \pm 0.586$
Walking - Treadmill 3mph - Treadmill 0	$4.349 \pm 0.588$
Walking - Treadmill 3mph - Treadmill 3 - light	$5.205 \pm 0.68$
Walking - Treadmill 3mph - Treadmill 6 - moderate	$6.159 \pm 0.786$
Walking - Treadmill 3mph - Treadmill 9 - hard	7.251 ± 1.211
kneeling	$1.511 \pm 0.399$
unknown	$3.177 \pm 1.269$
Carrying groceries	$4.544 \pm 1.242$
Doing dishes	$2.21 \pm 0.502$
Gardening	$3.185 \pm 0.959$
Ironing	$2.034 \pm 0.353$
Making the bed	$4.426 \pm 1.084$
Mopping	$3.571 \pm 0.746$
Playing videogames	$1.31 \pm 0.243$
Scrubbing a surface	$3.238 \pm 0.592$
Stacking groceries	$2.618 \pm 0.799$
Sweeping	$3.225 \pm 0.525$
Typing	$1.388\pm0.301$
Vacuuming	$3.072 \pm 0.471$
Walking around block	$4.149 \pm 1.31$
Washing windows	$3.24 \pm 0.784$
Watching TV	$1.269 \pm 0.273$
Weeding	$3.486 \pm 1.275$
Wiping/Dusting	$2.811 \pm 0.584$
Writing	$1.331 \pm 0.214$
taking out trash	$3.297 \pm 0.764$

Table B16-5: Average energy expenditure per activity in Kcal/min (Mean  $\pm$  STD) computed for all the subjects included in the MIT energy expenditure dataset. The average was computed over the activity duration after eliminating 33.3% (1/3) of the data at the beginning of the activity. In other words, non-steady state energy expenditure periods at the beginning of each activity were first eliminated.

	<b>TT</b>
Activity	Heart rate
	In BPM
	(Mean ± Std)
Bench weight lifting - hard	$108.849 \pm 18.292$
Bench weight lifting - light	$107.459 \pm 16.378$
Bench weight lifting - moderate	$105.803 \pm 14.457$
Bicep curls - hard	$103.969 \pm 21.227$
Bicep curls - light	$104.206 \pm 15.464$
Bicep curls - moderate	$104.839 \pm 16.127$
Calisthenics - Crunches	103.747 + 14.175
Calisthenics - Sit ups	129.052 + 16.831
Cycling - Cycle hard - Cycle 80rnm	124215 + 28577
Cycling - Cycle light - Cycle 100rpm	$147408 \pm 28.978$
Cycling Cycle light Cycle forpm	$110.627 \pm 20.926$
Cycling - Cycle light - Cycle 001pm	$127.22 \pm 25.002$
Cycling - Cycle nght - Cycle Solphi	$127.55 \pm 25.605$
Cycling - Cycle moderate - Cycle 80rpm	$132.088 \pm 28.317$
Lying down	$63.215 \pm 10.275$
Rowing - Rowing hard - Rowing 30spm	147.396 ± 18.017
Rowing - Rowing light - Rowing 30spm	$136.31 \pm 21.743$
Rowing - Rowing moderate - Rowing 30spm	$144.858 \pm 18.085$
Running - Treadmill 4mph - Treadmill 0	$131.842 \pm 29.207$
Running - Treadmill 5mph - Treadmill 0	$152.657 \pm 25.488$
Running - Treadmill 6mph - Treadmill 0	$159.805 \pm 22.842$
Sitting	$71.226 \pm 14.892$
Sitting - Fidget feet legs	$77.391 \pm 14.46$
Sitting - Fidget hands arms	78.595 ± 14.396
Stairs - Ascend stairs	$119.833 \pm 23.442$
Stairs - Descend stairs	115.07 ± 27.293
Standing	84.344 + 14.72
Walking - Treadmill 2mph - Treadmill 0	94 794 + 25 079
Walking - Treadmill 3mph - Treadmill 0	$103981 \pm 23966$
Walking - Treadmill 3mph - Treadmill 3 - light	$113.649 \pm 24.526$
Walking - Treadmill 3mph - Treadmill 6 - moderate	122 897 + 25 634
Walking Treadmill 3mph Treadmill 9 hard	$122.877 \pm 25.034$ $135.759 \pm 27.694$
Impoling	$70.720 \pm 12.207$
wheeling	19.729 ± 15.307
	97.78 ± 22.930
Carrying groceries	$103.357 \pm 10.957$
Doing dishes	84.161 ± 13.878
Gardening	93.015 ± 8.096
Ironing	$80.183 \pm 12.836$
Making the bed	99.07 ± 15.228
Mopping	96.53 ± 13.969
Playing videogames	$70.829 \pm 10.693$
Scrubbing a surface	$97.137 \pm 12.125$
Stacking groceries	$85.968 \pm 7.527$
Sweeping	92.062 ± 14.429
Typing	69.155 ± 10.716
Vacuuming	89.592 ± 10.741
Walking around block	$96.73 \pm 9.189$
Washing windows	$91.046 \pm 15.489$
Watching TV	67 345 + 11 389
Weeding	91 688 + 13 852
Wining/Dusting	84 183 + 13 725
Writing	$70.87 \pm 10.487$
taking out trash	$03.707 \pm 10.407$
taking out trash	$73.171 \pm 14.404$

Table B16-6: Average heart rate per activity in beats-per-minute (Mean  $\pm$  STD) computed for all the subjects included in the MIT energy expenditure dataset. The average was computed over the activity duration after eliminating 33.3% (1/3) of the data at the beginning of the activity. In other words, non-steady state heart rate periods at the beginning of each activity were first eliminated.

**Appendix B17: Data Collection Sessions Included in the MIT Energy Expenditure Dataset**
Subject	Gym Dataset	Cleaning Dataset	
MIT-001	Yes	Yes	
MIT-002	Yes	Yes	
MIT-003	Yes	Yes	
MIT-004	Yes	Yes	
MIT-005	No	No	
MIT-006	Yes	Yes	
MIT-007	Yes	No	
MIT-008	No	No	
MIT-009	Yes	No	
MIT-010	No	No	
MIT-011	Yes	Yes	
MIT-012	No	No	
MIT-013	Yes	Yes	
MIT-014	Yes	Yes	
MIT-015	No	Yes	
MIT-016	Yes	No	
MIT-017	Yes	Yes	
MIT-018	Yes	Yes	
MIT-019	Yes	Yes	
MIT-020	Yes	No	

Table A17-1: Datasets utilized in energy expenditure experiments. The datasets marked as No were left out from the analysis due to consistent low readings from the Cosmed K4b2 indirect calorimeter perhaps caused by an improper attachment of the face mask as identified by visual inspection of the datasets.

	Gym Dataset	Cleaning Dataset
Total datasets	15	12
Different subjects	16	

Table A17-1: Total number of datasets utilized in the energy expenditure experiments and total number of different subjects utilized in the experiments after eliminating datasets with suspicious Cosmed K4b2 indirect calorimeter readings identified by visual inspection.

**Appendix B18: Performance Measures for Energy Expenditure Estimation** 

Error Measure	Brief Description	Formula
Root Mean Squared Error (RMSE)	It is the most often used measure of error. It is also known as the Standard Error for the Estimate (SEE) in linear regression. The smaller the RMSE, the better the fit to the data. It shows the error in the same units and scale as the predicted values	$\sqrt{\frac{1}{N}\sum_{i=1}^{N}(p_i-a_i)^2}$
Mean Absolute Error (MAE)	Also known as the average absolute deviation. It is a weighted average of the absolute errors, similar to RMSE but less sensitive to large prediction errors and preferred over small datasets. It shows the error in the same units and scale as the predicted values	$\frac{1}{N}\sum_{i=1}^{N} \left  p_{i} - a_{i} \right $
Maximum Absolute Error Deviation (MAED)	It is the maximum absolute error between predicted and actual values over all the data points. It gives an idea of the maximum magnitude in the prediction error one can expect.	$\max( p_1 - a_1 ,,  p_N - a_N )$
Pearl's Correlation Coefficient (r)	Measures the linear relationship between the predicted and actual values. The value varies between [-1, +1], -1 indicating a decreasing linear relationship and +1 indicating an increasing linear relationship. The larger the absolute value of the number, the better the fit to the data.	$r = \frac{S_{PA}}{S_P S_A}$ where $S_{PA} = \frac{\sum_i (p_i - \overline{p})(a_i - \overline{a})}{N - 1}$ $S_P = \frac{\sum_i (p_i - \overline{p})^2}{N - 1}$ $S_A = \frac{\sum_i (a_i - \overline{a})^2}{N - 1}$
Squared Pearson's Correlation Coefficient (r <sup>2</sup> )	Also known as coefficient of determination. Its value varies from zero to one and it may be interpreted as the proportion of variance in the dependent variable that can be accounted for by the regression equation. For example, $r^2 = 0.62$ indicates that 62% of the variance in the dependent variable can be explained by the given regression equation. The remaining 38% is unexplained.	$r^2 = \left(\frac{S_{PA}}{S_P S_A}\right)^2$

 Table B18-1: Some popular performance measures utilized during energy expenditure estimation.

 The performance measures utilized in this work are shown in bold letters.

## **Appendix B19: Activity-dependent Regression Models**

Activity	RMSE	MAED
Bench weight lifting - hard	$0.60 \pm 0.26$	$1.32 \pm 0.65$
Bench weight lifting - light	$0.88 \pm 0.73$	$2.30 \pm 2.52$
Bench weight lifting - moderate	$1.29 \pm 0.92$	$3.17 \pm 2.80$
Bicep curls - hard	$0.55 \pm 0.18$	$1.28 \pm 0.38$
Bicep curls - light	$0.63 \pm 0.37$	$1.66 \pm 1.00$
Bicep curls - moderate	$0.54 \pm 0.40$	$1.39 \pm 1.50$
Calisthenics - Crunches	$1.53 \pm 0.79$	$4.05 \pm 2.92$
Calisthenics - Sit ups	$2.18 \pm 0.93$	$3.74 \pm 1.37$
Cycling - Cycle hard - Cycle 80rpm	$1.96 \pm 0.75$	$3.30 \pm 0.79$
Cycling - Cycle light - Cycle 100rpm	$1.41 \pm 0.79$	$2.85 \pm 1.04$
Cycling - Cycle light - Cycle 60rpm	$0.96\pm0.55$	$2.18\pm0.99$
Cycling - Cycle light - Cycle 80rpm	$1.24 \pm 0.50$	$2.50\pm0.73$
Cycling - Cycle moderate - Cycle 80rpm	$1.35 \pm 0.53$	$2.90 \pm 1.13$
Lying down	$0.20 \pm 0.08$	$0.41 \pm 0.15$
Rowing - Rowing hard - Rowing 30spm	$1.84 \pm 0.99$	$3.78 \pm 2.49$
Rowing - Rowing light - Rowing 30spm	$1.88 \pm 1.29$	$4.58 \pm 5.44$
Rowing - Rowing moderate - Rowing 30spm	$1.90 \pm 1.07$	$3.79 \pm 2.20$
Running - Treadmill 4mph - Treadmill 0	$1.43 \pm 0.58$	$2.88 \pm 0.95$
Running - Treadmill 5mph - Treadmill 0	$1.99 \pm 1.61$	$4.03 \pm 2.98$
Running - Treadmill 6mph - Treadmill 0	$1.96 \pm 1.68$	$3.17 \pm 2.79$
Sitting	$0.30 \pm 0.14$	$0.71 \pm 0.36$
Sitting - Fidget feet legs	$1.70 \pm 1.46$	$4.23 \pm 2.81$
Sitting - Fidget hands arms	$0.73 \pm 0.53$	$1.66 \pm 1.27$
Stairs - Ascend stairs	$1.03 \pm 0.19$	$2.64 \pm 0.75$
Stairs - Descend stairs	$1.52 \pm 0.32$	$2.93 \pm 0.64$
Standing	$0.31 \pm 0.13$	$0.72 \pm 0.47$
Walking - Treadmill 2mph - Treadmill 0	$1.10\pm0.48$	$2.65\pm0.98$
Walking - Treadmill 3mph - Treadmill 0	$1.26 \pm 0.44$	$2.34\pm0.92$
Walking - Treadmill 3mph - Treadmill 3 - light	$1.30 \pm 0.33$	$2.36\pm0.65$
Walking - Treadmill 3mph - Treadmill 6 - moderate	$1.60 \pm 0.51$	$2.76\pm0.66$
Walking - Treadmill 3mph - Treadmill 9 - hard	$2.10\pm0.83$	$3.22\pm0.88$
kneeling	$0.34\pm0.15$	$0.73\pm0.45$
Carrying groceries	$1.13\pm0.43$	$2.97\pm0.95$
Doing dishes	$0.62\pm0.36$	$1.98 \pm 2.05$
Gardening	$0.85\pm0.53$	$2.47 \pm 1.69$
Ironing	$0.75\pm0.47$	$2.29 \pm 2.55$
Making the bed	$1.15\pm0.59$	$2.95 \pm 1.83$
Mopping	$0.91 \pm 0.48$	$2.33 \pm 1.33$
Playing videogames	$0.43\pm0.44$	$1.81\pm2.60$
Scrubbing a surface	$0.79\pm0.40$	$1.90\pm0.84$
Stacking groceries	$1.10\pm0.48$	$3.18 \pm 1.97$
Sweeping	$1.08\pm0.79$	$3.00 \pm 1.87$
Typing	$0.31\pm0.13$	$1.13\pm0.96$
Vacuuming	$0.89\pm0.81$	$3.42\pm4.60$
Walking around block	$1.46\pm0.42$	$3.69 \pm 1.07$
Washing windows	$1.16 \pm 1.01$	$4.57 \pm 4.51$
Watching TV	$0.\overline{29 \pm 0.19}$	$0.\overline{98 \pm 1.21}$
Weeding	$1.42 \pm 1.38$	$4.93 \pm 7.79$
Wiping/Dusting	$0.89\pm0.5\overline{5}$	$2.84 \pm 2.43$
Writing	$0.26 \pm 0.10$	$0.69\pm0.4\overline{7}$
taking out trash	$0.90 \pm 0.32$	$3.09 \pm 1.16$

Table B19-1: Estimation of energy expenditure using activity-dependent linear regression models in a subject independent manner using the *invariant reduced* feature set during activity recognition and the *ACFFTPeaks* and a linear regression model per activity during energy expenditure estimation (ARSISI LR). Features were computed over sliding windows of 5.6s in length. Features are also computed over sensors at the hip, dominant wrist, and dominant foot.

Activity	RMSE	MAED
Bench weight lifting - hard	$0.56 \pm 0.23$	$1.19 \pm 0.45$
Bench weight lifting - light	$0.71 \pm 0.45$	$1.41 \pm 0.87$
Bench weight lifting - moderate	$0.85 \pm 0.49$	$1.66 \pm 0.84$
Bicep curls - hard	$0.57 \pm 0.30$	$1.04 \pm 0.78$
Bicep curls - light	$0.53 \pm 0.34$	$1.37 \pm 0.95$
Bicep curls - moderate	$0.49 \pm 0.28$	$1.12 \pm 1.10$
Calisthenics - Crunches	$1.45 \pm 0.65$	$2.63 \pm 1.42$
Calisthenics - Sit ups	$2.12 \pm 1.11$	$3.47 \pm 1.18$
Cycling - Cycle hard - Cycle 80rpm	$1.81 \pm 0.89$	$3.13 \pm 1.01$
Cycling - Cycle light - Cycle 100rpm	$1.18 \pm 0.76$	$2.59 \pm 1.09$
Cycling - Cycle light - Cycle 60rpm	$0.90 \pm 0.53$	$1.90 \pm 0.82$
Cycling - Cycle light - Cycle 80rpm	$1.15 \pm 0.51$	$2.23 \pm 0.66$
Cycling - Cycle moderate - Cycle 80rpm	$1.22 \pm 0.64$	$2.64 \pm 1.27$
Lying down	$0.19 \pm 0.08$	$0.38 \pm 0.15$
Rowing - Rowing hard - Rowing 30spm	$1.70 \pm 1.08$	$3.06 \pm 1.69$
Rowing - Rowing light - Rowing 30spm	$1.69 \pm 0.69$	$3.39 \pm 1.46$
Rowing - Rowing moderate - Rowing 30spm	$1.81 \pm 0.85$	$3.34 \pm 1.03$
Running - Treadmill 4mph - Treadmill 0	$1.33 \pm 0.55$	$2.72 \pm 1.19$
Running - Treadmill 5mph - Treadmill 0	$1.57 \pm 1.05$	$2.86 \pm 1.02$
Running - Treadmill 6mph - Treadmill 0	$1.45 \pm 1.26$	$2.32 \pm 1.37$
Sitting	$0.29 \pm 0.13$	$0.75 \pm 0.54$
Sitting - Fidget feet legs	$1.30 \pm 0.80$	$3.15 \pm 1.54$
Sitting - Fidget hands arms	$0.64 \pm 0.35$	$1.40 \pm 1.02$
Stairs - Ascend stairs	$0.98 \pm 0.16$	$2.34 \pm 0.77$
Stairs - Descend stairs	$1.47 \pm 0.33$	$2.81 \pm 0.77$
Standing	$0.30 \pm 0.12$	$0.74 \pm 0.50$
Walking - Treadmill 2mph - Treadmill 0	$1.07 \pm 0.49$	$2.58 \pm 1.01$
Walking - Treadmill 3mph - Treadmill 0	$1.15 \pm 0.51$	$2.07 \pm 0.85$
Walking - Treadmill 3mph - Treadmill 3 - light	$1.14 \pm 0.34$	$1.98 \pm 0.42$
Walking - Treadmill 3mph - Treadmill 6 - moderate	$1.48\pm0.50$	$2.30\pm0.50$
Walking - Treadmill 3mph - Treadmill 9 - hard	$2.09\pm0.77$	$3.02\pm0.82$
kneeling	$0.35\pm0.16$	$0.79\pm0.60$
Carrying groceries	$1.06\pm0.39$	$2.67\pm0.81$
Doing dishes	$0.66\pm0.44$	$1.62\pm0.99$
Gardening	$0.66\pm0.42$	$2.00 \pm 1.52$
Ironing	$0.71\pm0.37$	$1.88 \pm 1.22$
Making the bed	$1.06\pm0.41$	$2.47 \pm 1.22$
Mopping	$0.91 \pm 0.44$	$2.27 \pm 1.15$
Playing videogames	$0.34\pm0.23$	$1.38 \pm 1.44$
Scrubbing a surface	$0.65\pm0.23$	$1.50\pm0.56$
Stacking groceries	$0.91 \pm 0.40$	$2.31 \pm 1.12$
Sweeping	$1.02\pm0.92$	$2.14 \pm 1.30$
Typing	$0.33\pm0.18$	$1.29 \pm 1.28$
Vacuuming	$0.64\pm0.25$	$1.81\pm0.94$
Walking around block	$1.34\pm0.35$	$3.33\pm0.65$
Washing windows	$0.79\pm0.37$	$1.69\pm0.66$
Watching TV	$0.23 \pm 0.10$	$0.57 \pm 0.30$
Weeding	$0.87 \pm 0.29$	$2.01\pm0.9\overline{4}$
Wiping/Dusting	$0.81\pm0.3\overline{7}$	$2.28 \pm 1.41$
Writing	$0.25 \pm 0.12$	$0.65 \pm 0.59$
taking out trash	$0.85 \pm 0.25$	$2.76 \pm 1.03$

Table B19-2: Estimation of energy expenditure using activity-dependent models in a subject independent manner using the *invariant reduced* feature set during activity recognition and mean value predictions per activity during energy expenditure estimation (ARSISI Mean). Features were computed over sliding windows of 5.6s in length. Features are also computed over sensors at the hip, dominant wrist, and dominant foot.

Activity	RMSE	MAED
Bench weight lifting - hard	$0.59 \pm 0.19$	$1.22 \pm 0.57$
Bench weight lifting - light	$0.78 \pm 0.21$	$1.62 \pm 0.55$
Bench weight lifting - moderate	$0.91 \pm 0.36$	$1.95 \pm 0.89$
Bicep curls - hard	$0.72 \pm 0.37$	$1.18 \pm 0.37$
Bicep curls - light	$0.47 \pm 0.25$	$1.18 \pm 0.74$
Bicep curls - moderate	$0.50 \pm 0.29$	$1.07 \pm 0.58$
Calisthenics - Crunches	$2.01 \pm 1.25$	$5.46 \pm 4.60$
Calisthenics - Sit ups	$1.96 \pm 0.98$	$4.07 \pm 1.54$
Cycling - Cycle hard - Cycle 80rpm	$1.61 \pm 0.98$	$2.74 \pm 1.22$
Cycling - Cycle light - Cycle 100rpm	$1.13 \pm 0.40$	$2.36\pm0.67$
Cycling - Cycle light - Cycle 60rpm	$0.57\pm0.25$	$1.15 \pm 0.53$
Cycling - Cycle light - Cycle 80rpm	$0.92\pm0.37$	$2.17 \pm 1.02$
Cycling - Cycle moderate - Cycle 80rpm	$1.11 \pm 0.40$	$2.42 \pm 1.16$
Lying down	$0.20 \pm 0.08$	$0.47 \pm 0.25$
Rowing - Rowing hard - Rowing 30spm	$2.47 \pm 1.45$	$5.57 \pm 5.11$
Rowing - Rowing light - Rowing 30spm	$1.51\pm0.88$	$3.35 \pm 2.09$
Rowing - Rowing moderate - Rowing 30spm	$1.54\pm0.48$	$3.46 \pm 1.28$
Running - Treadmill 4mph - Treadmill 0	$1.26 \pm 0.60$	$2.52 \pm 1.07$
Running - Treadmill 5mph - Treadmill 0	$1.44 \pm 0.82$	$3.54 \pm 3.33$
Running - Treadmill 6mph - Treadmill 0	$1.32 \pm 0.61$	$2.34 \pm 1.00$
Sitting	$0.33 \pm 0.21$	$0.75 \pm 0.45$
Sitting - Fidget feet legs	$0.53 \pm 0.43$	$1.57 \pm 1.62$
Sitting - Fidget hands arms	$0.38 \pm 0.22$	$0.98 \pm 0.62$
Stairs - Ascend stairs	$0.95 \pm 0.28$	$2.14 \pm 0.89$
Stairs - Descend stairs	$1.51 \pm 0.31$	$3.04 \pm 0.68$
Standing	$0.32 \pm 0.20$	$0.74 \pm 0.52$
Walking - Treadmill 2mph - Treadmill 0	$0.53 \pm 0.34$	$1.49 \pm 1.21$
Walking - Treadmill 3mph - Treadmill 0	$0.74 \pm 0.26$	$1.91 \pm 0.67$
Walking - Treadmill 3mph - Treadmill 3 - light	$0.78 \pm 0.23$	$1.72 \pm 0.53$
Walking - Treadmill 3mph - Treadmill 6 - moderate	$0.90 \pm 0.44$	$2.17\pm0.81$
Walking - Treadmill 3mph - Treadmill 9 - hard	$1.00\pm0.57$	$2.18\pm0.93$
kneeling	$0.36\pm0.18$	$0.90\pm0.81$
Carrying groceries	$0.74\pm0.19$	$1.80\pm0.49$
Doing dishes	$0.33\pm0.13$	$0.98\pm0.40$
Gardening	$0.66\pm0.24$	$1.56\pm0.54$
Ironing	$0.48\pm0.28$	$1.22\pm0.73$
Making the bed	$1.02\pm0.31$	$2.42\pm0.53$
Mopping	$0.73\pm0.32$	$1.88\pm0.85$
Playing videogames	$0.28\pm0.20$	$0.90 \pm 1.22$
Scrubbing a surface	$0.82\pm0.31$	$2.46 \pm 1.91$
Stacking groceries	$0.73\pm0.19$	$1.83\pm0.76$
Sweeping	$0.62\pm0.24$	$1.48\pm0.72$
Typing	$0.27\pm0.11$	$0.80\pm0.42$
Vacuuming	$0.60\pm0.20$	$1.48\pm0.62$
Walking around block	$0.68\pm0.12$	$1.51\pm0.38$
Washing windows	$0.70\pm0.33$	$2.11 \pm 1.42$
Watching TV	$0.28\pm0.24$	$1.07 \pm 1.49$
Weeding	$0.73 \pm 0.39$	$1.45 \pm 0.80$
Wiping/Dusting	$0.51 \pm 0.14$	$1.23 \pm 0.35$
Writing	$0.26\pm0.09$	$0.71\pm0.45$
taking out trash	$0.57 \pm 0.18$	$1.43 \pm 0.37$

Table B19-3: Estimation of energy expenditure using activity-dependent linear regression models in a subject dependent manner using the *invariant reduced* feature set during activity recognition and the *ACFFTPeaks* and a linear regression model per activity during energy expenditure estimation (ARSDSI LR). Features were computed over sliding windows of 5.6s in length. Features are also computed over sensors at the hip, dominant wrist, and dominant foot.

Activity	RMSE	MAED
Bench weight lifting - hard	$0.40 \pm 0.21$	$0.78 \pm 0.45$
Bench weight lifting - light	$0.54 \pm 0.24$	$0.90 \pm 0.34$
Bench weight lifting - moderate	$0.55 \pm 0.33$	$0.82 \pm 0.45$
Bicep curls - hard	$0.41 \pm 0.40$	$0.69 \pm 0.50$
Bicep curls - light	$0.45 \pm 0.31$	$1.14 \pm 1.11$
Bicep curls - moderate	$0.46 \pm 0.17$	$0.79 \pm 0.19$
Calisthenics - Crunches	$1.29 \pm 0.91$	$2.68 \pm 2.00$
Calisthenics - Sit ups	$1.41 \pm 0.39$	$2.62 \pm 0.52$
Cycling - Cycle hard - Cycle 80rpm	$1.27 \pm 1.00$	$2.22 \pm 1.31$
Cycling - Cycle light - Cycle 100rpm	$0.93 \pm 0.37$	$2.05 \pm 0.90$
Cycling - Cycle light - Cycle 60rpm	$0.48 \pm 0.24$	$0.95 \pm 0.62$
Cycling - Cycle light - Cycle 80rpm	$0.77 \pm 0.32$	$1.78 \pm 0.74$
Cycling - Cycle moderate - Cycle 80rpm	$0.98 \pm 0.53$	$2.33 \pm 1.28$
Lying down	$0.20 \pm 0.08$	$0.49 \pm 0.38$
Rowing - Rowing hard - Rowing 30spm	$1.92 \pm 1.03$	$2.81 \pm 1.30$
Rowing - Rowing light - Rowing 30spm	$1.62 \pm 0.88$	$2.85 \pm 0.98$
Rowing - Rowing moderate - Rowing 30spm	$1.86 \pm 0.98$	$2.84 \pm 1.07$
Running - Treadmill 4mph - Treadmill 0	$1.14 \pm 0.46$	$2.33 \pm 0.99$
Running - Treadmill 5mph - Treadmill 0	$1.06 \pm 0.34$	$2.41 \pm 1.20$
Running - Treadmill 6mph - Treadmill 0	$1.02 \pm 0.57$	$1.88 \pm 1.26$
Sitting	$0.27 \pm 0.21$	$0.51 \pm 0.43$
Sitting - Fidget feet legs	$0.46 \pm 0.24$	$1.25 \pm 1.12$
Sitting - Fidget hands arms	$0.31 \pm 0.22$	$0.75 \pm 0.58$
Stairs - Ascend stairs	$0.91 \pm 0.22$	$2.01 \pm 0.76$
Stairs - Descend stairs	$1.40 \pm 0.33$	$2.50 \pm 0.75$
Standing	$0.31 \pm 0.23$	$0.71 \pm 0.67$
Walking - Treadmill 2mph - Treadmill 0	$0.44 \pm 0.28$	$1.19 \pm 0.85$
Walking - Treadmill 3mph - Treadmill 0	$0.56 \pm 0.17$	$1.51 \pm 0.46$
Walking - Treadmill 3mph - Treadmill 3 - light	$0.59 \pm 0.15$	$1.49 \pm 0.36$
Walking - Treadmill 3mph - Treadmill 6 - moderate	$0.72 \pm 0.20$	$1.72 \pm 0.37$
Walking - Treadmill 3mph - Treadmill 9 - hard	$0.76 \pm 0.30$	$1.86 \pm 0.64$
kneeling	$0.37 \pm 0.28$	$0.93 \pm 1.37$
Carrying groceries	$0.86 \pm 0.35$	$1.82 \pm 0.70$
Doing dishes	$0.36 \pm 0.16$	$0.88 \pm 0.34$
Gardening	$0.63 \pm 0.30$	$1.64 \pm 1.10$
Ironing	$0.51 \pm 0.31$	$1.30 \pm 0.82$
Making the bed	$1.01 \pm 0.34$	$2.24 \pm 0.59$
Mopping	$0.66 \pm 0.32$	$1.36 \pm 0.45$
Playing videogames	$0.26 \pm 0.22$	$0.93 \pm 1.36$
Scrubbing a surface	$0.62 \pm 0.24$	$1.39 \pm 0.62$
Stacking groceries	$0.58 \pm 0.20$	$1.20 \pm 0.25$
Sweeping	$0.53 \pm 0.13$	$1.12 \pm 0.29$
Typing	$0.27 \pm 0.13$	$0.73 \pm 0.44$
Vacuuming	$0.59 \pm 0.32$	$1.26 \pm 0.49$
Walking around block	$0.74 \pm 0.22$	$1.43 \pm 0.45$
Washing windows	$0.70 \pm 0.28$	$1.53 \pm 0.60$
Watching TV	$0.28\pm0.20$	$1.06 \pm 1.36$
Weeding	$0.65 \pm 0.33$	$1.44 \pm 0.69$
Wiping/Dusting	$0.52 \pm 0.25$	$1.46 \pm 0.94$
Writing	$0.22 \pm 0.07$	$0.54 \pm 0.42$
taking out trash	$0.60 \pm 0.19$	$1.39 \pm 0.27$

Table B19-4: Estimation of energy expenditure using activity-dependent models in a subject dependent manner using the *invariant reduced* feature set during activity recognition and mean value predictions per activity during energy expenditure estimation (ARSDSI Means). Features were computed over sliding windows of 5.6s in length. Features are also computed over sensors at the hip, dominant wrist, and dominant foot.



Figure B19-1: Estimation of energy expenditure using activity-dependent linear regression models in a subject independent manner using the *invariant reduced* feature set during activity recognition and the *ACFFTPeaks* and a linear regression model per activity during energy expenditure estimation (ARSISI LR). Features were computed over sliding windows of 5.6s in length. Features are also computed over sensors at the hip, dominant wrist, and dominant foot.



Figure B19-2: Estimation of energy expenditure using activity-dependent models in a subject independent manner using the *invariant reduced* feature set during activity recognition and mean value predictions per activity during energy expenditure estimation (ARSISI Mean). Features were computed over sliding windows of 5.6s in length. Features are also computed over sensors at the hip, dominant wrist, and dominant foot.



Figure B19-3: Estimation of energy expenditure using activity-dependent linear regression models in a subject dependent manner using the *invariant reduced* feature set during activity recognition and the *ACFFTPeaks* and a linear regression model per activity during energy expenditure estimation (ARSDSI LR). Features were computed over sliding windows of 5.6s in length. Features are also computed over sensors at the hip, dominant wrist, and dominant foot.



Figure B19-4: Estimation of energy expenditure using activity-dependent models in a subject dependent manner using the *invariant reduced* feature set during activity recognition and mean value predictions per activity during energy expenditure estimation (ARSDSI Means). Features were computed over sliding windows of 5.6s in length. Features are also computed over sensors at the hip, dominant wrist, and dominant foot.

## **Bibliography**

- [1] World Health Organization (WHO), *Obesity: Preventing and Managing the Global Epidemic*. Geneva: Stationery Office Books, 2001.
- [2] Centers for Disease Control and Prevention (CDC). "Overweight and Obesity." 2006 [cited October 31 2006]. Available from <u>http://www.cdc.gov/nccdphp/dnpa/obesity/</u>.
- [3] J. P. Foreyt and G. K. Goodrick, "The Ultimate Triumph of Obesity," *Lancet*, vol. 346, 8968, pp. 134-135, 1995.
- [4] Center for Disease Control. "Obesity Epidemic." 2008 [cited February 21st 2008]. Available from http://www.cdc.gov/nccdphp/dnpa/obesity/index.htm.
- [5] K. H. Schmitz, D. R. Jacobs, A. S. Leon, P. J. Schreiner, and B. Sternfeld, "Physical Activity and Body Weight: Associations over Ten Years in the CARDIA Study," *International Journal of Obesity*, vol. 24, 11, pp. 1475-1487, 2000.
- [6] J. O. Hill and J. C. Peters, "Environmental Contributions to the Obesity Epidemic," *Science*, vol. 280, 5368, pp. 1371-1374, 1998.
- [7] U.S Department of Health and Human Services; Centers for Disease Control and Prevention; National Center for Chronic Disease Prevention and Health Promotion, "Physical Activity and Health: A Report of the Surgeon General," Atlanta, GA, 1996.
- [8] R. W. Jeffery and S. A. French, "Epidemic Obesity in the United States: Are Fast Foods and Television Viewing Contributing?," *American Journal of Public Health*, vol. 88, 2, pp. 277-280, 1998.
- [9] M. B. Schulze, J. E. Manson, D. S. Ludwig, G. A. Colditz, M. J. Stampfer, W. C. Willett, and F. B. Hu, "Sugar-Sweetened Beverages, Weight Gain, and Incidence of Type 2 Diabetes in Young and Middle-Aged Women.," *Journal of the American Medical Association (JAMA)*, vol. 292, pp. 927-934, 2004.
- [10] L. R. Young and M. Nestle, "The Contribution of Expanding Portion Sizes to the US Obesity Epidemic," *American Journal of Public Health*, vol. 92, 2, pp. 246-249, 2002.
- [11] F. B. Hu, M. F. Leitzman, M. J. Stampfer, G. A. Colditz, W. C. Willett, and E. B. Rimm, "Physical Activity and Television Watching in Relation to Risk for Type 2 Diabetes Mellitus in Men," *Archives of Internal Medicine*, vol. 161, pp. 1542-1548, 2001.
- [12] A. Must and D. J. Tybor, "Physical Activity and Sedentary Behavior: A Review of Longitudinal Studies of Weight and Adiposity in Youth.," *International Journal of Obesity*, vol. 28, pp. 84-96, 2005.
- [13] D. Busch, "America On the Move Launches National Grassroots Initiative Will Inspire Healthy Eating and Active Living and Fight Obesity Epidemic," in *America On the Move HHS Press Office*. Boston, MA, 2003.
- [14] R. W. Jeffery, A. Drewnowski, L. H. Epstein, A. J. Stunkard, G. T. Wilson, and R. R. Wing, "Long-Term Maintenance of Weight Loss: Current Status," *Health Psychology*, vol. 19, 1, pp. 5-16, 2000.
- [15] D. F. Williamson, M. K. Serdula, R. F. Anda, A. Levy, and T. Byers, "Weight Loss Attempts in Adults: Goals, Duration, and Rate of Weight Loss," *American Journal of Public Health*, vol. 82, pp. 1251-1257, 1992.
- [16] S. A. French, R. W. Jeffery, and D. Murray, "Is Dieting Good for you?: Prevalence, Duration and Associated Weight and Behavior Changes for Specific Weight Loss Strategies Over Four Years in US Adults," *International Journal Of Obesity*, vol. 23, pp. 320-327, 1999.
- [17] T. A. Wadden and S. Phelan, "Behavioral Assessment of the Obese Patient," in *Handbook of Obesity Treatment*, T. A. Wadden and A. J. Stunkard, Eds. New York: Guilford Press, 2002, pp. 186-226.
- [18] M. L. Klem, R. R. Wing, M. T. McGuire, H. M. Seagle, and J. O. Hill, "A Descriptive Study of Individuals Successful at Long-Term Maintenance of Substantial Weight Loss.," *American Journal of Clinical Nutrition*, vol. 66, 2, pp. 239-246, 1997.
- [19] Nike and iPod. "Nike + iPod sport kit " 2006 [cited December 6th 2006]. Available from <u>http://www.apple.com/ipod/nike/</u>.

- [20] Adidas and Polar. "Adistar Fusion Running System." Adidas and Polar, 2006 [cited January 7th 2006]. Available from <u>http://www.adidas-polar.com/</u>.
- [21] S. N. Patel, J. A. Kientz, G. R. Hayes, S. Bhat, and G. D. Abowd, "Farther Than you May Think: An Empirical Investigation of the Proximity of Users to their Mobile Phones.," in *Proceedings of Ubicomp 2006*: Springer-Veralg, 2006, pp. 123-140.
- [22] Apple. "The iPhone." 2008 [cited January 28th 2008]. Available from http://www.apple.com/iphone/.
- [23] Nokia. "Nokia N95 Phone." 2008 [cited January 28th 2008]. Available from http://www.nokiausa.com/A4546019.
- [24] Samsung. "Samsung SCH-S310 Phone." 2008 [cited January 28th 2008]. Available from <u>http://www.samsung.com/AboutSAMSUNG/ELECTRONICSGLOBAL/InvestorRelations/NewsP</u> ublicDisclosure/PressRelease/PressRelease.asp?seq=20050112\_0000096403.
- [25] Nokia. "Nokia 5500 Phone." 2008 [cited January 28th 2008]. Available from <u>http://europe.nokia.com/5500</u>.
- [26] Nokia. "Activity Monitor." 2008 [cited January 28th 2008]. Available from http://research.nokia.com/projects/activity monitor.
- [27] B. J. Fogg, "Persuasive Computers: Perspectives and Research Directions," in *Proceedings of CHI* 98, 1998, pp. 225-232.
- [28] S. S. Intille, "A New Research Challenge: Persuasive Technology to Motivate Healthy Aging," in *IEEE Transactions of Information Technology in Biomedicine*, vol. 8, 2004.
- [29] E. Ravussin, "A NEAT Way to Control Weight," Science, vol. 307, 5709, pp. 530-531, 2005.
- [30] MyFoodPhone Nutrition Inc. "MyFoodPhone." 2006 [cited December 6th 2006]. Available from <u>http://www.myfoodphone.com/</u>.
- [31] Sensei. "Sensei." 2008 [cited February 21st 2008]. Available from http://www.sensei.com/senseipublic/Default.aspx.
- [32] MTI Actigraph. "GT1M Actigraph." 2008 [cited January 5th, 2008 2008]. Available from http://mtiactigraph.com/.
- [33] J. M. Jakicic, C. Winters, K. Lagally, J. Ho, R. J. Robertson, and R. R. Wing, "The Accuracy of the Tritrac-R3D Accelerometer to Estimate Energy Expenditure," *Medicine & Science in Sports & Exercise*, vol. 31, 5, pp. 747-754, 1999.
- [34] S. E. Crouter, K. G. Clowers, and D. R. Bassett, "A Novel Method for Using Accelerometer Data to Predict Energy Expenditure," *Journal of Applied Psychology*, vol. 100, pp. 1324-1331, 2006.
- [35] Muscle Dynamics Inc. "Caltrac." 2006 [cited October 31 2006]. Available from http://www.muscledynamics.net/.
- [36] Cambridge Neurotechnology Ltd. "Actiwatch." 2006 [cited October 30 2006]. Available from <u>http://www.camntech.com/</u>.
- [37] J. Lester, T. Choudhury, N. Kern, G. Borriello, and B. Hannaford, "A Hybrid Discriminative/Generative Approach for Modeling Human Activities," in *Proceedings of the International Joint Conference on Artificial Intelligence (IJCAI '05)*, 2005, pp. 766–772.
- [38] L. Bao and S. S. Intille, "Activity Recognition from User-Annotated Acceleration Data," in *Proceedings of the Second International Conference in Pervasive Computing (PERVASIVE '04)*. Vienna, Austria, 2004, pp. 1-17.
- [39] D. Olguin Olguin and A. Pentland, "Human Activity Recognition: Accuracy Across Common Locations for Wearable Sensors," in *Proceedings of The 10th IEEE International Symposium on Wearable Computing (Student Colloquium)*, Montreaux, Switzerland, 2006.
- [40] N. Ravi, N. Dandekar, P. Mysore, and M. L. Littman, "Activity Recognition from Accelerometer Data," in *Proceedings of the Seventeenth Innovative Applications of Artificial Intelligence Conference*. Pittsburgh, Pennsylvania, 2005.
- [41] MiniSun. "The Intelligent Device for Energy Expenditure and Activity (IDEEA)." 2006 [cited October 30 2006]. Available from <u>http://www.minisun.com/</u>.
- [42] BodyMedia Inc. "The BodyBugg Armband." September 16th, 2007 2007 [cited 2007]. Available from <u>http://www.bodybugg.com/</u>.
- [43] U. Maurer, A. Rowe, A. Smailagic, and D.Siewiorek, "Location and Activity Recognition Using eWatch: A Wearable Sensor Platform," in *Ambient Intelligence in Everyday Life*: Springer Berlin/Heidelberg, 2006, pp. 86-102.

- [44] T. Choudhury and G. Borriello, "The Mobile Sensing Platform: An Embedded System for Capturing and Recognizing Human Activities," in *IEEE Pervasive Computing Special Issue on Activity-Based Computing, to appear*, 2008.
- [45] M. L. Blum, *Real-Time Context Recognition*, M.S., Department of Information Technology and Electrical Engineering, Swiss Federal Institute of Technology Zurich (ETH), 2005.
- [46] A. Feldman, E. Munguia Tapia, S. Sadi, P. Maes, and C. Schmandt, "ReachMedia: On-the-move Interaction with Everyday Objects," in *Proceedings of the Ninth Annual IEEE International Simposium on Wereable Computers (ISWC '05)*. Osaka, Japan, 2005, pp. 52-59.
- [47] G. J. Welk, *Physical Activity Assessments for Health-Related Research*: Human Kinetics Publishers, 2002.
- [48] W. L. Haskell, M. C. Lee, A. Evans, and P. J. Irby, "Simultaneous Measurement of Heart Rate and Body Measurement to Quantitate Physical Activity," *Medicine & Science in Sports & Exercise*, vol. 25, pp. 109-115, 1993.
- [49] S. J. Strath, S. Brage, and U. Ekelund, "Integration of Physiological and Accelerometer Data to Improve Physical Activity Assessment," *Medicine & Science in Sports & Exercise*, vol. 37, pp. S563-571, 2005.
- [50] J. A. Levine, M. W. V. Weg, J. O. Hill, and R. C. Klesges, "Non-Exercise Activity Thermogenesis: The Crouching Tiger Hidden Dragon of Societal Weight Gain.," *Journal of American Heart Association*, vol. 26, 4, pp. 729-736, 2006.
- [51] J. A. Levine and N. L. Eberhardt, "Role of Non-exercise Activity Thermogenesis in Resistance to Fat Gain in Humans," *Science*, pp. 212-214, 1999.
- [52] N. Marmasse and C. Schmandt, "WatchMe: Communication and Awareness Between Members of a Closely-Knit," in *Proceedings of the International Conference in Ubiquitous Computing* (UBICOMP '04). Nottingham, UK, 2004, pp. 214-231.
- [53] P. Kaushik, *The Design and Evaluation of a Mobile Handheld Intervention for Providing Context-Sensitive Medication Reminders*, M.S, Media Lab, Massachusetts Institute of Technology, 2005.
- [54] A. K. Dey and G. D. Abowd, "CybreMinder: A Context-Aware System for Supporting Reminders," in *Proceedings of the International Symposium on Handheld and Ubiquitous Computing*, vol. 1927: Springer-Verlag London, UK 2000, pp. 172-186.
- [55] N. Marmasse, "ComMotion: A Context-Aware Communication System," in *Proceedings of the International Conference on Human Factors in Computing Systems (CHI '99)*, 1999, pp. 320 -321.
- [56] N. Kern and B. Schiele, "Multi-channel, Context-aware Notification on Wearable Devices," in Proceedings of the Workshop on Design and Evaluation of Notification Interfaces for Ubiquitous Computing, L. Bartram and M. Czerwinski, Eds., 2002.
- [57] S. S. Intille, J. Rondoni, C. Kukla, I. Anacona, and L. Bao, "A Context-aware Experience Sampling Tool," in *Proceedings of the Conference on Human Factors and Computing Systems: Extended Abstracts.* New York: ACM Press, 2003, pp. 972-973.
- [58] S. S. Intille, E. Munguia Tapia, J. Rondoni, J. Beaudin, C. Kukla, S. Agarwal, and L. Bao, "Tools for Studying Behavior and Technology in Natural Settings," *Proceedings of UBICOMP 2003*, vol. LNCS 2864, pp. 157-174, 2003.
- [59] M. Sung, C. Marci, and A. Pentland, "Wearable Feedback Systems for Rehabilitation," *Journal of NeuroEngineering, and Rehabilitation*, vol. 2, 17, 2005.
- [60] J. A. Paradiso, S. J. Morris, and A. Y. Benbasat, "Interactive Therapy with Instrumented Footwear," in *Proceedings of the International Conference on Human Factors in Computing Systems (CHI '04) Extended Abstracts*. Vienna, Austria, 2004, pp. 1341-1343.
- [61] E. A. Heinz, K. S. Kunze, M. Gruber, D. Bannach, and P. Lukowicz, "Using Wearable Sensors for Real-Time Recognition Tasks in Games of Martial Arts - An Initial Experiment," in *Proceedings* fo the 2006 IEEE Symposium on Computational Intelligence and Games, 2006, pp. 98-102.
- [62] K. Kunze, M. Barry, E. A. Heinz, P.Lukowicz, D. Majoe, and J. Gutknecht, "Towards Recognizing Tai Chi – An Initial Experiment Using Wearable Sensors," in *Proceedings of the Third International Forum on Applied Wearable Computing (IFAWC '06)*. Bremen, Germany: VDE Verlag, 2006, pp. 183-190.
- [63] M. Ermes, J. Parkka, J. Mantyjarvi, and I. Korhonen, "Detection of Daily Activities and Sports With Wearable Sensors in Controlled and Uncontrolled Conditions," in *Proceedings of the IEEE Transactions on Information Technology in Biomedicine* vol. 12, 2008, pp. 20-26.

- [64] J. Healey and B. Logan, *Wearable Wellness Monitoring Using ECG and Accelerometer Data*, Technical Report: HPL-2005-134, HP Laboratories Cambridge, Cambridge, MA 2005.
- [65] A. Steptoe, "Job Control, Perceptions of Control, and Cardiovascular Activity: An Analysis of Ambulatory Measures Collected Over the Working Day," *Journal of Psychosomatic Research*, vol. 50, 2, pp. 57-63, 2001.
- [66] J. B. Bussmann, W. L. Martens, J. H. Tulen, F. C. Schasfoort, H. J. van den Berg-Emons, and H. J. Stam, "Measuring Daily Behavior Using Ambulatory Accelerometry: The Activity Monitor," *Behavior Research Methods, Instruments, & Computers*, vol. 33, 3, pp. 349-56, 2001.
- [67] Nintendo. "Nintendo Wii." 2008 [cited January 20th 2008]. Available from <u>http://www.nintendo.com/wii</u>.
- [68] N. Crampton, K. Fox, H. Johnston, and A. Whitehead, "Dance Dance Evolution: Accelerometer Sensor Networks as Input to Video Games," in *Proceedings of the 2007 IEEE International Workshop on Haptic, Audio and Visual Environments and Games (HAVE '07)*, 2007, pp. 107 -112.
- [69] D. Ashbrook, T. Westeyn, and T. Starner, "Dancing in the Streets: Smartphones and Gaming," in Proceedings of the Workshop on Ubiquitous Entertainment and Games at the 7th International Conference on Ubiquitous Computing (UbiComp 2005). Tokio, Japan, 2005.
- [70] T. Westeyn, X. Bian, K. Vadas, T. Starner, and G. Abowd, "Recognizing Mimicked Autistic Self-Stimulatory Behaviors Using Hidden Markov Models " in *Proceedings of the Ninth IEEE International Symposium on Wearable Computers (ISWC '05)*. Osaka, Japan, 2005, pp. 164-169.
- [71] J. Kientz, G. Hayes, T. Westeyn, T. Starner, and G. Abowd. "Pervasive Computing and Autism: Assisting Caregivers of Children with Special Needs." *IEEE Pervasive Magazine Special issue on Pervasive Computing in Healthcare* 2007.
- [72] J. Ho, *Interruptions: Using Activity Transitions to Trigger Proactive Messages*, M.S, Department of Electrical Engineering and Computer Science, Massachusetts Institute of Technology, 2004.
- [73] N. Kern, S. Antifakos, B. Schiele, and A. Schwaninger, "A Model for Human Interruptability: Experimental Evaluation and Automatic Estimation from Wearable Sensors," in *Proceedings of the 8th International Symposium on Wearable Computing (ISWC)*. Washington DC, USA, 2004, pp. 158-165
- [74] N. Kern, B. Schiele, H. Junker, and P. Lukowicz, "Wearable Sensing to Annotate Meeting Recordings," in *Proceedings of the Sixth International Symposium on Wearable Computers* (ISWC): IEEE Press, 2002, pp. 186-193.
- [75] N. Kern, B. Schiele, H. Junker, and A. Schmidt, "Context Annotation for A Live Life Recording," in *Personal and Ubiquitous Computing*, 2005, pp. 251-263.
- [76] M. Blum, A. Pentland, and G. Troester, *inSensed: Wearable Collection of Multimedia Based on Interest*, Technical Report 599, ETH, Zurich Switzerland, 2005.
- [77] Wibree Alliance. "Wibree Standard." 2008 [cited February 21st 2008]. Available from <a href="http://www.wibree.com/">http://www.wibree.com/</a>.
- [78] Zigbee Alliance. "Zigbee Standard." 2008 [cited February 21st 2008]. Available from http://www.zigbee.org/en/index.asp.
- [79] G. Plasqui, A. M. Joosen, A. D. Kester, A. H. Goris, and K. R. Westerterp, "Measuring Free-living Energy Expenditure and Physical Activity with Triaxial Accelerometry," *Obesity Research*, vol. 13, 8, pp. 1363-1369, 2005.
- [80] W. J. Tharion, M. Yokota, M. J. Buller, J. P. DeLany, and R. W. Hoyt, "Total Energy Expenditure Estimated Using a Foot-contact Pedometer," *Medicine & Science in Sports & Exercise*, vol. 10, 9, pp. CR504-509, 2004.
- [81] K. F. Janz, "Use of Heart Rate Monitors to Assess Physical Activity," in *Physical Activity Assessments for Health-Related Research*, G. J. Welk, Ed. Ilinois: Human Kinetics Publishers, Inc., 2002, pp. 143-161.
- [82] H. K. Hiilloskorpi, M. E. Pasanen, M. G. Fogelholm, R. M. Laukkanen, and A. T. Manttari, "Use of Heart Rate to Predict Energy Expenditure from Low to High Activity Levels," *International Journal of Sports Medicine*, vol. 24, 5, pp. 332-336, 2003.
- [83] S. J. Strath, D. R. Bassett, A. M. Swartz, and D. L. Thompson, "Simultaneous Heart Rate-Motion Sensor Technique to Estimate Energy Expenditure," *Medicine & Science in Sports & Exercise*, vol. 33, 12, pp. 2118-23, 2001.

- [84] S. Brage, N. Brage, P. W. Franks, U. Ekelund, and N. J. Wareham, "Reliability and Validity of the Combined Heart Rate and Movement Sensor Actiheart," *Journal of Clinical Nutrition*, vol. 59, 4, pp. 561-570, 2005.
- [85] Omron Healthcare. "HJ-112 Premium Pedometer." 2006 [cited October 30 2006]. Available from http://www.omronhealthcare.com/enTouchCMS/app/viewCategory?catgId=34.
- [86] Yamax Inc. "Digi-Walker Pedometer Model SW-200." 2006 [cited October 31 2006]. Available from <u>http://www.new-lifestyles.com/DIGIWALKER/sw200.html</u>.
- [87] S. M. Patterson, D. S. Krantz, L. C. Montgomery, P. A. Deuster, S. M. Hedges, and L. E. Nebel, "Automated Physical Activity Monitoring: Validation and Comparison with Physiological and Self-report Measures," *Psychophysiology*, vol. 30, 3, pp. 296-305, 1993.
- [88] A. Sugimoto, Y. Hara, T. W. Findley, and K. Yoncmoto, "A Useful Method for Measuring Daily Physical Activity by a Three-direction Monitor," *Scandinavian Journal of Rehabilitation Medicine*, vol. 29, 1, pp. 37-42, 1997.
- [89] Polar. "Polar Heart Rate Monitor with WearLink Coded Chest Transmitter." 2008 [cited March 20 2008]. Available from <u>http://www.polarusa.com/</u>.
- [90] E. J. Berggren and E. H. Christensen, "Heart Rate as a Means of Measuring Metabolic Rate in Man," *Arbeitphysiologie*, vol. 14, pp. 255-260, 1950.
- [91] U. Ekelund, A. Yngve, and M. Sjöström, "Total Daily Energy Expenditure and Patterns of Physical Activity in Adolescents Assessed by Two Different Methods," *Scandinavian Journal of Medicine & Science in Sports*, vol. 9 5, pp. 257-264, 1999.
- [92] M. Massin, M. Lebrethon, D. Rocour, P. Gèrard, and J. Bourguignon, "Patterns of Physical Activity Determined by Heart Rate Monitoring Among Diabetic Children " *Archives of Disease in Childhool*, vol. 90, pp. 1223-1226, 2008.
- [93] Adidas and Polar. "Adistar Fusion Running System." Adidas and Polar, 2008 [cited March 7th 2008]. Available from <u>http://www.adidas-polar.com/</u>.
- [94] D. Andre, R. Pelletier, J. Farringdon, S. Safier, W. Talbott, R. Stone, N. Vyas, J.Trimble, D. Wolf, S. Vishnubhatla, S. Boehmke, J. Stivoric, and A. Teller, *The Development of the SenseWear Armband, a Revolutionary Energy Assessment Device to Assess Physical Activity and Lifestyle*, Technical Report, BodyMedia Inc, 2006.
- [95] BodyMedia Inc. "MyApex." 2008 [cited January 17th 2008]. Available from http://my.apexfitness.com/index.php.
- [96] J. M. Jakicic, M. Marcus, and K. Gallagher, "Evaluation of the SenseWear Pro Armband to Assess Energy Expenditure During Exercise," *Medicine & Science in Sports & Exercise*, vol. 36, 5, pp. 897-904, 2004.
- [97] K. Zhang, P. Werner, M. Sun, F. X. Pi-Sunyer, and C. N. Boozer, "Measurement of Human Daily Physical Activity," *Obesity Research*, vol. 11, 1, pp. 33-40, 2003.
- [98] A. Raj, A. Subramanya, D. Fox, and J. Bilmes, "Rao-Blackwellized Particle Filters for Recognizing Activities and Spatial Context from Wearable Sensors," in *Proceedings of the 10th International Symposium on Experimental Robotics (ISER '06)*. Rio de Janeiro, Brazil: Springer-Verlag, 2006.
- [99] A. Raj, J. Bilmes, and D. Fox, "Hierarchical Models for Activity Recognition," in *Proceedings of the IEEE Conference on Multimedia Signal Processing (MMSP '06)*. Victoria, BC, CA, 2006, pp. 233-237.
- [100] B. Ristic, S. Arulampalam, and N. Gordon, "Chapter 3: A Tutorial on Particle Filters," in *Beyond the Kalman Filter: Particle Filters for Tracking Applications* Artech House Radar Library, 2004.
- [101] P. Lukowicz, J. A. Ward, H. Junker, M. Stager, G. Troster, A. Atrash, and T. Starner, "Recognizing Workshop Activity Using Body Worn Microphone and Accelerometers," in *Proceedings of the International Conference in Pervasive Computing (PERVASIVE '04)*. Vienna, Austria: Springer Berlin/Heidelberg, 2004, pp. 18-32.
- [102] D. Minnen, T. Starner, J. A. Ward, P. Lukowicz, and G. Troster, "Recognizing and Discovering Human Actions from On-Body Sensor Data," in *Proceedings of the IEEE International Conference on Multimedia and Expo (ICME '05)*. Amsterdam, NL, 2005, pp. 1545-1548.
- [103] J. A. Ward, P. Lukowicz, and G. Troster, "Gesture Spotting Using Wrist Worn Microphone and 3-Axis Accelerometer," in *Proceedings of the Conference on Smart Objects and Ambient Intelligence*, vol. 121. Grenoble, France, 2005, pp. 99-104.

- [104] L. Rabiner and B. Juang, *Fundamentals Of Speech Recognition*. Upper Saddle River, NJ, USA Prentice Hall, 1993.
- [105] Ng and Jordan, "On Discriminative vs. Generative Classifiers: A Comparison of Logistic Regression and Naive Bayes," in *In Advances in Neural Information Processing Systems*, vol. 15, 2002.
- [106] R. Duda, P. Hart, and D. Stork, in *Pattern Classification*, 2nd ed: John Wiley & Sons, 2000.
- [107] U. Maurer, A. Smailagic, D. P. Siewiorek, and M. Deisher, "Activity Recognition and Monitoring Using Multiple Sensors on Different Body Positions," in *Proceedings of the International Workshop on Wearable and Implantable Body Sensor Networks (BSN'06)*: IEEE Computer Society 2006, pp. 113-116.
- [108] H. L. Chieu, W. S. Lee, and L. P. Kaelbling, "Activity Recognition from Physiological Data Using Conditional Random Fields," in *Proceedings of the Singapore-MIT Alliance (SMA) Computer Science Program.* Cambridge, MA, 2006.
- [109] P. Domingos and M. Pazzani, "Beyond Independence: Conditions for the Optimality of a Simple Bayesian Classifier," in *Proceedings of the Thirteenth International Conference on Machine Learning*, L. Saitta, Ed.: Morgan Kauffman, 1996, pp. 194-202.
- [110] P. Clark and T. Niblett, "The CN2 Induction Algorithm," *Machine Learning*, vol. 3, 4, pp. 261-283, 1989.
- [111] G. H. John and P. Langley, "Estimating Continuous Distributions In Bayesian Classifiers," in Proceedings of the Eleventh Conference on Uncertainty in Artificial Intelligence. San Mateo, 1995, pp. 338-345.
- [112] J. R. Quinlan, *C4.5: Programs for Machine Learning*. San Francisco, CA, USA: Morgan Kaufmann Publishers Inc, 1993.
- [113] J. Parkka, M. Ermes, P. Korpipaa, J. Mantyjarvi, J. Peltola, and I. Korhonen, "Activity Classification Using Realistic Data From Wearable Sensors," *IEEE Transactions on Information Technology in Biomedicine*, vol. 10, 1, pp. 119-128, 2006.
- [114] A. Y. Benbasat and J. A. Paradiso, "A Framework for the Automated Generation of Power-Efficient Classifiers for Embedded Sensor Nodes," in *Proceedings of the 5th ACM Conference on Embedded Networked Sensor Systems (SenSys '07)*. Sydney, Australia, 2007, pp. 219-232.
- [115] H. Junker, P. Lukowicz, and G. Troster, "Continuous Recognition of Arm Activities with Body-Worn Inertial Sensors," *Proceedings of the 8th IEEE International Symposium on Wearable Computers (ISWC '04)*, pp. 188-189, 2004.
- [116] N. Kern, B. Schiele, and A. Schmidt, "Multi-Sensor Activity Context Detection for Wearable Computing," in *European Symposium on Ambient Intelligence (EUSAI)*, vol. 2875/2003. Eindhoven, The Netherlands: Springer Berlin/Heidelberg, 2003, pp. 220-232.
- [117] U. Maurer, A. Rowe, A. Smailagic, and D. P. Siewiorek, "eWatch: A Wearable Sensor and Notification Platform," in *Proceedings of the IEEE Body Sensor Networks Workshop 2006*. Munchen, Germany, 2006, pp. 4-7.
- [118] C. C. Yang and Y. L. Hsu, "Developing a Wearable System for Real-Time Physical Activity Monitoring in a Home Environment," in *Proceedings of Telehealth 2007*. Montreal, QC, Canada, 2007.
- [119] T. Huynh and B. Schiele, "Analyzing Features for Activity Recognition," in *Proceedings of the Smart Objects and Ambient Intelligence*. Grenoble, France, 2005, pp. 159-163.
- [120] J. A. Levine, "Measurement of Energy Expenditure," *Public Health Nutrition*, vol. 8, 7A, pp. 1123-1132, 2005.
- [121] D. A. Schoeller, "Recent Advances from Application of Doubly Labeled Water to Measurement of Human Energy Expenditure," *The Journal of Nutrition*, vol. 129, 10, pp. 1765-1768, 1999.
- [122] B. E. Ainsworth, W. L. Haskell, M. C. Whitt, M. L. Irwin, A. M. Swartz, S. J. Strath, W. L. O'Brien, D. R. Bassett, Jr., K. H. Schmitz, P. O. Emplaincourt, D. R. Jacobs, Jr., and A. S. Leon, "Compendium of Physical Activities: An Update of Activity Codes and MET Intensities," *Medicine & Science in Sports & Exercise*, vol. 32, 9 Suppl, pp. S498-504, 2000.
- [123] A. K. Adams, R. A. Nelson, E. F. Bell, and C. A. Egoavil, "Use of Infrared Thermographic Calorimetry to Determine Energy Expenditure in Preterms Infants," *Journal of Clinical Nutrition*, vol. 71, 4, pp. 969.

- [124] M. Sun, G. W. Reed, and J. O. Hill, "Modification of a Whole Room Indirect Calorimeter for Measurement of Rapid Changes in Energy Expenditure," *Journal of Applied Physiology*, vol. 76, pp. 2686-2691, 1994.
- [125] COSMED Pulmonary Function Equipment. "K4b<sup>2</sup> Cardio Pulmonary Exercise Testing." 2008 [cited March 16th 2008]. Available from http://www.cosmed.it/products.cfm?p=1&fi=1&a=1&cat=1.
- [126] Parvo Medics. "Parvo Medics TrueOne 2400." 2008 [cited January, 2nd 2008]. Available from <u>http://parvo.com/Product.htm</u>.
- [127] Viasys Healthcare Inc. "Oxycon Mobile: Wireless Portable Ergospirometry System." 2008 [cited March 6th 2008].
- [128] MedGraphics. "The CPX Ultima Gas Exchange System." 2008 [cited March 30th 2008]. Available from http://www.medgraphics.com/datasheet\_ultimaCPX.html.
- [129] A. R. Ciminero, K. S. Calhoun, and H. E. Adam, *Handbook of Behavioral Assessment*. New York: Wiley, 1977.
- [130] A. S. Leon, J. Connett, D. R. Jacobs, and R. Rauramaa, "Leisure-time Physical Activity Levels and Risk of Coronary Heart Disease and Death: The Multiple Risk Factor Intervention Trial," *Journal of the American Medical Association*, vol. 258, 17, pp. 2388-2395, 1987.
- [131] S. N. Blair, W. L. Haskell, P. Ho, and ETAL, "Assessment of Habitual Physical Activity by a Seven-day Recall in a Community Survey and Controlled Experiments," *American Journal of Epidemiology*, vol. 122, 5, pp. 794-804, 1985.
- [132] A. L. Stewart, K. M. Mills, A. C. King, W. L. Haskell, D. Gillis, and P. L. Ritter, "CHAMPS Physical Activity Questionnaire for Older Adults: Outcomes for Interventions," *Medicine & Science in Sports & Exercise*, vol. 33, 7, pp. 1126-1141, 2001.
- [133] R. E. Taylor-Piliae, L. C. Norton, W. L. Haskell, "Validation of a New Brief Physical Activity Survey Among Men and Women Aged 60-69," *American Journal of Epidemiology*, vol. 164, 6, pp. 598-606, 2006.
- [134] D. A. Sch
  eon, *The Reflective Practitioner: How Professionals Think in Action*. New York: Basic Books, 1983.
- [135] A. A. Stone and S. Shiffman, "Ecological Momentary Assessment (EMA) in Behavioral Medicine," *Annals of Behavioral Medicine*, vol. 16, 3, pp. 199-202, 1994.
- [136] M. Richardson, B. Ainsworth, D. Jacobs, and A. Leon, "Validity of the Godin Physical Activity Questionnaire in Assessing Leisure Time Physical Activity," *Med Sci Sports Exerc*, vol. 28, S32, 1996.
- [137] G. R. Welk, R. W. Thompson, D. I. Galpher, and S. L. Dunn, "The Validity of the 7-day Physical Activity Recall Against the TriTrac-R3D Activity Monitor," *Med Sci Sports Exerc*, vol. 29, S44, 1997.
- [138] J. P. Robinson, "The Validity and Reliability of Diaries Versus Alternative Time Use Measures," in *Time Goods and Well-Being*, F. T. Juster and F. P. Stafford, Eds. Ann Arbor, MI, 1999, pp. 33-62.
- [139] N. Bolger, A. Davis, and E. Rafaeli, "Diary methods: Capturing life as it is Lived," Annual Review of Psychology, vol. 54, pp. 579-616, 2003.
- [140] J. Zuzanek, "Experience Sampling Method: Current and Potential Research Applications," in *Time-Use Measurement and Research: Report of a Workshop*, M. V. Ploeg, J. Altonji, N. Bradburn, J. DaVanzo, W. Nordhaus, and F. Samaniego, Eds.: The National Academies Press, 2000.
- [141] S. S. Intille, "Technological Innovations Enabling Automatic, Context-Sensitive Ecological Momentary Assessment," in *The Science of Real-Time Data Capture: Self-Report in Health Research*, A. A. Stone, S. Shiffman, A. A. Atienza, and L. Nebeling, Eds.: Oxford University Press, to appear, 2005.
- [142] M. Csikszentmihalyi and R. Larson, "Validity and Reliability of the Experience-Sampling Method," *The Journal of Nervous and Mental Disease*, vol. 175, 9, pp. 526-36, 1987.
- [143] W. L. Haskell, A. S. Leon, C. J. Caspersen, and et al., "Cardiovascular Benefits and Assessment of Physical Activity and Physical Fitness in Adults," *Medicine & Science in Sports & Exercise*, vol. 24, 6, pp. S201-220, 1992.
- [144] D. Basset and A. Cureton, "Measurement of Daily Walking Distance-Questionnaire Versus Pedometer," *Medicine & Science in Sports & Exercise*, vol. 32, 5, pp. 1018-1023, 2000.

- [145] P. S. Freedson, E. Melanson, and J. Sirard, "Calibration of the Computer Science and Applications (CSA), Inc. Accelerometer," *Medicine & Science in Sports & Exercise*, vol. 30, 5, pp. 777-781, 1998.
- [146] D. Hendelman, K. Miller, C. Bagget, E. Debold, and P. S. Freedson, "Validity of Accelerometry for the Assessment of Moderate Intensity Physical Activity in the Field," *Medicine and Science in Sports and Exercise*, vol. 32, pp. S442-S449, 2000.
- [147] J. F. Nichols, C. G. Morgan, L. E. Chabot, J. F. Sallis, and K. J. Calfas, "Assessment of Physical Activity with the Computer Science and Applications, Inc., Accelerometer: Laboratory Versus Field Validation," *Research Quarterly for Exercise and Sport.*, vol. 71, 1, pp. 36-43, 2000.
- [148] A. M. Swartz, S. J. Strath, D. R. Bassett, Jr., W. L. O'Brien, G. A. King, and B. E. Ainsworth, "Estimation of Energy Expenditure Using CSA Accelerometers at Hip and Wrist Sites," *Medicine & Science in Sports & Exercise*, vol. 32, 9 Supp, pp. S450-6, 2000.
- [149] D. P. Heil, "Predicting Activity Energy Expenditure Using the Actical Activity Monitor," *Research Quarterly for Exercise and Sport*, vol. 77, pp. 64-80, 2006.
- [150] K. H. Schmitz, M. Treuth, P. Hannan, R. McMurray, K. B. Ring, D. Catellier, and R. Pate, "Predicting Energy Expenditure from Accelerometry Counts in Adolescent Girls," *Medicine & Science in Sports & Exercise*, vol. 37, pp. 155-161, 2005.
- [151] K. Y. Chen and M. Sun, "Improving Energy Expenditure Estimation by Using a Triaxial Accelerometer," *Journal of Applied Physiology*, vol. 83, pp. 2112-2122, 1997.
- [152] M. P. Rothney, Advancing Accelerometry-based Physical Activity Monitors: Quantifying Measurement Error and Improving Energy Expenditure Prediction, Biomedical Engineering, Vanderbilt University, 2007.
- [153] Y. Jang, M. W. Jung, J. Kang, and H. C. Kim, "An Wearable Energy Expenditure Analysis System Based on the 15-Channel Whole-body Segment Acceleration Measurement," in *Proceedings of the 27th IEEE Engineering in Medicine and Biology*. Shangha, China: IEEE, 2005, pp. 3834-3836.
- [154] C. R. Noble, *Newtonian Posture & Activity Monitoring (NPAM)*, Technical Report, Ultimate Balance Inc, 2007.
- [155] Ambulatory Monitoring Inc. "Actillum." 2006 [cited October 30 2006]. Available from http://www.ambulatory-monitoring.com/.
- [156] Y. Kawahara, H. Kurasawa, and H. Morikawa, "Recognizing User Context Using Mobile Handsets with Acceleration Sensors," in *Proceedings of the IEEE International Conference on Portable Information Devices (PORTABLE '07)* Orlando, FL, 2007, pp. 1-5.
- [157] IM Systems. "ActiTrac Activity Monitor." 2006 [cited October 30 2006]. Available from http://www.imsystems.net/actitrac/index.html.
- [158] BodyMedia Inc. "BodyBugg Calorie Management System." 2006 [cited October 30 2006]. Available from http://www.bodymedia.com/products/index.jsp.
- [159] IM Systems. "DigiTrac Movemet Recorder." 2006 [cited 2006]. Available from <u>http://www.imsystems.net/DigiTrac.html</u>.
- [160] A. V. Rowlands, P. W. Thomas, R. G. Eston, and R. Topping, "Validation of the RT3 Triaxial Accelerometer for the Assessment of Physical Activity," *Medicine & Science in Sports & Exercise*, vol. 36, pp. 518-524, 2004.
- [161] P. C. Fehling, D. L. Smith, S. E. Warner, and G. P. Dalsky, "Comparison of Accerlometers with Oxygen Consumption in Older Adults During Exercise," *Medicine and Science in Sports and Exercise*, vol. 31, pp. 171-175, 1999.
- [162] K. Y. Chen, S. A. Acra, K. Majchrzak, C. L. Donahue, L. Baker, L. Clemens, M. Sun, and M. S. Buchowski, "Predicting Energy Expenditure of Physical Activity Using Hip and Wrist-worn Accelerometers," *Diabetes Technology and Therapeutics*, vol. 5, 2003.
- [163] K. Zhang, F. X. Pi-Sunyer, and C. N. Boozer, "Improving Energy Expenditure Estimation for Physical Activity," *Medicine & Science in Sports & Exercise*, vol. 36, 5, pp. 883-889, 2004.
- [164] S. E. Crouter, J. R. Churilla, and D. R. Basett, "Estimating Energy Expenditure Using Accelerometers," *European Journal of Applied Physiology*, vol. 98, pp. 601-612, 2006.
- [165] M. R. Puyau, A. L. Adolph, F. A. Vohra, and N. F. Butte, "Validation and Calibration of Physical Activity Monitors in Children," *Obesity Research*, vol. 10, 3, pp. 150-7, 2002.
- [166] MTI Actigraph. "GT1M Actigraph." 2005 [cited October 3rd, 2005 2005]. Available from <u>http://mtiactigraph.com/</u>.

- [167] Stayhealthy Inc. "CT1 Personal Calorie Tracker Kit and RT3 Tri-axial Research Tracker Kit." 2006 [cited October 30 2006]. Available from <u>http://www.stayhealthy.com/products/</u>.
- [168] Respironics. "Actiwatch." 2006 [cited October 30 2006]. Available from <u>http://actiwatch.respironics.com/</u>.
- [169] C. V. Bouten, W. P. Verboeket-van de Venne, K. R. Westerterp, M. Verduin, and J. D. Janssen, "Daily Physical Activity Assessment: Comparison Between Movement Registration and Doubly Labeled Water.," *Journal of Applied Physiology*, vol. 81, pp. 1019-1026, 1996.
- [170] M. B. Livingstone, P. J. Robson, and M. Totton, "Energy Expenditure by Heart Rate in Children: An Evaluation of Calibration Techniques," *Medicine & Science in Sports & Exercise*, vol. 32, pp. 1513-1519, 2000.
- [171] H. Kashiwazaki, "Heart Rate Monitoring as a Field Method for Estimating Energy Expenditure as Evaluated by the Doubly Labeled Water Method.," *Journal of Nutritional Science Vitaminol*, vol. 45, pp. 79-94, 1999.
- [172] P. S. Fredson and K. Miller, "Objective Monitor of Physical Activity Using Motion Sensors and Heart Rate," *Research Quarterly for Exercise and Sport*, vol. 71, pp. 21-29, 2000.
- [173] D. C. Nieman, *Exercise Testing and Prescription: A Health-Related Approach*. Mountain View, CA: Mayfield Publishing, 1999.
- [174] S. J. Strath, D. R. Bassett, A. M. Swartz, and D. L. Thompson, "Validity of the Simultaneous Heart Rate Motion Sensor Technique to Estimate Energy Expenditure," *Medicine & Science in Sports & Exercise*, vol. 34, pp. 888-894, 2002.
- [175] S. Brage, N. Brage, and P. W. Franks, "Branched Equation Modeling of Simultaneous Accelerometry and Heart Rate Improves Estimate of Directly Measured Physical Activity Energy Expenditure," *Journal of Applied Physiology*, vol. 96, pp. 343-351, 2004.
- [176] M. L. Fruin and J. W. Rankin, "Validity of a Multi-Sensor Armband in Estimating Rest and Exercise Energy Expenditure," *Medicine & Science in Sports & Exercise*, vol. 36, 6, pp. 1063-1069, 2004.
- [177] P. J. Cole, L. M. LeMura, T. A. Klinger, K. Strohecker, and T. R. McConnell, "Measuring Energy Expenditure in Cardiac Patients Using the Body Media Armband Versus Indirect Calorimetry: A Validation Study," *The Journal of Sports Medicine and Physical Fitness*, vol. 44, 3, pp. 262-271, 2004.
- [178] G. Welk and S. Blair, "A Comparative Evaluation of Three Accelerometery-based Physical Activity Monitors," *Medicine & Science in Sports & Exercise*, vol. 32, 9, pp. S489-S497, 2000.
- [179] S. W. Su, L. Wang, B. G. Celler, E. Ambikairajah, and A. V. Savkin, "Estimation of Walking Energy Expenditure by Using Support Vector Regression," in *Proceedings of the 27th IEEE Annual Conference in Engineering in Medicine and Biology (IEEE-EMBS '05)*. Shanghai, China, 2005, pp. 3526-3529.
- [180] J. H. Choi, J. Lee, H. T. Hwang, J. P. Kim, J. C. Park, and K. Shin, "Estimation of Activity Energy Expenditure Accelerometer Approach," *Proceedings of the 27th Annual Conference IEEE Engineering in Medicine and Biology* pp. 3830 - 3833, 2005.
- [181] M. P. Rothney, M. Neumann, A. Béziat, and K. Y. Chen, "An Artificial Neural Network Model of Energy Expenditure Using Nonintegrated Acceleration Signals," *Journal of Applied Physiology*, vol. 103, pp. 1427-1427, 2007.
- [182] K. Rennie, T. Rowsell, S. A. Jebb, D. Holburn, and N. J. Wareham, "A Combined Heart Rate and Movement Sensor: Proof of Concept and Preliminary Testing Study," *European Journal of Clinical Nutrition*, vol. 54, pp. 409-414, 2000.
- [183] R. G. Eston, A. V. Rowlands, and D. K. Ingledew, "Validity of Heart Rate, Pedometry, and Accelerometry for Predicting the Energy Cost of Children's Activities," *Journal of Applied Psychology*, vol. 84, 1, pp. 362-371, 1998.
- [184] A. Luke, K. C. Maki, N. Barkey, R. Cooper, and D. McGee, "Simultaneous Monitoring of Heart Rate and Motion to Assess Energy Expenditure," *Medicine & Science in Sports & Exercise*, vol. 29, 1, pp. 144-8, 1997.
- [185] J. K. Moon and N. F. Butte, "Combined Heart Rate and Activity Improve Estimates of Oxygen Consumption and Carbon Dioxide Production Rates," *Journal of Applied Physiology*, vol. 81, pp. 1754-1761, 1996.

- [186] M. S. Treuth, A. L. Adolph, and N. F. Butte, "Energy Expenditure in Children Predicted From Heart Rate and Activity Calibrated Against Respiration Calorimetry," *American Journal of Physiology*, vol. 38, pp. E12-E18, 1998.
- [187] B. J. Fogg, "Levels of Analysis for Persuasive Computers," 2001.
- [188] B. J. Fogg, *Persuasive Technology: Using Computers to Change What We Think and Do.* Ambsterdam Boston: Morgan Kaufmann Publishers, 2003.
- [189] T. Bickmore, A. Gruber, S. Intille, and D. Mauer, "A Handheld Animated Advisor for Physical Activity Promotion," in *Proceedings of the Annual Symposium of the American Medical Informatics Association*, 2006, pp. 855.
- [190] S. Consolvo, D. W. McDonald, T. Toscos, M. Y. Chen, J. Froehlich, B. Harrison, P. Klasnja, A. LaMarca, L. LeGrand, R. Libby, I. Smith, and J. A. Landay, "Activity Sensing in the Wild: A Field Trial of UbiFit Garden," in *Proceedings of the 2008 International Conference in Human Computer Interaction (CHI '08)*. Florence, Italy, 2007, pp. to appear.
- [191] J. O. Hill, H. R. Wyatt, G. W. Reed, and J. C. Peters, "Obesity and the Environment: Where Do We Go From Here?," *Science*, vol. 299, 5608, pp. 853-855, 2003.
- [192] J. A. Levine, L. M. Lanningham-Foster, S. K. McCrady, A. C. Krizan, L. R. Olson, P. H. Kane, M. D. Jensen, and M. M. Clark, "Interindividual Variation in Posture Allocation: Possible Role in Human Obesity," *Science*, vol. 307, pp. 584-586, 2005.
- [193] L. H. Epstein, R. A. Paluch, C. C. Gordy, and J. Dorn, "Decreasing Sedentary Behaviors in Treating Pediatric Obesity," *Archives of Pediatric & Adolescent Medicine*, vol. 154, pp. 220-226, 2000.
- [194] L. H. Epstein and J. N. Roemmich, "Reducing Sedentary Behavior: Role of Modifying Physical Activity," *Exercise and Sport Sciences Reviews*, vol. 29, pp. 103-108, 2001.
- [195] B. Sinclair, "Dance Dance Revolution Breaks Out in WV Schools," in *GameSpot*, 2006.
- [196] J. Nawyn, S. S. Intille, and K. Larson, "Embedding Behavior Modification Strategies into a Consumer Electronics Device: A Case Study," in *Proceedings of the International Conference on Ubiquitous Computing (UbiComp '06) 2006*: Springer-Verlag, 2006.
- [197] Contextual Computing Group. "Bluetooth Accelerometer." Georgia Institute of Technology, 2008 [cited January 16th 2008]. Available from <u>http://www.cc.gatech.edu/ccg/resources/btacc/</u>.
- [198]Sparkfun Electronics. "Wireless Accelerometer / Tilt Controller Version 2.5." 2008 [cited January<br/>16th 2008]. Available from<br/>http://www.sparkfun.com/commerce/product info.php?products id=254.
- [199] E. Munguia Tapia, S. S. Intille, L. Lopez, and K. Larson, "The Design of a Portable Kit of Wireless Sensors for Naturalistic Data Collection," in *Proceedings of the 4th International Conference on Pervasive Computing (PERVASIVE '06)*. Dublin, Ireland: Springer-Verlag, 2006.
- [200] D. R. Bassett Jr., B. E. Ainsworth, A. M. Swartz, S. J. Strath, W. L. O'Brien, and G. A. King, "Validity of Four Motion Sensors in Measuring Moderate Intensity Physical Activity," *Medicine & Science in Sports & Exercise*, vol. 32, 9 Suppl, pp. S471-80, 2000.
- [201] Omron Healthcare. "HJ-112 Premium Pedometer." 2008 [cited March 10 2008]. Available from http://www.omronhealthcare.com/enTouchCMS/app/viewCategory?catgId=34.
- [202] BodyMedia Inc. "The BodyBugg Armband." 2008 [cited March 10th 2008]. Available from <a href="http://www.bodybugg.com/">http://www.bodybugg.com/</a>.
- [203] COSMED Pulmonary Function Equipment. "K4b<sup>2</sup> Cardio Pulmonary Exercise Testing." 2008 [cited March 10th 2008]. Available from http://www.cosmed.it/products.cfm?p=1&fi=1&a=1&cat=1.
- [204] Polar. "Polar Heart Rate Monitor with WearLink Coded Chest Transmitter." 2008 [cited March 10 2008]. Available from <u>http://www.polarusa.com/</u>.
- [205] J. Karvonen, J. Chwalbinska-Moneta, and S. Saynajakangas, "Comparison of Heart Rates Measured by ECG and Microcomputer," *The Physician and Sports Medicine*, vol. 12, 65-69, 1984.
- [206] L. Leger and M. Thivierge, "Heart Rate Monitors: Validity, Stability, and Functionality," *The Physician and Sports Medicine*, vol. 16, pp. 143-151, 1988.
- [207] F. A. Treiber, L. Musante, S. Hartdagan, H. Davis, M. Levy, and W. B. Strong, "Validation of the Heart Rate Monitor with Children in Laboratory and Field Settings.," *Medicine & Science in Sports & Exercise*, vol. 21, 338-342, 1989.

- [208] J. E. McLaughlin, G. A. King, E. T. Howley, D. R. Basset, and B. E. Ainsworth, "Validation of the COSMED K4b2 Portable Metabolic System," *International Journal of Sports Medicine*, vol. 22, 280-284, 2001.
- [209] R. Duffield, B. Dawson, H. C. Pinnington, and P. Wong, "Accuracy and Reliability of a Cosmed K4b2 Portable Gas Analysis System.," *Journal of Science and Medicine in Sports*, vol. 7, 1, pp. 11-22, 2004.
- [210] J. Lester, T. Choudhury, G. Borriello, S. Consolvo, J. Landay, K. Everitt, and I. Smith, "Sensing and Modeling Activities to Support Physical Fitness," in *Workshop Proceedings of Ubicomp* (*Workshop: Monitoring, Measuring and Motivating Exercise: Ubiquitous Computing to Support Fitness*). Tokyo, Japan, 2005.
- [211] J. Mantyjarvi, J. Himberg, and T. Seppanen, "Recognizing Human Motion with Multiple Acceleration Sensors," in *Proceedings of the IEEE International Conference on Systems, Man, and Cybernetics*, vol. 2: IEEE Press, 2001, pp. 747-752
- [212] A. Y. Benbasat, *An Inertial Measurement Unit for User Interfaces*, M. S. Thesis, Program in Media Arts and Sciences, Massachusetts Institute of Technology, 2000.
- [213] O. Amft, H. Junker, and G. Troster, "Detection of Eating and Drinking Arm Gestures Using Inertial Body-Worn Sensors," in *Proceedings of The Ninth IEEE International Symposium on Wearable Computers (ISWC '05)*. Osaka, Japan: IEEE Computer Society, 2005, pp. 160-163.
- [214] L. R. Rabiner and B.-H. Juang, "A Theory and Implementation of Hidden Markov Models," in Fundamentals of Speech Recognition: Prentice Hall, 1993, pp. 321-389.
- [215] J. Friedman, T. Hastie, and R. Tibshirani, *Additive Logistic Regression: A Statistical View of Boosting*, Department of Statistics, Stanford, University Technical Report, Palo Alto, CA 1998.
- [216] C. J. C. Burges, "A Tutorial on Support Vector Machines for Pattern Recognition," *Data Mining and Knowledge Discovery*, vol. 2, pp. 121-167, 1998.
- [217] S.Thomas, J. Reading, and R. J. Shephard, "Revision of the Physical Activity Readiness Questionnaire (PAR-Q)," *Canadian Journal of Sport Sciences*, vol. 17, 4, pp. 338-345, 1992.
- [218] C. V. Bouten, K. T. Koekkoek, M. Verduin, R. Kodde, and J. D. Janssen, "A Triaxial Accelerometer and Portable Data Processing Unit for the Assessment of Daily Physical Activity," *IEEE Transactions on Bio-medical Engineering.*, vol. 44(3), pp. 136-47, 1997.
- [219] B. E. Ainsworth, W. L. Haskell, A. S. Leon, D. R. Jacobs, H. J. Montoye, J. F. Sallis, and R. S. Paffenbarger, "Compendium of Physical Activities: Classification of Energy Costs of Human Physical Activities.," *Medicine & Science in Sports & Exercise*, vol. 25, 1, pp. 71-80, 1993.
- [220] E. Munguia Tapia, S. Intille, and K. Larson, "Real-Time Recognition of Physical Activities and Their Intensities Using Wireless Accelerometers and a Heart Rate Monitor," in *Proceedings of the* 11th International Conference on Wearable Computers (ISWC '07). Boston, MA, 2007, pp. 37-40.
- [221] Tanita Corporation. "Tanita Body Composition Scale Model BC-536." 2006 [cited November 19, 2007]. Available from <u>http://www.tanita.com/IndexUS.shtml</u>.
- [222] S. S. Intille, K. Larson, E. Munguia Tapia, J. S. Beaudin, P. Kaushik, J. Nawyn, and R. Rockinson, "Using a Live-In Laboratory for Ubiquitous Computing Research," in *Proceedings of PERVASIVE* 2006. Berlin: Springer-Verlag, 2006.
- [223] Polar, "Polar Heart Rate Monitor," Visited 2001.
- [224] D. Minnen, T. Westeym, J. A. Ward, and P. Lukowicz, "Performance Metrics and Evaluation Issues for Continuous Activity Recognition," in *In Proceedings of Performance Metrics for Intelligent Systems (PerMIS)*,. Gaithersburg, MD, 2006.
- [225] J. A. Ward, P. Lukowicz, and G. Troster, "Evaluating Performance In Continuous Context Recognition Using Event-Driven Error Characterisation," in *Proceedings of the International Workshop in Location and Awareness (LoCA)*. Dublin, Ireland, 2006, pp. 239-255.
- [226] I. H. Witten and E. Frank, *Data Mining: Practical Machine Learning Tools and Techniques*, 2nd ed: Morgan Kaufmann, 2005.
- [227] The Mathworks Inc. "MATLAB." [cited March 10th 2008]. Available from <u>www.mathworks.com</u>.
- [228] E. Fix and J. Hodges, *Discriminatory Analysis; Non-parametric Discrimination: Consistency Properties*, November 1951.
- [229] P. Langley, W. Iba, and K. Thompson, "An Analysis of Bayesian Classifiers," in *Proceedings of the Tenth National Conference on Artificial Intelligence*. San Jose, CA: AAAI Press, 1992, pp. 223-228.

- [230] P. G. Weyand, M. Kelly, T. Blackadar, J. C. Darley, S. R. Oliver, N. E. Ohlenbusch, S. W. Joffe, and R. W. Hoyt, "Ambulatory Estimates of Maximal Aerobic Power From Foot -Ground Contact Times and Heart Rates in Running Humans," *Journal of Applied Physiology*, vol. 91, pp. 451-458, 2001.
- [231] J. A. Balogun, D. A. Martin, and M. A. Clendenin, "Calorimetric Validation of the Caltrac Accelerometer During Level Walking," *Physical Therapy*, vol. 69, pp. 501-509, 1989.
- [232] M. S. Bray, J. S. Morrow, J. M. Pivarnik, and J. T. Bricker, "Caltrac Validity for Estimating Caloric Expenditure with Children," *Pediatric Exercise Science*, vol. 4, pp. 166-179, 1992.
- [233] E. L. Melanson Jr. and P. S. Freedson, "Validity of the Computer Science and Applications, Inc. (CSA) activity monitor," *Medicine & Science in Sports & Exercise*, vol. 27, 6, pp. 934-40, 1995.
- [234] S. M. Ceesay, A. M. Prentice, K. C. Day, P. R. Murgatroyd, G. R. Goldberg, and W. Scott, "The Use of Heart Rate Monitoring in the Estimation of Energy Expenditure: A Validation Study Using Indirect Whole-Body Calorimetry.," *British Journal of Nutrition*, vol. 61, pp. 175-186, 1989.
- [235] S. Schulz, K. R. Westerterp, and K. Bruck, "Comparison of Energy Expenditure by the Doubly Labeled Water Technique with Energy Intake, Heart Rate, and Activity Recording in Man," *American Journal of Clinical Nutrition*, vol. 49, 1146-1154, 1989.
- [236] D. A. Winter, "Biomechanics and Motor Control of Human Movement," 2nd Edition ed. Toronto: John Wiley & Sons, 1990.
- [237] R. J. Quinlan, "Learning with Continuous Classes," in Proceedings of the 5th Australian Joint Conference on Artificial Intelligence. Singapore, 1992, pp. 343-348.
- [238] Y. Wang and I. H. Witten, "Induction of Model Trees for Predicting Continuous Classes," in Proceedings of the 9th European Conference on Machine Learning Poster Papers. Prague, Czech Republic, 1997.
- [239] A. J. Smola and B. Scholkopf, A Tutorial on Support Vector Regression, NeuroCOLT2, Technical Report NC2-TR-1998-030, 2003.
- [240] C. Hsu, C. Chang, and C. Lin, *A Practical Guide to Support Vector Classification*, Department of Computer Science National Taiwan University, Taipei, Taiwan 2007.
- [241] C. Chang and C. Lin. "A Library for Support Vector Machines." 2001 [cited. Available from <u>http://www.csie.ntu.edu.tw/~cjlin/libsvm/</u>.
- [242] J. H. Friedman, *Multivariate Adaptive Regression Splines*, Technical Report 102, Laboratory of Computational Statistics Stanford University, Palo Alto, CA.
- [243] B. L. Breiman, J. H. Friedman, R. A. Olshen, and C. J. Stone, *Classification and Regression Trees*. New York: Wadsworth, Inc, 1984.
- [244] C. Hsu, C. Chang, and C. Lin, A Practical Guide to Support Vector Classification, Technical Report Department of Computer Science and Information Engineering, National Taiwan University, 2003.
- [245] S. S. Keerthi, O. Chapelle, and D. DeCoste, "Building Support Vector Machines with Reduced Classifier Complexity," in *Journal of Machine Learning Research*, 2006, pp. 1493-1515.
- [246] A. G. Brooks, S. M. Gunn, R. T. Withers, C. J. Gore, and J. L. Plummer, "Predicting Walking METs and Energy Expenditure from Speed or Accelerometry," *Med Sci Sports Exerc*, vol. 37, 7, pp. 1216–1223, 2005.
- [247] J. A. Levine, P. A. Baukol, and K. R. Westerterp, "Validation of the Tracmor Triaxial Accelerometer System for Walking," *Medicine & Science in Sports & Exercise*, vol. 33, 9, pp. 1593-7, 2001.
- [248] D. P. Heil, B. K. Higginson, C. P. Keller, and C. A. Juergens, "Body Size as a Determinant of Activity Monitor Output During Over-ground Walking," *JEPonline*, vol. 6, pp. 1-11, 2003.
- [249] R. L. Blessey, H. J. Hislop, R. L. Waters, and D. Antonelli, "Metabolic Energy Cost of Unrestrained Walking," *Physical Therapy*, vol. 56, pp. 1019-1025, 1976.
- [250] W. J. Bubb, D. A. Martin, and E. T. Howley, "Predicting Oxigen Uptake During Level Walking at Speeds of 80-130m/min," *Journal of Cardiopulmonary Rehabilitation*, 5, pp. 462-465, 1985.
- [251] A. L. Cradock, J. L. Wiecha, K. E. Peterson, A. M. Sobol, G. A. Colditz, and S. L. Gortmaker, "Youth Recall and TriTrac Accelerometer Estimates of Physical Activity Levels," *Medicine & Science in Sports & Exercise*, vol. 36, 3, pp. 525-532, 2004.
- [252] M. Llyas and I. Mahgoub, "Chapter 5: Systems," in *Smart Dust: Sensor Network Applications, Architecture, and Design*: CRC Press, 2006, pp. 15-16.

- [253] D. Minnen, T. Westeyn, D. Ashbrook, P. Presti, and T. Starner, "Recognizing Soldier Activities in the Field," in *Proceedings of the Fourth International Workshop on Wearable and Implantable Body Sensor Networks (BSN '07)*, vol. 13. Aachen University, Germany: Springer Berlin Heidelberg, 2007, pp. 236-241.
- [254] K. V. Laerhoven and H. Gellersen, "Spine versus Porcupine: a Study in Distributed Wearable Activity Recognition," in *Proceedings of the Eight International Symposium on Wearable Computers (ISWC '04)*: IEEE Press, 2004, pp. 142-150.